

Final Program Environmental Impact Report for the Uptown Community Plan Update

Project No. 380611 SCH No. 2016061023 September 2016





FINAL ENVIRONMENTAL IMPACT REPORT

SCH No. 2013121076

SUBJECT: <u>UPTOWN COMMUNITY PLAN UPDATE</u>: CITY COUNCIL APPROVAL AND ADOPTION of an update to the Uptown Community Plan; Adoption of General Plan Amendments; Adoption of the Uptown Impact Fee Study; Amendments to the Land Development Code; Repealing the Mid-City Communities Planned District Ordinance (PDO); Repealing the West Lewis Street PDO; Rescinding the Interim Height Ordinance; and Rezoning of the Community Plan areas with Citywide zones.

FINAL DOCUMENT September 15, 2016:

In response to comments received during public review and City staff input subsequent to distribution of the Draft Program Environmental Impact Report (PEIR), minor revisions, clarifications and/or additions have been made to the document which do not change the conclusions of the Final PEIR regarding the project's potential environmental impacts and required mitigation. As defined in CEQA Section 15088.5, these revisions, clarifications or additions to the document – which are shown in strikeout/underline format, do not represent "significant new information" and therefore, recirculation of the Draft PEIR is not warranted. No new significant environmental impacts would occur from these modifications, and similarly, no substantial increase in the severity of environmental impacts would occur.

Additionally, in accordance with CEQA Section 15089, responses to comments received during the public review period of the Draft PEIR have been included in this final document and are located immediately after these Conclusions.

BACKGROUND:

The proposed Uptown Community Plan Update (proposed CPU) would be consistent with and incorporate relevant policies from the 2008 City of San Diego General Plan, as well as provide a long-range, comprehensive policy framework for growth and development in the Uptown community. The Uptown Community Plan was originally adopted in 1988 and last amended in 2008.

The Uptown Community Plan Update (CPU) can be found on the Planning Department's website at:

https://www.sandiego.gov/planning/community/profiles/uptown

The proposed Uptown CPU provides detailed policy direction to implement the General Plan with respect to the distribution and arrangement of land uses (public and private), the local street and transit network, the prioritization and provision of public facilities, community and site specific urban design guidelines, and recommendations to preserve and enhance natural open space and historic and cultural resources within the Uptown community.

CPU implementation requires adoption of the Uptown Community Plan, amendments to the General Plan to incorporate the CPU as a component of the General Plan Land Use Element, adoption of a Land Development Code (LDC) ordinance that would rezone the Planned District

Ordinance (PDO) areas within the CPU area with Citywide zones within the LDC and repeal the existing Mid-City Communities PDO, the West Lewis Street PDO, and Interim Height Ordinance. The project would also amend the mapped boundaries of the Uptown Community Plan Implementation Overlay Zone (CPIOZ) to include CPIOZ-Type A and CPIOZ-Type B areas that would limit building heights. A comprehensive update to the existing Impact Fee Study (IFS) (formerly known as the Public Facilities Financing Plan) is also proposed for adoption resulting in a new IFS for the Uptown community.

Uptown Community Plan Update

The Uptown Community Plan area consists of approximately 2,700 acres and lies just north of Downtown San Diego. It is bounded on the north by Mission Valley, on the east by Park Boulevard, and on the west and south by Old Town San Diego and Interstate 5. The Uptown community is located on a level mesa that is divided by numerous canyons and bordered by two major parks, Presidio and Balboa. The CPU area includes the neighborhoods of Mission Hills, Middletown, Hillcrest, the Medical Complex, University Heights, and Bankers Hill/Park West.

Applicant: City of San Diego Planning Department

ENVIRONMENTAL DETERMINATION:

Based on the analysis conducted for the project described above, the City of San Diego has prepared the following Environmental Impact Report (EIR) in accordance with the California Environmental Quality Act (CEQA). The analysis conducted identified that the project could result in significant impacts to the following issue area(s): **Transportation and Circulation, Noise (Ambient Noise and Construction), Historical Resources (Built Environment and Historic Districts), and Paleontological Resources (Ministerial Projects).**

The purpose of this document is to inform decision-makers, agencies, and the public of the significant environmental effects that could result if the project is approved and implemented, identify possible ways to minimize the significant effects, and describe reasonable alternatives to the project.

PUBLIC REVIEW DISTRIBUTION:

The following agencies, organizations, and individuals received a copy or notice of the draft EIR and were invited to comment on its accuracy and sufficiency. Copies of the Draft EIR, the Mitigation Monitoring and Reporting Program and any technical appendices may be reviewed in the offices of the Planning Department, or purchased for the cost of reproduction.

FEDERAL GOVERNMENT U.S. Environmental Protection Agency (19) U.S. Fish and Wildlife (23) U.S. Army Corps of Engineers (26) STATE OF CALIFORNIA California Department of Transportation, District 11 (31) California Department of Fish and Wildlife (32) California Department of Toxic Substances Control (39) California Regional Water Quality Control Board, Region 9 (44) State Clearinghouse (46A) California Coastal Commission (47) California Air Resources Board (49) California Transportation Commission (51) California Department of Transportation (51A) California Department of Transportation (51B) California Native American Heritage Commission (56) COUNTY OF SAN DIEGO Air Pollution Control District (65) County of San Diego Department of Planning and Land Use (68) County Water Authority (73) CITY OF SAN DIEGO Mayor's Office (91) Council President Lightner, District 1 Councilmember Zapf. District 2 Councilmember Gloria, District 3 Councilmember Cole, District 4 Councilmember Kersey, District 5 Councilmember Cate, District 6 Councilmember Sherman, District 7 Councilmember Alvarez, District 8 Council President Pro Tem Emerald, District 9 Theresa Quiroz, Planning Commissioner Planning Department K. Steinert A. Muto J. Murphy M. Pangilinan L. Gates B. Turgeon T. Galloway N. Bragado H. Greenstein G. Ghossain Planning Department – cont. S. Hajjiri D. Russell

R. Malone

M. Herrmann S. Osborn E. Vivero Ocampo F. Ianuarv S. Mercer K. Stanco S. Morrison M. Blake Development Services Department A. McPherson J. Quinn Transportation and Stormwater Department M. Stephens **CITY OF SAN DIEGO - continued** Fire and Life Safety Services (79) San Diego Fire – Rescue Department Logistics (80) Library Department (81) Central Library (81A) North Park Branch Library (81T) University Heights Branch Library (81JJJ) Historical Resources Board (87) Park & Recreation (89) Wetlands Advisory Board (91A) OTHER INTERESTED GROUPS, ORGANIZATIONS, AND INDIVIDUALS San Diego Association of Governments (108) San Diego County Regional Airport Authority (110) Metropolitan Transit System (112) San Diego Gas & Electric (114) Metropolitan Transit System (115) San Diego Unified School District (132) Sierra Club (165) San Diego Natural History Museum (166) San Diego Audubon Society (167) Mr. Jim Peugh (167A) California Native Plant Society (170) Wetland Advisory Board (171) Endangered Habitats League (182) Endangered Habitats League (182A) Citizens Coordinate for Century 3 (179) Carmen Lucas (206) South Coast Information Center (210) San Diego Archaeological Center (212) Save Our Heritage Organisation (214) Ron Christman (215) Clint Linton (215B) Frank Brown, Inter-Tribal Cultural Resources Council (216) Campo Band of Mission Indians (217) San Diego Archaeological Society Inc. (218) Kuumeyaay Cultural Heritage Preservation (223) Kuumeyaay Cultural Repatriation Committee (225) Native American Distribution (225A-S) Uptown Planners (498) North Park Planning Committee (363)

Golden Hill Community Planning Committee (259) Friends of Switzer Canyon (260) North Park Community Association (366) UCSD Physical & Community Planning (478) Middletown Property Owner's Association (496) Barry Hager, MISSION HILLS HERITAGE (497) Hillside Protection Association (501) Banker's Hill Canyon Association (502) Climate Action Campaign Allen Canyon Committee (504) Greater Golden Hill Community Development Corporation Walt Scott Chambers **David Swarens** Angela Landsberg Vicki Granowitz **Robert Barry** Stuart White, Mission Hills Business Improvement District Ann Wilson, Community Housing Works Kim Adler Ernestine Bonn Roy Dahl Anu Delouri **lim Frost** Ann Garwood Dave Gatzke Younger Glenn Robert Grinchuk Beth Jaworski Iohn Lamb Deidre Lee Don Liddell James Mellos III Janet O'Dea Jennifer Pesqueira Scott Sandel Michael Seidel Jake Sutton Andrew Towne Leo Wilson Tom Mullaney Gary Boner Eric Bowlby Rhett Butler Bruce Coons Ian Epley Neil Ferrier Tom Fox Sharon Gehl **Rich Gorin** Ionathan Hale Elizabeth Hannon **Richard Ledford** Bruce Leidenberger Joe Naskar Jeanne Rawlings Ken Tablang Gerrie Trussell

Chris Ward Tony Winney Kristin Harms Scott Kessler Angela Landsberg Susan Riggs-Tinsky Gary Weber David Swarens

RESULTS OF PUBLIC REVIEW:

- () No comments were received during the public input period.
- () Comments were received but did not address the accuracy or completeness of the draft environmental document. No response is necessary and the letters are incorporated herein.
- (X) Comments addressing the accuracy or completeness of the draft environmental document were received during the public input period. The letters and responses are incorporated herein.

Alynna

Alyssa Muto, Deputy Director Planning Department

June 10, 2016 Date of Draft Report

<u>September 15, 2016</u> Date of Final Report

Analyst: Kurtis Steinert, AICP / Denise Russell

Uptown Community Plan Update PEIR Letters of Comment and Responses

Letters of comment to the Draft PEIR were received from the following agencies, organizations, and individuals. Several comment letters received during the Draft PEIR public review period contained accepted revisions that resulted in changes to the final PEIR text. These changes to the text are indicated by strike-out (deleted) and underline (inserted) markings. The letters of comment and responses follow.

A1	Caltrans District 11	RTC-7
A2	SANDAG	RTC-10
A3	San Diego Unified School District	RTC-15
A4	State Clearinghouse	RTC-16
B1	Building Industry Association	RTC-20
B2	Climate Action Campaign (Coast Law Group on behalf of)	RTC-24
B3	Hillcrest History Guild	RTC-35
B4	Middletown Advisory Group	RTC-36
B5	Mission Hills Business Improvement District	RTC-40
B6	Mission Hills Heritage	RTC-44
B7	Mission Hills Heritage	RTC-46
B8	Mission Hills Town Council	RTC-74
B9	Rescue Hillcrest	RTC-76
B10	Rincon Band of Luiseno Indians	RTC-82
B11	San Diego Canyonlands	RTC-83
B12	San Diego County Archaeological Society	RTC-85
B13	San Diego County Board of Supervisors	RTC-86
B14	UCSD Physical & Community Planning	RTC-87
B15	Save Hillcrest	RTC-89
B16	Save Our Heritage Organisation	RTC-90
B17	Save Our Heritage Organisation (Brandt-Hawley Law Group on behalf of)	RTC-91
B18	Uptown Gateway Council (Allen Matkins on behalf of)	RTC-98
B19	Uptown United	RTC-264
B20	Uptown Planners	RTC-284
B21	Uptown Planners	RTC-288
B22	Uptown Planners	RTC-289
C1	Allen, Jeff	RTC-293
C2	Ashley, John	RTC-294
C3	Becker, Betty	RTC-297
C4	Becker, Elizabeth	RTC-298
C5	Becker, Gary	RTC-299
C6	Belinsky, Tina	RTC-300
C7	Biggs, Shannon & Bryan Liang	RTC-301
C8	Bilic, Lydia	RTC-306

C9	Bonner, Gary	RTC-307
C10	Bonner, Gary	RTC-308
C11	Buby, Lee	RTC-309
C12	Buompensiero, Joe & Marcia	RTC-312
C13	Burnett, Addie & Donald Strock	RTC-314
C14	Chavez, Joel	RTC-316
C15	Davis, Steve	RTC-317
C16	Dulay, Catrina	RTC-318
C17	Ebner, Joe	RTC-319
C18	Eddings, Ryan	RTC-320
C19	Eldred, Cynthia	RTC-335
C20	Emerick, Carol	RTC-337
C21	Epley, lan	RTC-348
C22	Epley, lan	RTC-349
C23	Epley, lan	RTC-351
C24	Evert, Matthew	RTC-353
C25	Fairbourn, Gary	RTC-354
C26	Gahagan, Tim	RTC-355
C27	Galetti, Jason	RTC-357
C28	Galetti, Tracee	RTC-358
C29	Gauld, Tom	RTC-359
C30	Gehl, Sharon	RTC-362
C31	Gorin, Richard	RTC-377
C32	Harms, Kristin	RTC-379
C33	Herbert, Vanessa	RTC-384
C34	Hilsen, Dene & Charles DeBarry	RTC-388
C35	Huntington, Karen	RTC-389
C36	Johnson, Roland	RTC-390
C37	Killian, Athena	RTC-391
C38	Laprath, Brian	RTC-393
C39	Le Tanter, Gaelle	RTC-394
C40	Lee, Deirdre	RTC-395
C41	Lee, Deirdre	RTC-396
C42	Lee, Deirdre	RTC-498
C43	Leicht, Mark & Katherine	RTC-401
C44	Lia, Marie Burke	RTC-402
C45	Lima, Paul	RTC-434
C46	MacMillian, Jill & Glen Feye	RTC-435
C47	Magnus, Pamela	RTC-436
C48	May, Gregory	RTC-437
C49	McComb, Roger	RTC-438
C50	Mercer, Patricia	RTC-439
C51	Mistry, Pretesh	RTC-440
C52	Moomjian, Scott	RTC-441
C53	Natale, Sandro	RTC-448
C54	Odom, Fred	RTC-449

ATTACHMENT 7

C55	Packer, Laura & O'Connor, J Robert & (e mail from Packer)	RTC-450
C56	Rice, Speedy & Judy Clarke	RTC-511
C57	Riddell, Daniel	RTC-514
C58	Robbie Robero	RTC-515
C59	Rosas, Maya & Michael Brennan, Dana Hook, Soheil Nkhahab	RTC-516
C60	Scheerer, Emily	RTC-518
C61	Scott, Doug	RTC-519
C62	Seisun, Verena	RTC-520
C63	Shanske, Donna	RTC-522
C64	Smith Adair, Christopher	RTC-523
C65	Spooner, Sarah	RTC-525
C66	Peter StClair	RTC-526
C67	Tagget-Burton, Dawn & Lorenzo Burton	RTC-536
C68	Tait, Laurie	RTC-538
C69	VA sdsolutions	RTC-541
C70	Vardan, A	RTC-542
C71	Webster, Becky	RTC-543
C72	Wong, Stephen & Sandy	RTC-544
C73	Wray, Francis	RTC-545

Master Response Regarding India Street Mitigation Measures

The following response is a master response addressing a number of comments that were received regarding India Street Improvements U17A and U17B identified in the Mobility Study (Appendix C of the PEIR) and corresponding mitigation measures TRANS 6.3-18 and TRANS 6.3-19 identified in Section 6.3 of the PEIR. It should be noted that the comments were primarily opposed to the implementation of the referenced mitigation measures, but did not raise a specific issue with regard to the adequacy of the PEIR.

Summary of Comments

There were a number of comment letters received for the Uptown CPU PEIR regarding both the Uptown CPU Mobility Study Improvements for India Street (U-17A and U-17B of the Mobility Study) and the related mitigation measures TRANS 6.3-18 and TRANS 6.3-19 in the Uptown CPU PEIR.

All of the comment letters expressed opposition to the implementation of the proposed Mobility Study Improvements and the associated mitigation measures presented in the Draft PEIR. In addition, many of the comments included opposition to the removal of the southbound lane of India Street between Sassafras Street and Redwood Street and stated that such a removal would negatively impact the local residents in accessing their homes. Many of comments went on to identify impacts to sidewalks, parking, and pedestrian safety that would result from road widening.

Several commenters requested the India Street improvements in the Mobility Study and the associated mitigation measures in the PEIR be deleted.

A number of comments addressed safety concerns for pedestrians, bicycles, and vehicles along the segment of India Street, particularly between Redwood Street and Vine Street.

Lastly, there were additional comments on the increase in the level of traffic using India Street as a result of the construction of the Rental Car Center for the San Diego International Airport and a request that traffic from the Rental Car Center be routed to Pacific Highway instead of India Street.

Response

Implementation of India Street Improvements (TRANS 6.3-18 and TRANS 6.3-19)

Improvements U17A and U17B in the Uptown Mobility Study correspond to mitigation measures TRANS 6.3-18 and TRANS 6.3-19 in the PEIR. As further detailed below, the proposed Candidate Findings ("Findings") included as an attachment to the Staff Report, provide a discussion of the infeasibility of these measures. As a result, these measures are not proposed for implementation. The following information is provided to further clarify the information included in the PEIR and Mobility Study (Appendix C of the PEIR) related to these mitigation measures.

India Street from Washington Street to Winder Street (Impact 6.3-18)

Implementation of mitigation measure TRANS 6.3-18 would reduce the significant impact along the segment of India Street from Washington Street to Winder Street (Impact 6.3-18) by restriping the roadway to as 2-lane collector with continuous left-turn lane.

A number of commenters objected to implementation of this measure as it would remove parking that supports adjacent businesses and would remove a buffer between the pedestrian walkway and the street, making the pedestrian environment less safe. The proposed Findings included as an attachment to the Staff Report shows that this improvement would conflict with the proposed CPU Mobility Element goals for "safe, walkable neighborhoods which utilize pedestrian connections and improved sidewalks to create a comfortable pedestrian experience." Mobility Element Policy MO-4.9 also supports implementing road diets and traffic-calming measures where appropriate to improve safety and quality of service, and increase walking and bicycling in Uptown. Mobility Element Policy MO-7.13 supports on-street parking on all streets to support adjacent uses and enhance pedestrian safety and activity. Thus, this measure would be infeasible because it would conflict with proposed Uptown CPU Mobility Element goals and policies.

India Street from Glenwood Drive to Redwood Street (Impact 6.3-19)

Implementation of mitigation measure TRANS 6.3-19 would reduce the significant impact along the segment of India Street from Glenwood Drive to Sassafras Street and From Sassafras to Redwood Street. From Glenwood Drive to Sassafras Street the measure includes widening the roadway to a 4-lane one-way collector and from Sassafras Street to Redwood Street the measure would widen the road to a 3-lane one-way collector. Implementation of these measures would change the configuration of India Street from two northbound one-way lanes to four northbound one-way lanes from Glenwood Drive to Sassafras Street. From Sassafras Street to Redwood Street, the measure would include widening the northbound portion of India Street to three lanes.

As discussed in the proposed Findings included as an attachment to the Staff Report, widening these roadway segments would increase crossing distance for pedestrians, require the removal on-street parking spaces that support adjacent businesses, and would impact residential and commercial structures by removing usable frontage for road purposes and potentially impacting structures. The proposed Findings show that the improvements would conflict with proposed CPU Mobility Element goals and policies including policy MO-4.9 that supports implementing road diets and traffic calming measures where appropriate to improve safety and quality of service, and increase walking and bicycling. Additionally, the proposed Findings show that the improvements on-street parking on all streets to support adjacent uses and enhance pedestrian safety and activity. Therefore, under the proposed Findings, the mitigation measures would not be implemented and impacts related to traffic on India Street would remain significant and unavoidable.

It should also be noted that the existing southbound lane along the segment of India Street from Sassafras Street to Redwood Street would remain. Some commenters were under the impression that the mitigation measure would remove the southbound lane; however that was not part of any mitigation measure. Table 13 of the Mobility Study showed a change from two lane collector (one

way) to a three lane collector (one way) where the southbound lane exists. However, the photo overlays showing improvements in this location did not include removal of the south bound lane (Figures 63 and 64 of the Mobility Study). These photos do show that to implement the improvement would require the removal of sidewalks and would encroach on the adjacent properties/buildings.

Safety Concerns

Many comments raised both exiting safety concerns and potential hazardous conditions that would result if the aforementioned improvements were implemented. Since implementation of mitigation measures TRANS 6.3-18 and TRANS 6.3-19 would not be implemented due to conflicts with the proposed Uptown CPU Mobility Element, implementation of these measures would not create or exacerbate any existing safety concerns in the area. Additionally, the proposed Uptown CPU Mobility Element includes a policy framework that promotes pedestrian and bicycle improvements, including enhancing sidewalks and bicycle lanes and retaining on-street parking to support adjacent businesses.

Removal of Measures from the PEIR

The referenced mitigation measures for India Street will not be removed from the PEIR or the Uptown Mobility Study because this would conflict with Section 15126.4 of the CEQA Guidelines which requires that an EIR discuss and consider measures that would minimize significant effects. Although the improvements are not recommended and the proposed Candidate Findings (included as an attachment to the Staff Report) show mitigation measures TRANS 6.3-18 and TRANS 6.3-19 to be infeasible, the measures are retained in the PEIR since they could reduce the potential impacts along these segments.

Rental Car Center Traffic

Regarding the request to reroute traffic from the Rental Car Center, the Rental Car Center is outside of the Uptown CPU area. Additionally, the San Diego County Regional Airport Authority prepared an EIR for the Airport Master Plan (State Clearinghouse Number 2005091105) that evaluated impacts associated with implementation of the Airport Master Plan including a consolidated rental car center that is now operational. Imposition of new mitigation measures to address Rental Car Center traffic is outside of the scope of the PEIR for the proposed Uptown CPU and associated discretionary actions.



Mr. Kurtis Steinert August 8, 2016 Page 2

SR-163 and 1-5, which include signalized intersections, in addition to other potentially significant impacts.

- A1-4 Caltrans supports the mitigation measures identified in the Transportation and Circulation section found in Table S-1 Summary of Significant Environmental Impacts, which include proposed mitigations listed on TRANS 6.3-1 through TRANS 6.3-39. The City should continue to coordinate with Caltrans to implement necessary improvements at intersections and interchanges where the agencies have joint jurisdiction, as well as coordinate with Caltrans as development proceeds and funds or "fair share" become available to ensure that the capacity of on-/olF-ramps is adequate.
- A1-5 Analysis of potential transit mitigation for development impacts is encouraged. Caltrans supports improved transit accommodation through the provision of park and ride facilities, bicycle and pedestrian access, signal prioritization for transit, bus on shoulders, ramp improvements, or other enhancements that can improve mobility and alleviate traffic impacts to State facilities.
 - The 1-8 Corridor Study that SANDAG recently completed in coordination with Caltrans
 offers examples of identified operational and multimodal improvements that can enhance
 the function of freeway corridors.
 http://www.sandag.org/index.asp?classid=13&subclassid=10&projectid=484&fuseaction=projecpr.detail.
 - SANDAG is currently working on the Uptown Bikeways Project, a multi-corridor bikeway project that will improve bicyclist mobility throughout the Uptown Community Caltrans will work with the City and SANDAG to address potential improvements proposed within Caltrans Right of Way. http://www.kcepsandiegomoving.com/RegionalBikeProjects/uptown_intro.aspx

Parks and Recreation

A1-6

- Uptown Mystic Park Concept: On March 7, 2013, Caltrans sent a comment letter to the City regarding a conceptual plan for the Uptown Park project located near SR-163. Please note that the park as proposed would impact the designated "Cabrillo Freeway Historic District"; Caltrans would not consider supporting any changes to the current Freeway Historic District.
 - For any proposed parks/park equivalencies on State Right of Way, further coordination is needed between Caltrans and the City to determine if the Right of Way can be declared excess and sold to a third party, and if not, to pursue an airspace lease. Caltrans policy allows for transportation compatible uses within the airspace defined as "any property within Right of Way limits of an existing highway... that is capable of other development without undue interference with the operation and foreseeable future expansion of the transportation corridor" and includes "area in cut or fill slopes (Caltrans Right of Way Manual, Section 15.01.01.01). Furthermore, Marler-Johnson Park Agreements (Section 15.04.01.05) allows a local agency to request use of Caltrans airspace for park or recreational purposes; Marler-Johnson agreements may be offered for a period of ten years with five-year extensions.

"Powerine is safe, norminable, ottegrated and Grauni transportance systems in verbance California's economy and biobility" A1-4 Comment noted. Section 15126.4 of the CEOA Guidelines states that mitigation measures must be fully enforceable through permit conditions, agreements, or other legally binding instruments, or otherwise incorporated into the associated plan or policy. While the PEIR includes a number of mitigation measures addressing freeway segments and meters, impacts to Caltrans facilities are considered significant and unavoidable because the City does not have approval authority over freeways and there is uncertainty related to the timing of implementation of the improvements and whether they will occur prior to the occurrence of impacts. However, policy MO-4.7 of the proposed Uptown CPU supports continued coordination between the City and Caltrans and the Final PEIR has been revised to clarify that the City will continue to work with Caltrans to identify options for fair-share contributions toward impacted segments where feasible.

- A1-5 Comment noted. This comment provides information provided regarding multi-modal and bikeway improvements and does not suggest an inadequacy or request a change in the PEIR.
- A1-6 Comment noted. Any action related to the proposed Mystic Park Concept will be closely coordinated with Caltrans, as an agreement between Caltrans and the City would be required. The recommended park feasibility study for this proposed conceptual park would address historic resource issues, vehicular, pedestrian and traffic circulation, reconfiguration of freeway on-ramps, new traffic signalization, community recreation needs, other issues to be determined, and a preliminary cost analysis.

ATTACHMENT 7

LETTER

Mr. Kurtis Steinert August 8, 2016 Page 3

A1-7

Caltrans appreciates the continued coordination with City staff on this Plan. If you have any questions, please contact Vanessa De La Rosa, Transportation Planner, at (619) 688-2510 or e-mail vanessa.delarosa@dot.ca.gov.

Sincerely,

Himlinly & Hodoon

KIMBERLY D. DODSON, Associate Transportation Planner

For

JACOB ARMSTRONG, Branch Chief Development Review Branch A1-7 This is a closing comment. All comments will become part of the public record.

RESPONSE

"Provide a safe, sustainable, integrated and efficient transportation system to enhance California's economy and livability"



LETTER

RESPONSE

- The majority of Fourth and Fifth Avenues are proposed as three-lane collectors with one lane dedicated for multi-modal transit. Consistent with Figure 6.3-5, the Uptown Bikeways Project is building a cycle track along the left side of Fourth and Fifth Avenues, which will use one of the three travel lanes. However, Figure 6.3-3 shows a streetcar and Rapid route proposed for Fourth and Fifth Avenues. Are the streetcar and Rapid both intended to share the right-most vehicle travel lane? If not, the figures may be inconsistent, as it may not be feasible to maintain the two vehicle travel lanes proposed in Figure 3-6 while also implementing the transit projects proposed in Figure 6.3-3 and the bike project proposed in Figure 6.3-5.
- In Figure 3-6, University Avenue (from Fourth Avenue to Park Boulevard) is proposed to remain a four-lane collector or major arterial, and in Figure 6.3-5, University Avenue is proposed to have a cycle track. The Uptown Bikeways Project will build the cycle track component (between SR 163 and Herbert Street) while maintaining the four vehicle travel lanes. However, streetcar, Rapid, and local bus services are also proposed for this segment, as seen in Figure 6.3-3. Would these transit services share the right-most vehicle travel lane? If not, the figures may be inconsistent, as it may not be feasible to maintain the four vehicle travel lanes proposed in Figure 3-6 while also implementing the transit projects and bike projects proposed in Figures 6.3-3 and 6.3-5.

Regional Bike and Pedestrian Infrastructure

A2-3 Other minor inconsistencies with SANDAG's Regional Bike Plan and the Uptown Bikeways Project exist within the Draft PEIR. SANDAG suggests clarifying these inconsistencies, described below.

> The Uptown Bikeways Project incorporates pedestrian improvements (e.g., high visibility crosswalks, bulb-outs) to provide pedestrian access along the corridors identified in the Banker's Hill "Walk the Walk" Plan. These improvements were prioritized after receiving feedback during community outreach meetings for the Uptown Bikeways Project. As such, the Project will include improvements to enhance east-west pedestrian connectivity across Fourth and Fifth Avenue along the following streets: Upas, Spruce, Quince, Nutmeg, Laurel, Juniper, Grape, and Elm. SANDAG suggests adding references to these corridors in both Section 6.3.1.6.c and Figure 16 in Appendix C.

A2-4 In Figure 6.3-5, Fourth Avenue (south of Laurel Street) is proposed as a cycle track. The Uptown Bikeways Project is building one-way cycle tracks, where feasible, along the majority of Fourth and Fifth Avenues. However, a cycle track on Fourth Avenue (south of Laurel Street) is not feasible due to the slope and resulting downhill bicycle speeds, in addition to considerations of sight lines and on-street parking. The Uptown Bikeways Project is planning to build a Class II buffered bike lane in this segment. SANDAG suggests updating the figure to reflect this information.

Smart Growth

A2-5 SANDAG appreciates that the City of San Diego has prioritized transit oriented development and land use changes in the project area that support both the goals of the Update as well as the Smart Growth Concept Map and Regional Plan. A key goal of the Regional Plan is to focus growth in smart growth opportunity areas. Five Smart Growth Opportunity Areas identified on the Smart Growth Concept Map are located in the planning area: two Existing/Planned Community

2

A2-3 Reference to the pedestrian improvements of the Uptown Bikeways Project have been added to Section 6.3.1.6c.

- A2-4 Figure 6.3-5 of the PEIR and Figure 3-2 of the proposed Uptown CPU have been updated to show Class II bicycle facilities on Fourth Avenue south of Laurel Street. Additionally, Figure 7 of the Mobility Study (Page 15), and Table 2 of the Mobility Study (Page 16) have been updated.
- A2-5 Comment noted.

ATTACHMENT 7

LETTER

RESPONSE

Centers (SD UP-2, SD UP-4), one Existing Planned Urban Center (SD UP-5), one Existing/Planned Mixed Use Transit Corridor (SD UP-1), and one Potential Mixed Use Transit Corridor (SD UP-3). The proposed project is currently well-served by a number of high-frequency local bus routes (Routes 1, 3, 7, 10, 11, and 120), as well as Trolley service (Green Line) and *Rapid* service (Route 215).

Other Considerations

A2-6

A number of edits are proposed in the attachment to this letter to address minor errors found in the Draft PEIR. Furthermore, SANDAG has a number of resources that can be used as plans are updated. These can be found on our website at www.sandag.org/igr.

When available, please send any additional environmental documents related to this project to:

Intergovernmental Review c/o SANDAG 401 B Street, Suite 800 San Diego, CA 92101

We appreciate the opportunity to comment on the Uptown Community Plan Update Draft PEIR. If you have any questions, please contact me at (619) 699-1943 or via e-mail at susan.baldwin@sandag.org.

3

Sincerely,

ma

SUSAN B. BALDWIN, AICP Senior Regional Planner

SBA/KHE/ais

Attachment: 1. Minor Edits, Uptown Community Plan Update Draft PEIR

A2-6 Comment noted. Responses to the attachment are provided in the following response to comments.

RESPONSE

Attachment 1 Minor Edits, Uptown Community Plan Update Draft PEIR

San Diego Forward: The Regional Plan

A2-7 Throughout the document, please replace any reference to the 2050 Regional Transportation Plan (RTP) with "San Diego Forward: The Regional Plan," as this was the most recently adopted Regional Plan (October 9, 2015).

On page 5-7, Section 5.1.5, please edit the section to read as follows:

"San Diego Forward: The Regional Plan (RP) combines two of the region's existing planning documents: the Regional Comprehensive Plan (RCP) and the Regional Transportation Plan and its Sustainable Communities Strategy (RTP/SCS). The RCP, adopted in 2004, laid out key principles for managing the region's growth while preserving natural resources and limiting urban sprawl. The plan covered eight policy areas, including urban form, transportation, housing, healthy environment, economic prosperity, public facilities, our borders, and social equity. These policy areas were addressed in the 2050 RTP/SCS and are now fully Integrated into the Regional Plan.

On April 24, 2015, SANDAG released the draft RP for public comment, with a closing date of July 15, 2015. A final RP was adopted by the SANDAG Board of Directors on October 9, 2015."

Rapid Transit

A2-8 When discussing Bus Rapid Transit throughout the document, please replace the existing term with "Rapid". In the same vein, please remove any references to "Arterial Rapid Transit" and replace with "Rapid". On pages 2-13 and 2-14, SANDAG suggests combining sections a, and c, into one section (using the header "Rapid") with the following description:

> "Rapid transit is corridor-level service providing fast and frequent transit services that are designed to take advantage of both freeway improvements, such High Occupancy Vehicles and managed lanes, and arterial improvements in order to serve longer distance regional trips. The Rapid service on arterials will operate on arterial roadways and provide limitedstop, high-speed service along several key corridors throughout the region, supplementing existing local bus services."

Freeway Segment Improvements

- A2-9 Please update pages 5-16, 5-17, 6.3-51, 6.3-52, and 11.6-7 to reflect improvements identified in the most recently adopted RP, as many of the freeway segment improvements listed are out of date. The changes include:
 - TRANS 6.3-33 This project is not in the RP. Please update this segment to include operational improvements on I-5 from I-15 to I-8 by 2050.
 - TRANS 5.3-34 The RP shows operational improvements on I-8 from I-5 to east of SR 125 by 2050.

A2-7 References to San Diego Forward: The Regional Plan have been updated throughout the document and the requested language added to Section 5.1.5 of the Final PEIR.

A2-8 The requested changes regarding Bus Rapid Transit were made to Section 2.3.3.2, Public Transportation.

A2-9 The corrections to the referenced measures have been made in the Final PEIR.

- TRANS 6.3-36 The RP shows Managed Lane improvements on I-805 from SR 15 to SR 163 by 2050.
- TRANS 6.3-37 The RP shows Managed Lane improvements on SR 94 from I-5 to I-805 by 2035. However, based on input from the community, Caltrans will evaluate additional alternatives as part of the environmental analysis for the SR 94 Express Lanes project.

Transportation Demand Management

A2-10 The SANDAG Transportation Demand Management (TDM) division supports the mobility and parking strategies included in the Uptown Community Plan, as well as the City's General Plan. SANDAG suggests referencing the TDM measures that exist within the City's Mobility Element to provide a better linkage between the two documents.

In order to further mitigate regional transportation impacts and increase mobility choices throughout the community, please consider integrating the following strategies:

- Promote the use of on-demand ridesharing (in addition to carshare and bikeshare) to reduce reliance on private automobiles and reduce the demand for parking. Allocation of designated curo space to facilitate convenient and seamless connections to shared mobility solutions may also be needed.
- Designate preferential and conveniently located parking spaces for carpools, vanpools, and other shared mobility options.
- Consider the feasibility of these additional parking management strategies: improved balance of on and off-street parking rates to assist with reducing cruising for parking; evaluation of extending priced parking periods (i.e., beyond 6 p.m.) as part of a dynamic or demand-based parking pricing implementation; and development of a comprehensive marketing and communication strategy to coincide with the development of an Uptown parking management plan.
- Encourage developers to incorporate TDM measures into developments through the entitlement process.
- Partner with the SANDAG TDM program, iCommute, to promote and incentivize regional services that encourage the use of transportation alternatives, such as the SANDAG Vanpool Program, online ridematching services, the Guaranteed Ride Home program, and bike encouragement programs (such as the GO by BIKE Mini-Grant program, free bike education courses, and the Walk, Ride, and Roll to School Mini-Grant and education program). Further, iCommute employer services can work with local businesses to offer commuter benefits programs that promote transportation alternatives to employees.

More information on the SANDAG TDM program can be accessed through www.iCommuteSD.com.

5

- A2-10 The comment provides recommendations for adding additional Transportation Demand Management (TDM) strategies to further mitigate regional transportation impacts and increase mobility choices throughout both communities, including:
 - Promoting the use of on-demand ride sharing;
 - Designating preferential and conveniently located parking spaces for carpools, vanpools, and other shared mobility options;
 - Consider additional parking management strategies;
 - Encourage developers to incorporate TDM measures into development through entitlement; and
 - Partner with the SANDAG TDM program, iCommute, to promote and incentivize regional services that promote alternative transportation.

The Mobility Element goals for both CPU's align with the recommendations provided by SANDAG. Specifically, one of the key Mobility Element goals in the Uptown CPU is for inter-agency coordination to implement comprehensive mobility strategies and project opportunities and identification of funding sources. The proposed CPU also incorporates specific policies consistent with SANDAG recommendations. For example, Policies MO-6.1 through MO-6.4 call for the City to encourage TDM strategies such as alternative work schedules and bicycle and ride sharing. Policies also support dedicated car-sharing parking spaces and providing electric vehicle charging stations. Thus, as discussed in Section 6.3.3 of the Draft PEIR under Issue 2, the proposed CPU and associated discretionary actions would be consistent with adopted policies, plans, or programs supporting TDM.

Additionally, language has been added to Section 6.3.5 of the Final PEIR to recognize that at the project-level, significant impacts at locations outside of the jurisdiction of the City could be partially mitigated in the form of TDM measures that encourage carpooling and other alternative means of transportation consistent with proposed CPU policies.

LETTER

	Letter A3		
	From: Hudson Sarah [mailto:shudson@sandi.net] Sent: Tuesday, July 12, 2016 12:13 PM To: Pangilinan, Marlon <mpangilinan@sandiego.gov> Subject: Comments regarding Uptown Community Plan Update PEIR</mpangilinan@sandiego.gov>		
	Mr. Pangilinan,		
A3-1	Please accept these comments and edits on behalf of San Diego Unified School District for the public review comment period ending August 8, 2016.	A3-1	Comment noted. The City appreciates the San Diego Unified School District's participation in the public review comment process.
A3-2	1) Page 6.12-4, Section 6.12.1.5 Schools. Grant is a K-8 school, not an elementary school.		
A3-3	2) Page 6.12-18, Table 6.12-5. The number of students shows for grades 6-8 in the middle column is incorrect. It shows 695; it should be 299. Please see attached letter which the correct information in Table 2. I believe someone accidentally used the North Park number instead of the Uptown number.	A3-2	The requested revision regarding Grant School has been made in the Final PEIR.
A3-4	3) Figure 6.12-1, Location of Public Services and Facilities – Uptown. Urban Discovery Charter school is on the map; however, it has moved to another location (840 14th Street, San Diego, CA 92101) outside of the Uptown area. It should be removed from the map.	A3-3	The number of students for grades 6-8 has been updated in the Final PEIR.
A3-5	In the future, I would be happy to provide letters in Word format, instead of PDF, so you or your staff does not have to retype the tables.	A3-4	Urban Discovery Academy Charter has been removed from
	Thank you.		Figure 6.12-1.
	Sarah Hudson Demographer, San Diego Unified School District Instructional Facilities Planning Department 4100 Normal Street, Annex 2, Noom 101 San Degn, CA 2010/2562 Telephone (619) 725-789 Fax (619) 725-789 shudson@sandi.net	A3-5	Comment noted.

ATTACHMENT 7

LETTER



SCH# Project Title Lead Agency	2016061023 Uptown Community Plan Update			
Туре	EIR Draft EIR			
Description	Note: Review Per Lead Note: Reference SCH #2013121076			
	The proposed update for the Uptown Community Plan would be consistent with and incorporate relevant policies from the 2008 City of San Diego General Plan, as well as provide a long range, comprehensive policy framework for growth and development in the Uptown community. The Uptown Community Plan was originally adopted in 1988. Uptown was last amended in 2008.			
Lead Agend	y Contact			
Name	Kurtis Steinert			
Agency	City of San Diego			
Phone	018-230-5200 Fax			
Address	1010 Second Ave, MS 413			
City	San Diego State CA Zip 92101			
Project Loc	ation			
County	San Diego			
City	San Diego			
Region				
Cross Streets				
Parcel No.	a la sector de la se			
Township	Range Section Base			
Proximity to	:			
Highways	163, 94			
Airports	SD Intl			
Railways	BNSF, SDIVR, MTS			
Schools	San Diego Kiver, Pacific Ocean			
Land Use	Various			
Project issues	Aesthetic/Visual; Air Quality; Archaeologic-Historic; Biological Resources; Drainage/Absorption; Geologic/Selsmic; Nolse; Population/Housing Balance; Public Services; Recreation/Parks; Schools/Universitiles; Solid Wasle; Toxic/Hazardous; Traffic/Circulation; Vegetation; Water Quality; Water Supply; Wildlife; Growth Inducing; Landuse; Cumulative Effects			
Reviewing Agencles	Resources Agency; Department of Fish and Wildlife, Region 5; Office of Historic Preservation; Department of Parks and Recreation; Department of Water Resources; Calitrans, Division of Aeronautics; Calitrans, District 6; Department of Housing and Community Development; Air Resources Board, Transportation Projects; Regional Water Quality Control Board, Region 9; Department of Toxic Substances Control; Native American Heritage Commission			
ate Received	06/10/2016 Start of Review 06/10/2016 End of Review 08/08/2016			
	Note: Blanks in data fields result from insufficient information provided by lead agency			

LETTER

RESPONSE

2016061623 STATE OF CALIFORNIA-CALIFORNIA STATE TRANSPORTATION AGENCY ROMLIND G. BROWN M. Geveraor DEPARTMENT OF TRANSPORTATION DISTRICT 11, DIVISION OF PLANNING Clear 4050 TAYLOR ST. M.S. 240 SAN DIEGO, CA 92110 08/08/160 PHONE (6191688-6950 Serieure drough FAX (619) 688-4299 Halp suve maier TTY 711 www.dot.ca.gov Governor's Office of Planning & Research August 8, 2016 AUG 08 2016 11-SD-5, 8, 94, 163 Draft Program Environmental Impact Report STATE CLEARINGHOUSE for the Uptown Community Plan Update Project No. 380611 Mr. Kurtis Steinert Senior Environmental Planner City of San Diego Planning Department 1010 Second Avenue, MS 413 San Diego, CA 92101 Dear Mr. Steinert: Thank you for including the California Department of Transportation (Caltrans) in the environmental review process for the City of San Diego's (City) Uptown Community Plan Update (Plan), area served by the Interstate 5 (1-5), Interstate 8 (1-8), State Route 94 (SR-94), and State Route 163 (SR-163). The mission of Caltrans is to provide a safe, sustainable, integrated, and efficient transportation system to enhance California's economy and livability. The Local Development-Intergovernmental Review (LD-IGR) Program reviews land use projects and plans to ensure consistency with our mission and state planning priorities of infill, conservation, and efficient development. To ensure a safe, efficient, and reliable transportation system, we encourage early consultation and coordination with local jurisdictions and project proponents on all development projects that utilize the multi-modal transportation network. Caltrans has reviewed the Draft Program Environmental Impact Report (DPEIR) dated June 10, 2016 for the Plan, and has the following comments: Caltrans recognizes that there is a strong link between transportation and land use. Development can have a significant impact on traffic and congestion on State transportation facilities. In particular, the pattern of land use can affect both total vehicle miles traveled and the number of trips. Caltrans strongly encourages local agencies to work towards a safe, functional, interconnected, multi-modal system. DPEIR The Plan's area consists of approximately 2,700 acres and lies just north of Downtown San Diego hordered by Mission Valley to the north, Park Boulevard on the east, and west and south by Old Town San Diego and 1-5. The Plan's DPEIR presents an analysis of existing conditions and the build-out conditions for Year 2035 and their respective impacts associated by cumulative impacts of future planned projects. Impacts within the community planning area include impacts to "Provide n safe, summable, integrated and efficient transportation system to enhance California's economy and loghility

ATTACHMENT 7

RESPONSE

LETTER

Mr. Kurtis Steinert August 8, 2016 Page 3

Caltrans appreciates the continued coordination with City staff on this Plan. If you have any questions, please contact Vanessa De La Rosa, Transportation Planner, at (619) 688-2510 or email vanessa.delarosa@dot.ca.gov.

Sincerely,

Himludy D Hodoan

KIMBERLY D. DODSON, Associate Transportation Planner

For

JACOB ARMSTRONG, Branch Chief Development Review Branch

> "Provide a sufe, sustainable, Tolegrated and efficient insurportation system to enhance California's economy and livebility"

LETTER

RESPONSE

		Letter B1		
B1-1	United in the state of the state o	August 8, 2016 Marlon Pangilinan Senior Planner City of San Diego Planning Department 1222 First Avenue MS-413 San Diego, CA 92101 mpangilinan@sandiego.gov Re: Building Industry Association of San Diego Comments on the Draft Program Environmental Impact Report Dear Mr. Pangilinan, The Building Industry Association of San Diego County represents 700 member companies and is the upice of aug 60 000 member advectors in the	B1-1	Comment noted. The City appreciates the Building Industry
	PAST CHAIRMAN David Poole Brookfield Residential	member companies and is the voice of over 60,000 men and women in the construction industry. We have reviewed the Draft update Uptown Community Plan Program Environmental Impact Report and offer the following comments and recommendations.		Association's (BIA's) participation in the public review comment process.
B1-2	PRESIDENT & C.E.O. Borre Winckel AFFILIATES California Building Industry Association National Association of Home Builders	San Diego, as elsewhere faces a severe housing crisis as inventory depletes and housing costs escalate to near historic highs. The San Diego Association of Governments concludes that the region must produce nearly 12,000 units annually to keep pace with population and employment demand. It has been a decade since the region experienced such production and while permits have improved, the region is currently off 8-percent compared to this time last year. This chronic shortage, coupled with expansive climate change regulations via AB 32 and SB 375 has brought renewed focus using the city's adopted Climate Action Plan to focus future growth to urban centers such as Uptown and its transit opportunities.	B1-2	Comment noted.
B1-3		 Areas of Concern Downzoning Potential Historic Districts Height Restrictions 	B1-3	Comment noted. This comment states that areas of concern are in downzoning, potential historic districts, and height restrictions, which are further detailed in the following comments.
		Building Industry Association of San Diego County 9201 Spectrum Center Blvd., Suite 110, San Diego, CA 92123-1407 P 858-450-1221 F 858-552-1445 www.biasandiego.org		

RESPONSE

Downzoning

B1-4

B1-5

The statement that *Uptown provides some of the highest densities in the region outside of downtown*, is misleading as the plan is in fact a substantial downzone of at least 2,000 units. The same can be said of the statement that the proposal provides the *opportunity for more than 9,500 additional housing units* over what is currently built when in actually there is a loss of total buildout units.

This is counterintuitive to the goals of the City of Villages and the Climate Action Plan that boasts of the need to maximize densities along transportation corridors as an essential component to greenhouse gas reduction efforts. The overall downzone in Uptown is inconsistent with the city's General Plan Goals and Policies which is to *'focus growth into mixed-use activity centers that are pedestrian-friendly, centers of community, and linked to the regional transit system*.' A stated reduction of 2,000 units means Uptown (a designated Transit Priority Area) will accommodate approximately 6,000 *fewer* residents under the proposed plan when compared to the current plan. This will further complicate the city's efforts to achieve its Regional Housing Needs Assessment (RHNA) objectives that totals 88,000 additional units by 2020. Moreover, to push approximately 6,000 people further out from the urban /transit centers will result in an increase in Vehicle Miles Traveled (VMT) as residents face longer commutes to work or shop due to housing and transit constraints.

Height Restrictions

The BIA has serious concerns with the proposed CPIOZ height restrictions that cap ministerial approvals at 65 feet in CPIOZ-A and discretionary approvals at 150 feet in CPIOZ-B as noted in figures 3-7 and 3-8. Such restrictions are the residue of the 'interim' height ordinance that was first established in 2008. The effect will be to impede the ability to achieve allowable densities and result in a *reduction* in the number of buildable units as economic feasibility becomes a factor. An appropriate remedy would be for the city to eliminate the height restrictions that impedes unit production along important transit corridors. Otherwise it must acknowledge that the project description is flawed because the downzone will be significantly larger than the 2,000 units currently stated in the project description.

B1-6 FAR is also being reduced, which is the amount of buildable mass that a developer is able to build. Reducing FAR effectively limits the size of the units, limiting bedrooms, which limits the potential for family housing. As an example, the current code for CL-6 is being reduced from 2.0 FAR by more than half down to .75.

The City does not agree that the proposed Uptown CPU is B1-4 inconsistent with the goals of the City of Villages strategy and the Climate Action Plan (CAP) because one of the main goals of the proposed Uptown CPU is to provide higher densities along transit corridors. The entire Uptown community is not a Transit Priority Area as this comment suggests; rather, portions of the Uptown community are designated as a Transit Priority Area. While the proposed Uptown CPU density distribution ultimately results in decreased residential densities in some areas, the highest densities are located where they will be best served by existing and planned transit, pedestrian, and bicycle facilities, balanced with commercial and mixed-use distributions that also support a multi-modal network. This land use pattern provided by the proposed CPU achieves the overall goals of the City of Villages and CAP. Refer to PEIR Section 6.1.3 for a discussion of project consistency with applicable plans including the City's General Plan and refer to Section 6.5.3 for a discussion of the proposed Uptown CPU's consistency with the CAP.

> While the expected build-out of the Uptown Community would involve approximately 2,000 less residential units than projected under the adopted Community Plan, the future population under build-out of the proposed CPU would be an estimated 55,700. This is not 6,000 fewer residents than the adopted Community Plan, as this comment suggests. Rather, build-out of the proposed Uptown CPU would result in approximately 3,000 less residents than the population estimate of 58,870 at build-out of the adopted Community Plan. It is important to note that though the proposed CPU would result in a lower population at build-out than the adopted Community Plan, it does not "push out" any existing residents and still allows for a population increase of almost 20,000 residents compared to the community's current population.

2

ATTACHMENT 7

Comment noted. The City does not agree that the project B1-5 description is flawed and does not overstate the amount of units that could be constructed under the proposed Uptown CPU. The planning estimate for the amount of housing units that could occur in the future was based on assumptions regarding what could reasonably develop in the future based on community plan land use designations for both the adopted and proposed community plan land uses. Generally, the analysis assumed that vacant parcels, parcels developed below the maximum residential density, and parcels along commercial mixed-use corridors had the greater potential for future development and the analysis assumed the maximum number of residential dwelling units per acre to determine the potential dwelling units. Building height was not a factor in limiting the maximum number of dwelling units unless existing parcels were already developed as mid- to high-rise, residential-only or mixed-used buildings with Type-1 construction (concrete and steel frame). Additionally, based on the analysis that was conducted for the Interim Height Ordinance the maximum residential densities in the commercial-mixed use areas of the Uptown community could be achieved for development projects with building heights that were 50 feet or greater. Comment noted. This comment does not relate to an inadequacy B1-6 with the PEIR; thus; a detailed response is not provided.

Gateway Specific Plan and CC.3-10 Zone

The BIA is supportive of the Uptown Gateway Council effort to create a Gateway District that would transform the commercial area into a true destination. Property owners within the 10 acres area in the heart of Hillcrest seek allowance for a Specific Plan and the establishment of CC.3-10 Zone unconstrained by height restrictions and supplemental residential regulations. It is critical to allow height in order to achieve attractive public realm, to provide density allowances that encourage smaller, more affordable housing units, and allow flexibility on parking standards that encourages the use of public transportation and compliance with the newly passed Climate Action Plan.

Potential Historic Districts

B1-7

- B1-8 The proposed potential historic district regulations are said to "provide supplemental protections until a more detailed historic district survey can be completed." This is viewed more as a means to impose still undefined historic designation regulations on private property by circumventing the traditional designation process. The result is that property owners would be guilty until proven innocent via a time consuming and undefined process to free themselves of unwarranted regulations.
- B1-9 A review of the 19 potential historic district boundaries as indicated on figure 6.7-4 appear to affect nearly 50% of the entire community and approximately 2,600 parcels. What efforts have been undertaken to notify the property owners within the potential district areas that their property will be subject to undefined 'interim protection,'? The resulting regulatory uncertainty could also negatively affect the total number of units that could be produced in Uptown further increasing the number of lost units.
- B1-10 The city should not impose 'interim regulations' on proprieties that may or may not meet historic standards.
- B1-11 The Building Industry Association of San Diego County appreciates the opportunity to comment on the PEIR. We consider the aforementioned modifications essential to comply with the goals and policies of the General Plan, City of Villages and especially the requirements of the Climate Action Plan.

Your favorable consideration of our comments and recommendations is greatly appreciated.

3

Sincerely atchew J. Adams Vice President

B1-7 Comment noted.

- B1-8 The supplemental development regulations to the Historical Resource Regulations are intended to protect the potential historic districts identified in the Historical Resources Survey and/or by the community and would only apply to structures that have been identified as being a contributing resource to a potential historic district. The regulations are not arbitrarily applied, rather they are applied only to specific properties that contribute to the character of the Potential Historic District. The traditional designation process would still occur in order to designate an official Historic District; however, the supplemental development regulations would protect potential historic districts until such time the formal designation review occurs.
- B1-9 The potential historic districts were identified through the efforts of the Historical Resources Survey and community outreach. The proposed Uptown CPU and associated discretionary actions including the supplemental development regulations have been made available to the public and through an extensive public review process.
- B1-10 Comment noted. This comment is an opinion of the BIA and does not suggest an inadequacy in the PEIR. Therefore, a detailed response is not required.
- B1-11 Comment noted.

LETTER



GHG emissions Citywide, but also mitigating the impacts of the City's General Plan. (CAP, p. 5). Eventually, this document will serve as a CEQA Qualified GHG Reduction Plan. In the interim, however, a project-level CAP consistency determination is an essential component of CEQA GHG impacts assessment. Inconsistency with a land use plan or policy intended to mitigate environmental impacts is likely to result in a finding of significant environmental impact. (See *Pocket Protectors v. City of Secramento* (2004) 124 Cal App.4th 903, 934 ["Because the land use policies at issue were adopted at least in part to avoid or mitigate environmental effects, we consider their applicability under the fair argument test with no presumption in favor of the City."]). B2-1 Please see responses to comments B2-2 through B2-9.

RESPONSE

B2-2 The comment states that the Climate Action Plan (CAP) does not currently serve as a CEQA Qualified GHG Reduction Plan, and that a project-level consistency determination is an essential component of CEQA GHG impacts assessment. The comment also states that inconsistency with a land use plan or policy is likely to result in a finding of significant environmental impact. The comment states that land use plans are an important part of achieving the GHG reductions identified in the CAP, and that the Uptown CPU fails to "ensure CAP consistency in 2020 and beyond."

The CAP was originally adopted in December 2015, and while it was anticipated that it would serve as a gualified GHG reduction plan for purposes of tiering under CEQA pursuant to CEQA Guidelines section 15183.5, it provided that future implementing actions were necessary in order to serve as such a plan. However, on July 12, 2016, the City Council adopted an amendment to the CAP, which included a CAP Consistency Checklist, and other amendments to the text of the CAP, which resulted in the CAP serving as a qualified GHG reduction plan. At that same time, the City Council also adopted a GHG Significance Determination Threshold (GHG Threshold). Following signature by the Mayor on July 19, 2016, the checklist and thresholds are being implemented immediately. The Uptown Community Plan Update (CPU) EIR tiers off of the GHG analysis set forth in the CAP Final EIR, which was certified on December 15, 2015, with an addendum certified on July 12, 2016 that specifically addressed the adoption of the GHG Threshold.

As discussed in PEIR Section 6.5, the proposed Uptown CPU is consistent with the adopted CAP, and contains goals and objectives that implement all of the five primary CAP strategies. Please see PEIR pages 6.5-7 through 6.5-11 for a discussion of consistency with the CAP strategies. It is concluded that the Uptown CPU would be consistent with each of the CAP strategies by:

B2-2 (cont.)

- Increasing the number of residential units and commercial development within the Transit Priority Areas (TPAs) within the community to support transit;
- Implementing transit-oriented development, particularly within and around two Community Villages and three Neighborhood Villages;
- Promoting pedestrian improvements in TPAs;
- Promoting sustainable building techniques for construction and operation of buildings that could include solar energy installations, electric vehicle charging stations, and solar water heating;
- Supporting waste reduction, recovery, and recycling;
- Encouraging the planting of native and drought-tolerant landscaping; and
- Increasing the tree canopy

Regarding the need to achieve overall compliance with the targets identified in the CAP, please also refer to CAP Chapter 3 which provides for annual monitoring and reporting to ensure CAP reduction targets are met. Please also see response to comment B2-3.

Rebecca Malone Climate Action Campaign CPU Comments July 8, 2016 Page 2

As the mechanism to achieve compliance with State reduction goals, the CAP requires vigilance and, in light of the looming 2020 reduction target, immediate implementation. Such implementation is especially important in the context of long-term land use plans such as Community Plan Updates (CPU). Unfortunately, the CPU EIRs fail to ensure the necessary CAP consistency in 2020 and beyond. As detailed below, the EIRs therefore reveal a significant environmental impact with respect to GHGs.

II. The EIRs Fail to Demonstrate Compliance with the Climate Action Plan

To determine whether impacts are significant under CEQA, all of the CPUs rely on a quantitative comparison of future buildout of current Community Plans with future buildout of the proposed CPUs. (See San Ysidro EIR, p. 5.4-16; North Park EIR, pp. 6.5-8-9; Golden Hill EIR, p. 7.5-8; Uptown EIR, pp. 6.5-7-8). Fundamentally, this analysis is improper.

First, the EIRs fail to address, much less analyze, environmental impacts pursuant to CEQA Guideline Section 15064.4(b). A lead agency should assess the significance of GHG emissions by considering the extent to which a project increases emissions compared to the existing environmental setting. (CEQA Guidelines §15064.4(b)(1)). All three Community Plan Update EIRs quantify existing emissions, as well as anticipated emissions for existing Community Plans at buildout, and emissions expected at buildout under the proposed CPUs.¹ (See Helix GHG Technical Report for San Ysidro CPU March 2016, pp. 15 and 27; RECON Supplemental Analysis to GHG Analysis for Uptown, North Park, and Golden Hill CPUs, May 16, 2016, pp. 6-8). Nonetheless, the EIRs fail to address the increase in emissions associated with the CPUs – especially in 2020 and 2035 when compared with the existing emissions – or explain why such increases are not significant.

Perhaps more importantly, the CPU EIRs and appendices do not put such increased emissions in context considering the Climate Action Plan reduction goals. The Climate Action Plan requires a **15 percent** reduction from 2010 baseline emissions by 2020, a **40 percent** reduction by 2030, and a **50 percent** reduction by 2035. (CAP, p. 21). Notwithstanding these ambitious CAP GHG reduction goals, and the CPUs' *quantitative* inconsistency with the CAP, the EIRs simply presume CAP consistency based on a *qualitative* analysis. The CPUs make this determination, in part, by claiming the CAP assumes growth based on the Community Plans in effect at the time the CAP was being developed. (See San Ysidro EIR, p. 5.4-8; Uptown EIR, p. 6.5-6; North Park EIR, p. 6.5-5; Golden Hill EIR, p. 7.5-5 ['The CAP assumes future population and economic growth based on the community plans that were in effect at the time the CAP was being developed. Therefore, community plan updates that would result in a

¹ The Helix GHG Technical Report for the San Ysidro CPU does not indicate in which year buildout occurs. Because construction emissions are annualized for thirty years, presumably buildout occurs in the next 30 years. (See Helix GHG Technical Report for San Ysidro CPU March 2016, p. 24).



B2-3 The commenter states that the Draft PEIR fails to assess the significance of GHG emissions by considering the extent to which a project increases emissions compared to the existing environmental setting. The existing GHG emissions are set forth on Draft PEIR pages 6.5-6 and 6.5-7, and specifically in Table 6.5-3. This information is also provided in Table 3 of the GHG Supplemental Report, which is included as Appendix E-2. To analyze the significance of GHG emissions, the Draft PEIR then estimates projected GHG emissions under the Uptown CPU as set forth in Draft PEIR Table 6.5-3 (this information is also provided in Table 7 of the GHG Supplemental Report, included as Appendix E-2). Table 6.5-3 of the Draft PEIR explicitly shows the increase from existing conditions and the proposed Uptown CPU (a total increase of 13,518 MT CO₂E increase over existing conditions).

RESPONSE

A two-step process was then used to determine whether the increase of 13,518 MT CO_2E in GHG emissions over existing conditions is significant. Whether that increase is significant was determined by (1) whether the Uptown CPU emissions would exceed the emissions in the Adopted Community Plan, and if so, whether the increase in GHG emissions is a direct result of implementing CAP strategies and the General Plan's City of Villages Strategy, and (2) whether the Uptown CPU is consistent with applicable policies and plans, including the CAP. Please see DEIR pages 6.5-6 through 6.5-11 for additional discussion.

As shown in the Draft PEIR, GHG emissions would increase over existing levels with build-out under both the Adopted Community Plan and proposed Uptown CPU due to the increase in development that would take place under both plans, but that the increase resulting from proposed Uptown CPU would be less than under the Adopted Community Plan. Looking at the Adopted Community Plan – not as a future baseline – but rather as a measure for determining significance of increased GHG emissions over existing emissions is instructive because it ensures that the

B2-3

B2-3 (cont.)

GHG emissions from the proposed Uptown CPU do not exceed the levels assumed in the CAP. Since implementation of the CAP is what ensures that the City meets Citywide GHG emissions reductions, it is important to look to whether any proposed changes to the assumptions in the CAP would affect the ability to achieve the CAP Citywide reductions. Because the proposed Uptown CPU would not increase emissions beyond what was assumed in the CAP – and in fact would reduce emissions – the proposed change in land uses would not significantly alter the assumptions in the CAP.

Additionally, with respect to Step 2 of the analysis, the Draft PEIR looked to see whether the proposed Uptown CPU would be consistent with the CAP and its strategies. Please see Draft PEIR pages 6.5-6 through 6.5-11 for additional discussion. Consistent with CAP Strategy 3, the Uptown CPU proposes increased density within TPAs in order to plan for reduced GHG emissions citywide. This necessary increase results in an increase in GHG emission levels in area, energy, waste, water, and construction emission sources (due to the increased density and new development); however, it results in a *decrease* in mobile emission sources. This decrease in mobile emissions is due to the continuing increase in regulations that improve vehicle efficiency. Additional decreases in mobile GHG emissions that are not reflected in the emission calculations would occur because density would increase in the TPAs, and trips would decrease due to increased use of alternative transportation modes. The document prepared by the California Air Pollution Control Officers Association (CAPCOA) entitled *Quantifying* Greenhouse Gas Mitigation Measures demonstrates that, by increasing transit accessibility, a shift in travel mode is facilitated along with reduced vehicle miles traveled (VMT). The effectiveness of these land-use strategies ranges from less than 1 percent up to a maximum 30 percent reduction in communitywide VMT and are not additive. For example, where high-density mixed-use development is located within a 5- to 10-minute walk from a transit station with

B2-3 (cont.)

high-frequency transit or bus service and is combined with walkable neighborhood design, a total VMT reduction up to 24 percent can be achieved. This is consistent with the CAP's GHG emissions reductions targets which are based on reductions in VMT from increasing the bicycling, walking, and transit mode shares within TPAs, and from decreasing commuter miles traveled, which results in a reduction in mobile emissions compared to the business as usual scenario. The Uptown CPU is consistent with the reductions estimates in the CAP because it promotes effective land use and implements the City of Villages strategy.

It is important to note that when modeling GHG emissions, the default CalEEMod trip generation rates and trip lengths were modeled for the existing condition, buildout of the Adopted Community Plan, and buildout of the proposed CPU. Actual trip lengths in San Diego County are shorter than these model default trip lengths. Additionally, as discussed, the CPU would reduce VMT due to the increased density in TPAs. These reductions are not reflected in the emission calculations presented in the DEIR. Thus, the GHG emission calculations included in the DEIR are conservative.

As shown in CAP Appendix A, the CAP VMT reductions in 2035 are Citywide reductions for labor force commuter trips. Some communities may have higher reductions, while some may have less due to a variety of factors, such as average commuter distance for a particular community. The CAP reductions are Citywide reductions, and due to the nature of community planning, are not always appropriate to be distributed equally amongst each community. For example, an increase in GHG emissions in one community may actually be necessary to alter the overall land use pattern in the City to achieve the reductions assumed for more effective land use Citywide.

B2-3 (cont.)

From a GHG perspective, increased density in a TPA correlates with lower GHG emissions. For example, CAPCOA's Quantifying Greenhouse Gas Mitigation Measures demonstrates that transit ridership increases with density, which justifies enhanced transit service. Higher density also allows City residents to take advantage of non-auto modes of transportation as such facilities become available. Therefore, focusing development inside TPAs rather than outside TPAs is consistent with CAP Strategy 3. This can be found on page 6 of the City of San Diego Climate Action Plan Consistency Checklist Technical Support Documentation. On the other hand, focusing development outside of a TPA would tend to be inconsistent with the CAP even though GHG emissions may not increase (because no new development would occur). Therefore, while looking at the increases or decreases in GHG emissions on a particular community plan update is instructive, it is not determinative as to overall Citywide consistency with the CAP.

In addition, the CAP recognizes that reductions can be achieved in multiple ways and that flexibility in implementation is necessary. As shown on pages 42 and 43 of the CAP, the annual monitoring and reporting would identify any potential deficiencies in reductions and the CAP could be amended to address those deficiencies. The annual monitoring and reporting program is the appropriate place to monitor Citywide GHG emissions reductions, not an individual community CPU EIR. Furthermore, new development within the Uptown CPU area that is subject to CEQA review would be required to complete the CAP Consistency Checklist to ensure project consistency with the CAP. As stated above, the City is implementing this requirement immediately for development projects.

Therefore, implementation of the Uptown CPU, in combination with implementation of the CAP overall, along with the CAP's annual monitoring and reporting, ensures achievement of the CAP's overall
B2-3 (cont.)

Citywide emissions reductions, and nothing in the land uses proposed in the Uptown CPU would be inconsistent with the promotion of effective land use to reduce VMT, or the ability to achieve the alternative mode shares assumed in the CAP.

Please also see response to comment B2-4 and B2-5.

- B2-4 The commenter states that the CAP relies on community plan updates to alter land use patterns and shift density to TPAs. The Uptown CPU is consistent with these CAP goals. Specifically, the commenter cites to CAP Strategy 3, which includes a supporting measure to locate a majority of all new residential development within TPAs. The Uptown CPU is consistent with this supporting measure in that it focuses new development and increased densities in two Community Villages (Hillcrest Core - West and Hillcrest Core - East) and three Neighborhood Villages (Mission Hills, Bankers Hill/Park West, and Middletown). These TPAs are served by several local and rapid bus routes, providing several options along Washington Street, University Avenue, Reynard Way, Fort Stockton Drive, First Avenue, Fourth Avenue, Fifth Avenue, Sixth Avenue, and Park Boulevard, as well as connections to the adjacent communities. Planned transit routes within the Uptown CPU area include BRT, light rail transit (LRT), and streetcar improvements. Please also see response to comment B2-3.
- Please see responses to comments B2-3 and B2-4. Regarding B2-5 modeling VMT reductions, please see DEIR Chapter 6.5 page 6.5-2 which discusses reductions in VMT. The Uptown CPU and associated discretionary actions propose an increase in multi-family residences. The VMT from residents of these new developments would be less due to the reduced trip lengths. Although this reduction was only counted for new development proposed under the proposed CPU and associated discretionary actions, this would reduce overall mobile emissions by 5.2 percent in the Uptown CPU area. This is supported by CAPCOA's Quantifying Greenhouse Gas Mitigation Measures measure LUT-4, Increase Destination Accessibility. Additionally, it is important to note that the GHG emission calculations did not take into account any reductions in vehicle miles traveled (VMT) that result from the transit-oriented land use pattern. For example, CAPCOA's Quantifying Greenhouse Gas Mitigation Measures identifies several features included in the

Rebecca Malone Climate Action Campaign CPU Comments July 8, 2016 Page 3

reduction in GHG at build-out compared to GHG emissions at build-out under the adopted Community Plan would result in further GHG reductions."). However, the phrase "2010 baseline emissions" cannot be read to mean a baseline defined by "emissions at buildout of Community Plans as they existed in 2010." This approach fails under the CAP and under CEQA.

Though the CAP assumed population growth in calculating *business-as-usual* emissions, nothing in the CAP or CAP appendices indicates GHG reduction modelling relied on existing Community Plans ever actually achieving this buildout. As such, the CPUs' reliance on full buildout at plan levels as a baseline is speculation and does not amount to substantial evidence. (Pub. Res. Code § 21082.2(c); CEQA Guidelines, § 15384(a) ['Argument, speculation, unsubstantiated opinion or narrative, evidence which is clearly erroneous or inaccurate, or evidence of social or economic impacts which do not contribute to or are not caused by physical impacts on the environment does not constitute substantial evidence.]).

Rather, the CAP's narrative goals and modelling appendices indicate the exact opposite is true: the CAP expects, and indeed relies on, Community Plan updates that will alter land-use patterns and shift density to Transit Priority Areas. The CAP includes goals to implement the City of Villages Strategy in Transit Priority Areas and promote effective land use to reduce vehicle miles traveled. (CAP, pp. 37-39). Specifically, a CAP supporting measure requires achievement of better walkability and transit-supportive densities "by locating a majority of all new residential development within Transit Priority Areas." (CAP, p. 39).

Parts of San Ysidro and the majority of Uptown, North Park, and Golden Hill are within Transit Priority Areas, but the EIRs and associated GHG analysis appendices fail to quartify: (i) how the CPUs implement the GHG emission reductions associated with CAP strategies, particularly increased density in TPAs; and, (ii) if such reductions meet the CAP 2020, 2030 and 2035 goals. Such quantitative consistency is particularly important here because to achieve the requisite reductions, the CAP relies heavily on Strategy 3, Bicycling, Walking, Transit and Land Use. Strategy 3 comprises one of the largest shares of local reduction actions. (CAP, p. 30). In the earlier years of the CAP, Strategy 3 is responsible for 36 percent of GHG emission reductions Citywide. Within Strategy 3, "Mass Transit" and "Promote Effective Land Use to Reduce Vehicle Miles Traveled" are two of the largest reduction sub-strategies. (*Id*).

Such modeling is achievable. The CAP models VMT (and associated GHG) reductions associated with each CAP strategy. (See CAP Appendix A, pp. A-31-A-38). Further, VMT reduction modeling was conducted as part of the CPU EIRs. Nonetheless, the EIRs fail to quantitatively bridge the analytical gap between: (i) the CPU VMT and associated GHG



B2-4

B2-5

RTC-31

B2-5 (cont.)

proposed Uptown CPU that would reduce VMT. CAPCOA measure LUT-1, Increase Density, is identified as means to reduce VMT and the corresponding GHG emission by up to 30 percent. By including a wide variety of land uses in the Hillcrest Core – West and Hillcrest Core - East Community Villages and the Mission Hills, Bankers Hill/Park West, and Middletown Neighborhood Villages, the CPU would achieve CAPCOA measure LUT-3, Increase Diversity of Urban and Suburban Developments (Mixed-Use), which is considered capable of reducing VMT and the corresponding GHG emission between 9 to 30 percent because residents would be in the same area as retail and office buildings. The concentration of development around the TPAs that are served by alternative transportation facilities would achieve CAPCOA measure LUT-5, Increase Transit Accessibility, which may result in up to a 24.6 percent reduction in VMT and corresponding GHG emissions. If the VMT reductions resulting from the inclusion of these factors into the proposed Uptown CPU were taken into account in the impact analysis, the reduction in GHG emissions in comparison with the Adopted Community Plan would have been even greater.

The commenter also notes that modeling for specific CAP goals is achievable. The City is continuing to explore a variety of ways to inform our data gathering and monitoring efforts for CAP implementation and GHG reductions. Rebecca Malone Climate Action Campaign CPU Comments July 8, 2016 Page 4

reductions; and, (ii) the correlating CAP GHG reductions. (See, for example, Uptown, North Park and Golden Hill CPU Appendix E.2. Attachment 1).²

This data is also a critical component of demonstrating CAP compliance. Without such data and analysis, numerous questions remain regarding CAP reduction measures. For example, if these four CPUs result in a net increase in emissions in both 2020 and 2035 compared to the 2010 baseline, and all other CPUs are similarly evaluated based only on an expected reduction in emissions compared to full buildout of adopted Community Plans – despite an increase from existing emissions – where will the reductions come from? If these four CPUs result in an increase in GHG emissions in 2020 and 2035, reductions from other future land use decisions will have to be even greater to make up for such increases, and it is unclear where such opportunities exist.

B2-6

B2-7

As the California Supreme Court recently found in *Center for Biological Diversity v*, Department of Fish & Wildlife ("Newhall Ranch") (2015) 62 Cal.4th 204, the EIRs here fail to bridge the analytical gap between the increase in CPU emissions and consistency with the CAP:

The analytical gap left by the EIR's failure to establish, through substantial evidence and reasoned explanation, a quantitative equivalence between the Scoping Plan's statewide comparison and the EIR's own project-level comparison deprived the EIR of its "sufficiency as an informative document" (Newhall Ranch, supra, 62 Cal.4th at 227, citing Laurei Heights Improvement Assn. v. Regents of University of California (1985) 47 Cal.3d 376, 392).

B2-8

As the planning mechanism to shape future development in these planning areas, the CPUs must result in CAP-mandated reductions *now*.⁵ Nevertheless, the EIRs contain no mention of the appropriate allocation of reduction measures attributable to CPU implementation. The CPUs' increase in GHG emissions is counterfactual to a CAP consistency determination. Because the EIRs fail to adequately address the "quantitative equivalence" between the City's CAP and the CPUs, the EIRs are insufficient and the CPUs will result in significant GHG impacts.

See also, Final Supplemental Environmental Impact Report for the Downtown San Diego Mobility Plan, SCH #2014121002, April 26, 2016, p.E-8,9 (reflecting achievement of active transportation mode share increases based on quantitative modeling).

³ The Supreme Court also posited that "a greater degree of reduction may be needed from new land use projects than from the economy as a whole" in light of the fact that new development is more easily designed to reduce GHG emissions. (New/hall Ranch, supra, 62 Cal.4th at 226).



The commenter asks how a community plan that increases GHG B2-6 emissions over existing conditions can result in GHG reductions. Please see response to comment B2-3. As discussed in response to comment B2-3, the reductions assumed from implementation of Strategy 3 come from a decrease in mobile source emissions tied directly to labor force commute trip length (see page A-31 through A-38 of Appendix A to the CAP). This increase in density in a community is anticipated to bring the labor force that is forecast to increase through 2035 to TPAs connected to employment centers in nearby communities. Implementation of the rest of the other CAP strategies would address the increase in other source emissions due to implementation of the CAP Strategy 3. In other words, any increases that result from the Uptown CPU also result in decreases in mobile source emissions. Therefore, even if a community plan increases overall GHG emissions within a particular community, if the community plan achieves mobile source reductions, that part of the assumed reductions in the CAP has been realized; implementation of the CAP overall is what would ensure that the City meets its targets identified in the CAP.

RESPONSE

It is also important to note that in the GHG emissions modeling done for the Adopted Community Plan and the Uptown CPU, the CalEEMod assumptions utilized to forecast GHG emissions were conservative and reflected the default from CalEEMod Version 2013.2.2. This approach to modeling does not take into account the emissions reductions of the Citywide ordinances and programs in the CAP to be implemented by the City, and which are not specifically relevant to the proposed CPU (i.e., Citywide energy, water or waste policies). For example, the first Goal under Strategy 2 of the CAP is to achieve 100 percent renewable energy Citywide by 2035. The CalEEMod energy default values are based on studies from the California Energy Commission, and not on achieving 100 percent renewable energy. Likewise, the Citywide efforts in CAP

LETTER

RESPONSE

B2-6 (cont.)

Rebecca Malone Climate Action Campaign CPU Comments July 8, 2016 Page 5

III. Conclusion

The current CPU EIRs fail to meet applicable CEQA mandates. The CPU EIRs must assess quantitative compliance with the Climate Action Plan, its reduction targets and goals. As drafted, the EIRs demonstrate a lack of compliance with Climate Action Plan goals because all four CPUs result in an increase in GHG emissions compared to baseline rather than a decrease of 15 percent by 2020, 40 percent by 2030, and 50 percent by 2035. Climate Action Campaign urges the City to conduct the requisite analysis and recirculate the EIRs for further public comment.

Thank you in advance for your consideration of our comments.

Sincerely,

COAST LAW GROUP LLP

Marco Gonzalez Livia Borak Attorneys for Climate Action Campaign

cc: Client

Strategy 1: Energy and Water Efficient Buildings would result in fewer emissions from sources associated with the provision of water, and CAP Strategy 4: Zero Waste would decrease the expected emissions from waste sources over what was accounted for in the CalEEMod modeling. In this manner, emissions projections for the Adopted Community Plan and for the Uptown CPU do not account for the GHG emissions reductions of the CAP. The emissions projections were produced to give a means of comparing the difference in land use emissions, i.e., the effect that changing the adopted land uses would have on the production of GHG emissions.

- B2-7 Please see response to comment B2-3.
- B2-8 Regarding the comment's footnote which suggests that greater GHG reductions may be needed for new development, please see the City's CAP Consistency Checklist, which is included as a CAP Appendix. The CAP Consistency Checklist provides for greater reductions from new development that is subject to CEQA. Regarding the Uptown CPU's overall consistency with the CAP, please see response to comment B2-3.
- B2-9 Please see response to comments B2-1 through B2-8.



LETTER

RESPONSE

Letter B3 Remembering the past... as we link to the future. hillcrest Marlon Pangilinan, Senior Planner The City of San Diego 1010 Second Avenue, MS-413 San Diego, CA 92101 Re: Uptown Community Plan Update (Project No. 380611 / SCH No. Pending) Comments on the Density Redistribution Alternative Dear Marlon, B3-1 This letter is to inform you that the Hillcrest History Guild supports the Density Redistribution B3-1 Alternative (DRA) which will move the proposed density in the community plan update from the heart of Hillcrest to Park Boulevard since this is where SANDAG has made an investment in the Rapid Bus. The Hillcrest History Guild (HHG) has participated in every phase of the Uptown community plan update (CPU), including providing recommendations throughout the process. Sincerely, Ann Garwood President, Hillcrest History Gruil www.Hillcrest History.org 501(c) (3) Nonprofit #16-1727118 info@HillcrestHistory.org (619) 260-1929 3065 Third Avenue #3 • San Diego, CA 92103

3-1 Comment noted. This comment states the Hillcrest History Guild's (HHG's) support of the Density Redistribution Alternative. The comment does not suggest an inadequacy in the Draft PEIR. The City appreciates the HHG's participation in the public review comment process and will consider all comments during the decision-making process.

	0	Email: Middletown92103@gmail.com		
	EXECUTIVE	August 2, 2016		
	COMMITTEE	RE: PROJECT NAME: Uptown Community Plan Update		
	Vanessa Harbert	PROJECT No. 21002568 / SCH No. Pending		
B4-1	neskis Webstor Matt Ramon Wallind Stirzzi Speedy Rice	COMMUNITY AREA: Uptown, District 3 Please note that the Uptown Community Plan Update has on its cover PROJECT PLAN 386611, but the notice for public comments on the Uptown	B4-1	The nui are diffe 210025 Final PE
		community Plan Update states to refer to PROJECT NO 2100250. The comments below refer to any Uptown Community Plan Update and the PEIR.		
		Draft Uptown Community Plan Update and PEIR		
B4-2		The Middletown Advisory Group became aware of the City's plan to update the Community Plan, which includes Middletown, in July 2016. To our knowledge, we are not aware that the residents of Middletown have been consulted in regards to their thoughts as to the needs of Middletown.	B4-2	Comme the ext
		The Middletown Advisory Group hereby submits its formal objection to the proposed expansion of India Street from West Olive Street and Washington Street as depicted in the June 10, 2016 (Uptown) Uptown Community Plan Update / Project No. 380611/2100258 Draft PEIR, including all appendices.		propos Outrea notes t
B4-3		India Street Traffic Concerns and Strong Opposition to Proposed Plan to Remove Parking and/or Sidewalks from India Street		Washin
		For many years, the traffic has been too dense, travels too fast and is a safety concern for the residents of Middletown as they enter or exit off of India Street by car, walk across India Street to the Pedestrian bridge over 1-5 or walk along the adjoining sidewalk on India Street. At various times different plans by the City of San Diego ("City") and other local agencies (e.g. San Diego Airport Authority and et al) show an elimination of parking along India Street and or a widening of India Street that would remove adjoining sidewalks and add an extra travel lane. Any adoptions of these plans will negatively impact the environment and quality of life for residents and		as a r recomn goals o respons the intr
		businesses along India Street and Middletown, The Middletown Advisory Group strongly oppose such plans. Accordingly, we request the City of San Diego issue a written statement and incorporate it by reference into the draft Uptown Community Plan Project No. 380611/2100258 Which states there are neither current nor will there be future plans for the removal of parking along India Street and/or sidewalks	B4-3	Comme India S
		from India Street from West Palm Street to Washington Street.		these r

- 4-1 The numbers referenced on the Uptown CPU and the Draft PEIR are different. The 380600 number is the project number, while the 2100258 number is reference to a billing number for City staff. The Final PEIR correctly refers to the project number.
- 4-2 Comment noted. Please refer to the Staff Report for a discussion of the extensive public outreach that has been done regarding the proposed Uptown CPU. Also refer to Section 4.2, Community Outreach and Plan Development, of the PEIR. This comment also notes that the Middletown Advisory Group (MAG) objects to the proposed expansion of India Street from West Olive Street and Washington Street. Although widening of India Street is identified as a mitigation measure in the PEIR, this measure is not recommended for implementation since it would conflict with the goals of the proposed Uptown CPU. Please refer to the master response regarding India Street mitigation measures included in the introduction to these responses to comments.
- 34-3 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

Middletown Advisory Group

The traffic on India Street between West Olive and Sassafras Streets as well as Sassafras Street between Pacific Highway and India Streets has increased significantly with the opening of the Rental Car Center (RCC). Since the RCC opened, traffic data indicates an increase of approximately 3,000 vehicles daily.

B4-4 Specific Notations and Comments regarding the Uptown Community Plan Update / Project No. 380611/2100258 / Draft PEIR, including all appendices related to India Street between West Redwood and Sassafras Streets that we object to and believe will be detrimental to the residents of Middletown

> Table S-1 Summary of Significant Environmental Impacts notes: India Street; Sassafras Street to Redwood Street (Impact 6.2-19)

Mitigation: TRANS 6.3-19: India Street (Impact 6.3-19), b. Sassafras Street to Redwood Street: Widen the roadway to a 3-lane one-way collector.

Middletown Comment: We strongly object to any widening of Iudia Street and to the inference that the two-way collector status between Sassafras and Redwood will be eliminated. The two-way collector is essential to residents of Spruce (and Redwood). If eliminated, Spruce residents will be required to travel to Laurel on Kettner and return up India for access to their homes. This will unnecessarily increase traffic on Laurel and Iudia Streets and add additional fuel use and emissions using the longer route.

B4-5 Table 6.3-7 Build-out Summary of Intersection Analysis - Uptown #20 & 21.

Middletown Comment: If this analysis was done before the San Diego International Airport Rental Car Center (SDIA- RCC) was opened and would be inaccurate. The SDIA RCC has added a substantial number of cars – daily- to the intersections of Sassafras Street at Kettner, as well as onto India Street from the I-5 North exiting onto India and the SDIA RCC traffic exiting the SDIA-RCC using Sassafras to travel to the I-5 North on ramp. The current traffic pattern has caused a significant increase traffic delays and regularly blocked intersections.

The Middletown Advisory Group has recommended the SDIA-RCC traffic exiting the SDIA-RCC be directed to use the northbound Pacific Highway to Washington Street to access the I-5 north. Given the size and length of Pacific Highway to Washington Street, this would reduce congestion and traffic delays. Additionally, traffic exiting the SDIA-RCC headed south could either continue to be directed onto Sassafras to Kettner to access the I-5 south on-ramp or directed south on Pacific Highway to access the downtown area of San Diego.

No improvements were made to India, Sassafras or Kettner Streets when the SDIA-RCC was open and its traffic directed onto these streets. Pacific Highway (north and south bound) and Washington Streets are much better equipped to handle the additional traffic from the RCC as opposed to India Street.

B4-6 Table 6.9-8 Buildout Summary of Roadway Segment Analysis: Uptown - India Street:

Middletown Comment: Same as above (6:3-7) but the India Street Buildout CPU figures demonstrate that adding an additional traffic lane will not solve the LOS problem and only make life much worse for local residents. Rerouting traffic from the Airport and RCC to Pacific Highway and Washington Street to access the I-5 north is the only viable, effective, and inexpensive short term solution

- B4-4 Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- B4-5 This comment makes reference to the recent opening of the San Diego International Airport Rental Car Center and that fact that the traffic counts conducted for the Traffic Impact Analysis would not have included the increased traffic from the Rental Car Center. Section 15125 of the CEQA Guidelines states that the baseline physical conditions by which a lead agency determines whether an impact is significant is normally established at the time the notice of preparation is prepared. The notice of preparation was issued on December 23, 2013, long before the 2016 opening of the Rental Car Center. Additional explanation regarding the baseline traffic counts is provided in Section 6.3.3, Impact Analysis, of the PEIR. Additionally, this comment requests that vehicles exiting the Rental Car Center be redirected to reduce traffic delays. However, this is outside of the boundaries of the Uptown community, and outside the scope of the Uptown CPU and PEIR.
- B4-6 This comment makes reference to the significant impacts to traffic identified along India Street. The PEIR appropriately discloses the findings of significant and unavoidable impacts to intersections and roadway segments. In addition, rerouting traffic from the airport and Rental Car Center, as suggested by the MAG, is outside the scope of this PEIR and would not be an enforceable mitigation measure because it is not related to an impact resulting from the proposed Uptown CPU and associated discretionary actions.

ō.

Middletown Advisory Group

Email: Middletown92103@gmail.com

Impact 6.3-19, page 6.3-38: "The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of India Street from Glenwood Drive to Redwood Street."

Middletown Comment: We agree that the changes will have a traffic impact that is detrimental for the residents living off India from Glenwood Drive to Redwood Street and oppose any changes widening India Street. Rerouting traffic from the Airport and RCC to Pacific Highway to Washington is the only short term solution as noted above.

B4-7 <u>6.3.5.2 Segments and 13.2.1.2 Roadways Segments: While the following intersection mitigation</u> measures would reduce potentially significant impacts, only TRANS 6.3-7d., TRANS 6.3-24a., and <u>TRANS 6.3-27</u> are proposed as part of the proposed Uptown CPU and associated discretionary actions.

> Middletown Comment: We support the exclusion of Trans/Impact 6.3-19 on page 6.3-50 and again at the Uptown Community Plan Update PEIR pages 11-3 / 4, with a request that this be clearly designated as not an option or proposal under consideration.

B4-8 Appendix C – Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions).

> Table 13 Summary of Improvement Evaluation Uptown: This references India Street between Redwood and Sassafras Streets as "one way Collector" and removal of parking and buildings.

Middletown Comment: We strongly object to any widening of India Street especially with such extreme damage to the neighborhood businesses, and to the inference that the two-way collector status between Sassafras and Redwood Streets will be eliminated. The two-way collector is essential to residents of Spruce and Redwood Streets. If eliminated, Spruce residents will be required to have! to Laure! Street on south Kettuer Street and return north up India Street for access to their homes. This will unnecessarily further increase traffic on India Street. We recognize that it says the Consultant does not recommend this change but that is not binding on the City and the wood changes are of considerable concern. We want this removed from consideration.

B4-9 Figure 64 of Appendix C: Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions).

> Middletown Comment: We strongly object to any widening of India Street. While recognizing that the Consultant says "These changes in Geometry are not recommended" and that recommendation is not binding on the City, the road changes are of considerable concern. We want our objection to this possible change clearly stated. We request the City to delete or issue a written statement, which will be incorporated by reference to the Uptown Plan stating that there are no current or future plans to remove parking and or sidewalks from India Street from Pain Street to Washington Street.

B4-10 Error Regarding Spruce Street, et al throughout Uptown Plan

The maps used throughout the U ptown Plan and appendices incorrectly depict Spruce Street at India Street and connecting streets off Spruce to show several alternative routes available to the 2004 residents living on Spruce and the adjoining streets (Ibis, Horton, Upas, Jackdaw) for ingress and egress. While it is unclear what effects on planning may have flowed from planners and consultants mistakenly believing that residents on Spruce et al. have alternative routes, it needs to correctly depict that these residents have no other ingress or egress but from India

- B4-7 Additional clarification regarding the measures not carried forward as part of the proposed Uptown CPU has been added to the applicable locations in Section 6.3, Transportation and Circulation, of the PEIR.
- B4-8 Comment noted. As noted in Table 13 of Appendix C, Mobility Study, the improvements listed for India Street are not recommended as part of the proposed Uptown CPU and are not proposed for implementation.
- B4-9 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- Comment noted. The proposed Uptown CPU maps appear to be B4-10 correct. West Spruce Street does connect to India Street. Access to all the other streets mentioned in the comment letter is solely from West Spruce/India Street. Because of the map's scale, the gap between West Spruce Avenue and Horton Avenue is difficult to see, but there is a gap in the figure and it was considered in the traffic analysis. However, the map in the Mobility Study does incorrectly show that a connection between West Spruce Avenue and Horton Avenue that will be corrected. While there was an error in the Mobility Study map, the model used in preparation of the analysis of potential impacts of the proposed Uptown CPU for traffic circulation did not include any connections with West Spruce Avenue that would provide additional ingress/egress to West Spruce Avenue other than India Street. Furthermore, the referenced Mobility Study Improvements (U17A and U17B) would be inconsistent with the proposed Uptown CPU polices and thus, would not be implemented due to infeasibility.

3

LETTER

RESPONSE

Middletown Advisory Group Email: Middletown92103@gmail.com

Street. Please correct this map error and do a proper re-analysis of the impact to the residents who rely solely on Spruce Street for access to their homes. The error and correction is atlached to show the seriousness of the map error.

Respectfully,

Middletown Advisory Group

4

LETTER

RESPONSE



B5-5 (cont.)

development area. For example, Policy MO-7.1 calls for the City to implement creative parking programs with new development such as in-lieu programs managed by the community parking district that would contribute to the construction of new parking structures. Additionally, Policy MO-7.2 requires the City to consider public parking structures with shared parking arrangements to supplement parking needs and serve Uptown businesses. Policy MO-7.3 encourages implementation of below-ground parking structures for new development with inadequate surface parking space and Policy MO-7.13 calls for the provision on-street parking on all streets to support adjacent uses. These, along with multiple other policies aimed at providing adequate parking for residents, visitors, and customers of businesses provided in the CPU, would help to ensure adequate parking for all land uses within the community.

- B5-6 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- B5-7 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- B5-8 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- B5-9 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- B5-10 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- B5-11 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- B5-12 Comment noted. All comments and responses will become part of the public record, and all comments will be considered during the decision-making process.

Page 2 of 2

- B5-6 Urban Design: The BID Board supports an eclectic variety of buildings in its commercial and mixed-use areas, as set forth in the proposed community plan update. Further, it agrees as commercial corridors continue to redevelop and add residential densities; the provision of public (and private) open space becomes more important; these areas can provide needed open space. The BID Board opposes the proposed height limitations in our BID and, instead, supports the 150 feet maximum from the adopted Uptown Community Plan of 1988.
- B5-7 Economic Prosperity: The BID Board supports full utilization of economic development tools and programs to attract and retain small businesses, through the maintenance and enhancement of commercial areas.
- B5-8 Recreation: The BID Board supports the proposed park sites within the BID.
- B5-9 Conservation: Many elements of conservation and sustainability have much broader geographic and political relationships and may be more suited to implement on a citywide or even regional basis.
- B5-10 Historic Preservation: The BID Board agrees the heritage of the community must be preserved. Further, it believes the current historic districts in Mission Hills and the City's historic review process are on sound footing and will protect the heritage of the neighborhood. At this time, the BID Board sees no need to increase or expand historic preservation areas in Mission Hills.
- B5-11 Implementation: The BID Board supports the proposed West Lewis Street Commercial Area implementation plan. The BID Board is opposed to any Community Plan Implementation Overlay Zone (CPIOZ) restriction limits where the City may require discretionary review. The BID Board recommends a height limit of 150 feet maximum for commercially-zoned and mixed-use zones, including in the Reynard Way Commercial Area Expansion.
- B5-12 In conclusion, the BID Board reached its decision based on its purpose and mission to improve the overall business climate of the commercial area and its continuing belief in the many neighborhood organizations whose membership will scrutinize any development within the BID boundaries with the end goal of achieving the best outcome for Mission Hills.

Thank you for this opportunity for the Mission Hills BID to share its thoughts and concerns on the Proposed Uptown Community Plan Update.

Sincerely,

Troy Curnett, President

Mission Hills Business Improvement District

attachments (2 pages)

cc: City of San Diego Mayor Kevin Faulconer City of San Diego Councilmember Todd Gloria City of San Diego Planning Director Jeff Murphy Mission Hills Heritage Chairperson Barry E. Hager Mission Hills Town Council President Belinda Smith Luctown Planners Chairperson Leo Wilson

525 West Washington #2228 + San Diego, CA 92101 + Mission Hilsbitzegman.com + Mission Hilsbitzegman.com

LETTER

RESPONSE



LETTER

RESPONSE



1999: Public Lot-On Goldfinch at Ft. Stockton facing East



1999: French Bakery is now 1Mission on Goldfinch



1Mission, the development occupying the block of Goldfinch, Falcon, Ft Stockton and West Washington, provides a broad spectrum of housing opportunities, promotes mixed-use occupancies along Mission Hills' transit corridors, and higher density residential development which activates our commercial center. 1Mission changed the community of Mission Hills for the better and would not be permitted under the proposed lower densities.

LETTER

Letter B6 325 W. Washington Street, Suite2, #221 San Diego, CA 92103 619-497-1193 www.MissionHillsHeritage.org

--A community organization dedicated to preserving the character. charm and historical resources of the Mission Hills neighborhood.

July 25, 2016

Kurtis Steinert, Senior Environmental Planner Marlon Pangilinan, Senior Planner The City of San Diego 1010 Second Avenue, MS-413 San Diego, CA 92101

Re: Draft Program Environmental Impact Report (PEIR)—Uptown Community Plan Update (Project No. 380611 / SCH No. *Pending*) Comments on the Density Redistribution Alternative

Dear Mr. Steinert and Mr. Pangilinan:

- B6-1 Mission Hills Heritage (MHH) is a non-profit organization representing the Uptown neighborhood of Mission Hills, with a mission to preserve and protect the charm, character and historic resources of Mission Hills. We have participated in every phase of the Uptown community plan update (CPU), including providing recommendations throughout the process.
- B6-2 This letter is to inform you that MHH supports the Density Redistribution Alternative (DRA) as described in the PEIR.
- B6-3 Our reasons for supporting the DRA include that this alternative more closely aligns with our previous recommendations that the maximum residential density along Reynard Way should be 29 du/ae, not the higher 44 du/ae set forth in the most recent draft of the Uptown CPU. The DRA also more closely aligns with input from residents of Hillcrest and other Uptown neighborhoods over the last several years. We agree that the DRA is the "environmentally superior alternative" as noted in the PEIR. We disagree, however, with the statement at page 10-26 of the PEIR that the DRA would reduce residential density along "transit commercial nodes" since this incorrectly assumes that Reynard Way is along a transit commercial node. While portions of Reynard Way are shown as a Transit Priority Area (TPA) in the TPA map dated May 31, 2016 included as Appendix B of the City's adopted Climate Action Plan, we believe these

B6-1 Comment noted. The City appreciates Mission Hills Heritage's (MHH's) participation in the CPU and public review comment process.

RESPONSE

B6-2 Comment noted.

B6-3

This comment suggests that the analysis of the Density Redistribution Alternative inaccurately concluded that the alternative would reduce residential density along transit commercial nodes because Reynard Way is not a transit commercial node. However, as discussed in the Introduction and Economic Prosperity Element of the proposed Uptown CPU, neighborhood pedestrian commercial uses are located along Reynard Way. While the proposed Uptown CPU does not identify Reynard Way as a "Neighborhood Center/Node", it is identified as a "Connector Street" and is the main north-south residential collector in the southern portion of the community. In addition, as noted in this comment, the roadway is identified in Appendix B of the City's CAP as within a San Diego Association of Governments Transit Priority Area (TPA). As described in Section 6.4, Transportation and Circulation, Reynard Way is lined with sidewalks and curbs on both sides of the street, is identified by the City as a Class III (Bike Route) facility, and is served by local and rapid bus routes. Several transit stops are located along Reynard Way, and the Middletown Trolley Station is accessible from Reynard Way by way of Laurel Street and Kettner Boulevard. Because TPAs are areas identified for focused funding and other policy tools to further promote non-vehicular transportation, and the proposed Uptown CPU policies will continue to improve the existing alternative transportation infrastructure, Reynard Way's success as a transit commercial node will only increase. In addition, the Density Redistribution Alternative would not only result in lower densities along Reynard Way, but also along other transit commercial nodes, further inhibiting the ability of the Density Redistribution Alternative to meet the CPU objectives.

LETTER

RESPONSE

Kurtis Steinert, Senior Environmental Planner Marlon Pangilinan, Senior Planner July 25, 2016 Page 2

areas are misclassified. The TPA Map shows that the northern portion of Reynard Way between Sutter Street and Curlew Way is within one-half mile of a Major Transit Stop located at Goldfinch Street and Washington Street. Relatively steep topography, however, makes it unlikely that residents within this area would actually walk to the transit stop. The TPA Map shows the southern portion of Reynard Way to be within one-half mile of a trolley stop to the west and transit stops along First Avenue to the cast. However, there are no connecting streets or pedestrian ways between Reynard Way and the trolley stop or First Avenue.

B6-4 Additionally, Reynard Way is not listed as an Existing/Planned or Potential Metropolitan Center, Urban Center, Town Center, Community Center, Mixed Use Transit Corridor, Special Use Center, or Rural Village on SANDAG's Smart Growth Concept Map (Draft April 2016). Thus, to the extent that the DRA would shift density from Reynard Way to Park Boulevard, the DRA would shift density from area that is <u>not</u> a transit corridor, onto an actual transit corridor.

B6-5 Thank you for the opportunity to review and comment on the PEIR. We will provide comprehensive comments to the PEIR in a separate letter.

Sincerely,

Mission Hills Heritage

Barry E. Hager, Board Chair

CC: Tait Galloway, Principal Planner, City of San Diego Adriana Martinez, Representative, Council District 3 Leo Wilson, Chair, Uptown Planners Belinda Smith, President, Mission Hills Town Council

- B6-4 See response to comment B6-3.
- B6-5 Comment noted.

I FTTFR

RESPONSE

		-	
	Letter B7 Sign Diego, CA 92103 619-497-1193 www.MissionHillsHeritage.org		
B7-1 B7-2	<text><text><text><text><text><text><text><text><text></text></text></text></text></text></text></text></text></text>	B7-1 B7-2	Comment noted. The City appreciates Mission Hills Heritage's (MHH's) participation in the CPU and public review comment process. The City does not agree there is a deficiency in the mitigation measures in the PEIR. While the policies and design guidelines of the community plan would not apply in a ministerial review process, the policies alone do not reduce the significance of impacts. Application of applicable zoning and Land Development Code (LDC) regulations would apply to future development and reduce the significance of impacts. Thus, development allowed under a ministerial process is restricted by its zoning and development regulations of the Land Development Code which would ensure changes to community character would be less than significant. Development within areas subject to CPIOZ-Type A regulations that does not meet the criteria under CPIOZ Type A would be required to meet findings for a Site Development Permit related to a proposed project and would be reviewed against proposed CPU policies. Additional detail has been added to Section 6.2 of the Final PEIR, under Issue 2 to further clarify what the anticipated physical changes would be in relation to the height of future development relation to the existing condition. As shown, impacts associated with future ministerial development would be less than significant with application of applicable zoning and LDC regulations.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 2

Overarching Deficiency of the PEIR

Throughout Chapter 6.0 Environmental Analysis – Uptown Community Plan Update, references are made to policies and/or design guidelines that supposedly ensure that impacts from development projects carried out in compliance with the Community Plan would be less than significant, and therefore no mitigation would be required. These policies and design guidelines are also cited in the Alternatives Section as arguments as to why the Proposed Project would have less impact than the other Alternatives.

Presumably the policies and design guidelines would be considered for any project that requires a discretionary Process 3 Site Development Permit in order to exceed the CPIOZ Type A Height Limits

However, no implementing mechanism or enforcement measure is included in the Uptown Community Plan Update (CPU) or associated discretionary actions (the "project") that would require the policies and/or design guidelines to be considered for the majority of development projects that would require only ministerial approval (e.g., building permit).

This deficiency can and should be remedied by incorporating the following feasible mitigation measure into the project:

All commercial and multi-family zoned areas shall be included in CPIOZ Type A pursuant to Municipal Code Chapter 13, Article 2, Division 14: Community Plan Implementation Overlay Zone. If the proposed development complies with the policies and design guidelines in the Uptown Community Plan, no discretionary permit would be required to address this issue (emphasis added). If the proposed development does not comply with the policies and design guidelines in the Uptown Community Plan, a Site Development Permit/Process Three shall be required in accordance with Municipal Code \$131.1402.

The above language should also be included in the Uptown Community Plan Update.

Specific Comments

Following are our specific comments on the Draft PEIR for the Uptown Community Plan Update and associated discretionary actions.

B7-3 Excerpt #1 - TABLE OF CONTENTS, Effects Not Found to be Significant, page ii:

"Energy "

Comment // I

Should read "Population and Housing."

B7-3 The Table of Contents of the PEIR has been updated in response to this comment.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 3

B7-4

Excerpt #2 - Executive Summary, S.3 Areas of Controversy, page S-3:

"Although there are no clear-cut areas of controversy"

Comment #2

The above statement ignores (or perhaps pre-dates?) the release of the PEIR. A major controversy arose when the June 2016 Final Draft of the Uptown Community Plan was released concurrently with the Draft Program EIR (PEIR), thus precluding input on the CP Final Draft that was analyzed by the PEIR. After working with the council-recognized community planning group. Uptown Planners, and other community organizations for over 7 years, the Planning Department released the June 2016 Final Draft differed substantially in several areas from previous drafts. At the June 27, 2016, meeting of the Uptown Planners, attended by numerous angry residents, the community planning group voted 8-1 (with two abstentions) to send the draft plan back to the City to have recommendations from a previous draft restored.

The PEIR must recognize this controversy.

B7-5 Excerpt #3 – Executive Summary, S,5 Summary of Significant Impacts and Mitigation Measures that Reduce the Impact. Table S-1, Summary of Significant Environmental Impacts, Visual Effects and Neighborhood Character, Page S-9:

"Environmental Issue

Would the project result in a substantial obstruction of a vista or scenic view from a public viewing area as identified in the community plan?

Results of Impact Analysis

The <u>policies</u> (emphasis added) of the proposed Uptown CPU and associated discretionary actions would enhance public view corridors through use of setbacks and design improvements along major roadways within the CPU area. Therefore, public view impacts would be less than significant, and no mitigation would be required."

Comment #3

No implementing mechanism is included in the project that would require the policies to be considered for any action, discretionary or ministerial (e.g., building permit). Therefore, it cannot he said that the policies would enhance public view corridors through use of setbacks and design improvements along major roadways within the CPU area. And, therefore, it cannot be said that public view impacts would be less than significant, and that no mitigation would be required.

A project, particularly a ministerial action, undertaken in accordance with the land use recommendations could result in significant impacts by substantially obstructing a vista or scenic view from a viewing area as identified in the community plan.

B7-4 Section S.3, Areas of Controversy, of the PEIR has been updated to identify stakeholder concerns and comments regarding the distribution of densities as an area of controversy.

The use of proposed policies to reduce potential impacts is B7-5 appropriate at this program-level review of the proposed CPU, which would guide all future development within the community. While the focus of the PEIR is on how the policies would minimize impacts, other existing regulations would also reduce potentially significant impacts such as the LDC that has requirements for setbacks and encroachments into environmentally sensitive lands which would apply to all development including ministerial projects. Though the PEIR analyzes the potential environmental effects associated with build-out of the proposed CPU and associated discretionary actions, it does not analyze or propose any one specific development project. Future development projects implemented in accordance with the proposed Uptown CPU would be subject to a separate project-level environmental review and would be required to be consistent with the proposed Uptown CPU land use plan, applicable policies, development regulations, and design guidelines. Therefore, projects implemented in accordance with the proposed Uptown CPU are not anticipated to result in significant impacts to visual effects and neighborhood character.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 4

The impact is significant and unmitigated. The measure presented in Comment #1 could mitigate the impact totally or partially depending upon the effectiveness of the policies employed.

B7-6

Excerpt #4 – Executive Summary, S.5 Summary of Significant Impacts and Mitigation Measures that Reduce the Impact. Table S-1, Summary of Significant Environmental Impacts, Visual Effects and Neighborhood Character, Page S-10:

"Environmental Issue

Would the project result in a substantial alteration (e.g. bulk, scale, materials or style) to ihe existing or planned (adopted) character of the area?

Results of Impact Analysis

The proposed Uptown CPU includes <u>policies</u> (emphasis added) that would encourage residential and mixed-use development that would be consistent with the existing neighborhood character and impacts would be less than significant. No mitigation would be required,"

Comment #4

No implementing mechanism is included in the project that would require the policies to be considered for any action, discretionary or ministerial (e.g., building permit). Therefore, it cannot be said that the policies would encourage residential and mixed-use development that would be consistent with the existing neighborhood character. And, therefore, it cannot be said that impacts to neighborhood character would be less than significant, and that no mitigation would be required.

A project, particularly a ministerial action, undertaken in accordance with the land use and density recommendations could result in significant impacts by substantially altering the existing or planned character of the area.

The impact is significant and unmitigated. The measure presented in Comment #1 could miligate the impact totally or partially depending upon the effectiveness of the policies employed.

See Excerpt #14 and Comment #14 for specific areas in Mission Hills that would be impacted the most.

B7-7 <u>Excerpt #5</u> – Executive Summary, S.5 Summary of Significant Impacts and Mitigation Measures that Reduce the Impact. Table S-1, Summary of Significant Environmental Impacts, Visual Effects and Neighborhood Character, Page S-10.

"Environmental Issue

B7-6 See response to comment B7-2 and B7-5. Additionally, it is not clear from the comment how one individual project could substantially alter the character of an area.

B7-7 See response to comment B7-2 and B7-5.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 5

Would the project result in a substantial change in the existing landform?

Results of Impact Analysis The proposed Uptown CPU includes <u>policies</u> (emphasis added) that would protect and preserve existing landforms (i.e., canvons and open space areas)."

Comment #5.

No implementing mechanism is included in the project that would require the policies to be considered for any action, discretionary or ministerial (e.g., building permit). Therefore, it cannot be said that the policies protect and preserve existing landforms (i.e., canyons and open space areas). And, therefore, it cannot be said that landform alternation would be less than significant, and that no mitigation would be required.

A project, particularly a ministerial action, undertaken in accordance with the land use recommendations could result in significant impacts by substantially altering the existing landform.

The impact is significant and unmitigated. The measure presented in Comment #1 could mitigate the impact totally or partially depending upon the effectiveness of the policies employed.

B7-8 <u>Excerpt #6</u> - Executive Summary, S.5 Summary of Significant Impacts and Mitigation Measures that Reduce the Impact. Table S-1, Summary of Significant Environmental Impacts, *Historical Resources*, Page S-26:

"Environmental Issue

Would implementation of the proposed project result in an alteration, including the adverse physical or aesthetic effects and/or the destruction of a historic building (including an architecturally significant building), structure, object, or site?

Results of Impact Analysis

Implementation of the proposed Uptown CPU and associated discretionary actions could result in an alteration of a historic building, structure, object, or site. This impact would be potentially significant.

Impact Level After Mitigation Significant and Unavoidable"

Comment #6

Mission Hills Heritage concurs that the impact is significant, but we do not concur that the impact is unavoidable. We have serious concerns over the past actions of the City that have hindered mitigating this impact. Moreover, the impact can be mitigated by implementing the

B7-8 Feasible mitigation (mitigation measure HIST 6.7-1) was applied. However, as discussed in Section 6.7.7, Significance of Impacts after Mitigation, even with implementation of the mitigation framework, the degree of future impacts and applicability, feasibility, and success of future mitigation measures cannot be adequately known for each specific future project at this program level of analysis. Therefore, while the proposed mitigation is anticipated to reduce impacts at the project-level, it cannot be certain until each project is designed and brought forward for consideration.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 6

identified potential historic districts concurrent with adoption of the plan. Please see Excerpt #18 and Comment #18 for further discussion.

B7-9 Excerpts #7 – Executive Summary, S.5 Summary of Significant Impacts and Mitigation Measures that Reduce the Impact. Table S-1, Summary of Significant Environmental Impacts, Public Services and Facilities, Pages S-46 and S-47:

"Environmental Issue

Would the project promote growth patterns resulting in the need for and/or provision of newer physically altered . . . parks or other recreational facilities, . . . the construction of which could cause significant environmental impacts in order to maintain service ratios . . . or other performance objectives? (page S-46)

Results of Impact Analysis

Regarding park and recreational facilities, there is an existing and projected deficit in population-based parks, which is an adverse impact, but not considered significant at the program level. Implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of population based parks in the CPU area, but does not propose construction of new facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to parks and recreation, and no mitigation is required. (pages S-46 and S-47)

Mitigation None Required (page S-46)

Impact Level After Mitigation Less Than Significant" (page S-46)

Comment #7

To say that the impact is "not considered significant at the program level" begs/ignores the question. Until adequate population based park sites have been identified, the impact relating to construction is potentially significant and unmitigated.

B7-10 Excerpt #8- Chapter 1 Introduction, third paragraph, page 1-1:

"The proposed Uptown CPU contains development <u>design guidelines</u> (emphasis added), as well as <u>policies</u> (emphasis added) related to a range of topics included in each section B7-9 Impacts related to parks and recreation facilities from implementation of the Uptown CPU would be less than significant because implementation of the proposed CPU includes policy support for increasing the acreage of population based parks in the CPU area, which is further supported by the proposed Uptown IFS. See response to comment B7-5 regarding the program-level analysis. The proposed Uptown CPU is not proposing any specific development project; rather, it creates a policy framework to guide future development and encourage implementation of the project's primary objectives within the Uptown community, which include increasing recreation opportunities and new public open spaces.

B7-10 See response to comment B7-2 and B7-5.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 7

Comment #8

No implementing mechanism is included in the project that would require the policies or design guidelines to be considered for any action, discretionary or ministerial (e.g., building permit). This is a fatal deficiency of the project.

B7-11 Reference #9 - Chapter 2 Environmental Setting, 2.3.1.2 Adopted Uptown Community Plan, page 2-9 briefly discusses the adopted 1988 Uptown Community Plan.

Comment #9

Please include the housing and population build-out projections as stated in the adopted 1988. Uptown Community Plan to serve as a comparison for the proposed CPU.

B7-12 Excerpt #10 - Chapter 3 Project Description, 3.4.2 Land Development Code Amendments. 3.4.2.2 Amendment to the Uptown Community Plan Implementation Overlay Zone Boundaries, Pages 3-18 and 3-19:

> "The mapped boundaries of the existing CPIOZ would be amended within the Uptown community... to address ministerial review of building height limits within Hillcrest and Mission Hills (CPIOZ-Type A)... CPIOZ-Type A identifies areas within the community where ministerial approval is granted for development that does not exceed 50 feet within Mission Hills and 65 feet in Hillerest and Bankers Hill/Park West."

Comment #10

The above statement implies that development within Mission Hills will not exceed 50 feet and that development within Hillcrest and Bankers Hill/Park West will not exceed 65 feet.

However, according to Municipal Code Article 2, Division 14, Table 132-14B (Community Plan Implementation Overlay Zone Applicability) it states in Column 1 (Type of Development Proposal):

"(3) Any development within the boundaries shown on a map identified in Section 132.1402, where the map shows "Type A" and the proposed development does not comply with the development standards or criteria in the applicable community plan."

and states in Column 3 (Required Permit Type/Decision Process):

"Site Development Permit/Process Three."

Thus, heights over 50 feet in Mission Hills and 65 feet in Hillcrest and Bankers Hill/Park West could be, and undoubtedly will be, permitted with a Site Development Permit under a Process

- B7-11 The housing and population build-out projections of the adopted Uptown Community Plan have been added to Section 2.3.1.2 of the PEIR in response to this comment.
- B7-12 While it is possible for heights over 50 feet in the Mission Hills neighborhood and 65 feet in Hillcrest and Bankers Hill/Park West neighborhoods may be permitted, the Site Development Permit review process would require a consistency review of the project with the adopted CPU; thus, consistency with the Urban Design element would ensure a significant impact to the existing scale of older neighborhoods would not result.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 8

Three. This increased height would result in a significant impact to the existing scale of these older neighborhoods.

The PEIR should acknowledge this possibility and consider it when analyzing Neighborhood Character.

B7-13 Excerpt #11 - Chapter 3 Project Description. 3.4.3 Zone Changes. 3.4.3.1 Citywide Rezoning, Page 3-19:

> "The implementation program for the Community Plan would replace the Mid-City Communities PDO and West Lewis Street PDO with Citywide zones and development regulations."

Comment #11

The development regulations of the citywide zones differ from those of the PDOs. In many cases these differences are substantial. For example the current CL Zone has a maximum height of 30 feet whereas its proposed replacement, CN-1-4, has a maximum height of 65 feet, thus more than doubling the allowed height. The current MR-800B Zone has a maximum Flor Area Ratio (FAR) of 1.25 whereas its proposed replacement, RM-3-9 Zone, has a maximum FAR of 2.25 thus almost doubling the allowed FAR. Therefore, the PEIR must include a matrix which compares the development regulations, including, but not limited to, maximum number of units, maximum density, maximum height, maximum FAR, maximum lot coverage, setbacks/yards, parking and architectural features for the Mid-City Communities PDO and West Lewis Street PDO with those of the Citywide zones.

The PEIR must acknowledge these differences and must consider them when analyzing Neighborhood Character and Traffic. The PEIR must also analyze the cumulative impacts that would result from the altered development regulations.

B7-14 Reference #12 - Chapter 3 Project Description, 3.6.1 Uptown Land Use Distribution at Plan Build-out. Table 3-8, page 3-30.

Comment #12

As is done on Tables 3-9 and 3-10 (page 3-31), please include existing land use (see Table 2-1 on page 2-6) to facilitate a better understanding of the land use changes that will occur.

B7-15 Reference #13 - Chapter 3 Project Description, 3.6.1 Uptown Land Use Distribution at Plan Build-out, Table 3-9 page 3-31:

> According to Table 3-9 there will be a loss of 2,020 single-family units between the Existing Development and the Proposed Plan Build-out.

B7-13 Table 3-3. Conversion to Citywide Zoning, lists the Mid-City Communities Planned District Ordinance (PDO), West Lewis Street PDO, and Residential Zones that would be replaced with citywide zoning. Section 3.4.3.2 Applicable Citywide Zones provides descriptions of the zones that would apply to the Uptown CPU area. These zones were primarily selected to be consistent with existing maximum allowed residential densities in similar PDO zones. To address differences in zoning development standards such as Floor Area Ratio (FAR), setbacks, lot coverage, etc., citywide zoning development standards were used since citywide zones represent the optimal correlation between residential density and development standards. Additionally, the Community Plan Implementation Overlay Zone (CPIOZ) is being used to implement building heights that were identified in the plan update process and to establish maximum building heights where none are provided under citywide zoning. As discussed in the PEIR, the proposed Uptown CPU and associated discretionary actions would result in higher densities in some areas and lower densities in others, and these resulting differences from existing conditions. Per Section 6.1.4, Significance of Impacts, the proposed change from the PDO to citywide zoning would not create any conflicts or inconsistencies with the adopted Land Development Code. See Section 6.2, Visual Effects and Neighborhood Character and Section 6.3, Transportation and Circulation, of the PEIR for a discussion of the potential impacts to neighborhood character and traffic, as well the cumulative effects related to those resources. The Final PEIR has also been revised to further expand on the anticipated land use changes in Section 6.2.3 under Issue 2.

RESPONSE

- B7-14 Table 2-1 provides acreages of existing land uses in the Uptown community.
- B7-15 Figure 2-4, Land Uses under Adopted Community Plan Uptown, depicts the current land uses, including Low Density Residential, under the adopted Community Plan. Figure 3-1, Proposed Land Use

RESPONSE

B7-15 (cont.)

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 9

Comment #13

Please include a graphic showing the location of the 2,020 single-family units that would be removed. The loss of single-family units will displace many families and individuals and is a potentially significant impact.

B7-16

Excerpts #14 Chapter 6 Environmental Analysis, 6.2 Visual Effects and Neighborhood Character, Issue 2 Neighborhood Character, Pages 6,2-4 and 6,2-5; and 6,2.4 Determination of Significance, 6,2.4,2 Neighborhood Character, page 6,2-8:

"Would the project result in a substantial alteration (e.g. bulk, scale, materials or style) to the existing or planned (adopted) character of the area?" (page 6.2-4)

"The proposed Urban Design Element of the proposed Uptown CPU would also provide <u>policies</u> (emphasis added) addressing commercial and mixed-use development, and residential in-fill development. <u>Policies</u> (emphasis added) are related to street wall articulation, ground-level uses, windows, building materials, lighting, signs, corners, architectural projections, roofbop and mechanical screening, public space, public art, street orientation and setbacks, sustainable building design, height and massing, development transitions, and canyons and open space. Implementation of these <u>policies</u> (emphasis added) would provide specific policy support to ensure that the bulk and scale of development is not out of character with the existing environment." (page 6.2-4)

"Additionally, the proposed Uptown CPU Urban Design Element includes <u>Design</u> <u>Guidelines</u> (emphasis added) by <u>Building Type</u> that would address building height, further protecting public view corridors and regulating the bulk and scale of development. " (page 6.2-4)

"With the implementation of the proposed Urban Design Element <u>policies</u> (emphasis added), new development would be consistent with the existing neighborhood character. Thus, impacts related to substantial alterations to the existing or planned character of the area would be less than significant. No mitigation would be required." (page 6,2-5)

"The proposed Uptown CPU includes <u>policies</u> (emphasis added) that would encourage residential and mixed-use development that would be consistent with the existing neighborhood character, and impacts would be less than significant. No mitigation would be required." (page 6.2-8)

Comment #14

No implementing mechanism is included in the project that would require the policies and design guidelines to be considered for any action, discretionary or ministerial (e.g., building permit). Therefore, it cannot be said that implementation of these policies would provide specific policy support to ensure that the development is not out of character with the existing environment.

- Uptown, depicts the proposed land uses under the proposed Uptown CPU. A comparison of these two figures can provide an accurate depiction of the levels of densities and land use types of the adopted Community Plan and the proposed CPU; however, specific location of these land uses changes cannot be provided because intensification of land uses will occur over time by private property owners where the land use plan allows these changes. While the proposed CPU envisions a reduction in the number of single family units, an increase of multi-family units would more than accommodate residences by increasing the overall availability of housing in the Uptown CPU area. Thus, while redevelopment of existing residential land uses, by its nature, causes a temporary displacement of residents, redevelopment with higher density housing increases the housing stock available to residences and helps the City meet its housing goals. Additionally, no specific project-level development is proposed at this time; the proposed Uptown CPU merely provides a framework in which redevelopment may occur.

B7-16 See response to comments B7-2 and B7-5. Proposed Uptown CPU policies related to height and massing in residential neighborhoods emphasize conforming to the predominant scale of the neighborhood, the incorporation of development transition, and designing the massing of combined lots to respond to the rhythm of both adjacent development and the prevailing development on the block. Required consistency with these policies would be similar to the consistency review required for discretionary projects under the West Lewis Street PDO regulations. Development allowed under ministerial processes is restricted by zoning and regulations of the Land Development Code, which would ensure that changes to community character are less than significant. Guidance on the screening of on-site parking is provided in the Mobility Element of the proposed Uptown CPU.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 10

And, therefore, it cannot be said that impacts to neighborhood character would be less than significant, and that no mitigation would be required.

A project, particularly one involving only a ministerial action, indertaken in accordance with the land use and density recommendations could result in significant impacts by substantially altering the existing or planned character of the area. This is most evident for the West Lewis Street Area between Stephens Street and Lark Street and Reynard Way between Upas Street and Eagle Street.

West Lewis Street Area

Currently the West Lewis Street Area, which is a small, commercial area with mostly early Twentieth Century buildings completely surrounded by a residential neighborhood, is regulated by the West Lewis Street Planned District. The project proposes to repeal the planned district regulations and zone the area CN-1-1. Unfortunately, the CN-1-1 Zone, does not address the unique needs of the West Lewis Street Area by ensuring that new structures and modifications to existing structures within the West Lewis Street Area complement the surrounding, architecturally stable, single-family development.

This potentially significant impact can and should be mitigated by incorporating the following provision into the Urban Design Element in order to carry over the review process set forth in the current West Lewis Street PDO at Section 1520,0202;

The West Lewis Street Area located on the north and south sides of Lewis Street between Stephens Street and Lark Street shall be shown on a CPIOZ Map as Type B.

The purpose of the review under a Site Development Permit/Process Three is to maintain compatibility of the existing commercial strip with the surrounding single-family residential area while permitting flexible and feasible commercial development and redevelopment options.

It is intended that new structures and modifications to existing structures within the West Lewis Street Area complement the surrounding, architecturally stable, single-family development.

In addition to the Site Development Permit Findings, the Hearing Officer shall make the following design considerations:

(a) New construction shall preserve and enhance the existing scale and character of the neighborhoad. New construction shall be compatible with adjacent properties, scale and proportion, rhythm and spacing, materials and texture, architectural detailing and rooflines and materials.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 11

(b) Rehabilitation of and/or additions to existing structures shall respect the original distinguishing qualities or character of the property.

(c) Parking shall be suitably screened and landscaped so as to mitigate any visual impacts.

Reference to West Lewis Street Area shall be added on page IM-211 of the Implementation Element under Community Plan Implementation Overlay Zone (CPIOZ).

Reynard Way

The proposed zoning map designates the area between Upas Street and Eagle Street as CN-1-4. Unfortunately, the CN-1-4 Zone permits a height of 65 feet which would result in structures substantially out of scale with the adjacent single-family homes of 30 feet or less.

This potentially significant impact can and should be mitigated by rezoning the area to CN-1-3; the CN-1-3 Zone has a height limit of 30 feet that would result in structures in character with the adjacent single-family area.

As an alternative measure to mutigate potentially significant impacts, the area should be included on Figure 3-7 as a CPIOZ Type A with a Maximum Building Height of 30 Feet.

If the above mitigation measures related to Reynard Way and West Lewis Street are not incorporated into the project, the impacts on these areas are significant and unmitigated.

Furthermore, to determine if the project would result in a substantial alteration (e.g. bulk, scale, materials or style) to the <u>planned (adopted)</u> (emphasis added) character of the area, the PEIR must analyze the impacts or changes that would result from applying the development regulations of the Citywide zones compared to the PDO regulations, which currently apply in the area.

B7-17 Excerpt #15 - Chapter 6 Environmental Analysis, 6.2 Visual Effects and Neighborhood Character, 6.2.3 Impact Analysis, Issue 4 Landform Alteration, Page 6.2-5 and 6.2.6; and 6.2.4 Determination of Significance, 6.2.4.4 Landform Alteration, page 6.2-8:

"Would the project result in a substantial change to the existing landform?" (page 6.2-5)

"While the proposed Uptown CPU would intensify some uses, the proposed CPU contains <u>policies</u> (emphasis added) to ensure that redevelopment takes into account existing development as well as the landform. Of particular importance are the proposed Uptown CPU Conservation Element and Urban Design Element <u>policies</u> (emphasis added) that would support conservation of existing landforms, canyon lands, and open space and would support the design of buildings that respects existing landforms." (pages 6,2-5 and 6,2-6)

B7-17 See response to comment B7-2 and B7-5.

K

LETTER

RESPONSE

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 12

> "Implementation of the proposed Uptown CPU and associated discretionary actions would result in less than significant impacts related to landform alteration based on implementation of proposed Uptown CPU polices (emphasis added) that require building form to be sensitive to topography and slopes, and existing protections for steep slopes (Environmentally Sensitive Lands) and grading regulations within the LDC. Thus, impacts related to landform alteration would be less than significant, and no mitigation would be required." (page 6.2-8)

Comment #15

No implementing mechanism is included in the project that would require the policies to be considered for any action, discretionary or ministerial (e.g., building parmit). Therefore, it cannot be said that the policies would support conservation of existing landforms, canyon lands, and open space and would support the design of buildings that respect existing landforms. And, therefore, it cannot be said that landform alteration would be less than significant, and that no mitigation would be required.

A project, particularly a ministerial action, undertaken in accordance with the land use recommendations could result in significant impacts by substantially altering the existing landform.

B7-18 <u>Excernt #16</u> - Chapter 6 Environmental Analysis, 6.5 Greenhouse Gas Emissions, 6.5.3 Impact Analysis, Issue 1 Greenhouse Gas Emissions, Page 6, 5-7:

> "GHG emissions were calculated for the existing (on the ground) land uses, the land uses at build-out of the adopted Community Plan (in 2035, and the land uses at build-out of the proposed Uptown CPU and associated discretionary actions (in 2035). Table 6.5-3 summarizes the GHG emissions under each scenario."

See Table 6.5-3 on page 6.5-7 for details.

And Reference #16 - Letter Dated May 16, 2016 Supplemental Analysis to the Greenhouse Gas Analysis for the Uptown, North Park, and Golden Hill Community Plan Updates, Project No. 30330/304032, SCH No. 2004651076 (RECON Number 6086.1), Page 6, Table 4.

Comment #16

Table 6.5-3 in the PEIR erroneously shows the year 2020 Uptown GHG Emissions (Table 4) and not the year 2035 Uptown GHG Emissions (Table 7). This error should be corrected here and elsewhere in the PEIR.

B7-18 Table 6.5-3, GHG Emissions for the Uptown Community Plan Area, of the PEIR has been updated with the 2035 emissions reported in the Supplement Analysis to the Greenhouse Gas Analysis (Appendix E-2 of the PEIR).

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 13

B7-19 Excerpt #17 Chapter 6 Environmental Analysis, 6.7 Historical Resources, 6.7.4 Impact Analysis, Issue 1 Historic Structures, Objects, or Sites, d Resources Identified through Public Outreach, page 6.7-15, second paragraph, last sentence.

> "In order to bring these four potential districts (Terrace, Avalon Heights, Hillcrest, and San Diego Normal School/San Diego City Schools Education Complex) forward for designation, additional, intensive-level research would be required to evaluate the district and define a precise boundary, period of significance, significance criteria, and contributing and non-contributing resources."

Comment #17

Mission Hills Heritage remains concerned over the lack of responsiveness by the City in designating Historic Districts. See attached letter addressing the problems in seeking historic designation for the Mission Hills Historic District Annex. As noted, the complete report was submitted to Historic Resources Board staff in September 2011. It did not move forward through the process until sometime in 2014, with the final hearing on June 26, 2014, thus, resulting in a 30-month delay.

Furthermore, despite requests from residents involved with the effort to nominate Inspiration Heights as a historic district, the City has refused to set up a meeting to proceed with the project for over 18 months.

Nevertheless, proper mitigation of the significant impact on historical resources described in this chapter should include performing the research and analysis to bring these four potential districts and the other identified potential historic districts forward for designation. All development shall be suspended until the research and analysis has been completed and the potential historic districts implemented.

B7-20 See Except #18 and Comment #18 for further discussion.

Excerpt #18 - Chapter 6 Environmental Analysis, 6.7 Historical Resources, 6.7.4 Impact Analysis, Issue 1 Historic Structures, Objects, or Sites, d. Resources Identified through Public Outreach, page 6.7-18:

"While the Municipal Code does provide for the regulation and protection of designated and potential historic resources, and while supplemental development regulations would provide additional protection for potential historic districts, it is impossible to ensure the successful preservation of all historic built environmental resources within the plan area. Therefore, impacts to the Potential Historic Districts and individual historic resources would be considered significant and unavoidable." B7-19 This comment is noted. Suspension of development is not required to protect potential historic districts because the proposed amendments to the Historical Resources Regulations include supplemental development regulations to assist in the preservation of specified potential historic districts until they can be intensively surveyed and brought forward for designation. See Section 6.7.4, Issue 1 of the PEIR which explains that in response to the identified lack in protections for potential historic districts, amendments to the Historical Resources Regulations are proposed to provide supplemental development regulations to address how and where modifications can be made on residential properties identified as potentially contributing to specified potential historic districts. Development that does not comply with the regulations of the supplemental development regulations would be subject to a Neighborhood Development Permit with deviation findings and mitigation. The amendments to the Historical Resources Regulations are scheduled to be brought to City Council with the proposed North Park CPU, prior to the Uptown CPU. However, ultimately the PEIR concludes that impacts to Potential Historic Districts would be significant and unavoidable because at a program level of analysis, the degree of future impacts and applicability, feasibility, and success of future mitigation measures cannot be adequately known for each specific future project at a program level of analysis and impacts to potential historic resources would be significant and unavoidable. In addition, Potential Historic Districts would not be fully protected until they are intensively surveys, verified, and designated. Therefore, this comment does not

RESPONSE

B7-20 See response to comment B7-19 regarding the mitigation framework and finding of significant and unavoidable impacts to potential historic districts. Additionally, potential historic districts must be evaluated against the City's Historical Resources Board's criteria for a historic district; have the required documentation

identify an impact not already analyzed in the PEIR.

B7-20 (cont.)

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 14

Comment #18

Mission Hills Heritage concurs with the comments and mitigation measures submitted by SOHO. In particular, the impact can be mitigated by implementing the identified potential historic districts (including the four community-identified potential historic districts) concurrently with adoption of the plan. On the other hand, the proposed supplemental development regulations are woefully inadequate and will not effectively mitigate the impact, for the following reasons; (i) the proposed regulations are not comprehensive, in that they exclude the community-identified potential historic districts, the Multiple Property Listings (MPLs) and the commercial properties located in the potential historic districts, (ii) the proposed regulations do not require projects involving potential contributors to follow the Secretary of Interior Standards of Treatment of Historic Properties (as are required for projects involving contributors to designated historic districts) and the regulations do not prohibit changes to windows and doors, all of which will lead to an erosion of historic fabric within potential historic districts, and (iii) unlike designated historic districts, the proposed regulations do not include any regulation of parcels deemed non-contributors, thereby allowing development that is incompatible with the potential district and negatively impacting the future implementation of the potential historic district. The proper and best mitigation strategy is to implement the identified potential historic districts and bring such forward for nomination concurrently with adoption of the plan.

B7-21 <u>Excerpts #19</u> - Chapter 6.0 Environmental Analysis, 6.12 Public Services, 6.12.3 Impact Analysis, Issue 1 Public Facilities, b. Park and Recreation, page 6.12-16 and 6.12-17;

"Build-out of the proposed Uptown CPU would add additional population to the CPU area and the CPU area would continue to have a deficit of population-based parks at build-out, which would be an <u>adverse impact</u> (emphasis added). Similarly, until the identified recreational facilities such as the aquatic complex are constructed, the deficit and the associated <u>adverse impact</u> (emphasis added) would continue." (page 6.2-16)

"Thus, although the existing and projected deficit in population-based parks is <u>adverse</u> (emphasis added), impacts associated with the construction of park facilities would be <u>less than significant</u> (emphasis added) at the program level. Implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of population-based parks in the CPU area, but does not propose construction of new facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a <u>less than significant impact</u> (emphasis added) associated with the construction of new facilities in order to maintain performance objectives for parks." (pages 6.2-6 and 6.2-7)

Comment #19

To say that the impact is adverse but "not considered significant at the program level" begs/ignores the question. Until adequate population based park sites have been identified, the impact relating to construction is potentially significant and unmitigated.

completed, including a Designation Request and Historical Report; and be discussed at two Historical Resources Board meetings. Additional noticing, site visits, and board hearing must also be completed prior to implementation of historical district boundaries. These requirements, while relatively extensive, are intended to ensure that historic districts are appropriately designated and do not impose unnecessary development restrictions while also ensuring the quality and significance of established historic districts within the City.

B7-21 Though there would be a deficiency in park and park equivalences at build-out of the proposed Uptown CPU, the existing conditions include a deficit in parks and park equivalencies. In addition, through the proposed Uptown CPU effort, 37.40 acres of proposed new population-based park land and park equivalency sites have been identified. The policy framework provided by the proposed Uptown CPU also supports acquisition and development of new public parks and park equivalencies, and encourages new private development to include recreational facilities. The project does not include construction of new facilities, but provides policy support for new parkland. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact associated with the construction of new facilities in order to maintain performance objectives for parks.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 15

To mitigate the impact, the Uptown Community Plan Update should include a program that ensures that existing deficiencies based on current population are remedied and that adequate facilities commensurate with future population are provided in conjunction with development projects that create the additional need.

B7-22 Excernt #20 - Chapter 7.0 Effects Found Not to be Significant, 7.3 Population and Housing, Page 7-3:

"The proposed Uptown CPU and associated discretionary actions would not displace people or existing housing . . . "

Comment #20

Reference Table 3-9 on page 3-31. The table shows a loss of 2,020 single-family units. Therefore, the proposed Uptown CPU and associated discretionary actions will displace people and existing housing.

This represents a 27% loss of existing single-family units and is potentially significant.

B7-23 Excerpt #21 – Chapter 10.0 Alternatives, 10.1 No Project (Adopted Community Plan Alternative), 10.1.1 Description, page 10-5, second paragraph:

"... the proposed Uptown CPU could have approximately 32,700 dwelling units at build-out, while the No Project Alternative could have approximately 34,600 dwelling units at build-out, or 1,900 more units compared to the proposed Uptown CPU."

Comment #21

First, please verify that the build-out year for both plans is 2035.

Second, what is the basis for saying that the No Project Alternative could have approximately 34,600 dwelling units at build-out?

This statement appears to be contradicted by the Adopted 1988 Community Plan, which states on page 37:

"The estimated build-out capacity for development is 25,410 dwelling units."

Please explain why there is a disparity of 9,190 units (34,600 - 25,410).

Furthermore, please explain why 34,600 in light of SANDAG's Series 13 growth forecast for Uptown of 29,223 in 2035. B7-22 See response to comment B7-15 regarding displacement.

B7-23 Build-out for both plans is 2035 (see Section 3.6, Plan Build-out, of the PEIR for explanation). The disparity of 9,190 dwelling units at build-out of the adopted Community Plan between what is stated in the PEIR and the adopted Community Plan is due to use of different development assumptions used at the time when the adopted Community Plan was approved. The dwelling unit capacity assumptions in Appendix J of the adopted Community Plan assumed that development would occur at the mid-range of the allowed residential density range in residential areas of the community. It also assumed that one half of the commercial areas would develop with residential units where ten percent of that area would be built at the maximum density permitted and that ninety percent of that area would be built at the lower "average" of the density range. The proposed Uptown CPU assumes that for areas likely to develop within residential and commercial areas, new projects would develop at the maximum of the density range, which more accurately reflects the development that has been occurring in the Uptown Community. This same assumption was used to calculate build-out estimates for the adopted Community Plan in order to appropriately compare it to the proposed Uptown CPU. Using the same assumptions and methodology, the proposed Uptown CPU would generate more housing units at build-out compared to the build-out assumed when the adopted community plan was approved.

Additionally, the estimated build-out of the adopted Community Plan differs from SANDAG's growth forecast because SANDAG uses actual population trajectory estimates, while the build-out of the Community Plan is simply based on complete build-out of all allowed land uses.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 16

B7-24 <u>Excerpt #22</u> – Chapter 10.0 Alternatives, 10.1 No Project (Adopted Community Plan Alternative), 10.1,2 Analysis of No Project Alternative, b. Visual Effects and Neighborhood Character, page 10-9:

> "Potential visual effects and impacts to neighborhood character under the No Project Alternative would be similar to those anticipated under the proposed Uptown CPU except that the No Project Alternative would allow reduced building heights when compared to the proposed Uptown CPU and associated discretionary actions. Particularly in Mission Hills and Hillcrest, development under the No Project Alternative would be limited to building heights of 50 and 65 feet, respectively. The reduced building heights could reduce potential neighborhood character impacts); however, the No Project Alternative would also not benefit from the proposed Uptown CPU <u>policies</u> (emphasis added) that are intended to ensure compatible development and design that enhances and is sensitivity (sie) to neighborhood character."

Comment #22

As previously stated there is no implementing mechanism included in the project that would require the policies to be considered for the majority of development projects that would only require ministerial approve (e.g., building permit). Unless an implementing mechanism is incorporated into the project, the No Project Alternative would have the same impact on neighborhood character as the proposed Uptown CPU.

B7-25 Excerpt #23.1 - Chapter 10.0 Alternatives, 10.1 No Project (Adopted Community Plan Alternative), 10.1 2 Analysis of No Project Alternative, e. Greenhouse Gas Emissions, page 10-10:

"The No Project Alternative would result in 535,684 metric tons of carbon dioxide equivalent (MT CO2E GHG) emissions, which would be slightly greater than the estimated 523,925 MT CO2E GHG emissions for the proposed Uptown CPU. The decrease in GHGs associated with the proposed Uptown CPU is a direct result of the proposed land use changes including an increase in commercial uses and decrease in residential, which would implement CAP Strategies and the General Plan's City of Villages Strategy. Since the No Project Alternative would not adjust the land use map or provide policies to implement these strategies, <u>GHG impacts of the No Project Alternative would be significant and unavoidable and greater than the proposed Uptown CPU (emphasis added).</u>"

Excerpt #23.2 – Chapter 6 Environmental Analysis, 6,5 Greenhouse Gas Emissions, 6,5.3 Impact Analysis, Issue 1 Greenhouse Gas Emissions, page 6,5-7, last paragraph:

"For the purposes of determining significance, GHG emissions attributable to the proposed Uptown CPU and associated discretionary actions at full build-out were compared to the adopted Community Plan GHG emissions. The reason this comparison is

- B7-24 See response to comment B7-5 regarding the program-level analysis.
- B7-25 Chapter 10, Alternatives, of the final PEIR has been updated to reflect the correct 2035 build-out greenhouse gas emission estimates. This correction does not change the result of the analysis. On July 12, 2016, the City Council adopted an amendment to the CAP, which included a CAP Consistency Checklist, and other amendments to the text of the CAP, which resulted in the CAP serving as a qualified GHG reduction plan. At that same time, the City Council also adopted a GHG Significance Determination Threshold (GHG Threshold) that is being implemented as of July 19. 2016. The PEIR tiers off of the GHG analysis set forth in the CAP Final EIR, which was certified on December 15, 2015, with an addendum certified on July 12, 2016 that specifically addressed the adoption of the GHG Threshold. See Section 6.5.2, Significance Determination Thresholds, for an explanation, which discusses that the proposed CPU and associated discretionary actions would have less-than-significant impacts if emissions from build-out are less than those generated by build-out of the adopted Community Plan, or if the increase in GHG emissions is a direct result of implementing CAP strategies and the City of Villages Strategy. Because build-out of the No Project Alternative would generate higher GHG emissions than the proposed CPU and would not implement land use changes consistent with CAP strategies and the City of Villages Strategy, the No Project Alternative would result in significant and unavoidable impacts associated with GHG emissions. Therefore, the excerpt from Section 10.1.2 of the PEIR, as referenced in this comment, is correct and no change in the level of significance is required.

appropriate is because the GHG emissions from the adopted Community Plan were used when developing the City's CAP GHG Inventory. Thus, if calculated future emissions of the proposed Uptown CPU would be consistent with, or less than, the emissions assumed in developing the CAP inventory, build-out of future land uses under the proposed Uptown CPU and associated discretionary actions would be consistent with the City's CAP emissions inventory and would not represent a significant impact related to GHG emissions."

Comment #23

First, as previously stated, the data presented is for 2020 and not for 2035.

Second, the impacts of the No Project Alternative would NOT be significant and unavoidable (as stated in Excerpt #23.1) simply because the calculated future emissions of the adopted Community Plan Alternative would be consistent with the emissions assumed in developing the CAP inventory (as stated in Excerpt #23.2).

Table 10-1 should be revised to indicate that the Greenhouse Gas Emissions for the No Project Alternative is Less than Significant but that the impact is greater than the Proposed Project.

B7-26 Excerpts #24 - Chapter 10.0 Alternatives, 10.2 Adopted Community Plan with Removal of the Interim Height Ordinance Alternative, 10.2.1 Description, page 10-15:

> "(T)he adopted Community Plan with Removal of the Interim Height Ordinance Alternative would allow taller buildings under ministerial review within the Mission Hills, Hillcrest, and Bankers Hill/Park West neighborhoods. The increased building height allowance... would have the potential to increase the intensity of development (emphasis added) with taller buildings compared to the proposed Uptown CPU and associated discretionary actions."

> "Specifically, as shown in Table 10-3, the proposed Uptown CPU would support 32,700 dwelling units at build-out, while the Adopted Community Plan would accommodate 34,600 dwelling units at build-out or 1,900 more units compared to the proposed Uptown CPU."

Comment #24

The first excerpt clearly acknowledges that a greater build-out would occur with the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative because of the taller buildings that would be allowed. Yet, the second excerpt suggests that the build-out would be the same as the Adopted Community Plan. Please explain.

B7-27 Excerpt #25 - Chapter 10.0 Alternatives, 10.2 Adopted Community Plan With Removal of Interim Height Ordinance Alternative, 10.2.2 Analysis of Adopted Community Plan with B7-26 The information in the referenced second excerpt has been removed from the PEIR.

B7-27 The data has been corrected in the final PEIR to reflect the 2035 estimates. See response to comment B7-26 regarding build-out of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative versus the No Project Alternative. See also response to comment B7-25 regarding inconsistency with the CAP and the associated significant and unavoidable impact.

LETTER

RESPONSE

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 18

Removal of the Interim Height Ordinance Alternative, e. Greenhouse Gas Emissions, page 10-17:

"The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in 535,684 MT CO2E GHG emissions which would be slightly greater than the estimated 523,925 MT CO2E GHG emissions for the proposed Uptown CPU. The decrease in GHGs associated with the proposed Uptown CPU is a direct result of the proposed land use changes including an increase in commercial uses and decrease in residential, which would implement CAP Strategies and the General Plan's City of Villages Strategy. Since the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not adjust the land use map or provide policies to implement these strategies, GHG impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be significant and unavoidable and greater than the proposed Uptown CPU."

Comment #25

First, as previously stated, the data presented is for 2020 and not for 2035.

Second, Excerpt #24 states that this alternative "would have the potential to <u>increase the intensity</u> of <u>development</u> (emphasis added) with taller buildings compared to the proposed Uptown CPU and associated discretionary actions." Presumably this would mean more development than the Adopted Community Plan. And yet it would generate the same level of emissions? Please explain.

Third, if the impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be the same as the Adopted Community Plan, then they would NOT be significant and unavoidable simply because the calculated future emissions of the adopted Community Plan Alternative would be consistent with the emissions assumed in developing the CAP inventory (as stated in Excerpt #23.2).

B7-28 Excerpts #26 - Chapter 10.0 Alternatives, 10.4 Density Redistribution Alternative, 10.4,1 Description, page 10-26 and 10-28:

> "Under this alternative, the density of future development would be lower along transit commercial nodes (emphasis added) ... Under this alternative, the reduction in density would be redistributed resulting in the same overall development potential as the proposed Uptown CPU. "(page 10-26)

> "When compared to the proposed Uptown CPU, the Density Redistribution Alternative reduces residential density development potential along . . . <u>Revnard Way</u> (emphasis added) . . . from 44 du/ac to 29 du/ac, ^a (page 10-28)

B7-28 See response to comment B6-3 regarding Reynard Way as a transit commercial node.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 19

Comment #26

The above incorrectly implies that Reynard Way is along a transit commercial node. The City's General Plan, SANDAG's Smart Growth Concept Map, and the City's Climate Action Plan do NOT support that implication as noted below.

A. The General Plan.

See Figure LU-1, the Village Propensity Map, on page 9 of the Land Use and Community Planning Element of the General Plan. According to that element. "This figure is based upon existing and community plan-designated land uses. By overlaying the facilities factors with the land uses, the Village Propensity Map illustrates existing areas that already exhibit village characteristics and areas that <u>may have a propensity to develop as village areas</u>." (emphasis added) Reynard Way clearly falls within the Low Propensity range. A cropped version of Figure I,U-1 is enclosed.

B. SANDAG's Smart Growth Concept Map.

Reference http://www.sandag.org/uploads/projectid/projectid_296_14006.pdf for Quick Reference: Smart Growth Areas by Place Type, October, 2014. Reynard Way is not listed as an Existing/Planned or Potential Metropolitan Center, Urban Center, Town Center, Community Center, Mixed Use Transit Corridor, Special Use Center, or Rural Village. Nor is it shown on the Smart Growth Concept Map, Mid-City and East County Subregion, October 2014 or Draft April 2016. A cropped version of Smart Growth Concept Map, Draft April 2016 is enclosed.

C. Climate Action Plan Transit Priority Area Map

The Transit Priority Area (TPA) Map (dated 5/31/16), which is included as Appendix B of the Climate Action Plan, shows that the northern portion of Reynard Way between Sutter Street and Curlew Way is within one-half mile of a Major Transit Stop located at Goldfinch Street and Washington Street. Relatively steep topography, however, makes it unlikely that residents within this area would actually walk to the transit stop. The TPA Map shows the southern portion of Reynard Way to be within one-half mile of a trolley stop to the west and transit stops along First Avenue to the east. However, there are no connecting streets or pedestrian ways between Reynard Way and the trolley stop or First Avenue.

B7-29 Excerpt #27 – Chapter 10.0 Alternatives, 10.4 Density Redistribution Alternative, 10.4.2 Analysis of Density Redistribution Alternative, a. Land Use, page 10-28:

"The Density Redistribution Alternative would retain the proposed Uptown CPU land uses, but further lowers density throughout the community with the exception of the Park. Boulevard transit corridor between Washington Street. University Avenue and Normal Street, Land use impacts under this alternative would be similar to the anticipated impacts to the proposed Uptown CPU. The Density Redistribution Alternative would facilitate B7-29 While the General Plan does not establish levels of density increases required to remain consistent with the City of Villages Strategy, the strategy aims to direct new development projects away from natural undeveloped lands into already urbanized areas and/or areas where conditions allow the integration of housing, employment, civic, and transit uses. It is a development strategy that mirrors regional planning and smart growth principles intended to preserve remaining open space and natural habitat and focus development in areas with available public infrastructure. Therefore, it is appropriate to assume that because the Density Redistribution Alternative would facilitate transit-oriented development and mixed-use development to a lesser degree than the proposed Uptown CPU, it would achieve consistency with the City of Villages Strategy to a lesser degree. As such, the City does not agree that the requested change should be made to the final PEIR.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 20

transit-oriented development and mixed use development but to a lesser degree than the proposed Uptown CPU due to reduced density near areas accessible to transit with the exception of the Park Boulevard transit corridor. Land use changes would be compatible with the implementation of the San Diego General Plan, but to a lesser degree. Like the proposed Uptown CPU, this alternative would not conflict with adopted land use plans, policies, or ordinances; however it would achieve consistency with the General Plan City of Villages strategy to a lesser extent (emphasis added). Thus, land use impacts of this alternative would be <u>slightly greater</u> (emphasis added) than the proposed Uptown CPU."

Comment #27

The General Plan does not establish levels of density increase that are required in order to be consistent with the City of Villages strategy. Therefore, it is fallacious to state that the Density Redistribution Alternative "would achieve consistency with the General Plan City of Villages strategy to a lesser extent" (comphasis added). Nor can it be said "that land use impacts of this alternative would be <u>slightly greater</u> (emphasis added) than the proposed Uptown CPU." These statements are certainly contrary to the findings that the air quality and traffic impacts of the Density Redistribution Alternative area less that the proposed Uptown CPU. Would a further doubling of density within these areas be considered to achieve consistency to a greater extent, particularly in light of the even greater air quality, traffic, and GHG emissions that would result?

The discussion of Land Use should be revised as follows:

"The Density Redistribution Alternative would retain the proposed Uptown CPU land uses, but further lowers density throughout the community with the exception of the Park Boulevard transit corridor between Washington Street, University Avenue and Normal Street. Land use impacts under this alternative would be similar to the anticipated impacts to the proposed Uptown CPU. The Density Redistribution Alternative would facilitate transit-oriented development and mixed use development but to a lesser degree than the proposed Uptown CPU due to reduced density near areas accessible to transit with the exception of the Park Boulevard transit corridor. Land use changes would be compatible with the implementation of the San Diego General Plan. Like the proposed Uptown CPU, this alternative would not conflict with adopted land use plans, policies, or ordinances."

On Table 10-1 (page 10-3) Land Use for the Density Redistribution Alternative should be shown as LS (=) (less than significant and Impact the same/similar to the Proposed Project).

B7-30 Excerpt #28 – Chapter 10.0 Alternatives, 10.4 Density Redistribution Alternative, 10.4.2 Analysis of Density Redistribution Alternative, c. Transportation, page 10-30, fourth paragraph:

"The Density Redistribution Alternative would contain the proposed Uptown CPU policies intended to promote a multimodal network that encourage walking, bicycling, and taking transit; however these goals would be achieved to a lesser extent due to the reductions in development potential within areas accessible to transit (emphasis added).

B7-30 The issue is not solely the provision of alternative transportation, it is the provision of alternative transportation that serves development. Therefore, reducing development along the alternative transportation network would not result in the same quality of a functioning multimodal network as would an alternative transportation network that serves higher densities. As such, the City does not agree that the requested change should be made to the final PEIR.
Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 21

> While the impacts to individual intersections and roadway segments would be less than the proposed Uptown CPU, like the proposed Uptown CPU these impacts would remain significant and unmitigated."

Comment #28

The issue is the provision of alternative modes of transportation. Regardless of the reduction in development potential within areas accessible to transit, the Density Redistribution Alternative would contain the proposed Uptown CPU policies intended to promote a multimodal network that encourage walking, bicycling, and taking transit

On Table 10-1 (page 10-3) Alternative Transportation for the Density Redistribution Alternative should be shown as LS (=) (less than significant and Impact the same similar to the Proposed Project).

B7-31

<u>Excerpt</u> #29 – Chapter 10.0 Alternatives, 10.4 Density Redistribution Alternative, 10.4.2 Analysis of Density Redistribution Alternative, e. Greenhouse Gas Emissions, page 10-31:

> "The Density Redistribution Alternative would decrease development potential over the proposed Uptown CPU, which would reduce GHG emissions; however the GHG efficiencies of providing development in proximity to transit would be lost, with the exception of the density provided along Park Boulevard. The decrease in development potential would accommodate approximately 1585 fewer units within the Density Redistribution Alternative in areas where residents would have convenient access to transit and commercial services. This would result in a potential conflict with the implementation of CAP Strategies and the General Plan's City of Villages Strategy. Decreasing residential and commercial density in transit corridors and Community Villages within a Transit Priority Area (TPA) would not support the City of San Diego in achieving the GHG emissions reduction targets of the CAP since these residents would need to find housing or employment elsewhere that may not have accessibility to transit. While speculative as to the availability and use of transit elsewhere, it is likely that the residents not within a transit priority area would make greater use of autos - either through not having transit available or longer commute distances - thereby increasing vehicle miles traveled as compared to the proposed Uptown CPU, and thus, GHG emission impacts associated with the Density Redistribution Alternative would be greater than the project."

Comment #29

The first statement should be revised to read:

"The Density Redistribution Alternative would decrease development potential over the proposed Uptown CPU, which would reduce GHG emissions." B7-31 The statement regarding the loss of GHG efficiencies of providing development in proximity to transit is not irrelevant; rather, it is necessary to appropriately portray the potential impacts associated with the Density Redistribution Alternative. See comment B7-25 regarding CAP consistency and subsequent CEQA GHG analyses.

LETTER

RESPONSE

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 22

All other statements are irrelevant and/or speculative and should be deleted.

The methodology for analyzing Greenhouse Gas Emissions is found in Section 6.5.1.1 of this PEIR on page 6.5-1. This section states:

"Annual GHG emissions due to the operation of build-out of the Liptown Community Plan area under the adopted and proposed plans were calculated using California Emissions Estimator Model (CalEEMod; CAPCOA 2013). The emissions sources include construction (off-road vehicles), mobile (on-road vehicles), area (fireplaces, consumer products [cleansers, aerosols, and solvents], landscape maintenance equipment, and architectural coatings), water and wastewater, and solid waste sources. Where project-specific data were not available, model inputs were based on information provided in the CalEEMod User's Guide (CAPCOA 2013)."

Thus, as noted, any discussion of consistency with or conflict with CAP, Transit Priority Area, and/or City of Villages are not factors in generating the GHG emissions.

Since the GHG Emissions would be less than the proposed Uptown CPU, they would also be less than the GHG Emissions for the Adopted Community Plan. Therefore, the impact would be less than significant.

On Table 10-1 the Greenhouse Gas Emissions for the Density Redistribution Alternative should be shown as LS (<) (less than significant and Impact less than the Proposed Project).

If any consideration is given to adopting this alternative as the Project, the City may want to consider doing a GHG Emissions analysis for the Density Redistribution Alternative.

B7-32 Excerpt #30 Chapter 10.0 Alternatives, 10.6 Environmentally Superior Alternative, page 10-40:

> "Based on a comparison of the alternatives' overall environmental impacts and their compatibility with the CPUs' goals and objectives, the Density Redistribution Alternative is the environmentally superior alternative for this Program EIR. While the Density Distribution Alternative would not be able to reduce the significant and unavoidable impacts of the proposed Uptown CPU, it would reduce impacts related to traffic circulation and air quality. At the same time, the Density Redistribution Alternative would not support the full implementation of the General Plan's City of Villages strategy of developing multi-modal centers that encourage walking, bicycling, and taking transit and contain a mixture of commercial and residential development because the density of future development under the Density Redistribution Alternative would be lower along transit continercial nodes except for the transit cortidor along Park Boulevard between University Avenue and Washington Street and Normal Street. The Density Redistribution Alternative could also conflict with the implementation of the City's Climate Action Plan

B7-32 See response to comments B7-29 through B7-31. Based on the reasons provided in these responses, the City does not agree that the requested change should be made to the final PEIR.

LETTER

RESPONSE

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 23

since the redistribution of density would result in a likely increase in greenhouse gas emission impacts and vehicle miles traveled."

Comment #30

Based on Comments #27, #28, and #29 the Environmentally Superior Alternative should read as follows:

"Based on a comparison of the alternatives" overall environmental impacts and their compatibility with the CPUs: goals and objectives, the Density Redistribution Alternative is the environmentally superior alternative for this Program EIR. While the Density Distribution Alternative would not be able to reduce the significant and unavoidable impacts of the proposed Uptown CPU related to noise, historical resources, and paleontological resources, it would reduce the significant impact related to traffic circulation and less than significant impacts to air quality, and Greenhouse Gas Emissions. The Density Redistribution Alternative would result in less than significant impacts similar to the Proposed Project for land use, visual effects and neighborhood character, biological resources, geologic conditions, hydrology/water quality, public services, public utilities, health and safety."

B7-33 Summary and Conclusion

As a summary statement, the Draft PEIR has serious deficiencies as cited above. Many of these deficiencies seem to rise out of an attempt to support the project while detracting from alternatives in direct violation of Municipal Code §128.0103 (b), which states:

"The Planning Director shall establish and maintain that degree of independence in the performance of these functions and duties as will assure the City Council, the City Manager, the Planning Commission, and the people of the City of San Diego that the review and analysis of the environmental consequences of projects, are in accordance with CEQA, are independent and wholly objective, and are not prepared for the purpose of either supporting or detracting from any project, plan, or position. (emphasis added) whether advanced by the City, any other governmental agency, or private interest."

These deficiencies must be corrected prior to certifying the PEIR.

B7-34 Furthermore, if the project is not modified to incorporate the feasible mitigation measures presented in italics in **Overarching Deficiency of the PEIR**, Comment #14 related to West Lewis Street Area and Reynard Way, and Comment #18 related to Historic Resources, the PEIR must be recirculated prior to certification pursuant to State CEQA Guidelines §15088.5 (a) which states: B7-33 This comment suggests that the PEIR is deficient, pointing to the reasons cited in this letter's previous comments. The City does not agree that the PEIR is deficient for the responses provided in the previous responses to this comment letter.

B7-34 See response to comments B7-16 and B7-20 regarding the program-level analysis and appropriateness of the mitigation framework included in the PEIR.

Kurtis Steinert, Senior Environmental Planner August 4, 2016 Page 24

"15088.5. RECIRCULATION OF AN EIR PRIOR TO CERTIFICATION

(a) A lead agency is required to recirculate an EIR when significant new information is added to the EIR after public notice is given of the availability of the draft EIR for public review under Section 15087 but before certification. As used in this section, the term "information" can include changes in the project or environmental setting as well as additional data or other information. New information added to an EIR is not "significant" unless the EIR is changed in a way that deprives the public of a meaningful opportunity to comment upon a substantial adverse environmental effect of the project or a feasible way to mitigate or avoid such an effect (including a feasible reject alternative) that the project's proponents have declined to implement. "Significant new information" requiring recirculation include, for example, a disclosure showing that:

 A new significant environmental impact would result from the project or from a new mitigation measure proposed to be implemented.

(2) A substantial increase in the severity of an environmental impact would result unless mitigation measures are adopted that reduce the impact to a level of insignificance.

(3) A feasible project alternative or mitigation measure considerably different from others previously analyzed would clearly lessen the environmental impacts of the project, but the project's proponents decline to adopt it. (emphasis added)

(4) The draft EIR was so fundamentally and basically inadequate and conclusory in nature that meaningful public review and comment were precluded."

We look forward to reviewing your responses to our comments, the Final PEIR, as well as revisions/additions to the Community Plan Update and associated discretionary actions. Please contact us if you have any questions. Thank you.

Sincerely,

Mission Hills Heritage

President im Reily.

Barry E. Hager, Board Cha

Enclosures: Letter to Interim Mayor Todd Gloria dated February 7, 2014 General Plan Figure LU-1, Village Propensity SANDAG Smart Growth Opportunity Areas

CC: Tait Galloway, Principal Planner, Planning Department Marlon Pangilinan, Senior Planner, Planning Department Mike Hansen, Director of Land Use and Environmental Policy, Mayor's Office Adriana Martinez, Representative, Council District 3 Leo Wilson, Chair, Uptown Planners Belinda Smith, President, Mission Hills Town Council

LETTER

ATTACHMENT 7

RESPONSE

Mission Hills Historic District Extension 4251 Arguello St San Diego CA 92103 619-295-3773

To: Interim Mayor Todd Gloria

CC: Chief Operating Officer Scott Chadwick

CC: City Attorney Jan Goldsmith

CC: Assistant Deputy Planning Director Cathy Winterrowd

Re: Mission IIills Ilistoric District Annex

February 7, 2014

Dear Mayor Gloria,

In September of 2011 we submitted all the required research to the Historic Resources Board (HRB) for the addition of 99 houses to be annexed to the existing Mission Hills Historic District. This completed submission reflected the culmination of five years of procedural work with the HRB. A dedicated group of volunteers from the community raised funds and did exhaustive research on the houses. We have had difficulty getting replies to communications with Kelley Stanco, the official contact in charge of overseeing the work for approval. We are requesting your assistance in bringing this effort to its conclusion, without further costs to us.

Our last contact with the HRB was a meeting with Kelley Stanco in July of 2013, just before she went on maternity leave. It was a procedural review with two members of the IIRB. Our District Annex Proposal was discussed with explanations of how the Annex District will be incorporated into the existing District. We are proposing no changes or amendments to their original descriptions and procedural methods. It was revealed during this meeting that the HRB has completed a house-by-house street review of our proposed district with our paperwork package in hand. The board members present then voted to continue along the approval process, and the meeting was adjourned.

2

Ms. Stanco laid out the following path of events:

- Homeownets will be contacted and a neighborhood workshop will be held to answer questions regarding becoming a Historic District.
- 2. A repeat polling will be done of all the neighbors affected, to ensure that there is community approval. This is apparently deemed necessary by the HRB because, through no fault of ours, it has taken the City and the HRB so long to complete the process, increasing costs to the city.
- 3. A date will be set for the first of usually two formal HRB hearings. The first hearing will be to reveal which houses are contested by the HRB for inclusion, and the second will be for presentation of our defense for houses upon which there is disagreement.
- 4. Neighborhood volunteers will then be required to research, update all information on 99 houses to reflect changes in ownership and possible alterations to individual homes that affect architectural integrity, again deemed necessary by the HRB because of the length of time our proposal has languished in their offices.

Needless to say, we would like to register our frustration and disappointment. Specifically, we object to the HRB doing a re-polling of the neighbors and requiring research to be repeated, at our expense, after so much time, money and effort has already been spent. The duplication of efforts, time and expense should be borne by the City, as all the delays and inattention rest at the City's doorstep. It is not the fault of the neighborhood that this has taken so long to process, and it is unfair to jeopardize our efforts after our work has been completed.

Again, we insist that if the HRB requires an update in the data, then HRB should be responsible for the incurred expenses and the completion of redoing the research since they are responsible for the dalays.

Sincerely,

Deborah Quillin

Jill Limber

LETTER



LETTER



LETTER

	Letter B8		
	Mission Hills		
	August 8th, 2016		
	Kurtis Steinert Senior Environmental Planner City of San Diego Planning Department 1010 Second Avenue, MS 413 San Diego, California 92101		
	Re: Comments on Draft Program Environmental Impact Report		
	Dear Mr. Steinert:		
B8-1	Mission Hills Town Council thanks you for the opportunity to comment on the Draft PEIR for the Uptown Community Plan Update dated June 10, 2016.	B8-1	Comment noted.
B8-2	We have been involved with the plan update process since our inception in 2007, and appreciate the work accomplished by City staff thus far. We have also worked with various other community groups & stakeholders including Uptown Planners, Mission Hills Heritage, Mission Hills BID, the Mission Hills-Hillcrest Library and Grant K-8. At various times during this process, we have also solicited formal input from the community, our members, and anecdotal input via conversations with residents, and leadership at the		comment process.
	Council.	B8-2	Comment noted.
B8-3	In summary, we agree with the analysis contained within the draft PEIR, and we support all that it includes in the Update. We also appreciate the broader improvements to open space, planting more trees public art,		
	improved parking, and multi-modal transportation, including pedestrian and bicycle safety and improvements.	B8-3	Comment noted.
B8-4	In particular we would like to address the following:		inadequacy in the P
	 Recreational Element: Thank you for hearing our comments on the need for more parks in Mission Hills, especially the 		
	open space for Mission Hills Trail. We appreciate the inclusion of a plan for a park on Reynard Way, and appreciate the opportunity to keep the building a working historic site if possible. Many of	B8-4	Comment noted.
	our residents have requested a dog park in the last 18 months, so perhaps space for a dog park element could be included in the Reynard Way Park design. Further, we appreciate the continued enhancement of West Lewis Street Pocket Park, and we are proceeding with Phase II by working with the City to identify funding. Lastly, we very much appreciate inclusion of the Grant K8 Joint U se		inadequacy in the P
	Facility. This multi-year, large scope project will be a much-needed improvement to the community, especially if we can build the multi-use field, gymnasium, and an essential area for community meeting space.	B8-5	Comment noted.
	Planned Historic Districts (PHDs)		inadequacy in the F
00-0	As we wrote in our letter to the City on 11/19/2015, we are open to examining the process for which PHDs are implemented. At this time however, there is no mechanism for communicating to the		has been in place
	nomeowners that their property zone is changing. We are concerned that over 50% of Mission Hills single family residences are slated to become PHDs without letting the owners vote on whether or		providing an oppor
	325 West Washington Street • Suite 2 #159 • San Diego, CA 92103		1

- 88-1 Comment noted. The City appreciates Mission Hills Town Council (Council's) participation in the CPU and public review comment process.
- 38-3 Comment noted. The comment does not suggest an inadequacy in the PEIR.
- B8-4 Comment noted. The comment does not suggest an inadequacy in the PEIR.
- B8-5 Comment noted. This comment does not suggest an inadequacy in the PEIR. An extensive public outreach process has been in place throughout the Uptown CPU process, providing an opportunity for the public to provide input and public comment on the potential historic districts.

Mission Hills

not they want these restrictions. We have asked the City for a community presentation directly, but have been told the City does not present on Draft PEIRs. We have also requested a notification letter be sent to all homeowners affected, and have been told there is no funding available. We are uncomfortable supporting this aspect of the Plan, without clear communication of PHD impacts to the residents of Mission Hills, and therefore cannot support it.

B8-6 Transforming Reynard Way

B8-9

To reiterate our letter of 3/28/13, many areas along Reynard Way are very dilapidated. Many buildings are deteriorating and open spaces appear abandoned. Without redevelopment, this area will continue to stagnate both from a residential and business standpoint. We support the the Plan recommendations for this area.

B8-7 In addition, as Reynard Way represents a major connector to Little Italy and Downtown, our Council reiterates its specific request that the area along Reynard Way be specifically evaluated in a traffic study to determine if the two-way left turn lane could be eliminated so that Reynard Way can become a true Complete Street with a Class 2 buffered bike lane & intermittent landscaped median which would not only beautify the area but help calm traffic flow.

B8-8 • Multimodal Transport:

We support the increased use of multimodal transport options including prioritizing pedestrian and cycling options in our community. Our community eagerly awaits the implementation of Washington Street as a Complete Street, with the addition of a bike lane and sidewalk.

 India Street Improvements: While India Street is not in Mission Hills, it is used by many Mission Hills residents. We support India Street improvements including traffic calming, and more pedestrian-oriented safety improvements from Palm Street to Washington Street. The Five Points business area is frequented by many of our residents, and making this area safer and therefore more utilized, is of utmost importance to us.

Sincerely,

Belinda Sun

Belinda Smith President Mission Hills Town Council

CC: Leo Wilson, Chair, Uptown Planners Todd Gloria, Councilmember D3 Marlon Pangilinan, Senior Planner, Planning Department Susan McNeil Schreyer, Executive Director, Mission Hills BID Jim Reilly, President, Mission Hills Heritage Vanessa Herbert, Middletown Advisory Group

325 West Washington Street • Suite 2 #159 • San Diego, CA 92103

B8-6 Comment noted. The comment does not suggest an inadequacy in the PEIR.

- B8-7 This comment requests that a traffic study be conducted for Reynard Way to determine if the two-way left-turn lane could be eliminated, supporting a Class II buffered bicycle lane and landscaped median. The comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required. Implementation of a Class II bicycle facility along Reynard Way would require project-level environmental review.
- B8-8 Comment noted. The comment does not suggest an inadequacy in the PEIR.
- B8-9 Comment noted. The comment does not suggest an inadequacy in the PEIR.

LETTER

Letter B9 HILLCREST August 8, 2016 Senior Planner Marton Pangilinan City of San Diego Planning Department 202 C St San Diego, CA 92101 Dear Marlon, On behalf of myself and Rescue Hillcrest, a group of residents, business owners, and B9-1 B9-1 Comment noted. The City appreciates Rescue Hillcrest's stakeholders dedicated to protecting the quality of life in Hillcrest, I am writing to comment on the proposed Uptown Community Plan and its associated Preliminary participation in the Uptown CPU and public review comment Environmental Impact Report. process. IN GENERAL The proposed UCP in many ways fails to represent the considerable input from the B9-2 Comment noted. The comment does not suggest an inadequacy in B9-2 community over the past seven years on density and heights, especially as expressed in the PEIR. See Chapter 4.0, History of Project Changes Related to recommendations made by the Uptown Planners: CEQA, of the PEIR for a description of the community outreach A.) Per City Council Policy 600-24, the City Council recognizes one official planning B9-3 group in each community, which in Uptown is the Uptown Planners (Article III, Section undertaken. Also see the Staff Report for a discussion of the 2). Therefore, it is problematic if not fatal that there has been significant documentation that the proposed UCP countermands the input from Uptown Planners on the June 2015 stakeholder involvement and outreach efforts. and January 2016 drafts in deference to "other stakeholders." especially the private economic interests represented by The Atlantis Group. B9-3 Comment noted. This comment does not raise an inadequacy B.) It is damning that the city has not repudiated the documented efforts of the Hillcrest B9-4 Business Improvement Association to suborn the Uptown CPU process. The HBIA is regarding the PEIR. required to obey all laws as a condition of their contract to receive tax revenue. Yet they have boasted about actively undermining the legal authority of the recognized CPG, in violation of CP 600-24, and grossly overstepping the activities allowed under their enabling ordinance. Council Resolution R-273937. And they have admitted to lobbying B9-4 Comment noted. This comment does not raise an inadequacy on behalf of themselves and other private economic interests without registering as a regarding the PEIR. This City has considered input from all Uptown lobbyist, in violation of SDMC §27.4007. An article that reports these admissions, "How Hillcrest sidestepped Uptown Planners," published in The San Diego Reader on July 20. stakeholders, including residents, business owners, community 2016, can be found at the following hyperlink and is incorporated by reference with this email: http://www.sandlegoreader.com/news/2016/jul/20/citylights-rules-are-bentleaders, public officials, and other interested parties. hillcrest/. -1-



- B9-5 C.) The proposed Uptown CPU is evidence of the adage, "garbage in, garbage out." The population growth-driven mandates in it are even more arbitrary and unfounded as those which were proposed, and unrealized, in the <u>1988 UCP</u>. The 1988 UCP stated the population then as "approximately 36,500" (page 5), and projected a population of "approximately 38,700" in twenty years (page 147). The <u>proposed UCP</u> notes the current population is only 36,750, yet estimates "a future population of 55,700 people and 32,680 dwelling units" (page IN-11). Although in reality Uptown has experienced a net loss of 250 people in the past 28 years, the proposed UCP assumes an increase of the population or 18,950 or 52%, which would rival the growth prompted by WWII. Does anyone seriously believe this? Yet without justification, these figures drive the entire rationale of the proposed UCP.
- B9-6 D.) But even if such growth occurs, instead of having thresholds that trigger density and height increases as population levels actually increase, elected officials and planning staff have proposed development guidelines that require the situation to become critical before anything is done. So there is no concomitant strategy for mobility, recreasion, or public services to match the projected demand. This once again indicates that the proposed UCP is geared more toward serving private economic interests than the public good.
- B9-7 E.) The city has repeatedly used the zoning allowances from the 1988 UCP instead the existing conditions on the ground when making comparisons to the projections and allowances for the proposed UCP. Yet CEQA requires a "ground to plan" analysis. The continued justification for the proposed UCP by reference to 'what was allowed' under the 1988 UCP is not only facetious but illegal.
- B9-8 F.) Despite the community's great effort in establishing the Interim Height Ordinance and its continued support for keeping 50' ministerial and 65' discretionary limits, the proposed UCP would arbitrarily implement heights of 100' and 120' by means of CPIOZ. Yet the city has failed to demonstrate that build-out within the IHO limits would not be able meet the goals of the Climate Action Plan.
- B9-9 G.) Regarding CPIOZ, it is cause for concern that it is proposed to govern heights community-wide, which is an unprecedented expansion of its statutory purpose of "providing supplemental development guidelines that are tailored to specific sites within community plan areas" (§132,1401). But it is full-till alarming that the CPIOZ areas mapped in the proposed Uptown CPU do not match those shown in the existing municipal code for Uptown, which was last updated in 2000. (Reference §132,1402, Diagram 132-14K, Uptown CPIOZ, reproduction of Map No. C-780.2. Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000; amended 6-19-2000 by O-18814 N.S.) And it is uncertain that this contradiction will be resolved before the approval of the final UCP.

Comment noted. This comment does not raise an inadequacy regarding the PEIR. The future residential population was estimated based on the total number of dwelling units at full buildout of the proposed CPU land uses, and not necessarily on regional growth predictions alone. In other words, the estimated population at future build-out of the proposed CPU equals the total estimated capacity of the proposed CPU residential land uses. This comment states that under the adopted Community Plan, Uptown's population at 20 years following plan adoption in 1988 would be approximately 38,700. However, the adopted Community Plan at 20 years following adoption is not same as the adopted Community Plan at build-out, and should not be used to compare against the proposed CPU population at build-out. The build-out of the adopted Community Plan would support an estimated population of 58,870, which is greater than the estimated population at build-out of the proposed CPU (55.700).

RESPONSE

- B9-6 Comment noted. This comment does not raise an inadequacy regarding the PEIR. The Mobility Element. Public Facilities, Services and Safety Element and Recreation Element of the proposed Uptown CPU have identified capital improvement needs and include policies for the continued provision and enhancement of transportation infrastructure, recreation, and public services.
- B9-7 The PEIR includes a plan-to-ground analysis identifying the anticipated impacts associated with implementation of the proposed Uptown CPU and associated discretionary actions compared to existing ground conditions. The proposed CPU is also compared with the adopted Community Plan to provide context and background for the analysis.

-2-

RESPONSE

The Community Plan Implementation Overlay Zone (CPIOZ) is B9-8 intended to ensure that development proposals are reviewed for consistency with the use and development criteria that have been adopted for specific sites. It would not arbitrarily implement heights of 100 feet and 120 feet, as this comment suggests. Rather, the CPIOZ would require discretionary review in certain areas related to building height to ensure consistency with the proposed Uptown CPU design and development criteria. Under the high-rise building policies in the proposed Uptown CPU's Design Guidelines by Building Type, areas within the Uptown CPU area could be permitted to develop with buildings up to 100 feet in height. However, areas also covered by a CPIOZ Type A would be subject to ministerial review for development that does not exceed 50 feet within Mission Hills and 65 feet in Hillcrest and Bankers Hill/Park West. CPIOZ Type B areas would be subject to a discretionary review process that would implement the proposed Uptown CPU policies and recommendations, particularly those related to building height consistent with the Urban Design Element. Ensuring consistency with the proposed Urban Design Element policies through this process is necessary to preserve the existing neighborhood character and avoid potential impacts. The PEIR is not required to demonstrate that the height allowed under the current Interim Height Ordinance (IHO) would be adequate to meet the goals of the Climate Action Plan (CAP). The proposed Uptown CPU was developed over a lengthy process of stakeholder input and evaluation of consistency with the City of Villages strategy and the City's CAP. Increasing density in the areas subject to the proposed CPIOZ areas would further implement the City of Villages strategy and the City's CAP by increasing density in areas with transit access. B9-9 Comment noted. This comment does not raise an inadequacy regarding the PEIR. See response to comment B9-8 regarding the use of the CPIOZ as a tool to ensure development consistency with the proposed Uptown CPU design and development criteria.

LETTER



RESPONSE

LETTER

+ HILLCREST Cultural resources require immediate attention, as the 45-year review standard for architecture is inadequate for preserving buildings of significance in time. (Page HP-172) So as a community stakeholder, I am exercising my right to request that the implementation of the proposed Hillcrest historic district should receive priority under historic preservation Policies HP-2.2 and HP-2.3. (Page HP-173) 2.) Again, although I understand staff has already done so, I want to ensure that Policies B9-14 B9-14 Comment noted. This comment does not suggest an inadequacy in EP-2.1, 2.2, and 2.3 as written in the proposed UCP are struck from the final UCP, and that the BIAs are not included in any policies, as it is inappropriate for the city to the PEIR and does not require a detailed response. Policies EP-2.1, designate public-private partnerships of which it itself is a partner at the policy level. This EP-2.2, and EP-2.3 have been removed from the proposed Uptown is also relevant to removing the endorsement of the HBIA from EP-1.4. (Page 108) CPU and replaced with policy EP-2.1 that calls for the support of 3.) Insist that the Normal Street Linear Park and other park areas be developed under B9-15 programs and strategies for attracting, supporting, and retaining city control, such as the Parks and Recreation Department, rather than relinguish control of public land and public right-of-ways to private groups. This is in accordance with small businesses in Uptown. Policies MO-4.12 and RE-1.6. (Pages MO-52 and RE-135 THEREFORE: B9-15 Comment noted. Parks and recreation facilities included in Table 7-As I stated at the beginning, the proposed Uptown CPU in many ways fails to represent B9-16 1 of the proposed Uptown CPU that are within public lands and the considerable input from the community over the past seven years, especially as expressed in recommendations made by the Uptown Planners. However, if a choice public right-of-ways would be under control of the City. must be made from among the proposed Uptown CPU and the PEIR alternatives, and given the circumstances described in this letter, then the Density Redistribution Alternative represents the most fair and appropriate compromise among all B9-16 Comment noted. This comment identifies the Density concerned, has been determined the Environmentally Superior Alternative under Redistribution Alternative, which is the environmentally superior CEQA, and therefore should be incorporated into the final UCP. Under no circumstances is the proposed UCP as-is considered reasonable or acceptable. alternative, as the Council's preferred alternative. Finally, while I am on the board of the Uptown Planners and a member of the Hillcrest B9-17 Business Improvement Association, these remarks are not made on their behalf. And B9-17 Comment noted. these comments are not meant in any way to reflect negatively on you, Marlon, as I respect the professionalism with which you represent the city's positions (whether I agree with them or not), and believe you to fairly answer all questions asked. Thank you for your consideration. -4-

LETTER





	Letter B11		
	San Diego Canyonlands • 3552 Bancroft Street San Diego, CA 92104 • 619-284-9399 • • www.sdcanyonlands.org •		
	August 8, 2016		
	Marlon I. Pangilinan Senior Planner City of San Diego Planning Department 1222 First Avenue MS-413 San Diego, CA 92101 Email: <u>mpangilinan@candiego.gov</u> Email: <u>PlanningCEOA@sandiego.gov</u>		
	Re: Uptown Community Plan Update (June 2016), Focus on Maple Canyons		
	Dear Marlon,		
B11-1	These comments, for the Uptown Community Plan Update ("Update"), specifically address Maple Canyon and request some degree of flexibility for potential trail or canyon enhancement needs that may not be currently addressed in the Update. San Diego Canyonlands ("SDCL") is a non-profit organization that works to foster education and ongoing community involvement in stewardship and advocacy related to San Diego's canyons in collaboration with other community-based organizations.	B11-1	Comment noted. The City appreciates San Diego Canyonlands' (Canyonlands') participation in the public review comment process.
B11-2	Maple Canyon in Bankers Hill The Recreation Element in the Update accurately describes the existing trail through the bottom of Maple Canyon from W. Maple Street col-de-sac in the south to the trail that elimbs the slope on the west side adjacent to the Quince Street footbridge, (Figure 7.1 RE-134). The trail through the bottom also serves as a utility access road for Maple Canyon. Proposed trail improvements mentioned in the Update include "Design and construct trail amenities, such as benches, overlooks, interpretive signs, protective fencing, native landscaping, trash and recycling containers as determined and approved by City (RE-131)."	B11-2	Implementation of trails identified within the proposed Uptown CPU would be evaluated on a project level as each trail improvement is proposed. Erosion would be considered as part of this project level
	It will be important to address the impacts of watershed hydromodification, particularly the severe crosion in the main, ephemeral stream channel in the bottom of Maple Canyon. before any new trails are designed or amenities are installed.		review. In general, addressing erosion issues would be an important factor for future trail implementation.
311-3	Consistent with the Conservation Element in the Update, goals and policies, specifically CE-2.21 "Identify opportunities for additional hydromodification management measures to protect natural drainages from erosion and other problems", SDCL and the Friends of Maple Canyon are working in coordination with the City of San Diego to design and implement solutions to stabilize ephemeral stream channels in Maple Canyon. SDCL has funding from the California Coastal Conservancy for the conceptual design phase. Since specific improvements to Maple Canyon that may be needed as part of a streambed rehabilitation project are not known at this time, trails and other potential amenities in the canyon bottom should occur subsequent to the streambed stabilization project.	B11-3	See response to comment B11-2.
B11-4	Conservation Element (CE-2.7) states, "Repair and retrofit storn drain discharge systems to prevent erosion and improve water quality by adequately controlling flow and providing filtration. Storm drain outfalls should limit the use of concrete in favor of more natural, vegetated designs."	B11-4	Comment noted. This comment does not raise an issue with regard to the adequacy of the PEIR. The City will be coordinating planning

B11-5 Comment noted. This comment does not raise an issue with regard to the adequacy of the PEIR. SDCL and the Friends of Maple Canyon very much appreciate the content of this element and we have made B11-6 Comment noted. The information provided about the SDCL's identical comments to the City Transportation and Storm Water Department ("TSW") that is planning to repair or Canyon Enhancement Planning (CEP) stakeholder process is replace 14 storm drains leading into Maple Canyon in fall of 2017. City Planners and the Uptown Planners official community advisory group should follow up with TSW regarding this important element of the Update. acknowledged. While new trailhead locations might be identified prior to the installation of these 14 new storm drains, trail routes and trail designs should only occur subsequent to their installation. B11-7 Comment noted. The City acknowledges the efforts of the CEP B11-5 Conservation Element Policy CE-2.6 states, "Habitat restoration efforts should aid wildlife movement by providing vegetative cover and controlling and directing access to designated trails." stakeholders in siting a trailhead at the West Maple Canyon Pocket Park. Directing access to designated trails is an important solution toward reducing erosion of Maple Canyon's slopes caused by unapproved, unsustainable "social" trails. Currently there are no designated trails on the east side of Maple Canyon, and there is only one trailhead on the west side (at 3rd Ave. and Quince St.) that leads from the top B11-8 Figure 7-1 in the proposed Uptown CPU displays the approximate of the slope to the bottom of Maple Canyon. location of existing and proposed parks and recreation facilities, SDCL is in the middle of a Canyon Enhancement Planning stakeholder process for Maple Canyon in which B11-6 and open space including trails. Trail locations shown on the figure community stakeholders are discussing a number of potential trailheads for access to Maple Canvon, none of which are decided at this point. We hope to make presentations to both the Bankers Hill Community Group and are approximate, and are provided to illustrate general trail Uptown Planners in the coming months. alignments and connections to the community. Final alignments will The Recreation Element does identify a proposed trailhead at the "West Maple Canyon Pocket Park" (RE-B11-7 be determined as specific trail improvement projects are 130) on the east side of the canvon. This is one, of many trailheads being considered by the Canvon Enhancement Planning stakeholders group for Maple Canyon, but again, nothing is decided at this point. implemented. The proposed Uptown CPU does not preclude additional trails from being proposed and developed in the future. B11-8 SDCL and the Friends of Maple Canyon respectfully request language and/or a defined process within the Update that will make allowances for additional or improved trails or other approved amenities, without Trails may be proposed and implemented without an amendment requiring a Community Plan Amendment, under the following conditions: to the community plan. However, any new or improved trail facilities 1. The enhancement plans do not conflict with other City-approved land use plans and policies, including the Update: that are intended to meet the park equivalency requirement would 2. The Uptown Planners approves any proposed enhancements; 3. The enhancement plans will be considered under a separate CEQA review process (if CEQA review is require an amendment to the community plan to record the park required). equivalency credit and to be included in the Impact Fee Study. Subject to the foregoing three conditions, and if approved by the City, trails and amenities could be designed and installed where needed and appropriate for the trail type, without requiring a Community Plan Amendment. Trails and park amenities such as benches, interpretive signs, Going forward, SDCL will prioritize our Canvon Enhancement Planning process in other City canvons to B11-9 fencing, etc. may also be proposed and installed without a complement all of the City's community plan updates. Considering the dozens of canyons SDCL serves requirement for a community plan amendment. Within the throughout the City that are in need of enhancement, restoration, and improved access, requiring a Community Plan Amendment to approve every aspect of these specific kinds of enhancements would be unnecessarily proposed Uptown CPU, Table 7-1, Population-Based Parks and cumbersome. We hope you agree. Recreation Facilities Inventory and Recommendations include Thank you for your consideration of these comments. flexibility for the trail improvements. As an example, the following Sincerely. language provides necessary flexibility: "Design and construct trail Inic Soulle amenities such as benches, interpretive signs, protective fencing, Eric Bowlby, native landscaping, trash and recycling containers, overlooks, etc., Executive Director where needed and appropriate for the trail type, as determined and San Diego Canvonlands Canyons - The Geographic DNA of San Diego approved by City." 2 B11-9 Comment noted. The City appreciates the efforts of the CEP group's efforts to compliment the City's CPUs. Refer to response B11-8.



- B12-1 Comment noted. The City appreciates San Diego County Archaeological Society, Inc.'s participation in the CPU and public review comment process.
- B12-2 Comment noted.

LETTER

Letter B13 RON ROBERTS CHAIRMAN SUPERVISOR, FOURTH DISTRICT SAN DIEGO COUNTY BOARD OF SUPERVISORS August 3, 2016 Kurtis Steinert Senior Environmental Planner City of San Diego Planning Department 1010 Second Ave, MS 413 San Diego, CA 92101 Subject: UPTOWN COMMUNITY PLAN UPDATE, PROJECT NO. 21002568 Dear Mr. Steinert: B13-1 B13-1 Comment noted. Refer to the Master Response Regarding India As Chainman of the San Diego County Board of Supervisors and a resident of the Uptown community, I would like to express my strong opposition to the India Street improvement options included in the Uptown Street Mitigation Measures included in the introduction to these Community Plan Update. response to comments (pages RTC-4 through RTC-6). Project U-17A proposes to convert India Street between Washington and Winder streets from a two-lane B13-2 collector into a two-lane collector with a continuous left-turn lane. It further converts India Street between Glenwood Drive and Sassafrass Street from a one-way two-lane collector into a one-way four-lane collector. Comment noted. Refer to the Master Response Regarding India B13-1 Project U-17B converts India Street between Sassafrass and Redwood Street from a one-way two-lane collector into a one-way three-lane collector. These modifications would result in a significant loss of on-Street Mitigation Measures included in the introduction to these street parking spaces and require additional right-of-way from the adjacent residences and businesses. response to comments (pages RTC-4 through RTC-6). Projects U-17A and U-17B appear as options in both the Uptown Community Plan Draft Environmental B13-3 Impact Report (EIR) and the Mobility Study completed by Kimley Horn, Appendix C to the Draft EIR. The improvements are laid out in Figure 63 of the Mobility Study, which includes the notation: "These changes B13-1 Comment noted. Refer to the Master Response Regarding India in geometry are not recommended." Table 13 of the Mobility Study, "Summary of Improvement Evaluation, Uptown," specifies 25 parking spaces, 11 businesses and five residences would be impacted, while also Street Mitigation Measures included in the introduction to these noting that these improvements are not recommended. Finally, the Draft EIR, in identifying which mitigation measures will be included in the community plan update and the Impact Feasibility Study, does not response to comments (pages RTC-4 through RTC-6). recommend these projects for inclusion. B13-4 While it is encouraging these improvements are not included in the draft of the Uptown Community Plan B13-1 Comment noted. Refer to the Master Response Regarding India Update, I want to go on record registering my opposition to these projects being added to the plan or considered in the future. Thank you for your efforts to develop long range plans to help our community grow Street Mitigation Measures included in the introduction to these in a sustainable way. response to comments (pages RTC-4 through RTC-6). RON ROBER Chairman San Diego County Board of Supervisors COUNTY ADMINISTRATION CENTER . 1800 PACIFIC HIGHWAY, ROOM 335 - SAN DIRUG, CALIFORNIA 92101-2470 (619) 531-5544 · Fax (619) 531-6262 · E-Mail: Ron Roberts Bedounty.ca.gov raberia con @ Printed on recycled paper

LETTER



Ltr. to City of San Diego Planning Department August 4 2016 Page 2

- B14-4 The KOA study provides several options to create better ingress and egress for the Hillcrest Medical Center (study attached). Existing circulation routes to the Hillcrest Medical Center are circuitous and not direct; as a result the primary reason for improved circulation to the Medical Center is to improve access and address life safety concerns such as emergency transport delays, pedestrian safety, traffic circulation, visitor and patient way-finding/driver confusion, and public transit.
- B14-5 UC San Diego has shared the KOA study with the City in 2013 and discussed the circulation concerns described above. The public review draft Uptown Community Plan Update EIR does not however reflect the plan and need for improved access in the area of the UC San Diego Medical Center. UC San Diego requests that the City discuss and include the KOA findings in the Uptown Community Plan Update and EIR. This would enable the future project to move forward in a way that would be consistent with the vision of the community plan.

Thank you for your consideration.

Sincerely,

An

Robert Clossin Director Physical and Community Planning

With electronic copies to:

Jeff Murphy Marlon Pangilinan Samir Hajjiri Gary Pence

- B14-4 Comment noted. This comment in informational in nature and does not suggest an inadequacy in the PEIR.
- B14-5 Policies included in the proposed Uptown CPU support traffic operational improvements to facilitate ingress and egress to and from the UC San Diego Medical Center in Hillcrest. Any future project proposed by the UC San Diego Medical Center would require submittal of a transportation technical study to Development Services Department and Transportation and Storm Water Department for review and approval. Discussion of the referenced KOA report in the proposed Uptown CPU is not needed since there is an existing policy framework supportive of facilitating ingress and egress to and from the UC San Diego Medical Center.

LETTER

	Letter B15 Warlon Pangilinan Senior Planner The City of San Diego 1010 Second Avenue, MS-413 San Diego, CA 92101		
B15-1	This letter is to inform you that Save Hillcrest supports the Density Redistribution Alternative (DRA). In addition to our support of this alternative we request that historic districts be returned to	B15-1	Comment noted. The City appreciates Save Hillcrest's participation in the public review process.
B15-2 B15-3	An auditor to our support of this areanative we request that historic districts de retified to the plan. Save Hillcrest maintains that community input was overlooked throughout the process and that the final draft is seriously flawed. Sincerely, Nancy Moors Chair	B15-2 B15-3	This comment requests that historic districts be returned to the plan. It is unclear what is meant by this comment; however, historic districts and associated policies are included in the Historic Preservation Element of the proposed Uptown CPU. The City conducted an extensive outreach program as detailed in Section 4.2, Community Outreach and Plan Development, of the PEIR and discussed in the Staff Report. While the City incorporated stakeholder input in the proposed Uptown CPU, it is impossible to meet all needs of every stakeholder involved. This comment also suggests that the PEIR is flawed; however, no specific examples or issues are provided that would allow the City to provide a detailed response.
	501(c)3 26-3034556 425 Robinson Street, San Diego, CA 92103 (619) 379-3416		

ER	TAGEORO					Letter B16
SAVE OUR	CID SATION.	§ave	Our	Heri	tage O Saving S	「ganisation San Diego's Past for the Futur
00	NDED 190					
	Tuesday, July City of San Di c/o Kurtis Stei 1010 Second / San Diego, Ca	26, 2016 iego Planning De nert, Senior Envi Avenue, MS 413 lifornia 92101	epartment ironmental P	lanner		
	Re: Hillcrest H	listoric District l	mplementati	on		
	Mr. Steinert an	nd the Planning	Division,			
B16-1	On June 27, U potential histo July 2015 dra Uptown Com	Jptown Planner pric district (wh ft of the Uptown munity Plan.	s voted (8-1- ich includes n update, and	2) to urge the applying the l l for this to be	City Staff to impl LGBTQ context), done concurrent	lement the Hillcrest as identified in the ly with the Final
B16-2	SOHO suppor historic distri- resources.	rts this motion a cts, concurrent v	and strongly with the plar	believes that n, is the only e	implementation of ffective mitigation	f all the potential n for historic
	Thank you for International Bruce Coons Executive Dir Save Our Her	r the opportunit	y to commer	nt,		
M. Wayn	Jaye MacAskill, Pr e Donaldson • Erik Har	esident • David Gol Ison • Paul Johnson	BOARI dberg, Vice Pro • Nancy Moors Bruce Coo	O OF DIRECTOF esident • Jessica • John Rush • Sco ns, Executive Di	IS McGee, Treasurer • . tt Sandel • Sandor Shap rector	John Eisenhart, Secretary pery • Kiley Wallace • Elizabeth Weems
2476 S	San Diego Avenu	e • San Diego (CA 92110 ·	www.sohosar	ndiego.org • 619/	297-9327 • 619/291-3576 fa)

- B16-1 Comment noted. The City appreciates the Save Our Heritage Organisation's participation in the public review process. The LGBTQ Historic Context Statement is a separate document on a different track than the Uptown CPU. Once completed, the LGBTQ Historic Context Statement will guide the identification, evaluation and preservation of LGBTQ resources Citywide, including the Uptown planning area, and will be used in conjunction with all other applicable contexts and surveys when evaluating resources in Uptown.
- B16-2 Comment noted. As discussed in Section 6.7.4, Impact Analysis, while the Hillcrest Potential Historic District survey was not initially identified as a potential historic district by the 2004 and 2006 survey work, the area may be eligible under Historical Resources Board Criteria A and C. As discussed in the proposed Uptown CPU, in order to bring the Hillcrest Potential Historic District forward for designation, additional, intensive level research would be required to evaluate the district and define a precise boundary, period of significance, significance criteria, and contributing and noncontributing resources. The PEIR discloses potentially significant and unavoidable impacts to potential historic districts in Section 6.7, Historical Resources, of the PEIR and provides a mitigation framework including supplemental development regulations that would protect potential historic districts in the interim until they can be formally designated.



LETTER

	EIR Comments July 28, 2016 Page 2	B17-3	Mitigation measures HIST 6.7-1 calls for avoidance, which is preferred, or site-specific mitigation of historic resources impacts for any development implemented in accordance with the proposed CPU. The proposed CPU provides adequate flexibility and incentive for preservation of historic resources. In addition
B17-3	 2. Please also revise the EIRs to identify and analyze the following mitigations to facilitate historic preservation and rehabilitation: a. Exclude historic resources from calculations of floor area ratio, increasing density for projects that retain an historic resource; b Exclude historic resources from parking calculations, allowing reduced parking when a project retains an historic resource; c Include city-wide transferable development rights (TDRs), enabling property owners to buy and sell rights to encourage projects near transit and amenities. 		the Municipal Code currently provides incentive opportunities, including Conditional Use Permits to facilitate adaptive reuse and Planned Development Permits to allow for deviations from development standards to achieve a better project, such as one that preserves and incorporates a designated historic resource. Inclusion of the measures recommended in this comment is not needed to further reduce significant historical resources impacts. Even if those measures were added, the degree of future impacts and applicability, feasibility, and success of future mitigation measures would not be known for each specific future project at
B17-4	3. The EIRs state that plan impacts to historic resources are "significant and unavoidable." Supplemental development regulations proposed to mitigate impacts to potential historic districts require additional implementation measures to be ineffective: a Will the EIRs be revised to impose timelines for implementation of the regulations, establishing priorities for potential historic districts? If not, why not? 	B17-4	a program level of analysis and impacts would remain significant and unavoidable. The proposed supplemental development regulations are not proposed as a mitigation measure; rather they are part of the project. Implementation of the supplemental development regulations would occur concurrent with approval of the
B17-5	\hat{b} Will the EIRs be revised to consider feasible funding for the regulations? If not, won't the mitigation be unenforceable?		proposed Uptown CPU. Thus, the protections for potential historic districts would be in place immediately with adoption of the proposed CPU and a timeline for implementation of the
B17-6	<i>c</i> Will the EIRs be revised to consider amendments to the supplemental development regulations so that they comprehensively apply to all resources within identified potential historic districts? If not, why not?		regulations is not needed. A draft work program for intensively surveying and processing the potential historic districts has been developed with input from the community, and will be further
B17-7	<i>d</i> Will the EIRs be revised to apply regulatory protections to community- identified potential historic districts that the City agrees are eligible, the Multiple Property Listings (MPLs), and the commercial properties located within potential districts? If not, why not?		refined as it is implemented. Significant and unavoidable impacts are identified even after implementation of the mitigation framework because the degree of future impacts and applicability, feasibility, and success of future mitigation

B17-4 (cor	nt.) measures cannot be adequately known for each specific future project at a program level of analysis. Mitigation measure HIST 6.7-1 and CPU policies protecting historic resources will be implemented to avoid or reduce impacts resulting from development to the greatest extent feasible. Policies included in the proposed CPU would be implemented at the time of CPU adoption.
B17-5	As stated in response B17-4 above, the supplemental development regulations (amendments to the Historical Resources Regulations) are a project feature and will be implemented ahead of the adoption of the CPU as they are scheduled to go before the City Council prior to the proposed CPU. Thus, the supplemental regulations would be enforceable as it would become a part of the Historical Resources Regulation upon approval and would be implemented accordingly.
B17-6	The Draft PEIR did consider the proposed supplemental development regulations in the analysis; therefore, the PEIR does not require revision. The amendments to the Historical Resources Regulations are identified as part of the project in Chapter 3. Refer to Table 3-1 which identifies adoption of zoning amendments to the Historical Resources Regulations and amendments to the Neighborhood Development Permit regulations to address Potential Historic Districts as project components.
	Applicability of the supplemental development regulations is detailed in the proposed code language (https://www.sandiego.gov/sites/default/files/draft_potential_hist oric_district_regulations_05312016.pdf) which specifies that the regulations would apply to single dwelling unit or multiple dwelling unit development on a premises within a potential

B17-6 (cont.)

historical district as specified in a land use plan when the premises has been identified as a potential contributing resource to the potential historical district. The regulations do not apply to all buildings within a potential historic district unless they are specifically identified as a contributing resource. The regulations would not apply to structures that are not identified as contributing resources to the potential historic district because non-contributing resources do not add any value to the potential historic district and their alteration would not further detract from the Potential Historic District.

B17-7 See response to comment B17-6 regarding applicability of the supplemental development regulations. The PEIR will not be revised. As proposed, the PEIR identifies the community identified potential historic districts in an effort to fully disclose the potential environmental impact. However, the supplemental development regulations will only apply to the potential historic districts identified during the reconnaissance survey efforts. The decision makers will ultimately decide what potential historic districts will be subject to the supplemental regulations. Multiple Property Listings (MPLs) are not potential historic districts and are protected through current regulations requiring evaluation of resources 45 years old or older.

B17-8	EIR Comments July 28, 2016 Page 2	B17-8	The Draft PEIR already considered the proposed amendments to the Historical Resources Regulations and additional revisions are not required. See response to comment B17-4; potential historic districts will be immediately protected with the proposed amendments to the Historical Resources Regulations (https://www.sandiego.gov/sites/default/files/draft_potential_hist oric_district_regulations _05312016.pdf). These regulations provide immediate protections until a formal Historic
B17-9	f Will the EIRs be revised to consider requiring homeowners of historic properties to obtain permits for changes to windows and doors? The HRB staff consistently determines that historic resources are not eligible for designation when original windows and doors have been replaced. However, since the supplemental regulations do not require permits for changes to windows and doors that maintain the same size and location (the issue being a change in material), the HRB staff does not have a way to oversee this type of detrimental change to historic fabric, the removal of fabric is not documented by the HRB staff, and this change is then used as leverage that the resource is not historic. Will the EIRs address this problem? If not, why not?		Designation determination can be made in accordance with policies HP-2.1 and HP-2.2. Projects that do not comply with the requirements of the supplemental regulations will be required to process a discretionary Neighborhood Development Permit, at which time project-specific mitigation may be identified. The Historical Resources Board is an appointed body with authority over historical resources in the City and are well-practiced in designating individual historical sites, establishing historical districts, and reviewing development projects that may affect historical resources. At least 4 of the Board members meet the
B17-10	g Will the EIRs consider enforceable protections to facilitate historic property homeowners' compliance with the <i>Secretary of the Interior's</i> <i>Standards</i> and the retention of historic fabric? While we understand code compliance violations within potential historic districts are near the top of the priority list, the repercussions when compliance is not met are little more than a slap on the wrist. The regulations also need to be proactive toward developers that act first and seek forgiveness afterwards.		U.S. Secretary of the Interior's Professional Qualification Standards, and all Board members receive training on a yearly basis on the identification and preservation of historic resources. Historical Resources staff members also meet the U.S. Secretary of the Interior's Professional Qualification Standards, and attend professional seminars, trainings and conferences.
B17-11	\hbar Will the EIRs consider infill guidelines to effectively protect the potential districts from incompatible change, i.e. scale, bulk, rhythm, and materials, for parcels that do not include a historic resource, but are located within a potential district? Isn't it true that infill guidelines are necessary to ensure the potential historic district remains intact until such time when the district is brought forward?	B17-9	The PEIR will not be revised as requested in this comment. Window replacements within the original openings, which are the only window modifications exempt from a permit, do not in and of themselves preclude a building from contributing to a historic
B17-12	<i>i.</i> Will the EIRs consider the following suggestions? If not, why not?: The regulations' use of the term "original footprint" is ambiguous and will lead to trivial interpretive disputes; this term should be eliminated or clearly defined in the policy document. The limitation to keep exterior		district. Thus, the proposed supplemental development regulations do not require revisions and PEIR revisions are not warranted. Additionally, the proposed CPU include policies to better inform and educate the public, including businesses, on the merits of historic preservation as well as to promote the maintenance and restoration of privately owned historical resources through incentive programs.

B17-10	This comment does not suggest an inadequacy in the analysis of the PEIR. The amended Land Development Code regulations would be enforceable through Neighborhood Code Compliance. A permit would not be issued without compliance and consistency with all applicable regulations.
B17-11	The PEIR will not be revised as requested in this comment. Infill guidelines applicable to non-contributing resources would not be needed because existing zoning and land development code requirements would provide adequate regulations for bulk and scale appropriate to each specific Potential Historic District
B17-12	This comment does not suggest an inadequacy in the analysis of the PEIR, rather it addresses the proposed amendments to the Historical Resources Regulations. This comment suggests that corner properties would not be appropriately protected. Since circulation of the draft PEIR for public review, the City has prepared revisions to the supplemental regulations for potential historic districts to specifically address corner lots, and to define the term "original primary façade." The City does not agree that alterations to the rear third of a building would facilitate inharmonious change and awkward projections, as this comment suggests as the entirety of the structure would still be subject to applicable zoning limitations and would generally not be visible from the street.

RESPONSE

LETTER

EIR Comments July 28, 2016

Page 2

visible alterations to the rear 1/3 of a building will facilitate inharmonious change and awkward projections, while simply requiring compliance with the *Secretary of the Interior's Standards* would allow property owners more flexibility and provide for a more harmonious project that is compatible with the surrounding neighborhood. Corner properties, having two full facades visible to public rights-of-way, are not appropriately protected under these supplemental regulations. Additional provisions need to ensure that these most visible properties are treated with extra sensitivity, as corner properties, have especially large visual impacts within historic districts.

Thank you for your attention to these comments.

Sincerely, Susan Brandt-Hawley Attorney for SOHO

LETTER

RESPONSE

Letter B18 Steinert, Kurtis Lewis, Parnela <plewis@allenmatidins.com> Monday, August 08, 2016 3:31 PM Steinert, Kurtis Chine, Jaffrey; Riley, Heather Comments on the Uptown Community Plan Update Program EIR 201608081513.pdf From: Sent: To: Cc: Subject: Attachments: Please see attached correspondence and confirm receipt. The link to the attachment will be toward the bottom of this email. Thank you. Pamela T. Lewis Legal Secretary to Joe M. Davidson; Michael J. Holmes; Heather S. Riley; Hina Gupta Allen Matkins Leck Gamble Mallory & Natsis LLP 501 West Broadway, 15th Floor, San Diego, CA 92101-3541 (619) 233-1155 (main) (619) 233-1158 (fax) Allen Matkins As of August 1, 2016, the address of our downtown Los Angeles office is: 865 South Figueroa Street, Suite 2800, Los Angeles, CA 90017-2543 Confidentiality Notice: The information contained in this electronic e-mail and any accompanying. attachment(s) is intended only for the use of the intended recipient and may be confidential and/or privileged. If any reader of this communication is not the intended recipient, unauthorized use, disclosure or copying is strictly prohibited, and may be unlawful. If you have received this communication in error, please immediately notify the sender by return e-mail, and delete the original message and all copies from your system. Thank you.

family residential uses outside near canyons and where transit

Allen Matkins Leck Gamble Mallory & Natsis LLP Allen Matkins B18-1 Comment noted. The City appreciates the Uptown Gateway Attomevs at Law 501 West Broadway, 15th Floor | San Diego, CA 92101-3541 Telephone; 619,233,1155 | Facsimile: 619,233,1158 Council's (Council's) participation in the public review process. www.allenmatkins.com Jeffrey A, Chine B18-2 Comment noted. E-mail: jchine@allenmatkins.com Direct Dial: 619.235.1525 File Number: 374268-00005/SD844741.01 Via Email/Hand Delivery B18-3 This comment expresses concerns with the proposed Uptown August 8, 2016 CPU's ability to meet the goals of the General Plan and the Climate Action Plan (CAP). It is important to note that Kurtis Steinert Senior Environmental Planner implementation of the CAP is not based solely on an increase in City of San Diego Planning Department 1010 Second Avenue, MS 413 residential densities; but focuses on providing residential San Diego, California 92101 density within transit priority areas. The proposed Uptown CPU Re: Comments on the Uptown Community Plan Update Program expresses the goals of the CAP by providing high residential Environmental Impact Report, Project No. 21002568 densities within transit priority areas compared to existing Dear Mr. Steinert: conditions. While densities are reduced compared to the B18-1 Thank you for the opportunity to comment on the Uptown Community Plan Update ("CPU") existing plan in some areas, the proposed Uptown CPU Program Environmental Impact Report ("PEIR"). These comments are submitted on behalf of the maintains the transit-oriented development focus with the Uptown Gateway Council, the Association of Commercial Property Owners in Hillcrest ("Council"). highest densities allowed within transit priority areas. The Introduction proposed Uptown CPU also expresses General Plan policies through site-specific recommendations to both implement The goal of this letter is to assist the City of San Diego ("City") in its efforts to implement B18-2 the General Plan and the Climate Action Plan ("CAP"). The Council wholeheartedly supports this citywide goals and policies and address community needs. endeavor as the 1988 Uptown Community Plan is severely outdated. At the outset, we wish to Lower residential densities in some areas are required to acknowledge the hard work of City staff and the challenges inherent in updating any community plan, particularly in a community such as Uptown with multiple stakeholders possessing disparate ensure that the bulk and scale of development maintain the visions for their community. existing neighborhood character as well as public views of B18-3 The Uptown Community Plan area presents a unique opportunity to further the laudable canyons and open space. The proposed land uses locate the goals of the General Plan and the CAP. Regrettably, the proposed CPU fails to capitalize on that potential. Rather than reflect smart growth planning principles, the CPU would effectively highest intensity uses along transit corridors where existing and "downzone" the Uptown Community Plan area. The CPU would displace people and eliminate homes from one of the most transit rich and development ready areas in the City. The CAP, with its future commercial, residential, and mixed-use development can recently enacted implementation measures, and the City of Villages strategy of the General Plan support existing and planned transit investments in the chart a bold but necessary course for future development within the City. Yet the forward-thinking vision contained in these planning documents will never be achieved if the City fails to implement community. Commercial uses are also used strategically by the these fundamental policies through the community plan update process. proposed Uptown CPU to encourage commercial uses along transit corridors. This transit-oriented development pattern is Los Angeles | Orange County | San Diego | Century City | San Francisco necessary to meet the goals of the General Plan's City of Villages Strategy and the CAP. Therefore, placing lower-density, single-

B18-3 (cont.)

and mixed uses are generally less common, and placing higherdensity residential uses along main transit corridors and near mixed-use commercial and employment areas would further the goals of the City of Villages Strategy and the CAP. Additionally, while redevelopment, by its nature, causes temporary displacement, the proposed Uptown CPU would not result in the permanent displacement of residences. See also Section 6.1.3 and Section 6.5.3 for discussions on the proposed CPU's consistency with the General Plan and CAP.

	Allen Matkins Leck Gamble Mallory & Natsis LLP Attorneys at Law		
	Kurtis Steinert August 8, 2016 Page 2		The proposed liptown CDL would result in a pat loss in total
B18-4	In this regard, the CPU as currently proposed is inconsistent with the General Plan and the CAP. The CPU abandons key City policies by reducing rather than increasing densities. As a result, the Council cannot support the proposed CPU. We urge the City to pause and reconsider its vision for Uptown in a manner that will accommodate population growth while reducing greenhouse gas ("GHG") emissions. To do so, the City will need to significantly revise the CPU to address the issues raised in this letter.	618-4	residential units compared to the adopted Community Plan and would reduce densities along some transit corridors where existing height limits would allow development that is out of character with the existing setting. However, the proposed
B18-5	For the same reasons, the PEIR in its current form fails to meet the minimum requirements of the California Environmental Quality Act ("CEQA"). The PEIR also must be revised to address the various deficiencies outlined below.		Uptown CPU places the highest residential densities within close proximity of transit and commercial services and near job
B18-6	Attached hereto are a number of exhibits prepared by the Council's experts. These reports from the Atlantis Group Land Use Consultants ($\underline{Exhibit A}$), Safdie Rabines Architects ($\underline{Exhibit B}$) and Scott Moomjian ($\underline{Exhibit C}$) are hereby incorporated into this letter. This letter also incorporates by reference the following correspondence previously submitted to the City:		centers, which furthers the City of Villages Strategy and goals of the CAP. For more discussion on the proposed CPU's consistency with the General Plan and CAP, see Section 6.5.3, Impact Analysis
	Letter from Marcela Escobar-Eck to Marlon Pangilinan, dated December 1, 2015;		inpact Analysis.
	• Letter from the Council to Jeff Murphy, dated June 8, 2016.	B18-5	Comment noted. This comment introduces deficiencies
	For the City's convenience, these letters are attached hereto as <u>Exhibit D</u> and <u>Exhibit E</u> , respectively. We respectfully request City responses to the comments contained in all of the attachments.		identified in the PEIR, which are specified in later comments.
D10.7 The primary purpose of CEOA is	CEQA The primary purpose of CEOA is to "filnform governmental decision makers and the public	B18-6	Comment noted. Responses to comments included in the exhibits are provided beginning with B18A-1 and on.
DIO /	about the potential, significant environmental effects of proposed activities." ¹ To ensure that public agencies consider the full environmental consequences of a project, the EIR is considered to be the "heart" of CEQA. ² An EIR "serves not only to protect the environment but also to demonstrate to the public that the environment is being protected." ³ In addition, an EIR "demonstrate[s] to an apprehensive citizenry that the agency has, in fact, analyzed and considered the ecological implications of its action." ⁴	D10 7	Comment noted. This comment is informational in nature are
		D10-7	does not suggest an inadequacy in the PEIR.
	 14 Cal. Code Regs. § 15002(a)(1). 14 Cal. Code Regs. § 15003(a). 14 Cal. Code Regs. § 15003(b). 14 Cal. Code Regs. § 15003(d). 		
ATTACHMENT 7

	Allen Matkins Leck Gamble Mallory & Natsis LLP Attorneys at Law		
	Kurtis Steinert August 8, 2016 Page 3		
B18-8	Toward that end, the California Supreme Court has explained that:	B18-8	Comment noted. This comment is informational in nature and
	Because the EIR must be certified or rejected by public officials, it is a document of accountability. If CEQA is scrupulously followed, the public will know the basis on which its responsible officials either approve or reject environmentally significant action, and the public, being duly informed, can respond accordingly to action with which it disagrees. The EIR process protects not only the environment but also informed self-government. ⁵		does not suggest an inadequacy in the PEIR.
B18-9	The law is well-settled that an EIR "should be prepared with a sufficient degree of analysis to provide decisionmakers with information which enables them to make a decision which intelligently takes into account [] environmental consequences." ⁶ Although "CEQA does not require technical perfection in an EIR," the courts do look for "adequacy, completeness, and a good faith efforts at full disclosure. A court does not pass upon the correctness of an EIR's environmental conclusions, but only determines if the EIR is sufficient as an informational document." ⁷	B18-9	Comment noted. This comment is informational in nature and does not suggest an inadequacy in the PEIR.
B18-10	An EIR "must contain facts and analysis, not just the bare conclusions of a public agency. An agency's opinion concerning matters within its expertise is of obvious value, but the public and decision-makers, for whom the EIR is prepared, should also have before them the basis for that opinion, so as to enable them to make an independent, reasoned judgment." ⁸ Failure to comply "with the information disclosure provisions of [CEQA] which precludes relevant information from being presented to the public agencymay constitute a prejudicial abuse of discretion" ⁹	B18-10	Comment noted. This comment is informational in nature and does not suggest an inadequacy in the PEIR.
	Project Description		
B18-11	"An accurate. stable and finite description of a project is basic to an informative and legally sufficient EIR. A curtailed or distorted description of the project may stultify the objectives of the reporting process." ¹⁰ An inaccurate project description renders the resulting environmental analysis invalid. ¹¹	B18-11	Comment noted. This comment is informational in nature and does not suggest an inadequacy in the PEIR.
B18-12	The PEIR defines the CPU as "the comprehensive update to the Uptown Community Plan, which is intended to guide development through 2035 build-out of the Community Plan The	B18-12	This comment restates information from Chapter 3, Project Description, of the PEIR and does not identify a deficiency in the PEIR: therefore, a detailed response is not required.
	 ⁵ Laurel Heights Improvement Association v. Regents of the University of California (1988) 47 Cal.3d 376, 390. ⁶ 14 Cal. Code Regs. § 15151. ⁷ 14 Cal. Code Regs. § 15003(i). ⁸ Kings County Farm Bureau v. City of Hanford (1990) 221 Cal.App.3d 692, 736. ⁹ Pub. Resources Code § 21005(a). ¹⁰ Kings County Farm Bureau, supra, 221 Cal.App.3d at p. 738. ¹¹ Communities for a Better Environment v. City of Richmond (2010) 184 Cal.App.4th 70, 80-89. 		

Allen Matkins Leck Gamble Mallory & Natsis LLP Attorneys at Law

Kurtis Steinert August 8, 2016 Page 4

proposed CPU provides detailed policy direction to implement the General Plan with respect to the distribution and arrangement of land uses (public and private); local street and transit network; prioritization and provision of public facilities, community, and site-specific urban design guidelines; and recommendations to preserve and enhance natural open space and historic and cultural resources within the Uptown community.¹² A specific project objective is to "[m]aintain or increase the housing supply through designation of higher residential densities along major transit corridors.¹³

- B18-13 Towards that end, the PEIR states that residential density "would be increased from the adopted Community Plan in some areas and reduced in some areas to help achieve [project] objectives."¹⁴ The next sentence of the PEIR, however, reveals a basic flaw in the CPU, which seriously undermines the City's efforts to further the General Plan policies. The PEIR indicates that the "CPU results in an overall communitywide reduction of future housing units at Community Plan build-out when compared to the adopted Community Plan."¹⁵ This information is buried in the Land use Element text of the Project Description chapter.
- B18-14 The PEIR subsequently includes an analysis of Uptown Land Use Distribution at Plan Build-Out.¹⁶ Table 3-9 purports to "describe[] the existing and proposed residential development anticipated from application of land uses shown on the proposed Uptown Land Use Map on vacant and underutilized parcels, according to analysis undertaken for the CPU."¹⁷ Table 3-9 indicates that, contrary to the statement cited above with regard to the ultimate reduction in future housing units, vacant and underutilized parcels would see a 41% increase in total housing units between Existing Development and Proposed Plan Build-out.¹⁸ There is no analysis demonstrating how the City arrived at that conclusion.
- B18-15 If Table 3-9 is attempting to compare the number of residential units that exist today to the number of residential units that are anticipated at build-out of the CPU, such analysis is not helpful since it presents an apples to oranges comparison. If, instead, Table 3-9 is attempting to compare anticipated residential build-out of the existing Community Plan to anticipated residential build-out of the CPU, that analysis contradicts the statement cited above that the "CPU results in an overall community wide reduction of future housing units at Community Plan build-out when compared to the adopted Community Plan."¹⁹ These apparently conflicting statements create confusion and

PEIR, p. 3-3.
 Ibid. PEIR, p. 3-4.
 Ibid; emphasis added. PEIR, pp. 3-30 - 3-31.
 PEIR, p. 3-30.
 PEIR, p. 3-31.

¹⁹ PEIR, p. 3-4.

- B18-13 See response to comment B18-4 regarding the reallocation of densities in support of a mixed-use, transit-oriented land use pattern.
- B18-14 This comment suggests a discrepancy in the PEIR; however, the statements in question are not in error. The PEIR evaluates impacts of the project against existing conditions and provided information about build-out of the proposed Uptown CPU for context. Existing development is not the same as build-out of the adopted Community Plan. Existing development refers to the land uses as they exist today; build-out of the adopted Community Plan refers to the land uses at full implementation of the Community Plan. For purposes of the PEIR, build-out of the adopted Community Plan was assumed to be at year 2035. See Section 2.3.1.1, Existing Land Use, and Section 2.3.1.1 and 2.3.1.2 of the PEIR for a discussion on the existing land uses and adopted Community Plan. Table 2-2, Existing Land Use and Population versus Adopted Community Plan, was added to this section to provide details on these buildout numbers. As shown, the adopted community plan would result in an increase in total housing units compared to existing development in the community.
- B18-15 Section 15125 of the CEQA Guidelines states that the environmental setting for which a lead agency determines whether an impact is significant is the physical environmental conditions as they exist at the time the notice of preparation is published. That said, the PEIR provides information on the existing conditions and build-out of the adopted CPU for comparison against the proposed Uptown CPU (Tables 2-2, 3-9 and 3-10). See response to comment B18-14 regarding the difference between the statements related to existing development and those related to build-out of the adopted Community Plan.

	Allen Matkins Leck Gamble Mallory & Natsis LLP Attorneys at Law		
	Kurtis Steinert August 8, 2016 Page 5		
B18-16 B18-17	highlight the basic problem of the CPU – it fails to provide density required to implement the General Plan and CAP. Further, Table 3-9 is directly contradicted by the GHG Emissions chapter of the PEIR. According to Table 6.5-2 of the PEIR, the CPU would result in 32,680 dwelling units (5,500 single- family units; 27,180 multi-family units). ²⁰ In contrast, the current Community Plan would result in 34,600 dwelling units (5,540 single-family units; 29,060 multi-family units). ²¹ Based on this table, CPU build-out would result in a total of 1,920 less units (40 less single-family; 1,880 less multi- family units) than build-out of the existing Community Plan. ²² In addition, the CPU itself notes that when compared to the existing Community Plan, the project will result in "a reduction in residential densities." ²³ In light of this acknowledgement, coupled with the information included in the GHG Emissions chapter, it appears that the single sentence buried in the text of the Project Description is accurate – the CPU would in fact reduce the sumphere of dural line units in Juterum.	B18-16 B18-17 B18-18	The statements identified in this comment are not PEIR inconsistencies, as the comment suggests. See response to comment B18-14 regarding the difference between the statements related to existing development and those related to build-out of the adopted Community Plan. Comment noted. Comment noted. See the previous response to comments in this latter for responses to the specific Project Description concerns.
B18-18	In summary, the PEIR's Project Description is flawed – either it is internally inconsistent or simply inaccurate. In any event, the Project Description fails to comply with the information disclosure requirements of CEQA as it does not provide a clear description of the proposed project, and it must be revised.		The project description is not internally flawed. The project does meet the objective of maintaining or increasing the housing supply through the designation of higher residential densities along major transit corridors compared to existing conditions
B18-19	 Land Use The law is well-settled that a project is consistent with an applicable land use plan if, "considering all its aspects, it will further the objectives and policies of the general plan and not obstruct their attainment."²⁴ CEQA requires that an EIR "shall discuss any inconsistencies between the proposed project and applicable general plans¹²⁵ Despite that clear direction, the PEIR fails to address the project's inconsistencies with the General Plan. Rather, the PEIR takes the position that the CPU "would build upon the goals and strategies in the General Plan aims to direct new development projects away from natural undeveloped lands into already urbanized areas and/or areas where conditions allow the PEIR, p. 6-5-7; see also, PEIR Appendix E-1, p. 7 (Table 2); PEIR Appendix E-2, p. 2 (Table 2). <i>1 Ibid.</i> <i>1 Did.</i> <i>Friends of Lagoon Valley v. City of Vacaville</i> (2007) 154 Cal.App.4th 807, 817. FEIR, p. 3-2. 	B18-19	The City does not agree that the PIER does not address consistency with the General Plan. PEIR section 6.1.3 of the PEIR, Issue 1, provides an analysis of conflicts with applicable plans including the General Plan. The analysis concludes that the proposed Uptown CPU would be consistent with the General Plan.

LETTER

RESPONSE

	Allen Matkins Leck Gamble Mallory & Natsis LLP Attorneys at Law		
	Kurtis Steinert August 8, 2016 Page 6		
B18-20	 integration of housing, employment, civic, and transit uses. It is a development strategy that mirrors regional planning and smart growth principles intended to preserve remaining open space and natural habitat and focus development in areas with available public infrastructure.²⁷ Since the CPU would <i>reduce</i> residential development in Uptown, then the CPU is .nconsistent with the General Plan's City of Villages strategy. Attached hereto as Exhibit A is an analysis prepared by the Atlantis Land Use Group that identifies the General Plan goals and policies that the CPU fails to implement. As can be seen, the CPU does <i>not</i> "place an emphasis on directing growth into mixed-use activity centers that are pedestrian-friendly and linked to an improved regional transit system."²⁸ Instead, the CPU's density reductions stand in the way of General Plan implementation. As a result, the proposed project is <i>not</i> consistent with the General Plan as stated 	B18-20 B18-21	See response to comment B18-3. Also see response to the referenced Exhibit A in responses B18A-1 through B18A-9. Project consistency with SANDAG's San Diego Forward: The
B18-21	For the same reasons, the project is not consistent with the goals of the San Diego Association of Governments' San Diego Forward: The Regional Plan, which seeks to "develop compact, walkable communities close to transit connections and consistent with smart growth principles." ³⁰ The reduced densities that will result with adoption of the CPU would <i>not</i> "establish a pedestrian-oriented, urban, and mixed-use community village that would reduce reliance on the automobile and promote walking and [the] use of alternative transportation" in Uptown. ³¹		Regional Plan is addressed in the PEIR section 6.1.3 of the PEIR, Issue 1. As discussed in that section, the CPU proposes to establish a pedestrian-oriented, urban, and mixed-use community village that would reduce reliance on the automobile and promote walking and use of alternative
B18-22 B18-23	Moreover, attached hereto as Exhibit B is an analysis prepared by Safdie Rabines Architects, which identifies the complications that will arise with implementation of the CPU. As explained therein, the development regulations proposed in Uptown would further inhibit development, pushing residential densities even lower than acknowledged in the PEIR. Ultimately, the PEIR does not satisfy the informational disclosure requirements of CEQA because it fails to recognize the land use impacts that will occur with implementation of the CPU. As a result, the PEIR must be revised.		transportation. Policies contained within the proposed Uptown CPU Land Use and Mobility Elements serve to promote bus transit use as well as other forms of adopted plan, the proposed Uptown CPU would assign the most intensive land uses in areas proximate to transit, consistent with smart growth principles. See also response to comment B18-3.
		B18-22	' Comment noted. Responses to Exhibit B are provided beginning with response to comment B18A-1.
	 PEIR, p. 5-1. PEIR, p. 6.1-8. PEIR, pp. 6.1-2 - 6.1-11. PEIR, p. 6.1-13. <i>Ibid.</i> 	B18-23	Comment noted. Refer to responses B18-3 and B18-14 regarding land use analyses.

	Allen Matkins Leck Gamble Mallory & Natsis LLP Attomeys at Law Kurtis Steinert August 8, 2016 Page 7	D10 74	Comment noted. This comment is informational in nature and
	Greenhouse Gas Emissions	D10-24	does not require a detailed response.
B18-24	The PEIR's GHG emissions analysis is likewise flawed. The PEIR states:		
B18-25	If emissions from build-out of the proposed Uptown CPU and associated discretionary actions are less than those that would be generated by build-out of the adopted Community Plan, impacts related to GHG emissions would be less than significant provided the proposed Uptown CPU and associated discretionary actions implement the land use strategies identified in the [CAP]. If emissions are greater than those of the adopted Community Plan, impacts related to GHG emission could still be less than significant if the increase in GHG emissions is a direct result of implementing CAP strategies and the General Plan's City of Villages Strategy. ³² Based on the reduced densities that will occur with implementation of the CPU, the project is not quantitatively or qualitatively consistent with the recently adopted CAP. Thus, the project's GHG impacts are potentially significant.	B18-25	See response to comment B2-3 and B18-3 for a detailed discussion of the methodology used to calculate emissions and for a discussion of project consistency with the CAP. Also refer to Appendix E-1 and E-2 of the PEIR for further detail about the methodology used to calculate emissions.
	Despite the fact that the GHG analysis recognizes that the CPU would reduce residential		
B18-26	densities in Uptown, the PEIR improperly determine that implementation of the CPU would result in a VMT reduction when compared to build-out of the existing Community Plan. ³³ Therefore, the PEIR concludes that vehicle emissions would decrease from 380,530 metric tons of carbon dioxide equivalent ("MTCO ₂ E") to 372,922 MTCO ₂ E, a 7,608 MTCO ₂ E reduction. The PEIR does not explain how this alleged reduction would be achieved in the face of the CPU's density reduction. In fact, the CPU's density reductions will force development out of a recognized Transit Priority Area ("TPA") and into natural, undeveloped lands – in direct contrast to the goals of the CAP. Moving development away from existing public infrastructure and away from the employment opportunities available in Uptown and nearby Downtown would result in a VMT increase. Pushing residents out of Uptown threatens to take the City back to the days before the City of Villages strategy and CAP were adopted, when development boomed in the suburbs, far from transit and mixed-use development opportunities. The PEIR's conclusion that the CPU would decrease vehicle emissions is not adequately explained, and if emissions would increase under the CPU, then the proposed project would severely undermine the City's ability to meet the CAP's ambitious GHG emission reduction goals. ³⁴ This concern must be thoroughly vetted so that the public and decisionmakers can fully understand the issue.	B18-26	The City does not agree with the statements made in this comment. The proposed CPU would not force development into natural, undeveloped lands nor will it push current residents out of the Uptown community. Rather, any new development or redevelopment consistent with the proposed Uptown CPU would occur on infill sites. As explained in Section 6.5.3, Impact Analysis, of the PEIR, vehicle miles traveled (VMT) can be reduced by decreasing the planned number of single-family and multi-family residences, as well as increasing residential density within proximity to transit and commercial services. The proposed Uptown CPU implements both of these strategies: decreases residential densities in some areas while increasing
	 ³² PEIR, p. 6.5-5. ³³ PEIR, Appendix E-1, p. 60; PEIR, Appendix E-2, p. 4. ³⁴ PEIR, p. 6.5-7. 		residential density in other areas, located near planned or exiting transit infrastructure and commercial uses combined with policies that promote a walkable and bicycle-friendly neighborhood design. See also response to comment B18-3 regarding proposed CPU consistency with the CAP.

RTC-106

Matkins Leck Gamble Mallory & Natsis LLP ys at Law	
Kurtis Steinert August 8, 2016	B18-28
Page 8	
Moreover, the CPU also would result in a qualitative inconsistency with the CAP. The CPU	

- B18-27 Mo density reduction frustrates compliance with Action 3.1 of the CAP, as the project is not consistent with the General Plan.³⁵ Moreover, the project would not comply with Action 3.2 of the CAP as the CPU would reduce residential densities, which in turn would suppress pedestrian improvements in a TPA and decrease commuter walking opportunities.³⁶ And, contrary to Action 3.6, the CPU's residential reduction would not implement transit-oriented development in the Uptown Community area.37
- The PEIR's failure to address these issues results in an inadequate GHG emissions analysis. B18-28 The City must revise the PEIR to accurately reflect the impact caused by the CPU to ensure that the public and decisionmakers have access to all of the necessary information.

Historical Resources

- B18-29 Next, the PEIR's historical resources analysis is flawed. It improperly assumes historicity for a significant portion of the Uptown Community Plan area, despite a lack of substantial evidence to support the assumption. As explained in detail in Exhibit C, there are numerous flaws in the Uptown Community Plan Area Draft Historic Resources Survey Report ("Survey Report") prepared for the PEIR.
- B18-30 First, the overly inclusive nature of the Survey Report will in practical effect "actually increase[] the potential for the designation of properties which were formerly determined by the survey to be ineligible for determination."³⁸ Second, the potential historic district and Bungalow and Apartment Court Multiple Property Listing analyses are inadequate.³⁹ As a result, the PEIR cannot and should not rely on the Survey Report.⁴⁰ The Council urges the City to not adopt the Survey Report and reconsider the Historical Resources chapter of the PEIR.

Population and Housing

- B18-31 Contrary to the statement made in the PEIR that "the proposed Uptown CPU and associated discretionary actions would not displace people or existing housing," that is what will happen with project implementation.⁴¹ The PEIR's unsupported conclusion that the CPU will not have any potential population and housing impact is erroneous.
 - ³⁵ PEIR, p. 6.5-10.
 - ³⁶ *Ibid.*
 - ³⁷ *Ibid.*

- ³⁸ Exhibit C, pp. 2-3.
- ³⁹ *Id.* at pp. 3-7.
- ⁴⁰ PEIR, pp. 6.7-8 6.7-18.
- ⁴¹ PEIR, p. 7-3.

- D10 27 Comment noted. See response to comment B18-26.
 - The City does not agree that the GHG emissions analysis presented in the PEIR is inadequate. See response to comment See response to comment B2-3, B18-3, and B18-26 for a detailed discussion of the methodology used to calculate emissions and for a discussion of project consistency with the CAP. Also refer to Appendix E-1 and E-2 of the PEIR for further detail about the methodology used to calculate emissions.
 - This comment makes reference to the historical resources B18-29 analysis of the PEIR and states that assumed historic resources in the Uptown community lack evidence to support their historic value. The Uptown Community Plan Area Historic Resources Survey was prepared consistent with standard preservation practice, and included research and preparation of a Historic Context Statement; fieldwork to identify individual properties and historic districts which may be significant and eligible for designation; a preliminary inspection and assignment of a California Historical Resource Status Code: and documentation in a database. A full discussion of the survey methods and results is available in the Historic Resources Survey document. Refer to responses B18C-1 and on for responses regarding the stated "flaws" in the Historic Resources Survey Report.
- B18-30 The Uptown Community Plan Area Historic Resources Survey was prepared consistent with standard preservation practice. All individual resources, potential historic districts and Multiple Property Listings (MPLs) were evaluated for potential significance against the City's designation criteria based upon the historic context statement and input from community cultural and historical interest groups. Reconnaissance surveys are intended to indicate where historic resources may be present, and are never intended to provide detailed evaluation

B18-30 (cont.)
or final resolution regarding the historic status of a property, as the comment suggests. They are an informational tool which serves as a base-line for future property-specific and sometimes intensive evaluation. In addition, the inclusive nature of the Historical Resources Survey Report supports policies of the General Plan and concerns of the community for historic preservation. Refer to response C52-7 for responses regarding the corrections to the Status Codes.
B18-31 While redevelopment, by its nature, causes temporary displacement of housing, the proposed Uptown CPU would not result in the permanent displacement of residences.

ATTACHMENT 7

RESPONSE

Allen Matkins Leck Gamble Mallory & Natsis LLP Attorneys at Law Kurtis Steinert August 8, 2016 Page 9 B18-32 See response to comment B18-3. As shown herein, the density reduction that will occur in Uptown will force residents out of B18-32 the urban core and into areas of the City lacking public transit and job centers. Therefore, the project may result in potential direct and indirect population and housing impacts and those impacts must be evaluated in the PEIR. By not providing this analysis, the PEIR prevents the public and decisionmakers from understanding the true impacts of the project. Alternatives B18-33 Comment noted. This comment is informational in nature and B18-33 CEQA requires that an EIR describe a "range of reasonable alternatives to the project."42 does not suggest an inadequacy in the PEIR. "An EIR need not consider every conceivable alternative to a project. Rather it must consider a reasonable range of potentially feasible alternatives that will foster informed decisionmaking and public participation.¹¹⁴³ An alternative need not reduce every impact, it just has to substantially reduce *any* significant impact.⁴⁴ That is because "it is practically impossible to imagine an alternative that would provide substantial environmental advantages in all respects.⁴⁵ Further, an agency is not precluded from including an alternative that would substantially reduce some impacts, but increase others, so long as the alternative's significant impacts are also analyzed in the EIR.⁴⁶ B18-34 As discussed in Chapter 6.0, the proposed Uptown CPU and The PEIR's alternative analysis fails to comply with these principles as it does not include an B18-34 associated discretionary actions would result in significant increased density alternative.47 Since the PEIR includes a flawed Project Description, the environmental analysis fails to adequately analyze the project's significant land use and GHG and/or cumulative environmental impacts related to emissions impacts, among others. An increased density alternative, which would place more homes in Uptown and advance the far-reaching goals of the General Plan and the CAP, would alleviate the transportation, noise, historical resources, and paleontological land use and GHG emissions impacts outlined in this letter. In addition, an increased density resources. The range of alternatives considered in Chapter 10.0, alternative - such as the one proposed in the Council's June 8, 2016 letter (Exhibit E) - would satisfy the project objective that seeks to maintain or increase the housing supply along transit Alternatives, includes appropriate alternatives that would corridors.48 reduce these significant impacts, consistent with Section 15126.6 of the CEQA Guidelines. Two of the alternatives provide for increased density, including the Adopted Community Plan with Removal of the Height Ordinance Alternative and the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative. As discussed in previous response to 42 14 Cal. Code Regs. § 15126.6(a). comments in this letter, the proposed Uptown CPU is consistent ⁴³ Ibid. ⁴⁴ 14 Cal. Code Regs. § 15126.6(c). with the General Plan and the CAP. See response to comment 45 Sierra Club v. City of Orange (2008) 163 Cal.App.4th 523, 546. B18-3. Therefore, an alternative that would reduce significant ⁴⁶ 14 Cal. Code Regs. § 15126.6(d). ⁴⁷ The Analysis of Adopted Community Plan with Removal of the Interim Height Ordinance impacts due to inconsistency with these planning documents is Alternative is not the same as an increased density alternative.

⁴⁸ PEIR, p. 3-3; 14 Cal. Code Regs. § 15126.6(a).

RTC-109

not required.

LETTER

	Allon Metkins Look Comble Mellow & Natois LLD		
	Attorneys at Law		
	Kurtis Steinert August 8, 2016		
	Page 10		
B18-35	Failure to include such an alternative, particularly in light of the Council's repeated requests for an increase in density in Uptown, results in an alternatives analysis that is "manifestly unreasonable." ⁴⁹ The City's error is further compounded by the fact that the PEIR does not include a discussion of alternatives that were considered but excluded from the analysis. ⁵⁰	B18-35	See response to comment B18-34.
	Conclusion		
B18-36	For the reasons outlined above, the Council does not support the proposed CPU. We encourage the City to produce a revised CPU, which implements the goals of the General Plan and the CAP.	B18-36	Comment noted.
	We look forward to receiving written responses to the items raised in this letter and the exhibits. Please do not hesitate to contact me with any questions. Very truly yours, Jeffrey A phine		
	 cc: Mayor Kevin Faulconer Councilmember Todd Gloria Jeff Murphy, Director, Planning Department Tom Tomlinson, Assistant Director, Planning Department Nancy Bragado, Deputy Director, Long Range Planning Alyssa Muto, Deputy Director, Environment and Policy Analysis Tait Galloway, Program Manager, Long Range Planning Brian Schoenfisch, Program Manager, Long Range Planning Marlon Pangilinan, Senior Planner Bob Vacchi, Director, Development Services Department Elyse Lowe, Deputy Director, Project Submittal and Management Division Jerry Sanders, President and CEO, San Diego Regional Chamber of Commerce Matt Adams, Vice-President, Building Industry Association of San Diego Bennet Greenwald Uptown Gateway Council 		
		1	

LETTER



LETTER



	Atlantis Group	MARCELA ESCOBAR-ECK, Principal 2488 Historic Decatur Road Suite 220 • San Dieg	o, CA 92106 • (619) 523-1930			
318A-3	SUMMARY Marcela Escobar- of Development S she founded 3 ye the City of Cartsb- the Naval Trainin several unique d community, to bui	Eck has almost 30 years of service with local governme Services for the City of San Diego — and currently has a th rars ago. During her tenure with the City of San Diego a ad, she led several multi-year community engagement pi g Center and several other complex policy initiatives. lialogue workshops between the City and various inter ild better relationships and resolve conflicts.	nt agencies – most recently as Director nriving small business land use firm that and in her role as Planning Director for rograms including the redevelopment of She also designed and implemented rest groups, such as the development	B18A-3	Comment noted.	
	Marcela is certifie through the Natio training sessions	ed by the National Charrette Institute as a Charrette Ma onal Conflict Resolution Center where she also served for incoming mediators. She is a native Spanish speake	inager. She is also a trained mediator on various committees and conducted er.			
	PRINCIPAL Atlantis Group		10/2007 TO PRESENT San Diego, CA			
	Atlantis Group is t with our private an use planning, con not-for-profit com and team develop partnerships with and land use mat	the premier land use and strategic planning consulting firm nd public agency clients ensuring their success. Our inn munications, strategic planning and government relation panies and public agencies coordinate with outside cons oment. We also provide effective marketing communicati key audiences and the surrounding community, and anal ters, including interpretation of regulatory documents.	n. We specialize in forming partnerships ovative and award winning work in land ns assists large and small for-profit and sultants and other agencies on projects ons and community relations, to create ysis and expertise on zoning, permitting			
	Atlantis Group se or small business regulatory docum Land Use Plan Permit Process Due Diligence, Expert Witness	rves both public and private clients with everything from , to analysis and expertise on zoning, permitting and lan- ents. ming and Government Relations sing/Public Policy Assistance , Zoning, Permitting, Case History s	advocating on behalf of a large corporation d use matters, including interpretation of			
	DIRECTOR City of San Diego Deve	elopment Services Department	10/2006 то 10/2007 San Diego, CA			
	 Management of 6 Oversee the staff permit and devel plans for complia California Environ issues. 	55 professional and administrative staff with a \$68.9 mill that review all development submissions for building cor opment information in all fields including fire, engineeri ance with development regulations as well as redevelop mmental Quality Act review; issue permits; conduct insp mental Quality Act review; issue permits; conduct in	ion dollar budget (FY 2007) de and life safety requirements; provide ng, and planning; review development pment and coastal standards; conduct pections and resolve code compliance			
	PLANNING DIRECTOR City of Carlsbad Planni	ing Department	9/2005 то 10/2006 Carlsbad, CA			
	 Manage the daily Coastal Program. 	y operations and the implementation of the General P , Habitat Management Plan and the California Environme	Plan, Growth Management Plan, Local ental Quality Act.			
	CHIEF DEPUTY DIRECTO City of San Diego Deve	R, CUSTOMER SUPPORT AND INFORMATION DIVISION elopment Services Department	1/2005 то 9/2005 San Diego, CA			
	 Management of 2 Manage daily ope development per systems, custome 	233 professional and administrative staff with a \$24.2 mill erations of the division including evaluation of staff perform rmits, submittal, records and information systems, te er accounts and the overall department personnel and but 1	ion dollar budget mance in the following areas: managing chnology and geographic information udget functions.			

DEP City	JTY DIRECTOR, PROJECT MANAGEMENT DIVISION of San Diego Development Services Department	3/2001 то 12/2004 San Diego, CA
•	Management of 85 professional and administrative staff who are dire development permits within the City of San Diego Administer conflict resolution sessions with customers, staff and comm Prepare and present written and oral reports to citizen groups, commis Management representative at all Planning Commission and City development projects, redevelopment interface and planning policy iss City's lead negotiator on Development Agreements	ectly responsible for the management of all nunity groups ssions, boards and City Council y Council hearings relating to land use, sues
Nav. City	AL TRAINING CENTER (NTC), REDEVELOPMENT PROJECT DIRECTOR of San Diego Planning Department	10/1999 то 3/2001 San Diego, CA
•	City's lead negotiator on the most complex Disposition and Dex Redevelopment Agency. The development of the property resulted in a will provide housing, employment and educational centers as well as million in Tax Increment will be generated to the Redevelopment Ager City's liaison with the Coastal Commission for the entitlement of the pri Navy for the conveyance of NTC. Negotiations resulted in a no-cost of	velopment Agreement negotiated for the a \$500 million redevelopment project which s an Arts and Culture Center. Over \$200 ncy over the life of the project operty, and with the U.S. Department of the conveyance of NTC to the City.
Pro City	SRAM MANAGER, PROJECT MANAGEMENT DIVISION of San Diego Planning and Development Review Department	1/1999-10/1999 San Diego, CA
•	Management of 35 professional and 25 administrative staff dired development permits within the City of San Diego. Developed and mo	ctly responsible for the management of onitored budget
Priv City	ate Development Manager/Fiscal Personnel Manager of San Diego Community and Economic Development Department	11/1996-1/1999 San Diego, CA
•	City's Project Manager for the 5,000-acre Black Mountain Ranch Ma managed all aspects of the entitlement, engineering and construction on all aspects of the Development Agreement and subsequent amend Trained staff and community planning groups on land use and process Managed and supervised staff related to all private and public develo operations and finance Managed special projects/assignments for Department Director and C	Inster Planned Community. Processed and n document permits. City's lead negotiator lments. s related issues spment work. Directed complex activities in ity Manager's Office
Cou City	ncil Liaison of San Diego Manager's Office	6/1994-3/1995 San Diego, CA
•	Reviewed and approved all managerial items prior to requesting consi Prepared and conducted briefings for Manager, and staff briefings for Sat with the City Manager at Council hearings and responded to Coun Responded to and coordinated appropriate responses to Council direct	ideration on the Council docket the Mayor and City Council ncil inquiries ctives for the City Manager
Sen City	or PLANNER/ASSOCIATE PLANNER/ASSISTANT PLANNER of San Diego Development Services/Planning Department	8/1987-11/1996 San Diego, CA
•	Conducted community/business outreach training regarding City proce Authored, co-authored and edited several elements of citywide legislat Conducted conflict resolution forums on adopted legislation, section of Technical advisor to city staff, citizens and developers on city requiren	essing requirements tion and guidelines perations and work programs nents and codes
	2	

<section-header><section-header><section-header><section-header><text><text><text><text><text><text><text><text></text></text></text></text></text></text></text></text></section-header></section-header></section-header></section-header>			
<text><text><text><text><text><text><text><text><text><text><text><list-item><list-item><section-header><section-header></section-header></section-header></list-item></list-item></text></text></text></text></text></text></text></text></text></text></text>	EDUCATION		
<text><text><text><text><text><text><text><list-item><list-item><list-item><list-item><section-header><section-header><text></text></section-header></section-header></list-item></list-item></list-item></list-item></text></text></text></text></text></text></text>	BACHELOR OF SCIENCE IN LANDSCAPE AR WITH AN EMPHASIS IN URBAN AND REGIONA University of California, Davis Member Signa Lambda Alpha (Nationa	HITECTURE ⊾PLANNING Landscape Architecture Honor Society)	June 1987
<text><text><text><text><section-header><list-item><list-item><list-item><list-item><list-item><list-item><list-item><text></text></list-item></list-item></list-item></list-item></list-item></list-item></list-item></section-header></text></text></text></text>	SPECIAL SKILLS MEMBERSHIPS		
<text><text><text><section-header><list-item><list-item><list-item><list-item><list-item><list-item><list-item><section-header><section-header><section-header><text></text></section-header></section-header></section-header></list-item></list-item></list-item></list-item></list-item></list-item></list-item></section-header></text></text></text>	Language: Native speaker, English and	Spanish	
<text><section-header><section-header><section-header><section-header><section-header><section-header><section-header><section-header><section-header></section-header></section-header></section-header></section-header></section-header></section-header></section-header></section-header></section-header></text>	Certifications: NCI Complete Charrette	Manager ™	
<section-header><section-header><section-header><section-header><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></section-header></section-header></section-header></section-header>	Mediation: Community Mediation Cente	r of San Diego - Volunteer Mediator	1990-1992
<list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item>	Memberships/Affiliations		
3	Memberships/Affiliations American Planning Association (AP, Building Industry Association (BIA) City of San Diego Code Monitoring Lambda Alpha International (LAI) - San Diego County Taxpayers Assoc South County Economic Developme Urban Land Institute (ULI)) eam: Small Business Member [he Honorary Society for the Advancement of Land Ec Policy Committee – Member (formerly Co-Chair of Hou iation (SDCTA): Public and Community Assets Comm nt Council (representative to the Wildlife Advisory Grou	conomics using Committee) hittee Member bup)
3			
3			
3		2	
		3	

LETTER

	Atlantis Group 2488 Historic Decatur Road, Suite 220 San Diego, CA 92106 Phone: (619) 523-1930 www.AtlantisSD.com		
B18A-4	This exhibit includes a synthesis of City of San Diego General Plan Goals, Policies, and Discussions, which the Draft CPU fails to implement. A PowerPoint presentation presented by the City of San Diego Planning Department to Uptown Planners on February 2, 2016, the Adequate Sites Inventory Grid #66, which includes the Hillcrest Core, and the Adequate Housing Sites Inventory Summary by Community Plan Area 2010-2020 table from the Housing Element of the General Plan are also included.	B18A-4	Comment noted. This comment does not suggest a specific inadequacy of the PEIR; therefore, a detailed response is not required.
B18A-5	The Goals and Policies from the Housing and Land Use Elements of the General Plan on the following two pages repeatedly emphasize the importance of either maintaining or increasing density in infill, transit-oriented locations. However, the Draft CPU states that the update process has created "changes that have resulted in a reduction in residential densities to address preservation of neighborhood character within older, historic neighborhoods, a lack of public facilities, and changes in the development vision for certain neighborhoods within the community. During this process areas that could support increased residential density were also identified" (LU-31).	B18A-5	Comment noted. This comment does not suggest a specific inadequacy of the PEIR; therefore, a detailed response is not required.
B18A-6	Yet, the density reductions far outnumber any increases in density, as the Draft CPU overall reduces development capacity by 1,900 units across the Community Planning Area. This includes many areas in close proximity to transit, as is shown in slide 21 of the PowerPoint presentation. Meanwhile, parking requirements favor automobile ownership, even in Transit Areas. As demonstrated in Table 142-05C of the Land Development Code, one bedroom units require 1.25 spaces and two bedroom units require 1.75 spaces in Transit Areas, reducing the incentive to commute via transit or active transportation as is needed to fulfil the mode share targets of the CAP.	B18A-6	See response to comment B18-3 regarding the proposed Uptown CPU's reduction in density in some areas and increase in others. Regarding parking allowances for one- and two- bedroom units, it is true that reduced parking incentivizes transit and active transportation use over vehicular transit.
B18A-7	In addition, the Adequate Sites Inventory map of Hillcrest demonstrates that nearly the entire Hillcrest Core is identified as Potential Future Infill sites in the Housing Element's identification of "housing potential on land suitable for residential development adequate to accommodate the City's housing allocation of 88,096 total units" (HE-148), while the table reveals that 8,812 Potential Future Infill out of the City's identified 34,288 Potential Future Infill units are located in Uptown.		Proposed Uptown CPU policies provide allow for shared parking agreements (MO-7.6) and parking in-lieu fees (MO-7.9) to allow for flexibility in terms of parking management. A reduction in parking requirements for development would require an amendment to the Land Development Code that is not proposed as part of the project.
	1.	B18A-7	Comment noted. This does not suggest a specific inadequacy in the PEIR; therefore, a detailed response is not required.

B18A-8

RESPONSE

Atlantis Group 2488 Historic Decatur Koad, Suite 220 San Diego, CA 92106 Phone: (619) 523-1930 www.AtlantisSD.com

General Plan Element	Goal	Policies	Analysis
Housing	Goal 1 – Ensure the provision of sufficient housing for all income groups to accommodate San Diego's anticipated share of regional growth over the next housing element cycle, 2013 – 2020, in a manner consistent with the development pattern of	HE-A.2 Community Plans should be updated on a regular basis and provide for certainty in the development process. Through these updates the City shall continue to identify areas appropriate for increased infill residential and mixed-use development. Programmatic EIR's should be adopted with Community Plan Updates to allow EIR tiering for individual projects consistent with the updated Plan.	In the Draft CPU, the Hillcrest Core, Mission Hills Community Village, and the Park Boulevard Mixed Use Corridor remain at the same development capacity or are downzoned, with height restrictions added through CPIO2 that did not exist within the current Community Plan, resulting in a net loss of 1,900 housing units. As a built out community, the vast majority of the units lost come from downzoning potential multif-family housing, which is the more affordable, workforce housing addressed in policy HE-A-7.
	the Sustainable Communities Strategy (SCS), that will help meet regional GHC targets by improving transportation and land use coordination and jobs/housing balance, creating more transit- oriented, compact and walkable communities, providing more housing capacity for all income levels, and protecting resource areas. HE-A.7 Work to develop a comprehensive strategy for all income levels, and protecting resource areas.	HE-A.3 Through the community plan update process, designate land for a variety of residential densities sufficient to meet its housing needs for a variety of household sizes, with higher densities being focused in the vicinity of major employment centers and transit service.	The loss of development capacity in mixed use Transit Priority Areas is inconsistent with policies HE-A.2, HE-A.3, and HE-A.7, which specifically call for "increased infill," "higher densities being focused in the vicinity of major employment and transit service," and "workforce housing proximate to employment
		and/or multimodal transportation facilities." For these reasons, the Draft CPU fails to apply the Housing Element Goals and Policies and is therefore inconsistent with the General Plan.	

B18A-8

This comment claims that the proposed Uptown CPU is inconsistent with Goal 1 and associated policies of the Housing Element in the City's General Plan regarding provision of sufficient housing for all income groups. However, the proposed Uptown CPU includes Policy LU-1.1, which calls for the provision of land use types to accommodate both affordable and market rate housing and commercial opportunities, and Policy LU-2.3, which requires the development of adequate housing for those with special needs, including low-income residents. Though the proposed CPU would result in reductions in densities in some areas and a net reduction in total housing units compared to build-out of the adopted Community Plan, the reduction in density would not disproportionately affect affordable housing.

2

RTC-117

B184-9	
DTOA-J	
	2488 Histr

Seneral Plan Sement	Goal	Policies	Analysis
Land Use	Mixed-use villages located throughout the City and connected by high-quality transit.	LU-A.7b Achieve transit-supportive density and design, where such density can be adequately served by public facilities and services (see also Mobility Element, Policy ME-B-9). Due to the distinctive nature of each of the community planning areas, population density and building intensity will differ by each community.	The Hillorest Core's proximity to Downtown San Diego, access to six bus lines and current and upcoming bicycle facilities on at "and 5". Avenues make it a prime location for increased densities. Uptown is uniquely situated to maximize use of its transit and bicycle infrastructure, as well as achieve the Climate Action Plan's targets for transit and active transportation mode share. The Draft CPU does not "achieve transit-supportive density" because of its height limitations and lack of increased density within the Hilfrest Core, which is a Transit Profit Area. The stagnant or decreased densities and CPIOZ height limitations over Uptown's villages and transite homes. It was not up to the son to "decreased benows, it does not "maintain of Uptown to provide affordable homes, it does not "maintain to Uptown to provide affordable homes. It does not "maintain to up town to provide affordable homes. It does not "maintain to up town to provide affordable homes. It does not "maintain to up town to provide affordable homes. It does not "maintain to provide affordable homes. It does not "maintain to the shelling to the shelling homes. It does not "maintain to up town to provide affordable homes. It does not "maintain to prove height provide affordable homes. It does not "maintain to prove height provide affordable homes. It does not "maintain to prove the provide affordable homes. It does not "maintain to prove the provide affordable homes. It does not "maintain to prove the provide affordable homes. It does not "maintain to prove the provide affordable homes. It does not "maintain to prove the provide affordable homes. It does not "maintain to prove the provide affordable homes. It does not "maintain to prove the prov
	Community plans that maintain or increase planned density of residential land uses in appropriate locations.	LU-C.3. Maintain or increase the City's supply of land designated for various residential densities as community plans are prepared, updated, or amended.	
	Ensure diverse and balanced neighborhoods and communities with housing available for households of all income levels.	LU-H.3 Provide a variety of housing types and sizes with varying levels of affordability in residential and village developments.	
		LU-H.7. Provide a variety of different types of land uses within a community in order to offer opportunities for a diverse mix of uses and to help create a balance of land uses within a community (see also LU-A.7),	or increase the City's supply of land designated for various residential densities," provide "varying levels of affordability in residential and village developments," or "offer opportunities for a diverse mix of uses," as policies LU-A.7b, LU-C.3, LU-H.3, and LU-H.7 require.
			For these reasons, the Draft CPU fails to meet the Land Use Element Goals and Policies and is therefore inconsistent with the General Plan.

B18A-9

Atlantis Group

San Diego, CA 92106 Phone: (619) 523-1930

www.AtlantisSD.com

This comment claims that the proposed Uptown CPU is inconsistent with goals and associated policies of the Land Use Element in the City's General Plan regarding mixed-use villages, increased density, and diverse and balanced neighborhoods. See response to comment B18-3 regarding the reduction in densities in some areas and increase in others and proposed CPU policies that support development of affordable housing.

RESPONSE

Regarding the Community Plan Implementation Overlay Zone (CPIOZ), the amendment is intended to regulate specific building heights primarily along the transit corridors to ensure development proposals are reviewed for consistency with the use and development criteria that have been adopted through community plan updates. Under the high-rise building policies, areas within the Uptown CPU area could be permitted to develop with buildings up to 100 feet in height. Areas also covered by a CPIOZ Type A would be subject to ministerial review for development that does not exceed 50 feet within Mission Hills and 65 feet in Hillcrest and Bankers Hill/Park West. CPIOZ Type B areas would be subject to a discretionary review process that would implement the proposed Uptown CPU policies and recommendations, particularly those related to building height consistent with the Urban Design Element. Ensuring consistency with the proposed Urban Design Element policies through this process is necessary to preserve the existing neighborhood character and avoid potential impacts. Therefore, the CPIOZ would not inhibit the ability of the proposed Uptown CPU to increase densities along transit corridors or provide affordable housing.





RTC-120









RTC-123

EAST sign Hills Proposed Land Use **PROPOSED LAND USE** 15, June 2015 Draft Land Use া ধ্যাপথ ×SA **Current Land Use** otown Community an Update SD SD

ATTACHMENT 7





RESPONSE







RESPONSE











RESPONSE

RTC-134



ATTACHMENT 7





RTC-137

ATTACHMENT 7
Community Commercial 0-109 du/ac **A Proposed Land Use PROPOSED LAND USE** une 2015 Draft Land Use Community Commercial 0-73 du/ac **Current Land Use** ptown Community lan Update Mixed-Use 73-110 du/ac SD

LETTER

ATTACHMENT 7

RESPONSE

LETTER

Estimated Community Plan Build-Out

	Existing (2014)	Current Community Plan	Community Plan without Incentive Zoning	June 2015 Draft Community Plan	January 2016 Draft Community Plan
Housing Units	23,160	34,600	32,190	32,540	32,700
Non-Residential Floor Area (Sq. Ft.)	7,229,000	7,303,000	7,483,400	7,476,000	7,476,000
Household Population	36,750	58,870	54,890	55,430	55,700

The January 2016 Draft has 1,900 fewer housing units than the 1988 Community Plan.

SD

LETTER

RESPONSE

San Diego Municipal Code	Chapter 14: General Regulations
(5-2016)	

Table 142-05C Minimum Required Parking Spaces for Multiple Dwelling Units and Related Accessory Uses

Multiple Dwelling Unit Type and Related and Accessory Uses	Automobile Spaces Required Per Dwelling Unit (Unless Otherwise Indicated)			Motorcycle Spaces Required Per Dwelling Unit	Bicycle ⁽³⁾ Spaces Required Per
	Basic (1)	Transit Area ⁽²⁾	Parking Impact ⁽¹⁾		Dwening Onli
Studio up to 400 square feet	1.25	1.0	1.5	0.05	0.3
1 bedroom or studio over 400 square feet	1.5	1.25	1.75	0.1	0.4
2 bedroomes	2.0	1.75	2.25	0.1	0.5
3-4 bedrooms	2.25	2,0	2.5	0.1	0.6
5+ bedrooms	2.25	2.0	(See footnote 6)	0.2	1.0
Affordable Housing dwelling units regulated by Section 142.0527	N/A	N∕A	0.25 beyond that required in Section 142.0527	(See footnote 3)	(See footnote 3)
Condominium conversion ⁽³⁾					
1 bedroom or studio over 400 Square feet	1.0	0.75	1.25	N/A	N/A
2 bedrooms	1.25	1.0	1.5	N/A	N/A
3 + bedrooms	1.5	1.25	1.75	N/A	N/A
Rooming house	1.0 per tenant	0.75 per tenant	1.0 per tenant	0.05 per tenant	0.30 per tenan
Boarder & Lodger Accommodations	1.0 per two boarders or lodgers	1.0 per two boarders or ledgers	1.0 per two boarders or lodgers, except 1.0 per boarder or lodger in beach impact area	N/A	N/A
Residential care facility	1 per 3 beds	1 per 4 beds or per	I per 3 beds or	N/A	N/A

Ch. Art. Div. 14 2 5 9

RTC-140

LETTER



RESPONSE

LETTER

Table 47: Adequate Housing Sites Inventory Summary by Community Plan Area 2010-2020 Military Housing Planned Communit Planning Area Process With Plan Adequate Housi Sites Inventor Units Completed Net Units Affordable Project Units Torrey Hills Infill/Vacant >=30 DU/AC n D. . Net Units Affordable Project Units Torrey Pines Infill/Vacant >=30 Ó DU/AC 1,440 2.341 Net Units Affordable Project Units .0 University Infill/Vacant >=30 DU/AC ù. D 8,812 10,012 Net Units Affordable Projec Units D Uptown Infill/Vacant >=30 DU/AC 8,812 8,977 ADOPTION DRAFT City of San Dicgo Housing Element HE-156

LETTER



LETTER

	925 Fort Stockton Drive, San Diopo Collionnia, 92103-1817 p. 61-2922-046a - t. 619,299,6072, www.saidicrabinas.com	B18B-1	Comment noted. The proposed Uptown CPU does not propose or bring forward any specific development project. Rather, it provides a policy framework to guide future development. No project at Sixth Avenue and Robinson Avenue is proposed at this time. The proposed Uptown CPU proposes to designate the land at Sixth Avenue and Robinson Avenue as Community Commercial, with the CC-3-9 zone.
	Mr. Bennet Greenwald The Greenwald Company Suite A 2929 Canon Street San Diego, California 92106 Reference: 6 th and Robinson Avenue Project	B18B-2 B18B-3	Comment noted. See response to comment B18B-1. Additionally, this comment does not suggest an inadequacy of the PEIR; therefore, a detailed response is not required. Comment noted. This comment does not suggest an inadequacy of the PEIR; therefore, a detailed response is not required.
B18B-1	Dear Mr. Greenwald, Based on the proposed Community Plan Update, we have studied your site at 6 th and Robinson. The aim is to understand how the proposed update impacts your ability to develop a viable project, one that is consistent with the goals and objectives of both Climate Action Plan as well as the Community Plan Update which encourage the use of residential density to promote vibrant, sustainable, pedestrian oriented communities with access to parks, open space and public transit	B18B-4	Comment noted. This comment does not suggest an inadequacy of the PEIR; therefore, a detailed response is not required.
B18B-2	We looked at your site at 6 th and Robinson as a case study from two perspectives. Option 1 looks at how the project could use the maximum FAR with commercial development at grade level to understand what kind of residential density could be achieved, whereas Option 2 follows the CPU's maximum Residential Density to understand its overall impact on the achievable FAR.	B18B-4	Comment noted. This comment does not suggest an inadequacy of the PEIR; therefore, a detailed response is not required.
B18B-3 B18B-4 B18B-5	Option 1 Based on the allowable FAR of 6.0, we arrive at a total gross floor area of 414,000 square feet. Assuming 34,500 SF of ground floor commercial area and a modest 2-bedroom residential unit size of 1,100 square feet, it is possible to achieve 345 units or 218 dwelling units / acre (exceeding the 109 dwelling unit density limit in the ordinance). Option 2 Using the maximum allowable residential density of 109 dwelling units / acre, we arrive at a total of 173 units. Maintaining the same assumptions as Option 1, the total gross floor area is 224,424 square feet, which represents a FAR of only 3.25 (far below the allowable 6.0). Our interpretation of the Draft CC-3-9 zoning as proposed in the Community Plan Update (CPU) is not promoting the right mix of residential density for several reasons. From a broad over-arching perspective the zone is in conflict with the stated goals of the Climate Action Plan. Specifically the goal of implementing increased residential density within Transit Priority areas is being constrained by incongruities between the maximum allowable residential density and the Floor Area Ratio (FAR) regulations. The housing density limitations make it infeasible for a project from ever being able to maximize the allowable FAR. It appears the draft zone is encouraging development that favors more commercial development versus residential. The only way to achieve	B18B-5	Comment noted. This comment suggests that the proposed Uptown CPU is not consistent with the CAP and does not allow for higher density development within transit priority areas because the proposed CC-3-9 zone encourages development that favors commercial over residential. Zone CC-3-9 is designated as Office – Commercial, with residential uses permitted to promote mixed-use development. The zone is intended for local convenience shopping, civic sues, and services serving an approximate 3-mile radius, permitting office uses and housing up to a very high residential density. Therefore, it is reasonable to assume that the CC-3-9 zone encourages commercial development as that is its intent. See response to comment B18-3 regarding the proposed Uptown CPU's consistency with the CAP and justification for the allocation of housing densities.

RESPONSE

STREET STREET GES STREET GES

925 Fort Stockton Drive, San Diogo, California - 92103-1817 p. 619 197 0153 - f. 619 299 5072 - envec saidlombinas co-r

the maximum FAR will be to create large residential dwelling units and / or supplement the delta by adding more commercial space which does not support affordability or sustainability (and not in alignment with the Climate Action Plan). In addition, maximum height limits imposed by the CPU of 120' will force a lower more compact / dense development on the block, putting pressure to maximize building footprint at street levels, which works against other ideals of the CPU which encourage enhancements such as open space and pedestrian oriented activities at the street / grade levels.

We hope this is helpful as you work toward developing your strategies for development at 6th and Robinson. Please refer to the attached summary table. Should you have any questions or concerns, please contact us.

Kind Regards,

Chim

Ricardo Rabines Safdie Rabines Architects

LETTER

ATTACHMENT 7



LETTER



LETTER

RESPONSE

Scott A. Moomjian Attorney at Law 5173 Waring Road, #145 San Diego, California 92120 Telephone (619) 230-1770 Facsimile (619) 785-3340 smoomjian@earthlink.net

August 6, 2016

Mr. Kurtis Steinert San Diego Senior Environmental Planner City of San Diego Planning Department 1010 Second Avenue, MS 413 San Diego, CA 92101

Sent Via Electronic (E-mail) Transmittal <u>PlanningCEOA(a)sandiego.gov</u>

Re: Uptown Community Plan Update Project Number: 21002568; SCH No. Pending

Dear Mr. Steinert:

I am writing this letter in order to express my strong concerns regarding the Uptown Community Plan Update ("Plan Update") Program Environmental Impact Report ("PEIR") as they relate to the proposed treatment, processing, consideration, and disposition of potential historical resources within the Uptown community.

My background in the field of historic resources is extensive. Over the past twenty six years, I have worked on hundreds of projects involving historic properties. In the past sixteen years, I have represented owners of historic properties achieve their objectives with local, state and federal government agencies that supervise or regulate such properties. Where appropriate, I have nominated them to local and national historic registers. I have also prepared or consulted on historical reports for historic properties throughout the County. A significant portion of my work has involved facilitating the rehabilitation of buildings or the redevelopment of sites containing historic resources. I am a qualified historical consultant by the City of San Diego, and my professional qualifications meet the Secretary of the Interior's *Standards and Guidelines for Archaeology and Historic Preservation (1995)* in the disciplines of Architectural History, Historical Preservation, and History.

I have reviewed the relevant environmental documents associated with historical resources prepared in conjunction with the project, including but not limited to, the Plan Update

Please refer to response to comment letter C53 which provides responses to the August 6, 2016 letter from Scott Moomjian.

and PEIR dated June 10, 2016 ("Historical Resources, Section 6.7.2.2); the Uptown Community Plan Area Draft Historic Resources Survey Report ("Survey Report") dated November 2015 and revised May 2016 (Appendix G-2) with Appendices A-G; the undated City of San Diego, Planning Department "Potential Historic Districts Fact Sheet" for the Uptown, North Park, and Golden Hill communities; and proposed San Diego Municipal Code section revisions. I am also well familiar with previous historic surveys conducted in the Uptown area, including the *Historic Resources Inventory for "Uptown Area," San Diego California* (1981), and the draft *Uptown Historic Architectural & Cultural Landscape Reconnaissance Survey* (2007) ("Draft Uptown Survey"). It should be noted that to date, that no historic surveys or historic resource inventories for the Uptown community have been formally reviewed or adopted by the City of San Diego.

The scope of my comments herein presented will be limited to problems associated with (1) the proposed Draft Historic Resources Survey Report; (2) the proposed regulatory framework for potential historic districts (PHDs); and (3) Multiple Property Listings (MPLs), specifically the Bungalow and Apartment Court MPL. Collectively, each of my arguments substantiate the inherent deficiencies and flaws in the Survey Report and proposed City action. As a result, they should be rejected in their entirety.

(1) Proposed Draft Historic Resources Survey Report

As an initial matter, there are thousands of properties located within the boundaries of the Uptown Community Plan Area and the geographic area is massive. According to the Survey Report, the Planning Area encompasses nearly 2,700 acres and contains the communities of Park West, Middletown, Mission Hills, Hillcrest, the Medical Complex area, as well as the western half of University Heights. While the earlier Draft Uptown Survey (2007) surveyed 11,104 properties and identified 2,192 properties as potentially significant (59 of which were located in potential historic districts), the new Survey Report identified 11,109 properties, and found that 2,134 are potentially eligible for designation as individually significant properties, including properties identified as part of potential MPLs. An additional 1,454 properties were found to be potential contributing resources to 23 potential historic districts. Finally, 6,808 properties were identified and documented in the survey, but were not determined potentially historic upon initial visual inspection. While not directly cited in the Survey Report, there are therefore, a total of approximately 3,588 properties which exist in the Uptown community, either as potentially significant individual resources, or as potentially significant contributors to a historic district. The Survey Report, however, fails to account for the true number of buildings which may be potentially significant in the Uptown community because it identifies only the number of properties (i.e. by parcels and address), and not the actual number of structures on a property (see discussion of bungalow/residential courts within the MPL below).

According to the Survey Report, the Uptown Historical Context and Oral History Report prepared for the Draft Uptown Survey (2007) was "discarded in its entirety" and replaced by a new historic context statement prepared by City Planning Staff. Further, due to the fact that the assignment of Status Codes (which provide "a summary assessment of the resource") undertaken as part of the Draft Uptown Survey were "flawed," new Status Codes within the Survey Report were assigned.¹ More specifically, those Uptown properties which were determined ineligible for local designation (assigned a "6Z" code) by the Draft Uptown Survey, were reclassified as "Identified in Reconnaissance Level Survey: Not evaluated" (reassigned as a "7R" code).

In effect, by changing a prior determination of ineligibility to a new determination of potential eligibility through a non-evaluation code, the Survey Report has essentially eliminated a former presumption of insignificance in favor of future potential significance.² By effectuating a change in Status Codes, the Survey Report has subtly undermined some of the prior evaluations undertaken as part of the Draft Uptown Survey, and cast a new "net" over these properties as potentially significant. Properties captured in this manner are now presumed to be potentially significant, rather than presumed to be ineligible for local designation. The change in Status Codes actually increases the potential for the designation. Such action interjects less assurance and more uncertainty and cost for property owners in the historic review and historic designation process.³ Based upon the foregoing deficiencies associated with the Survey Report, it should not be adopted by the City for use in the Uptown Community Plan Update.

(2) Proposed Regulatory Framework For Potential Historic Districts (PHDs)

The Survey Report states that the "The 2007 Draft Uptown Survey identified nineteen (19) potential historic districts that meet one or more of the City's local designation criteria for historical sites. In addition, City staff and members of the Uptown Community have identified four (4) additional historic districts – Allen Terrace, Avalon Heights, Hillcrest and the San Diego Normal School/San Diego City Schools Education Complex historic districts – that also appear to meet one or more of the City's local designation criteria." In total, therefore, the Survey Report proposes a total of 23 potential historic districts for the Uptown community, with a total of 1,454 properties that were found to be potential contributing resources to the 23 potential historic districts (PHDs) and their contributors generally appear to be inadequate in terms of historic methodology, historic source material, and scholarly/academic historic interpretation and analysis.

3

¹ The Survey Report strongly implies that the Draft Uptown Survey, which was commissioned and paid for by the City, with oversight and input provided by the Planning Department, was inherently deficient. According to the Survey Report, "based upon the limited level of the survey work and the quality of the original Historic Context Statement on which survey was based, assignment of such a Status Code [6Z] was not appropriate."

² This is supported by the statement in the Survey Report that "[w]hile these properties have not been identified as potentially significant as part of this Survey Report, they *have not been cleared as not historic*, and would be evaluated in the future for historic significance at the parcel level consistent with the requirements of the City's Municipal Code. *Italics added*.

³ "Based on the results of the Initial Determination, if there is evidence that the site contains a historical resource, preparation of a historic evaluation is required." With the change in Status Codes, the owner of an Uptown property who submits a development application will not be able to rely upon the previous determination of ineligibility as "evidence" that the property is not significant. In such a case, since the property is included in the Survey Report, but "not evaluated" a property owner will inevitably have to spend more money and time for the preparation of a site-specific historic study.

According to the PEIR, "to further increase protection of potential resources specifically potential historic districts—the City is proposing to amend the Historical Resources Regulations to include supplemental development regulations to assist in the preservation of specified potential historic districts until they can be intensively surveyed and brought forward for designation."⁴ The City proposes to "protect" potential, historic districts by amending the San Diego Municipal Code as part of the Community Plan Update process (specifically Sections 143.0210-143.0255, *Potential Historical District*, Section 126.0402-126.0404, *Neighborhood Development Permit Requirement*; and Section 132.1601-132.1604, *Potential historic District Overlay Zone*). Such amendments would, in part, "limit modifications [to the affected property] within the front 2/3rds of the original building footprint" and allow changes to occur only "in the rear 1/3rd of the building footprint or accessory buildings." By establishing a new regulatory framework of this type, merely for the specific protection of potential, historic district shat may never come to fruition, the City will severely and adversely affect an Uptown property owner's right to otherwise develop property in a reasonable and appropriate manner.

In terms of the time and cost associated with the processing of the 23 PHDs, City Staff has indicated that each historic district would take 1-2 years to process at a cost of approximately \$85,000 per district.⁵ Accordingly, it would take approximately 23-46 years to process all 23 PHDS at a cost of approximately \$1,955,000. In addition, if each eligible contributing historic district property (1,454 total properties) were designated and subject to a future Mills Act agreement, the City could anticipate a loss to the General Fund of approximately over \$3.3 million.⁶ The overt development restrictions, lengthy processing time, and enormous costs involved in such a process certainly outweigh the alleged "benefits" that come from the "protection" of contributing properties to any potential, future historic districts. Based upon the overly restrictive regulations resulting from the establishment of PHDs in the Uptown community, as well as the time and costs associated with those policies and procedures, the City should reject the proposed regulatory framework for PHDs. To do otherwise would be both arbitrary and capricious.

(3) Bungalow and Apartment Court Multiple Property Listing (MPL)

The Survey Report introduces the concept of "Multiple Property Listings" (MPLs) which are defined as "a group of related significant properties with shared themes, trends, and patterns

4

⁴ "The proposed potential historic district regulations would provide supplemental protections until a more detailed historic district survey can be completed" (City of San Diego, Planning Department "Fact Sheet").

⁵ Kelley Stanco, Senior Planner & HRB Liaison, "Interim Protections For Potential Historic Districts," Presentation, March 2016.

⁶ This amount was conservatively estimated by determining the median sale price for homes within the Uptown (92103 area code) at an amount of \$572,000 (www.fulia.com). Assuming an average Mills Act property tax reduction of 40% (per City Staff) based upon property taxs of \$5,720 per year (at 1% rate), results in a property tax savings of \$2,288. The amount of this reduction times the number of contributing properties with a Mills Act agreement (1,454 total properties) results in a loss to the General Fund of approximately \$3,326,752. Note that this amount is substantially higher if it were to include those individually significant Uptown properties (2,134) and MPLs subject to a Mills Act agreement (an additional \$4,882,592 loss to the General Fund). A total financial loss could exceed \$8.2 million

of history." The Survey Report identifies three "thematically related property groupings" that appear to be significant as MPLs, including the "Bungalow and Apartment Court" MPL. This MPL is defined as a "discontinuous grouping of approximately 150 residential courts" located throughout the Uptown survey area. The Survey Report indicates that these properties derive significance under Historical Resources Board (HRB) Criterion A (Community Development) "as special elements of the Uptown Community's social history related to multi-family housing, and its architectural development associated with local transportation patterns," as well as Criterion C (Architecture) for "distinctive characteristics of courtyard design." However, these assertions are not thoroughly supported or justified by any new or meaningful historical evidence.

Over the past several years, City of San Diego HRB Staff has entertained a certain fascination and admiration over local bungalow/residential courts as "significant" property types. The genesis behind the history of San Diego's bungalow courts occurred with the publication of "Bungalow Courts in San Diego: Monitoring a Sense of Place" (Spring 1988) in the *Journal of San Diego History*. Subsequently, documentation of bungalow courts as a housing type within the City was discussed further in the Draft Uptown Survey (2007). In addition, one consensual HRB historic designation involving a bungalow court occurred in 2007, and three involuntary HRB historic site designations involving bungalow courts occurred between 2007-2008.

According to "Bungalow Courts in San Diego: Monitoring a Sense of Place," bungalow courts were "well-designed, small houses carefully arranged around a planned open space." They were primarily built in the *hundreds (if not thousands)* throughout Southern California during the 1920s and 1930s and ceased to be built around 1940. Most courts were built along new streetcar lines of the period in some variation of the Mediterraean/Mission style and covered with bougainvillea. The typical bungalow court came to feature a group of six to ten small, individual houses placed around a communal garden. Usually two standard lots were enough. According to the article, bungalow courts were classified into four categories, based upon spatial arrangement. These classes included the (1) detached, full court - the "classic" court consisting of individual cottages arranged around a long, narrow, garden-like walkway (3) attached, full court - when two or more of the bungalows share a common wall, and (4) attached, narrow court. Since the term "court" implies an enclosed, designed space, in all cases the building arrangement included an end structure and a proper garden.

In reliance upon the above cited article, the Draft Uptown Survey (2007) identified a potential "Bungalow & Apartment Court Thematic Historic District" within the Uptown community. Although it should be noted that no present "Bungalow & Apartment Court Thematic Historic District" exists within Uptown or any other part of the City, the survey identified a total of 144 bungalow and apartment courts which were determined to be potentially significant as *district contributors only*, not individually significant, and not as MPLs.

According to the Draft Uptown Survey, which has been essentially adopted as part of the Survey Report, bungalow courts feature well-designed, small houses carefully arranged around a planned open space. The typical bungalow court consisted of a group of six to ten individual houses around a communal garden. Most bungalow courts in San Diego sit on two regular (50' X 100') lots. In several instances, the courts were built in two phases, with one side completed first, and the other side constructed when the land became available.

In August 2007, the "Dr. Chester Tanner Office Bungalow Court" was designated by the City of San Diego's Historical Resources Board (HRB) under HRB Criterion C as "an excellent example of both the Spanish Eclectic architectural style and as an example of a unique 1927-1935 Spanish Eclectic Office Bungalow Court." The property, located in the Uptown community, was determined to be significant due to the fact that it was identified in the draft survey; were constructed as medical office buildings (rather than residential structures); and featured many characteristics of the Spanish Eclectic architectural style.

In 2007, two bungalow court properties were referred to the HRB for involuntary historic site designation. The first property, located at 104-118 Dickinson Street in the Uptown community, was referred to the HRB for designation consideration under HRB Criterion C (Architecture) in November 2007 by City Staff on the basis that it, "drawing heavily from the Minimal Traditional style" was a "a good example of Streamline Modern architecture expressed in the apartment courtyard building type." When considered by the HRB, several Board Members found the property to be more Modern Minimal in style. The HRB refused to designate the property. Similarly, the second property, located at 7522-7534 Herschel Avenue in the La Jolla community, was referred to the HRB for designation consideration under HRB Criterion C (Architecture) in November 2007. City Staff believed the property to be significant on the basis that it was "a very good example of a Minimal Traditional apartment courtyard." Again, the HRB failed to designate the property.

In March 2008, another bungalow court property was referred to the City of San Diego's Historical Resources Board (HRB) for involuntary historic site designation. This property, located at 7417-7427 Olivetas Avenue in the La Jolla community, was referred to the HRB for designation consideration under HRB Criterion A (Community Development) as "the only Contemporary style bungalow court in La Jolla, a limited building type in the community" and under HRB Criterion C (Architecture) "as a very good example of a post-WWI, Contemporary style bungalow court with high integrity." At the hearing, the HRB designated the property, pursuant to the Staff Recommendation, despite a wealth of information supporting the conclusion that the property was appealed to the San Diego City Council and the designation was overturned on the basis that factual errors in materials and information were presented to the HRB at the time of hearing, and upon the submittal of new information indicating that the property was not significant.

The fundamental problem with the present Survey Report is that it alleges that bungalow/residential courts derive significance from their very *nature* as a property type (defined as a "grouping of grouping of individual properties based on shared physical or associative characteristics"). This theory essentially holds that the bungalow/residential court is

RESPONSE

significant because of its multi-family residential use within the Uptown community. By logical extension, would a single-family residence in the Uptown community be considered significant, in and of itself, because it was originally built as a single-family home and has maintained this use over the years? In no instance does the Survey Report establish precisely why the location, design, or the use of the bungalow/residential court as "discontinuous groupings" are any more significant than other similarly-situated multi-family structures, single-family homes, commercial buildings, or other structures built throughout Uptown from the 1920s-1960s.7 Moreover, the Survey Report does not include any additional, substantial information regarding bungalow/residential courts above and beyond much of the information previously generated as part of the Draft Uptown Survey. The Survey Report also fails to explain why the concept of a bungalow/residential court MPL has been advanced when the earlier Draft Uptown Survey proposed the establishment of a potential "Bungalow & Apartment Court Thematic Historic District" within the Uptown community. Finally, the Survey Report is misleading when it asserts that "approximately 150 residential courts" located throughout the Uptown survey area would be included within the MPL and ultimately be "designated as part of a city-wide MPL of San Diego residential courts." This is especially true when one considers the fact that each bungalow/residential court, by definition, has between 6-10 individual homes on each parcel, thereby bringing the total number of actual structures eligible for designation to between 900-1,500. If designated, each eligible bungalow/residential court property subject to a Mills Act agreement could potentially cost the City's General Fund hundreds of thousands of dollars in lost revenue. Based upon the foregoing deficiencies associated with the Survey Report, it should not be adopted by the City for use in the Uptown Community Plan Update.

In conclusion, I appreciate the opportunity to comment on the Plan Update and the PEIR. I look forward to receiving written responses to the issues I have raised in this letter. Please do not hesitate to contact me should you have any questions or need any additional information.

Sincerely,

Scott A. Moonjian Attorney at Law

⁷⁷ Additionally, the Survey Report does not explain or reconcile why the "period of significance" of bungalow/residential courts was extended to 1960, when all other prior authoritative sources have conclusively determined that the construction of bungalow/residential courts generally ended in 1940 (prior to the Second World War).

LETTER



RESPONSE

LETTER

THE OPLOWIN GALEWAY COUNCIL The Association of Commercial Pronerty Owners in Hillcrest			
December 1, 2015			
Mr. Marlon Pangilinan City of San Diego Planning Department 1222 First Avenue, MS 413 San Diego, CA 92101			
Subject: Uptown Community Plan Update June 2015 draft The Uptown Gateway Council			
Dear Mr. Pangilinan,			
We represent the Uptown Gateway Council, an association of commercial property owners in the Uptown Community Planning Area of the City of San Diego. The coalition, which has been recently formed and continues to grow, includes The Greenwald Company, owner of properties on the 3700 block of 6 th Avenue as well as Pernicano family, in the proposed <i>Uptown Gateway</i> District (Attachment 1)	B18D-1	Comment noted.	
To date the Uptown Gateway Council represents 12 individual property owners of over seven acres of commercial land in the core of Hillcrest. The proposed <i>Uptown Gateway District</i> is bounded by Washington Street to the north and Pennsylvania Avenue to the south, 7 th Avenue to the east and 4 th Avenue to the west (Attachment 2).	B18D-2	Comment noted.	
We will fully participate in the community planning process, as our constituency represents the large commercial property owners of Uptown integral participants in this process. We were aware of the June update to the Uptown Community Plan Update that proposed significant downzoning to our properties; however, because it was presented without changes being tracked, we anticipated additional opportunities to fully review and comment on an additional draft. As you are aware, due to the intervening holidays, we requested a minor extension of time to comment on the June draft. We are disappointed no extension was granted, although we understand the pressure the City has placed on the Planning Department to hold to tight schedules.	B18D-3	Comment noted.	
We will continue to vigorously participate in the process from this point forward, presenting our ideas to promote a more forward thinking update to our community plan. Therefore, please consider this letter as our initial statement of general thoughts and ideas. More detailed and additional strategic approaches to responsible planning for this vital area will be forthcoming	B18D-4	Comment noted.	
due to the interveni draft. We are disapp has placed on the Pl We will continue to ideas to promote a consider this letter additional strategic	ng holidays, we requested a minor extension of time to comment on the June bointed no extension was granted, although we understand the pressure the City anning Department to hold to tight schedules. It wigorously participate in the process from this point forward, presenting our more forward thinking update to our community plan. Therefore, please as our initial statement of general thoughts and ideas. More detailed and approaches to responsible planning for this vital area will be forthcoming	ng holidays, we requested a minor extension of time to comment on the June bounded no extension was granted, although we understand the pressure the City lanning Department to hold to tight schedules. It wigorously participate in the process from this point forward, presenting our more forward thinking update to our community plan. Therefore, please as our initial statement of general thoughts and ideas. More detailed and approaches to responsible planning for this vital area will be forthcoming	ng holidays, we requested a minor extension of time to comment on the June pointed no extension was granted, although we understand the pressure the City anning Department to hold to tight schedules. A vigorously participate in the process from this point forward, presenting our more forward thinking update to our community plan. Therefore, please as our initial statement of general thoughts and ideas. More detailed and approaches to responsible planning for this vital area will be forthcoming B18D-4 Comment noted.

	Mr. Pangilinan December 1, 2015 Page 2 of 5		
	shortly.		
B18D-5	Historically, our property owners have understood that they have a long standing recognition of their ability to densify. The fact that nothing has been built does not diminish this historical reliance on the established regional planning efforts and State mandates. Deals have been transacted on that reliance. We cannot just stand still and wait for the rug to be pulled from under them, and with it their property values, dreams and aspirations for a better Uptown.	B18D-5	Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
B18D-6	This draft plan does not accomplish what we anticipate all residents and property owners of Uptown would desire: the activation of the streets with people, commerce, and entertainment. There is a disconnect in this draft plan, in that somehow this activation can be realized while simultaneously diminishing the height and density of future projects. The key to activation of the ground plane and to increasing the public realm is to build higher and denser. If we can achieve this several things happen:	B18D-6	Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
	• Projects become feasible. Our member's projects cannot revitalize the <i>Uptown Gateway District</i> without building higher and denser;		
	• The architecture can be dramatically more interesting and creative;		
	• The residential elements can be more highly amenitized;		
	• There will be an opportunity to create walkable neighborhoods, public spaces, and artistic and cultural amenities at the ground level by opening up the ground plane.		
B18D-7	These goals are in regional, City, and neighborhood interests. It is fundamental to understand that the economics of real estate development dictate that goals for a more livable community are, frankly, unachievable without the flexibility to develop higher and denser projects. Please see Attachment 3 – Financial Feasibility of Development Alternatives prepared by The London Group.	B18D-7	Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
B18D-8	 The other overarching concern that we have is that the draft plan, as written today, is inconsistent with regional planning goals, the goals stated in the 2008 General Plan and accompanying 2013-2020 Housing Element, and the draft Climate Action Plan, which the City of San Diego will adopt this month, among others, to wit: The City has committed to SANDAG certain housing goals as its fair share of regional housing accommodation. Yet, the Uptown Community Plan Update as currently envisioned eliminates over 2,300 units, or 20% of its expected future inventory, where SANDAG projections anticipate that more than 12,000 housing units will be built by 2050. These SANDAG projections are based upon a reliance on the City of San Diego's stated plans and goals. If these units are eliminated, the units must then be reallocated to other communities if the City is to deliver on its commitment and contract with the region. 	B18D-8	Comment noted. The City does not agree that the proposed Uptown CPU is inconsistent with the City's General Plan or CAP. Rather, the proposed Uptown CPU builds upon the goals and strategies of the General Plan and more specifically, of the City of Villages Strategy. The site-specific land use designations and policies of the proposed CPU would also further the goals of the CAP. See response to comment B18-3 for further detail regarding the proposed CPU consistency with the General Plan and CAP and reallocation of housing densities.

LETTER

Mr. Pangilinan December 1, 2015 Page **3** of **5**

- Hillcrest has long served as a "Gateway" community, which means that it is one of the very few places throughout the City of San Diego, which is supposed to be designed to accept density.
- Sixth Avenue is in fact a "gateway" to the City of San Diego. Consider the route taken by
 most major special events, parades and community gatherings.
- The City has also committed to Transit Oriented Development (TOD) which recognizes
 that places that are well served with multiple transportation nodes should be encouraged to
 more densely develop. The Uptown Gateway District is now served by six bus routes, and
 is already close to the region's largest employment center (Downtown), Balboa Park, and
 shopping.
- Uptown is a community that can deliver housing which is affordable to a larger cross section of our community. This is not another expensive community such as Downtown, La Jolla, Carmel Valley or Point Loma. This is a community that historically has housed middle class residents. This demographic will be maintained with future projects, if we are able to achieve the necessary density.
- Lowering density creates an economic productivity issue. Diminishing, rather than enhancing, the economic productivity of this area ultimately translates into a regional failure. The net result of the draft plan as written is that new development doesn't work and not much will change for the better - this in an area where, as you have been notified, economic activity is currently scaling down. If not much is built or rehabilitated, this makes the City worse. Maintaining some sort of Uptown "status quo", or creating something even less than the status quo, is a stab in the heart of economic productivity.
- B18D-9

Therefore, on behalf of our members and the Uptown Gateway Council, we are requesting the Uptown Community Plan Update be revised with the following:

- Re-designate the proposed Community Village area on figure 2-5 to an Urban Village area;
- Maintain the land use designation as high density as per the current community plan and the base zoning density at one unit per 400 square feet (the proposed of CC-3-6 is completely unacceptable and constitutes a dramatic devaluation of all the properties);
- Allow projects that "significantly" improve and enhance the public realm to achieve

B18D-9 Comment noted. This comment requests changes to the proposed Uptown CPU. Refer to response to comment B18-3 regarding the appropriate residential density allocation and mixed uses of the proposed CPU and its ability to further the goals of the General Plan's City of Villages Strategy and CAP. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.

¹ "Significantly" is term that we hope to define in this update process. We believe there is a way to develop some objective criteria to enhance the public realm in exchange for density and height increases. We currently have grave concerns with the Incentive Zoning Analysis dated September 25, 2014 both from a design and construction

LETTER

RESPONSE

Mr. Pangilinan December 1, 2015 Page **4** of **5**

densities of one unit per 200 square feet and heights over 200 feet;

- Eliminate the application of CPIOZ provide increased certainty in the development process and do not turn policy documents into quasi regulatory documents;
- Reduce the number of Urban Design Element policy redundancies and conflicts in order to more clearly align these guidelines with existing regulations. Many of the policies listed are not practical or enforceable at an implementation stage (Attachment 4);
- Allow the concept of vertical flexibility to focus on the ground plane;

The above request is based upon several factors including the City Council's off stated need for additional housing within the City and the region, the direct proximity to Downtown San Diego, access to six transit lines and a future streetcar, and the need for flexibility to encourage future development.

- B18D-10 For the reasons stated above, we request that the Uptown Community Plan Update create the *Uptown Gateway District* and maintain the Land Use Designation as high density within the district, consistent with the adopted community plan and the General Plan Housing Element. Further, we request that the property owners, all the current signatories of the Uptown Gateway Council, along with the Atlantis Group, be added to the notification list for all future Stakeholder, Working Group, Planning Group, and all other meetings related to the Uptown Community Plan Update.
- B18D-11 We look forward to working with you and your staff to effectuate the above requested change. We also hope to be a resource for you as an association of commercial property owners. We are excited to be participating in the upcoming Uptown Planners Community Plan Update meetings and discussions.

If you have any questions or need additional information, please let me know at (619) 523-1930 or at <u>mescobareck@atlantissd.com</u>,

- B18D-10 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required. However, see response to comment B18-3 for further detail regarding the proposed CPU consistency with the General Plan and CAP. The City will add the Council and Atlantis Group to future stakeholder workshops, outreach, or other input meetings related to the Uptown community. Pease note that at this time, the public outreach effort to gain input from community stakeholders has ceased, and the proposed Uptown CPU would be adopted at the time of the Final PEIR.
- B18D-11 Comment noted. The remaining attachments to this letter do not raise an issue with regard to the adequacy of the PEIR, thus a response is not required.

implementation stand point as well as from a market & feasibility standpoint.

LETTER

RESPONSE

ATTACHMENT 7

Mr. Pangilinan December 1, 2015 Page **5** of **5**

Sincerely,

Haicelo tseobai-tck

- Marcela Escobar-Eck Principal Atlantis Group Land Use Consultants On behalf of the Uptown Gateway Council
- cc: City of San Diego:

Honorable Councilmember Todd Gloria Jeff Murphy, Director, Planning Department Tait Galloway, Program Manager, Planning Department

Development Team: Sherm Harmer, Urban Housing Partners Gary London, The London Group Jeannine Savory, The Savory Group Ricardo Rabines, Safdie Rabines Architects Jeff Chine, Allen Matkins

Attachments:

- 1. Letter from the current Uptown Gateway Council
- 2. Uptown Gateway District boundary map
- 3. Financial Feasibility of Development Alternatives
- 4. Initial Comments on June 2015 Draft Uptown Community Plan

RESPONSE

Attachment 1

The Uptown Gateway Council

The Association of Commercial Property Owners in Hillcrest

November 10, 2015

RE: Proposed Down Zoning of Hillcrest

To Whom It May Concern:

As you know, the City and the community of Hillcrest have been engaged in a process that will lead to a community plan update next spring. We applaud their efforts, fully recognizing that our community is at a crossroads in terms of the need to modernize not just the plan, but the community's infrastructure, accommodating new businesses and new development of all kinds.

No one has a greater stake in our community than the signatories to this letter. Together, we own a majority of property located within the business "heartland" of Hillcrest. The drawing appended to this letter shows our ownership and the potential positive influence development of these properties offers to the community and the region over time.

This letter is a general description of our aspirations and expectations for not just our properties but for the future of the Hillcrest community. We address all of our neighbors, as well as, the Uptown Planning Group and the City of San Diego.

As the City and community work together to complete this effort to sculpt a new plan please think beyond the short term and the present. Formulate a plan that will accommodate and encourage a better future for generations to come in the community.

We suggest major planning principles which include:

- The San Diego region is growing. It will continue to grow providing future economic opportunities for our children and grandchildren. There is significant demand for new facilities in every neighborhood; especially those close to transportation nodes. Therefore, do not downsize the community's ability to accept new development, either through limitations on height, density or development type. We cannot support any effort to downsize future development through height restrictions, recessive Floor Area Ratio (FAR) requirements, parking restrictions or any other regulatory tool. These tools restrict our community's ability to accept future housing units, retail, office and hotel uses. Building on the existing infrastructure and existing uses in each of these categories, we need to increase our facilities in these land use areas, not reduce them.
- Create certainty for future development and let the plan encourage change rather than be an inhibitor of progress. The standards contained within the current draft Uptown Community Plan would best serve the community if they were written, worded, and dedicated to the proposition that new development which accommodates community facilities is encouraged. The plan should establish a "baseline" for scheduling development approvals rather than building in mechanisms employed by future "objectors" to cause lengthy approval process. In other communities, inhibiting plans eventually result in less inspired developments and less desirable communities. Invite development, redevelopment and creativity in Hillcrest.

- Preserve that which is worthy, but don't go overboard. Let's be realistic about what
 exists. Much of Hillcrest needs renewal and redevelopment. Not everything that is old is
 worthy of preservation. We can celebrate our rich history while still encouraging
 innovative development. We are a community of mostly older buildings. Many of those
 buildings will eventually need replacement. Few are worthy of historic preservation. We
 want the "bar" to be set very high on standards of preservation, so this tool is not misused
 to prevent new development in places that it is warranted.
- Create contemporary incentives to encourage excellent architecture and "green" projects. We encourage the new Plan to provide a method by which certain tools can be employed to achieve greater height and density, through the tradeoff mechanism of bonuses and incentives. Those tools include "green" incentives for energy efficiencies, pocket parks, electric car stations, etc., which are being applied downtown and implemented by Civic San Diego.
- Allow flexibility in the plan. It is likely to be in place for a long time. Create a process not a frozen set of regulations. No one really knows what the future will bring. Standards of today may become anachronisms tomorrow. Needs that exist today may not exist tomorrow. A plan should be written that provides for alternative ways to achieve stated policy goals. The process, and the very plan it is based on, should recognize that. For instance, we believe that current parking standards shouldn't be applicable, as auto drive counts in urban places are dropping with changing auto use propensities and alternative transportation choices. We may even see transformational changes in auto usage as autonomous autos are introduced and become the new standard. The point is, the process needs to be able to encourage flexibility, accommodate change, not treat it as an "exception" to the plan.
- Recognize Hillcrest is an urban "Hub" and a gateway to the City. Recognize that the pressure for growth is real. Take advantage of what that pressure can provide for Hillcrest, the City and the region. We want Hillcrest to continue to be an example of how the City intends to accommodate and implement its "City of Villages". This community should not run from change, growth and opportunity. Any bird's eye view of Hillcrest shows that we are San Diego's "Uptown" hub. We can accommodate new urban households, businesses and the supporting land uses that come with it. Let's plan to create a lively, even more exciting and inviting community. Let's embrace this better future for Hillcrest.

The undersigned property owners do not necessarily have near term development plans for our respective properties, except to maintain them and keep our tenants happy. However, as your partners in the community, we simply wish to preserve our property values and our rights to redevelop in future years.

(SIGNATURES ON THE FOLLOWING PAGE)

Page | 2

Signed and Agreed,

NAME OF PROPERTY OWNER OR REPRESENTATIVE:	PROPERTY LOCATION OR ADDRESS:
Bennet Greenwald	3715-3795 6 th Avenue
Nick Totah	3935-3941 4 th Avenue
Lucy Burni	1202 University Avenue
Charlie Jadallah	3864 5 th Avenue, 441 University Avenue, 3917 4 th Avenue and 3850 4 th Avenue
Ron Pelman	3900 5 th Avenue
Gary Pernicano	3818-3840 6th Avenue and 3835 5th Ave
KG Ventures	501-535 University Avenue
Lyda Cohen	3825 5 th Avenue
Bob LaFever	635 Robinson Avenue
Roger Arko	3796 5th Ave and 3845 5th Ave
	· · · · · · · · · · · · · · · · · · ·
÷	

Page | 3

RTC-163

LETTER

	Hand and Agord, Name of PECONSTITUTION AND HAMARAN TATION Beneric Desale unog for (Har wein Jeen Ben- (Har wein	1710-1117 LOGATION (SIL ADDILITES 3716-3795 Bill ref AVADELIE 5905 HAN AIR, 121453 905 Aleco (1) S.(1) 9:213.7 3854 5" Auf 1471 Unsersty / 3117 4 RAN 1570 7" Aig 3590 5" Augustus 7210 3 3840 6th Augustus 7210 3 3796 5th Clar 3845 574 Aug 9:103 3845 574 Aug 9:103 3845 574 Aug 9:103	
×	Hynd ad Agord, Naves of Fillower (MARTA AND Barner Bright UND) romer rense (Martin Jack) Cer y Charlie Idelta De Pertman Lander Jack	10000000000000000000000000000000000000	
	Kliger arko	3845 5# Que 9403	
	е :	Ange (3	
		•	
	e ^t e	ал ж	

LETTER

ATTACHMENT 7



LETTER



LETTER



LETTER



LETTER

RESPONSE



Source: The London Group Realty Advisors

To determine feasibility, we analyzed construction costs and projected revenues for each development alternative. This formed the basis for a calculation of residual land value, which is the price a developer would be willing to pay for the properties to achieve an acceptable level of profit.¹

Ultimately, the lower density development options (existing zoning and downzoning) do not achieve sufficient land value to justify redevelopment of the property. This residual land value analysis is identical to an analysis that any prospective developer or their capital partner would make as part of their due diligence in evaluating the development opportunity.

The overarching conclusion is that your property is not economically feasible to develop under these conditions.

The following bullet points highlight the key elements of our analysis:

- The price a developer would pay for the land is lower than the value of the existing improvements and income stream currently generated by the property.
- 2. Greater density is required to achieve the higher land residuals necessary to facilitate redevelopment. Zoning of 1 unit per 200 SF of land achieves this, assuming there are no other restrictions such as affordable housing requirements or open space requirements which would reduce land value.
- Construction costs will increase with density and height due to the type of construction (concrete and steel) and the inclusion of underground structured parking. Ultimately, the density must be sufficient to support these higher costs as well as create a higher land residual.

¹ Current return metrics range from 6.0% yield on cost (net operating income divided by costs) or an IRR well above 15%.

Page 2 of 4

LETTER

RESPONSE



Sixth & Robinson Financial Feasibility of Alternatives

4. The optimal zoning would allow for a Floor Area Ratio (FAR) of 6.0 or greater. This would permit more innovative urban design, including increasing open space and better activate the ground level, a notable community benefit. Compared to the existing commercial structures, such a development would dramatically benefit the community.

Should you have any questions regarding this analysis, please contact us.

Sincerely,

Joy H. Tool Gary H. London President

Nathan Morden

Nathan Moeder Principal

Page 3 of 4

RESPONSE

Sixth & Robinson Financial Feasibility of Alternatives

CORPORATE PROFILE

THE LONDON GROUP Realty Advisors

REPRESENTATIVE SERVICES

Market and Feasibility Studies	Development Services	Litigation Consulting
Financial Structuring	Fiscal Impact	Workout Projects
Asset Disposition	Strategic Planning	Valuation
Government Processing	Capital Access	Economic Analysis

The London Group is a full service real estate investment and development consulting, capital access and publishing firm. We determine the answers to the questions: Should 1 purchase the property? If so, how much should 1 pay and what is my potential rate of return? What type of project should 1 invest in or develop? What type of deal should 1 structure?

To answer these questions we conduct market analysis, feasibility studies, provide financial structuring advice and general economic consulting. Often we 'package' the deal and provide access to capital sources. We also have capabilities in pre-development consulting including asset management and disposition and in providing team coordination, processing and disposition services (packaging and promotion).

We cyber publish The Real Estate & Economic Monitor a newsletter providing market trend analysis and commentary for the serious real estate investor. It is available and regularly updated on the World Wide Web at the following address: www.londongroup.com.

As the former West Region Director of the Price Waterhouse Real Estate Consulting Group, Gary H. London, President, brings acknowledged credentials and experience as an advisor and analyst to many successful projects and assignments throughout North America.

The London Group also draws upon the experience of professional relationships in the development, legal services, financial placement fields as well as its own staff.

Clients who are actively investigating and investing in apartment projects, retail centers and commercial projects have regularly sought our advice and financial analysis capabilities.

We have analyzed, packaged and achieved capital for a wide variety of real estate projects including hotels, office buildings, retail shopping centers and residential housing communities. We are generalists with experiences ranging from large scale, master planned communities to urban redevelopment projects, spanning all land uses and most development issues. These engagements have been undertaken throughout North America for a number of different clients including developers, investors, financial institutions, insurance companies, major landholders and public agencies.

Page 4 of 4

RESPONSE

Attachment 4

Land Use Element

The Land Use Element's goal pertaining to commercial development states the desire for "active commercial districts that benefit from a sustainable level of residential density and multiple mobility options." The following comments are designed to assist in meeting that goal:

Initial Comments on June 2015 Draft Uptown Community Plan

- Policy LU-2.5, "Preserve and enhance the special character of specific, well-defined, low-density neighborhoods from encroachment by incompatible, higher density residential or commercial development" does not reference which specific, well-defined neighborhoods are to be preserved and does not explain what would qualify as encroachment.
- 2. Figure 2-1 contradicts LU-2.5 by placing higher density commercial (0-73 Du/Ac) along University Avenue east of SR 163 adjacent to lower density residential (30-44 Du/Ac) and lower density commercial (0-44 Du/Ac) along 6th Avenue north of Pennsylvania with higher density residential (45-73 Du/Ac). Density should be increased in the Uptown Gateway District to match the higher residential density.
- Figure 2-5: Conflict in densities identified for Community Village and Neighborhood Village and General Plan designations in GP Table LU-4. The Community Village Area in the Uptown Gateway Council should have higher commercial density than the Neighborhood Village Areas.
- 4. LU-3.8: Permit high intensity pedestrian-oriented commercial and mixed-use development in the Hillcrest Neighborhood Center/Node surrounding University and Fifth Avenues.
 - a. This area is not zoned for high intensity commercial, other parts of Uptown are zoned higher, yet the Hillcrest Neighborhood Center/Node is the only area specifically mentioned in the policy language for high intensity development. This policy and the categorization as a Community Village show that the area should have a higher permitted commercial and residential density.

Urban Design Element

The Urban Design Element's stated intent is to "set forth broad urban design concepts... as well as more specific principles and related design guidelines..." (Pg. UD-60 Intro). The following recommendations are proposed to align these policies with regulatory tools to build within the community's intent:

- 5. The 'Hillcrest Core' is described in the Existing Context and Urban Form section (UD-61) and appears to be identified on Figure 4-2 Neighborhood Centers and Nodes map as the largest of several identified 'center and node' areas. This broad urban design concept of Community and Neighborhood Cores, Centers, and Nodes needs to be clarified as these terms are interchangeable throughout the element and not specific to the General Plan's Village Place Type (LU Element Section 2.3). In addition, the 'Hillcrest Core' is identified by SANDAG Smart Growth Concept Map as an 'Urban Center,' and Strategic Framework Element "Commercial Village Center." We recommend identifying the' Hillcrest Core' as an Urban Village Center on Figure 4-2;
- 6. Landmarks and Gateways are described in Existing Context and Urban Form section (UD-61) as "distinct areas," and are identified differently on Figure 4-3 Landmarks and Gateways map as singular Gateway signs, Bridges, and Buildings. This broad urban design concept should incorporate 'Gateway Areas' at key places on Figure 4-3 to announce the entry into a neighborhood or commercial districts to demarcate key historic, cultural, civic, and

1

LETTER

shopping areas. This recommendation would implement UD-4.31 "At Gateway Locations..." and 4.56 "mark key gateways," to more clearly identify the Core, Center, and Nodes concept;

- 7. The Urban Design Element is organized by 4.3 Streets and the Public Realm, and 4.4 Development Form (private realm), however, 4.4 also includes a Public Space (UD-79) and Public Art (UD-80) sections, which are redundant and should be edited to more clearly define public realm versus private realm expectations for new development applicants.
- 8. More specific design guidelines redundancies requiring editing includes:
- Guidelines for "articulating building façade," are repeated in Street Wall Articulation (UD-4.1, 4.3, 4.4), Ground Level Uses (4.10), Windows (4.16), Building Transparency (4.32), Public Space (4.51), Street Orientation (4.63), Setbacks (4.65), Height and Massing in Neighborhood Centers and Nodes (4.69), and Height and Massing Residential Neighborhoods (4.71) sections;
- Guidelines for "high-quality materials" are repeated in Building Materials (4.18), Signs (4.26), Architectural Projects (4.37), and Street Orientation (4.64).
- Two (2) of the ninety (90) policies are 'Required' (shall) while all others are 'should' policies. Requiring a narrow range of floor-to-floor heights (4.8), and the design of all rooftops in an expressive manner creates conflicts. For more specific design guidelines for "Rooftops and Rooffines, and Mechanical Equipment" sections requiring screening (4.44, and 4.45) and complimentary roof design as opposed to expressive (4.40, 4.43, and 4.70) (hese policies should be combined and/or eliminated where existing land development codes already address these items);
- Bay Window projections are outlined in sections (4.4, and 4.38), and in conflict with more general window guidelines (4.14, 4.16, 4.30, 4.32 – 4.35, and 4.51).
- Eliminate policies UD-4.22, 4.37 3rd bullet, and 4.64 as untenable polices.
- 9. The 4.4 Development Form section begins with general-to-all policies and guidelines (4.1-4.66) and concludes with three important sections, Height and Massing in Neighborhood Centers and Nodes (refer to Figure 4-2), Height and Massing in Residential Neighborhoods (these are undefined in UD Element), and Transitions. We understand the issue of transitions between existing and new development, mixed-use and single-use development, and lower-density to higher-density areas is of utmost importance to the community during this update process. Therefore, we recommend defining Residential Neighborhood areas on Figure 4.2 or creating a new map defining these areas to link to the policies guidelines (4.71-4.78). These important policy guidelines would be listed in the proper hierarchical order of policies over the more general police. More specific design guidelines redundancies requiring editing includes:
- Edit 4.69 "simple, yet varied, massing..." as this is impossible to interpret;
- Edit 4.69 and 4.71 as these are redundant to general policies 4.4, 4.37 4.41;
- Edit 4.76 as this in conflict with regulatory implementation;
- Edit 4.77 windows and 4.78 roof breaks as in conflicts with above mention general polices
- Edit Building Types (4.79 4.81) to refer to Neighborhood Centers and Nodes and Residential Neighborhood areas.

2
LETTER



RESPONSE

The Uptown Gateway Council The Association of Commercial Property Owners in Hillcrest June 8, 2016 Comment noted. The City appreciates the Council's participation B18E-1 Jeff Murphy in the public review process. **Director Planning Department** City of San Diego Sent Via Email B18E-2 Comment noted. The comment requests a new zone be added **RE: Uptown Community Plan Update** to the proposed Uptown CPU, but does not identify a specific deficiency or impact it aims to correct or mitigate in the PEIR. B18E-1 The Uptown Gateway Council (UGC), a coalition of commercial property owners in and surrounding the This comment does not suggest an inadequacy in the PEIR; Hillcrest Core, applauds the work of the City of San Diego Planning Department for their tremendous therefore, a detailed response is not required. effort in updating the Uptown Community Plan. The effort of the Planning Department has shown a thoughtful approach in their pursuit of a balanced plan between the unique needs and interests of the Uptown Community, and City and State's legally binding plans and legislation. B18E-3 Comment noted. The UGC wants a thriving Uptown and Hillcrest with flourishing businesses that support a vibrant B18E-2 community. We envision a healthy local ecosystem called the Uptown Gateway where people of all B18E-4 The CC-3-9 zone calls for Community Commercial land uses, in walks of life can live, work, and play in the core of Hillcrest. Based on the UGC's decades of cumulative experience as property owners redeveloping communities across the country, we urge the City to adopt which residential development is permitted. This would allow a new zone for the Uptown Gateway, CC-3-10 (Attachment 1). This is the only way that a vibrant for a mixed-use redevelopment of areas within the Hillcrest Hillcrest Core will ever be achieved. neighborhood zoned as CC-3-9. B18E-3 The UGC became more actively involved in the Uptown Community Plan Update process after the Draft June 2015 Plan was released and the affected property owners learned that their properties were being downzoned, some by more than half the density they are currently allowed by right. Since then, UGC The Land Use Element of the proposed Uptown CPU included B18E-5 members have attended every regular and special Uptown Planners meeting and have written letters, policies specific to the goals listed in this comment. For collectively and individually, sharing their industry knowledge of how the Plan will affect the most important commercial hub in Uptown over the next 20 to 30 years. The UGC believes it is not only a example, Policy LU-2.8 calls for the provision of incentives for crucial stakeholder in the Update process, but has extremely relevant and valuable insight to be utilized mixed residential/commercial development at appropriate in developing a Community Plan Update that fosters the highest level of success for the Uptown locations and Policy LU-2.9 requires higher density residential Community, benefiting both commercial property owners and residents. B18E-4 development to be located appropriately to promote safer and Although the City now proposes maintaining some of the existing densities, redevelopment in the Hillcrest Core cannot be achieved with the proposed development standards in CC-3-9. Therefore, we livelier commercial districts. Policies MO-1.1 through MO-1.16 of propose a new zone based on market economic analysis and best practice urban design principles, many the Mobility Element support the enhancement of pedestrian of which are replicated today in places like the Pearl District of Portland, Oregon. facilities and creation of a walkable network. Additionally, B18F-5 After careful review of the March 2016 revisions to the Proposed Zoning Matrix, Community Plan Implementation Overlay Zone (CPIOZ) Maps, and Draft Commercial Development Regulations that were policies included in the Economic Prosperity Element further presented at the March 24th Planning Department workshop, as well as the January 2016 revisions to these goals. For example, Policy EP-1.1 required the the Land Use Element, more must be done in order to meet the January 2016 draft Land Use Element's improvement of pedestrian, bicycle and transit infrastructure in own goals (page 2), specifically: Uptown's commercial districts and Policy EP-1.2 supports the 1 revitalization of alleys in commercial mixed-use areas to improve aesthetics and safety.

The Uptown Gateway Council The Association of Commercial Property Owners in Hillcrest Active commercial districts that benefit from a sustainable level of residential density and multiple mobility options; Continued revitalization of commercial districts Uptown Community Plan Update draft; and . Active pedestrian-oriented commercial areas. B18F-6 B18E-6 In addition, the following General Plan policies must be adhered to in the Community Plan Update and the UGC proposals only strengthen the adherence to the City of Villages strategy: • LU-H.3: Provide a variety of housing types and sizes with varying levels of affordability in residential and village developments. LU-H.7: Provide a variety of different types of land uses within a community in order to offer opportunities for a diverse mix of uses and to help create a balance of land uses within a community LU-I.2: Balance individual needs and wants with the public good. LU-I.10: Improve mobility options and accessibility for the non-driving elderly, disabled, lowincome and other members of the population b. Increase the supply of housing units that are in close physical proximity to transit and to everyday goods and services, such as grocery stores, medical offices, post offices, and drug stores. LU-I.11: Implement the City of Villages concept for mixed-use, transit-oriented development as a way to minimize the need to drive by increasing opportunities for individuals to live near where they work, offering a convenient mix of local goods and services, and providing access to high quality transit services. B18E-7 Finally, the recently adopted Climate Action Plan Appendix B makes the Hillcrest Core a Transit Priority B18E-7 Area. Within Strategy 3: Bicycling, Walking, Transit & Land Use, the following support our recommendation: · Goal: Promote effective land use to reduce vehicle miles traveled. Action 3.6: Implement transit-oriented development within Transit Priority Areas. Target: Reduce average vehicle commute distance by two miles through implementation of the City of Villages Strategy by 2035. Supporting Measures: o Achieve better walkability and transit-supportive densities by locating a majority of all new residential development within Transit Priority Areas. In the attachments you will find the proposed CC-3-10 zone, Community Plan Implementation Overlay Zone (CPIOZ) Maps, Uptown Gateway Specific Plan section of the Uptown Land Use Element, and the financial analysis that demonstrates, in detail, the need for a new zone for the Uptown Gateway. We look forward to supporting the Community Plan Update that includes the CC-3-10 zone in the Uptown Gateway, and to a future in Uptown with a diversity of successful businesses and housing options that support Uptown. 2

- 18E-6 Comment noted. The comment does not raise an inadequacy in the PEIR. The proposed Uptown CPU adheres to these listed General Plan policies. Specifically, Policies LU-H.3 and LU-H.7 of the General Plan is supported by Policies LU-1.1 and LU-1.2 of the proposed Uptown CPU; Policy LU-1.2 of the General Plan is supported by Policies LU-2.1 through LU-2.4 and multiple other policies of the proposed Uptown CPU; Policy LU-1.10 of the General Plan is supported by Policy LU-2.3 and multiple Mobility Element policies that promote pedestrian, bicycle, and transit facilities of the proposed Uptown CPU; and Policy LU-1.11 of the General Plan is supported by Policies LU-3.1 through LU-3.7.
- B18E-7 Comment noted. The proposed Uptown CPU supports the listed items from Appendix B of the CAP, including promoting effective land use to reduce vehicle miles traveled (Policy CE-3.1), implementing transit-oriented development within Transit Priority Areas (Policy LU-2.6), and implementing the City of Villages Strategy (Policy LU-3.1 through LU-3.7).

LETTER

	The Uptown Gateway Council The Association of Commercial Property Owners in Hillcrest		
	Sincerely,		
	Fluid D		
	On behalf of the Uptown Gateway Council		
	Comprised of the Undersigned Commercial Property Owners		
	CC: Mayor Kevin Faulconer Mike Hansen, Director of Land Use and Environmental Policy Councilmember Todd Gloria <u>Planning Department</u> Nancy Bragado, Deputy Director, Long Range Planning Tait Galloway, Program Manager Marlon Pangilinan, Senior Planner		
B18E-8	<section-header><section-header><section-header><section-header><list-item><list-item><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></list-item></list-item></section-header></section-header></section-header></section-header>	B18E-8	The attachments provided are noted. Regarding the letter dated May 18, 2016 prepared by the London Group Realty Advisors (Attachment 4 to Exhibit E); this letter provides a financial analysis of future development within the CC-3-9 zone. This economic analysis is noted; however it does not raise an inadequacy with regard to the content of the PEIR. The densities provided in the proposed land use plan for the Uptown CPU would meet City and State mandated housing requirements, which does not require the City to demonstrate the financial feasibility of development at the densities provided.

LETTER

RESPONSE

The Uptown Gateway Council

The Association of Commercial Property Owners in Hillcrest

Signed and agreed upon by the following members of the Uptown Gateway Council:

Hillcrest 5th Avenue, LLC By: Carleton Management, Inc. its Manager

Jeffrey H. Silberman, President & CEO

Hillcrest Retail, LLC By: Carleton Management, Inc. its Manager

Jeffrey H. Silberman, President & CEO

Roger Orko

Diana Strauss Casey Diana. Strauss Casey Strauss Properties + Danube Properties, Inc.

K6 Ventures eg Morgan H. Allman, goneral partner 501-535 Umversity twe.

Hundle sumand

4

LETTER



RESPONSE

San Diego Munie (1-2016)	icipal Code (Proposed Changes 3-17-2016) ATTACHMENT 1 Chapter 13: Zones
	 CN-1-5 allows <i>development</i> with a pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 600 square feet of <i>lot</i> area (Added 12-9-1997 by 0-18451 N.S.; effective 1-1-2000.) (Amended 7-10-2015 by 0-20512 N.S.; affective 8-9-2015.)
	[Editors Note: Amendments as adopted by O-20512 N. S. will not apply within the Coastal Overlay Zone until the California Coastal Commission certifies it as a Local Coastal Program Amendment. Click the link to view the Strikeout Ordinance highlighting changes to prior language http://docs.sandiego.gov/municode_strikeout_ord/O-20512-SO.pdf]
§131.0503	 Purpose of the CR (CommercialRegional) Zones (a) The purpose of the CR zones is to provide areas for a broad mix of business/professional office, commercial service, retail, wholesale, and limited manufacturing uses. The CR zones are intended to accommodate large-scale, high intensity <i>developments</i>. Property within these zones will be primarily located along major <i>streets</i>, primary arterials, and major public transportation lines. (b) The CR zones are designed for auto-oriented <i>development</i> and are differentiated based on the uses allowed as follows: CR-1-1 allows a mix of regional serving commercial uses and residential uses with an auto orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area CR-2-1 allows regional serving commercial and limited industrial uses with an auto orientation but no residential use (Added 12-9-1997 by 0-18451 N.S.; effective 1-1-2000.) (Amended 7-10-2015 by 0-20512 N.S.; effective 8-9-2015.) [Editors Note: Amendments as adopted by 0-20512 N. S. will not apply within the Coastal Overlay Zone until the California Coastal Commission certifies it as a Local Coastal Program Amendment. Click the link to view the Strikeout Ordinance highlighting changes to prior language http://docs.sandiego.gov/municode_strikeout_ord/O-20512-SO.pdf]

Т

RESPONSE

110)				
§131.0504	Purpo	se of t	he CO (CommercialOffice) Zones	
	(a)	The p limite zones areas	urpose of the CO zones is to provide areas for ed, complementary retail uses and residential are intended to apply in large-scale activity of where a full range of commercial activities is	r employment uses with use as specified. The CO centers or in specialized s not desirable.
	(b)	The C allow	O zones are differentiated based on the uses ed as follows:	and development scale
		(1)	The following zones allow residential development	lopment:
			 CO-1-1 is intended to accommodate residential uses with a neighborhoo permits a maximum <i>density</i> of 1 <i>dw</i> square feet of <i>lot</i> area 	e a mix of office and d scale and orientation and <i>velling unit</i> for each 1,000
			 CO-1-2 is intended to accommodate residential uses that serve as an emp a maximum <i>density</i> of 1 <i>dwelling u</i> feet of <i>lot</i> area 	e a mix of office and ployment center and permits nit for each 1,500 square
		(2)	The following zones prohibit residential de	velopment:
			 CO-2-1 is intended to accommodate neighborhood scale and orientation 	e office uses with a
			CO-2-2 is intended to accommodate employment center	e office uses that serve as an
		(3)	The following zones allow residential development;	lopment in a pedestrian
			 CO-3-1 is intended to accommodate residential uses and permits a maximumit for each 800 square feet of lot 	e a mix of office and mum <i>density</i> of 1 <i>dwelling</i> area
			 CO-3-2 is intended to accommodate residential uses and permits a maxi- unit for each 600 square feet of lot 	e a mix of office and mum <i>density</i> of 1 <i>dwelling</i> area
	(Adde (Amer	d 12-9- nded 7-	1997 by O-18451 N.S.; effective 1-1-2000.) 10-2015 by O-20512 N.S.; effective 8-9-2015	i.)
				Che Art Div
				13 1 5 3

RESPONSE

	[Editors Note: Amendments as adopted by O-20512 N. S. will not apply within the Coastal Overlay Zone until the California Coastal Commission certifies it as a Local Coastal Program Amendment. Click the link to view the Strikeout Ordinance highlighting changes to prior language http://docs.sandiego.gov/municode_strikeout_ord/O-20512-SO.pdf]	
131.0505	Purpose of the CV (CommercialVisitor) Zones	
	(a) The purpose of the CV zones is to provide areas for establishments catering to the lodging, dining, and recreational needs of both tourists and the local population. The CV zones are intended for areas located near employment centers and areas with recreational resources or other visitor attractions.	
	(b) The CV zones are differentiated based on <i>development</i> size and orientation as follows:	
	 CV-1-1 is intended to accommodate a mix of large-scale, visitor- serving uses and residential uses and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area 	
	 CV-1-2 is intended to accommodate a mix of visitor-serving uses and residential uses with a pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of <i>lot</i> area 	
	(Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.) (Amended 7-10-2015 by O-20512 N.S.; effective 8-9-2015.)	
	[Editors Note: Amendments as adopted by O-20512 N. S. will not apply within the Coastal Overlay Zone until the California Coastal Commission certifies it as a Local Coastal Program Amendment. Click the link to view the Strikeout Ordinance highlighting changes to prior language http://docs.sandiego.gov/municode_strikeout_ord/O-20512-SO.pdf]	
Art. Div. 1 5 4		

1

RESPONSE

San Diego Mu	micipal (Code	(Proposed Changes 3-17-2016)	ATTACHMENT 1 Chapter 13: Zones
1-2016)				
§131.0506	Purp	ose of	the CP (Commercial–Parking) Zone	
	The j autor comr (Add	ourpose nobiles nercial ed 12-9	of the CP zone is to provide off-street parking . The CP zone is intended to be applied in con- areas to provide needed or required off-street p <i>p-1997 by O-18451 N.S.; effective 1-1-2000.</i>)	; areas for passenger junction with established parking.
§131.0507	Purp	ose of	the CC (CommercialCommunity) Zones	
	(a)	The p commintent for a stree of th zone trans	purpose of the CC zones is to accommodate co- mercial services, retail uses, and limited industu- sisty and small to medium scale. The CC zones range of <i>development</i> patterns from pedestrian <i>is</i> to shopping centers and auto-oriented strip c e CC zones may include residential <i>developme</i> . s will be primarily located along collector <i>street</i> portation lines.	mmunity-serving rial uses of moderate s are intended to provide 1-friendly commercial commercial streets. Some int. Property within the CC ets, major streets, and public
	(b)	The follo	CC zones are differentiated based on the uses a ws:	llowed and regulations as
		(1)	The following zones allow a mix of commu- uses and residential uses:	nity-serving commercial
			CC-1-1 is intended to accommodate commercial characteristics and perm 1 dwelling unit for each 1,500 square	development with strip nits a maximum density of e fect of lot area
			 CC-1-2 is intended to accommodate intensity, strip commercial character maximum density of 1 dwelling unit of lot area 	development with high ristics and permits a for each 1,500 square feet
			 CC-1-3 is intended to accommodate orientation and permits a maximum for each 1,500 square feet of <i>lot</i> area 	: development with an auto a <i>density</i> of 1 <i>dwelling unit</i> a
		(2)	The following zones allow community-servi uses:	ing uses with no residential
			CC-2-1 is intended to accommodate commercial characteristics	development with strip
			CC-2-2 is intended to accommodate intensity, strip commercial character	development with high ristics
				Ch. Art. Div.

RESPONSE

San Diego Municipal Code (1-2016)	(Proposed Changes 3-17-2016)	Chapter 13: Zones
	CC-2-3 is intended to accommodate de orientation	evelopment with an auto
	 CC-2-4 is intended to accommodate de pedestrian orientation 	evelopment with a
	 CC-2-5 is intended to accommodate de intensity, pedestrian orientation 	evelopment with a high
(3)	The following zones allow a mix of pedestriar serving commercial uses and residential uses:	n-oriented, community-
	 CC-3-4 is intended to accommodate de pedestrian orientation and permits a m dwelling unit for each 1,500 square fee 	evelopment with a naximum <i>density</i> of 1 et of <i>lot</i> area
	 CC-3-5 is intended to accommodate de intensity, pedestrian orientation and pe density of 1 dwelling unit for each 1,50 	evelopment with a high ermits a maximum 00 square fect of <i>lot</i> area
	 CC-3-6 is intended to accommodate de intensity, pedestrian orientation and pe density of 1 dwelling unit for each 1,00 CC-3-7 is intended to accommodate de intensity, pedestrian orientation and pe density of 1 dwelling unit for each 800 CC-3-8 is intended to accommodate de intensity, pedestrian orientation and pe density of 1 dwelling unit for each 600 CC-3-9 is intended to accommodate de intensity, pedestrian orientation and pe density of 1 is intended to accommodate de intensity, pedestrian orientation and pe density of 1 dwelling unit for each 400 	evelopment with a high ermits a maximum 00 square feet of <i>lot</i> area <i>evelopment</i> with a high ermits a maximum 9 square feet of <i>lot</i> area <i>evelopment</i> with a high ermits a maximum 9 square feet of <i>lot</i> area <i>evelopment</i> with a high ermits a maximum 9 square feet of <i>lot</i> area
(4)	The following zones allow heavy commercial uses:	uses and residential
	commercial characteristics and permits 1 <i>dwelling unit</i> for each 1,500 square f	s a maximum <i>density</i> of feet of <i>lot</i> area
	 CC-4-2 is intended to accommodate de intensity, strip commercial characteris maximum density of 1 dwelling unit fo of lot area 	evelopment with high stics and permits a or each 1,500 square feet
Ch. Art. Div.		

LETTER

San Diege Municipal Code (Proposed Lnanges 3-17-2016) Chapter 13: Zone (1-2016) • CC-4-3 is intended to accommodate development with an autorientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area • CC-4-4 is intended to accommodate development with a pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area • CC-4-4 is intended to accommodate development with a pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area • CC-4-5 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area • CC-4-6 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area • CC-4-6 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,000 square feet of lot area • CC-4-6 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,000 square feet of lot area	2 <u>5</u>) f
 CC-4-3 is intended to accommodate development with an autorientation and permits a maximum density of 1 dwelling unifor each 1,500 square feet of lot area CC-4-4 is intended to accommodate development with a pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area CC-4-5 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area CC-4-6 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,500 square feet of lot area CC-4-6 is intended to accommodate development with a high intensity, pedestrian orientation and permits a maximum density of 1 dwelling unit for each 1,000 square feet of lot area 	
 CC-4-4 is intended to accommodate <i>development</i> with a pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area CC-4-5 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area CC-4-6 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area CC-4-6 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,000 square feet of <i>lot</i> area (5) The following zones allow a mix of heavy commercial and limited 	
 CC-4-5 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area CC-4-6 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,000 square feet of <i>lot</i> area (5) The following zones allow a mix of heavy commercial and limited 	
 CC-4-6 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,000 square feet of <i>lot</i> area (5) The following zones allow a mix of heavy commercial and limited 	
(5) The following zones allow a mix of heavy commercial and limited	
industrial uses and residential uses:	
 CC-5-1 is intended to accommodate <i>development</i> with strip commercial characteristics and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area 	
 CC-5-2 is intended to accommodate <i>development</i> with high intensity, strip commercial characteristics and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area 	
 CC-5-3 is intended to accommodate development with an auto orientation and permits a maximum density of 1 dwelling uni for each 1,500 square feet of lot area 	ř.
 CC-5-4 is intended to accommodate <i>development</i> with a pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area 	
 CC-5-5 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,500 square feet of <i>lot</i> area 	
 CC-5-6 is intended to accommodate <i>development</i> with a high intensity, pedestrian orientation and permits a maximum <i>density</i> of 1 <i>dwelling unit</i> for each 1,000 square feet of <i>lot</i> area 	
(Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.) (Amended 7-10-2015 by O-20512 N.S.; effective 8-9-2015.)	
Ch. Art. Div. 13 1 5	7

LETTER



n Diego Mu 2016)	nicipal Coo	de (Proposed Changes 3-17-2016)	Chapter 13: Zones
§131.0520	Use Re	egulations of Commercial Zones	
	The reg in Tabl	gulations of Section 131.0522 apply in the con le 131-05B.	nmercial zones where indicated
	(a)	The uses permitted in any commercial zone m following:	nay be further limited by the
		 Section 131.0540 (Maximum permitter residential regulations); 	ed residential <i>density</i> and other
		(2) Use limitations applicable to the Airpo Overlay Zone (Chapter 13, Article 2,	ort Land Use Compatibility Division 15); -
		 The presence of <i>environmentally sens</i>. 14, Article 3, Division 1 (Environmer Regulations); or 	<i>itive lands</i> , pursuant to Chapter ntally Sensitive Lands
		(4) Any other applicable provision of the	San Diego Municipal Code.
	(b)	Within the commercial zones, no <i>structure</i> or thereof, shall be constructed, established, or a used or maintained except for one or more of in Table 131-05B. It is unlawful to establish, for any purpose or activity not listed in this so	improvement, or portion altered, nor shall any <i>premises</i> be the purposes or activities listed maintain, or use any <i>premises</i> ection or Section 131.0522.
	(c)	All uses or activities permitted in the commer entirely within an enclosed building unless th conducted outdoors.	rcial zones shall be conducted he use or activity is traditionally
	(d)	Accessory uses in the commercial zones may Section 131.0125.	be permitted in accordance with
	(e)	Temporary uses may be permitted in the com period of time with a Temporary Use Permit Chapter 12, Article 3, Division 4.	mercial zones for a limited in accordance with
	(f)	For any use that cannot be readily classified, determine the appropriate use category and us Section 131.0110.	the City Manager shall se subcategory pursuant to
	(Addea (Amena (Amena	d 12-9-1997 by O-18451 N.S.; effective 1-1-20 ded 10-25-2011 by O-20047 N.S.; effective 1- ded 6-18-2013 by O-20261 N.S.; effective 7-19	100.) 1-2012.) 9-2013.) <u>Ch. An. Div.</u> 13 1 5 9
	(Addea (Amena (Amena	d 12-9-1997 by O-18451 N.S.; effective 1-1-20 ded 10-25-2011 by O-20047 N.S.; effective 1- ded 6-18-2013 by O-20261 N.S.; effective 7-19	100.) 1-2012.) 9-2013.) [13 1 5 9

RESPONSE

LETTER



§131.0522 Use Regulations Table for Commercial Zones

The uses allowed in the commercial zones are shown in Table 131-05B.

Legend for Table 131-05B

Symbol in Table 131-05B	Description of Symbol
Р	Use or use category is permitted. Regulations pertaining to a specific use may be referenced.
L	Use is permitted with limitations, which may include location limitations or the requirement for a use or <i>development permit</i> . Regulations are located in Chapter 14, Article 1 (Separately Regulated Use Regulations).
N	Neighborhood Use Permit Required. Regulations are located in Chapter 14, Article 1 (Separately Regulated Use Regulations).
С	Conditional Use Permit Required. Regulations are located in Chapter 14, Article 1 (Separately Regulated Use Regulations).
-	Use or use category is not permitted.



LETTER

	Lone				Lone	s			
[See Section 131 0112 for an	Designator		-		1			-	-
explanation and descriptions of	15t & 21tu >>	CN(1)-	C	R-		CO-		CV-	CP-
the Use Categories,	3rd >>	1-	1-	2-	1-	2-	3-	1-	1-
Subcategories, and Separately Regulated Uses]	4th >>	12345	1	1	1 2	1 2	1 2	1 2	r
Open Space			-	-		-	1	1	r
Active Recreation			-	-	-	-			-
Passive Recreation		-	-	-	-		-	-	-
Natural Resources Preservation			-	-	-		-	-	-
Park Maintenance Facilities		.+	-	1.0		-	-	-	-
Agriculture			T	-	1	-	1	1	1
Agricultural Processing			-	-	-	-	-	-	-
Dairies			-	-	-		-	-	
Horticulture Nurseries & Green	houses	-	-	-		-	-	-	-
Raising & Harvesting of Crops	nouses		1	-	1.5	-	-	-	-
Raising, Maintaining & Keeping Animals	g of		1		-	Ľ.	-	4	2
Separately Regulated Agriculture	Jses						-	-	
Agricultural Equipment Repair	Shops	-	P	P	- 1	-	1.4	-	1.4
Commercial Stables		μ.	-6	-	5		~	L	-
Community Gardens		L	L	L	L	L		L	L
Equestrian Show & Exhibition	Facilities	-	1	-	1.4	-	-	C	-
Open Air Markets for the Sale of Agriculture-related Products &	of Flowers	-	-	-	4	-	-	•	-
Residential			-	_		-			
Mobilehome Parks		-	-	1941	1.1	-	1.5	1	 *
Multiple Dwelling Units		P(2)	P(2)	-	P(2)	-	P(2)	P(2)	
Rooming House [See Section 131.0112(a)(3)(A)]		P(2)	P	1	P	2	P	P(2)	15
Shopkeeper Units		P ⁽²⁾	P(2)		P ⁽²⁾	-	P(2)	P ⁽²⁾	$T \sim 1$
Single Dwelling Units		÷	- 2-1	4	1.5	- 1	-	-	-
Separately Regulated Residential U	Ises			_				-	-
Boarder & Lodger Accommoda	ations	L ⁽²⁾	L	-	L	- 2-	1	L(2)	-
Companion Units		2.1	- (=)	\sim	1-14	(- G) (1.14	1.4	-
Employee Housing:			-		-	-	-	1	1
6 or Fewer Employees			1	-		-		-	~

RESPONSE

Categories/Subcategories	Zone Designator	Ú. —			Zone	s			
[See Section 131.0112 for an explanation and descriptions of	1st & 2nd	CN(1)-	C	R-		CO-		CV-	CP-
the Use Categories,	3rd >>	1-	1-	2-	1-	2-	3-	1-	1-
Subcategories, and Separately Regulated Uses]	4th >>	12345	1	1	1 2	1 2	12	1 2	1
12 or Fewer Employees		10 40		-	247			1.21	- e . 1
Greater than 12 Employees		×	-	-	1.4	-	-		-
Fraternities, Sororities and Stud Dormitories	ent	C ⁽²⁾	c		С	1	-	C ⁽²⁾	-
Garage, Yard, & Estate Sales				-	- A	$= \mathbf{G}_{\mathbf{r}} =$	1.4	- P	-
Guest Quarters			-	14.	-	1.42	1 9	-	-
Home Occupations		L	L	1.47	L		L	L	-
Housing for Senior Citizens		C ⁽²⁾	C	-	С	-	C	C ⁽²⁾	14
Live/Work Quarters		1 4	L	1.40	1.4	4	L	-	\sim
Residential Care Facilities:									
6 or Fewer Persons		P ⁽²⁾	P	4	Р		P	P ⁽²⁾	
7 or More Persons		C ⁽²⁾	C	-	С		C	C ⁽²⁾	1.4
Transitional Housing:		1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	_		-				
6 or Fewer Persons		P ⁽²⁾	P		Р	1.	P	P ⁽²⁾	
7 or More Persons		C ⁽²⁾	C	1.4	С	$<22^{\prime},$	C	C ⁽²⁾	
Watchkeeper Quarters				L	1.16	- (H)	-	-	- -
Institutional			_	2	_				
Separately Regulated Institutiona	Uses					_			
Airports			C	C	C	C	1.2	C(10)	-
Botanical Gardens & Arboretur	ns	-	P	P	С	C	-	P	-
Cemeteries, Mausoleums, Crem	atories		C	C	C	C		C(10)	-
Correctional Placement Centers			C	C	C	C	-	C(10)	
Educational Facilities:			1						
Kindergarten through Grade	12	C (10)	C	C	С	C	C	C(10)	E.
Colleges / Universities		1.1.4	C	C	С	C	C	C(10)	
Vocational / Trade School		-	P	P	P	P	C	-	-
Energy Generation & Distribution	acilities	C ⁽¹⁰⁾	P	C	Р	P	P	P(10)	-
Exhibit Halls & Convention Facilit	ies	12.410.1	P	P	C	C	1.00	P	2-5
Flood Control Facilities		L	L	L	L	L	L	L	
Historical Buildings Used for Purp Otherwise Allowed	oses Not	C ⁽¹⁰⁾	C	C	C	C	C	C ⁽¹⁰⁾	1



1

RESPONSE

(1.5010)											
Jse Categories/Subcategories	Zone Designator			Zones							
[See Section 131.0112 for an	1st & 2nd	(1)				00		CV-	cm		
the Use Categories	2.122	CNO-	1	K-	1	20-	1 2		1- 1-		
Subcategories, and Separately	Srd >	1-	1-	4-	1-	1 2-	3-	1-			
Regulated Uses]	4th >>	2 3 4 3	1.0	1	1 2	1 2	12	1 2	- 1		
Homeless Facilities:				-							
Congregate Meal Facilities		C(10)	C	-	С	C	C	C ⁽¹⁰⁾	(h)		
Emergency Shelters		C(10)	C	-	С	C	C	C(10)			
Homeless Day Centers	C ⁽¹⁰⁾	C	-	С	C	C	C(10)	-			
Hospitals, Intermediate Care Faci Nursing Facilities	ities &	~	P	Р	C	C	C	P ⁽¹⁰⁾	-		
Interpretive Centers		1 (n.e.)		÷.	1	1.1	1	1.00	-		
Museums	Â.		P	P	С	C	C	Р	-		
Major Transmission, Relay, or Communications Switching Static		C	C	C	С	C	C ⁽¹⁰⁾	-			
Satellite Antennas		L	L	L	L	L	L	L	L		
Social Service Institutions	-	C	C	С	С	C	C(10)				
Solar Energy Systems	L	L	L	L	L	L	L	L			
Wireless Communication Facility			1	0.11			1.1.1		- 21		
Wireless communication facili public right-of-way with subte equipment adjacent to a non-r	L	L	L	L	L	L	L	L			
Wireless communication facil public right-of-way with sub equipment adjacent to a reside	N	N	N	N	N	N	N	N			
Wireless communication facil public right-of-way with above equipment	Ç	C	С	С	С	C	C	С			
Wireless communication facili the public right-of-way	L	L	L	Ľ	L	L	L	L			
Retail Sales	1		-								
Building Supplies & Equipmen	P(11)	P(11)	P(11)		н.	-	140	-			
Food, Beverages and Grocerie	P(11)	P(11)	$P^{(11)}$	P ⁽¹¹⁾	P(11)	P(11)	P(11)	÷			
Consumer Goods, Furniture, A Equipment	P(11)	P(11)	P ⁽¹¹⁾	P ^(3,11)	P ^(3,11)	P(3,11)	P(13)	Å			
Pets & Pet Supplies	· · · · · · · · · · · · · · · · · · ·	P(11)	P(11)	P(11)		-	-	÷.	-		
Sundries, Pharmaceutical, & O Sales	onvenience	P ⁽¹¹⁾	P(11)	P ⁽¹¹⁾	P ⁽¹¹⁾	P ⁽¹¹⁾	P ⁽¹¹⁾	P(11)	16		
Wearing Apparel & Accessorie	28	P ⁽¹¹⁾	P(11)	P(11)	-	1.61	114	P(11)	÷		
			_	_							

Ch. Art. Div.

RESPONSE

lse Categories/Subcategories	Zone Designator				Zone	s			
[See Section 131.0112 for an explanation and descriptions of	1st & 2nd >>	CN ⁽¹⁾ -	C	R-	1	CO-		CV-	CP-
the Use Categories, Subcategories, and Separately Regulated Uses]	3rd >> 4th >>	1- 1 2 3 4 5	1-	2-	1- 1 2	2-	3-	1- 1 2	<u>1-</u> 1
Agriculture Related Supplies	& Equipment		P	Р	-	-	-	-	
Alcoholic Beverage Outlets		L	L	L	L	L	L	L	1.4
Farmers' Markets					-				
Weekly Farmers' Markets		L	L	L	L	L	L	L	L
Daily Farmers' Market Sta	unds	L	L	L	L	L	L	L	-
Plant Nurseries	-	Р	Р	Р	-	0-31		4	3-34
Retail Farms		L	L	L	L	L	L	-	thèn:
Retail Tasting Stores		L	L	L	L	L	L	L	147
Swap Meets & Other Large O Facilities	utdoor Retail	1.1	С	С	1		-	C(10)	120
Commercial Services									
Building Services		1.187-1	P	Р	P(6)	P(6)	P(6)	1.61	
Business Support		Р	Р	P	P(5)	P ⁽⁷⁾	P(7)		-
Eating & Drinking Establishm	ents	P ⁽⁴⁾⁽¹⁶⁾	P(16)	P ⁽¹⁶⁾	P(5,16)	P(5,16)	P(5,16)	P ⁽¹⁶⁾	-
Financial Institutions		Р	Р	P	Р	P	P	1.	
Funeral & Mortuary Services	-		P	Р	1 - Q	(A)	4	$m \geq 0$	1.30
Instructional Studios		Р	P	Р	Р	P	P	P(12)	1.00
Maintenance & Repair		Р	P	P	P(6)	P(6)	P(6)		-
Off-site Services		-	P	P	->	3.4	-	-	-
Personal Services	- it	Р	P	Р	1	1991	-	P	
Radio & Television Studios			P	P		140	0.41	-	4
Tasting Rooms	1	· · · · · · · · · · · · · · · · · · ·	(m)	1.98	- e'	-	-		-
Visitor Accommodations	1	P	P	P	1.0		-	P	19
Separately Regulated Commercia Jses	Il Services								
Adult Entertainment Establishmen	ts:								
Adult Book Store		L	L	L	÷		-	1	-
Adult Cabaret	1		L	L	- <u>F</u>			L	. e.
Adult Drive-In Theater			L	L		-	-	L	-
Adult Mini-Motion Picture Th	neater		L	L	1.1	-	-	L	-
Adult Model Studio		L	L	L			-	L	12
Adult Motel			L	L	12-1	-	-	L	
Adult Motion Picture Theater	s		L	L			-	L	
Adult Peep Show Theater	1		L	L	1.14	4	-	L	
Adult Theater	1	-	L	L	-	-	-	L	1.4

Ch. Art. Div.

RESPONSE

Jse Categories/Subcategories	Zone Designator				Zon	les			
[See Section 131.0112 for an	1st & 2nd	CN(I)	0	n	1	00		CN	cn
the Use Categories.	3rd >>		1.	2-	I.	2-	3-	LV-	L-
Subcategories, and Separately	Jugas	12345	1	1	12	12	112	12	1
Regulated Uses]	4th >>				1 -			1	
Body Painting Studio		L	L	L	1040	-		L	
Massage Establishment		L	L	L			-	1.6	1.4
Sexual Encounter Establishme	nt	L	L	L	104.0	1.14	(AL	L	1.1417
Assembly and Entertainment Uses Places of Religious Assembly	, Including	L ⁽¹⁰⁾	L	L	L	L	L	L ⁽¹⁰⁾	-
Bed & Breakfast Establishments:					-				
1-2 Guest Rooms		1	P	Р	-	1941	1	P	E.
3-5 Guest Rooms			Р	Р		.н.	-	P	- 24
6+ Guest Rooms			P	P	104	L×1	~	P	
Boarding Kennels/Pet Day Care		L	L	L	N	N	-	N ⁽¹⁰⁾	· •
Camping Parks			C	C	C	C	-	C	× 1
Child Care Facilities:		1			-	_			
Child Care Centers		L	L	-	L	L	L	L(10)	7.001
Large Family Child Care Hom	es	L	L	-	L	L	L	L(10)	
Small Family Child Care Hom	es	L	L	1.40	L	L	L	L	- Y.
Eating and Drinking Establishmen Drive-in or Drive-through Compos	ts with a nent	- C		P	Р	Р	-	P -	
Fairgrounds		-	C	C	1.1.4	114-1	1.4	C	
Golf Courses, Driving Ranges, an Putt Courses	d Pitch &	1	C	C	C	C		C	
Helicopter Landing Facilities		$\mathbf{r} = \mathbf{L}_{\mathbf{r}} =$	C	C	C	C	C	C(10)	*
Massage Establishments, Speciali	zed Practice	L	L	L	~	1.4	()	L(14)	×
Medical Marijuana Consumer Co	operatives			C		1.41	$\{ \langle \cdot \rangle_{i} \}$	1	
Mobile Food Trucks		L(15)	L ⁽¹⁵⁾	L(15)	L ⁽¹⁵⁾				
Nightclubs & Bars Over 5,000 Sq in Size	uare Feet	1 - 5	С	C	C	C	C	C	
Parking Facilities as a Primary Us	ie:	11				-			
Permanent Parking Facilities		-	P	Р	C	C		C	Р
Temporary Parking Facilities		11	N	N	- C -	C	C	C	N
Private Clubs, Lodges and Fraterna	al	P ⁽¹⁰⁾	P	P	P	Р	P	P(10)	-
Organizations				-	-		-	1	
Privately Operated, Outdoor Recre Facilities over 40,000 Square Feet	ation in Size ⁽⁹⁾	÷	P	P	C	C	r	C	1.0
Pushcarts:	C						-	-	
Pushcarts on Private Property		L	L	L	L	L	L	L	
Pushcarts in Public Right-of-M	ay	N	N	N	N	N	N	N	-

RESPONSE

se Categories/Subcategories	Zone				Zone	es			
[See Section 131.0112 for an	1st & 2nd	(D)		n.	1	00		au	m
the Use Categories	2rd~	1	1	R-	1.	2.	3.	UV-	1-
Subcategories, and Separately	Sid 2	12345	1	1	112	112	112	112	1-
Regulated Uses]	4th >>				1 2			12	1
Recycling Facilities:			-	_					
Large Collection Facility		N	N	N	N	N	N	N ⁽¹⁰⁾	-
Small Collection Facility		L	L	L	L	L	L	L(10)	- 3-8
Large Construction & Demoli Recycling Facility	ition Debris	1.5	1	1	1	-	-	1	1
Small Construction & Demoli Recycling Facility	ition Debris	1.282	-	1	1		1		
Drop-off Facility		L	L	L	L	L	L	L	14
Green Materials Composting	Facility	III ARECTED	-	1.00			1.4	-	-
Mixed Organic Composting F	acility	1	-	-		1.5	-	-	17.
Large Processing Facility Acc Least 98% of Total Annual W Recyclables from Commercia Traffic	cepting at /eight of l & Industrial		-	1		Ì			
Large Processing Facility Acc Types of Traffic	cepting All	1.24	1	1	1		-	2	1
Small Processing Facility Acc Least 98% of Total Annual W Recyclables From Commercia Industrial Traffic	cepting at reight of al &		Ĩ	Ĩ	-			•	
Small Processing Facility Acc	cepting All	2	12	1	1	5	-		-
Reverse Vending Machines		T	T	T	T	T	L	T	1.1
Tire Processing Facility		L.	1 D	L	<u>ь</u>	12	10	14	
Sidewalk Cafes		L	L	L	L	L	L	L	-
Sports Arenas & Stadiums	a (C.a. (C.))	-	C	C	C	C	-	C	-
Theaters that are Outdoor or Over Square Feet in Size	5,000		C	C	С	C	-	C	-
Urgent Care Facilities		N	N	N	N	N	N	N(10)	-
Veterinary Clinics & Animal Hos	pitals	L	L	L	N	N	N	-	(A)
Zoological Parks		-	1-	1.00	1.2	-	1.2	1.2	-
Offices				-	-			1.	
Business & Professional		P(7)	P	P	Р	P	P	-	1 1-1
Government		Р	P	P	Р	Р	P	-	-
Medical, Dental & Health Pract	titioner	Р	P	P	Р	P	P	P(10)	-
Regional & Corporate Headqua	irters	Р	P	P	- P	P	P	-	14

RESPONSE

Use Categories/Subcategories	Zone				Zone	s			
See Section 131.0112 for an	1st & 2nd		T		<u> </u>	-			
explanation and descriptions of	>>	CN(1)-	C	R-	_	CO-		CV-	CP-
the Use Categories,	3rd >>	1-	1-	2-	1-	2-	3-	1-	1-
Regulated Uses]	4th >>>	1 2 3 4 5	1	1	1 2	1 2	12	1 2	1
Separately Regulated Office Uses			1.			1.7	1.		
Real Estate Sales Offices & Ma	odel Homes	L	L	-	L	L	L	L	17.1
Sex Offender Treatment & Cou	inseling	L	L	L	L	4	L	Lim	2-2
Vehicle & Vehicular Equipment S Service	ales &								
Commercial Vehicle Repair &			P	Р		1.55		17	10
Commoniel Vakiele Sales & Der	tala	-	D	D			-		
Commercial Vehicle Dangin & Main	tans	-	P P	D		-			-
Personal Vehicle Seles & Dontok	tenance	-	D	D	-	-	-	-	-
Vehicle Equipment & Supplies S Rentals	ales &	- e'	P	P	-	1.5		1	1.20
Separately Regulated Vehicle & V	ehicular		-				-		
Automobile Service Stations			IN	IN	C	C	IC	C	-
Outdoor Storage & Display of New Unregistered Motor Vehicles as a	v, Primary		C	C	-	-	-		-
Distribution and Storage		-	-	1	-	-			
Fauinment & Materials Storage	Varde		1	P			1.		
Moving & Storage Facilities	Tarus		-	P	-	1.5	1	-	-
Distribution Facilities		-	-	D(8)	-	- a2	-	-	-
Separately Regulated Distribution	and		-	14			4		
Impound Storage Yards		1.20	15	C	-	-	1 -	-	
Junk Yards			1	-	2	1,21	1.4	-	
Temporary Construction Storage Y Located Off-site	ards	L	L	L	Ľ.	L	L	L	
Industrial									1
Heavy Manufacturing		4	-	1.8.0		147	1.4	-	-
Light Manufacturing			-	P ⁽⁸⁾	Ξ	-	-	-	-
Marine Industry			-	-	-	-			Dell.
Research & Development	1.45.1	7	P	P	Р	Р	P		
Trucking & Transportation Te	rminals	-	P	P	1	-	-	-	-
Separately Regulated Industrial U	ses								
Artisan Food and Beverage Produ	cer	N	=	3	- 5-	3	- e	1	12
Hazardous Waste Research Facilit	у		-	-	1.1.2.11	1.41	-	1.4	1.611

LETTER

RESPONSE

Use Categories/Subcategories	Zone				Zone	5			
[See Section 131.0112 for an explanation and descriptions of	1st & 2nd	CN ⁽¹⁾ -	C	R-		CO-		CV-	CP-
the Use Categories,	3rd >>	1-	1-	2-	1-	2-	3-	1-	1-
Subcategories, and Separately Regulated Uses]	4th >> 1	2 3 4 5	1	1	1 2	1 2	1 2	1 2	1
Hazardous Waste Treatment Faci	lity	-	14	-	-		1.	-	-
Marine Related Uses Within the O Overlay Zone	Coastal		С	C	С	L	2	C	7
Mining and Extractive Industries		-	5-1	-		1.04	167	-	.41
Newspaper Publishing Plants			С	C	C	С	12	C(10)	1
Processing & Packaging of Plant & Animal By-products Grown Of premises	Products f-		1	-	-	2	e.		0
Very Heavy Industrial Uses			-		2	1.0-3	-	-	-
Wrecking & Dismantling of Moto Vehicles	or	- 5	1		15	1	-	-	÷
Signs						-	1.1		
Allowable Signs	1	Р	Р	Р	Р	P	P	Р	Р
Separately Regulated Signs Uses									
Community Entry Signs		L	L	L	L	L	L	L	L
Neighborhood Identification Sign	S	A.	- A-	-	1.04/	154	1.4.1	-	(a)
Comprehensive Sign Program		N	Ν	N	Ν	N	N	N	N
Revolving Projecting Signs		N	Ν	N	N	N	N	N	N
Signs with Automatic Changing C	Copy	N	N	N	N	N	N	N	N
Theater Marauges			N	N	-	1.1-1	1.4	N	3-3



1

se Categories/Subcategories	Zone		Zones							
[See Section 131.0112 for an	Designator		66							
explanation and descriptions of	1st & 2nd >>	1		CC-						
the Use Categories,	3rd >>>	1-	2-	3-	4-	5-				
Regulated Lines	4th >>	123	12345	456789	123456	123456				
Regulated Oses		11								
Open Space		10.00		1						
Active Recreation		-	-		-	-				
Passive Recreation				-	-	-				
Natural Resources Preservation	on		1. 1.	1.00	1 (190 - I	1.08.0				
Park Maintenance Facilities			-	-		1.1.4				
griculture										
Agricultural Processing			1.0	-						
Aquaculture Facilities				10	1-1					
Dairies				.:	1.401					
Horticulture Nurseries & Gree	enhouses	-	-	1. 340	1,0+2,1	1.040				
Raising & Harvesting of Crop	S	1.41	1.00.1	1.0		1.14				
Raising, Maintaining & Keepi	ng of Animals	1.200	1.141.1		11,0-11					
eparately Regulated Agriculture	Uses									
Agricultural Equipment Repair S	hops	Р	Р	1.1.1	Р	Р				
Commercial Stables		1.00				1. S.C				
Community Gardens		L	L	L	L	L				
Equestrian Show & Exhibition F	acilities	1.1.4.10	-1	1.00	t dia 11	1.1.4				
Open Air Markets for the Sale of Related Products & Flowers	f Agriculture-			1.14		4				
lesidential		1.0								
Mobilehome Parks		-	-		1.4					
Multiple Dwelling Units		P(2)		P(2)	P ⁽²⁾	P(2)				
Rooming House [See Section 131.0112(a)(3)(A)]		Р	4	Р	Р	Р				
Shonkeeper Units		P	-	·P	р	Р				
Single Dwelling Units		-	1.401	1.040	1.14.15	C. Same				
separately Regulated Residential	Uses									
Boarder & Lodger Accommodat	tions	L	-	L	L	L				
Companion Units	1.1		10.411	11.1.2	1.42.11	1.14				
Employee Housing:		1								
6 or Fewer Employees		14.	1.4.1	1.1.1	-	-				
12 or Fewer Employees		1.0	-		-	L. Lat				
Greater than 12 Employees	5	1	1-1-1	-	2	A				
Fraternities, Sororities and Stude	nt Dormitories	С	1	C	С	С				
Garage Vard & Estate Sales										



RESPONSE

Jse Categories/Subcategories [See Section 131.0112 for an Desig	Zone gnator			Zon	ies	
explanation and descriptions of 1st & 2	nd >>		_	CC	k	
the Use Categories,	3rd >>	1-	2-	3-	4-	5-
Regulated Uses]	4th >> 1	231	2345	456789	9123456	5123456
Guest Quarters		-			S	1
Home Occupations		L	-	L	L	L
Housing for Senior Citizens		C	-	С	C	C
Live/Work Quarters		L	-	L	L	L
Residential Care Facilities:						2 7 7 11
6 or Fewer Persons	- 34	P	-	Р	Р	Р
7 or More Persons	- · · · · ·	C	-	C	C	C
Transitional Housing:		-				
6 or Fewer Persons		P		Р	Р	Р
7 or More Persons		C	+	С	C	C
Watchkeeper Quarters		-	L		-	-
Institutional	1					
Separately Regulated Institutional Uses			_			_
Airports		C	С	С	С	С
Botanical Gardens & Arboretums		C	С	С	C	C
Cemeteries, Mausoleums, Crematories	- <u>1</u> ,5	C	С	С	C	C
Correctional Placement Centers		C	С	С	C	C
Educational Facilities:						
Kindergarten through Grade 12		C	С	C	C	C
Colleges / Universities		C	С		C	C
Vocational / Trade School		P	Р		Р	P
Energy Generation & Distribution Faciliti	es	P	С	С	C	P
Exhibit Halls & Convention Facilities	1.11	C	С	C	C	C
Flood Control Facilities		L	L	L	L	L
Historical Buildings Used for Purposes N Otherwise Allowed	ot	C	С	С	С	C
Homeless Facilities:						
Congregate Meal Facilities		C	-	С	C	C
Emergency Shelters		C	-	С	C	C
Homeless Day Centers	-	C	-	C	C	C
Hospitals, Intermediate Care Facilities & Nursing Facilities		С	С	C	C	C
Interpretive Centers		-				~
Museums		C	С	P	C	C
Major Transmission, Relay, or Communications Switching Stations		С	С	Ĉ	C	C
Satellite Antennas		L	L	L	L	L. L
h Art Din		~	-	0	1 4	

RESPONSE

Use Categories/Subcategories Z [See Section 131.0112 for an Design	Lone	-	Zon	es	⁻
explanation and descriptions of 1st & 2n	d >>		CC		
the Use Categories, 3r	d>> 1-	2-	3-	4-	5-
Subcategories, and Separately Regulated Uses] 4t	h >> 1 2 2	31234	5456789	123456	12345
Social Service Institutions	C	C	C	C	C
Solar Energy Systems	L	L	L	L	L
Wireless Communication Facility:					-
Wireless communication facility in the public right-of-way with subterranean equipment adjacent to a non-residential	L	L	L.	L	L
Wireless communication facility in the public right-of-way with subterranear equipment adjacent to a residential use	n N	N	N	N	N
Wireless communication facility in the public right-of-way with above ground equipment	C C	С	C	С	С
Wireless communication facility outside the public right-of-way	e L	L	L.	L	L
Retail Sales	111 14 14				
Building Supplies & Equipment	P(11)	P(11)	1	P(11)	P(11)
Food, Beverages and Groceries	P(11)	P ⁽¹¹⁾	P ⁽¹¹⁾	P ⁽¹¹⁾	P(11)
Consumer Goods, Furniture, Appliance Equipment	s, P ⁽¹¹⁾	P ⁽¹¹⁾	P ⁽¹¹⁾	P ⁽¹¹⁾	P (11)
Pets & Pet Supplies	P(11)	P(11)	P ⁽¹¹⁾	P(11)	P(11)
Sundries, Pharmaceutical, & Convenien Sales	ce P ⁽¹¹⁾	P ⁽¹¹⁾	P(11)	P ⁽¹¹⁾	P(11)
Wearing Apparel & Accessories	P(11)	P(11)	P(11)	P(11)	P(11)
Separately Regulated Retail Sales Uses	1.5	-			
Agriculture Related Supplies & Equipme	nt -	÷	-	Р	P
Alcoholic Beverage Outlets	L	L	L	L	L.
Farmers' Markets					
Weekly Farmers' Markets	L	L	L	L	L
Daily Farmers* Market Stands	L	L	L	L	L
Plant Nurseries	P	Р	Р	Р	Р
Retail Farms	L	L	L	L	L
Retail Tasting Rooms	L	L	L	L	L
Swap Meets & Other Large Outdoor Ret Facilities	ail -	5		1	C
Commercial Services					
Building Services	1. 1.	1.1	Page 2.	Р	P
Business Support	P	Р	P	Р	P

 Ch. Art. Div.

 13
 1
 5
 21

ONSE

Categories/Subcategories	Zone		-	Zone	s	
ee Section 131.0112 for an	Designator					
planation and descriptions of	1st & 2nd >>	1	_	CC-		
e Use Categories,	3rd >>	1-	2-	3-	4-	5-
ibcategories, and Separately egulated Uses]	4th >>	1 2 3	12345	456789	123456	12345
Eating & Drinking Establishm	ents	P ⁽¹⁶⁾	P(16)	P(16)	P(16)	P(16)
Financial Institutions		Р	Р	Р	Р	P
Funeral & Mortuary Services		Р	Р	Р	Р	Р
Instructional Studios		Р	Р	Р	Р	Р
Maintenance & Repair		Р	Р	Р	Р	Р
Off-site Services					Р	Р
Personal Services		Р	Р	Р	Р	Р
Radio & Television Studios		P	Р	Р	Р	Р
Tasting Rooms		_(17).	_(17)	_(17)	_(17)	P ⁽¹⁷⁾
Visitor Accommodations		P	P	Р	Р	Р
Separately Regulated Commer Uses	rcial Services	2				
dult Entertainment Establishmen	ts:					
Adult Book Store		L	L	L	L	L
Adult Cabaret		L	L	L	L	L
Adult Drive-In Theater	Drive-In Theater			L	L	L
Adult Mini-Motion Picture	Mini-Motion Picture Theater			L	L	L
Adult Model Studio	ult Model Studio			L	L	L
Adult Motel		L	L	L	L	L
Adult Motion Picture Theate	er	L	L	L	L	L
Adult Peep Show Theater		L	L	L	L	L
Adult Theater		L	L	L	L	L
Body Painting Studio		L	L	L	L	L
Massage Establishment		L	L	(m)	1911	L
Sexual Encounter Establishr	nent	L	L	L	L	L
ssembly and Entertainment Uses aces of Religious Assembly	nd Entertainment Uses, Including eligious Assembly			Ľ	L	L
ed & Breakfast Establishments:		1				
1-2 Guest Rooms	1-2 Guest Rooms			Р	Р	Р
3-5 Guest Rooms		Р	Р	Р	Р	Р
6+ Guest Rooms		P	Р	P	Р	Р
oarding Kennels/Pet Day Care		L	L	L	\mathbf{L}	L
amping Parks		С	С	С	С	С
hild Care Facilities:						
hild Care Facilities:	oing Parks I Care Facilities: Child Care Centers				L	

Ch. Art. Div.

RESPONSE

	1					
Use Categories/Subcategories [See Section 131.0112 for an	Zone Designator	-		Zone	s	
explanation and descriptions of	1st & 2nd >>			CC-	1	1.11.1
the Use Categories,	3rd >>	1-	2-	3-	4-	5-
Subcategories, and Separately Regulated Uses]	4th >>	1 2 3	12345	456789	123456	12345
Small Family Child Care Hon	nes	L	-	L	L	L
Eating and Drinking Establishme Drive-in or Drive-through Comp	ents with a onent	Р	Р	<u>C</u>	Р	Р
Fairgrounds		C	С		C	С
Golf Courses, Driving Ranges, an Courses	d Pitch & Putt	С	С	С	С	С
Helicopter Landing Facilities		C	C	С	C	С
Massage Establishments, Speciali	zed Practice	L	L	L	L	Ĺ
Medical Marijuana Consumer Co	operatives		С	-	Delle 1	
Mobile Food Trucks		L(15)	L(15)	L(15)	L(15)	L(15)
Nightclubs & Bars Over 5,000 Sq Size	uare Feet in	С	С	C	С	C
Parking Facilities as a Primary U.	se:	1211		100		-
Permanent Parking Facilities		P	С	Р	P	Р
Temporary Parking Facilities		N	С	N	N	N
Private Clubs, Lodges and Fraterr Organizations	nal	Р	Р	Р	Р	Р
Privately Operated, Outdoor Recr Facilities Over 40,000 Square Fee	eation et in Size ⁽⁹⁾	C	C	Ċ	C	C
Pushcarts:						1.0
Pushcarts on Private Property	U 1 1 1 1 1 1 1	L	L	L	L	L
Pushcarts in Public Right-of-1	Way	N	N	N	N	N
Recycling Facilities:						
Large Collection Facility		N	N	N	N	N
Small Collection Facility		L	\mathbf{L}	L	L	L
Large Construction & Demoli Recycling Facility	ition Debris	-			7	1-2-
Small Construction & Demol Recycling Facility	ition Debris				7	1.7
Drop-off Facility		L	L	L	L	L
Green Materials Composting	Facility					1.1.42
Mixed Organic Composting F	acility	1 mil 1				

Ch. Art. Div.

 13
 1
 5
 23

LETTER

Use Categories/Subcategories	Zone			Zone	25	
[See Section 131.0112 for an	Designator			00		
the Lies Cotegories	1st & 2nd >>		2	- CC.		-
Subcategories and Separately	3rd >>	1 22	2-	3-	4-	1221
Regulated Uses]	4th >>>	1 2 3	12345	4 2 6 7 8 9	123430	12345
Large Processing Facility Acc Least 98% of Total Annual W Recyclables from Commercia Industrial Traffic	epting at eight of l &					
Large Processing Facility Acc Types of Traffic	epting All			100	्रेट	1
Small Processing Facility Acc Least 98% of Total Annual W Recyclables From Commercia Industrial Traffic	epting at eight of Il &	1			С	C
Small Processing Facility Acc Types of Traffic	epting All	\sim		10.5	С	С
Reverse Vending Machines	se Vending Machines Processing Facility Cafes				L	L
Tire Processing Facility						- etc.
Sidewalk Cafes					L	L
Sports Arenas & Stadiums	1.	С	С	С	C	С
Theaters that are Outdoor or Over Square Feet in Size	are Outdoor or Over 5,000 a Size			C	С	C
Urgent Care Facilities		N	N	N	N	N
Veterinary Clinics & Animal Hosp	pitals	L	L	L	L	L
Zoological Parks		14	1 (A) (1.1.4	(m)	
Offices		1.1	1.1.1			
Business & Professional		Р	Р	Р	Р	Р
Government		Р	Р	Р	р	Р
Medical, Dental & Health Prac	titioner	Р	Р	Р	Р	Р
Regional & Corporate Headqu	arters	Р	Р	Р	Р	Р
Separately Regulated Office Uses	111-11	1.7				
Real Estate Sales Offices & M	Real Estate Sales Offices & Model Homes			L	L	L
Sex Offender Treatment & Co	unseling	L	L	L,	L	L
Vehicle & Vehicular Equipment S Service	ales &			-		
Commercial Vehicle Repair &	Maintenance	1.1	4	11. j. in	P	P
Commercial Vehicle Sales & R	tentals		1	1	Р	Р
Personal Vehicle Repair & Ma	intenance	Р	Р	1.1.1	Р	Р
Personal Vehicle Sales & Rent	als	Р	Р	14	P	Р

Ch. Art. Div.

LETTER

se Categories/Subcategories	Zone	6 F.		Zone	25		
See Section 131.0112 for an	Designator						
the Lise Cotegories	Ist & 2nd >>	1 1	2			-	
Subcategories and Separately	3rd >>	1-	2-	3-	4-	2-	
Regulated Uses]	4th >>	1 23	2343	4 2 0 / 0 9	123430	123430	
Vehicle Equipment & Supplies S Rentals	Sales &	Р	Р		Р	Р	
eparately Regulated Vehicle & Ve	hicular	-					
quipment Sales & Service Uses							
Automobile Service Stations		N	N	N	N	N	
Outdoor Storage & Display of N	lew.	C	С	-	С	C	
Unregistered Motor Vehicles as	a Primary						
Use			1.1	1 · · · · · · · · · · · · · · · · · · ·		1	
istribution and Storage	~~~~···						
Equipment & Materials Storage	Yards	1.0			1.00		
Moving & Storage Facilities		1.	1.0	1	Р	Р	
Distribution Facilities		-			1.24	P(8)	
eparately Regulated Distribution a ses	and Storage						
Impound Storage Yards				L. T	С	С	
Junk Yards		-	1.901	11 A.		4 (4) 1	
Temporary Construction Storag Located Off-site	e Yards	L	L	L	L	L	
dustrial		-					
Heavy Manufacturing	1.11	-	-		12	200	
Light Manufacturing			-	1	- 54	P(8)	
Marine Industry			-		-	Contraction of the	
Research & Development		P	Р	-	Р	P	
Trucking & Transportation Ter	minals	-				-	
enarately Regulated Industrial Us	PS	-					
Artisan Food and Beverage Pro	ducer	L	L	L	L	L	
Hazardous Waste Research Fac	ility	-	-	1	-		
Hazardous Waste Treatment Fa	cility	-			-		
Marine Related Uses Within the Overlay Zone	Coastal	С	С	C	С	С	
Newspaper Publishing Plants		C	C	C	C	P	
Processing & Packaging of Plan	t Products &	-			-		
Animal By-products Grown Off	-premises						
very Heavy Industrial Uses	tin Maliat	-	-			-	
wrecking & Dismantling of Mc	tor Vehicles				-	-	
igns		-	-				
Allowable Signs		Р	Р	P	Р	Р	

RESPONSE

ATTACHMENT 7

ATTACHMENT I San Diego Municipal Code (Proposed Changes 3-17-2016) Chapter 13: Zones (1-2016)Use Categories/Subcategories Zone Zones [See Section 131.0112 for an Designator 1st & 2nd >> explanation and descriptions of CCthe Use Categories, 3rd >> 1-2-3-4-5-Subcategories, and Separately 23123454 23456123456 4th>Regulated Uses] Community Entry Signs Neighborhood Identification Signs 1.1 -Comprehensive Sign Program N N N N N N N N N Revolving Projecting Signs N N Signs with Automatic Changing Copy N N N N N Theater Marquees N N N N

Footnotes to Table 131-05B

- ¹ Uses shall not begin operating before 6:00 a.m. or continue operating later than 12:00 midnight in Commercial-Neighborhood (CN) zones.
- See Section 131.0540.
- ³ Only office furniture, appliances, and equipment establishments are permitted. The gross floor area occupied by these uses shall not exceed 2,500 square feet for each *premises*.
- ⁴ Live entertainment and the sale of intoxicating beverages other than beer and wine are not permitted in the Commercial-Neighborhood (CN) zones, unless approval of a deviation is granted via a Planned Development Permit in accordance with Section 126.0602(b)(1). <u>Within the North Park Community</u> Plan area, full alcohol sales are permitted in the CN zones.
- ⁵ The sale of alcoholic beverages is not permitted as a primary use.
- ⁶ The gross floor area occupied by these uses shall not exceed 2,500 square feet for each premises.
- 7 Hiring halls are not permitted.
- ⁸ These activities shall be located solely within an enclosed building that does not exceed 7,500 square feet of gross floor area. Activities that would require a permit from the Hazardous Materials Management Division of the County of San Diego or from the San Diego Air Pollution Control District are not permitted.
- ⁹ The 40,000 square feet includes all indoor and outdoor areas that are devoted to the recreational use; it does not include customer parking areas.
- ¹⁰ This use is not allowed within the Coastal Overlay Zone, except that assembly and entertainment uses may be incorporated as an *accessory use* to visitor accommodations.
- Development of a large retail establishment is subject to Section 143.0302.
- ¹² Within the Coastal Overlay Zone, instructional studios are not permitted on the ground *floor* in the CV-1-1 or CV-1-2 zone.
- ¹³ Permitted in CV zones where the gross floor area occupied by an individual retail sales establishment would not exceed 2,500 square feet.
- ¹⁴ Specialized practice massage establishments are permitted only as an accessory use in the CV-1-1 and CV-1-2 zones.
- ¹⁵ This use is permitted as a limited use subject to a mobile food truck permit in accordance with Section 123,0603 and the limited use regulations in Section 141,0612.
- ¹⁶ Eating and drinking establishments abutting residential development located in a residential zone may operate only between 6:00 a.m. and 12:00 midnight.
- ¹⁷ Tasting rooms are only permitted as an *accessory use* to a beverage manufacturing plant.



LETTER

RESPONSE

	ATTACHMENT 1		
San Diego Municipal Code (Proposed Changes 3-17-2016) (1-2016)	Chapter 13: Zones		
(Amended 6-12-2001 by O-18948 N.S.; effective 12-12-2001.)			
(Amended 3-1-2006 by O-19467 N.S.; effective 8-10-2006.) (Amended 8-10-2004 by O-19308 N.S.; effective 4-11-2007.)			
(Amended 6-15-2007 by O-19624 N.S.; effective 7-15-2007.)			
(Amended 4-25-2008 by O-19759 N.S.; effective 3-25-2008.) (Amended 11-13-08 by O-19799 N.S; effective 12-13-2008.)			
(Amended 11-13-08 by O-19803 N.S; effective 12-13-2008.) (Amended 11-13-08 by O-19804 N.S: effective 12-13-2008.)			
(Amended 7-6-2011 by O-20065 N.S.; effective 8-5-2011.)			
(Amended 8-4-2011 by O-20081 N.S.; effective 10-6-2011.) (Amended 2-22-2012 by O-20141 N.S.; effective 3-23-2012.)			
(Amended 6-18-2013 by O-20261 N.S.; effective 7-19-2013.)	1 12 27 20141		
(Retified to "Use Regulations Table for Commercial Zones" and O-20356 N.S.; effective 4-24-2014.)	amenaed 3-23-2014 by		
(Amended 4-3-2014 by O-20357 N.S.; effective 10-15-2014.) (Amended 5-5-2015 by O-20481 N.S.; effective 6-4-2015.)			
(Amended 7-10-2015 by O-20512 N.S.; effective 8-9-2015.)			
Editors Note: Amendments as adopted by O-20512 N. S. will n	ot apply within the Coastal		
Overlay Zone until the California Coastal Commission certifies it	as a Local Coastal		
Click the link to view the Strikeout Ordinance highlighting change	es to prior language		
http://docs.sandiego.gov/municode_strikeout_ord/O-20512-SO.pd	<u>lf]</u>		
(Amended 8-7-2015 by O-20555 N.S.; effective 9-6-2015.)			
[Editors Note: Amendments as adopted by O-20555 N.S. will not	ot apply within the Coastal		
Overlay Zone until the California Coastal Commission certifies it Program Amendment.	as a Local Coastal		
Click the link to view the Strikeout Ordinance highlighting change	es to prior language		
http://docs.sandicgo.gov/muncode_surkcout_ord/0-20555-50.p			
	Ch. Art. Div.		



RESPONSE

ATTACHMENT 7

ATT: San Diego Municipal Code (Proposed Changes 3-17-2016) Cha (1-2016) Changes 3-17-2016)

ATTACHMENT 1 Chapter 13: Zones

§131.0531 Development Regulations Tables for Commercial Zones

The following development regulations apply in each of the commercial zones as shown in Tables 131-05C, 131-05D, and 131-05E.

(a) CN Zones

Table 131-05C

Development Regulations for CN Zones

Development Regulations	Zone Designator	11.2		Zones		2.21
See Section 131.0530 for	1st & 2nd >>			CN-		
Development Regulations of	3rd >>>	1-	1-	l-	1-	1-
Commercial Zones	4th >>	1	2	3	4	5
Max Permitted Residential Density(1)	3,000	1,500	1,500	1,000	600
Supplemental Residential Regulation Section 131.0540]	ns [See	applies	applies	applies	applies	applies
Lot Area						
Min Lot Area (sf)		2,500	5,000	5,000	2,500	2,500
Max Lot Area (ac)		0.3	10	10	0.3	0.3
Lot Dimensions						
Min Lot Width (ft)		25	50	50	25	25
Min Street Frontage (ft)		25	50	50	25	25
Min Lot Depth (ft)		100				
Setback Requirements	1.11	1				
Min Front Setback (ft)						-
Max Front Sethack (ft)		10(2)	-	5(2)	5(2)	5(2)
[See Section 131.0543(a)]						
Min Side Setback (ft)		10	10	10	10	10
Optional Side Setback (ft)	111	0	0	0	0	0
[See Section 131.0543(b)]					_	
Side Setback Abutting Residentia	il	applies	applies	applies	applies	applies
[See Section 131.0543(c)]					_	-
Min Street Side Setback (ft)						
Max Street Side Sethack (ft)		10(2)		10(2)	10(2)	10(2)
[See Section 131.0543(a)				-		-
Min Rear Setback (ft)		10	10	10	10	10
Optional Rear Setback (ft)	_	0	0	0	0	0
[See Section 131.0543(b)]	110					
Section 131.0543(c)]	al[See	applies	applies	applies	applies	applies
Max Structure Height (ft)		30	30	30	65	65
Max Floor Area Ratio		1.0(3.4)	1.0(3,4)	1.0(3,4)	1.0(3.4)	1.0(3.4)
Floor Area Ratio Bonus for Resi Use [See Section 131.0546(a)]	dential Mixed	0,5	0.75	0.75	1.2	1.2

Ch. Art. Div.

RESPONSE

ATTACHMENT 7

San Diego Municipal Code (Proposed Changes 3-17-2016) Chapter 13: Zones (1-2016)Zone Development Regulations [See Section 131.0530 for Zones Designator 1st & 2nd >> CN-Development Regulations of 3rd >> 1-1-1-1-1-Commercial Zones] 4th >> 1 2 3 4 5 Minimum Floor Area Ratio for Residential 0.38 0.5 0.38 0.6 0.6 Use

ATTACHMENT 1

applies

inter.

applies

--

applies

applies

applies

applies

applies

applies

applies

applies

applies

Footnotes for Table 131-05C

Ground-floor Height [See Section 131.0548]

Building Articulation [See Section 131.0554]

Refuse and Recyclable Material Storage [See

Loading Dock and Overhead Door Screening

Pedestrian Paths [See Section 131.0550]

Transparency [See Section 131.0552]

Regulations [See Section 142.1030]

Section 142.08051

I)

- One dwelling unit per specified minimum square footage of lot area as determined in accordance with Section 113.0222.
- ² See Section 131.0543(a)(2).
- Within the Kearny Mesa Community Plan area, the maximum *floor area ratio* is 0.50 and the portion of the maximum allowed gross *floor area* that may be occupied by retail sales or eating and drinking establishments shall not exceed 70 percent.

-

applies

applies

applies

applies

applies

Within the Otay Mesa Community Plan area, the maximum floor area ratio is 0.30.

Ch. Art. Div. 13 1 5 30

LETTER

RESPONSE

(b)	CR, CO, (Developn	CV, an nent R	d CP Zo Tal egulatio	ones ble 131- ons for	-05D CR, C0	0, CV,	CP Zoi	ies			
Development Regulations	Zone Designator					Ze	ones				
[See Section 131.0530 for Development	1st & 2nd >>	CR-			C	:0-			¢	CV-	CP-
Regulations of Commercial Zones]	3rd >> 4th >>	1-2-	1	1-	1-	2-	1	3-	1	1-	1-
Max Pormitted Decidenti	al Dancity (D)	1 500	1.000	1.500		-	800	600	1 500	1.500	
Supplemental Residentia	al Densuy ···· I	applies	annlies	applies	annlies	annlies	annlies	annlies	annlies	annlies	
Regulations [See Section Lot Area	131.0540]	appires	appres	Tabbues	Tappires	Tabbues	appries	oppines	appires	appres	
Min Lot Area (sf)		15,000	5,000	5,000	5,000	5,000	5,000	5,000	15,000	5,000	10-00
Max. Lot Area (ac) Lot dimensions				-		-	-	-	-		
Min Lot Width (ft)		100	50	50	50	50	50	50	100	50	
Min Street Frontage Min Lat Depth (ft)	e (ft)	100	50	50	50	50	50	50	100	50	
Setback Requirements	1	1	1	1 200	1	1.000	1			1	
Min Front Setback (Max Front Setback (ft) (ft) 43(a)1	10	10 25 ⁽²⁾	10	10 25 ⁽²⁾	10	10(2)	10(2)	10 	10(2)	10
Min Side Setback (f	t)	10	10	10	10	10	10	10	10	10	10
Side Setback Abutti Residential [See Sec	ng tion	applies	applies	applies	applies	applies	s applies	applies	applies	applies	applies
Min Street Side Sei Max Street Side Sei ISee Section 131.05	tback (ft) tback (ft) 43(a)	10 	10 25 ⁽²⁾	10	10 25 ⁽²⁾	10	10(2)	10(2)		10(2)	-
Min Rear Setback (1	t)	10	10	10	10	10	10	10	10	10	10
Rear Setback Abutti Residential [See Sec 131.0543(c)]	ng tion	applies	applies	applies	s applies	applies	s applies	applies	applies	applies	applies
RESPONSE

(1-2010)									Chupte	15. 201	
Development Regulations	Zone Designator		Zones								
[See Section 131.0530 for Development	1st & 2nd	CR-			С	0-			C	V-	CP-
Regulations of	>> 3rd >>	1- 2-	1	1-	3	-	1	3-	1	1-	1-
Commercial Zones]	4th >>	-1-	1	2	1-	2-	1	2	1	2	1
Max Structure Height (ft)	60	45	60	45	65	65	70	60	45	30
Min Lot Coverage (%)	-				in the second se					35	
Max Floor Area Ratio	2	1.0(4,5)	0.75(4,5)	1.5(4,5)	0.75(4.5)	1.5(4.5)	2.0(4,5)	2.0(4,5)	2.0(4,5)	2.0(4,5)	1.0(4.5)
Floor Area Ratio E Residential Mixed Section 131.0546(a	Bonus for Use [See)]	1.0	1.0	1.5		-	2.5	2.5	-	æ	-
Minimum Floor Ar Residential Use	0.5	1.0	0.75		<u>.</u>	1.5	1,5	1943	-	1	
Floor Area Ratio B Child Care [See See 131.0546(b)]	applies	÷	applies	÷	applies	applies	applies	-	÷	-	
Ground-floor Height [Se 131.0548]	e Section	applies	applies	applies	applies	applies	applies	applies	applies	applies	-
Pedestrian Paths [See Se 131.0550]	ction	applies	applies	applies	applies	applies	applies	applies	applies	applies	
Transparency [See Section	on 131.0552]	-	applies		applies		applies	applies	-	applies	
Building Articulation [Se 131.0554]	e Section	applies	applies	applies	applies	applies	applies	applies	applies	applies	~
Street Yard Parking Rest Section 131.0555]		applies	-	applies	~	applies	applies		-		
Parking Lot Orientation 131.0556]	applies	-	applies		applies	÷.	+	applies	applies	(Ŧ	
Refuse and Recyclable N Storage [See Section 142.	applies	applies	applies	applies	applies	applies	applies	applies	applies	applies	
Loading Dock and Over Screening Regulations [5 142.1030]	applies	applies	applies	applies	applies	applies	applies	applies	applies	applies	

Ch. Art. Div.

RTC-210

ATTACHMENT 7

RESPONSE

ATTACHMENT 7

ATTACHMENT 1 San Diego Municipal Code (Proposed Changes 3-17-2016) Chapter 13: Zones (1-2016)

Footnotes for Table 131-05D

- ¹ One dwelling unit per specified minimum square footage of lot area as determined in accordance with Section 113.0222.
- ² See section 131.0543(a)(2).
- ³ See section 131.0543(b).
- ⁴ Within the Kearny Mesa Community Plan area, the maximum *floor area ratio* is 0.50 and the portion of the maximum allowed *gross floor area* that may be occupied by retail sales or eating and drinking establishments shall not exceed 70 percent.
- ⁵ Within the Otay Mesa Community Plan area, the maximum *floor area ratio* is 0.30.

(c) CC Zones

Table 131-05E Development Regulations for CC Zones

Development Regulation [See	Zone Designator	Zones CC-					
Section 131.0530 for	1st & 2nd >>						
Development Regulations of	3rd >>	1-2-4-5-	1-2-4-5-	1-2-4-5-	2-3-4- 5-		
Commercial Zones]	4th>>>	1	2	3	4		
Max Permitted Residential Densit	y ⁽¹⁾	1,500	1,500	1,500	1,500		
Supplemental Residential Regulat 131.0540]	ions [See Section	applies	applies	applies	applies		
Lot Area							
Min Lot Area (sq. ft.)		5,000	5,000	5,000	2,500		
Max Lot Area (ac)		1.00	*		-		
Lot Dimensions							
Min Lot Width (ft)		50	50	100	25		
Min Street Frontage (ft)		50	50	100	25		
Min Lot Depth (ft)	100	100					
Max Lot Depth (ft)		150	150	(mm)	*		
Setback Requirements							
Min Front Setback (ft)			~		-		
Max Front Setback (ft)		100(2,3)	100(2,3)		5(2)		
[See Section 131.0543(a)]							
Min Side Setback (ft)		10	10	10	10		
Optional Side Setback (ft)		0	0	0	0		
[See Section 131.0543(b)]	Supplier and and	1					
Side Setback Abutting Reside 131.0543(c)]	ential [See Section	applies	applies	applies	applies		

Ch. Art. Div. 13 1 5 33

RESPONSE

San Diego Municipal Code (Propo (1-2016)	ised Changes 3-17-2	1016)		Cha	pter 13: Zone:		
Development Regulation [See	Zone Designator	Zones					
Section 131.0530 for	1st & 2nd >>	1.1.1.1.1.1.1		CC-			
Development Regulations of	3rd >>	1-2-4-5-	1-2-4-5-	1-2-4-5-	2-3-4- 5-		
Commercial Zones]	4th >>	1	2	3	4		
Min Street Side Setback (ft)		-	-				
Max Street Side Setback (ft) [See Section 131.0543(a)]	-	-	*	-	10(2)		
Min Rear Setback (ft)	10	10	10	10			
Optional Rear Setback (ft) [See Section 131.0543(b)]	0	0	0	0			
Rear Sethack Abutting Residen 131.0543(c)]	tial [See Section	applies	applies	applies	applies		
Max Structure Height (ft)		30	60	45	30		
Min Lot Coverage (%)		-	-		35		
Max Floor Area Ratio		0.75(4.5)	2,0(4,5)	0.75(4,5)	1.0(4,5)		
Floor Area Ratio Bonus for Re. [See Section 131.0546(a)]	0.75	12	0.75	0.5			
Minimum Floor Area Ratio for	0,56	-	0.56	0.25			
Ground-floor Height [See Section 13		applies	applies				
Pedestrian Paths [See Section 131.05	applies	applies	applies	applies			
Transparency [See Section 131.0552		(and)		annlies			



1

RTC-212

ATTACHMENT 7

1

RESPONSE

				ATTA	CHMENT 1	
San Diego Municipal Code	Proposed Changes 3-1	7-2016)		Cha	pter 13: Zones	
(1-2016)						
Development Regulation [See	Zone Designator		Ż	ones		
Section 131.0530 for	1st & 2nd >>	CC-				
Development Regulations of	3rd >>>	1-2-4-5-	1-2-4-5-	1-2-4-5-	2-3-4- 5-	
Commercial Zones]	4th >>	1	2	3	4	
Building Articulation [See Section	applies	applies	applies	applies		
Parking Lot Orientation [See Se	ction 131.0556]	applies	applies	applies	-	
Refuse and Recyclable Material 142.0805]	applies	applies	applies	applies		
Loading Dock and Overhead D Regulations [See Section 142.10]	applies	applies	applies	applies		

Development Regulation [See	Zone Designator	Zones CC					
Section 131.0530 for	1st & 2nd >>						
Development Regulations of	3rd >>	2-3-4-5-	3-4-5-	3-	3-	3- 9	3-
Commercial Zones]	4th >>	5	6	7	8		10
Max permitted Residential Densit	y(1)	1,500	1,000	800	600	400	200
Supplemental Residential Regular 131.0540]	ions [See Section	applies	applies	applies	applies	applies	applies
Lot Area							1
Min Lot Area (sq. ft.)		2,500	2,500	2,500	2,500	2,500	2,50
Max Lot Area (ac)					-		7
Lot Dimensions		-					1
Min Lot Width (ft)		25	25	25	25	25	25
Min Street Frontage (ft)	25	25	25	25	25	25	
Min Lot Depth (ft)	- 26	-		-		-	
Max Lot Depth (ft)			-			\sim	
Setback Requirements							1
Min Front Sethack (ft) Max Front Sethack (ft)		10(2)		5(2)	5(2)	5(2)	13
[See Section 131.0543(a)]		10.7	2.	2	211	2.	
Min Side Setback (ft)		10	10	10	10	10	10
Optional Side Setback (ft) [See Section 131.0543(b)]	0	0	0	0	0	0	
Side Setback Abutting Reside 131.0543(c)]	applies	applies	applies	applies	applies	-	
Min Street Side Setback (ft) Max Street Side-Setback (ft) [See Section 131.0543(a)]	10(2)	10(2)	10(2)	10(2)	10(2)	1.1	
Min Rear Setback (ft)		10	10	10	10	10	10

Ch. Art. Div. 13 1 5 35

RESPONSE

(, 2010)							
Development Regulation [See	Zone Designator	1		Zones			
Section 131.0530 for	1st & 2nd >>	San Car		CC			
Development Regulations of	3rd >>	2-3-4-5-	3-4-5-	3-	3-	3-	3-
Commercial Zones]	4th >>	5	6	7	8	9	10
Optional Rear Setback (ft) [See Section 131.0543(b)]	0	0	0	0	0	0	
Rear Setback Abutting Reside 131.0543(c)]	applies	applies	applies	applies	applies	*	
Max Structure Height (ft)	100	65	65	100		-	
Min Lot Coverage (%)		35	35	35	35	35	-
Max Floor Area Ratio		2.0(4,5)	2.0(4,5)	2.0(4,5)	2.0(4,5)	2.0(4,5)	2.045
Floor Area Ratio Bonus for Re [See Section 131.0546(a)]	2.0	2.0	2.5	2.5	3.0	3.0	
Minimum Floor Area Ratio fo	r Residential Use	1.0	1.0	1.5	1.5	2.0	2.0
Ground-floor Height [See Section 1	applies	applies	applies	applies	applies	applies	
Pedestrian Paths [See Section 131.0	applies	applies	applies	applies	applies	applies	
Transparency [See Section 131.055	applies	applies	applies	applies	applies	applies	
Building Articulation [See Section	applies	applies	applies	applies	applies	applies	
Parking Lot Orientation [See Section		1.1	-	-		-	
Refuse and Recyclable Material St	applies	applies	applies	applies	applies	applies	

applies

applies

applies

applies applies

applies

Footnotes for Table 131-05E

142.0805]

1

- ¹ One dwelling unit per specified minimum square feet of lot area as determined in accordance with Section 113.0222.
- ² See Section 131.0543(a)(2).

Loading Dock and Overhead Door Screening Regulations [See Section 142.1030]

- ³ See Section 131.0543(a)(3).
- ⁴ Within the Kearny Mesa Community Plan area, the maximum *floor area ratio* is 0.50 and the portion of the maximum allowed gross *floor area* occupied by retail sales or eating and drinking establishments shall not exceed 70 percent.
- ⁵ Within the Otay Mesa Community Plan area, the maximum *floor area ratio* is 0.30.
- 6 131.0540(d) does not apply to CC-3-10

(Amended 6-12-2001 by O-18948 N.S.; effective 12-12-2001.) (Amended 11-13-08 by O-19799 N.S; effective 12-13-2008.) (Amended 8-4-2011 by O-20081 N.S.; effective 10-6-2011.) (Amended 4-11-2014 by O-20361 N.S.; effective 5-18-2014.)

Ch. Art. Div.

-

ATTACHMENT 7

LETTER

	ATTACHMENT 1	
t,	San Diego Municipal Code (Proposed Changes 3-17-2016) Chapter 13: Zones	
	(Retitled from "Development Regulations Tables of Commercial Zones" to "Development Regulations Tables for Commercial Zones" and amended 7-10-2015	
	by O-20512 N.S.; effective 8-9-2015.)	
	[Editors Note: Amendments as adopted by O-20512 N. S. will not apply within the	
	Coastal Overlay Zone until the California Coastal Commission certifies it as a Local	
	Click the link to view the Strikeout Ordinance highlighting changes to prior language	
	http://docs.sandiego.gov/municode_strikeout_ord/O-20512-SO.pdf]	
	Ch der Div	
	$\begin{bmatrix} 13 & 11 & 51 \\ 13 & 1 & 5 & 37 \end{bmatrix}$	



LETTER



1	San Diego Municipal Code (Proposed Changes 3-17-2016) ATTACHMENT 1 (1-2016) Chapter 13: Zones	
	§131.0543 Setback Requirements for Commercial Zones	
1	Refer to Section 131.0556(b) and are subject to the following exceptions and additional regulations:	
	(a) Front and <i>Street</i> Side <i>Setback</i> Requirements	
	 Off-street parking in all commercial zones may be located within the required front <i>yard</i> and required street side yard adjoining the required landscaped strip abutting <i>public rights-of-way</i>. 	
ı T	(2) In the CO, CV, and CC zones (except the CC-3 zones) with a maximum front or street side setback as shown in Tables 131-05C, 131-05D, and 131-05E, the maximum setback shall apply to only 70 percent of the street frontage. The remaining 30 percent is not required to observe the maximum setback and may be located farther from the property line. See Diagram 131-05B.	
	(3) In the CN and CC-3 zones with a maximum front or street side setback as shown in Tables 131-05C, 131-05D, and 131-05E, the maximum setback shall apply to only 90 percent of the street frontage. The remaining 10 percent is not required to observe the maximum setback and may be located farther from the property line. See Diagram 131-05C.	
Ī	Ch. Art. Div. 13 1 5 40	

LETTER

t	San Diego Municipal Code	(Proposed Changes 3-17-2016)	ATTACHMENT 1 Chapter 13: Zones

LETTER



LETTER



LETTER



LETTER

San Diego Municipal Code	IPropo	sed Changes 3-17-2016)	ATTACHMENT 1 Chapter 13: Zones
(1-2016)	P. SP.		
(1)	Comme propert acre sha that abu comply Section	creial development abutting low dens ies with a permitted density of less th all provide a 10-foot minimum setbat its low density residential zoned prop with additional step back requireme 131.0543(c)(3).	ity residentially zoned an 15 dwelling units per ck for any side or rear yard perty. The structure shall ints in accordance with
(2)	Comme resident 15 dwe and loc 131.054	ercial development abutting medium i tially zoned properties with a permitt lling units or more per acre that prov ate the structure at the <i>property line</i> i 43(b) shall comply with the following	to high <i>density</i> ted <i>density</i> of ride no side or rear <i>setback</i> as provided for by Section g:
	(A)	Minimum step back for structures pl line	laced at the side property
		Any portion of the structure of shall be stepped back from th feet, or 10 percent of the lot feet, whichever is less.	exceeding 15 fect in height he side <i>property line</i> 10 width but not less than 5
		(ii) Each 15 feet in height above back at least 3 feet from the portion of the structure immediate	e 30 feet shall be stepped minimum <i>setback</i> of that ediately below.
	(B)	Minimum step back for structures pl line.	laced at the rear property
		 Any portion of the structure of shall be stepped back from the feet, or 10 percent of the lot of feet, whichever is less. 	exceeding 15 feet in height he rear <i>property line</i> 10 depth but not less than 5
		(ii) Each 15 feet in height above back at least 3 feet from the portion of the structure immediate	e 30 feet shall be stepped minimum <i>sethack</i> of that ediately below.
(3)	For side from th stepped of the s	e and rear yards, if the structure is se e property line, each 15 fect in heigh l back at least 3 feet from the minimu tructure immediately below.	et back 10 feet or more it above 30 feet shall be um <i>setback</i> of that portion
(Added 12-9	-1997 by (O-18451 N.S.; effective 1-1-2000.)	
			Ch. Art. Div. 13 1 5 43

RESPONSE

San Diego Mui (1-2016)	ucipal Cod	le (Proposed Changes 3-17-2016)	Chapter 13: Zones
	(Amend (Amend	led 8-4-2011 by O-20081 N.S.; effective 10-6-20 led 6-18-2013 by O-20261 N.S.; effective 7-19-2	011.) 2013.)
§131.0546	Maxim	um Floor Area Ratio	
	Maxim	um <i>floor area ratio</i> is specified in Tables 131-0 to the following additional regulations:	5C, 131-05D, 131-05E and is
	(a)	Floor Area Ratio Bonus for Mixed Use	
		(1) A floor area ratio bonus is provided in s indicated in Tables 131-05C, 131-05D, uses that are developed as a part of a mi minimum required residential floor area and must be applied toward the resident remainder of the bonus may be used for residential uses.	some commercial zones, as and 131-05E, for residential xed-use <i>-development</i> . A <i>vratio</i> is shown in the tables, ial portion of the project. The either commercial or
		(2) If an underground parking structure is p development, a floor area ratio bonus sl gross floor area ratio of the undergrour premises, but not to exceed a floor area floor area ratio must be applied toward development.	rovided as part of a mixed- use hall be provided equal to the d parking <i>structure</i> on the <i>ratio</i> of 1.0. The additional the residential portion of the
	(b)	Floor Area Ratio Bonus for Child Care Facilitie	28
	(Added (Amend	In the CR-1-1, CR-2-1, CO-1-2, CO-2-2, CO-3 area ratio bonus over the otherwise maximum permitted at the rate of 4 square feet of addition square foot of gross floor area devoted to the c the total area of the premises when determining development. The area designated for the child child care for a minimum of 10 years and must requirements of Section 141.0606 (Child Care 1 12-9-1997 by O-18451 N.S.; effective 1-1-2006 led 7-10-2015 by O-20512 N.S.; effective 8-9-20	-1, and CO-3-2 zones, a <i>floor</i> allowable <i>gross floor area</i> is al <i>gross floor area</i> for each 1 <i>hild care facility</i> to be added to the <i>floor area ratio</i> for a <i>care facility</i> must be used for be in compliance with the Facilities).
	[Editor Coastal Coastal Click th http://do	rs Note: Amendments as adopted by O-20512 Overlay Zone until the California Coastal Cor Program Amendment. he link to view the Strikeout Ordinance highligh ocs.sandiego.gov/municode_strikeout_ord/O-20	N. S. will not apply within the mission certifies it as a Local ting changes to prior language 512-SO.pdf]
Ch. Art. Div. 13 1 5 44			

ATTACHMENT 7

RESPONSE



ATTACHMENT 7

LETTER



LETTER

RESPONSE



§131.0552 Transparency

Where transparency is required by Tables 131-05C, 131-05D, or 131-05E, a minimum of 50 percent of *street wall* area between 3 and 10 feet above the sidewalk shall be transparent, with clear glass visible into a commercial or residential use. Windows or other transparent materials that provide visibility into a garage or similar area do not count toward the required transparency. See Diagram 131-05E.



(Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)

§131.0554 Building Articulation

Where building articulation is required by Tables 131-05C, 131-05D, or 131-05E, the following regulations apply.

(a) All building elevations fronting a *public right-of-way* shall be composed of offsetting planes that provide relief in the *building facade* by insetting or projecting surfaces (planes) of the building. The minimum number of offsetting planes and the minimum horizontal separation between planes is based on the length of the new *building facade*, as shown in Table 131-05F.

Ch. Art. Div.

LETTER



Table 131-05F Offsetting Plane Requirements

Length of New Building Façade	Number of Offsetting Planes Required
25 ft or less	2 with a minimum separation of 3 inches
More than 25 ft. but less than or equal to 50 ft	4 planes consisting of : 2 with a minimum separation of 3 inches, and 2 with a minimum separation of 8 inches
More than 50 ft. but less than or equal to 100 ft	6 planes consisting of: 2 with a minimum separation of 3 inches, and 2 with a minimum separation of 8 inches, and 2 with a minimum separation of 3 feet
More than 100 ft	 6 planes consisting of: 2 with a minimum separation of 3 inches, and 2 with a minimum separation of 8 inches, and 2 with a minimum separation of 3 feet, and plus 1 additional plane for each 50 feet of <i>building</i> <i>facade</i> length over 100 feet (maximum of 3 additional planes required with a minimum separation of 5 feet).

(1) For purposes of this section, the area of a plane may include separate surfaces that are non-contiguous but which all lie in the same plane. Each numbered surface of the building shown in Diagram 131-05F represents a different plane of the building facade. The sum of the area of each plane labeled with the same number in Diagram 131-05F is the total area of that plane.



LETTER



LETTER

1	San Diego Mu (1-2016)	nicipal Code	(Proposed Changes 3-17-2016)	Chapter 13: Zones	
		(3)	A cornice equal to at least 70 percent of <i>facade</i> that has a minimum offset of 1 <i>wall</i> ; and	of the length of the <i>building</i> foot, located along the <i>street</i>	
		(4)	Structural cantilevers with combined l percent of the length of the <i>building fe</i> at least 2 fect, located along the <i>street</i> as offsetting planes.	engths totaling at least 30 acade, with an average offset of wall. Roof eaves do not count	
		(Added 12-9	0-1997 by O-18451 N.S.; effective 1-1-20	00.)	
	§131.0555	Parking Re	striction		
		In the CO-1 front and str	-1, CO-3-1, and CO-3-2 zones, parking is eet side yard.	s not permitted in the required	
		(Added 12-9 (Amended 7	-1997 by O-18451 N.S.; effective 1-1-20 -10-2015 by O-20512 N.S.; effective 8-9-	00.) 2015.)	
		[Editors No Coastal Ove Coastal Prog Click the lin http://docs.s	te: Amendments as adopted by O-2051 relay Zone until the California Coastal C gram Amendment. k to view the Strikeout Ordinance highlig andiego.gov/municode_strikeout_ord/O-	2 N. S. will not apply within the ommission certifies it as a Local ghting changes to prior language 20512-SO.pdf]	



LETTER



LETTER

	Article 1: Separately Regulated Use Regulations	
	Division 10: Industrial Use CategorySeparately Regulated Uses (Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)	
§141.1001	Artisan Food and Beverage Producer	
	Artisan Food and Beverage Producer applies to establishments that engage in commercial on-site production of food and/or beverage products. Typical products may include, but are not limited to micro-breweries, coffee roasting, ice cream, baked goods, confectioneries, alcoholic and non-alcoholic beverages, and other foodstuffs and may be permitted as a Limited Use or subject to a Neighborhood Use Permit indicated with either an "L" or "N" in the Use Regulation Table 131-05B in Chapter 13, Article 1 (Base Zones) subject to the following regulations.	
	 (a) These uses shall be limited to 10.000 square feet of ground floor area. (b) All storage shall be within an enclosed building or screened from the public right-of-way by fences or walls and landscaping. Stored items shall not be stacked to a height that exceeds the height of the screening. (c) Hours of operation shall be limited to 6:00 a.m. until 10:00 p.m. so that neighboring residential development is not disturbed by noise and lights. (d) Distribution facilities are not permitted adjacent to residentially zoned property. 	
<u>§141.1002</u>	Hazardous Waste Research Facilities	
	This section regulates <i>structures</i> , improvements on the land, and all contiguous land used for research related to the treatment, transfer, storage, resource recovery, disposal, or recycling of <i>hazardous waste</i> .	
	Hazardous waste research facilities may be permitted with a Conditional Use Permit decided in accordance with Process Four in the zones indicated with a "C" in the Use Regulations Tables in Chapter 13, Article 1 (Base Zones) subject to the following regulations.	
	(a) At the time of application for a Conditional Use Permit, the <i>applicant</i> shall provide proof of approval of a Research, Development and Demonstration Permit for <i>Hazardous Waste</i> Treatment from the Environmental Protection Agency, or any other agency of the United States Government, pursuant to the Federal Resources Conservation and Recovery Act.	
	(b) The applicant shall provide the City with documentation of the activities that will take place on the site and a plan describing the safeguards the applicant	

LETTER

	ATTACHMENT I	
San Diego Mur	hicipal Code (Proposed Changes 3-17-2016) Chapter 14: General Regulations (Added 12-9-1997 by 0-18451 N.S.; effective 1-1-2000.)	
§141.100 <u>3</u> 3	Hazardous Waste Treatment Facilities	
	This section regulates <i>structures</i> , improvements on the land, and all contiguous land used for the treatment, transfer, storage, resource recovery, disposal, or recycling of <i>hazardous waste</i> .	
	Hazardous waste treatment facilities may be permitted with a Conditional Use Permit decided in accordance with Process Five in the zones indicated with a "C" in the Use Regulations Tables in Chapter 13, Article 1 (Base Zones) subject to the following regulations.	
(a) At le file a notice Assis	ast 90 calendar days before applying for a Conditional Use Permit, the <i>applicant</i> shall e of intent to apply for the permit with the City Manager and with the Office of Permit tance in the State Office of Planning and Research. The notice of intent shall contain a complete description of the nature, function, and scope of the <i>development</i> .	

LETTER

San Diego M (1-2016)	unicipal C	odelerc	prosed Changes 2+1/-2010) Chapter 14: General Regulations
	(b)	With Conc coop San I propo appli	in 90 calendar days of the date on which a notice of intent to apply for a litional Use Permit is filed, the Office of Permit Assistance will, in eration with the City Manager, convene a public meeting in the City of Diego to inform the public about the nature, function, and scope of the seed project and the procedures that are required for approving cations for the <i>development</i> .
	(c)	With Conc mem asses the C repre com Gove Cour in pe after the p time	in 90 calendar days of the date on which a notice of intent to apply for a litional Use Permit is filed, the City Council shall appoint a seven- ber local assessment committee. The membership of the local sment committee shall be broadly constituted to reflect the makeup of ity and shall linclude three representatives of the City at large, two sentatives of environmental or public interest groups, and two sentatives of affected businesses and industries. Members of the nittee shall have no direct financial interest, as defined in the California imment Code, Section 87103, in the proposed <i>development</i> . The City cil shall provide staff resources to assist the local assessment committee forming its duties. The local assessment committee shall cease to exist final administrative action by state and local agencies has been taken on ermit applications. The local assessment committee shall, within the period prescribed by the City Council, do all of the following:
		(1)	Adopt rules and procedures that are necessary to perform its duties;
		(2)	Represent generally the interest of the residents of the City and the interests of adjacent communities in negotiations with the <i>applicant</i> ;
		(3)	Negotiate with the <i>applicant</i> on the detailed terms, provisions, and conditions for project approval that would protect the public health, safety, and general welfare and the environment of the City and adjacent communities, and would promote the fiscal welfare of the City through special benefits and compensation;
		(4)	Receive and expend, subject to the approval of the City Manager and authorization of the City Council, the technical assistance grants made available by the Office of Permit Assistance in the State Office of Planning and Research to enable the local assessment committee to hire an independent consultant to assist the committee in reviewing the <i>development</i> and negotiating terms, provisions, and conditions with the <i>applicant</i> ; and
<i>Ch. drt. Div.</i> 14 1 10	2	(5)	Advise the City Manager, Planning Commission, and City Council of the terms, provisions, and conditions for approval that have been agreed upon by the committee and the <i>applicant</i> and of any additional information that the committee deems appropriate. The City Manager, Planning Commission, and City Council may use this advice for their independent consideration of the <i>development</i> .

LETTER

San Die	go Municip	Dal Code (Proposed Changes 3-17-2016) Chapter 14: General Regulat	1 ons
(1-20	16)		
	(0	d) The City Manager will notify the Office of Permit Assistance in the State Office of Planning and Research within 10 <i>business days</i> of the date on wh an application for a Conditional Use Permit is <i>deemed complete</i> by the City	ch
	(e	e) Within 60 calendar days of receipt of this notice, the Office of Permit Assistance in the State Office of Planning and Research will convene a meeting of the lead agency and responsible agencies for the <i>development</i> , the <i>applicant</i> , the local assessment committee, and the interested public, to be held in the City of San Diego, to determine the issues that concern the agencies that are required to approve the project and the issues that concern the public. The City Manager shall provide notice to the public of the date, time, and place of the meeting.	c
	(f	f) Following the meeting required by Section 141.10032(e), the local assessment committee and the <i>applicant</i> shall meet and confer on the proposal for the purpose of establishing the terms, provisions, and conditions under which the <i>development</i> would be acceptable to the community. If the local assessment committee and the <i>applicant</i> cannot resolve any differences through the meetings, the Office of Permit Assistance in the State Office of Planning and Research may recommend the use of a mediator.	nt e i
	(E	g) Pursuant to the California Health and Safety Code, Section 25199.7(g), the applicant shall pay a fee, established by the Office of Permit Assistance in the State Office of Planning and Research, equal to the cost of hiring independent consultants to review the development. The Office of Permit Assistance in State Office of Planning and Research may use this money to make technic assistance grants to the local assessment committee to enable the committee hire an independent consultant to assist the committee in reviewing the development and negotiating terms, provisions, and conditions with the applicant.	ne nt he J to
	(I	h) Pursuant to California Health and Safety Code Section 25199.7(h), the applicant shall pay one-half of the costs of any mediation process that may recommended by the Office of Permit Assistance in the State Office of Planning and Research. The remaining costs will be paid, upon appropriati by the legislature, from the State General Fund.	oc on
	(i	 Within 60 calendar days of the date on which the application is <i>deemed</i> complete, and after a noticed public hearing, the City Council shall, at the request of the <i>applicant</i>, issue an initial written determination on whether th proposed <i>development</i> is consistent with the following documents: The <i>land use plan</i> and zoning ordinances in effect at the time the application was received, and 	e
		Ch. Art. D 14 1 1	3

LETTER

(_	San Diego Mun (1-2016)	icipal Code	e(Proposed Changes 3-17-2016)	Chapter 14: General Regulations			
		C	(2) The County Hazardous Waste California Health and Safety C Section 25135), if such plan is	Management Plan authorized by ode Article 3.5 (commencing with in effect at the time of the application.	*		
i i		(j) T P	The City Manager shall send a copy of pursuant to Section 141.100፺금(i), to th	the written determination, made the <i>applicant</i> .			
1		(k) T C a d ii	The determination required by Section Council from making a different detern approve or deny the Conditional Use P determination is based on information initial determination was made.	141.10032(i) does not prohibit the City mination when the final decision to Permit is made, if the final that was not considered at the time the			
		(I) T a dt A w d (Added I	The decision of the City Council regard approval, or denial of a Conditional Us applicant or an <i>interested person</i> to the the Governor's designee, pursuant to C Article 8.7 (commencing with Section within 30 calendar days after the date of denies the Conditional Use Permit. 12-9-1997 by O-18451 N.S.; effective	ding the approval, conditions of se Permit is final unless appealed by the e Governor of the State of California, or California Health and Safety Code 25199) to Chapter 6.5 of Division 20, on which the City Council approves or 1-1-2000.)			
1	§141.100 <u>4</u> 3	Marine-	-Related Uses in the Coastal Zone				
		Marine-r a "P" in r related u Permit d "C" in th the follo	related uses in the Coastal Overlay Zo the Use Regulations Tables in Chapte uses in the Coastal Overlay Zone may decided in accordance with Process Fo he Use Regulations Tables in Chapter owing regulations.	ne are permitted in zones indicated with r 13, Article 1 (Base Zones). Marine- be permitted with a Conditional Use ur in the zones indicated with a 13, Article 1 (Base Zones) subject to			
		(a) A a d	All storage, service, and repair areas sh are not visible, or shall be screened so development and public rights-of-way.	hall be located on the site so that they that they are not visible, from adjacent			
		(b) C v	Off-street parking shall be provided at without impacting adjacent or nearby p	a level sufficient to serve the facility property.			
		(c) A	A litter control program is required be	fore approval of the permit.			
		(Added 1 (Amende	12-9-1997 by O-18451 N.S.; effective ed 5-5-2015 by O-20481 N.S.; effectiv	1-1-2000.) e 6-4-2015.)			
(Ch. Art. Div.						

LETTER



LETTER

		(A) All required permits for the construction, landscaping, or related land improvements have been approved by a public agency in accordance with applicable provisions of state law and locally adopted plans and ordinances, including, but not limited to Public Resources Code, Division 13 (commencing with Section 21000).
		(B) The lead agency's approval of the construction project included consideration of the onsite excavation and onsite earthmoving activities pursuant to Public Resources Code, Division 12 (commencing with Section 21000)
		(C) The approved construction project is consistent with the general plan or zoning of the <i>premises</i> .
		(D) Surplus materials shall not be exported from the <i>premises</i> unless and until actual construction work has commenced and shall cease if it is determined that construction activities have terminated, have been indefinitely suspended, or are no longer being actively pursued.
	(7)	Operation of a plant site used for mineral processing, including associated onsite structures, equipment, machines, tools, or other materials, including the onsite stockpiling and onsite recovery of mined materials, subject to all of the following conditions:
		(A) The plant site is located on lands designated for industrial or commercial uses in the applicable county or city general plan.
		(B) The plant site is located on lands zoned industrial or commercial, or are contained within a zoning category intended exclusively for industrial activities by the applicable city or county.
		(C) None of the minerals being processed are being extracted onsite.
		(D) All reclamation work has been completed pursuant to the approved reclamation plan for any mineral extraction activities that occurred onsite after January 1, 1976.
Ch. Art. Drv. 14 1 10 6	(8)	Emergency <i>excavations</i> or <i>grading</i> conducted by the California Department of Water Resources or the Reclamation Board for the purpose of averting, alleviating, repairing, or restoring damage to property due to imminent or recent floods, disasters, or other emergencies.

I FTTFR

RESPONSE

ATTACHMENT 1 **Chapter 14: General Regulations**

San Diego Municipal Code (Proposed Changes 3-17-2016) (1-2016)

- (9) Surface mining operations conducted on lands owned or leased, or upon which easements or rights-of-way have been obtained, by the California Department of Water Resources for the purpose of the State Water Resources Development System or flood control, and surface mining operations on lands owned or leased, or upon which easements or rights-of-way have been obtained, by the Reclamation Board for the purpose of flood control, if the California Department of Water Resources adopts, after submission to and consultation with, the California Department of Conservation, a reclamation plan for lands affected by these activities, and those lands are reclaimed in conformance with the standards specified in regulation of the board adopted pursuant to the California Surface Mining and Reclamation Act of 1975. The California Department of Water Resources shall provide an annual report to the California Department of Conservation by the date specified by the California Department of Conservation on these mining activities.
- (10) Excavations or grading for the exclusive purpose of obtaining materials for roadbed construction and maintenance conducted in connection with timber operations or forest management on land owned by the same person or entity. This exemption is limited to excavation and grading that is conducted adjacent to timber operation or forest management roads and shall not apply to onsite excavation or grading that occurs within 100 feet of a Class One watercourse or 75 feet of a Class Two watercourse, or to excavation for materials that are, or have been sold for commercial purposes.
 - (A) The exemption set forth in Section 141.10054(a)(10) applies only if slope stability and erosion are controlled in accordance with Section 3704(f) and Section 3706(d) of Title 14 of the California Code of Regulations and, upon closure of the site, the person closing the site implements, where necessary, revegetation measures and postclosure uses in consultation with the California Department of Forestry and Fire Protection.
- (11) Excavations, grading, or other earthmoving activities in an oil or gas field that are integral to, and necessary for, ongoing operations for the extraction of oil or gas that comply with all of the following conditions:
 - The operations are being conducted in accordance with Public (A) Resources Code, Division 3 (commencing with Section 3000).



LETTER

			(B) The operations are consistent with any general plan or zoning	
			 (C) The earthmoving activities are within oil or gas field properties under a common numer or operator. 	
			(D) No excavated materials are sold for commercial purposes.	
	(b)	Vested	d Rights	
		(1)	Any person who obtained a vested right to conduct a <i>surface mining</i> operation before January 1, 1976, shall not be required to secure a Conditional Use Permit pursuant to the provisions of Section 141.100 <u>5</u> 4, as long as the vested right continues and no substantial change is made in the operation except in accordance with the provisions of Section 141.100 <u>5</u> 4. Any substantial change in the <i>surface mining</i> operation subsequent to January 1, 1976, may be permitted only with a Conditional Use Permit in accordance with Section 141.1004.	
			(A) A person is deemed to have a vested right if, before January 1, 1976, that person has in good faith and in reliance upon a permit or other authorization, if a permit or other authorization was required, diligently commenced <i>surface mining</i> operations and incurred substantial liabilities for work and necessary materials. Expenses incurred in obtaining the enactment of a resolution in relation to a particular operation or the issuance of a permit shall not be deemed liabilities for work or materials that would create a vested right.	
		(2)	A person with vested rights who has continued <i>surface mining</i> in the same disturbed area after January 1, 1976, and who did not receive approval for his or her reclamation plan prior to July 1, 1990, shall cease continuation of the surface mining operation until a reclamation plan has been submitted to the City Manager and approved in accordance with Process One. All <i>reclamation</i> plans submitted to the City Manager for vested operations that are conducted after January 1, 1976, shall be accompanied by the applicable deposit.	
		(3)	Any person who has obtained a vested right to conduct <i>surface mining</i> operations shall obtain a grading permit and be subject to the same frequency of inspection as those mining operators required to obtain a Conditional Use Permit pursuant to Section 141.100 <u>5</u> 4.	
Ch. Art. 14 1	Div. 10 8			

RESPONSE

ATTACHMENT 1 Chapter 14: General Regulations

San Diego Municipal Code (Proposed Changes 3-17-2016) (1-2016)

- (4) Nothing in Section 141.10054 shall be construed as requiring the filing of a reclamation plan for, or the reclamation of, mined lands on which surface mining operations were conducted prior to, but not after, January 1, 1976.
- (c) The applicant shall submit a reclamation plan, financial assurances and grading plans, in accordance with the provisions set forth in Section 141.1004; the California Surface Mining and Reclamation Act of 1975, Article 5 including Section 2772, 2773 and 2773.1; applicable provisions of Chapter 8, Division 2, Title 14 of the California Code of Regulations including Sections 3500-3505 and 3700-3713; and procedures established by the City Manager. The Conditional Use Permit, reclamation plan, financial assurance, and grading plan shall be processed as a consolidated action.
- (d) The Director of the California Department of Conservation shall be notified by the City Manager of the filing of a Conditional Use Permit application pursuant to Section 141,100<u>5</u>4.
- (e) In accordance with Public Resources Code section 2772, any person who owns, leases, or otherwise controls or operates on all or any portion any mined lands, or who plans to conduct *surface mining* operations on the lands, shall submit a *reclamation* plan for approval by the City Manager. The *reclamation* plan shall be submitted in a format specified by the City Manager. The reclamation plan shall include all information and documentation set forth in Public Resources Code sections 2772 (c) and 2773(a).
- (f) The mining operator shall file an annual surface mining report on forms provided by the State Mining and Geology Board with the California Department of Conservation and the City Manager no later than the anniversary date established by the Director of the California Department of Conservation, or as otherwise required by the Conditional Use Permit.
- (g) Reclamation plans, reports, applications, and other documents submitted in accordance with Section 141.10054 are public records unless it can be demonstrated to the satisfaction of the City Attorney that the release of this information would reveal production, reserves, or rate of depletion that is entitled to protection as proprietary information. The City Attorney shall identify the proprietary information as a separate part of each application. A copy of all permits, reclamation plans, reports, applications, and other documents submitted in accordance with Section 141.10054, including proprietary information, shall be furnished to the Director of the California Department of Conservation by the City Manager. Proprietary information shall be made available to persons other than the Director of Department of Conservation only when authorized by the surface mining owner in accordance with Public Resources Code section 2778.



LETTER

RESPONSE

ATTACHMENT 1 Chapter 14: General Regulations

- San Diego Municipal Code(Proposed Changes 3-17-2016) (1-2016)
 - As a condition of approval for the Conditional Use Permit or the reclamation (h) plan, or both, the applicant shall agree to allow the City, upon notice of inspection, to enter the site to inspect and evaluate continuing compliance with the Conditional Use Permit and the reclamation plan. The inspections shall occur no less frequently than once in any calendar year, in accordance with Public Resources Code section 2774(b). The inspection shall be conducted by a state-registered geologist, state registered civil engineer, state licensed landscape architect or state registered forester, who is experienced in land reclamation and who has not been employed by the surface mining operation in any capacity during the twelve month prior to the inspection. The inspection shall be conducted using a form provided by the California Department of Conservation and subject to review and approval by the City Manager. The completed inspection form and an inspection report shall be submitted to the City Manager within fifteen days of the inspection. All costs related to the inspections and report shall be borne solely by the operator. The City Manager shall notify the California Department of Conservation within thirty days of completion of the inspection that the inspection has been conducted; the City Manager shall also forward a copy of the notice, the completed inspection form and any necessary supporting documentation, to the applicant.
 - (i) As a result of the annual inspection, if the City Manager finds that the surface mining operator is not following the provisions of the reclamation plan, the surface mining operator shall be given notice to comply within a given time not to exceed ninety calendar days. A copy of the notice shall be given to the owner of the land upon which the surface mining operations are located. If at the end of the stated time the operator is not in compliance, the City Manager may revoke or suspend the Conditional Use Permit or the reclamation plan or both until the surface mining operator complies or obtains approval of a revised reclamation plan.
 - (j) In accordance with the provisions of Section 141.10054, Public Resources Code section 2773.1 and as a condition of approval of the Condition Use Permit or the reclamation plan or both, the surface mining operator shall submit financial assurances to ensure compliance with the surface mining operation's reclamation plan, including revegetation and landscaping requirements, restoration of aquatic or wildlife habitat, restoration of water bodies and water quality, slope stability and erosion and drainage control, disposal of hazardous materials, annual adjustments for disturbance to new lands and those anticipated for the upcoming calendar year, inflation and other measures, as necessary.

Ch. Art. Div.

1

LETTER

 (1) Cost estimates shall be prepared in accordance with the procedures bard is "financial assurances diadlines" and shall be submitted to the City Manager for review and approval prior to the surface mining periods recently financial assurances. (2) A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. (3) Revisions to financial assurances shall be submitted to the City Manager each year prior to the anning experiods of the financial assurances. The annual adjustment shall take into account or the wind and structures shall be into account or the financial assurances. The annual adjustment shall take into account or the review and disturbed by surface mining, operations, changes with respect to environmental conditions affected by mining operations, changes with respect to environmental conditions affected by mining operations, changes in the laws and regulations affecting surface mining, inflation and reclamation of lands accomplished in accordance with the reclamation of and scenario of lands accomplished in accordance with the reclamation particles of the california Department of Conservation, that the surface mining operations, the accordance with the reclamation particles of the california Department of Conservation, that the surface mining period or defined as surface and the california period period the work in accordance with the provisor of the colivorian Use Period tand has completed the work in accordance shall be male surface mining operators and the California Department of Conservation, that the surface mining operations, the accordance with the provisor of the colivorian surface according the surface mining operators of the colivorian surface accordance with the provisor of the colivorian surface acomplete different accordance with the provisor of the co	<text><list-item><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></list-item></text>	<text><list-item><list-item><list-item><list-item><list-item></list-item></list-item></list-item></list-item></list-item></text>	90 Municipal Co.	de (j	Proposed Changes 3-17-2016)	ATTACHMENT 1 Chanter 14: General Regulations
 Cost estimates shall be prepared in accordance with the procedures outlined in the most recent edition of the State Mining and Geology Board's "Financial Assumace Guidelines" and shall be submitted to the City Manager for review and approval prior to the surface mining operations occuring financial assumances. A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. Revisions to financial assumances shall be submitted to the City Manager each year prior to the annives and the for approval of the financial assumances. The annual adjustments shall the into account new lands disturbed by surface mining operations, new information concerning mining reclamation on the reclamation of subject mining distances with the provide of the surface mining, inflation and reclamation of a subject mining face mining, inflation and reclamation of lands accomplished in accordance with the reclamation plan, thanges in the laws and regulations affecting augree mining, inflation and reclamation of lands accomplished in accordance with the reclamation plan. The financial assumances shall be made payable to the City of San Diego and the California Department of Conservation and may be any of those listed below. The financial assurances that be released, upon written notification from the City Manager to the surface mining operator and the California Department of Conservation, that the surface mining operator and the California Department of Conservation, the account of the following: A deposit of money or negotiable bonds of the kind approved for securing deposites of public moneys; A deposit of money or negotiable bonds of the kind approved for securing leposites of public moneys; A deposit of money or negotiable bonds of the kind approved for securing leposites of public moneys; A deposit of money or negotiable bonds of the kind approved for securing securing leposites of public	 Cost estimates shall be prepared in accordance with the procedures outlined in the most recent efficient of the State Mining and Geology Board * Financial Assurance Guideline* and shall be submitted to the City Manager for review and approval prior to the surface mining operator securing financial assurances. A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. Revisions to financial assurances shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assurances. The annual alguisments shall the submitted to the City Manager each year prior to the anniversary date for approval of the financial assurances. The annual alguisments shall the submitted in the account new lands disturbed by surface mining operations, changes with respect to environmental conditions affected by mining operations, new information concerning mining reclamation of the reclamation of subject mined lands, modifications of the reclamation of and here and reclamation of a subject or equipments and take and may be any of those listed below. The financial assurances shall be released, upon written notification of from the City Manager to the surface mining operator and the California Department of Conservation, that the surface mining operator and the California Department of Conservation shall be released, upon written notifications from the City Manager to the surface mining operator and the California Department of Conservation shall be released, upon written for the surface mining operator and the California Department of Conservation and may be any of the following: (A) A bond or bonds by one or more duly authorized corporate security disputible bonds of the kind approved for securing deposits of public moneys; (B) A deposit of money or negotiable bonds of the kind approved for securing adminest or federal government and pledging that the fuds necessary to carry out th	 Cost estimates shall be prepared in accordance with the procedures outlined in the most recent edition of the State Mining and Geology Board's "Financial Assumaces and approval prior to the surface mining operation securing financial assumaces. A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. Revisions to financial assumaces shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assumaces. The annual adjustments shall take into account new hards disturbed by surface mining operations, changes with respect to environmental conditions affected by mining operations, new information concerning mining <i>reclamation</i> or the reclamation of subject mined lands, modifications of the reclamation and metamation of lands accomplished in accordance with the reclamation plan. The financial assumaces shall be made payable to the City of San Diego and the California Department of Conservation, that the surface mining operations, fraining operations, fraining operations, fraining operations, fraining operations of the California Department of Conservation, that the conditional Up Permit and has compliable to the sufface mining operations of the California Department of Conservation, that the surface mining operations of the California Department of Conservation, that the conditional Up Permit and has compliable to the sufface mining operations of the California Department of Conservation, that the surface mining operations of the following: A hond or bonds by one or more financial institutions subject to regulation by one core more financial institutions subject to regulation of 1981 complic to and the California Department of Conservation and the observation of the financial assumaces. A chepsi of money or negoriable bonds of the kind approved for securing deposits of public moneys: A negosi of money or negoriable bonds of th	(1-2016)	ue je	Toposed Changes 5-17-2010	Chapter 14, General Regulations
 (2) A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. (3) Revisions to financial assurances shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assurances. The annual adjustments shall take into account new lands disturbed by <i>surface mining</i> operations, new miniformation concerning mining <i>reclamation</i> of the <i>reclamation</i> of <i>reclamation</i> of the <i>reclamation</i> plan. Financial assurances may be any of the following: (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or fedral government and <i>pledging</i> that the funds necessary to carry out the plan are on <i>deposit</i> of <i>reclamation</i> plan. The <i>reclamation</i> of <i>recla</i> government and <i>reclamat</i>	 (2) A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. (3) Revisions to financial assumaces shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assumaces. The annual adjustments shall take into account new hands disturbed by <i>suface iming</i> operations, here elementation of a subject mined hands, modifications of the reclamation of a subject mined hands, modifications of the reclamation plan, changes sin the laws and regulations affected by minor fination and reclamation of the reclamation plan, changes in the laws and regulations affecting and/nec mining, inflation and reclamation of lands accomplished in accordance with the reclamation plan, the composition of the relamation plan. (4) The financial assurances shall be made payable to the City of San Diego and the California Department of Conservation, that the suffice mining operator is in compliance with the previsions of the suffice mining operator of the City Manager to the suffice mining operator of the City for San Diego and the California Department of Conservation, that the suffice mining operator is in compliance with the previsions of the Conditional Use Permit and has completed the work in accordance with the approved reclamation plan. Financial assurances may be any of the following: (A) A boad or bonds by one or more duly authorized corporate securities: (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys: (C) An instrument of credit from one or more financial institutions subject to regulation by the state or fideral jovernment and pledging that the finan encessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and dequite to ensure reclamation in accordance with the California Surface Mining and Reclamation	 (2) A copy of the cost estimates will be forwarded to the State California Department of Conservation for review. (3) Revisions to financial assurmaces shall be submitted to the City Manage each year prior to the anniversary date for approval of the financial assurmaces shall be mining operations, heave on information concerning mining reclanation of the reclamation of subject mining definitions, faces of the reclamation of an evolutions affecting any face mining information and many and affecting any face mining information and many and infecting any face mining information and many and infecting any face mining influince and reclamation of ands accomplished in accordance with the reclamation plan. (4) The financial assurmaces shall be made payable to the City of San Ogo and the California Department of Conservation and may be any of those listed below. The financial sourances shall be released, upon with no tofficiation from the City Manager othat angle mining prior and may be any of those listed below. The financial assurances with the provisions of the Contificinal Use Permit and has completed the work in accordance with the approved reclamation plan. Financial assurances may be any of these listed below. The financial assurances may be any of the following: (A) A bond or bonds by one or more duly authorized corporate securities: (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or folder government and plediging that the finance of the financial assurances in the plan are on deposit and guaranteed for payment; or the filternia by available and adequate to ensure reclamation in accordance with the California beget metas plan and the california beget metas pla		(1)	Cost estimates shall be prepared i outlined in the most recent edition Board's "Financial Assurance Gu the City Manager for review and operator securing financial assura	n accordance with the procedures n of the State Mining and Geology uidelines" and shall be submitted to approval prior to the <i>surface mining</i> neces.
 (3) Revisions to financial assurances shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assurances. The annual adjustments shall take into account new lands disturbed by surface mining operations, changes with respect to environmental conditions affected by mining operations, changes in the laws and regulations of the reclamation of subject minel dands, modifications of the reclamation and reclamation of plan. (4) The financial assurances shall be made payable to the City of San Diego and the California Department of Conservation and may be any of those listed below. The financial assurances shall be readed, upon written notification from the City Manager to the surface mining operator and the California Department of Conservation and may be any of those listed below. The financial assurances value the provisions of the Condonace with the provisions of the Condonace with the surface mining operator is in compliance with the provisions of the Condonace with the surface mining operator is no compleximent of Conservation, that the surface mining operator is no compliance with the provisions of the Conditional US Permit and has completed the work in accordance with the approved reclamation plan. Financial assurances may be any of the following: (A) A bond or bonds by one or more financial institutions (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for securing which he State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Department or decomplexitor. (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate	 (3) Revisions to financial assumances shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assumances. The annual algustment shall take into account new lands disturbed by <i>sunface mining</i> operations, changes with respect to environmental conditions affected by mining operations, new information concerning mining <i>reclamation</i> of the <i>reclamation</i> of the <i>schamation</i> plan, changes in the laws and regulations affecting <i>surface mining</i>, inflation and <i>reclamation</i> flands, noclifications of the <i>reclamation</i> of a lands accomplished in accordance with the reclamation plan. (4) The financial assurances shall be made payable to the City of San Digge and the California Department of Conservation, that the <i>surface mining</i> operator and the California Department of Conservation, that the <i>surface mining</i> operator is in compliance with the provisions of the Conditional Use Permit and has completed the work in accordance with the approved <i>reclamation</i> plan. Financial assurances may be any of the following: (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for security which the state or federal government and pledging that the fund second payment; or (C) An instrument of credit from one or more financial institutions subject to regulation by subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and dequate to ensure reclamation Action of 1975. 	 (3) Revisions to financial assumaces shall be submitted to the City Manager each year prior to the anniversary date for approval of the financial assumaces. The annual adjustments shall take into account new lands disturbed by <i>surface mining</i>, orelations, new information conterning mining <i>reclamation</i> or the <i>reclamation</i> of as subject mined lands, modifications of the reclamation of plan, changes in the laws and regulations affecting surface mining, inflation and recelamation of lands accomplished in accordance with the reclamation plan, changes in the laws and regulations affecting surface mining, inflation and recelamation of lands accomplished in accordance with the reclamation plan. (4) The financial assurances shall be made payable to the City of San Diggs and the California Department of Conservation and may be any of floxe listed below. The financial assurances shall be released, upon written notification from the City Manager to the <i>surface mining</i> operators and the California Department of Conservation, that the <i>surface mining</i> operator is in compliance with the provisions of the Conditional Use Permit and has completed the work in accordance with the approved <i>reclamation</i> plan. Financial assurances may be any of the following: (4) A bond or bonds by one or more dual suthorized corporate securities; (5) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (c) An instrument of credit from one or more financial institutions subject to regulation with the california and california and government and pleiging that the financia according with the dargevernet and pleiging that the finan security with the California beam finance with the california beam for a subject or regulation the three california beam finance with the california beam finance with		(2)	A copy of the cost estimates will Department of Conservation for r	be forwarded to the State California eview.
 (4) The financial assurances shall be made payable to the City of San Diego and the California Department of Conservation and may be any of those listed below. The financial assurances shall be released, upon written notification from the City Manager to the <i>surface mining</i> operator and the California Department of Conservation, that the <i>surface mining</i> operator is in compliance with the provisions of the Conditional Use Permit and has completed the work in accordance with the approved <i>reclamation</i> plan. Financial assurances may be any of the following: (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securiting deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation of 1975. 	 (4) The financial assurances shall be made payable to the City of San Diego and the California Department of Conservation and may be any of those listed below. The financial assurances shall be released, upon written notification from the City Manager to the <i>surface mining</i> operator and the California Department of Conservation, that the <i>surface mining</i> operator is in compliance with the provisions of the Conditional Use Permit and has completed the work in accordance with the approved <i>reclamation</i> plan. Financial assurances may be any of the following: (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the finds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 	 (4) The financial assurances shall be made payable to the City of San Diego and the California Department of Conservation and may be any of those listed below. The financial assurances shall be released, upon written notification from the City Manager to the surface mining operator and the California Department of Conservation, that the surface mining operator is in compliance with the provisions of the Conditional Use Permit and has completed the work in accordance with the approved reclamation plan. Financial assurances may be any of the following: (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the fords necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation of 1975. 		(3)	Revisions to financial assurances Manager each year prior to the an financial assurances. The annual new lands disturbed by <i>surface m</i> to environmental conditions affect information concerning mining <i>ra</i> subject mined lands, modification the laws and regulations affecting <i>reclamation</i> of lands accomplished plan.	shall be submitted to the City universary date for approval of the adjustments shall take into account <i>ining</i> operations, changes with respect ted by mining operations, new <i>eclamation</i> or the <i>reclamation</i> of as of the <i>reclamation</i> plan, changes in g <i>surface mining</i> , inflation and ed in accordance with the reclamation
 (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 	 (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 	 (A) A bond or bonds by one or more duly authorized corporate securities; (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. Ch. dn. Dn. 14 1 10 17 		(4)	The financial assurances shall be Diego and the California Departn of those listed below. The finance written notification from the City operator and the California Depar <i>surface mining</i> operator is in com Conditional Use Permit and has c with the approved <i>reclamation</i> pl of the following:	made payable to the City of San nent of Conservation and may be any ial assurances shall be released, upon Manager to the <i>surface mining</i> truent of Conservation, that the upliance with the provisions of the completed the work in accordance an. Financial assurances may be any
 (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 	 (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. Ch. der. Div. 11 	 (B) A deposit of money or negotiable bonds of the kind approved for securing deposits of public moneys; (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 			 (A) A bond or bonds by one of securities; 	or more duly authorized corporate
 (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 	 (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 	 (C) An instrument of credit from one or more financial institutions subject to regulation by the state or federal government and pledging that the funds necessary to carry out the plan are on deposit and guaranteed for payment; or (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. 			(B) A deposit of money or neg for securing deposits of p	gotiable bonds of the kind approved ublic moneys;
 (D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. Ch. dr. Dr. 10 	(D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. Ch. Art. Dm. 14 1 10 11	(D) Other security which the State Mining and Geology Board determines are reasonably available and adequate to ensure reclamation in accordance with the California Surface Mining and Reclamation Action of 1975. Ch. dr. Dr. 11			(C) An instrument of credit fr subject to regulation by th pledging that the funds no deposit and guaranteed fo	om one or more financial institutions te state or federal government and ecessary to carry out the plan are on r payment; or
					(D) Other security which the 5 determines are reasonably reclamation in accordance and Reclamation Action c	State Mining and Geology Board available and adequate to ensure e with the California Surface Mining of 1975. <u>Ch. Arr. Dr.</u> <u>14</u> <u>1</u> <u>10</u> <u>11</u>

LETTER

 (3) Default of financial assurances shall comply with the procedures established by the City Manager, as amended from time to time. (4) Whenever any surface mining operation or portion of a surface mining operation that is subject to Section 141.1002/4 is sold, assigned, conveyed, exchanged, or otherwise transferred, the successor in interest shall be bound by the provisions of the Conditional USe Permit, reclamation plan, the provisions of Section 141.1002/4 and the California Surface Mining and Reclamation Act of 1975. (6) In accordance with Public Resources Code section 2770, and as further provided in Section 141.1002/4, whenever any surface mining operation becomes <i>ilde</i>, the surface mining operation shall alumit a proposed interim management plan (<i>IMP</i>) to the City Manager. For review and approval. The IMP shall be calonitized within ninety days of the operation becoming <i>ilde</i> on forms provided by the City Manager. The view and approval of the IMP shall be carried out in accordance with Public Resources Code section 2770(h). Upon receip of a complete proposed IMP, the City Manager shall forward it to the California Department of Conservation for review. (m) Devitations from the approved <i>reclamation</i> plan, financial assurances and the Conditional Use Permit have been approved by the decision maker in accordance with Public Resources to by the decision maker in accordance with Process Four, or the Substantial Conformance Review process where applicable. (n) In the OR-1-2 zone, the following regulations apply. (1) Processing and other related mining activities (such as asphaltic processing) are permitted only within the allowable 25 percent development area. (2) All mining and other related mining activities must be consistent with the Species Conservation Program Plan, the California Surface Mining and Reclamation Act of 1975, the regulations in Chapter 14, Article 3, Division 1 (Environmentaly Sensitive Lands Regulations) and other applicab	-2016)	Contraction of Changes S 17-2010	Chapter 14, General Regulations
 (k) Whenever any <i>surface mining</i> operation or portion of a <i>surface mining</i> operation that is subject to Section 141.10024 is sold, assigned, conveyed, exchanged, or oftenvise transfered, the successor in interest shall be bound by the provisions of Section 141.10024 and the California Surface Mining and Reclamation Act of 1975. (i) In accordance with Public Resources Code section 2770, and as further provided in Section 141.10024, and the California Surface Mining and Reclamation Act of 1975. (ii) In accordance with Public Resources Code section 2770, and as further provided in Section 141.10024, whenever any <i>surface mining</i> operation becomes <i>idle</i>, the <i>surface mining</i> operation shall submit a proposed interim management plan (<i>MP</i>) to the City Manager for review and approval. The IMP shall be submitted within ninety days of the operation becoming <i>idle</i> on forms provided by the City Manager. Review and approval. The IMP shall be carried out in accordance with <i>Public Resources</i> Code section 2770(h). Upon receipt of a complete proposed IMP, the City Manager shall forward it to the California Department of Conservation for review. (m) Deviations from the approved <i>reclamation</i> plan, including an IMP, are not permitted unless amendments to the <i>reclamation</i> plan, financial assurances and the Conditional USe Permit have been approved by the decision maker in accordance with Process Four, or the Substantial Conformance Review processing) are permitted only within the allowable 25 percent development area. (i) In the OR-1-2 zone, the following regulations apply. (ii) Processing and other related mining activities must be consistent with the objectives, guidelines, and recommendations in the Multiple Species Conservation Program Plan, the California Surface Mining and Reclamation Act of 1975, the regulations in Chapter 14, Arkile 3, Division 1 (Environmentally Sensitive Lands Regulations) and other applicable state and local laws and regulations. (ii		(5) Default of financial assur established by the City M	nces shall comply with the procedures nager, as amended from time to time.
 (1) In accordance with Public Resources Code section 2770, and as further provided in Section 141.100<u>5</u>4, whenever any <i>surface mining</i> operator in management plan (<i>IMP</i>) to the City Manager for review and approval. The IMP shall be submitted within ninety days of the operation becoming <i>idle</i> on forms provided by the City Manager. Review and approval of the IMP shall be earried out in accordance with Public Resources Code section 2770(h). Upon receipt of a complete proposed IMP, the City Manager shall forward it to the California Department of Conservation for review. (m) Deviations from the approved <i>reclamation</i> plan, including an IMP, are not permitted unless amendments to the <i>reclamation</i> plan, financial assurances and the Conditional Use Permit have been approved by the decision maker in accordance with Process Four, or the Substantial Conformance Review process where applicable. (n) In the OR-1-2 zone, the following regulations apply. (1) Processing and other related mining activities (such as asphaltic processing) are permitted only within the allowable 25 percent development area. (2) All mining and other related mining activities must be consistent with the objectives, guidelines, and recommendations in the Multiple Species Conservation Program Plan, the California Surface Mining and Reclamation Ator 1975, the regulations in Chapter 14, Article 3, Division 1 (Environmentally Sensitive Lands Regulations) and other applicable state and local laws and regulations. (3) Any sand removal activities should be monitored for noise impacts to surface mining operation the breeding and needing and needin	(k)	Whenever any <i>surface mining</i> op operation that is subject to Sectio exchanged, or otherwise transfer by the provisions of the Conditio provisions of Section 141.100 <u>5</u> 4 Reclamation Act of 1975.	ration or portion of a <i>surface mining</i> 141.100 <u>5</u> 4 is sold, assigned, conveyed, 2d, the successor in interest shall be bound al Use Permit, <i>reclamation</i> plan, the and the California Surface Mining and
 (m) Deviations from the approved reclamation plan, including an IMP, are not permitted unless amendments to the reclamation plan, financial assurances and the Conditional Use Permit have been approved by the decision maker in accordance with Process Four, or the Substantial Conformance Review process where applicable. (n) In the OR-1-2 zone, the following regulations apply. (1) Processing and other related mining activities (such as asphaltic processing) are permitted only within the allowable 25 percent development area. (2) All mining and other related mining activities must be consistent with the objectives, guidelines, and recommendations in the Multiple Species Conservation Program Plan, the California Surface Mining and Act of 1975, the regulations in Chapter 14, Article 3, Division 1 (Environmentally Sensitive Lands Regulations) and other applicable state and local laws and regulations. (3) Any sand removal activities should be monitored for noise impacts to surrounding sensitive habitats, and all new sediment removal or surface mining operations proposed in proximity to the MHPA or changes in existing operations, must include noise reduction methods that take into consideration the breeding and nesting seasons of sensitive bird species. 	(1)	In accordance with Public Resou provided in Section 141.10054, v becomes <i>idle</i> , the <i>surface mining</i> management plan (<i>IMP</i>) to the C IMP shall be submitted within ni forms provided by the City Mana be carried out in accordance with Upon receipt of a complete prop- to the California Department of C	these Code section 2770, and as further henever any <i>surface mining</i> operation operator shall submit a proposed interim by Manager for review and approval. The ety days of the operation becoming <i>idle</i> on ger. Review and approval of the IMP shall Public Resources Code section 2770(h). sed IMP, the City Manager shall forward it onservation for review.
 (n) In the OR-1-2 zone, the following regulations apply. (1) Processing and other related mining activities (such as asphaltic processing) are permitted only within the allowable 25 percent development area. (2) All mining and other related mining activities must be consistent with the objectives, guidelines, and recommendations in the Multiple Species Conservation Program Plan, the California Surface Mining and Reclamation Act of 1975, the regulations in Chapter 14, Article 3, Division 1 (Environmentally Sensitive Lands Regulations) and other applicable state and local laws and regulations. (3) Any sand removal activities should be monitored for noise impacts to surrounding sensitive habitats, and all new sediment removal or surface mining operations, must include noise reduction methods that take into consideration the breeding and nesting seasons of sensitive bird species. 	(m)	 Deviations from the approved reprinted unless amendments to and the Conditional Use Permit I accordance with Process Four, or process where applicable. 	<i>amation</i> plan, including an IMP, are not he <i>reclamation</i> plan, financial assurances we been approved by the decision maker in the Substantial Conformance Review
 Processing and other related mining activities (such as asphaltic processing) are permitted only within the allowable 25 percent development area. All mining and other related mining activities must be consistent with the objectives, guidelines, and recommendations in the Multiple Species Conservation Program Plan, the California Surface Mining and Reclamation Act of 1975, the regulations in Chapter 14, Article 3, Division 1 (Environmentally Sensitive Lands Regulations) and other applicable state and local laws and regulations. Any sand removal activities should be monitored for noise impacts to surrounding sensitive habitats, and all new sediment removal or <i>surface mining</i> operations proposed in proximity to the <i>MHPA</i> or changes in existing operations, must include noise reduction methods that take into consideration the breeding and nesting seasons of sensitive bird species. 	(n)	In the OR-1-2 zone, the followin	regulations apply.
 (2) All mining and other related mining activities must be consistent with the objectives, guidelines, and recommendations in the Multiple Species Conservation Program Plan, the California Surface Mining and Reclamation Act of 1975, the regulations in Chapter 14, Article 3, Division 1 (Environmentally Sensitive Lands Regulations) and other applicable state and local laws and regulations. (3) Any sand removal activities should be monitored for noise impacts to surrounding sensitive habitats, and all new sediment removal or <i>surface mining</i> operations, must include noise reduction methods that take into consideration the breeding and nesting seasons of sensitive bird species. 		 Processing and other rela processing) are permitted development area. 	ed mining activities (such as asphaltic only within the allowable 25 percent
 (3) Any sand removal activities should be monitored for noise impacts to surrounding sensitive habitats, and all new sediment removal or <i>surface mining</i> operations proposed in proximity to the <i>MHPA</i> or changes in existing operations, must include noise reduction methods that take into consideration the breeding and nesting seasons of sensitive bird species. 		(2) All mining and other relative objectives, guidelines Species Conservation Pro- and Reclamation Act of 1 Division 1 (Environment applicable state and local	ed mining activities must be consistent with and recommendations in the Multiple gram Plan, the California Surface Mining 975, the regulations in Chapter 14, Article 3, Ily Sensitive Lands Regulations) and other aws and regulations.
		(3) Any sand removal activit surrounding sensitive hat surface mining operation changes in existing opera that take into consideratii sensitive bird species	es should be monitored for noise impacts to tats, and all new sediment removal or proposed in proximity to the <i>MHPA</i> or ions, must include noise reduction methods in the breeding and nesting seasons of
	Ch. An. Div. 14 1 10 12	sensuive bird species.	

LETTER

				ATTACHMENT 1
San Diego Mi (1-2016)	inicipal	Code (Proposed Changes 3-17-2016)	Chapter 14: General Regulations
		(4)	All existing and future mined lan shall be reclaimed in accordance and Reclamation Act of 1975 an biologically to the <i>MHPA</i> . Nati as possible, No invasive non-nat into areas adjacent to the MHPA Any permitted <i>surface mining</i> ac must consider changes and impa	nds adjacent to or within the <i>MHPA</i> with the California Surface Mining id should be designed to contribute we habitats should be restored as much tive plant species shall be introduced a stivity, including <i>reclamation</i> of sand, tets to surface water and groundwater
	(0) (Am (Am (An	The G grad inade nended 6- nended 1 nended 3-	quality, water table level, fluvial upstream and downstream and m City Manager may suspend or revo ling permit for violation of the term equate financial assurances, or Mun -12-2001 by O-18948 N.S.; effectiv 1-28-2005 by O-19468 N.S.; effective -1-2005 by O-19468 N.S.; effective	hydrology, flooding, and habitats nust provide adequate mitigation. oke a Conditional Use Permit or ns and conditions of the permit, sicipal Code violations. <i>e</i> 12-12-2001.) five 2-9-2006.) 4-1-2006.)
§141.100 <u>6</u> 5	Nev	vspaper	Publishing Plants	
	New deci Use foll	vspaper j ided in ad Regulat owing re	publishing plants may be permitted coordance with Process Three in the tions Tables in Chapter 13, Article 1 gulations.	with a Conditional Use Permit e zones indicated with a "C" in the I (Base Zones) subject to the
	(a)	The c	design of the structures shall incorp tents that help to diminish building b	porate a variety of architectural bulk.
	(b)	Large mini inten	er <i>structures</i> , areas of high activity imize impacts to surrounding <i>develo</i> ise.	and parking areas shall be located to opment that is smaller in scale and less
	(c)	Acce	ess to the facility shall be as direct as ials and shall avoid residential stree	is possible from <i>freeways</i> and primary <i>ets</i> .
	(d)	Off-s with	street parking shall be provided at a out impacting adjacent or nearby pro	level sufficient to serve the facility operty.
	(c) (Ad	All st are n deven ded 12-9	torage, service, and repair areas sha tot visible, or shall be <i>screened</i> so the <i>lopment</i> and <i>public rights-of-way</i> . 1-1997 by O-18451 N.S.; effective 1-	Il be located on the site so that they hat they are not visible, from adjacent -1-2000.)
				Ch. Art. Div. 14 1 10 13
LETTER



LETTER

RESPONSE

				ATTACHMENT 1
I.	San Diego Mun (1-2016)	icipal C	ode (Proposed Changes 3-17-2016)	Chapter 14: General Regulations
		(d)	The facility shall be designed to protect use	rhy streams and hodies of water
		(d) (Adde	from runoff related to the operation of the f ed 12-9-1997 by O-18451 N.S.; effective 1-1-2	acility. 2000.)
Ĩ.	§141.100 <u>9</u> 8	Wrec	king and Dismantling of Motor Vehicles	
		Wrec Use F "C" in the fo	king and dismantling of motor vehicles may b Permit decided in accordance with Process For n the Use Regulations Tables in Chapter 13, A Illowing regulations.	e permitted with a Conditional ar in the zones indicated with a Article 1 (Base Zones) subject to
		(a)	All storage shall be within an enclosed built right-of-way by fences or walls and landsca stacked to a height that exceeds the height of	ling or <i>screened</i> from the <i>public</i> ping. Stored items shall not be <i>f</i> the <i>screening</i> .
		(b)	A litter control program is required before	approval of the permit.
		(c)	A pest control program is required before	approval of the permit.
		(d)	Hours of operation shall be limited so that r development is not disturbed by noise and 1	ights.
		(e)	Measures shall be taken to ensure that the g affected by the increase of impermeable sur the facility.	round water table is not adversely faces due to the <i>development</i> of
Ĩ		(f) (Adde	Measures shall be taken to ensure that the w affected by runoff of fuel, lubricants, or oth ed 12-9-1997 by O-18451 N.S.: effective 1-1-2	vater quality is not adversely er pollutants. 2000.)
Ť.				
				Ch. Art. Div. 14 1 10 15

RESPONSE

LETTER

CPIOZ A -Building Heights by Right ATTACHMENT 2 (Ministerial Approval) a ecomount N' ARADA DA -* MOR.20070 対岸 michile --an an an an - -----CHANNEL HAND VANHERBATC -MANANCES HERA AVE WHELE N N GEALLER ST BALBOA PARK 104 LEGEND Maximum Building Height: 65 feet by right approval 🛄 Maximum Building Height: 50 feet by right approval 61.001 Community Plan Boundary D - THE AND AND AND STREETS Uptown Gateway Specific Plan Area Chaptering 4 - 2012 has a 30 - Ad Register media in A laws on Provinging states on a state has found 130 filling a forware company or galaxing a Modern film.

LETTER



RESPONSE

Uptown Community Plan—Land Use Element Uptown Gateway Specific Plan

2.4 Uptown Gateway Specific Plan

Discussion

In order to implement the General Plan's City of Villages Strategy, village areas are planned throughout the Uptown Community. The Uptown Gateway urban village is the most significant commercial center and largest village place-type in Uptown. The Uptown Gateway is envisioned as a compact, active area that creates sustainable and efficient land use patterns and includes a variety of residential, commercial, and civic spaces. Based on its location, the Uptown Gateway is an ideal pedestrian-friendly, transit-oriented mixed use urban village where people of all walks of life can live, work, and play. In addition, the Uptown Bikeways projects include large investments into protected bikeways through the Uptown Gateway on 4th Avenue, 5th Avenue, and University Avenue. Transit is an integral part of village development in Uptown, with multiple transit lines along the major north-south and east-west corridors connecting activity centers and employment centers. Application of transit-oriented development design principles are intended to support increased transit use. Further details on village area and transit supportive design are contained in the Urban Design Element.

Specific Plan

The Uptown Gateway will be redeveloped through the Uptown Gateway Specific Plan, which will provide specific design guidelines for the development of Uptown's urban village. The Specific Plan will contain design details for commercial, residential, and mixed use development, public spaces, the precise location of public facilities and amenities, streetscape improvements, and implementation plans that will provide infrastructure improvements and facilities as development occurs.

The Community Plan and General Plan provide specific direction and guidance for the development of Specific Plan. The Specific Plan should be privately sponsored and developed in collaboration with the City of San Diego. It will be considered an amendment to the Community Plan, to add implementation of the Uptown Gateway district.

The Uptown Gateway Specific Plan must create sustainable and efficient land use patterns, and must meet all of the criteria within policies LU-4.1 and LU-4.2 below and must demonstrate consistency with the General Plan policies, specifically the Urban Design Element Sections C and E, and the Mobility Element Sections A, ME-B.9, and ME-B.10, ME-C.3, and Table ME-1 to provide further guidance to determine consistency with City policies. The implementation program must include the phased provision of infrastructure and public facilities.

Policies and Recommendations

LU-4.1 Require the Uptown Gateway Specific Plan and any required rezoning to be consistent with the policies of this plan.

LU-4.2 Achieve sustainable and efficient land use patterns with comprehensive urban village development that:

a. Provide a land use map that illustrates the detailed land use designations. The specific plan land use map will refine the Uptown Community Plan Land Use Map as part of the specific plan approval process.

5/16/16

1

RESPONSE

Uptown Community Plan—Land Use Element Uptown Gateway Specific Plan ATTACHMENT 3

- Illustrate the complete circulation system and indicate how the system will relate to the overall Uptown circulation system.
- c. Illustrate a separate system of pedestrian and bicycle facilities and pathways that create uninterrupted north-south and east-west links to the surrounding areas.
- d. Identify specific locations for public spaces.
- e. Link public spaces to one another with pathways to increase connectivity and enhance sense of community.
- f. Incorporate a diversity of housing types that includes market rate and affordable housing. Encourage inclusionary housing on-site.
- g. Include an appropriate amount of housing consistent with the projections provided in the plan.
- h. Provide development at densities that support transit as an integral component of the Uptown Gateway urban village and Transit Priority Area.
- Require a mixed-use residential/commercial component to be included within village core areas, with neighborhood-serving commercial uses such as food markets, restaurants, and other small retail shops.
- Include a detailed design plan for the mixed-use urban core that identifies retail, convenience uses, and public spaces.
- k. Provide refined architecture, urban design, and streetscape guidelines consistent with the policies in the Uptown Community Plan and the relevant General Plan policies.
- I. Include guidelines and illustrations for height, bulk, and scale of buildings and their relation to each other.
- m. Provide a street tree plan that utilizes species within the Street Tree Plan within the Urban Design Element of the Uptown Community Plan.
- Require a phasing plan to ensure timely provision of necessary public facilities to serve the proposed development.

5/16/16

2

LETTER



RESPONSE

and and a

Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

Our overarching conclusion is that the City is not likely to meet the State mandated housing elements due to constraints imposed by existing zoning CC-3-9, which limits the actual densities required to achieve project feasibility. Absent a zoning upgrade, Hillerest and the remaining "urban ring" will stagnate. Properties that would normally be coveted and targeted for redevelopment and revitalization are foreclosed by the current zoning. Nothing will be developed.

The following summarizes the main conclusions of our research and analysis:

- Existing zoning in these "Urban Ring" neighborhoods are not sufficiently dense, which discourages new development. Goals to build infill housing, whether market rate or affordable, are not likely to be achieved in the coming years.
 - a. Based on our analysis of development achieved under CC-3-9 zoning (1 unit per 400 SF of land), new development will work at land values ranging from \$125 to \$150 per square foot.
 - b. However, there are few opportunities to acquire property in this price range. As a result, higher density development will be required to achieve a new "land value basis" ranging from \$250 to \$300 per square foot. It is this range of value which will compel land owners to sell to a developer. The development value of the land must be higher than the current use of the property otherwise a landowner will not sell for development purposes. The only way to achieve a higher land value is to increase the allowable density to a level much higher than 109 units per acre.
 - c. The density bonus does little to encourage the inclusion of affordable housing. In Table 1, we analyze the impact of land values, profitability and affordable housing. By including affordable housing, the land value increases just slightly by \$25 per square foot. However, this is due to reduced parking ratios which reduces construction costs. The inclusion of additional units does not result in a more profitable project. In fact, it lowers the net operating income on a per unit basis (NOI/Unit), which is the key metric developers consider. Essentially, the goal should be for the density bonus to increase the bottom line profitability (NOI/Unit). But the density bonuses studied in our report actually result in a "wash", at best, for the developer.
- 2. To meet housing goals for both market rate and affordable units, the density required must approach 200 units per acre, particularly in popular high traffic areas with an already existing commercial base. This is because the value of land in already existing commercial areas achieves a range of \$200 to \$300 per square foot (some properties are more expensive). To encourage redevelopment of these areas and properties, the City must allocate density so the land is worth more as a development project.

LETTER

RESPONSE



Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

- a. In Table 2 we studied higher density scenarios of 1 unit per 275 SF and 225 SF of land. This density represented projects similar to those recently constructed in East Village that are Type III modified construction of six stories or fewer. Our conclusion is that land values of \$250 to \$325 per square foot can be achieved. However, because the projects cannot exceed six stories without changing to more expensive Type I construction, it essentially results in an "urban wall" with the building pushed out to every corner of the property. This would be a highly undesirable result.
- In Table 3 we studied density of 1 unit per 200 SF of land, which is consistent with our recommended zone of CC-3-10.
 - i. At this density, the most expensive Type I construction method must be utilized. As a result, development needs to be "vertical" to achieve higher rental premiums through unit views and orientation. This "stacking" of development has a direct benefit to the community because it opens up the ground floor of the site to public open space, parks, plazas and gathering places.
 - ii. A vertical project also will result in differentiating unit types: shopkeeper units, garden-style and tower units. The project as a whole shifts the local neighborhood because it naturally facilitates place making and improved building design (compared to the squared six story boxed developments characteristic of recent Downtown San Diego projects).
 - iii. The achievable land value, with or without affordable housing, ranges from \$250 to \$300 per square foot. Most importantly, the NOI/Unit is not significantly changed when affordable housing is added (reduction of 5% or less). While still a "wash", institution investors are more likely to support the inclusion of affordable housing at this level of density.
 - iv. It is important to note that the parking ratios assumed in this proposed zoning is 1.0 per residential unit, and no parking for commercial. This is critical because the high cost of underground parking can render these projects infeasible. This is a recognition that reduced parking standards, or even the elimination of parking standards, are required in high traffic commercial zones that are identified as transit areas.

Density and design flexibility are required to meet housing goals. Increasing the allowable maximum building height is not the cure unless it is accompanied by techniques to increase overall density (units per acre). The key to building both market rate and affordable housing will be to entitle development to reach the range of 200+/- units per acre, as a minimum.

Recently built projects that have included affordable housing have not actually benefited from the density bonus, Rather, it is the *deviations* from the code or other allowances (density deviation, parking regulations, setbacks, etc.) that have resulted in much higher density. These mechanisms have so far proven to be more important to developers because they create a path to

LETTER

RESPONSE



Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

maximize density. But these are only a "patch up". What is clearly required is a comprehensive re-evaluation of zoning and density within the City of San Diego.

Should you have any questions regarding this analysis, please contact us.

Sincerely,

Joy H. Tool Gary H. London President

Nathan Mordin Nathan Moeder Principal

LETTER

RESPONSE



Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

RESPONSE

C.

Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

Approach to Analysis & Methodology

Our study focused on the area of San Diego known as the "Urban Ring", considered prime for accommodating a large share of the future residential growth of the City of San Diego. This area includes those neighborhoods surrounding Downtown to the north and the east: Uptown, Hillerest, University Heights, Normal Heights, North Park, South Park and Golden Hill.

The purpose of this section is to detail the assumptions and methodologies of our comprehensive financial analysis. The purpose of our analysis is to answer two key questions:

- What level of density is required to achieve a land value that compels property owners to sell their land for development purposes?
- 2. How does the density bonus from affordable housing impact housing development?

Property Land Values

The first question is the most critical because property owners must be compelled to sell their property to developers, otherwise redevelopment will not occur. Recently, the City has received incorrect data and conclusions regarding the level of density required to encourage housing development.² The first step is to understand what the current market values are for properties based on land square footage.

The following table details the 2015 property sales statistics in the 92103, 92104 and 92116 zip codes (e.g. Hillcrest/Uptown, North Park and Normal Heights). In 2015, there were 1,236 transactions with an average sale price of \$122.89. The average values increase to \$163.11 per square foot if R1 zoning is excluded, which are single family neighborhoods that are not likely to be redeveloped into higher density multifamily.

² For example, the 2014 study issued by the consulting firm AECOM (ittled "Incentive Zoning Analysis" pages 7-9 which relied on land value and construction cost data that does not accurately reflect current market conditions or actual development scenarios.

I FTTFR

RESPONSE

Summary of Prop Zip Codes	perty Transactions 92103, 92104 and	- 2015 Sales 92116
Zone	# Sales	Av. \$/S.F of Land
С	38	\$205.03
R1	650	\$86.57
R2	229	\$150.51
R3	260	\$164.02
R4	39	\$180.89
S	20	\$181.45
Total	1,236	\$122.89
Total (excl. R1)	586	\$163.11

Source: San Diego County Tax Assessor; Realist

It is important to note that the majority of redevelopment projects, or infill housing, seek opportunities in established or up-and-coming neighborhoods. These neighborhoods represent the upper threshold of property values because they are relatively safe or gentrification has already begun. Therefore, while the average transactional property value is \$163 per square foot of land (in 2015), we anticipate infill housing opportunities to be in areas where the land values are higher, sometimes significantly. This also coincides with our recent advisory work for property owners in the "Urban Ring" where property values are \$200 to \$300 per square foot of land.

Another factor which must be recognized is that a landowner would expect the developer value to be much higher than the current as-is use of the property. For example, consider a commercial structure with a tenant that is valued at \$200 per square foot of land. The landowner can sell the property for the as-is use in relatively quick fashion with a normal escrow period. There is also a higher level of certainty that the buyer would close on the property since it is a typical commercial building.

However, in the development world, builders require longer escrow periods to achieve entitlements/permits and remove other contingencies related to development. There is also greater uncertainty related to timing. It's possible that builders could try to renegotiate the sale price based on their due diligence inspections. This results in the property being tied up, or off the market, for a longer period of time. Due to this additional risk and exposure, the seller would expect a higher sale price than the current as-is use value of the property, which is a cleaner and quicker path to closing a transaction. So a property that may have an existing as-is value of \$200 per square foot, would need to sell for considerably higher (e.g. \$250 to \$300 per square foot) to incentivize the property owner to sell it for development purposes.

in the second se

Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

Financial Analysis: Inputs & Assumptions

The main goal of our financial analysis is to determine the level of development intensity, or density, required to achieve a land value that is well in excess of \$200 per square foot of land. The purpose of this section is to detail the key assumptions of our financial analysis.

From a zoning perspective, it is important to analyze the code in the context of rental apartments to ensure that land values and economic feasibility is achieved in the conservative scenario, which is rentals.

Condominium development represents a higher achievable land value because the for-sale value of units is typically greater than apartments on a rental basis. However, condominium development is cyclical. Therefore, it is important to be certain that the zoning code works for both rental apartments and condominium development to ensure housing is delivered regardless of the market cycle.

Revenues and Costs

The following bullet points summarize the key inputs and assumptions related to revenues and costs:

- The monthly rents utilized in our analysis are based on the same rents being achieved in the East Village area of Downtown. The total average monthly rent is approximately \$2,500 or \$3.15 to \$3.25 per square foot among these projects.
- The East Village projects are typically five to six stories above retail and represent Type III modified construction.
- In projecting our rents for conceptual development in the Urban Ring, we utilized the same level of rents as East Village, but adjusted the overall rent level by the number of stories in the project. For example, a three story building would have an overall lower rent than a six story building.
- Overall, our analysis is conservative by utilizing East Village rent levels because it results in the maximum residual land value possible (due to the assumption of high rents). Today, this higher level of rent may only be achievable in the most desirable locations of the Urban Ring, such as the proposed Uptown Gateway District, downtown Hillcrest or parts of North Park. However, we anticipate this level of rent to be achieved in other parts of the Urban Ring as redevelopment evolves in the coming years.
- For construction costs, we have utilized hard costs that are currently being experienced in the market (2016). Construction costs estimates where provided by Ledcor, Safdie Rabines Architects and De Bartolo and Rimanic Design Studio.
- For each density scenario we considered the corresponding building height, which dictates the type of construction and hard costs required to build the project. We utilized Type V Modified construction for buildings under six stories. Type III Modified was utilized for

RESPONSE



Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

projects that are around six stories in height. Type I construction was utilized for projects well in excess of six stories.

Other Key Assumptions

The following bullet points detail the other key factors and assumptions related to our financial proformas:

- All parking is assumed to be subterranean. In an effort to achieve the maximum residual land value, we have assumed zero parking spaces for commercial space, which is a deviation from the current zoning code.
- Annual inflation of rents is forecasted at 4.0% per year.
- Disposition capitalization rates are assumed at 5.0% five years from today for both apartments and rental. The asset is assumed to be sold in Year 5.
- Financing is assumed at 70% debt / 30% equity.

Affordable Housing

Our analysis considers the density bonus achievable under State Law (AB 2222). In an effort to maximize land values, we assumed the following scenarios in our analyses:

- ✤ 10% Low Income Housing (80% AMI); 20% density bonus
- ✤ 5% Very Low Income Housing (50% AMI); 20% density bonus

These scenarios represent the greatest upside to developers (greatest density bonus for the fewest number of affordable housing units). Also, important to our analysis is that we didn't pursue higher densities because it would alter the construction method and increase costs across the entire project. Therefore, to evaluate the true impact of affordable housing, it is critical to stay within the same construction method and cost structure of a given project.

Project Feasibility

To determine the residual land value of a property, the project must be feasible, which means that the investment generates an acceptable minimum profit for its investors.

Project feasibility is ultimately determined by the rate of return that is achieved for the equity invested. The minimum required returns are determined by the equity markets, large institutions such as pension funds and insurance companies. Locally, this would include institutions such as

LETTER

RESPONSE

G

Analysis of Zone CC-3-9 and Density Bonuses ATTACHMENT 4

CalPERS, SDCERS and SDCERA. These institutions invest part of their portfolio into real estate to generate a reliable source of income to pay for projected benefits and claims.³

For apartment development projects in San Diego County, institutions require an annual return of 6.0% to 6.5% on project costs (known as Yield On Cost). This is the minimum return required for institutions to be able to afford their projected benefits and claims. A project is not feasible and will not be built if this benchmark return is not achieved.

² Every institution has a return requirement. For example, if CalPERS invests \$50 million into an apartment asset and has a return requirement of 6.5%, then it needs to receive \$3.25 million per year to meet that goal. If the goal cannot be achieved, they would seek development opportunities in other markets.

LETTER



LETTER

RESPONSE

	TAI	BLE 1	
		CC-3-9 (1 Unit per 400 SF of	Lot Area)
	100% Market Rate	10% Low Income (80% AMI); 20% Density Bonus	5% Very Low Income (50% AMI); 20% Density Bonus
Density (DU/Ac.)	109	131	131
Bldg Height	40'	50'	50'
# Units	75	90	90
Construction Type	V-Mod.	V-Mod.	V-Mod.
Hard Costs PSF	\$160	\$160	\$160
Parking Costs (\$/Space)	\$40,000	\$40,000	\$40,000
Land Value PSF	\$125	\$150	\$150
Land Value Per Unit	\$50,000	\$50,000	\$50,000
Average Rental Rates			
Market Rate	\$2,230	\$2,238	\$2,237
Affordable	\$0	\$1,276	\$810
Total Blended Rent	\$2,230	\$2,152	\$2,173
Total Blended \$/S.F.	\$2.86	\$2.79	\$2.80
% Diff to Market Rate		-2.4%	-2.0%
NOI Per Unit	\$24,633	\$23,179	\$23,518
% Diff to Market Rate	· · · · ·	-5,9%	-4.5%
Profitability Over 5 Years			
Profit/Unit	\$177,953	\$145,248	\$170,969
% Diff to Market Rate		-18.4%	-3.9%
Yield On Cost	6.2%	6.3%	6.3%
IRR	24%	22%	24%
Residential Value Per Unit (YR 5)	\$439,001	\$394,755	\$422,219
% Diff to Market Rate		-10.1%	-3.8%

Notes:

-Increase in land value is primarily due to lower required parking ratios, not necessarily density bonus.

-Parking ratios also assume zero spaces for commercial.

Conclusions:

-Land values ranging from \$125 to \$150 psf are too low to encourage redevelopment of existing property.

LETTER



City of San Diego August 5, 2016 Page 2 of 5

The EIR fails to adequately analyze land use impacts. Indeed, a goal of the City's B19-4 Climate Action Plan ("CAP") - "Promote effective land use to reduce vehicle miles traveled" - is implemented by Action 3.6: "Implement transit-oriented development within Transit Priority Areas." CAP at 39. Furthermore, the General Plan's Strategic Framework Element provides: "By directing growth primarily toward village centers, the strategy works to preserve established residential neighborhoods and open space, and to manage the City's continued growth over the long term." General Plan at SF-6. In other words, both the CAP and General Plan acknowledge the appropriateness of encouraging development in areas of employment and transit simultaneously preserves existing neighborhoods and open space while reducing greenhouse gas emissions. Yet the Community Plan does not guarantee such growth and development. For example, Policy LU-2.8 provides: "Preserve and provide incentives for mixed residential/commercial development at appropriate locations." Community Plan at LU-32. Likewise, Policy LU-2.9 provides: "Locate higher density residential development in appropriate areas that are situated to promote safer and livelier commercial districts." Id. There is no definition of what constitutes "appropriate" and no indication that such development should occur within Transit Priority Areas.

B19-5 Section 2.3 of the Community Plan does reference "mixed-use transit corridors," but again there is no reference to Transit Priority Areas. *Id.* at LU-38 – 39. Nor is any consideration given to the fact that several areas currently do not have transit meeting the definition of a Transit Priority Area (a copy of an April 11, 2016 letter from Dave Potter on this issue is attached and hereby incorporated by reference). The EIR fails to acknowledge the impacts associated with development that is inconsistent with the CAP and General Plan requirements.

B19-6 Also, the EIR fails to consider shade and shadow impacts, despite the fact that it would result in a 74 percent increase in multi-family residential units.

B19-7 Additionally, the EIR acknowledges "an adverse impact because of the population-base park land deficit," but claims "impacts to land use would be less than significant." EIR at 6.1-10. However, the Community Plan acknowledges the "General Plan standard is to provide a minimum of 2.8 usable acres of public park land per 1,000 residents." Community Plan at RE-127. It states that existing parks and "park equivalencies" constitute 18.21 acres. *Id.* at RE-133. But with population in 2010 of 36,750 people (*id.* at IN-11), the General Plan standard required a <u>minimum</u> of 102.9 acres of park land. The Community Plan projects <u>some</u> additional park land and "park equivalencies" for a total of 55.06 acres, but this results in a shortage of over 100 acres when accounting for the project more than doubling of population. *Id.* at RE-133. In other words, at build-out, the Project would result in slightly more than one-third of the General Plan <u>minimum</u> standard for park land. There is no legitimate way to characterize this as anything other than a significant impact.

7.1

B19-4 Section 6.1 of the PEIR analyzes potential impacts to land use resulting from build-out of the proposed Uptown CPU, including potential conflicts with applicable local and regional plans and policies. As discussed in the PEIR, the proposed Uptown CPU is intended to further the goals of the General Plan and Climate Action Plan (CAP) through policies specific to the individual community and neighborhood characterizes and was found to be consistent with the General Plan and CAP. The policies noted in this comment refer to "appropriate" locations. To determine appropriateness, the goals listed for each element of the proposed Uptown CPU would be consulted. For example, goals of the proposed Land Use Element include, but are not limited to: active commercial districts that benefit from a sustainable level of residential density and multiple mobility options; compatibility of uses within established neighborhoods; preservation of structures with potential historic significance; and active pedestrian-oriented commercial areas; retention of residential neighborhood character.

RESPONSE

B19-5 This comment notes that several areas planned for mixed-use currently do not have transit meeting the definition of a Transit Priority Area. However, the designation of a Transit Priority Area does not preclude other areas from being served by transit and areas not designated Transit Priority Areas may otherwise be corridors for pedestrian, bicycle, or transit travel. This comment refers to an attachment which is a letter from David Potter regarding the Climate Action Plan Transit Priority Area Map. The following e-mail response from Nancy Bragado was provided in response to this letter. Although the letter was not written regarding the City on May 2, 2016 is included here for informational purposes only:

B19-5 (cont.)
From: Bragado, Nancy
Sent: Monday, May 02, 2016 3:02 PM
To: 'NOTICE' < <u>davidapott@aol.com</u> >
Cc: Murphy, Jeff < <u>Murphyl@sandiego.gov</u> >; Hansen, Mike < <u>MHansen@sandiego.gov</u> >; Graham,
David < <u>GrahamD@sandlego.gov</u> >
Subject: RE: Climate Action Plan Transit Priority Area Map
Dear Dave,
This is in response to your letter to Mayor Faulconer and Councilmember Alvarez
dated April 11, 2016. In your letter you questioned the accuracy of the Transit
Priority Area (TPA) map included as Appendix B of the City's Climate Action Plan
(CAP).
We reviewed your analysis and the resources you consulted, and found that you
based your conclusions on a SANDAG map showing transit lines with ten minute or
better all-day service (see attached). In contrast to prepare the TPA man staff
consulted SANDAG data identifying transit lines with minimum 15 minute
frequency during merning and afternoon peak commute periods. The 15 minute
requency during morning and alternoon peak commute periods. The 15 minute
standard is what is included in the SB 743 (California Public Resources Code
Sections 21099 and 21064.3) definition incorporated into the CAP.
In addition, you questioned whether TPAs that span park areas, and other
locations without proposed or permitted housing, should be included on the TPA
map. The City is not proposing residential development in parks. It is however,
desirable to provide transit services to highly-frequented destinations including
parks. The General Plan City of Villages strategy calls for growth to be focused in
mixed use villages connected by high-quality transit. To implement the General
Plan and CAP mode share goals staff recommends focusing housing employment
and civic uses into TPAs. General Plan Policy 11-4 6 states that "some villages may
have an amplement constraint while others may be major charping
have an employment orientation, while others may be major shopping
destinations, or primarily residential in nature." The appropriate mix, intensity and
location of uses is to be determined at the community plan level. Please note that
the CAP TPA map is intended to serve as a citywide illustrative and does not
replace the need to consult the appropriate community plan for land use
recommendations.
We are in the process of updating the TPA map to reflect the revised transit system
included in San Diego Forward: The Regional Plan, adopted by the SANDAG Board in
October 2015. We would be happy to share with you the source data we
requested from SANDAG for the man undate
Please feel free to contact me if you would like to discuss further
Sinceraly
Sincerely,
Nancy
Nancy Nancy Bragado
Deputy Director
City of San Diego
Planning Department
(619) 533-4549
www.sandiego.gov

B19-5 (cont.)

This comment also expresses concerns with the proposed Uptown CPU's ability to meet the requirements of the General Plan and the CAP. The proposed Uptown CPU is consistent with the General Plan policies as it presents site-specific recommendations to both implement Citywide goals and policies and address community needs. While the proposed Uptown CPU would reduce residential density in some areas, it would also increase density in others. Lower residential densities in some areas are required to ensure that the bulk and scale of development maintain the existing neighborhood character as well as public views of canyons and open space. The proposed land uses locate the highest intensity uses along transit corridors where existing and future commercial, residential, and mixed-use development can support existing and planned transit investments in the community. Commercial uses are also used strategically by the proposed Uptown CPU to encourage commercial uses along transit corridors. This transitoriented development pattern is necessary to meet the goals of the General Plan's City of Villages Strategy and the CAP. Therefore, placing lower-density, single-family residential uses outside near canyons and where transit and mixed uses are generally less common, and placing higher-density residential uses along main transit corridors and near mixed-use commercial and employment areas would further the goals of the City of Villages Strategy and the CAP. Additionally, while redevelopment, by its nature, causes temporary displacement, the proposed Uptown CPU would not result in the permanent displacement of residences. See also Section 6.1.3 and Section 6.5.3 for discussions on the proposed CPU's consistency with the General Plan and CAP.

- B19-6 Issue 2 Neighborhood Character of Section 6.2.3, Impact Analysis, of the PEIR discusses the proposed Uptown CPU's potential impacts associated with substantial alteration (e.g., bulk, scale materials, or style) to the existing or planned character of the community. At a program-level of analysis, it is not possible to evaluate site specific shade and shadow impacts of future development and the height, design and specifications of future development is not known. However, the proposed PEIR does address compatibility between mixed-use development and single family land uses. Specifically, Section 6.2 of the PEIR, under Issue 2 addresses neighborhood character and discusses that the proposed Urban Design Element policies that would ensure compatibility with regard to bulk and scale. Additional detail was added in this section to explain how the proposed CPU Urban Design Element policies would ensure compatible transitions between higher density areas and lower density areas and avoid creation of excessive shade or shadows (e.g., by applying building setbacks and upper-story stepbacks, for example).
- B19-7 Community parks and park equivalencies are discussed in Section 6.12, Public Services and Facilities. Though there would be a deficiency in park and park equivalences at build-out of the proposed Uptown CPU, the existing conditions include a deficit in parks and park equivalencies. In addition, through the proposed Uptown CPU effort, 37.40 acres of proposed new population-based park land and park equivalency sites have been identified. The policy framework provided by the proposed Uptown CPU supports acquisition and development of new public parks and park equivalencies, and encourages new private development to include recreational facilities. At this program-level analysis, it is appropriate to assume that policy support would increase the acreage of population-based parks in the CPU area at build-out. Lastly, the project does not include construction of new recreational facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact associated with the construction of new facilities in order to maintain performance objectives for parks.

B19-8

B19-9

RESPONSE

City of San Diego August 5, 2016 B19-9 Page 3 of 5 The EIR's discussion of impacts to public services and facilities is likewise flawed, as it fails to acknowledge the significant impacts associated with failing to meet the minimum park land standard. EIR at 6.12-16-17 & 20. The EIR fails to adequately analyze traffic impacts. The EIR acknowledges that the Project would result in significant impacts. EIR at 6.3-53. Indeed, many intersections, roadway segments and ramps are operating at LOS E or worse. SB743 will soon be a requirement for California. The study does not provide for a Vehicle Miles B19-10 Traveled ("VMT") analysis. A project of this size should include a VMT calculation at a minimum. The EIR notes that the traffic study recommended "improvements that would B19-10 mitigate or reduce roadway segment and intersection impacts." EIR at 6.3-48. And the traffic study notes (p. 6-1) that the recommended mitigation "would restore operations to LOS D or better at all locations." However, the EIR notes that only four mitigation measures were adopted, but there is no showing that such mitigation is infeasible. It further states: "There is no funding mechanism for the remaining measures not included within the IFS." EIR at 6.3-53. CEQA's "substantive mandate" requires agencies to refrain from approving projects with significant effects where there are feasible mitigation measures or alternatives that can lessen or avoid those effects. Mountain Lion B19-11 Foundation v. Fish and Game Comm. (1997) 16 Cal.4th 105, 134. "[T]he Legislature has I declared it to be the policy of the state 'that public agencies should not approve projects as proposed if there are feasible alternatives or feasible mitigation measures available which would substantially lessen the significant environmental effects of such projects" Uphold Our Heritage v. Town of Woodside (2007) 147 Cal.App.4th 587. 597 - 98 (citations omitted).

The EIR and traffic study fail to analyze impacts after mitigation for freeway B19-11 segments and ramp meters. However, the traffic study (p. 6-1) acknowledges that "none of the freeway impacts would be fully mitigated by the" Project. Again, there is no analysis of feasible mitigation to address these impacts.

B19-12 The EIR claims: "No mitigation measures are identified for impacts to freeways because freeway improvements are not within the authority of the City." EIR at 6.3-46. The California Supreme Court has ruled that an agency may not avoid analyzing and mitigating the impacts of its project merely on the basis that it did not have the authority to address such impacts:

> [I]f campus expansion requires that roads or sewers be improved, the Trustees may do the work themselves on campus, but they have no authority to build roads or sewers off campus on land that belongs to others. Yet the Trustees are not thereby excused from the duty to mitigate or avoid [the project's] off-campus effects on traffic or wastewater management, because CEOA requires a public agency to mitigate or avoid its projects' significant effects not just on the agency's own property but

B19-8 See response to comment B18-7.

- The City is in the process of considering how to integrate Vehicle Miles Traveled (VMT) into its transportation analyses pursuant to Senate Bill 743. However, at this time a final methodology and approach has not been adopted by the City and is not part of the City's CEQA Thresholds. Thus, a VMT analysis was not provided.
- The referenced measures that are not recommended are included within the Draft PEIR for purposes of identifying what measures *could* be implemented that would reduce the identified significant transportation impacts to a less than significant level. Section 6.3.5, Mitigation Framework, of the Final PEIR has been revised to clarify that these mitigation measures would be inconsistent with the proposed Uptown CPU.
- Pursuant to Section 15126.4 of the CEQA Guidelines, mitigation measures must be fully enforceable. The mitigation measures identified to reduce impacts to freeway segments are not carried forward as recommended measures because they are not within the authority of the City and are therefore infeasible.
- The City has not avoided analyzing and mitigating impacts of its B19-12 projects based on its inability to implement freeway segment mitigation measures. Potential impacts to freeway segments resulting from the proposed Uptown CPU are appropriate disclosed in Issue 1 Traffic Circulation, f) Freeway Segments of Section 6.3.3, Impact Analysis, and Section 6.3.4 Significance of Impacts. In addition, mitigation measures that would reduce potentially significant impacts to freeway segments were identified and appropriately disclosed in Section 6.3.5, Mitigation Framework, of the PEIR. However, as previously stated, Section 15126.4 of the CEQA Guidelines requires that mitigation measures be fully enforceable. Measures are included for each significant

B19-12 (cont.)

impact; however only those measures included within the SANDAG RP could be feasible to implement because only those improvements are supported and scheduled for future funding and implementation. Thus, as future development is proposed within the Uptown CPU area, developers could contribute fair share contributions towards those specified improvements. Language has been added to the Final PEIR as follows:

At the project-level, significant impacts at locations outside of the jurisdiction of the City could be partially mitigated in the form of transportation demand management (TDM) measures that encourage carpooling and other alternative means of transportation consistent with proposed CPU policies. Fair share contributions could also be provided toward the construction of the following projects that are included in the SANDAG's Regional Plan (RP):

- Operational improvements along I-8 between I-5 to SR-15 (TRANS 7.3-15)
- Construction of managed lanes along SR-15 between I-805 and SR-94 (TRANS 7.3-16)
- Construction of managed lanes along I-805 between SR-8 to SR-163 (TRANS 7.3-17)
- Construction of managed lanes along SR-94 between I-5 to I-805 (TRANS 7.3-18)

B19-12 cont. B19-13	 City of San Diego August 5, 2016 Page 4 of 5 <u>"on the environment</u>" with "environment" defined for these purposes as "the physical conditions which exist <u>within the area which will be affected</u> <u>by a proposed project.</u>" <i>City of Marina v. Board of Trustees of the California State Univ.</i> (2006) 2006 39 Cal.4th 341, 360 (emphases in original) (citations omitted). The EIR fails to adequately analyze greenhouse gas emission impacts. On April 29, 2015, Governor Brown issued Executive Order B-30-15, which establishes a "new interim statewide greenhouse gas emission reduction target to reduce greenhouse gas emissions to 40 percent below 1990 levels by 2030" The EIR does not address compliance with Executive Order B-30-15. <i>See also Center for Biological Diversity v.</i> 	B19-13	This comment makes reference to Executive Order (EO) B-30-15 (2030 Statewide GHG Emissions Goal). Section 5.5.2.2 of the PEIR provides the regulatory background for EO B-30-15. Significance thresholds used in the evaluation of greenhouse gas (GHG) impacts are discussed in Section 6.5.2, Significance Determination Thresholds. As discussed in the PEIR, implementation of the City's Climate Action Plan (CAP) would result in Citywide GHG reductions consistent with its proportionate share of Statewide GHG emissions targets. Because the proposed Uptown CPU is consistent with the City's CAP, it is consistent with EO B-30-15. As such, the City disagrees that the PEIR failed to adequately analyze greenhouse gas emission impacts.
B19-14	Dept. of Fish and Wildlife (2015) 62 Cal.4 st 204, 229. The EIR fails to adequately analyze water supply impacts. There is an inadequate showing of water supply for the Project. The California Supreme Court recently identified three "principles for analytical adequacy under CEQA": (1) "CEQA's informational purposes are not satisfied by an EIR that simply ignores or assumes a solution to a problem of supplying water to a proposed land use project"; (2) "an adequate environmental impact analysis for a large project, to be built and occupied over a number of years, cannot be limited to the water supply for the first stage or the first few years"; and (3) "the future water supplies identified and analyzed must bear a likelihood of actually proving available An EIR for a land use project must address the impacts of <u>likely</u> future water sources, and the EIR's discussion must include a reasoned analysis of the circumstances affecting the likelihood of the water's availability." <i>Vineyard Area Citizens for Responsible Growth, Inc. v. City of Rancho</i> <i>Cordova</i> (2007) 40 Cal.4 th 412, 430 – 32 (emphasis in original) (citations omitted). The EIR fails to comply with these mandates.	B19-14	Potential impacts to water supply are analyzed in Issue 1 Water Supply of Section 6.13.3, Impact Analysis. Appendix K, Water Supply Assessment, concludes that there is sufficient water supply to serve the proposed Uptown CPU's water demands in normal, single-dry year, and multiple-dry year forecasts. As discussed in Appendix K, the projected level of water use associated with the proposed Uptown CPU was determined to be within the regional water resources planning documents of the City, Water Authority, and Municipal Water District, which identify current and future
B19-15	The EIR fails to adequately analyze growth inducement impacts. The EIR claims the Project would not result in growth inducement because "the population in the proposed Uptown CPU area will grow whether or not the proposed" Project is adopted. EIR at 8-3. But this statement conflicts with the analysis of alternatives, which specifically acknowledges that the Lower-Density Alternative would result in "2,650 persons less than the proposed" Project. <i>Id.</i> at 10-34.		water supplies and necessary actions to develop these supplies. Build-out projections for the proposed Uptown CPU are consistent with the growth projections used for the City's 2015 Urban Water Management Plan, and once adopted, the proposed Uptown CPU would be considered in the next cycle of the City's water supply planning. Therefore, the PEIP identifies and discloses
B19-16	CEQA requires that an EIR "produce information sufficient to permit a reasonable choice of alternatives so far as environmental aspects are concerned." San Bernardino Valley Audubon Society v. County of San Bernardino (1984) 155 Cal.App.3d 738, $750 - 51$. "[T]he discussion of alternatives shall focus on alternatives to the project or its location which are capable of avoiding or substantially lessening any significant effects of the project, even if these alternatives would impede to some degree the attainment of the project objectives, or would be more costly." CEQA Guidelines § 15126.6(b). "Without meaningful analysis of alternatives in the EIR, neither the courts nor the public can fulfill their proper roles in the CEQA process." Laurel Heights		that an adequate water supply would be available to support build-out of the proposed Uptown CPU.

B19-15	While the Lower-Density Alternative would result in a reduced residential population than the proposed CPU, there would still be population growth in the community under that alternative. The fact that an alternative would result in a lower population does not contradict the conclusions of the growth inducement discussion of Chapter 8.
B19-16	discussion of Chapter 8. This comment is informational in nature and does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.

City of San Diego August 5, 2016 Page 5 of 5

- B19-16 *Improvement Assoc, v. University of California* (1988) 47 Cal.3d 376, 404. The EIR fails to comply.
- B18-17 CEQA contains a "substantive mandate" that agencies refrain from approving a project with significant environmental effects if "there are feasible alternatives or mitigation measures" that can substantially lessen or avoid those effects. Mountain Lion Foundation v. Fish and Game Comm. (1997) 16 Cal.4th 105, 134; Pub. Res. Code § 21002. It "requires public agencies to deny approval of a project with significant adverse effects when feasible alternatives or feasible mitigation measures can substantially lessen such effects." Sierra Club v. Gilroy (1990) 222 Cal.App.3d 30, 41. Here, the Density Redistribution Alternative meets most of the Project objectives while being the environmentally superior alternative. EIR at 10-40.
- B19-18 The EIR fails to propose feasible mitigation and alternatives that can lessen or avoid the significant Project impacts. *City of Marina v. Board of Trustees of the California State Univ.* (2006) 2006 39 Cal.4th 341, 360; *see also* CEQA Guidelines § 15126.6(b). For example, the EIR acknowledges significant raffic impacts because the timing of traffic improvements is unknown. EIR at 6.3-53. Likewise, the Project will result in an over 100-acre shortfall of parks. Community Plan at RE-133. Yet the EIR fails to consider requiring the timing of aspects of Project development to await the construction of necessary traffic and park improvements.
- B19-19 Furthermore, the Project and its objectives are defined too narrowly, thereby resulting in a narrowing of the consideration of alternatives to the Project. *City of Santee v. County of San Diego* (1989) 214 Cal.App.3d 1438, 1455.
- B19-20 The EIR is sufficiently lacking that the only way to fix these issues is to revise it and recirculate an adequate report.

For the foregoing reasons, Uptown United urges you to reject the Project and EIR as drafted. Thank you for your consideration of these concerns.

Enc.

B19-17 This comment is informational in nature and does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.

RESPONSE

B19-18 Pursuant to Section 15126.6, alternatives considered must avoid or substantially lessen *any* of the significant effects of a project (e.g. alternatives are not required to avoid or lessen *all* significant impacts). Therefore, the range of alternatives considered was sufficient, as each alternative analyzed would result in a reduction in the severity of at least one significant impact of the proposed CPU. Significant impacts to transportation and circulation were not able to be avoided by any alternative. The No Project Alternative, in which the adopted Community Plan would continue to guide development and no CPU would be adopted, would still result in a significant impact (and a slightly greater impact) to transportation, particularly to individual roadways segments, intersections, and freeway segments, intersections, and ramp meters. Under the proposed Uptown CPU, the Final PEIR recognizes that as future development proceeds, each individual development projects would be required to pay development impact fees (DIFs) to fund improvements identified in the IFS. Additionally, the Final PEIR has been revised to state that the City will continue to coordinate with Caltrans and SANGAG, as future project-level developments proceed, to develop potential "fair share" multi-modal mitigation strategies for freeway impacts, as appropriate (refer to section 6.3.5.3 and 6.3.5.4 of the Final PEIR). Thus, through implementation of the IFS and payment if DIF by future development projects as development proceeds, funding would be provided to proportionally fund the identified needs for public facilities such as parks and transportation improvements.

B19-19	This comment suggests that the project objectives are defined too narrowly. However, Community Plans, by their nature, must be specific to the individual community it governs. In addition, the Uptown CPU must implement the General Plan's City of Villages Strategy and the CAP through site-specific recommendations Therefore, the project objectives are appropriately narrow and intended to serve the specific needs of the Uptown community.
B19-20	Comment noted.

Prom:	NOTICE < davidapott@aol.com >
Sent:	Monday, April 11, 2016 11:59 AM
To:	kevinfaulconer@sandiego.gov; davidalvarez@sandiego.gov
Ce:	sherrilightner@sandiego.gov; ToddGloria@sandiego.gov; loriezapf@sandiego.gov; myrtlecole@sandiego.gov; MarkKersey@sandiego.gov; ChrisCate@sandiego.gov; scottsherman@sandiego.gov; martiemerald@sandiego.gov; joe.lacava@gmail.com; murphyj@sandiego.gov; nsbragado@sandiego.gov diversection and the Teoret December
Subject:	LTD TDA adf. Contification htm
Attachments:	LIR IPA.pat, _Certificationhtm
Dear Mayor Faulcone	er and Councilmember Alvarez:
As a planning consult resident, I strongly su established by the ad Working Group. I also two documents toget	tant (and former Deputy Planning Director) in San Diego since 1968 and as a apport the strategies to achieve attainable greenhouse gas reduction targets lopted Climate Action Plan (CAP) and the creation of the CAP implementation o support the adopted Strategic Framework Element of the General Plan. These her will guide San Diego's growth in the future.
Of prime interest was map, along with the s villages with high-der	the Transit Priority Area (TPA) Map included as Attachment B in the CAP. This Strategic Framework Element, will serve during the initial phase of locating nsity housing and mixed use.
A cursory review of t and 21064.3 of the C for the Transit Priority	the TPA Map, however, identified areas that do not comply with Section 21099 california Public Resources Code; these sections were established as the basis y Area Map.
Based on my finding other Transit Priority	during the cursory review, I undertook a more detailed evaluation of a number of Areas.
The results of that ev have any questions.	valuation are included in the attached letter. Please review and contact me if you
Thank you for your c	ourteous attention.
Sincerely,	
David A. Potter 4975 Milton Street San Diego, CA 9211 (619) 275-5120	0

ATTACHMENT 7

RESPONSE

POTTER & ASSOCIATES

4975 Milton Street, San Diego, CA 92110-1252 tel: (619) 507-1415 e-mail: davidapott@aol.com

VIA E-MAIL

April 11, 2016

Mayor Kevin L. Faulconer City of San Diego City Administration Building 202 C Street, 11th Floor San Diego, CA 92101

Councilmember David Alvarez, District 8 Chair, CAP Implementation Working Group City of San Diego City Administration Building 202 C Street, 10th Floor San Diego, CA 92101

Re: Climate Action Plan Transit Priority Area Map

Dear Mayor Faulconer and Councilmember Alvarez:

As a planning consultant (and former Deputy Planning Director) in San Diego since 1968 and as a resident, I strongly support the strategies to achieve attainable greenhouse gas reduction targets established by the adopted Climate Action Plan (CAP) and the creation of the CAP implementation Working Group. I also support the adopted Strategic Framework Element of the General Plan. These two documents together will guide San Diego's growth in the future.

Of prime interest was the Transit Priority Area (TPA) Map included as Attachment B in the CAP. This map, along with the Strategic Framework Element, will serve during the initial phase of locating villages with high-density housing and mixed use.

A cursory review of the TPA Map, however, identified areas that do not comply with Section 21099 and 21064.3 of the California Public Resources Code; these sections were established as the basis for the Transit Priority Area Map.

Based on my finding during the cursory review, I undertook a more detailed evaluation of a number of other Transit Priority Areas.

ATTACHMENT 7

Mayor Faulconer and Councilmember Alvarez Climate Action Plan Transit Priority Area Map April 11, 2016	Page 2		
Basis for the Transit Priority Area Map			
A key land use strategy of the CAP is to reduce vehicle miles tra- all new residential development within Transit Priority Areas. The Area and Major Transit Stop as follows:	veled by locating a majority of e CAP defines Transit Priority		
SB 743 established Section 21099 of the California Public Ro states: " <u>Transit priority area</u> " means "an area within one-half is existing or planned, if the planned stop is scheduled to be of horizon included in a Transportation Improvement Program 450.216 or 450.322 of Title 23 of the Code of Federal Regula	sources Code (CPRC), which mile of a major transit stop that completed within the planning idopted pursuant to Section ations."		
Major Transit Stop, as defined in CPRC Section 21064.3, me existing rail transit station, a ferry terminal served by either a intersection of two or more major bus routes each having a fr minutes or less during the morning and afternoon peak comm	ans: a site containing an bus or rail transit service, or the requency of service of 15 nute periods.		
According to the CAP the Transit Priority Area (TPA) Map inclu the adopted SANDAG 2050 Regional Transportation Plan (RTP	ided as Appendix B is based on).		
Resources Consulted			
In reviewing the Transit Priority Area Map, the following resour	ces were consulted:		
Existing Transit			
 Existing Bus Routes: <u>http://www.sdmts.com/schedules-re routes</u> 	eal-time-maps-and-routes/bus-		
Existing Trolley: <u>http://www.sdmts.com/schedules-real-t</u>	ime-maps-and-routes/trolley		
Planned Transit			
 Future Transit: SANDAG 2050 Regional Transportation Constrained Transit Network, page 2-7 	, Figure 1.1 2050 Revenue		
 Future High Frequency Bus Service: SANDAG 2050 Re Figure A.7 2035 High Frequency Local Bus Routes, pa 	egional Transportation Plan, Ige A-45.		
Areas Not Qualifying for Transit Priority Area			
A review of the areas shown in Attachment B revealed a numbe routes that do not qualify as a Transit Priority Area. The primar qualify is because one or both bus routes do not have a frequence	r of areas around intersecting bus y reason these areas do not y of service of 15 minutes or less		
		RTC-277	

RESPONSE

LETTER

Mayor Faulconer and Councilmember Alvarez Climate Action Plan Transit Priority Area Map April 11, 2016

Page 3

during the morning and afternoon peak commute periods. These areas should not be identified as a Transit Priority Area and should be deleted from Attachment B. See areas enclosed in red on the attached map and identified by the corresponding number.

Non-qualifying areas are discussed below.

• #1 - Route 50 / 105 Intersect (Clairemont Drive and Burgener Boulevard)

The Transit Priority Area Map designates the area around Clairemont Drive and Burgener Boulevard as a Transit Priority Area. However, under current conditions neither Route 50 nor the intersecting Route 105 has a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. And according to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," neither route is slated to have a high frequency of service in the future.

 # 2 - Route 105 / Route "Not Numbered" Intersect (Clairemont Mesa Boulevard and Moraga Avenue)

The Transit Priority Area Map designates the area around Clairemont Mesa Boulevard and Moraga Avenue as a Transit Priority Area. Under current conditions the "Not Numbered" Route on Moraga does not exist and the intersecting Route 105 does not have a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. According to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," the "Not Numbered" Route on Moraga Avenue is slated to have a high frequency of service in the future; however, Route 105 is not slated to have a high frequency of service in the future.

3 - Route 41 / Route 105 Intersect (Governor Drive and Genesee Avenue)

The Transit Priority Area Map designates the area around Governor Drive and Genesee Avenue as a Transit Priority Area. However, under current conditions neither Route 41 nor the intersecting Route 105 has a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. And according to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," neither route is slated to have a high frequency of service in the future.

• # 3.1 - Genesee Avenue Between Rose Canyon and South of Route 52

There are no qualifying areas along Genesee Avenue between Rose Canyon on the north and ¹/₂ mile north of Clairemont Mesa Boulevard on the south. Furthermore, this area encompasses Rose Canyon and San Clemente Canyon (Marian Bear Park), which are preserved as dedicated open space where no residential uses are permitted.

RESPONSE

LETTER

Mayor Faulconer and Councilmember Alvarez Climate Action Plan Transit Priority Area Map April 11, 2016

Page 4

• # 4 - Route 83 / Route 10 Intersect (Washington Street and Hawk Street)

The Transit Priority Area Map designates the area around Washington Street and Hawk Street as a Transit Priority Area. Under current conditions Route 10 does have a frequency of service of 15 minutes during the morning and afternoon peak commute periods and will continue to do so in the future. However, Route 83 has a frequency of service of one hour during the morning and afternoon peak commute periods, and according to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," the route is not slated to have a high frequency of service in the future.

• #5 - Route 854 / Route 115 Intersect (Lake Murray Boulevard and Jackson Drive)

The Transit Priority Area Map designates the area around Lake Murray Boulevard and Jackson Drive as a Transit Priority Area. However, under current conditions neither Route 854 nor the intersecting Route 115 has a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. And according to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," neither route is slated to have a high frequency of service.

6 - Route 854 / Route 115 Intersect (Lake Murray Boulevard and Navajo Road)

The Transit Priority Area Map designates the area around Lake Murray Boulevard and Navajo Road as a Transit Priority Area. However, under current conditions neither Route 854 nor the intersecting Route 115 have a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. And according to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," neither route is slated to have high frequency of service in the future.

• #7 - Route 35 / Route 923 Intersect (Cable Street and Voltaire Street)

The Transit Priority Area Map designates the area around Cable Street and Voltaire Street as a Transit Priority Area. Under current conditions Route 35 has a frequency of service of 15 minutes during the afternoon peak commute period but not during the morning peak commute period. The intersecting Route 115 does not have a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. According to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," Route 35 is slated to have a high frequency of service in the future; however, Route 923 is not slated to have a high frequency of service in the future.

+ #8 - Route 28 / Route 84 Intersect (Rosecrans Street and Canon Street)

The Transit Priority Area Map designates the area around Rosecrans Street and Canon Street as a Transit Priority Area. Under current conditions neither Route 28 nor the

LETTER

RESPONSE

Mayor Faulconer and Councilmember Alvarez Climate Action Plan Transit Priority Area Map April 11, 2016

Page 5

intersecting Route 84 has a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. According to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," Route 28 is slated to have high frequency of service in the future; however, Route 84 is not slated to have a high frequency of service in the future.

#9 - Route 28 / Route 923 Intersect (Rosecrans Street and Nimitz Boulevard)

The Transit Priority Area Map designates the area around Rosecrans Street and Nimitz Boulevard as a Transit Priority Area. Under current conditions neither Route 28 nor the intersecting Route 923 has a frequency of service of 15 minutes or less during the morning or afternoon peak commute periods. According to the 2050 RTP, and specifically Figure A.7 "2035 High Frequency Local Bus Routes," Route 28 is slated to have a high frequency of service in the future; however, Route 923 is <u>not</u> slated to have a high frequency of service in the future.

• # 10 - Rosecrans Street Between Canon Street and Midway Drive

Based on the determination that Route 28 / Route 84 Intersect and Route 28 / Route 923 Intersect are not Major Transit Stops, there should be no designation of Transit Priority Area on Rosecrans Street between ½ mile southwest of Canon Street and ½ mile southwest of Midway Drive.

This was not a comprehensive evaluation of all of the Transit Priority Areas; hence, there may also be other non-qualifying areas.

Other Areas Not Qualifying for Transit Priority Area by Virtue of Their Location

There are intersecting bus lines and transit stops that meet the technical definition of a Major Transit Stop; however, they do not further the cause of reducing greenhouse gas emissions because of their inability to support housing at that particular location. These areas should not be identified as a Transit Priority Area and should be deleted from Attachment B. See areas enclosed in red on the attached map and identified by the corresponding letter.

Areas not qualifying by virtue of location are discussed below.

 A - Route 8 / Route 9 Intersect (Ingraham Street and West Mission Bay Drive/Sea World Drive)

This Major Transit Stop is located with Mission Bay Park where housing is neither proposed nor permitted.

RESPONSE

ATTACHMENT 7

Mayor Fa Climate April 11,	aulconer and Councilmember Alvarez Action Plan Transit Priority Area Map , 2016	Page 6
	B - Transit Stop #1 in Balboa Park	
	This Major Transit Stop is located on Park Boulevard at Preside where housing is neither proposed nor permitted.	ent's Way in Balboa Park
•	C - Transit Stop #2 in Balboa Park	
	This Major Transit Stop is located on Park Boulevard at the Zo housing is neither proposed nor permitted.	o in Balboa Park where
•	D - Harbor Drive Between Pacific Highway and Harbor Island	d Drive
	San Diego International Airport is on the north side of Harbor I on the south side of Harbor Drive. Housing is neither permitted area.	Drive, and port lands are I nor proposed in either
	E – Old Town San Diego State Historic Park	
	Although located adjacent to the Old Town Transit Center, the intended to preserve and recreate the Mexican and early Ameri housing is not compatible with the historic character of the Par	State Historic Park is ican periods. High-density k.
•	F – East Side of Sixth Avenue between Upas Street and I-5	
	This area is located in Balboa Park where housing is neither pr	oposed nor permitted.
Conse	equences of Including Non-qualifying Transit Priority Areas	
Increa area a effecti gas en increa planni chara	asing density in areas that are not consistent with the definiti and major transit stop would be contrary to the CAP goal, w ive land use to reduce vehicle miles traveled for the purpose nissions. In fact, it would have just the opposite effect by pro- ased auto use in areas not sufficiently served by transit. Furt ing under the Strategic Framework Element would be faulty cterized.	ion of transit priority hich is to promote of reducing greenhouse moting growth and hermore, land use y if transit is inaccurately
Showi	ing Transit Priority Areas in areas where housing is either n es a distorted view of where growth should occur.	ot proposed or permitted
Recon	nmendations	
The fo	ollowing measures are recommended to update the Transit Prior	ity Area Map.
LETTER

RESPONSE

ayor Faulconer and Councilmember Alvarez limate Action Plan Transit Priority Area Map pril 11, 2016
Reevaluate each and every Transit Priority Area as shown on the map in Appendix B, correct the errors, and publish a revised version.
Include the following notation on the revised version of the Transit Priority Area Map in Appendix B:
"See larger scale, community-specific Transit Priority Areas in updated community plans."
Work with community planning staff and prepare larger scale community-specific Transit Priority Area Map to be included in each updated community plan.
a practice, of course, it will be necessary to do more that just identify the "availability" of ansit. Truly assessing a reduction in vehicle miles traveled would include (among other hings): assessing when and how transit is actually used, whether there are practical incentives to neourage transit usage, and whether the land uses that are proposed for a particular area provide an appropriate mix of uses that would allow transit to be truly effective.
hope this letter serves to improve the effectiveness of the Climate Action Plan. Please contact ne if you have any questions.
Thank you, and I look forward to hearing about efforts undertaken to update the Transit Priority Area Map.
Sincerely,
Que al A Potter
David A. Potter, AICP
 Council President Sherri Lightner (D1) Councilmembers Zapf (D2), Gloria (D3), Cole (D4), Kersey (D5), Cate (D6), Sherman (D7), and Emerald (D9) Jeff Murphy, Planning Director Nancy Bragado, Deputy Director Joe LaCava, Community Advocate, CAP Implementation Working Group
Attachment: Transit Priority Area Map with Non-Qualifying Areas Enclosed in Red
walk Botton



ATTACHMENT 7



RESPONSE

corridor, to be served by the existing Rapid Bus Route and the planned Mid-City trolley line.

5. The Density Redistribution Alternative has been identified as the "Environmentally Superior Alternative".

- B20-3 6. It is important to provide appropriate transitions between different types of development.
- B20-4 Resolution
 - 1. We recommend the adoption of the Density Redistribution Alternative.

2. We recommend that additional provisions be included in the Community Plan and zoning which will improve the transitions between different densities and heights.

B20-5 B. RESOLUTION ON MOBILITY

It is recommended that more of the mitigation measures proposed by the traffic consultant be added to the Mobility Element. The objective is to reduce the number of intersections and roadway segments that are projected to have "cumulative traffic related impacts".

B20-6 C. RESOLUTIONS ON RECREATION

1. <u>Previous Resolutions.</u> We support the prior resolutions from Uptown Planners, dated 11/19/15, 2/24/16, and 5/3/16.

2. <u>Recreation plan not acceptable.</u>The draft Community Plan Recreation Element, at the overall level, is quite inadequate as a plan for meeting the public recreation needs of the Uptown Community.

B20-7 The plan can be viewed in two parts.

a. The two Recreation $\,$ Centers and the Aquatic Complex appear adequate to serve the projected population.

b. The planned park acreage is completely inadequate to serve the projected population. The proposed new parks and equivalencies are not enough to keep up with the additional development, and make no reduction in the existing deficit.

B2-3 Comment noted. This comment does indicate an inadequacy in the PEIR. The proposed Uptown CPU includes policies addressing building compatibility and transitions between new and existing development (e.g., Policies UD-4.71, UD-4.75, UD-4.80, UD-4.88, and Transition Plane Guidelines shown in Figure 4-11 of the proposed Uptown CPU). Thus, the proposed Uptown CPU does provide appropriate transitions between different types of development.

- B20-4 Comment noted. This comment does indicate an inadequacy in the PEIR. This comment offering the Uptown Planner's support of the Density Redistribution Alternative is noted. Refer to response B20-2 regarding development transitions.
- B20-5 Comment noted. As described in Section 6.3, Transportation and Circulation, many of the mitigation measures identified in the Traffic Impact Study and the Draft PEIR are not recommended for implementation because they would conflict with proposed Uptown CPU mobility element goals and policies related to providing a multi-modal transportation system that supports all types of movement, including pedestrian, bicycle and transit. Only those measures proposed in the Uptown IFS would be consistent with the proposed Mobility Element vision.
- B20-6 Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.
- B20-7 Comment noted. This comment does indicate an inadequacy in the PEIR, but makes a comment regarding the proposed Uptown CPU recreation element. The draft PEIR does disclose a park deficiency in the Uptown community in Section 6.12, Public Services and Facilities. Though there would be a deficiency in park and park equivalences at build-out of the proposed Uptown CPU, the existing conditions include a deficit in parks and park equivalencies. In addition, through the proposed Uptown CPU effort, 37.40 acres of proposed new population-based park land and park equivalency sites have been identified. The policy framework provided by the proposed Uptown CPU also supports acquisition and development of new public parks and park equivalencies, and encourages new private development to include recreational facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in less than significant impacts related to parks.

² of 4

B20-8

U-8 3. Serving increased population. The new Community Plan must provide, at a minimum, sufficient new parks to "keep pace with the Uptown population growth", as stated in the goal on page RE-126. That target can be easily calculated:

	Current	Projected	Increase
Population	36,750	55,700	18,950
Park standard	1	1	2.8 acres per 1000
Additional park land required to serve population increase			53.1 acres
Proposed parks and equivalencies	_		36.85 acres
Additional parks needed		II	16.25 acres

- B20-9 4. Equivalencies. The above figure for "Proposed parks and equivalencies" includes a substantial count of "parks on paper". The Uptown Planners have repeatedly objected to the "double counting" of Balboa Park as part of Uptown parks. Under no circumstances can more of Balboa Park be counted as "equivalencies" for Uptown parks. The additional parkland of 16.25 acres must be actual parkland.
- B20-10 5. <u>Park standards.</u> It should be noted that the City's standard of 2.8 acres of population-based parks (per 1000 residents) is a very low standard. [Nationally, the most common standard for neighborhood and community parks is 5.0 acres per 1000. When the California Quimby Act was established, it provided a standard of 3.0 to 5.0 acres. Therefore, the City of San Diego General Plan standard of <u>2.8 acres</u> should be considered an <u>absolute minimum</u>. To accept less would be to condemn future San Diegons to park-deficient communities.]
- B20-11 6. <u>Substitution</u>. If a specific site shown in the Community Plan is determined to be infeasible, or its acquisition is delayed indefinitely, one or more substitute sites should be identified.
- B20-12 7. <u>Park Needs Assessment Survey</u>. We strongly object to this study to its design and conclusions. The survey was designed to lead to certain conclusions, yet those conclusions are not believable.
- B20-13 8. <u>DMV site for Recreation</u>. Policy RE-1.14 and the accompanying diagram show the current DMV site as proposed "Mixed Use". Pursuant to CA Govt Code 54220, we request that any property released by the State Department of General Services be dedicated by the city for an Aquatics Center and/or Recreation Center.

3 of 4

B20-8 Comment noted.

- B20-9 Comment noted. Parkland equivalencies are appropriate in this area due to Balboa Park's recreational value, use and function, and public accessibility related to the Uptown community. Further, there is not adequate land available outside of Balboa Park to provide for the community's needs.
- B20-10 Comment noted. This comment does not raise an issue with the adequacy of the PEIR. The 2.8 acre per 1,000 resident standard is consistent with the City's General Plan.
- B20-11 Comment noted. Policy RE-1.2 of the proposed Uptown CPU requires the City to pursue land acquisition for the creation of new parks and recreation facilities as opportunities arise. Therefore, potential future parks are not limited to only those identified in the proposed Uptown CPU.
- B20-12 Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.
- B20-13 Request noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.

			T	
B20-14	 <u>Normal Street Linear Par</u> public right of way as a publi & Rec Department. 	<u>k.</u> We support the development of the Normal Street c linear park under the management of the city's Park	B20-14	Comment noted.
	D. QUESTIONS from Upto	own Planners regarding draft EIR		
	Regarding the Alter	natives:		
B20-15	I. The <u>population differentian</u> Lower Density Alternative 2650 in the draft EIR. P Buildout population figure	al between the proposed Community Plan and the e changed. It increased from 810 in February 2016 to Tease explain the reasons for the change in the es.	B20-15	Alternatives considered prior to public review may have differed from alternatives selected for evaluation in the Draft PEIR.
B20-16	2. What is the difference in I Community Plan and the	octential housing units between the proposed Redistribution Alternative?	B20-16	As shown in Table 10-4, the Density Redistribution Alternative would result in 1,585 fewer units than the proposed Uptown CPU.
	Regarding the Clim	ate Action Plan (CAP):	D00 47	
B20-17	1. Tell us how the draft Mob	ility Element meets the CAP. Include timelines.	B20-17	Action Plan (CAP) is analyzed in detail in Section 6 5 Creenhouse
B20-18	20-18 2. Complete the CAP Checklist for the proposed Uptown Community Plan.	Gas Emissions, of the PEIR. The Mobility Element of the proposed		
	Sincerely yours,			bicycling and transit opportunities which is consistent with the
	Leo Wilson			CAP's Strategy 3 (Bicycling, Walking, Transit & Land Use).
	Leo Wilson			
	Chair, Uptown Flanners	Uptown Planners letter, Community Plan Resolutions-A.doex.	B20-18	The City's CAP Checklist only applies to individual development projects subject to discretionary review, and would not apply to a Community Plan that requires a program level review. Note that the measures identified in the CAP Checklist would only be enforceable at the project level. As future development within the Uptown community occurs, individual projects would be required to prepare the CAP Checklist, as applicable.
		4 of 4		

LETTER

RESPONSE

Letter B21 UPTOWN PLANNERS San Diego, CA 92103 UptownPlanners@org (619) 822-5103 August 4, 2016 Mr. Jeff Murphy, Planning Director Mr. Kurtis Steinert, Sr. Environmental Planner City of San Diego 1010 Second Avenue, MS 413 San Diego, California 92101 **Re: Uptown Planners Recommendations Regarding the Draft** Environmental Report Reference to Removing Parking on India Street At the August 2, 2016 meeting, Uptown Planners voted to approve the following motion regarding sections of the draft Environment Impact Report that discuss the possible removal of parking along a section of India Street: Move that the City delete all references in the June 2016 Community Plan Update (the CPU) to the removal of parking and/or sidewalks from India Street in the section from Palm Street to Washington Street, and issue a written statement to be incorporated in the final CPU that there are neither current nor will there be future plans for such removals. The City will not act on those mitigations stated in the CUP Mobility Element Appendix C since they are not recommended. The deletions shall include but not be limited to references contained in: the CPU Impact 6.3-19 (b), 6.3-38, 6.3-50; and Mobility Appendix C Table 13, and Figures 63 & 64. Sincerely yours, Lea Wilson Leo Wilson Chair, Uptown Planners

B21-1

B21-1 Removal of sidewalks and parking along India Street are not recommended as part of the proposed Uptown CPU and associated discretionary actions. Note that the proposed Uptown CPU does not include recommendations for removal of parking on India Street. Policy MO-1.4 supports pedestrian improvements that promote a safe connection along Washington Street between Hawk Street and India Street. Refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).



RESPONSE

	The Uptown Community has been served for the last 30 years by the Mid-City Communities Planned District Ordinance (PDO). The replacement of the PDO with the citywide zoning ordinance does not "provide for development compatible with the pattern of the existing neighborhoods" as specified in the current PDO.		
	The proposed zoning is markedly different from the existing zoning. In many cases the proposed zoning has different height, setbacks, FAR, etc. Even where no change to density is proposed, the proposed new zones would introduce major changes in allowable development.		
	In addition to the quantitative items like height and setbacks, many textual provisions in the PDO have not been addressed in the proposed zoning. An example is Architectural Features, section 1512.0304.		
	The proposed zoning represents an unwarranted shift in the "PDO elimination" plan, which started before the year 2000. The Planning Dept and DSD stated that the purpose was standardization, and that all provisions in the PDO would be carried out in standard zones. Where an existing zone wasn't equivalent, assurances were given that a new zone would be created.		
	THEREFORE, WE RESOLVE that the current PDO continue to provide zoning regulations for the Uptown community. or that new zones be created in the citywide zoning ordinance to reflect all the current regulations in the PDO.		
B22-4	C. PUBLIC FACILITIES (adopted 10/22/15)	B22-4	Comment noted.
	Motion 1. Adopted 10/22/15, to reaffirm the Uptown Planners motion from May 4, 2010. "In order to promote a useful dialogue with the City officials in the Community Plan Update planning process about the kind of development that is wanted in Uptown, Uptown Planners states its oppositionin principle to any substantial increase in density which exceeds existing developmentbeing incorporated into the updated Uptown Community Plan where necessary infrastructure improvements are not adequate to support the development."		
B22-5	<u>Motion 2.</u> Uptown Planners supports the position of Mission Hills Heritage that the concept of "incentive zoning provisions" should be deleted [from the City's draft CP Public Facilities, Services and Safety element] as a proposed solution for addressing the infrastructure deficit.	B22-5	Comment noted. in the PEIR; theref
B22-6	Motion 3.	B22-6	Comment noted.
	<u>Whereas:</u> 1. Adequate public facilities are essential to the urban quality of life in Uptown.		in the PEIR; theref
	2 of 4		

Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.

- 322-5 Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.
- B22-6 Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.

RESPONSE

2. The Uptown community has endured decades of infrastructure deficiencies. These include a lack of neighborhood and community parks, lack of recreation centers, outdated fire stations and libraries, and inadequate transportation infrastructure.

3. Expansion of public facilities has not kept pace with the growing demand since the 1988 community plan was adopted. In particular, parks have become increasingly overburdened, and traffic congestion has worsened.

4. A previous resolution from Uptown Planners on 5/4/10 stated the principle that substantial increases in density, over the existing development, should not be allowed where the infrastructure is inadequate.

Resolution:

1. The Uptown Planners support the goals and principles of the General Plan relating to adequate facilities, in particular the following:

a. Adequate public facilities available at the time of need. (PF-14)

b. Require development proposals to fully address impacts to public facilities and services. (PF-15)

c. Reserve the right and flexibility to use the City's police powers and fiscal powers to impose timing and sequencing controls on new development to regulate the impacts and demands on existing or new facilities and services. (PF-15).

B22-7 Motion 4

The updated Uptown Community Plan must provide the following:

a. A phasing or threshold system which links the approval of new development to the adequacy of public facilities and services.

b. Adequate public facilities means that the facilities and services meet specific thresholds or level-of-service, at the time of need.

c. "At the time of need" means that the facilities and services must be in place to serve new development, or guaranteed, that is, planned, scheduled and funded.

d. The link between project approval and facilities adequacy could be accomplished by various procedures, for example, requiring relevant findings as part of the approval process. B22-7 Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required.

3 of 4

LETTER

RESPONSE

e. Provisions which ensure that the community is involved in decisions about facility financing. An example is the use of Uptown DIF funds for storm sewer replacement, instead of parks and recreation centers.

Sincerely yours. Leo Wilson

Leo Wilson Chair, Uptown Planners

Uptown Flammers letter, 8-4-16.does

4 of 4

	Letter C1			
	From: Jeff Allen <jallen6150@hotmail.com> Sent: Finday, August 05, 2016 11:45 PM To: PLN_PlanningCEQA Subject: Uptown Community Plan Update, PROJECT No. 21002568 / Project No. 38061</jallen6150@hotmail.com>			
C1-1	Hello, I reside near Columbia and Palm streets in Middletown. I am strongly opposed to the proposal to remove the south bound lane on India St. and plans to turn all lanes into a northbound one-way. This will severley impact access into our neighborhood. I urge you to look into directing rental car return traffic onto Pacific Hiway and preserve India St for local residents/businesses and to implement traffic control measures on India St. so that pedestrians can safely cross the street to the T-5 overpass. Thank you for your consideration.	C1-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).	
	Jeff Allen			
	Ĭ			

RESPONSE

ATTACHMENT 7

	Letter C2		
	Ashley Development and Real Estate Lic. 884089 3002 Columbia St, San Diego, CA 92103 Phone: (619) 295-7113 Fax: (619) 683-3737		
	08/04/2016 Kurtis Steiner Senior Environmental Planner City of San Diego Planning Department 1010 Second Ave, MS 413 San Diego, CA 92101 Re: Uptown Community Plan Update: Project No. 21002568/SCH No. pending;		
	Community Area: Uptown; Council District: 3 Dear Mr. Steiner		
C2-1	My name is John Ashley I have lived in Middletown San Diego for 38 years. I have built and own properties on India St, West Palm, and Columbia St, all of which are affected by these planning proposals.	C2-1	Introductory comment noted. The City appreciates individual participation in the public review comment process.
C2-2	First: The removal of the southbound lane from Sassafras to Redwood is absolutely not acceptable. This in affect would funnel all Middletown residents down to a bottleneok at Laurel St making traffic congestion much worse than it presently is. I strongly oppose this proposal.	C2-2	Comment noted. Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	Second: The removal of parking and sidewalk on India St as well as several feet of Building frontage, this in affect would be a disaster. For example, my properties at 2921 India St the front entrance doors are only 11 feet from the back of the sidewalk. These doors are also down steps and in fact are 2 feet below the back of the sidewalk level. Removing much-needed parking spaces from residents who need them badly is thoughtless.	C2-3	This comment makes reference to expenses incurred by a property owner for improvements along a property's frontage. As detailed in the Findings included as an attachment to the Staff Report the mitigation measures TRANS 6.3-18 and TRANS 6.3-19 are infeasible and will not be implemented. Property improvements such as
C2-3	Please note: Example, if a small owner, developer wants to develop a small 100' deep by 50' frontage lot the Planning Dept. will insist that the driveway width kept to a bare minimum to provide street parking by the owner. In addition, the planning dept. will require the following:		sidewalks, curb, and gutter along India Street would not be impacted by the identified mitigation measures.
	JIFage		

RTC-294

LETTER

RESPONSE

	 Drive way and sidewalk will have to comply with up to date handicap requirements Thus, the sidewalk kerbs, and gutters be demolished required by the city, and new driveway, sidewalk, kerbs, and gutters will be required to city codes and paid for by the owner. New landscaping be provided per city requirements and paid for by owner If the lot is situated on a corner then a handicap ramp will be required per city code and paid for by the owner All this is a great expense and inconvenience to the owner 		
C2-4	The City Planning Dept. now wants to take this parking away; I regard this as shameless and irresponsible. In addition, all cars parked along the east side of India St. acts as a buffer from northbound trucks and cars speeding along India St at 50 mph to 70 mph. It is easy to image how dangerous and scary it would be for all the pedestrians and small business owners' customers along the India St., and just 3 feet away from pedestrians.	C2-4	Comment not India Street M these respons
C2-5	Question: Are we now "The Middletown parking lot"?	C2-5	This comment
	For the last 25 to 30 years the City of San Diego and the Port District has gradually been encouraging parking lots, and parking Structures into Middletown without any thought to residents, and much to the stress of owners.		the PEIR.
C2-6	Note: Structure now known as 'Wally Park"	C2-6	This comment
	I can't believe that the City planning Dept. allowed a huge parking structure to have the <u>entrance and exit</u> 30' from the level crossing tracks for the trolley, and the coaster. The Wally Park shuttle buses drive dangerous across two traffic lanes to the opposite side parking lot entrance which also about 30' from Trolley tracks. Wow! This causes a major traffic jam including drivers having to stop on the trolley tracks.		not suggest an
C2-7	Note: Beautiful Car Rental Structure	C2-7	Comment not
	The City Planning Dept. gave the Port Authority Planning permission to build a huge car rental structure on the west side of pacific highway. The port authority has or had plenty of land, and could have built their own exit roadway North to Washington St. to relieve the traffic situation. However, the port of authority built a roadway extending Sassafras as egress and ingress from this huge structure, which empties 14 rental cars at a time onto Sassafras St that effectively has to cross three traffic lights including a trolley stop all cars are heading to India St causing extreme traffic congestion. In fact, the traffic lights from the Car Rental at Pacific Highway, the Sassafras St, and the India St are currently not effectively moving the traffic to prevent these congestions.		India Street M these respons circulation iss outside of the
	2 1 - 0 2 =		

- 2-4 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C2-5 This comment does not suggest an inadequacy in the analysis of the PEIR.
- C2-6 This comment references the Wally Park parking structure and does not suggest an inadequacy in the analysis of the PEIR.

C2-7 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6). Traffic circulation issues associated with the rental car return traffic is outside of the scope of this PEIR.

C2-8 Conclusion

Is it now time for the City to recompense Middletown owners and Residents

The Southbound lane should be left in place to give Middletown residents a reasonable access to their properties. A traffic light can be provided at Redwood St, which will provide safe crossing from the southbound lane to W Redwood St allowing ingress and egress from India to Redwood and Redwood to India, also providing additional U-turn access to W Spruce. The light would also reduce the speed of traffic coming north from freeway.

C2-9 West Palm

A pedestrian traffic light placed at West Palm Street will allow safe crossing to go over the bridge to Kettner Blvd to reach the Middletown Trolley Station. This light would also slow down crazy people driving from the freeway, and the airport at 70 mph on India St.

Respectfully,

John Ashley

C2-8 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

> This comment also suggests installing a traffic light on Redwood Street. The Draft PEIR for the proposed Uptown CPU is a planninglevel document. The goals stated in the proposed CPU's Mobility Element are to create "safe, walkable neighborhoods, which utilize pedestrian connections and improved sidewalks to create a comfortable pedestrian experience". The City proposes Mobility Element Policy MO-4.9 which would implement road diets and traffic calming measures where appropriate to improve safety and quality of service, and increase walking and bicycling in Uptown, and Mobility Element Policy MO-7.13 which supports on-street parking on all streets in order to support adjacent uses and enhance pedestrian safety and activity. As future development occurs these policies would be implemented through construction improvements, such as new traffic lights, that would provide safer crossing for pedestrians that would be consistent with the abovementioned policies in the proposed CPU Mobility Element.

C2-9 This comment suggests a traffic light on West Palm Street. Refer to the second paragraph in response to comment C2-9.

3 [P a g =

ATTACHMENT 7

		Γ]		
	From	Both Bocker chettyberker@checkelabal.pain	Letter C3		
	Sent: To: Subject:	Saturday, August 06, 2016 4 50 PM PLN_PlanningCEQA Uptown Community Plan Update; #21002568/SCH No. Pending			
C3-1	Traffic has increase do not make it worse Washington and Lau eliminating access to negatively impact of	d significantly in Middletown since the opening of the Rental C e by changing India Street into a one-way 3 lane thoroughfare urel St. Taking away parking and sidewalks, existing building to o homes by removing the southbound lane from Sassafras to b or community.	ar Center Please between fronts and Redwood would	C3-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	Use what you have-	-Pacific Highway-for access. Do not destroy			
	Thank you for your t Betty Becker 3306 Hawk St., San	Diego, CA 92103			

	· · · · · · · · · · · · · · · · · · ·		
	Letter C4		
	3327 Hawk St. San Diego, CA 92103 August 3, 2016		
	Kurtis Steiner Senior Environmental Planner, City of San Diego Planning Department 1010 Second Ave., MS 413 San Diego, CA 92101		
	Re: Project Name: Uptown Community Plan Update Project # 21002568/SCH No. Pending Community Area: Uptown Council District: 3		
C4-1	Please read this letter! I am very concerned about the India Street proposal between Washington and Laurel. Middletown is a wonderful neighborhood. Help to keep it that way,	C4-1	Introductory comment noted. The City appreciates individual participation in the public review comment process.
C4-2	The building of the Rental Car Center on Pacific Highway has caused a significant increase in traffic in our area. Pacific Highway is just that—a highway, built to handle a large volume of traffic. India Street is a residential/small business street. It is an integral part of Middletown. The proposal to remove sidewalks and parking on India as well as several feet of existing building fronts and residences would be devastating to our neighborhood and is not necessary. Pacific Highway is the logical choice of entrance and exit to and from the Rental Car Center. On the west side, Pacific Highway and Washington Street go directly to Interstate 5 North. On the east side, Pacific Highway leads directly to downtown San Diego and Interstate 5 South.	C4-2	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	The idea to remove the southbound lane on India Street from Sassafras to Redwood is not feasible. The purpose of this road is to provide access to the many homes in Middletown. If you will look at the width of Redwood St. on your maps, you will see that it is very wide. It is built to handle all of the residential traffic for this area. At the top, Horton Ave. and Hawk St. are also very wide, merging into Redwood, for this purpose. All other ways to access our homes are small, narrow, winding residential streets. Attempting to funnel large numbers of automobiles through them would not be safe or acceptable.		
	A sign on Sassafras Street says "Making our neighborhoods better". Turning India Street into a one-way, 3 lane thoroughfare is not the answer. Use the Highway, Pacific Highway, for the purpose it was intended.		
	Sincerely,		
	Clizabetu W. Beeker		
	Ellzabeth Becker Middletown Homeowner		
	RTC	-298	

LETTER

	Letter C5		
	Mr. Kurtis Steiner 1010 Second Ave MS 413 San Diego, ca 92101		
	Mr. Steiner,		
	Uptown Community Plan Update Project Number 21002568/SCH Uptown Council District 3		
	Re; India Street between Washington and Laurel.		
C5-1	I realize we are just a small area in the grand scheme of greater San Diego but each and every community is important. Let me ask, how can we have an impact on this project that will greatly effect our area?	C5-1	Comment noted. The Draft PEIR is distributed for review to the public for the purpose of providing comments "on the sufficiency of the document in identifying and analyzing the possible impacts on
C5-2	The number of houses and businesses that use the southbound lane on India Street for access has to be great. The inconvenience, time, extra mileage and added traffic on residential streets to access their homes or destinations would be a constant problem.		the environment and ways in which the significant effects of the project might be avoided or mitigated" (Section 15204, CEQA Guidelines).
C5-3	The job of finding an alternate route for the airport traffic is yours and I wish you well or I feel there will be a very big uproar.	C5-2	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C5-4	If you go through with the planned project your bigger headache will be handling the traffic in the neighborhood streets		
	I am opposed to making India Street a throughfare.	C5-3	This comment references alternate routes to airport traffic, but does not suggest an inadequacy in the analysis of the PEIR.
	Sincerely Qary Becker 3327 Hawk Street San Diego, ca 92103	C5-4	This comment does not suggest an inadequacy in the analysis of the PEIR.

	Letter C6		
	From: Tina Belinsky <tina@villagehatshop.com> Sent: Thursday, July 28, 2016 12:11 PM To: PLN_PlanningCEQA Subject: re: Uptown Community Update</tina@villagehatshop.com>		
	Mr Kurtis Steinert,		
C6-1	I own the building on 3443 India Street, San Diego, CA 92103. It has come to our attention that you plan to remove all parking on our one way street, in addition to taking up to 5 extra feet of sidewalk space. I reviewed the photos and your plans. This plan is completely untenable, and as a Business woman who owns property in the Mid-town area I must reject you plan for the following reasons?	C6-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C6-2	 I have a large parking lot adjacent to my building. My business will not be impacted as much as all the rest of the businesses that line India Street. They have no parking whatsoever. Where are there customer supposed to park? Are you suggesting that they all quit or move their businesses? 	C6-2	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C6-3	My employees and customer that exit my parking lot would be subject to speeding traffic whizzing by. It is already a hazard to exit my parking lot with a parking lane as a buffer. My removing the parking lane you are endangering lives.	C6-3	Comment noted. Please refer to the Master Response Regarding
C6-4	3. You plan to remove 5 feet of sidewalk would put the traffic right up to most of the buildings that line the street, and appears from your pictures even venture into some of the buildings, which would require demolition. I have never seen a street where the traffic is that close to buildings before. What are you thinking?		India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C6-5	4. Are you aware of the volume walking traffic that uses the sidewalks on India Street? I have recently read the health and welfare of a community is related to the availability of sidewalks. How would people travel on foot? It is not as if they could walk up the next parallel street. The next parallel street is private, very steep, and not continuous. San Diego is a city that requires you own a car, because the public transportation options are s limited. I was under the impression we were moving to improve this over time, not make it worse.	C6-4	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C6-6	5. I have also witness numerous accidents right outside my window on the 2 nd floor. More speed and more traffic will increase the possibility for accidents, and probably increase accidents that crash right into the buildings tha line the street. Please feel free to contact me, or visit me at Village Hat Shop anytime.	C6-5	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	Thank you, Tina Belinsky	C6-6	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	Village Hat Shop		
	I http://www.villagehatshop.com http://www.berets.com vilhat@villagehatshop.com		

I FTTFR

IMG 0068.JPG; middletownnotice pdf

S B < Ilbcsbigos@amail.com> Saturday, July 30, 2016 6:11 AM

PLN_PlanningCEQA

From:

Sent:

Attachments:

To: Subject:

C7-1

Letter C7

Comment noted. Please refer to the Master Response Regarding C7-1 India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

RESPONSE

- C7-2 Comment noted.
- C7-3 Comment noted.
- C7-4 Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6). The goals stated in the proposed CPU's Mobility Element are to create "safe, walkable neighborhoods, which utilize pedestrian connections and improved sidewalks to create a comfortable pedestrian experience". The City proposes Mobility Element Policy MO-4.9 which would implement road diets and traffic calming measures where appropriate to improve safety and quality of service, and increase walking and bicycling in Uptown, and Mobility Element Policy MO-7.13 which supports on-street parking on all streets in order to support adjacent uses and enhance pedestrian safety and activity. As future development occurs these policies would be implemented, and construction improvements, such as "pedestrian zebra crossings", would provide safe crossing for pedestrians and would be consistent with the above-mentioned policies in the proposed CPU Mobility Element.
- C7-5 Comment noted. This comment does not suggest an inadequacy in the analysis of the PEIR. However, please refer to the master response regarding India Street mitigation measures included in the introduction to these responses to comments. Also see the Staff Report for a discussion of the extensive public outreach that has been done regarding the proposed Uptown CPU. Also refer to Section 4.2, Community Outreach and Plan Development, of the PFIR.

involving expanding India to turn it into an Interstate 5 ramp for the airport rental car facility. In fact, we cannot understand why the city does not easily see that, if the city instead required the airport authority to build a real off ramp and took a few traffic calming measures, this area would be prime for development as an extremely close, affordable and walkable section of town. The area the proposal is turning into a ramp is less than a tenth of a mile from the Blue Line Trolley, less than C7-2 a quarter mile from Little Italy, less than a mile from downtown, less than a mile from the airport, less than a mile from the bay, and right next to Interstate 5. India also flows right from downtown, through well developed Little Italy, and up to Washington. Because it historically served partially as a ramp for exiting the 5 to reach Washington, it is a busy street and significantly underdeveloped. As I already stated, this area abuts Little Italy, Banker's Hill and Downtown and is part of the last mile of an C7-3 already constructed trolley line. Yet, in order to take public transportation from Middletown, most of the residents, who live above this noisy, unsafe, busy area, scale a steep hill, and cross two extremely hazardous streets (India and Ketter). This plan not only ignores primed undeveloped land at the bottom of the hill, it is about to make both of those C7-4

Fwd: Uptown Community Plan Project No. 21002568

I would like to take a moment to register the strong opposition my husband and I have to the current plans

roads even busier. Please note, there aren't even any pedestrian zebra crossings on Kettner from the highway's pedestrian bridge. Residents have to be extremely adept at real life Frogger if they want to take the Blue Line.

The city of San Diego is spending millions of dollars building new trolley lines, focusing affordable housing C7-5 along the transportation corridors we have, and attempting to convince reluctant San Diegan's to take public transportation. Yet, for some reason, they are totally blind to a walkable area just lying right under their noses because it is next to a highway that is not shielded.

This prime area of the city is significantly underdeveloped right now. Because of the current traffic, airport noise and the lack of protection from the 5, India in this area is lined with small old buildings, empty lots and businesses hanging onto the edge. While the city cannot build up because of the airport, it can certainly better utilize this district for density.

Instead of tossing away a residential/business district so close to downtown and so perfectly primed for smart growth, our city should be focusing on calming the traffic so close to a prime trolley stop, installing a sound barrier next to the highway, widening the sidewalk, placing a bike lane down India to connect Little Italy and Washington, and creating incentives for a mixture of restaurants/businesses on the first floor and mid-level and affordable housing above. There are such possibilities, and they will all be lost.

Finally, the interface between Sassafras, Kettner and India isn't working right now, and I can't see how it will work simply by making India wider and faster. I have attached a picture taken at 10:30 am at the corner of Kettner and Sassafras. This is the new norm for this intersection at all times of the day. Cars from the rental car

1.1

facility trying to turn left onto India back up all the way down the hill and through the Kettner intersection. These guys need a real on ramp.

There is still time right now for the city to take a step back and really look at the area the project is addressing to better maximize its use. Please make the airport build its own off ramp from Pacific Highway, and instead of turning India into one, invest the money in building a sound wall, putting in a bike lane, and promoting Middletown's growth.

I would also like to point out that I am extremely concerned by a pattern in which the city is claiming extensive outreach without telling residents what is going on and lying to its residents during meetings they do find out about by stating publicly that a project is not happening the way residents fear it will even though the documents available on the county web site directly contradict this information. A similar situation arose when master plan was considering closing a portion of University that would have sealed off one of the few egresses from Middletown, if there is a "Let's just pretend our residents are too lazy to pay attention and or to look things up," memo circulating, it isn't working.

I have included the local media in this email. For their information I have also scanned and included an update posted by a local community group in my neighborhood. That is the only outreach we've received, by the way. If it weren't for these neighbors, we wouldn't even know what was going on.

I really hope the media pick up both the story of the imminent lost opportunities on India, as well as the very different but just as important story about repeated fake proclamations of community outreach that never happened and the soft shoe shuffle of falsehoods peddled at meetings.

z

I also hope the message of the residents gets through. This is a bad plan. There is still time to pull back.

Shannon Biggs and Bryan Liang 2920 Union St ATTACHMENT 7

Middletown Update - July 26, 2016

Important Action Needed before August 8, 2016

Any questions, please email us at Middletown92103@gmail.com

Over 150 Middletown residents crowded into the Uptown Community Planning meeting July 5th only to be told there was no intention to remove parking on India Street and the one-way access to West Spruce and Redwood Streets. However, what was never mentioned at this meeting and is buried deep within the draft Uptown Community Plan, is a mobility study which reports a much different scenario. In fact, the plan to remove parking and sidewalks on India Street, as well as several feet of existing building fronts is *depicted in charts* and photographs found in incorporated attachments to the Plan. Additionally, tables refer the proposed removal of the southbound lane of India Street. All of this is **contrary to what was represented** at the Uptown Planners Meeting and **more importantly, public comment regarding the Plan is due by August 8th.**

This report once adopted, dictates city planning for many years, and is very difficult to amend or change. It is during this draft and public comment period that change is possible.

Not once during this meeting did officials at the meeting bring this impending deadline or information to the attention of residents. The information is publically available, buried deep within appendix C to a Mobility Study which is an incorporated attachment the Uptown Plan.

Residents have been asking for time on the agenda of Uptown Planners to provide input to the new plan, however Uptown Planners has said their agenda is too full and August is the last meeting prior to the deadline to submit comments for this draft report. NEXT UPTOWN PLANNERS MEETING 8/2/16 @ 6PM, we need another good turnout.

Residents have also provided an effective way and cost efficient way to route the traffic from the residential area of Middletown to the business area. The business owners of Middletown support the proposal.

The irony of all of this is that the existing Community Plan for Middletown which is about 5 pages and adopted February 2, 1988 included the following objectives:

Reduce auto/pedestrian conflicts. Preserve views on the western slopes. Improve the appearance of businesses along India Street and the public right of way. Maintain and strengthen the pedestrian-oriented environment within the Washing and India Street commercial zone.

Public comment is due by August 8 on the City's Plan. Written comments should be sent to:

Kurtis Steiner, Senior Environmental Planner, City of San Diego Planning Department, 1010 Second Avenue, MS 413, San Diego, CA 92101 Or e-mail your comments to PlanningCEQA@sandiego.gov With the Project Name and Number in the subject line.

Please note only written comments, received either via US Mail, hand-delivered, or via email, will be considered official comments in the Final EIR. PROJECT NAME: Uptown Community Plan Update PROJECT No. 21002568 / SCH No. Pending COMMUNITY AREA: Uptown COUNCIL DISTRICT: 3°

Please write or email your objections before the 8 August Deadline.

RESPONSE

Sources:

Appendix C: Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions). The removal of the southbound lanc on India Street from Sassafras to Redwood Street is included within tables 10 and 13 which show India Street from Sassafras to Redwood Streets changing from what they refer to in Table 10 as a three lane collector (two way) to a proposed three lane collector (one way) within Table 13.

The removal of parking and sidewalks on India, as well as several feet of building fronts, are depicted in charts and photographs found in the incorporated attachment:

Appendix C: Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions). Figures 63 and 64



LETTER



Letter C8

 From:
 Lydia Bilic <lbilic@cox.net>

 Sent:
 Wednesday, August 03, 2016 10:52 AM

 To:
 PLN_PlanningCEQA

 Subject:
 objection to PROJECT NAME: Uptown Community Plan Update/21002568 / SCH No. Pending

I object:

- C8-1 Table 13 Summary of Improvement Evaluation Uptown: This references India St between Redwood and Sassafras as "one way Collector" and removal of parking and buildings.
- C8-2 Also, I object to any widening of India Street. The two-way collector is essential to residents of Spruce (and Redwood). the maps used throughout the Uptown plan and appendices incorrectly depict Spruce at India and connecting streets off Spruce to show several alternative routes available to the 200+ residents living on Spruce and it's spurs (lbis, Horton, Upas, Jackdaw) for ingress and egress.

1

Lydia Bilic

COLDWELL BANKER WEST CA RE BROKER / REALTOR LIC#: 01782343 WWW.LYDABILCHOMES.COM P. (619) 549-4042 LBILIC@COX.NET

- C8-1 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C8-2 This comment references the changes to India Street. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

This comment also suggests that the maps provided in the proposed Uptown CPU incorrectly depict alternative routes on Spruce Street. The proposed Uptown CPU maps appear to be correct. West Spruce Street does connect to India Street. Access to all the other streets mentioned in the comment letter is solely from West Spruce/India Street. Because of the map's scale, the gap between West Spruce Avenue and Horton Avenue is difficult to see, but there is a gap in the figure and it was considered in the traffic analysis. However, the map in the Mobility Study does incorrectly show that a connection between West Spruce Avenue and Horton Avenue that will be corrected. While there was an error in the Mobility Study map, the model used in preparation of the analysis of potential impacts of the proposed Uptown CPU for traffic circulation did not include any connections with West Spruce Avenue that would provide additional ingress/egress to West Spruce Avenue other than India Street. Furthermore, the referenced Mobility Study Improvements (U17A and U17B) would be inconsistent with the proposed Uptown CPU polices and thus, would not be implemented due to infeasibility.

	Letter C9		
From: Sent: To: Subject:	Pangilinan, Marlon Thursday, June 23, 2016 9.04 AM Steinert, Kurtis FW: PEIR Alternatives		
From: Gary Sent: Thurss To: Pangilin: Cc: Leo Wils mat92103@ Subject: PEI Marlon, C9-1 After review believe the Update held housing uni projected in C9-2 On page 10- Alternative i C9-3 In this day o regarding th C9-4 What assum Alternative i Kind Regard Gary Bonnel	Bonner (mail totgarybonner@cox.net) (ay, June 32, 2016 6:33 AM an, Marion < MPangilinan@sandiego.gov) on kies kursterm@sbeglobal.net>; TMullaney@aol.com: rg-dariel@cox.net; rdahl@cox.net; vahoc.com B Alternatives ang the PEIR I have a question regarding the population figures supplied for the Lower Density Alternative, 1 Jolan was previously referred to as the Community Plan without incentive Zoning, At the Community Plan on 2/2/2016 you provided data showing the Community Plan without incentive Zoning, would provide of 32,109 and a projected population of 54,880. This was a population of 53,050 for the Lower Density while the January 2016 Draft Community Plan without incentive Zoning would provide while the data remains the same for the Proposed Uptown CPU, a projected population of 55,700. Propulerized data this is not a math error. Fappears the methodology has changed since February e Lower Density Alternative, but the assumptions did not change for the Proposed CPU. spino in methodology have changed since February in calculation the data for the Lower Density and why were they changed? s	C9-1 C9-2 C9-3 C9-4	This comment does not suggest an inadequacy in the PEIR, rather is questioning changes that occurred from previous versions of the draft Uptown CPU in relation to the version that was released for public review. See response to comment C9-4. See response to comment C9-4. The assumptions used to estimate the community plan build-out for the draft 2015 Community Plan without Incentives that was presented at the January 2, 2016 Uptown Planners meeting initially assumed that all parcels within former incentive areas could redevelop. In determining the community plan build out for the Lower Density Alternative, the assumptions used assumed that all parcels were likely to redevelop in former incentive areas except those that were fully developed such as multi-family residential and mixed-use development near the maximum or exceeding the adopted plan density; condominiums; mid or high rise buildings with steel or concrete frame construction; service stations, schools, hospitals, churches and places of worship; parks and open space, etc. These assumptions were similarly used for the currently Proposed Community Plan and represent a more realistic approach to determining community plan build out for the Lower Density Alternative.

	Letter C10		
	From: Gary Bonner [mailto:garybonner@cox.net] Sent: Saturday, August 06, 2016 12:31 PM To: Pangilinan, Marlon Subject: Draft EIR		
	Marlon,		
C10-1	Section 6.3.1.1 Roadway Networks, states First Avenue, Fourth Avenue and Fifth Avenue are posted at 30 miles per- hour. This is incorrect all three Avenues have large segments through Bankers Hill that are posted as 25 mph. First Avenue also has a section with diagonal parking. It is not all parallel parking.	C10-1	Clarification has been added to Section 6.3.1.1, Roadway Networks, of the Final PEIR regarding the posted speed limits and parking.
	Gary Bonner		
	- CO		

		Letter C11		
	From: Sent: To: Subject:	Lee Buby <isbuby@gmail.com> Wednesday, July 27, 2016 12 37 PM PLN, PlanningCEDA PROJECT NAME: Uptown Community Plan Update; No.: 21002568 / SCH No. Pending</isbuby@gmail.com>		
	Intended for Mr. Kurt	tis Steiner.		
C11-1	Dear Mr. Steiner, As a resident of Midd Plan Update, project a request that my points for clarification if nec meeting. Thank you,	lletown for 13 years, I have concerns regarding the subject project, Uptown Community number 21002568. I am a CPA prone to objectivity and logical analysis and I would s below be considered when finalizing this Update. Please do not hesitate to contact me ressary, or if I may be of better use presenting these issues and solutions at an appropriate	C11-1	Introductory comment noted. The City appreciates individual participation in the public review comment process.
	Lee S. Buby, CPA 619-750-1110			
C11-2	Regarding the Routin (via the Sassafras Esi Sassafras and Redwoo	tegarding the Routing of Northbound I-5 Rental Car Center Traffic onto India Street and down Sassafras Street via the Sassafras Exit) and possible removal of the two-block long Southbound lane of India Street between assafras and Redwood Streets: . India Street borders a large residential area of Uptown and provides a vital access point to this rea. Because of the one-way northbound nature of the majority of India Street, the two-block section of outbhound lane from Sassafras Street to Redwood Street provides essential access to this area for anyone orning from 1-5 South. The removal of this lane would not only add significant commute time to the residents f this area, it would also contribute significantly to the already-congested traffic at intersections of Laurel treet and Kettner Blvd./India St., the next available point at which a resident would be forced to make necessive left turns to reach their homes. This would also increase traffic attempting to access this area via 'olumbia Street from Laurel Street, an already traffic-stressed street due to its extremely narrow lanes where eople often wait at the end of a block for opposing traffic to clear before risking an accident in passing each ther. This would greatly affect the property values of this area as it would go from being considered a entrally located, highly-accessible neighborhood to a relatively very inaccessible one.	C11-2	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	 India Street borda area. Because of the soutbbound lane from coming from 1-5 Sout of this area, it would a Street and Kettner BU successive left turns t Columbia Street from people often wait at th other. This would gree centrally located, high 			
	 Northbound traff Street extremely busy unsafe speeds. Routh situation in which dri airport traffic, but ero Street. This is not on 	ic from the airport is already routed down India to reach the I-5 onramp. This makes India and these vehicles consistently treat this entire street section as an onramp, accelerating to ng Car Rental Center traffic coming from the South onto the Sassafras Exit produces a vers unfamiliar with the area must not only quickly merge with northbound, accelerating iss through this traffic in order to turn left and head down a steep hill on Sassafras ly confusing, it is dangerous.		
	 There is NO imp "crossing" of slowing with airport traffic ac- short sections of west Highway and the freq orthodox operation of traffic on Sassafras ar 	rovement that can be made to India Street that compensates for the aforementioned rental car center traffic (non-residents unfamiliar with terrain and where they are going) celerating and heading for the northbound I-5 onramp or, <i>perhaps even more troubling</i> , the bound Sassafras Street that are punctuated by traffic lights at Kettner Blvd, and Pacific uent interruption of traffic by railroad crossing controls, including the prevention of traffic lights at the Sassafras Stetture timerinersection during those times. The back-up of ad its effects on India Street traffic are unavoidable and would only be worsened by		

RTC-309

eliminating the southbound section of India Street in favor of an additional collector lane to turn left onto Sassafras, the ability to do so would be impossible without further shortening the eastbound lanes of Sassafras, already a source of congestion at this intersection.

4. There are no fueling options available to car renters being routed to the rental car center directly from the freeway through India Street to Sassafras Street. This only ensures that many drivers attempting to return their vehicles with full tanks will begin meandering around the area looking for fueling options, creating unnecessary traffic and confusion.

5. Any increases in traffic should naturally be routed through existing industrial and commercial areas before considering residential corridors.

6. The City has long neglected the maintenance of streets in both Middletown and in the industrial areas between it and the rental car center where it has chosen to route traffic, offering a final impression of blight and disrepair to visiting tourists.

The only logical routing and queuing area for returning to the rental car center from the South is Pacific Highway:

a. It is underused, particularly in that area.

b. It is desperately in need of maintenance, landscaping, and frontage aesthetics regardless of where the traffic is being routed from if the City wishes to leave tourists with an appropriate unpression of the area.

e. It can accommodate long queues, has long sections without fraffic signals and leaves very little to confuse unfamiliar drivers.

d. It can easily accommodate the same traffic volume that was originally routed to N. Harbor Dr. from the Hawthorne Street exit.

e. It offers at least one refueling option between exiting and reaching the rental car center.

Regarding the Proposed removal of Parking Spaces and Other Changes to the India Street Commercial node near Washington Street:

1. Parking in this area is at a premium, not just for the commercial businesses, but for the residents in this area as well. This results in a highly pedestrian-centric corridor that is only possible because of the slow traffic along this section of India Street.

2. Traffic is only slowed because of the availability of parking spaces here.

3. Drivers wishing to by-pass this congestion (i.e. not wishing to access the residential or commercial areas reachable from this section of India Street) can easily do so by veering left at the "Y" where Northbound lanes of San Diego Avenue begin. This serves as a natural separation of the high-speed traffic wishing to continue to Washington Street or the next I-5 Northbound onramp. *However*, there is NO signage directing through-traffic to veer left at this "Y" and for whatever reason, the bypass of pedestrian and parking spot-seeking drivers available by this underused section of street is not obvious to most drivers.

2

RESPONSE

C11-13 4. The businesses here are extremely desirable and would be greatly affected, possibly eventually eliminating this ever-more-popular node of restaurants that serves as the only such resource reachable by foot by residents, business, and the Washing Street trolley stop in this area. C11-14 5. Potential logical solutions: a. Eliminate or reduce through-traffic in this area by: i. Installing clear signage for drivers heading to Washington Street to veer left onto San Diego Avenue, and/or ii. Install delineators that prevent left turns onto Washington Street from India Street, and/or iii. Closing this section of India Street to Northbound traffic from the "Y" to Washington Street, making the entire section Southbound and accessible only from Washington Street with primary egress via a right-turn-only intersection at the "Y" with San Diego Avenue and secondary egress Eastbound into the residential streets, which would improve traffic flow.

3

ATTACHMENT 7

	Letter C	12	
	From: Marcia Buompensiero <lonzahn2@cox.net> Sent: Monday, August 01, 2016 5:35 AM To: Steinert, Kurtis Subject: Uptown Community Flan Update/Final EIR</lonzahn2@cox.net>		
	July 31, 2016		
	Kurtis Steinert, Senior Environmental Planner City of San Diego Planning Department 1010 Second Ave., MS 413 San Diego, CA 92101		
	RE: Uptown Community Plan Update/Final ElR Project No. 21002568/SCH No. Pending Community AREA: Uptown Council District: 3		
	Dear Mr. Steinert:		
C12-1	We are very concerned about the devastating impact that will result if the proposed Uptown Community Plar (referenced above) is adopted without modifications that address safety concerns and issues along the India Corridor as proposed. Specifically, we protest the current proposal to:	Street C12-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6)
	 Remove the southbound lane on India Street from Sassafras to Redwood Street Remove parking and sidewalks on India, as well as several feet of building fronts 		
C12-2	Over a twenty year period, we have witnessed a steady increase in car traffic on India Street. This is due in me a result of the heavy volume of traffic, routed from the airport via Harbor Drive to Laurel Street, then onto Ind order to access the I-S North on-ramps.	ost part as dia Street in	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6)
C12-3	Posted traffic signs indicate a speed limit of "35" miles per hour. This, apparently, is viewed as merely a "suggestion" as most traffic speeds are clocked in excess of 40-45 mph and above. Residents have complained to City officials for years about the increased volume of traffic and the lack of effective traffic control (i.e., blinking caution lights, stop signs, traffic signals). To date, nothing has been done to curb the volume and speed of traffic. This is a tragedy waiting to happen!	estion" as	
		signs, C12-3	Comment noted. This comment does not identify an inadequacy of the PEIR.
C12-4	Now, with the opening of the new Rental Car Center at the foot of Sassafras Street, the volume of traffic that	moves east	
	on Sassafras, turning left onto India has further increased the congestion on both Sassafras and India Streets.	C12-4	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C12-5	On July 5, 2016, over 150 residents of the affected communities (Little Italy, Middletown, Mission Hills, and Fi crowded into the Joyce Beers Center to address the Uptown Advisory Planning Council concerning this matter	ve Points) r.	
	Our voices were heard and the UPC adopted two resolutions:		
	 Resolution 1: To recommend to the San Diego City Council that immediate action be taken with responsed Uptown Community Plan: 	ct to the C12-5	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6). The
	 That the proposal to remove the southbound lane on India Street from Sassafras to Redwood Street be eliminated; and 	Street be	
	3		implementation of traffic calming and safety measures with an emphasis on pedestrian and bicycle mobility within the community, including along India Street.

RESPONSE

- b. That the proposal to remove parking and sidewalks on India, as well as several feet of building fronts be eliminated.
- c. Further, that, effective immediately, the City of San Diego implement safety measures on India Street to slow the traffic and install signage to reroute traffic leaving the RCC to Pacific Highway so as to access the Washington Street/San Diego Avenue on-ramp to I-5 North.
- 2. Resolution 2: To recommend to the San Diego City Council that ANY long-range plans for the affected areas be devised to include routing traffic along the Pacific Highway corridor by constructing permanent improvements (such as a new on-ramp) to route traffic safely to I-5 North and any other measures that will improve safe traffic flow that does not negatively impact the safety and quality of life for the residents in the affected areas.
- C12-6 My question is: Has the City of San Diego revised the proposed Uptown Community Plan to include ANY of the recommendations from the Uptown Community Advisory Council? If so, what are they? If not, what other options are available to the citizens of the negatively impacted communities so that we may have our voices heard?

Respectfully,

Joe and Marcia Buompensiero 3116 Horton Ave. San Diego, CA 92103

 Hon, Mayor Kevin Faulconer and Members of the San Diego City Council (Hon. Sherri Lightner, Hon. Marti Emerald, Hon. Lorie Zaph, Hon. Todd Gloria, Hon. Myrtle Cole, Hon. Mark Kersey, Hon. Chris Cate, Hon. Scott Sherman, Hon. David Alvarez)

Hon. Ron Roberts, District 4, San Diego County Board of Supervisors

C12-6 Development of the proposed Uptown CPU was developed with considerable public input. This public review process and subsequent public hearings are available for further public comment and input.

I FTTFR

		Letter C13		
	AUG	UST 4, 2016		
	MIS SEN CIT 101 SAN	TER KURTIS STEINER IOR ENVIRONMENTAL PLANNER Y OF SAN DIEGO PLANNING DEPT. O SECOND AVENUE, MS 413 DIEGO, CA 92101		
	DEA	R MR. STEINER:		
	REI	UPTOWN COMMUNITY PLAN UPDATE PROJECT NO. 21002568/SCH NO. PENDING COMMUNITY AREA: UPTOWN, COUNCIL DISTRICT 3		
C13-1	NE LOC ONE AND CON STR TUR PRO	ARE WRITING AS PROPERTY OWNERS OF TWO 4-PLEX APARTMENT COMPLEXES, ONE ATED AT 1218 WEST SPRUCE STREET, SAN DIEGO, CA 92103(OWNER-OCCUPIED) AND LOCATED AT 1520-1526 CHALMERS STREET (BOTH COMPLEXES SITUATED ONE BLOCK TWO BLOCKS OFF INDIA STREET RESPECTIVELY). WE WISH TO EXPRESS OUR CERNS OVER PROPOSED CHANGES TO INDIA STREET DUE TO THE RECENT RENTAL CAR UCTURE TRAFFIC. PLANS TO REMOVE SIDEWALKS, PARKING, AND THE SOUTHBOUND IN LANE ON INDIA STREET TO ACCESS SPRUCE AND REDWOOD STREETS PRESENT MANY BBLEMS AND HARDSHIPS TO OUR MIDDLETOWN NEIGHBORHOODS.	C13-1	Introductory comment noted. The City appreciates individual participation in the public review comment process.
	THE	FOLLOWING ARE SOME OF THE ISSUES WE WOULD BE FACING AS PROPERTY/		
C13-2	1.	INGRESS/EGRESS. THERE IS NO OUTLET IN SPRUCE CANYON (OUR SPRUCE STREET PROPERTY). IF THE INDIA STREET SOUTHBOUND TURN LANE TO SPRUCE AND REDWOOD STREETS WAS DONE AWAY WITH, SPRUCE CANYON RESIDENTS WOULD HAVE TO DRIVE TO ALREADY OVERCROWDED LAUREL STREET AND BACKTRACK.	C13-2	Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6)
C13-3	2.	SPRUCE CANYON PARKING WILL BECOME A BIG BIG ISSUE IF INDIA STREET PARKING IS REMOVED TO BUSINESSES, SPECIFICALLY BETWEEN REDWOOD AND SASSAFRAS. WE HAVEN'T COUNTED APARTMENTS, CONDO UNITS, HOUSES, MONTESSORI SCHOOL FOR STUDENTS/ITEACHERS. A HIGH PERCENTAGE DO NOT HAVE OFF-STREET PARKING. THE PERCENTAGE GOES HIGHER WHEN COUNTING HOMES, CONDOS AND APARTMENTS THAT HAVE ONE-CAR GARAGES AND TWO OR MORE CARS DUE TO FAMILY SIZE AND ROOMMATES.	C13-3	Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6)
C13-4	з.	THIS IS A HIGH RISK FACTOR FOR BUSINESS OWNERS WHO MAY LOSE CUSTOMERS AND ULTIMATELY THEIR BUSINESSES.		these response to comments (pages tric-4 through tric-o).
C13-5	4,	PROPERTY VALUES ARE AT RISK FOR DEVALUATION BECAUSE OF INDIA STREET TRAFFIC INGRESS/EGRESS. RENTAL PROPERTIES CAN LOSE ENHANCEMENT VALUE TO POTENTIAL RENTERS WHO WERE ATTRACTED TO THE LOCATION, LOCATION, LOCATION, AND TROLLEY ACCESS TO DOWNTOWN AND FREEWAY ACCESS NORTH, SOUTH AND EAST.	C13-4	Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C13-6	5,	SASSAFRAS STREET IS VERY STEEP. I HAVE BEEN HERE IN THIS NEIGHBORHOOD SINCE 1973 AND HAVE WITNESSED CAR ACCIDENTS AT INDIA AND SASSAFRAS, NOT TO MENTION THE MANY BIG RIGS THAT HAVE JACK-KNIFED AT THE SAME LOCATION TYING UP TRAFFIC FOR HOURS. - 1 -	C13-5	Comments t. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6)
			C13-6	Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

- C13-7 6. RESIDENTS GETTING OUT OF THE AREA IN A CRISIS OR EMERGENCY VEHICLES REACHING DESTINATIONS IN LIFE THREATENING SITUATIONS (POLICE, FIRE),
- C13-8 WHEN WE ATTENDED THE JULY UPTOWN COMMUNITY MEETING, WE WERE ALL MADE TO UNDERSTAND THERE WAS NO INTENTION TO REMOVE PARKING AND ONE-WAY ACCESS TO WEST SPRUCE AND REDWOOD ONLY TO FIND OUT THIS IS NOT THE TRUTH OR REALITY. WE CANNOT HELP BUT WONDER IF WE ARE GETTING RAILROADED AND/OR RAMRODDED WITHOUT ANY CONSIDERATION OF THE UPCOMING PLIGHT. (EXCUSE MY SKEPTICISM, BUT I HAVE PAPERWORK ON AN UNRESOLVED ERODING HILLSIDE MATTER ALL THE WAY BACK TO MAYOR MAUREEN O'CONNOR TO THE PRESENT WITHOUT RESOLVE.) INPUT IS NECESSARY FOR ACHIEVEMENTS TO BE MADE, SMART INPUT AND SMART ACHIEVEMENTS. THOUGHT BEFORE ACTION.
- C13-9 WE WONDER IF KETTNER STREET WOULDN'T BE A BETTER ROAD ON MUCH OF THIS CHANGE. THE VINE STREET SUBSTATION WILL BE THERE. BUSINESS IS SPORADIC, AND THE HOWELESS ON THAT STREET IN THAT AREA IS THRIVING MUCH TO ALL OF OUR DISMAY. PLEASE RECONSIDER RENTAL CAR TRAFFIC PLANNING THAT HAS BECOME THE NIGHTMARE FROM HELL FOR MIDDLETOWN/INDIA STREET.
- C13-10 WE HAVE HEARD MANY SUGGESTIONS. RON ROBERTS, WHO LIVES IN SPRUCE CANYON (A COMMONSENSE, MATTER OF FACT (CAN THINK ON HIS FEET KIND OF MAN) SEEMED TO HAVE SOME IDEAS AT THE MEETING IN JULY. SOME THOUGHTS THAT HAVE BEEN BATTED AROUND: REROUTING RENTAL CAR EXIT AND RETURNS TOWARD WASHINGTON STREET, BUILDING A RAMP THAT WOULD DROP RETURNS TOWARD WASHINGTON STREET, BUILDING A RAMP THAT WOULD DROP RETURNS TOWARD WASHINGTON STREET, BUILDING A RAMP THAT WOULD DROP RETURNS TOWARD WASHINGTON STREET, BUILDING A RAMP THAT WOULD DROP RETURNS TOWARD WASHINGTON STRUCTURE FROM SOUTHBOUND WASHINGTON STREET INSTEAD OF SASSAFRAS. I DO KNOW THE STEEPNESS OF SASSAFRAS AT KETTWER (BLIND SPOT) IS BAD (JACK-KNIFING AND ACCIDENT HAZARDS AT SASSAFRAS AND INDIA STREET. MATHE THE TROLLEY SHOULD GO UNDERGROUND AT LAUREL OR PALM OR SASSAFRAS SOMEHOW. I DO KNOW THAT PACIFIC HIGHWAY IS IN NEED OF UPDATING AND THAT FRONTAGE ROAD (UGH) SEENS POORLY LAID OUT. I REMEMBER THE AREA IN 1958. IT IS NO LONGER 1958.
- C13-11 DON'T QUITE UNDERSTAND WHY ALL OF THIS DIDN'T GET FIGURED AT THE TIME OF THE RENTAL CAR PLANNING STAGE. IT DEFINITELY SEEMS THE CART GOT AHEAD OF THE HORSE.

THANK YOU FOR YOUR ATTENTION TO THESE SERIOUS ISSUES. PLEASE KEEP US APPRISED WELL IN ADVANCE WOULD BE A NICE GESTURE.

- 2 -

DONALD W. STROCK

6863 GLENROY STREET

SAN DIEGO, CA 92120 TEL. (619) 817-1464

VERY SINCERELY,

ADDIE L. BURNETT DATE

1218 W. SPRUCE STREET, B SAN DIEGO, CA 92103 TEL. (619) 299-5617

abm

CC: RON ROBERTS (HAND-DELIVERED) TODD GLORIA (HAND-DELIVERED)

- C13-7 Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C13-8 Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C13-9 Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C13-10 Comments noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C13-11 Comment noted. This comment does not identify an inadequacy of the PEIR; therefore, a detailed response is not required.

Letter C14

From: Sent: To: Subject: |chave99 <|chave99@gmail.com> Saturday_Juty 30, 2016 7:09 PM PLN_PlanningCECA Uptown community Plan Update / Council district 3 / PROJECT No. 21002568

To Kurtis Steiner,

C14-1 As a 20 year resident of Middletown, residing on 3035 Columbia street, 1 would like to express my strong Opposition to this ill conceived plan, hidden in section Appendix C, to remove parking and sidewalks on India street, as well as several feet of existing building fronts, as found and plainly depicted in charts and photographs, found in "incorporated attachments" to the Plan. Furthermore, the tables and documents additionally mention the proposed removal of the southbound lane of India street. This information, and plan which i am highlighting is concerning to me because it is different than what was said to us, the residents, at the Uptown Planners Meeting, while in front of 150 witnesses and residents, who showed up at this public gathering.

> This proposal from the city is ill conceived and does not take into account the negative impacts on the community, its businesses and all its residents. I am 100% opposed to it, and will further express this publicly.

Joel Chavez 3035 Columbia street San Diego, CA. 92103

Happy Connecting, Sent from my optimi Tablet.

C14-1 This comment does not identify an inadequacy in the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

+

RESPONSE

Letter C15

From: Sent: To: Subject: Steve Davis <stevendavisinsd@gmail.com> Monday, August 08, 2016 12 01 PM PLN_PlanningCEQA Project Name_Uptown Community Plan Update

Kurtis Steiner Senior Environmental Planner City of San Diego Planning Department

Mr. Steiner:

C15-1

am writing to you with regard to the Uptown Community Plan Update, Project No. 21002568, Community Area: Uptown, Council District: 3,

Lam a long time resident of Middletown. During the nearly 20 years I have lived in this community, access to my home going south on interstate 5 has always been the Kettner offramp, then the one way access on India St. to Redwood, where I turn left and go up the hill to my home. The Uptown Community Plan contains within it, as a possible proposal, the elimination of the one way access on India St. Eliminating that one way access would make it very, very difficult to get to Middletown. It would require going all the way downtown and then backtracking or taking Washington Street, way out of the way, to come in from the east. Any other access than India Street would be extremely inconvenient.

I attended the most recent meeting of the Uptown Planners advisory committee, and I understand that the elimination of the one way on India St. is not currently planned or proposed. Nonetheless, I am concerned that the City, will re-visit this matter at some date in the future and may decide to eliminate the one-way India access. I want to voice my very strong opposition to any such action.

Steve Davis

C15-1 Comment noted. This comment does not identify an inadequacy in the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
Letter C16

	From: Sent: To:	Catrina Dulay <catrinadulay@gmail.com> Monday, August 01, 2016 9/20 AM PLN_PlanningCEQA</catrinadulay@gmail.com>							
	Subject:	Concerns regarding UPTOWN COMMUNITY PLAN UPDATE / PROJECT NO. 21002568							
	Dear Mr. Steiner								
C16-1	l am contacting y with the traffic h	you because 1 am a concerned and upset resident living on India Street, so 1'm very familiar azards and parking difficulties that take place in this neighborhood every day.							
C16-2	You don't need t conditions are a l know that the paa work, most likely you can't find pa have spaces to pa that next to the le down these narro occurs on India S provides access t Street route to Re their homes (mys	You don't need to live in this residential (emphasis on "residential") neighborhood to understand that traffic conditions are a hazard. You simply have to be a visitor at one of the many great businesses on India Street to know that the parking and traffic conditions are awful. Imagine that you're driving home from a long day at work, most likely experiencing traffic on the freeway, and the frastration follows you all the way home becau you can't find parking on your own street because business patrons have taken your parking because they don have spaces to park. Imagine this scenario taking place every time you leave home and come back. Now take that next to the level and imagine every resident experiencing the same inconvenience, going in circles, up an down these narrow streets, looking for parking that's already scarce. On top of that, the amount of speeding the occurs on India Street is ridiculous and unsafe for anyone attempting to walk on the pedestrian bridge that provides access to the Middletown trolley station on the other side. In addition, closing the southbound India Street route to Redwood Street would cause a huge inconvenience for residents who take that route to get to their homes (myself included).							
	get to work on time. Not if businesses need to deal with the lack of parking for their customers. Not if cyclists relying on their bicycles for transportation are concerned for their safety while sharing the road with speeding drivers. Not if my fellow neighbors can't safely get to their homes and their loved ones.								
	As a native San I around me and th concerns and the asking you not to	Diegan who takes pride in calling this city my home, I want what's best for the people living to visitors who come to enjoy the businesses on my street. I am asking you to listen to my concerns of my neighbors. I do not approve of these changes that were proposed and I am o move forward with them.							
	Thank you for your consideration.								
	Sincerely, Catrina Dulay								
		4							

- C16-1 Comment noted. The City appreciates individual participation in the public review comment process.
- C16-2 Comment does not suggest an inadequacy in the analysis of the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

		Letter C17		
	From: Joe Ebner <joe com="" ebner@gmail=""> Sent: Monday, August 08, 2016 2:29 PM To: PLN<planningceqa< th=""> Subject: Uptown Community Plan Update - Project #21002568/SCH No Pend</planningceqa<></joe>	ling - Uptown Dist 3		
	Hi Kurtis Steiner,			
C17-1	I've owned a condo at 1232 W Redwood since 2000. I think the proposed changes on In Laurel and Washington would be terrible for both residential homes and commercial bus mention knocking out the one way street between Redwood and Sassafrass going south y traffic for everyone who lives in the area.	lia Street between inesses. Not to vould really mess up	C17-1	This comment does not identify an inadequacy in the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to
	Tm 100% AGAINST all of these changes. Plus knocking out parking for all the businese sales for everyone further north on India.	es would really impact		comments (pages RTC-4 through RTC-6).
	My vote is 100% NO.			
	Joe Ebner			
	ι.			

				Letter C18		
	Ryan Eddings & / 426 Brookes Ave San Diego, CA 92	Ana Širović nue 103				
	August 7, 2016					
	Kurtis Steinert, S City of San Diego 1010 Second Ave San Diego, CA 92	enior Enviro Planning De enue, MS 41 101	nmental Planner epartment 3			
	RE: Draft PEI Project n VIA E-M/	R for the Up number 2100 AlL	otown Community Plan Update 02568			
	Dear Mr. Steiner	t			640.4	
C18-1	We respectfully : Plan Update. We bioacoustics.	submit comr e are resider	ments and questions regarding the draft PEIR for the of Hillcrest and researchers in the areas of t	or the Uptown Community Irban studies and	C18-1	Comment noted.
C18-2	 2 The main goal of these comments and questions is to ensure that the city adequately and transparently follows relevant laws and regulations, conducts comprehensive and unbiased environmental analyses, identifies and mitigates all significant effects on the environment, and uses the community plan update process to identify and mitigate problems created as a result of the current, adopted community plan. The comments and questions center on three substantial problems identified in the draft PEIR: Inadequacy of the draft PEIR; Inconsistency between the proposed Uptown Community Plan, the General Plan, the Notice of Preparation (NOP), and the draft PEIR; and Procedural shortcomings. 				C18-2	This introductory comment is noted and responded to in the following responses.
	and time-consum	ning litigatio	n.			
	Inadequacy of th	e Draft PEII	3		610.2	
C18-3	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question	C18-3	concern are detailed in comments C18-4 through C18-15 and
	2.0 – Environmental Setting	2-1-2- 62	Many subsections within Chapter 2.0 do not describe current conditions, as CEQA Guidelines 15125 require. Some of these deficiencies are noted in comments below. Please remove all unnecessary information that does not describe existing conditions from these subsections and be sure that existing conditions (on the ground) are adequately described.	CEQA Guidelines 15125; 15125(a): "The description of the environmental setting shall be no longer than is necessary to an understanding of the significant effects of the proposed project and its alternatives."		responses are provided to those comments below.

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question
C18-4	2.3.1.1 - Existing Land Uses	2-6 - 2-9	The land use descriptions from page 2-6 to page 2-9 seem to be describing conditions from the adopted community plan, not <u>existing</u> conditions. Table 2-1, and Figure 2- 4 also seem to describe land uses according to the adopted community plan, not according to physical conditions (although Table 2-1 may be existing conditions). Additionally, there is no appropriate map showing existing (on the ground) densities and uses to be able to substantiate the claims made on pages 2-8 and 2-9. According to CEQA Guidelines, descriptions in this section should be of the physical environment when the NOP is published. This should be corrected since this is the baseline for comparing the significance of impacts. Additionally, this error may call into question impact findings in later chapters. Please justify the adequacy of this section, given CEQA Guidelines 15125.	According to CEQA Guideline 15125(a), "An EIR must include a description of the physical environmental conditions in the vicinity of the project, as they exist at the time the notice of preparation is published, or if no notice of preparation is published, at the time environmental analysis is commenced, from both a local and regional perspective. This environmental setting will normally constitute the baseline physical conditions by which a lead agency determines whether an impact is significant."
C18-5	2.3.1.1 - Existing Land Uses	2-6 - 2-9	Where did the information for Table 2-1, Figure 2-4, and subsections a, b, c, and d come from? The contradictions between the table, figure, and subsections are too numerous to name outright, but how, for example, does one reconcile the fact that Figure 2-4 shows the area around 5th and University as Mixed Used (73-110 du/ac) when, in fact, there are very few residential units in this area and, furthermore, that Table 2-1 does not even list Mixed Use as a category?	CEOA Guideline 15125(a)

C18-4 The PEIR evaluates potential impacts against the appropriate baseline of existing conditions. Section 2.3.1 of the PEIR describes the land uses within the Unterwork CPUL The land uses described in

baseline of existing conditions. Section 2.3.1 of the PEIR describes the land uses within the Uptown CPU. The land uses described in Table 2-1 are from the City of San Diego Planning Department and represents the actual number of acres associated with actual land uses on the ground. The subheader in that table that states "General Plan Land Use Category" has been revised to state "Land Use" to clarify that the land uses and acreages shown are actual "on the ground" land use not General Plan Land Uses. The Draft PEIR included a Figure 2-4 that provided a graphic representation of land uses based on the adopted Uptown Community Plan. While this figure generally provides information about land uses that are on the ground, since the land use designations generally represent existing conditions, there are some areas where the figure did not represent existing conditions. To better describe existing conditions, Figure 2-4 has been revised to show existing land uses from SANDAG, not land use designations. The text based discussions that followed Figure 2-4 in the Draft PEIR do represent a description of on the ground land uses, as they provide a more detailed narrative of actual land uses for each type of use (residential, commercial/mixed-use, etc.).

C18-5 Table 2-1 in the PEIR represents the number of acres associated with actual land uses on the ground. Figure 2-4 that was provided in the Draft PEIR was revised in the Final PEIR to show existing land uses based on SANDAG data. The original Figure 2-4 included in the Draft PEIR represented the land uses envisioned through implementation of the adopted Community Plan and thus, showed the area around University Avenue and 5th as Mixed-Use with a residential density of 75-110 du/ac. Regardless of the updated figure, both the Draft and Final PEIR provide an adequate representation of existing physical conditions in the Uptown CPU area and the environmental analysis is based on the appropriate baseline of existing physical conditions.

ATTACHMENT 7

RESPONSE

PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question		
2.3.1.1 - Existing Land Uses	2-6 - 2-9	The Land Use - Existing Conditions subsection (6.1.1) of the Environmental Analysis Chapter (6.0) of the PEIR states, "The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0 respectively." Given that Chapter 2.0 lacks an adequate description of the existing environmental land use setting and that Chapter 6.0 does not include such a description, what environmental land use information did the city use as its baseline for the environmental impact findings in Chapter 6.07	CEQA Guideline 15125(a): "This environmental setting will normally constitute the baseline physical conditions by which a lead agency determines whether an impact is significant." CEQA Guideline 15125(c): "The EIR must demonstrate that the significant environmental impacts of the proposed project were adequately investigated and discussed and it must permit the significant effects of the project to be considered in the full environmental context."	C18-6	The City used an adequate baseline in its environmental analysis in Chapter 6.0. In most cases, the information in Chapter 2.0, Environmental Setting is a general description of existing conditions and more specific environmental baseline information is included within the specific subject area analysis section in Chapter 6 and/or is included with a technical report, if applicable.
2.3.1.1 - Existing Land Uses	2-9	The PEIR states, "Descriptions of the categories from the City's General Plan Land Use and Community Planning Element (Table LU-4) that are applicable to the Uptown community are presented in Table 5-1, General Plan Land Use Categories. Application of these categories for consistency with the General Plan Land Use and Community Planning elements is accomplished with approval of individual community plan updates." As the categories alluded to above do not represent existing conditions, how does this information fulfill the requirements of CEQA Guidelines 15125?	CEQA Guidelines 15125	C18-7	The stated information is not provided as existing condition information but is provided as a cross reference to the reader to explain that the existing land uses described in the chapter are distinguishable from the General Plan categories that are described in Table 5-1.
		updates." As the categories alluded to above do not represent existing conditions, how does this information fulfill the requirements of CEQA Guidelines 15125?			
	PEIR Section 2.3.1.1 - Existing Land Uses 2.3.1.1 - Existing Land Uses	PEIR SectionPage(s)2.3.1.1 - Existing Land Uses2-6 - 2-92.3.1.1 - Existing Land Uses2-9	PEIR Section Page(s) Comment/Question 2.3.1.1 = Existing 2-6 - 2-9 The Land Use - Existing Conditions subsection (6.1.1) of the Environmental Analysis Chapter (6.0) of the PEIR states, "The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0 respectively." Given that Chapter 2.0 lacks an adequate description of the existing environmental land use setting and that Chapter 6.0 does not include such a description, what environmental land use information did the city use as its baseline for the environmental impact findings in Chapter 6.0? 2.3.1.1 - Existing Land Uses 2-9 The PEIR states, "Descriptions of the categories from the City's General Plan Land Use and Community Planning Element (Table LU-4) that are applicable to the Uptown community are presented in Table 5-1, General Plan Land Use Categories for consistency with the General Plan Land Use and Community Planning elements is accomplished with approval of individual community plan updates." As the categories alluded to above do not represent existing conditions, how does this information fulfill the requirements of CEOA Guidelines 15125?	PER Section Page(s) Comment/Question Legal Basis for Comment/Question 2.3.1.1 - Existing Land Uses 2-6 - 2-9 The Land Use - Existing Conditions subsection (6.1.1) of the Environmental Analysis Chapter (6.0) of the PER states, "The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0 respectively." Given that Chapter 2.0 lacks an adequate description of the existing environmental agency determines whether an impact is significant." CEOA Guideline 15125(c): "The EIR must demonstrate that the significant environmental land use information did the city use as its baseline for the environmental land use information did the city use as its baseline for the environmental land use information did the city use as its baseline for the environmental land use information did the proposed project were a dequately investigated and discussed and it must permit the significant environmental land use information did the significant environmental land use information did the significant environmental land use information did the significant environmental context." CEOA Guidelines 15125 2.3.1.1 - Existing Land Uses 2-9 The PEIR states, "Descriptions of the categories from the City's General Plan Land Use and Community are presented in Table 5-1, General Plan Land Use categories for consistency with the General Plan Land Use and Community Planning elements is accomplished with approval of individual community plan updates: "As the categories alluded to above do not represent existing conditions, law does this information fulfil the requirements of CEOA Guidelines 151257	PBR Section Page(s) Comment/Question Legal Basis for Comment/Question 2.3.1.1 - Existing Land Uses 2-6 - 2-9 The land Use - Existing Conditions subsection (5.1.1) of the Environmental subsection (5.1.1) of the Environmental subsection (5.1.1) of the Environmental environmental setting and that Chapter 2.0 and 50 respectively." Given that Chapter 2.0 lacks an adequate description of the existing environmental land use setting and that Chapter 6.0 does not include such a description, what environmental land use information (ddt the significant environmental land use information (ddt the significant environmental limpact findings in Chapter 6.07 CEDA Guideline 15125(c): The EIR must demonstrate that the significant environmental impact 5 of the proposed project were adequately investigated and discussed and it must permit the significant environmental context." 2.3.1.1 - Existing Land Uses 2-9 The PEIR states, "Descriptions of the categories for consistency with the General Plan Land Use Categories Application of these categories for consistency with the General Plan Land Use and Community Planning Element (Table LU-4) that are applicable to the Uptown community are presented in Table 5-3, General Plan Land Use Categories Application of these categories for consistency with the General Plan Land Use community plan updates." As the categories Biolide to above do not represent existing conflictions, how does this information fulfill the requirements of CEQA Guidelines 151257 C18-7

Г

RESPONSE

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question	C18-8	The discussion of built form and development is intended as an
8-8	2.3.2.1 - Existing Context and Urban Form	2-9 - 2- 13	Characteristics of the urban form, such as existing heights, setbacks, street widths, densities, styles and vegetation, are missing from the description of the Urban Form. Furthermore, these characteristics must be described in sufficient detail to differentiate the various neighborhoods and sections of neighborhoods where these characteristics vary. For example, heights throughout areas that are currently commercially zoned in Hillcrest generally do not exceed three stories, with limited exceptions. Multi-family areas of Hillcrest have heights are generally in the three to four story range. These descriptions are necessary for environmental impact analysis. Subsection b. simply states what has influenced the built form; it is <i>not</i> a description of the existing built form. Please justify the adequacy of this section, given CEQA Guidelines 15125.	CEQA Guidelines 15125	(18.9	overarching description of the Uptown CPU area. The City recognizes that individual areas within the community differ greatly. The PEIR incorporates by reference the proposed Uptown CPU, which includes more extensive descriptions of the built environment and form for each of the Uptown neighborhoods. Refer to the proposed Uptown CPU, Section 1.1 Community Plan Profile for these detailed descriptions.
8-9	2.3.2.1 - Existing Context and Urban Form	2-9 - 2- 13	The Visual Effects and Neighborhood Character - Existing Condions subsection (6.2.1) of the Environmental Analysis chapter (6.0) of the PEIR states, "The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0 respectively." Given that Chapter 2.0 lacks an adequate description of the existing environmental context and urban form setting and that Chapter 6.0 does not included a description of the existing environmental setting, what environmental context and urban form information did the city use as its baseline for the environmental impact findings in Chapter 6.0?	CEOA Guideline 15125(a): "This environmental setting will normally constitute the baseline physical conditions by which a lead agency determines whether an impact is significant." CEOA Guideline 15125(c): "The EIR must demonstrate that the significant environmental impacts of the proposed project were adequately investigated and discussed and it must permit the significant effects of the project to be considered in the full environmental context."	010-9	keler to response to comments C18-6 and C18-8.
3-10	2.3.3 - Transportation and Circulation	2-13	PEIR states, "In addition, SR-163 traverses the Uptown Community Plan area, resulting in the use of local roads for trucking and transport of goods between freeways." Is SR-163 considered a "local road" in this context or is this suggesting that trucks use	CEQA Guidelines 15125(c)	C18-10	This sentence has been clarified to read: "Truck transport of goods occurs within the CPU area on these freeways and on local roads."

and the second s	1.1.1.1.1.1.1		Comment/Question		
		surface streets to transport goods between freeways?			
2.3.3.2 - Public Transportation	2-13 - 2- 14	The description is not of existing physical conditions, but rather of planned conditions (though it is not clear which plan is referenced). For example, BRT and streetcar services do not exist in Uptown. Furthermore, there is no description of the current bus service, lines, and frequencies that exist today. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection, given CEOA	CEQA Guidelines 15125	- C18-11	The introductory paragraph of Section 2.3.3.2 indicates that the Uptown CPU area is served by the trolley (adjacent to the CPU area) and bus service. The remaining information in this section provides information about these different types of transit services and is included to provide context to the analysis included in Section 6.3.3, Issue 2. Additional information about these existing conditions related to transit is provided in Section 6.3.1 of the PEIR.
2.3.3.4 - Bicycle Facilities	2-14 - 2- 15	Guidelines 15125. The description is not of existing physical conditions in Uptown, but rather is one of the types of bicycle facilities that exist in general. Furthermore, there is no. description of the current bicycle facilities in Uptown, including their locations, distances, and conditions. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection or justify the adequacy of this subsection, given CEQA Guidelines 15125.	CEQA Guidelines 15125	C18-12	The information in this section provides context to the analysis related to bicycle facilities provided in Section 6.3.3, Issue 2 of the PEIR. Additionally, Section 6.3, Transportation and Circulation of the PEIR includes additional information about the existing conditions related to bicycle facilities (Section 6.3.1.6.b.), indicating that Class II (Bicycle Lanes) and III (Bicycle Route) facilities are provided on Fourth, Fifth, and Sixth avenues, as well as on portions of downtown streets, and there are no bicycle connections north to Mission Valley and Class III bicycle routes
2.3.3 - Transportation and Circulation	2-13	PEIR states, "The Uptown Community Plan area is identified in the General Plan's Land Use and Street System map Figure LU-2." This is not true. In fact, no community plan areas are identified in the referenced map. Appropriate maps with roads, transit lines, and bike lanes are necessary to understand the location of all transportation infrastructure in Uptown and how circulation is impacted. Please update this statement or justify the adequacy of the statement and the adequacy of section 2.3.3, given CEQA Guidelines 15125.	CEQA Guidelines 15125	C18-13	The Uptown CPU area is included within the referenced General Plan Figure LU-2; however, the figure does not specifically call out the boundaries of the Uptown CPU area. The text in the PEIR Section 2.3.3 does provide existing conditions information relative to freeways within the CPU area and describes the main roadways in Section 2.3.3.1. Refer to response to comments C18-
	2.3.3.2 - Public Transportation 2.3.3.4 - Bicycle Facilities 2.3.3 - Transportation and Circulation	2.3.3.2 - Public 2-13 - 2-13 Transportation 14 2.3.3.4 - Bicycle 2-14 - 2-15 Facilities 15 2.3.3 - Transportation and Circulation 2-13	2.3.3.2 - Public Transportation2-13 - 2 14The description is not of existing physical conditions, but rather of planned conditions (though it is not clear which plan is referenced). For example, BRT and streetcar services do not exist in Uptown. Furthermore, there is no description of the current bus service, lines, and frequencies that exist today. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection or justify the adequacy of this subsection, given CEOA Guidelines 15125.2.3.3.4 - Bicycle Facilities2-14 - 2 15The description is not of existing physical conditions in Uptown, but rather is one of the types of bicycle facilities that exist in general. Furthermore, there is no description of the current bicycle facilities in Uptown, including their locations, distances, and conditions. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection or justify the adequacy of this subsection or justify the adequacy	Comment/Question 2.3.3.2 - Public 2-13 - 2 Transportation The description is not of existing physical conditions, but rather of planned conditions (though it is not clear which plan is referenced). For example, BRT and streetcar services do not exist in Uptown. Furthermore, there is no description of the current bus service, lines, and frequencies that exist today. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection or justify the adequacy of this subsection, given CEOA Guidelines 15125. CEQA Guidelines 15125 2.3.3.4 - Bicycle 2-14 - 2- 15 The description is not of existing physical conditions in Uptown, but rather is one of the types of bicycle facilities that exist in general. Furthermore, there is no description of the current bicycle facilities in Uptown, including their locations, distances, and conditions. This should be changed to reflect current conditions to the extent that description or justify the adequacy of this subsection, given CEQA Guidelines 15125. CEQA Guidelines 15125 2.3.3 - Transportation and Circulation 2-13 PEIR states, "The Uptown Community Plan area is identified in the General Plan's Land Use and Street System map Figure LU-2." CEQA Guidelines 15125 2.3.3 - Transportation and Circulation 2-13 PEIR states, "The Uptown Community Plan area is identified in the General Plan's Land Use and Street System map Figure LU-2." CEQA Guidelines 15125 2.3.3, given CEQA Guidelines 15125, 2.3.3, given CEQA Guidelines 15125. CEQA Guidelines 15125. <td>2.3.3.2 - Public 2-13 - 2 Transportation The description is not of existing physical conditions, but rather of planned conditions, but rather of planned conditions (though it is not clear which plan is reference). For example, BRT and streets rerevices do not exist in Uptown. Furthermore, there is no description of the current bus service, lines, and frequencies that exist today. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection or justify the adequacy of this subsection given CEOA Guidelines 15125. CEOA Guidelines 15125 C18-12 2.3.3.4 - Bicycle 2.14 - 2: The description is not of existing physical conditions in Uptown, burtarber is no of the types of bicycle facilities that exist in general. Furthermore, there is no description of the current bicycle facilities in Uptown, including their locations, distances, and conditions. This should be changed to reflect current conditions to the extent that descriptions given CEOA Guidelines 15125. CEOA Guidelines 15125 C18-12 2.3.3 - Transportation and Circulation 2-13 PHR states, The Uptown Community Plan area is identified in the General Plan's Land Use and Street System map Figure LU-2/ This is not true. In fact, no community plan areas are identified in the referenced map. Appropriate maps with roads, transt lines, and bike lanes are necessary to understand the location of all transportation infrastructure in Uptown and how circulation is impacted. Please update this statement on Usify the adequacy of the statement and the adequacy of section 2.3.3, given CEOA Guide</td>	2.3.3.2 - Public 2-13 - 2 Transportation The description is not of existing physical conditions, but rather of planned conditions, but rather of planned conditions (though it is not clear which plan is reference). For example, BRT and streets rerevices do not exist in Uptown. Furthermore, there is no description of the current bus service, lines, and frequencies that exist today. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection or justify the adequacy of this subsection given CEOA Guidelines 15125. CEOA Guidelines 15125 C18-12 2.3.3.4 - Bicycle 2.14 - 2: The description is not of existing physical conditions in Uptown, burtarber is no of the types of bicycle facilities that exist in general. Furthermore, there is no description of the current bicycle facilities in Uptown, including their locations, distances, and conditions. This should be changed to reflect current conditions to the extent that descriptions given CEOA Guidelines 15125. CEOA Guidelines 15125 C18-12 2.3.3 - Transportation and Circulation 2-13 PHR states, The Uptown Community Plan area is identified in the General Plan's Land Use and Street System map Figure LU-2/ This is not true. In fact, no community plan areas are identified in the referenced map. Appropriate maps with roads, transt lines, and bike lanes are necessary to understand the location of all transportation infrastructure in Uptown and how circulation is impacted. Please update this statement on Usify the adequacy of the statement and the adequacy of section 2.3.3, given CEOA Guide

	PEIR Section	Page(s)	Comment/Question	Legal Basis for		
C18-14	2.3.4 – Air Quality	2-16 - 2-18	The air quality subsection describes regional air quality from a location that is more than 1.5 miles from the nearest Uptown border. The subsection neither describes general air quality in Uptown, nor specific areas of concern like alleys where sensitive receptors like residences are in very close proximity to substantial sources of pollution like heavy diesel trucks. This should be changed to reflect current conditions to the extent that descriptions can serve as a baseline for environmental impact analysis. Please update this subsection, given CEQA Guidelines 15125. Additionally, please describe the environmental setting that was used as a baseline for the environmental analyses in Chapter 6.0.	CEQA Guidelines 15125	. (18-14	A description of the conditions in the air basin is appropriate because air quality is regulated at a basin level and factors across the basin including meteorological conditions and pollutants emitted in other locations affect basin-wide air quality. Also, as stated in Section 2.3.4.3, there is no current methodology for directly measuring diesel particulate concentrations but the California Air Resources Board estimates diesel particulate emissions could add an additional 420 in one million to the ambient cancer risk levels in San Diego County. Nonetheless, the analysis contained in Section 6.4 of the PEIR does address exposure of sensitive receptors and includes an analysis of potential localized carbon monoxide hot spot impacts under Issue 3 of Section 6.4.3 of the PEIR.
C18-15	2.3.6 - Noise	2-20 - 2- 24	The entire noise section is a listing of noise and sound definitions and does not contain any description of the current noise environment in Uptown. There are no quantitative, physical measures of noise characteristics presented in this section. This should be changed to describe current conditions and nothing more. Please update this subsection or justify the adequacy of this subsection, given CEQA Guidelines 15125. Additionally, please describe the environmental setting that was used as a baseline for the environmental analyses in Chapter 6.0.	CEQA Guidelines 15125	C18-15	The existing conditions used as the baseline for the noise analysis is included in Section 6.6.1 of the PEIR. The information included in Section 2.3.6 is background information intended to provide additional context to the analysis. A clarifying statement was added to the Final PEIR, Section 2.3.6 to refer the reader to the existing conditions discussion for noise in Section 6.6.1 of the PEIR.
C18-16	3.0 – Project Description	3-3-3- 18	Contrary to CEQA Guidelines 15150, subsections 3.4.1.2 (Mobility Element), 3.4.1.3 (Urban Design Element), 3.4.1.4 (Economic Prosperity Element), 3.4.1.5 (Public Facilities, Services, and Safety Element), 3.4.1.6 (Recreation Element), 3.4.1.7 (Conservation Element), 3.4.1.8 (Noise Element), 3.4.1.9 (Historical Preservation Element), and 3.4.1.10 (Implementation) attempt to incorporate the Uptown CPU by reference, instead of actually describing these crucial elements of the CPU. As a result Chapter 3.0	CEQA Guidelines 15150(f): "Incorporation by reference is most appropriate for including long, descriptive, or technical materials that provide general background but do not contribute directly to the analysis of the problem at hand."	C18-16	The proposed Uptown CPU is a main project component analyzed in the Draft PEIR. It would not be appropriate to restate the information within the Uptown CPU within the body of the PEIR, as this would make the document extremely lengthy and difficult to navigate. Thus, incorporation by reference is an appropriate approach in this case. The Uptown CPU itself does not contribute to the analysis contained within the PEIR; rather it is a component of the project analyzed in the PEIR.

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question	C18-17
			(Project Description) is inadequate for supporting the environmental analyses presented in Chapter 6.0. Please update these subsections with descriptions of the CPU or justify their adequacy according to CEQA Guidelines 15124 and 15150(f).		
C18-17	3.0 – Project Description	3-3-3- 18	The wording of the subsections suggests that the project under consideration by the EIR is the production of a CPU document, as opposed to the production of an urban form with environmental consequences. The wording should explicitly describe the project's intended results to inform the environmental analyses, as opposed to describing the general content of the CPU document.	CEQA Guidelines 15124: "The description of the project shall contain the following information but should not supply extensive detail beyond that needed for evaluation and review of the environmental impact"	C18-18
C18-18	6.1.3 – Land Use, Impact Analysis	6.1-2 - 6.1-12	The PEIR states that the Uptown CPU, the General Plan, and the City of Villages Strategy are consistent. Please see the section below for comments and questions regarding the inconsistency of the Uptown CPU with the other documents.		C18-19
C18-19	6.2.3– Neighborhood Character, Impact Analysis	6.2-4 - 6.2-5	Given the lack of an adequate description of existing conditions regarding urban form (see comment for 2.3.2.1 - Existing Context and Urban Form), the Impact Analysis is inadequate for determining changes in neighborhood character. Characteristics of the urban form, such as existing heights, setbacks, street widths, densities, styles and vegetation, are missing from the descriptions of existing condition and expected conditions resulting from the CPU. The impact analysis indicates that there are no significant impacts, though a comparison of existing conditions on the ground to the urban design element of the CPU suggests that there will be substantial increases in building heights, a reduction in setbacks in many places like alleys where potentially noisy and polluted urban canyons may develop, and a substantial change in floor area ratios. Please describe		

RESPONSE

The PEIR project description includes various sections to describe the various elements of the project. The sections of the document referenced by the comment are intended to describe the policies laid out in each element of the proposed Uptown CPU. Section 3.6 of the project description describes the result of build-out of the proposed CPU in Tables 3-8 through 3-10. Tables 3-9 and 3-10 describe the anticipated change in residential units and commercial square footage within the plan area compared to the existing condition as a result of build-out of the Uptown CPU area.

C18-18 Comment noted.

C18-19 Refer to response to comment C18-8. A reference was added in Section 6.2.3 of the PEIR, under Issue 2, to direct the reader to Section 1.1 of the proposed Uptown CPU for specific discussions of the characteristics of each neighborhood within the Uptown CPU area. The analysis does recognize that the proposed Uptown CPU would result in an intensification of land uses and would be subject to growth and change. Section 6.2.3 Issue 2 of the PEIR was also revised to add additional descriptions of how the proposed CPU would change the physical characteristics of the CPU area. Essentially, most change would occur on sites that are undeveloped or underdeveloped, which would limit the extent that implementation of the proposed Uptown CPU would change the character of an entire area due to the built-out nature of the plan area. New development with increases in height, bulk, and scale would be expected to occur in areas of the community that have been already identified for higher intensity development and have already been developing at a higher intensity than the surrounding existing development as part of the existing or evolving character of the area, such as in areas within Bankers Hill/Park West along Fifth and Sixth avenues where the adopted and proposed CPU allow High to Very High

C18-19 (cont.)

Residential densities and building heights up to 150 feet. The analysis describes how the potential impacts of these changes and intensification would be addressed by the requirement that future development comply with proposed Uptown CPU policies intended to ensure neighborhood compatibility such as through the application of building transitions and upper-story stepbacks and through application of Design Guidelines by Building Type included in the proposed Urban Design Element. At a program level of analysis, the PEIR finds these project elements would reduce potentially significant aesthetics impacts to less than significant.

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question
1			the objective measures, existing conditions, and planned conditions used to determine that there would be no significant impact, or alternatively, please update the section to reflect an adequate impact analysis.	
-20	6.6.1.1 Noise Measurements	Table 6.6-1	Sampling for data presented in this table was not sufficient to "provide a characterization of the variability of noise throughout the Uptown CPU area." 15 min recordings during one time are not sufficient to understand the level of understand the level of	
			variability in noise that may occur. Also, 15 min measure is too short to capture well true 1-h Leq. Given that these measurements seem aimed primarily at traffic noise, lack of measurements during rush hour is notable. On the other hand, selected measurement locations do not represent the full variability of noise conditions that are relevant for the community. For example, no measurements were collected in central, mixed use areas around University/ Robinson or in alleys in commercial areas, or at noise-sensitive receptors, even though some of these are likely areas with substantial noise contributions from various stationary sources of noise (deliveries, trash services, HVAC use, etc.).	
-21	Section 6.6.1.5. Existing Stationary Noise	6.64	The PEIR states, "Stationary sources of noise within the Uptown CPU area are due to the normal activities associated with a given land use Noises from these types of activities would be considered normal environmental noises that would be expected to occur within these types of land uses and are not typically considered significant sources of noise." Flawed logic in this paragraph suggests that noises that occur from normal use cannot be sources of significant noise. For example, a neighbor's party can substantially increase	

ATTACHMENT 7

RESPONSE

The six noise measurement locations are an adequate sample size to construct an acceptable noise model because they are representative of the range of noise environments that occur throughout the project area The noise measurement locations were chosen to represent the general noise environment in the project area and are sufficient to identify major noise sources and to characterize typical noise levels in the project vicinity. The measurement locations provide an adequate sample of traffic noise, which is the dominant noise source in the CPU area. Based on noise measurement guidance published by Caltrans, a noise measurement representing an hourly equivalent noise level does not need to last the entire hour. As long as noise levels do not change significantly, a shorter time period is sufficient to represent the entire hour of interest. The recommended length of measurements depends on traffic volumes and how much the noise level fluctuates, and generally ranges from 10 to 30 minutes and is an acceptable procedure. Because vehicle traffic noise in the CPU area is a relatively steady noise source, 15 minutes is a sufficient time to establish that the measured value adequately represents the noise source. The 15-minute duration is adequate for representing a 1-hour average noise level. Furthermore, the noise measurements taken in the CPU area primarily function as a tool to calibrate and validate the traffic noise model. Selected measurements included traffic counts which were required to validate the vehicle classification mix used in the analysis. This measurement should not be mistaken for representing the baseline ambient noise levels of the site. Although noise measurements were not conducted during peak hour, the vehicle classification mix observed is representative of the peak hour. The analysis of future vehicle traffic noise impacts is not based on the existing noise measurements; rather, it is based on the future daily traffic volumes on the roadways. These volumes were used to calculate the community noise equivalent level, which is a time-weighted 24-hour average noise level.

C18-20

C18-21 The PEIR identifies stationary sources of noise that are typical of given land uses. These examples of noise sources are not intended to be all inclusive. The City regulates excessive and annoying noise within City limits through enforcement of the Noise Abatement and Control Ordinance of the City's Municipal Code. As discussed in the FEIR, regulations in the Noise Abatement and Control Ordinance are in place to control noise and reduce noise impacts between various land uses. Given implementation of these policies and enforcement of the Noise Abatement and Control Ordinance of the Municipal Code, impacts would be less than significant.

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question		
			noise levels above what is allowed. Or idling delivery truck can easily surpass regulated noise levels. Both events would require mitigation or could result in nuisance complaints. Lack of consideration of stationary noise sources throughout the Noise Analysis for Community Plan Updates could be a result of this flawed thinking and needs to be amended.		C18-22	Comment noted Annendix E of the FEIR includes the detailed
8-22		Figure 6.6-2 (Page 6.6,-5)	Figure caption misrepresents what is shown. Contours plotted are not "existing noise contours"; they are modeled contours calculated using assumptions that were based on currently available traffic patterns and FHWA Traffic Noise Model.			methodology for creation of the noise contours. The figure label is intended to describe that the contours shown represent existing conditions. No revision to the PEIR is required.
					C18-23	More technical information regarding noise model assumptions
8-23	6.6.4 Impact Analysis	6.6-7- 6.6.24	No details are presented to understand assumptions that went into the 2035 traffic noise analysis models. Noise Analysis document does not present how they differ from current conditions. Therefore it is not possible to evaluate how the results presented relate to the CPU. Noise Ordinance Compliance: There were no measures or models presented to evaluate potential impact of new plan on likelihood of noise ordinance compliance. This would be of particular concern in residential areas that are bordering commercial and other uses area. Why were mitigation measures only considered in terms of construction? Why not consider mitigation of other noise sources?			Are included within a technical appendix to the PEIR (Appendix F). Modeling at a program level of analysis to determine compliance with property line limits is not possible because this is an analysis that must occur at a project level considering the applicable land uses, existing and proposed structures and noise levels. The analysis provided in Section 6.6.4, Issue 4 is adequate because it demonstrates that noise policies, as contained in the General Plan Noise Element, the proposed Uptown CPU, and regulations in the Noise Ordinance are in place to control noise and reduce noise impacts between various land uses. Mitigation was not required for this issue because the existing regulatory
8-24	6.6.4 – Impact Analysis	6.6.7- 6.6.24	Cumulative impacts of noise were not considered quantitatively. The models all separated traffic, rail, and airplane noise. No consideration of commercial use impacts or stationary impacts were conducted, and combined contributions of multiple noise sources at any given location were also not quantified.		C18-24	The entire analysis in Section 6.6.4 is cumulative in nature because it is based on traffic noise levels at build-out of the proposed CPU and traffic is the main noise contributor in the CPU area. Airport noise and rail noise is discussed separately because these noise sources occur within discrete areas of the CPU area and different methodologies are required.

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question	C18-25			
C18-25	Appendix F, Table 3	29.	Table 3 purports to present "City of San Diego noise and land use compatibility guidelines" but is not consistent with such Table 4-K in California Environmental Quality Act Significance Determination Thresholds from the City of San Diego	California Environmental Quality Act, Significance Determination Thresholds, City of San Diego				
	Inconsistency b Preparation (No	etween the j OP), and the	I proposed Uptown Community Plan, the Gene draft PEIR:	ral Plan, the Notice of				
C18-26	Preparation (NOP), and the draft PEIR: Various parts of the EIR erroneously state that the land use plan and land use designations in the Uptown Community Plan Update are consistent with the General Plan (for example, see PEIR Section 3.1, Introduction, p. 3-2, and PEIR Section 6.1.3, Impact Analysis, p. 6.1-2). The inconsistency between the General Plan and Uptown CPU is evidenced by the fact that much of the land in Uptown has been designated as Neighborhood Commercial and Community Commercial with overlapping designations as Neighborhood or Community Villages in the CPU. These overlapping commercial zones and Village designations, however, conflict with the General Plan's policies and goals, as will be demonstrated below. Thus, the Planning Department's use of these designations in Uptown is contrary to the City of							
C18-27	The City of Villa inconsistencies not change com of land use, den planners have fi of Villages Strat and General Pla General Plan sta into site-specific demonstrates ti General Plan, ho	ges Strategy between the imunity plan isity, and inter ree reign over egy and Gene in over comm ates, "It will b c recommend hat communi owever, does	(Strategic Framework) states, "there are no l General Plan and any given community plan b land use designations. The community plans a ensity." This does not explicitly mean, however r land use designations. Such an interpretatio eral Plan Policies meaningless. The authority o hunity plans is demonstrated by the fact that t be the role of the community plans to refine G lations that will guide the development of eac ty plans may refine goals and policies of the G i it state that Community Plans can contradict	land use or zoning because the General Plan does are the final arbiter on issues r, that the city and community in would render both the City of the City of Villages Strategy he Land Use section of the eneral Plan goals and policies th community" (p. LU-13). This eneral Plan. Nowhere in the the General Plan.	C18-27			
C18-28	Contradicting p amendment of Uptown CPU, T CPU (for examp "Maintain consi City's comprehe plan, action mu manner that is o	olicies in Com land use polit he proposed le, see PEIR S istency betwe ensive plan. It st be taken to consistent wi	munity Plan Updates and the General Plan sh cres in the General Plan, which was not recom amendment to the General Plan only includes tection 3.4, Project Description, p. 3-3). Gener cen community plans and the General Plan, as in the event of an inconsistency between the G o either: 1) amend the community plan, or 2) a th the General Plan's Guiding Principles."	ould be reconciled through an mended as part of the s incorporation of the Uptown ral Plan Policy LU-C.1.c. states, together they represent the ieneral Plan and a community amend the General Plan in a	C18-28			
C18-29	General Plan Po and deviating fr	olicy LU-B,1. e rom General I	explicitly describes the latitude planners are giv Plan Designations:	ven in designating land uses				
					C18-29			

Table 3 summarizes the noise and land use compatibility

- guidelines established in the 2008 General Plan. It is also noted in Appendix F that in 2015, the City Council approved a General Plan amendment to the Noise Element to change the guidelines for park uses. Table 3 has been updated to reflect the most recent General Plan amendments. The City is using the General Plan Noise Element for determining noise and land use compatibility. These compatibility levels are consistent with the levels shown in Table K-4.
- C18-26 Comment noted. The City does not agree that the use of the Neighborhood Commercial and Community Commercial designations are contrary to the City of Villages strategy. The Community Commercial and Neighborhood Commercial land use designations in the proposed community plan are consistent with the City of Villages Strategy because these land use designations provide the flexibility for stand-alone commercial uses to continue to provide goods and services within Uptown's community and neighborhood village areas as well as the opportunity for creating mixed-use development.
- C18-27 Comment noted. These comments do not raise an issue with regard to the adequacy of the PEIR. Please refer to response to comment C18-26 for a discussion of the proposed CPU's consistency with the City of Villages strategy.
- C18-28 Comment noted. The City does not agree that there is a conflict between the proposed Uptown CPU and the General Plan. Refer to Section 6.1, Land Use, and Section 6.5, Greenhouse Gas Emissions, which provide discussions on the proposed Uptown CPU's consistency with the General Plan.
- C18-29 Comment noted. These comments do not raise an issue with regard to the adequacy of the PEIR. The General Plan does not require compliance with the land use designations in Table LU-4;

Use the recommended Community Plan Designations identified on Table LU-4 so that over time, all community plans will use a common nomenclature to describe similar land uses and densities.

a. Use community plan text and graphics to provide greater specificity than is provided on Table LU-4, as needed.

 Identify the lower and upper ends of the allowable density ranges in community plans, with environmental review.

2. Allow community plans to analyze and define a more narrow residential density range within the ranges established on Table LU-4.

3. Use icons to identify various types of institutional uses.

4. Establish standards for population density and building intensity for each land use designation as community plans are updated.

Policy LU-B.1. thus states that residential densities must remain within the ranges established in Table LU-4. At least one zone proposed for Uptown deviates from the Designations in Table LU-4:

CPU Zone	CPU Density (du/ac)	General Plan Designation	General Plan Density (du/ac)
CC-3-9	109	Community Commercial	0 - 74

C18-30 General Plan Land Use Policy LU-A.1.c. states that the designation of Neighborhood, Community, and Urban Village Centers must be consistent with the goals of the General Plan. Community and Neighborhood Villages are described in the following way (see General Plan p. LU-7):

> Community and Neighborhood Village Centers should be located in almost every community plan area. They are community- and neighborhood-oriented areas with local commercial, office, and multifamily residential uses, including some structures with office or residential space above commercial space. Village Centers will contain public gathering spaces and/or civic uses. Uses will be integrated to the maximum extent possible in order to encourage a pedestrian-oriented design and encourage transit ridership. Community and Neighborhood Village Centers range in size from just a few acres to more than 100 acres. Community Village Centers are intended to serve a larger area than Neighborhood Village Centers. Community Village Centers may also have a more significant employment component than a neighborhood village.

> In Table LU-4, a Neighborhood Village is further described as providing "...housing in a mixed-use setting and convenience shopping, civic uses as an important component, and services serving an approximate three mile radius." A Community Village is described as providing "...housing in a mixed-use setting and serves the commercial needs of the community-at-large, including the industrial and business areas. Integration of commercial and residential use is emphasized; civic uses are an important component. Retail, professional/administrative offices, commercial recreation facilities, service businesses, and similar types of uses are allowed." Table LU-4 also describes a new "Multiple Use" Designation, which both Neighborhood Village and Commercial Village fall under. In both cases, residential development im Villages is required.

C18-31 Each of these village descriptions tends to favor a community and housing orientation over a commercial orientation. By contrast, a Community Commercial Designation, "Provides for shopping areas with retail, service, civic, and office uses for the community at large within three to six miles. It can also be applied to Transit Corridors where multifamily residential uses could be added to enhance the viability of existing commercial uses." This designation explicitly places a preference on commercial C18-29 (cont.)

rather they are listed as "Recommended Community Plan Land Use Designation." Additionally, the footnote in Table LU-4 General Plan and Community Plan Land Use Categories under General Plan Density Range (du/ac) indicates that residential density ranges will be further refined and specified in each community plan. Furthermore, General Plan Policy LU-B.1.a and LU-B-1.a.1 state respectively to use community plan text and graphics to provide greater specificity than is provided on Table LU-4, as needed and identify the lower and upper ends of the allowable density ranges in community plans, with environmental review. The CC-3-9 zone is being used to tailor particular areas within the Uptown Community designated for Community Commercial – areas that are characterized with community commercial serving uses and mixed-use development that allows very high residential density.

C18-30 Consistency with the City of San Diego General Plan City of Villages Strategy is provided in Section 6.1.3, Issue 1 of the PEIR. Refer to response C18-26.

C18-31 Comment noted. The City does not agree that inappropriate and unnecessary information was included in the PEIR.

- uses over residential uses. Furthermore, Commercial Land Use is described in detail in the Economic Prosperity Element of the General Plan (pp. EP-13 to EP-16) and is generally associated with regional and subregional employment zones.
- C18-32 Even if one successfully argues that commercial Designations are appropriate for Uptown, it is reasonable to also argue that the current extent of commercial designations in Uptown is excessive and that the majority of areas labeled as Villages in the CPU should be designated multiple use instead of commercial. Commercial designations are contrary to the City of Villages strategy, which defines a village as "the mixed-use heart of a community where residential, commercial, employment, and civic uses are all present and integrated" (General Plan, p. SF-3). Further support of Village Designations in more community-oriented areas versus explicitly commercial areas is the suggested placement of Multiple Use zones throughout the more established urbanized neighborhoods in San Diego on General Plan Figure LU-2. Areas with commercial designations in Figure LU-2 include Mission Valley and the Midway District, which have far more intensive commercial uses than any part of Uptown.
- C18-33 In order to support Village development in San Diego, the General Plan explicitly requires planners and the City Council to define, adopt, and apply a new "Multiple Use" Designation. The General Plan states that the Multiple Use category "will emerge within the central urbanized communities and downtown area (Additional areas of multiple use development would be anticipated as community plans are updated and village sites are designated.)" (General Plan p. LU-5). As of now, the Multiple Use Designation has not yet been adopted into the Land Use Code, though as the General Plan suggests, this should not stop planners from defining and implementing the designation. As a starting point, the following description could be used for Multiple Use zones in the Land Use Code:

The purpose of the multiple use zones is to provide housing, services, civic uses, and open spaces in a mixed-use commercial and residential setting with a pedestrian orientation and emphasis on alternative means of transportation to mitigate negative transportation impacts of growth. The multiple use zones also explicitly recognize the sharing of space between residential and commercial uses and include mechanisms like appropriate setbacks and development incentives to mitigate conflicts between commercial and residential interests.

C18-34 Given the inconsistencies between the Uptown Community Plan Update, the General Plan, and City of Villages Strategy described above, please further clarify how the PEIR can claim that these documents are consistent in their current form.

> Furthermore, please clarify how Community Commercial and Neighborhood Commercial Designations can be used consistently with Village Designations?

Finally, please describe the process that led to the inclusion of the Commercial designations in the CPU over Multiple Use designations. Was the Multiple Use designation suggested to the community for use in the Community Plan Update? If so, when was it suggested and why was the suggestion disregarded?

Procedural Shortcomings:					
PEIR Section	Page(s)	Comment/Question	Legal Basis for		
	0.000	Contracting Market of 1960	Comment/Question		

- C18-32 This comment does not suggest an inadequacy of the PEIR. However, the City does not agree that commercial designations are contrary to the City of Villages Strategy. Rather, commercial land uses strategically along transit and pedestrian/bicycle corridors and near residential uses supports mixed-use communities.
- C18-33 This comment does not suggest an inadequacy of the PEIR. While no specific "Multiple Use" designation is called for in the proposed Uptown CPU, several land use designations, such as Community Commercial, allow for mixed residential uses, thus supporting multiple use zones.

C18-34 The City does not agree that there are inconsistencies between the proposed Uptown CPU, General Plan, and City of Villages Strategy. Refer to Section 6.1, Land Use, and Section 6.5, Greenhouse Gas Emissions, which provide discussions on the proposed Uptown CPU's consistency with the General Plan and the General Plan's City of Villages Strategy. See response C18-33 regarding multiple use zones.

	PEIR Section	Page(s)	Comment/Question	Legal Basis for Comment/Question
C18-35	Entire document		CEQA Guidelines suggest that draft EIRs of unusual scope and complexity should normally be less than 300 pages. Although the period to respond to the draft EIR was extended to 60 days, the length and inclusion of inappropriate and unnecessarily burdensome review process. Although this issue is not necessarily contrary to CEQA rules or guidelines, it does call into question the motives and qualifications of the city in preparing a clear, concise, and accurate EIR document.	CEQA Guidelines 15141: "The text of draft EIRs should normally be less than 150 pages and for proposals of unusual scope or complexity should normally be less than 300 pages."

Thank you for consideration of these comments and questions. Please let us know if you would like further clarification on any of the items described above.

Sincerely, Ryan Eddings

Ana Širović, Ph.D

C18-35 Due to the complexity of the proposed Uptown CPU, the EIR length, though longer than the length suggested by the CEQA Guidelines, was appropriate to disclose all potential environmental impacts. Additionally, the City provided the public with additional time beyond the required 45 days to review and consider the information contained within the PEIR.

Letter C19

THOMAS A. MAY, ESQ. AND CYNTHIA L. ELDRED, ESQ. 4303 ALTAMIRANO WAY SAN DIEGO, CA 92103 <u>TOM@ELREDLAW.COM</u>; <u>CINDY@ELDREDLAW.COM</u> (619) 233-7366

VIA ELECTRONIC MAIL ONLY

July 17, 2016

City Council of the City of San Diego Planning Commission of the City of San Diego Mayor Kevin Faulconer 202 "C" Street San Diego, CA 92101

- Re: Uptown Community Plan Update Proposed Designation of a Presidio Hills Potential Historic District
- C19-1 Dear Members of the City Council, Members of the Planning Commission, and Mayor Faulconer:

We have lived at 4303 Altamirano Way in Mission Hills since 1999. We and many of our neighbors oppose designation of 59 homes located on Altamirano Way, Marilouise Way, Presidio Drive, Arista Street, and Cosoy Way as the "Presidio Hills Potential Historic District". Further, we and many of our neighbors oppose designation of these homes as a "Historic District". Finally, we oppose designation of our "heavily altered" home as a "potential contributor" to a "potential historic district".

- C19-2 City regulations currently in place already burden homes over 45 years old with building and development review not applicable to other homes within the City. These regulations have the potential to result in an involuntary or voluntary designation of any individual home as "historic", thereafter limiting repair, maintenance, and improvement of the home.
- C19-3 If the "Presidio Hills Potential Historic District" were approved as provided in the draft Uptown Community Plan Update (the "Update") and the accompanying draft regulations, a home determined not to be historic during an over-45-year review, would nevertheless be subject to additional review to determine whether it is a "potential contributor" to a "potential historic district". If the City made that determination, the proposed repairs, maintenance, and/or improvements to the front 2/3rds of the home would be limited. As

C19-1 Comment regarding the opposition of the Presidio Hills Potential Historic District and the identification of the home at 4303 Altamirano Way as a potential contributor structure is noted. However, it should be understood that neither the Presidio Hills Potential Historic District, nor the property at 4303 Altamirano Way will be designated as a result of the proposed CPU or the establishment of the supplemental regulations for potential historic districts.

- C19-2 Comment noted. This comment does not raise an issue with regard to the adequacy of the PEIR. The existing Historical Resources Regulations provide protections for properties 45 years old or older which appear eligible for designation as individually significant resources. The proposed supplemental development regulations to the Historical Resource Regulations are proposed to avoid significant and irreversible impacts to potential historic districts, which are not protected by the existing regulations. The proposed supplemental regulations would only apply to residential structures that have been identified as a contributing resource to the Potential Historic District. Properties that have been significantly altered would likely be found to be non-contributing, and the proposed supplemental regulations would therefore not apply.
- C19-3 This comment does not identify an inadequacy of the PEIR. Contributors to a historic district must be constructed within the identified period of significance of the historic district, relate to the theme for which the district was identified as being significant, and retain sufficient integrity to convey the significance of the district. Therefore, contributing structures would not arbitrarily include nonhistoric homes that do not contribute to the historic significance of a district. As such, while a structure within a potential historic district that is over 45 years of age may be subject to additional evaluation, if it is determined to not be historic it would not be considered a contributor.

a result, non-historic homes would be subject to historic regulations, all without substantive or procedural due process considerations, and for an unlimited period of time.

- C19-4 The City through its Historical Resources Board (the "HRB") already has vehicles in place pursuant to which individual homeowners may obtain historical designation status for their homes. We do not oppose those existing regulations. By the same token, the City through its HRB already has vehicles in place pursuant to which property owners within an area may work together to provide a majority vote to obtain "Historic District" status when the City determines by substantial evidence and through a publicly noticed process that the proposed area qualifies for that status. We do not oppose that voluntary process.
- C19-5 We do oppose designation of a "Presidio Hills Potential Historic District" based on no more than the preliminary analysis provided in the Update and the draft environmental impact report for the Update. We oppose the establishment of the 59 homes as a "potential historic district" without the consent of at least a majority of the homeowners, and without the thorough analysis that must be completed prior to designation of any area as a "Historic District".
- C19-6 We also oppose designation of our home as a "potential contributor" to a "potential historic district". The Draft Historic Resources Survey Report (the "Draft Report") appended to the Update lists our home as a "potential contributing resource", while at the same time describing it as "heavily altered". The Draft Report classifies the architectural style of our home as "Minimal Traditional" and does not list this style among the six "resource types" that are "contributing resources" to the proposed "Presidio Hills Potential Historic District".

City Historical Resources Staff have reviewed documentation that we have provided and told us that despite the language of the Draft Report, our home is not a "potential contributing resource". Nevertheless, Staff have told us that the Draft Report is in final form for Planning Commission and City Council decision and that the Status Code for our home (5S3; Appears to be individually eligible for local listing or designation through survey evaluation) will not be corrected to indicate that our home is not a "potential contributing resource". The Planning Commission should not recommend approval and the City Council should not approve any documentation with known inaccuracies.

C19-7 We respectfully request that the City remove the "Presidio Hills Potential Historic District" from any consideration as such in the Update.

Very truly yours,

Cynthia L. Eldred, Esq.

2

Thomas A. May, Esq.

cc: Marie Lia, Esq.

- C19-4 Comment noted. This comment does not identify an inadequacy of the PEIR. The process to apply for and establish a Historic District is not proposed to change.
- C19-5 Comment noted. The Presidio Hills Potential Historic District will be designated as a result of the CPU or the establishment of the supplemental regulations for potential historic districts. Designation of the district will require an intensive level survey and processing consistent with the requirements of the HRB's District Procedures, which includes an opportunity for property owners to voice their position on the designation of the district, and a separate public hearing process.
- C19-6 Comment noted. Neither the Presidio Hills Potential Historic District, nor the property at 4303 Altamirano Way will be designated as a result of the CPU or the establishment of the supplemental regulations for potential historic districts. The Historic Resources Survey will be used as an informational tool and baseline for future, property-specific analysis at the time a building permit application is submitted. If it is determined based on that property-specific analysis that the building does not contribute to the potential significance of the district – either due to alterations or other factors – then it would not be subject to the new supplemental development regulations.

C19-7 Comment noted.

LETTER

	Letter C20		
	Date: August 5, 2016		
	To: Kurtis Steinert, Senior Environmental Planner, City of San Diego Planning Department		
	RE: Uptown Community Plan Update, Project NO: # 21002568/SCH NO. <i>Pending</i> (# on public notice) #380611 (# on document)		
	From: Carol Emerick		
C20-1	My comments address the issues listed in the introduction to the DPEIR for the Uptown Community Plan Update.	C20-1	Comment noted.
	PEIR Introduction: "The purpose of this document is to inform decision-makers, agencies, and the public of the significant environmental effects that could result if the project is approved and implemented, identify possible ways to minimize the significant effects, and describe reasonable alternatives to the project."		
C20-2	Now that I am informed, I would like to I. address the significant environmental	C20-2	Comment noted.
	minimize the significant effects and III . describe reasonable alternatives to the project.		
C20-3	 Significant environmental effects from approval of the UCP June 2016 are seen in the areas of 1. Land Use and Character of the Neighborhoods 2. Transportation 3. Noise 4. Historical Resources 5. Parks and Libraries 	C20-3	Comment noted. Contrary to the comment, the PEIR identified impacts related to land use and community character (Section 6.1, Land Use) and parks and libraries (Section 6.12, Public Services and Facilities) would be less than significant.

C20-4	1. Land Use and Character of the Neighborhoods	C20 4	Commont noted
	"The amount of area in each land use designation under the CPU is shown on Table 3-8. The predominant land use designation in Uptown would remain residential, with single-family residential at 726 acres mirroring the current prevalence of single-family	C20-4	Comment noted.
	houses. The multi-family designation would comprise 380 acres." DPEIR Land Use "3.6.1 Uptown Land Use Distribution at Plan Build-out"		

	Table 3-8 DPEIR for Uptown Community Plan Update, June 2016		
	PROPOSED LAND USE IN Uptown Build-Out 2035		
	ACRES % of TOTAL ACREAGE		
	Residential - Single 726 27%		
	Residential - Multi 380 14%		
	Residential Total 1,106 41%		
	<u>^</u>		
	Compare the build-out in 2035, above, to the 2016 build-out below.		
	TABLE 2-2 Uptown Community Plan Draft, June 2016, p. LU22		
	EXISTING LAND USE IN Uptown June 2016		
	ACRES % OF TOTAL ACREAGE		
	Single-family Residential 852 32%		
	Multifamily Residential 277 10% Residential Total 1.129 42%		

		C20-5	This comment restates information contained in the PEIR. It is
C20-5	As the DPIR states, single family homes continue to have more acres than multi-family structures. but		correct that there would be more multi-family residential units under the proposed Uptown CPU, but less overall acreage of this
	there are 126 fewer acres of single family homes in 2035 than in 2016, and		designation. The information contained in the Draft PEIR disclosed
	there are 103 more acres of multi-family homes in 2035 than in 2016 and		this information (refer to Tables 3-8 through 3-10). However, to
	there are 23 <i>fewer</i> acres of residential land in 2035 than in 2016.		further clarify the proposed land use changes, an additional statement was added to Section 3.6.1, Uptown Land Use Designation at Plan Build-out, to state: "Overall, implementation of the proposed Uptown CPU and associated discretionary actions is anticipated to result in a shift from single-family residential units to multi-family units. Specifically, the number of single-family units is anticipated to decrease by 2,020 units and the number of multi-family units is anticipated to increase by 11,560 units (refer to Table 3-9)."

C20-6 In the PEIR UCPU, "Table 3-9 describes the existing and proposed residential development anticipated from This comment restates information contained in the Draft PEIR application of land uses shown on the proposed Uptown Land Use Map on vacant and underutilized parcels, C20-6according to analysis undertaken for the CPU". Uptown Community Plan Update PEIR Page 3-31 and does not identify an inadequacy in the PEIR. The comment makes a statement that single-family homes will not be prevalent In the PEIR UCPU, Table 3-9 Residential Development: Existing and at Proposed CPU Build-out in 2035 in Uptown. However, to be clear, existing single-family **Existing Development** Proposed Plan Build-out areas such as those that existing in Mission Hills and University (27) % Decrease Single-Family Units 7,540 33% 5,520 17% (2,020) Heights are not expected to change significantly since these areas **Multi-Family Units** 15,620 67% 27,180 83% 11,550 74% INCREASE are largely built-out and the proposed Uptown CPU would apply Total Housing Units 23,160 100% 32,700 100% 9,540 41% INGREASE Household Population 36,750 55,700 18,950 52% INCREASE residential-low land use designation. Notes: 1 Includes detached single-family, multiple-unit single-family, 2 Includes residential units in mixed-use development. Source: City of San Diego 2016" According to the proposed 2035 build-out, Uptown will change from single family units to predominantly multi-family units . C20-7 C20-7 Comment noted. Refer to response to comment C20-6. Single-Single-family homes will not be prevalent in 2035 in Uptown. family areas are not expected to change to multi-family areas 5.1.1.1 Land Use and Community Planning Element under the proposed CPU because the existing single-family areas "Community Plans are important because they contain specific policies in the CPU area are largely fully developed parcels that could not that protect community character. Future public and private projects will be accommodate new multi-family development. Additionally, evaluated for consistency with policies in the Community Plans." Uptown Community Plan redevelopment of single-family areas is not anticipated to occur Update PEIR Page 5-4 since it would not make financial sense to tear down existing The predominant character of Uptown is single-family homes in favor of a small multi-family development not currently multi-family units. that could be accommodated under a residential-low land use designation. In the PEIR UCPU, Table 6.1-1 Proposed Uptown CPU Policies Related to Land Use LU-2.4 Preserve existing single-family homes and neighborhoods as a distinct housing choice as well as for their contribution to the historic character of the community. LU-2.5 Preserve and enhance the special character of specific, well-defined, low-density neighborhoods from encroachment by incompatible, higher density residential or commercial development.

C20-8 3.6 Plan Build-out Comment noted. The comment does not make any statements C20-8 "The proposed CPU does not specify or anticipate when build-out would occur ... that would conflict with the information in the Draft PFIR. build-out is assumed to occur in 2035. "DPEIR Land Use " Earlier Uptown Community plans stated build-out occurred in 1930. "Uptown had been largely built out by the 1930's" UCP Draft June 2015, p. HP-142 Since build-out occurred in 1930 in Uptown, it would now be more accurate to say build-UP for the future density that is predicted. There is no more out, only UD. "Future development realized under the proposed land use map is referred to as build-out. The proposed CPU does not specify or anticipate when build-out would occur, as long-range demographic and economic trends are difficult to predict. However, for facility planning, technical evaluation, and environmental review purposes, build-out is assumed to occur in 2035. UCPU PEIR, Page 3-30 Comment noted. The comment does not make any statements C20-9 C20-9 **NEW ZONES** that would conflict with the information in the Draft PFIR. In order to achieve higher density the planning department added new zones that allow buildings to be 150' high. Build-out occurred in 1930. All that is left is to build up. The chart entitled UPTOWN-RECOMMENDED ZONE DESIGNATIONS-MARCH 2016 is a perfect example of major zone changes that allow build-up . The chart shows the very-high zone designation of RM-4-10 has no height limit. The CPOZ charts show 150' heights will be allowed in some neighborhoods.

RTC-340

- C20-10 C20-10 2016 March In a chart published by the City of San Diego, entitled Uptown-Recommended Zone Designations-March 2016 there are 21 zones listed. New zones have been added to the development tool box. There are **11 footnotes** that explain the zone designations. Footnote #9 states "New zones to achieve mixed use density of 73 and 109 du/ac." This designation was given to three zones, RM3-9, CC3-8 and CC3-9. Two zones, RM-4-10 and CC3-9 had a 0 maximum height. How high can those buildings go? Footnote #11 states "Maximum building height and allowances will be dictated by CPIOZ for particular areas within the Uptown Community." CPIOZ allows 150' heights. So much for city-wide zoning. Hillcrest, is the particular area within the Uptown Community, that will get the 150 foot buildings, high in the sky. C20-11 2016 June Now, there are only 2 footnotes at the end of the MARCH chart NOW entitled Uptown-Recommended Zone Designations-June 2016. In the box heading the column entitled "Max Height" there is Footnote #2 that states "Maximum building heights will be superseded by CPIOZ." CPIOZ allows for 150 foot heights. The only people who benefit from 150 foot heights are the developers. There is less land cost and more profit per unit. There is more bang for their buck. Now, 2016, the new goal is to increase units on each lot by building up into the sky. The taller the structure, the more profit is generated. See Incentive Zoning Analysis, the report prepared by AECOM for the city of San Diego, September 25,2014 C20-11
 - The comment does not identify an inadequacy in the PEIR. As discussed in Section 6.2.3, Impact Analysis, of the PEIR, under the high-rise building policies, areas within the Uptown CPU area could be permitted to develop with buildings up to 100 feet in height; however, these areas would be covered by a Community Plan Implementation Overlay Zone (CPIOZ) Type B (see Figure 6.1-3 in Section 6.1, Land Use). Within the CPIOZ areas, projects would be subject to a discretionary review process that would implement the proposed Uptown CPU policies and recommendations, particularly those related to building height consistent with the Urban Design Element. Specially, CPIOZ-Type A identifies areas within the community where ministerial approval is granted for development that does not exceed 50 feet within Mission Hills and 65 feet in Hillcrest and Bankers Hill/Park West. CPIOZ-Type B identifies areas within the community where discretionary approval is granted through a Process 3 Site Development Permit for development that does not exceed 150 feet in Bankers Hill/Park West, 120 feet in central Hillcrest, and 100 feet in Hillcrest east of the SR-163. Proposed Uptown CPU Urban Design Element provides design guidelines by building types to control massing and ensure compatible transitions. Building setbacks and upper-story stepbacks are recommended to address massing and compatibility where more intense development is located adjacent to lower height buildings (refer to Urban Design Element policies related to development transitions). These policies and guidelines would ensure taller buildings would not adversely impact surrounding lower intensity properties through neighborhood incompatibility or through creation of excessive shade or shadows.
 - C20-11 Allowing higher buildings in certain areas furthers the goals of the General Plan's City of Villages Strategy and the City's Climate Action Plan (CAP) by increasing residential density along pedestrian, bicycle, and transit corridors. The City is obligated to implement both the City of Villages Strategy and the CAP as a tool to reduce greenhouse gas emission; this strategy also reduces traffic congestion and increase housing availability in the City.

C20-12 Table S-1 Summary of Significant Environmental Impacts

"Would the project result in a substantial alteration (e.g. bulk, scale materials or style) to the existing or planned (adopted) character of the area?

The proposed Uptown CPU includes policies that would encourage residential and mixed-use development that would be <u>consistent with the existing neighborhood character</u> and impacts would be less than significant. "

Really ?????

C20-13 The June 2016 UCP introduces zoning for buildings that are **150'**. **That would not be consistent with the existing neighborhood character**. In Hillcrest, where 30 foot buildings are average, 65 foot buildings are being introduced, but 150' buildings are not consistent with the neighborhood.

> 150' buildings are NOT consistent with the existing neighborhood character. Impacts will be significant.

What happened to the desire to preserve neighborhood character that is discussed in the 1974, 1988, 2015 and 2016 community plans?

Please see the sections in the DPEIR for the Uptown Community Plan Update on Transportation, Parks, Historical Resources and Noise, to understand the significant impacts of 150' buildings and higher density. C20-12 Comment noted. The comment does not make any statements that would conflict with the information in the Draft PEIR.

As stated in response to comment C20-10, proposed Uptown CPU C20-13 Urban Design Element provides design guidelines by building types to control massing and ensure compatible transitions. Building setbacks and upper-story stepbacks are recommended to address massing and compatibility where more intense development is located adjacent to lower height buildings (refer to Urban Design Element policies related to development transitions). These policies and guidelines would ensure taller buildings would not adversely impact surrounding lower intensity properties through neighborhood incompatibility or through creation of excessive shade or shadows. 150-foot buildings under the CPIOZ-Type B would require discretionary review prior to development permit approval by the City, which would include a review of the Urban Design Element and require building consistency with the design policies included.

RESPONSE

C20-14	In the DPEIP there is another element of a description of a zone that does not appear in the UCPU, June 2016 zone matrix.	C20-14	See response to comment C20-15.
	+ Land Use Designations with a +		
	Table 5-1 General Plan Land Use Categories		
	Residential – Very High None Provides for multi-family housing in the highest density range. 75+ du/ac		
	"Whenever a plus (+) sign is identified next to a density number, the upper limit may be further specified in a Community Plan without causing the need for amending the General Plan, upon evaluation of impacts."		
	Uptown Community Plan Update PEIR Page 5-2		
C20-15	There is no asterisk in the UPTOWN COMMUNITY PLAN June 2016	C20-15	Table 5-1 of the PEIR refers to land uses under the General Plan
	UPTOWN COMMUNITY PLAN June 2016		and Table 2-3 of the proposed Uptown CPU refers to land uses
	TABLE 2-3: UPTOWN COMMUNITY PLAN LAND USE DESIGNATIONS		Relationship to General Plan, of the PEIR for a discussion on how
	Residential – Very High None Provides for multifamily housing with a very high density range. 74-109 RM-4-10 zone 3.60 FAR	the General Plan and Community Plan work together to guide development in the community.	
	111-28		
	Why do the zone designation descriptions keep changing ????		

I FT	TFR

C20-16	2. Transportation Transportation and Circulation "Would the project result in an increase in p the existing traffic load and capacity of the intersections, freeway segments, interchan Uptown Community Plan Update PEIR Page	projected traffic, which is substantial in relation to street system including roadway segments, ges, or freeway ramps? " S-11 = Page S-18 Eight pages of impacts,	C20-16	Comment noted. This comment does not identify an inadequacy of the PEIR.
	Impact Level After Mitigation	Significant and Unavoidable		
C20-17	3. Noise "Would the project result in or create a sign Would the project result in an exposure of p levels which exceed standards established i Uptown Community Plan Update PEIR Page Impact Level After Mitigation "Would the project result in the exposure of construction noise? "	ifficant increase in the existing ambient noise levels? beople to current or future transportation noise in the Noise Element of the General Plan?" S-21 <u>Significant</u> and Unavoidable f people to significant temporary	C20-17	Comment noted. This comment does not identify an inadequacy of the PEIR.
	Uptown Community Plan Update PEIR Page	S-24		
C20-18	Impact Level After Mitigation 4. Historical Resource "Would implementation of the proposed pr physical or aesthetic effects and/or the des architecturally significant building), structur Uptown Community Plan Update PEIR Page	Significant and Unavoidable CES oject result in an alteration, including the adverse truction of a historic building (including an re, object, or site?" 5-26	C20-18	Comment noted. This comment does not identify an inadequacy of the PEIR.
	Impact Level After Mitigation "Would implementation of the project resu of a prehistoric archeological resource, a re human remains, including those interred ou Uptown Community Plan Update PEIR Page Impact Level After Mitigation	<u>Significant</u> and Unavoidable It in a substantial adverse change in the significance ligious or sacred use site, or disturbance of any itside of formal cemeteries? " S-28 - Page S-39 <u>Significant</u> and Unavoidable		

new or physically altered public facilities (including police

ratios, response times, or other performance objectives?"

5. Public Services and Facilities

C20-19 Comment noted. This comment does not identify an inadequacy of the PEIR.

Park and Recreation

C20-19

"Regarding park and recreational facilities, there is an existing and projected deficit in population based parks, which is an adverse impact, but not considered significant at the program level. Implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of population based parks in the CPU area, but does not propose construction of new facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to parks and recreation, and ho mitigation is required."

"Would the project promote growth patterns resulting in the need for and/or provision of

protection, parks or other recreational facilities, fire/life safety protection, libraries, schools, or maintenance of public facilities including roads), the construction of which

could cause significant environmental impacts in order to maintain service.

Uptown Community Plan Update PEIR Page S-46 - S-47

Impact Level After Mitigation

Less than Significant

C20-20 Wow ! So nice to hear the city will provide policy support for increasing the acreage of population based parks, but disappointing to learn the city does not propose construction of new facilities or parks. Therefore, the impact is less than significant since there won't be any.

> That explains why one of the top priority items in the 2003 Uptown Public Facilities Financing Plan, 6 acres of park space, has not become a reality for Uptown. See page 14 of the Uptown Public Facilities Financing Plan, 2003.

So the options for new parks are

1. DIF funds, that cover a fraction of what is considered adequate, or

2 How about naming rights for green spaces? There could be Gateway Park or Atlantis Green Space. Maybe a Pernicano's Park. C20-20 Comment noted, the comment does not identify an inadequacy in the PEIR. Community parks and park equivalencies are discussed in Section 6.12, Public Services and Facilities. Though there would be a deficiency in park and park equivalences at build-out of the proposed Uptown CPU, the existing conditions include a deficit in parks and park equivalencies. In addition, while the proposed Uptown CPU does not propose any individual project, 37.40 acres of proposed new population-based park land and park equivalency sites have been identified through the proposed Uptown CPU effort. The policy framework provided by the proposed Uptown CPU supports acquisition and development of new public parks and park equivalencies, and encourages new private development to include recreational facilities.

C20-21	1 Iburuina	C	20-21	Comment noted. This comment does not identify an inadequacy
2021	"Although a new library is planned for the Uptown CPU area, the proposed Uptown CP associated discretionary actions does not include construction of library facilities. Deve a new facility would be subject to separate environmental review at the time design pla available. Therefore, impacts related to library facilities would be less than significant, a mitigation is required.	PU and elopment of lans are and no		of the PEIR.
	Uptown Community Plan Update PEIR Page S-47			
	Impact Level After Mitigation Less than Signific	icant		
	That explains why another one of the top priority items in the 20	003		
	Uptown Public Facilities Financing Plan, p.14, a library, has not	t		
	become a reality for Uptown.			

C20-22	One more item	C	20-22	Comment noted. This comment does not identify an inadequacy of the PEIR.
	The 1974 Uptown Community Plan, page 92, states			
	undergrounding of utilities should be done "immediately".	100		
	We are still waiting, 42 years later. We have been paying on our	r utility		
	bill for undergrounding, since 1972, 44 years.			
	The most current information on the city web site states that	root will		
	occur in 2024, 50 years after the community plan was adopted	1.		
	"the proposed Uptown CPU contains policies supporting utility underground undergrounding is currently underway in the Uptown community." Uptown Community Plan Update PEIR Page 6.13-5	ding and		
C20-23	Good to know.	C	20-23	Comment noted. The proposed Uptown CPU is intended to provide a long-range guide for future physical development of the
	Mitigation of significant impacts can take many decades and			community for decades to come. Mitigation would be enforced at
	generations.			the project level as individual projects are proposed and
				implemented in accordance with the proposed Uptown CPU.

RESPONSE

LETTER

C20-24	II. Minimize the significant effects Please take the next few years to discuss the changes in density that have carelessly and recklessly been added, willy nilly, here and there since June 2015, with no consideration for the detrimental effects, of the significant and unavoidable impacts, the increased density will cause in Hillcrest.	C20-24	Comment noted. This comment does not identify an inadequacy of the PEIR. While significant and unavoidable impacts to transportation and traffic, noise, historical resources, and paleontological resources were identified, all feasible mitigation at the program level was identified and recommended. Future projects would be subject to the applicable mitigation identified in the PEIR, and may require subsequent environmental review pursuant to CEOA to analyze potential impacts at the project level.
C20-25	Until the transportation elements of the 1974 Uptown Community Plan listed on pages 55-72 are addressed, as well as all the other significant impacts affecting Uptown, historical resource issues, libraries and parks, and noise please delay build- UP .	C20-25	Comment noted. See response to comment C20-24.
C20-26	III. Reasonable Alternative Please adopt the Density Redistribution Alternative for the Uptown Community Plan. It is a reasonable alternative to the project, the June 2016 plan. Thank you for consideration of my concerns. Carol Emerick	C20-26	Comment noted.

	Letter C21		
	Planning Department Uptown Community Plan Update EIR		
	RE: Project No. 21002568 response		
	Planning Department,		
C21-1	Comments and responses are provided below for the EIR. Project Name: Uptown Community Plan Updates	C21-1	Comment noted. The City appreciates your participation in the public review process.
C21-2 C21-3 C21-4 C21-5	 Final PEIR Comments the stated project above: 1. We support the main project plan densities in the PIER along with the higher densities listed in "10.4 Density Re-distribution Alternative", (the multiple "City of Villages" core concept). 2. We feel the main project addresses the Climate Action Plan goals and objectives. 3. To address the crisis in affordable housing height and densities need to be in place accommodate below market units. 4. Keeping densities at an appropriate level is essential to place-making and smart growth. 3. We support the vision of the Gateway Council for the Hillcrest Core and the heights and densities required for that vision to happen. Ian Epley i.e., Design and Hillcrest CDC 3939 First Avenue San Diego, CA 92103 	C21-2 C21-3 C21-4 C21-5	Comment noted. Comment noted. This comment does not identify an inadequacy of the PEIR. The proposed Uptown CPU Policy LU-1.1 provides guidance for providing affordable housing by stating the following: "Provide a variety of land use types to accommodate both affordable and market rate housing and commercial opportunities." Comment noted. This comment does not identify an inadequacy of the PEIR. Comment noted.

C22-1

C22-2

C22-3

C22-4

C22-5

C22-6

C22-7

RESPONSE C22-1 Comment noted. The City appreciates your participation in the public review process. Letter C22 **Planning Department** C22-2 Comment noted. The comment does not raise an inadequacy in the Uptown Community Plan Update EIR PEIR. The Historical Resources Survey prepared for the project identified 19 potential historic districts containing a total of RE: Project No. 21002568 response approximately 2,600 properties and roughly 2,000 contributing Planning Department, resources. Four additional Potential Historic Districts were identified by the community through the public outreach process. However, Comments are provided with an EIR question at the end of comments. these potential historic districts have not been designated, and Due process, it is one of the fundamentals of our society. would only be designated after they "are intensively surveyed, So when the Historical Resource Staff over stepped and overreached to recommend creating verified, and brought forward for designation consistent with City Potential Historic districts without due process, it would have the same effect of shutting down progress in a community. And while this may be fine for nimbyists and no-growthers, the rest of regulations and procedures (6.7-25), which include workshops, the community should not be penalized. Everyone seems to talk about certainty that historic public hearings and noticing." Suspension of development would districts would bring, but because of the potential aspect of historic districts, building would be not be required to protect potential historic districts because held in limbo, because the rules for building are the same, whether an established district or potential. amendments to the Historical Resources Regulations would provide We have, ad nauseam, illustrated why historic districts are bad planning policy from nonclear and consistent supplemental development regulations to sustainability, drives up housing costs, only benefits the wealthy, circumvents housing for assist in the preservation of specified potential historic districts until families and affordable housing, etc. they can be intensively surveyed and brought forward for While a Historic Resources Board can be constructive working with historic resources, trying to designation. See Section 6.7.4, Issue 1 of the PEIR. Development save virtually everything is tantamount to supporting no growth. that does not comply with the development requirements of the The Climate Action Plan, (CAP) adopted by the City Council and the Uptown Planners board supplemental development regulations may still be processed with requires a policy toolbox that is agile enough to let a community progress for a vital, sustainable a Neighborhood Development Permit. future. There is no room for policies based on romantic notions of old or that prosper the preservation C22-3 Comment noted. The comment does not raise an inadequacy in the industry. There is already an established process for proposed districts to go through to be adopted or not. PEIR. The following is a response to the recommendations put forward by the Uptown Planners Historic Resources C22-4 Comment noted. This comment does not raise an inadequacy of the Historic policies taken out of the EP should remain out, there is no evidence of PEIR. historicity and economic prosperity, in fact there are many studies to proof the opposite. C22-5 Comment noted. The comment does not raise an inadequacy in the · We recommend that the City Council, as in the past, not adopt the Uptown Historical PEIR. The project would be consistent with the City CAP as described survey, due the subjectivity of identifying potential historic resources That the Uptown HRC drop the idea of a historic district in Hillcrest. Various in Section 6.5.3 c. of the PEIR. organizations have already established that the National Main Street program is the vehicle for historic resources.

LETTER

- That the board reject recommending the City Council to adopt an interim/conservation zone. This would keep the community in stasis.
- In the 'Overlay Strategy for Potential Historic Districts: Concerns and Solutions' section, we recommend rejecting all 5 solutions based from the concern that these are draconian and would be disastrous for any progress in Uptown especially Hillcrest.
- Historic districts do nothing to promote benefits of a community's future. They just fulfil the needs of preservation's hobby.

lan Epley, Former Chair of Uptown Planners

lepley1@gmail.com

C22-6 Comment noted. The comment does not raise an inadequacy in the PEIR.

RESPONSE

C22-7 Comment noted. The comment does not raise an inadequacy in the PEIR.

LETTER

RESPONSE

Letter C23 Planning Department Uptown Community Plan Update EIR RE: Project No. 21002568 response Planning Department, Comment noted. The City appreciates your participation in the C23-1 C23-1 Comments are provided with an EIR question at the end of comments. public review process. C23-2 As a former chair of the Uptown Planners Group, I would like to make an argument against the policy of Potential Historic Districts (PHDs) that have been written into the Historic Resources C23-2 Comment noted. The comment does not raise an issue with regard Element of the Uptown Community Plan Update. The Historic Resources staff is overreaching in its attempt to establish PHDs' in many areas of the Uptown Community. The problem with this to the adequacy of the PEIR. Designation of a Potential Historic policy is that it goes against the City of San Diego's current due process of determining a District would not carry the same rules and restrictions as if a historic district. The potential aspect of the PHD would carry the same rules and restrictions as Historic District was approved. Suspension of development would if the historic district were already adopted. Waiting for a PHD to be adopted into a bone fide historic district (which may or may not happen), would remove certainty to business owners not be required to protect potential historic districts because and homeowners and leave them in limbo when it comes to improving one's property. This has supplemental development regulations would assist in the been the reason for the Uptown Historical Survey not being adopted due to the subjective nature of determining the potential nature of historicity. preservation of specified potential historic districts until they can be intensively surveyed and brought forward for designation. See C23-3 We are appreciative that both the mayor's office and Uptown Planners have taken the lead Section 6.7.4, Issue 1 of the PEIR which explains that Historical initiative for the Climate Action Program (CAP). (The Uptown Planners also recommended adoption). The goals of the CAP would be much easier to achieve without the restrictions that Resources Regulations would provide supplemental development would be placed on our neighborhoods by a PHD. One of the consequences of the restrictive regulations to address how and where modifications can be made PHDs would result in re-investment in our neighborhoods virtually coming to a halt. Adaptive re-use or strict historic restoration is very expensive and without agile adaptive planning tools, on residential properties identified as potentially contributing to there would be less incentive to accomplish the goals of the CAP. Another consequence of the specified potential historic districts. Development that does not PHD would be the lack of affordable housing construction. Affordable housing and multi-family comply with the development requirements of the supplemental housing are essential components in lowering the carbon foot-print of the CAP, and accommodating the City of San Diego General Plan, Smart Growth planning, and Transit development regulations may still be processed with a Oriented Development (TOD). As an example, San Francisco has infamously suffered from the Neighborhood Development Permit. effect of Historic Districts, brought to our attention through documentaries and planning. studies. C23-3 Comment noted. The comment does not raise an issue with regard C23-4 PHDs also go to the respect of property rights. Imagine a homeowner across the street from an to the adequacy of the PEIR. None of the potential historic districts area that has been determined to be a PHD. While he/she can repair a front door or window without a City of San Diego building permit, the person across the street in the PHD not only are located within designated Transit Priority Areas; thus, they has to pay for a permit, but also must obtain the approval of the historic resources police to could not cause an inconsistency with the CAP. make sure the door is appropriate (another subjective decision). This is analogous to the City of San Diego acting as a "Home-Owners Association" (HOA), with more sustainable and energy efficient ones, thus making it more difficult to achieve the carbon foot-print goals of the CAP. C23-4 Comment noted. The comment does not raise an issue with regard According to the Hillcrest Business Association (HBA), virtually all of the business and property to the adequacy of the PEIR. owners are against the implementation of a PHD. The HBA, in lieu of PHDs, is already working

RESPONSE

the National Main Street program, a program that allows a community to have the toolbox to adapt to the future while embracing the past. The program has had great success in North Park.

- C23-5 There is a detailed process at the City of San Diego in place to address the adoption of a historic district. If there is a groundswell of neighborhood support to the adoption said district, then that's how it should play out, not from the top down but from the neighborhood up. We would recommend the concept of PHDs be removed from the Uptown Community Update Draft and emphasize the process for establishing historic districts that presently exists through the Historic Resources Board.
- C23-6 Question: Can or will the financial impact be studied or quantified in the EIR due to the loss of property tax to specific historic districts due to the increase in Mills Act applicants thus reducing funds for neighborhood infrastructure improvement?

lan Epley, past chair of Uptown Planners i.e., Design 3939 1st Avenue San Diego, CA 92103 Ph. 619 780.8992 Fax 619 299.4250 Cell 619 757.4888 iepley1@gmail.com

- C23-5 Comment noted. The comment does not raise an issue with regard to the adequacy of the PEIR.
- C23-6 Comment noted. The comment does not raise an issue with regard to the adequacy of the PEIR. Implementation of the Uptown CPU would not require or directly result in an increase in Mills Act applicants; thus, such an analysis is not warranted.

RESPONSE

l etter	C24
	C27

From: Sent: To: Subject:

Matthew Event <mjevent@yahoo.com> Monday, August 08, 2016 7:14 PM PLN_PlanningCEQA Uptown.community.plan.update.21002568/sch-

Hi city of San Diego planning dept,

C24-1 I am writing on behave of my myself and my tenants on living on Columbia St. I live in the section of Middletown that is really only accessible by the two way section of India at that services redwood at and Columbia street from the north side. It is my understanding that the city is considering removing this section of roads and all the street parking on India st. I feel this is extremely unfair to residents who have been living or have owned homes in the area since before the airport was constructed and since before more traffic has developed with the moving of the parking lots to the east side of the airport. We already have to car the city parking abatement at least twice a month to report people parking their cars for 5-21 days in our neighborhoods while they travel out of the city via out airport. Having even less parking available will only make this worse. Additionally, traffic is already bad on kettner to get to laurel st, sending locals that way will only further congest that bottleneck and further frustrate he local residents. Please consider other alternatives that don't involve further adding to the frustrations of the local Middletown community and promote a better place for all people to live and not just the benefactors of this proposed change to India street.

- Matthew Evert Columbia and Palm St.

Sent from my iPhone

C24-1 Comment noted. The comment does not identify an inadequacy in the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
ATTACHMENT 7

		Letter C25		
	From: Sent: To: Cc: Subject:	G Fairbourn «garyfairbourn@cox.net» Sunday, July 31, 2016 8:55 AM PLN: PlanningCEOA prnader@cox.net Objections to Uptown Community Plan Update Project No. 21002568.		
C25-1 C25-2 C25-3	I am in strong objection to every day and its removal v l am also in strong objection Winder St. This is a slow m it will severely impact the b Instead of trying to make N slow down the cars that spi I would like to know what y Thanks, Gary Fairbourn 2855 Union Street garyfairbourn@cox.net	the removal of the South bound lane on India from Sassafras to Redwood. I use this lane would severely impact how I access my home on Union Street. In to the removal of parking on the short segment of India between Washington St. and toving segment of road and allows parking for the local businesses. If you remove the parking pusinesses. I bet you have not provided an alternate parking scheme for these businesses. For the bound India all one way with more lanes for the cars, why don't you focus on trying to and through this 35MPH zone. Don't make it easier for them to speed. your final plan would look like. Can you send me some information?	C25-1 C25-2	Comment noted. The comment does not identify an inadequacy in the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6). The recommendation provided in this comment is consistent with Policy MO-1.14 of the proposed Uptown CPU, which supports traffic calming treatments on residential streets where excessive speeding occurs.
			C25-3	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

		Letter C26		
C26-1 C26-2 C26-3 C26-4	From: Sent: To: Cc: Subject: Mr. Jeff Murphy, Pla Mr. Kurtis Steinert, City of San Diego Pla J010 Second Avenue San Diego, CA 92101 Re: Uptown Commun The Mobility Study fo The Proposed Comm Diego's " City of Villa deemed "Transporta Why were so many i By what method, we Given the increased method did the city of density? And what r alternatives? Increased density is the current mobility of the existing transit ri- house gasses will im Funding has not bee trolley. Shouldn't the comes, in order to in Regards Tim Gahagan 3794 Herbert Street, 619-315-7058	In gahagan stgahagan (@yahoo come Friday, August 06, 2016 240 PM Murphy, Jeff, PLN_PlanningCEQA Tim Gahagan Uptown Community Plan Update PROJECT No. 21002568 anning Director Sr. Environmental Planner anning Department, MS 413, http Plan Update PROJECT No. 21002568 or Uptown included many mobility mitigation ideas that were left out of munity Plan. Efficient mobility is a necessary underpinning of San ages" concept and for justifying increased density in existing urban areas tion Priority Areas". mportant mitigation solutions left out of the proposed Community Plan? re the decisions made to leave out these identified mobility solutions? congestion possible in Uptown's already clogged streets, by what conclude that Green House Gasses would decrease with increased methods for greenhouse gas production in much of the world. Given conditions and minimally planned and funded improvements, and given dership trajectories in San Diego County, isn't it also possible that green crease, from the existing base with the city's plans for increased density? In secured for all the Proposed Plan mobility solutions such as the e city ensure these mobility solutions be provided BEFORE the density crease the chances that greenhouse gas reduction will actually occur? San Diego, CA, 92103	C26-1 C26-2 C26-3	All identified mitigation measures were evaluated pursuant to Section 15126.4 of the CEQA Guidelines and chosen based or feasibility and ability to reduce identified significant impacts. Refer to Appendix C, Mobility Study, of the PEIR for the feasibility of the identified improvements. Policies included in the proposed Uptown Mobility Element are elements of the project; and thus not included as mitigation measures. For example, the proposed Uptown Mobility Element has a focus on multi-moda improvements that would benefit pedestrian, transit, and bicycle commute options in the community. A number of specific improvements are identified in the proposed Uptown Infrastructure Fee Study (IFS) and the proposed Uptown CPU provides the policy support for implementation of these improvements. Section 6.5.1.1, Methodology and Assumptions, of the PEIR and Section 4.2, Methodology and Assumptions, of the Greenhouse Gas Analysis (Appendix E-1) provide a discussion of the method used to determine impacts related to greenhouse gas emissions The same methodology used to assess impacts of the proposed Uptown CPU were used to assess impacts of the alternatives Please refer to response B2-2 for details regarding the methodology for assessing GHG emission impacts. See response to comment C26-2. As discussed in Section 6.5 Greenhouse Gas Emissions, of the PEIR, increasing residentia density and commercial uses along pedestrian, bicycle, and transi corridors can decrease vehicle miles traveled (VMT) as trips between land use types are shorter and may be accommodated by alternative modes of transportation.
		1		

or funding for trolley improvements. However, the prope	ent,
Uptown CPU includes a number of policies that would supp	ort
coordination between the City and SANDAG and the Metropol	itan cioc
MO-3.1 to MO-3.12). Additionally, the City's CAP includes goals	for
GHG reductions that are monitored by the City annually.	CAP
Chapter 3 which provides for annual monitoring and reporting ensure CAP reduction targets are met. As shown on pages 42	g to and
43 of the CAP, the annual monitoring and reporting would ider	ntify
any potential deficiencies in reductions and the CAP could	be
amended to address those deficiencies.	

C27-1 Comment noted. The City appreciates your participation in the public review process. The comments do not indicate a specific
C27-1 Comment noted. The City appreciates your participation in the public review process. The comments do not indicate a specific
C27-1 Comment noted. The City appreciates your participation in the public review process. The comments do not indicate a specific
public review process. The comments do not indicate a specific
Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C27-2 Comment noted. Pacific Highway is outside of the Uptown CPU area. Please refer to the Master Response Regarding India Street
to comments (pages RTC-4 through RTC-6) which includes a
discussion on the Rental Car Center
discussion on the Kental Car Center.

From: Tracee Galetti stracee galetti@yahoo come Sent: Tuesday, August 02, 2016 4:41 PM To: PLN. FlarmingCEQA Subject: Uptown Community Plan Update Project No. 21002568/SCH No. Pending To whom it may concern: C28-1 C28-1 Lam a resident and homeowner in Middletown and currently reside in the Terracina complex off of the	
To: PLN. PlanningCEQA Subject: Uptown Community Plan Update Project No. 21002568/SCH No. Pending To whom it may concern: C28-1 Lam a resident and homeowner in Middletown and currently reside in the Terracina complex off of the C28-1 Lam a resident and homeowner in Middletown and currently reside in the Terracina complex off of the	
To whom it may concern: C28-1 Lam a resident and homeowner in Middletown and currently reside in the Terracina complex off of the C28-1 Comment noted. The City appreciates your participati	
C28-1 Lam a resident and homeowner in Middletown and currently reside in the Terracina complex off of the C28-1 Comment noted. The City appreciates your participation	
Intersection India and West Spruce. public review process.	ion in the
C28-2 This email is to document as a matter of record, that as a citizen of Middletown 92103 community I definitely Object to the recommendations of this proposal. I do not agree with the study findings. C28-2 Comment noted. This comment does not indicate	a specifi
The current study proposal to turn India into a three-lane one-way highway is Extremely concerning and inconsiderate as it is already difficult to turn right begins up the the sneed of testic with 2 inadequacy of the PEIR. Please refer to the Master	Response
lanes and someone's nearly impossible to get all the way over into the left turn lane already. I can't even imagine an extra lane to turn out and try to be in the immediate left. This will become even more dangerous and hazardous to the citizens that live in this community. Regarding India Street Mitigation Measures include introduction to these response to comments (pages RTC-	ed in the -4 through
C28-3 The proposal plan seems to be without the consideration of the existing community which includes homeowners and businesses that have been around for many years. I am a born and raised in San Diego and the recent traffic with the rental car access and freeway sions to exit Sassafrass has	
already made it hard enough but I cant Imagine this change. C28-3 Comment noted. Pacific Highway is outside of the Up	town CPL
Also, why cant Pacific Highway be better utilized as it is already equipped for heavier traffic? Please take the neighborhood and businesses into consideration that would be VERY negatively affected by the change if this occurs.	ndia Stree e response
Thank you for taking the time to read this email to comments (pages RTC-4 through RTC-6), which i	ncludes a
Concerned 92103 homeowner, discussion on the Rental Car Center.	
Tracee Galetti	

RESPONSE

C29-1 Comment noted. Letter C29 C_{29-2} Comment noted. This comment does not suggest an inadequacy From: Tom Gauld <tomgauld@gmail.com> Monday, August 08, 2016 1:27 PM Sent: in the PEIR. Mayor Kevin Faulconer, Councilmember Todd Gloria, PLN PlanningCEQA, Pangillinan To: Marlon Subject: Objection to Proposed Final Plan for Uptown Neighborhood and Hillcrest C_{29-3} Refer to PEIR Section 6.2. As discussed in the PEIR, the Urban Dear Elected Officials and City Planners: Design Element of the proposed Uptown CPU includes numerous The Uptown Planning Group has worked tirelessly over the past 7-years to analyze our neighborhood and policies that would direct future development in a manner than C29-1 discuss the community needs with residents. ensures the physical attributes of the community will be retained All this hard work and due diligence resulted in the submittal of a 2015 draft Community Plan that preserves and enhanced, and views of open space would be maintained. our historic neighborhood, its sense of community and walkability score. Policies UD-4.70 through UD-4.92 of the proposed Uptown CPU Sadly, as a result of lobbyists and absentce property owners who have blighted the neighborhood with poorly C29-2 maintained buildings and lack of investment, our neighborhood is facing the potential of having high density address development height, massing, and transitions that would forced into an already over-congested traffic area. The property owners will leave town with their profits and leave the residents to deal with the traffic congestion and the pollution. guide future development to be compatible with the existing surrounding development. More intense development would be The mountain views to the east and the natural light that flows through historic Hillcrest are key attractions to C29-3 living, dining, shopping and recreating in the neighborhood. These qualities are now under threat by the subject to setbacks and upper-story stepbacks to address massing proposed 100-200 foot buildings. This is a density that is more suited to downtown San Diego and not a pedestrian orientated historic neighborhood. and compatibility where adjacent to lower height buildings. In Hillcrest already deals with severe traffic congestion during peak-time rush hours. Traffic flow from east to addition, Policies UD-1.1 through UD-1.11 address the C29-4 west is regularly gridlocked during rush hour and the recent installation of additional stop signs along Robinson preservation of views, canyons, and natural open space in the Avenue has exacerbated the issue. Uptown Community. Zoning and Land Development Code This traffic congestion reduces the critical response time for the emergency vehicles that service the numerous C29-5medical facilities that drive the coonomic engine of the neighborhood. High rise buildings with underground regulations would further ensure development occurs in keeping parking garages and multiple ingresses/egresses will add to the traffic congestion and potentially have with the character of the community. Based on these catastrophic consequences by preventing emergency medical and fire vehicles from reaching their destination in a time critical manner. considerations and the numerous other design policies included It is admirable that the City of San Diego is looking to enhance public transportation and biking routes. C_{29-6} in the proposed Uptown CPU, impacts related to substantial However, Hillerest is, and always will be, a neighborhood where the majority of residents work in the professional employment centers that are not served by viable public transport options (UTC, Torrey Pines alterations to the existing character and natural views of the area Mesa, Sorrento Valley, Miramar). Adding high density to the core of Hillcrest will dramatically increase the would be less than significant. number of cars and associated pollution and traffic congestion. The back-up of traffic will ultimately overflow onto the 163 and 5 freeways and this will have huge consequences for traffic flow on 163, the 5 and 805. Trespectfully request that you preserve the physical and community character of historic Hillcrest by C29-4 Comment noted. This comment does not suggest an inadequacy C29-7 SUPPORTING THE PLAN SUBMITTED BY THE UPTOWN PLANNING GROUP and saying NO to the lobbyists and absentee property owners that have blighted our neighborhood. Please don't allow the absentee in the PFIR. property owners walk away from our neighborhood with huge profits while residents are left to live with the catastrophic consequence of over-development, pollution and traffic congestion Section 6.12.3, Impact Analysis, discusses potential impacts to C29-5 Sincerely emergency response times resulting from the proposed Uptown Tom Gauld 475 Redwood Street, Unit 408 CPU. As discussed in this section, implementation of the proposed San Diego, CA 92103 Uptown CPU and associated discretionary actions would result in an increase in overall population, which could result in a change in

C29-5 (cont.)

response times. However, future facilities would be planned based on adopted General Plan Public Facilities Element standards detailed in Chapter 5.0, Regulatory Framework (Section 5.12.1.3) of the PEIR. The proposed Uptown CPU and associated discretionary actions do not propose the construction of fire/life safety facilities. However, the proposed Uptown CPU contains a policy framework that addresses maintaining the high level of fire protection throughout the Uptown community. Additionally, as future development is proposed within the Uptown CPU area, individual projects would be subject to payment of Development Impact Fees (DIF), which would provide facilities financing in accordance with Municipal Code Section 142.0640. The Uptown GPU CPU includes a comprehensive update to the existing Impact Fee Study (IFS) that will define applicable DIF fees for future development, including funding for fire/life safety facilities.

C29-6 The proposed Uptown CPU assigns density to plan for growth that will complement existing land use patterns and encourage use of alternative forms of transportation. The PEIR does recognize significant and unavoidable impacts to freeway facilities would occur with buildout of the CPU. The Final PEIR has been revised to recognize that significant impacts to freeway facilities could be partially mitigated by transportation demand management (TDM) measures that encourage carpooling and other alternative means of transportation consistent with proposed CPU policies. Fair share contributions could also be provided toward the construction of the projects that are included in the SANDAG's San Diego Forward: the Regional Plan, including:

 C29-6 (cont.) Operational improvements along I-8 between I-5 to SR-125 (TRANS 6.3-34) Construction of managed lanes along SR-15 between I-5 and I-805 and between I-8 and SR-163 (TRANS 6.3-35) Construction of managed lanes along I-805 between SR-15 and SR-163 (TRANS 6.3-36) Construction of managed lanes along SR-94 between I-5 and SR-125 (TRANS 6.3-37)
C29-7 Comment noted. All comments will be considered during the decision-making process.

	Letter C30 SHARON L. GEHL 4301 Hermosa Way San Diego, CA 92103 (619) 299-9606 sigehl@cox.net August 1, 2016		
	Kurtis Steinert, Senior Environmental Planner Marlon Pangilinan, Senior Planner The City of San Diego 1010 Second Avenue, MS-413 San Diego, CA 92101		
	Re: Draft Program Environmental Impact Report (PEIR)—Uptown Community Plan Update (Project No. 380611 / SCH No. <i>Pending</i>) Comments on the Density Redistribution Alternative		
	Dear Mr. Steinert and Mr. Pangilinan:	C30-1	Comment noted.
C30-1	I have closely followed and participated in the process of Updating the Uptown Community Plan from the beginning seven years ago and provided recommendations.	C30-2	This comment does not raise an issue with regard to the adequacy of the PEIR. The City does not agree that the proposed
	Because I think neither the proposed Uptown Plan Update, nor the current Adopted Plan adequately meet the objectives of the General Plan and the City's Climate Action Plan, I support the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative described in the PEIR, with the addition of the increase of density along Park Boulevard between University Avenue and Washington Street and Normal Street from 73 du/ac to 109 and 145 du/ac.		Uptown CPU would reduce residential densities at village centers and along transit corridors. While the proposed Uptown CPU would reduce residential density in some areas, it would also increase density in others. Lower residential densities in some areas are required to ensure that the bulk and scale of
C30-2	l would like to comment further on the adequacy of the Uptown community Plan Update: <u>Uptown Community Plan, June 2016 Draft, Would Reduce Supply of New</u> <u>Housing</u>		development maintain the existing neighborhood character as well as public views of canyons and open space. These areas are also generally less served by transit and mixed uses. The proposed land uses locate the highest intensity uses along transit
	The stated intent of updating the 1988 Uptown Community Plan is to conform with the recently adopted General Plan, which calls for a "City of Villages" that encourages future growth in village centers that combine residential and commercial, encourage walking, biking, and the use of public transportation as a way to reduce greenhouse gas emissions, promote local businesses, and create a sense of community.		corridors where existing and future commercial, residential, and mixed-use development can support existing and planned transit investments in the community. Commercial and other employment-generating uses are also used strategically by the
	The Uptown Community Plan proposes instead to reduce residential densities around the village centers and along the main transit corridors as a way to protect the upper middle class		proposed Uptown CPU to encourage commercial uses along transit corridors. This transit-oriented development pattern is necessary to meet the goals of both the General Plan's City of
	1		Villages Strategy and the City's Climate Action Plan (CAP).
	RTC	-362	

- "Neighborhood Character" from new residents who would live in multifamily middle class housing in Uptown.
- C30-3 Lowering residential density is intended to <u>reduce</u> the growth in the housing supply, and as we know, housing is about supply and demand. There is demand for more middle class housing in Uptown and Mission Hills, more people do want to live here. The intent of downzoning the community is to keep new people from moving to the area by reducing the supply of new housing. The whole proposed Uptown Community Plan is based on discrimination against people who are a different social class or younger or have different aesthetic tastes.
- C30-4 The last time the city updated the Uptown Community Plan in 1988, they also reduced residential densities. The result since the 1990 census has been an average growth rate of only .4% for the community during a period when the City as a whole had a .9% growth rate. While the plan did succeed in shifting some new residential growth away from the canyons and to the commercial areas and transportation corridors, no one anticipated the sharp drop in the average number of residents per housing unit that reduced the growth rate as a whole. The current adopted Land Use densities have not been high enough to produce the infill development that the City of San Diego General Plan and Climate Action Plan call for. Reducing densities further as a way to prevent an increase in the housing supply, would likely result in near zero population growth for the community.
- C30-5 Reductions in residential densities in Uptown would be contrary to the City of San Diego's recently adopted General Plan goal of increasing infill urban development. The City is not proposing that we restrict the growth of the Medical complex or other job centers in Uptown, or that we try to discourage new commercial development; only that new employees and customers live outside of the area and drive in. This is not good city planning, it is contrary to the recommendations in the City's Climate Action Plan, it is not good for the environment, for property values, or for the City's tax income. Reducing the number of units that can be built on a property reduces the value of that property and city revenue from property tax. Reducing population growth also reduces local City sales tax income.
- C30-6 To fulfill the General Plan goal of encouraging infill development rather than sprawl, densities in the urban areas need to be increased enough to make infill development cheaper than sprawl.

2

- C30-3Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. The proposed Uptown CPU and associated discretionary actions are intended to provide guidance on orderly growth and redevelopment in accordance with smart growth principles. Through the placement of higher density residential development in areas in and around transit and commercial corridors, the proposed CPU would reinforce a mixed-use urban environment that supports transit and pedestrian activity and would allow for an increase in residential density over what currently exists. The proposed reduction in density in some areas would ensure that the neighborhood visual character is maintained, as detailed in response to comment C30-2. Lastly, the proposed Uptown CPU would designate land uses to accommodate growth, although additional housing units would not be built without demand.
- C30-4 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. The proposed Uptown CPU is intended to serve as a long-term plan for the physical development of the Uptown community and to manage and address future growth through build-out of the community.
- C30-5 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. Development in the Uptown CPU area will generally occur as infill projects, focusing on vacant or under-utilized parcels or previously utilized lots rather than on undeveloped land with high natural resource values. The proposed Uptown CPU would plan for growth within the community and would allow development of additional units beyond what currently exists. See response to comment C30-2 regarding the proposed Uptown CPU's consistency with the General Plan's City of Villages Strategy and the CAP. See also PEIR Section 6.1.3 and Section 6.5.3 for discussions on the proposed CPU's consistency with the General Plan and CAP.
- C30-6 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. See response to comment C30-2 regarding the distribution of densities in the proposed Uptown CPU.

ATTACHMENT 7

Adopted Con	mmunity Pla	n Statistics			C30-7	Comment noted. The comment does not raise an issu regard to the adequacy of the PIER.
Residential Population						
Uptown District						
Census	1990	2015 Pe	ercent Change per Year			
	35,167	38,868	.4%			
1	San	Diego	1			
Census	1990	2015 Pe	ercent Change per Year			
	1,110,549	1,368,061	,9%			
Residential I	Housing Unit	ts				
l'	2015 Upt	own Community P	lanning Area			
Total Population	Total Housing Units	; Percent Single Family	Percent Single Family Multiple	Percent Multiple Family		
38,868	23,197	20%	13%	67%		
ŕ.		2015 City of San D	iego	1		
Total Population	Total Housin Units	g Percent Single Family Detached	Percent Single Family Multiple	Percent Multiple Family		
1,368,061	526,762	39%	14%	45%		
Data from SANDAG online "Profile Warehouse", and SANDAG staff. Note that Uptown has added less than 100 housing units per year with the cur Community Plan Land Use densities. Another way to look at it is, we've averag per month for the last 25 years. That hasn't been enough to keep up with natu increases or normal demand for new housing, which is estimated to be an ave annually in highly desirable Uptown.				ear with the current low 1988 5, we've averaged only 8 unit 2p up with natural population 2d to be an average of 1 ½ %	1 5 6 7 7 7	
			з			

LETTER

RESPONSE

- We should allow enough housing to be built for the number of new babies, less the number of people who die each year; and for normal demand for housing from people who weren't born in Uptown, but want to live here.
- C30-8 Intentionally lowering densities and heights so that less than 8 units can be built each month in the future is just wrong. We should instead put the densities back to what they were before the 1988 Plan lowered them to keep new people out.

Housing Demand

- C30-9 Since the current Adopted Land Use regulations have resulted in little growth in the last 25 years, it can be assumed that lower allowed residential density would result in a further reduction in the growth of housing units.
- C30-10 There isn't a lack of demand for housing in Uptown, rather the rise in the cost of living shows that supply hasn't kept up with demand. City proposals to reduce residential densities further in the Plan Update would likely lower the future growth rate closer to zero. Uptown is a job center where we should allow market rate housing that can pencil out, where we should allow people to live close to jobs in the medical complex, and close enough to walk or bike to neighborhood businesses. Pushing housing out of the area would create more commuter traffic and be contrary to the City's Climate Action Plan which calls for more housing in "Transit Priority Areas" within one-half mile of major transit stops. Instead this Draft Plan proposes limiting the growth of housing within one-half mile of major transit stops in Uptown.

Some of the changes that we've seen in Uptown since the current community plan took effect in 1989 have made the community better. Projects like the Vibra Hospital and Village Hillcrest Shopping Complex on 5th Ave. have added jobs, stores, restaurants, and much needed public parking in the core of Hillcrest. We've also added housing near our village centers and public transit. There is no reason to stop making changes that have worked.

But because current city regulations for Uptown have resulted in little actual growth, proposals to reduce the amount of new housing and jobs near our village centers would likely lower the future annual growth rate closer to zero. We actually need to increase the amount of new housing near jobs, public transit, and stores; for the sake of the environment and to lower the cost of housing. More people living close enough to walk or bike to stores, restaurants and work, would mean less traffic and therefore less greenhouse gases.

C30-11 General Plan goals may say that we want infill development, but if Land Use regulations require that new development be built at existing densities, or in the case of Uptown, at lower densities yet, infill won't pencil out; and new development will continue to be in the suburbs.

4

- C30-8 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. See response to comment C30-2 regarding the distribution of densities in the proposed Uptown CPU.
- C30-9 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. The CPU and associated discretionary actions would designate planned land uses and zoning that would accommodate future development within the CPU area.
- C30-10 Though single- and multi-family residences would be reduced at build-out of the proposed CPU compared to build-out of the adopted Community Plan at year 2035, the proposed Uptown CPU is not proposed to reduce or inhibit population growth in the community. Compared to existing land uses, while build-out of the proposed Uptown CPU would result in a reduced number of single-family residences, it would increase the number of multifamily residences located in proximity to transit. Increased commercial uses along transit and pedestrian corridors would also result compared to both existing conditions and build-out of the adopted Community Plan. Development in the Uptown community would generally occur as infill projects, focusing on vacant or under-utilized parcels, along major transportation corridors.

As discussed in response to comment C30-2, the reduction in densities in some locations is not intended to reduce population growth. Rather, the residential density distribution included in the proposed Uptown CPU is consistent with the goals of the General Plan's City of Villages Strategy and the CAP by promoting a mix of uses and higher residential densities in close proximity to transit and pedestrian corridors.

C30-11 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER.

Mission Hills

- C30-12 One area where the 1988 Plan has apparently created a problem is in the residential areas on Reynard Way that would be reduced in density. While six projects were built along that street in the ten years before density was reduced in 1988, nothing has been built there in the 24 years since. The current allowed density level and building heights are too low to make it economically feasible to redevelop old apartment buildings along that street. Reducing the level further would simply compound the problem. The City actually needs to raise the allowed density and building heights to make it worthwhile to build the hillside retaining walls that are now required for any new construction on many Reynard Way lots. Failure to do so will result in these older buildings continuing to decline.
- C30-13 Density reductions in multi-family areas remove an incentive to redevelop older, unattractive apartment buildings. Reduction in residential density especially along Reynard (which is a public transit route) would diminish opportunities to revitalize this portion of Mission Hills.
- C30-14 Bus Route 83 runs right down Reynard Way to downtown in one direction and the Hospital complex in the other direction, with a number of stops along the street. Residents can get to work in 10 minutes no matter which direction they go. This is a perfect place to put new housing. The only problem with Route 83 is that it only runs once an hour. Over 10 years ago it ran every 20 minutes, but the frequency was cut because there weren't enough customers to warrant more frequent after service after densities after the 1988 plan reduced population growth in Mission Hills to essentially zero.
- C30-15 Another place where proposals to downzone properties will be particularly damaging is on the south side of University Ave. between Eagle and Ibis St. The properties on this heavily traveled street would be rezoned from the already too low "duplex", to "single family". University is used by commuters as a freeway bypass to avoid Hwy 8. This is not the place to require single family housing.

The "duplex" zoned housing around the higher density Village Center of Mission Hills should be maintained as a transition between the downtown area and the single family zoned areas further out.

5

Uptown

C30-16

The proposed down-zoning of Uptown is an effort to prevent normal population growth. It is contrary to good city planning principles, contrary to the City of San Diego's City of Villages Master Plan and planning policy, and contrary to state law. C30-12 Comment noted. The comment does not raise an issue with regard to the adequacy of the PIER. See response C30-12.

- C30-13 Multi-family residential densities along Reynard Way, a noted transit route, would not be reduced under the proposed Uptown CPU. As shown on Figure 3-4, Proposed Uptown Land Use South, of the PEIR, Reynard Way between Sutter Street and Curlew Street would be surrounded by Residential Medium High (30-44 dwelling units per acre [du/ac]) land uses. Figure 2-4, Land Uses under Adopted Community Plan Uptown, of the PEIR shows that the same area is designated as Medium High Density Residential (29-44 du/ac). Other land uses along Reynard Way in the proposed Uptown CPU include Neighborhood Commercial (which allows 0-44 du/ac), Residential Medium (16-29 du/ac), and some Residential Low (5-9 du/ac). These uses would not result in lower densities compared to build-out of the adopted Community Plan as shown on Figure 2-4.
- C30-14 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-15 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-16 See response to comment C30-2 and C30-9.

RESPONSE

- C30-17 The proposed down-zoning of Uptown will lower the property values for over 1,000 property owners and will therefore also lower the future property tax income for the City of San Diego. No planned growth means no developer impact fees to spend in Uptown and no need for the city to spend money on upgrading Uptown facilities. If the population is supposed to go somewhere else, then money for upgrading community facilities should go somewhere else too.
- C30-18 What is particularly inexplicable is the proposal to down-zone many streets and neighborhoods beyond the average existing densities, making the majority of buildings on many streets non-conforming.
- C30-19 No one is proposing that we limit of jobs or commercial businesses in Uptown, all the City Planning Department is proposing is that new employees and future customers not be allowed to live close by, that they have a to drive into the neighborhood, adding more traffic to our major streets. We actually need to add more housing close to jobs to lessen future traffic, especially housing close to the hospital complex, so that more people can walk to work.
- C30-20 Limiting the supply of housing close to public transit, jobs and stores in Uptown is not good for the environment or the poor. There will be more traffic, pollution, and parking problems if more people have to drive long distances to jobs in Uptown, rather than walk to work or a restaurant. Limiting housing supply will, over time, push people with less money and those on fixed incomes out of the community.

Mobility Element

- C30-21 The City's Climate Action Plan assumes that each Community will do its part to reduce greenhouse gas emissions, by increasing the amount of housing within one-half mile of public transit. Public transit needs customers, which means people need to live, work and/or shop near enough to transit stops to use it. Public transit also needs to be convenient enough to use, Increasing rather than decreasing the amount of housing and jobs within one-half mile of a major public transit stop, as the CAP proposes, would increase transit use. This is another reason that densities should be increased rather than decreased. Reducing densities will mean fewer customers to support public transportation.
- C30-22 If the new Community Plan intends future growth in population to be primarily outside of the Uptown district, emphasis should be placed on the mode of transportation that those communing in from outside of the district use the most, the automobile. The mobility element needs to look for ways to accommodate the planned increase in automobile traffic in Uptown. Sufficient public parking should be provided to meet increased future needs.

- C30-17 Comment noted. This comment does not suggest an inadequacy in the PEIR. See response to comment C30-10. Build-out of the proposed Uptown CPU would allow for an increase in Uptown's population and would not inhibit development in the community.
- C30-18 Comment noted. This comment does not suggest an inadequacy in the PEIR. See response to comment C30-2 regarding the pattern of residential densities proposed in the CPU.
- C30-19 Comment noted. This comment does not suggest an inadequacy in the PEIR. See response to comment C30-2 regarding the pattern of residential densities and other land uses proposed in the CPU.
- C30-20 Comment noted. See response to comment C30-2 regarding the pattern of residential densities and other land uses proposed in the CPU. This density distribution is intended to reduce traffic, pollution, and parking problems by encouraging transit and pedestrian travel through the location of higher densities and commercial uses along transit corridors.

The City General Plan states that the City of Villages emphasis on transit-oriented development, among other City of Villages and citywide strategies, is consistent with environmental justice goals. The proposed CPU is intended to implement the City of Villages strategy, and furthers the goals specified under Section I, Environmental Justice, of the General Plan's Land Use and Community Planning Element by providing a combination of land uses that improve mobility, emphasize the existing diversity of the community, and support future growth and prosperity in the plan area. In addition, Policy LU-2.3 of the proposed Uptown CPU addresses the development of adequate housing for those with special needs such as those with low income.

- C30-21 See response to comment C30-2 regarding the pattern of residential densities and other land uses proposed in the CPU as a method to implement the General Plan's City of Villages Strategy and the CAP. In addition, the Mobility Element of the proposed Uptown CPU includes numerous policies (Policies MO-1.1 through MO-3.12) aimed at strengthening the Uptown community's pedestrian, bicycle, and transit facilities and service.
- C30-22 Policies MO-4.1 through MO-6.4 of the proposed CPU's Mobility Element address the need for a safe and efficient street and freeway system within the Uptown community and between neighboring communities. Additionally, Policies MO-7.1 through MO-7.21 address parking management strategies specific to the community's needs.

LETTER

RESPONSE

- C30-23 Sustainable Design Strategy
 The Sustainable Design Strategy should recommend that older energy inefficient housing near
 "Village Centers" and transportation corridors be replaced with energy efficient multifamily
 housing. Just as older cars that get low mileage are bad for the environment, older buildings
 with little or no insulation and single pane windows are bad for the environment. Just as new
 cars are designed to use less gasoline, newer buildings are designed to use energy and produce
 less greenhouse gas.

 C30-24 The City's *Climate Action Plan* and the United States Environmental Protection Agency (EPA).
- C30-24 The City's Climate Action Plan and the United States Environmental Protection Agency (EPA) Smart Growth recommendations, such as those on Location Efficiency and Housing Type, should be used to reduce the impact of development on Global Climate Change by planning for energy efficient multifamily housing near public transit. The EPA states that, "How and where we construct our communities has an enormous effect on our energy consumption. Buildings and transportation together account for about 70 percent of energy use in the U.S. and are responsible for 62% of U.S. Greenhouse gas emissions." http://www.epa.gov/dced/location_efficiency_BTU.htm

Building Heights

- C30-25 The City of San Diego should not reduce building heights below what they were before the Interim Height ordinances. Reducing heights is another way of reducing densities and encouraging sprawl. Density is better for the environment. Uptown already has tall buildings, more of the same would fit in esthetically with what already exists. Because Uptown is on a high mesa, the views can be delightful. It makes sense to build something tall here. Taller buildings, like the ones in downtown, can be more affordable per unit.
- C30-26 The Vibra Hospital and Village Hillcrest Shopping Complex on 5th Ave., is a good example of why we need flexibility in building heights. The Vibra Hospital is tall like the Mercy Hospital across the street, while the stores on 5th are at a lower scale appropriate to 5th Ave., and under it all is public parking that benefits the whole area. The complex also includes a public open space, restaurants, a movie theater, and recreation, (i.e. a gym). That's a lot of infrastructure added to the community.
- C30-27 Without the tall hospital, we would have had fewer jobs and customers in Uptown, and the public parking garage would not have penciled out. We need flexibility in building heights in Uptown to make everything work for the benefit of the community.

- C30-23 Comment noted. The proposed Uptown CPU contains several policies in its Conservation Element that address sustainable development in the community. However, the policies do not address existing buildings with regarding to energy efficiency as the CPU would guide future development and redevelopment rather than existing structures. The City's CAP does include measures to address energy efficiency of existing homes.
- C30-24 Comment noted.

- C30-25 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-26 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-27 Comment noted. This comment does not suggest an inadequacy in the PEIR.

LETTER

Public Facilities, Services & Safety Element

- C30-28 The City has said that future population growth can be shifted from Uptown to Downtown and Mission Valley where there are, presumably, sufficient public facilities like parks and libraries. When we consider public facilities, we should consider all facilities that serve the public.
- C30-29 There are a large number of underutilized public facilities in Uptown and Mission Hills. Our local businesses, restaurants, religious institutions, and bus system don't have enough customers; they could use more. If we do need to upgrade or repair any infrastructure, it's cheaper to repair and expand what we have than to build everything brand new in a suburb 50 miles away.
- C30-30 Saying that we can't allow housing to grow because of a "lack of infrastructure", is just an excuse to prevent growth. There may have been a problem with facilities in the 1980s, but the City has been steadily replacing water and sewer lines in Uptown for years.
- C30-31 All new projects pay DIF fees toward infrastructure and facilities improvements. Most of the money for the new Lewis Street pocket park in Mission Hills came from the One Mission project (Paseo de Mission Hills). If we say that new projects and population should go somewhere else, than DIE money and new parks should go somewhere else too.

Potential Historic Districts

- C30-32 The City of San Diego is proposing putting 26 Potential Historic Districts (PHDs) in the Community Plans for Uptown, North Park, and Golden Hill. These would not be actual legal historic districts, only "potential" districts. The City would then apply supplemental regulations to them that would create de facto historic districts; without first doing the research necessary to establish whether an area actually meets city standards for a historic district, or allowing, property owners to vote on whether or not they want to be in a historic district and subject to the additional regulations.
- C30-33 The City historic staff says that if they don't apply restrictions now, before they do the necessary studies, some property owners might chose to make changes to their property that could preclude them from being a "contributing property" to a possible future historic district. The result could be that some of the PHDs might no longer have the minimum 50% contributing properties needed for a future historic district.
- C30-34 The reality is, of the 26 PHDs encompassing over 3,300 properties; only 2, with about 62 properties combined, are close to tipping from plus 50% to negative 50%. Those two PHDs have about 62 properties total and are not in Uptown. Why should the owners of over 3,300 properties be encumbered because the City historic staff is worried that they might not be able to someday get control of 62 properties in North Park?

C30-28 Comment noted. This comment does not suggest an inadequacy in the PEIR.

- C30-29 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-30 Comment noted. This comment does not suggest an inadequacy in the PEIR. As future development is proposed within the Uptown CPU area, individual projects would be subject to applicable Development Impact Fees for public facilities financing in accordance with Municipal Code Section 142.0640. In addition, the proposed Uptown CPU Public Facilities, Services, and Safety Element includes goals to provide and maintain infrastructure and public services for future growth without diminishing services to existing development. Specific policies regarding public facilities financing include public facilities and services prioritization as well as fire-rescue, police, wastewater, storm water infrastructure, waste management and recycling, libraries, schools, public utilities, and healthcare services and facilities, all included within the proposed Uptown CPU.
- C30-31 Comment noted. See response to comment 30-30.
- C30-32 Comment noted. The Historical Resources Survey identified 19 new potential historic districts and the community identified four potential historic districts that were determined to meet the National Register standards for determining district boundaries and that appear to meet at least one of the City's local designation criteria for historical sites. Refer to Appendix G-2, Historical Resources Survey Report, for the results of the research conducted on these districts. The amended Historical Resources Regulations are intended to provide supplemental development regulations for the potential historic districts until they are

C30-32 (d	cont.) formally evaluated and designated. These supplemental development regulations would address how and where modifications can be made on residential properties identified as potentially contributing to specified Potential Historic Districts. Without these amended regulations, future development has the potential to substantially degrade or destroy resources potentially contributing to a potential historic district, which would result in
	significant and irreversible impacts. Should a potential historic district or potentially contributing property be evaluated and found ineligible, the protections would not apply.
C30-33	Comment noted. This comment does not suggest an inadequacy in the PEIR.
C30-34	Comment noted. This comment does not suggest an inadequacy in the PEIR. See response to comment C30-32.
c	130-33 130-34

	LETTER	RESPONSE			
C30-35	The City does not need to put PHDs in the community plan and apply restrictions to over 3,300 properties; they have time to wait and do things the normal way. But <u>if</u> meanwhile, a property owner did make changes that a preservationist would not approve of, that means that they are voting to not be in a future historic district. They should have the freedom to make that choice.	C30-35	Comment noted. This comment does not suggest an inadequacy in the PEIR.		
C30-36	There would be <u>no financial benefit</u> to being in a PHD. Because they would not be actual historic districts, properties would not qualify for Mills Act tax subsidies to compensate for the additional restrictions; owners couldn't even put up a plaque. But because of the supplemental PHD regulations, PHDs would still have all of the financial <u>disadvantages</u> of being in a real historic district.	C30-36	Comment noted. This comment does not suggest an inadequacy in the PEIR.		
C30-37	The proposed restrictions on properties in the PHDs would likely lower their value. Because there would be less owners could do to upgrade them to keep them functional, fewer people would want to buy them. Owners would be required to tell potential buyers that the property was in a PHD and subject to supplemental regulations.	C30-37	Comment noted. See response to comment C30-32.		
C30-38	The regulations would also add to the cost of getting a building permit. City staff would decide, based on an unofficial preliminary survey, whether or not owners would be required to give them an historic report on their property that would cost thousands of dollars. Staff would then use that report to decide if the regulations that they wrote would apply to a property and how they would apply. If an owner did not agree with staff's decision, they would have additional costs to hire experts and lawyers and appeal what the city staff decided.	C30-38	Comment noted. See response to comment C30-32.		
C30-39	Experts have estimated that it could take over 20 years before necessary historic studies could be done on all of the 26 PHDs and it could be determined whether or not the areas would actually qualify as historic districts. Only then would property owners be allowed to vote on whether they actually want their property to be in a historic district and subject to restrictions.	C30-39	Comment noted. See response to comment C30-32.		
C30-40	Many people like to live in old houses that have been upgraded with modern amenities to meet modern needs, but few people want to live in museums. That's why the <u>majority</u> of property owners in Uptown, North Park, and Golden Hills who have done major remodels or built new houses over the years, did not choose to do them in a way that would be allowed by the PHD regulations. This is precisely why the City historic staff wants to put PHDs in the Community Plans and apply regulations without allowing owners to vote. They know that the majority of owners would not do things their way if they had the freedom to choose.	C30-40	Comment noted. See response to comment C30-32.		
C30-41	Most people are against forcing historic designation on others against their will. The City should be against it too. They should not put PHDs in the Community Plans and apply supplemental regulations to them that would create de facto historic districts. The financial damage they would do to the owners of thousands of properties far outweigh any possible reason to put them in the Plans.	C30-41	Comment noted. See response to comment C30-32.		
	9				

C30-42	The PHDs would also mean lowered property values, less tax revenue for the City, less economic activity, les new construction, and more money spent on Mills Act tax subsidies which lose money for the city, despite what some people might wish.	C30-42	Comment noted. See response to comment C30-32.
	Potential Historic District Statistics		
C30-43	There are 26 total Potential Districts in the Uptown, Greater North Park, and Greater Golden Hill Community Plans, with approximately 3319 total structures, of which a total of 1807 are potentially contributing, and a total of 870 are "unknowns" that haven't been studied yet. If the 2677 potential contributing properties plus "unknowns" all got Mills Act contracts, at a typical \$10,000 yearly subsidy, it would cost taxpayers throughout the state \$26,770,000 a year.	C30-43	Comment noted. See response to comment C30-32. This comment does not suggest an inadequacy in the PEIR.
C30-44	The cost to the City of doing the historic studies on the Potential Historic Districts will be substantial. Since the largest proposed district has 458 properties, while the smallest has 11, we can assume that the cost will vary. City staff estimates have varied from \$80,000 to \$170,000 each. If it costs \$80,000 to do the historic studies of each potential district, times 26 potential districts, the cost would be \$2,080,000. If it costs \$170,000 to do the historic studies of each potential district, times 26 potential districts, the cost would be \$4,420,000.	C30-44	Comment noted. This comment does not suggest an inadequacy in the PEIR.
C30-45	If the 3319 properties in the Potential Historic Districts are worth a minimum of \$500.000 each, the value of the properties negatively affected by these regulations would be a minimum of \$1,659,500,000.	C30-45	Comment noted. This comment does not suggest an inadequacy in the PEIR.
	Economics of Historic Preservation		
C30-46	While everyone likes the idea of "historic preservation"; the problem is most people don't like it enough to spend money on it, which is why historic preservation is rarely sustainable financially. The overwhelming majority of historic museums throughout the United States don't have enough customers to pay the bills. That's why they are always asking for donations, for	C30-46	Comment noted. This comment does not suggest an inadequacy in the PEIR.
	volunteers, and for taxpayer money. The likelihood of losing money is also why it is difficult to impossible to give away an old building that needs to be moved to another site, like the Henry B. Jones Residence located on the grounds of Scripps Mercy Hospital. The land is what has value, not the old building.		
C30-47	volunteers, and for taxpayer money. The likelihood of losing money is also why it is difficult to impossible to give away an old building that needs to be moved to another site, like the Henry B. Jones Residence located on the grounds of Scripps Mercy Hospital. The land is what has value, not the old building. When the City forces historic designation on a building it lowers the value of the land, because it limits what can be done with the property. Few owners would voluntarily put historic designation on their property in San Diego if they weren't bribed with a large Mills Act tax subsidy. Only 5 to 6 properties came before the Historic Review Board annually before Mills Act subsidies became a popular option. The City, County, and State of California all lose money on these subsidies, and on Historic Districts.	C30-47	Comment noted. This comment does not suggest an inadequacy in the PEIR.
C30-47	volunteers, and for taxpayer money. The likelihood of losing money is also why it is difficult to impossible to give away an old building that needs to be moved to another site, like the Henry B. Jones Residence located on the grounds of Scripps Mercy Hospital. The land is what has value, not the old building. When the City forces historic designation on a building it lowers the value of the land, because it limits what can be done with the property. Few owners would voluntarily put historic designation on their property in San Diego if they weren't bribed with a large Mills Act tax subsidy. Only 5 to 6 properties came before the Historic Review Board annually before Mills Act subsidies became a popular option. The City, County, and State of California all lose money on these subsidies, and on Historic Districts.	C30-47	Comment noted. This comment does not suggest an inadequacy in the PEIR.

RESPONSE

- C30-48 Second point; The City does not get any value in exchange for these large tax subsidies for high income housing. The average person walking or driving through Mission Hills or any other part of Uptown can't tell which houses have historic designation or are part of a Historic District unless they see a plaque. It all looks the same. The Tax subsidies do not make any difference. Most people with million dollar houses take care of their properties; they don't need tax subsidies for maintenance. It is questionable whether anyone who can afford to own a house should be subsidized by the City, when the City can't afford to give subsidies to all of the low income residents who need help.
- C30-49 Third point; historic preservationists and property owners have different goals. Preservationists want to turn buildings into museums that reflect a "historically significant period", or person, which they can use to illustrate a story. (Note it is not necessary that the story be accurate, only that someone will pay to hear it; hence the prevalence of Ghost Tours.)
- C30-50 While preservationist want drafty single pane windows left in place as part of a museum display, homeowners want homes that are comfortable and functional for their families. They might want to replace drafty windows with energy effect double pane windows, add a dishwasher, a second bathroom, a third bedroom, or a bigger closet. Changing and adapting buildings to meet new needs is what people have done throughout history. While pretending that a building is "historic" and has never changed, (hide that dishwasher), may be fashionable among preservationists; a building that changes over time to meet new needs is actually more accurate historically.
- C30-51 Fourth Point; old buildings are bad for the environment, because they need more energy to heat and cool than would an equivalent sized new house that meets modern Title 24 energy requirements. The environmental review should estimate the environmental damage caused by the high percentage of energy inefficient old structures in the Plan Update communities and question all proposals to discourage or prevent property owners from replacing or remodeling them.
- C30-52 Last point; there would be no need to have historic districts if owners actually wanted to live in museum houses. The City is proposing to put "Potential Historic Districts" in the Community Plan precisely because they know that the majority of owners don't want to live in historic districts.
- C30-53 When property owners extensively remodel, rebuild, or replace old buildings with something more energy efficient, their properties increase in value, they raise the value of nearby houses, and pay more in property taxes. Preventing owners from rebuilding or replacing energy inefficient blighted houses is bad for the environment, bad for neighborhood character, bad for the local economy, bad for workers, bad for property values, and bad for the City's tax base.

11

- C30-48 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-49 Comment noted. This comment does not suggest an inadequacy in the PEIR.
- C30-50 Comment noted. This comment does not suggest an inadequacy in the PEIR. Note that window replacements within the original openings of historic buildings are exempt from requiring a building permit. In addition, development that does not comply with the supplemental development regulations would be possible with issuance of a Neighborhood Development Permit with deviation findings and mitigation.
- C30-51 A historic building designation would not necessarily prevent a property owner from making the energy efficiency retrofits that owners of non-eligible older buildings are able to make. As noted in response to comment C30-50, window replacements within the original openings of historic buildings do not require a building permit, and are therefore not subject to the new supplemental development regulations. In addition, development that deviates from the Historical Resources Regulations would not be prohibited, but would require a Neighborhood Development Permit, which would include reasonably feasible measures to protect and preserve the integrity of the potential historic district.
- C30-52 The City must adhere to the federal, state, and local laws and regulations pertaining to the protection historic resources described to Section 5.7, Historical Resources, of the PEIR. In addition, Section 5024.1(g) of the Public Resources Code states that even a resource that is not listed in, or determined eligible for listing in, the California Register, not included in a local register, or not deemed significant in a historical resource survey may nonetheless be historically significant for purposes of CEQA. The

C30-52 (cont.)
City is proposing to implement supplemental development regulations for potential historic districts as part of the CPU in order to protect these potential historic resources.
C30-53 Comment noted. See response C30-51.

C30-54	The City's <i>Climate Action Plan</i> stresses the need to update buildings to be more energy efficient. Historic preservation is contrary to that, therefore Potential Historic Districts should not be included in this Community Plan nor in any other Community Plan.	C30-54	Comment noted. See response C30-51.
	Conclusion		
C30-55	The City of San Diego should not lower existing residential densities to keep new people from moving into Uptown. It would be discrimination based on social class and/or age. The City of San Diego should not support discrimination.	C30-55	Comment noted. See response to comment C30-20.
C30-56	Many areas in Uptown should actually have their allowed residential densities increased near community centers and public transit lines to meet present and future community needs. Urban areas like Uptown are where we should allow market rate housing that can pencil out, where we should allow people to live close to jobs in the Medical Complex, and close enough to walk or blke to neighborhood businesses. We do not need more commuter traffic.	C30-56	Comment noted. See response to comment C30-2.
C30-57	Instead of trying to keep new people from moving to Uptown, the City should allow Uptown to share responsibility to provide housing that accommodates normal growth and shifts in population with other communities.	C30-57	Comment noted. See responses to comments C30-9 and C30-10.
C30-58	Using City laws and regulations to keep new people from moving into the neighborhood by preventing the supply of housing from keeping up with the demand for housing may be good for some people's definition of upper middle class "neighborhood character", but it is bad for the environment, bad for traffic, and just plain wrong. We need to allow middle class people who work and shop in Uptown to live here too. Telling them to drive in from somewhere else just makes traffic worse.	C30-58	Comment noted. See responses to comments C30-9 and C30-10.
C30-59	As we learned in Economics class, the cost of housing, like any other commodity, is a matter of supply and demand. When cities pass laws to prevent the construction of more middle class multifamily housing, it not only pushes young families out of the area, it makes the cost of renting or buying housing more expensive everyone in San Diego County. That means more of our money goes to housing costs and less is left for the rest of our life.	C30-59	Comment noted. See responses to comments C30-2, C30-9, and C30-10.
C30-60	We can have both nice neighborhoods and new neighbors. There is plenty of room for new apartments and condos in San Diego within walking distance of public transit, jobs, and local restaurants; if only we allow it.	C30-60	Comment noted.
	Questions on PEIR:		
C30-61	Can the City of San Diego legally "Preserve the neighborhood character" of Uptown by preventing the construction of market rate middle class housing?	C30-61	While the City aims to preserve the neighborhood character of Uptown through the proposed CPU, it would not prevent construction of market rate, middle-class housing. The proposed Uptown CPU includes policies that address both the needs of neighborhood character and housing demand.

	Letter C31	C31-1	Comment noted. Refer to the following responses addressing the suitability of the suggested mitigation.
	Comments on Mobility element of Uptown Community Plan Update	C31-2	This comment suggests how Robinson Avenue could be converted to a one-way eastbound street to mitigate traffic impacts. However, the
	Streets		suggestion is not accompanied by any specific studies or justification as to how it would address an impact. Any recommended
C31-1	The obvious and most effective mitigation is not included in the current draft: make University and Robinson a one-way pair through the core of Hillerest. Here's how it might work:		operational improvement projects would require analysis at a project level. A transportation technical study would need to be
C31-2	Convert Robinson Avenue to one-way eastbound traffic from Front Street to 5th Avenue.		conducted for review and approval by Development Service Department and Transportation and Storm Water Department.
	 Reverse the stop signs at Front and Robinson so that traffic proceeding south(cast) on Front will have right-of-way when turning left onto Robinson. 		
	 Redirect the #10 bus eastbound to Albatross-University-Front-Robinson. 	C31-3	This comment suggests how University Avenue could be
	 Redirect the #11 bus eastbound to turn right at Robinson instead of University. 		converted to a one-way westbound street to mitigate traff
	 Remove the bus stop at the northeast corner of 1ⁿ & Robinson and replace with one on Robinson just east of 1st Avenue. 		impacts. However, the suggestion is not accompanied by any specific studies or justification as to how it would address an
	 Create a bus stop at 4th and Robinson. 		impact. Any recommended operational improvement projects would require analysis at a project level. A transportation technical study would need to be conducted for review and approval by Development Services Department and Transportation and Storm Water Department.
	 Route the #10 and #11 buses to turn left at 5th Avenue, stop at the current stop for the #1 and #120 buses at 5th and University, and turn right on University and continue with their current routes. 		
C31-3	Convert University Avenue to one-way westbound service between 5 th Avenue and Front Street.		
	 Remove the eastbound bus stops at 3rd and University and 5th and University. 		
	 Westbound #10 and #11 bus service will continue on current routes and stops. 	C31-4	As proposed, the Uptown CPU is not recommending the
	 Recovered street real estate can be made available for bicycle and or parking enhancements as needed. 		implementation of a continuous left turn lane along First Avenue between Laurel Street and Washington Street because it would
	By diverting some of University Avenue traffic to Robinson, you accomplish the following:		increase pedestrian crossing distance and impact sidewalks which would conflict with the proposed Uptown CPU pedestrian oriented policies that support a pedestrian scale environment and enhanced pedestrian amenities. Additionally, it would conflict with Mobility Element Policy MO-7.13 that supports on-street parkin, on all streets to support adjacent uses and enhance pedestrian safety and activity. Additionally, the proposed bicycle networ recommends a Class III bike route along the entire length of Firs Avenue within the community planning area. A Class III bike route is located on shared roadways that accommodate vehicles and bicycles in the same travel lane.
	 One-way streets provide real estate to make left and right turns free of conflicting oncoming traffic. 		
	 Pedestrian safety enhanced by knowing which way to look for oncoming cars. 		
	 Enhanced commercial value of properties fronting on Robinson. 		
C31-4	Reconsider mitigation on 1 st Avenue between Washington and Laurel.		
	The street is not wide enough for a continuous two-way left turn lane. Creating three narrow lanes will make it impossible for cars to pass bicycles without lane violation and maintain the 3 foot separation now mandated.		

C31-5 Parking

Eliminate the time limit on parking meters. Require drivers to check in and check out when they arrive and leave. Instead of making overtime an infraction, charge users for the difference between check-in and check-out time. In order to promote turnover, increase the rate after an initial period so that a person who parks at a meter all day will pay more than s/he would at the nearest commercial off-street facility.

- This will eliminate the unnecessary shaming that parking violations cause, and if a luncheon or a medical appointment runs a bit overtime, it will not invoke draconian penalties.
- This will also reduce the need for parking enforcement personnel.
- It offers a potentially more fair treatment of disabled and non-disabled use of metered spaces. Disabled and non-disabled vehicles could pay the same rate for time up to the base period and either be allowed to stay for the day or be charged the base rate for overtime, whichever is considered the more equitable.
- C31-5 Comment noted. This comment does not identify an inadequacy in the PEIR; therefore, a detailed response is not required. All comments will be considered during the decision-making process.

Г

	Letter C32		
	August 8, 2016		
	Kurtis Steinert Senior Environmental Planner City of San Diego Planning Department 1010 Second Avenue, MS 413 San Diego, CA 92101		
	Subject: Comments of Draft Program Environmental Impact Report for PROJECT NAME: Uptown Community Plan Update PROJECT NO. 21002568 / SCH No. Pending		
	Dear City Staff and Decision Makers:	C32-1	Comment noted. The City appreciates your participation in the
C32-1	We have reviewed the Draft Program Environmental Impact Report for the Uptown Community Plan Update (UCPU) dated June 10, 2016 ("PEIR") and submit the following comments:		public review process.
	COMMENTS ON EXECUTIVE SUMMARY		
C32-2	Section S.3: Areas of Controversy	C32-2	Comment noted. General clarification regarding comments
	The PEIR states that there are no clear-cut areas of controversy. This is false. The following land use recommendations for University Heights were approved by Uptown Planners on June 7, 2016:	CJ2-2	received has been added to Section S.3, Areas of Controversy the Final PEIR. Regarding land use recommendation #2,
	 <u>Change</u> the zoning for the San Diego Unified School District site and the multi-family residential areas between Park, Maryland, Adams and Tyler from RM-2-5 to RM-1-3 to reflect current zoning (MCCPD-MR1500) and to provide for a height limit of 30 feet that is compatible with the pattern of the existing neighborhoods. 		related zoning has been changed to CC-3-4 which allows the same density and height as the CN-1-3 zone, but will allow community serving uses.
	 <u>Change</u> the zoning for the small parcel located at the northwest corner of Park Blvd, and EI Cajon Blvd. from CC-3-5 to CN 1-3 that is compatible with the pattern of the existing neighborhoods. 		
C32-3	In a letter to Marlon Pangilinan dated December 1, 2015, UHCA states:	C32-3	The City does not agree that the proposed zoning would result
	"A small parcel of land along Park Blvd adjoining this larger piece of land is currently used as office space and designated "Office/Residential" allowing 4-5 du/ac. The new plan proposes 0-74 du/ac for that parcel. A vast majority of our residents want to keep the current designation for the school district property and amend the smaller property to a 0- 29 du/ac designation,"		incompatible development. As discussed in Section 6.2.3 of the PEIR under Issue 2, proposed Uptown CPU Policies specifically address development height, massing, and transitions that would guide future development to be compatible with the existing
	This record of opposition to key aspects of the Uptown Community Plan Update (UCPU) should have been noted in Section S.3.		surrounding development. Refer to the proposed Uptown CPU policies UD-4.70 through UD-4.92. See also response to comment C32-2 above that addresses the referenced "smaller property."

COMMENTS ON CHAPTER 3: PROJECT DESCRIPTION

3.4.2 Land Development Code Amendments

 Section 3.4.3.2 b. RM Zones states that "The purpose of the RM zones is to provide for multiple dwelling unit development at varying densities and provide for residential development that is compatible with the pattern of existing neighborhoods." Yet, the conversion of the current Mid-City PDO MR-1500 zone to the new Citywide RM-2-5 zone, which applies to several neighborhoods in UH, allows up to 33% more in height and 145% in Floor Area Ratio, which is not compatible with the pattern of these existing neighborhoods.

COMMENTS ON CHAPTER 6: ENVIRONMENTAL ANALYSIS

Section 6.1: Land Use

- C32-4 Section 6.1.3 Impact Analysis does not provide a "plan-to-ground" analysis as stated on page 1-5 of the PEIR. The goals and recommendations of the UCPU are concluded on page 6.1-5 of the PEIR to be "consistent with development design guidelines, other mobility and eivic guidelines, and programs in accordance with the general goals stated in the General Plan" without any supporting documentation or analysis.
- C32-5 Consistency with the General Plan policies in Section 6.1.3.a City of San Diego General Plan should include detailed analysis of issues related to General Plan policies listed in Section 5.1.1.1 on page 5-4, including that planned density of residential land uses is not completely within appropriate locations. As examples, an analysis will reveal the following:
 - The UCPU is not consistent with the third listed goal for "Community plans that maintain
 or increase planned density of residential land uses in appropriate locations" because
 stakeholders have informed the City during the planning process that the density
 proposed for the residential areas between Park, Maryland, Adams and Tyler is not
 appropriate, nor is the density proposed for the small parcel located at the northwest
 comer of Park Blvd, and El Cajon Blvd.
 - The UCPU is also not consistent with Project Objective 4 on page 3-3, which states "Preserve the neighborhood character and design relationships between neighborhoods within each community through the development of transitions and design policies." The RM-2-5 zoned proposed for specific areas carved out between Park, Maryland, Adams and Tyler allow for a height up to 40 feet. However, the proposed zones for the commercial corridor along Park and the east-west corridors along Madison, Meade and Monroe only allow up to 30 feet in height.

2

- The PEIR does present a plan-to-ground analysis. The comment C32-4 suggests that the PEIR makes a conclusion about consistency with the General Plan without any supporting documentation or analysis. However, following Table 6.1-2, there are several pages of discussion of each element of the proposed Uptown CPU with discussion regarding how the relevant element and proposed policies are consistent with the General Plan. Regarding a plan-toground analysis, the threshold for this issue area requires an evaluation of consistency with the applicable plans in order to identify if any indirect or secondary environmental impacts could occur, but the baseline remains existing conditions. Since the proposed Uptown CPU was shown to be consistent with the General Plan, no physical environmental impacts would result related to inconsistency with the General Plan and no indirect or secondary environmental impacts would occur.
- C32-5 The proposed Uptown CPU is consistent with the General Plans' goal to maintain or increase planned density of residential land uses in appropriate locations. The CPU furthers this goal by proposing changes in densities that are intended to promote a multi-modal transportation network and meet the needs of the community. The City does not agree that the Uptown CPU is not consistent with the project objectives. The proposed Uptown CPU is also consistent with the CPU objectives to preserve neighborhood character as discussed in response to comment C32-6.

C32-6 Section 6.2: Visual Effects and Neighborhood Character

Section 6.2.3 Impact Analysis, Issue 2 Neighborhood Character, lacks a meaningful analysis of the impacts of the Citywide rezoning on existing neighborhood character. These aspects of the UCPU will indeed result in a "substantial alteration (e.g. bulk, scale, materials or style)" to the existing character of University Heights, and the resulting significant impacts must be disclosed to decision makers and stakeholders. As examples, an analysis will reveal the following:

- On page 6.2-4 Neighborhood Character, the PEIR states that "The proposed Urban Design Element of the proposed Uptown CPU would also provide policies addressing commercial and mixed-use development, and residential in-fill development. Implementation of these policies would provide specific policy support to ensure that the bulk and scale of development is not out of character with the existing environment. However, structures built to the allowable height of 40 feet and a Floor Area Ratio of 1.35 in an area of historic-age single-family homes and two-story apartment complexes would by definition result in a substantial change in bulk and scale to the existing character. This is a significant impact no amount of setbacks or design features can avoid.
- On page 6.2-7, the PEIR offers as rationale for non-significance "since the CPU areas are already urbanized and include existing development of the type that would be further developed under the CPUs. Future development in accordance with the CPU areas is likely to take place on infill sites in previously developed locations. Each proposed CPU (Golden Hill, North Park, and Uptown) contains policies to ensure that any new development is consistent with the existing character and protects public views. The proposed policies address consistency in setbacks, height and bulk, landscaping, design, historic character, and natural features such as canyons and hillsides."

While it is true that much of the UCPU area is already developed, the residential buildings typically do not exceed two stories in height and many are of historie age, so they reflect consistent low-rise scale, are made of historic materials such as stuceo and irreplaceable types of wood, and are built in Craftsman or Revival styles. New development on the very few vacant sites available (what it is assumed is meant by "infil") would likely not be consistent with the historic materials, style, low-rise scale and detail of neighboring structures. This is particularly true in the residential areas of University Heights that are proposed for conversion from the MR-1500 zone to the new RM-2-5 zone, which allows another 33% in height and 145% in Floor Area Ratio. Redevelopment on occupied sites would create similar incompatibilities with the added impact of demolishing existing structures. This is a significant impact.

C32-7 On page 6.2-8, Section 6.2.4.2 Neighborhood Character, states that "The proposed Uptown CPU includes policies that would encourage residential and mixed-use development that would be consistent with the existing neighborhood character, and impacts would be less than significant. No mitigation would be required." Again, the proposed RM-2-5 zone proposed for several neighborhoods in University Heights would increase the height by 33% and the Floor Area Ratio by 145%. These are significant impacts.

3

This comment suggests that the analysis of impacts to C32-6 Neighborhood Character is inadequate. However, the City does not agree. As discussed in Issue 2 of Section 6.2.3, Impact Analysis, all future development projects would be implemented in accordance with the City's General Plan, Land Development Code, and Zoning that applies restrictions on development to ensure it is consistent with surrounding character (e.g. Floor Area Ratios and setbacks). Additionally, all discretionary development would be subject to review against proposed CPU policies. The Urban Design Element of the proposed Uptown CPU includes policies that would direct future development in a manner than ensures the physical attributes of the community will be retained and enhanced, both relative to public spaces and streetscape and private development. More intense development would be subject to setbacks and upper-story stepbacks to address massing and compatibility where adjacent to lower height buildings. Based on these considerations and the numerous other design policies included in the proposed Uptown CPU, impacts related to substantial alterations to the existing character of the area would be less than significant. However, in response to public comments received, additional discussion was added under Issue 2 in Section 6.2.3 to further characterize the anticipated land use changes.

C32-7 See response to comment C32-6.

Resources Guidelines, through the subsequent project review

C32-8	Section 6.3: Transportation and Circulation	C32-8	Maryland Street south of Meade Avenue and Lincoln Avenue east
	 Table 6.3-8, Buildont Summary of Roadway Segment Analysis, fails to analyze how buildout will affect Maryland south of Meade and Lincoln cast of Maryland. Maryland Street is used frequently as a pass-through for cars from outside the neighborhood and should have been included in the analyses. 		of Maryland Street between Maryland Street and Washington Street are not classified roadways. The Community Plan Circulation Element identifies only roadways that are classified as
C32-9 C32-10	Section 6.7: Historical Resources		collector or higher. Classified streets (collector or higher) serve the needs of the entire community; whereas, local streets (non- classified) serve the needs of the residents along those streets. However, Lincoln Avenue between Washington Street and Park Boulevard is a classified roadway and was analyzed and included in the study area.
	 Section 6.7.4 Impact Analysis of the PEIR lacks quantification and detail that should be provided under a plan-to-ground assessment. Examples of deficiencies that should be corrected in order for the PEIR to provide a good faith effort at full disclosure required by CEQA Guidelines Section 15151 are discussed below. The focus of the historic resources impacts analysis is on designated buildings and districts and on properties deemed eligible for individual listing. However, on page 6.7-7, the PEIR states, "The City of San Diego's CEQA Significance Determination Thresholds define a significant historic resource as one which qualifies for the California Register of Historical Resources or is listed in a local historic register or deemed significant in a historical resource survey, as provided under Section 5024.1(g) of the Public Resources Code; though even a resource that is not listed in, or determined eligible for listing in, the California Register, not included in a local register, or not deemed significant in a historical resource survey may nonetheless be historically significant for purposes of CEQA [emphasis added]. The significance determination threshold for impacts related to historical resources is stated on page 6.7-7 as, "An alteration, including the adverse physical or aesthetic effects and/or the destruction of a historic age structures potentially affected by policies, zoning, and programs proposed by the UCPU. However, the impact analysis in Section 6.7.4 Issue 1 Historic Structures, Objects, or Sites is cursory and policy-focused, and does not quantify potential impacts from proposed zoning on community resources of historic age. Examples of specific impacts that should be analyzed and disclosed are: 		
		C32-9	This comment makes reference to the fact that a resource not listed or determined eligible for listing may be historically significant for purposes of CEQA. The PEIR analysis is not deficient and provides an analysis appropriate for a program EIR. As future developments proceed within the Uptown CPU area, specific projects would be subject to review under the Historical Resources Regulations. Additionally, any project that requires
			referenced in this comment (Section 5024.1(g) of the Public Resources Code).
	 The San Diego Normal School/San Diego City Schools Education Complex was identified during public outreach as a potential historic district but, as such, does not qualify for the interim historic protection measures proposed by the City. The underlying zoning proposed by the City for this site is RM-2-5, which would allow up to 29 du/ac and 40 feet in height. The PEIR should analyze the impact that full build-out encouraged by this zoning would have on the multiple unique historic resources on this site. 	C32-10	The level of evaluation of historic resources in the PEIR is adequate for a program EIR. The PEIR considers the effect of implementing the proposed Uptown CPU and associated discretionary actions would have on historic resources at the plan level not the project level As future development is proposed
	 Similarly, there are a half dozen historic apartment houses in the 4200 block of Campus Avenue that were built anticipation of the Panama-California Exposition in 1915. The proposed zoning for this block is RM-3-7, which allows up to 44 du/ac and 40 feet in height. The PEIR should analyze the impact that full build-out encouraged by this zoning would have on multiple unique historic resources in this block. 		within the Uptown CPU area, all development projects with the potential to affect historical resources—such as designated historical resources; potentially historical buildings, districts landscapes, objects, and structures; important archaeologica sites; tribal cultural resources, and traditional cultura
	4		properties—would be subject to site-specific review in accordance with the City's Historical Resources Regulations and Historical

process.

 The Swedenborgian Church, built in 1927 by architect Louis Gill, is located at 4144 Campus Avenue and is also zoned as RM-3-7. The PEIR should analyze the impact that full build-out encouraged by this zoning would have on this unique historic resource.

Sincerely,

Kristin Harme

Kristin Harms, University Heights

William Ellig, University Heights

Distribution:

Councilmember Todd Gloria University Heights Community Association University Heights Development Chris Ward Marlon Pangilinan, City of San Diego C32-10 (cont.)

Regarding the San Diego Normal School/San Diego City Schools Education Complex, the proposed Uptown CPU recognizes the historic potential of this site in land use policy LU-2.17 that states, "Consider the reuse of the San Diego Unified School District Education Center at Park Boulevard and Normal Street which could include medium-high residential development, the potential for mixed-use development, public space, and/or the rehabilitation and reuse of buildings such as the Teachers Training Annex." As stated above, any future development would be subject to evaluation for impacts to historical resources under the existing regulatory framework. Additionally, as indicated in the proposed Uptown CPU, the structure at 4345 Campus Avenue is currently included on the National Register of Historic Places and would be protected through this designation. All future development on existing designated historic resources and potential historic resources would be protected through the existing historical resources regulations. In addition, mitigation measures HIST-6.7-1 and HIST 6.7-2 provide a framework that would be required of all development projects with the potential to impact significant historical resources. Therefore, though specific historic resources have the potential to be impacted by future development, the City's development regulations and policies of the proposed Uptown CPU would minimize adverse impacts. A significant and unavoidable impact to historical resources was disclosed in the PEIR because at the program-level of review, the degree of future impacts and applicability, feasibility, and success of future mitigation cannot be adequately known for each specific future project.

5



Vanessa A. Herbert 3042 State Street San Diego, CA 92103 (619) 249-4348

I am in favor of the City beautifying this empty lot, however the projected cost of \$970,000 and planned concept for the property is ridiculous. The lot has been an evesore for the 19 years I have been a resident. The property is located on a very busy corner at the entrance to the underpass of the I-5 North. A children's play and picnic area would be a complete waste of taxpayers' dollars. This lot is not a desirable place to picnic or let children play. A better plan would be for the City to sell or lease the property to build or furnishing a park in a more desirable location.

C33-7 India Street Traffic Concerns and Strong Opposition to Proposed Plan to Remove Parking and/or Sidewalks from India Street

> For many years, the traffic has been too dense, travels too fast and is a safety concern for the residents of Middletown as they enter or exit off of India Street by car, walk across India Street to the Pedestrian bridge over I-5 or walk along the adjoining sidewalk on India Street. At various times different plans by the City of San Diego ("City") and other local agencies (e.g. San Diego Airport Authority and et al) show an elimination of parking along India Street and or a widening of India Street that would remove adjoining sidewalks and add an extra travel lane. Any adoptions of these plans will negatively impact the environment and quality of life for residents and businesses along India Street and Middletown. Accordingly, I would request the City of San Diego Issue a written statement and incorporate it by reference into the draft Uptown Community Plan Project No. 380611/2100258 which states there are neither current nor will there be future plans for the removal of parking along India Street and/or sidewalks from India Street from West Palm Street to Washington Street.

The traffic on India Street between West Olive and Sassafras Streets as well as Sassafras Street between Pacific Highway and India Streets has increased significantly with the opening of the Rental Car Center (RCC). Since the RCC opened, traffic data indicates an increase of approximately 3,000 vehicles daily.

Specific Notations and Comments regarding the Uptown Community Plan Update / Project No. 380611/2100258 / Draft PEIR, including all appendices related to India Street between West Redwood and Sassafras Streets that we object to and believe will be detrimental to the residents of Middletown

Table 5-1 Summary of Significant Environmental Impacts notes: India Street: Sassafras Street to Redwood Street (Impact 6.3-19)

Mitigation: TRANS 6.3-19: India Street (Impact 6.3-19), b. Sassafras Street to Redwood Street: Widen the roadway to a 3-lane one-way collector.

Comment: I strongly object to any widening of India Street and to the inference that the two-way collector status between Sassafras and Redwood will be eliminated. The two-way collector is essential to residents of Spruce (and Redwood). If eliminated, Spruce residents will be required to travel to Laurel on Kettner and return up India for access to their homes. This will unnecessarily further increase traffic on Kettner, Laurel and India Streets and add additional fuel use and emissions using the longer route.

Table 6.3-7 Build-out Summary of Intersection Analysis - Uptown #20 & 21.

Comment: If this analysis was done before the San Diego International Airport Rental Car Center (SDIA-RCC) was opened and would be inaccurate. The SDIA RCC has added a substantial number of cars – dailyto the intersections of Sassafras Street at Kettner as well as onto India Street from the I-5 North exiting onto India and the SDIA RCC traffic exiting the SDIA-RCC using Sassafras to travel to the I-5 North on ramp. C33-7 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

RESPONSE

2

Vanessa A. Herbert 3042 State Street San Diego, CA 92103 (619) 249-4348

The current traffic pattern has caused a significant increase traffic delays and regularly blocked intersections.

The Middletown Advisory Group has recommended and other Community organizations agree the SDIA-RCC traffic exiting the SDIA-RCC be directed to use the northbound Pacific Highway to Washington Street to access the I-5 north. Given the size and length of Pacific Highway to Washington Street, this would reduce congestion and traffic delays. Additionally, traffic exiting the SDIA-RCC headed south could either continue to be directed onto Sassafras to Kettner to access the I-5 south on-ramp or directed south on Pacific Highway to access the downtown area of San Diego.

No improvements were made to India. Sassafras or Kettner Streets when the SDIA-RCC was open and its traffic directed onto these streets. Pacific Highway (north and south bound) and Washington Streets are much better equipped to handle the additional traffic from the RCC as opposed to India Street.

Table 6.3-8 Buildout Summary of Roadway Segment Analysis: Uptown - India Street:

Comment: Same as above (6.3-7) but the India Street Buildout CPU figures demonstrate that adding an additional traffic lane will not solve the LOS problem and only make life much worse for lacal residents. Rerouting traffic from the Airport and RCC to Pacific Highway and Washington Street to access the I-5 north is the only viable, effective, and inexpensive short term solution

Impact 6.3-19, page 6.3-38: "The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of India Street from Glenwood Drive to Redwood Street."

Comment: I agree that the changes will have a traffic impact that is detrimental for the residents living off India from Glenwood Drive to Redwood Street and oppose any changes widening India Street. Rerouting traffic from the Airport and RCC to Pacific Highway to Washington is the only short-term solution as noted above.

Appendix C – Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions).

Table 13 Summary of Improvement Evaluation Uptown: This references India Street between Redwood and Sassafras Streets as "one way Collector" and removal of parking and buildings.

Comment: I strongly object to any widening of India Street especially with such extreme damage to the neighborhood businesses, and to the inference that the two-way collector status between Sassafras and Redwood Streets will be eliminated. The two-way collector is essential to residents of Spruce and Redwood Streets. If eliminated, Spruce residents will be required to travel to Laurel Street on south Kettner Street and return north up India Street for access to their homes. This will unnecessarily further increase traffic on India Street. I recognize that it says the Consultant does not recommend this change but that is not binding on the City and the road changes are of considerable concern. I request that this be removed from consideration.

Figure 64 of Appendix C: Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions).

3

Vanessa A. Herbert 3042 State Street San Diego, CA 92103 (619) 249-4348

Comment: I strongly object to any widening of India Street. While recognizing that the Consultant says "These changes in Geometry are not recommended" and that recommendation is not binding on the City, the road changes are of considerable concern. I request the City to delete or issue a written statement, which will be incorporated by reference to the Uptown Plan stating that there are no current or future plans to remove parking and or sidewalks from India Street from Palm Street to Washington Street.

C33-8 Error Regarding Spruce Street, et al throughout Uptown Plan

The maps used throughout the Uptown Plan and appendices incorrectly depict Spruce Street at India Street and connecting streets off Spruce to show several alternative routes available to the 200+ residents living on Spruce and the adjoining streets (lbis, Horton, Upas, Jackdaw) for ingress and egress. While it is unclear what effects on planning may have flowed from planners and consultants mistakenly believing that residents on Spruce et al, have alternative routes, it needs to correctly depict that these residents have no other ingress or egress but from India Street. Please correct this map error and do a proper reanalysis of the impact to the residents who rely solely on Spruce Street for access to their homes. The error and correction is attached to show the seriousness of the map error.

Respectfully,

Vanessa Herbert

C33-8 Comment noted. The proposed Uptown CPU maps appear to be correct. West Spruce Street does connect to India Street. Access to all the other streets mentioned in the comment letter is solely from West Spruce/India Street. Because of the map's scale, the gap between West Spruce Avenue and Horton Avenue is difficult to see, but there is a gap in the figure and it was considered in the traffic analysis. However, the map in the Mobility Study does incorrectly show that a connection between West Spruce Avenue and Horton Avenue that will be corrected. While there was an error in the Mobility Study map, the model used in preparation of the analysis of potential impacts of the proposed Uptown CPU for traffic circulation did not include any connections with West Spruce Avenue that would provide additional ingress/egress to West Spruce Avenue other than India Street. Furthermore, the referenced Mobility Study Improvements (U17A and U17B) would be inconsistent with the proposed Uptown CPU polices and thus, would not be implemented due to infeasibility.

		Letter C34		
	From: Sent: To: Cc: Subject:	Derie Hilsen <dhilsen333@gmail.com> Sunday, August 07, 2016 9:39 AM PLN_PlanningCEQA middletown92103@gmail.com Project Name: Uptown Community Plan Update, Project No, 21002568/SCH No. Pending</dhilsen333@gmail.com>		
	Project Name: Uptown Co	ommunity Plan Update		
	Project No. 21002568/SCI	H No. Pending		
	Community Area: Uptown	n		
	Council District 3			
	Dear Mr. Steiner,			
C34-1	We are writing to you regs bound lane that goes from homes. If this section is re	arding some proposed changes to India Street. We are very much opposed to the removal of the south- Sassafins to Redwood. There are hundreds of residents that use this as the only ingress and egress to their emoved our options are limited to:	C34-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to
	 Exiting the freeway a Washington and put more of the street only 1 car at a neighborhood, (via courtes) 	It Washington St. and then taking a 2-mile convoluted route to our area. This would increase traffic on traffic on residential streets. Many of the streets in this area are 2-way and if cars are parked on both sides i time is able to proceed. With limited traffic the residents in this area can maneuver around our sy and yielding to each other) but not if these streets become more congested.		these response to comments (pages RTC-4 through RTC-6).
	 Taking the Airport ex neighborhood. Traffic at I designed for a volume of t 	at and proceeding onto Kettner, passing Sassafras and then going to Laurel and backtracking through the Laurel and Kettner is already a nightmare and the streets in the neighborhood, as stated before are not brough traffic.		
	We are also concerned wit parking already. Removin begun The dangers of thi accidents as residents after Middletown residents after	In the proposals to remove parking and sidewalks along sections of India Street. There is very limited g the parking and adding another lane would only increase the volume of speeding traffic that has already s (especially now that more traffic is being routed to India Street for the new Rental Car facility) are; more mpt to turn onto India Street from Palm, Redwood and other streets as well as (pedestrian safety) as mpt to corros India at Palm to get to the trelley this is extremely hazardous and of paramount concern!		
	Concerns regarding these been offered. These prope auto/pedestrian conflicts, r	issues have been expressed at the Uptown Community Meetingsreasonable and low-cost alternatives have sails are not in Reeping with the stated goals in the Uptown Community Plan Update of reducing maintaining the character of the neighborhood and improving access to mass transit:		
	We urge you to consider th	he impact these changes would have for the residents and business owners in our neighborhood		
	Sincerely,			
	Charles DeBerry and Den	e Hilsen		
	Property Owners at 2920 I	Union St.		
	Middletown, 92103			

		Letter C35		
	From: Sent: To: Subject:	Karen Huntington skhunbington eng@gmail.com> Thursday, August 04, 2016 1:47 PM PLN_PlanningCEQA Uptown Community Plan Update		
	To: Kurtis Steiner, Senior Environmental Plat RE: Uptown Communit	nner, City of San Diego Planning Department y Plan Update		
	PROJECT No. 2100256	87 SCH No. Pending		
	COMMUNITY AREA:	Uptown		
	COUNCIL DISTRICT:	3"		
C35-1	Please do not turn India Spruce and Redwood. T Midtown area and it is h of or neighborhood and	Street into a 3-lane highway and get rid of the single southbound lane feeding onto here is already such a huge increase in traffic due to the Rental Car Center in our arder and harder to get in and out of our neighborhood. Traffic should be re-routed out onto Pacific Highway.	C35-1	Comment noted. The comment does not suggest an inadequacy in the analysis of the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through
	Thank you,			RTC-6), which includes a discussion on the Rental Car Center.
	Karen Huntington			
	4-year resident of Midto	wn		
	1030 W. Redwood Stree	et		
	San Diego, CA 92103			
		ĩ		
		Letter C36		
-------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------
	From: Sent: To: Cc: Subject:	Roland johnson ≼rjcsandiego@sbcglobal net> Saturday, July 30, 2016 5,01 PM PLN. PlanningCEQA Middletown92103@gmail.com Uptown Community Plan Update Project Number 21002568		
C36-1 C36-2 C36-3	As a resident of Middletown I a vetted by the community. The community Parking is already traffic up and clown Sassafiras S non-standard gradient (which I Street has 3 traffic lights, a rail light. Rental car traffic does no Streets which are already comm	an vehemently against the proposed modifications to India Street. This plan has never been properly e elimination of parking and sidewalks and the change of traffic flow will adversely affect the entire a tremendous problem, and it is to be eliminated on India Street?? The current routing of rental car itreet has caused major traffic problems already. Sassafnas Street is probably in excess of 12% grade, a believe would not be allowed to be built in this manner today.) In the space of three blocks, Sassafnas road/trolley crossing and an excessive grade going Northbound to India Street ending in a traffic ot belong on Sassafras and India Streets. It belongs on Kettner, Pacific Coast Highway and Washington neretal, not residential streets. You are going to get some people killed!	C36-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	Roland Johnson 3018 Columbia St San Diego, Ca 92103		C36-2	Comment noted. Please refer to the Staff Report for a discussion of the extensive public outreach that has been done regarding the proposed Uptown CPU. Also refer to Section 4.2, Community Outreach and Plan Development, of the PEIR. Note that the Draft PEIR is distributed for review to the public for the purpose of providing comments "on the sufficiency of the document in identifying and analyzing the possible impacts on the environment and ways in which the significant effects of the project might be avoided or mitigated" (Section 15204, CEQA Guidelines).
			C36-3	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

Г

RESPONSE

	From: Sent: To: Subject:	Athena killian <athenakillian@gmail.com> Monday, August 08, 2016 3:01 PM PLN. PlanningCEOA Fwd: Project Name_Uptown Community Pan Update Project Number_21002568/SCH No Pending</athenakillian@gmail.com>		
	Kurtis Steiner			
	Senior Environmenta	al Planner, City of San Diego Planning Department		
	1010 Second Ave. N	/IS 413		
	San Diego, CA 9210	93		
	Project Name: Uptov	wn Community Pan Update		
	Project Number: 210	002568/SCH No. Pending		
	Dear Mr. Steiner,			
C37-1	I have owned a hom for the last 12 years, a longtime dream of your plans to, aid eg and removing the So personally impact my place. Your plans to would be more sever	te just off of the India Street pedestrian bridge on West Palm Street in Middletown lived in the Hillcrest/Mission Hills neighborhood for almost 20. This location was mine to be able to afford and own property. I am now extremely concerned that ress to and from the new Rental Car Center, by way of removing street parking buthbound India Street lane from Sassafras to Redwood will negatively and y life and property value of a home I have worked so hard to obtain in the first inject more and more traffic into an area that is already heavily congested, but rely impacted than it already is makes no sense.	C37-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	I can appreciate the but the decisions that by them. India Streel parking and, with the congested and unsa sign on India at Wes pedestrian bridge du	city trying to make the most of improvements that will collect tourism revenue, at have been made have been made without consulting the people most affected t and the streets that run perpendicular to it in Middletown are already short on a implementation of the RCC, our neighborhood has become even more fe just trying to get in and out of our homes. As it stands, we already need a stop of Palm as it is nearly impossible to cross the street at West Palm to access the re to traffic going as fast as they do on the 5.		
	Since the opening of	f the Rental Car Center, I have already seen a significant impact to my commute		

to and from work. Headed west on Sassafras, trying to make a right on Pacific Highway, has become difficult because the westbound lane is not wide enough for traffic going both straight and right. The cars going straight (into the RCC) are not aware, because they are tourists, that there are cars that need to make a right on Pacific Highway because there has been no signage implemented to indicate such. Going East on Sassafras is far worse, with people not knowing which lanes to use to get to 5 North, often using the wrong one, and worse, backing up the traffic through the Sassafras/Kettner intersection... sometimes through the railroad tracks. In addition, I use the southbound section of India Street multiple times a day to get to my home and without that access to Redwood St. the egress to my whole neighborhood is extremely limited.

I propose using the Hawthorne Street exit from 5 North to get to the RCC as it is already used for all Northbound airport traffic. And on exit of the RCC, the use of Pacific Highway North to get to 5 North makes far more sense than injecting all of this new, erratic traffic into a "residential" zone on India Street, while proposing to close the only parking portion of India Street available to residents in an area that is already short on parking. Pacific Highway is a much more equipped to handle the incurred traffic from the RCC and that option would be far more considerate of the residents here who are already impacted and would be forced into a decreased quality of life, vs catering to the tourists, who are here for a brief amount of time. What makes more sense?

It's hard to understand why this was not given more thought prior to the construction of the Rental Car Center. But... to not give ANY consideration to the residents here in South Mission Hills, who have either already paid a high price to purchase their properties, or pay high rents to live here, is extremely unjust. I can only hope that you will take the residents of this "residential area" into consideration before you make a decision that will negatively impact them forever:

÷.

Thank you for your time. Muma Killing

Letter C38 From: Brian Laprath <blaprath@gmail.com> Sent: Monday, August 08, 2016 9.55 PM PLN_PlanningCEQA To: PROJECT NAME: Uptown Community Plan Update, PROJECT No. 21002568 / Project No. Subject: 38061 To Whom it May Concern, I have been a resident in Middletown for seven years. I am writing to petition your organization not to move C38-1 Comment noted. Please refer to the Master Response Regarding C38-1 forward with the overhaul of India St to include the access road running from south from Sassafras St to India Street Mitigation Measures included in the introduction to Redwood St. these response to comments (pages RTC-4 through RTC-6). In the last few years, Columbia St has experienced an increase in curb-parked vehicles. The corner of Columbia C38-2 and Palm in particular has less curb space than it needs to accommodate residents' vehicles. Displacing parked cars from India St will turn an increasingly inconvenient situation into a an unmanageable one. This also has the Comment noted. Please refer to the Master Response Regarding C38-2 potential to significantly lower home values in the area. India Street Mitigation Measures included in the introduction to In my experience living here, traffic on Laurel St is far less efficient than traffic on India northbound. Simply C38-3 put, overhauling India St will yield insignificant gains to India, and make an already bad situation worse on these response to comments (pages RTC-4 through RTC-6). Laurel -- especially since the additional stop lights were installed in front of the train tracks. Laurel is a major thoroughfare for the airport and Balboa Park. I assume adding more to an existing bottleneck is not in the best interests of the neighborhood, nor the airport. Comment noted. Please refer to the Master Response Regarding C38-3 I know I speak for the neighborhood when I say this is a very important issue. Thank you very much for your C38-4 India Street Mitigation Measures included in the introduction to time and consideration with our thoughts this project! these response to comments (pages RTC-4 through RTC-6). Regards. This is a closing comment. The City appreciates your participation C38-4 -Brian Laprath in the public review process. c: 707.685.7987 . .

			Letter C39		
	From: C Sent: V To: F Subject: L	Saelle tait <gaelletait@yahoo.com> Nednesday, July 27, 2016 11:44 AM 2N. PlanningOEQA Jptown Community Plan Update</gaelletait@yahoo.com>			
	Draft Program Environ	mental Impact Report.			
	Wednesday, July 27				
	PROJECT : #21002568 UPTOWN Community a To: Kurtis Steinert (Ser	8 / SCH (pending#) area_District council #3 nior Environmental Planner)			
C39-1	My name is Gaelle Le Ti I write to you today beca the 'Draft Program Env If you look at the Appen sidewalks on India is ins sidewalk will also force e Actually like to walk even and feel quite secure to of Midletown we need to kettner st). The fact of lo residential won't be able	anter. I leave and own a condominium at 3033 India s isomental Impact Report". dix C:Mobility Study . Figure 63 and 64 . The remova ane! It is already difficult to find a space for parking at r yever one to walk up hill and down hill to get to Little Ita rywhere . India street is the only flat street to get from I walk at anytime of the day. Not only the fact that to get walk on India and trying to get across the freeway 5 (c ising the parking to add an additional line for the traffic	treet. Inking and sidewalk in al of parking and hight but losing the ly or Downtown. I ittle Italy to Oldtown t to the trolley station rossing India st. and will mean that all the	C39-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C39-2	The San Diego City Pla loma and even further to to push the people to ge environmental and greer Sincerely , Best regards Gaelle Le Tanter.	nners should be more concern about extend the trolley La Jolla to avoid the increase of traffic . They should ut t more in public transportation and being seen as a firs n.	to the airport, Point come with better ideas t city in California to be	C39-2	This comment expresses concern with public transit options, but does not raise an issue with the adequacy of the PEIR. The Mobility Element of the proposed Uptown CPU includes numerous policies aimed at improving transit service within the Uptown community.
		1			

	Letter C40		
	From: Deirdre Lee [mailto:deirdresjungle@cox.net] Sent: Sunday, July 10, 2016 12:10 PM To: Mayor Kevin Faulconer <kevinfaulconer@sandiego.gov>; Councilmember Todd Gloria <toddgloria@sandiego.gov>; Pangilinan, Marlon <mpangilinan@sandiego.gov>; Galloway, Tait <tgalloway@sandiego.gov>; Murphy, Jeff <murphyj@sandiego.gov> Subject: Fwd: Uptown Community Plan Hijacked</murphyj@sandiego.gov></tgalloway@sandiego.gov></mpangilinan@sandiego.gov></toddgloria@sandiego.gov></kevinfaulconer@sandiego.gov>		
C40-1	Dear Mayor, Council Member, and Planners, I am resending this letter that I sent in February. I did not get a response from any one of you. The community is still trying to get back to the plan that they worked on. We now have the additional concern that we now have an employee of the developers' lobbyist sitting on the plan review committee. This is a clear conflict of interest.	C40-1	Thank you for your comment. The City appreciates your participation in the public review process. Please see responses to comment letter C41 for detailed responses to your letter dated February 15, 2016.

 C41-1 After six years of hard work by informed community members, multiple meetings and workshops, input from every conceivable stake holder, the draft plan has been revamped at the last minute to satisfy the demands of outsider developers and greedy property owners. Money is the pure objective here, not the betterment of the community. C41-2 To claim that this last minute doubling of height limits and densities in Hillcrest is necessary to satisfy the CAP, and that nothing has been built in the years, is absurd and laughable. Obviously these people have not been living here, and the health of the planet is not their priority. Our streets are gridlocked now and what they are building is far from affordable. I have been commuting by bike for 37 years and it has become more and more difficult, even with improved bike lanes, because of traffic. Thousands of people are not giving up their cars. C41-3 Hillcrest is and always has been dense in population. The draft plan before it was hijacked already allowed for considerable increased density and heights in a formula that would allow growth but protect the neighborhood from becoming University City. This is what our hard working planners and community members developed through this lengthy process. Why did we work this hard if it was to be thrown out to profit hugh development companies? And this money does not stay in the community. C41-4 No one is trying to protect strip malls, just the sunlight on our streets, the diversity in the character of the neighborhood, the human scale. The pro-super-density people say look at North Park, but North Park has no 100°-200° buildings. They say Hillcrest should be the mext Little Italy I recall that Little haly roled back their density/height allowances, as their streets became totally shadowed etc. (And why would we want to be Little Italy? There are many services available in our vibrant Hillcrest beyond eating and drinking.) C41-6 Einally, I am really disturbed		February 15, 2016 To: Mayor Kevin Falconer, Councilman Todd Goria, Planner Marion Pangilinan, Tate Galloway, Jeff Murphy
 C41-2 To claim that this last minute doubling of height limits and densities in Hillcrest is necessary to satisfy the CAP, and that nothing has been built in ten years, is absurd and laughable. Obviously these people have not been living here, and the health of the planet is not their priority. Our streets are gridlocked now and what they are building is far from affordable. I have been commuting by bike for 37 years and it has become more and more difficult, even with improved blike lanes, because of traffic. Thousands of people are not giving up their cars. C41-3 Hiltcrest is and always has been dense in population. The draft plan before it was hijacked already allowed for considerable increased density and heights in a formula that would allow growth but protect the neighborhood from becoming University City. This is what our hard working planners and community members developed through this lengthy process. Why did we work this thard if it was to be thrown out to profit hugh development companies? And this money does not stay in the community. C41-4 No one is trying to protect strip malls, just the sunlight on our streets, the diversity in the character of the neighborhood, the human scale. The pro-super-density people say look at North Park, but North Park has no 100°-200° buildings. They say Hillcrest should be the next Little Italy 1 recall that Little Italy rolled back their densityheight allowances, as their streets became totally shadowed etc. (And why would we want to be Little Italy? There are many services available in our vibrant Hillcrest beyond eating and drinking.) C41-5 Uptown and Hillcrest are deficient in park space and have no recreation center or aquatics facility. Stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure. C41-6 Finally, 1 am really disturbed by and disappointed in this sneaky last minute revamp by our City Planners. We deserve better than this. Deirdr	C41-1	After six years of hard work by informed community members, multiple meetings and workshops, input from every conceivable stake holder, the draft plan has been revamped at the last minute to satisfy the demands of outsider developers and greedy property owners. Money is the pure objective here, not the betterment of the community.
 C41-3 Hillcrest is and always has been dense in population. The draft plan before it was hijacked already allowed for considerable increased density and heights in a formula that would allow growth but protect the neighborhood from becoming University City. This is what our hard working planners and community members developed through this lengthy process. Why did we work this hard if it was to be thrown out to profit hugh development companies? And this money does not stay in the community. C41-4 No one is trying to protect strip malls, just the sunlight on our streets, the diversity in the character of the neighborhood, the human scale. The pro-super-density people say look at North Park, but North Park has no 100' -200' buildings. They say Hillcrest should be the next Little Italy. I recall that Little Italy rolled back their density/height allowances, as their streets became totally shadowed etc. (And why would we want to be Little Italy? There are many services available in our vibrant Hillcrest beyond eating and drinking.) C41-5 Uptown and Hillcrest are deficient in park space and have no recreation center or aquatics facility. Stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure. C41-6 Finally, I am really disturbed by and disappointed in this sneaky last minute revamp by our City Planners. We deserve better than this. Deirdre Lee 244 W. Brookes Avenue San Diego, CA 92103 	C41-2	To claim that this last minute doubling of height limits and densities in Hillcrest is necessary to satisfy the CAP, and that nothing has been built in ten years, is absurd and laughable. Obviously these people have not been living here, and the health of the planet is not their priority. Our streets are gridlocked now and what they are building is far from affordable. I have been commuting by bike for 37 years and it has become more and more difficult, even with improved bike lanes, because of traffic. Thousands of people are not giving up their cars.
 C41-4 No one is trying to protect strip malls, just the sunlight on our streets, the diversity in the character of the neighborhood, the human scale. The pro-super-density people say look at North Park, but North Park has no 100'-200' buildings. They say Hillcrest should be the next Little Italy. I recall that Little Italy rolled back their density/height allowances, as their streets became totally shadowed etc. (And why would we want to be Little Italy? There are many services available in our vibrant Hillcrest beyond eating and drinking.) C41-5 Uptown and Hillcrest are deficient in park space and have no recreation center or aquatics facility. Stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure. C41-6 Finally, I am really disturbed by and disappointed in this sneaky last minute revamp by our City Planners. We deserve better than this. Deirdre Lee 244 W. Brookes Avenue San Diego, CA 92103 	C41-3	Hillcrest is and always has been dense in population. The draft plan before it was hijacked already allowed for considerable increased density and heights in a formula that would allow growth but protect the neighborhood from becoming University City. This is what our hard working planners and community members developed through this lengthy process. Why did we work this hard if it was to be thrown out to profit hugh development companies? And this money does not stay in the community.
C41-5 Uptown and Hillcrest are deficient in park space and have no recreation center or aquatics facility. Stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure. C41-6 Finally, I am really disturbed by and disappointed in this sneaky last minute revamp by our City Planners. We deserve better than this. Deirdre Lee 244 W. Brookes Avenue San Diego, CA 92103	C41-4	No one is trying to protect strip malls, just the sunlight on our streets, the diversity in the character of the neighborhood, the human scale. The pro-super-density people say look at North Park, but North Park has no 100' -200' buildings. They say Hillcrest should be the next Little Italy. I recall that Little Italy rolled back their density/height allowances, as their streets became totally shadowed etc. (And why would we want to be Little Italy? There are many services available in our vibrant Hillcrest beyond eating and drinking.)
C41-6 Finally, I am really disturbed by and disappointed in this sneaky last minute revamp by our City Planners. We deserve better than this. Deirdre Lee 244 W. Brookes Avenue San Diego, CA 92103	C41-5	Uptown and Hillcrest are deficient in park space and have no recreation center or aquatics facility. Stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure.
	C41-6	Finally, I am really disturbed by and disappointed in this sneaky last minute revamp by our City Planners. We deserve better than this. Deirdre Lee 244 W. Brookes Avenue San Diego, CA 92103

- C41-1 Comment noted. The comment does not identify an inadequacy in the PEIR.
- C41-2 Comment noted. The comment does not identify an inadequacy in the PEIR.
- C41-3 Comment noted. The comment does not identify an inadequacy in the PEIR.
- Comment noted. The comment does not identify an inadequacy in the PEIR. The proposed Uptown CPU Urban Design Element provides design guidelines by building types to control massing and ensure compatible transitions. Building setbacks and upperstory stepbacks are recommended to address massing and compatibility where more intense development is located adjacent to lower height buildings (refer to Urban Design Element policies related to development transitions). These policies and guidelines would ensure taller buildings would not adversely impact surrounding lower intensity properties through neighborhood incompatibility or through creation of excessive shade or shadows.
- C41-5 Comment noted. The comment does not identify an inadequacy in the PEIR. Future community parks under the Uptown CPU are discussed in Section 6.12, Public Services and Facilities. Though there would be a deficiency in park and park equivalences at buildout of the proposed Uptown CPU, the existing conditions include a deficit in parks and park equivalencies. In addition, while the proposed Uptown CPU does not propose any individual project, 35.31 acres of proposed new population-based park land and park equivalency sites have been identified through the proposed Uptown CPU effort. The policy framework provided by the proposed Uptown CPU supports acquisition and development of new public parks and park equivalencies, and encourages new private development to include recreational facilities. At this program-level

C41-5 (cont.)

analysis, it is appropriate to assume that policy support would increase the acreage of population-based parks in the CPU area at build-out. Lastly, there is a less-than-significant impact associated with the construction of new facilities in order to maintain performance objectives for parks because the project does not include construction of new recreational facilities.

This comment states that "stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure" but does not identify what infrastructure needs would not be met beyond park facilities, which are addressed in the paragraph above. Impacts related to Public Services and Facilities is discussed in Section 6.12 and impacts related to Public Utilities are discussed in Section 6.13. Both sections determined that impacts would be less than significant and no mitigation would be required.

C41-6 Comment noted.

	Letter C42 To: Mayor Kevin Falconer, Councilman Todd Gloria, Planner Marlon Pangilinan, Tate Galloway, Jeff Murphy and all involved with reviewing the draft UCP Subject: Untown Community Plan Update	C42-1	Thank you for your comment. The City appreciates your participation in the public review process and acknowledges your support of the Density Redistribution Alternative and Lower- Density Alternative. Please see responses below to your specific comments on the proposed Uptown CPU.
C42-1	I am strongly apposed to the June 2016 Uptown Community Plan draft update as revised by the Planning Department at the behest of a coalition of developers. The update should return to the June 2015 draft. In the draft EIR, the *environmentally superior" Density Redistribution Alternative or the Lower-Density Alternative generally support the community's lune 2015.	C42-2	Comment noted. The comment does not identify an inadequacy in the PEIR.
C42-2	plan. GREED?	C42-3	Comment noted. The comment does not identify an inadequacy in the PEIR.
C42-2	workshops, input from every stake holder, the draft plan was revamped at the last minute to satisfy the demands of outsider developers and greedy property owners. (The irony—property owners allowing their vacant parcels to deteriorate for decades, wanting an up- zone so they can make more money.) Money is the pure objective here, not the betterment of the community. The jobs are temporary and most of that money leaves the community.	C42-4	Comment noted. The comment does not identify an inadequacy in the PEIR. Implementation of the CAP is not based solely on an increase in residential densities; the CAP aims to reduce GHG
C42-3	WHERE WENT HILLCREST?? The developers show pretty visuals that look like somewhere else entirely and they have made it clear that they are not interested in parks or history or even adequate parking. They want no encumbrances on their freedom to build whatever they want.		appropriate land use patterns. Assigning higher-density residential uses along main transit corridors and near mixed-use commercial and employment areas and placing lower-density single-family
C42-4	CAP?? To claim that this last minute doubling of height limits and densities in Hillcrest is necessary to satisfy the CAP, make any profit, and that nothing has been built in ten years due to 1HO, is untrue. Look around. Are those non-profits building the five story projects in Bankers Hill? Hardly, Obviously the developers and lobbyists have not been living here, and the health of the planet is not their priority. Our streets are gridlocked now and they they are not promising affordable housing.		residential land uses within existing lower-density, single-family neighborhoods would further the goals of the City of Villages Strategy and the CAP. See also the PEIR, Section 6.1.3 and Section 6.5.3 for discussions on the proposed Uptown CPU's consistency with the General Plan and CAP.
C42-5	DID I MENTION TRAFFIC?? Commuting by bike has become more and more difficult, even with improved bike lanes, because of traffic. Thousands of people are not giving up their cars. They have no choice. Most employment centers are far from here by transit. Our few freeway access points are backed up a good part of the day. The core of Hillcrest is largely gridlocked.	C42-5	Comment noted. The comment does not identify an inadequacy in the PEIR. The PEIR for the Uptown CPU identifies significant and unavoidable impacts related to transportation and traffic. The
C42-6	PARKS?? LIBRARY?? THE USUAL PUBLIC FACILITIES?? Uptown and Hillcrest are deficient in park space and have no recreation center or aquatics facility. Stuffing 20,000 more people in the area is excessive with little hope for much improved infrastructure. Is our long awaited library funded? Under grounding postponed repeatedly. We need the facilities in the plan and funded.		proposed Uptown CPU also provides policy support for improved bicycle mobility within the CPU area.

Funding for specific facility improvements is provided in the C42-6 proposed Uptown Infrastructure Fee Study (IFS). The proposed Uptown CPU provides the policy support for specific improvements. The need for parks, libraries and other infrastructure is discussed in the PEIR Section 6.12, Public Services and Facilities. As discussed in this section, an additional library is not required to meet the library service requirements of the proposed Uptown CPU. While not required, there are plans to build an approximately 25,000-square-foot new library, which would result in an exceedance of the recommended minimum branch library size requirement of 15,000 square feet. The new library would proceed as a separate action from the proposed Uptown CPU and associated discretionary actions and would be required to undergo its own environmental review. The proposed CPU Public Facilities, Services, and Safety Element policy framework supports expanded library facilities, which the new Mission Hills/Hillcrest Branch Library would address. The proposed CPU Recreation Element Tables 7-1 and 7-2 summarize the existing and future parks, park equivalencies, and recreation facilities that have been identified in Uptown Community to supplement their existing population-based park and recreation facilities inventory. In addition to neighborhood and pocket parks, the table also includes recommendations for joint use of school property, new trails and improvements to existing trails, as well as recommendations generated by the community and City.

- C42-7
 NEW DEVELOPMENT IS GOOD...IN MODERATION Hillcrest is and always has been dense in population. It satisfied CAP long ago. The June 2015 draft plan already allowed for considerable increased density and heights in a formula that would allow growth but protect the neighborhood from becoming Century City.

 C42-8
 We like the sunlight on our streets, the diversity in the character of the neighborhood, the human scale, the sense of history. The pro-super-density people say look at North Park, so many new bars, but the North Park core has no 100'-200' buildings in their plan. We are not Little Italy. We need and want many services beyond eating and drinking.
- C42-9 OH, DID I MENTION TRAFFIC??

The draft UCP needs to be revisited with an emphasis on the community, without the paid lobbyists running the show.

Deirdre Lee 244 W. Brookes Avenue San Diego, CA 92103

- C42-7 Comment noted. The comment does not identify an inadequacy in the PEIR.
- C42-8 Comment noted. The comment does not identify an inadequacy in the PEIR. The PEIR evaluates potential impacts of the plan related to neighborhood character. The proposed Uptown CPU Urban Design Element provides design guidelines by building types to control massing and ensure compatible transitions. Building setbacks and upper-story stepbacks are recommended to address massing and compatibility where more intense development is located adjacent to lower height buildings (refer to Urban Design Element policies related to development transitions). These policies and guidelines would ensure taller buildings would not adversely impact surrounding lower intensity properties through neighborhood incompatibility or through creation of excessive shade or shadows.
- C42-9 Comment noted. The comment does not identify an inadequacy in the PEIR.

RESPONSE

ATTACHMENT 7

			1	1
	From: Sent: To: Cc: Subject:	Letter C43 Mark Leicht <travelingtoes2@hotmail.com> Friday, July 29, 2015 7:00 PM PLNL PlanningCEQA Katherine Leicht Uptown Community Plan Update, Project No. 21002568 / SCH No. Pending, Uptown, District 3rd</travelingtoes2@hotmail.com>		
C43-1	Attn: Kurtis Steiner, Having received a "h appalled by the fact Sassafrass to Redwo and 13 respectively, You and the board in Community Planning the reference Tables forum, I doubt serio very least. We the people, com Senior Environment, the affected commu- your teams actions a attendees Shamel i've never considere by the very nature o honestly say vou hau be so proud of yours i'm very much again Mark & Katherine Le Sent from my iPad	Sr. Environmental Planner City of San Diego Planning Department India Street Update" from our Middletown92103@gmail advocate, I can honestly say that I'm that the Information on closure of "The removal of the southbound lane on India Street from sold Streets" was never discuss, nor any of the the referenced Appendix C: Mobility Study, Tables 10 during my 2-hour attendiance at July 5, 2016 Uptown Community Planning Meeting genesentatives, I seem to recall 234 voting members in attendance, representing the Uptown g committee, not once was this highly controversial subject brought forth, nor placing on the ease is 10 & 13. Had you and your colleagues been as transparent as what should be expected in such a usity the meeting would have been as rational and respectful as you encountered. Shameful at the amunity advocates, property owners, future generations depend on our governing officials, especially al Planners such as yourself to provide these forums, with accurate forthright information so that inities and landholders can make valued, intelligent responses. I'm personally affronted by you and and those on the committee that chose, yes knowing chose to not bring this subject to the 200- ful. d myself a person to get to emotional over governmental proceedings and or procedural actions, but of the way this was and is being managed by the City of San Diego Planning Department, I can ve now added another name to the Middletown advocacy community, congratulations, you should self and your colleagues. Is the closure. Sith closure noted within, sacrificing property and sidewalks. light, homeowners in Middletown for 16 years.	C43-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

ATTACHMENT 7



1

RESPONSE

The public has had numerous opportunities to be a part of the development of the proposed Uptown CPU and proposed potential historic districts. Additionally, the supplemental development regulations would only apply to properties that have been identified as having some historic significance and would not be an arbitrary application.

C44-1 Comment noted. The City appreciates your participation in the public review process.

C44-2 Comment noted. This comment does not suggest an inadequacy of the PEIR. Note, the proposed potential historic districts and associated amendments to the historical resources regulations would not be effective until adopted by City Council and would be an amendment to the Municipal Code.

Comment noted. This comment does not identify an inadequacy in the PEIR. However, it should be noted that the proposed potential historic districts and associated amendments to the Historical Resources Regulations would regulate certain modifications to residential structures identified as contributing resources and would not severely restrict property rights. The identified potential historic districts meet the National Register standards for determining district boundaries and appear to meet at least one of the City's local designation criteria for historical sites. Refer to Appendix G-2, Historical Resources Survey Report, for the results of the research conducted on these districts. The amended Historical Resources Regulations are intended to provide supplemental development regulations to protect potential historic districts until they are formally evaluated and designated. These supplemental development regulations would address how and where modifications can be made on residential properties identified as potentially contributing to specified potential historic districts. Without these amended regulations, future development has the potential to substantially degrade or destroy resources potentially contributing to a potential historic district, which would result in significant and irreversible impacts. Should a potential historic district or potentially contributing property be evaluated and found ineligible, the protections would not apply.

- C44-4 The adoption of these three Updates would establish a Potential Historic District Overlay Zone (PHDOZ) in order to provide "interim protection" to the privately owned residential properties within such Potential Historic Districts (PHDs). The adoption would also amend the Municipal Code to allow such Potential Historical Districts. These properties would be "protected" from their owners' exercise of the rights that other private property owners possess under the Municipal Code. Those rights are to modify or redevelop their properties as they deem necessary after complying with the City's current "Over 45 Potential Historical Review" process and other land use regulations. The properties that would be protected from the exercise of these rights are located in the 36 Potential Historical Districts to be included in the PHDOZ. These districts contain 4843 privately owned residential parcels. Our documentation of the number of parcels affected by these proposed restrictions is based on the Tables contained in these three Plan Updates as set forth in Exhibit A to this letter.
- C44-5 In making these comments, I am relying on my thirty-four years of experience in representing public and private property owners of older properties within the City and my review of these proposed Updates and their supporting documents. Since the 1980s, I have noticed that the Planning Department is bifurcated, with one segment devoted to slowing or preventing change in the City and another segment devoted to managing change in the City. Somehow the two segments have continued to co-exist all these years and both contain dedicated public servants.
- C44-6 I will begin with an explanation of the City current regulatory process for addressing modifications proposed for properties with structures more than 45 years of age. It is necessary to understand how the current system works in order to understand the impact of the restrictions that will be imposed on the "protected" Potential Historical District properties.

The Current "Over 45 Potential Historical Review" System

C44-7 The current system is governed by the Municipal Code's Historical Resources Regulations in Chapter 14 and the City's Information Bulletin 580. Whenever a property owner applies for a building permit, City staff determines whether the affected structure is more than 45 years of age and, if so, whether it is a potential historical resource. If City staff requires more information on this topic, the property owner can participate in a Single Discipline Preliminary Review process under Information Bulletin 580 during which the Historical Resources Board (HRB) staff can review and possibly clear the property as not a Potential Historical Resource. At that point, this clearance is good for five years and the property owners are free to exercise their property rights. Since the current system went into place, HRB staff has documented these Preliminary Review clearances and their clearance rates have often been in the 70% range. This process has allowed

2

- C44-4 Comment noted. This comment does not identify an inadequacy of the PEIR. An owner's rights to modify or redevelop their properties would not be completely lost under the amended Historical Resources Regulations. Modifications to structures that potentially contribute to a potential historic district which comply with the requirements of the supplemental development regulations may be processed through a ministerial building permit, while modifications which do not comply with the supplemental development regulations would still be allowed with issuance of a Neighborhood Development Permit with deviation findings and mitigation, as appropriate.
- C44-5 Comment noted. This comment does not identify an inadequacy of the PEIR.
- C44-6 Comment noted. This comment does not identify an inadequacy of the PEIR.
- C44-7 Comment noted. This comment does not identify an inadequacy of the PEIR. The comment outlines the current review process for structures 45 years old or older under Municipal Code Section 143.0212. This review process addresses only resources which appear eligible for designation as individual resources, and does not provide any level of protection for properties that are not individually significant, but contribute to the significance of a potential historic district. The proposed supplemental development regulations for potential historic districts would add a new prong to the existing review process to protect these resources until the potential historic district can be intensively surveyed and brought forward for designation.

RESPONSE

property owners to exercise of their property rights in a fairly reasonable fashion and thereby increase the amount and quality of in-fill housing allowed in the City's older neighborhoods.

- C44-8 If the HRB staff cannot resolve the Potential Historical Resource issue through this Preliminary Review process, they will offer the property owner two options. One option is to revise their development proposal so that it meets the Secretary of the Interior's Standards for Rehabilitation, which are intended to guide rehabilitation projects for already designated properties. The other option is for the property owner to prepare and submit a Historical Resources Research Report on the property that complies with the City's Guidelines for such Reports. After the full Report is submitted, HRB staff may clear the property as not a Potential Historical Resource and, again, that clearance is good for five years. If Staff cannot clear the property, and the owners still wishes to proceed with their development, the property can referred to the HRB for a designation consideration. If the HRB votes not to designate, the property owners are free to proceed with their development. If the HRB votes to designate, the property owner can appeal that decision to the City Council under specific grounds for appeal in the Municipal Code, or the property owner can apply for a Site Development Permit to either relocate the structure or demolish the structure. A Site Development Permit for either option can be issued if the findings required by Chapter 14 of the Municipal Code can be made by the Planning Commission. Those findings are, primarily, based upon the economic infeasibility of retention of the structure on site in its current form.
- C44-9 As the result of the pre-2000 and post-2000 "Potential Historical Review" systems, dating back to 1967, the City of San Diego has designated approximately 2,798 historical resources.¹ This number includes resources designated as contributing structures to 17 established Historical Districts, the establishment of which will be discussed below. Local historical designation is desirable because it can result in a substantial reduction in property taxes. Consequently, under the current system there will be no shortage of voluntary designations in the future.
- C44-10 In November of 2014, the City adopted a new Historic Resource Designation Process, under which the HRB will consider 4 voluntary designation nominations per month and, since the Board meets 11 months per year, the maximum number of voluntary designations per year is 44. Between June of 2015 and June of 2016, the Board designated 42 properties in response to the owners' requests; therefore, the new process is working. The other items heard by the Board include the involuntary designation applications related to permit applications, National and State Register recommendations, historic district designations and other items within the Board's jurisdiction. Because of the great demand for property tax relief, it is reasonable to assume that

3

C44-8 Comment noted. This comment does not identify an inadequacy of the PEIR. Please see response to C44-7 above.

- C44-9 Comment noted. This comment does not identify an inadequacy of the PEIR.
- C44-10 Comment noted. This comment does not identify an inadequacy of the PEIR.

¹ This number was determined by reviewing the City's Historical Resources Board records and CHRID.

ATTACHMENT 7

RESPONSE

the Board will continue to hear approximately 44 voluntary designation applications in each of the coming years. And, because of the great demand for new housing and in-fill development in the City, it is also reasonable to assume that the Board's agendas will also include a comparable number of involuntary designation actions in instances where new development is proposed for the sites.

C44-11 These factors pertaining to the Board's current work load are relevant when considering the new work load that would be created with the adoption of the Potential Historic District Overlay Zone (PHDOZ), discussed below.

The role of Historic Reconnaissance Surveys

- C44-12 Within the last two decades, a number of historic reconnaissance surveys have been conducted by consultants to the City in each of these three Community Plan areas. These surveys include drive-bys and additional research to identify "potential individual historical resources" and "potential historical districts" to "inform" the city-wide community planning process. Although these surveys are public documents, and identify thousands of potential historical properties, few property owners are aware of them.
- C44-13 These Surveys, some of which date to 2006, are the basis for these proposed Community Plan Updates, which will amend the Municipal Code and create new regulatory procedures to restrict development in 36 Potential Historical Districts (PHDs) containing 4843 parcels.
- C44-14 These Community Plan Updates also reference other surveyed properties that are treated under a new category for San Diego as "Multiple Property Listings" and include Victorian Era Resources, Bungalow and Apartment Courts and properties associated with Kate Sessions. These properties are not referenced in this letter because they will continue to be regulated by the existing Potential Historical Review system and they are not impacted by the PHDs.

Potential Historical Review System proposed by these new Community Plan Updates

C44-15 As acknowledged in the Planning Department's Fact Sheet on these Potential Historical Districts, attached as Exhibit B, about 95% of the residential buildings within these 36 PHDs are over 45 years old. Under the City's current historic regulations, buildings over 45 years of age must be reviewed for individual historic significance and if they do not meet the City's criteria for designation, they can be cleared and are no longer subject to historical regulatory restrictions. The "slow change" Planning Department's concern is that while a building may not be individually significant today, sometime in the future interest may develop to create a historic district in the neighborhood. Contributors to historic districts are properties that may not be

- C44-11 Comment noted. This comment does not identify an inadequacy of the PEIR.
- C44-12 Comment noted. This comment does not identify an inadequacy of the PEIR. However, it should be noted that the Historic Resources Survey prepared in support of the Community Plan Update was prepared consistent with Federal and State guidance and historic preservation best practices. The survey process, like the CPU process, has involved extensive outreach and publicly noticed meetings. In addition, all public hearings associated with the CPU adoption are noticed as required.
- C44-13 Comment noted. This comment does not identify an inadequacy of the PEIR. Please see response to C44-12 above.
- C44-14 Comment noted. This comment does not identify an inadequacy of the PEIR.
- C44-15 Comment noted. This comment does not identify an inadequacy of the PEIR. The supplemental development regulations for potential historic districts are proposed to provide protection for the potential districts until they can be intensively surveyed and brought forward for designation. Included in the CPU package is a work program which anticipates processing of all potential historic districts in Uptown within 11 years.

ATTACHMENT 7

significant in their own right but may collectively convey significance for a district designation. The City has created 17 historic districts under their existing programs. However, these districts were created with the consent of a majority of the "contributors" in advance. ² The "slow change" Planning Department is afraid that standard historical districts could not be created today because property owners are aware of the City's housing shortage that has resulted in increased property values throughout the City, and the fact that the "manage change" Planning Department has been supporting in-fill density in the older neighborhoods that are close to job opportunities. These Potential Historical Districts are the "slow change" Planning Department's pre-emptive strike to prevent new development in these neighborhoods.

- C44-16 The entire reason for creating these Potential Historical Districts is to restrict the property rights of owners of structures that today do not merit local historical designation, on the chance that a historie district *might* be proposed in the neighborhood in the future, and that these non-significant structures *might* be found to be potential contributors to such a district. These proposed restrictions will go in effect when these Community Plans are adopted, before any actual historic districts are created and before any of these 4843 individual structures are evaluated as potential contributors to such actual districts.
- C44-17 During the time frame that these restrictions are in place, from the adoption of the Community Plans until a property is designated as a contributor to one of the actual historical districts, the affected property owners will only be allowed to make improvements to the rear one third of their homes, if they are able to obtain a Neighborhood Use Permit for that purpose. During the time frame that these restrictions are in place, the properties cannot be sold because buyers could not be assured of what the future "designation" status of the property would be and no one could be assured if or when any of the properties would qualify for the Mills Act property tax benefits. The property owners' rights will be frozen, they cannot make the improvements they want to meet their family's needs and they cannot sell the property. Their only real option would be to turn the property into an Airbnb or Vacation Rental by Owner, to earn enough money to buy a home elsewhere where they will have standard property rights.
- C44-18 Time Frame: How long would the affected property owners' rights be restricted?

In the Planning Department's Power Point on this topic dated March 2016 and credited to Kelley Stanco, Senior Planner and HRB Liaison, attached as Exhibit C, it is estimated that the Intensive

² The current system for designating historic districts is discussed as one of the Unanswered Questions below in this letter.

5

- C44-16 Comment noted. The City's reason for proposing supplemental development regulations for potential historic districts through the proposed amendments to the Historical Resources Regulations is to minimize significant impacts to historical resources.
- C44-17 Comment noted. However, property owners of potential contributing structures to potential historic districts that propose to maintain or restore the front two thirds and alter only the rear third of their structures simply require a Process One construction permit (not a Neighborhood Development Permit). If the modifications exceed the rear third and do not meet the other criteria for a Process One approval in accordance with Section 143.0255(b) of the Land Development Code, then the modifications are subject to a Neighborhood Development Permit. Property owners' rights would not be frozen, nor would the proposed amendments to the Historical Resource Regulations incentivize property owners to list their properties as short-term vacation rentals, as this comment suggests.
- C44-18 Comment noted. This comment does not identify an inadequacy of the PEIR. However, it should be noted that included in the CPU package is a work program which anticipates processing of all potential historic districts in Uptown within 11 years.

ATTACHMENT 7

LETTER

RESPONSE

Survey Work and Processing required to convert these Potential Historic Districts into actual Historic Districts would be two years per District at a cost of \$85,000 each.

Uptown Community Plan Update

19 + 4 Proposed Historic Districts = 23 x 2 years = 46 years 23 Proposed Historic Districts x \$85,000 = \$1,955,000

North Park Community Plan Update

6 + 5 Proposed Historic Districts = 11 x 2 = 22 years 11 Proposed Historic x \$\$5,000 = \$935,000

Greater Golden Hill Plan Update

2 Proposed Historic Districts = 2 x 2 years = 4 years 2 Proposed Historic Districts x \$85,000 = \$170,000

Total costs to the City = \$3,060,000. Total Time Frame: 46 + 22 + 4 = 72 years

However, this time frame doesn't take into consideration the fact that the Historical Resources Board can only process a specific number of voluntary and involuntary designation actions per year, not including these 36 PHDs. In order to stick to the 72 year schedule, additional staff and a second Historical Resources Board could be required.

C44-19 Series of Unanswered Questions

Lack of Notice

It is important to understand that the relatively simple action of adopting these three Community Plan Updates will restrict the property rights of the owners of 4843 residential properties for decades to come. It is unclear whether these owners will be notified of this action or its consequences in advance. As far as we know, no comprehensive notice of what is proposed has been distributed to the 4843 property owners to date. Because of the great number of parcels that would be restricted under this proposal, notice to the individual property owners may not be required. Instead, a Legal Notice would be published in the local newspaper. This lack of notice may be legally justified under other land use processes, such as zoning changes, when all like properties in a specific zone, such as single family residential uses, are identically treated. But this lack of notice may be questionable when only certain properties within a given zone will be impacted.

6

C44-19 Comment noted. This comment does not identify an inadequacy of the PEIR. However, it should be noted that the Community Plan Update process has involved extensive outreach and publicly noticed meetings. In addition, all public hearings associated with the CPU adoption are noticed as required.

C44-20 If the City Attorney's Office has opined that this lack of notice is permissible in this situation, C44-20 Comment noted. This comment does not identify an inadequacy the Council Members from the impacted Council Districts may want to take a second look at this of the PEIR. Please see response to C44-19 above. issue, as they will be the ones receiving the angry phone calls from their constituents. Beyond the legality of this proposed lack of notice to the 4843 property owners, this proposal does not appear to meet the principle of "transparency" in local government that we have heard so much about in recent years. Compliance with existing Historical Resources Board's Policies Comment noted. This comment does not identify an inadequacy C44-21 C44-21 The Historical Resources Board Policy 4.1, Procedure on Establishing Historic Districts, last amended on October 27, 2011 reads as follows. of the PEIR. However, in response to the question regarding the soliciting of support and opposition from property owners and Designation Request: "If the request is brought forth by the City, based on comprehensive historical studies available to the City, staff shall include information neighbors, the requirements cited relate to the designation of a showing the extent of community support and involvement in the preparation of the historic district, which is not proposed with the CPU package. The Historic District Document. For all nominations, staff will solicit the level of support and required outreach efforts would be conducted at the time a opposition from property owners within the proposed district boundary consistent with the process outlined in the District Nomination Guidelines, and shall provide that potential historic district is brought forward for nomination, information to the Board." consistent with all Historical Resources Board procedures and This Policy Statement is attached as Exhibit D and illustrates the inconsistency of the proposed Municipal Code requirements. Potential Historical Districts with the current regulations in effect. This issue is also addressed in Exhibit E, which is from the current Historical Resources Board website under Frequently Asked Questions: "What if some of my neighbors object to being a part of a historical district?" "If the district criteria are met and a majority of contributors consent to designation, the Board can establish the district, despite minor objections." Under Policy 4.1, historic districts can only be created by the Historical Resources Board and only after staff has solicited the level of support and opposition from property owners within the proposed district boundary. In this case, that would mean the property owners within the proposed 36 district boundaries. Is it legally possible to create 36 historic districts without complying with this procedure and soliciting the level of support and opposition from affected property owners? 7

C44-22 How did the Historical Resources Board lose their Municipal Code authority to establish Historical Districts?

According to Chapter 12: Land Development Reviews, Article: Zoning, Division 2: Designation of Historical Resources Procedures, Section 123.0201 through 123.0206, only the Historical Resources Board is authorized to review and "make a decision on whether to designate a historical resource," including a historical district, "based on the criteria specified in and consistent with the procedures of the Historical Resources Guidelines of the Land Development Manual." See Exhibit F.

C44-23 I appreciate this opportunity to participate in the City's consideration of these Community Plan Updates that would impact so many owners of older residences in these Plan Areas. I realize that my comments are only tangentially related to the adequacy of these Draft PEIRs, but I wanted to raise these historical resource issues at the earliest possible opportunity in the public review of these Plan Updates to ensure that these issues are addressed in the public dialogue that will follow.

Respectfully submitted,

Marie Burke Lia Attorney at Law

8

- C44-22 Comment noted. This comment does not identify an inadequacy of the PEIR. Note that potential historic districts are not the same and are not offered the same protections as designated Historic Districts.
- C44-23 Comment noted. All comments will be considered during the decision-making process.

ATTACHMENT 7



Exhibit A

This estimate is based on the information contained in each Community Plan Update as cited below.

Uptown Community Plan area:

- This Plan area includes 340 parcels with already designated local historical resources subject to the benefits and burdens of local designation. (Plan Table 10-2) These parcels are not included in our total count of affected properties.
- 2) The Uptown Plan proposes 19 potential historic districts that contain a total of 2821 parcels. (Plan Table 10-3)
- This Plan proposes to include Bungalow and Apartments Courts in a single Multiple Property Listing containing 150 parcels. (Plan Table 10-4)
- 4) Properties associated with Kate Olivia Sessions are to be included in a single Multiple Property Listing containing an unknown number of parcels. No list of such properties is included in this data base so there is no way of knowing how many parcels will be affected for category. (Plan Table 10-5)
- Victorian Era Resources within this Plan area, that were identified in the 2006 Survey, are to be included in a single Multiple Property Listing containing 458 parcels (Plan Table 10-6)
- Public Outreach has resulted in 27 parcels being identified as potentially significant individual resources for designation. (Plan Table 10-7)
- Public Outreach as also resulted in 4 additional Potential Historic Districts being identified, containing 354 parcels. (Plan Table 10-8)

North Park Community Plan area:

- This Plan area includes 105 parcels with already designated local historical resources subject to the benefits and burdens of local designation. (Plan Table 10-2) These parcels are not included in our total count of affected properties.
- This Plan has identified 47 new potentially individually significant resources. (Plan Table 10-3)

. . .

RESPONSE

- This Plan has identified 6 new potential historic districts containing 269 parcels. (Plan Table 10-4)
- This Plan has proposed a new MPL for Residential Courts, containing 92 parcels. (Plan Table 10-5)
- 5) Public Outreach has identified 21 potentially significant individual resources in this area. (Plan Table 10-6)
- 6) Public Research has identified 5 additions to the Potential Residential Courts to be included in the MPL for this category. (Plan Table 10-7)
- Public Outreach has identified 5 additional Potential Historic districts, containing 754 parcels. (Plan Table 10-8)

Greater Golden Hill Community Plan Area:

- This Plan area includes 77 parcels with already designated local historical resources subject to the benefits and burdens of local designation. (Plan Table 10-1) These parcels are not included in our total count of affected properties.
- This Plan has identified new potentially individually significant resources in this area affecting 52 parcels. (Plan Table 10-2)
- This Plan has proposed a new MPL for Residential Courts, containing 11 parcels. (Plan Table 10-3)
- Public Research has identified additional potentially significant individual resources, affecting 7 parcels. (Plan Table 10-4)
- Public Outreach has identified 2 additional Potential Historic districts, containing 645 parcels. (Plan Table 10-5)

The above list of parcels to be restricted in all three districts results in a total of 4,843 parcels.



ATTACHMENT 7

LETTER

RESPONSE



Communities of Uptown, North Park & Golden Hill

Potential Issue of Concern

Over time, homes of historic importance have been modified or improved to the point where they are no longer 'individually significant,' but still may 'contribute' to the significance of a potential historic district. Additionally, homes which may not be distinctive enough to meet criteria for individual designation may nevertheless contribute to the significance of a potential historic district. The loss of these potential contributing structures could significantly and adversely impact the district's eligibility for future historic designation.



Potential Historic Districts Identified

In support of the CPU process, a detailed reconnaissance survey was completed that identified 26 potential historic districts (PHD) in the three communities (19 in Uptown, 6 in North Park and 1 in Golden Hill). The analysis included a basis for the PHDs potential significance and identification of potential contributing and non-contributing resources. Additionally, a historic survey completed in 1996 identified a PHD in South Park and in west Golden Hill (Culverwell & Taggart's Addition). These identified PHDs will be subject to the supplemental regulations identified below.

Eleven additional PHDs were identified by community members (4 in Uptown and 5 in North Park) based on a cursory windshield survey. However, a detailed reconnaissance survey would need to occur in order to properly identify potential contributing and non-contributing resources. Until further analysis can be completed, these additional PHDs are not being considered for further regulations as part of the CPU process.

The LGBTQ Historic Context Statement is currently underway and will verify the presence of a
potential historic district within Hillcrest. Future intensive level survey work can build off of the
results of the LGBTQ Historic Context Statement. (Continue to page 2)

1010 Second Ave. Suite 1200, San Diego, CA 92101 | (619) 235-5200 | planning@sandiego.gov | www.sandiego.gov/planning

ATTACHMENT 7

RESPONSE

LETTER



Additionally, the San Diego Normal School/San Diego City Schools Education Complex Historic
District, which is institutional in use, was identified by the community through survey
evaluation by an independent consultant. The San Diego Normal School/San Diego City
Schools Education Complex Historic District is owned by the San Diego Unified School District
and is not subject to the City's zoning and overlay requirements.

SD

Proposed Supplemental Regulations Identified

45-year review

An estimated 95% of the structures within the PHDs are currently subject to review for 'individual significance' under the City's Historic Ordinance and new development for potentially significant structures is reviewed for consistency with the US Secretary of the Interior Standards. No changes are proposed to the current ordinance.

· Applicable to residential structures only

It is estimated that over 93% of the structures within the PHDs are low-density residential properties. Conversely, an estimated 95% of the proposed new growth is focused in commercial areas and higher density residential.

The draft Supplemental Regulations will only apply to single and multi-family residential structures within the PHDs that are currently subject to 45-year review. Commercial properties will not be subject to the Supplemental Regulations, but will continue to be subject to the 45-year review. In terms of impact, there are an estimated 2,700 residential structures in the consultant identified PHDs. Since 2003, less than 400 building permits were issued in those areas.

Supplemental Regulations

Residential structures that are not individually significant, but are potential contributors (pursuant to the criterion described in the historic survey) will be subject to the following requirements.

- No modifications allowed to the front 2/3 of the original building footprint unless the modification will repair existing historic materials or restore the building to its historic appearance.
 - Exception: Improvements exempt from building permits pursuant to SDMC 129.0203, as well as improvements identified in SDMC 143.0212(a)(1)-(4) (same standard as applied to 45-year review).
 - Exception: Deviation may be approved though a Process 2 Neighborhood Development Permit. Projects will be reviewed for consistency with the US Secretary of the Interior Standards (similar to 45-year review) and the following findings must be made.
 - ✓ All feasible measures to protect and preserve the integrity of the potential historic district have been provided; and,
 - The proposed deviation is the minimum necessary to afford relief and accommodate the development and all feasible measures to mitigate for any impacts to the potential historic district have been provided; and,
 - The proposed project will not result in a loss of integrity within the potential historic district which would render it ineligible for historic designation.

· · · · ·	
EXHIBIT C	



RTC-417

LETTER

ATTACHMENT 7

D Planning

How Will the Interim Protection Measures Work?

Policy Considerations

Significance Not Yet Established

- Intensive Survey Work and Processing Required to Designate
 1-2 Years & Approx \$85,000 Per District
- Inequitable to Hold Potential Resources to Designated Resource Standard

Regulatory Balance

- Clarity & Certainty for Public
 Quantitative Requirements
 - Easy to Understand
- Work within Building Permit Process for Compliant Projects
- · Provide Discretionary Process for Deviations



sandlego.gov

EXHIBIT D

RESPONSE



THE CITY OF SAN DIEGO Historical Resources Board

POLICY 4.1 - ADOPTED BY HSB ON JANUARY 7, 1977 AMENDED BY HRB ON AUGUST 28, 2000 AMENDED BY HRB ON APRIL 25, 2002 AMENDED BY HRB ON OCTOBER 27, 2011

PROCEDURE ON ESTABLISHING HISTORIC DISTRICTS

1. PURPOSE AND INTENT

This policy is intended to guide the designation of historical districts within the City of San Diego. A Historical District is a significant concentration, linkage, or continuity of sites, buildings, structures, or objects that are united historically, geographically, or aesthetically by plan or physical development and that have a special character, historical interest, cultural or aesthetic value, or that represents one or more architectural periods or styles in the history and development of the City (SDMC 113.0103.) Designated historic districts are those districts designated by the Historical Resources Board pursuant to Land Development Code Chapter 12, Article 3, Division 2, and included in the City of San Diego Historical Resources Board Register, or listed in or determined to be eligible for listing in the California Register of Historic Resources or the National Register of Historic Places, Individuals or groups interested in submitting a historic district nomination for designation should review the Guidelines for Preparing a Historic District Nomination in Consultation with Staff (District Nomination Guidelines) for direction on how to proceed.

BACKGROUND 2.

The City of San Diego Historical Resources Board (previously the Historical Sites Board) adopted its first policy on historical districts on January 7, 1977. The original policy did not distinguish different district types and provided twelve guidelines that were to be used in assessing a district's historical significance. The policy was updated on August 28, 2000, and it was at this time that districts were first separated into distinct categories. These categories included Geographic, Thematic, and Emerging. On April 25, 2002 the district policy was revised again to include the Archaeological District and the Voluntary/Traditional Historical District types.

Since the policy revision in 2002 that resulted in five district types, several issues with the district policy have been identified. Among these issues are the confusion created by different district types; the difficulty in regulating the various district types with the existing Historical Resources Regulations; the difficulty in establishing some district types; the difficulty in preserving the integrity of other district types; and the fact that the existing policy accommodated too many varying levels of community support and degrees of research and survey work. In addition, while the City is permitted to adopt its

own district policy based on local preservation needs, the existing policy was not consistent with State and National Register district policies and procedures.

As a result of these issues, the District Policy was amended in 2011 with the goal of simplifying the designation and regulation of historical districts. The district types were reduced from five to one, a standard geographic historical district. Resources that are thematically related but not located within a geographically limited and defined boundary shall be addressed through a Multiple Property Submission, consistent with National Park Service Bulletin 16B, *How to Complete the National Register Multiple Property Documentation Form*. Archaeological districts will now be processed as a standard geographical historic district under this policy.

3. EXISTING DISTRICTS

All but one of the existing districts were established as standard geographic districts, meaning they were confined to a defined geographic area, and all identified contributing properties were designated at the time the district was established. This amended district policy will not impact the way these districts are regulated. However, the Ocean Beach Cottage Emerging Historical District was established with only a context statement and period of significance to establish the significance of the district, as well as a few properties which fell within that context and period of significance that were eligible for designation under HRB Criterion F. A complete intensive survey was never completed, and therefore all eligible contributing properties are not known. Owners of properties which fall within the context statement and period of significance may bring their properties forward for designation as contributors to the district. Only those properties identified and designated as contributors are currently regulated.

Because the Ocean Beach Cottage district does not have a full intensive survey, is based on a context statement and period of significance, and is limited to those properties that fall within the context and period that are volunteered by the property owner for designation, conversion of this district to a standard geographic district is not feasible. In addition, the district's long history as a historic district precludes conversion to a Multiple Property Submission. Therefore, the district will continue to be regulated under the prior policy. Property owners may continue to bring properties forward for designation under the established context and period of significance, and the district shall remain voluntary in nature. However, no new districts will be processed under this district type.

4. CRITERIA

The district must be evaluated and determined to be significant under one or more of the adopted Designation Criteria A-E. Contributing resources are designated under Criterion F. The Board's adopted *Guidelines for the Application of Historical Resources Board Designation Criteria* provides guidance on the proper application of the designation criteria:

Criterion A - Exemplifies or reflects special elements of the City's, a community's or a neighborhood's historical, archaeological, cultural, social, economic, political, aesthetic, engineering, landscaping or architectural development.

Criterion B - Is identified with persons or events significant in local, state or national history.

Page 2 of 7

Criterion C - Embodies distinctive characteristics of a style, type, period or method of construction or is a valuable example of the use of natural materials or craftsmanship.

Criterion D - Is representative of a notable work of a master builder, designer, architect, engineer, landscape architect, interior designer, artist or craftsman.

Criterion E - Is listed or has been determined eligible by the National Park Service for listing on the National Register of Historic Places or is listed or has been determined eligible by the State Historical Preservation Office for listing on the State Register of Historical Resources.

Criterion F - Is a finite group of resources related to one another in a clearly distinguishable way or is a geographically definable area or neighborhood containing improvements which have a special character, historical interest or aesthetic value or which represent one or more architectural periods or styles in the history and development of the City.

Research Considerations: When evaluating the district under the adopted designation criteria above, the following research considerations may be useful in developing a context and establishing significant themes:

- Common Heritage: An area associated with groups of existing or former residents who, because of their common employment or heritage have contributed significantly to the City's development. Such an area will usually contain structures of architectural interest identified with common heritage and traditional functions.
- <u>Traditional Activity</u>: An area or district associated with traditional activity, such as a central market, an educational or transportation facility, wharves, or warehousing. Such an area may also be remarkable for the particular architectural styles or method of construction associated with its original or traditional activity. Often a traditional activity has significantly shaped the history of the community which it served adding to its historic significance. If the traditional function exists in the present, it serves to illustrate the similarities and differences between past and present.
- <u>Rare Past</u>: A district which was once representative of common existence during a specific historic era but is now rare or unusual. Such as: an example of architecture, artistry, or design once common, now rare, or a function or use once common, now rare.
- <u>Development Progression</u>: Neighborhoods or districts illustrating the progressive development of style and changes in architectural and cultural taste.
- Consistent Plan: Districts illustrating the development of coherent or consistent planning and design, or innovations in planning philosophy.
- <u>Public Works</u>: Districts which illustrate the development of public works and other significant engineering achievements. During all historical periods structural aspects have been important, but after 1850, systems or construction employing steel and masonry contributed greatly to the evolution of commercial, industrial, and public buildings and therefore take a large part in the study of architecture of late periods.

Page 3 of 7

<u>Features of Daily Living</u>: Districts which illustrate the details of daily living during a previous period. Equipment or mechanical devices such as call bells, speaking tubes, dumbwaiters, fans, or similar systems are examples.

Industrial Evolution: Districts which illustrate the evolution of an industrial era and its effects on humanity. Examples include company towns, glassworks, factories, manufacturing processes, and marketing developments. Aspects of these which have been instrumental in changing modes of work, altering working conditions, improving living standards, and generally affecting the social order may also manifest significance.

<u>Craftsmanship</u>: Examples of workmanship, craftsmanship, artistry, or design which would today be economically infeasible or difficult to reproduce and/or are of benefit to the contemporary community as significant reminders of the past.

<u>Building Groupings</u>: Building groupings where the significance and importance of the individual structures is increased because of their relationship to a grouping or row of other significant structures, which may or may not be of a similar period or design style.

Landmark Supportive: District of quality buildings or sites, often made up of individual landmark structures supported by other structures of somewhat lesser importance. Such districts are normally easily definable and have a significance over and above the sum of the values of each historic site because of the total historic environment.

5. OBJECTIVE

The objective of a District is to maintain the scale and basic character of the subject district and other designated historic districts through:

- a. Protection and preservation of the basic characteristics and salient architectural details of structures insofar as these characteristics and details are compatible with the historical district, as defined by the Board in the Statement of Significance adopted for each district.
- b. Affording the widest possible scope for continuing vitality through private renewal and architectural creativity, within appropriate controls and standards. The Board intends to foster a climate in which each district may continue to exist as a living, changing neighborhood and not a static museum.
- Encouraging development of vacant property and redevelopment of incompatibly developed properties in accordance with the character of the area.
- d Encouraging continuous research into San Diego's human past and culture for the benefit of future generations.

Page 4 of 7

6. DOCUMENTATION

The following information is required to designate a historical district. These requirements are discussed in greater detail in the District Nomination Guidelines.

Geographic Boundaries: Specific geographic boundaries are established to encompass the historically significant area, sites and features.

A Context and Statement of Significance: Historic contexts are those patterns or trends in history by which a specific occurrence, property, or site is understood and its meaning (and ultimately its significance) within history or prehistory is made clear. The nomination must place the district within its historic context and develop a statement of significance that describes how the district is significant within that context under one or more of the adopted HRB Criteria A-E. For a more detailed discussion regarding historic contexts, refer to the District Nomination Guidelines.

Site Surveys: All properties within the district boundaries are evaluated and identified as contributing to the historical significance of the district, or as non-contributing sites. State of California Department of Parks and Recreation Forms (DPR-523 Forms) are provided for all properties within the District boundaries.

Contributing Sites: Contributing sites are those that meet the significance characteristic of the District and are specifically designated historical resources. These sites shall be eligible for all the benefits and responsibilities of historic designation, including the application of the Historical Building Code, Tax Code Incentives, and US Secretary of Interior Standards for development.

Non-contributing Sites: Non-contributing sites are those that have been substantially modified so that they no longer contribute to the historical integrity of the district, or sites that were developed subsequently and have no inherent historical significance or features. These sites are not eligible for benefits resulting from historical designation, except if the owner subsequently restores original historic fabric and features making it a contributing site to the district. In these cases, the Board shall hold a public hearing where the status of the site from non-contributing to contributing can be assessed and approved.

Vacant Parcels: Vacant parcels within the boundaries of a District will be deemed to have significance if they relate to the quality and character of the district, otherwise they will be classified as noncontributing sites. Alteration of vacant sites will be reviewed in a manner consistent with the applicable historical district's approved development guidelines.

District Features: The Board will determine upon designation of a Historic District those features and characteristics deemed essential to the maintenance of the district's architectural and/or historic integrity.

Development Guidelines: If deemed necessary to maintain the historical and/or architectural integrity of a historical district the Board will prepare and adopt a set of development guidelines to be used in development project review for contributing and non-contributing sites, and other district features such as streets, etc., to provide an appropriate context for the application of US Secretary of Interior Standards.

Demolition and Alteration: Alteration to a contributing site within a historical district must comply with the applicable provisions of the San Diego Land Development Code. For example, a Site

Page 5 of 7

ATTACHMENT 7

Development Permit is required for substantial alteration within historical districts and a recommendation of the Historical Resources Board is required prior to Planning Commission decision on a Site Development Permit. Furthermore, if a deviation for demolition or removal of a contributing structure within a historical district is approved, the applicant must obtain approval for new development before the issuance of a Demolition/Removal Permit. (San Diego Land Development Code sections 126.0502, 126.0503, 143.0250, and 143.0260(c).

Inter-Department and Agency Support: Participation of appropriate City departments will be sought to assist with the preparation of a historical district. Upon designation of a historical district, actions of Departments which could affect said district, including proposed changes in land use, changes in traffic patterns and street closing, public improvements, encroachment permits, etc. should be forwarded to the Board for review and recommendation, consistent with the applicable historical district development guidelines and Section 111.0206d(4) of San Diego Land Development Code which empowers the Board "To adopt standards and guidelines to be used by the Board in reviewing applications for development permits involving designated historical resources."

7. BOARD ACTION

The following actions are required for the City of San Diego Historical Resources Board to establish a historical district:

- a. Designation request: Any organization, or individual can bring forth a request for historical district designation, as detailed in the District Nomination Guidelines. The request should also include a petition endorsed by a substantial number or a majority of the affected property owners. If the request is brought forth by the City, based on comprehensive historical studies available to the City, staff shall include information showing the extent of community support and involvement in the preparation of the Historic District Document. For all nominations, staff will solicit the level of support and opposition from property owners within the proposed district boundary consistent with the process outlined in the District Nomination Guidelines, and shall provide that information to the Board.
- b Historical Report: The request shall include a Historical Report with information about the proposed historical district, including a Methods section, Context, Statement of Significance, Period of Significance, boundaries or area of effect, DPR-523 Forms, and Development Guidelines as needed. Information will be submitted to staff of the Historical Resources Board for review.
- d. Board Review: Two meetings of the Board are required to establish a historical district. Upon receipt of a complete historical district designation package, the Historical Resources Board staff shall schedule the item for review by the Board. At this time the Board shall evaluate the completeness and adequacy of the information submitted establishing the significance of the proposed historical district at a regularly scheduled Board meeting. If the information submitted is found adequate, a second noticed public hearing shall be scheduled for the next available Board hearing.

Page 6 of 7
- e Noticing: Notices will be mailed as required by the San Diego Land Development Code Section 123.0202 (b) to all affected property owners, and community planning groups, neighborhood associations, historical societies, and other interested parties.
- f. **Site Visit:** Historical Resources Board members are required to physically visit the district area and view the sites within the district's boundary before taking any action.
- g. Board Hearing: The Board will hear public testimony on the establishment of the historical district, and take appropriate action. The action of the Board to designate a historical district may be appealed to the City Council as established by the San Diego Land Development Code Section 123.0203.
- h. Implementation: Upon Board designation of a historical district the boundaries of said district shall be transmitted to all affected City departments so they may be aware of the Board's interest and involvement in any actions that could potentially affect the historical integrity and significance of the district. The Board shall review any development request affecting a significant historical resource as established by the various sections of the City of San Diego Land Development Code, to provide the appropriate recommendations to the decision maker. Additionally, contributing sites within a historical district shall be eligible for the Mills Act Program provided they meet the standards of the program.

Page 7 of 7



Frequently Asked Questions

What is a historical district?

A historical district is a significant concentration, linkage or continuity of sites, buildings, structures or objects that are united historically, geographically, or aesthetically by plan or physical development and that have a special character, historical interest, cultural or aesthetic value, or that represent one or more architectural periods or style in the history and development of the city as defined in San Diego Municipal Code Section <u>113.0103</u>.

What if some of my neighbors object to being part of a historical district?

As the Historical Survey Report is being prepared, those developing the district should be establishing local support for designation. Widespread public support is most critical to establish a Geographic/Traditional District because the designation applies to ALL of the properties within the district boundaries. If significant controversy is noted at the public hearing, the Board may recommend that further public support be solicited prior to establishing the district. If district criteria are met and a majority of contributors consent to designations, the Board can establish the district, despite minor objections. Finally, historical district designations are always subject to City Council appeal, per section <u>123.0203</u> of the San Diego Municipal Code.

How long does it take to get a district designated?

Forming a district could take as much as one to two years after the documentation is submitted to the City, depending upon how many properties are included, whether professionals or volunteers are doing the work, the availability of Historical Resources Board staff for technical assistance and final review and processing, local controversy, and the Historical Resources Board docket schedule.

What are Development Guidelines?

All designated contributors to historical districts adhere to the <u>Secretary of the Interior's</u> <u>Standards for Preservation, Rehabilitation, Restoration, and Reconstruction</u> (link is external). Developed by the <u>National Parks Service</u> (link is external), these are well-recognized standards for the treatment of historical properties. Broadly written, they afford the widest possible scope for continuing neighborhood vitality through private renewal and architectural creativity.

Per the <u>Historical District Policy</u>, more restrictive guidelines may be developed if the property owners so desire or if the Historical Resources Board determines that development guidelines are appropriate. The guidelines may include standards on plantings or objects (sidewalks, lighting fixtures, street trees, etc.) within public right of way, or fencing or other aspects of the front yard on private property visible from the street. Exterior paint colors and materials can be

https://www.sandiego.gov/planning/programs/historical/fag

RESPONSE

1 12 4

regulated, as well as the scale and location of new additions on both contributors and noncontributors. Finally, development guidelines may regulate the size, scale, design and use of new infill on existing vacant lots, or where demolition of non-contributors has occurred.

Example Guidelines:

- Sherman Heights & Grant Hill Park Historic Districts:
- o Design Criteria and Guidelines

https://www.sandiego.gov/planning/programs/historical/fag



San Diego Municipal Code (12-2006) Chapter 12: Land Development Reviews

Article 3: Zoning

Division 2: Designation of Historical Resources Procedures (Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)

§123.0201 Purpose of Historical Resource Designation Procedures

The purpose of these procedures is to establish a process to identify and designate for preservation those *historical resources* that embody the special elements of the city's architectural, artistic, cultural, engineering, aesthetic, historical, political, social, and other heritages.

(Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)

§123.0202 Designation Process for Historical Resources

- (a) Nominations. Nominations of a historical resource to become a designated historical resource may originate from the Historical Resources Board, the City Manager, the City Council, or any member of the public including the property owner by submitting a research report or similar documentation, as identified in the Historical Resources Guidelines of the Land Development Manual, to the Board's administrative staff for consideration by the Board. Nominations from the City Manager may originate as a result of a sitespecific survey required for the purpose of obtaining a construction or development permit consistent with Section 143.0212.
- (b) Public Notice to Owner. The owner of a property being considered for designation by the Historical Resources Board shall be notified at least 10 *business days* before the Board hearing. Notice to the owner shall contain information about the potential impacts of designation and a request to contact the Board's administrative staff regarding information for making a presentation to the Board on the proposed designation. No action shall be taken by the Board to designate a *historical resource* except at a public hearing that provides all interested parties an opportunity to be heard.
- (c) Adequacy of Research Report. The decision on whether or not to designate a *historical resource* shall be based on the information in a research report, as specified in the Historical Resources Guidelines of the Land Development Manual. If the Board determines, either by public testimony or other documentary evidence presented to it, that the research report is not adequate to assess the significance of the *historical resource*, the Board may continue its consideration of the property for up to two regular meetings and direct that a research report be prepared by the applicant with specific direction from staff as to the inadequacies of the original report. The revised research report



a 14 e San Diego Municipal Code (12-2006) **Chapter 12: Land Development Reviews** may be prepared by City staff or volunteers, with a copy provided to the owner at least 10 business days before the next Board meeting at which the designation will be considered. If a final decision is not made within 90 calendar days of receipt of a nomination for designation, the consideration of the property by the Board shall terminate unless a continuance has been granted at the request of the property owner. (d) Continuance. At the request of the property owner, the Historical Resources Board shall grant a continuance of one scheduled Board meeting after the motion has been made to designate a historical resource. (e) Historical Resources Board Decision. The Historical Resources Board shall review the Research Report and shall make a decision on whether to designate a historical resource based on the criteria specified in, and consistent with the procedures of the Historical Resources Guidelines of the Land Development Manual. The action to designate shall require the affirmative vote by six members of the Board. Findings. The decision to designate a historical resource shall be based on (f) written *findings* describing the historical significance of the property. (g) Re-initiation of Designation Proceedings. Designation procedures may not be re-initiated within 5 years without owner consent, absent significant new information. (Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.) (Amended 12-15-2006 by O-19557 N.S.; effective 1-14-2007.) §123.0203 **Appeal From Historical Resources Board Decision** (a) The action of the Historical Resources Board in the designation process is final 11 business days following the decision of the Board unless an appeal to the City Council is filed with the City Clerk no later than 10 business days after the action of the Board. The decision of the Historical Resources Board may be appealed by an applicant or an interested person. An appeal shall be in writing and shall specify wherein there was error in the decision of the Board. The City Council may reject designation on the basis of factual errors in materials or information presented to the Board, violations of bylaws or hearing procedures by the Board or individual member, or presentation of new information. 12 3 2 2

San Diego Municipal Code (12-2006) Chapter 12: Land Development Reviews

(b) Upon the filing of the appeal, the City Clerk shall set the matter for public hearing as soon as is practicable and shall give written notice to the property owner and the appellant of the time and date set for the hearing. At the public hearing on the appeal, the City Council may by resolution affirm, reverse, or modify the determination of the Board and shall make written *findings* in support of its decision.

(Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)

§123.0204 Recordation of Designated Historical Resources

No later than 90 calendar days following the final decision to designate a historical resource, the City Manager shall record the designation with the County Recorder. (Added 12-9-1997 by 0-18451 N.S.; effective 1-1-2000.)

§123.0205 Amendment or Recision of Historical Resource Designation

The Historical Resources Board may amend or rescind any designation of a *historical resource* in the same manner and procedure as was followed in the original designation. This action may be taken only if there is new information, the discovery of earlier misinformation, or a change in circumstances surrounding the original designation.

(Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)

§123.0206 State and National Register

The City Council shall consider endorsing the nomination of a *historical resource* for inclusion in the California Register of Historic Resources and the National Register of Historic Places upon recommendation of the Historical Resources Board. (Added 12-9-1997 by O-18451 N.S.; effective 1-1-2000.)





RESPONSE

ATTACHMENT 7

C4-1 Comment noted. Please refer to the Master Response Regarding. India Street Milingation Measures included in the introduction to these response to additional to the service of the street of					
August 2, dub August 2, dub Registry 2, dub Shall		August 7 2015	Letter C46		
Ret Uptoon Community Plan Update: Project No. 2003SM/SCH Ro. Pending Community Veritigions Count District 3 C461 Weak writing as project youngs on this Street (spuce Caryon) to wake our concerns over the misponed changes to that Street due to the receive field of Car Structure Traffic. The plan to remove participation of the state way output to the schedule output to the sched		August 2, 2016 Kurtis Steiner Senior Environmental Planner City of San Diego 1010 2 nd Avenue, MS 413 San Diego, CA 92101			
C46-1 We are writing as property towns on bits Street (spruce Caryon) to wice our concerns were the induced street to be the core Neted Car Street (spruce Caryon). The vice our concerns were the induced street to be the core Neted Car Street Neted Car Street Caryon. C46-1 Comment noted. Please refer to the Master Response Regarding. India Street Mitigation Measures included in the introduction to these were worked bases to be the core Neted Car Street Neted Car Street Nete Caryon. C46-1 Comment noted. Please refer to the Master Response Regarding. India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6), which include a discussion on the Rental Car Center. Note the south base to the access we would be feeling as property/builtones on the 320 biols of bias incerns the south base to the south base on the south street in the ingrup of the south base on the south three three the south base on the south three three the south three		Re: Uptown Community Plan Update; Project N Uptown; Council District: 3	lo. 21002568/SCH No. Pending; Community Area:		
 C46-1 We are writing as property owners on this Street (Snuce Canyon Its visice are concerns over the interview individual street due to the street from take Street (Snuce Canyon Its visice and bardhips to our Mediation It digitification It digitification Its street would be facing as property/business owners: India Street St		Dear Mr. Steiner:		C46-1	Comment noted. Please refer to the Master Response Regarding
Following are only some of the issues we would be facing so progently/business owner. 1. Ingrest/grades to Spruce 10 spruce to this a considerable distance to entre Spruce SL, which turns into this. 2. Ibis 52, privide sound later to the acconsiderable distance to entre Spruce SL, which turns into this. 3. Ibis 52, privide sound later to brack or Splute SL brack or Distancess, etc. (ep. bracker), property (30%) (10% brack on Distancess, etc. (ep. bracker), property values could decrease because of the traiticing resciences and are a 2-car family. 4. Business covers would be at its to loss cover of the raiticing resciences and are a 2-car family. 5. Business covers would be at its loss covers even light of the covers would be at its loss covers and are a 2-car family. 6. Business covers would be at its loss cover and are a 2-car family. 7. BudietSum Property Values could decrease because of the traiticing resciences on India 5. and other streets in the neighborhoad. At the last Uptown community meeting, we were told there was no intention to remove parking and the one vary access Divers Spruce and Resciences on thre. We urg you to by CRYSINUSUY reconsider these plans. These dides are by no means IMPROVENENTS to this area. In addition, we urge you to abor cornicity in the streets in the neighborhoad. C46-2 Thank you for your consideration to these very grave issues. Improve the attribution of the dide to a difference is the street street in the neighborhoad. Tank you for your consideration to these very grave issues. Im MacMilian, Property Owner <tr< td=""><td>C46-1</td><td>We are writing as property owners on Ibis Stree proposed changes to India Street due to the re- sidewalks/parking and the Southbound turn lar Middletown Neighborhoods; especially, Spruce</td><td>et (Spruce Canyon) to voice our concerns over the cent Rental Car Structure Traffic. The plan to remove te on India St. pose many problems and hardships to our Canyon.</td><td></td><td>India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6), which include a discussion on the Rental Car Center.</td></tr<>	C46-1	We are writing as property owners on Ibis Stree proposed changes to India Street due to the re- sidewalks/parking and the Southbound turn lar Middletown Neighborhoods; especially, Spruce	et (Spruce Canyon) to voice our concerns over the cent Rental Car Structure Traffic. The plan to remove te on India St. pose many problems and hardships to our Canyon.		India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6), which include a discussion on the Rental Car Center.
At the last Uptown community meeting, we were told there was no intention to remove parking and the one-way access to West Spruce and Redwood only to find out that this was not true. We urge you to VERY SERIOUSIV reconsider Rentae Car traffic planning that has become a nightmare for Middletown/India St. An few ideas would be to re-route Rental Car traffic planning that has become a nightmare for Middletown/India St. An few ideas would be to re-route Rental Car Scits and Returns toward Washington St. build a ramp that would drop returns right into the structure and re-route returns into structure from Southbound Washington St. instead of Sassafras. CC46-2 Thank you for your consideration to these very grave issues. Glen Feye, Property Owner Glen Feye, Property Owner 3242 bis St., Middletown Cc: Ron Roberts; Todd Gloria		 Following are only some of the issues we would Ingress/Egress to Spruce/Ibis Streets. would have to drive a considerable dist Ibis St. parking will become an issue if I between Redwood & Sassafras). There alone. Fifty-one percent (51%) of those when considering the houses that have Business owners would be at risk to lor Property values could decrease becaus Emergency vehicles would take more to threatening emergency. Middletown/Spruce Canyon residents v India St. and other streets in the neight 	I be facing as property/business owners: There is no other outlet in Spruce Canyon. Residents ance to enter Spruce St. which turns into Ibis. India parking is removed for businesses, etc. (esp. are 31 total Units/Houses on the 3200 block of Ibis e do not have off-street parking. This % goes even higher e only 1 parking space and are a 2-car family. se some of their customers! e of the traffic ingress/egress from India. time to reach residences in case of fire or a life would not be able to walk to and from businesses on sorhood.		
C46-2 This is a closing comment. The City appreciates your participation in the public review process.		At the last Uptown community meeting, we we one-way access to West Spruce and Redwood o VERY SERIOUSLY reconsider these plans. These In addition, we urge you to also reconsider Ren Middletown/India St. An few ideas would be to Washington St.; build a ramp that would drop n structure from Southbound Washington St. insi	The told there was no intention to remove parking and the only to find out that this was not true. We urge you to i ideas are by no means IMPROVEMENTS to this area, tal Car traffic planning that has become a nightmare for o re-route Rental Cars Exits and Returns toward eturns right into the structure and re-route returns into tead of Sassafras.		
Jill MacMillan, Property Owner Glen Feye, Property Owner 3242 Ibis St., Middletown 3230-36 Ibis St., Middletown Cc: Ron Roberts; Todd Gloria	C46-2	Thank you for your consideration to these very	grave issues.	C46-2	in the public review process.
Cc: Ron Roberts; Todd Gloria		Jill MacMillan, Property Owner 3242 Ibis St., Middletown	Glen Feye, Property Owner 3230-36 Ibis St., Middletown		
		Cc: Ron Roberts; Todd Gloria			

		Letter C47		
	From: Sent: To: Subject:	pamela magnus <pamela@architectsmagnus.com> Sunday, July 31, 2016 4 15 PM PLN, PlanningCEQA Uptown Community Plan Update, PROJECT No. 21002568 / Project No. 380611</pamela@architectsmagnus.com>		
	Attn: Kurtis Steinert, Senior Environmental P City of San Diego Plant	lanner, ing Department		
	RE: Public comment on	draft PEIR of Uptown Community Plan Update related to changes on India Street		
C47-1	I am one of many conce and Redwood) to access School, and a popular b of at the India/Spruce of	rmed residents that depends on a small 2-way portion of India Street (between Sassafras s Spruce Street to get home. Spruce Street services nearly 120 residences, a Montessori ar with no other access. Hazards here are daily as children and adults are being dropped orner with no signal, stop sign or keep clear markings on India.	C47-1	Comment noted. This comment does not indicate an inadequacy of the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these
	Page 6.3-50 of the draft TRANS 6.3-19: India S lane one-way collector. ³ *Note: the current two- this access due to no off	PEIR recommends widening India Street at this intersection for increased traffic: treet (Impact 6.3-19): "Sassafras Street to Redwood Street: Widen the roadway to a 3 way access road is not mentioned and I would like to make clear 1) the importance of ter access and 2) safety measures that need to be added at this intersection.		response to comments (pages RTC-4 through RTC-6).
	Page 6.3-49 of the draft they are not proposed in *Note: I appreciate that but what then is the mea	PEIR states that "though these measures would reduce potentially siginifcant impacts, the Uptown CPU." India Street widening and removal of the 2-way is not proposed in the Uptown CPU, usure to address increased traffic?		
C47-2	Page MO-48 of the Upt be wonderful. In reality	own Community Plan shows India Street as a future bicycle lane. In theory, that would , traffic on India Street has become too hazardous.	C47-2	the PEIR. The Mobility Element of the proposed Uptown CPU also
C47-3	Middletown residents h directed to a street that n input from the neighbor CPU contirms they will alternate routes, such as be in place before increa	ave made it clear that the added airport and rental car center traffic should not be residents depend on. There needs to be accountability for increasing traffic without hood directly affected. If the PEIR confirms that traffic will increase and the Uptown not widen the roadway, then there should be a revised Traffic Impact Study to find Pacific Highway or dedicated freeway on/off ramps. Appropriate infrastructure should using traffic.		includes policies addressing pedestrian and bicycle safety, which would be implemented with any future bicycle lane plans along India Street.
	Thank you, Pamela Magnus 3370 Jackdaw Street		C47-3	Comment noted. The Traffic Impact Study looked at all possible solutions to mitigate increases in traffic due to build-out of the proposed Uptown CPU, Please refer to the Master Response
	San Diego, CA 92103			Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6), which includes a discussion on the Rental Car Center, for further detail.

		Letter C48		
	From: Sent: To: Subject:	Gregory May <gregory_may_jr@yahoo.com> Wednesday, June 29, 2016 11:09 AM Steinert, Kurtis Uptown Planners Community plan comment</gregory_may_jr@yahoo.com>		
C48-1	I just want to g June 2015 draft hard work went i it. Besides I c MY Hillcrest! V it. If we don'	go on record and say that I want the city to return to the plan without density bonuses. Approximately 7 years of into it and it would be a complete waste of time to ignore don't like the bully-property-owners/developers hijacking we have an agreed upon plan, and we should stick with t then our democratic process would be pointless.	C48-1	Comment noted. All comments will be considered during the decision-making process.
C48-2	with that in mir no matter what i we have already mandatory to pre facades should b designating ceri idea of redevel working with our designs	nd we MUST preserve what is left of our historic resources their imperfect conditions are. We are at a point where lost enough historic resources, and should make it eserve the community character. Buildings, or at least be restored, not demolished. I agree with the tain streets/blocks as a historic district. I like the opment, but I feel that a good designer can do wonders, ' historic resources and have them compliment their	C48-2	Comment noted. This comment does not identify an inadequacy in the PEIR. The proposed Uptown CPU includes numerous policies to address preservation of historical resources and identifies a number of potential historic districts. The PEIR evaluates potential impacts associated with implementation of the plan relative to
C48-3	I run a Facebook and time and tim resources is ext	k page called <u>vintage San Diego</u> , with over 37,777 likes, me again, people agree that preservation of our historic rremely important.		historical resources and concludes that even with implementation of all feasible mitigation measures; impacts to historical resources
	Sincerely, Gregory May			would be significant and unavoidable.
	1220 ROBINSON AV SAN DIEGO CA 921	/E APT 103 .03-4410	C48-3	Comment noted.
	(619) 291-4734			
		1		

From:	Roger McComb <rogeramccomb@gmail.com></rogeramccomb@gmail.com>
Sent:	Saturday, July 30, 2016 4:08 PM
To:	PLN PlanningCEQA
Subject:	Uptown Community Plan Update. Project No. 21002568

C49-1 Istrongly oppose the potential removal of the southbound lane on India Street from Sassafras to Redwood Street. Removal of the southbound lane would provide only one point of access to the residents on the 1000 and 1100 blocks of West Palm and the surrounding area and that would be from Laurel Street. Residents heading home from Pt Loma or areas north of the airport would be forced to go out of their way to get home instead of being able to drive down Kettner and turn up Sassafras. The city should not be dumping on it's residents because of the new Rentall Center.

Regards,

Roger McComb

C49-1 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

RESPONSE

Letter C50

 From:
 Tricla Mercer <tmercer12@gmail.com>

 Sent:
 Monday, August 01, 2016 10:11 AM

 To:
 PLN_PlanningCEQA

 Subject:
 Uptown Community Plan Update 21002568 / SCH No. Pending

To whom it may concern:

C50-1 I am a resident of Middletown and currently reside on Columbia Street near Sassafras. The current plans to turn India into a three lane highway is incredibly concerning especially with the recent issues that have been occurring due to the opening of the rental car center and the traffic being routed up Sassafras to India Street. In the past two months, I have had to slam on my breaks to avoid being side swiped while crossing India street heading West on Sassafras toward Pacific Highway. I was also inches away from being struck by a car while walking across India with my sister and my dog.

The issue is the steep Sassafras hill we are now having tourists navigate to reach the 5 freeway entrance. Locals that frequently travel that hill have experience with traffic flow and recognize that they need to yield when turning left onto India. I believe tourists are so uncomfortable with the steep hill that they immediately turn left without looking for oncoming traffic or pedestrians crossing the street.

Why isn't Pacific Highway being utilized? It's equipped for heavy traffic. Have the rental car exit turn left onto Pacific Highway, then right on Washington, then left on San Diego Ave to access the 5 Freeway. Why make changes to a local neighborhood to accommodate rental car traffic, when instead the business roads that have already been established could be utilized.

Please take into consideration that you're disrupting an entire neighborhood to accommodate tourists when there are well equipped roads in place.

Thank you for your time.

Concerned resident.

Patricia Mercer

Ps. Please install a lefthand light signal on Sassafras and Pacific Highway. I have sat at that light for three light cycles due to the heavy rental cars traffic exiting and crossing over Pacific Highway. I'm just trying to get to work. I have found a new route to work, but installing a lefthand turn signal will eliminate the congestion as well. Thank you.

C50-1 Comment noted. This comment does not identify an inadequacy of the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6), which includes a discussion on the Rental Car Center.

	Letter C51		
	From: Pretash Mistry <premistry@gmail.com> Sent: Tuesday, August 02, 2016 8:40 PM To: PLN_PlanningCEQA Subject: Uptown Community Plan Update - Project No. 21002568</premistry@gmail.com>		
C51-1	I am writing to formally state my concern with the current plan in place to remove parking and sidewalks on india street. Parking is difficult as it is on the side streets off India and this would only make that situation worse.	C51-1	Comment noted. This comment does not indicate in inadequacy of the PEIR. Please refer to the Master Response Regarding India
C51-2	I'd like to encourage the planning department to make India street safer and more beautiful for the residents and visitors that travel through the area.		Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C51-3	1. The pedestrian crossing across India to get to the bridge over the 5 freeway is impossible to cross because traffic does not stop and is speeding at highway speeds. This should be made a formal crosswalk requiring traffic to stop while people safely cross the street to get to the bridge. Same goes for Kettner to get to the trolley station. This should be made safer for residents trying to get to the trolley.	C51-2	Comment noted.
C51-4	2. Traffic on India moves too fast for posted speed limit signs. A light or stop sign should be placed at or before redwood to slow traffic down and make it easier to traffic to turn onto Redwood and Spruce Streets. There is a school steps away from India street. This is a tragedy waiting to happen. Signs should be on india street stating it is a school zone and all applicable laws should apply.	C51-3	Comment noted. This comment does not indicate in inadequacy of the PEIR, which analyzed impacts of the proposed CPU compared
C51-5	3. India street needs to be repayed with bike awareness		to existing conditions. Policies MO-1.1, MO-1.2, MO-1.6, and MO-
	Thanks for taking this into account regarding the future plan for the Middletown area.		and crossings in the Uptown community.
	Pretesh Mistry, Middletown Resident		
		C51-4	Comment noted. This comment does not indicate in inadequacy of the PEIR; therefore, a detailed response is not required. However, all comments will be considered during the decision-making process. The Mobility Element of the proposed Uptown CPU addresses the need for traffic calming improvements in the community.
		C51-5	Comment noted. This comment does not indicate in inadequacy of the PEIR; therefore, a detailed response is not required. However, all comments will be considered during the decision-making process. The Mobility Element of the proposed Uptown CPU contains numerous policies aimed at enhancing the bicycle network of the community.

RESPONSE

	Scott A. Moomjian Attorney at Law 5173 Waring Road, #145 San Diego, California 92120 Telephone (619) 230-1770 Facsimile (619) 785-3340 smoomjian@earthlink.net		
	August 6, 2016		
	Mr. Kurtis Steinert San Diego Senior Environmental Planner City of San Diego Planning Department 1010 Second Avenue, MS 413 San Diego, CA 92101 Sent Via Electronic (E-mail) Transmittal PlanningCEOA@sandiego.gov		
	Re: Uptown Community Plan Update Project Number: 21002568; SCH No. Pending		
	Dear Mr. Steinert:		
C52-1	I am writing this letter in order to express my strong concerns regarding the Uptown Community Plan Update ("Plan Update") Program Environmental Impact Report ("PEIR") as they relate to the proposed treatment, processing, consideration, and disposition of potential historical resources within the Uptown community.	C52-1	Comment noted.
C52-2	My background in the field of historic resources is extensive. Over the past twenty six years, I have worked on hundreds of projects involving historic properties. In the past sixteen years, I have represented owners of historic properties achieve their objectives with local, state and federal government agencies that supervise or regulate such properties. Where appropriate, I have nominated them to local and national historic registers. I have also prepared or consulted on historical reports for historic properties throughout the County. A significant portion of my work has involved facilitating the rehabilitation of buildings or the redevelopment of sites containing historic resources. I am a qualified historical consultant by the City of San Diego, and my professional qualifications meet the Secretary of the Interior's <i>Standards and Guidelines for Archaeology and Historic Preservation (1995)</i> in the disciplines of Architectural History, Historical Preservation, and History.	C52-2	This comment includes background information on the commenter. The City appreciates your participation in the public review process.
C52-3	I have reviewed the relevant environmental documents associated with historical resources prepared in conjunction with the project, including but not limited to, the Plan Update I	C52-3	Comment noted. Appendix G-2, Uptown Community Plan Area Historical Resources Survey Report has been prepared by the City of San Diego Planning Department, and will be adopted as an appendix to the proposed Uptown CPU.

ATTACHMENT 7

and PEIR dated June 10, 2016 ("Historical Resources, Section 6.7.2.2); the Uptown Community Plan Area Draft Historic Resources Survey Report ("Survey Report") dated November 2015 and revised May 2016 (Appendix G-2) with Appendices A-G; the undated City of San Diego, Planning Department "Potential Historic Districts Fact Sheet" for the Uptown, North Park, and Golden Hill communities; and proposed San Diego Municipal Code section revisions. I am also well familiar with previous historic surveys conducted in the Uptown area, including the *Historic Resources Inventory for "Uptown Area," San Diego California* (1981), and the draft *Uptown Historic Architectural & Cultural Landscape Reconnaissance Survey* (2007) ("Draft Uptown Survey"). It should be noted that to date, that no historic surveys or historic resource inventories for the Uptown community have been formally reviewed or adopted by the City of San Diego.

C52-4 The scope of my comments herein presented will be limited to problems associated with (1) the proposed Draft Historic Resources Survey Report; (2) the proposed regulatory framework for potential historic districts (PHDs); and (3) Multiple Property Listings (MPLs), specifically the Bungalow and Apartment Court MPL. Collectively, each of my arguments substantiate the inherent deficiencies and flaws in the Survey Report and proposed City action. As a result, they should be rejected in their entirety.

C52-5 (1) <u>Proposed Draft Historic Resources Survey Report</u>

As an initial matter, there are thousands of properties located within the boundaries of the Uptown Community Plan Area and the geographic area is massive. According to the Survey Report, the Planning Area encompasses nearly 2,700 acres and contains the communities of Park West, Middletown, Mission Hills, Hillcrest, the Medical Complex area, as well as the western half of University Heights. While the earlier Draft Uptown Survey (2007) surveyed 11,104 properties and identified 2,192 properties as potentially significant (59 of which were located in potential historic districts), the new Survey Report identified 11,109 properties, and found that 2,134 are potentially eligible for designation as individually significant properties, including properties identified as part of potential MPLs. An additional 1,454 properties were found to be potential contributing resources to 23 potential historic districts. Finally, 6,808 properties were identified and documented in the survey, but were not determined potentially historic upon initial visual inspection. While not directly cited in the Survey Report, there are therefore, a total of approximately 3,588 properties which exist in the Uptown community, either as potentially significant individual resources, or as potentially significant contributors to a historic district. The Survey Report, however, fails to account for the true number of buildings which may be potentially significant in the Uptown community because it identifies only the number of properties (i.e. by parcels and address), and not the actual number of structures on a property (see discussion of bungalow/residential courts within the MPL below).

C52-6

According to the Survey Report, the Uptown Historical Context and Oral History Report prepared for the Draft Uptown Survey (2007) was "discarded in its entirety" and replaced by a new historic context statement prepared by City Planning Staff. Further, due to the fact that the assignment of Status Codes (which provide "a summary assessment of the resource") undertaken as part of the Draft Uptown Survey were "flawed," new Status Codes within the Survey Report

2

- C52-4 Comment noted. Detailed responses to the noted concerns are provided in the following responses to comments.
- This comment makes reference to the Historic Resources Survey C52-5 Report and notes that the inventory of potential historic properties may miscount the total number of potential historic buildings because bungalow court properties could include multiple buildings. However, Section 6.7.4, Impact Analysis, of the PEIR discloses potential direct impacts due to substantial alteration, relocation, or demolition of historic buildings, structures, objects, sites, and districts. At this program level of analysis, it is appropriate to inventory potentially historic properties, as was done in the Historic Survey Report. Providing a building specific inventory for all structures within a property is not necessary to appropriately disclose potential impacts. The mitigation framework combined with the proposed Uptown CPU policies promoting the identification and preservation of historical resources would reduce the program-level impact related to historical resources of the built environment. For example, prior to issuing any individual development permit, Mitigation Measure HIST 6.7-1 would require an historic evaluation of any building or structure over 45 years of age that may be impacted by the development. Thus, this comment does not identify a deficiency in the PEIR as both feasible mitigation and disclosure of potentially significant impacts related to historic structures was included in the Final PEIR.
- C52-6 Comment noted. This comment does not suggest an inadequacy of the PEIR.

C52-7

C52-8

RESPONSE

were assigned.¹ More specifically, those Uptown properties which were determined ineligible for local designation (assigned a "6Z" code) by the Draft Uptown Survey, were reclassified as "Identified in Reconnaissance Level Survey: Not evaluated" (reassigned as a "7R" code).

In effect, by changing a prior determination of ineligibility to a new determination of potential eligibility through a non-evaluation code, the Survey Report has essentially eliminated a former presumption of insignificance in favor of future potential significance.² By effectuating a change in Status Codes, the Survey Report has subtly undermined some of the prior evaluations undertaken as part of the Draft Uptown Survey, and cast a new "net" over these properties as potentially significant. Properties captured in this manner are now presumed to be potentially significant, rather than presumed to be ineligible for local designation. The change in Status Codes actually increases the potential for the designation. Such action interjects less assurance and more uncertainty and cost for property owners in the historic review and historic designation process.³ Based upon the foregoing deficiencies associated with the Survey Report, it should not be adopted by the City for use in the Uptown Community Plan Update.

(2) Proposed Regulatory Framework For Potential Historic Districts (PHDs)

The Survey Report states that the "The 2007 Draft Uptown Survey identified nineteen (19) potential historic districts that meet one or more of the City's local designation criteria for historical sites. In addition, City staff and members of the Uptown Community have identified four (4) additional historic districts – Allen Terrace, Avalon Heights, Hillcrest and the San Diego Normal School/San Diego City Schools Education Complex historic districts – that also appear to meet one or more of the City's local designation criteria." In total, therefore, the Survey Report proposes a total of 23 potential historic districts for the Uptown community, with a total of 1,454 properties that were found to be potential contributing resources to the 23 potential historic districts (PHDs) and their contributors generally appear to be inadequate in terms of historic methodology, historic source material, and scholarly/academic historic interpretation and analysis.

This comment does not raise an issue with the adequacy of the C52-7 PEIR. The comment makes reference to the Historic Resources Survey Report and takes issue with revisions related to assignment of California Historical Resource Status Codes, and states that the revised status codes "cast a new 'net' over properties as potentially significant." An early draft of the survey prepared in 2006-2007 assigned a status code of "6Z" or "6L" to properties that did not appear eligible based upon initial visual inspection during the reconnaissance survey. This survey was not "discarded in its entirety" as stated by the commenter. However, because the reconnaissance survey did not include sufficient detail to address potential significance unrelated to architecture, the use of the "6Z" and "6L" status codes (which indicate a lack of significance) was not appropriate per National and State guidance related to conducting surveys and assigning status codes. The "6Z" and "6L" status codes were corrected to "7R" indicating that they were identified in a reconnaissance survey, but not evaluated. Reconnaissance surveys are intended to indicate where historic resources may be present, and are never intended to provide detailed evaluation or final resolution regarding the historic status of a property, as the comment suggests. They are an informational tool which serves as a base-line for future propertyspecific and sometimes intensive evaluation. This is made clear in the State's User's Guide to the California Historical Resource Status *Codes*, which states, "users of the California Historic Resource Status Codes should keep in mind that the status codes are broad indicators which, in most cases, serve as a starting place for further consideration and evaluations. Because the assigned status code reflects an opinion or action taken at a specific point in time, the assigned status code may not accurately reflect the resource's eligibility for the National Register, California Register, or local listing or designation at some later time."

C52-8 This comment does not raise an issue with the adequacy of the PEIR. The City does not agree that the historical documentation related to the potential historic districts is inadequate.

¹ The Survey Report strongly implies that the Draft Uptown Survey, which was commissioned and paid for by the City, with oversight and input provided by the Planning Department, was inherently deficient. According to the Survey Report, "based upon the limited level of the survey work and the quality of the original Historic Context Statement on which survey was based, assignment of such a Status Code [6Z] was not appropriate."

² This is supported by the statement in the Survey Report that "[w]hile these properties have not been identified as potentially significant as part of this Survey Report, they have not been cleared as not historic, and would be evaluated in the future for historic significance at the parcel level consistent with the requirements of the City's Municipal Code. Italics added.

³ "Based on the results of the Initial Determination, if there is evidence that the site contains a historical resource, preparation of a historic evaluation is required." With the change in Status Codes, the owner of an Uptown property who submits a development application will not be able to rely upon the previous determination of ineligibility as "evidence" that the property is not significant. In such a case, since the property is included in the Survey Report, but "not evaluated" a property owner will inevitably have to spend more money and time for the preparation of a site-specific historic study.

³

- C52-9 According to the PEIR, "to further increase protection of potential resources specifically potential historic districts—the City is proposing to amend the Historical Resources Regulations to include supplemental development regulations to assist in the preservation of specified potential historic districts until they can be intensively surveyed and brought forward for designation."⁴ The City proposes to "protect" potential, historic districts by amending the San Diego Municipal Code as part of the Community Plan Update process (specifically Sections 143.0210-143.0255, *Potential Historical District*; Section 126.0402-126.0404, *Neighborhood Development Permit Requirement*; and Section 132.1601-132.1604, *Potential historic District Overlay Zone*). Such amendments would, in part, "limit modifications [to the affected property] within the front 2/3rds of the original building footprint" and allow changes to occur only "in the rear 1/3nd of the building footprint or accessory buildings." By establishing a new regulatory framework of this type, merely for the specific protection of potential, historic districts that may never come to fruition, the City will severely and adversely affect an Uptown property owner's right to otherwise develop property in a reasonable and appropriate manner.
- C52-10 In terms of the time and cost associated with the processing of the 23 PHDs, City Staff has indicated that each historic district would take 1-2 years to process at a cost of approximately \$85,000 per district.⁵ Accordingly, it would take approximately 23-46 years to process all 23 PHDS at a cost of approximately \$1,955,000. In addition, if each eligible contributing historic district property (1,454 total properties) were designated and subject to a future Mills Act agreement, the City could anticipate a loss to the General Fund of approximately over \$3.3 million.⁶ The overt development restrictions, lengthy processing time, and enormous costs involved in such a process certainly outweigh the alleged "benefits" that come from the "protection" of contributing properties to any potential, future historic districts. Based upon the overly restrictive regulations resulting from the establishment of PHDs in the Uptown community, as well as the time and costs associated with those policies and procedures, the City should reject the proposed regulatory framework for PHDs. To do otherwise would be both arbitrary and capricious.

(3) Bungalow and Apartment Court Multiple Property Listing (MPL)

C52-11

The Survey Report introduces the concept of "Multiple Property Listings" (MPLs) which are defined as "a group of related significant properties with shared themes, trends, and patterns

4

- C52-9 This comment does not raise an issue with the adequacy of the PEIR. The City does not agree that the proposed amendments to the historical resources regulations would severely and adversely affect Uptown property owners right to develop property. The intent of the amended Historical Resources Regulations is to minimize significant impacts to potential historic districts. Without the amended regulations, development consistent with the proposed Uptown CPU could result in substantial deterioration or loss of unevaluated historic resources in the community.
- C52-10 Comment noted. This comment does not raise an issue with regard to the adequacy of the PEIR.

C52-11 This comment does not raise an issue with the adequacy of the PEIR. Details on the historic value of Bungalow and Apartment Court Multiple Property Listings are provided in the Historical Resources Survey Report. These property types reflect the distinctive characteristics of courtyard design and elements of the community's social history related to multi-family, as well as architectural development associated with local transportation patterns. This meets the City of San Diego local designation Criteria A and C by definition.

⁴ "The proposed potential historic district regulations would provide supplemental protections until a more detailed historic district survey can be completed" (City of San Diego, Planning Department "Fact Sheet"). ⁵ Kelley Stanco, Senior Planner & HRB Liaison, "Interim Protections For Potential Historic Districts," Presentation, March 2016.

⁶ This amount was conservatively estimated by determining the median sale price for homes within the Uptown (92103 area code) at an amount of \$572,000 (www.trulia.com). Assuming an average Mills Act property tax reduction of 40% (per City Staff) based upon property taxes of \$5,720 per year (at 1% rate), results in a property tax savings of \$2,288. The amount of this reduction times the number of contributing properties with a Mills Act agreement (1,454 total properties) results in a loss to the General Fund of approximately \$3,326,752. Note that this amount is substantially higher if it were to include those individually significant Uptown properties (2,134) and MPLs subject to a Mills Act agreement (an additional \$4,882,592 loss to the General Fund). A total financial loss could exceed \$8.2 million

RESPONSE

of history." The Survey Report identifies three "thematically related property groupings" that appear to be significant as MPLs, including the "Bungalow and Apartment Court" MPL. This MPL is defined as a "discontinuous grouping of approximately 150 residential courts" located throughout the Uptown survey area. The Survey Report indicates that these properties derive significance under Historical Resources Board (HRB) Criterion A (Community Development) "as special elements of the Uptown Community's social history related to multi-family housing, and its architectural development associated with local transportation patterns," as well as Criterion C (Architecture) for "distinctive characteristics of courtyard design." However, these assertions are not thoroughly supported or justified by any new or meaningful historical evidence.

- C52-12 Over the past several years, City of San Diego HRB Staff has entertained a certain fascination and admiration over local bungalow/residential courts as "significant" property types. The genesis behind the history of San Diego's bungalow courts occurred with the publication of "Bungalow Courts in San Diego: Monitoring a Sense of Place" (Spring 1988) in the *Journal of San Diego History*. Subsequently, documentation of bungalow courts as a housing type within the City was discussed further in the Draft Uptown Survey (2007). In addition, one consensual HRB historic designation involving a bungalow court occurred between 2007-2008.
- C52-13 According to "Bungalow Courts in San Diego: Monitoring a Sense of Place," bungalow courts were "well-designed, small houses carefully arranged around a planned open space." They were primarily built in the *hundreds* (*if not thousands*) throughout Southern California during the 1920s and 1930s and ceased to be built around 1940. Most courts were built along new streetcar lines of the period in some variation of the Mediterraean/Mission style and covered with bougainvillea. The typical bungalow court came to feature a group of six to ten small, individual houses placed around a communal garden. Usually two standard lots were enough. According to the article, bungalow courts were classified into four categories, based upon spatial arrangement. These classes included the (1) detached, full court the "classic" court consisting of individual cottages arranged around a spacious central garden (2) detached, full court when two or more of the bungalows share a common wall, and (4) attached, narrow court. Since the term "court" implies an enclosed, designed space, in all cases the building arrangement included an end structure and a proper garden.
- C52-14 In reliance upon the above cited article, the Draft Uptown Survey (2007) identified a potential "Bungalow & Apartment Court Thematic Historic District" within the Uptown community. Although it should be noted that no present "Bungalow & Apartment Court Thematic Historic District" exists within Uptown or any other part of the City, the survey identified a total of 144 bungalow and apartment courts which were determined to be potentially significant as *district contributors only*, not individually significant, and not as MPLs.
- C52-15 According to the Draft Uptown Survey, which has been essentially adopted as part of the Survey Report, bungalow courts feature well-designed, small houses carefully arranged around a

5

- C52-12 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- C52-13 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.

- C52-14 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- C52-15 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.

ATTACHMENT 7

planned open space. The typical bungalow court consisted of a group of six to ten individual houses around a communal garden. Most bungalow courts in San Diego sit on two regular (50' X 100') lots. In several instances, the courts were built in two phases, with one side completed first, and the other side constructed when the land became available.

- C52-16 In August 2007, the "Dr. Chester Tanner Office Bungalow Court" was designated by the City of San Diego's Historical Resources Board (HRB) under HRB Criterion C as "an excellent example of both the Spanish Eclectic architectural style and as an example of a unique 1927-1935 Spanish Eclectic Office Bungalow Court." The property, located in the Uptown community, was determined to be significant due to the fact that it was identified in the draft survey; were constructed as medical office buildings (rather than residential structures); and featured many characteristics of the Spanish Eclectic architectural style.
- C52-17 In 2007, two bungalow court properties were referred to the HRB for involuntary historic site designation. The first property, located at 104-118 Dickinson Street in the Uptown community, was referred to the HRB for designation consideration under HRB Criterion C (Architecture) in November 2007 by City Staff on the basis that it, "drawing heavily from the Minimal Traditional style" was a "a good example of Streamline Modern architecture expressed in the apartment courtyard building type." When considered by the HRB, several Board Members found the property to be more Modern Minimal in style. The HRB refused to designate the property. Similarly, the second property, located at 7522-7534 Herschel Avenue in the La Jolla community, was referred to the HRB for designation consideration under HRB Criterion C (Architecture) in November 2007. City Staff believed the property to be significant on the basis that it was "a very good example of a Minimal Traditional apartment courtyard." Again, the HRB failed to designate the property.
- C52-18 In March 2008, another bungalow court property was referred to the City of San Diego's Historical Resources Board (HRB) for involuntary historic site designation. This property, located at 7417-7427 Olivetas Avenue in the La Jolla community, was referred to the HRB for designation consideration under HRB Criterion A (Community Development) as "the only Contemporary style bungalow court in La Jolla, a limited building type in the community" and under HRB Criterion C (Architecture) "as a very good example of a post-WWII, Contemporary style bungalow court with high integrity." At the hearing, the HRB designated the property, pursuant to the Staff Recommendation, despite a wealth of information supporting the conclusion that the property was appealed to the San Diego City Council and the designation was overturned on the basis that factual errors in materials and information were presented to the HRB at the time of hearing, and upon the submittal of new information indicating that the property was not significant.
- C52-19 The fundamental problem with the present Survey Report is that it alleges that bungalow/residential courts derive significance from their very *nature* as a property type (defined as a "grouping of grouping of individual properties based on shared physical or associative characteristics"). This theory essentially holds that the bungalow/residential court is

6

- C52-16 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.
- C52-17 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.

C52-18 Comment noted. This comment does not suggest an inadequacy in the PEIR; therefore, a detailed response is not required.

C52-19 This comment does not raise an issue with the adequacy of the PEIR. While bungalow and apartment courts, by their nature, are potentially significant historic structures, a formal evaluation would be required to determine significance. As detailed in the previous comments, bungalow and apartment courts are not always designated as a significant historical resource when brought forward for evaluation.

significant because of its multi-family residential use within the Uptown community. By logical extension, would a single-family residence in the Uptown community be considered significant, in and of itself, because it was originally built as a single-family home and has maintained this use over the years? In no instance does the Survey Report establish precisely why the location, design, or the use of the bungalow/residential court as "discontinuous groupings" are any more significant than other similarly-situated multi-family structures, single-family homes, commercial buildings, or other structures built throughout Uptown from the 1920s-1960s.7 Moreover, the Survey Report does not include any additional, substantial information regarding bungalow/residential courts above and beyond much of the information previously generated as part of the Draft Uptown Survey. The Survey Report also fails to explain why the concept of a bungalow/residential court MPL has been advanced when the earlier Draft Uptown Survey proposed the establishment of a potential "Bungalow & Apartment Court Thematic Historic District" within the Uptown community. Finally, the Survey Report is misleading when it asserts that "approximately 150 residential courts" located throughout the Uptown survey area would be included within the MPL and ultimately be "designated as part of a city-wide MPL of San Diego residential courts." This is especially true when one considers the fact that each bungalow/residential court, by definition, has between 6-10 individual homes on each parcel, thereby bringing the total number of actual structures eligible for designation to between 900-1,500. If designated, each eligible bungalow/residential court property subject to a Mills Act agreement could potentially cost the City's General Fund hundreds of thousands of dollars in lost revenue. Based upon the foregoing deficiencies associated with the Survey Report, it should not be adopted by the City for use in the Uptown Community Plan Update.

C52-20

In conclusion, I appreciate the opportunity to comment on the Plan Update and the PEIR. I look forward to receiving written responses to the issues I have raised in this letter. Please do not hesitate to contact me should you have any questions or need any additional information.

Sincerely,

Scott d. Moong Scott A. Moomjian Attorney at Law

⁷⁷ Additionally, the Survey Report does not explain or reconcile why the "period of significance" of bungalow/residential courts was extended to 1960, when all other prior authoritative sources have conclusively determined that the construction of bungalow/residential courts generally ended in 1940 (prior to the Second World War). C52-20 Comment noted.

ATTACHMENT 7

	Letter C53		
	From: Sandro Natale <sandro@bigblockrealty.com> Sent: Wednesday, July 27, 2016 6:45 PM To: PLN_Planning CEOA Cc: middletum/92 103/08 gmail.com Subject: Uptown Community Plan Update - Proj #21002568</sandro@bigblockrealty.com>		
	Kevin Steiner or To Whom It May Concern;		
C53-1	On Tuesday July 5th 100+ Middletown residents met with the City Planner, the Airport Authority and our elected representatives at the Joyce Beers Center for the monthly town meeting. During that meeting we had a lengthy discussion regarding the unsafe and unacceptable amount of traffic on India street due to the Airport and Rental Car Center. It was our understanding that the city planning commission in cooperation with the airport authority would immediately modify the signage so that Airport & Rental Car Center traffic would be routed North on Pacific Highway toward the Northbound 5 freeway entrance at Washington street.	C53-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
	Recently it was discovered that this topic was far from over and that a plan to modify India street was still very much in the works. This email is to serve as my formal comment regarding the matter.		
	 We OPPOSE the use of India Street as the route for cars leaving the Airport & Rental Car Center to get to the 5 freeway North. We OPPOSE the widening of India Street to accommodate Airport traffic. We OPPOSE the removal of the Southbound portion of India Street between Sassafras & West Redwood. We APPROVE the use of Pacific Highway for cars leaving the Airport & Rental Car Center traveling to the 5 freeway North. 		
C53-2	In closing, India street is a quiet, slow moving residential street which serves as the ONLY option for many residents living in the western portion of Mission Hills from Laurel to Washington street to get in and out of their neighborhood. What we're asking is simple, please move the "5 Freeway" signage that currently directs Airport & RCC traffic West on Laurel/Sassafras then North on India Street. We're asking for those same signs to be placed at Pacific Highway which would direct traffic North on Pacific Highway (a 3 Lane Non Residential Street) up to the 5 North entrance at Washington Street.	C53-2	Comment noted. This comment references alternate routes to airport traffic, but does not suggest an inadequacy in the analysis of the PEIR.
	Respectfully,		
	Sandro Natale BROKER - BRA L T Y SDD BEST Real Estate Brokerage* U-T Readers Poll 2012 - 2016 Sandro BigBlockRealty.com		

ATTACHMENT 7

		Letter C54		
	From: Sent: To: Subject:	Fred Odom <fred@odoms.net> Wednesday, July 27, 2016 3:48 PM PLM_PlanningCEQA Proposal to Eliminate Parking Lene on India Street Northbound</fred@odoms.net>		
C54-1	Sent: To: Subject: I travel Indiaa Street North there, I turn right (East) or because 95% of the traffic If the parking is eliminated essentially useless. Fred Odom 3405 tark Street San Diego, CA 92103 619 980-5749 fred@odoms.net	Wednesday, July 27, 2016 3-48 PM PLAL Planning GEQA Proposal to Eliminate Parking Lene on India Street Northbound bound to Sassafras three times weekly, and I can see no reason to eliminate the parking lane of sassafras to want to be in the right lane. There is never much traffic in the right lane is merging left to either turn left onto Sassafras or continue on to the 1-5 North ramo. It to create another traffic lane, I predict that no one will use that new lane it will be	C54-1	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

	Letter C55 J. ROBERT O'CONNOR, III LAURA W. PACKER 4304 Altamirano Way San Diego, Ca. 92103 Telephone (619) 298-9034 August 4, 2016 Hon. Mayor Kevin Faulconer Hon city Council Members City Planning Department and All Staff	C55-1 The City appreciates your participation in the public review process and has received your letter documenting that 33 homeowners of properties within the Presidio Hills Potential Historic District are opposed to this potential designation. This comment does not suggest an inadequacy of the PEIR; therefore, a detailed response is not required. However, the Presidio Hills Potential Historic District will not be designated as a result of the proposed CPU or the establishment of the supplemental regulations for potential historic districts. Additionally, no action to designate Presidion Hills as a Historic District would be taken prior to fulfilling all Historical
C55-1	RE: OPPOSITION TO PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT As part of the Uptown Community Plan Update, the Presidio Hills Potential Historic District ("PHD") is comprised of 59 homes located on four (4) streets: Altamirano Way, Cosoy	including public outreach efforts to gather input from homeowners within the Presidio Hills Potential Historic District.
	Way, Marilouise Way and Presidio Drive. At present, homeowners representing 33 of the 59 homes have signed a Petition Against the Presidio Hills PHD. More Petitions are being gathered. The Petition also states that these homeowners in the future would not vote for an "actual" Historic District for the Presidio Hills neighborhood.	The supplemental development regulations that would apply to contributing resources within a potential historic district would not require only "historic" materials be used. The regulations would
C55-2	Attached is a summary of the homeowners who have signed not only the Petition, but also, letter to each of you objecting to the Presidio Hills Potential Historic District. The actual signed Petitions and letters will be e-mailed to each of you in the next day or two. These homeowners will vigorously oppose being included in any PHD, and request that the Presidio Hills PHD be removed from the Uptown Community Plan Update.	not apply to modifications to the rear one-third of contributing resources, and the following modifications to the front two-thirds of the structure would not be limited:
	Very truly yours, Adduct O Connor & J. Robert O'Connor, III Laura W. Packer Enclosure	 Modifications that would repair existing historic materials or restore the building to its historic appearance; Modifications or repairs that are limited to an electrical or plumbing/mechanical permit that would not change the exterior; In kind roof repair and replacement; In kind foundation repair and replacement, except for structures with decorative block or cobblestone foundation; Replacement windows in existing window openings that do not require any changes to the exterior wall; Installation of fences that are 6 feet in height or less; Painting.

C55-1 (cont.)

PRESIDIO HILLS POTENTIAL DISTRICT DOCUMENTS

Petitions

Marilouise Way

2405 - Pavel

2440 - Elzy

2449 - Gillick

2464 - Saba

2425 - Fisher

Presidio Dr.

2409 - Needham

2430 - Celentino

2526 - Potts/Diamond

2429 - Moore 2435 - Wetzel

Altamirano Way

- 21

4285 - McKenzie 4224 - Golden 4255 - Tuttle 4262 - Piacquadio 4265 - Lawrence 4272 - Delue 4285 - Williams 4290 - Vano 4304 - May/Eldred 4304 - O'Connor/Packer 4314 - Cavignac 4322 - O'Connor 4330 - Casebolt 4343 - Castillo 4350 - Dunn 4356 - Comer 4361 - Scott 4365 - Scarvelis 4366 - Nelson 4369 - Dunn Cosoy Way

4269 - Pasha 4277 - Filderman 4287 - Ottenheimer

TOTAL 33

If a modification that is not exempt from the regulations is proposed in the front 2/3rds of the original building footprint, a Neighborhood Development Permit (NDP) would be required. Thus, the proposed supplemental development regulations are not anticipated to result in deferred maintenance to homes.

C55-2 Comment noted. We have received the attached petitions and letters referenced in this comment. Refer to response to comment C55-1 for the City's response to the concerns raised in the letters.

RESPONSE

Attachments to Letter C55

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: WMMCKinjie Signaturellaidere H. McKenzie Print Name William McKEnzie Print Name: Marilyon H. McKenzie Addres AZIZ ALTAMIRANO WAY Date: Juley 31, 2016

RESPONSE

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature:

•••••

.

Print Name: DAVID E. GOLDEN

Address: 4224 ALTAMIRANO WA7 San Diego, CA. 92103

Date: 07.25 , 2016

RTC-453

ATTACHMENT 7

RESPONSE

ATTACHMENT 7

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Much & Stall

••••

.

.

.

Address: 4224 ALTAMI RANO WAY Date: 7.25_, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

GEOLOF TUTTLE	Drint News
dress: 4255 ALTAMIRANO San Diego CA 92103	Date: $7/27$, 2016
San Diego, CA. 92103	

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

1.1

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Print Name: SARAH TUTTLE Smith mirano Way, Date: July 28 Address: _, 2016 San Diego, CA.

RESPONSE

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Address: <u>4262 A Hamirano</u> Way San Diego, CA. 92103

Signature: <u>Kathlean Placqued</u>ico Signature: <u>Manuel Fracqued</u>ico Print Name: <u>Jane Fracqued</u>ico Date: 7/26 2016

ATTACHMENT 7

ATTACHMENT 7

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature Signature: Print Name: ZACK LAWSENCe Print Name:

Address: 4265 ALTAMICANO WAY San Diego, CA. 92103 Date:

,2016

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL HISTORIC DISTRICT for the Presidio Hills neighborhood.

Print Name: Rawae Delug Address: 4272 ALTANIKANO WM Date: 7 29, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Lourus mi

κ.

Signature:

Print Name: LAUREN WILLIAMS

Print Name:

Address: 4285 Alfamirano Way San Diego, CA. 92103

Date: 07 28 , 2016

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature:

Print Name: George Vano

.

.

Address: 4290 Altamicano Way Date: 7/24 San Diego, CA. 92103

Signature: <u>Allison L. Vano</u> Print Name: <u>Allison L. Vano</u>

, 2016

RTC-461
PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Thomas A May Print Name: Thomas A May Print Name: Thomas A May Print Name Chuthia L. Eldred

Address: 4303 Altanizare Wkg Date: 7/24, 2016 San Diego, CA. 92103

RESPONSE

ATTACHMENT 7

RESPONSE

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: A. Holeert O'Conna & Signature: Jon H. Lobe Print Name: J. Robert O'Connor "Print Name: Loura W. Pocker

Address: <u>4364 AlfamirANo WAG</u> Date: <u>August 1</u>, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

Signature: AC Print Name: JEFF CALLGAGC Cavignae Print Name Address: San D

RESPONSE

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

1

:

.

Signature: Print Name: Day O'CONNOR Address: 4322 ALTAMIRANO Date: 7/23, 2016 San Diego, CA. 92103 WAY

ATTACHMENT 7

RESPONSE

ATTACHMENT 7

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Katherine S. Casebolt Print Name: KATHERINE S. CASEBOLT

-

.

:

-

2

.

Address: 4330 ALTAMIRANO WAY San Diego, CA. 92103

Date: 09-25-, 2016

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Hawra Castillo Print Name: Lavra Castillo Signature: Print Name: Jose Castillo

Address: <u>1243</u> <u>Alfamiliano</u> Way San Diego, CA. 92103

Date: An. 1 ,2016

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL HISTORIC DISTRICT for the Presidio Hills neighborhood.

Print Name: JOIAN DUNN AND DEADNA PAKER Address: 4350 ACTAMIRAND Wy Date: Aug 1, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature

.

Signature:

Print Name: PATRICK PAUL COMER

Print Name: ELLIE ComER

Address: 4356 ALTAMIRANO WAY Date: July 25, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

Signature: Cliff Signature: Constant Signature: Constant Signature: Constant Constant Print Name: Rebetach S Soft Address: <u>4361 Altamirano Ula</u> Date: 7/26, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Thomas R. Jellen Nel Signature: Mauren O. Jellen - Helen Signature:

Print Name: Thomas R. Yellen-Nelson

Print Name: Maureen O. Yellen-Nelson

Address: 4366 Altamirano Way San Diego, CA. 92103

Date: August 1, , 2016

.

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Print Name: John AUNN & DEANNA BAKER Address: 4369 ALTANI RAND Wy Date: Aug 1, 2016 San Diego, CA. 92103

RTC-472

ATTACHMENT 7

ATTACHMENT 7

Portfolio/Unrealized GL View - Wells Fargo Advisors, LLC

Page 4 of 4

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

-

Name: Marilyn M. Schorman RORERT J FILDERMAN Address: 4277 COSOY WAY

San Diego, CA. 92103

https://wfadvisors.wellsfargo.com/eBrokerage/eb.do?GRETEL=0AFC3F1F9F842FB8C8DF51B2A38F30AC4... 7/13/2016

RESPONSE

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

.

Signature: Eduard Delterhours, J. Signature: Largiano Orozos Print Name: EDWAND J. OTTENI HEINER, Print Name: Georgi ANA Orozos The Address: 4287 COSOY WAY San Diego, CA. 92103 Date: 4287 Cosog N. 2016 San Diego, CA. 92103

ATTACHMENT 7

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

÷

Signature: Madelaine Pavel Signature: Frank Pavel

Print Name: MAPELEWE PAVEL Print Name: BRANK PAVEL

Address: 2405 MARILOUISE WAY Date: 7-19-16, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

ignature:	Signature:	
rint Name: Bruck FISHSA	Print Name:	
ddress: 27 21 Marilouise Dry San Diego CA 92103	Date:	, 2016
San Diego, CA. 92105		

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

Signature: Signature: Print Name: JALE M. WETLEL Print Name:

Address: 2435 MARILOUISE WAY Date: 2/18/2016 San Diego. CA. 92103



PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

Signature: <u>Steven R. Ely</u> Signature: <u>Signature</u> Print Name: <u>Steven R. Ely</u> Print Name: <u>Address: 2445 Marilouise WayDate</u>, 2016 San Diego, CA. 92103

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

ə:
- 28-16,2016

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

Signature: Dovald Sabar Print Name: Dovald Sabar Print Name: Dovald Saba Print Name: BARBARA A, SABA

Address: 2464 Maril Ouise Way Date: 7/7 . 2016 San Diego, CA. 92103

RESPONSE

ATTACHMENT 7

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

1 OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Print Name: Jesnie Needh

Address: 2109 Maicho Dave San Diego, CA. 92103

2

.

14

2 ÷.

Date: 1/28, _, 2016



PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown

Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: _______ Signature: _______ Print Name: ______ Print Name: ______

...

Address: 2430 Presidio De Date: 7-19, 2016 San Diego, CA. 92103

RESPONSE

ATTACHMENT 7

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature: Jour Print Name: EARLE M. Por-

• •

:

.

.

Address: 2500 PRESIDIO JR. Date: Juny 20, 2016 San Diego, CA. 92103

RESPONSE

PETITION AGAINST PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

As a Presidio Hills neighborhood homeowner I am AGAINST the Presidio Hills Potential Historic District contained in the Uptown Community Plan Update.

I OBJECT TO THE FACT THAT I CANNOT VOTE against the Presidio Hills Potential Historic District, which violates my rights and DUE PROCESS.

In addition, I WOULD NOT VOTE FOR AN ACTUAL

HISTORIC DISTRICT for the Presidio Hills neighborhood.

Signature HALCONDA Print Name: <u>Krishe Diamond</u> Address: <u>2526 Preside</u> Date: <u>June 27</u>, 2016

1

;

.

-.

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

1 am a homeowner in the Presidio Hills neighborhood of Mission Hills. 1 am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature: En Pash Signature: John Pasha Printed Name: ERIN PASHA Printed Name: John Pasha Address: 4269 Cosoy Way Dated: Aug 3, 2016

RTC-485

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Date

Sincerely, Signature Printed Name

Signature: Printed Name:

Address: San Diego, CA. 92103

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

÷

.

• .

Signature: <u>Hilleut O'Connor</u> Signature: <u>Jour M. Pocher</u> Printed Name: <u>J. Robert O'Connor</u> <u>Itt</u> Address: <u>4304 Altamirano</u> <u>WAPated: Aug. 1</u>, 2016

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature: Printed Name: Bruck FISHER

Signature:

Printed Name:

Address:

2425 Maribuike US Date: ,2016

San Diego, CA. 92103

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

4

- -

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely, Signature: <u>Marlips III. Ellerman</u> Printed Name: <u>COPERT J FILDERMAN</u> Printed Name: <u>MATALYN M FILDERMAN</u> Address: <u>4277 Cos by WAY</u> Dated: <u>JULY 28</u>, 2016

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature:____

Signature:

Printed Name: Tack VAWSENCe

Printed Name:

Address: 4265 ALLAM, RAND WAY San Diego, CA. 92103

Date:_____, 2016

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Signature:	Signature:
Printed Name: CEOLGE	TuttLE Printed Name:
Address: 4255 ALTA	MIRONO WS 7/27 2016
San Diego, CA. 92103	,2010

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature: DaMue Printed Name: Kathleen Plaquadio

Address: <u>Hale Altamirano Way</u> San Diego, CA. 92103



I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature:

Signature:

Printed Name: LAUREN WILLIAMS

Address: 4285 Altanirano Way San Diego, CA. 92103

Date: 07/2 2016

Printed Name:

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature:

Signature:

Printed Name: PANOALL Delug

Address: 4272 ALTAMIRANO WAY San Diego, CA. 92103

Date: 7/29/ 2016

Printed Name:

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit,

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature: Printed Name: Jose Castal

Signature: Kalla Castille Printed Name: LAWA CASHILO

Address: 4343 Altamivano Will San Diego, CA. 9210

Date: Avg. 1 ,2016

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

2

•.

Signature: Rohert D. Hansen Signature: Jure N. Hansen Printed Name: Rohert D. Hausen Printed Name: Karen S. Hansen Address: 2490 Presidio Drive Dated: 3/29, 2016

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

.

z

5

Signature: WM MCKEMie Signature: WM MCKEMie Signature: WM MCKEMie Signature: Northern H LOCKCREZER Printed Name: William McKEMizic Printed Name: Havilyn K McKemice Address: 4212 ALTAMIZAND WAY Dated Sule 31, 2016
I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature Printed Nam

Signature: Printed Name:

Address: San Diego, CA. 92103

2016

LAM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

40

2

•

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely Signatu

Printed Name:

Signature: Printed Name: Equi

Address: 252 San Diego, CA. 9210.

15TIE DAMEN

Dated: Juny 24, 2016

RTC-499

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Council Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature: <u>Storm PElgo</u> Signature: Ozzie Island LLCO Printed Name: <u>Steven R. Elzy</u> Printed Name:

Address: 2445 Mar	i louise Wa Date: July 25	, 2016
San Diego, CA. 92103		

LAM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit,

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

. .

Signature: <u>Alla</u> Signature: <u>Barbara</u> Baba Printed Name: <u>Banald Saba</u> Address: <u>2464 Marilouise</u> Wy Dated: <u>7/17</u>, 2016

LAM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature:

Signature:

Printed Name:

Printed Name MINE H. WEREL

Address: 2935 MARILOUISE San Diego, CA. 92103

Dated:

LAM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

2

Υ.

Signature Connie Napodham Printed Name:

Address: 2409 Presidio Drive San Diego, CA. 92103

July 28, 2016 Dated:

Signature: Jerevice & Merdham

Printed Name: Terence E. Need ham

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature:

Signature:

Printed Name:

Address: 4255 All San Diego, CA. 92103

Dated: 2016

Printed Name:

RTC-504

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Signature

Sincerely,

ę

•

Signature:

Printed Name: PATRICK PAUL COMEN

Address: 4356 ALTAMIRANO WAY San Diego, CA. 92103

Dated: July 25, 2016

Printed Name: ELLIE COMER



RESPONSE

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely.

3

÷

1

Signature: <u>Katherine S. Casebolt</u> Printed Name: <u>Katherine S. Casebolt</u> Printed Name: <u>Katherine S. Casebolt</u> Printed Name: <u>Katherine S. Casebolt</u>

Address: <u>4335 ALTANICAND</u> WAY Dated: 07-25, 2016 San Diego, CA. 92103

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Signature

Sincerely,

...

.

÷

•

Signature:

Printed Name: DAVID E. GOLDEN

Address: 4224 ALTAMIRANO WAY San Diego, CA. 92103

Dated: 07.25 , 2016

Printed Name: LINDAK

I AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

...

.

2

•.

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

Signature: Julion Signature: Allison Z Vano Printed Name: George Vano Printed Name: Allison L. Vano Address: 4290 Altomirano Way Dated: 7/23, 2016

RTC-508

RESPONSE

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

14.1

5

1 AM OPPOSED TO THE PRESIDIO HILLS POTENTIAL HISTORIC DISTRICT

San Diego City Planning Department San Diego Development Services Department and Staff Members

RE: Presidio Hills Potential Historic District

Department and Staff Members:

I am a homeowner in the Presidio Hills neighborhood of Mission Hills. I am very much opposed to the Presidio Hills Potential Historic District set out in the proposed Uptown Community Plan Update.

I am opposed because this Potential Historic District is severely infringing upon my home ownership rights without allowing any voter input in this matter. The fact that the City planning staff wants to "freeze" their subjectively determined "character" of the Presidio Hills neighborhood for an indeterminable length of time is a substantial violation of due process.

The Potential Historic District regulations will prevent improvements to the front 2/3 of the "contributor" homes, requiring every homeowner to disclose this status to any potential buyer. Requiring that only "historic" materials be used to repair and replace items such as windows, roofs and doors could cause homes to experience deferred maintenance due to the high cost of such out-dated building materials. In short, the neighborhood could suffer - not benefit.

There is no justifiable reason to impose these burdens on homeowners, who have significant pride of ownership in their Presidio Hills neighborhood. The present review of homes 45 years and more is sufficient to protect the individual historic significance of these home.

Sincerely,

÷

۰.

Signature: Madelane Parel Signature: FRAMA Parel Printed Name MARELEINE PAVEL Printed Name: FRAMA PAVEI

Address 2405 MARILOUISE WAY Dated: 7-19, 2016 San Diego, CA. 92103

Letter C56

 From:
 Speedy Rice Gmail <speedynce@gmail.com>

 Sent:
 Saturday, July 30, 2016 2:48 PM

 To:
 PLN-PlanningECA

 Subject:
 PROJECT NAME: Uptown Community Plan Update; PROJECT No. 21002568 / Project No. 380611

 Attachments:
 Map for India Traffic study.pdf

Kurtis Steinert, Senior Environmental Planner, City of San Diego Planning Department, 1010 Second Avenue, MS 413, San Diego, CA 92101

Dear Mr. Steinert:

- C56-1 First, please note that the Uptown Community Plan Update has on its cover PROJECT PLAN 380611, but the notice for public comments on the Uptown Community Plan Update states to refer to PROJECT NO 2100258. The comments below refer to the Uptown Community Plan Update and the PEIR whichever number is correct.
- C56-2 Second, the maps used throughout the Uptown plan and appendices incorrectly depict Spruce at India and connecting streets off Spruce to show several alternative routes available to the 200+ residents living on Spruce and it's spurs (Ibis, Horton, Upas, Jackdaw) for ingress and egress. While it is unclear what effects on planning may have flowed from planners and consultants mistakenly believing that residents on Spruce et al, have alternative routes, they need to correctly considered that we have no other ingress or egress but India. Please correct the map error and do a proper re-analysis of the impact to the families that rely solely on Spruce. The error and correction is attached to show the seriousness of the map error.

Uptown Community Plan Update and PEIR:

C56-3 My wife and I would like to object to any expansion of India St between Palm and Washington St, as depicted in the June 10, 2016 (Uptown) Uptown Community Plan Update / Project No. 380611/2100258 Draft PEIR, including all appendices. As a general comment: Traffic is already too dense and travels too fast for local residents to safely enter or exit off of India by car, walk across India to the Pedestrian bridge over 1-5, walk along the adjoining sidewalk on India with any sense of security from traffic, or use the sidewalk under I-5 on India to walk to Little Italy. At various times different plans by the City and local agencies show an elimination of parking along India and or a widening of India that would take away adjoining sidewalks and add an extra travel lane. Any adoptions of those plans will negatively impact the environment and quality of life for residents along India St.

Specific Notations and Comments the Uptown Community Plan Update / Project No. 380611/2100258 / Draft PEIR, including all appendices related to India Street between

- C56-1 Comment noted. The Project Plan number refers to the proposed Uptown CPU number and the project number on the public notice refers to the PEIR for the proposed Uptown CPU and associated discretionary actions.
- This comment suggests that the maps provided in the proposed C56-2 Uptown CPU incorrectly depict alternative routes on Spruce Street. The proposed Uptown CPU maps appear to be correct. West Spruce Street does connect to India Street. Access to all the other streets mentioned in the comment letter is solely from West Spruce/India Street. Because of the map's scale, the gap between West Spruce Avenue and Horton Avenue is difficult to see, but there is a gap in the figure and it was considered in the traffic analysis. However, the map in the Mobility Study does incorrectly show that a connection between West Spruce Avenue and Horton Avenue that will be corrected. While there was an error in the Mobility Study map, the model used in preparation of the analysis of potential impacts of the proposed Uptown CPU for traffic circulation did not include any connections with West Spruce Avenue that would provide additional ingress/egress to West Spruce Avenue other than India Street. Furthermore, the referenced Mobility Study Improvements (U17A and U17B) would be inconsistent with the proposed Uptown CPU polices and thus, would not be implemented due to infeasibility.
- C56-3 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

Sassafras Street and Redwood Street that we object to and believe will be environmentally detrimental to the residents who must use Spruce for ingress and egress.

Table S-1 Summary of Significant Environmental Impacts notes: India Street: Sassafras Street to Redwood Street (Impact 6.3-19)

Mitigation: TRANS 6.3-19: India Street (İmpact 6.3-19), b. Sassafras Street to Redwood Street: Widen the roadway to a 3-lane one-way collector. Comment: We object to any widening of India Street and to the inference that the two-way collector status between Sassafras and Redwood will be eliminated. The two-way collector is essential to residents of Spruce (and Redwood). If eliminated, Spruce residents will be required to travel to Laurel on Kettner and return up India for access to their homes. This will unnecessarily increase traffic on India and add additional fuel use and emissions using the longer route.

PAGE 6.3-10, India Street between Sassafras Street and Redwood Street (LOS E). Comment: The Level of Service method is outdated and will be outlawed in 2019. The Governor's Office of Planning and Research (OPR) states that Vehicle Miles Traveled is a better measure of environmental impact than LOS and is the most appropriate measure to replace LOS. It is legally permissible to use VMT now and not use LOS. The average daily traffic volumes should be measured against VMT.

If LOS is used, the traffic on India Street between Sassafras Street and Redwood Street must be measured again since the opening of the Rental Car Center. Traffic data indicates an increase of approximately 3000 vehicles daily on this segment since the RCC opened.

Table 6.3-7 Build-out Summary of Intersection Analysis – Uptown #20 & 21. Comment: If this analysis was done before the RCC was opened it is highly inaccurate. The SDIA RCC has added a substantial number of new cars – daily- to the intersections of Sassafras at Kettner and India causing increased traffic delays and regularly blocked intersections. This Analysis should be redone.

Table 6.3-8 Buildout Summary of Roadway Segment Analysis: Uptown - India St: Comment: Same as above (6.3-7) but the India street Buildout CPU figures demonstrate that adding an additional traffic lane will not solve the LOS problem and only make life much worse for local residents. Rerouting traffic from the Airport and RCC to Pacific Highway to Washington is the only short term solution and new airport specific on/off ramps connected to 1-5 is the only long term solution.

Impact 6.3-19, page 6.3-38: "The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of India Street from Glenwood Drive to Redwood Street."

Comment: We agree that the changes will have a traffic impact that is detrimental for the residents living off India from Glenwood Drive to Redwood Street and oppose any changes widening India St. Rerouting traffic from the Airport and RCC to Pacific Highway to Washington is the only short term solution and new airport specific on/off ramps connected to I-5 is the only long term solution.

2

6,3.5.2 Segments and 13.2.1.2 Roadways Segments: While the following intersection mitigation measures would reduce potentially significant impacts, only TRANS 6,3-7d., TRANS 6,3-24a., and TRANS 6,3-27 are proposed as part of the proposed Uptown CPU and associated discretionary actions.

Comment: We support the exclusion of Trans/Impact 6.3-19 on page 6.3-50 and again at the Uptown Community Plan Update PEIR pages 11-3 / 4, with a request that this be clearly designated as not an option or proposal under consideration.

In Appendix C – Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions).

Table 13 Summary of Improvement Evaluation Uptown: This references India St between Redwood and Sassafras as "one way Collector" and removal of parking and buildings.

Comment: We object to any widening of India Street especially with such extreme damage to the neighborhood businesses, and to the inference that the two-way collector status between Sassafras and Redwood will be eliminated. The two-way collector is essential to residents of Spruce (and Redwood). If eliminated, Spruce residents will be required to travel to Laurel on Kettner and return up India for access to their homes. This will unnecessarily increase traffic on India and add additional fuel use and emissions using the longer route. We recognize that it says the Consultant does not recommend this change but that is not binding on the City and the road changes are of considerable concern. We want our objection to this possible change clearly stated.

Figure 64 of Appendix C: Mobility Study (Uptown, North Park & Golden Hill Community Plan Update Mobility Study for Future (2035) Conditions).

Comment: We object to any widening of India Street especially with such extreme damage to the neighborhood businesses. While recognizing that the Consultant says "These changes in Geometry are not recommended" that recommendation is not binding on the City and the road changes are of considerable concern. We want our objection to this possible change clearly stated.

Thank you

Speedy Rice & Judy Clarke 1020 West Upas St San Diego, CA 92103 speedyrice@gmail.com

PROJECT NAME: Uptown Community Plan Update PROJECT No. 21002568 / SCH No. Pending COMMUNITY AREA: Uptown COUNCIL DISTRICT: 3"

	From: Daniel <driddell1@cox.net> Sent: Monday, August 03, 2016 4:42 PM To: PLN_PlanningCEQA Subject: Uptown Community Plan Update, project no. 21002568 To: City of San Diego Planning Department. Subject: Uptown Community Plan Update, project no. 21002568. Date: August 8, 2016 Date: August 8, 2016</driddell1@cox.net>		
C57-1	I am writing to urge changes to the Uptown Community Plan Update, project no. 21002568. Specifically, I am opposed to the following recommendations made regarding India Street in the Uptown, North Park and Golden Hill Plan Updates Traffic Impact Study June 2015. 1. The removal of the southbound lane on India Street from Sassafras Street to Redwood Street. 2. Changing this portion of India Street from a three Iane collector (two way) to a three Iane collector (one-way). 3. The removal of any current parking spaces, sidewalks or building fronts located on India Street between Laurel	 7-1 Comment noted. Please refer to India Street Mitigation Measures these response to comments (page 7-2 This comment references the R 	the Master Response Regarding included in the introduction to es RTC-4 through RTC-6). ental Car Center and suggests
C57-2 C57-3	Street and Washington Street. As a resident of West Redwood Street my safety, and the safety of my neighbors and visitors, has been severely impacted by designating india Street (a residential/mixed use street) as the ingress and egress for the new Rental Car Center located at the San Diego International Airport. The increased risk to life and limb for the residents of the neighborhood served by India Street is of shocking concern to me. The logistical barriers to traffic flow caused by the I-5 overpass, Kettner Street stoplight (at Sassafras) and the train/trolley tracks on Sassafras Street will not be abated by any changes to India Street. Sincerely, Daniel Riddell 1042 West Redwood Street San Diego, California	increased traffic on India Street I The Draft PEIR for the proposed document. The goals stated in Element are to create "safe, walka pedestrian connections and im comfortable pedestrian experience Element Policy MO-4.9 which we traffic calming measures where ap quality of service, and increase w and Mobility Element Policy MO parking on all streets in order enhance pedestrian safety and occurs these policies would be construction improvements.	busier and unsafe for residents. Uptown CPU is a planning-level the proposed CPU's Mobility ble neighborhoods, which utilize proved sidewalks to create a ce". The City proposes Mobility buld implement road diets and opropriate to improve safety and valking and bicycling in Uptown, -7.13 which supports on-street to support adjacent uses and activity. As future development e implemented through future
		7-3 Comment noted. Please refer to India Street Mitigation Measures these response to comments (page	the Master Response Regarding included in the introduction to es RTC-4 through RTC-6).

Letter C58

Robbie Robero ≺robbierobero@gmail.com≻ Friday August 05, 2016 6:58 PM PLN_PlanningCEQA Uptown Community Plan Update - 21002568

To whom it may concern:

From:

Sent:

Subject:

To:

C58-1 I would like to express some concerns about this sidewalk removal project that has been proposed. I have worked on India st and use it daily since I live very close. I strongly believe making India st a 3 lane highway is a safety hazard as well as a huge inconvenience for so many residents that live in this area. The sidewalk that runs along India is my only path to the all the food and necessities further down toward Rubicon. I don't own a car and rely on that pathway to get me to Washington blvd safely. This project will effect so many businesses and residents on a daily basis and won't do much in the end to solve any real problems. My biggest concern is safety and I've already seen 2 terrible accidents. I watched someone get crushed by a speeding car exiting the 5 trying to turn onto India first hand and I don't believe making this street wider is gonna help that issue! People fly off the freeway onto India and that's dangerous to all the residents turning off their streets. Furthermore, cars are constantly slowing down to pull over or coming to abrupt stops trying to find an address. As a result of this project, they won't even have a shoulder to safely pull over to, with all these speeding cars I mentioned earlier behind them. I see trucks use those shoulders to deliver goods all the time, what do you say to all those business owners that would be affected by a lengthy and costly construction project like this? I believe if you listen to the people that use this street frequently you will find my point of views valid. I feel this is a huge mishandling of our city tax dollars that could be used for much more important projects. Please consider this letter in your decision.

C58-1 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

The goals stated in the proposed CPU's Mobility Element are to create "safe, walkable neighborhoods, which utilize pedestrian connections and improved sidewalks to create a comfortable pedestrian experience". The City proposes Mobility Element Policy MO-4.9 which would implement road diets and traffic calming measures where appropriate to improve safety and quality of service, and increase walking and bicycling in Uptown, and Mobility Element Policy MO-7.13 which supports on-street parking on all streets in order to support adjacent uses and enhance pedestrian safety and activity. As future development occurs these policies would be implemented through future construction improvements.

RESPONSE

ATTACHMENT 7

	August 8, 2015		
	Jeff Murphy, Planning Director Kurtis Steinert, Senior Environmental Planner City of San Diego Transmitted via e-mail		
	Re: Uptown Community Plan Update and Draft EIR Recommendations		
C59-1	We, the undersigned, are Uptown community members and professionals in the built environment. We are architects, engineers, and city planners who have all made the decision to live in Uptown with our loved ones. We are devoted to making our great community even better, which is why we all individually decided to give back with our knowledge and expertise by running for and being elected by our peers to the Uptown Planners Community Planning Group.	C59-1	Comment noted. The City appreciates your participation in the public review process.
C59-2	We support the Draft Uptown Community Plan Update (CPU) and the Draft Environmental Impact Review (EIR) released in June 2016, as reflected in our votes as Board Members at the July 19 th Special Uptown Planners meeting. We voted against the Uptown Planners recommendations to support the Density Redistribution Alternative (DRA) from the Draft EIR and to reject the City's Draft Uptown CPU.	C59-2	Comment noted.
C59-3	We believe that a thriving city needs to provide housing for all, and to ensure that Uptown residents and businesses thrive means allowing more market rate and affordable homes to be built throughout the community, especially within Transit Priority Areas. We would like to see more housing affordability and parks, improved infrastructure and transit, but decreasing development potential does not move Uptown closer to achieving those important reals.	C59-3	Comment noted.
C59-4	The Draft CPU downzones Uptown by 1,900 homes from the development potential allowed in the current 1988 Community Plan. No alternatives in the Draft EIR analyzed increasing development potential from the 1988 Community Plan. Yet, between the Draft CPU and the Draft EIR alternatives, the Draft CPU is superior urban planning and therefore we as community members and professionals in the built environment support the City's proposal.	C59-4	Comment noted. The No Project Alternative as analyzed in Chapter 10, Alternatives, of the PEIR would include higher overall residential densities (and therefore higher development potential) than the proposed Uptown CPU.
	Conclusion	650 F	
C59-5	The Draft CPU incorporates years of community input from a wide array of community members and as a result, offers the best vision of the future of Uptown. It distributes density throughout the many unique urban core villages that we are fortunate to have in our community, ensuring growth is near the many transit lines in Uptown. It also focuses the most intense growth along 4 th , 5 th , and 6 th Avenues through Hillcrest and Bankers Hill where the \$200 million Uptown Bikeways investment will install protected cycle tracks connecting our community to Downtown. While we believe that more can be done to make Uptown the best it can be, specifically increasing the housing supply, we support the Draft CPU and Draft EIR.	C59-5	Comment noted.
	Thank you,		
	1 of 2		
		I	

RESPONSE

August 8, 2016

All.an

Michael Brennan Principal Landscape Architect Carson Douglas Landscape Architecture Hillcrest

D_C. gll

Dana C. Hook, PE Vice President CDM Smith Bankers Hill/Park West

Soheil Nakhshab, PE Principal, CEO Nakhshab Development & Design, Inc. Bankers Hill/Mission Hills

Maya Rosas Associate Land Use Consultant Atlantis Group Hillcrest/Mission Hills

2 of 2

RESF

ATTACHMENT 7

ATTACHMENT 7

RESPONSE

Letter C60

From:	Emily Scheerer <emily@sdhomes.com></emily@sdhomes.com>
Sent:	Tuesday, August 02, 2016 12:27 PM
To:	PLN PlanningCEQA
Subject:	Uptown Community Plan Update Project No. 21002568/ SCH No. Pending
Subject.	optown community Fian opdate Project No. 21002000 SOFTNO. Pen

To Whom It May Concern,

C60-1 I am writing on behalf of my company, SDHOMES, in regards to the proposed changes being recommended for India Street. We run our business our of a commercial building located at 3351 India St. San Diego CA 92103. We also rent residential units to tenants located at 3351A, 3351B, 3353 and 3355 India St. San Diego CA 92103. If the city chooses to adopt the environmental impact report that will eliminate parking on India, cut back sidewalks and infringe upon our commercial business' property line, it will be a great disservice. There is very limited parking along India St. that is permitted for over 2 hours as it is. Our employees move their cars several times a day to avoid ticketing in 2 hour parking. The surrounding residents are already protective and irritated by the commercial businesses on India using their side streets to park, they suffer from limited off street parking as well. Our driveway is designated for residential use and no customers are able to easily access our business as the street stands today. It would be impossible to be a successful business if the parking is reduced, let alone eliminated. This action will hurt small businesses, residences and threaten the viability of our entire business. We chose to work on India Street because of the great freeway access and signage potential. We have been impressed with the improvement the area has shown and have been happy to contribute to the development of Mid-Town. It is with great sincerity that we humbly ask the city to take careful consideration when voting on whether or not to expand India Street into the fronts of our businesses.

_

--

Warm Regards,

Emily Scheerer Project Manager Notary Public Comm. #2014146

3351 India St. San Diego, CA 92103 P: <u>619-291-2229</u> F: 866-223-5528 emily@sdhomes.com www.SDHOMES.com BRE #01940997

• The area finding during the number requirements that is confidential index preparaty. Our recall is not invested invested in the reconstruction of the interview of the int

C60-1 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

ATTACHMENT 7

	From: Sent: To: Cc: Subject:	Douglas Scott <plumspot1@gmail.com> Sunday, June 19, 2016 6:42 PM PLN_PlanningCEQA UptownPlanners@yahoo.com Hillcrest LGBT historic district</plumspot1@gmail.com>	61		
C61-1	To whom it may concern. I urge you to restore the I Plan. Doug Scott Hillcrest Property Owner 619.550.9575	Hillcrest LGBT historic district proposed in the June 2015 draft of the Uptown Com	nmunity C6	61-1 (Comment noted. The City appreciates your participation in the public review process. This comment requests the inclusion of a Hillcrest LGBT historic district in the proposed Uptown CPU. The City initiated preparation of a San Diego LGBTQ Context Statement n October 2015, after completion of the Golden Hill and North Park Context Statements and Surveys. The San Diego LGBTQ Context Statement will not be finalized until September 30, 2016. The San Diego LGBTQ Historic Context Statement will identify the chemes significant to the LGBTQ community throughout San Diego. Once finalized, the San Diego LGBTQ Historic Context Statement will be used to assist in the identification of potential individually significant resources, both through historic designation nominations and potential historic resource reviews associated with permit applications. In addition, Policy EP-1.7 of the proposed CPU promotes the LGBTQ historic heart of Hillcrest's Entertainment District.

- C62-1 Letter C62 From: Verena Seisun, Hillcrest Resident To: Leo Wilson, Chair, Uptown Planners Marlon Pangilinan, Senior Planner Jeff Murphy, Planning Director, City of San Diego Kurtis Steinert, Sr. Environmental Planner. Date: 7/28/16 Re Uptown Community Plan Update and Draft EIR C62-2 As a resident, I would like to address the following items of concern C62-1 1. Traffic: There are no plans for specially designated lanes for public transportation! If we want to entice the residents to use public transportation, there should be a distinct advantage over private cars. Being stuck in a bus in the same lane as all the cars will not make it attractive for people to abandon their own vehicles. C62-2 a. Why does the Mobility Element for Uptown ignore the need for C62-3 designated bus lanes? C62-3 b. A population increase of 19-20,000 would spell chaos and would significantly diminish the quality of life for all residents. Why didn't the city consider an alternative which matches the capacity of the transportation grid, with perhaps one-half that much increase? C62-4 c. Since the streets are already impacted by traffic, why doesn't the proposed Community Plan contain realistic measures for mitigating additional traffic impacts? C62-5 d. The draft EIR projects 50% of all roadway segments to be impacted by worsening traffic. Does the city view that as acceptable? C62-4 C62-6 2. Height limits: Mid-rise is ideal. Hillcrest business leaders refer to Little Italy as a model for redevelopment. But Little Italy has mostly 1-2 story buildings on their main commercial street, and 4-6 story buildings on other streets. Couldn't Hillcrest accommodate a substantial amount of growth if it retained the current 65 ft height limit? Verena Seisun
 - C62-1 Comment noted. The proposed Uptown CPU and associated discretionary actions would support implementation of local transit improvements by providing policies that support prioritizing the transit system and improving efficiency of transit services. For example, a number of transit-focused Mobility Element Policies are included in the proposed Uptown CPU that would support efforts to develop planned transit facilities. The intent of the proposed Uptown CPU is not to identify specific transit improvements for implementation.
 - C62-2 Comment noted. This comment does not identify an inadequacy in the PEIR. Designated bus lanes are not specifically identified in the Uptown CPU; however, the CPU includes policies promoting coordination with MTS on transit improvements, which may include dedicated bus lanes where feasible.
 - C62-3 Comment noted. This comment does not identify an inadequacy in the PEIR. The proposed Uptown CPU is intended to guide future development to serve the needs of the community and allow for orderly growth. The Lower-Density Alternative described and analyzed in Chapter 10, Alternatives, of the PEIR includes lower residential densities, and therefore a lower population at build-out, than the proposed Uptown CPU. Additionally, many of the failing roadway segments, intersections and freeways are currently failing in the existing condition.
 - C62-4 Comment noted. This comment does not identify an inadequacy in the PEIR. The Mobility Element of the proposed Uptown CPU contains multiple policies aimed at reducing traffic congestion and improving circulation within the community and between adjacent communities. Of the mitigation measures identified to reduce significant impacts to transportation and circulation, all feasible measures are ultimately included in the Mitigation Monitoring and

C62-4 ((cont.) Reporting Program included as an attachment to the Staff Report. As discussed in Section 6.3, Transportation and Circulation, of the PEIR, many measures identified to reduce impacts are inconsistent with the mobility goals and policies of the proposed Uptown CPU. Refer to the Findings included as an attachment to the Staff Report for details on the feasibility of mitigation measures.
C62-5	Comment noted. Section 6.3, Transportation and Circulation, of the PEIR disclosed a significant and unavoidable impact to transportation and circulation, particularly to intersections, roadway segments, and freeway segments. Refer to the attachment to the Staff Report for the proposed Statement of Overriding Considerations that explain why the City would adopt the proposed CPU despite significant and unavoidable impacts.
C62-6	Comment noted. This comment does not identify an inadequacy in the PEIR, rather is a comment about the proposed Uptown CPU.

Letter C63 Uptown Community Plan Update from the City of San Diego (6/2016) - EIR Component Comment C63-1 C63-1 I am a Bankers Hill resident, having lived in the Uptown District (Mission Hills/Hillcrest/BH) for 30 years. The Mobility Study in the latest City of San Diego offering for our Uptown Community Plan's EIR is my main concern in that we are already impacted with the current building "boom" in Bankers Hill. We have 550 units recently completed or nearing completion in a 3 X 8 block area (4th to 6th and Maple to Upas Sts.). This means at least an additional 1,200 cars + friends, family and service vehicles will be added to the neighborhood in the very near future - with an unknown number of additional properties ready for demolition (4th and Maple) and several sites for sale as well. C63-2 C63-2 Since I've lived in Bankers Hill the past five years, any available daytime street parking is FULL - from the west side of the Balboa Park Bridge on 6th Ave. all the way down to Interstate 5. Workers in doctors' office, hair salons, restaurants, 7-11's are unable to regularly park to get to work - as the automobile still dominates the Uptown neighborhoods as in ALL areas of 5an Diego. This current Uptown Community Plan (6/2016) as offered to Uptown residents will find our neighborhood being impacted with even more traffic congestion - probably like "December Nights" traffic every day of the year without even a mention of a City-built parking structure to accommodate the additional vehicles in Bankers Hill in EIR mitigation. C63-3 The EIR's meager/minor mobility mitigation for traffic (while increasing the density of Uptown from its current 36,750 residents to 55,700 over the next 20 years) will do very little to ease the current traffic C63-3 gridlock. The SD Planning Department made decisions over the years to model San Diego over Los Angeles, i.e., expanding housing onto less expensive suburban land for increasing developers' profits. Hence, we currently have a "reverse commute" that the EIR is unable to address or come close to resolving for residents. [Only 4% of downtown SD residents live/work downtown (per the U.T., June 2016); to be able to afford to live in Downtown/Uptown, most residents go North to UTC, Sorrento Valley, Carmel Heights or out to Poway for the good-paying jobs (per the U.S., July 2016). The downtown jobs in the courts, some legal, accounting, investments are taken by residents living in the suburbs OR for hotels, restaurants, minor retail positions- by our "contingent Tijuana workforce". Hence, our freeways are "no-ways" between 7.5 to 8 hours/day.] C63-4 The 30+ number of Roadway Segments + 6 Freeway Segments alterations/mitigations mentioned in the EIR consist mostly of re-striping street pavements on the Uptown streets that were built in the 1940's/1950's when we had 1/20th of the population. As mentioned in the EIR, even these minor mitigations to accommodate the increased number of residents "cannot be guaranteed because funding. sources are not guaranteed nor is the timing of their implementation" Regarding Freeway Segment policies. mitigation: "mitigation measures identified to address impacts to freeway segments would not be implemented due to a lack of guaranteed funding sources and the City's lack of jurisdiction over freeway facilities. Similarly, the timing and responsibility for mitigation monitoring, enforcement and reporting C63-4 are currently unknown, although it is assumed that both the City of San Diego and Caltrans would be involved in mitigation monitoring, enforcement and reporting,....impacts to intersections and roadway segments would remain significant and unavoidable." In summary, mitigation suggestions may help with mobility around the neighborhoods but there is currently no funding for even these minor Uptown IFS. improvements. C63-5 And finally, due to the "reverse commute" situation in San Diego, bike-ride-walk "looks good from 40,000 feet" but in reality, there are hardly any cyclists on 5th Avenue going through Bankers Hill (other C63-5 than mostly recreational riders on weekends), as they cannot get to any functional transit stop that will

- C63-1 Comment noted. This comment does not identify an inadequacy of the PEIR. The proposed Uptown CPU would not induce or inhibit growth; rather, community plans are intended to guide inevitable growth in an orderly manner.
- C63-2 Comment noted. Section 6.3, Transportation and Circulation, of the PEIR disclosed a significant and unavoidable impact to intersections, roadway segments, and freeway segments in the Uptown community. Additionally, parking is not an environmental issue that requires evaluation under the California Environmental Quality Act. However, the proposed Uptown CPU does include policies related to parking management.
- C63-3 Comment noted. This comment does not identify an inadequacy of the PEIR. While even with implementation of the feasible mitigation measures identified in the PEIR, significant and unavoidable transportation impacts would result from implementation of the proposed Uptown CPU and associated discretionary actions. The proposed Uptown CPU does includes multiple policies in its Mobility Element aimed at reducing traffic congestion through increased pedestrian, bicycle, and transit use, an improved street and freeway system, and transportation demand management policies.
- C63-4 Comment noted. This comment restates information contained within the PEIR but does not identify an inadequacy of the PEIR. Funding for specific improvements is identified in the proposed Uptown IFS.
- 63-5 Comment noted. This comment does not identify an inadequacy of the PEIR.

ATTACHMENT 7

LETTER

RESPONSE

take them North or Northeast to their real jobs in a timely fashion. The bus lines' riders going through Uptown are about 95% poorer County residents (going to low-end-paying retail jobs in the neighborhood), the elderly, the disabled, a few students and tourists.

- C63-6 Please give the EIR more attention and at least money for street-widening in all the aforementioned roadway/freeway segments, as adding an additional 20,000 residents to our neighborhoods will make living here highly undesirable not to even mention the unaffordability of Uptown condos here already.
- C63-7 Also, the Density Redistribution Alternative to the City's 6/2016 CPU is favored by all Uptown residents who are aware of the City's current offering for our neighborhood's development over the next 20 years.

Thank you for your consideration,

Donna Shanske 475 Redwood St., Unit 310 San Diego, CA 92103

- C63-6 Comment noted. Funding for specific improvements is identified in the proposed Uptown IFS.
- C63-7 Comment noted.

C64-1	From: Sent: To: Subject: As a resident of Hillcres quietly removed from th neighborhood, and I car Thank you, Christopher M. Smith	Christopher Smith Adair correstphreimth@gmail.com> Thursdey, June 23, 2016 6:45 AM PLN_PlanningCEQA Uptown Corrimunity Plan Update PROJECT No. 3806(11/SCH No. Pending st, I am deeply concerned and surprised to learn that the historical district plan has been he neighborhood's community plan. I value the unique character and history of my n only hope this was a genuine oversight that will be quickly corrected.	C64-1	Comment noted. This comment does not identify an inadequacy of the PEIR. The comment notes concern with the removal of the historical district plan for Hillcrest from the proposed CPU. While Hillcrest was not identified as a potential historic district during the reconnaissance survey for Uptown, it was identified as a potential
				historic district by the community. Both survey-identified and community-identified resources are discussed in the Historic Preservation Element (HPE) of the proposed Uptown CPU, and the Hillcrest potential historic district is included in the City's work program for processing potential historic districts in the coming years. The HPE also includes numerous policies that address the protection of the community's historic resources. Figure 10-2 and Figure 10-5 of the proposed CPU depict the locations of the registered historic districts and identified potential historic districts, respectively, within the Uptown community. The amended Historical Resources Regulations would provide supplemental development regulations that would protect contributing resources within potential historic districts identified by City Council until they are formally evaluated for designation. Finally, Mitigation Measure HIST 6.7-1 would help to avoid or mitigate potential impacts to potential historically significant structures resulting from development or redevelopment.

ATTACHMENT 7

	Letter C65		
	From: Sarah Spooner <sarahrspooner@gmail.com> Sent: Tuesday, August 02, 2016 9:34 AM To: PLN_PlanningCEQA Subject: India Street Proposal</sarahrspooner@gmail.com>		
	To whom it may concern,		
C65-1	My name is Sarah Spooner and I am a resident of Middletown and currently reside on Columbia Street near Sassafras.	C65-1	Introductory comment noted. The City appreciates your participation in the public review process.
C65-2	Suggested Project No. 21002568 to turn India Street into a three lane highway greatly impacts our neighborhood adversely. This is especially true when there are other, easier solutions to problems related to the new traffic flow in our neighborhood since the new Rental Car Center has opened.	C65-2	Comment noted. Please refer to the Master Response Regarding
C65-3	Why isn't Pacific Highway being utilized? It's equipped for heavy traffic. Have the rental car exit turn left onto Pacific Highway, then right on Washington, then left on San Diego Ave to access the 5 Freeway. Why make changes to a local neighborhood to accommodate rental car traffic, when instead already established roads could be utilized.	India Street Mitigation Measures included these response to comments (pages RTC-4 t	India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
C65-4	Our community is united against this decision, which only benefits the car rental business and hurts our local businesses.	C65-3	This comment references alternate routes to airport traffic, but does not suggest an inadequacy in the analysis of the PEIR
	Thank you for taking the time to read this.		
	Sincerely,	C65-4	This comment does not suggest an inadequacy in the analysis of
	Sarah Spooner		the PEIR; therefore, a detailed response is not required.

ATTACHMENT 7

RESPONSE

	From: Peter StClair <phstc2@gmail.com> Sent: Monday, August 09, 2016 12:23 PM To: PLN PlanningCEQA Subject: Uptown CP Update Project 21002568 Uptown District 3</phstc2@gmail.com>		
C66-1	Please address my comments on the Uptown CP Update PEIR I. Overarching comments: I. The conclusions are subjective, There is very little data analyzed in this PEIR. It largely restates the CPU, GP and City codes. I cannot determine how the how the writers came to their conclusions, specifically that the Density Redistribution option is the Environmentally Superior option. Is for the analyzed in this CPU, GP and City codes. I cannot determine how the writers came to their conclusions, specifically that the Density Redistribution option is the Environmentally Superior option.	C66-1	Comment noted. As discussed in Section 10.6 of the PEIR, the Density Redistribution Alternative was identified as the environmentally superior alternative because it would reduce impacts related to transportation and circulation and air quality. Discussions of the Density Redistribution Alternative's potential impacts to transportation and circulation and air quality are provided in Section 10.4.2, Analysis of Density Redistribution Alternative.
00-2	achievement of CAP goals. Since CAP and GHG are, from a legal perspective, (see California Jegislation, Executive Orders, California Supreme Court and Appellate Court decisions upholding the primacy of these measurements in evaluating General Plan and specific project contributions) the most important aspects of a project's impacts, the Density Redistribution option cannot, on its face, be the superior option.	C66-2	When identifying an environmentally superior alternative, all impacts that would be reduced by the alternative must be considered.
C66-3	It is amusing that after 10 years of community battles and millions of dollars in expense, the CPU and the alternatives considered in the PEIR are all remarkably close in terms of final buildont foreseen for Uptown.	C66-3	Comment noted. This comment does not suggest an inadequacy of the PEIR.
00-4	The City has already shown a willingness to adopt transportation options like the 5th Ave bake lane without consulting Uptown Planners and without the need for a CPU. So the proposed benefit of the CPUits transportation/transit and design review, can already be implemented. The only real benefit of a new plan is to get rid of the Interim Height Ordinance, which has just about killed off Hillerest as a viable economic force in city and regional retailing and affordable housing.	C66-4	Comment noted. This comment does not suggest an inadequacy of the PEIR.
C66-5	2. The PEIR accepts on face the conclusion in the CPU that population in the plan area will increase from 36,700 today to 55,700 at "build out". The PEIR assumes buildout occurs in 2035. That would be 19 years from the date the final Draft Uptown Plan was circulated. Based on the proposed land uses and zoning contained in the CPU, a projection is made as to total dwelling units at build out. (See Table 3-9 page 3-31)	C66-5	Comment noted. This comment does not suggest an inadequacy of the PEIR.
C66-6 C66-7 C66-8	These projections assume the number of persons/DU will increase from 1.59 today to 1.70 in the future. There is no justification given for accepting the assumption. In fact, occupancy/DU has been decreasing for a long time, especially in urban areas like Uptown. The plan calls for more compact development. As SFR is replaced with multifamily, unit sizes will decrease. Housing will become more costly, not less costly. The PEIR notes there will continue to be an imbalance, with demand for housing exceeding supply. DIF will add to cost.	C66-6	This comment does not suggest an inadequacy of the PEIR. The PEIR makes no assumptions regarding what the applied persons per household rate should be at community plan build-out. SANDAG projections indicate there is growth and a need for additional housing.
	1)	C66-7	Comment noted. This comment does not suggest an inadequacy of the PEIR. Refer to response to comment C66-6.
		C66-8	Comment noted. This comment does not suggest an inadequacy of the PEIR.

- C66-9 Therefore, it is highly likely the Uptown plan will not accommodate population growth. Rather than reduce GHG and help meet the goals of CAP by providing relatively dense, more affordable urban infill housing, this plan seems designed to produce very high cost, unaffordable housing. (See 5.5.3.3 Page 5-22) C66-10 The actual difference in housing production among the alternatives considered in the PEIR is miniscule. And no alternative, according to the PEIR, will meet demand. C66-11 I believe the provision for new housing is less than anticipated in the Regional Plan, which set goals for transit and which supported state rules on GHG and the City's CAP. Various factions appearing at Uptown Planners meetings said the city had previously committed to "density transfer" out of Uptown to other areas (downtown?) in the General Plan. But the Regional Plan and CAP assume further densification and intensity of land use. Therefore, the Uptown Community Plan Update fails to contribute enough to city-wide and regional goals. C66-12 The PEIR should point that out. CPU does not allow enough housing. And none of it may be affordable. C66-13 Where housing and mixed uses is encouraged, FAR limitations, setback requirements from the ground floor up, and DIF will reduce actual production, or create housing totally unaffordable to most people in Uptown. It is impossible to see how Affordable Housing can be produced under the strictures of the CPU and DIP. C66-14 It does not allow enough flexibility in land use and circulation to attract new high paying jobs, specifically in the Medical Complex area, which continues to be hampered in its expansion and appeal to new R&D and technology firms by the LU elements in the CPU that insist on protecting existing and potentially eneroaching residential uses and which fail to remedy circulation problems. C66-15 It does not seem to do anything to encourage revitalization of the retail/entertainment core in Hillerest. C66-16 Indeed, many of the non-residential activities in the Uptown Community are regional in character. The Medical Complex. The entertainment and retail areas. The medical office buildings. There is no "tolerance" in the CPU for this integration of Uptown into the greater City of San Diego or our region. Rather than measure impacts on a city-wide or regional basis, they are measured corner by corner. Parochial. C66-17 Does the CPU contribute to the City's General Plan and CAP goals or not? Unanswered in this PEIR. C66-18 The PEIR tends to repeat vast amounts of the actual text of the CPU or the City General Plan without performing any independent analysis. C66-19 3. I cannot determine whether the Mitigations outlined in the PEIR are mandatory or potentially applicable to individual projects. In some cases, such as impacts on Historic Resources and Paleontology, existing local, state and federal law already provides for the mitigations that are recommended. Therefore, there is no reason to restate code or the law. C66-20 In other cases, such as road and circulation improvements, the writers seem to state that mitigations will occur on a project by project basis. C66-21 But the premise of the CPU is that once it is approved, most projects will be permitted as a right. Only certain projects whose scope, scale or specifics differ from what is allowed in the CPU and Citywide Zoning will go through any kind of discretionary review. 2
- C66-9 Comment noted. While the proposed Uptown CPU would reduce residential densities in some areas, it would increase residential densities in others. It strategically places higher residential densities and mixed-uses along major transit corridors. In addition, the Mobility Element of the proposed CPU includes numerous policies that promote pedestrian, bicycle, and transit use. Therefore, the proposed Uptown CPU furthers the General Plan's City of Villages Strategy and the Climate Action Plan (CAP).
 - C66-10 Section 15126.6 of the CEQA Guidelines requires an analysis of a reasonable range of alternatives that meet most of the basic project objectives. These objectives, outlined in Section 3.3, Project Objectives, of the PEIR, were tailored specific to the needs of the Uptown community and to implement the City of Villages Strategy and the City's CAP.
 - C66-11 The Regional Plan and CAP do not rely on densification and intensity of land use alone. As previously stated, while the proposed Uptown CPU would reduce residential density in some areas, it would also increase density in others. Lower residential densities in some areas are required to ensure that the bulk and scale of development maintain the existing neighborhood character as well as public views of canyons and open space. These areas are also generally less served by transit and mixed-uses. The proposed land uses locate the highest intensity uses along transit corridors where existing and future commercial, residential, and mixed-use development can support existing and planned transit investments in the community. Commercial and other employment-generating uses are also used strategically by the proposed Uptown CPU to encourage commercial uses along transit corridors. Therefore, the transit-oriented development pattern provided by the proposed Uptown CPU—and not an arbitrary densification of all land uses throughout the entire community—is consistent with the goals of both the General Plan's City of Villages Strategy, the CAP, and Regional Plan. See Section 6.1, Land Use, for a discussion on the proposed CPU's consistency with applicable regional and local planning documents.

C66-12	This comment indicates that the proposed Uptown CPU may not provide adequate affordable housing. However, Policy LU-1.1 of the proposed Uptown CPU aims to "provide a variety of land use types to accommodate both affordable and market rate housing and commercial opportunities." Policy LU23 of the proposed CPU also addresses the need for adequate housing for those with special needs, including low income. The proposed Uptown CPU plans for growth and would accommodate an increase in residential units in the CPU area compared to existing conditions. Additionally, all residential development would be subject to affordable housing regulations.
C66-13	Comment noted. This comment does not suggest an inadequacy of the PEIR.
C66-14	Comment noted. This comment does not suggest an inadequacy of the PEIR. This comment suggests concern with the proposed CPU's ability to create jobs. However, policies included in the proposed Uptown CPU address the need for continued job creation in the community. Specifically, LU-1.6 supports the expected employment growth in the health sector within the Uptown community. The proposed CPU also includes a land use pattern with an additional emphasis of retail and employment uses in order to balance the predominantly residential community.
C66-15	Comment noted. This comment does not suggest an inadequacy of the PEIR. This comment expresses concern with the proposed Uptown CPU's ability to encourage revitalization of the Hillcrest core. However, Policy LU-3.2 specifically addresses this by encouraging high intensity pedestrian-oriented commercial and mixed-use development in the Community Village – Hillcrest Core West area. Policy LU-3.3 also encourages active commercial business uses on the ground floor levels in this area. Policy UD-4.25 addresses the incorporation of architectural design features to highlight the Hillcrest Core and other gateway locations.
C66-16	Comment noted. This comment does not suggest an inadequacy of the PEIR. The proposed Uptown CPU is intended to further regional planning goals, such as those of the General Plan, with site-specific policies tailored to the community's needs.

C66-17	The proposed Uptown CPU's consistency with and contribution to the City's General Plan and CAP goals are discussed in Section 6.1.3 and Section 6.5.3 of the PEIR. In short, the proposed Uptown CPU would further the goals of the City of Villages strategy and the City's CAP.
C66-18	This comment suggests that the PEIR lacks any independent analysis from the CPU. The analysis presented in the PEIR is a program-level review of the physical changes in the environment that would result from adoption and implementation of the proposed CPU and other associated discretionary actions. As such, it would be impossible to conduct an environmental analysis of the CPU independent of the CPU itself.
C66-19	All mitigation measures recommended and included in the Mitigation Monitoring and Reporting Program will be legally required and enforceable by the City. This comment suggests that the Historical Resources and Paleontological Resources mitigation measures are already required by existing law. These mitigation measures go above and beyond existing regulatory requirements to ensure that impacts are avoided or minimized and provide a framework to ensure future development implements identified mitigation.
C66-20	At a program level of review, it is appropriate to include mitigation measures that would be implemented by future individual development projects. For traffic, the mitigation measures are identified to meet the CEQA requirements for identifying mitigation, but the measures would not be feasible to implement at a project level as discussed in the Candidate Findings included as an attachment to the staff report.
C66-21	Some projects implemented in accordance with the proposed Uptown CPU would be subject to ministerial review while others would be subject to discretionary review pursuant to the City's Land Development Code.

C66-22	Therefore, the PEIR must rely on imposition of Development Impact Fees on every new project in order to say impacts are mitigated,	C66-22	This comment suggests that Development Impact Fees are the only tool to ensure mitigation is completed. However, the PEIR includes
C66-23	However, the DIF Study is only that: a study. It is flawed and its cost estimates are potentially very inaccurate.		other measures that would be implemented at the project level. For example, Mitigation Measure HIST 6.7-1 requires project level
C66-24	By far the largest component of DIF is a \$250,000,000 fund to acquire parkland.		evaluation and potential mitigation or avoidance of potentially
C66-25	So DIF are set at \$12,000 per DU for residential projects.		historic resources prior to the City's issuance of a development
C66-26	It is unclear whether DIF payments are lumped in a single DIF pool for Uptown, or whether the fees are broken down into their separate components and an individual fund created for each component.	666.00	
C66-27	If the latter is the case, then at the outset of permitting under the new CPU, there will be inadequate funds available for road improvements that are somehow linked to each project. Since most projects are unlikely to go through discretionary review, there will be no objective basis to assess road and other impacts.	C66-23	the PEIR.
C66-28	If the former is the case, improvements can be phased to meet needs (based on LOS studies, etc.) and sufficient funding may be available. But, this means the massive amount of money needed for park land acquisition may not be available until the end of the time frame considered in the PEIR (year 2035 according to the PEIR)	C66-24	Comment noted.
C66-29	Therefore, I cannot tell whether the proposed mitigations (which apply similarly to each alternative as I read the PEIR) are in fact coincident with need (except for park land).	C66-25	Comment noted.
C66-30	If not, are they indeed mitigations?	C66-26	Comment noted. Development Impact Fees (DIF) are collected within a single DIF fund for the Uptown community.
C66-31	It would be cynical to think that the massive amount of money deemed necessary to acquire park land was only put in the DIF in order to generate fees large enough to actually do the required (road and circulation) work.	C66-27	Comment noted. This comment does not suggest an inadequacy in
C66-32	In disregarding the potential affect of (high) DIF on the rate of new (residential) development, and the absolutely chilling affect on affordable housing, the PEIR makes subjective and unwarranted assumptions about the ability of the CPU to meet GIIG and CAP goals.		the PEIR.
	II. Specific comments:	C66-28	Comment noted. This comment does not suggest an inadequacy in the PEIR.
C66-33	 Table S-1 page S-10, Light and Glare. There is no reason to believe City will in fact assure that lighting does not create excessive brilliance and glare. 		
C66-34	The lights of the Decorative Street Lamp project in Mission Hills create a great deal of light trespass (too much light falling on building facades and up into the air and too much light cast on streets themselves).	C66-29	This comment is unclear as to which mitigation measure is referred to as being unclear on its "coincident with need." All mitigation
C66-35	This may be a result of improper contract administration.		measures proposed are intended to reduce potentially significant
C66-36	The approved plans for the project indicated the lumieres would be the same as used in Talmadge. In fact, the plans were taken directly from the Talmadge project and not even re-titled for Mission Hills. But, there is a tremendous difference in light trespass between the communities. Talmadge decorate lights seem to function as set forth in the plan. The Mission Hills lumieres are far brighter and east light in many more directions other		Uptown CPU. Refer to the Mitigation Monitoring and Reporting Program of the Final PEIR for more detail.
	than "down". City has tried to remedy this problem by installing individual shields on lumieres. A sad statement about the efficacy of our contracting process.	C66-30	See response to comment C66-29.
	à	C66-31	Comment noted. This comment does not suggest an inadequacy in the PEIR.
		C66-32	See response to comment C66-11 regarding the proposed Uptown CPU's consistency with the General Plan and CAP.

C66-33	As discussed in Issue 5 of Section 6.2.3, Impact Analysis, of the PEIR, development implemented in accordance with the proposed Uptown CPU may contribute to existing conditions of light and glare. However, the Urban Design Element of the proposed CPU addresses lighting in the community to ensure there is no unnecessary adverse light and glare, such as Policies UD-3.12, UD-3.13, and UD-3.15. In addition, development would be subject to the citywide lighting policies included in the Land Development Code.
C66-34	This comment references an existing condition and does not suggest an inadequacy of the analysis presented in the PEIR. Lighting policies within the proposed Uptown CPU Urban Design Element would support pedestrian-oriented street lighting with appropriate shielding and low heights to minimize light spillage.
C66-35	Comment noted.
C66-36	This comment references an existing condition and does not suggest an inadequacy of the analysis presented in the PEIR. Lighting policies within the proposed Uptown CPU Urban Design Element would support pedestrian-oriented street lighting with appropriate shielding and low heights to minimize light spillage.

RTC-531

C66-37 This comment suggests that the proposed Uptown CPU calls for C66-37 street lighting that casts light upward. On the contrary, lighting It also seems City has withdrawn from the Dark Skies program. The kind of lighting called for in the CPU (non-hooded) casts light upward. The light is of a spectrum that disrupts astronomers and wildlife (migrating policies within the proposed Uptown CPU Urban Design Element birds). The lumieres can be ordered with "shields" that prevent light being thrown upward. But at least in Mission Hills, that was not the case. would support pedestrian-oriented street lighting with appropriate shielding and low heights to minimize light spillage. Specifically, C66-38 2. Traffic mitigations should include two other options that provide vastly more congestion relief than proposed in the PEIR, including Trans 6.3-7: Policy UD-3.15 calls for street lighting to focus on illuminating the pedestrian zone, which would mean lighting would be cast C66-39 -- A bus pocket/turnout on the north side of Washington St. between 5th and 4th Avenues. This is currently a bank branch and before that was a fast food restaurant. downward ---Extension of Front St. through Florence School. Florence School can be relocated and may be closed. It C66-40 forms what should be the heart of West Hillcrest. Extending Front St. from the Medical Complex to Robinson C66-38 See responses to comments C66-39 and C66-40. will greatly relieve congestion on Washington St. and probably allow Front St. to be re-striped to two ways north of Washington. This also creates a "center of town" and enable many of the CPU's goals--walkability. bicycling, transit, parks and recreation centers. C66-39 Comment noted. Policies in the Mobility Element of the proposed C66-41 --Rapid re-sequencing of lights along Washington and University after emergency vehicles pass. CPU would support transit facility improvements. The specific 3. Page S-21. Noise. Does not seem to consider the increasing number of hybrid and electric cars and buses in mitigation measure would not be appropriate for a program-level C66-42 use or the decrease in auto traffic resulting from more walking, biking and transit use. analysis, but could be considered as individual developments are C66-43 4. Page S-26, HIST 6.7-1, 2, etc. There is no evidence given that impacts are not fully mitigated by existing proposed in the area. Additionally, the comment does not provide codes and laws. The rest of this section restates the CPU, provides no additional analysis and should be any support or evidence that the measure would in fact reduce removed. significant traffic impacts. C66-44 5. Page S-42 PALEO 6.10 describes North Park. That is not part of Uptown. 6. 1 am not sure changes in transportation such as ride sharing and potentially the driverless car (all of which C66-45 C66-40 This comment suggests mitigation that would require the closure should accelerate over the horizon of the PEIR--to the year 2035) are factored into these analyses. of an existing school and road extension. The specific mitigation C66-46 7. Page 6.5-11 and 6.5.3 Cumulative Impacts. 1 am not sure what the writers are trying to say. The City cannot measure would not be appropriate for a program-level analysis, quickly modify land use plans. The CPU itself is the manifest example of the inability to deal with change. This plan update has taken nearly 10 years and there is still a virtually 50/50 split in Uptown but could be considered as individual developments are proposed community members over its direction. Therefore to say we can be responsive is not borne out by facts. The in the area. Additionally, the comment does not provide any plan needs to get things right NOW, not later. (There is no feature for "Adaptive Management" in a Community Plan. It is a sorely lacking feature). support or evidence that the measure would in fact reduce C66-47 significant traffic impacts. 8. 6.7.1. Impacts on Historic Resources are speculative. We have codes and laws in place. 9. 6.11.4. Is LID in place in the City of San Diego? My sense from the Stormwater PEIR ffasco was City C66-48 treated LID as a "...goal rendered infeasible due to our clay soils and the built-out nature of most of our urban C66-41 Policy MO-5.1 of the proposed Uptown CPU calls for the utilization communities." of Intelligent Transportation System improvements, which include C66-49 Are stormwater BMP's inn place? Again, my sense was city staff and their consultants rejected the central ideas traffic signal coordination. Inclusion of the referenced mitigation of LID and Green Streets in favor of highly engineered (steel, concrete and electricity) treatment facilities. See measure would not be appropriate for a program-level analysis, plans for Maple Canyon erosion controls. but could be considered as individual developments are proposed C66-50 10, 6.12.1.2. This says it all! We need 155.96 acres of parkland. We have 18,21 acres. There is no feasible plan for bridging the gap. DIF says it will cost more than \$250,000,000. in the area. C66-42 Refer to Section 6.6, Noise, of the PEIR for a detailed discussion and analysis of the ambient noise impacts due to vehicular traffic, trolley service, and land use patterns associated with build-out of the proposed Uptown CPU. Additionally, a potential shift in noise from use of electric and hybrid vehicles would be speculative to assume in the analysis.

C6	66-43 6	Refer to Section 6.7, Historical Resources, for a detailed discussion and analysis of the potential impacts to historical resources, existing regulations, mitigation measures, and significance of potential impacts after mitigation.
C6	56-44 ⁻ I	The Final PEIR has been corrected to reference Uptown, and not North Park.
C6	56-45 	Impacts to transportation and circulation were assessed based on a description of future community build-out conditions for the Uptown CPU area compared to existing conditions. Benefits of ride sharing and self-driving cars on the long-term transportation setting of the entire community are difficult to quantify and, therefore, not included.
C6	56-46 i i i i i i i i i i i i i i i i i i i	This comment makes reference to the following statement: "If implementation of the proposed Uptown CPU and associated discretionary actions cumulatively with other CPUs would be inconsistent with the CAP or other plans/policies for the reduction of GHG, the City could amend land use plans to reflect more aggressive strategies for GHG reduction and to ensure consistency with the adopted CAP." This statement references the fact that City policies, plans, and codes will be evaluated as needed to ensure CAP greenhouse gas emissions targets are met as intended. This does not mean that the City's land use plans and policies that are currently being updated in accordance with the CAP, such as the proposed Uptown CPU, are not expected to meet greenhouse gas emissions targets. However, many circumstances play into the effectiveness of various plans and policies and if, in the future, it is determined that these targets are not being met as planned, then the City will need to update land use plans and policies to reflect more aggressive strategies to ensure the targets can be met in the future.
C6	56-47 (i i	Comment noted. The PEIR concluded a significant and unavoidable impact to potential historic districts because the degree of future impacts and applicability, feasibility, and success of future mitigation measures cannot be adequately known for each specific future project at this program level of analysis.
RESPONSE

- C66-48 Comment noted. This comment does not indicate an inadequacy in the PEIR.
- C66-49 This comment does not indicate an inadequacy in the PEIR. Storm water best management practices are required on a project basis.
- C66-50 Comment noted. The comment does not identify an inadequacy in the PEIR. The PEIR does identify that the community should be served by 155.96 acres of parkland based on General Plan standards for population-based parks. The Final PEIR has been corrected to state that 14.66 acres of population-based parkland currently exists in the community, not 18.21 acres. While there is a deficiency in population-based parks, this is not considered significant at the program level because implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of populationbased parks in the Uptown CPU area. Additionally, the Recreation Element of the proposed Uptown CPU includes numerous policies addressing the need for additional parkland and park equivalencies in the community.

ATTACHMENT 7

RESPONSE

C66-51	The imposition of (high) DIF has consequences. It will greatly impact Affordable Housing.	C66-51	C
C66-52	11. Chapter 8 Growth Inducement. The writers of the PEIR flatly state that Uptown's population will grow irrespective of the CPU.	C66-52	S
	Therefore, given the significant limitations on growth in the CPU (Height, FAR, setbacks, DIF) rents and prices will rise. This makes Uptown less affordable and less attractive to many potential employers. It reduces the likelihood CPU will achieve GHG and CAP goals, much less General Plan objectives.		r
C66-53	12. Project Alternatives.		
	In fact, the existing Adopted Plan, freed from the Interim Height Ordinance significantly increases the possibility of build out by 2035 thus reducing GHG and boosting the goals of the CAP. It concentrates higher density along transit corridors and strengthens employment (at higher wages than in retail/entertainment)	C66-53	C
C66-54	13, 10.2.2 (d) and (e) may be incorrect. With increased density cities normally experience less pollution and GHG emissions and greater use of transit. That is the whole point of our General Plan, City of Villages and the CAP.	00-54	v a
C66-55	14. 10.4.1 Density Redistribution Alternative. The PEIR states this option allows for the same development potential. I think that is incorrect. I cannot see it in the land use maps or Table 10-4. It appears to change zoning in the Medical Complex that eliminates mixed uses in favor of purely residential uses. Table 10-4 shows less population and fewer DU's under this alternative. Something is amiss.	C66-55	T s
C66-56	15. 10.6. Environmentally Superior Alternative. Not logical and without quantitative support. GHGs are not less than other alternatives and CAP is not as well-supported.		C
C66-57	16. 11.2 b Mitigation for Transportation and Circulation.	C66-56	S
	Are these mandatory mitigations? Suggested mitigations? The PEIR suggests their phasing and funding will be dependent on actual building permit applications with DIF applied. DIF is a study and contains many unchallenged assumptions.	C66-57	e A
	There should be a bus pocket on the north side of Washington St. between Fifth and Fourth Avenue.		Ν
	Front St. should be extended through Florence School, the school relocated and this land become the center of west Hillcrest, providing transit hub, park and recreation center.		N T
C66-58	17. 11.2.3 and 11.2.4. This restates codes and the law. Why is that necessary?		ii C
	Peter H. StClair 2341 Whitman Street San Diego CA 92103		د ۱
	619-260-1307	00-58	r
	5		

- C66-52 See response to comment C66-12 regarding the proposed Uptown CPU's guidance for the provision of affordable housing. See also response to comment C66-1 regarding the proposed Uptown CPU's consistency with the General Plan and CAP.
 - C66-53 Comment noted.

comment noted.

- C66-54 By its nature, increased density is associated with an increase in vehicle miles traveled, which would result in increased air pollution and greenhouse gas emissions.
- C66-55 The Density Redistribution Alternative would increase densities in some areas from the proposed Uptown CPU and decrease densities in others, resulting in a similar development potential.
- C66-56 See responses to comments C66-1 and C66-2 regarding the environmentally superior alternative.
- C66-57 All mitigation measures recommended and included in the Mitigation Monitoring and Reporting Program will be mandatory. Note that many mitigation measures identified in Section 6.3, Transportation and Circulation, are not carried forward due to inconsistencies with the proposed CPU and/or the City's inability to enforce.
- C66-58 As previous stated, these measures go above and beyond existing regulations and code to reduce potentially significant impacts.

Letter C67 From: Dawn Taggett <dawndburton@outlook.com> Sent: Friday, August 05, 2016 7 46 PM PLN. PlanningCEQA To: Dawn Taggett, Lorenzo Burton Cc: Comments on Uptown Community Plan Update: 21002568 / SCH No. Pending / Uptown Subject: District 3 Dear Mr. Steinert. C67-1 My name is Dawn Taggett-Burton and I live with my husband Lorenzo Burton, on Albatross Street in the Uptown Hillcrest district. I would like to officially register my support for the Lower-Density Alternative development plan (page 518 of the EIR), which incorporates land uses proposed in the June 2015 Draft Community Plan. This plan is the closest plan to the community plan negotiated over several years by the Uptown Planners and supported by the Uptown Community. The lower-density alternative plan also does not include the corresponding density bonus incentives, and according to the Uptown Planners and the EIR, it is also the "environmentally superior alternative" as shown on p. 524 of the EIR C67-2 Even though I am in support of the lower-density alternative (p. 518), I would like add that any plan adopted by the city must include rigid, unwavering requirements for real community improvement including street repair, traffic easing, public open space (not pseudo open space within the proposed developments, but real parks and meeting spaces). It has not escaped our community that Hillcrest is being asked to accept all of the new development; however, none of the new park space is within Hillcrest. We find this troubling and it must be rectified. Additionally, as seen with previously adopted plans, community improvement funds are always part of the approved deals' however, those funds are then reallocated outside of the district, and Hillcrest is left with nothing more than greater density and traffic jams. C67-3 Recent studies have found that although the development plans mainly focus on the urban core areas, the new jobs are being created in the northern portions of our county (Sorrento Valley/La Jolla/Del Mar), including inland areas just west of 115. Studies have reported that individuals moving into the urban areas (Downtown, Uptown, North Park, etc ...) are then commuting into the areas where the jobs actually are (not in urban areas). If the city is serious about implementing the Climate Action Plan, then proposed development should focus mainly on providing housing (affordable, middle-class, and high-end) near where people work. By providing housing near where the jobs are, commuting can be accomplished using greener alternatives (cycling, walking, bus, trolley, etc.,), rather than moving more people into the Uptown/Downtown areas without adequate long range transportation options, forcing them onto the IS, I8, and I15 where they sit in traffic and increase levels of greenhouse gases. C67-4 We all realize that housing is a problem within San Diego. Affordable housing is an even greater problem. However, I wholly reject the idea that if San Diego approves the super high density development plans proposed by the developers who have been lobbying the city and showing up at the Uptown Planners meetings, that this will somehow result in affordable housing. It defies logic. They want us to believe that if a 15 story building is approved, it will somehow result in cheaper rents. It hasn't in the past, so why would it now? Uptown is a community that commands higher rents and purchase prices, so that is exactly what the developers have demanded in the past and will continue to demand for the foreseeable future. Higher density will only mean that developers will have more units to rent or sale at unaffordable prices. I want someone to tell me how many units will have to be

C67-1 Thank you for your comment. The City appreciates your participation in the public review process and expressing support for the Lower-Density Alternative.

- C67-2 As described in Section 3.4.4, the Uptown CPU includes adoption of an Impact Fee Study (IFS) (formerly known as a Public Facilities Financing Plan [PFFP]) that addresses the need for public facilities associated with the identified needs of the Uptown CPU area. Funds collected under the Uptown IFS are maintained for use within the Uptown community. Additionally, the proposed Uptown CPU Public Facilities, Services, and Safety Element includes goals to provide and maintain infrastructure and public services for future growth without diminishing services to existing development. Specific policies regarding public facilities financing include public facilities and services prioritization as well as fire-rescue, police, wastewater, storm water infrastructure, waste management and recycling, libraries, schools, public utilities, and healthcare services and facilities, all included within the proposed Uptown CPU. It should be noted that future growth would occur throughout the CPU area and would not be limited to Hillcrest. As shown on Figure 6.12-2, the Uptown CPU proposes to introduce several park facilities within Hillcrest, including Mystic Park, Normal Street Linear Park, Maryland Street Pocket Park, Sixth Avenue Pocket Park, and First & Robinson Avenue Pocket Park.
- C67-3 The proposed Uptown CPU is one of 52 community plans within the City. Future updates of other community plans covering La Jolla, Sorrento Valley, and neighboring jurisdictions would plan for housing in those areas to address the jobs/housing balance within the northern region of the City (It should be noted, that Del Mar is an incorporated city). Consequently, the Uptown CPU is one of numerous community plans that collectively address regional needs within the City.

built before we reach the magic supply demand ratio where there is so much supply, that consumers can demand lower prices. Would there be a single tree or greenspace left in Uptown if we even tried to achieve that magical point? The developers attending the meetings could not answer this question, because to do the math and answer it would demonstrate to all listening, that Uptown cannot responsibly increase density to the point of oversupply, because to do so would require that we rid our area of most of the historic homes and replace them with multi units and highrises. Canyons would be peppered with multifamily units, and greenspace would be scarce.

2

C67-5

So again, I support the lower-density alternative recommended by the Uptown Planners. However, I hope that the city considers the history of broken promises made by developers; history of re-routing funds meant for community improvement; the conspicuous absence of dedicated park space for Uptown; lack of any real plans for traffic easing; lack of real plans for preserving our historic homes/buildings; and lack of a vision for developing in such a way that will revitalize the character of the community without stripping our neighbors of their access to natural sunlight.

Thank you for your consideration, Dawn Taggett-Burton/Lorenzo Burton

Sent from Mail for Windows 10

C67-4 The Uptown CPU would provide for affordable housing through Land Use Policy LU-1.1, which would "provide a variety of land use types to accommodate both affordable and market rate housing and commercial opportunities." It is difficult to predict future prices within the volatile Southern California housing market, but adherence to Land Use Policy LU-1.1 would ensure that the Uptown CPU introduces a variety of housing options, including affordable housing.

C67-5 Comment noted. The comment does not raise an issue with the adequacy of the PEIR.

RESPONSE

	From: Sent: To: Cc: Subject:	Tait, Laurie <itait@ndii.com> Tuesday, August 02, 2016 3:30 PM PLN_PlanningCEQA Laurie&Gaelle TAIT PERSO (Itait@cox.net) Uptown Community Plan Update: Project No. 21002568</itait@ndii.com>	Letter C68		
		Laur	ie Thomas Tait		
		3033	3 India Street #3		
		San	Diego, CA 92103		
		Aug	ust 2 nd 2016		
	Kurtis Steiner				
	Senior Environmental Planne	r.			
	City of San Diego Planning De	epartment,			
	1010 Second Avenue, MS 413	3,			
	San Diego, CA 92101				
	PlanningCEQA@sandlego.gov	v			
	RE:				
	Project Name: Uptown Comn	nunity Plan Update			
	Project No. 21002568 / SCH N	No. Pending			
	Community Area: Uptown				
	Council District: 3"				
	Dear Mr Steiner,				
C68-1	l am writing to you to voice n it will have a significant impa certain hit to the value of my	ny strong objection to the proposed Mobility Study Project No. 2 ct to the safety of my community, an impact to the amenities of property.	1002568. The proposal as I understand my neighborhood, and an almost	C68-1	Introductory comment noted. The City appreciates individual participation in the public review comment process.
C68-2	As I understand it the design	of street systems should encompass:			
	Safety- for both vehicular a	nd pedestrian traffic.		C68-2	Comment noted.
	Efficiency of Service- for all	users.			
		1			

RESPONSE

· Livability or Amenities- especially as affected by traffic elements in the circulation system.

Elements in the local circulation system should not have to rely on extensive traffic regulations in order to function efficiently and safely.

Local streets should be designed to discourage excessive speeds.

C68-3 In terms of safety, my main concern is that if India Street is widened to add an additional lane, the added volume (as shown in Uptown Community Plan Update PEIR Page 6.3-33) of traffic and increased speed will make entering/exiting my driveway outright dangerous if not impossible, It will also make walking my dog very unsafe (not to mention the fact) will not be able to walk him down to the amenities of Middletown Neighborhood Village, prominently known for its various restaurants located along India Street also known as International Restaurant Row). A study by Farouki (1976) and Moore (1969) show that the mean free speed of cars in roads increases linearly with the roadway width. If the speed along India increases by this factor we will need extensive traffic regulation to discourage excessive and dangerous speed. What are the Planning Department proposing in terms of regulation & policing?

C68-4 As part of the Project No. 21002568 Mobility Study, has the City of San Diego conducted a Traffic Impact Study specifically outlining the safety impact to this community? The impact study I read (<u>https://www.sandiego.gov/sites/default/files/appendix b-1.traffic Impact study.gd</u>) seemed to be mainly concerned with traffic volume, traffic flow and delays, capacity etc. Nowhere did I see any mention of safety impacts. Surely Safety must be at the top of the San Diego Senior Environmental Planner's priorities when proposing an environmental plan of this nature?

C68-5 The San Diego County General Plan

(http://www.sandiegocounty.gov/content/dam/sdc/pds/gpupdate/docs/GP/MobilityElement.pdf) outlines several Goals and Policies that conflict with the proposals in project No. 21002568.

The Mobility Element balances competing goals of accommodating trips generated by land use, while striving to retain a transportation network that complements, rather than impacts, the character of communities, which is generally rural in much of the unincorporated County. Therefore, widening of roads, which can dramatically change the character of a community, should be pursued only after environmental and community character impacts are also considered. The need to widen roads is minimized when trip vehicle miles traveled are reduced, the performance of the existing network is optimized, and the use of alternative modes of travel is maximized.

Accommodate all users of the road right-of-way — The Mobility Element also supports the concept of complete streets that are designed and operated to **enable safe access** for all users and for all modes of travel including non-motorized users and transit riders. This includes users of all ages and abilities such as the elderly, children, and people with disabilities.

GOALS AND POLICIES

GOAL M-1 Balanced Road Network. Policies A safe and efficient road network that balances regional travel needs with the travel requirements and preferences of local communities. M 1.1 Prioritized Travel within Community Planning Areas. Provide a public road network that accommodates travel between and within community planning areas rather than accommodating overflow traffic from State highways and freeways that are unable to meet regional travel demands.

0

C68-6 In terms of **amenities** the removal of all sidewalks to the North of my property along India Street will have a great impact on the walkable shops, bars and restaurants my neighborhood currently offers. Implementation of the proposed Uptown CPU and associated discretionary actions would restrict & impede pedestrian connectivity. It seems that this plan is in direct opposition to The City of San Diego Pedestrian Master Plan.

- C68-3 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).
- C68-4 Comment noted. This comment does not identify an inadequacy of the PEIR. The purpose of the Traffic Impact Study was to identify potential transportation and circulation impacts resulting from the proposed Uptown CPU and associated discretionary actions related to traffic operations. However, safety is a primary concern that plays a role in determining which identified mitigation measures are feasible to implement. The Mobility Element of the proposed Uptown CPU includes numerous policies addressing safety of streets within the community, including Policy MO-4.1, MO-4.3, MO-4.9, MO-4.10, MO-4.11, and MO-5.1.
- C68-5 Comment noted. This comment does not identify an inadequacy of the PEIR.
- C68-6 Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

RESPONSE

The Project Working Group and the consultant team prepared an overall vision statement for the PMP: "To create a safe, accessible, C68-7 C68-7 Comment noted. This comment does not identify an inadequacy of connected and waikable pedestrian environment that enhances neighborhood quality and promotes waiking as a practical and attractive means of transportation in a cost-effective manner." the PFIR. C68-8 In addition the Institute of Traffic Engineers guidelines point out that "Sidewalks should ordinarily be provided along streets used for pedestrian access to schools, parks, shopping areas, and transit stops." Comment noted. This comment does not identify an inadequacy of C68-9 The concept of protecting neighborhoods by ensuring that local streets serve their residential function is usually supported by local C68-8 ordinances. the PEIR. The Mobility Element of the proposed CPU includes numerous policies that address pedestrian safety. C68-10 Lastly the Uptown CPU draft outlines the noise impact on my neighborhood as Significant and Unavoidable. This alongside the increased traffic volume, the safety issued of entering/exiting my property due to increased volume and speed of traffic, the loss of Comment noted. This comment does not identify an inadequacy of C68-9 amenities will certainly have a negative impact on property values in this neighborhood. What does San Diego City Planning department propose in terms of compensation due as a direct result of this project? the PEIR. Yours Sincerely, C68-10 This comment does not identify an inadequacy of the PEIR. A significant and unavoidable impact associated with an increase in Laurie Tail. ambient noise levels was identified due to build-out of the proposed Uptown CPU. However, ambient noise increases are anticipated to occur with any level of growth and would also occur under the adopted Uptown Community Plan. While there are no feasible mitigation measures to reduce noise impacts to existing residences, the Noise Element proposed Uptown CPU contains numerous policies that address noise compatibility in the community that will guide future development to locate noisegenerating and noise-sensitive uses appropriately, require noise attenuation measures from new development, reduce unnecessary noise, and encourage other noise controls in the community. 3

RESPONSE

		Letter C69		
From: Sent: To: Subject:	V <va.sdsolutions@gmail.com> Friday, August 05, 2016 6:26 PM PLN_PlanningCEQA Project Name: Uptown Community Plan Update/ P</va.sdsolutions@gmail.com>	roject No. 21002568 / SCH No. Pending		
Community Area: Uptown Council District:3*				
2nd Email due to Deliv	very Failure.			
To whom it may conce	ern,			
l am writing to you as a around the India St. ar southbound St. and the going to make that play off at "Camp Run-A-Mu If you cut out and get r to safely turn on the cr any foot access across speeding down as if it"shoulder/sidewalk St. higher speeds making streets. Personally, if ever I'm of the "Camp Run-A-Mutt parking or a shoulder to someone is going to re makes it safer and easi restaurants at the end seriously fear for my sa arca, the parking, shou	a concerned member of the community that fi rea that is proposed to have a major constructi e sidewalk/parking and shoulder on India and cc a complete danger zone. I frequently visit t' utt', turn right on the cross streets that lay on ross streets, pull over to the shoulder for drop that long stretch of India. People already com 's a highway and this project will make it look Individuals will potentially be driving down t it extremely dangerous for those to turn onto visiting a friend in Mission Hills off India or dr t' grooming/dog day care area it's going to be to pull over, park or slowly turn on those cros car end me when I'm turning but the comfort c ier. It also allows walking access from locatior of the block (Gelato Vero, Irish Hub, El Indio, aafety and the safety of those in similar circum ilder and the sidewalk.	equently visits, dines and shops on project. Getting rid of the turning it into a three-lane St. is estarlight restaurant, drop my dog India St. from Laurel to Washington. rea for many community members offs and pick ups and will not have e off the 5 North Sassafras exit exactly as such with 3-lane no nat proposed 3-lane highway at India or turn right on any of the cross opping off/picking up my dog from an extreme safety issue not having s streets. I already feel as though f the shoulder/parking on the side s such as the dog day care to the Rubicon Deli, etc.). If removed, I stances such as mine who utilize that	C69-1	Comment noted. Please refer to the Master Response Regardin India Street Mitigation Measures included in the introduction these response to comments (pages RTC-4 through RTC-6).
I'm also amongst the m necessary locations lik extra length by having	nany that use the southbound St. on India to g te those mentioned above. Cutting this route w to get off near downtown and make my way h	et to many of my favorite and ill extend my trips to an unnecessary ack to find India St.		
I believe this project w members on India St. I there are many other o such a great inconveni community who oppos	vill cause a huge safety issue for the residents, I believe its an unnecessary cost for the city to community areas in the city need major fixing ience to so many people of that community wh ses this project.	business owners and community spend on such a huge project when I also deeply know it's going to cause ich is why I am a part of the		
Thank you				
	1			

Letter C70

From: V A <vardan.a@hotmail.com> Sent: Friday, August 05, 2016 6:09 PM PLN_PlanningCEQA Uptown Community Plan Update: Project #21002568/sch No. pending Subject:

Hello,

To:

C70-1 I am writing because I work as a delivery driver for a parts and service company who often makes drop offs and pick ups at the ATS Transmission Shop on India St. Cutting off this the shoulder/parking is going to be a complete nightmare! I won't be able to pull to the shoulder or be able to briefly park to drop off inoperable automobiles with a tow-truck and I definitely have no clue I will wheel anything out of there without the safety of a shoulder lane with a huge tow tuck. I also know a lot of people in the automotive industry that do drop/offs and pickups at that location and they are able to pull to the side parking in front and complete those tasks. I don't know how this business, and many other businesses will survive effectively if this project goes into motion. Loppose this project 100%!

Thank you,

Community Member of the Automotive Transmission Industry

Comment noted. Please refer to the Master Response Regarding C70-1 India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

Letter	C71
Letter	C/ 1

From:
Sent:
To:
Cubiant.

beckle@shearperadise.com Monday, August 08, 2016 7:11 AM PLN PlanningCEQA Uptown Community Plan Update 21002568/SCH No Pending

Mr. Steiner.

My thoughts and opinions have already been stated with a letter from the Middletown Advisory Committee.

C71-1 However, what is equally disturbing is how matters for India Street were decided with complete disregard to the many residents and businesses that have been in the Middletown area for decades. It truly feels as if the city doesn't realize that Middletown is a neighborhood, that India Street isn't an on ramp to the freeway. The fact that the proposed changes to India Street were even written down tells me and all other residents that your office didn't do proper research, never earne to the actual area to look at the situation, doesn't think through a problem and just picks what looks like the option that requires no thought.

With the opening of the RCC, and a dumping of ALL the traffic it has created in our neighborhood, did you really think we wouldn't notice? Just to give you an idea of what we are dealing with, just imagine 3,000 confused, late/in a hurry, un-familiar with steep hills, drivers in un-familiar cars on your street daily. Not a pleasant thought is it?

India Street is used by downtown, which has grown, by Little Italy which has grown, and the airport which has grown. And soon, more hotels will be opening in the Harbor Island area. So India Street will have even, more traffic. I have lived on Columbia Street above India for 27 years, the traffic has gotten worse every year but the RCC sent it over the top.

The solution is clear, a dedicated on/off ramp for the RCC, airport, Harbor Island traffic. If your office had been thinking that plan would've been started years ago certainly to keep up with the growing area.

The cluster which is Sasafrass, Kettner is a frightening ingress/ egress mess for the RCC. Did you think we wouldn't notice that either? Many of us do our shopping in the Midway area. Your lack of proper planning is ruining our neighborhood, our property values have gone down and it's un-safe to live here.

Please fix the mess you have allowed to happen.

Beckie Webster

C71-1 Comment noted. This comment does not identify an inadequacy of the PEIR. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

	From: Sent: To: Subject: Dear Mr. Stein	Letter C72 Stephen Wong <swong7664@gmail.com> Monday, August 01, 2016 2.06 AM PLN_PlanningCEQA Project Name: Uptown Community Plan Update Project No. 21002568 Project Name: Uptown Community Plan Update Project No. 21002568</swong7664@gmail.com>		
C72-1	My family resi and would like amendments to My son attende India Street on modify the traf amendment to changes to Indi The portion we between Sassai deletes the prir to exit to Laure to the traffic or significantly. Please reconside Sincerely, Stephen & San	des at 3054 Union Street within the Uptown Community to express our concerns about some of the proposed of the Uptown Community Plan. et the last meeting on the subject of traffic amendments to July 5th and was told there really was no intention to The pattern. It is now my understanding that this the Community Plan (Appendix C) specifically includes the is Street. e specifically object to is the removal of the southbound lane fras Street to Redwood Street. Elimination of this lane mary access to our home on Union Street forcing my family el Street and return on India Street northbound thus adding a Laurel and India and increasing our travel time der your proposal.	C72-1	Introductory comment noted. The City appreciates individual participation in the public review comment process. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

		1	
C73-1 C73-2 C73-3	From: Francis Wray < Mray20@gmail.comp.	C73-1 C73-2	Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6). Comment noted. Please refer to the master response regarding India Street mitigation measures included in the introduction to these responses to comments. The goals stated in the proposed CPU's Mobility Element are to create "safe, walkable neighborhoods, which utilize pedestrian connections and improved sidewalks to create a comfortable pedestrian experience". The City proposes Mobility Element Policy MO-4.9 which would implement road diets and traffic calming measures where appropriate to
	Please remove this proposal of at least consider some alternatives. Some have suggested utilizing the Pacific Coast Highway for car rental center and airport traffic. That would greatly reduce the number of residents and businesses effected. Thank you for your time on this matter. Sincerely, Prancis Wray 3020 Horton ave	C73-3	 improve safety and quality of service, and increase walking and bicycling in Uptown, and Mobility Element Policy MO-7.13 which supports on-street parking on all streets in order to support adjacent uses and enhance pedestrian safety and activity. As future development occurs these policies would be implemented through future construction improvements. Comment noted. Please refer to the Master Response Regarding India Street Mitigation Measures included in the introduction to these response to comments (pages RTC-4 through RTC-6).

TABLE OF CONTENTS

List	of Acro	onyms and Abbreviations	vii
Exec	utive	Summary	ES-1
1.0	Intro	oduction	1-1
	1.1	PEIR Purpose and Intended Uses	1-2
	1.2	PEIR Legal Authority	1-2
	1.3	EIR Type, Scope and Content, and Format	1-4
	1.4	PEIR Process	1-8
2.0	Envi	ronmental Setting	
	2.1	Regional Location	2-1
	2.2	Project Location	2-4
	2.3	Existing Physical Characteristics	2-6
3.0	Proj	ect Description	
	3.1	Introduction	3-1
	3.2	Relationship to the General Plan	3-2
	3.3	Project Objectives	3-3
	3.4	Project Description	3-3
	3.5	Environmental Design Considerations	3-26
	3.6	Plan Build-out	3-30
4.0	Hist	ory of Project Changes Related to CEQA	4-1
	4.1	NOP and Project Initiation	4-1
	4.2	Community Outreach and Plan Development	4-1
	4.3	Changes Based on Comments on the Draft Community Plans	4-2
5.0	Regu	ulatory Framework	5-1
	5.1	Land Use	5-1
	5.2	Visual Effects and Neighborhood Character	5-7
	5.3	Transportation and Circulation	5-8
	5.4	Air Quality	5-10
	5.5	Greenhouse Gas Emissions	5-16
	5.6	Noise	5-24
	5.7	Historical Resources	5-31
	5.8	Biological Resources	5-35
	5.9	Geologic Conditions	5-40
	5.10	Paleontological Resources	5-41
	5.11	Hydrology/Water Quality	5-42
	5.12	Public Services and Facilities	5-45

	5.13	Public Utilities	5-48
	5.14	Health and Safety5	5-49
6.0	Envii	ronmental Analysis – Uptown Community Plan Update	6-1
	6.1	Land Use	.1-1
	6.2	Visual Effects and Neighborhood Character6.	.2-1
	6.3	Transportation and Circulation6.	.3-1
	6.4	Air Quality	.4-1
	6.5	Greenhouse Gas Emissions	.5-1
	6.6	Noise	.6-1
	6.7	Historical Resources	.7-1
	6.8	Biological Resources	.8-1
	6.9	Geologic Conditions	.9-1
	6.10	Paleontological Resources 6.1	0-1
	6.11	Hydrology/Water Quality6.1	1-1
	6.12	Public Services and Facilities6.1	2-1
	6.13	Public Utilities	3-1
	6.14	Health and Safety6.1	4-1
7.0	Effec	ts Not Found to be Significant9	<u>7</u> -1
	7.1	Agricultural Resources	.7-1
	7.2	Mineral Resources	.7-2
	7.3	EnergyPopulation and Housing	.7-3
8.0	Grov	vth Inducement	8-1
90	Signi	ificant Unavoidable Impacts/Significant Irreversible Environment	al
2.0	Char	nges/Energy Conservation	9-1
	9.1	Significant and Unavoidable Impacts	.9-1
	9.2	Significant Irreversible Environmental Impacts	.9-1
	9.3	Energy Conservation	.9-3
10.0	Upto	own Alternatives	0-1
	10.1	No Project (Adopted Community Plan) Alternative	0-4
	10.2	Adopted Community Plan with Removal of the Interim Height	•
	1012	Ordinance Alternative)-13
	10.3	Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative)-21
	10.4	Density Redistribution Alternative)-26
	10.5	Lower-Density Alternative)-34
	10.6	Environmentally Superior Alternative)-41

11.0	Mitigation Monitoring and Reporting Program	-1
12.0	References	-1
13.0	Individuals and Agencies Consulted13-	-1
14.0	Certification	-1

APPENDICES

- A: Notice of Preparation and Comment Letters
- B-1: Traffic Impact Study
- B-2: Traffic Update Letter
- B-3: Traffic Memo Lower Density Alternative
- C: Mobility Study
- D: Air Quality Analysis
- E-1: Greenhouse Gas Analysis
- E-2 Supplemental Greenhouse Gas Analysis
- F: Noise Analysis
- G-1: Prehistoric Cultural Resources Constraints Analysis
- G-2: Historic Resources Survey
- H: Biological Resources Report
- I: Geotechnical Report
- J: Correspondence with Public Service Providers
- K: Water Supply Assessment
- L: GeoTracker Results

FIGURES

2-1:	Regional Location	2-2
2-2:	Uptown Community – Vicinity Map	2-3
2-3:	Uptown Community Plan Update Area	2-5
2-4:	Existing Land UseLand Uses under Adopted Community Plan – Uptown	2-7
2-5:	Neighborhood Centers and Nodes – Uptown	
2-6:	Landmarks and Gateways – Uptown	
2-7:	Sensitive Biological Resources- Uptown	2-32
2-8:	Uptown Watersheds	2-47
3-1:	Proposed Land Use – Uptown	3-5
3-2:	Proposed Uptown Land Use – West	3-6
3-3:	Proposed Uptown Land Use – East	3-7
3-4:	Proposed Uptown Land Use – South	3-8
3-5:	Uptown Village Locations and Mixed-Use Corridors	
3-6:	Future Roadway Classifications	
3-7:	Uptown Proposed CPIOZ – Type A	
3-8:	Uptown Proposed CPIOZ – Type B	
3-9:	Uptown Proposed Zoning	
6.1-1:	SDIA Airport Influence Area – Uptown	6.1-16

FIGURES (cont.)

6.1-2:	SDIA Threshold Siting Surface Profile – Uptown	6.1-17
6.1-3:	SDIA Noise Contour Map – Uptown	6.1-18
6.1-4:	Residential Use Allowed Under the General Community Plans Inside SDIA 70 dB CN	IEL 6.1-19
6.1-5:	SDIA Safety Compatibility Zones – Uptown	6.1-21
6.1-6:	Residential Use Allowed Under the General/Community Plans Inside the SDIA Safet	.y
	Compatibility Zones – Uptown	6.1-22
6.1-7:	SDIA Airspace Protection Boundary –Uptown	6.1-23
6.1-8:	SDIA Overflight Area Boundary – Uptown	6.1-24
6.2-1:	Uptown Canyons and Public Views	6.2-3
6.3-1:	Existing Roadway Functional Classification – Uptown	6.3-2
6.3-2:	Existing Roadway Segment ADT Volumes – Uptown	6.3-11
6.3-3:	Planned Transit Services – Uptown	6.3-19
6.3-4:	Regional Bicycle Plan – Uptown	6.3-20
6.3-5:	Existing and Planned Bicycle Facilities – Uptown	6.3-21
6.3-6:	Build-out Proposed Land Use Roadway Segment ADT Volumes – Uptown	6.3-27
6.4-1:	2035 Annual PM Concentrations from Freeway Operations – Uptown	6.4-10
6.6-1:	Noise Measurement Locations	6.6-3
6.6-2:	Existing Traffic Noise Contours for the Uptown CPU Area	6.6-5
6.6-3:	Future (2035) Traffic Noise Contours for the Uptown CPU Area	6.6-17
6.6.4:	Airport Noise Contours	6.6-22
6.6-5:	Quieter Home Program Participation for the Uptown CPU Area	6.6-23
6.7-1:	Location of City Register Designated Historic Resources	6.7-2
6.7-2:	Location of City Register Designated Historic District	6.7-3
6.7-3:	Cultural Sensitivity Areas – Prehistoric Resources	6.7-5
6.7-4:	Potential Historic Districts Identified by the Historic Resources Survey	6.7-14
6.7-5:	Location of Potential Historic Districts Identified by the Community	6.7-16
6.8-1:	Existing Vegetation Communities and Land Cover Types	6.8-2
6.8-2:	Location of Open Space	6.8-5
6.8-3:	Location of Sensitive Biological Resource Impacts	6.8-8
6.8-4:	Location of MHPA Boundary Line Adjustments	6.8-12
6.9-1:	Geology – Uptown	6.9-2
6.9-2:	City of San Diego Geologic Hazards and Faults – Uptown	6.9-3
6.9-3:	Geologic Hazards – Uptown	6.9-4
6.12-1:	Location of Public Services and Facilities	6.12-2
6.12-2:	Parks, Recreation Facilities, and Open Space	6.12-7
10-1:	No Project Alternative: Adopted Uptown Community Plan	10-6
10-2:	Maximum Building Heights under the Existing Interim Height Ordinance	10-7
10-3:	Maximum Building Heights of Existing Base Zones (Removal of the Interim Height	
	Ordinance)	10-14
10-4:	Density Redistribution Alternative	10-28
10-5:	Lower-Density Alternative	10-36

TABLES

S-1:	Summary of Significant Environmental ImpactsS-8
2-1:	North Park Existing Land Uses2-6
<u>2-2:</u>	Existing Land Use and Population versus Adopted Community Plan
2- <u>3</u> 2:	Regional Corridor Classification System
2- <u>4</u> 3:	Summary of Air Quality Measurements Recorded at the San Diego -
	1110 Beardsley Street Monitoring Station
2- <u>5</u> 4:	California GHG Emissions by Sector in 1990, 2008, and 2012
2- <u>6</u> 5:	Typical Sound Levels in the Environment and Industry
2- <u>7</u> 6:	Sensitive Plant Species Known or with the Potential to Occur in the
	Uptown Community
2- <u>8</u> 7:	Sensitive Wildlife Species Known to Occur in the Uptown Community Plan Update Area 2-37
2-9 8 :	San Diego County GHG Emissions by Sector in 2010
2- <u>109</u> :	Historic Imported, Local, and Recycled Water Demands to Public Utilities Department 2-54
3-1:	Project Components
3-2:	Land Use Classifications and Permitted Densities/Intensities
3-3:	Conversion to Citywide Zoning
3-4:	Proposed OR Zone within CPU Area
3-5:	Proposed RM Zones within CPU Area
3-6:	Proposed CN Zones within CPU Area
3-7:	Proposed CC Zones within CPU Area
3-8:	Proposed Land Use Classifications in Uptown
3-9:	Residential Development: Existing and at Proposed CPU Build-out
3-10:	Non-Residential Development: Existing and at Proposed CPU Build-out
3-11:	Potential Future Discretionary Actions Associated with Plan Build-out
5-1:	General Plan Land Use Categories5-2
5-2:	Ambient Air Quality Standards
5-3:	City of San Diego land Use – Noise Compatibility Guidelines (Table NE-3)
5-4:	San Diego Property Line Noise Level Limits
5-5:	Airport Noise Compatibility Criteria
5-6:	Deployment Measures to Address Future Growth by Population Density per Square Mile 5-47
5-7:	Deployment Measures to Address Future Growth by Population Clusters
6.1-1:	Proposed Uptown CPU Policies Related to Land Use
6.1-2:	Uptown Community Plan Proposed Land Use Designations
6.1-3:	Recommended Zone Designations – Uptown
6.3-1:	City of San Diego Roadway Segment Capacity and Level of Service
6.3-2:	Level of Service Criteria for Intersections
6.3-3:	Level of Service Criteria for Freeway Segment Analysis
6.3-4:	Existing Freeway Segment Level of Service
6.3-5	Existing Freeway Ramp Metering
6.3-6:	Significance Criteria for Facilities in Study Area 6.3-23
6.3-7:	Build-out Summary of Intersection Analysis - Uptown
6.3-8:	Build-out Summary of Roadway Segment Analysis – Uptown
6.3-9:	Summary of Freeway Segment Level of Service
6.3-10:	Peak Hour Ramp Metering Analysis – Horizon Year Conditions
6.4-1:	Air Quality Impact Screening Levels

TABLES (cont.)

6.4-2:	Sample Daily Construction Emissions	6.4-5
6.4-3:	Total Operational Emissions for the Uptown CPU Area	6.4-7
6.4-4:	Maximum Build-out CO Concentrations in the Uptown CPU Area	6.4-8
6.5-1:	San Diego Gas & Electric Intensity Factors	6.5-3
6.5-2:	Land Use Distribution	6.5-7
6.5-3:	GHG Emissions for the Uptown Community Plan Area	6.5-7
6.6-1:	Noise Measurements – Uptown	6.6-2
6.6-2:	Increases in Ambient Noise for the Uptown CPU Area	6.6-8
6.6-3:	Future Vehicle Traffic Contour Distances for the Uptown CPU Area	6.6-14
6.6-4:	Railway Noise Levels	6.6-20
6.7-1:	Potential Historic Districts Identified in the Uptown Historic Resource Survey	6.7-9
6.8-1:	Vegetation Communities and Land Cover Types within the Uptown CPU Area	6.8-1
6.8-2:	Proposed Open Space for Uptown	6.8-6
6.8-3:	Modifications to Vegetation Communities and Land Cover Types as a	
	Result of the MHPA Boundary Line Corrections at Uptown	6.8-11
6.9-1:	Deterministic Seismic Hazard Parameters – Uptown	6.9-7
6.9-2:	Probabilistic Seismic Hazard Parameters – Uptown	6.9-7
6.12-1:	Deployment Measures for San Diego City Growth	6.12-4
6.12-2:	Response Time Measures	6.12-4
6.12-3:	Population-Based Parks and Recreation Facilities	6.12-8
6.12-4:	Summary of Existing and Proposed Population-Based Parks and Recreation	
	Facilities	. 6.12-18
6.12-5	Uptown Student Generation Rates from Existing Housing Units	.6.12-20
6.12-6:	Uptown Potential Student Generation Rates from Future Additional Housing	
	Units	. 6.12-21
6.14-1:	Hazardous Materials Sites in Uptown	6.14-2
10-1:	Matrix Comparison of Project Alternatives and Proposed Uptown CPU	10-3
10-2:	Amendments to the 1988 Uptown Community Plan	10-5
10-3:	Buildout Comparison of the No Project (Adopted Community Plan) Alternative and	
	the Proposed Uptown CPU	10-8
<u>10-4:</u>	Build-out Comparison of the Density Redistribution Alternative and the	
	Proposed Uptown CPU	<u> 10-27</u>
10- <u>5</u> 4:	Buildout Comparison of the Lower-Density Alternative and the Proposed Uptown CP	U 10-35

List of Abbreviated Terms

°F	degrees Fahrenheit
µg/m3	micrograms per cubic meter
AAQS	Ambient Air Quality Standards
AB	Assembly Bill
ACOE	U.S. Army Corps of Engineers
ADA	Americans with Disabilities Act
ADT	average daily traffic
AFY	acre-feet per year
AIA	Airport Influence Area
Airport Authority	San Diego County Regional Airport Authority
ALUC	Airport Land Use Commission
ALUCP	Airport Land Use Compatibility Plan
AMSL	above mean sea level
APCD	Air Pollution Control District
APS	Alternative Planning Strategy
ART	Arterial Rapid Transit
Basin Plan	Water Quality Control Plan for the San Diego Basin
BAU	business as usual
BMP	best management practice
BRT	bus rapid transit
CAA	Clean Air Act
CAAQS	California Ambient Air Quality Standards
CAFE	Corporate Average Fuel Economy
Cal EPA	California Environmental Protection Agency
CAL FIRE	California Department of Forestry and Fire
CalARP	California Accidental Releases
CalEEMod	California Emissions Estimator Model
CalGreen	California Green Building Standards Code
CalRecycle	California Department of Resources Recycling and Recovery
Caltrans	California Department of Transportation
CAP	Climate Action Plan
CAPCOA	California Air Pollution Control Officers Association
CARB	California Air Resources Board
CASAC	Clean Air Scientific Advisory Committee
CBC	California Building Code

CCAP	U.S. Climate Change Action Plan
ССС	California Coastal Commission
CCR	California Code of Regulations
CDC	California Department of Conservation
CDFW	California Department of Fish and Wildlife
CDP	Coastal Development Permit
CEC	California Energy Commission
CEQA	California Environmental Quality Act
CERCLA	Comprehensive Environmental Response, Compensation, and Liability Act
CFR	Code of Federal Regulations
CH ₄	Methane
CIP	Capital Improvement Project
City	City of San Diego
City BMP	City of San Diego Bicycle Master Plan
CNDDB	California Natural Diversity Data Base
CNEL	Community Noise Equivalent Level
CO	Carbon monoxide
CO ₂	Carbon dioxide
CPIOZ	Community Plan Implementation Overlay Zone
CPU	Community Plan Update
CPUAC	Community Plan Update Advisory Committee
CPUC	California Public Utilities Commission
CRC	California Residential Code
CRHR	California Register of Historic Resources
CTC	California Transportation Commission
CUPA	Certified Unified Program Agency
CWA	Clean Water Act
dB	decibel
dB(A)	A-weighted decibels
DIF	Development Impact Fees
DPM	diesel particulate matter
DTSC	California Department of Toxic Substances Control
du/ac	dwelling units per acre
EO	Executive Order
EPA	U.S. Environmental Protection Agency
EPCRA	Emergency Planning Community Right-to-Know Act
ESA	Endangered Species Act
ESA	Environmental Site Assessment
ESD	Environmental Services Department
ESL	Environmentally Sensitive Lands
FAA	Federal Aviation Administration
FAR	floor area ratio
FEMA	Federal Emergency Management Act

FHWA	Federal Highway Administration
FIRM	Flood Insurance Rate Map
FTA	Federal Transit Administration
GHC	Geologic Hazard Category
GHG	greenhouse gas
GIS	geographic information system
GWP	global warming potential
gdp	gallons per day
H&SC	California Health and Safety Code
HAs	Hydrologic Areas
HAZMIT	2010 San Diego County Multi-Jurisdictional Hazard Mitigation Plan
HCH	hexachlorocyclohexane
HCM	Highway Capacity Manual
HFC	hydrofluorocarbons
HMBP	Hazardous Materials Business Plan
HMD	Hazardous Materials Division
HMTA	Hazardous Materials Transportation Act
HOV	High Occupancy Vehicle
HRB	Historical Resources Board
HSAs	Hydrologic Subareas
HUs	Hydrologic Units
HVAC	heating, ventilation, and air conditioning
Hz	Hertz
I-5	Interstate 5
I-8	Interstate 8
I-805	Interstate 805
ICLEI	International Council for Local Environmental Initiatives
IFS	Impact Fee Study
in/sec	inches per second
IPCC	Intergovernmental Panel on Climate Change
ITS	Intelligent Transportation Systems
IWRP	Integrated Water Resources Plan
IWRP	Integrated Water Resources Plan
JRMP	Jurisdictional Runoff Management Plan
JURMP	Jurisdictional Urban Runoff Management Program
kHz	kilo-Hertz
LBP	lead-based paint
LCFS	Low Carbon Fuel Standard
LCP	Local Coastal Plan
LCPA	Local Coastal Plan Amendment
LCS	lead-containing surfaces
LDC	Land Development Code
LDM	Land Development Manual

L _{dn}	day-night equivalent level
L _{eq}	Average sound level
LEV III	Low Emission Vehicle III
LGBT	lesbian, gay, bisexual, and transgender
LGBTQ	lesbian, gay, bisexual, transgender, and queer
LID	Low Impact Development
LOS	Level of Service
Lpw	sound power
LRA	Local Responsibility Area
LRT	light rail transit
LUE	Land Use Element
LUP	Land Use Plan
LUST	leaking underground storage tanks
MBTA	Migratory Bird Treaty Act
MHCP	Multiple Habitat Conservation Program
MHMP	Multi-Jurisdictional Hazard Mitigation Plan
MHPA	Multi-Habitat Planning Area
MMRP	Mitigation Monitoring and Reporting Program
MMT CO ₂ E	million metric tons of CO ₂ equivalent
MOE	measurement of effectiveness
mpg	miles per gallon
mph	miles per hour
MPL	Multiple Property Listing
MPO	Metropolitan Planning Organization
<u>MS4</u>	Municipal Separate Storm Sewer System
MSCP	Multiple Species Conservation Program
MSL	mean sea level
MT CO ₂ E	metric tons of CO ₂ equivalent
MW	megawatt
MWD	Metropolitan Water District of Southern California
MWh	megawatt hour
N ₂ O	Nitrous oxide
NAAQS	National Ambient Air Quality Standards
NAHC	Native American Heritage Commission
NCCP	Natural Community Conservation Plan
NCHRP	National Cooperative Highway Research
NCP	National Contingency Plan
NCWRP	North City Water Reclamation Plant
NDP	Neighborhood Development Permit
NFIP	National Flood Insurance Program
NFPA	National Fire Protection Association
NHTSA	National Highway Traffic and Safety Administration
NMFS	National Marine Fisheries Service

NO ₂	nitrogen dioxide
NOP	Notice of Preparation
NO _x	oxides of nitrogen
NPDES	National Pollutant Discharge Elimination System
NRC	Nuclear Regulatory Commission
NRHP	National Register of Historic Places
O ₃	ozone
OES	Office of Emergency Services
OES	Office of Emergency Services
OPR	Office of Planning and Research
OSHA	Occupational Safety and Health Administration
PAHS	polynuclear aromatic hydrocarbons
pb	lead
PCB	polychlorinated biphenyls
PCBs	polychlorinated biphenyls
PCE	Tetrachloroethylene
PDO	Planned District Ordinance
PDP	Planned Development Permit
PEIR	Program Environmental Impact Report
PFC	perfluorocarbons
PFFP	Public Facilities Financing Plan
PLWTP	Point Loma Wastewater Treatment Plan
PM	particulate matter
PM_{10}	particulate matter less than 10 microns in diameter
PM _{2.5}	particulate matter less than 2.5 microns in diameter
ppb	parts per billion
pphm	parts per hundred million
ppm	parts per million
PPV	peak particle velocity
PRD	planned residential developments
PSE	Public Safety Element
PUD	Public Utilities Department
PWD	Public Works Department
RAC	Regional Advisory Committee
RAQS	Regional Air Quality Strategy
RBP	Regional Bicycle Plan
RCP	Regional Comprehensive Plan
RCRA	Federal Resource Conservation and Recovery Act
RHNA	Regional Housing Needs Assessment
RM	Ready Made
RME	Resource Management Element
ROG	Reactive organic gas
RPS	Renewable Portfolio Standard

RTP	Regional Transportation Plan		
RUWMP	Regional Urban Water Management Plan		
RWMG	Regional Water Management Group		
RWQCB	Regional Water Quality Control Board		
SAM	Site Assessment and Mitigation		
SANDAG	San Diego Association of Governments		
SARA	Superfund Amendments and Reauthorization Act		
SB	Senate Bill		
SBWRP	South Bay Water Reclamation Plant		
SCAB	South Coast Air Basin		
SCH	State Clearinghouse		
SCS	Sustainable Communities Strategy		
SDAB	San Diego Air Basin		
SDCWA	San Diego County Water Authority		
SDG&E	San Diego Gas & Electric		
SDIA	San Diego International Airport		
SDMC	San Diego Municipal Code		
SDPL	San Diego Public Library		
sec/veh	seconds per vehicle		
sf	square feet		
SF ₆	sulfur hexafluoride		
SFHA	Special Flood Hazard Area		
SHPO	State Historic Preservation Office		
SIP	State Implementation Plan		
SNPMS	Sustainable North Park Main Street		
SO ₂	sulfur dioxide		
SoCalGas	Southern California Gas Company		
SOI	Sphere of Influence		
SR-15	State Route 15		
SR-163	State Route 163		
SR-94	State Route 94		
SSMP	Sewer System Management Plan		
SSO	Sanitary Sewer Overflows		
STAA	Surface Transportation Assistance Act		
STC	sound transmission class		
SUSMP	Standard Urban Stormwater Mitigation Plan		
SWPPP	storm water pollution prevention plan		
SWRCB	State Water Resources Control Board		
T&SW	Transportation and Storm Water Department		
TAC	toxic air contaminants		
TAIC	Technology Associates International Corporation		
TCM	transportation control measures		
TDM	Transportation Demand Management		

TDC	
IDS	total dissolved solid <u>s</u>
TERPS	Terminal Instrument Procedures
TMDL	total maximum daily load s
TPA	Transit Priority Area
TSS	Threshold Siting Surface
UDC	Unified Disaster Control
URMP	Urban Runoff Management Plan
USACE	United States Army Corps of Engineers
U.S.C.	United States Code
U.S. DOT	United States Department of Transportation
USFWS	United States Fish and Wildlife Service
UST	underground storage tank
UWMP	Urban Water Management Plan
v/c	volume to capacity
VMT	vehicle miles travelled
VOC	volatile organic compounds
WMP	Waste Management Plan
WQIP	Water Quality Improvement Plan s
WRCC	Western Regional Climate Center
WSA	Water Supply Assessment



Executive Summary

S.1 Proposed Project

Project Location and Setting

The Uptown Community Plan Update (CPU) area is centrally located to the north of Downtown San Diego and south of the Mission Valley community. The Uptown Community Plan area forms the western boundary and a portion of the northern boundary of Balboa Park.

The Uptown Community Plan area consists of approximately 2,700 acres (approximately 4.2 square miles) and lies just north of Downtown San Diego. It is bounded on the north by the steep hillsides of Mission Valley, on the east by Park Boulevard, and on the west and south by Old Town San Diego and I-5. The Uptown community is located on a level mesa that is divided by numerous canyons and bordered by two major parks, Presidio and Balboa. The CPU area includes the neighborhoods of Mission Hills, Middletown, Hillcrest, the Medical Complex, University Heights, and Bankers Hill/Park West.

Uptown's overall physical structure reflects its geography and development patterns. Most of the street system uses a grid pattern. The CPU area is traversed by three major east-west streets; Washington Street and University Avenue in the northern portion of the community and Laurel Street in the southern portion. Park Boulevard, which services as the community's eastern boundary, as well as First Avenue are important two-way north-south streets along with Fourth and Fifth avenues, which are one-way south- and northbound streets, respectively. Other significant streets are the one-way northbound India Street and one-way westbound Hawthorne Street.

Project Description

The project includes the comprehensive update to the Uptown Community Plan, which is intended to guide development through 2035 build-out of the Community Plan. For facility planning, technical evaluation, and environmental review purposes, build-out is assumed to occur in 2035. The

Community Plan also addresses changes in conditions since 1988, when the Uptown Community Plan was last adopted. The proposed CPU provides detailed policy direction to implement the General Plan with respect to the distribution and arrangement of land uses (public and private); local street and transit network; prioritization and provision of public facilities, community, and site-specific urban design guidelines; and recommendations to preserve and enhance natural open space and historic and cultural resources within the Uptown community.

CPU implementation requires adoption of the Uptown Community Plan, amendments to the General Plan to incorporate the proposed CPU as a component of the General Plan Land Use Element, adoption of a Land Development Code (LDC) ordinance that would rezone the Planned District Ordinance (PDO) areas within the CPU area with Citywide zones within the LDC and repeal the existing Mid-City Communities PDO, the West Lewis Street PDO, and Interim Height Ordinance. The project would also amend the mapped boundaries of the Uptown Community Plan Implementation Overlay Zone (CPIOZ) to include CPIOZ-Type A and CPIOZ-Type B areas that would limit building heights. A comprehensive Impact Fee Study (IFS) (formerly known as the Public Facilities Financing Plan) is also proposed for adoption.

The overall vision of the Uptown Community Plan is to guide, over the next 20 to 30 years, future infill development that is transit supportive per the General Plan and is also protective of desired community character and resources. The proposed land use plan locates the highest intensity land uses within the community along transit corridors where existing and future commercial, residential and mixed-use development can support existing and planned transit investments. The Land Use Element defines Village Districts and key corridors where future growth is targeted within the community in order to fulfill the General Plan's City of Villages strategy.

The proposed Uptown CPU includes an Introduction and Implementation chapter, and includes the following elements: Land Use; Mobility; Urban Design; Economic Prosperity; Public Facilities, Services and Safety; Recreation; Conservation; Noise; and Historic Preservation. Chapter 11 of the proposed Uptown CPU describes available financing methods for public improvement projects.

S.2 Project Objectives

In accordance with California Environmental Quality Act (CEQA) Guidelines Section 15124, the following objectives were identified to outline the underlying purpose for the project. These objectives will be used to assist the lead agency in developing a reasonable range of alternatives to be evaluated in this PEIR and ultimately aid decision-makers in preparing findings and overriding considerations, if necessary. The primary objectives for the project are:

- Develop a multi-modal transportation network emphasizing active transportation measures for walkable and bicycle-friendly streets, and transit-related measures supporting transit operations and access.
- Maintain or increase the housing supply through the designation of higher residential densities focusing along major transit corridors.

- Provide for increased economic diversification through land use to increase employment and economic growth opportunities.
- Preserve the neighborhood character and design relationships between neighborhoods within each community through the development of transitions and design policies.
- Identify significant historic and cultural resources within the community and provide for their preservation, protection, and enhancement.
- Provide increased recreation opportunities and new public open spaces.
- Preserve, protect, and enhance the community's natural landforms, including canyons and environmentally sensitive lands.
- Include financing strategies that can secure infrastructure improvements concurrent with development.

S.3 Areas of Controversy

<u>Areas of controversy include community concerns, generally related to the distribution of residential</u> <u>densities, development intensity, building heights, and the use of potential historic districts. Refer to</u> <u>Section 4.3, Changes Based on Comments on the Draft Community Plans, of this PEIR.</u> <u>Although</u> <u>there are no clear-cut areas of controversy, e</u><u>E</u>nvironmental impacts classified as significant and unavoidable <u>that may generate controversy</u> have been identified in the resource topics of transportation and circulation, noise, historical resources, and paleontological resources, which are described in Chapters 6.3, 6.6, 6.7, and 6.10, respectively.

S.4 **Project Alternatives**

In order to fully evaluate the environmental effects of proposed projects, CEQA mandates that alternatives to the proposed project be analyzed. Section 15126.6 of the state CEQA Guidelines requires the discussion of "a range of reasonable alternatives to the project, or to the location of the project, which would feasibly attain most of the basic objectives of the project but would avoid or substantially lessen any of the significant effects of the project" and the evaluation of the comparative merits of the alternatives. The alternatives discussion is intended to "focus on alternatives to the project or its location which are capable of avoiding or substantially lessening any significant effects of the project," even if these alternatives would impede to some degree the attainment of the project objectives.

Alternatives to the proposed CPU are evaluated in Chapter 10 of this PEIR for the Uptown CPU. The evaluations analyze the ability of each alternative to further reduce or avoid significant environmental effects of the proposed CPU. Each major issue area included in the impact analysis of this PEIR has been given consideration in the alternatives analysis. This PEIR evaluates four <u>five</u> alternatives to the proposed Uptown CPU and associated discretionary actions including: (1) No Project (Adopted Community Plan) Alternative; (2) Adopted Community Plan with Removal of the Interim Height Ordinance Alternative;

(3) Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative; (4) <u>Density Redistribution Alternative</u>, and (4<u>5</u>) Lower-Density Alternative.

No Project (Adopted Community Plan) Alternative

Under the No Project Alternative, the adopted Uptown Community Plan would continue to guide development and would be implemented with the zoning program, which includes Mid-City Communities Plan District, West Lewis Plan District, and the Interim Height Ordinance.

The No Project Alternative would consist of the adopted Community Plan land use designations as they apply today, including all amendments to the Community Plan from its original adoption in 1988 to the most recent amendment in 2008. Adopted community plan land use designations seek to promote a balance of land uses. The majority of the land use is designated as Low-Density Residential at 5 to 10 units per acre. The adopted plan locates higher residential density away from the single-family neighborhoods and focuses development on the major transportation corridors: Washington; University; Park Boulevard; and 4th, 5th, and 6th avenues. Mixed-use development is encouraged in selected areas with residential use over street-level retail use. In Uptown, the Hillcrest and Bankers Hill neighborhoods are identified for the highest intensity within the community with up to 110 dwelling units per acre (du/ac) along 5th and 6th avenues and within the Hillcrest core. Institutional and Schools/Public Facilities are designated for City-owned and other public/quasipublic facilities.

Areas of proposed land use change are concentrated throughout the community where the proposed Uptown CPU would generally facilitate lower intensity mixed-use development compared to the existing Community Plan. Specifically, the proposed Uptown CPU could have approximately 32,700 dwelling units at build-out, while the No Project Alternative could have approximately 34,600 dwelling units at build-out or 1,900 more units compared to the proposed Uptown CPU.

Adopted Community Plan with Removal of the Interim Height Ordinance Alternative

This alternative would apply the adopted Uptown Community Plan and zoning program including the Mid-City Communities Plan District and West Lewis Plan District with the exception that the Interim Height Ordinance (O-20329) that limits structure heights in specific areas to 50 and 65 feet would not be applied. Height limits of the base zones would be applied. As a result, those areas now subject to the Interim Height Ordinance would allow buildings up to the height permitted by the Mid-City Communities Plan District. In the case of areas in Mission Hills currently limited to 50 feet, structures would be permitted up to 150 feet. In the areas of Hillcrest limited to 65 feet, structures would be permitted to 200 feet (refer to Figure 10-3).

Compared to the proposed Uptown CPU that would include new structure height regulations in certain areas through implementation of the CPIOZ (depicted on Figures 3-7 and 3-8 of this PEIR), the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would allow taller buildings under ministerial review within the Mission Hills, Hillcrest, and Bankers Hill/Park West neighborhoods. The increased building height allowance combined with slightly higher density

under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would have the potential to increase the intensity of development with taller buildings compared to the proposed Uptown CPU and associated discretionary actions.

Areas of proposed land use change are concentrated throughout the community where the proposed Uptown CPU would generally facilitate lower intensity mixed-use development compared to the adopted Community Plan. Specifically, the Adopted Community Plan would accommodate 34,600 dwelling units at build-out or 1,900 more units compared to the proposed Uptown CPU.

Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would use the adopted Uptown Community Plan land use map. The alternative would address neighborhood character issues by implementing the new proposed urban design policies that address objectives such as creating development transitions between new development and existing neighborhoods, increasing the urban tree canopy, and supporting sustainable development. Under this alternative, the current zoning program, which includes the Mid-City Communities Plan District and the West Lewis Plan District, would be retained with the exception of the Interim Height Ordinance (O-20329), which would be rescinded. Figure 10-2 shows the maximum building heights in areas affected by the Interim Height Ordinance that would apply under this alternative. The proposed project CPIOZ would reduce heights in areas of Mission Hills and Hillcrest compared to building heights that would be allowed under the Proposed CPU Policies with the Adopted Community Plan Land Use Map Alternative.

The build-out assumptions and land use map would be identical to the No Project (Adopted Community Plan) Alternative. Similar to the proposed project, this alternative would also address potential historical resource impacts by amending the Historical Resources Regulations in the Land Development Code to provide supplemental development regulations pertaining to potential historic districts. Application of the proposed Uptown CPU policies related to urban design and mobility under this alternative would provide design guidance including development transitions to new development and would support multi-modal transportation choices.

Density Redistribution Alternative

The Density Redistribution Alternative uses land uses proposed in June 2015 Draft Community Plan without the corresponding density bonus incentives originally proposed with this land use scenario. Under this alternative, the density of future development would be lower along transit commercial nodes except for the transit corridor along Park Boulevard between University Avenue and Washington Street and Normal Street. Under this alternative, the reduction in density would be redistributed resulting in the same overall development potential as the proposed Uptown CPU. The locations and associated density decreases from the proposed Uptown CPU are described below:

- 1. India Street (Neighborhood Commercial 0-29 du/ac)
- 2. Reynard Way (Residential Medium 16-29 du/ac and Neighborhood Commercial 0-29 du/ac)

- 3. 4th Avenue between Upas and Spruce (Office Commercial 0-29 du/ac)
- 4. 4th Avenue between Laurel and Grape (Office Commercial 0-29 du/ac)
- 5. Bankers Hills/Park West Neighborhood west of 1st Ave (Residential Medium 16-29 du/ac)
- 6. Medical Center Complex (Neighborhood Office Commercial 0-44 du/ac)
- 7. Washington Street near Dove (Community Commercial 0 44 du/ac)
- 8. Central Hillcrest (Community Commercial 0-44 du/ac)
- 9. South of Pennsylvania in Hillcrest (Community Commercial 0-73 du/ac)

When compared to the proposed Uptown CPU, the Density Redistribution Alternative reduces residential density development potential along India Street, Reynard Way, the 4th Avenue Commercial Office areas, and Bankers Hills/Park West Neighborhood from 44 du/ac to 29 du/ac. The Density Redistribution Alternative reduces areas of the Medical Center Complex, Washington Street near Dove Street, and areas within Central Hillcrest from 73 du/ac to 44 du/ac. Additionally, the core Central Hillcrest area is reduced from 109 du/ac to 44 du/ac and density in Hillcrest, South of Pennsylvania, is reduced from 109 du/ac to 74du/ac. The Normal Street corner lot along Park Boulevard is reduced to Community Commercial 0-44 du/ac. The Density Redistribution Alternative increases transit corridor density along Park Boulevard between University Avenue and Washington Street and Normal Street from 73 du/ac to 109 and 145 du/ac.

Lower-Density Alternative

The Lower-Density Alternative incorporates the land uses proposed in June 2015 Draft Community Plan without the corresponding density bonus incentives originally proposed with this land use scenario. The Lower-Density Alternative would be the same as the Density Redistribution Alternative with the exception that density would not increase along the Park Boulevard generally between Washington Street, University Avenue, and Normal Street. The Lower-Density Alternative would reduce multi-family development potential and result in a slight increase in single-family development potential. The total projected population under the Lower-Density Alternative would be 2,650 persons fewer than under the proposed Uptown CPU.

Environmentally Superior Alternative

CEQA Guidelines section 15126.6(e)(2) requires an EIR to identify the environmentally superior alternative. If the No Project Alternative is the environmentally superior alternative, the EIR must identify an environmentally superior alternative from the other alternatives.

Based on a comparison of the alternatives' overall environmental impacts and their compatibility with the CPU's goals and objectives, the Density Redistribution Alternative is the environmentally superior alternative for this Program EIR. While the Density Distribution Alternative would not be able to reduce the significant and unavoidable impacts of the proposed Uptown CPU, it would reduce impacts related to traffic circulation and air quality. At the same time, the Density Redistribution Alternative would not support the full implementation of the General Plan's City of Villages strategy of developing multi-modal centers that encourage walking, bicycling, and taking transit and contain a mixture of commercial and residential development because the density of future development under the Density Redistribution Alternative would be lower along transit commercial nodes except for the transit corridor along Park Boulevard between University Avenue

and Washington Street and Normal Street. The Density Redistribution Alternative could also conflict with the implementation of the City's Climate Action Plan since the redistribution of density would result in a likely increase in greenhouse gas emission impacts and vehicle miles traveled.

S.5 Summary of Significant Impacts and Mitigation Measures that Reduce the Impact

Table ES-1 summarizes the results of the environmental analysis including the potentially significant environmental impacts of the proposed CPU and proposed mitigation measures to reduce or avoid these impacts. Impacts, including analysis of cumulative impacts, and mitigation measures are organized by issue in Chapter 6.0, Environmental Analysis. Chapter 6.0 also includes discussions of proposed policies that would reduce identified impacts.

Table S-1 Summary of Significant Environmental Impacts				
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
Land Use Would the proposed project conflict with the environmental goals, objectives, or guidelines of a General Plan or Community Plan or other applicable land use plan or regulation and as a result, cause an indirect or secondary environmental impact?	The proposed Uptown CPU and associated discretionary actions are consistent with the General Plan and the City of Villages strategy. Furthermore, the policies developed for the proposed Uptown CPU associated with each of the elements were drafted in a manner that is consistent with the General Plan and San Diego Forward – the Regional Plan. Proposed amendments to the Land Development Code and zoning amendments would implement the proposed CPU and would be consistent with applicable environmental goals, objectives and guidelines of the General Plan. The proposed change from the PDO to Citywide zone and implementation of the CPIOZ to regulate height would not create any conflicts or inconsistencies with the adopted Land Development Code. Future development in accordance with the proposed Uptown CPU would be required to comply with Environmentally Sensitive Lands (ESL) regulations. As the proposed Uptown CPU and associated discretionary actions would be consistent with applicable environmental goals, objectives, or guidelines of a General Plan, no indirect or secondary environmental impact would result and impacts would be less than significant. No mitigation is required.	None Required	Less than Significant	
Would the proposed project lead to the development or conversion of General Plan or Community Plan designated open space or prime farmland to a more intensive land use, resulting in a physical division of the community?	The proposed Uptown CPU and associated discretionary actions would not result in the conversion of open space or physically divide an established community. Community connectivity would be enhanced by provisions in the proposed Uptown CPU that improve pedestrian and transit amenities. Impacts would be less than significant; therefore, no mitigation would be required.	None Required	Less than Significant	

Table S-1				
Summary of Significant Environmental Impacts				
Environmental Issue Would the project conflict with the provisions of the City's Multiple Species Conservation Program (MSCP) Subarea Plan or other approved local, regional, or state habitat conservation plan?	Results of Impact Analysis Implementation of the proposed Uptown CPU and associated discretionary actions would not have significant impacts on the Multi-Habitat Planning Area (MHPA) because ESL Regulations would limit development encroachment into sensitive biological resources and would be consistent with the MSCP. Therefore, impacts related to conflicts with the MSCP Subarea Plan would be less than significant and no mitigation is required.	Mitigation None Required	Impact Level After Mitigation Less than Significant	
Would the project result in land uses which are not compatible with an adopted Airport Land Use Compatibility Plan (ALUCP)?	Although the Uptown community is within the San Diego International Airport (SDIA) Airport Influence Area (AIA), the proposed Uptown CPU and associated discretionary actions would not result in conflicts with the adopted Airport Land Use Compatibility Plan (ALUCP). Future projects would be required to receive Airport Land Use Commission consistency determinations, as necessary which would ensure future projects are reviewed for consistency with the SDIA ALUCP. As a result, the proposed Uptown CPU and associated discretionary actions would not result in land uses that are incompatible with an adopted Airport Land Use Compatibility Plan. Impacts would be less than significant and no mitigation is required.	None Required	Less than Significant	
Visual Effects and Neighborhood C	haracter			
Would the project result in a substantial obstruction of a vista or scenic view from a public viewing area as identified in the community plan?	The implementation of the proposed Uptown CPU and associated discretionary actions would not result in substantial obstruction of public views from view corridors, designated open space areas, public roads, or public parks. New development within the community would take place within the constraints of the existing urban framework and development pattern, thereby not impacting view corridors. The policies of the proposed Uptown CPU and associated discretionary actions would enhance public view corridors through use of setbacks and design improvements along major roadways within the CPU area. Therefore, public view impacts would be less than significant, and no mitigation would be required.	None Required	Less than Significant	

Table S-1 Summary of Significant Environmental Impacts				
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
Would the project result in a substantial alteration (e.g. bulk, scale materials or style) to the existing or planned (adopted) character of the area?	The proposed Uptown CPU includes policies that would encourage residential and mixed-use development that would be consistent with the existing neighborhood character and impacts would be less than significant. No mitigation would be required.	None Required	Less than Significant	
Would the project result in the loss of any distinctive or landmark tree(s), or stand of mature trees identified in the community plan?	The implementation of the proposed Uptown CPU and associated discretionary actions would not result in the loss of any distinctive or landmark trees or any stand of mature trees; therefore no impacts would result.	None Required	Less than Significant	
Would the project result in a substantial change in the existing landform?	Implementation of the proposed Uptown CPU and associated discretionary actions would not result in significant landform alteration impacts based on the developed nature of the CPU area and compliance with existing regulations in place that would protect steep slope and canyon areas from development. The proposed Uptown CPU includes policies that would protect and preserve existing landforms (i.e., canyons and open space areas). In addition, future development would be evaluated to ensure compliance with the City's grading ordinance and significance thresholds related to grading quantities. Therefore, impacts would be less than significant and no mitigation would be required.	None Required	Less than Significant	
Would the project create substantial light or glare which would adversely affect daytime and nighttime views in the area?	Impacts relative to lighting and glare would be less than significant. No mitigation would be required.	None Required	Less than Significant	

Table S-1			
Summary of Significant Environmental Impacts			
			Impact Level
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation
Transportation and Circulation			
Would the project result in an increase in projected traffic, which is substantial in relation to the existing traffic load and capacity of the street system including roadway segments, intersections, freeway segments, interchanges, or freeway ramps?	 The Uptown CPU would result in the following cumulative impacts to intersections, roadway segments, freeway segments and ramp meters: a. Intersections Washington Street & Fourth Avenue (<i>Impact 6.3-1</i>) Washington Street & Eighth Avenue/ SR-163 Off-Ramp (<i>Impact 6.3-2</i>) Washington Street/ Normal Street & Campus Avenue/ Polk Avenue (<i>Impact 6.3-3</i>) University Avenue & Sixth Avenue (<i>Impact 6.3-4</i>) Elm Street & Second Avenue (<i>Impact 6.3-6</i>) Boadway Segments First Avenue: Washington Street to University Avenue (<i>Impact 6.3-7</i>) First Avenue: University Avenue to Robinson Avenue (<i>Impact 6.3-7</i>) First Avenue: Robinson Avenue to Grape Street (<i>Impact 6.3-7</i>) Fourth Avenue: Arbor Drive to Washington Street (<i>Impact 6.3-8</i>) Fourth Avenue: Washington Street to University Avenue (<i>Impact 6.3-7</i>) First Avenue: Washington Street to University Avenue (<i>Impact 6.3-7</i>) First Avenue: Robinson Avenue to Grape Street (<i>Impact 6.3-7</i>) Fourth Avenue: Arbor Drive to Washington Street (<i>Impact 6.3-8</i>) Fourth Avenue: Washington Street to University Avenue (<i>Impact 6.3-9</i>) Fifth Avenue: Robinson Avenue to Laurel Street (<i>Impact 6.3-10</i>) Sixth Avenue: University Avenue to Laurel Street (<i>Impact 6.3-11</i>) Sixth Avenue: Laurel Street to Elm Street (<i>Impact 6.3-11</i>) 	The following mitigation measures were identified to reduce significant impacts; however as discussed in Chapter 6.3 of this PEIR, not all measures would be feasible and only specified measures are included in the proposed Impact Fee Study (IFS), as indicated below. Intersections TRANS 6.3-1: Washington Street & Fourth Avenue (Impact 6.3-1): Widen Fourth Avenue in the southbound direction to add a second left-turn lane. Restripe the southbound approach to be two left-turn lanes, one through lane, and one right-turn lane. TRANS 6.3-2: Washington Street & Eighth Avenue/SR-163 Off-Ramp (Impact 6.3-2): Widen Washington Street in the eastbound direction to four lanes and the westbound direction to three lanes. Widen the SR-163 Off-ramp to two lanes. TRANS 6.3-3: Washington Street/Normal Street & Campus Avenue/Polk Avenue (Impact 6.3-3): Widen Washington Street in the northeast direction to add an exclusive right-turn lane. TRANS 6.3-4: University Avenue & Sixth Avenue (Impact 6.3-4): Widen Sixth Avenue in the southbound direction to add a second left-turn lane.	Significant and Unavoidable
Table S-1			
----------------------------------------------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------
Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
	 Ninth Avenue: Washington Street to University Avenue (<i>Impact 6.3-12</i>) Campus Avenue/ Polk Avenue: Washington Street to Park Boulevard (<i>Impact 6.3-13</i>) Cleveland Avenue: Tyler Street to Richmond Street (<i>Impact 6.3-14</i>) Fort Stockton Drive: Sunset Boulevard to Goldfinch Street (<i>Impact 6.3-15</i>) Grape Street: First Avenue to Third Avenue (<i>Impact 6.3-16</i>) Grape Street: Third Avenue to Sixth Avenue (<i>Impact 6.3-16</i>) Hawthorn Street: First Avenue to Third Avenue (<i>Impact 6.3-16</i>) Hawthorn Street: Third Avenue to Sixth Avenue (<i>Impact 6.3-16</i>) Hawthorn Street: Third Avenue to Sixth Avenue (<i>Impact 6.3-17</i>) Hadthorn Street: Washington Street to Winder Street (<i>Impact 6.3-17</i>) India Street: Glenwood Drive to Sassafrass Street (<i>Impact 6.3-19</i>) India Street: Columbia Street to Redwood Street (<i>Impact 6.3-20</i>) Lincoln Avenue: Washington Street to Park Boulevard (<i>Impact 6.3-21</i>) Park Boulevard: Mission Avenue to El Cajon Boulevard (<i>Impact 6.3-23</i>) Richmond Street: Cleveland Avenue to Upas Street (<i>Impact 6.3-23</i>) Richmond Street: Cleveland Avenue to Upas Street (<i>Impact 6.3-24</i>) Robinson Avenue: First Avenue to Third Avenue (<i>Impact 6.3-25</i>) 	 TRANS 6.3-5: Elm Street & Sixth Avenue (Impact 6.3-5): Widen Elm Street in the westbound direction to add a second right- turn lane. This improvement project is identified in the Uptown IFS. TRANS 6.3-6: Cedar Street & Second Avenue (Impact 6.3-6): Install a traffic signal at this intersection. This intersection is located outside the boundaries of the CPU. Roadway Segments TRANS 6.3-7: First Avenue (Impact 6.3-7) a. Washington Street to University Avenue: Restripe the roadway to a 2-lane collector with continuous left-turn lane. b. University Avenue to Robinson Avenue: Widen the roadway to a 4-lane collector with continuous left-turn lane. c. Robinson Avenue to Laurel Street: Restripe the roadway to a 2-lane collector with continuous left-turn lane. d. Laurel Street to Hawthorn Street: Restripe the roadway to a 2-lane collector with continuous left-turn lane. d. Laurel Street to Grape Street: Restripe the roadway to a 2-lane collector with continuous left-turn lane. This improvement project is identified in the Uptown IFS. e. Hawthorn Street to Grape Street: Restripe the roadway to a 2-lane collector with continuous left-turn lane. TRANS 6.3-8: Fourth Avenue from Arbor Drive to Washington Street (Impact 6.3-8): 	

Table S-1			
Summary of Significant Environmental Impacts			
Environmental Issue	Table S-1 Summary of Significant Environmental In Results of Impact Analysis • Robinson Avenue: Third Avenue to Eighth Avenue (Impact 6.3-25) • • San Diego Avenue: Hortensia Street to Pringle Street (Impact 6.3-26) • • State Street: Laurel Street to Juniper Street (Impact 6.3-27) • • University Avenue: Ibis Street to Fifth Avenue (Impact 6.3-28) • • University Avenue: Sixth Avenue to Eighth Avenue (Impact 6.3-29) • • University Avenue: Normal Street to Park Boulevard (Impact 6.3-30) • • Washington Street: Fourth Avenue to Sixth Avenue •	Mitigation Widen the roadway to a 4-lane collector with continuous left-turn lane. TRANS 6.3-9: Fourth Avenue from Walnut Avenue to Laurel Street (Impact 6.3-9): Restore the roadway to a 3-lane one-way collector for vehicles and remove the dedicated multi-modal lane. TRANS 6.3-10: Fifth Avenue from Robinson Avenue to Walnut Avenue (Impact 6.3-10): Restore the roadway to a 3-lane one-way collector for vehicles and remove the	Impact Level After Mitigation
	 Washington Street: Fourth Avenue to Sixth Avenue (<i>Impact 6.3-31</i>) Washington Street: Richmond Street to Normal Street (<i>Impact 6.3-32</i>) Freeway Segments I-5 from Old Town Avenue to Imperial Avenue (Impact 6.3-33) I-8 from Hotel Circle West to SR-15 (Impact 6.3-34) SR-15 from I-805 to SR-94 (Impact 6.3-35) I-805 from I-8 to SR-15 (Impact 6.3-36) SR-94 from 25th Street to SR-15 (Impact 6.3-37) SR-163 from I-8 to I-5 (Impact 6.3-38) d. Ramp Meters Hancock Street to I-5 southbound on-ramp in the PM peak period (6.3-40) Fifth Ave to I-5 southbound on-ramp in the PM peak period (6.3-41) 	 dedicated multi-modal lane. TRANS 6.3-11: Sixth Avenue (Impact 6.3-11) a. Washington Street to University Avenue: Widen the roadway to a 6-lane prime arterial. b. University Avenue to Laurel Street: Widen the roadway to a 4-lane major arterial. c. Laurel Street to Elm Street: Widen the roadway to a 4-lane collector. TRANS 6.3-12: Ninth Avenue from Washington Street to University Avenue (Impact 6.3-12): Restripe the roadway to a 2-lane collector with continuous left-turn lane. TRANS 6.3-13: Campus Avenue/ Polk Avenue from Washington Street to Park Boulevard (Impact 6.3-13): Restripe the roadway to a 2-lane collector with continuous left-turn lane. TRANS 6.3-14: Cleveland Avenue from Tyler Street to Richmond Street (Impact 6.3-14): 	

Table S-1			
	Summary of Significant Environmental I	mpacts	
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		Restripe the roadway to a 2-lane collector with continuous left-turn lane.	
		TRANS 6.3-15: Fort Stockton Drive from Sunset Boulevard to Goldfinch Street (impact 6.3-15): Restripe the roadway to a 2-lane collector with continuous left-turn lane.	
		TRANS 6.3-16: Grape Street from First Avenue to Sixth Avenue (Impact 6.3-16): Restripe the roadway to a 2-lane collector with continuous left-turn lane.	
		TRANS 6.3-17: Hawthorn Street from First Avenue to Sixth Avenue (Impact 6.3-17): Restripe the roadway to a 2-lane collector with continuous left-turn lane.	
		TRANS 6.3-18: India Street from Washington Street to Winder Street (Impact 6.3-18): Restripe the roadway to a 2-lane collector with continuous left-turn lane.	
		TRANS 6.3-19: India Street (Impact 6.3-19)	
		 a. Glenwood Drive to Sassafrass Street: Widen the roadway to a 4-lane one-way collector. b. Sassafrass Street to Redwood Street: Widen the roadway to a 3-lane one-way collector. 	
		TRANS 6.3-20: Laurel Street from Columbia Street to Sixth Avenue (Impact 6.3-20): Widen the roadway to a 4-lane collector.	

Table S-1				
	Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
		TRANS 6.3-21: Lincoln Avenue from Washington Street to Park Boulevard (Impact 6.3-21): Restripe the roadway to a 2-lane collector with continuous left-turn lane.		
		TRANS 6.3-22: Park Boulevard from Mission Avenue to El Cajon Boulevard (Impact 6.3-22): Widen the roadway to a 4-lane one-way collector.		
		TRANS 6.3-23: Park Boulevard from Robinson Avenue to Upas Street (Impact 6.3- 23): Widen the roadway to a 4-lane one-way collector.		
		TRANS 6.3-24: Richmond Street (Impact 6.3-24)		
		 a. Cleveland Avenue to Robinson Avenue: Restripe the roadway to a 2-lane collector with continuous left-turn lane. This improvement project is identified in the Uptown IFS. b. Robinson Avenue to Upas Street: Restripe the roadway to a 2-lane collector with continuous left-turn lane. 		
		TRANS 6.3-25: Robinson Avenue (Impact 6.3-25)		
		a. First Avenue to Third Avenue: Restripe the roadway to a 2-lane collector with continuous left-turn lane.b. Third Avenue to Eighth Avenue: Widen the roadway to a 4-lane collector.		

Table S-1			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		TRANS 6.3-26: San Diego Avenue from Hortensia Street to Pringle Street (Impact 6.3- 26): Restripe the roadway to a 2-lane collector with continuous left-turn lane.	
		TRANS 6.3-27: State Street from Laurel Street to Juniper Street (Impact 6.3- <u>2627</u>): Restripe the roadway to a 2-lane collector with continuous left-turn lane. This improvement project is identified in the Uptown IFS.	
		TRANS 6.3-28: University Avenue from Ibis Street to Fifth Avenue (Impact 6.3-28): Widen the roadway to a 4-lane collector.	
		TRANS 6.3-29: University Avenue from Sixth Avenue to Eighth Avenue (Impact 6.3-29): Widen the roadway to a 4-lane major arterial and install a raised median.	
		TRANS 6.3-30: University Avenue from Normal Street to Park Boulevard (Impact 6.3- 30): Widen the roadway to a 4-lane collector.	
		TRANS 6.3-31: Washington Street from Fourth Avenue to Sixth Avenue (Impact 6.3- 31): Widen the roadway to a 6-lane major arterial.	
		TRANS 6.3-32: Washington Street from Richmond Street to Normal Street (Impact 6.3-32): Restripe the roadway to a 6-lane prime arterial and remove on-street parking.	

1	Table S-1			
	Summary of Significant Environmental Impacts			
	Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
			Freeway Segments	
			TRANS 6.3-33: 1-5 northbound and southbound from Old Town Avenue to Imperial Avenue: <u>SANDAG's 2050 Revenue</u> Constrained Regional Transportation Plan (RTP) includes operational improvements along 1-5 between Old Town Avenue and Imperial Avenue. This project is expected to be constructed by year 2050. This measure provides partial mitigation, since it improves freeway operation in the vicinity of the project. No improvements are identified for this segment in SANDAG's San Diego Forward, The Regional Plan (RP) (Impact 6.3-	
			TRANS 6.3-34: I-8 eastbound and westbound from Hotel Circle (W) to SR-15: SANDAG's <u>2050 Revenue Constrained RP</u> T includes operational improvements along I-8 between Hotel Circle (W) and SR-15 I-5 and SR-125. This project is expected to be constructed by year 2050. This measure provides partial mitigation since it improves freeway operation in the vicinity of the project. (Impact 6.3-34)	
			TRANS 6.3-35: SR-15 northbound and southbound from I-805 to SR-94: SANDAG's <u>2050 Revenue Constrained RPTP</u> proposes the construction of managed lanes along SR- 15 between I-805 and SR-94from I-5 to I-805 and from I-8 to SR-163. Between I-8 and SR- 163, the project is expected to be constructed	

Table S-1				
	Summary of Significant Environmental Impacts			
			Impact Level	
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation	
		by 2035, between SR-94 and I-805, the project		
		is expected to be constructed by 2035, and		
		between I-5 and SR-94, the project is		
		expected to be constructed by 2050. This		
		project is expected to be constructed by year		
		2035. This measure provides partial		
		mitigation, since it reduces the traffic		
		demand on the freeway general purpose		
		lane. (Impact 6.3-35)		
		TRANS 6.3-36: I-805 northbound and		
		southbound from I-8 to SR-15: SANDAG's		
		2050 Revenue Constrained RPTP proposes		
1		the construction of managed lanes along I-		
		805 between ISR-158 and SR-1635. This		
		project is expected to be constructed by year		
		205 3 0. This measure provides partial		
		mitigation, since it reduces the traffic		
		demand on the freeway general purpose		
		lane. Additionally, Caltrans is studying buses		
		on shoulder options along the I-805 corridor		
		<u>on an interim basis. (</u> Impact 6.3-36)		
		TRANS 6.3-37: SR-94 eastbound and		
		westbound from 25th Street to SR-15:		
		SANDAG's 2050 Revenue Constrained RPTP		
		proposes the construction of managed lanes		
		along SR-94 between 25th Street and SR-15 -		
		5 and SR-125I-805. This project is expected to		
		be constructed by year 2020Between I-5 and		
		I-805, this project is expected to be		
		constructed by year 2035. In 2050, the project		
		is expected to be constructed between I-805		
		and SR-125. Caltrans is evaluating		

Table S-1				
	Summary of Significant Environmental Impacts			
			Impact Level	
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation	
		alternatives to this measure as part of the		
		environmental analysis for the SR-94 Express		
		Lanes Project. This measure (or an alternative		
		measure) would provides partial mitigation,		
		since it reduces the traffic demand on the		
		freeway general purpose lanes. (Impact 6.3-		
		37)		
		TRANS 6.3-38: SR-163 northbound from I-8		
		to Robinson Avenue and SR-163 southbound		
		from I-8 to I-5: No improvements are		
		identified for this state route segment in		
		SANDAG's 2050 R <u>PTP</u> . (Impact 6.3-38)		
1		Ramp Meters		
		TRANS 6.3-39: The City of San Diego shall		
		coordinate with Caltrans to address ramp		
		capacity at impacted on-ramp locations.		
		improvements could include additional lanes,		
		Interchange reconfiguration, etc.; nowever,		
		specific capacity improvements are suit		
		improvements that must be defined more		
		over time. Furthermore, implementation of		
		freeway improvements in a timely manner is		
		beyond the full control of the City since		
		Caltrans has approval authority over freeway		
		improvements. At the project level, significant		
		impacts at locations outside of the		
		jurisdiction of the City could be partially		
		mitigated in the form of fair share		
		contribution or TDM measures that		
		encourage carpooling and other alternative		
		means of transportation consistent with		

Table S-1				
	Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
		proposed CPU policies. Fair share contributions may be provided at the project level for impacted ramps where the impacted facility is included in the SANDAG RP; however, at this time none of the impacted ramps are included in the SANDAG RP. (Impacts 6.3-39 – 6.3-41)		
Would the project conflict with adopted policies, plans, or programs supporting alternative transportation?	The proposed Uptown CPU and associated discretionary actions would be consistent with adopted policies, plans, or programs supporting alternative transportation. The proposed Uptown CPU and associated discretionary actions would provide policies that support improvements to pedestrian, bicycle, and transit facilities. Thus, the project would have a less than significant impact related to conflicts with adopted policies, plans or programs supporting alternative transportation.	None Required	Less than Significant	
Air Quality				
Would the project conflict or obstruct implementation of the applicable air quality plan?	Future operational emissions from the build-out of the Uptown CPU would be less than anticipated for future operational emissions under the adopted community plan. Thus, emissions associated with the proposed Uptown CPU are already accounted for in the RAQS, and adoption of the proposed Uptown CPU and associated discretionary actions would not conflict with the Regional Air Quality Strategy (RAQS). Thus impacts related to conflicts with applicable air quality plans would be less than significant.	None Required	Less than Significant	
Would the project result in a violation of any air quality standard or contribute substantially to an existing or projected air quality violation?	Regarding operational emissions under Issue 2, build-out of the CPU area would exceed the City's project-level thresholds for the proposed Uptown CPU; however the Uptown CPU would emit fewer pollutants than would occur under the adopted Community Plan. Therefore, the air emissions from build-out of the proposed Uptown CPU would not increase air pollutants in the region,	None Required	Less than Significant	

Table S-1 Summary of Significant Environmental Impacts				
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
	would not further increase the frequency of existing violations of federal or state Ambient Air Quality Standards (AAQS), or would not result in new exceedances. Therefore, operational air quality impacts associated with the adoption of the proposed Uptown CPU and associated discretionary actions would be less than significant.			
Would the project expose sensitive receptors to substantial pollutant concentrations, including toxins?	Implementation of the proposed Uptown CPU and associated discretionary actions would not result in any CO hotspots. Additionally, carcinogenic risks associated with diesel fueled vehicles operating on local freeways would be less than the applicable threshold and non-carcinogenic risks from diesel particulate matter would be below the maximum chronic hazard index. Thus, air quality impacts to sensitive receptors would be less than significant and no mitigation is required	None Required	Less than Significant	
Would the project create objectionable odors affecting a substantial number of people?	Odor impacts would be less than significant as the proposed Uptown CPU and associated discretionary actions does not propose land uses associated with generation of adverse odors. No mitigation is required.	None Required	Less than Significant	
Greenhouse Gas Emissions		-		
Would the project generate GHG emissions, either directly or indirectly, that may have a significant impact on the environment?	Potential impacts related to greenhouse gas (GHG) emissions from implementation of the proposed Uptown CPU and associated discretionary actions would be less than significant as the GHG emissions from the Uptown CPU would be less than those assumed for the Uptown CPU area in the CAP GHG Inventory. Thus, the proposed Uptown CPU and associated discretionary actions would be consistent with the Climate Action Plan (CAP) and would result in a less than significant impact related to GHG emissions.	None Required	Less than Significant	

Table S-1			
	Summary of Significant Environmental Ir	npacts	
Environmental Issue Would the project conflict with an	Results of Impact Analysis The proposed Uptown CPU would implement the General Plan's	Mitigation None Required	Impact Level After Mitigation Less than
applicable plan, policy or regulation adopted for the purpose of reducing the emission of GHGs?	City of Villages Strategy and include policies for the promotion of walkability and bicycle use, polices promoting transit-supportive development, and thus, is consistent with the CAP and the General Plan. Impacts related to conflicts with applicable plans and policies addressing GHG emissions would be less than significant and no mitigation is required.		Significant
Noise			
Would the project result in or create a significant increase in the existing ambient noise levels?	An increase in ambient vehicular traffic noise in the Uptown CPU area would result from continued build-out of the proposed Uptown CPU and associated discretionary actions and increases in traffic due to regional growth. A significant increase would occur adjacent to several street segments in the Uptown CPU area that contain existing noise sensitive land uses. The increase in ambient noise levels could result in the exposure of existing noise sensitive land uses to noise levels in excess of the compatibility levels established in the General Plan, and impacts would be significant (Impact 6.6-1).	No feasible mitigation has been identified at the program level to reduce impacts 6.6-1 and 6.6-2 to less than significant.	Significant and Unavoidable
	For new discretionary development, there is an existing regulatory framework in place that would ensure future projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions would not be exposed to ambient noise levels in excess of the compatibility levels in the General Plan. Thus, noise impacts to new discretionary projects would be less than significant.		
	However, in the case of ministerial projects, there is no procedure to ensure that exterior noise would be adequately attenuated. Therefore, exterior noise impacts for ministerial projects located in areas that exceed the applicable land use and noise compatibility level would be significant and unavoidable (Impact 6.6-2).		

Table S-1			
	Summary of Significant Environmental Ir	npacts	
Environmental Issue Would the project result in an exposure of people to current or future transportation noise levels which exceed standards established in the Noise Element of the General Plan?	Summary of Significant Environmental InResults of Impact AnalysisIn the Uptown CPU area, noise levels for all land uses would be incompatible (i.e., greater than 75 A-weighted decibel [dB(A)]Community Noise Equivalent Level [CNEL]) closest to the freeways and specific segments of Sixth Avenue and Grape Street.A mitigation framework exists for new discretionary development in areas exposed to high levels of vehicle traffic noise.Implementation of the policies in the proposed Uptown CPU and General Plan would preclude or reduce traffic noise impacts because they would be required to demonstrate that exterior and interior noise levels would be compatible with City standards.	Mitigation No feasible mitigation has been identified at the program level to reduce impact 6.6-3 to less than significant as there is no mechanism to require exterior noise analysis and attenuation for these ministerial projects.	Impact Level After Mitigation Significant and Unavoidable
	Noise compatibility impacts associated with future discretionary projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions would be less than significant with implementation of existing regulations and noise standards. However, in the case of ministerial projects, there is no procedure to ensure that exterior noise is adequately attenuated. Therefore, exterior noise impacts for ministerial projects located in areas that exceed the applicable land use and noise compatibility level would be significant and unavoidable (Impact 6.6-3).		
Would the project result in the exposure of people to noise levels which exceed property line limits established in the Noise Abatement and Control Ordinance of the Municipal Code?	Mixed-use areas would contain residential and commercial interfaces. Mixed-use sites and areas where residential uses are located in proximity to commercial sites would expose sensitive receptors to noise. Although noise-sensitive residential land uses would be exposed to noise associated with the operation of these commercial uses, City policies and regulations would control noise and reduce noise impacts between various land uses. In addition, enforcement of the federal, state, and local noise regulations would control impacts. With implementation of these policies and enforcement of the Noise Abatement and Control Ordinance of the Municipal Code, impacts would be less than significant and no mitigation is required at the program level.	None Required	Less than Significant

Table S-1			
Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Impact Level Mitigation After Mitigation	
Would the project result in the exposure of people to significant temporary construction noise?	a. Construction Noise Construction activities related to implementation of the Uptown CPU and associated discretionary action would potentially generate short-term noise levels in excess of 75 dB(A) L _{eq} at adjacent properties. While the City regulates noise associated with construction equipment and activities through enforcement of noise ordinance standards (e.g., days of the week and hours of operation) and imposition of conditions of approval for building or grading permits, there is a procedure in place that allows for variance to the noise ordinance. Due to the highly developed nature of the CPU area with sensitive receivers potentially located in proximity to construction sites, there is a potential for construction of future projects to expose existing sensitive land use to significant noise levels. While future development projects would be required to incorporate feasible mitigation measures, due to the close proximity of sensitive receivers to potential construction sites, the program-level impact related to construction noise would remain significant and unavoidable (Impact 6.6-4).	 NOISE 6.6-1: At the project level, future discretionary development projects will be required to incorporate feasible mitigation measures. Typically, noise can be reduced to comply with City standards when standard construction noise control measures are enforced at the project site and when the duration of the noise-generating construction period is limited to one construction season (typically one year) or less. Construction activities shall be limited to the hours between 7:00 A.M. and 7:00 P.M. Construction is not allowed on legal holidays as specified in Section 21.04 of the San Diego Municipal Code, with exception of Columbus Day and Washington's Birthday, or on Sundays. (Consistent with Section 59.5.0404 of the San Diego Municipal Code). Equip all internal combustion enginedriven equipment with intake and exhaust mufflers that are in good condition and appropriate for the equipment. Locate stationary noise-generating equipment (e.g., compressors) as far as possible from adjacent residential receivers with temporary noise barriers. 	

Table S-1			
	Summary of Significant Environmental Ir	mpacts	
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		Utilize "quiet" air compressors and other stationary noise sources where technology exists.	
		• The contractor shall prepare a detailed construction plan identifying the schedule for major noise-generating construction activities. The construction plan shall identify a procedure for coordination with adjacent residential land uses so that construction activities can be scheduled to minimize noise disturbance.	
		• Designate a "disturbance coordinator" who would be responsible for responding to any complaints about construction noise. The disturbance coordinator will determine the cause of the noise complaint (e.g., bad muffler, etc.) and will require that reasonable measures be implemented to correct the problem.	
Would the project result in the exposure of people to significant temporary construction noise? (cont.)	b. Vibration - Construction By use of administrative controls, such as scheduling construction activities with the highest potential to produce perceptible vibration to hours with least potential to affect nearby properties, perceptible vibration can be kept to a minimum and as such would result in a less than significant impact with respect to perception. However, pile driving within 95 feet of existing structures has the potential to <u>the</u> exceed 0.20 inch per second <u>PPV threshold</u> , and <u>therefore</u> , impacts would be potentially significant (Impact 6.6-5).	NOISE 6.6-2: For discretionary projects where construction would include vibration- generating activities, such as pile driving, within 95 feet of existing structures, site- specific vibration studies shall be conducted to ensure the development project would not adversely affect adjacent properties to the satisfaction of the Chief Building Official. Such efforts shall be conducted by a qualified structural engineer and could determine the area of impact and to present appropriate	Significant and Unavoidable

Table S-1 Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		mitigation measures that may include the following:	
		 Identify sites that would include vibration compaction activities such as pile driving and have the potential to generate groundborne vibration and the sensitivity of nearby structures to groundborne vibration. This task shall be conducted by a qualified structural engineer. 	
		 Develop a vibration monitoring and construction contingency plan to identify structures where monitoring would be conducted; set up a vibration monitoring schedule; define structure-specific vibration limits; and address the need to conduct photo, elevation, and crack surveys to document before and after construction conditions. Construction contingencies would be identified for when vibration levels approach the limits. 	
		• At a minimum, mMonitor vibration during initial demolition activities and during pile-driving activities. Monitoring results may indicate the need for more or less intensive measurements.	
		When vibration levels approach limits, suspend construction and implement contingencies to either lower vibration levels or secure the affected structures.	

Table S-1 Summary of Significant Environmental Impacts			
			Impact Level
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation
		• Conduct post-survey on structures where either monitoring has indicated high levels or complaints of damage have been made. Make appropriate repairs or compensation where damage has occurred as a result of construction activities.	
Would the project result in the exposure of people to significant temporary construction noise? (cont.)	c. Vibration - Operation Post-construction operational vibration impacts could occur as a result of commercial operations that are implemented in accordance with the proposed Uptown CPU and associated discretionary actions. The commercial uses that would be constructed under the proposed Uptown CPU and associated discretionary actions would include uses such as retail, restaurants, and small offices that would not require heavy mechanical equipment that would generate groundborne vibration or heavy truck deliveries. Residential and civic uses do not typically generate vibration. Thus, operational vibration impacts associated with the proposed Uptown CPU implementation and associated discretionary actions would be less than significant. No mitigation is required.	None Required	Less than Significant
Historical Resources			
Would implementation of the proposed project result in an alteration, including the adverse physical or aesthetic effects and/or the destruction of a historic building (including an architecturally significant building), structure, object, or site?	Implementation of the proposed Uptown CPU and associated discretionary actions could result in an alteration of a historic building, structure, object, or site. This impact would be potentially significant.	HIST 6.7-1: Historic Buildings, Structures, and ObjectsPrior to issuance of any permit for a development project implemented in accordance with the proposed North ParkUptown CPU that would directly or indirectly affect a building/structure in excess of 45 years of age, the City shall determine whether the affected building/structure is	Significant and Unavoidable

Table S-1					
	Summary of Significant Environmental Impacts				
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation		
		 historically significant. The evaluation of historic architectural resources shall be based on criteria such as: age, location, context, association with an important person or event, uniqueness, or structural integrity, as indicated in the Guidelines. Preferred mitigation for historic buildings or structures shall be to avoid the resource through project redesign. If the resource cannot be entirely avoided, all prudent and feasible measures to minimize harm to the resource shall be taken. Depending upon project impacts, measures shall include, but are not limited to: Preparing a historic resource management plan; Adding new construction which is compatible in size, scale, materials, color and workmanship to the historic resource (such additions, whether portions of existing buildings or additions to historic districts, shall be clearly distinguishable from historic fabric); Repairing damage according to the Secretary of the Interior's Standards for Rehabilitation; Screening incompatible new construction from view through the use of berms, walls and landscaping in keeping with the historic period and character of the resource; and 			

	Table S-1		
	Summary of Significant Environmental I	mpacts	
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		Shielding historic properties from noise generators through the use of sound walls, double glazing and air conditioning.	
		Specific types of historical resource reports, outlined in Section III of the Historical Resources Guidelines, are required to document the methods to be used to determine the presence or absence of historical resources, to identify potential impacts from a proposed project, and to evaluate the significance of any historical resources identified. If potentially significant impacts to an identified historical resource are identified these reports will also recommend appropriate mitigation to reduce the impacts to below a level of significance, where possible. If required, mitigation programs can also be included in the report.	
		To further increase protection of potential resources – specifically potential historic districts – the City is proposing to amend the Historical Resources Regulations to include supplemental development regulations to assist in the preservation of specified potential historic districts until they can be intensively surveyed and brought forward for designation.	
Would implementation of the project result in a substantial adverse change in the significance of a prehistoric archeological resource, a religious or sacred	Implementation of the proposed Uptown CPU and associated discretionary actions could adversely impact a prehistoric archeological resource including religious or sacred use sites and human remains. This impact would be potentially significant.	HIST-6.7-2: Archaeological and Tribal Cultural Resources Prior to issuance of any permit for a future development project implemented in	Significant and Unavoidable

Table S-1				
	Summary of Significant Environmental Impacts			
				Impact Level
	Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation
1	use site, or disturbance of any		accordance with the proposed North	0
	human remains, including those		ParkUptown CPU that could directly affect an	
	interred outside of formal		archaeological or tribal cultural resource, the	
	cemeteries?		City shall require the following steps be taken	
			to determine: (1) the presence of	
			archaeological or tribal cultural resources	
			and (2) the appropriate mitigation for any	
			significant resources which may be impacted	
			by a development activity. Sites may include,	
			but are not limited to, residential and	
			commercial properties, privies, trash pits,	
			representing the contributions of people	
			from diverse socio-economic and ethnic	
			hackgrounds. Sites may also include	
			resources associated with prehistoric Native	
			American activities.	
			Initial Determination	
			The environmental analyst will determine the	
			likelihood for the project site to contain	
			historical resources by reviewing site	
			photographs and existing historic	
			information (e.g. Archaeological Sensitivity	
			Maps, the Archaeological Map Book, and the	
			City's "Historical Inventory of Important	
			Architects, Structures, and People in San	
			Diego") and may conduct a site visit, as	
			needed. If there is any evidence that the site	
			consistent with the City Guidelines would be	
			required. All individuals conducting any	

	Table S-1				
	Summary of Significant Environmental Impacts				
En	vironmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
			phase of the archaeological evaluation program must meet professional qualifications in accordance with the City Guidelines. Step 1: Based on the results of the Initial Determina- tion, if there is evidence that the site contains a historical resource, preparation of a historic evaluation is required. The evaluation report would generally include background research, field survey, archaeological testing and analysis. Before actual field reconnaissance would occur, background research is required which includes a record search at the SCIC at San Diego Museum of Man. A review of the Sacred Lands File maintained by the NAHC must also be conducted at this time. Information about existing archaeological collections should also be obtained from the San Diego Archaeologicaly Center and any tribal repositories or museums. In addition to the record searches mentioned above, background information may include, but is not limited to: examining primary sources of historical information (e.g., deeds and wills), secondary sources (e.g., local histories and genealogies), Sanborn Fire Maps, and historic cartographic and aerial photograph sources; reviewing previous archeological research in similar areas		

Table S-1					
	Summary of Significant Environmental Impacts				
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation		
	, <u>,</u>	models that predict site distribution, and archaeological, architectural, and historical site inventory files; and conducting informant interviews. The results of the background information would be included in the evaluation report.			
		Once the background research is complete, a field reconnaissance must be conducted by individuals whose qualifications meet the standards outlined in the City Guidelines. Consultants are encouraged to employ innovative survey techniques when conducting enhanced reconnaissance, including, but not limited to, remote sensing, ground penetrating radar, and other soil resistivity techniques as determined on a case-by-case basis. Native American participation is required for field surveys when there is likelihood that the project site contains prehistoric archaeological resources or traditional cultural properties. If through background research and field surveys historical resources are identified, then an evaluation of significance, based on the City Guidelines, must be performed by a qualified archaeologist.			
		Step 2 <u>:</u>			
		Where a recorded archaeological site or Tribal Cultural Resource (as defined in the Public Resources Code) is identified, the City would be required to initiate consultation with identified California Indian tribes			

Table S-1				
Summary of Significant Environmental Impacts				
			Impact Level	
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation	
		pursuant to the provisions in Public		
		Resources Code Section 21080.3.1 and		
		21080.3.2., in accordance with Assembly Bill		
		52. It should be noted that during the		
		consultation process tribal representative(s)		
		will be directly involved in making		
		recommendations regarding the significance		
		of a tribal cultural resource which also could		
		be a prehistoric archaeological site. A testing		
		program may be recommended which		
		requires reevaluation of the proposed project		
		in consultation with the Native American		
		representative which could result in a		
		combination of project redesign to avoid		
		and/or preserve significant resources as well		
		as mitigation in the form of data recovery		
		and monitoring (as recommended by the		
		qualified archaeologist and Native American		
		representative). The archaeological testing		
		program, if required will<u>shall</u> include		
		evaluating the horizontal and vertical		
		dimensions of a site, the chronological		
		placement, site function, artifact/ecofact		
		density and variability, presence/absence of		
		subsurface features, and research potential.		
		A thorough discussion of testing		
		methodologies, including surface and		
		subsurface investigations, can be found in		
		the City Guidelines. Results of the		
		consultation process will determine the		
		nature and extent of any additional		
		archaeological evaluation or changes to the		
		proposed project.		

Table S-1				
	Summary of Significant Environmental Impacts			
			Impact Level	
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation	
		The results from the testing program shall be		
		evaluated against the Significance Thresholds		
		found in the Guidelines. If significant		
		historical resources are identified within the		
		Area of Potential Effect, the site may be		
		eligible for local designation. However, this		
		process would not proceed until such time		
		that the tribal consultation has been		
		concluded and an agreement is reached (or		
		not reached) regarding significance of the		
		resource and appropriate mitigation		
		measures are identified. When appropriate,		
		the final testing report must be submitted to		
		Historical Resources Board staff for eligibility		
		determination and possible designation. An		
		agreement on the appropriate form of		
		mitigation is required prior to distribution of		
		a draft environmental document. If no		
		significant resources are found, and site		
		conditions are such that there is no potential		
		for further discoveries, then no further action		
		is required. Resources found to be non-		
		significant as a result of a survey and/or		
		assessment will require no further work		
		beyond documentation of the resources on		
		the appropriate Department of Parks and		
		Recreation (DPR) site forms and inclusion of		
		results in the survey and/or assessment		
		report. If no significant resources are found,		
		but results of the initial evaluation and		
		testing phase indicates there is still a		
		potential for resources to be present in		
		portions of the property that could not be		

Table S-1				
	Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
		tested, then mitigation monitoring is required.		
		Step 3:		
		Preferred mitigation for historical resources is to avoid the resource through project redesign. If the resource cannot be entirely avoided, all prudent and feasible measures to minimize harm shall be taken. For archaeological resources where preservation is not an option, a Research Design and Data Recovery Program is required, which includes a Collections Management Plan for review and approval. When tribal cultural resources are present and also cannot be avoided, appropriate and feasible mitigation will be determined through the tribal consultation process and incorporated into the overall data recovery program, where applicable or project specific mitigation measures incorporated into the project. The data recovery program shall be based on a written research design and is subject to the provisions as outlined in CEQA, Section 21083.2. The data recovery program must be reviewed and approved by the City's Environmental Analyst prior to distribution of a draft CEQA document and shall include the results of the tribal consultation process. Archaeological monitoring may be required during building demolition and/or construction grading when significant		
		Archaeological monitoring may be required during building demolition and/or construction grading when significant resources are known or suspected to be		

Table S-1				
	Summary of Significant Environmental Impacts			
			Impact Level	
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation	
		present on a site, but cannot be recovered	Jucci micou	
		prior to grading due to obstructions such as.		
		but not limited to, existing development or		
		dense vegetation.		
		A Native American observer must be retained		
		for all subsurface investigations, including		
		geotechnical testing and other ground-		
		disturbing activities, whenever a Native		
		American Traditional Cultural Propertytribal		
		cultural resource or any archaeological site		
		located on City property or within the Area of		
		Potential Effect of a City project would be		
		impacted. In the event that human remains		
		are encountered during data recovery and/or		
		a monitoring program, the provisions of Public		
		Resources Code Section 5097 must be		
		followed. In the event that human remains are		
		discovered during project grading, work shall		
		halt in that area and the procedures set forth		
		in the California Public Resources Code		
		(Section 50987.98) and State Health and Safety		
		Code (Section 7050.5), and in the federal, state,		
		and local regulations described above shall be		
		undertaken. These provisions will be outlined		
		in the Mitigation Monitoring and Reporting		
		Program (MMRP) included in a subsequent		
		project-specific environmental document. The		
		Native American monitor shall be consulted		
		during the preparation of the written report, at		
		which time they may express concerns about		
		the treatment of sensitive resources. If the		
		Native American community requests		
		participation of an observer for subsurface		

Table S-1			
	Summary of Significant Environmental I	mpacts	
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		investigations on private property, the request shall be honored.	
		Step 4:	
		Archaeological Resource Management reports shall be prepared by qualified professionals as determined by the criteria set forth in Appendix B of the Guidelines. The discipline shall be tailored to the resource under evaluation. In cases involving complex resources, such as traditional cultural properties, rural landscape districts, sites involving a combination of prehistoric and historic archaeology, or historic districts, a team of experts will be necessary for a complete evaluation.	
		Specific types of historical resource reports are required to document the methods (see Section III of the Guidelines) used to determine the presence or absence of historical resources; to identify the potential impacts from proposed development and evaluate the significance of any identified historical resources; to document the appropriate curation of archaeological collections (e.g. collected materials and the associated records); in the case of potentially significant impacts to historical resources, to recommend appropriate mitigation measures that would reduce the impacts to below a level of significance; and to document the results of mitigation and monitoring programs, if required.	

Table S-1			
Summary of Significant Environmental impacts			
Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
	Archaeological Resource Management reports shall be prepared in conformance with the California Office of Historic Preservation "Archaeological Resource 		
	Table S-1 Summary of Significant Environment Results of Impact Analysis	Table S-1 Summary of Significant Environmental Impacts Mitigation Archaeological Resource Management reports shall be prepared in conformance with the California Office of Historic Preservation "Archaeological Resource Management Reports: Recommended Contents and Format' (see Appendix C of the Guidelines), which will be used by Environmental staff in the review of archaeological resource reports. Consultants must ensure that archaeological resource reports are prepared consistent with this checklist. This requirement will standardize the content and format of all archaeological technical reports submitted (under separate cover) along with historical resources reports for archaeological sites and tribal cultural resource maps and records search information gathered during the background study. In addition, a Collections Management Plan shall be prepared for projects which result in a substantial collection of artifacts and must address the management and research goals of the project and the types of materials to be collected and curated based on a sampling strategy that is acceptable to the City. Appendix D (Historical Resources Report Form) may be used when no archaeological resources were identified within the project boundaries.	

Table S-1 Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
		Step 5:	
		For Archaeological Resources: All cultural materials, including original maps, field notes, non-burial related artifacts, catalog information, and final reports recovered during public and/or private development projects must be permanently curated with an appropriate institution, one which has the proper facilities and staffing for insuring research access to the collections consistent with state and federal standards, unless otherwise determined during the tribal consultation process. In the event that a prehistoric and/or historic deposit is encountered during construction monitoring, a Collections Management Plan would be required in accordance with the project MMRP. The disposition of human remains and burial related artifacts that cannot be avoided or are inadvertently discovered is governed by state (i.e., Assembly Bill 2641 [Coto] and California Native American Graves Protection and Repatriation Act of 2001[Health and Safety <u>Code 8010-8011</u>]) and federal (i.e., Native American Graves Protection and Repatriation Act [U.S.C. 3001-3013]) law, and must be treated in a dignified and culturally appropriate manner with respect for the deceased individual(s) and their descendants. Any human bones and associated grave goods of Native American origin shall be turned over to the appropriate Native American group for	
		American origin shall be turned over to the appropriate Native American group for repatriation.	

Table S-1			
	Summary of Significant Environmental I	mpacts	
			Impact Level
Environmental Issue	Results of Impact Analysis	Mitigation	After Mitigation
		Arrangements for long-term curation of all	
		recovered artifacts must be established	
		between the applicant/property owner and	
		the consultant prior to the initiation of the	
		field reconnaissance. When tribal cultural	
		resources are present, or non-burial-related	
		artifacts associated with tribal cultural	
		resources area suspected to be recovered,	
		the treatment and disposition of such	
		resources will be determined during the	
		tribal consultation process. This information	
		must then be included in the archaeological	
		survey, testing, and/or data recovery report	
		submitted to the City for review and	
		approval. Curation must be accomplished in	
		accordance with the California State Historic	
		Resources Commission's Guidelines for the	
		Curation of Archaeological Collection (dated	
		involved Title 26 of the Code of Federal	
		Pagulations, Part 70 of the Enderal Pagistor	
		Additional information regarding curation is	
		provided in Section II of the Guidelines	
		provided in Section if of the Guidelines.	
Biological Resources			
Would the project result in a	Implementation of the proposed Uptown CPU and associated	None Required	Less than
substantial adverse impact, either	discretionary actions would result in land use changes that would		Significant
directly or through habitat	affect primarily developed areas. Thus, impacts to sensitive		
modifications, on any species	species would not be anticipated to occur since any sensitive		
identified as a candidate,	species that could occur within the CPU area are likely to occupy		
sensitive, or special status species	canyon bottoms that would not be subject to development due to		
in the MSCP or other local or	their designation as Open Space and/or MHPA. Additionally, any		
regional plans, policies or	impact to sensitive vegetation communities would be subject to		

Table S-1			
	Summary of Significant Environmental Ir	npacts	
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
regulations, or by the California Department of Fish and Wildlife (CDFW) or U.S. Fish and Wildlife Service (USFWS)?	the City's ESL regulations, which would ensure any impacts to vegetation communities and potential sensitive species that may occupy those communities would addressed. Thus, based on the lack of sensitive species anticipated to occur in the developable areas of the CPU area in addition to the regulatory framework in place that protects sensitive species, impacts to wildlife species would be less than significant and no mitigation would be required.		
Would the project result in a substantial adverse impact on any Tier I Habitats, Tier II Habitats, Tier IIIA Habitats, or Tier IIIB Habitats as identified in the Biology Guidelines of the Land Development Manual or other sensitive natural community identified in local or regional plans, policies, regulations, or by the CDFW or USFWS?	Implementation of the proposed Uptown CPU and associated discretionary actions has a low potential to impact any of the five sensitive plant species previously recorded in the Uptown community. As described previously, implementation of the proposed Uptown CPU and associated discretionary actions would result in land use changes that would affect primarily developed areas. The potential for sensitive plant species to still occur is low due to the extent of development that has taken place within the CPU area and along the urban- canyon interface. Impacts to sensitive plant species would be less than significant and no mitigation would be required.	None Required	Less than Significant
Would the project result in a substantial adverse impact on wetlands (including, but not limited to, marsh, vernal pool, riparian, etc.) through direct removal, filling, hydrological interruption, or other means?	No wetland habitats have been identified within the Uptown CPU area. Thus, impacts to wetlands would be less than significant and no mitigation would be required.	None Required	Less than Significant
Would the project interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native	The proposed MHPA boundary line correction would increase the amount of protected open space in canyons, which would be beneficial for wildlife movement in canyon areas. Thus, no impact to wildlife corridors would occur. Impacts to wildlife nursery sites, particularly migratory birds,	None Required	Less than Significant

Table S-1			
	Summary of Significant Environmental Ir	npacts	
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
resident or migratory wildlife corridors, including linkages identified in the MSCP Plan, or impede the use of native wildlife nursery sites?	would be avoided through compliance with the Migratory Bird Treaty Act (MBTA) in addition to compliance with protections afforded to lands within and adjacent to MHPA lands. Development on lands adjacent to MHPA lands would be required to avoid impacts to wildlife nursery sites in adjacent habitat areas as detailed further under Issue 5 below. Thus, with the existing regulatory framework in place, potential impacts to wildlife nursery sites would be less than significant.		
Would the project conflict with the provisions of an adopted Habitat Conservation Plan, Natural Conservation Community Plan, or other approved local, regional, or State habitat conservation plan or local policy protecting biological resources, either within the MSCP plan area or in the surrounding region?	The proposed Uptown CPU and associated discretionary actions would be consistent with the City's MHPA Land Use Adjacency Guidelines and Municipal Code (Section 142.0740) requirements relative to lighting adjacent to the MHPA. Additionally, in complying with the MHPA Land Use Adjacency Guidelines requirements, landscape plans for future projects would require that grading would not impact environmental sensitive land, that potential runoff would not drain into MHPA land, require that toxic materials used on a development do not impact adjacency sensitive land, that development includes barriers that would reduce predation by domestic animals, that landscaping does not contain exotic plants/invasive species. In addition, the MHPA Land Use Adjacency Guidelines direct development so that any brush management activities are minimized within the MHPA and contains requirements to reduce potential noise impacts to listed avian species. Compliance with the City's MHPA Land Adjacency Guidelines and adherence to the policies in the Conservation Element of the Uptown CPU would reduce potential impacts of the proposed CPU to less than significant.	None Required	Less than Significant
Geologic Conditions			
Would the project expose people or structures to potential substantial adverse effects, including the risk of loss, injury, or	Based on the Geotechnical Report prepared by GEOCON, Inc., the proposed Uptown CPU and associated discretionary actions would not have direct or indirect significant environmental impacts with respect to geologic hazards, because future	None Required	Less than Significant

Table S-1			
Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
 death involving: Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault, Strong seismic ground shaking, Seismic-related ground failure, including liquefaction, or Landslides? 	development would be required to occur in accordance with the San Diego Municipal Code (SDMC) and California Building Code (CBC). This regulatory framework includes a requirement for site- specific geologic investigations to identify potential geologic hazards or concerns that would need to be addressed during grading and/or construction of a specific development project. Thus, impacts would be less than significant and no mitigation is required.		
Would the project result in substantial soil erosion or the loss of topsoil?	Adherence to the SDMC grading regulations and construction requirements and implementation of the recommendations and standards of the City's Geotechnical Study Requirements would preclude significant impacts related to erosion or loss of topsoil. Thus, impacts would be less than significant and no mitigation is required.	None Required	Less than Significant
Would the project be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse?	Future development within the Uptown CPU area would be subject to requirements of the CBC and SDMC, which include preparation of a site-specific geotechnical investigation and implementation of any geotechnical recommendations to ensure geologic instability hazards are avoided. Thus, with compliance with the CBC and SDMC, geologic instability impacts associated with future development within the Uptown CPU area would be less than significant.	None Required	Less than Significant

Table S-1			
Summary of Significant Environmental Impacts			
Environmental Issue Would the project be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risks to life or property?	Results of Impact Analysis A site-specific Geotechnical Investigation required for future projects within the CPU area would be required to identify the presence of expansive soils and provide recommendations to be implemented during grading and construction to ensure potential hazards associated with expansive soils are minimized. Thus, with implementation of the recommendations included in site-specific geotechnical investigations required under the CBC and SDMC, potential impacts associated with expansive soils would be less than significant.	Mitigation None Required	Impact Level After Mitigation Less than Significant
Paleontological Resources		1	I
Would the project result in development that requires over 1,000 cubic yards of excavation in a high resource potential geologic deposit/formation/rock unit or over 2,000 cubic yards of excavation in a moderate resource potential geologic deposit/formation/rock unit?	Because of high sensitivity for paleontological resources within the San Diego, Pomerado Conglomerate, and Mission Valley Formations, grading into these formations could potentially destroy fossil resources. Therefore, implementation of future discretionary and ministerial projects within the proposed Uptown CPU area within these formations has the potential to result in significant impacts to paleontological resources.	PALEO 6.10: Prior to the approval of subsequent discretionary development projects implemented in accordance with the proposed North ParkUptown CPU, the City shall determine the potential for impacts to paleontological resources within a high sensitivity formation based on review of the project application submitted, and recommendations of a project-level analysis completed in accordance with the steps presented below. Future projects shall be sited and designed to minimize impacts on paleontological resources in accordance with the City's Paleontological Resources Guidelines and CEQA Significance Thresholds. Monitoring for paleontological resources required during construction activities shall be implemented at the project level and shall provide mitigation for the loss of important fossil remains with future subsequent development projects that are subject to environmental review.	Discretionary Projects Less than Significant with Mitigation <u>Ministerial</u> Projects Significant and Unavoidable

Table S-1 Summary of Significant Environmental Impacts			
Environmental Issue	Summary of Significant Environmental I Results of Impact Analysis	Mitigation I. Prior to Project Approval A. The environmental analyst shall complete a project-level analysis of potential impacts on paleontological resources. The analysis shall include a review of the applicable United States Geological Survey Quad maps to identify the underlying geologic formations, and shall determine if construction of a	Impact Level After Mitigation
		 Require over 1,000 cubic yards of excavation and/or a 10-foot, or greater, depth in a high resources potential geologic deposit/formation/ rock unit. Require over 2,000 cubic yards of excavation and/or 10-foot, or greater, depth in a moderate resource potential geologic deposit/formation/rock unit. Require construction within a known fossil location or fossil recovery site. Resource potential within a formation is based on the Paleontological Monitoring Determination Matrix. B. If construction of a project would occur within a formation with a moderate to high resource potential, monitoring during construction would be required. 	

Table S-1 Summary of Significant Environmental Impacts			
Environmental Issue	Table S-1 Summary of Significant Environmental In Results of Impact Analysis	 Mitigation Monitoring is always required when grading on a fossil recovery site or a known fossil location. Monitoring may also be needed at shallower depths if fossil resources are present or likely to be present after review of source materials or consultation with an expert in fossil resources (e.g., the San Diego Natural History Museum). Monitoring may be required for shallow grading (<10 feet) when a site has previously been graded, and/or unweathered geologic deposits/formations/rock units are present at the surface. Monitoring is not required when 	Impact Level After Mitigation
Hydrology and Water Quality		 Wontoring is not required when grading documented artificial fill. When it has been determined that a future project has the potential to impact a geologic formation with a high or moderate fossil sensitivity rating, a Paleontological Mitigation Monitoring and Report Program shall be implemented during construction grading activities. 	
Would the project result in flooding due to an increase in impervious surfaces, changes in absorption rates, drainage patterns, or the rate of surface runoff?	All development is subject to drainage and floodplain regulations in the SDMC and would be required to adhere to the City's Drainage Design Manual and Storm Water Standards Manual. Therefore, with future development, the volume and rate of overall surface runoff within the proposed Uptown CPU and associated discretionary actions would either remain the same as	None Required	Less than Significant

Table S-1			
	Summary of Significant Environmental Ir	npacts	
Environmental Issue	Results of Impact Analysis the existing condition or would be reduced when compared to the existing condition. Impacts would be less than significant and mitigation is not required.	Mitigation	Impact Level After Mitigation
Would the project result in an increase in pollutant discharge to receiving waters and increase discharge of identified pollutants to an already impaired water body?	New development under the proposed Uptown CPU and associated discretionary actions would be required to implement LID and storm water BMPs into project design to address the potential for transport of pollutants of concern through either retention or filtration. The implementation of LID design and storm water BMPs would reduce the amount of pollutants transported from Uptown to receiving waters. Impacts would be less than significant and no mitigation would be required.	None Required	Less than Significant
	Future development would adhere to the requirements of the MS4 permit for the San Diego Region and the City's Storm Water Standards Manual, water quality conditions, both surface and groundwater, are not expected to have an adverse effect on water quality. Additionally, the City has adopted the Master Storm Water Maintenance Program to address flood control issues by cleaning and maintaining the channels to reduce the volume of pollutants that enter the receiving waters. Impacts would be less than significant, and no mitigation would be required.		
Would the project deplete groundwater supplies, degrade groundwater quality, or interfere with ground water recharge?	Groundwater within the San Diego Mesa is exempt from municipal and domestic supply beneficial use and does not support municipal and domestic supply. Groundwater within the Mission San Diego area of the Lower San Diego portion of the San Diego Hydrologic Unit has a potential beneficial use for municipal and domestic supply. Storm water regulations that encourage infiltration of storm water runoff and protection of water quality would also protect the quality of groundwater resources and support infiltration where appropriate. Thus, implementation of the proposed Uptown CPU and associated discretionary actions	None Required	Less than Significant
Table S-1 Summary of Significant Environmental Impacts			
-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	---------------	----------------------------------
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
	would result in a less than significant impact on groundwater supply and quality.		
Public Services and Facilities			
Public Services and Facilities Would the project promote growth patterns resulting in the need for and/or provision of new or physically altered public facilities (including police protection, parks or other recreational facilities, fire/life safety protection, libraries, schools, or maintenance of public facilities including roads), the construction of which could cause significant environmental impacts in order to maintain service ratios, response times, or other performance objectives?	 Police Protection Regarding police protection, the proposed Uptown CPU and associated discretionary actions do not include construction of new police facilities. As population growth occurs and the need for new facilities is identified, any future construction of police facilities would be subject to a separate environmental review at the time design plans are available. Therefore, implementation of the proposed Uptown CPU and associated discretionary actions would result in less than significant environmental impacts associated with the construction of new facilities in order to maintain service ratios, response times, or other performance objectives related to police services, and no mitigation is required. Park and Recreation Regarding park and recreational facilities, there is an existing and projected deficit in population based parks, which is an adverse impact, but not considered significant at the program level. Implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of population based parks in the CPU area, but does not propose construction of new facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to 	None Required	Less than Significant
	parks and recreation, and no mitigation is required.		
	Fire/Life Safety Protection		
	Regarding fire/life safety protection, implementation of the proposed Uptown CPU and associated discretionary actions would result in an increase in overall population which could		

Table S-1 Summary of Significant Environmental Impacts			
			Impact Level
Environmental Issue	Results of Impact Analysis result in a change in fire-rescue response times and a demand for new or expanded facilities. However, any expansion construction of existing facilities or the development of a new facility would be subject to separate environmental review at the time design plans are available. Therefore, at the impacts associated with police/life safety facilities would be less than significant, and no mitigation is required.	Mitigation	After Mitigation
Would the project promote growth patterns resulting in the need for and/or provision of new or physically altered public facilities (including police protection, parks or other recreational facilities, fire/life safety protection, libraries, schools, or maintenance of public facilities including roads), the construction of which could cause significant environmental impacts in order to maintain service ratios, response times, or other performance objectives? (cont.)	Libraries Although a new library is planned for the Uptown CPU area, the proposed Uptown CPU and associated discretionary actions does not include construction of library facilities. Development of a new facility would be subject to separate environmental review at the time design plans are available. Therefore, impacts related to library facilities would be less than significant, and no mitigation is required. Schools Regarding school facilities, future residential development that occurs in accordance with the proposed Uptown CPU and associated discretionary actions would be required to pay school fees as outlined in Government Code Section 65995, Education Code Section 53080, and Senate Bill 50 to mitigate any potential impact on district schools. The City is legally prohibited from imposing any additional mitigation related to school facilities through implementation of Senate Bill 50, and the school district would be responsible for potential expansion or development of new facilities. Therefore, impacts to schools would be less than significant, and no mitigation is required. The proposed Uptown CPU contains policies to address the maintenance and improvement of public facilities. Impacts would therefore be less than significant, and no mitigation is required.	None Required	Less than Significant

Table S-1				
	Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation	
Public Utilities				
Would the project use excessive amounts of water beyond projected available supplies?	Based on the findings of the Water Supply Assessment (WSA), there is sufficient water supply to serve existing and projected demands of the proposed Uptown CPU and associated discretionary actions, and future water demands within the Public Utilities Department's (PUD's) service area in normal and dry year forecasts during a 20-year projection. Therefore, no significant impacts to water supply are anticipated for the implementation of the CPU.	None Required	Less than Significant	
Would the project promote growth patterns resulting in the need for and/or provision of new or physically altered utilities, the construction of which could cause significant environmental impacts in order to maintain service ratios, or other performance objectives?	Storm Water Future projects would be required to exercise strict adherence to Subsequent projects would be subject to existing regulations in place at the time projects are implemented existing storm water regulations and conformance with General Plan and Uptown CPU policies. Project-specific review under CEQA and storm water regulations in place at the time future projects are proposed would assure that significant adverse effect to the City's storm water system, as well as significant impacts associated with the installation of storm water infrastructure, would be avoided.	None Required	Less than Significant	
	Sewer and Water Distribution			
	The proposed Uptown CPU acknowledges that upgrades to sewer lines are an ongoing process. These upgrades are administered by the Public Works Department (PWD) and are handled on project-by-project basis. Because future development of properties with the proposed Uptown CPU and associated discretionary projects would likely increase demand, there may be a need to increase sizing of existing pipelines and mains for both wastewater and water. The proposed Uptown CPU takes into consideration the existing patterns of development, and the update is a response to the community's needs and goals for the future. The necessary infrastructure improvements to storm			

Table S-1			
Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
	water, wastewater, and water infrastructure would be standard practice for new development to maintain or improve the existing system in adherence to sewer and water regulations and conformance with General Plan and proposed Uptown CPU policies. Additionally, <u>subsequent projects would be subject to</u> <u>existing regulations in place at the time projects are</u> <u>implemented</u> future discretionary projects would be required to <u>undergo project-specific review_under CEQA_</u> that would <u>asen</u> sure that-impacts associated with the installation of storm water infrastructure would be reduced to below a level of significance. Therefore, impacts to sewer and water utilities would be less than significant.		
	Communications		
	Given the number of private utility providers available to serve the proposed Uptown CPU area there is capacity to serve the area. Impacts would be less than significant.		
Would the project result in impacts to solid waste management, including the need for construction of new solid waste landfills; or result in a land use plan that would not promote the achievement of a 75 percent waste diversion as targeted in AB 341 and the City's Climate Action Plan?	To ensure waste generation and recycling efforts during construction and post-construction future land use occupancy and operation (i.e., residential, commercial, industrial, mixed-use, etc.) are addressed, a Waste Management Plan (WMP) shall be prepared for any project proposed under the proposed Uptown CPU and associated discretionary actions exceeding the threshold of 40,000 square feet or more. Implementation of these WMPs would ensure that future development project impacts would be considered less than significant. Non-discretionary projects proposed under the proposed Uptown CPU and discretionary actions, and discretionary projects that would fall below the 60 ton thresholds, would be required to comply with the San Diego Municipal Code sections addressing construction and demolition debris, waste and recyclable materials storage, and recyclable materials (and in the future organic materials) collection. Therefore, at this program level of review, the proposed Uptown	None Required	Less than Significant

Table S-1 Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
	CPU and associated discretionary actions would not require increased landfill capacity, and impacts associated with solid waste would be less than significant.		
Health and Safety			
Would the project expose people or structures to a significant risk of loss, injury, or death involving wildland fires, including when wildlands are adjacent to urbanized areas or where residences are intermixed with	Existing policies and regulations would help reduce, but not completely abate, the potential risks of wildland fires. The General Plan and CPU contain goals and policies to be implemented by the City's Fire-Rescue Department, and through land use compatibility, training, sustainable development, and other measures, these goals and policies are aimed at reducing the risk of wildland fires.	None Required	Less than Significant
wildlands?	Continued monitoring and updating of existing development regulations and plans also would assist in creating defensible spaces and reduce the threat of wildfires. Public education, firefighter training, and emergency operations efforts would reduce the potential impacts associated with wildfire hazards. Additionally, future development would be subject to conditions of approval that require adherence to the City's Brush Management Regulations and requirements of the California Fire Code. As such, impacts relative to wildland fire hazard would be less than significant		
Would the project result in hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within a quarter-mile of an existing or proposed school?	The proposed Uptown CPU and associated discretionary actions would not result in hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within a quarter-mile of and existing or proposed school. Impacts to schools would be less than significant. No mitigation is required.	None Required	Less than Significant

Table S-1 Summary of Significant Environmental Impacts			
Environmental Issue	Results of Impact Analysis	Mitigation	Impact Level After Mitigation
Would the project impair implementation of, or physically interfere with, an adopted emergency response plan or emergency evacuation plan?	The proposed Uptown CPU would not impair implementation of, or physically interfere with, an adopted emergency response plan or emergency evacuation plan; therefore, impacts are less than significant, and no mitigation would be required.	None Required	Less than Significant
Would the project be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, creates a significant hazard to the public or environment?	Although there are closed Leaking Underground Storage Tank (LUST) and Cleanup Program sites and there is one open LUST and two open Cleanup Program sites within the Uptown community, there are local, State, and Federal regulations and programs in places that minimize the risk to sensitive receptors on or adjacent to hazardous materials sites. Adherence to these regulations would result in less than significant impacts relative to hazardous materials sites and no mitigation is required.	None Required	Less than Significant
Would the project expose people or structures to a significant risk of loss, injury or death from off- airport aircraft operational accidents?	Impacts relative to safety hazards related to being located within an airport influence area less than significant. No mitigation is required.	None Required	Less than Significant

1

Chapter 1.0 Introduction

This draft Program Environmental Impact Report (PEIR) for the proposed Uptown Community Plan Update (proposed CPU or project area) and other associated discretionary approvals (collectively referred to throughout this PEIR as the project) has been prepared on behalf of the City of San Diego (City) in compliance with the California Environmental Quality Act (CEQA) Statute and Guidelines (Public Resources Code, Section 21000 et seq. and California Code of Regulations, Title14, Section 15000, et seq.) and in accordance with the City's Environmental Impact Report Guidelines (EIR Guidelines; City of San Diego 2005) and the City's California Environmental Quality Act Significance Determination Thresholds (Significance Determination Thresholds) (2011).

The project analyzed within this PEIR includes a number of legislative actions to be considered by the City Council but primarily is a comprehensive update of the 1988 Uptown Community Plan. The proposed Uptown CPU reflects citywide policies and programs developed in the City of San Diego General Plan Update of 2008 (General Plan) and are consistent with the General Plan for the proposed CPU area. The proposed Uptown CPU contains nine elements, as well as an Introduction and Implementation section. The elements are as follows: Land Use; Mobility; Urban Design; Economic Prosperity; Public Facilities, Services and Safety; Recreation; Conservation; Noise; and Historic Preservation.

The proposed Uptown CPU contains a specific vision embodied in its guiding principles, as well as key goals. The proposed Uptown CPU contains development design guidelines, as well as policies related to a range of topics included in each section such as mobility options, environmental conservation, recreation opportunities, neighborhood character, and historic preservation, in accordance with the general goals stated in the General Plan. The proposed CPU serves as the basis for guiding a variety of other future implementing actions, such as parkland acquisitions and mobility options.

1.1 PEIR Purpose and Intended Uses

In accordance with the CEQA Guidelines Section 15121, the purpose of this PEIR is to provide public agency decision-makers and members of the public with detailed information about the potential significant environmental effects of the project, possible ways to minimize its significant effects, and reasonable alternatives that would reduce or avoid any identified significant effects. This PEIR is informational in nature and is intended for use by decision-makers, Responsible or Trustee Agencies as defined under CEQA, other interested agencies or jurisdictions; and the general public. The PEIR includes recommended mitigation measures which, when implemented, would lessen project impacts and provide the City, the Lead Agency as defined in Article 4 of the CEQA Guidelines (Sections 15050 to 15051), with ways to substantially lessen or avoid significant effects of the project on the environment, whenever feasible. Alternatives to the proposed CPU are presented to evaluate alternative land use scenarios, policies, and/or regulations that would further reduce or avoid significant impacts associated with the proposed CPU and associated discretionary actions.

In accordance with CEQA Guidelines Section 15168, a PEIR may serve as the EIR for subsequent activities or implementing actions, including future development of public and private projects, to the extent it contemplates and adequately analyzes the potential environmental impacts of those subsequent projects. If, in examining future actions for development within the CPU area, the City finds no new effects could occur, or no new mitigation measures would be required other than those analyzed and/or required in the PEIR, the City can approve the activity as being within the scope covered by this PEIR, and no new environmental documentation would be required. If additional analysis is required, it can be streamlined by tiering from this PEIR pursuant to CEQA Guidelines, Sections 15152, 15153, and 15168 (e.g., through preparation of a Mitigated Negative Declaration, Addendum, or EIR).

1.2 PEIR Legal Authority

1.2.1 Lead Agency

The City of San Diego is the Lead Agency for the project pursuant to Article 4 (Sections 15050 and 15051) of the CEQA Guidelines. The Lead Agency, as defined by CEQA Guidelines Section 15367, is the public agency which has the principal responsibility and authority for carrying out or approving a project. On behalf of the Lead Agency, the City's Planning Department, Environmental Analysis Section, since reorganized under the Environmental and Planning Analysis Division of the Planning Department, conducted a preliminary review of the project and decided that an EIR was required. The analysis and findings in this document reflect the independent, impartial conclusions of the City.

1.2.2 Responsible and Trustee Agencies

State law requires that all EIRs be reviewed by Responsible and Trustee Agencies. A Responsible Agency, defined pursuant to CEQA Guidelines Section 15381, includes all public agencies other than the Lead Agency which have discretionary approval power over the project. A Trustee Agency is defined in Section 15386 of the CEQA Guidelines as a state agency having jurisdiction by law over

natural resources affected by a project that are held in trust for the people of the state of California. Implementation of the project would require subsequent actions or consultation from Responsible or Trustee Agencies. A brief description of some of the primary Responsible or Trustee Agencies that may have an interest in the project is provided below.

1.2.2.1 U.S. Army Corps of Engineers

The U.S. Army Corps of Engineers (USACE) has jurisdiction over development in or affecting the navigable waters of the United States. All permits issued by the USACE are subject to consultation and/or review by the United States Fish and Wildlife Service (USFWS) and the United States Environmental Protection Agency (EPA). Drainages and canyons occurring in the CPU area may contain streams and wetlands, which may be classified as jurisdictional waters of the United States. No permits from USACE are required at this time; however, future development projects, particularly improvements to infrastructure such as water and sewer lines that could occur with implementation of the proposed CPU and associated discretionary actions, may require review and/or USACE permits in the future.

1.2.2.2 California Department of Transportation

The proposed CPU area is adjacent to the California Department of Transportation (Caltrans) facilities, including Interstate 5 (I-5), State Route 163 (SR-163), and SR-94. No permits from Caltrans are required at this time; however, Caltrans approval would be required for any encroachments or construction of facilities in a Caltrans right-of-way associated with future projects within the CPU area.

1.2.2.3 California Department of Fish and Wildlife

An Agreement Regarding Proposed Stream or Lake Alteration (Streambed Alteration Agreement) with an agency or private party proposing to alter the bed, banks, or floor of any watercourse/stream, is under the authority of the California Department of Fish and Wildlife (CDFW) pursuant to Section 1600 et seq. of the State Fish and Game Code. The purpose of code Sections 1600-1616 is to protect and conserve fish and wildlife resources that could be substantially adversely affected by a substantial diversion or obstruction of natural flow of, or substantial change or use of material from the bed, bank, or channel of, any river, stream, or lake. Drainages and canyons occurring in the CPU area may contain streams and wetlands. No permits from CDFW are required at this time; however, development projects, particularly improvements to infrastructure such as water and sewer lines that could occur with implementation of the proposed CPU and associated discretionary actions, may require review and/or Streambed Alteration Agreements in the future.

1.2.2.4 San Diego Regional Water Quality Control Board

The San Diego Regional Water Quality Control Board (RWQCB) regulates water quality through the Federal Clean Water Act Section 401 certification process and oversees the National Pollutant Discharge Elimination System (NPDES) Permit No. CAS0109266, which consists of wastewater

discharge requirements, as well as Waste Discharge Requirements Program, which regulates point discharges not subject to the Federal Water Pollution Control Act Amendments. The RWQCB is responsible for implementing permitting, compliance, and other activities to reduce pollutants in municipal, construction, and industrial storm water runoff, including overseeing the development and implementation of Water Quality Improvement Plans as required by the Regional MS4 Permit for parts of the San Diego region, which includes the City, as well as ensuring that all other MS4 permit requirements are met. No permits from RWQCB are required at this time; however, future development projects within the proposed CPU area may require review and/or Section 401 certifications.

1.2.2.5 San Diego County Regional Airport Authority

The San Diego County Regional Airport Authority (Airport Authority) operates the San Diego International Airport (SDIA). The Airport Authority also serves as San Diego County's Airport Land Use Commission (ALUC) and is responsible for land use planning as it relates to public safety surrounding the region's airports. As a Responsible Agency, the Airport Authority, acting as the ALUC, would review future development proposals within the proposed CPU area and make "consistency determinations" with the provisions and policies set forth in the SDIA Airport Land Use Compatibility Plan (ALUCP) up until the time the ALUC determines the CPU and zoning consistent with the ALUCP for SDIA. Future development projects within the CPU area would be subject to the noise, safety, overflight, and airspace protection policies in the ALUCP for SDIA, which also include the Code of Federal Regulations, Part 77 requirement to provide notification to Federal Aviation Administration (FAA) as addressed in the ALUCP for SDIA.

1.3 EIR Type, Scope and Content, and Format

1.3.1 Type of EIR

This EIR has been prepared as a Program EIR (PEIR), as defined in Section 15168 of the CEQA Guidelines. In accordance with CEQA, this PEIR examines the environmental impacts of the proposed CPU, which are comprised of a series of actions. The combined actions can be characterized as one large project for the purpose of environmental review in this PEIR and are herein collectively referred to as the "proposed CPU or the project." The PEIR focuses on the physical changes in the environment that would result from adoption and implementation of the proposed CPU and other associated discretionary actions described in Chapter 3.0, Project Description, including anticipated general impacts that could result during future construction and operation.

1.3.2 PEIR Scope and Content

The scope of analysis for this PEIR was determined by the City as a result of initial project review, as well as consideration of comments received in response to the Notice of Preparation (NOP) circulated December 23, 2013, and a scoping meeting held on January 9, 2014, at Balboa Park (Santa Fe Room), 2150 Pan American Road, San Diego, California 92101. The NOP for analysis of the project, related letters received, and comments made during the scoping meeting are included as

Appendix A of this PEIR. Through these scoping activities, the project was determined to have the potential to result in significant environmental impacts to the following subject areas:

- Land Use
- Visual Effects and Neighborhood Character
- Transportation and Circulation
- Air Quality
- Greenhouse Gas Emissions
- Noise
- Historical Resources

- Biological Resources
- Geologic Conditions
- Paleontological Resources
- Hydrology/Water Quality
- Public Services and Facilities
- Public Utilities
- Health and Safety

It should be noted that the NOP for the PEIR included the project as well as the proposed Community Plan Updates for the North Park and Golden Hill community plan areas. As a result of timing related to stakeholder input, the environmental analysis for the Uptown CPU was separated from the analysis of the North Park and Golden Hill Community Plan Updates. The North Park and Golden Hill Community Plan Updates. The North Park and Golden Hill Community Plan Updates. The North Park and Golden Hill Community Plan Updates are analyzed in a separate PEIR circulated for public review from May 31, 2016, to July 29, 2016. The State Clearinghouse number assigned with issuance of the NOP (SCH #2013121076) is being used for the North Park/Golden Hill CPU PEIR and a new State Clearinghouse number will be assigned for the Uptown CPU PEIR at the start of public review.

The intent of this PEIR is to determine whether implementation of the proposed CPU and associated discretionary actions would have a significant effect on the environment through analysis of each issue identified during the scoping process. The Environmental Analysis for the proposed Uptown CPU and associated discretionary actions is presented in the Environmental Analysis section in this PEIR (Chapter 6.0). Each environmental issue area presented in this chapter includes presentation of threshold(s) of significance for the particular issue area under evaluation based on the CEQA Guidelines and the City's Significance Determination Thresholds (2011); identification of an issue statement; an assessment of any impacts including cumulative impacts; a summary of any project impacts; and recommendations for mitigation measures and mitigation monitoring and reporting, as appropriate, for each significant issue area.

Pursuant to CEQA Guidelines Section 15126, all phases, or in the case of this project, discretionary actions associated with the proposed CPUs are considered in this PEIR when evaluating potential impacts on the environment, including the construction of future development and operational phases to the extent possible at the program-level. Impacts are identified as direct or indirect, short-term or long-term, and are assessed on a plan-to-ground basis. The plan-to-ground analysis addresses the changes or impacts that would result from implementation of the proposed CPU and associated discretionary actions compared to existing ground conditions. The proposed CPU is also compared with the current Community Plan to provide context and background for the analysis.

The PEIR includes all mandatory contents of EIRs as required pursuant to CEQA Guidelines Sections 15120 to 15132. A Cumulative Impacts analysis is presented within each specific environmental issue area of Chapter 6.0. Chapter 7.0, Effects Found Not to Be Significant, presents a brief discussion of environmental effects that were evaluated as part of the initial scoping and review process for the project and were found not to be potentially significant. Chapter 8.0 presents a discussion of Growth Inducement, and Chapter 9.0 presents a discussion of Significant Unavoidable Impacts, Significant Irreversible Environmental Changes, and Energy Conservation.

Chapter 10.0 of this PEIR includes a discussion of Alternatives that could avoid or reduce potentially significant environmental effects associated with implementation of the proposed CPU and associated discretionary actions. Alternatives discussed in the PEIR include the No Project (Adopted Community Plan) Alternative, Adopted Community Plan with Removal of Interim Height Ordinance Alternative, Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative, Density Redistribution Alternative, and the Lower-Density Alternative. For the purposes of this PEIR, the No Project Alternative would be the continued implementation of the adopted Community Plan with the same land uses as identified in that Community Plan.

1.3.3 PEIR Format

The format and order of contents of this PEIR follow the direction in the City's EIR Guidelines. A brief overview of the various chapters of this PEIR is provided below:

- Executive Summary (CEQA Guidelines Section 15123). Provides a summary of the PEIR, a brief description of the project, identification of areas of controversy, issues to be resolved by the decision-makers, and inclusion of a summary table identifying significant impacts, proposed mitigation measures, and significance of impact after mitigation. A summary of the project alternatives and comparison of the potential impacts of the alternatives with those of the project is also provided.
- Chapter 1.0, Introduction. Contains an overview of the legal authority, purpose, and intended uses of the PEIR, as well as its scope and content.
- Chapter 2.0, Environmental Setting (CEQA Guidelines Section 15125). Provides a description of the project's regional context, location, and existing physical characteristics and land use within the proposed CPU area. An overview of available public infrastructure and services, as well as relationship to relevant plans, is also provided in this section. The Environmental Setting chapter is detailed, providing background information relevant to each environmental issue area further addressed in Chapter 6.0. Within the CPU impact analysis chapter, the applicable environmental setting discussion contained in Chapter 2.0 is referenced to avoid repetition.
- Chapter 3.0, Project Description (CEQA Guidelines Section 15124). Provides a detailed discussion of the project, including background, objectives, key features, and environmental design considerations.
- Chapter 4.0, History of Project Changes. Provides a summary of the process of developing the proposed CPU.
- Chapter 5.0, Regulatory Setting. Originally the PEIR included analysis of three CPUs (Uptown, North Park, and Golden Hill). This chapter was written to reduce the amount of redundant description of the regulations associated with individual environmental topics that would be the same for each CPU area (e.g., noise regulations). While the Uptown CPU is now a

separate document, this chapter has been retained. Within the CPU impact analysis chapter (Chapter 6.0), the applicable regulatory setting discussion contained in Chapter 5.0 is referenced.

- Chapter 6.0, Environmental Analysis (CEQA Guidelines Section 15126). This chapter provides a detailed community-specific evaluation of potential environmental impacts associated with the project for environmental issues determined through the initial review and public scoping processes to be potentially significant. Chapter 6.0 begins with the issue of land use, followed by the remaining issues in order of significance. The analysis of each issue begins with a reference to the environmental setting and regulatory framework provided in Chapters 2.0 and 5.0, respectively, and a statement of specific thresholds used to determine significance of impacts, followed by an evaluation of potential impacts, including cumulative impacts. If significant impacts are identified, feasible mitigation measures to avoid or reduce any significant impacts are identified. Where mitigation measures are required, a statement regarding the significance of the impact after mitigation is provided.
- Chapter 7.0, Effects Found Not to Be Significant. Identifies all of the issues determined in the scoping and preliminary environmental review process to be not significant for the proposed CPU and associated discretionary actions, and briefly summarizes the basis for these determinations. For the project, it was determined that environmental issues associated with agriculture, mineral resources, and population and housing would not be significant, and, therefore, are summarized in Chapter 7.0.
- Chapter 8.0, Growth Inducement (CEQA Guidelines Section 15126.2(d)). Evaluates the potential influence the proposed CPU and associated discretionary actions may have on economic or population growth within the proposed CPU area, as well as the region, either directly or indirectly.
- Chapter 9.0, Significant Unavoidable Impacts/Significant Irreversible Environmental Changes/Energy Conservation (CEQA Guidelines Section 15126(b), 15126(c), and 15126.4 (a)(1)) provides a summary of any significant unavoidable impacts of the project as detailed in Chapter 6.0. This chapter also describes the potentially significant irreversible changes that may be expected and addresses the use of nonrenewable resources and energy use anticipated during project implementation.
- Chapter 10.0, Alternatives (CEQA Guidelines Section 15126.6). Provides a description of alternatives to the project, including the No Project (Adopted Community Plan) Alternative, Adopted Community Plan with Removal of Interim Height Ordinance Alternative, Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative, Density Redistribution Alternative, and the Lower-Density Alternative.
- Chapter 11.0, Mitigation Monitoring and Reporting Program. Documents all the mitigation measures identified in the PEIR for the project.
- Chapter 12.0, References. Lists all of the reference materials cited in the PEIR.

- Chapter 13.0, Individuals and Agencies Consulted (CEQA Guidelines Section 15129). Identifies all of the individuals and agencies contacted during preparation of the PEIR.
- Chapter 14.0, Certification. Identifies all of the agencies, organizations, and individuals responsible for the preparation of the PEIR.

Technical reports, used as a basis for much of the environmental analysis in the PEIR, have been summarized in the PEIR, and are included as appendices to this PEIR. The technical reports prepared for the project and their location in the PEIR are listed in the table of contents. Availability of the Draft PEIR and the technical appendices is discussed in Section 1.4.1, Draft PEIR.

1.3.4 Incorporation by Reference

As permitted by CEQA Guidelines Section 15150, this PEIR has referenced several technical studies and reports. Information from these documents has been briefly summarized in this PEIR, and their relationship to this PEIR is described. These documents are included in Chapter 12.0, References, are hereby incorporated by reference, and are available for review at the City Planning Department, located at 1010 Second Avenue, Suite 1200, San Diego, California 92101. Included within the list of materials incorporated by reference into this PEIR are the following:

- City of San Diego General Plan (City of San Diego 2008)
- City of San Diego Program Environmental Impact Report for the General Plan (Final PEIR) (City of San Diego 2007)
- City of San Diego Housing Element FY2013-FY2020 (City of San Diego 2013)
- City of San Diego Municipal Code (City of San Diego 2008)
- City of San Diego Uptown Community Plan, as amended (City of San Diego 1988)
- Uptown Community Plan Area Historic Resources Survey (City of San Diego 2015)

1.4 PEIR Process

The City, as Lead Agency, is responsible for the preparation and review of this PEIR. The PEIR review process occurs in two basic stages. The first stage is the Draft PEIR, which offers the public the opportunity to comment on the document, while the second stage is the Final PEIR.

1.4.1 Draft PEIR

In accordance with the City's Municipal Code Section 128.0306 and CEQA Guidelines Section 15105, the Draft PEIR is distributed for review to the public and interested and affected agencies for a review period of 45 days. The purpose of the review period is to allow the public an opportunity to provide comments "on the sufficiency of the document in identifying and analyzing the possible impacts on the environment and ways in which the significant effects of the project might be

avoided and mitigated" (Section 15204, CEQA Guidelines). City Municipal Code Section 128.0307 allows the Planning Director to approve requests for additional public review time from the affected officially recognized community planning group, in this case the Uptown Community Planning Group. Approval of additional review time shall not exceed 14 calendar days. The Uptown Planning Group has requested additional public review time and the request has been granted by the Planning Director.

In accordance with Sections 15085 and 15087 (a) (1) of the CEQA Guidelines, upon completion of the Draft PEIR, a Notice of Completion is filed with the State Office of Planning and Research and Notice of Availability of the Draft PEIR issued in the *San Diego Daily Transcript*, a newspaper of general circulation in the area.

The Draft PEIR and all related technical studies are available for review during the public review period at the offices of the Planning Department, located at 1010 Second Avenue, Suite 1200, San Diego, California 92101, and on the Planning Department website for CEQA Policy and Review:

http://www.sandiego.gov/planning/programs/ceqa/

The Uptown Community Plan Update website is:

http://www.sandiego.gov/planning/community/profiles/uptown/index.shtml

Electronic copies of the Draft PEIR are also available at the following public libraries:

San Diego Central Library	University Heights Branch Library
330 Park Boulevard	4193 Park Boulevard
San Diego, California 92101	San Diego, California 92103

Mission Hills Branch Library 925 West Washington Street San Diego, California 92103 North Park Branch Library 3795 31st Street San Diego, California 92104

1.4.2 Final PEIR

Following the end of the public review period, the City, as Lead Agency, will provide written responses to comments received on the Draft PEIR per CEQA Guidelines Section 15088. All comments and responses will be considered in the review of the PEIR. Detailed responses to the comments received during public review, a Mitigation Monitoring and Reporting Program (MMRP), Findings of Fact, and a Statement of Overriding Considerations for impacts identified in the PEIR as significant and unavoidable, will be prepared and compiled as part of the PEIR finalization process. The culmination of this process is a public hearing where the City Council will determine whether to certify the Final PEIR, which includes the MMRP, Findings, and Statement of Overriding Considerations, as being complete and in accordance with CEQA. The Final PEIR will be available for public review at least 14 days before the City Council public hearing in order to provide commenters the opportunity to review the written responses to their comment letters.

2

Chapter 2.0 Environmental Setting

At the time of the release of the Notice of Preparation (NOP), the Program Environmental Impact Report (PEIR) was to discuss the potential impacts of implementing three specific Community Plan Updates (CPUs; i.e., Uptown, North Park, and Golden Hill). Because the three Community Plan areas are adjacent to each other, many topics typically discussed as part of the Environmental Setting chapter have common elements across the three communities. However, since issuance of the NOP, the analysis of the proposed Uptown CPU was separated from the North Park and Golden Hill analyses (Chapter 4.0, History of Project Changes). The current chapter discusses the Uptown community's setting; however, because of the other community plan areas are adjacent to the Uptown community, there remains some discussion of the two other Community Plan areas as they relate to the Uptown Community Plan.

2.1 Regional Location

The Uptown CPU area (Uptown community or Uptown) is centrally located to the north of Downtown San Diego and south of the Mission Valley community (Figures 2-1 and 2-2). The Uptown Community Plan area forms the western boundary and a portion of the northern boundary of Balboa Park.

To the north, Uptown is bordered by the south slope of the Mission Valley community, which, in combination with the varying topography, provides an open area between the Uptown and Mission Valley communities. To the south, Uptown is adjacent to Balboa Park, Interstate 5 (I-5), and Downtown. To the east, Uptown is adjacent to Balboa Park and North Park. To the west, Uptown is adjacent to I-5 and Midway–Pacific Highway and the topographical difference between Uptown and neighboring Old Town San Diego.



Uptown Community Plan Boundary

FIGURE 2-1 Regional Location





FIGURE 2-2 Uptown Community Vicinity Map The Uptown community is situated within the same landform that is part of a broad mesa interspersed with many natural and/or semi-developed canyons, allowing a distinctive combination of outward views and interaction with open space along most community edge points. The canyons,

which geographically connect to Mission Valley to the north and interconnect Uptown to the North Park CPU community, are present throughout the Uptown community. Canyons offer relief from the built environment while also creating a barrier to pedestrian, bicycle, vehicular, and intra/intercommunity connections. The canyon landform also creates a sense of seclusion from the surrounding City not uncommon for San Diego's neighborhoods and helps support the interconnectedness between the two communities located on the broad mesa landform.

Uptown and the communities to the east and southeast surround regionally significant and historic Balboa Park. Major transportation corridors traverse the communities, connecting downtown San Diego to other communities in the City, as well as the region. As development radiated out from Downtown along streetcar lines, later forming commercial districts along arterial streets and major crossings, traditional storefronts associated with small and sole-proprietor businesses remain. A grid pattern of streets has developed in Uptown and the associated communities. Vehicular access is affected at many "pinch points" in the communities where street widths narrow or access is "funneled" due to canyon and freeway interfaces.

The CPU area is urbanized and generally characterized as a mix of residential, commercial, and institutional areas. The Uptown community has also been part of one of the longest historical development periods in the region due to its central location and various land use plans and zoning programs, which has left a variety of building forms and architectural styles as well as potential historic resources. The Uptown community developed prior to current Citywide public facilities standards. As a result, locating and financing new facilities, such as parks, is difficult due to lack of available land as well as a limited rate of new development. Aging infrastructure in the community often needs to be upgraded and/or replaced.

2.2 **Project Location**

The Uptown Community Plan area consists of approximately 2,700 acres (approximately 4.2 square miles) and lies just north of Downtown San Diego. It is bounded on the north by the steep hillsides of Mission Valley, on the east by Park Boulevard, and on the west and south by Old Town San Diego and I-5 (see Figure 2-2). The Uptown community is located on a level mesa that is divided by numerous canyons and bordered by two major parks, Presidio and Balboa. The CPU area includes the neighborhoods of Mission Hills, Middletown, Hillcrest, the Medical Complex, University Heights, and Bankers Hill/Park West (Figure 2-3).

Uptown's overall physical structure reflects its geography and development patterns. Most of the street system uses a grid pattern. The CPU area is traversed by three major east-west streets; Washington Street and University Avenue in the northern portion of the community and Laurel Street in the southern portion. Park Boulevard, which services as the community's eastern boundary, as well as First Avenue are important two-way north-south streets along with Fourth and Fifth avenues, which are one-way south- and northbound streets, respectively. Other significant streets are the one-way northbound India Street and one-way westbound Hawthorne Street.



FIGURE 2-3 Uptown Community Plan Update Area

2.3 Existing Physical Characteristics

2.3.1 Land Use

2.3.1.1 Existing Land Uses

Uptown has a limited amount of vacant parcels. As shown in Table 2-1, single-family land use make up approximately 852 acres or 32 percent of the total acres within the community and are the predominant land use within the Uptown community. Multi-family use accounts for approximately 277 acres or 10 percent of the total acreage in the community. Commercial uses, including employment, retail, and services, cover approximately 109 acres or four percent of the total area within the community. The largest retail concentration is in the Hillcrest core where Fourth and Fifth avenues intersect with Washington Street, University Avenue, and Robinson Avenue. Retail also extends in a more linear orientation along Washington Street west of the core, University Avenue east of the core, and along Fourth and Fifth avenues south of the core. Smaller, neighborhood-scale retail nodes also exist in Uptown's residential neighborhoods, such as on Park Boulevard and on West Lewis Street.

The concentration of hospitals and medical support uses in the Medical Complex, and the distribution of office uses along Fourth and Fifth avenues contribute to the identity to these north-south corridors. The existing land uses and distribution are depicted in Figure 2-4, summarized in Table 2-1, and discussed below.

Table 2-1 Existing Land Uses – Uptown			
General Plan Land Use Category	Acres		
Agriculture (Community Garden)	0.5		
Education	30		
Industrial	1		
Institutional	97		
Multi-Family Residential	277		
Office Commercial	57		
Open Space	410		
Parking	28		
Parks	28		
Recreational	3		
Retail Commercial	109		
Roads	761		
Single-Family Residential	852		
Visitor Commercial	6		
Vacant	26		
Total Acreage	2,656		



FIGURE 2-4 Existing Land Use

a. Residential

Residential land uses form the basis and majority of land use acreage in the community. Residential densities vary throughout the community. Very High Residential density areas are located along Sixth Avenue between Upas Street and Laurel Street, along the Fifth Avenue commercial corridor between Pennsylvania Avenue and Maple Street, and within the central Hillcrest commercial core generally bounded by State Route 163 (SR-163), Pennsylvania Avenue, Front Street, and Washington Street. High Residential density areas are located along Second and Third avenues between Maple Canyon and Maple Street, Sixth Avenue between Laurel Street and Elm Street, First and Third avenues between University Avenue and Thorn Street, and within the residential areas in the Medical Complex neighborhood. The Low Residential density areas of the community include stable single-family neighborhoods and are located generally around the central, eastern, and western ends of the community and to the north where they are adjacent to open space.

b. Commercial/Mixed Use

Commercial land uses are located primarily along the community's transportation corridors including: The San Diego Avenue–India Street corridor between the Old Town San Diego community and Olive Street; Reynard Way between Maple Street and Juniper Street; along the Washington Street, University Avenue, Robinson Avenue, and Park Boulevard commercial corridors; and along Fourth and Fifth avenues in the area south of Maple Canyon in the Bankers Hill/West Park neighborhoods. The central Hillcrest commercial core is generally bounded by SR-163, Pennsylvania Avenue, Front Street, and Washington Street. There are neighborhood commercial areas along West Lewis Street and the intersection of Redwood Street and Reynard Way. Areas that include a mixture of commercial and residential uses are also along sections of the Washington Street between Ibis Street and Third Avenue; along Fourth Avenue between Robinson Avenue and Maple Street; along First and Third avenues between Washington Street and University Avenue; Robinson Avenue between Seventh Avenue and SR-163; Seventh and Ninth avenues between Washington Street and University Avenue; within the area bounded by Front Street, Buchanan Place, Montecito Way, and Arbor Place; Third Avenue and Fifth Avenue between Lewis Street and Washington Street; the area bounded by Cleveland Avenue, Richmond Street, and SR-163; University Avenue between SR-163 and Park Boulevard; Park Boulevard between Normal Street and Lincoln Avenue; and Lincoln Street between Washington Street and Cleveland Avenue.

c. Institutional

Institutional uses provide either public or private facilities that serve a public benefit. These uses may serve the community or a broader area. Typically, the larger or more significant public uses such as schools, fire stations, and hospitals are identified on the land use map. Major institutional land uses within the community consist mainly of <u>hospitals</u>. Fire Stations 3, 5, and 8; the Mission Hills and University Heights Branch Libraries; and several public and private schools. Private institutional uses often require a Conditional Use Permit or other type of discretionary permit per the San Diego Municipal Code.

d. Parks and Open Space

Parks and open space areas fulfill a variety of important purposes in the community including active and passive recreation, conservation of resources and protection of views, and providing visual relief in a built-out urban environment. In the Uptown community, oOpen space <u>primarily consists of the steep</u>, undeveloped canyons that also provide opportunity for recreational trail use. Just outside of the southeastern boundary of the CPU area is the extensive active open space/recreational areas of Balboa Park. Refer to Figure 6.12-2 in the Public Services and Facilities section for the location of <u>existing parks</u>. is generally free from development or may be developed with limited, low-intensity uses in a manner that respects the natural environment and conserves sensitive environmental resources.

Protection of resources within lands designated as Open Space affects multiple property owners (including the City of San Diego) and is accomplished primarily through application of various development regulations of the Municipal Code, particularly the Environmentally Sensitive Lands (ESL) Regulations. The City has pursued acquisition of private parcels or acquisition of easement as a means of conserving open space resources and protecting environmentally sensitive areas from development.

Table 2-1, Uptown Existing Land Use provides the acreage of land area covered by land use category for the existing conditions. Descriptions of the categories from the City's General Plan Land Use and Community Planning Element (Table LU-4) that are applicable to the Uptown community are presented in Table 5-1, General Plan Land Use Categories. Application of these categories for consistency with the General Plan Land Use and Community Planning elements is accomplished with approval of individual community plan updates.

2.3.1.2 Adopted Uptown Community Plan

The adopted Uptown Community Plan (1988) covers approximately 2,700 acres. The adopted Community Plan provides more detailed land use, design, roadway, and implementation information than what is found at the General Plan level. The adopted community plan identifies key issues in the community and enumerates a set of objectives to achieve the community's vision. Specific goals, objectives, and policies to implement the adopted Uptown Community Plan are contained in its elements: Residential, Commercial, Transportation, Community Facilities, and Services; Open Space and Recreation; Conservation, Cultural, and Heritage Resources; and Urban Design. The adopted Uptown Community Plan would be replaced by the proposed Uptown CPU. Table 2-2 lists the land uses and population at build-out of the adopted Uptown Community Plan compared to existing conditions.

Table 2-2				
Existing Land Use and Pe	opulation versus Ad	lopted Community Plan		
		Adopted Community Plan		
Land Use	<u>Existing</u>	<u>Build-Out (2035)</u>		
Residential (dwelling units)				
Single-Family	<u>7,540</u>	<u>5,540</u>		
<u>Multi-Family¹</u>	<u>15,620</u>	<u>29,060</u>		
SUBTOTAL ²	<u>23,160</u>	<u>34,600</u>		
Non-Residential (square feet)				
Commercial	<u>4,184,170</u>	<u>4,783,000</u>		
<u>Industrial</u>	<u>19,710</u>	<u> </u>		
<u>Institutional</u>	<u>2,627,550</u>	<u>2,314,900</u>		
<u>Hotels</u>	<u>366,460</u>	<u>174,000</u>		
<u>Recreation</u>	<u>31,110</u>	<u>31,100</u>		
<u>SUBTOTAL²</u>	<u>7,229,000</u>	7,229,000 7,303,000		
Population of Community	Population of Community			
Population	<u>36,750</u> <u>58,870³</u>			
¹ All dwelling units that are not single-family were counted as multi-family. This				
includes dwelling units on other land uses such as commercial and institutional.				
² Total area may not match the sum of listed areas due to rounding.				
³ Estimated population under the adopted community plan at year 2035.				

2.3.2 Visual Effects and Neighborhood Character

2.3.2.1 Existing Context and Urban Form

a. Neighborhood Centers and Nodes

Urban design is influenced by land use; residential is the predominant land use in Uptown, but there are also several nodes of retail, office, and mixed-use, creating centers within each of Uptown's neighborhoods. These centers are generally located along the major transportation corridors. These neighborhood centers form a basis for locating village place types identified by the General Plan.

In the Uptown CPU area, the most significant concentration of the village-like development is in the Hillcrest core, where several major transportation corridors intersect. University Avenue is the anchor corridor of the urban village, which is characterized largely by commercial services and retail development. Key intersections (between First Avenue and Fifth Avenue) within this center act as additional nodes where pedestrians and commercial uses activate the environment along the street. The Hillcrest core extends to from Robinson Street to Washington Street between First and Fifth avenues, and includes residential uses and a variety of commercial use including retail, restaurants, and medical facilities.

Washington Street west of the Hillcrest core, centered at the intersection of Washington and Goldfinch, functions as a center for the Mission Hills neighborhood. This center includes more recent multi-unit, mid-rise residential buildings, many of which include pedestrian-oriented retail on the ground floor.

Smaller neighborhood-scale community centers also exist in Uptown's residential neighborhoods, such as on Park Boulevard and Adams Avenue in University Heights, Fifth Avenue and Laurel Street in Bankers Hill/Park West, and along India Street in Middletown. Within these mixed-use areas, pedestrian-oriented streets and building frontages create public space, which serves the adjacent residential areas and attracts visitors. Neighborhood centers and nodes are illustrated in Figure 2-5.

The concentration of hospitals and medical support uses in the Medical Complex neighborhood forms a community center with an important employment component. While the medical developments themselves have a distinct physical form and are visible landmarks, the distribution of commercial office uses along Fourth and Fifth avenues also provides a distinct personality to these north–south corridors.

Buildings such as St. Paul's Cathedral, the Fifth Avenue Financial Building, Village Hillcrest, and the Teachers Training annex are among those that serve as identifiable landmarks. The community's gateways and bridges are also landmarks. These include Uptown's pedestrian bridges (Quince, Spruce, and Vermont streets bridges), the First Avenue Bridge over Maple Canyon, the historic gateway signs (Hillcrest, Mission Hills, and University Heights), and the River Rock neighborhood identity markers indicating entrance into University Heights. Landmarks and gateways are important components of urban design, because they create discernible markers of neighborhood distinction. Landmarks and gateways in Uptown are illustrated in Figure 2-6.

b. Built Form and Development

Uptown's physical form and architectural character is a product of its history. Uptown has been valued for its proximity to Downtown and its unobstructed views of the harbor, and includes a variety of architectural styles and mature landscapes dating to the City's early history. Infill development and the replacement and modification of buildings have occurred during past decades.

c. Canyons and Views

Due to its diverse topography, Uptown has prominent view corridors, offering views to Downtown, Balboa Park, Mission Valley, the San Diego Bay, and Mission Bay. Public view corridors are located along public streets and transportation corridors with views of areas such as the San Diego Bay, Mission Bay, Balboa Park, Mission Valley, and open spaces areas and canyons.

2.3.3 Transportation and Circulation

The Uptown Community Plan area is identified in the General Plan's Land Use and Street System Map (contained in the Land Use and Community Planning Element, Figure LU-2). Traffic circulation patterns within the Uptown community are reflective of the fact that freeways and/or highways form the western and southern (I-5) boundaries of the Uptown community and another freeway (Interstate 8 [I-8]) is just to the north of the Uptown planning area. In addition, SR-163 traverses the Uptown Community Plan area., resulting in the use of local roads for <u>T</u>trucking and transport of goods <u>occurs within the CPU area on these between the</u> freeways and on local roads.

Map Source: SanGIS



 $\label{eq:FIGURE 2-5} FIGURE \ 2\text{-}5$ Neighborhood Centers and Nodes – Uptown

Map Source: SanGIS



FIGURE 2-6 Landmarks and Gateways – Uptown

2.3.3.1 Roadways and Access

Freeway and/or highway access in the vicinity of the Uptown planning area is provided via I-5, Interstate 15 (I-15), Interstate 805 (I-805), and SR-163, which are north-south routes and State Route 94 (SR-94), which is an east-west route. I-8 is an east-west freeway located just north of the Uptown community. These highways improved regional accessibility and separate the Uptown community from central San Diego. Due to the topography of the Uptown community, in many places these facilities are below-grade to the surrounding developed land uses.

Major roadways within the Uptown community generally run in an east-west direction. The most prominent are University Avenue and Washington Street in the northern part of Uptown and Laurel Street in the southern part. Prominent north-south roadways include First, Fourth, and Fifth avenues and Park Boulevard. Traffic on several roadway segments within the Uptown community currently exceeds acceptable levels as defined by City thresholds.

2.3.3.2 Public Transportation

The City works with local agencies to provide transportation systems for its residents and visitors. Bus (including Bus Rapid Transit) and trolley service, as well as commuter rail stations, are served by the San Diego Metropolitan Transit System (MTS) and the North County Transit District. The Uptown community is served by the San Diego trolley (light rail) line and bus service operated by MTS. The trolley, which parallels I-5, has transit stops adjacent to the Uptown CPU area.

a. Bus-Rapid Transit-(BRT)

BRT Rapid transit is corridor-level service providing fast and frequent transit services that are designed to take advantage of <u>both</u> freeway improvements, such as High Occupancy Vehicle and managed lanes, <u>and arterial improvements</u> in order to serve longer distance regional trips. The Rapid Transit service on arterials will operate on arterial roadways and provide limited-stop, high-speed service along several key corridors throughout the region, supplementing existing local bus service.

b. Light Rail Transit (LRT)

LRT is a type of transit vehicle and service that uses steel wheels and operates over railroad tracks. LRT systems generally serve stations averaging one mile apart, are not remotely controlled, and can operate in a separated right-of-way or on public streets. The San Diego Trolley is a LRT system.

c. Rapid Bus (also known as Arterial Rapid Transit)

Rapid Bus or Arterial Rapid Transit (ART) provides rapid and frequent transit service along arterials that use signal priority and queue jumper lanes at major intersections.

<u>c</u>d. Streetcar

Streetcars are electric-powered rail vehicles designed for short-distance trips with station spacing every few blocks or every quarter mile on average. Typical speeds are up to the speed limit of the street they operate on, generally averaging 12 miles per hour (with stops). They are designed for dense urban areas, such as downtown areas, and they integrate well with street traffic, signals, and pedestrians. They operate either in mixed traffic with automobiles or on a dedicated right-of-way and would accommodate up to 100 passengers per car.

2.3.3.3 Rail

In addition to the local light rail system, the San Diego and Imperial Valley Railroad operates at night along separate tracks paralleling the trolley tracks, and the Burlington Northern Santa Fe Railroad operates freight trains on separate tracks located west of Harbor Drive (City of San Diego 2013).

2.3.3.4 Bicycle Facilities

Types of bicycle facilities include bicycle boulevards, bicycle paths (Class I), bicycle lanes (Class II), bicycle routes (Class III), and cycle tracks (Class IV). Bicycle boulevards and cycle tracks are additional facilities that are not defined by the California Department of Transportation (Caltrans) and are not part of the existing bicycle network in the Uptown community (Table 2-<u>3</u>2).

ATTACHMENT 7 2.0 Environmental Setting

Table 2-3 Regional Corridor Classification System

Cycle Tracks

A cycle track is a hybrid type bicycle facility that combines the experience of a separated path with the on-street infrastructure of a conventional bike lane. Cycle tracks are bikeways located in roadway right-of-way but separated from vehicle lanes by physical barriers or buffers. Cycle tracks provide for one-way bicycle travel in each direction adjacent to vehicular travel lanes and are exclusively for bicycle use. Cycle tracks are not recognized by Caltrans Highway Design Manual as a bikeway facility. Development of cycle track on segments of the regional corridor system is proposed through experimental, pilot projects.



Bicycle Boulevards

Bicycle boulevards are local roads or residential streets that have been enhanced with traffic calming and other treatments to facilitate safe and convenient bicycle travel. Bicycle boulevards accommodate bicyclists and motorists in the same travel lanes, typically without specific vehicle or bicycle lane delineation. These roadway designations prioritize bicycle travel above vehicular travel. The treatments applied to create a bike boulevard heighten motorists' awareness of bicyclists and slow vehicle traffic, making the boulevard more conducive to safe bicycle and pedestrian activity. Bicycle boulevard treatments include signage, pavement markings, intersection treatments, traffic calming measures and can include traffic diversions. Bicycle boulevards are not defined as bikeways by Caltrans Highway Design Manual; however, the basic design features of bicycle boulevards comply with Caltrans standards.



2.3.4 Air Quality

The Uptown planning area is located within the San Diego Air Basin (SDAB) of the San Diego Air Pollution Control District (SDAPCD), between 0.5 mile and 2.4 miles northeast of the San Diego Bay. Air quality conditions and local climate are described in this section.

2.3.4.1 Climate

The San Diego region, including the Uptown community, is influenced by proximity to the Pacific Ocean and semi-permanent high-pressure systems that result in warm, dry summers and mild, occasionally wet winters. The Uptown CPU area is subject to frequent offshore breezes. The dominant meteorological feature affecting the region is the Pacific High Pressure Zone, which produces the prevailing westerly to northwesterly winds blowing pollutants away from the coast toward inland areas.

The Uptown community, like the rest of San Diego County's coastal areas, has a Mediterranean climate characterized by warm, dry summers and mild, wet winters. The mean annual temperature at San Diego International Airport, recorded near downtown San Diego and Uptown, is 64 degrees Fahrenheit (°F). The average annual precipitation for the area is approximately 10 inches, falling primarily from November to April. Winter mean low temperatures average 49°F, and summer mean high temperatures average 74°F based on the measurements taken at the San Diego International Airport.

The dominant meteorological feature affecting the region is the Pacific High Pressure Zone, which produces the prevailing westerly to northwesterly winds. These winds tend to blow pollutants away from the coast toward the inland areas. Consequently, air quality near the coast is generally better than what occurs at the base of the coastal mountain range.

Fluctuations in the strength and pattern of winds from the Pacific High Pressure Zone interacting with the daily local cycle produce periodic temperature inversions that influence the dispersal or containment of air pollutants in the San Diego Air Basin (SDAB). Beneath the inversion layer pollutants become "trapped" as their ability to disperse diminishes. The mixing depth is the area under the inversion layer. Generally, the morning inversion layer is lower than the afternoon inversion layer. The greater the change between the morning and afternoon mixing depths, the greater the ability of the atmosphere to disperse pollutants.

Throughout the year, the height of the temperature inversion in the afternoon varies between approximately 1,500 and 2,500 feet above mean sea level (MSL). In winter, the morning inversion layer is about 800 feet above MSL. In summer, the morning inversion layer is about 1,100 feet above MSL. Therefore, air quality generally tends to be better in the winter than in the summer.

The prevailing westerly wind pattern is sometimes interrupted by regional "Santa Ana" conditions. A Santa Ana occurs when a strong high pressure develops over the Nevada to Utah area and overcomes the prevailing westerly coastal winds, sending strong, steady, hot, dry northeasterly winds over the mountains and out to sea.

Strong Santa Ana winds tend to blow pollutants out over the ocean, producing clear days. However, at the onset or during breakdown of these conditions or if the Santa Ana is weak, local air quality may be adversely affected. In these cases, emissions from the South Coast Air Basin to the north are blown out over the ocean, and low pressure over Baja California draws this pollutant-laden air mass southward. As the high pressure weakens, prevailing northwesterly winds reassert themselves and send this cloud of contamination ashore in the SDAB. When this event does occur, the combination of transported and locally produced contaminants produce the worst air quality measurements recorded in the basin.

2.3.4.2 Existing Air Quality

Air quality at a particular location is a function of the kinds, amounts, and dispersal rates of pollutants being emitted into the air locally and throughout the basin. The major factors affecting pollutant dispersion are wind speed and direction, the vertical dispersion of pollutants (which is affected by inversions), and the local topography.

Air quality is commonly expressed as the number of days in which air pollution levels exceed state standards set by the California Air Resources Board (CARB) or federal standards set by the U.S. Environmental Protection Agency (EPA). The San Diego Air Pollution Control District (APCD) maintains 11 air quality monitoring stations located throughout the greater San Diego metropolitan region. Air pollutant concentrations and meteorological information are continuously recorded at these 11 stations. Measurements are then used by scientists to help forecast daily air pollution levels.

The air quality monitoring station nearest the Uptown CPU area is the San Diego-Beardsley Street monitoring station that is located at 1110 Beardsley and monitors the following pollutants: ozone (O_3) , carbon monoxide (CO), nitrogen dioxide $(NO_2)^{\circ}$ and particulate matter less than 10 microns in diameter (PM₁₀) and particulate matter less than 2.5 microns in diameter (PM_{2.5}). The sulfur dioxide (SO_2) monitors were decommissioned in 2012, as this pollutant is less of a concern in the SDAB. Table 2-<u>4</u> provides a summary of measurements of O₃, CO, SO₂, NO₂, PM₁₀, and PM_{2.5} collected at the Beardsley Street monitoring station for the years 2010 through 2014.

2.3.4.3 Regional Background Toxic Air Pollutants

The San Diego APCD samples for toxic air contaminants at the El Cajon and Chula Vista monitoring stations. Excluding diesel particulate emissions, data from these stations indicate that the background cancer risk in 2008 due to air toxics was 135 in one million in Chula Vista and 150 in one million in El Cajon. There is no current methodology for directly measuring diesel particulate concentrations. Based on CARB estimates, diesel particulate emissions could add an additional 420 in one million to the ambient cancer risk levels in San Diego County.

Thus the combined background ambient cancer risk due to air toxics in the urbanized areas of San Diego County could potentially range from around 555 to 570 in one million. As such, diesel particulate matter is the air toxic of primary concern on a regional basis.

^aThe SO₂ monitor was decommissioned on June 30, 2011. ^bCalculated days. Calculated days are the estimated number of days that a measurement would have been greater than the level of the standard had measurements been collected every day. Particulate measurements are collected every six days. The number of days above the standard is not necessarily the number of violations of the standard for the year.

2.3.5 Greenhouse Gas Emissions

The Uptown Community Plan area is currently a source of anthropogenic greenhouse gas (GHG), with emissions generated by vehicular traffic and by the energy use, water use, and solid waste disposal practices of existing development.

2.3.5.1 State and Regional GHG Inventories

a. CARB Inventory

The CARB performs statewide GHG inventories. The inventory is divided into nine broad sectors of economic activity: agriculture, commercial, electricity generation, forestry, high global warming potential (GWP) emitters, industrial, recycling and waste, residential, and transportation. Emissions are quantified in million metric tons of CO_2 equivalent (MMT CO_2E). Table 2-<u>5</u>4 shows the estimated statewide GHG emissions for the years 1990, 2008, and 2012.

As shown in Table 2-<u>5</u>4, statewide GHG source emissions totaled approximately 427 MMT CO_2E in 1990, 487 MMT CO_2E in 2008, and 459 MMT CO_2E in 2012. Many factors affect year-to-year changes in GHG emissions, including economic activity, demographic influences, environmental conditions such as drought, and the impact of regulatory efforts to control GHG emissions. CARB has adopted multiple GHG emission reduction measures, and most of the reductions since 2008 have been driven by economic factors (recession), previous energy-efficiency actions, and the Renewables Portfolio Standard. Transportation-related emissions consistently contribute the most GHG emissions, followed by electricity generation and industrial emissions. The forestry sector is unique because it not only includes emissions associated with harvest, fire, and land use conversion (sources), but also includes removals of atmospheric carbon dioxide (CO_2 ; sinks) by photosynthesis, which is then bound (sequestered) in plant tissues.

Table 2-5				
California GHG Emissions by Sector in 1990, 2008, and 2012				
	1990 ¹	2008 ³	2012	
	Emissions in	Emissions in	Emissions in	
	MMT CO ₂ E	MMT CO ₂ E	MMT CO ₂ E	
Sector	(% total) ²	(% total) ²	(% total) ²	
Sources				
Agriculture	23.4 (5%)	37.99 (8%)	37.86 (8%)	
Commercial	14.4 (3%)	13.37 (3%)	14.20 (3%)	
Electricity Generation	110.6 (26%)	120.15 (25%)	95.09 (21%)	
High GWP		12.87 (3%)	18.41 (4%)	
Industrial	103.0 (24%)	87.54 (18%)	89.16 (19%)	
Recycling and Waste		8.09 (2%)	8.49 (2%)	
Residential	29.7 (7%)	29.07 (6%)	28.09 (6%)	
Transportation	150.7 (35%)	178.02 (37%)	167.38 (36%)	
Forestry (Net CO ₂ flux)	-6.69			
Not Specified	1.27			
TOTAL	426.6	487.10	458.68	
SOURCE: California Energy Commission (CEC) 2014, CARB 2007 & 2014a				
¹ 1990 data was retrieved from the CARB 2007 source.				
² Percentages may not total 100 due to rounding.				
² 2008 and 2012 data was retrieved from the CARB 2014a source.				
Reported emissions for key sectors. The inventory totals for 2008 and 2012 did not				
include Forestry or Not Specified sources.				

b. City of San Diego CAP Inventory

A San Diego regional emissions inventory prepared as part of the City of San Diego's Climate Action Plan reported GHG emissions totaling approximately 13 MMT CO₂E in 2010. Similar to the statewide emissions, transportation-related GHG emissions contributed the most Citywide, followed by emissions associated with energy use.

2.3.6 Noise

Existing conditions related to the noise environment are included in Section 6.6.1 of the PEIR. The following background information provides additional context related to evaluating the noise environment.

2.3.6.1 Existing Noise Environment

Noise sensitive receptors are land uses for which the associated primary activities, whether indoor or outdoor, are susceptible to disruption by loud noise events. The most common noise sensitive uses include: residences, hospitals, nursing facilities, intermediate care facilities, educational facilities, libraries, museums, places of worship, child-care facilities, and certain types of passive recreational parks and open space. Existing noise sources in the CPU area include motor vehicle and
stationary sources. Stationary noise sources include industrial and commercial operations. Noise from these sources can conflict with existing noise sensitive receptors.

2.3.6.2 Fundamentals of Noise

Sound propagation (i.e., the passage of sound from a noise source to a receiver) is influenced by several factors including the distance from the source, geometric spreading, ground absorption and atmospheric effects, as well as shielding by natural and/or manmade features. Noise is unwanted or disturbing sound.

The noise descriptors used in the environmental analysis (Chapter 6.0) are the decibel (dB), A-weighted decibel (dBA), 1-hour average-equivalent noise level (L_{eq}), and the community noise equivalent level (CNEL). The hourly equivalent sound level (L_{eq}) is the average dBA sound level over a 1-hour period. A-weighting is a frequency correction that often correlates well with the subjective response of humans to noise. Similar to Leq, the CNEL is a 24-hour average A-weighted decibel sound level. However, CNEL also incorporates a 5 dBA penalty to sound levels occurring between 7:00 p.m. and 10:00 p.m., and 10 dBA penalty to sound levels occurring between 10:00 p.m. and 7:00 a.m. The additional 5 dBA and 10 dBA penalties during evening and nighttime hours, respectively, are intended to account for the added sensitivity of humans to noise during these time periods. For example, although a noise level of 60 dBA is typically considered acceptable during the day, during rest hours that same 60 dBA noise level may be considered a nuisance. CNEL values are typically used in land use planning to evaluate the compatibility of adjacent land uses.

The subsections below further describe elements and measures of noise.

a. Frequency and Hertz

A continuous sound can be described by its frequency (pitch) and its amplitude (loudness). Frequency relates to the number of pressure oscillations per second. Low-frequency sounds are low in pitch, like the low notes on a piano, whereas high-frequency sounds are high in pitch, like the high notes on a piano. Frequency is expressed in terms of oscillations, or cycles, per second. Cycles per second are commonly referred to as Hertz (Hz). High frequencies are sometimes more conveniently expressed in units of kilo-Hertz (kHz) or thousands of Hertz. The extreme range of frequencies that can be heard by the healthiest human ear spans from 16 to 20 Hz on the low end to about 20,000 Hz (or 20 kHz) on the high end.

b. Sound Pressure Levels and Decibels

The amplitude of a sound determines its loudness. Loudness of sound increases and decreases with its amplitude. Sound pressure levels are described in units called the decibel. Decibels are measured on a logarithmic scale that quantifies sound intensity in a manner similar to the Richter scale used for earthquake magnitudes. Thus, a doubling of the energy of a noise source, such as doubling of traffic volume, would increase the noise level by 3 dB; a halving of the energy would result in a 3 dB decrease.

c. A-weighted Decibels

The human ear is not equally sensitive to all frequencies within the sound spectrum. Human hearing is limited not only in the range of audible frequencies but also in the way it perceives the sound in that range. In general, the healthy human ear is most sensitive to sounds between 1,000 Hz and 5,000 Hz, and it perceives a sound within that range as more intense than a sound of higher or lower frequency with the same magnitude. To approximate the frequency response of the human ear, a series of sound level adjustments is usually applied to the sound measured by a sound level meter.

The A-scale weighting network approximates the frequency response of the average healthy ear when listening to most ordinary sounds. When people make judgments of the relative loudness or annoyance of a sound, their judgments correlate well with the A-scale sound levels of those sounds. Noise levels for traffic noise reports are typically reported in terms of A- weighted decibels [dB(A)]. All sound levels discussed in the PEIR analysis (Chapter 6.0) are A-weighted. Examples of typical noise levels for common indoor and outdoor activities are depicted in Table 2-<u>6</u>5.

Under controlled conditions in an acoustics laboratory, the trained, healthy human ear is able to discern changes in sound levels of 1.5 dB(A) under certain conditions. Outside such controlled conditions, the average healthy ear can barely perceive changes of 3 dB(A), a change of 5 dB(A) is readily perceptible; and an increase (decrease) of 10 dB(A) sounds twice (half) as loud.

Table 2-6 Typical Sound Levels in the Environment and Industry				
Common Outdoor Activities	Noise Level [dB(A)]	Common Indoor Activities		
_	110	Rock band		
Jet fly over at 300 m (1000 feet)	100	_		
Gas lawn mower at 1 m (3 feet)	90	_		
Diesel truck at 15 m (50 feet), at 80 km/hr (50 mph)	80	Food blender at 1 m (3 feet) Garbage disposal at 1 m (3 feet)		
Noisy urban area, daytime Gas lawn mower at 30 m (100 feet)	70	Vacuum cleaner at 3 m (10 feet)		
Commercial area Heavy traffic at 90 m (300 feet)	60	Normal speech at 1 m (3 feet)		
Quiet urban daytime	50	Large business office Dishwasher next room		
Quiet urban nighttime	40	Theater, large conference room (background)		
Quiet suburban nighttime	30	Library		
Quiet rural nighttime	20	Bedroom at night, concert hall (background)		
_	10	Broadcast/recording studio		
Lowest threshold of human hearting	0	Lowest threshold of human hearting		
SOURCE: Caltrans 2013				

d. Noise Descriptors

The two noise metrics used in the analysis (Chapter 6.0) are the equivalent noise level (L_{eq}) and the CNEL.

Equivalent Noise level (Leq)

The equivalent sound level (L_{eq}) is also referred to as the time-average sound level. It is the equivalent steady state sound level, which in a stated period of time would contain the same acoustical energy as the time-varying sound level during the same time period. The period of time averaging may be specified; $L_{eq}(3)$ would be a three-hour average. When no period of time is specified, a one-hour average is assumed. The one-hour A-weighted equivalent sound level is the energy average of the A-weighted sound levels occurring during a one-hour period. It is important to understand that noise of short duration, that is, times substantially less than the averaging period, is averaged into ambient noise during the period of interest. Thus, a loud noise lasting many seconds or a few minutes may have minimal effect on the measured sound level averaged over a one-hour period.

Community Noise Equivalent Level (CNEL)

People are generally more sensitive and annoyed by noise occurring during the evening and nighttime hours. Thus, the CNEL was introduced. The CNEL scale represents a time-weighted 24-hour average noise level based on the A-weighted sound level. CNEL accounts for the increased noise sensitivity during the evening (7:00 P.M. to 10:00 P.M.) and nighttime hours (10:00 P.M. to 7:00 A.M.) by adding five and ten decibels, respectively, to the average sound levels occurring during these hours.

2.3.6.3 Vibration

Groundborne vibration consists of oscillatory waves that propagate from the source through the ground to adjacent structures. The frequency of a vibrating object describes how rapidly it is oscillating. The number of cycles per second of oscillation is the vibration frequency, which is described in terms of hertz. The normal frequency range of most groundborne vibration that can be felt generally ranges from a low frequency of less than 1 Hz to a high of about 200 Hz.

While people have varying sensitivities to vibrations at different frequencies, in general they are most sensitive to low-frequency vibration. Vibration in buildings caused by construction activities may be perceived as motion of building surfaces or rattling of windows, items on shelves, and pictures hanging on walls. Vibration of building components can also take the form of an audible low-frequency rumbling noise, which is referred to as groundborne noise.

Vibration energy spreads out as it travels through the ground, causing the vibration level to diminish with distance away from the source. High-frequency vibrations reduce much more rapidly than low frequencies, so that low frequencies tend to dominate the spectrum at large distances from the source. When vibration encounters a building the overall vibration level is typically reduced;

however, under certain circumstances, vibration can be amplified due to structural resonances of the floors and walls.

Vibration levels are usually expressed as single-number measure of vibration magnitude, in terms of velocity or acceleration, which describes the severity of the vibration without the frequency variable. The peak particle velocity (PPV) is defined as the maximum instantaneous positive or negative peak of the vibration signal, usually measured in inches per second. Since it is related to the stresses that are experienced by buildings, PPV is often used in monitoring of blasting vibration. Although PPV is appropriate for evaluating the potential of building damage, it is not suitable for evaluating human response since it takes some time for the human body to respond to vibrations.

2.3.7 Historical Resources

Historical resources (also referred to as cultural resources) are physical features, both natural and constructed, which reflect past human existence and are of historical, archaeological, scientific, educational, cultural, architectural, aesthetic, or traditional significance. These resources may include such physical objects and features as archaeological sites and artifacts, buildings, groups of buildings, structures, districts, street furniture, signs, cultural properties, and landscapes. Historical resources in the San Diego region span a timeframe of at least the last 10,000 years and include both the prehistoric and historic periods. For purposes of the PEIR, historical resources consist of archaeological sites, tribal cultural resources, and built environment resources that are determined to be significant under California Environmental Quality Act (CEQA).

Archaeological resources include prehistoric and historic locations or sites where human actions have resulted in detectable changes to the area. This can include changes in the soil, as well as the presence of physical cultural remains. Archaeological resources can have a surface component, a subsurface component, or both. Historic archaeological resources are those dating after European contact. These resources may include subsurface features such as wells, cisterns, or privies. Other historic archaeological remains include artifact concentrations, building foundations, or remnants of structures.

2.3.7.1 Prehistory

The prehistoric cultural sequence for what is now San Diego County is generally thought of as three basic periods: Paleoindian, locally characterized by the San Dieguito complex; Archaic, characterized by the cobble and core technology of the La Jollan and Pauma complexes; and Late Prehistoric, marked by the appearance of ceramics, small arrow points, and cremation burial practices. Late Prehistoric materials in southern San Diego County, known as Yuman I and Yuman II, are believed to represent the ancestral Kumeyaay (AECOM 2015).

By the time Spanish colonists began to settle in Alta California in 1769, the areas that are now part of the Uptown community were within the territory of the Kumeyaay people, a group of exogamous, nontotemic territorial bands with patrilineal descent. The Kumeyaay had a hunting and gathering economy based primarily on various plant resources. For people in the area that is now Uptown, grass seeds were probably the primary food, supplemented by various other seeds such as sage (*Salvia* spp.), sagebrush (*Artemisia californica*), lamb's quarters (*Chenopodium album*), and pine nuts

(*Pinus* sp.). Small game was a major source of protein, but deer were hunted as well. Coastal bands ate a great deal of fish, taking them with lines, nets, and bows and arrows. Balsas or reed boats were used. Shellfish and other littoral resources were important to coastal people, too. Settlements were moved seasonally to areas where wild foods were in season. For example, inland bands might have moved into desert areas in the spring to gather agave (*Agave deserti*), then to higher-altitude areas in the fall to gather acorns. Coastal bands lived in more or less permanent villages focused on more seasonally stable inshore and littoral resources. However, they often traveled to the area that is now Torrey Pines and La Rumarosa (in northern Baja California) to harvest pine nuts, for example, and to Cuyamaca and Mount Laguna for acorns (AECOM 2015).

There is one named Kumeyaay village identified in the vicinity of the community of Uptown, the village of *Cosoy/Kosaii/Kosa'aay*. Villages and campsites were generally located in areas where water was readily available, preferably on a year-round basis. The San Diego River, which is located approximately 0.5 mile from the Uptown Community Plan area, provided an important resource not only as a reliable source of water, but as a major transportation corridor through the region. Although the actual location of the village is unknown, it is reported that a site called *Cosoy/Kosaii/Kosa'aay* by the Native Americans was in the vicinity of Presidio Hill and Old Town, located less than one mile west of the Uptown Community Plan area boundary. Additionally, two named Kumeyaay villages or *rancheria* may lie to the southeast of the Uptown Community Plan area, in the vicinity of Golden Hill. The village, or *rancheria of Los Choyas*, was located near the mouth of Los Chollas Creek. The village of *Pu-Shuyi* was located near the foot of modern-day Market Street (AECOM 2015).

2.3.7.2 History

In the mid-19th century, San Diego had approximately 650 residents. However, new arrivals were transforming the small Mexican community into a growing commercial center. In 1867, Alonzo Erastus Horton acquired nearly 1,000 acres of land two miles south of "Old Town", where downtown San Diego sits today. Dubbed "New San Diego", Horton orchestrated the creation of a new city center, relocating the city's first bank, main newspaper, and several government buildings to this site. Thus Old Town was supplanted as the city's primary commercial center. The arrival of the railroad in the 1880s linked San Diego with the eastern United States and sparked its first building boom. By 1887, San Diego's population had spiked to 40,000, and large tract of new development began to appear on the hills immediately adjacent to downtown.

By 1892, substantial infrastructure improvements were underway, including public utilities, street paving, sewer systems, and the electrification of the streetcar system. These improvements would be critical to the development of new suburbs surrounding downtown and the 1,400-acre City Park (Balboa Park), including present-day Uptown and the communities of North Park and Golden Hill.

The completion of a transcontinental rail line in 1885 was a catalyst for the first notable wave of development in Uptown. At the time, speculation still abounded, but a substantial number of homes were constructed near the southern border of Uptown, in present-day Park West. Over the next two decades, new development shifted north towards present-day Hillcrest and University Heights, due in large part to the construction of several public transit lines. Development at this time was

primarily residential, but by the early 1900s the area was also home to several businesses, a state Normal School, and a popular public park.

Development activity accelerated once more in anticipation of the much awaited 1915 Panama-California Exposition. By the 1920s, both Park West and Hillcrest were almost entirely developed, and the more distant communities of University Heights and Mission Hills were nearly built out by the 1930s. Following the Great Depression and World War II, Uptown was the target of several redevelopment efforts and witnessed a considerable amount of physical change. Despite being bisected by Interstate 5 and State Route 163, Uptown still contains cohesive blocks of historic structures, especially in Park West, Hillcrest, and University Heights. In addition, Mission Hills has retained its historic fabric and contains a sizable concentration of single-family homes dating from the 1910s, 1920s and 1930s.

In the years following the Great Depression, Uptown experienced marked physical change. Residential construction essentially ceased, and many business ventures failed along established commercial thoroughfares. It was United States' entrance into World War II that effectively ended the economic downturn and boosted the regional economy. This was particularly true in San Diego; with its extensive military or manufacturing facilities now devoted to the defense industry, the city had received the highest per capita share of war contracts in the state. Like other large cities, San Diego's wartime and postwar population growth far outpaced its ability to provide sufficient services and housing. In response, City officials rezoned large sections of the Community Plan area to accommodate high-density residential development.

In Uptown, unimproved lots in established neighborhoods were infilled with single-family homes and residential courts inspired by Federal Housing Administration (FHA) designs. Developers of multi-family housing favored higher densities over the residential courts of the pre-war period. The result was the proliferation of the two-story stucco box apartment building, designed to maximize the number of units and provide the required the parking on a single residential lot. Development from this era reflected Post-War American values and design trends, such as automobile oriented commercial development and Modern design in both residential and commercial buildings.

As the economy slowly began to rebound, new businesses occupied existing storefronts along established commercial corridors, often renovating their facades with more contemporary details. The modernization of storefronts occurred along Main Streets and commercial corridors throughout California, and included new large display windows which allowed merchandise to be visible to passing motorists. Such changes reflect the evolution of a thriving commercial core.

The suburbanization of the Post-War period left Uptown with an aging population and deteriorating building stock by the late 1960s. The relative safety and affordability this presented attracted members of the Lesbian, Gay, Bisexual, Transgender and Queer (LGBTQ) community, who established businesses and support and advocacy groups catering to the gay community beginning in the 1970s. In most instances, existing building stock was utilized and adaptively reused. The affordable housing, particularly the bungalow and apartment courts, was also attractive to those seeking a sense of community. The investment of the LGBTQ community in Uptown has led to a renaissance over the last several decades that has made Uptown a vibrant, walkable community.

Today, Uptown is best characterized in terms of its diversity. In addition to housing people from a wide variety of income levels and ethnic groups, the community boasts a built environment that is equally as eclectic, reflecting the rich history – both shared and unique – of some of San Diego's oldest neighborhoods.

Some of the key historical themes in the evolution of the Uptown community include:

- The Railroad Boom and Early Residential Development: 1885-1909
- The Panama-California Exposition and Streetcar Suburbs: 1909-1929
- Great Depression and World War II: 1929–1948
- Postwar Development, Suburbanization, the Automobile & Modernism: 1948-1970
- Neighborhood Revitalization and the LGBT Community: 1970-Present

2.3.8 Biological Resources

Uptown is one of the urban communities in the City of San Diego and is essentially completely built out. Most of each of the community plan areas are developed and consist of ornamental and nonnative vegetation within the urbanized portions. Native vegetation generally occurs within the canyons and areas designated as open space where development has not occurred.

2.3.8.1 Soils

The U.S. Department of Agriculture mapped the following soil series in the Uptown area: Gaviota fine sandy loam, Huerhuero loam, Olivenhain cobbly loam, Riverwash, Redding-Urban Land complex, Redding cobbly loam, terrace escarpments, made land, and urban land. Most of the Uptown area is covered by urban lands; the canyons are mostly covered by Huerhuero loam.

2.3.8.2 Topography

The Uptown Community Plan area consists of the generally flat San Diego Mesa incised by steepsided canyons draining into Mission Valley and/or the San Diego Bay basin. Current land use in the CPU area consists of developed residential communities and commercial buildings on the mesa tops, and undeveloped areas generally located on natural canyon hillsides and in canyon bottoms. The gradient of natural canyon slopes is variable but are locally steeper that 2:1 (horizontal to vertical). Manufactured slopes are locally present and where steeper than 1 ½:1 up to eight feet high or greater than eight feet high and steeper than 2:1 are considered existing non-confirming slopes.

2.3.8.3 Botanical Resources

A general description of vegetation communities and land cover types mapped within the Uptown community is described below. There are seven vegetation communities and land cover types present: coastal sage scrub, chaparral, grassland, riparian scrub, eucalyptus woodland, disturbed land, and urban/developed. Acreages of vegetation communities and land cover types mapped within the CPU area is described within Section 6.8.

a. Wetland Vegetation Communities

Wetland vegetation communities are dominated by plant species adapted to soils that have periods of prolonged saturation. Wetland vegetation communities are considered sensitive and regulated by the U.S. Army Corps of Engineers (ACOE), U.S. Fish and Wildlife Service (USFWS), the California Department of Fish and Wildlife (CDFW), Regional Water Quality Control Board (RWQCB), and the City of San Diego. One wetland community, riparian scrub, occurs in the CPU area.

Riparian scrub is considered a sensitive wetland habitat under Environmentally Sensitive Lands (ESL) and the City of San Diego's Biology Guidelines. This vegetation community may vary from open to dense and is typically dominated by broad-leaved, winter deciduous trees and/or shrubs. It may contain an understory consisting of sub-shrubs or herbaceous species, although denser stands may prevent the development of understory vegetation. Tree species may include willows (*Salix* spp.), Fremont cottonwoods (*Populus fremontii*), and/or western sycamores (*Platanus racemosa*). Scrubs are generally dominated by riparian shrubs such as mule fat (*Baccharis salicifolia*). Riparian scrub is typically found along major drainages, but also may occur in smaller drainages.

b. Upland Communities

Upland vegetation communities do not support wetland species. These native vegetation types occur on the drier areas of the mesa, slopes, and canyons in the CPU area. There are three vegetation communities and three land cover types in this category as described below.

Grassland

Grassland is characterized by a dense to sparse cover of native and non-native annual grasses, which may include numerous native wildflowers, particularly in years of high rainfall. Grasslands contain species including, but not limited to, needle grasses, bromes, wild oats, ryegrasses, and fescues. Typically, this community includes at least 50 percent cover of the entire herbaceous layer attributable to annual non-native grass species, although other native and non-native plant species may be intermixed.

These annual plants germinate with the onset of the rainy season and set seeds in the late winter or spring. Grassland is typically found on fine-textured, usually clay, soils that range from being moist or waterlogged in the winter to being very dry during the summer and fall. This community is found in valleys and foothills throughout much of California at elevations below 3,000 to 4,000 feet.

Coastal Sage Scrub

Coastal sage scrub is a plant community comprised of low-growing, aromatic, drought-deciduous, soft-woody shrubs that have an average height of approximately three to four feet. The plant community is typically dominated by facultatively drought-deciduous species such as California sagebrush (*Artemisia californica*), California buckwheat, and coyote bush (*Baccharis pilularis*) with non-native herbs and grasses growing between and within the shrubs. The vegetation community typically is found on low moisture-availability sites with steep, xeric slopes or clay rich soils that are slow to release stored water. These sites often include drier south- and west-facing slopes and

occasionally north-facing slopes, where the coastal sage scrub can act as a successional phase of chaparral development.

Chaparral

Chaparral is a plant community typically dominated by broad-leaved sclerophyllous shrubs or small trees that typically range in height range from four to 10 feet tall. Chaparral is typically dominated by blue-colored lilacs including Ramona lilac (*Ceanothus tomentosus* var. *olivaceus*), chaparral whitethorn (*C. leucodermis*), and hairy ceanothus (*C. oliganthus*) and may include manzanita (*Arctostaphylos* spp.), toyon (*Heteromeles arbutifolia*), sugar bush (*Rhus ovata*), and mission manzanita (*Xylococcus bicolor*). Chaparral typically is found in coastal foothills of San Diego County at elevations below 3,000 feet. It usually occupies canyon slopes or ravines where mesic conditions are present. The vegetation is usually dense, with little or no understory cover, but may include patches of bare soil. Many species in this community are adapted to repeated fires by their ability to stump sprout.

c. Other Land Cover Types

Three other land cover types are present within the Uptown CPU area. All result from some sort of development, encroachment, or other human disturbance.

Urban/Developed

Areas mapped as urban/developed include locations with residential housing, commercial, and industrial land uses. Additionally, urban/developed includes ornamental areas that have been landscaped with non-native species and are actively maintained. This land cover type is found over the majority of the Uptown CPU area.

Disturbed Land

Disturbed land includes undeveloped areas where vegetation has been removed and supports primarily non-native plant species. These lands may have also been modified by activities such as off-road vehicle use. Disturbed land is typically located along the interface between the urban habitat areas and undeveloped canyons.

Eucalyptus Woodland

Eucalyptus woodland is comprised of stands of eucalyptus trees (*Eucalyptus* spp.). These trees are not native to the area and are considered invasive species because of their rapid growth rate, broad cover, and allelopathic chemicals contained in their leaf litter that prevents understory species from growing. Once established, eucalyptus groves often form dense canopies that displace native habitats over time.

2.3.8.4 Sensitive Biological Resources

Biological resources are considered sensitive if they are: (1) covered species or narrow endemic species under the City of San Diego's Multiple Species Conservation Program (MSCP) Subarea Plan and Biology Guidelines, (2) listed by state or federal agencies as threatened or endangered or are proposed for listing; (3) on California Rare Plant Rank 1B (considered endangered throughout its

range) or California Rare Plant Rank 2 (considered endangered in California but more common elsewhere) of the California Native Plant Society (CNPS) Inventory of Rare and Endangered Vascular Plants of California (2012); or (4) considered rare, endangered, or threatened by the California Natural Diversity Data Base (CNDDB; State of California 2014) or local conservation organizations or specialists. Noteworthy plant species are considered to be those that are on California Rare Plant Rank 3 (more information about the plant's distribution and rarity needed) and California Rare Plant Rank 4 (plants of limited distribution) of the CNPS Inventory. Sensitive vegetation communities are those identified by the CNDDB, the Jepson Online Interchange, or identified by the City of San Diego (2012). Assessments for the potential occurrence of sensitive species are based upon review of species occurrence records from the CNDDB, known ranges, and habitat preferences for the species relative to habitat types present in the Uptown CPU area.

a. Sensitive Vegetation Communities

Sensitive vegetation communities are those communities that are of highly limited distribution. These communities may also support concentrations of sensitive plant or wildlife species. Within the City of San Diego's Biology Guidelines, upland vegetation communities have been divided into four tiers of sensitivity. Upland vegetation communities that are classified as Tier I (rare uplands), Tier II (uncommon uplands), or Tier III (common uplands) are considered sensitive by the City. Tier IV (other uplands) vegetation communities are not considered sensitive. The sensitive vegetation community tiers present in the Uptown CPU area are shown in Figure 2-7 and are summarized below.

- Coastal sage scrub, in pristine or disturbed condition, is considered sensitive by federal and state resource agencies due to the scarcity of this vegetation community and the number of sensitive species associated with it. This vegetation community is categorized as a Tier II vegetation community.
- Chaparral is categorized as a Tier IIIA vegetation community. Tier IIIA communities, although common, are considered sensitive as they may support a variety of rare plant and animal species.
- Grassland is classified as a Tier IIIB community. Tier IIIB habitat is considered less valuable than native habitat, but still provides foraging habitat for many species, particularly raptors, and may support a variety of rare plant and animal species.
- Riparian scrub is considered a sensitive wetland habitat by the City of San Diego and resource agencies.

b. Sensitive Plant Species

The sensitive plant species below are known to occur within the Uptown CPU area based on information obtained from CNDDB. Precise locations of sensitive plant species is not available at the program-level analysis conducted for this PEIR and would be identified through on-site reconnaissance and project-level analysis in conjunction with any proposed future development projects. Table 2-<u>76</u> lists the sensitive plant species with known occurrences in the Uptown CPU area. General descriptions of these sensitive plant species are described below.



Table 2-7						
Sensitive Plant Species Known or with the Potential to Occur in the Uptown Community						
		CNPS				
	State/	Rare				
	Federal	Plant	City of San			
Species	Status	Ranking	Diego	Habitat/Blooming Period		
	ANGIC	SPERMS:	DICOTS			
CACTACEAE	CACTUS FA	MILY	1			
<i>Ferocactus viridescens</i> San Diego barrel cactus	-/-	2.1	MSCP	Succulent; chaparral, coastal sage scrub, valley and foothill grassland, vernal pools; blooms May–June; elevation less than 1,500 feet.		
CRASSULACEAE	STONECRO	P FAMILY	T			
<i>Dudleya variegata</i> Variegated dudleya	_/_	1B.2	NE, MSCP	Perennial herb; openings in chaparral, coastal sage scrub, grasslands, vernal pools; blooms May–June; elevation less than 2,000 feet.		
FAGACEAE	OAK FAMIL	Y				
<i>Quercus dumosa</i> Nuttall's scrub oak	_/_	1B.1	-	Evergreen shrub; closed-cone coniferous forest, coastal chaparral, coastal sage scrub, sandy and clay loam soils; blooms FebMarch; elevation less than 1,300 feet.		
LAMIACEAE	MINT FAM	ILY				
Acanthomintha ilicifolia San Diego thornmint	CE/FT	1B.1	NE, MSCP	Annual herb; chaparral, coastal sage scrub, and grasslands on friable or broken clay soils; blooms April–June; elevation less than 3,100 feet.		
ANGIOSPERMS: MONOCOTS						
THEMIDACEAE						
Bioomeria [=Muilia] clevelandii San Diego goldenstar	_/_	2.1	MSCP	Perennial herb (bulbiferous); chaparral, coastal sage scrub, valley and foothill grassland, vernal pools, clay soils; blooms May; elevation 170–1,500 feet.		
See NOTES on next page.						

	Table 2-7				
Sensitive Plant Species Known or with the Potential to Occur in the Uptown Community					
FEDER	RAL C	CANDIDATES AND LISTED PLANTS	STATE LISTED PLANTS		
FE =	Fee	derally listed endangered	CE = State listed endangered		
FT =	Fee	derally listed threatened			
	JF 54	- Narrow ondomic			
	_	- Natrow endernic	vered energies		
INISCP	-	 Multiple Species Conservation Program Col 	rered species		
CALIF	ORN	IA NATIVE PLANT SOCIETY RARE PLANT RAI	NKINGS		
1B	=	Species rare, threatened, or endangered in	California and elsewhere. These species are eligit		
. –		for state listing.			
2	=	Species rare, threatened, or endangered	in California but more common elsewhere. The		
		species are eligible for state listing.			
3	=	Species for which more information is need	ded. Distribution, endangerment, and/or taxonon		
		information is needed.	-		
4	=	A watch list of species of limited distribution	n. These species need to be monitored for chang		
		in the status of their populations.			
.1	=	Species seriously threatened in California	(over 80% of occurrences threatened; high degr		
		and immediacy of threat)			
.2	=	Species fairly threatened in California (20-8	30% occurrences threatened; moderate degree a		
		immediacy of threat)			
.3	=	Species not very threatened in California	(<20% of occurrences threatened; low degree a		
		immediacy of threat or no current threats k	nown)		

c. Listed and MSCP-Covered Plant Species

The sensitive plant species discussed below have known historical occurrences within the Uptown CPU area based on information obtained from CNDDB. Precise locations of sensitive plant species are not available at the plan-level analysis conducted for this PEIR and would be identified through on-site reconnaissance in conjunction with future projects with the potential to impact sensitive biological resources. The distribution of suitable habitat within the Uptown CPU area was used to determine the potential for occurrence of sensitive plant species for the plan level of analysis. Potential areas of effect to sensitive plant species were identified in remnant native habitat existing at the interface of development and the adjacent urban canyons. Native habitat also exists within the canyons. The remaining portion of the Uptown CPU area is built out and does not support sensitive biological resources.

The geographic information system (GIS) analysis showed that only very small areas (less than 0.1 acre per lot) of native habitat may remain on individual lots adjacent to canyon edges that may be impacted by edge effects (e.g., brush management zone 1). Therefore, it was determined that sensitive plant species have a low potential to occur within these areas. The GIS analysis also showed that sensitive plant species have the potential to occur further downslope within the relatively undisturbed native habitats. However, these areas are located where development is not expected to occur. Sensitive plant species could potentially occur within relatively undisturbed native habitats in the canyon areas of the community plans. However, the project involves little or no change to the open space or Multi Habitat Planning Area (MHPA) designations in the urban canyons.

Potentially occurring sensitive species would be conserved in accordance with ESL regulations, the Biology Guidelines, and the provisions of the MSCP Subarea Plan.

San Diego thornmint (*Acanthomintha ilicifolia***).** San Diego thornmint is federally listed as threatened and state-listed as endangered. It is considered a narrow endemic under the MSCP and has a California Native Plant Society (CNPS) Rare Plant Ranking of 1B.1 (rare, threatened, or endangered in California and elsewhere; seriously endangered in California). This annual herb in the mint family (Lamiaceae) flowers from April through June. It is known to occur at elevations between 30 and 3,200 feet in San Diego County and in northern Baja California. Preferred habitat is friable or cracked clay soil in grassy openings within chaparral and coastal scrub. This species has known occurrences within the Uptown community.

San Diego goldenstar (Bloomeria [=Muilla] clevelandii). San Diego goldenstar is a covered species under the MSCP and has a CNPS Rare Plant Ranking of 1B.1 (rare, threatened, or endangered in California, but more common elsewhere; seriously endangered in California). San Diego goldenstar is a bulbiferous herb of the Brodiaea family (Themidaceae). This species is found only in southwestern San Diego County and northern Baja California, where it occurs on clay soils in coastal sage scrub, chaparral, and grassland habitats. It is a perennial bulb threatened by loss, degradation, and conversion of habitat. This species has known occurrences within the Uptown community.

Snake cholla (*Cylindropuntia* [=*Opuntia*] *californica var. californica*). Snake cholla is considered a narrow endemic species under the MSCP and has a CNPS Rare Plant Ranking of 1B.1 (rare, threatened, or endangered in California and elsewhere; seriously endangered in California). It is a generally prostrate cactus (Cactaceae family) that may grow up to 9 feet and blooms with yellow or green-yellow flowers in April and May. This variety grows only in southern San Diego County and Baja California, with the northernmost known location in Florida Canyon in Balboa Park. Snake cholla occurs in coastal sage scrub and chaparral habitats between100 and 500 feet elevation, most often on dry hillsides. It is associated with Huerhuero loam, Gaviota fine sandy loam, and Redding cobbly loam soils. This variety can be distinguished from *C. californica* var. *parkeri* by its range, prostrate form, and shorter tubercle and longer central spine. There is a low potential for occurrence of this species within the Uptown CPU area.

Variegated dudleya (*Dudleya variegata***)** Variegated dudleya is considered a narrow endemic species under the MSCP and has a CNPS Rare Plant Ranking of 1B.2 (rare, threatened, or endangered in California and elsewhere; fairly endangered in California). This small succulent perennial in the stonecrop family (Crassulaceae) emerges from a corm in spring and produces yellow flowers in May and June. Its range extends from southwestern San Diego County to Baja California. It occurs in coastal sage scrub, grassland, and chaparral habitats below 500 feet. It usually grows in stony places lacking shrub cover, on isolated rocky substrate in grasslands, and on mima mounds near vernal pools. It often occurs on gravelly loam soils. This species can be distinguished from many-stemmed dudleya (*D. multicaulis*) by its spoon-shaped, rather than linear, leaves and from Blochman's dudleya (*D. blochmaniae* ssp. *blochmaniae*) by its yellow rather than white flowers. This species has known occurrences within the Uptown community.

Palmer's goldenbush [=Palmer's ericameria] (*Ericameria palmeri var. palmeri* [=*E. palmeri* ssp. *palmeri*]). Palmer's goldenbush is a CNPS List 1B.1 species (rare, threatened, or endangered in California and elsewhere; seriously endangered in California) and is a MSCP-covered species. This

shrub in the sunflower family (Asteraceae) may grow to 5 feet tall and flowers from September to November. Its range extends from San Diego County south into Baja California; the northernmost occurrence is reported from Carmel Valley with most reports from near Jamul and Jamacha. It prefers seasonally moist sites, such as coastal drainages or mesic chaparral, but may occur in coastal sage scrub. It is associated with sandy loam soils. There is a low potential for occurrence of this species within the Uptown CPU area.

San Diego barrel cactus (Ferocactus viridescens). San Diego barrel cactus is a covered species under the MSCP and has a CNPS Rare Plant Ranking of 2B.1 (rare, threatened, or endangered in California, but more common elsewhere; seriously endangered in California). This globular succulent in the cactus family (Cactaceae) grows to 1 foot tall and flowers in May and June. It is found only in coastal San Diego County and Baja California. Although found as far north as Oceanside coastally and Poway inland, the largest populations of coast barrel cactus occur in Otay Mesa and Otay Valley, Point Loma, and Marine Corps Air Station Miramar. This species occurs in sandy and rocky areas in coastal sage scrub and grassland habitats below 500 feet elevation. It is the only barrel cactus found in coastal areas. This species has known occurrences within the Uptown community.

d. Other Sensitive Plant Species

California adolphia (*Adolphia californica*). California adolphia has a CNPS Rare Plant Ranking of 2B.1 (rare, threatened, or endangered in California, but more common elsewhere; seriously endangered in California). This small shrub in the buckthorn family (Rhamnaceae) flowers from December to April and loses its leaves in late summer and fall. Its spiny stems are identifiable at close range year-round, however. This species generally occurs in Diegan coastal sage scrub, near the edge of chaparral, particularly in dry canyons or washes. It is associated with San Miguel and Friant soils. Its range is limited to San Diego County and northern Baja California at elevations below 1,000 feet. In San Diego County, it is found from the Carlsbad area south into the Proctor Valley and the Otay area. There is a low potential for occurrence of this species within the Uptown CPU area.

Decumbent goldenbush (*Isocoma menzezii* **var.** *decumbens***).** Decumbent goldenbush has a CNPS Rare Plant Ranking of 1B.2 (rare, threatened, or endangered in California and elsewhere; fairly endangered in California). This shrub is a member of the Asteraceae family that blooms from April through November. It ranges from Orange County to Baja California, with known occurrences on San Clemente and Santa Catalina islands. Decumbent goldenbush occurs in chaparral and coastal scrub habitats, often preferring sandy substrate and disturbed areas at elevations from 30 to 400 feet above mean sea level. There is a low potential for occurrence of this species within the Uptown CPU area.

Nuttall's scrub oak (*Quercus dumosa***).** Nuttall's scrub oak has a CNPS Rare Plant Ranking of 1B.1 (rare, threatened, or endangered in California and elsewhere; seriously endangered in California). This evergreen shrub in the oak family (Fagaceae) grows less than 10 feet tall and blooms from February to April. This species is found near the coast in Santa Barbara, Orange, and San Diego Counties; and in Baja California, at elevations below 1,300 feet. It grows in chaparral, coastal sage scrub, and closed-cone coniferous forest habitats, preferring coastal chaparral with a relatively open canopy in flat areas, but growing in dense stands on north-facing slopes. In San Diego County it is known to grow as far inland as Camp Elliot and Otay Mesa, being replaced by the similar scrub oak

(*Q. berberidifolia*) in higher, drier locations. Nuttall's scrub oaks can be distinguished from the scrub oak, with which it may hybridize, by its acorn, which is less than 0.4 inch wide, moderately tuberculed, with a thin cup, and by its leaves, which tend to be smaller, spinier, more undulated, and have densely matted gray hairs. This species has known occurrences within the Uptown CPU area. However, there is a low potential for occurrence of this species within Uptown CPU area.

Singlewhorl burrobrush (*Ambrosia monogyra* **[=Hymenoclea monogyra]).** Singlewhorl burrobrush is a CNPS List 2B.2 species. This shrub in the sunflower family (Asteraceae) has slender stems, narrow leaves, and large inflorescences that bloom from August to November. Singlewhorl burrobrush is found in the southwestern United States from California to Texas as well as within northern Mexico. This species occurs in washes and dry riverbeds. There is a low potential for occurrence of this species within the area affected by the Uptown CPU.

e. Sensitive Wildlife Species

The sensitive wildlife species discussed below are known to occur within the CPU area based on information obtained from CNDDB. Precise locations of sensitive wildlife species are not available for this program-level analysis and would be identified through on-site reconnaissance in conjunction with future projects. Table 2-<u>8</u>7 lists the sensitive wildlife with known occurrences in the Uptown area. These sensitive wildlife species are described below.

Table 2-8					
Sensitive Wildlife Sp	ecies Known to Occu	r in the Uptown CPU Area			
Species	Status	Habitat/Comments			
BIRDS (Nomenclature from American Ornithologists' Union 2013 and Unitt 2004)					
TROGLODYTIDAE – Wrens					
Coastal cactus wren Campylorhynchus brunneicapillus	CSC, MSCP, *	Maritime succulent scrub, coastal sage scrub and desert scrub with <i>Opuntia</i> thickets. Rare localized resident.			
MAMMALS (Nomenclature from Baker et al. 2003 and Hall 1981)					
PHYLLOSTOMIDAE – New World Le	eaf-nosed Bats				
Mexican long-tongued bat	CSC	Sightings in San Diego County			
Choeronycteris mexicana		very rare. Migratory.			
STATUS CODES					
CSC = California Department of Fish and Game species of special concern					
MSCP = Multiple Species Conservation Program covered species					
* = Taxa listed with an aste	* = Taxa listed with an asterisk fall into one or more of the following categories:				
 Taxa considered endangered or rare under Section 15380(d) of CEQA guidelines 					
Taxa that are biologically rare, very restricted in distribution, or declining throughout					
their range					
• Population(s) in California that may be peripheral to the major portion of a taxon's range					
but which are threatened with extirpation within California					
• Taxa closely associated with a habitat that is declining in California at an alarming rate					
(e.g., wetlands, ripa	arian, old growth forests,	desert aquatic systems, native grasslands)			

The GIS analysis showed that only very small areas (less than 0.1 acre per lot) of native habitat may remain on individual lots adjacent to canyon edges that may be impacted by edge effects (e.g., brush management zone 1). Therefore, it was determined that sensitive wildlife species have a low potential to occur within these areas. The GIS analysis also showed that sensitive wildlife species have the potential to occur further downslope within the relatively undisturbed native habitats. However, these areas are outside of any potential plan level impacts (i.e., development is not expected to occur); therefore, no significant impacts to sensitive wildlife species are anticipated to occur.

Sensitive wildlife species could potentially occur within relatively undisturbed native habitats in the canyon areas of the community plans. However, the plan update involves little or no change to the open space or MHPA designations in the urban canyons. Potentially occurring sensitive species would be conserved in accordance with ESL regulations, the Biology Guidelines, and the provisions of the MSCP Subarea Plan.

f. Sensitive Birds

Coastal California gnatcatcher (*Polioptila californica californica*). The coastal California gnatcatcher is federally listed as threatened, a CDFW species of special concern, and an MSCP-covered species. The coastal California gnatcatcher is a nonmigratory, resident species found on the coastal slopes of southern California, ranging from Ventura County southward through Los Angeles, Orange, Riverside, and San Diego counties into Baja California. Coastal California gnatcatchers typically occur in or near sage scrub habitat, although chaparral, grassland, and riparian woodland habitats are used where they occur adjacent to sage scrub. Breeding occurs from February through August, and nests are constructed most often in California sagebrush. The coastal California gnatcatcher diet consists mainly of sessile small arthropods, such as leafhoppers, spiders, beetles, and true bugs. The primary cause of decline in the coastal California gnatcatcher is due to habitat loss and degradation. There is a low potential for occurrence of this species within any of the areas affected by the Uptown CPU

Coastal cactus wren (*Campylorhynchus brunneicapillus couesi***).** The coastal cactus wren is a CDFW species of concern and an MSCP-covered species. This species ranges from southern Orange County through San Diego County into extreme northwestern Baja California. Year-round residents, coastal cactus wrens inhabit coastal lowlands containing thickets of cholla and prickly pear cactus in coastal sage and maritime succulent scrub. Coastal cactus wrens build their nests in the cactus and males often build secondary nests, used for roosting by adults and fledglings and nesting for subsequent broods. Nesting occurs from March through July; fledglings remain in the nest until September. Their diet consists mainly of grasshoppers, beetles, ants, wasps, butterflies, moths, spiders, and occasionally vegetation, reptiles, and amphibians. The primary cause for the decline of this species is degradation and loss of breeding habitat loss due to urbanization. This species has known occurrences within the Uptown community. However, the potential for occurrence of this species within Uptown CPU area is low as suitable habitat in the form of cactus thickets are not likely present.

g. Sensitive Mammals

Mexican long-tongued bat (*Choeronycteris mexicana***).** The Mexican long-tongued bat is a CDFW species of special concern. This species' distribution extends from the southern United States, through Mexico and Central Mexico, to northern South America. It has been reported as recently as 1999 in a number of urban locations in San Diego County, including Mount Helix and the San Diego Zoo. In other states, it has been reported in desert and montane riparian habitats, succulent scrub, and pinyon-juniper woodlands, and it roosts in caves, mines, and buildings. This bat is a colonial breeder from May to August. Their diet consists mainly of moths, but they eat other insects such as flies and beetles. Threats to this species include recreational caving; natural or intentional mine closures, renewed mining, mine reclamation, and loss of food resources. Indirectly, development, prescribed fire, or grazing could potentially have negative impacts on food plants. This species has known occurrences within the Uptown community. However, the potential for occurrence of this species the Uptown CPU area is low due to the lack of suitable habitat such as caves and mines, which are not present in the CPU area.

2.3.8.5 Jurisdictional Waters/Wetlands

Agencies with jurisdictional authority over wetlands and other jurisdictional water resources include U.S. Fish and Wildlife Service (USFWS), ACOE, California Department of Fish and Wildlife (CDFW), Regional Water Quality Control Board (RWQCB), and the City of San Diego. Wetland definitions applicable to each agency are described below. A general description of each agencies regulatory authority over jurisdictional waters is provided in Chapter 5.0, Regulatory Framework. Approximately 3.3 acres has been mapped as a wetland (e.g., riparian scrub) within the Uptown community within the bottom of an urban canyon.

a. U.S. Army Corps of Engineers

As stated in the federal regulations for the Clean Water Act, wetlands are defined as:

Those areas that are inundated or saturated by surface or groundwater at a frequency and duration sufficient to support, and that under normal circumstances, do support a prevalence of vegetation typically adapted for life in saturated soil conditions (EPA, 40 Code of Federal Regulations [CFR] 230.3 and CE, 33 CFR 328.3).

Wetlands are delineated using three parameters: hydrophytic vegetation, wetland hydrology, and hydric soils. According to ACOE, indicators for all three parameters must be present to qualify an area as a wetland.

In accordance with Section 404 of the Clean Water Act, ACOE regulates the discharge of dredged or fill material into waters of the U.S. The term "waters of the United States" is defined as:

- All waters currently used, or used in the past, or may be susceptible to use in interstate or foreign commerce, including all waters subject to the ebb and flow of the tide;
- All interstate waters including interstate wetlands;

- All other waters such as intrastate lakes, rivers, streams (including intermittent streams), mudflats, sandflats, wetlands, sloughs, prairie potholes, wet meadows, playa lakes, or natural ponds; the use, degradation, or destruction of which could affect foreign commerce including any such waters: (1) which could be used by interstate or foreign travelers for recreational or other purposes; or (2) from which fish or shellfish are, or could be taken and sold in interstate or foreign commerce; or (3) which are used or could be used for industries in interstate commerce.;
- All other impoundments of waters otherwise as defined as waters of the United States under the definition;
- Tributaries of waters identified above;
- The territorial seas; and
- Wetlands adjacent to waters (other than waters that are themselves wetlands) identified in the paragraphs above [33 CFR Part 328.3(a)].

ACOE also requires the delineation of non-wetland jurisdictional waters. These waters must have strong hydrology indicators such as the presence of seasonal flows and an ordinary high watermark. An ordinary high watermark is defined as:

... that line on the shore established by the fluctuations of water and indicated by physical characteristics such as [a] clear, natural line impressed on the bank, shelving, changes in the character of soil, destruction of terrestrial vegetation, the presence of litter and debris, or other appropriate means that consider the characteristics of the surrounding areas (33 CFR Part 328.3).

Areas delineated as non-wetland jurisdictional waters may lack wetland vegetation or hydric soil characteristics. Hydric soil indicators may be missing, because topographic position precludes ponding and subsequent development of hydric soils. Absence of wetland vegetation can result from frequent scouring due to rapid water flow. These types of jurisdictional waters are delineated by the lateral and upstream/downstream extent of the ordinary high watermark of the particular drainage or depression.

b. U.S. Fish and Wildlife Service

Under Sections 7 and 10 of the Endangered Species Act, USFWS has regulatory authority over federally listed endangered or threatened plant and animal species. Specifically, Section 7 requires agencies to ensure that their activities are not likely to jeopardize the continued existence of listed species or impact designated critical habitats through consultation with the Service. Under Section 7, the USFWS issues a Biological Opinion that serves as the incidental take permit (ITP) associated with a 404 permit authorized by the ACOE. Under Section 10(a)1(A), the USFWS requires the preparation of a habitat conservation plan that accompanies the ITP to ensure that the authorized take is adequately mitigated and minimized.

c. California Department of Fish and Wildlife

Under sections 1600–1607 of the Fish and Wildlife Code, CDFW regulates activities that would divert or obstruct the natural flow or substantially change the bed, channel, or bank of any river, stream, or lake that supports fish or wildlife. CDFW has jurisdiction over riparian habitats (e.g., riparian scrub) associated with watercourses. Jurisdictional waters are delineated by the outer edge of riparian vegetation or at the top of the bank of streams or lakes, whichever is wider.

d. RWQCB Jurisdiction

The San Diego RWQCB is a state agency responsible for protecting water quality in California's San Diego Region (Region 9). The jurisdiction of this agency includes all waters of the state and all waters of the United States as mandated by both the federal Clean Water Act and the California Porter-Cologne Water Quality Control Act. State waters are "any surface water or groundwater, including saline waters, with the boundaries of the state" [Water Code Section 13050(e)].

e. City of San Diego

According to the City of San Diego's Municipal Code (City of San Diego 2012), wetlands are areas which are characterized by any of the following conditions: (1) all areas persistently or periodically containing naturally occurring wetland vegetation communities characteristically dominated by hydrophytic vegetation; (2) areas that have hydric soils or wetland hydrology and lack naturally occurring wetland vegetation communities have removed the historic wetland vegetation; and (3) areas lacking wetland vegetation communities, hydric soils, and wetland hydrology due to non-permitted filling of previously existing wetlands.

2.3.8.6 Wildlife Movement Corridors

Habitat linkages and wildlife corridors are defined as areas that connect suitable wildlife habitat areas in a region otherwise fragmented by rugged terrain, changes in vegetation, or human disturbance. Natural features such as canyon drainages, ridgelines, or areas with vegetation cover provide corridors for wildlife travel. Habitat linkages and wildlife corridors are important because they provide access to mates, food, and water; allow the dispersal of individuals away from high population density areas; and facilitate the exchange of genetic traits between populations. Wildlife movement corridors are considered sensitive by the City of San Diego and resource and conservation agencies.

Within the Uptown CPU area, several canyons occur. However, these canyons are isolated by development from and are not part of a major wildlife corridor system. Although not part of a significant regional corridor, the canyons provide for local wildlife movement, such as birds and small mammals, and serve as a stepping-stone for wildlife species movement between other local canyon systems and into major off-site habitat areas.

2.3.9 Geology

2.3.9.1 Soils and Geologic Formations

The Uptown CPU area is underlain by four surficial soil deposits and three geologic formations. The surficial soils include artificial fill (unmapped), topsoil/colluvium, alluvium (unmapped), and very old terrace deposits (formerly Lindavista Formation). The geologic formations include San Diego Formation, Pomerado Conglomerate, and Mission Valley Formation. Uptown CPU area geology s described and illustrated in Section 6.9. A general discussion of the surficial soils and geologic formations is presented below.

a. Artificial Fill (Unmapped)

Artificial fill is likely present in many areas of the Uptown CPU area. The location, extent, and suitability of the fill would need to be determined during site-specific geotechnical investigations. Artificial fills in older neighborhoods could possibly contain soils environmentally impacted by burn dumps, cesspools, etc.

b. Topsoil And Colluvium (Unmapped)

Varying thickness of topsoil likely blankets the level portions of the Uptown CPU area. Colluvium is present on sloping and natural hillsides within the Community Plan areas. Topsoil and colluvium are generally soft, loose, and/or expansive.

c. Alluvium (QAL)

Alluvial soils are mapped in canyon bottoms. These soils consist of soft sandy to silty clay and interfingers or grades with topsoil and slopewash along the outer edges of canyons. Depth of alluvial materials is anticipated to range from approximately 5 feet in smaller drainages to in excess of 20 feet in major drainages.

d. Very Old Terrace Deposits (QVOP)

Pleistocene age very old terrace deposits (formerly Lindavista Formation) are present at the surface across most of the San Diego Mesa. The very old terrace deposits are described by Kennedy and Tan (2008) as poorly sorted, red brown, interfingered siltstone, sandstone, and conglomerate.

Reed (1991) describes a mudstone unit (proposed, therein, as the Normal Heights Mudstone member of the Lindavista formation) lying on top of the very old terrace deposits. The Normal Heights Mudstone typically ranges from a few feet thick to approximately 10 feet thick, or greater, in localized areas. This mudstone unit displays a "wide variation in structural performance." The mudstone is typically highly expansive.

e. San Diego Formation (TSD)

The Pliocene-age San Diego Formation is exposed on slopes along drainages within the Uptown Community Plan area and underlies the very old terrace deposits. The San Diego Formation consists of dense, yellow-brown, fine- to medium-grained, poorly indurated micaceous sandstone. It is readily eroded and forms uniform slopes along the sides of narrow canyons in the study area. The San Diego Formation is typically massive, and is considered to be flat lying, and exhibits a favorable geologic structure for gross slope stability. Soils derived from this formation are low expansive and have relatively good shear strength characteristics and, as such, can provide good capping materials for pads and higher strength soils for construction of fill slopes. Portions of the San Diego Formation are cohesionless and can erode readily where they are exposed on non-conforming slope faces.

f. Pomerado Conglomerate (TP)

Tertiary-age Pomerado Conglomerate is mapped on the north-facing slopes primarily in the northern portions of the Uptown CPU area. The Pomerado Conglomerate is typically a cobble conglomerate embedded in a silty to clayey sand soil matrix. The Pomerado Conglomerate is favorable for overall slope stability.

g. Mission Valley Formation (TMV)

Tertiary-age Mission Valley Formation is exposed in the canyons and north-facing slopes in the northern portions of the Uptown CPU area. The Mission Valley Formation is composed of light gray, friable, fine to medium grained sandstone with occasional cobble conglomerate tongues. The Mission Valley Formation is generally flat-lying or nearly horizontally bedded and is favorable for overall slope stability.

2.3.9.2 Faulting and Seismicity

a. Geologic Hazard Category

Review of the 2008 City of San Diego Seismic Safety Study, Geologic Hazards and Faults, indicates the majority of the Uptown CPU area is mapped as Geologic Hazard Category (GHC) 52. Category 52 is "other level areas, gently sloping to steep terrain, favorable geologic structure, low risk". The northern boundary of the Uptown CPU area is designated as GHC 53, which is "level or sloping terrain, unfavorable geologic structure, low to moderate risk". The south end of the CPU area is mapped within the downtown special fault zone, GHC 13. Figure 6.9-2 shows the Uptown Community Plan area boundary superimposed on the 2008 City of San Diego Seismic Safety Study.

b. Faulting

Review of published geologic literature indicates the Uptown CPU area is located on the east margin of the Rose Canyon Fault Zone (see Figure 6.9-2). The Rose Canyon Fault Zone is characterized by a zone of north- trending, strike-slip faults, portions of which are deemed active by the State of California.

The *City of San Diego Seismic Safety Study, Geologic Hazards and Faults* (2008) Grid Tiles 20 and 21 map faults crossing the northwestern portion of the planning area. These faults are described as "potentially active, inactive, presumed inactive, or activity unknown". These faults have been named as the Old Town and Mission Bay fault segments of the Rose Canyon Fault Zone. Some researchers (Rockwell 2010 in GEOCON 2015) deem faulting in Old Town, near the Mormon Battalion Historic Site and the Presidio Hills Golf Course, to be active.

Kennedy (Kennedy 1975 in GEOCON 2015) indicates the Old Town fault has vertically offset sediments approximately 100,000 years old by more than 20 meters, indicating late Quaternary activity. Typically, building set-backs are not required on potentially active or inactive faults. However, considering the proximity of these faults to the Rose Canyon fault, site-specific fault studies should be performed where development extends across the identified fault zones. Additionally, these faults are considered to have a potential for surface rupture, unless site-specific studies demonstrate otherwise.

2.3.9.3 Groundwater

Near surface groundwater (less than 20 feet deep) is unlikely in geologic formations within the Uptown community. Subsurface water may be present at depth in alluvial soils deposited in canyon drainage channels.

2.3.10 Paleontology

Paleontological resources, or fossils, are the remains and/or traces of prehistoric plant and animal life. Fossils provide direct evidence of ancient organisms and document the patterns of organic evolution and extinction that have characterized the history of life. Fossil remains, such as bones, teeth, shells, and wood, are found in the geologic deposits (formations) within which they were originally buried. Paleontological resources contain not only the actual fossil remains, but also the localities where those fossils are collected and the geologic formations containing the localities. Fossil remains are important, as they provide indicators of the earth's chronology and history. They represent a limited, nonrenewable, and sensitive scientific and educational resource.

The potential for fossil remains at a location can be predicted through previous correlations that have been established between the fossil occurrence and the geologic formations within which they are buried. Geologic formations possess a specific paleontological resource potential wherever the formation occurs based on discoveries made elsewhere in that particular formation. To evaluate paleontological resources in the proposed CPU area, the presence and distribution of geologic formations, and the respective potential for paleontological resources must be evaluated.

Geologic formations located within the Uptown CPU area include San Diego Formation, Pomerado Conglomerate, and Mission Valley Formation, described in Section 2.3.9, Geology, above. A paleontological resource sensitivity matrix is provided in Table 2-<u>98</u> that identifies the geologic formation, location of potential occurrence, and its sensitivity rating. Paleontological resource sensitivity of geologic formations is typically rated from high to zero. The sensitivity of the paleontological resource determines the significance of a paleontological impact. The specific criteria applied for each sensitivity category are summarized below.

Table 2-9 Paleontological Determination Matrix				
Geological Deposit/Formation/Rock		Sensitivity		
Unit	Potential Fossil Localities	Rating		
Alluvium (Qsw, Qal, or Qls)	All communities where unit occurs	Low		
Ardath Shale (Ta)	All communities where unit occurs	High		
Bay Point/Marine Terrace (Qbp) ¹	All communities where unit occurs	High		
Cabrillo Formation (Kcs)	All communities where unit occurs	Moderate		
Delmar Formation (Td)	All communities where unit occurs	High		
Friars Formation (Tf)	All communities where unit occurs	High		
Granite/Plutonic (Kg)	All communities where unit occurs	Zero		
	Mira Mesa/Tierrasanta	High		
Lindavista Formation (QIn, QIb) ²	All other areas	Moderate		
Lusardi Formation (Kl)	Black Mountain Ranch/Lusardi Canyon Poway/Rancho Santa Fe	High		
	All other areas	Moderate		
Mission Valley Formation (Tmy)	All communities where unit occurs	High		
	Rose Canvon	High		
Mt. Soledad Formation (Tmv)	All other areas where unit occurs	Moderate		
Otay Formation (To)	All communities where unit occurs	High		
Point Loma Formation (Kp)	All communities where unit occurs	High		
	Scripps Ranch/Tierrasanta			
Pomerado Conglomerate (Tp)	All other areas	High		
River/Steam Terrace Deposits (Qt)	South Eastern/Chollas Valleys/ Fairbanks Ranch/Skyline/Paradise Hills/Otay Mesa, Nestor/San Ysidro	Moderate		
	All other areas	Low		
San Diego Formation (Qsd)	All communities where unit occurs	High		
Santiago Peak Volcanics (Jsp) Metasedimentay	Black Mountain Ranch/La Jolla Valley, Fairbanks Ranch/Mira Mesa/ Peñasquitos	Moderate		
Santiago Peak Volcanics (Jsp) Metavolcanic	All other areas	Zero		
Scripps Formation (Tsd)	All communities where unit occurs	High		
Stadium Conglomerate (Tst)	All communities where unit occurs	High		
Sweetwater Formation	All communities where unit occurs	High		
Torroy Sandstone (Tf)	Black Mountain Ranch/Carmel Valley	High		
Torrey Sandstone (11)	All other areas	Low		
Sensitivity RatingGrading Thresholds for Required MonitoringHigh=>1,000 cubic yards and 10 feet+ deepModerate=>2,000 cubic yards and 10 feet+ deepZero-Low=Monitoring not requiredBaypoint ¹ – Broadly correlative with Qop 1-8 of Kennedy and Tan (2008) new mapping nomenclature.Lindavista ² – Broadly correlative with Qvop 1-13 of Kennedy and Tan (2008) new mapping nomenclature.				
*Monitoring is always required when grading on a fossil recovery site or near a fossil recovery site in the same				
geologic deposit/formation/rock unit as the project site as indicated on the Kennedy Maps. **Monitoring may be required for shallow grading (i.e., <10ft) when a site has previously been graded and/or				

**Monitoring may be required for shallow grading (i.e., <10ft) when a site has previously been graded and/or unweathered geologic deposits/formations/rock units are present at the surface.

***Monitoring is not required when grading documented or undocumented artificial fill.

SOURCE: City of San Diego CEQA Significance Thresholds 2011.

- High Sensitivity These formations contain a large number of known fossil localities. Generally, highly sensitive formations produce vertebrate fossil remains or are considered to have the potential to produce such remains.
- Moderate Sensitivity These formations have a moderate number of known fossil localities. Generally, moderately sensitive formations produce invertebrate fossil remains in high abundance or vertebrate fossil remains in low abundance.
- Low and/or Unknown Sensitivity These formations contain only a small number of known fossil localities and typically produce invertebrate fossil remains in low abundance. Unknown sensitivity is assigned to formations from which there are no known paleontological resources, but which have the potential for producing such remains based on their sedimentary origin.
- Very Low Sensitivity Very low sensitivity is assigned to geologic formations that, based on their relative youthful age and/or high-energy depositional history, are judged to be unlikely to produce any fossil remains.

2.3.11 Hydrology and Water Quality

2.3.11.1 Drainage

The Uptown CPU area is located on a mesa top incised with a complex network of canyons. Drainage occurs in two directions. The northern portion of the mesa drains through the canyons and storm drains to the San Diego River, located within Mission Valley to the north. The southern portion of the mesa drains via the canyon systems and storm drains to San Diego Bay (City of San Diego 2015).

The Uptown CPU area is located in the San Diego Basin Planning Area Hydrologic Unit and lies within two Hydrologic Subareas of the San Diego River watershed. The northern portion of the Uptown community is located within the Mission San Diego Hydrologic Subarea 907.11 (Figure 2-8). With a land area of approximately 440 square miles, the San Diego River watershed is the second largest hydrologic unit in San Diego County. The watershed's is approximately 475,000 residents and contains portions of the cities of San Diego, El Cajon, La Mesa, Poway, and Santee and several unincorporated areas. Approximately 58.4 percent of the San Diego River watershed is currently undeveloped. Important hydrologic resources in the watershed include five water storage reservoirs, a large groundwater aquifer, extensive riparian habitat, coastal wetlands, and tide pools.

The southern portion of the Uptown community is located within the Pueblo San Diego Hydrologic Subarea 908.21. The Pueblo San Diego watershed is the smallest hydrologic unit in San Diego County, encompassing approximately 60 square miles of predominantly urban landscape in the cities of San Diego, La Mesa, Lemon Grove, and National City. The watershed contains the smallest proportion of unincorporated area (0.3 percent) of the hydrologic units within the county. The population of the Pueblo San Diego watershed is approximately 500,000 residents, making it the county's most densely populated watershed. Approximately 75 percent of the watershed is developed. Due to the high level of existing urbanization in the watershed, only small amounts of additional land is projected for development over the next 15 years (Project Clean Water 2016).

ATTACHMENT 7



FIGURE 2-8 Uptown Watersheds

2.3.11.2 Water Quality

Urban runoff is surface water runoff generated from developed or disturbed land associated with urbanization. The increase in impervious surfaces and fewer opportunities for infiltration within the landscape increase storm flows and provides a source for sediment and other pollutants to enter receiving waters. Urban runoff is a major component of urban flooding and is a particular problem for management of watersheds. Urban runoff is the largest pollution source of Southern California's coastal beaches and near-shore waters. Urban runoff control programs typically focus on managing the effect that new impervious surfaces have on stream channels but may also provide remediation of existing problems. The northern portion of the community is within the San Diego Watershed, which <u>comprises-includes</u> the San Diego River, and the southern portion is within the Pueblo San Diego Watershed, which ultimately discharges into San Diego Bay.

a. San Diego River

The San Diego River generally flows to the west from the Uptown CPU area and discharges into the Pacific Ocean just north of the Ocean Beach community. The San Diego River has been listed as an "impaired" body under Section 303(d) of the Clean Water Act due to fecal coliform, low dissolved oxygen, phosphorus, total dissolved solids, and toxicity. Major impacts to this watershed include surface water quality degradation, habitat degradation and loss, sediment, invasive species, eutrophication, and flooding. Sources of impacts include urban runoff, agricultural runoff, mining operations, sewage spills, and sand mining.

b. San Diego Bay

The majority of the Uptown CPU drains to San Diego Bay. The beneficial uses of the inland surface waters in the Pueblo San Diego watershed are limited to contact (potential use activities involving a significant risk if ingestion of water, including wading by children and swimming) and non-contact (aquatic recreation pursuits not involving a significant risk of water ingestion, including fishing and limited body contact incidental to shoreline activity) recreation, warm freshwater habitat, and wildlife habitat. The San Diego Bay receiving water supports an extensive array of beneficial uses (RWQCB 1994).

The existing coastal beneficial uses identified for San Diego Bay include industrial service supply, navigation, contact water recreation, non-contact water recreation, commercial and sport fishing, preservation of biological habitats of special significance, estuarine habitat, wildlife habitat, rare, threatened, or endangered species, marine habitat, migration of aquatic organisms, spawning, reproduction, and/or early development, and shellfish harvesting (RWQCB 1994).

The watershed drainage consists of a group of relatively small local creeks and pipe conveyances, many of which are concrete-lined and drain directly into San Diego Bay. The creeks in the watershed are highly impacted by urban runoff, and Chollas Creek and the mouth of the creek in San Diego Bay are listed as 303(d)-impaired water bodies for various trace metals parameters and aquatic toxicity. Several sites in San Diego Bay that are impacted by runoff from the Pueblo San Diego watershed have been identified as hot spots by California's Bay Protection Toxic Cleanup Program.

Although much of the Uptown CPU area drains directly to San Diego Bay, as one of the major conveyances of water to the bay Chollas Creek may include runoff from part of the Uptown CPU area. Impairments from multiple pollutants have led to establishment of Chollas Creek total maximum daily loads (TMDLs). Five TMDLs have been adopted for Chollas Creek: the pesticide (diazinon) TMDL (with a final compliance date of December 31, 2010), the dissolved metals TMDLs (for copper, lead and zinc), and an indicator bacteria TMDL. Multiple agencies, including the City of San Diego, the Cities of La Mesa and Lemon Grove, the County of San Diego, the San Diego Unified Port District, Caltrans, and the U.S. Navy, were among those identified as having responsibility in reducing pollutants to mandated levels. The indicator bacteria TMDL is being re-evaluated based upon new scientific data. Implementation Plans are designed to meet the requirements of the metals and bacteria TMDLs over a 20-year period, with phased incremental reductions required.

2.3.11.3 Groundwater

Groundwater within the San Diego Mesa is exempt from municipal and domestic supply beneficial use by the 1989 Regional Water Quality Control Board's Resolution No. 89-33, as it was determined that this area does not support municipal and domestic supply. Groundwater within the Mission San Diego area of the Lower San Diego portion of the San Diego Hydrologic Unit has a potential beneficial use for municipal and domestic supply and existing beneficial uses for agricultural supply, industrial service supply, and industrial process supply (RWQCB 1994).

2.3.12 Public Infrastructure

The Uptown community is served by a variety public facilities and services, including utilities such as water and sewer, and solid waste disposal. The infrastructure needs for these services are managed through the City's Capital Improvements Projects (CIP) program. The City conducts a biannual review of public services, facilities, and utilities implementation in conjunction with the budget/CIP review cycle. As part of this review process, the City assesses the need for new or expanded services and public facilities in order to provide appropriate services and infrastructure commensurate with population increase.

2.3.12.1 Public Services and Facilities

Existing public services and facilities, including parks, recreation centers, libraries, schools, fire, emergency medical, and police, serve the residents and businesses within the Uptown and surrounding communities. The following provides a discussion of the existing and planned public services and facilities that are, or will be, available to the CPU area. The information provided below is based on communications with the service providers during preparation of this PEIR.

a. Police Protection

Police services are provided by the San Diego Police Department. The Police Department does not staff individual stations based on population ratios. The goal Citywide is to maintain 1.45 officers per 1,000 population ratio, which the Police Department is currently meeting based on a 2010 census-estimated residential population of 1,376,173. The Police Department currently uses a five-level

priority dispatch system, which includes, in descending order: Priority E (Emergency), One, Two, Three, and Four.

Police protection for the Uptown community is provided by the Central Divisions and Western Division of the Police Department. Located at 2501 Imperial Avenue, Central Division serves a population of 103,524 people and encompasses 9.7 square miles. Western Division is located at 5215 Gaines Street serving a population of 129,709 people and encompassing 22.7 square miles. The Central Divisions serves the neighborhoods of Balboa Park, Barrio Logan, Core-Columbia, Cortez, East Village, Gaslamp, Golden Hill, Grant Hill, Harborview, Horton Plaza, Little Italy, Logan Heights, Marina, Memorial, Banker's Hill/Park West, Petco, Sherman Heights, South Park and Stockton. The Western Division serves the neighborhoods of Hillcrest, La Playa, Linda Vista, Loma Portal, Midtown, Midway District, Mission Hills, Mission Valley West, Morena, Ocean Beach, Old Town, Point Loma Heights, Roseville-Fleetridge, Sunset Cliffs, University Heights and Wooded Area.

b. Parks and Recreation

The Uptown community is served by the Mission Hills Park, Old Trolley Barn Park, a small portion of which is located in the North Park community plan area, and West Lewis Street Pocket Park, as well as joint-use facilities at Birney Elementary School (which is shared with North Park) and Roosevelt Middle School. Mission Hills Park, which includes Pioneer Memorial Park, consists of passive recreation amenities, such as multi-purpose turf areas, a children's play area, seating, picnicking, walkways, and landscaping. Old Trolley Barn Park consists of passive recreation amenities, such as multi-purpose turf areas, a children's play area, seating, picnicking, walkways, and landscaping. West Lewis Street Pocket Park consists of passive recreational amenities, a trail, public art, interpretive signage, and seating. Two regional parks border Uptown: Balboa Park (described below) and Presidio Park. Presidio Park, which is located in Old Town San Diego community plan area, encompasses approximately 40 acres and includes Junípero Serra Museum, picnic areas, small venue space, restrooms, monuments, and open lawn space for active and passive recreation.

c. Fire Protection

Fire facilities serve multiple neighborhoods and therefore need to be located on major roads accessible to neighborhoods, and adjacent to freeways when practicable. Fire Station No. 5, located at 3902 Ninth Avenue, Fire Station No. 8, located at 3974 Goldfinch Street, and Fire Station No. 3, located at 725 W. Kalmia Street, provide primary fire protection and advanced life support services to the Uptown community and the surrounding area. All fire department engines and trucks are full Advanced Life Support units and are equipped and capable of managing medical emergencies. The construction of a new fire station was specifically identified by the current Public Facilities Financing Program (PFFP) for the Uptown CPU area and is included in the IFS for the update.

Emergency medical services are also provided to the Uptown community and throughout the City through a public/private partnership between the City's Emergency Medical Services (EMS) and Rural Metro Corporation, which provides additional personnel and some ambulances. EMS has ambulances, paramedics, and emergency medical technicians (EMTs) who respond to emergency calls. Calls are prioritized from Level 1 (most serious) to Level 4 (non-emergency).

d. Libraries

Library services are provided by the San Diego Public Library (SDPL) and its branch locations. Per the City's *Guiding Principles for Library Facilities* (July 2001), the minimum branch library size should be 15,000 square feet. The Library System Improvements Program for the SDPL originally included a new Central Library (completed in 2014) and 23 branch libraries. Nine libraries have been completed with either new construction or expansion. Three branches are in the SDPL five-year plan for either expansion or new construction: Mission Hills/Hillcrest, Skyline Hills, and San Ysidro. Others are in planning and design phases, on hold due to lack of funding, or the projects will be closed until funding is identified.

The Uptown community is served by two branch locations of the San Diego Library system: Mission Hills Branch Library located in Uptown and the University Heights Branch Library located in North Park.

e. Schools

The Uptown community is located within the jurisdiction of the San Diego Unified School District (SDUSD). The Uptown community is served by three public elementary schools: Florence, Alice Birney, and Grant Elementary Schools; Roosevelt Middle School; and San Diego High School.

In 2012, voters approved funding of two bond measures, Propositions S and Z, to fund repairs, and renovate and revitalize schools within the SDUSD. Bond projects build off improvements that were started with Prop MM funding and include classroom technology, safety and security upgrades, Americans with Disabilities Act (ADA) upgrades, new/renovated facilities, temporary classrooms replaced by permanent classrooms, air conditioning, upgrades to ADA improvements in athletic facilities, turf fields, and other capital improvements at traditional and charter schools throughout the district.

All development projects within the City are required to pay school fees in accordance with the requirements of the SDUSD, and as mandated by state law, to accommodate the needs of public schools serving existing and future students.

g. Roadways

The City's Public Works Department provides a full range of engineering services for the City's capital investment in various types of infrastructure, including roadways, and provides traffic engineering services to the communities. The department is responsible for the planning, design, project management, and construction management of public improvement projects, and also for providing traffic operations and transportation engineering services.

Operation and maintenance of roadways are managed by the Streets Division of the City's Transportation and Storm Water Department. The Streets Division is responsible for the maintenance of roadways, bridges, sidewalks, traffic control devices, street lighting, and urban forestry.

h. Water and Sewer Infrastructure

The Uptown Community Plan area is located in the City of San Diego Public Utilities Department (PUD) service area. The PUD serves more than 1.3 million residents in the City and in certain surrounding areas, including both retail and wholesale customers. The PUD relies on imported water as its major water supply source and is a member agency of the San Diego County Water Authority (SDCWA), which is in turn a member agency of the Metropolitan Water District of Southern California (MWD). The PUD currently purchases approximately 85 to 90 percent of its water from the SDCWA, which supplies the water (raw and treated) through two aqueducts consisting of five pipelines. In addition, the PUD uses three local supply sources to meet or offset potable demands: local surface water, conservation, and re-cycled water. The PUD water system extends over 404 square miles, including 324 square miles in the City, and includes potable and recycled water facilities.

Wastewater in the CPU area is managed by the San Diego PUD Wastewater Branch, which operates the two components of the City's wastewater system: the Metropolitan Sewerage System and the Municipal Wastewater Collection System. The metropolitan system treats wastewater for a service area of 450 square miles, stretching from Del Mar and Poway in the north to Alpine and Lakeside in the east and the border of Mexico in the south. The service area includes the City of San Diego and 15 other cities and districts. The system serves a population of about 2.2 million and treats an average of 180 million gallons of wastewater per day.

The Municipal Wastewater Collection System is responsible for the collection and conveyance of wastewater from residences and businesses in the City of San Diego, serving a 330 square mile area with a population of 1.3 million people. The Municipal Wastewater Collection System consists of over 2,894 miles of sewer lines, nine major pump stations, and 75 smaller pump stations. Wastewater is conveyed via the pump stations to North City Water Reclamation Plant, the Point Loma Wastewater Treatment Plant, and the South Bay Water Reclamation Plant. Treated effluent is discharged to the Pacific Ocean through either the Point Loma Ocean Outfall or the South Bay Ocean Outfall.

Current City sewer infrastructure is in need of continued upgrade and replacement to maintain the system. Planned improvements to existing facilities would increase City wastewater treatment capacity to serve an estimated population of 2.9 million through the year 2050, when nearly 340 million gallons per day of wastewater are anticipated to be generated. During the early 1900s, as the City of San Diego developed, sewer lines were added in the canyons to utilize gravity flow to transport sewage to the west for treatment. Of the 2,894 miles of sewer lines in the City, 253 miles are currently situated in canyons and other environmentally sensitive areas. These pipelines and manholes have historically had limited cleaning, because the original maintenance paths to these facilities were not adequately maintained. As a result, a number of sewer spills have occurred within urban canyons or other inaccessible areas over the years. In 2001, the City initiated the Long-Term Canyon Sewer Maintenance Program, which focuses on evaluating each of the City's sewer lines in canyons and environmentally sensitive areas for long-term maintenance access needs. In January of 2002, the City Council adopted two council policies related to this purpose.

Council Policy 400-13 identifies the need to provide maintenance access to all sewers in order to reduce the potential for spills. The policy requires that environmental impacts from access paths in

environmentally sensitive areas should be minimized to the maximum extent possible through the use of sensitive access path design, canyon-proficient maintenance vehicles, and preparation of plans that dictate routine maintenance and emergency access procedures.

Council Policy 400-14 outlines a program to evaluate the potential to redirect sewage flow out of canyons and environmentally sensitive areas to an existing or proposed sewer facility located in City streets or other accessible locations. The policy includes an evaluation procedure that requires both a physical evaluation and a cost-benefit analysis. Based on the analysis, if redirection of flow outside the canyon is found to be infeasible, a Long-Term Maintenance and Emergency Access Plan is required. The plan would be specific to the canyon evaluated, and would prescribe long-term access locations for routine maintenance and emergency repairs along with standard operating procedures identifying cleaning methods and inspection frequency. Additionally, the CPU contains policies that address maintenance and repair of the existing sewer network (CPU PF-1.9).

2.3.12.2 Public Utilities

Public utilities include public water, energy, sewer, storm water, and solid waste collection and recycling that are available to serve the Uptown CPU area. A description of the existing conditions of each of these public utilities is provided below. Potential impacts to public utilities from implementation of the Uptown CPU are discussed in Section 6.13.

a. Water Supply

City of San Diego

The City of San Diego PUD provides water service to more than 1.3 million residents over 404 square miles of developed land in the south central portion of San Diego County, including the Uptown CPU area. In the past, the City relied on water from MWD for 95 percent of its supply. During years of drought this made the City extremely vulnerable to water supply shortages, such as in 1991 when a drought forced MWD to cut its deliveries to San Diego by 30 percent. As a result, SDCWA has implemented a strategy to aggressively diversify its water supply portfolio through the introduction of new local and imported water supplies, so that by 2014 MWD deliveries accounted for around 49 percent of the total supply with new sources and conservation efforts accounting for the remaining 51 percent.

SDCWA secured new imported water supplies through a long-term (45-75 year) water conservation and transfer agreement with the Imperial Irrigation District, which provided approximately 100,000 acre-feet of water from the Colorado River in 2014 and will double by 2021. SDCWA has a separate 110-year agreement to receive approximately 80,000 acre-feet of water from the Colorado River by lining parts of the Coachella and All-American canals.

SDCWA is also in the final stages of executing a \$3.1 billion Capital Improvements Program that involves 50 different projects, including new reservoirs, pipelines, pumping stations, a new regional water treatment facility, and a project to raise the San Vicente Dam to allow for additional local storage. Other strategies involve collaboration with SDCWA's 24 local member retail agencies, and include: promoting water conservation through water use efficiency programs, and the introduction of supplies from groundwater, recycled water, and seawater desalination. Additional information about SDCWA water supply diversification projects is provided in SDCWA's 2010 Urban Water Management Plan (UWMP).

The City PUD receives the majority of its water supply from MWD through the Water Authority. Historic imported water deliveries from the Water Authority to the PUD and local surface water, conservation savings, and recycled water deliveries are shown in Table 2-<u>10</u>9.

Table 2-10 Historic Imported, Local, and Recycled Water Demands to Public Utilities Department					
	Imported	Local Surface		Recycled	
	Water	Water	Conservation ¹	Water	Total ²
Fiscal Year	(acre-feet)	(acre-feet)	(acre-feet)	(acre-feet)	(acre-feet)
1990	233,158	22,500			255,658
1995	162,404	59,204	8,914		230,342
2000	207,874	39,098	17,410	3,250	267,632
2005	204,144	26,584	29,410	4,294	264,432
2010	188,337	13,117	34,317	12,173	247,944
¹ Conserved water is from savings and is not a direct supply.					
² Total includes water supplied and conserved.					

The City water system consists primarily of nine surface water reservoirs with over 408,000 acre-feet of storage capacity, three water treatment plants, 31 treated water storage facilities, and more than 3,213 miles of transmission and distribution lines. The local surface raw water storage facilities are connected directly or indirectly to the City's water treatment operations: Otay Water Treatment Plant, Alvarado Water Treatment Plant, and Miramar Water Treatment Plant. These three plants have a total capacity of 294.4 million gallons per day.

The City's two recycled water facilities, North City Water Reclamation Plan (NCWRP) and South Bay Water Reclamation Plant (SBWRP), were built to treat wastewater to a level approved for landscaping irrigation, manufacturing, and other specified non-potable uses. These recycled water facilities not only provide water to City residents and business, but also to other jurisdictions and water districts, including the City of Poway and the Olivenhain Municipal Water District. As part of the City's water resource strategy, the Water Purification Demonstration Project is examining the use of advanced water purification technology to provide additional water supply. The Demonstration Project will determine the feasibility of a full-scale reservoir augmentation project, which would diversify San Diego's water supply and reduce its dependence on imported water.

The PUD emphasizes the importance of water conservation to minimize water demand and avoid excessive water use. The PUD's Water Conservation Program, established in 1985, accounts for approximately 73,000 acre-feet of potable water savings per year. These savings have been achieved through creation of a water conservation ethic and implementation of programs, policies, and ordinances designed to promote water conservation practices, including irrigation management. In accordance with Municipal Code Section 147.04, all residential, commercial, and industrial buildings, prior to a change in ownership, are required to be certified as having water-conserving plumbing fixtures in place. The PUD also examines new water saving technologies and annually checks

progress toward conservation goals, working collaboratively with the MWD and Water Authority to formulate new conservation initiatives.

The City developed a Long-Range Water Resources Plan (2002–2030) in order to address the projected need for additional water supplies. This plan detailed existing water supplies, new water supply opportunities, objectives and performance measures, and ultimately conclusions and recommendations. The plan is to be implemented in three phases in order to meet the City's growing demands and to make adjustments as necessary. The three phases are 2010, 2020, and 2030.

In May 2011, the City issued a draft 2010 UWMP that addresses the City's water system, water supply sources, historic and projected water use, and provides a comparison of water supply to water demands during average, single-dry, and multiple-dry year periods. The UWMP was prepared in accordance with the Urban Water Management Act (as amended, California Water Code, Sections 10610 through 10656), which requires every urban water supplier that provides water for municipal purposes to more than 3,000 connections or supplying more than 3,000 acre-feet of water annually to adopt and submit a plan every five years to the California Department of Water Resources.

In accordance with the Conservation Element of the City's General Plan (Policy CE-A.11), development projects shall implement sustainable landscape design such as planting "deciduous shade trees, evergreen trees, and drought-tolerant native vegetation, as appropriate, to contribute to sustainable development goals" and using "recycled water to meet the needs of development projects to the maximum extent feasible" to aid in water conservation (City of San Diego 2008a).

The area of Uptown is served by existing six-inch- to 36-inch-diameter public water mains located in a grid pattern within the connecting streets. Water is distributed to businesses and residences through private water lines that connect to the public water main.

Metropolitan Water District of Southern California

The MWD was formed in 1928 to develop, store, and distribute supplemental water in southern California for domestic and municipal purposes. The MWD is a wholesale supplier of water to its member agencies, which include the SDCWA. It obtains supplies from local sources as well as the Colorado River via the Colorado River Aqueduct, which it owns and operates, and the Sacramento-San Joaquin Delta via the State Water Project. Planning documents such as the Regional Urban Water Management Plan (RUWMP) and Integrated Water Resources Plan (IWRP) help ensure the reliability of water supplies and the infrastructure necessary to provide water to southern California.

MWD's 2010 RUWMP documents the availability of these existing supplies and additional supplies necessary to meet future demands, includes the resource targets included in the IWRP, and contains a water supply reliability assessment that includes a detailed evaluation of the supplies necessary to meet demands over a 25-year period in average, single-dry year, and multiple-dry year periods. MWD's recently adopted IWRP (2010) identifies a mix of resources (imported and local) that, when implemented, will provide 100 percent reliability for full-service demands. Service demands will be met through the attainment of regional targets set for conservation, local supplies, State Water Project supplies, Colorado River supplies, groundwater banking, and water transfers through year 2035.

San Diego County Water Authority

The SDCWA purchases water from the MWD that is delivered to the region through two aqueducts. Of the MWD's 26 cities and member agencies, the SDCWA is the largest member agency in terms of deliveries and purchases, with about 25 percent of all the water that MWD delivered in fiscal year 2007. As a retail member agency of the SDCWA, the PUD purchases water from the SDCWA for retail distribution within its service area. As discussed above, in 2014 MWD deliveries accounted for around 49 percent of the total supply with new sources and conservation efforts accounting for the remaining 51 percent.

The SDCWA's 2010 UWMP was adopted by the SDCWA Board on June 23, 2011, in accordance with state law and the RUWMP. The plan contains a water supply reliability assessment that identified a diverse mix of imported and local supplies necessary to meet demands over the next 25 years in average, single-dry year, and multiple-dry year periods. The UWMP documents that no shortages are anticipated within its service area. The SDCWA also prepared an annual water supply report for use by its members that provides updated documentation on existing and projected water supplies.

PUD Water Supply Assessment and Verification

Senate Bill (SB) 221 and SB 610 went into effect January 2002, with the intention of linking water supply availability to land use planning by cities and counties. SB 610 requires water suppliers to prepare a Water Supply Assessment (WSA) report for inclusion by land use agencies during the CEQA process for new developments subject to SB 221. SB 221 requires water suppliers to prepare written verification that sufficient water supplies are planned to be available prior to approval of large-scale subdivision of land under the State Subdivision Map Act. As defined in SB 221 and SB 610, large-scale projects include residential development projects of more than 500 residential units and/or shopping centers or businesses employing more than 1,000 people or having more than 500,000 square feet of floor space.

The City's PUD prepared WSA reports for the project (May 2015), which are included as Appendix K to this PEIR. The WSA reports were prepared for the project to assess whether sufficient water supplies are, or will be, available to meet the projected water demands associated with the proposed land use scenarios. Because no subdivision of land is proposed as part of this project, the WSA reports were prepared in compliance with the requirements of SB 610. The WSA reports include, among other information, identification of existing water supply entitlements, water rights, water service contracts, or agreements relevant to the identified water supply for the project; and quantities of water received in prior years pursuant to those entitlement, rights, contracts, and agreements.

b. Water, Sewer, and Storm Water Infrastructure

Wastewater in the Uptown Community Plan area is managed by the PUD Wastewater Branch, which operates the two components of the City's wastewater system: the Metropolitan Sewerage System and the Municipal Wastewater Collection System. The metropolitan system treats wastewater for a service area of 450 square miles, stretching from Del Mar and Poway in the north to Alpine and Lakeside in the east and the border of Mexico in the south. The service area includes the City of San

Diego and 15 other cities and districts. The system serves a population of about 2.2 million and treats an average of 180 million gallons of wastewater per day.

The Municipal Wastewater Collection System is responsible for the collection and conveyance of wastewater from residences and businesses in the City of San Diego, serving a 330-square-mile area with a population of 1.3 million people. The Municipal Wastewater Collection System consists of over 2,894 miles of sewer lines, nine major pump stations, and 75 smaller pump stations. Wastewater is conveyed via the pump stations to North City Water Reclamation Plant, the Point Loma Wastewater Treatment Plant, and the South Bay Water Reclamation Plant. Treated effluent is discharged to the Pacific Ocean through either the Point Loma Ocean Outfall or the South Bay Ocean Outfall.

The largest pump stations in the collection system are pump stations #1 and #2. Pump Station #1, located on East Harbor Drive, collects all of south San Diego's wastewater and has an average daily flow of 75 million gallons. It sends the wastewater flow north via the 8-mile South Metro Interceptor to Pump Station #2, which is located on North Harbor Drive. The average daily flow into Pump Station #2 is approximately 180 million gallons. This station pumps the wastewater to the Point Loma Wastewater Treatment Plant through two 87-inch force mains.

The Point Loma Wastewater Treatment Plant, located on the coast, processes approximately 175 million gallons a day of wastewater generated by 2.2 million residents and workers. The plant has a treatment capacity of 240 million gallons per day. The plant discharges to the Point Loma Ocean Outfall, a 4.5-mile-long outfall that ends at a depth of 320 feet. The current modified National Pollutant Discharge Elimination System permit for the Point Loma Wastewater Treatment Plant and outfall was renewed in 2010.

The PUD also operates the Metro Biosolids Center, a state-of-the-art regional biosolids treatment facility, which turns waste into dewatered biosolids that are currently used as soil amendments, landfill, and landfill cover, but which also may be used to promote growth of agricultural crops. Skim from the Point Loma Wastewater Treatment Plant is transported through the 17-mile Miramar Sludge Pipeline for treatment at the Biosolids Center along with solids from the North City Water Reclamation Plant. Any remaining wastewater from the treatment process is returned to the Point Loma Wastewater Treatment Plant.

The San Diego PUD anticipates that planned improvements to the wastewater system will increase capacity to serve a population of 2.9 million, or 340 million gallons of wastewater per day, by the year 2050. Beginning in 2007, the City increased water and sewer rates to replace and improve both the water and sewer systems infrastructure. Some pipelines have been in operation for a hundred years and need to be replaced. The City of San Diego Water Department's Capital Improvement Program Guidelines and Standards provides the framework for the design and construction of new water facilities and address water efficiency, conservation, recycled and reclaimed water, cost effectiveness, and timely construction.

The City also monitors and maintains the water and sewer system on an ongoing basis because of the age of the water and sewer infrastructure in the older communities. In a continuing replacement program, outmoded concrete sewer mains and cast iron water mains are being replaced on a Citywide basis through the annual Capital Improvements Program. Replacement is currently scheduled based on breaks or blockages in the mains.
The Transportation and Storm Water Department is responsible for the operation and maintenance of streets, sidewalks, and storm drains; leads efforts to protect and improve the water quality of rivers, creeks, bays, and the ocean; performs traffic and transportation system engineering; manages the utilities undergrounding program; and plans and coordinates work in the public right-of-way. Storm drains are designed to handle normal water flow, but occasionally during heavy rain, flooding will occur. Storm drain infrastructure within the community's streets often discharges into the natural canyon areas, causing erosion. Storm water pollution affects people as well as aquatic plant and animal life. Oil and grease from parking lots and roads, leaking petroleum storage tanks, pesticides, cleaning solvents, and other toxic chemicals can contaminate storm water and be transported into water bodies and receiving waters.

While storm drain infrastructure within public streets in the community still needs to be upgraded, new regulations require storm water flow to be controlled within individual sites. The City's Municipal Separate Storm Sewer System Permit (MS4 Permit), issued by the San Diego RWQCB, requires all development and redevelopment projects to implement storm water source control and site design practices to minimize the generation of pollutants. Additionally, the permit requires new development and significant redevelopment projects that exceed certain size threshold to implement Structural Storm Water Best Management Practices (Structural BMPs) to reduce pollutant in storm water runoff and control runoff volume. There is also an increased reliance on Low Impact Development (LID) strategies to meet the MS4 Permit <u>and total maximum daily load</u> requirements, and total maximum daily load as well. Examples of LID techniques <u>are include</u> bioretention cells, green roofs, permeable pavement, infiltration basins, and biofiltration planters.

c. Solid Waste

The City provides refuse, recycling, and yard waste collection and disposal services to some residents under the People's Ordinance (Municipal Code Section 66.0127), adopted in 1919. The free solid waste collection services provided by the City are to primarily single-family homes, and some multi-family and commercial/business customers through General Fund monies. Most multi-family residences are not served and are required to fund and contract directly with private haulers for trash and recycling collection.

Solid waste generated in the Uptown CPU area is collected by private franchised haulers and taken to one of three active landfills permitted to accept solid waste: West Miramar Sanitary Landfill, Otay Landfill, and Sycamore Sanitary Landfill. The Miramar and Sycamore landfills are both located in the City, while Otay Landfill is located in the County of San Diego. Based on current and projected disposal rates and permitted disposal limits, the San Diego region is anticipated to exceed the ability of existing landfills to accept waste within the next 10 years unless landfill expansions are approved.

The Miramar Landfill is permitted to receive 8,000 tons per day, and on average, it receives less than 1,000,000 tons per year. The anticipated closure date for the landfill is 2022. The Sycamore Landfill is permitted to receive a maximum of 3,965 tons per day, although the permit and the facility franchise are inconsistent. The owner/operator is currently proposing a significant increase in throughput, together with a major expansion of the height and footprint of the facility. The Sycamore Landfill, based on a 3,965-ton-per-day limit, is expected to operate until 2031. In order to meet the region's long-term (year 2050) solid waste needs, the Sycamore Landfill expansion has been proposed. The

Sycamore Landfill Master Plan proposes to increase the landfill capacity to 157 million cubic yards, which would allow an increase from 3,965 tons per day to approximately 11,450 tons per day. With the proposed expansion, the landfill would be operational until approximately 2050. This increase in landfill capacity is not currently approved or permitted, and therefore cannot be guaranteed to be completed at this time. The Otay Landfill is permitted to receive 5,830 tons per day. Permits were recently modified, which reduced the overall height of the landfill with no loss of capacity. The Otay Landfill is expected to serve the region through 2021.

In an effort to address landfill capacity and solid waste concerns, the California Legislature passed the Integrated Waste Management Act in 1989 (Assembly Bill 939), which mandated that all cities reduce waste disposed in landfills from generators within their borders by 50 percent by the year 2000. In response, the City Environmental Services Department developed the Source Reduction and Recycling program that outlines waste management policies and programs to meet the City's long-term disposal needs and achieve the mandated waste reduction. Since 2004, the City has diverted more than 50 percent of its generated waste stream from disposal. The City adopted the Recycling Ordinance in November 2007, and phased implementation of the ordinance over the next two years.

The State enacted Assembly Bill 341 in 2011, which established a policy goal for California that no less than 75 percent of solid waste generated be source reduced, recycled, or composted by 2020. Additionally, California Department of Resources Recycling and Recovery's (CalRecycle's) Strategic Directive 6.1 (CalRecycle 2015) calls for a 50-percent reduction in organic waste disposed by 2020. Compliance with and implementation of the above state regulations and policy goals could potentially extend the life of existing landfills. On July 13, 2015, the City adopted a Zero Waste Plan, which would result in 70 percent waste diversion by 2020, 90 percent waste diversion by 2035 and 100 percent diversion by 2040.

A report was prepared by CalRecycle and issued in May 2012, detailing strategies to achieve Assembly Bill 341 goal primarily through recycling. In July 2012, the City updated the Recycling Ordinance to lower the exemption threshold for required recycling, thereby requiring all privately serviced businesses, commercial/institutional facilities, apartments, and condominiums generating four or more cubic yards of trash per week to recycle.

Relative to development activities, pursuant to the City's Significance Determination Thresholds, any land development project that may generate approximately 60 tons of waste or more during construction and/or operation is required to prepare a project-specific Waste Management Plan to address disposal of waste generated during short-term project construction and long-term post-construction operation. The WMP is required to identify how the project would reduce waste and achieve target reduction goals and must include: projected waste generation calculations and identification of the types of waste materials generated; description of how materials would be reused on- site; identification of source separation techniques for recycling; and identification of recycling and reuse facilities where waste would be taken if not reused on-site. The WMP reduces solid waste impacts to below a level of significance. In tandem with the WMP, all new development projects must comply with the City's Construction and Demolition Ordinance and Section 142.08 of the LDC, which outlines the requirements for refuse and recyclable materials storage.

d. Energy

Electricity

San Diego Gas & Electric (SDG&E) is the owner and operator of electricity transmission, distribution, and natural gas distribution infrastructure in San Diego County, and currently provides gas and electric services to the Uptown community. SDG&E is regulated by the California Public Utilities Commission (CPUC). The CPUC sets the gas and electricity rates for SDG&E and is responsible for making sure that California utilities customers have safe and reliable utility service at reasonable rates, protecting utilities customers from fraud, and promoting the health of California's economy.

There are two major operating power plants in San Diego County: the Encina Power Plant and the San Onofre Nuclear Generating Station. However, it should be noted that the reactors at the San Onofre Nuclear Generating Station have been offline since January 2012. There are also a number of smaller generating plants in the county that are used as backup during times of peak power demand. These in- region assets are currently capable of generating approximately 2,360 megawatts (MW) of electricity, about 55 percent of the region's summer peak demand. However, San Diego's older in-region resources typically run at partial capacity (1,628 MW) due to air quality, high fuel cost, and other reasons. Power generation and power use are not linked geographically. Electricity generated is fed into the statewide grid and is generally available to any users statewide. SDG&E purchases electricity from this statewide grid through various long-term contracts.

Along with traditional utilities, private generating companies, and state agencies, the California Independent System Operator (ISO) is a component of the state's electricity industry. The ISO is a not-for-profit public benefit organization that operates the state's wholesale power grid. The California ISO strives to make sure California's electricity needs are met.

Natural Gas

Natural gas is imported into the San Diego region by pipeline after being produced at any of several major supply basins located from Texas to Alberta, Canada. Although the San Diego region has access to all of these basins by interstate pipeline, the final delivery into the SDG&E system is dependent on just one Southern California Gas Company (SoCalGas) pipeline which enters San Diego County from Orange County located along I-5.

Natural gas consumption by sector varies somewhat each year. In general, power plants account for the highest percentage of natural gas consumption in the San Diego region. Residential consumption of natural gas for heating and cooking is the second highest percentage, followed by cogeneration, commercial and industrial consumption, and natural gas fueled vehicles.

Solar Energy

In San Diego, solar energy can be used as an alternative to fossil-fuel energy via private on-site installation/generation or through earmarked purchase of green power from SDG&E. The California Energy Commission (CEC) mandated SDG&E to provide 20 percent of its total energy from solar or other renewable energy sources by the year 2010. While SDG&E missed this goal in 2010, the *Renewables Portfolio Standard Quarterly Report, 1st and 2nd Quarter 2012,* issued by CPUC, states that

SDG&E, the region's primary energy provider, "served 20.8 percent of its 2011 retail sales with RPSeligible renewable energy", thereby meeting the 2010 goal. SDG&E is on track to meet a 25 percent goal by 2016, as well as the long-term goal of 33 percent by 2020.

Currently, there are no mandated standards or ordinances requiring reliance on alternative energy by new developments. However, the City's Climate Action Plan (CAP) establishes a goal to achieve 100 percent renewable energy on the Citywide electrical grid by 2035. Additionally, Title 24 of the California Public Resources Code does contain mandated energy efficiency requirements for all new developments.

e. Communications

Communications systems for telephone, computers, and cable television are serviced by utility providers such as AT&T, Cox, Time Warner, and other independent cable companies. In addition, television services are available from the two satellite services, Direct TV and Dish. Facilities are located above and below ground within private easements. In recent years, the City has initiated programs to promote economic development through the development of high-tech infrastructure and integrated information systems. The City also works with service providers to underground overhead wires, cables, conductors, and other overhead structures associated with communication systems in residential areas in accordance with proposed development projects. Individual development projects consisting of more than four lots are subject to San Diego Municipal Code Section 144.0240, which requires privately owned utility systems and service facilities to be placed underground.

2.3.13 Health and Safety

A hazardous material is any item or agent (biological, chemical, radiological, and/or physical), which has the potential to cause harm to humans, animals, or the environment, either by itself or through interaction with other factors. Hazardous materials are defined and regulated in the United States primarily by laws and regulations administered by the U.S. EPA, the U.S. OSHA, the U.S. DOT, and the U.S. NRC. Each agency has its own definition of a "hazardous material." Some common definitions are included below.

2.3.13.1 Hazardous Materials

Hazardous materials are substances with certain physical or chemical properties that could pose a substantial present or future hazard to human health or the environment when improperly handled, disposed, or otherwise managed. Title 22 of the California Code of Regulations, Division 4.5, Chapter 11, Article 3 groups hazardous materials into the following four categories based on their properties: toxic (causes human health effects), ignitable (has the ability to burn), corrosive (causes severe burns or damage to materials), and reactive (causes explosions or generates toxic gases). Hazardous materials are commonly used in commercial, agricultural and industrial applications as well as in residential areas to a limited extent.

2.3.13.2 Hazardous Waste

A hazardous waste is any waste that may (1) cause, or significantly contribute to an increase in mortality or an increase in serious irreversible, or incapacitating reversible, illness, or (2) pose a substantial present or potential hazard to human health or the environment, due to factors including, but not limited to, carcinogenicity, acute toxicity, chronic toxicity, bio-accumulative properties, or persistence in the environment, when improperly treated, stored, transported, or disposed of, or otherwise managed (California Health and Safety Code, Section 25141). Hazardous materials and wastes can result in public health hazards if improperly handled, released into the soil or groundwater, or released into the air through vapors, fumes, or dust.

2.3.13.3 Hazardous Materials Sites

Hazardous materials are used for a variety of purposes including service industries, various small businesses, medical uses, schools, and households. Many chemicals used in household cleaning, construction, dry cleaning, film processing, landscaping, and automotive maintenance and repair are considered hazardous. Businesses that handle/generate hazardous materials within the City are monitored by the U.S. EPA. Small quantity hazardous waste generators include facilities such as automotive repair, dry cleaners, and medical offices.

2.3.13.4 Wildfire Hazards

Extended droughts characteristic of the City's Mediterranean climate result in large areas of dry vegetation, particularly in late summer and fall, when Santa Ana winds blow in from the desert and dry out the vegetation. Potential wildfire risk zones within the Uptown CPU area are areas that have steep slopes, limited precipitation, and plenty of available vegetation fuel. Uptown contains undeveloped land in the form of canyons that are occupied by a variety of native and non-native plant communities. Due to the amount of natural, unmaintained open space, there exists a high risk for wildfires.

Current City regulations require that brush management zones be established adjacent to development to reduce the risk from wildland fires. The purpose of such a program is to reduce the risk of wildfire while minimizing visual, biological, and erosion impacts to natural areas. In all the areas requiring brush management, a combination of two brush management zones occurs. Zone 1 consists of paving or ornamental plantings, which would be located within the development pad of each residential lot. Zone 2 involves the selective thinning and pruning of native vegetation and is considered impact neutral.

2.3.13.5 Emergency Preparedness

The County of San Diego Office of Emergency Services (OES) coordinates the overall county response to disasters. OES is responsible for: notifying appropriate agencies when a disaster occurs; coordinating all responding agencies; ensuring that resources are available and mobilized; developing plans and procedures for response to and recovery from disasters; and developing and providing preparedness materials for the public.

OES staffs the Operational Area Emergency Operations Center (EOC), a central facility that provides regional coordinated emergency response, and also acts as staff to the Unified Disaster Council (UDC), its governing body. The UDC, established through a joint powers agreement among all 18 incorporated cities and the County of San Diego, provides for coordination of plans and programs countywide to ensure protection of life and property.

In 2010, the County and 18 local jurisdictions, including the City of San Diego, adopted the Multi-Jurisdictional Hazard Mitigation Plan (MHMP). The MHMP is a countywide plan that identifies risks and ways to minimize damage by natural and manmade disasters. The plan is a comprehensive document that serves many purposes, including creating a decision tool for management, promoting compliance with state and federal program requirements, enhancing local policies for hazard mitigation capability, and providing interjurisdictional coordination.

The City of San Diego's disaster prevention and response activities are conducted in accordance with U.S. Department of Homeland Security Office of Domestic Preparedness requirements and incorporate the functions of planning, training, exercising, and execution. The City's disaster preparedness efforts include oversight of the City's EOC, including being responsible for maintaining the EOC in a continued state of readiness, training City staff and outside agency representatives in their roles and responsibilities, and coordinating EOC operations when activated in response to an emergency or major event/incident.



Chapter 3.0 Project Description

3.1 Introduction

The project analyzed in this Draft Program Environmental Impact Report (PEIR) is the Uptown Community Plan Update (proposed CPU) as well as several discretionary actions listed in Table 3-1, Project Components. The proposed CPU and associated regulatory documents and actions form the "project" for this PEIR and are referred to throughout the PEIR as the project. The project description contained within this section provides the basis for the environmental analysis in this PEIR for the proposed CPU and the associated discretionary actions.

Table 3-1			
Project Components			
Certification of PEIR			
Adoption of the Uptown Community Plan			
Adoption of the General Plan Amendments to Amend Community Plan			
Adoption of the Uptown Impact Fee Study			
Repeal the Mid-City Communities Planned District Ordinance (PDO)			
Repeal the West Lewis Street PDO			
Rescind the Interim Height Ordinance			
Rezone PDO areas with Citywide zones			
Amend the boundaries of the Uptown Community Plan Implementation Overlay Zone (CPIOZ)			

The proposed CPU and associated regulatory documents are available for review at the City and at the following website:

Uptown CPU:

https://www.sandiego.gov/planning/community/profiles/uptown

The existing Uptown Community Plan was last updated in 1988. The proposed CPU, and associated actions, will ensure consistency of the CPU with and incorporate relevant policies from the City of San Diego General Plan (General Plan), as well as provide a long-range, comprehensive policy framework and vision for growth and development in the Uptown community through 2035.

The Uptown CPU provides a long-range guide for the future physical development of the community. The CPU process started in 2009 with a public outreach effort centered around community meetings that included Uptown's stakeholder committees and neighborhood associations, workshops on key topics, a multi-day charrette, and meetings of the Uptown Planning Committees and the City's recognized community planning groups.

The Uptown CPU was updated in parallel with the North Park and Golden Hill CPUs in order to address key issues and propose solutions as they relate to attributes shared by each of the communities, including those relating to urban design, historic preservation, open space, and mobility. The City worked with community members to identify locations that would support compact, pedestrian-friendly, mixed-use village areas linked by transit and also developed community-specific policies that support infill development. Shared planning solutions were developed with refinements appropriate to individual community and neighborhood contexts. Background information regarding development of the proposed CPU, including project changes and community outreach, is described in Chapter 4.0, History of Project Changes Related to CEQA.

3.2 Relationship to General Plan

The General Plan, adopted in 2008, did not change the Community Plan land use designations or zoning on individual properties, but rather provided policy direction for future CPUs, discretionary project review, and implementation programs. The General Plan provided the Citywide vision and comprehensive policy framework for how the City should grow and develop, provide public services, and maintain the qualities that define the City as a whole.

The proposed Uptown CPU would build upon the goals and strategies in the General Plan. The proposed CPU is intended to further express General Plan policies through the provision of site-specific recommendations and policies that implement Citywide goals and policies at the Community Plan-level, address community needs, and guide zoning. The General Plan and Community Plan work together to establish the policy framework for growth and development in the CPU area. The Land Development Code within the Municipal Code implements the Community Plan policies and recommendations through zoning and development regulations. Specific General Plan policies are referenced within the proposed CPU to emphasize its relevance and applicability in the Uptown community. This PEIR provides analysis and evaluation of all relevant land use and environmental issues associated with the project.

3.3 **Project Objectives**

In accordance with California Environmental Quality Act (CEQA) Guidelines Section 15124, the following objectives were identified to outline the underlying purpose for the project. These objectives will be used to assist the lead agency in developing a reasonable range of alternatives to be evaluated in this PEIR and ultimately aid decision-makers in preparing findings and overriding considerations, if necessary. The primary objectives for the project are:

- Develop a multi-modal transportation network emphasizing active transportation measures for walkable and bicycle-friendly streets, and transit-related measures supporting transit operations and access.
- Maintain or increase the housing supply through the designation of higher residential densities focusing along major transit corridors.
- Provide for increased economic diversification through land use to increase employment and economic growth opportunities.
- Preserve the neighborhood character and design relationships between neighborhoods within each community through the development of transitions and design policies.
- Identify significant historic and cultural resources within the community and provide for their preservation, protection, and enhancement.
- Provide increased recreation opportunities and new public open spaces.
- Preserve, protect, and enhance the community's natural landforms, including canyons and environmentally sensitive lands.
- Include financing strategies that can secure infrastructure improvements concurrent with development.

3.4 **Project Description**

The project includes the comprehensive update to the Uptown Community Plan, which is intended to guide development through 2035 build-out of the Community Plan. For facility planning, technical evaluation, and environmental review purposes, build-out is assumed to occur in 2035. The Community Plan also addresses changes in conditions since 1988, when the Uptown Community Plan was last adopted. The proposed CPU provides detailed policy direction to implement the General Plan with respect to the distribution and arrangement of land uses (public and private); local street and transit network; prioritization and provision of public facilities, community, and site-specific urban design guidelines; and recommendations to preserve and enhance natural open space and historic and cultural resources within the Uptown community.

CPU implementation requires adoption of the Uptown Community Plan, amendments to the General Plan to incorporate the CPU as a component of the General Plan Land Use Element,

adoption of a Land Development Code (LDC) ordinance that would rezone the Planned District Ordinance (PDO) areas within the CPU area with Citywide zones within the LDC and repeal the existing Mid-City Communities PDO, the West Lewis Street PDO, and Interim Height Ordinance. The project would also amend the mapped boundaries of the Uptown Community Plan Implementation Overlay Zone (CPIOZ) to include CPIOZ-Type A and CPIOZ-Type B areas that would limit building heights. An Impact Fee Study (IFS) (formerly known as the Public Facilities Financing Plan) is also proposed for adoption resulting in a new IFS for the Uptown community. Each of these project elements is discussed further below.

While the proposed CPU sets forth procedures for implementation, it does not establish regulations or legislation nor does it, on its own, rezone property. Zoning and development regulations place controls on development and use of public and private property.

The proposed Uptown CPU includes an Introduction and Implementation chapter, and includes the following elements: Land Use; Mobility; Urban Design; Economic Prosperity; Public Facilities, Services and Safety; Recreation; Conservation; Noise; and Historic Preservation. Chapter 11 of the proposed Uptown CPU describes available financing methods for public improvement projects. Each element of the proposed CPU is described below.

3.4.1 Community Plan Elements

3.4.1.1 Land Use Element

The Land Use Element establishes the land use framework for the Uptown community and defines the distribution of proposed land uses on a map. The land use framework for the proposed CPU is depicted on the proposed Community Plan land use maps (Figures 3-1 through 3-4). The maps designate the proposed general location, distribution, and extent of land uses. The land use classifications are meant to be broad enough to give the City flexibility in implementation but clear enough to provide sufficient direction to carry out the goals of the proposed CPU. The figures are to be used and interpreted only in conjunction with the text and other figures contained in the proposed CPU.

The land use plans locate the highest intensity land uses within each community along transit corridors where existing and future commercial, residential, and mixed-use development can support existing and planned transit investments. Residential density is proposed to be increased from the adopted Community Plan in some areas and reduced in some areas to help achieve these objectives. The CPU results in an overall communitywide reduction of future housing units at Community Plan build-out when compared to the adopted Community Plan.

Community Plan land use designations that would be applied within the CPU area are described below. Future development within each land use designation would be subject to the CPU policies applicable to each designation. Table 3-2 provides a summary of land use classifications within the CPU area and permitted densities/intensities.



FIGURE 3-1 Proposed Land Use - Uptown



FIGURE 3-2 Proposed Uptown Land Use – West

ATTACHMENT 7



FIGURE 3-3 Proposed Uptown Land Use – East

Map Source: SanGIS

Map Source: SanGIS

ATTACHMENT 7



 $FIGURE \ 3-4$ Proposed Uptown Land Use – South

Table 3-2			
	Land Use Classifications and Permitted Densities/Intens	sities	
		Residential	
		Density	
Community Plan Land Use	Description	(du/ac)	Development Intensity
Residential	·		· · · ·
Residential – Very High	Provides for multi-family housing within a very-high density range. Commercial uses are also allowed, but not required.	74-109	3.60 FAR
Residential – High	Provides for multi-family housing within a high density range. Commercial uses are also allowed, but not required.	45-73	2.70 FAR
Residential – Medium High	Provides for multi-family housing within a medium-high density range. Commercial uses are also allowed, but not required.	30 to 44	1.80 FAR
Residential – Medium	Provides for both single-family and multi-family housing within a medium density range.	16 to 29	1.35 FAR
Residential – Low Medium	Provides for both single-family and multi-family housing within a low- medium density range.	10 to 15	0.75 FAR
Residential – Low	Provides for both single-family and multi-family housing within a low- density range.	5 to 9	Varies; see Table 131-04 in Municipal Code
Commercial Employment, R	etail & Services		
Neighborhood Commercial	Provides for shopping areas with retail, service, civic, and office uses	0-15	1.0 FAR
	for a neighborhood. Housing is allowed as part of a mixed use project.		
Neighborhood Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-29	1.0 FAR
Neighborhood Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-44	1.0 FAR
Community Commercial	Provides for shopping areas with retail, service, civic, and office uses	0-29	2.0 FAR
Community Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-73	2.0 FAR
Community Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-109	2.0 FAR
Office Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-29	0.75 FAR
Office Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-44	2.0 FAR
Office Commercial	Provides for shopping areas with retail, service, civic, and office uses for the community. Housing is allowed as part of a mixed use project.	0-73	2.0 FAR

Table 3-2 Land Use Classifications and Permitted Densities/Intensities			
		Residential	
		Density	
Community Plan Land Use	Description	(du/ac)	Development Intensity
Office Commercial	Provides for shopping areas with retail, service, civic, and office uses	0-109	2.0 FAR
	for the community. Housing is allowed as part of a mixed use project.		
Institutional and Public and Semi-Public Facilities			
Institutional	Provides a designation for uses that are identified as public or semi-	Not Applicable	Not Applicable
	public facilities in the Community Plan, including but not limited to		
	schools, libraries, and fire stations.		
Park, Open Space and Recreation			
Open Space	Applies to land or water areas generally free from development or	1	Limited
	developed with very low-intensity uses that respect natural		
	environmental characteristics.		
Population-based Parks	Provide for passive and/or active recreational uses, such as community	Not Applicable	Not Applicable
	parks and neighborhood parks.		
Notoci			

Notes:

du/ac = dwelling units per acre

1. FAR – Floor Area Ratio and represents total allowed FAR, as follows:

- For Neighborhood Commercial and Community Commercial: FAR includes only non-residential uses. Zones applied allow additional FAR for residential mixed-use.
- For Residential only uses: Projects would need to comply with both density and FAR standards.
- 2. New residential development is required to be within the density range (both maximum and minimum) specified in the applicable designation as shown in Table 2-3 of the proposed Uptown CPU. Residential density is applied to overall parcel area, excluding land that is not developable because of steep slopes or other natural constraints. Clustering is permitted in all residential designations to encourage open space conservation and preservation of natural topography; this may result in portions of a site developed at a density higher than the applicable density range, which is acceptable as long as the density for the overall development site is not exceeded.

a. Land Use Designations

The Community Plan land use designations are indicated on the proposed land use maps and are differentiated by color. The land use designations are described below.

Residential

Residential - Very High

Residential – Very High allows for multi-family housing in the highest density range (75 to 109 dwelling units per acre [du/ac]).

Residential - High

Residential – High allows for multi-family housing within a high density range (45 to 73 du/ac).

Residential - Medium High

Residential – Medium–High allows for multi-family housing within a medium-high density range (30 to 44 du/ac).

Residential - Medium

Residential – Medium allows for both single-family and multi-family housing within a medium density range (16 to 29 units du/ac).

Residential - Low Medium

Residential – Low–Medium provides for both single-family and multi-family housing within a low-medium density range (10 to 15 du/ac).

Residential - Low

Residential – Low is intended for areas with predominantly single-family residential development, with a low density range of 5 to 9 du/ac.

Commercial and Employment

Neighborhood Commercial - Residential Permitted

Neighborhood Commercial - Residential Permitted focuses on commercial uses and provides for shopping areas with retail, service, civic, and office uses for the community at large within 3 miles. Residential use between 0-15 du/ac, 0-29 du/acre, and 0-44 du/acre; office, public, and community gathering spaces are also allowed.

Community Commercial - Residential Permitted

Community Commercial - Residential Permitted focuses on commercial uses, and provides for shopping areas with retail, service, civic, and office uses for the community at large within 3 to 6 miles. Residential use between 0-29 du/ac, 0-44 du/ac, 0-73 du/acre, and 0-109 du/acre; office, public, and community gathering spaces are also allowed.

Institutional and Public/Semi-Public Facilities

Institutional

The Institutional designation provides for uses that are identified as public or semi-public facilities in the CPU, including but not limited to, hospitals, schools, libraries, and police and fire stations.

Park, Open Space, and Recreation

Open Space

Open Space applies to land generally free from development, developed with very low-intensity uses that respect natural environmental characteristics. Open Space lands consisting of canyons, slopes, and other natural land forms are located throughout the City. This Open Space is intended to preserve and protect native plants and animals, while providing public access and enjoyment by the use of hiking, biking, and equestrian trails.

Population-Based Parks

Population-based parks provide for passive and/or active recreational uses, such as community parks, neighborhood parks, and recreation centers to meet the recreational needs of the community as defined by the future Recreation Element. Population-based parks (commonly known as Neighborhood and Community parks), facilities, and services are intended to serve the daily needs of the neighborhood and community. When possible, they adjoin schools in order to share facilities, and ideally are within walking distance of the residences within their service area.

b. Neighborhood Centers and Villages

The proposed Uptown CPU identifies Neighborhood Centers and Villages where growth is focused into mixed-use activity centers that are pedestrian-friendly and linked to an improved regional transportation system. These areas would implement the General Plan's City of Villages Strategy and are envisioned to have an integrated mixture of uses, accessible and attractive streets, and public spaces. The proposed CPU identifies specific policies applicable to development in the area. Refer to Figure 3-5 for the location of proposed community villages and mixed-use corridors.



FIGURE 3-5 Uptown Village Locations and Mixed-Use Corridors

Hillcrest Core – West

Hillcrest Core – West is located west of the State Route 163 (SR-163) to Front Street provides a variety of commercial–retail businesses and eating and drinking establishments. Hillcrest Core – West is characterized by its narrower, grid-patterned streets and small business storefronts and has served as the traditional heart of the Hillcrest neighborhood.

Hillcrest Core – East

Hillcrest Core – East located east of SR-163 along Washington Street and University Avenue to Park Boulevard. Hillcrest Core – East has grown in importance through the years as the second Community village core in Uptown. This Community Village core contains the Uptown District – a large pedestrian-oriented, mixed-use retail center and. This Community Village core is characterized by University Avenue as the main transit corridor marked by both auto-oriented strip commercial, as well as pedestrian-oriented businesses.

Mission Hills

The Neighborhood Village in Mission Hills focused around Goldfinch Street and Washington Street provides a variety of convenience goods and services to adjacent single-family and multi-family residential neighborhoods that surround the Neighborhood Village core.

Bankers Hill/Park West

The Neighborhood Village in Bankers Hill/Park West is characterized by offices; commercial and residential development, quality shopping, eating and drinking establishments; high-rise mixed-use developments; and its proximity to Balboa Park.

Middletown

Within Middletown, the Neighborhood Village is prominently known for its various restaurants located along India Street also known as International Restaurant Row.

3.4.1.2 Mobility Element

The proposed Uptown CPU Mobility Element provides direction on how to achieve mobility goals through a balanced, multi-modal transportation network in the Community Plan area. This element is closely linked to the Land Use and Urban Design Elements. The Mobility Element describes existing and future conditions related to streets; vehicles and parking; as well as bicycles, pedestrians, and public transit, including recommended mobility improvements to achieve adequate capacity and improved access.

The proposed CPU identifies specific policies applicable to pedestrians, bicycling, and transit and identifies priority routes for each mode. Policies applicable to the street system are provided in addition to roadway classifications. Street system policies focus on providing a complete street network throughout the communities to accommodate all modes. Future roadway classifications proposed for the Uptown CPU area is shown in Figure 3-6.

Map Source: SanGIS

ATTACHMENT 7



FIGURE 3.6 Future Roadway Classifications The proposed Uptown CPU includes policies related to Intelligent Transportation Systems (ITS), such as coordinated traffic signals and use of Transportation Demand Management (TDM) to reduce single-occupancy vehicle trips. The proposed CPU also includes policies related to parking that address issues such as the design and placement of parking areas and compatibility with bicyclists and motorcycles. The Mobility Element is contained within Chapter 3 of the proposed Uptown CPU.

3.4.1.3 Urban Design Element

The proposed Uptown CPU Urban Design Element describes community character and provides goals and policies related to urban form, including public spaces and mixed-use design, neighborhood and community gateways and linkages, building types and massing, streetscape and pedestrian orientation, urban forestry, and other unique aspects of the Uptown community. This element presents the proposed urban form of the Community Plan area and highlights opportunities for urban design in the community. The Urban Design Element is contained within Chapter 4 of the proposed Uptown CPU.

3.4.1.4 Economic Prosperity Element

The proposed Uptown CPU Economic Prosperity Element links economic prosperity goals with land use distribution and employment land use policies, including specific policies aimed at supporting existing and new businesses to preserve and create job opportunities for residents, primarily through new commercial and office development where appropriate. This element seeks to enhance economic opportunity in the Community Plan area, building on significant growth opportunities along the CPU's main commercial corridors. The Economic Prosperity Element is contained within Chapter 5 of the proposed Uptown CPU.

3.4.1.5 Public Facilities, Services, and Safety Element

The proposed Uptown CPU Public Facilities, Service, and Safety Element identifies public facilities and services intended to serve existing and future residents, including educational facilities, public safety services, and infrastructure systems. This element provides policies regarding police and fire services, schools and public libraries, public utilities, geological and seismic hazards, flooding hazards, fire hazards, and hazardous materials. The Public Facilities, Services, and Safety Element is contained within Chapter 6 of the proposed Uptown CPU.

3.4.1.6 Recreation Element

The proposed Uptown CPU Recreation Element provides goals and policies and identifies opportunities to create a more comprehensive park strategy. The proposed CPU calls for the acquisition and development of new parks and associated facilities, improving existing parks in order to expand active and passive recreational use, and provide access to trails and open spaces. This element identifies existing parks, proposed parks, and park equivalencies to provide additional recreation opportunities. Proposed park sites may be acquired and/or developed as park land by the City. Where undeveloped land is limited, unavailable or cost-prohibitive, the General Plan allows for the application of park equivalencies to be determined by the community and City staff. Park

equivalencies include joint use facilities, trails, privately owned publicly accessible parks, nontraditional parks (such as rooftop or indoor recreation facilities), portions of resource-based parks, and park facility expansions or upgrades. The Uptown community is an urbanized community where park equivalencies are appropriate for satisfying some of the community's population-based park needs. The Recreation Element is contained within Chapter 7 of the proposed Uptown CPU.

3.4.1.7 Conservation Element

The proposed Uptown CPU Conservation Element provides goals and policies to effectively manage, preserve, and enhance natural resources in the community. The element addresses open space and landform preservation policies, urban runoff management, water resource management, air quality, and waste diversion. This element supports sustainability through policies and land use guidance that provide for economic resiliency, resource conservation, renewable energy, and enhancement of habitat and the urban forest. Strategies included in the Conservation Element address development and use of sustainability and energy generation types, including reuse or recycling of building material, adaptively retrofitting and reusing existing buildings, constructing energy-efficient buildings with healthy and energy-efficient interior environments, creating quality outdoor living spaces, improving materials recycling programs, and promoting local initiatives for local sources and environmentally sustainable goods and services.

Development in the Community Plan area will generally occur as infill projects, focusing on vacant or under-utilized parcels or previously utilized lots rather than on undeveloped land with high natural resource values. The proposed Conservation Element is contained within Chapter 8 of proposed Uptown CPU.

3.4.1.8 Noise Element

The proposed Uptown CPU Noise Element provides goals and policies addressing noise compatibility, including commercial, traffic, and airport noise and identifies future noise contours from freeways and major roads in the community. The Noise Element is contained within Chapter 9 of the proposed Uptown CPU.

3.4.1.9 Historic Preservation Element

The proposed Uptown CPU Historic Preservation Element describes the archaeological and historic context and history of the built environment in Uptown. The Historic Preservation Element focuses on the protection of the community's historical and cultural resources, and supports educational opportunities and incentives to highlight, maintain, and preserve historic resources. This element provides a framework for evaluating individual historic properties and districts for the National Register of Historic Places, California Register of Historic Places, and the San Diego Register of Historic Resources. Specific policies for the Community's historic resources.

The proposed Historic Preservation Element identifies Potential Historic District Boundaries within the Uptown community that are intended to provide protection measures to prevent the loss of the overall integrity of Potential Historic Districts. Additional detail about implementation of Potential Historic Districts is discussed in Section 3.4.2.23 below. The Historic Preservation Element is contained within Chapter 10 of the proposed Uptown CPU.

3.4.1.10 Implementation

The proposed CPU includes an Implementation chapter that describes future actions that would implement the Community Plan. Future implementation actions are described below and detailed in Chapter 11 of the proposed Uptown CPU.

- Approve and regularly update an IFS identifying the capital improvements and other projects necessary to accommodate present and future community needs as identified throughout the CPU.
- Fund and construct facilities and other public improvements in accordance with the IFS.
- Pursue additional funding sources, such as grant funding, to implement unfunded needs identified in the IFS.
- Apply and implement the Community Plan's urban design policies and recommendations during review of developments projects including administration of the Uptown CPIOZ.
- Seek longer-term implementation strategies that could be considered toward meeting the Community Plan's identified improvement projects.

3.4.2 Land Development Code Amendments

3.4.2.1 Repeal of Planned District Ordinances

The project would repeal the existing Mid-City Communities PDO, the West Lewis Street PDO, the Interim Height Ordinance, and rezone those PDO parcels with existing Citywide zones to implement the proposed land use plan designations.

3.4.2.2 Amendment to the Uptown Community Plan Implementation Overlay Zone Boundaries

The CPIOZ is applied within the boundaries of the Uptown Community Plan per Chapter 12, Article 2, Division 14 of the Municipal Code to regulate specific building heights primarily along the transit corridors within the neighborhoods of Hillcrest, Mission Hills, and Bankers Hill/Park West. According to Chapter 13, Article 2, Division 14 of the Municipal Code, the purpose of the CPIOZ is to supplement the Municipal Code by providing development regulations that are tailored to specific circumstances and/or sites within the community. The mapped boundaries of the existing CPIOZ would be amended within the Uptown community to replace CPIOZ-Type A, related to retail parking requirements for the Thackery Gallery structure in Hillcrest, and CPIOZ-Type B, related to discretionary review of office uses in the Medical Complex neighborhood with new boundaries to address ministerial review of building height limits within Hillcrest and Mission Hills (CPIOZ-Type A)

and discretionary review of building height limits within Hillcrest and Bankers Hill/Park West (CPIOZ-Type-B). CPIOZ-Type A identifies areas within the community where ministerial approval is granted for development that does not exceed 50 feet within Mission Hills and 65 feet in Hillcrest and Bankers Hill/Park West. CPIOZ-Type B identifies areas within the community where discretionary approval is granted through a Process 3 Site Development Permit for development that does not exceed 150 feet in Bankers Hill/Park West, 120 feet in central Hillcrest, and 100 feet in Hillcrest east of the SR-163. Maps depicting areas where CPIOZ-Type A and CPIOZ-Type B would be applied to address building heights are in the proposed Uptown CPU Urban Design Element and are shown in Figures 3-7 and 3-8.

3.4.2.3 Historical Resources Regulations

The project analysis addresses the amendments to the Historical Resources Regulations (Municipal Code Sections 143.0210 et seq.) as they relate to implementation of the proposed Uptown CPU. These amendments have been proposed and analyzed as part of the PEIR for the North Park and Golden Hill CPUs and would provide supplemental development regulations for Potential Historic Districts as adopted by the City Council (see Figure 6.7-4 of this PEIR). These regulations would provide protections to the integrity of the Potential Historic Districts by requiring an evaluation of proposed modifications to single- and multi-family residential structures identified as contributing resources within the boundaries of the proposed Potential Historic Districts.

Applicable residential structures would be subject to the following requirements:

- No modifications allowed to the front 2/3 of the original building footprint unless the modification will repair existing historic materials or restore the building to its historic appearance.
 - Exception: Improvements exempt from building permits pursuant to SDMC 129.0203, as well as improvements identified in SDMC 143.0212(a)(1)-(4) (same standard as applied to 45-year review).
 - Exception: Deviation may be approved though a Process 2 Neighborhood Development Permit. Projects will be reviewed for consistency with the US Secretary of the Interior Standards (similar to 45-year review) and the following findings must be made.
 - ✓ All feasible measures to protect and preserve the integrity of the potential historic district have been provided; and,
 - ✓ The proposed deviation is the minimum necessary to afford relief and accommodate the development and all feasible measures to mitigate for any impacts to the potential historic district have been provided; and,
 - ✓ The proposed project will not result in a loss of integrity within the potential historic district which would render it ineligible for historic designation.

<u>Projects subject to the supplemental development regulations would be allowed to deviate from the</u> <u>regulations with approval of a Neighborhood Development Permit (NDP). Amendments to the NDP</u> <u>regulations are included and analyzed as part of the PEIR for the North Park and Golden Hill CPUs.</u> These amendments would add the requirement that a NDP is required for development impacting single dwelling unit and multiple dwelling unit structures on a parcel containing a potential contributing resource within the City Council specified Potential Historic District. The NDP revisions would add supplemental findings applicable to these projects.

3.4.3 Zone Changes

3.4.3.1 Citywide Rezoning

The implementation program for the Community Plan would replace the Mid-City Communities PDO and West Lewis Street PDO with Citywide zones and development regulations. Conversion from Planned District zoning to Citywide zoning is summarized in Table 3.3 below. Certain commercial PDO zones listed in the table show multiple compatibility with Citywide zones as a result of varying densities allowed based on lot size. The implementation program for the Community Plan also includes residential zones identified in Table 3₋₋3 that would be converted to an open space-residential zone to preserve privately owned property that is designated in the Community Plan as open space with limited development. The proposed zoning for the Uptown CPU area is depicted on Figure 3-9.

Table 3-3 Conversion to Citywide Zoning			
Mid-City Communities Planned District	Citywide Zone		
MR-3000	RM-1-1		
MR-1500	RM-2-5		
MR-1000	RM-3-7		
MR-800B	RM-3-9		
MR-400	RM-4-10		
CV-4, CL-5, CL-6	CN-1-3		
CN-3, CN-4, CN-2A, CV-3, CL-2	CN-1-4		
NP-3	CC-1-3		
NP-3 <u>, CN-1</u>	CC-3-4		
CN-1	CC-3-5		
NP-2 <u>, CN-2A</u>	CC-3-6		
CN-1, CN-1A, CL-2, NP-1	CC-3-8		
CN-1, CN-1A, CV-1	CC-3-9		
West Lewis Street Planned District	Citywide Zone		
WLSPD	CN-1-1		
Residential Zones	Citywide Zone		
RS-1-1, RS-1-2, RS-1-4, RS-1-5	OR-1-1		



FIGURE 3-7 Uptown Proposed CPIOZ-Type A



ATTACHMENT 7



FIGURE 3-8 Uptown Proposed CPIOZ- Type B



FIGURE 3-9 Uptown Proposed Zoning

3.4.3.2 Applicable Citywide Zones

The following zones would apply within the Uptown CPU area and are described below.

a. OR Zone

The purpose of the OR zone is to preserve privately owned property that is designated as Open Space in a land use plan for such purposes as preservation of public health and safety, visual quality, sensitive biological resources, steep hillsides, and control of urban form, while retaining private development potential. The following OR zone, described in Table 3-4, would be applied in the CPU area:

• OR-1-1 is intended to allow open space with limited private residential development.

Table 3-4			
Proposed OR Zone within CPU Area			
	Maximum	Maximum	Maximum
Zone	Density	Height	FAR
OR-1-1	1 du/ac	30 feet	0.45

b. RM Zones

The purpose of the RM zones is to provide for multiple dwelling unit development at varying densities and provide for residential development that is compatible with the pattern of existing neighborhoods. The following RM zones, described in Table 3-5, would be applied in the CPU area:

- RM-1-1 is intended to allow a mix of Low to Medium residential density (up to 15 dwelling units per acre).
- RM-2-5 is intended to allow Medium residential density (up to 29 dwelling units per acre).
- RM-3-7 is intended to allow a mix of Medium residential density (up to 44 dwelling units per acre) with limited ground floor neighborhood serving commercial uses with a pedestrian orientation.
- RM-3-9 is intended to allow a mix of High residential density (up to 73 dwelling units per acre) with limited ground floor neighborhood serving commercial uses with a pedestrian orientation.
- RM-4-10 is intended to allow High density multiple dwelling units with limited commercial uses, with a maximum density of up to 109 dwelling units per acre.

Table 3-5 Proposed RM Zones within CPU Area			
	Maximum	Maximum	Maximum
Zone	Density	Height	FAR
RM-1-1	15 du/ac	30 feet	.75
RM-2-5	29 du/ac	40 feet	1.35
RM-3-7	44 du/ac	40 feet	1.80
RM-3-9	73 du/ac	60 feet	2.70
RM-4-10	109 du/ac	50 feet	3.60

c. CN Zones

The purpose of the CN zones is to provide residential areas with access to a limited number of convenient retail and personal service uses. The CN zones are intended to provide areas for smaller scale, lower intensity developments that are consistent with the character of the surrounding residential areas. The zones in this category may include residential development. Property within the CN zones will be primarily located along local and collector streets. The following CN zones, described in Table 3-6, would be applied in the CPU areas:

- CN-1-1 is intended to allow development of a limited size with up to 15 dwelling units per acre as part of a pedestrian-oriented mixed-use development.
- CN-1-3 is intended to allow for neighborhood commercial with up to 29 dwelling units per acre as part of a pedestrian-oriented mixed-use development.
- CN-1-4 is intended to allow neighborhood commercial development with up to 44 dwelling units per acre as part of pedestrian-oriented mixed-use development.

Table 3-6 Proposed CN Zones within CPU Area			
		Maximum	Maximum
Zone	Maximum Density	Height	FAR
CN-1-1	15 du/ac	30 feet	1.0
CN-1-3	29 du/ac	30 feet	1.0
CN-1-4	44 du/ac	65 feet	1.0

d. CC Zones

The purpose of the CC zones is to accommodate community-serving pedestrian-oriented commercial services, retail uses in a mixed-use setting. The CC zones are intended to provide for a range of development patterns from pedestrian-friendly commercial streets to shopping centers. All of the CC zones in the Uptown community include residential development. Property within the CC zones will be primarily located along collector streets, major streets, and public transportation lines. The following CC zones, described in Table 3-7, would be applied in the CPU area:

- CC-1-3 is intended to accommodate development with an auto orientation and a Medium density.
- CC-3-4 is intended to accommodate development with a pedestrian orientation and a Medium density.
- CC-3-5 is intended to accommodate development with a high intensity, pedestrian orientation and Medium density.
- CC-3-6 is intended to accommodate development with a high intensity, pedestrian orientation and Medium High density.
- CC-3-8 is intended to accommodate development with a high intensity, pedestrian orientation and High density.
- CC-3-9 is intended to accommodate development with a high intensity, pedestrian orientation and Very High density.

Table 3-7 Proposed CC Zones within CPU Area			
	Maximum	Maximum	Maximum
Zone	Density	Height	FAR
CC-1-3	29 du/ac	45 feet	0.75
CC-3-4	29 du/ac	30 feet	1.0
CC-3-5	29 du/ac	100 feet	2.0
CC-3-6	44 du/ac	65 feet	2.0
CC-3-8	73 du/ac	100 feet	2.5 2.0
CC-3-9	109 du/ac	unlimited	<u>3.02.0</u>

3.4.4 Impact Fee Study

The project includes adoption of an IFS (formerly known as a Public Facilities Financing Plan [PFFP]) that addresses the need for public facilities associated with the identified needs of the Uptown CPU area. City Council adopted the current PFFP in 2003. The existing PFFP sets forth the major public facilities needs in the areas of transportation (streets, sidewalks, storm drains, traffic signals, etc.), libraries, park and recreation facilities, and fire stations that are needed to serve the community. The proposed IFS for Uptown would be used to determine the public facilities needs associated with the proposed CPU area. It includes potential funding sources for public facility financing, particularly development impact fees

The IFS identifies and prioritizes improvements to public facilities. Improvements vary widely in their range and scope; some could be implemented incrementally as scheduled street maintenance occurs, and others would require significant capital funding from city, state, regional, and federal agencies, or are not feasible until significant new development occurs. A complete list of projects is included in the proposed IFS.

3.4.5 MHPA Boundary Line Corrections

The project includes a comprehensive communitywide Multi-Habitat Planning Area (MHPA) boundary line correction within the Uptown CPU area. A comprehensive, systematic approach was developed in order to evaluate areas of existing developed land that should be removed, as well as areas where biological resources should be added. The boundary line corrections generally removed existing developed areas in addition to the 35-foot brush management zone 1 area as required in accordance with the City's Land Development Code, Section 142.0412. The comprehensive MHPA boundary corrections for the Uptown CPU area would result in removal of acreage of existing developed lands from the MHPA and an addition of sensitive habitats including coastal sage scrub and chaparral. For specific acreage of vegetation communities/land cover proposed for addition and removal from the MHPA, refer to Chapter 6.8.

3.5 Environmental Design Considerations

Several environmental design considerations, beyond compliance with mandatory existing regulations, have been incorporated into the proposed CPU as recommendations within policies to avoid or reduce environmental impacts. These are described below.

3.5.1 Sustainability

Sustainable building concepts and practices have been incorporated into the proposed policies within various elements of the proposed CPU. Implementation of these policies will serve to reduce or avoid potential environmental effects associated with water and energy consumption, consumption of non-renewable or slowly renewing resources, and urban runoff.

3.5.2 Village Districts and Transit Corridors

Development completed in accordance with the proposed CPU would occur in an existing urbanized area with established public transportation infrastructure, including existing and future transit service. Most future development is expected to occur within proximity of areas served by transit, which may reduce vehicle trips and miles traveled. In addition, implementation of the policies contained in the Land Use, Mobility, Recreation, and Conservation elements of the proposed CPU would improve mobility within the Community Plan area, by promoting development of a balanced, multi-modal transportation network, including better pedestrian and bicycle mobility. Policies that support walking and bicycling as transportation choices could also reduce vehicle trips and vehicle miles traveled.

3.5.3 Transit

While the intent of the proposed Uptown CPU Mobility Element is to provide a more cohesive transportation network, it contains policies that specifically address transit services and facilities, including improving the environment surrounding transit stops, and working with the San Diego Metropolitan Transit System (MTS) to incorporate transit priority measures.

3.5.4 Recreation

The proposed Uptown CPU Recreation and Conservation elements contain policies aimed at creating a sustainable park system that meets the needs of Uptown residents and visitors by increasing the quantity and quality of recreation facilities within the CPU area.

3.5.5 Urban Runoff/Water Quality

Urban runoff management policies located in the proposed Uptown CPU Conservation Element seek to reduce potential runoff/water quality impacts by encouraging the use of Low Impact Development (LID) techniques and materials that slow water runoff and absorb pollutants from roofs, parking areas, and other urban surfaces; incorporating bio-swales or other design practices where there are sufficient public rights-of-way throughout the community; and encouraging private property owners to design or retrofit landscaped areas to better capture storm water runoff.

3.5.6 Diversity and Affordability of Housing

The land use plan for the proposed CPU proposes a range of single- and multi-family housing densities intended to provide a range of housing types, including moderate and high densities that typically could allow a mix of market rate and affordable multi-family units. This could enable a wider range of economic levels and age groups to live within these communities. The Uptown Land Use Element contains policies related providing land use types to accommodate both affordable and market rate housing; providing a diverse mix of housing type; enabling rental and ownership opportunities in all types of housing including alternative housing units such as companion units, live/work studios, and shopkeeper units; and developing adequate housing for those with special needs such as the elderly, disabled persons, low income, and those who need nursing care.

3.5.7 Bicycle Network

In order to reduce reliance on fossil fuels and encourage alternative modes of transportation, the proposed Uptown CPU aims to provide a safe and convenient bicycle network that connects community destinations and links to surrounding communities and the regional bicycle network. In support of this goal, the Uptown Mobility Element includes bicycle policies in Section 3.2 of the Mobility Element. Specifically, implementation of Uptown Mobility Element Policies in this section would support and implement bicycle priority streets and facilities that would connect Uptown to neighboring communities with emphasis on constructing missing bikeways in the bicycle network, implementing and building upon the San Diego Bicycle Master Plan. The Mobility Element Policy MO-2.3 calls for increasing bicycle comfort and accessibility for all levels of bicycle rides with improvements such as signage, marking, and wayfinding for bicycles, directing them to points of interest within Uptown and adjacent communities, actuated by signal timing for bicycles, priority parking for bicycles, wider bike lanes, and separated bicycle facilities, where feasible.

3.5.8 Access to Outdoor and Active Spaces

The proposed CPU addresses existing and planned access to outdoor and active spaces, and provides for additional outdoor recreation opportunities, including land acquisition for creation of public parks within the community. On-site open space within new multi-family development is also recommended. Access is to be improved per policies for better pedestrian and bicycle access to open space within canyons as well as Balboa Park.

Strategies to expand programming within existing public spaces to reduce the existing parkland deficit in the CPU area are also included in the proposed CPU. The Recreation Element includes policies to provide parkland to meet the needs of the community through CPU build-out; provide for preservation, protection, and enhancement of existing and planned parkland facilities; ensure accessibility of parkland to all residents and visitors; and to preserve, protect, and enhance/restore resources associated with existing and proposed open space.

3.5.9 Improved Transportation Network and Increased Alternative Modes of Transportation

The proposed CPU includes several policies intended to improve the existing transportation network, as well as encouraging alternative modes of transportation to reduce impacts related to traffic/circulation and air quality. The proposed Mobility Element would support and help implement the General Plan at the Community Plan-level by including specific policies and recommendations that will improve mobility through the development of a balanced, multi-modal transportation network. Specifically, the Mobility Element includes policies addressing walkability that would promote and encourage new construction and upgrades to existing pedestrian pathways. Transit Policies of the proposed Mobility Element would support improving access to public transit facilities (i.e., San Diego Trolley); Intelligent Transportation System policies would promote smart parking technology; and Bicycle Policies would promote a continuous network of bicycle facilities connecting the CPU areas to the Citywide bicycle network and bicycle parking facilities. In support of General Plan Policies UD-D-1 through D-3, the proposed Land Use Element Sections 2.2 and 2.3 focus the highest intensity development (residential and non-residential) on the community's commercial-transit corridors and village areas to capitalize on access to transit, boost transit ridership, and reduce reliance on driving.

3.5.10 Energy Efficiency in Buildings

The Urban Design and Conservation elements of the CPU include policies to reduce air, water, and land pollution, and other environmental impacts associated from energy production and consumption. The Urban Design Element encourages new infill buildings and retrofitting of existing buildings to take into account energy-efficient design. In particular, the Uptown Urban Design Element provides policies to incorporate building features that reduce energy consumption, reduce solar heat gain, and restore and adaptively reuse older structures. Specifically, policies within the Urban Design Element address sustainable building design, access to light and air, and adaptive reuse. The Uptown Conservation Element policy CE-1.1, along with proposed Mobility Element policies, addresses energy efficiency and sustainable building design by encouraging new development to build upon the community's existing street grid network to create a more functional environment for pedestrians and bicyclists and reduce local dependence on automobile transportation; encourage the reduction in development of project-level greenhouse gas emissions to acceptable levels by incorporating sustainable building and development practices, applying site-specific mitigation measures, and adhering to standardized measures outlined in the City's Climate Action Plan; and as part of a comprehensive energy-reduction strategy, promote the continued use or adaptive reuse of existing buildings in conjunction with any needed upgrades to their energy-use efficiency. This would include preserving existing buildings with important architectural or historic character as valued community assets and preserving structures that meet the Historical Resources criteria for designation and adaptive reuse, if necessary, to maintain their economic viability.

3.5.11 Air Quality

The Conservation Element includes policies to reduce the CPU's impact on air quality and climate change. The Conservation Element includes Air Quality policy CE-3.4, which encourages alternative modes of transportation, street tree and private tree planting programs throughout the community to increase absorption of carbon dioxide and pollutants, and the relocation of incompatible uses that contribute to poor air quality. The implementation of Sustainability Development policies in section 4.4 Development Form of the Urban Design Element also aim to reduce project-level greenhouse gas emissions to acceptable levels through project design, application of site-specific mitigation measures, or adherence to standardized measures outlined in an adopted Citywide Climate Action Plan.

3.5.12 Urban Agriculture, Urban Forestry, and Sustainable Landscape Design

The proposed Conservation Element includes policies in Section 8.1 Sustainable Development that would support sustainable food practices, locally sourced goods and services, and would seek opportunities for community agricultural use of property including community gardens. Urban Forestry policies in Section 4.4 Development Form of the Urban Design Element of the proposed Uptown CPU discuss and encourage the implementation of programs for enhancing the urban forest and supporting urban forestry efforts by incorporating shade-producing street trees that are suited to the San Diego climate along all streets and roadways as well as maximizing tree shade canopy. Additionally, policies in the Urban Design Element advocate for the incorporation of sustainable landscape treatments like drought-tolerant and climate appropriate plant species. Proposed policies in the Conservation Element encourage the use of water-wise practices with new development and building retrofits such as recycled/gray water systems, low water use vegetation in public spaces, and incorporating water-efficient landscape design in community greening projects.
3.6 Plan Build-out

Future development realized under the proposed land use map is referred to as build-out. The proposed CPU does not specify or anticipate when build-out would occur, as long-range demographic and economic trends are difficult to predict. However, for facility planning, technical evaluation, and environmental review purposes, build-out is assumed to occur in 2035.

3.6.1 Uptown Land Use Distribution at Plan Build-out

The amount of area in each land use designation under the CPU is shown on Table 3-8. The predominant land use designation in Uptown would remain residential, with single-family residential <u>acreage</u> at 726 acres mirroring the current prevalence of single-family houses. Tand the multi-family designation would acreage compriseing 380 acres. <u>Overall, implementation of the proposed Uptown CPU and associated discretionary actions is anticipated to result in multi-family development at higher densities along transit corridors. Specifically, compared to the existing condition, the number of single-family units is anticipated to decrease by 2,020 units and the number of multi-family units is anticipated to increase by 11,560 units (refer to Table 3-9).</u>

Table 3-8 Proposed Land Use Classifications in Untown							
Community Plan Land Use	Acres	Percent					
Residential							
Residential - Single	726	27%					
Residential - Multi	380	14%					
Residential Total	1,106	41%					
Commercial and Office							
Visitor and Retail Commercial	181	8%					
Office Commercial	32	1%					
Commercial, Employment Total	213	9%					
Institutional and Educational Facilities							
Institutional	100	4%					
Education	29	1%					
Institutional and Education Total	129	5%					
Open Space and Parks							
Open Space	398	14%					
Population-based Parks	51	3%					
Parks and Open Space Total	449	17%					
Roads							
Roads	761	28%					
Agriculture							
Agriculture	0.5	0.0% ¹					
Total	2,658.5	100%					
SOURCE: City of San Diego 2016.							
¹ Existing community garden. Percentage accounts for least	ss than 1% of	f the total					
acreage in the community							

Table 3-9 describes the existing and proposed residential development anticipated from application of land uses shown on the proposed Uptown Land Use Map on vacant and underutilized parcels, according to analysis undertaken for the CPU. Table 3-10 shows the same for existing and proposed non-residential development.

Table 3-9 Residential Development: Existing and at Proposed CPU Build-out								
	Existing Development Proposed Plan Build-out Difference							
	Residential	Percent of	Residential	Percent of		Change		
Residential Development	Units	Total	Units	Total	Change	(%)		
Single-Family Units ¹	7,540	33 %	5,520	17%	(2,020)	(27) %		
Multi-Family Units ²	15,620	67 %	27,180	83%	11,560	74 %		
Total Housing Units	23,160	100%	32,700	100%	9,540	41 %		
Household Population	36,750		55,700		18,950	52 %		
Notes:								

¹Includes detached single-family, multiple-unit single-family.

²Includes residential units in mixed-use development.

SOURCE: City of San Diego 2016.

Table 3-10 Non-Residential Development: Existing and at Proposed CPU Build-out									
	Existing Devel	Existing Development Proposed Plan Build-out							
Non-Residential Development	Non-Residential Building (square feet)	Percent of Total	Non-Residential Building (square feet)	Percent of Total	Change	Change (%)			
Commercial/Retail	1,875,780	26%	3,186,500	43%	1,309,720	70 %			
Visitor Commercial	366,460	5%	174,000	2%	(192,460)	(53)%			
Office	2,308,390 32		1,598,700	21%	(709,690)	(31)%			
Industrial	19,710	1%	0	0%	(19,700)	(100)%			
Education	413,100	5%	364,200	5%	(48,900)	(12)%			
Institutional	2,214,450	30%	2,121,500	28%	(92,950)	(4)%			
Recreational	31,110	1%	31,100	1%	0	0%			
Total Non-Residential Development	7,229,000	100%	7,476,000	100%					
SOURCE: City of San Die	ego 2016.								

3.6.2 Future Actions Associated with Plan Build-out

Due to the nature of an amendment to a Community Plan and a lack of site-specific development proposals associated with the proposed CPU, site-specific environmental analyses of future development anticipated within the CPU area are not undertaken within this PEIR. However, the analysis anticipates that future development would occur within CPU area and would be subject to applicable development regulations and requirements of the CPU and this PEIR. Future development within the CPU would involve subsequent approval of public and private development proposals through both ministerial in accordance with the zoning and development regulations and discretionary reviews in accordance with the zoning and development regulations and proposed Uptown CPU policies. These subsequent activities may be public (i.e., road/streetscape improvements, parks, public facilities) or private projects, and are referred to as future development or future projects in the text of the PEIR. A non-inclusive list of discretionary actions that would occur as the CPU is implemented are shown on Table 3-11.

Table 3-11
Potential Future Discretionary Actions Associated with Plan Build-out
City of San Diego
Subdivision Maps
Discretionary Permits
Site Development Permits
Establishment of Public Facilities Financing Mechanisms
Conditional Use Permits
Neighborhood Development Permits
Neighborhood Use Permits
Planned Development Permits
Variances
Street Vacations, Release of Irrevocable Offers of Dedication, and Dedications
Water and sewer infrastructure and road improvements
State of California
Caltrans Encroachment Permits
Section 1602/1603 Streambed Alteration Agreements
Water Quality Certification Determinations for Compliance with Section 401
Department of Education approval of school sites
Federal Actions
U.S. Army Corps of Engineers Section 404 Permits
USFWS Section 7 or 10 (a)
Other Agencies
SDG&E/Public Utilities Commission approvals of power line relocations or undergrounding

4

Chapter 4.0 History of Project Changes Related to CEQA

4.1 NOP and Project Initiation

The City initiated the process of updating the Uptown, North Park and Golden Hill Community Plans in 2009, when the planning team began its analysis of existing conditions. The Notice of Preparation (NOP) for the Program Environmental Impact Report (PEIR) was issued on December 23, 2013 (State Clearinghouse No. 2013121076, a new State Clearinghouse No. to be assigned for the Uptown Community Plan Update [CPU] PEIR at the start of public review). A public scoping meeting was held on January 9, 2014, to gather agency and public input on the scope and content of the PEIR. Written comments were also received during the 30-day public comment period and are included as Appendix A of this PEIR. Potentially significant concerns and issue areas were defined based on the initial analysis of environmental setting and baseline conditions, and comments on the NOP, and are analyzed as part of this PEIR.

4.2 Community Outreach and Plan Development

Between 2009 and 2016, an extensive outreach program was undertaken to solicit input from residents, business owners, community leaders, public officials, and other interested parties. The outreach program included multiple Community Plan Update Advisory Committee meetings on various land use topics, historic resources and mobility open house events, and a cluster workshop involving participants from each of the three communities to discuss urban design. Multi-day workshops or "charrettes" focusing on land use, areas of change and stability, urban design, mobility, historic resources, and recreation were conducted for each of the Community Plan Update (CPU) areas culminating in an urban design framework that would set the foundation for developing land use policies and recommendations. Additionally, "Open Mic Night" events were hosted by the City in an effort for community members to consider various perspectives from stakeholder organizations such as those representing local business districts, neighborhood-level organizations, historic preservation societies, planning and architectural organizations, and hospitals, as well as

walkability, open space, and housing advocates. The policies and details of the CPUs were developed and shaped through this process.

4.3 Changes Based on Comments on the Draft Community Plans

Subsequent to the NOP in December 2013, the stakeholders in the Uptown CPU area continued to have comments and concerns regarding the recommended edits to the CPU, whereas the community groups for North Park and Golden Hill had largely completed their review of their individual CPUs and voted to proceed with key components of their respective CPU. To maintain the overall progress and not unnecessarily delay all CPUs, the City Planning Department made the decision to remove Uptown CPU from the North Park/Golden Hill PEIR, moving forward with a separate Uptown CPU PEIR. Chapter 2.0 (Environmental Setting) and Chapter 5.0 (Regulatory Framework) have retained some discussions related to the adjacent CPUs to reflect background information of these concurrent planning efforts. The North Park/Golden Hill CPUs PEIR was sent out for public review under separate cover on May 31, 2016. When the Uptown CPU PEIR is released for public review, a State Clearinghouse Number will be assigned to the project, separate from the North Park/Golden Hill CPUs PEIR.

5

Chapter 5.0 Regulatory Framework

The regulatory framework applicable to each subject area included within this Program Environmental Impact Report (PEIR) is included in this chapter.

5.1 Land Use

Included within Section 3.0, Project Description, of this PEIR are descriptions of the existing land use plans that currently apply to the proposed Uptown Community Plan Update (CPU) area. The following expands the discussion of applicable plans and development regulations, including the General Plan, pertinent San Diego Municipal Code (SDMC) regulations, the City Multiple Species Conservation Program (MSCP) Subarea Plan, and the Airport Land Use Compatibility Plan.

5.1.1 City of San Diego General Plan

A comprehensive update of the City's General Plan was adopted in 2008, incorporating the City of Villages strategy, which in turn was developed and adopted as part of the Strategic Framework Element in 2002. The Strategic Framework Element represented the City's new approach for shaping how the City will grow while attempting to preserve the character of its communities and its most treasured natural resources and amenities. It was developed to provide the overall structure to guide the General Plan update and future CPUs and amendments, as well as the implementation of an action plan. Table 5-1 summarizes the general land use categories that will be applied within the Uptown CPU.

Under the City of Villages strategy, the General Plan aims to direct new development projects away from natural undeveloped lands into already urbanized areas and/or areas where conditions allow the integration of housing, employment, civic, and transit uses. It is a development strategy that mirrors regional planning and smart growth principles intended to preserve remaining open space and natural habitat and focus development in areas with available public infrastructure.

ATTACHMENT 7

5.0 Regulatory Framework

Table 5-1							
		General Pla	an Land Use Categories				
General Plan Land Use	Recommended Community Plan Designation	Use Considerations	Description	General Plan Density Range (du/ac ¹)			
Open Space, and Recreation	Open Space	None	Provides for the preservation of land that has distinctive scenic, natural, or cultural features; that contributes to community character and form; or that contains environmentally sensitive resources. Applies to land or water areas that are undeveloped, generally free from development, or developed with very low- intensity uses that respect natural environmental characteristics and are compatible with the open space use. Open Space may have utility for: primarily passive park and recreational uses; conservation of land, water, and other natural resources; historic or scenic purposes; visual relief; or landform preservation.	N/A			
Parks,	Population-based Parks	None	Provides for areas designated for passive and/or active recreational uses, such as community parks and neighborhood parks. It will allow for facilities and services to meet the recreational needs for the community as defined by the Community Plan.	N/A			
	Residential – Low	None	Provides for both single-family and multi-family housing within a Low-density range.	5-9 du/ac			
	Residential – Low–Medium	None	Provides for both single-family and multi-family housing within a Low–Medium-density range.	10-14 du/ac			
ential ¹	Residential – Medium	None	Provides for both single-family and multi-family housing within a Medium-density range.	15-29 du/ac			
Reside	Residential – Medium–High	None	Provides for multi-family housing within a Medium–High-density range.	30-44 du/ac			
	Residential – High	None	Provides for multi-family housing within a High- density range.	45-74 du/ac			
	Residential – Very High	None	Provides for multi-family housing in the highest density range.	75+ du/ac			
etail, and	Neighborhood Commercial	Residential Permitted	Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Housing may be allowed only within a mixed-use setting.	0-44 du/ac			
ial Employment, R Services ^{1, 2,3}	Community Commercial	Residential Permitted	Provides for shopping areas with retail, service, civic, and office uses for the community at large within three to six miles. It can also be applied to Transit Corridors where multi-family residential uses could be added to enhance the viability of existing commercial uses.	0-74 du/ac			
Commerci	Office Commercial	Residential Permitted	Provides for office employment uses with limited, complementary retail uses. Residential uses may occur only as part of a mixed-use (commercial/residential) project.	0-44 du/ac			

5.0	Regu	latory	Framework
-----	------	--------	-----------

Table 5-1 General Plan Land Use Categories							
General Plan Land Use	Recommended Community Plan Designation	Use Considerations	Description	General Plan Density Range (du/ac ¹)			
Institutional and Public and Semi-Public Facilities ⁴	Institutional	None	Provides a designation for uses that are identified as public or semi-public facilities in the Community Plan and which offer public and semi-public services to the community. Uses may include but are not limited to: airports, military facilities, community colleges, university campuses, landfills, communication and utilities, transit centers, water sanitation plants, schools, libraries, police and fire facilities, cemeteries, post offices, hospitals, park-and-ride lots, government offices, and civic centers.	N/A			
du/ac = dwe	lling units per acre	further refined and en	asific in each Community Plan. Desidential densities	mal also bo			

¹Residential density ranges will be further refined and specific in each Community Plan. Residential densities mal also be narrowed within the density ranges established for the Commercial Employment, Retail, and Services General Plan land use category in this table. Community Plans may also establish density minimums where none are specified in the Commercial Employment, Retail, and Services General Plan land use category. Calculation of residential density is to be rounded to the nearest whole number if the calculation exceeds a whole number by 0.50 or more in most cases. In all other remaining instances, such as in the coastal areas, calculation of density is to be based on established policies and procedures. Whenever a plus (+) sign is identified next to a density number, the upper limit may be further specified in a Community Plan without causing the need for amending the General Plan, upon evaluation of impacts. For uses located within an airport influence area, the density ranges should be consistent with the Airport Land Use Compatibility Plan and Air Installation Compatible Use Zone study or steps should be taken to overrule the Airport Land Use Commission.

²Consult the Economic Prosperity Element for policies related to the commercial and industrial land use designations. ³Commercial land use designations may be combined to meet community objectives.

⁴Community Plans will further define the specific institutional uses allowed on a particular site.

The General Plan includes 10 elements that are intended to provide guidance for future development. These are listed here and discussed in more detail below: (1) Land Use and Community Planning Element; (2) Mobility Element; (3) Urban Design Element; (4) Economic Prosperity Element; (5) Public Facilities, Services, and Safety Element; (6) Recreation Element; (7) Conservation Element; (8) Noise Element; (9) Historic Preservation Element; and (10) Housing Element. The Housing Element, which must be updated every five years under state law, was last updated in 2014 and is provided under separate cover due to the need for more frequent updates. It is required to be consistent with the General Plan goals and City of Villages strategies.

5.1.1.1 Land Use and Community Planning Element

The Land Use and Community Planning Element provides overarching policies to integrate the City of Villages strategy and guides the provision of public facilities while accommodating planned growth. Policies within this element, in combination with other elements, also ensure consistency with zoning regulations (e.g., SDMC).

The Land Use and Community Planning Element of the City's General Plan is largely seen as the structure and framework for developing Community Plans. When appropriate, policies call for Community Plans to further identify appropriate land uses to meet the goals set by the General Plan and City of Villages strategy. The policies also indicate that mixed-use areas, villages, and

community-specific policies are developed with public input and involvement.

The Land Use and Community Planning Element contains five goals related to community planning. These goals are to provide:

- 1. Community plans that are clearly established as essential components of the General Plan to provide focus upon community-specific issues.
- 2. Community plans that are structurally consistent yet diverse in their presentation and refinement of Citywide policies to address specific community goals.
- 3. Community plans that maintain or increase planned density of residential land uses in appropriate locations.
- 4. Community plan updates that are accompanied by updated Impact Fee Study (IFS; formerly known as Public Facilities Financing Plan [PFFPs]).
- 5. Community plans that are kept consistent with the future vision of the General Plan through comprehensive updates or amendments.

Community Plans are important because they contain specific policies that protect community character. Future public and private projects will be evaluated for consistency with policies in the Community Plans.

Environmental Protection/Environmental Justice. The General Plan Land Use and Community Planning Element also provides direction regarding balanced communities, equitable development, and environmental justice. The U.S Environmental Protection Agency (EPA) defines Environmental Justice as fair treatment and meaningful involvement of all peoples, regardless of race, color, national origin, or income, with respect to development, implementation and enforcement of environmental laws, regulations, and policies. The City of Villages strategy and emphasis on transit system improvements, transit-oriented development, and the Citywide prioritization and provision of public facilities in underserved neighborhoods are consistent with environmental justice goals.

5.1.1.2 Urban Design Element

The Urban Design Element of the General Plan includes goals and policies specific to mixed-use villages and commercial areas. The element emphasizes the integration of compatible land uses. In addition, this element anticipates the creation of transit-focused, walkable village centers, the provision of high-quality public spaces and civic architecture, and the enhancement of the visual quality of office and industrial development.

5.1.1.3 Economic Prosperity Element

The Economic Prosperity Element contains policies that are intended to improve the economic prosperity. This is accomplished by ensuring that the economy grows in ways that strengthen San Diego industries, retail and create good jobs with self-sufficient wages, increase average income, and stimulate economic investment in the community.

5.1.1.4 Noise Element

The focus of the Noise Element is to minimize excessive noise effects and improve the quality of life of people working and living in the City. The Noise Element identifies goals and related policies with regard to noise and land use compatibility, motor vehicle traffic noise, and trolley and train noise that are relevant to the CPUs. While the Noise Element articulates the City's goals, the enforcement mechanism to control noise is the City's Noise Ordinance, which is discussed in Section 5.6.

5.1.2 Land Development Code Regulations

Chapters 11 to 15 of the SDMC are referred to as the Land Development Code (LDC), as they contain the City's planning, zoning, subdivision, and building regulations that regulate how land is to be developed within the City. The LDC contains Citywide base zones that specify permitted land use, density, floor area ratio (FAR), and other development requirements for given zoning classifications, as well as overlay zones and supplemental regulations that provide additional development requirements.

Development of the proposed CPU area is subject to the development regulations of the LDC. As part of the LDC, certain geographic areas of the City, known as Planned Districts, are governed by specific Planned District Ordinances (PDOs), as identified in Chapter 15 of the LDC. Planned district means any legally described geographic area, (1) which has historical significance or serves as an established neighborhood or community, or (2) which is at the time of adoption developing or substantially undeveloped and for which a program of phased growth is desirable, and (3) which has been designated a Planned District by the City Council. The District shall be wholly within the boundaries of a precise plan or coterminous with the boundaries of a Community Plan. PDOs provide the means to adopt plans for certain areas of the City that provide land use, capital improvements, and public facilities controls in lieu of conventional zoning to accomplish the following goals:

- 1. To preserve and enhance the cultural, aesthetic or economic value of neighborhoods having special importance due to their historical significance or because of their being part of older, established communities and neighborhoods; and
- 2. To systematically implement a comprehensive plan for the phased growth of developing and undeveloped areas of the City.

To implement the proposed CPU and included as part of the project analyzed within this PEIR, the City is proposing the deletion of existing zoning established by PDO and would apply Citywide zoning across the community.

5.1.2.1 General Development Regulations

Chapter 14 of the LDC includes the general development regulations, supplemental development regulations, building regulations, and electrical/plumbing/mechanical regulations that govern all aspects of project development. The grading, landscaping, parking, signage, fencing, and storage

requirements are all contained within the Chapter 14, General Regulations. Also included within the general regulations of Chapter 14 are the Environmentally Sensitive Land (ESL) Regulations, discussed below.

5.1.2.2 Environmentally Sensitive Lands Regulations

According to Section 143.0110 of the LDC, ESL Regulations apply to areas with any of the following: sensitive biological resources, steep hillsides, coastal beaches, sensitive coastal bluffs, and special Flood Hazard Areas. Development on a site containing environmentally sensitive lands requires a Site Development Permit in accordance with Section 125.0502 of the LDC. Future development on environmentally sensitive lands within the Uptown CPU area would be subject to the ESL Regulations where steep hillsides and sensitive biological resources occur.

5.1.2.3 Historical Resources Regulations

The purpose of the City's Historical Resources Regulations, found in Section 143.0251 of the LDC is to protect, preserve, and, where damaged, restore the historical resources of San Diego, which include historical buildings, historical structures or objects, important archaeological sites, historical districts, historical landscapes, and traditional cultural properties. These regulations are intended to assure that development occurs in a manner that protects the overall quality of historical resources. The Historic Resources Regulations require that development affecting designated historical resources or historical districts shall provide full mitigation for the impact to the resource, in accordance with the Historical Resources Guidelines of the Land Development Manual (LDM), as a condition of approval. If development cannot, to the maximum extent feasible, comply with the development regulations for historical resources, then a project would require a permit.

5.1.3 Multiple Species Conservation Program

The Multiple Species Conservation Program is discussed below in Section 5.8.

5.1.4 Airport Land Use Compatibility Plan

The San Diego County Regional Airport Authority, serving as the Airport Land Use Commission, is required by state law to prepare an Airport Land Use Compatibility Plan (ALUCP) for the San Diego International Airport (SDIA). The Uptown CPU is within the Airport Influence Area (AIA) for SDIA. The AIA serves as the boundary for the ALUCP. The AIA is divided into to two review areas. Review Area 1 is defined by the combination of the 60-decibel (dB) Community Noise Equivalent Level (CNEL) noise contour, the outer boundary of all safety zones, and the airspace Threshold Siting Surfaces. All policies and standards in the ALUCP apply within Review Area 1. Review Area 2 is defined by the combination of the airspace protection and overflight boundaries beyond Review Area 1. Only airspace protection and overflight policies and standards apply within Review Area 2.

The ALUCP contains policies and criteria that address land use compatibilities concerning noise and safety aspects of airport operations and land uses, heights of buildings, residential densities and residential intensities and the disclosure of aircraft overflight. The adopted ALUCP for SDIA contains

policies that limit residential uses in areas experiencing noise above 60 dB CNEL by placing conditions on residential uses within the 60 dB CNEL contour. Residential uses in such areas may require sound attenuation to reduce interior noise levels to 45 dB. Since the Airport Land Use Commission does not have land use authority, the City implements the compatibility plan through land use plans, development regulations, and zoning regulations.

5.1.5 San Diego Forward: The Regional Plan

SANDAG is the regional authority that creates regional-specific documents to provide guidance to local agencies, as SANDAG does not have land use authority. SANDAG's San Diego Forward: The Regional Plan (RP) combines two of the region's existing planning documents: is an update of the Regional Comprehensive Plan for the San Diego Region (RCP) and the 2050 Regional Transportation Plan/Sustainable Communities Strategy (2050 RTP/SCS), combined into one document. The Regional Plan<u>RCP</u>, adopted in 2004, laid out key principles for managing the region's growth while preserving natural resources and limiting urban sprawl. The plan covered eight policy areas, including urban form, transportation, housing, health environment, economic prosperity, public facilities, our borders, and social equity. These policy areas were addressed in the 2050 RTP/SCS and are now fully integrated into the Regional Plan.

On April 24, 2015, SANDAG released the draft RP for public comment, with a closing date of July 15, 2015. A final RP was adopted by the SANDAG Board of Directors on October 9, 2015. provides a blueprint for San Diego's regional transportation system in order to effectively serve existing and projected workers and residents within the San Diego region. In addition to the 2050 RTP, the Regional Plan includes an SCS, in compliance with Senate Bill (SB) 375. The SCS aims to create sustainable, mixed-use communities conducive to public transit, walking, and biking by focusing future growth in the previously developed, western portion of the region along the major existing transit and transportation corridors. The purpose of the SCS is to help the region meet the greenhouse gas (GHG) emissions reductions set by the California Air Resources Board (CARB). The Regional Plan has a horizon year of 2050 and projects regional growth and the construction of transportation projects over this time period. The Regional Plan was adopted by the San Diego Association of Governments (SANDAG) Board on October 9, 2015.

5.2 Visual Effects and Neighborhood Character

5.2.1 California Scenic Highways Program

Recognizing the value of scenic areas and the value of views from roads in such areas, the California State Legislature established the California Scenic Highway Program in 1963. This legislation sees scenic highways as "a vital part of the all-encompassing effort...to protect and enhance California's beauty, amenity and quality of life." Under this program, a number of state highways have been designated as eligible for inclusion as scenic routes. The one-mile portion of State Route 163, known as the Cabrillo Freeway, between the north and south boundaries of Balboa Park, is an Officially Designated State Scenic Highway.

5.2.2 City Of San Diego General Plan

The General Plan includes Citywide design goals and policies regarding visual elements that complement the goals for pedestrian-oriented and walkable villages from the City of Villages strategy. A village environment includes high-quality public spaces, civic architecture, and the enhancement of visual quality of all types of development.

The Urban Design Element of the General Plan establishes a set of design principles from which future physical design decisions can be based. Policies call for respecting San Diego's natural topography and distinctive neighborhoods, providing public art, and encouraging the development of walkable, transit-oriented communities.

In its introduction, the Urban Design Element of the General Plan states:

As the availability of vacant land becomes more limited, designing infill development and redevelopment that builds upon our existing communities becomes increasingly important. A compact, efficient, and environmentally sensitive pattern of development becomes increasingly important as the City continues to grow. In addition, future development should accommodate and support existing and planned transit service (City of San Diego 2008).

The General Plan Urban Design Element policies relevant to planning at the Community Plan level involve architectural and landscape elements, as well as the design of transit, parking, and residential. As part of community planning, this element also contains policies related to public spaces and cultural amenities that contribute to the character of neighborhoods.

5.3 Transportation and Circulation

This section summarizes existing regulations that apply to the transportation system.

5.3.1 State Regulations

5.3.1.1 California Department of Transportation (Caltrans)

Caltrans is the primary state agency responsible for transportation issues. One of its duties is the construction and maintenance of the state highway system. Caltrans has established standards for street traffic flow and has developed procedures to determine if intersections require improvements. For projects that may physically affect facilities under its administration, Caltrans requires encroachment permits before any construction work may be undertaken. For projects that would not physically affect facilities but may influence traffic flow and levels of services at such facilities, Caltrans may recommend measures to mitigate the traffic impacts of such projects. In addition, Caltrans must review proposals to signalize any freeway ramp interchanges through their Intersection Control Evaluation process (Caltrans Traffic Operations Policy Directive #13-01).

5.3.1.2 California Transportation Commission (CTC)

The CTC consists of nine members appointed by the California Governor. CTC is responsible for the programming and allocating of funds for the construction of highway, passenger rail, and transit improvements throughout the state. CTC is responsible for adopting the State Transportation Improvement Program and the State Highway Operation and Protection Program.

5.3.1.3 AB 1358 – California Complete Streets Act of 2008

Supporting some of the previously referenced regulations/requirements, the California Complete Streets Act of 2008 (Assembly Bill [AB] 1358) requires circulation elements as of January 1, 2011, to accommodate the transportation system from a multi-modal perspective, including public transit and walking and biking, which have traditionally been marginalized in comparison to autos in contemporary American urban planning.

5.3.2 Local Regulations

5.3.2.1 San Diego Forward: The Regional Plan

See Section 5.1.5 for discussion of San Diego Forward: The Regional Plan.

5.3.2.2 SANDAG Regional Bike Plan

The San Diego Regional Bike Plan adopted by SANDAG supports implementation of the RP. It provides a regional strategy to make riding a bike a useful form of transportation for everyday travel. The plan will help San Diego meet its goals to reduce greenhouse gas emissions and improve mobility. Goals of the Regional Bike Plan include: increase levels of bicycling; improve bicycling safety; encourage complete streets; support reductions in emissions; and increase community support. In September 2013, the SANDAG Board of Directors approved funding to implement the Regional Bike Plan Early Action Program, which focuses on the region's highest priority projects. Priority is chosen in part based on proximity to smart growth areas, taking into account the fact that bikeways would be used more often if they connect high-density activity hubs within a short distance of each other, and on whether a project would fill key gaps in the regional bike networks.

5.3.2.23 City of San Diego General Plan

The Mobility Element of the City of San Diego General Plan defines the policies regarding traffic flow and transportation facility design. The purpose of the Mobility Element is "to improve mobility through development of a balanced, multi-modal transportation network." The main goals of the Mobility Element pertain to walkable communities, transit first, street and freeway system, intelligent transportation systems (ITS), Transportation Demand Management (TDM), bicycling, parking management, airports, passenger rail, goods movement/freight, and regional transportation coordination and financing.

a. Uptown Adopted Community Plan Mobility Element

The purpose of the adopted Uptown Community Plan Mobility Element is to establish goals and policies to guide future street network and design, street classification, Level of Service (LOS), transit facilities and service, pedestrian and bicycle accommodations, and facility improvements needed to support future travel needs within the Community Plan area. This element would be replaced by the Mobility Element of the CPU if adopted.

b. City of San Diego Bicycle Master Plan (Update December 2013)

The City's Bicycle Master Plan Update (City of San Diego 2013) provides a framework for making cycling a more practical and convenient transportation option for a wider variety of San Diegans with varying riding purposes and skill-levels. The plan update evaluates and builds on the 2002 Bicycle Master Plan so that it reflects changes in bicycle user needs and changes to the City's bicycle network and overall infrastructure.

5.4 Air Quality

Motor vehicles are San Diego County's leading source of air pollution. In addition to these sources, other mobile sources include construction equipment, trains, and airplanes. Emission standards for mobile sources are established by state and federal agencies, such as the CARB and the U.S. EPA. Reducing mobile source emissions requires the technological improvement of existing mobile sources and the examination of future mobile sources, such as those associated with new or modification projects (e.g., retrofitting older vehicles with cleaner emission technologies). The State of California has developed statewide programs to encourage cleaner cars and cleaner fuels. Since 1996, smog-forming emissions from motor vehicles have been reduced by 15 percent, and the cancer risk from exposure to motor vehicle air toxics has been reduced by 40 percent. The regulatory framework described below details the federal and state agencies that are in charge of monitoring and controlling mobile source air pollutants and the measures currently being taken to achieve and maintain healthful air quality in the San Diego Air Basin (SDAB).

In addition to mobile sources, stationary sources also contribute to air pollution in the SDAB. Stationary sources include gasoline stations, power plants, dry cleaners, and other commercial and industrial uses. Stationary sources of air pollution are regulated by the local air pollution control or management district, in this case the San Diego Air Pollution Control District (APCD).

The State of California is divided geographically into 15 air basins for managing the air resources of the state on a regional basis. Areas within each air basin are considered to share the same air masses and, therefore, are expected to have similar ambient air quality. If an air basin is not in either federal or state attainment for a particular pollutant, the basin is classified as a moderate, serious, severe, or extreme non-attainment area for that pollutant (there is also a marginal classification for federal non-attainment areas). Once a non-attainment area has achieved the air quality standards for a particular pollutant, it may be redesignated to an attainment area for that pollutant. To be redesignated, the area must meet air quality standards and have a 10-year plan for continuing to meet and maintain air quality standards, as well as satisfy other requirements of the Clean Air Act. Areas that are redesignated to attainment are called maintenance areas.

5.4.1 Federal Regulations

Ambient Air Quality Standards represent the maximum levels of background pollution considered safe, with an adequate margin of safety, to protect the public health and welfare. The federal Clean Air Act (CAA) was enacted in 1970 and amended in 1977 and 1990 [42 United States Code (USC) 7401] for the purposes of protecting and enhancing the quality of the nation's air resources to benefit public health, welfare, and productivity. In 1971, in order to achieve the purposes of Section 109 of the CAA [42 USC 7409], the U.S. EPA developed primary and secondary national ambient air quality standards (NAAQS).

Six criteria pollutants of primary concern have been designated: ozone (O₃), carbon monoxide (CO), sulfur dioxide (SO₂), nitrogen dioxide (NO₂), lead (Pb), and respirable particulate matter (PM₁₀ and PM_{2.5}). The primary NAAQS "...in the judgment of the Administrator, based on such criteria and allowing an adequate margin of safety, are requisite to protect the public health..." and the secondary standards "...protect the public welfare from any known or anticipated adverse effects associated with the presence of such air pollutant in the ambient air" [42 USC 7409(b)(2)]. The primary NAAQS were established, with a margin of safety, considering long-term exposure for the most sensitive groups in the general population (i.e., children, senior citizens, and people with breathing difficulties). The NAAQS are presented in Table 5-2.

Table 5-2						
		Ambier	nt Air Quality S	tandards		
Pollutant	Averaging	California	Standards ¹	Driman ^{3,5}	National Standa	ards ²
Ozone ⁸	1 Hour	0.09 ppm (180 µg/m ³)	Ultraviolet	-	Same as	Ultraviolet
020110	8 Hour	0.07 ppm (137 µg/m³)	Photometry	0.070 ppm (137 μg/m³)	Standard	Photometry
Respirable Particulate Matter (PM ₁₀) ⁹	24 Hour Annual Arithmetic Mean	50 μg/m³ 20 μg/m³	Gravimetric or Beta Attenuation	<u>150 μg/m³</u> -	Same as Primary Standard	Inertial Separation and Gravimetric Analysis
Fine Particulate	24 Hour	No Separate State	e Standard	35 µg/m³	Same as Primary Standard	Inertial Separation and
Matter (PM _{2.5}) ⁹	Annual Arithmetic Mean	12 µg/m³	Gravimetric or Beta Attenuation	12 µg/m³	15 µg/m³	Gravimetric Analysis
	1 Hour	20 ppm (23 mg/m ³)		35 ppm (40 mg/m ³)	-	
Carbon Monoxide	8 Hour	9.0 ppm (10 mg/m ³)	Non-dispersive Infrared	9 ppm (10 mg/m ³)	-	Non-dispersive Infrared
(CO)	8 Hour (Lake Tahoe)	6 ppm (7 mg/m³)	Photometry	-	-	Photometry
Nitrogen	1 Hour	0.18 ppm (339 µg/m³)	Gas Phase	100 ppb (188 µg/m³)	-	Gas Phase
Dioxide (NO ₂) ¹⁰	Annual Arithmetic Mean	0.030 ppm (57 μg/m³)	Chemi- luminescence	0.053 ppm (100 μg/m ³)	Same as Primary Standard	Chemi- luminescence
	1 Hour	0.25 ppm (655 μg/m³)		75 ppb (196 μg/m³)	-	
Sulfur	3 Hour	-	Ultraviolat	-	0.5 ppm (1,300 μg/m³)	Ultraviolet Fluorescence;
Dioxide (SO ₂) ¹¹	24 Hour	0.04 ppm (105 μg/m³)	Fluorescence	0.14 ppm (for certain areas) ¹⁰	-	photometry (Pararosaniline
	Annual Arithmetic Mean	-		0.030 ppm (for certain areas) ¹⁰	-	Method)
	30 Day Average	1.5 μg/m ³		-	-	
Lead ^{12,13}	Calendar Quarter	-	Atomic Absorption	1.5 μg/m ³ (for certain areas) ¹²	Same as	High Volume Sampler and Atomic
	Rolling 3-Month Average	_		0.15 µg/m ³	Standard	Absorption
Visibility Reducing Particles ¹⁴	8 Hour	See footnote 13	Beta Attenuation and Transmittance through Filter Tape			
Sulfates	24 Hour	25 µg/m³	lon Chroma- tography	No National S	tandards	
Hydrogen Sulfide	1 Hour	0.03 ppm (42 μg/m ³)	Ultraviolet Fluorescence			
Vinyl Chloride ¹²	24 Hour	0.01 ppm (26 µg/m³)	Gas Chroma- tography			
see joolnoles on next page.						

	Table 5-2					
	Ambient Air Quality Standards					
þ	pm = parts per million; ppb = parts per billion; µg/m ³ = micrograms per cubic meter; – = not applicable. California standards for ozone, carbon monoxide (except 8-hour Lake Tahoe), sulfur dioxide (1 and 24 hour),					
	exceeded. All others are not to be equaled or exceeded. California ambient air quality standards are listed in the Table of Standards in Section 70200 of Title 17 of the California Code of Regulations					
2	National standards (other than ozone, particulate matter, and those based on annual arithmetic mean) are not to be exceeded more than once a year. The ozone standard is attained when the fourth highest 8-hour concentration					
	measured at each site in a year, averaged over three years, is equal to or less than the standard. For PM ₁₀ , the 24- hour standard is attained when the expected number of days per calendar year with a 24-hour average concentration above 150 µg/m ³ is equal to or less than one. For PM _{2.5} , the 24-hour standard is attained when 98 percent of the daily concentrations, averaged over three years, are equal to or less than the standard. Contact the					
3	U.S. EPA for further clarification and current national policies.					
	based upon a reference temperature of 25°C and a reference pressure of 760 torr. Most measurements of air quality are to be corrected to a reference temperature of 25°C and a reference pressure of 760 torr; ppm in this					
4	Any equivalent measurement method which can be shown to the satisfaction of the Air Resources Board to give					
5	National Primary Standards: The levels of air quality necessary, with an adequate margin of safety to protect the nublic health					
6	National Secondary Standards: The levels of air quality necessary to protect the public welfare from any known or anticipated adverse effects of a pollutant.					
7	Reference method as described by the U.S. EPA. An "equivalent method" of measurement may be used but must have a "consistent relationship to the reference method" and must be approved by the U.S. EPA.					
8	On October 1, 2015, the national 8-hour ozone primary and secondary standards were lowered from 0.075 to 0.070 ppm.					
9	On December 14, 2012, the national annual PM _{2.5} primary standard was lowered from 15 μ g/m ³ to 12.0 μ g/m ³ . The existing national 24-hour PM _{2.5} standards (primary and secondary) were retained at 35 μ g/m ³ , as was the annual secondary standards of 15 μ g/m ³ . The existing 24-hour PM ₁₀ standards (primary and secondary) of 150 μ g/m ³ also were retained. The form of the annual primary and secondary standards is the annual mean, averaged over 3					
10	years. To attain the 1-hour national standard, the 3-year average of the annual 98th percentile of the 1-hour daily					
	maximum concentrations at each site must not exceed 100 ppb. Note that the national standards are in units of parts per billion (ppb). California standards are in units of parts per million (ppm). To directly compare the national standards to the California standards the units can be converted from ppb to ppm. In this case, the national standard of 100 ppb is identical to 0.100 ppm.					
11	On June 2, 2010, a new 1-hour SO ₂ standard was established and the existing 24-hour and annual primary standards were revoked. To attain the 1-hour national standard, the 3-year average of the annual 99 th percentile of the annual 91 th percentile o					
	standards (24-hour and annual) remain in effect until one year after an area is designated for the 2010 standard, except that in areas designated nonattainment for the 1971 standards, the 1971 standards remain in effect until					
	Note that the 1-hour national standard is in units of parts per billion (ppb). California standards are in units of parts per million (ppm). To directly compare the 1-hour national standard to the California standard the units can					
12	be converted to ppm. In this case, the national standard of 75 ppb is identical to 0.075 ppm. The ARB has identified lead and vinyl chloride as 'toxic air contaminants' with no threshold level of exposure for adverse health effects determined. These actions allow for the implementation of control measures at levels					
13	below the ambient concentrations specified for these pollutants. The national standard for lead was revised on October 15, 2008 to a rolling 3-month average. The 1978 lead					
	standard (1.5 μg/m ³ as a quarterly average) remains in effect until one year after an area is designated for the 2008 standard, except that in areas designated nonattainment for the 1978 standard, the 1978 standard remains in effect until implementation plans to attain or maintain the 2008 standard are approved.					
14	In 1989, the ARB converted both the general statewide 10-mile visibility standard and the Lake Tahoe 30-mile visibility standard to instrumental equivalents, which are "extinction of 0.23 per kilometer" and "extinction of 0.07					
SC	per kilometer" for the statewide and Lake Tahoe Air Basin standards, respectively. DURCE: CARB 2015.					

5.4.2 State Regulations

5.4.2.1 Criteria Pollutants

The U.S. EPA allows states the option to develop different (stricter) standards. The State of California has developed the California Ambient Air Quality Standards (CAAQS) and generally has set more stringent limits on the criteria pollutants (see Table 5-2). In addition to the Federal criteria pollutants, the CAAQS also specify standards for visibility-reducing particles, sulfates, hydrogen sulfide, and vinyl chloride (see Table 5-2). The California CAA, also known as the Sher Bill or California AB 2595, was signed into law on September 30, 1988, and became effective on January 1, 1989. The California CAA requires that districts implement regulations to reduce emissions from mobile sources through the adoption and enforcement of transportation control measures. The California CAA also requires that a district must:

- 1. Demonstrate the overall effectiveness of the air quality program;
- 2. Reduce non-attainment pollutants at a rate of five percent per year, or include all feasible measures and expeditious adoption schedule;
- 3. Ensure no net increase in emissions from new or modified stationary sources;
- 4. Reduce population exposure to severe non-attainment pollutants according to a prescribed schedule;
- 5. Include any other feasible controls that can be implemented, or for which implementation can begin, within ten years of adoption of the most recent air quality plan; and
- 6. Rank control measures by cost-effectiveness. The SDAB is a non-attainment area for the State O_3 standards, the State PM_{10} standard, and the State $PM_{2.5}$ standard.

5.4.2.2 Toxic Air Contaminants

The public's exposure to toxic air contaminants (TACs) is a significant public health issue in California. In 1983, the California Legislature enacted a program to identify the health effects of TACs and to reduce exposure to these contaminants to protect the public health (AB 1807: Health and Safety Code Sections 39650–39674). The Legislature established a two-step process to address the potential health effects from TACs. The first step is the risk assessment (or identification) phase. The second step is the risk management (or control) phase of the process.

The California Air Toxics Program establishes the process for the identification and control of toxic air contaminants and includes provisions to make the public aware of significant toxic exposures and for reducing risk. Additionally, the Air Toxics "Hot Spots" Information and Assessment Act (AB 2588, 1987, Connelly Bill) was enacted in 1987 and requires stationary sources to report the types and quantities of certain substances routinely released into the air. The goals of the Air Toxics "Hot Spots" Act are to collect emission data, to identify facilities having localized impacts, to ascertain health risks, to notify nearby residents of significant risks, and to reduce those significant risks to

acceptable levels. The Children's Environmental Health Protection Act, California <u>Senate Bill (SB)</u> 25 (Chapter 731, Escutia, Statutes of 1999), focuses on children's exposure to air pollutants. The act requires CARB to review its air quality standards from a children's health perspective, evaluate the statewide air monitoring network, and develop any additional air toxic control measures needed to protect children's health. Locally, toxic air pollutants are regulated through the San Diego APCD's Regulation XII.

Of particular concern statewide are diesel-exhaust particulate matter (DPM) emissions. DPM was established as a TAC in 1998 and is estimated to represent a majority of the cancer risk from TACs statewide (based on the statewide average). Diesel exhaust is a complex mixture of gases, vapors, and fine particles. This complexity makes the evaluation of health effects of diesel exhaust a complex scientific issue. Some of the chemicals in diesel exhaust, such as benzene and formaldehyde, have been previously identified as TACs by the CARB and are listed as carcinogens either under the State's Proposition 65 or under the Federal Hazardous Air Pollutants program.

Following the identification of diesel particulate matter as a TAC in 1998, CARB has worked on developing strategies and regulations aimed at reducing the risk from diesel particulate matter. The overall strategy for achieving these reductions is found in the *Risk Reduction Plan to Reduce Particulate Matter Emissions from Diesel-fueled Engines and Vehicles* (State of California 2000). A stated goal of the plan is to reduce the cancer risk statewide arising from exposure to diesel particulate matter 85 percent by 2020.

5.4.2.3 State Implementation Plan

State Implementation Plan (SIP) is a collection of documents that set forth the state's strategies for achieving the NAAQS. In California, the SIP is a compilation of new and previously submitted plans, programs (such as monitoring, modeling, permitting, etc.), district rules, state regulations, and federal controls. The CARB is the lead agency for all purposes related to the SIP under state law. Local air districts and other agencies, such as the Department of Pesticide Regulation and the Bureau of Automotive Repair, prepare SIP elements and submit them to CARB for review and approval. The CARB then forwards SIP revisions to the U.S. EPA for approval and publication in the Federal Register. All of the items included in the California SIP are listed in the Code of Federal Regulations (CFR) at 40 CFR 52.220.

The San Diego APCD is responsible for preparing and implementing the portion of the SIP applicable to the SDAB. The San Diego APCD adopts rules, regulations, and programs to attain State and federal air quality standards, and appropriates money (including permit fees) to achieve these objectives.

5.4.2.4 Regional Air Quality Strategy

The San Diego APCD prepared the 1991/1992 Regional Air Quality Strategy (RAQS) in response to the requirements set forth in AB 2595. The draft was adopted, with amendments, on June 30, 1992 (County of San Diego 1992). Attached, as part of the RAQS, are the Transportation Control Measures (TCMs) for the air quality plan prepared by the SANDAG in accordance with AB 2595 and adopted by SANDAG on March 27, 1992, as Resolution Number 92-49 and Addendum. The required triennial

updates of the RAQS and corresponding TCMs were adopted in 1995, 1998, 2001, 2004, and 2009. An update is currently being prepared based on the revised 8-hour ozone standard. The RAQS and TCMs set forth the steps needed to accomplish attainment of the CAAQS.

5.5 Greenhouse Gas Emissions

In response to rising concern associated with increasing GHG emissions and global climate change impacts, several plans and regulations have been adopted at the national, state, and local levels with the aim of reducing GHG emissions. Important federal, state, and local plans and regulations are summarized below.

5.5.1 Federal

5.5.1.1 Federal Clean Air Act

The U.S. Supreme Court ruled on April 2, 2007, in *Massachusetts v. U.S. Environmental Protection Agency* that carbon dioxide (CO₂) is an air pollutant, as defined under the CAA, and that the U.S. EPA has the authority to regulate emissions of GHGs. The U.S. EPA announced that GHGs (including CO₂, methane [CH₄], nitrous oxide [N₂O], hydrofluorocarbons [HFC], perfluorocarbons [PFC], and sulfur hexafluoride [SF₆]) threaten the public health and welfare of the American people. This action was a prerequisite to finalizing the U.S. EPA's GHG emissions standards for light-duty vehicles, which were jointly proposed by the U.S. EPA and the United States Department of Transportation's National Highway Traffic Safety Administration (NHTSA). The standards were established on April 1, 2010 for 2012 through 2016 model year vehicles and on October 15, 2012 for 2017 through 2025 model year vehicles (U.S. EPA 2010, 2012).

5.5.1.2 Climate Change Action Plan

Adopted in 1993, the U.S. Climate Change Action Plan (CCAP) consists of voluntary actions to reduce all significant GHGs from all economic sectors. Backed by federal funding, the CCAP supports cooperative partnerships between the government and the private sector in establishing flexible and cost-effective ways to reduce GHG emissions. The CCAP encourages investments in new technologies, but also relies on previous actions and programs focused on saving energy, reducing transportation emissions, improving forestry management, and reducing waste. With respect to energy and transportation-related GHG emissions reductions, the CCAP includes the following:

 Energy Demand Actions to accelerate the use of existing energy saving technologies and encourage the development of more advanced technologies. Commercial actions focus on installing efficient heating and cooling systems in commercial buildings and upgrading to energy-efficient lighting systems (the Green Lights program). The State Buildings Energy Incentive Fund provides funding to states for the development of public building energy management programs. Residential actions focus on developing new residential energy standards and building codes and providing money-saving energy efficient options to homeowners.

- 2. Energy Supply Actions to reduce emissions from energy supply. These actions focus on increasing the use of natural gas, which emits less CO₂ than coal or oil, and investing in renewable energy sources, such as solar and wind power, which result in zero net CO₂ emissions. Energy supply strategies also focus on reducing the amount of energy lost during distribution from power plants to consumers.
- 3. Transportation Actions to reduce transportation-related emissions are focused on investing in cleaner fuels and more efficient technologies, and reducing vehicle miles traveled (VMT). In addition, the U.S. EPA and Department of Transportation (U.S. DOT) are to draft guidance documents for reducing VMTs for use in developing local clean air programs.

5.5.1.3 Corporate Average Fuel Economy Standards

The federal Corporate Average Fuel Economy (CAFE) standards determine the fuel efficiency of certain vehicle classes in the U.S. While the standards had not changed since 1990, as part of the Energy and Security Act of 2007, the CAFE standards were increased in 2007 for new light-duty vehicles to 35 miles per gallon (mpg) by 2020. In May 2009, President Obama announced further plans to increase CAFE standards to require light-duty vehicles to meet an average fuel economy of 35.5 mpg by 2016. With improved gas mileage, fewer gallons of transportation fuel would be combusted to travel the same distance, thereby reducing nationwide GHG emissions associated with vehicle travel.

5.5.2 State

The State of California has adopted a number of plans and regulations aimed at identifying statewide and regional GHG emissions caps, GHG emissions reduction targets, and actions and timelines to achieve the target GHG reductions.

5.5.2.1 Executive Order S-3-05 – Statewide GHG Emission Targets

This executive order (EO), signed on June 1, 2005, established the following GHG emission reduction targets for the state of California:

- by 2010, reduce GHG emissions to 2000 levels;
- by 2020 reduce GHG emissions to 1990 levels;
- by 2050 reduce GHG emissions to 80 percent below 1990 levels.

This EO also directs the secretary of the California EPA to oversee the efforts made to reach these targets, and to prepare biannual reports on the progress made toward meeting the targets and on the impacts to California related to global warming, including impacts to water supply, public health, agriculture, the coastline, and forestry. With regard to impacts, the report shall also prepare and report on mitigation and adaptation plans to combat the impacts. The first Climate Action Team Assessment Report was produced in March 2006 and has been updated every two years.

5.5.2.2 Executive Order B-30-15 – 2030 Statewide GHG Emission Goal

This EO, issued by Governor Brown on April 29, 2015, established an interim GHG emission reduction goal for the state of California: by 2030, reduce GHG emissions to 40 percent below 1990 levels. This EO also directed all state agencies with jurisdiction over GHG emitting sources to implement measures designed to achieve the new interim 2030 goal as well as the pre-existing long-term 2050 goal identified in EO S- 3-05 (see discussion above). Additionally, this EO directed CARB to update its AB 32 (Nuñez) mandated Scoping Plan (see discussion above) to address the 2030 goal. Therefore, in the coming months, CARB is expected to develop statewide inventory projection data for 2030 as well as commence its efforts to identify reduction strategies capable of securing emission reductions that allow for achievement of the EO's new interim goal.

5.5.2.3 AB 32 – California Global Warming Solutions Act

In response to EO S-3-05, the California legislature passed AB 32, the California Global Warming Solutions Act of 2006, which was signed on September 27, 2006. It requires the CARB to adopt rules and regulations that would reduce GHG emissions to 1990 levels by 2020. The CARB is also required to publish a list of discrete GHG emission reduction measures. As required by AB 32, CARB has established a statewide GHG emissions cap for 2020, and adopted reporting rules for large industrial sources and a Climate Change Scoping Plan (Scoping Plan).

5.5.2.4 Climate Change Scoping Plan

As directed by AB 32, the Scoping Plan prepared by CARB in December 2008 includes measures to reduce statewide GHG emissions to 1990 levels by 2020. These reductions are what CARB identified as necessary to reduce forecasted "Business As Usual" (BAU) 2020 emissions. CARB will update the Scoping Plan at least once every five years to allow evaluation of progress made and to correct the Scoping Plan's course where necessary.

The 2008 Scoping Plan estimated annual BAU 2020 emissions to reach 596 million metric tons of CO₂ equivalent (MMT CO₂E). Thus, to achieve 1990 emissions levels of 427 MMT CO₂E, a 169 MMTCO₂E reduction was thus determined to be needed by 2020. The majority of reductions are directed at the sectors with the largest GHG emissions contributions— transportation and electricity generation—and involve statutory mandates affecting vehicle or fuel manufacture, public transit, and public utilities. The CARB list of reductions is included in the technical GHG analysis in Appendix E. The Scoping Plan also lists several other recommended measures that will contribute toward achieving the 2020 statewide reduction goal, but whose reductions are not (for various reasons, including the potential for double counting) additive with the measures listed in Table 8 of Appendix E. These include state and local government operations. The Scoping Plan reduction measures and complementary regulations are described further in the following sections, and are grouped under the two headings of Transportation-related Measures and Non-Transportation-Related Measures as representative of the sectors to which they apply.

Approved in May 2014, the First Update to the Scoping Plan defines CARB's priorities for the next five years and sets the groundwork to reach long-term goals set forth in EO S-3-05. The First Update describes advancements in climate science such as the quantification of the impacts of temperature change, further understanding of the mechanisms of climate pollutants (black carbon, methane, and hydrofluorocarbons), and improvements to GHG monitoring. The First Update also describes progress made since the original Scoping Plan including implementation of a more comprehensive Cap-and-Trade Program, the Low Carbon Fuel Standard (LCFS), a 33 percent Renewable Portfolio Standard, and an Advanced Clean Cars program that has been adopted at the Federal level.

5.5.2.5 AB 1493 – Vehicular Emissions of Greenhouse Gases

AB 1493 (Pavley) directed CARB to adopt vehicle standards that lowered GHG emissions from passenger vehicles and light-duty trucks to the maximum extent technologically feasible, beginning with the 2009 Model Year. CARB has adopted amendments to its regulations that would enforce AB 1493 but provide vehicle manufacturers with new compliance flexibility. Pavley standards are currently divided into two phases. Standards that regulate vehicles model years 2009 through 2016 are termed "Pavley I", standards for Model Years 2017 through 2025 were originally termed "Pavley II".

With these actions, it is expected that Pavley I will reduce GHG emissions from California passenger vehicles by a total of 31.5 MMT CO₂E counted toward the total pre- economic downturn statewide reduction target on the capped sector of 146.7 MMT CO₂E (CARB Scoping Plan). CARB adopted a second phase of the Pavley regulations, termed "Pavley II," which are now called the Low Emission Vehicle III (LEV III) Standards. LEV III covers Model Years 2017 to 2025. These reductions are to come from improved vehicle technologies such as small engines with superchargers, continuously variable transmissions, and hybrid electric drives.

5.5.2.6 Executive Order S-01-07 – Low Carbon Fuel Standard

This executive order directed that a statewide goal be established to reduce the carbon intensity of California's transportation fuels by at least ten percent by 2020 through a LCFS. CARB adopted the LCFS as a discrete early action measure pursuant to AB 32 in April 2009 and includes it as a reduction measure in its Scoping Plan.

The LCFS is a performance standard with flexible compliance mechanisms intended to incentivize the development of a diverse set of clean, low-carbon transportation fuel options. Its aim is to accelerate the availability and diversity of low-carbon fuels such as biofuels, electricity, and hydrogen, by taking into consideration the full life cycle of GHG emissions. A ten-percent reduction in the intensity of transportation fuels is expected to equate to a reduction of 16.5 MMTCO₂E in 2020. However, in order to account for possible overlap of benefits between LCFS and the Pavley GHG standards, CARB has discounted the contribution of LCFS to 15 MMT CO₂E.

5.5.2.7 Senate Bill 375—Regional Emissions Targets

The SB 375 was signed in September 2008 and requires CARB to set regional targets for reducing passenger vehicle GHG emissions in accordance with the Scoping Plan measure described above. Its

purpose is to align regional transportation planning efforts, regional GHG reduction targets, and land use and housing allocation to reduce GHG emissions by promoting high-density, mixed-use developments around mass transit hubs.

The Scoping Plan prepared pursuant to AB 32 by CARB in 2008 and updated in 2014 identifies reduction targets for all sources of GHG emissions in the state. While the transportation sector is responsible for the greatest GHG reductions (nearly 30 percent of the total reductions), most of these reductions will come from higher fuel efficiency vehicles per the Pavley standards (18 percent) and a more diverse fuel mix per the low carbon fuel standards (9 percent). statewide, RTPs prepared by metropolitan planning organizations, such as SANDAG, are responsible for less than 3 percent of the GHG reductions. SB 375 is the mechanism that establishes GHG emission reduction targets for each regional agency.

SANDAG's SB 375 target is to reduce GHG emissions from cars and light trucks by 7 percent, per capita, by 2020, and by 13 percent by 2035, using a 2005 baseline. The RP, encompassing both the RTP and SCS, shows that the region will exceed these targets by pursuing the following strategies: using land in ways to make developments more compact, conserving open space, and investing in a transportation system that provides people with alternatives to driving alone. The CARB, in consultation with Metropolitan Planning Organizations (MPOs), was required to provide each affected region with passenger vehicle GHG emissions reduction targets for 2020 and 2035 by September 30, 2010. The SANDAG is the San Diego region's MPO. On August 9, 2010 CARB released the staff report on the proposed reduction target, which was subsequently approved by CARB on September 23, 2010. The San Diego region will be required to reduce GHG emissions from cars and light trucks seven percent per capita by 2020 and 13 percent by 2035. The reduction targets are to be updated every eight years, but can be updated every four years if advancements in emissions technologies affect the reduction strategies to achieve the targets.

Once reduction targets are established, each of California's MPOs must prepare and adopt a SCS that demonstrates how the region will meet its GHG reduction targets through integrated land use, housing, and transportation planning. Enhanced public transit service combined with incentives for land use development that provides a better market for public transit will play an important role in the SCS. After the SCS is adopted by the MPO, the SCS will be incorporated into that region's federally enforceable RTP. San Diego's MPO, SANDAG, completed and adopted its 2050 RTP in October 2011, the first such plan in the state that included a SCS.

CARB is also required to review each final SCS to determine whether it would, if implemented, achieve the GHG emission reduction target for its region. If the combination of measures in the SCS will not meet the region's target, the MPO must prepare a separate Alternative Planning Strategy (APS) to meet the target. The APS is not a part of the RTP.

5.5.2.8 Million Solar Roofs Program

The Million Solar Roofs Program was created by SB 1 in 2006 and includes the California Public Utilities Commission's (CPUC's) California Solar Initiative and California Energy Commission's (CEC's) New Solar Homes Partnership. It requires publicly owned utilities to adopt, implement, and finance solar-incentive programs to lower the cost of solar systems and help achieve the goal of installing

3,000 megawatts (MW) of new solar capacity by 2020. The Million Solar Roofs Program is one of CARB's GHG reduction measures identified in the 2008 Scoping Plan. Achievement of the program's goal is expected to equate to a reduction of 2.1 MMT CO₂E in 2020 statewide BAU emissions.

5.5.2.9 California Energy Code

The California Code of Regulations, Title 24, Part 6 is the California Energy Code. This code, originally enacted in 1978 in response to legislative mandates, establishes energy- efficiency standards for residential and non-residential buildings in order to reduce California's energy consumption. The Energy Code is updated periodically to incorporate and consider new energy-efficiency technologies and methodologies as they become available. The most recent amendments to the Energy Code, known as 2013 Title 24, or the 2013 Energy Code, became effective July 1, 2015. The 2013 Title 24 requires energy use reductions of 25 to 30 percent above the former 2008 Title 24 Energy Code. By reducing California's energy consumption, emissions of statewide GHGs may also be reduced.

New construction and major renovations must demonstrate their compliance with the current Energy Code through submission and approval of a Title 24 Compliance Report to the local building permit review authority and the CEC. The compliance reports must demonstrate a building's energy performance through use of CEC-approved energy performance software that shows iterative increases in energy efficiency given selection of various heating, ventilation, and air-conditioning (HVAC); sealing; glazing; insulation; and other components related to the building envelope. Title 24 governs energy consumed by the built environment by the major building envelope systems such as space heating, space cooling, water heating, some aspects of the fixed lighting system, and ventilation. Non-building energy use, or plug-in energy use (such as appliances, equipment, electronics, plug-in lighting), are independent of building design and are not subject to Title 25.

5.5.2.10 California Green Building Standards

California Code of Regulations, Title 24, Part 11 are the California Green Building Standards. Beginning in 2011, California Green Building Standards Code (CalGreen) instituted mandatory minimum environmental performance standards for all ground-up new construction of commercial and low-rise residential buildings, state-owned buildings, schools, and hospitals. It also includes voluntary tiers (I and II) with stricter environmental performance standards for these same categories of residential and non-residential buildings. Local jurisdictions must enforce the minimum mandatory requirements and may adopt CalGreen with amendments for stricter requirements.

The mandatory standards require:

- 20 percent mandatory reduction in indoor water use relative to specified baseline levels;
- 50 percent construction/demolition waste diverted from landfills;
- mandatory inspections of energy systems to ensure optimal working efficiency; and
- requirements for low-pollutant emitting exterior and interior finish materials such as paints, carpets, vinyl flooring, and particle boards.

The voluntary standards require:

- Tier I 15 percent improvement in energy requirements, stricter water conservation requirements for specific fixtures, 65 percent reduction in construction waste, ten percent recycled content, 20 percent permeable paving, 20 percent cement reduction, cool/solar reflective roof; and
- Tier II 30 percent improvement in energy requirements, stricter water conservation requirements for specific fixtures, 75 percent reduction in construction waste, 15 percent recycled content, 30 percent permeable paving, 30 percent cement reduction, cool/solar reflective roof.

Similar to the compliance reporting procedure described above for demonstrating code compliance under Title 24, Part 6, in new buildings and major renovations, compliance with the CalGreen water reduction requirements must be demonstrated through completion of water use reporting forms for new low-rise residential and non-residential buildings. The water use compliance forms must demonstrate a 20 percent reduction in indoor water use by either showing a 20 percent reduction in the overall baseline water use as identified in CalGreen or a reduced per-plumbing-fixture water use rate.

The CARB Scoping Plan includes a Green Building Strategy with the goal of expanding the use of green building practices to reduce the carbon footprint of new and existing buildings. Consistent with CalGreen, the Scoping Plan recognized that GHG reductions would be achieved through buildings that exceed minimum energy-efficiency standards, decrease consumption of potable water, reduce solid waste during construction and operation, and incorporate sustainable materials. Green building is thus a vehicle to achieve the Scoping Plan's statewide electricity and natural gas efficiency targets, and lower GHG emissions from waste and water transport sectors.

In the Scoping Plan, CARB projects that an additional 26.3 MMT CO₂E could be reduced through expanded green building standards. However, this reduction is not counted toward the BAU 2020 reduction goal to avoid any double counting, as most of these reductions are accounted for in the electricity, waste, and water sectors. Because of this, CARB has assigned all emissions reductions that occur because of green building strategies to other sectors for meeting AB 32 requirements, but will continue to evaluate and refine the emissions from this sector.

5.5.2.11 Senate Bill 97—CEQA GHG Amendments

SB 97 (Dutton), passed by the legislature and signed on August 24, 2007, required the Office of Planning and Research on or before July 1, 2009 to prepare, develop, and transmit to the Resources Agency amendments to the CEQA guidelines (Guidelines) to assist public agencies in the evaluation and mitigation of GHGs or the effects of GHGs as required under CEQA, including the effects associated with transportation and energy consumption. SB 97 required the Resources Agency to certify and adopt those guidelines by January 1, 2010. Proposed amendments to the state CEQA Guidelines for GHG emissions were submitted on April 13, 2009, adopted on December 30, 2009, and became effective March 18, 2010.

Section 15065.4 of the amended Guidelines includes the following requirements for determining the significance of impacts from GHG emissions:

- (a) The determination of the significance of greenhouse gas emissions calls for a careful judgment by the lead agency consistent with the provisions in section 15065. A lead agency should make a good-faith effort, based to the extent possible on scientific and factual data, to describe, calculate, or estimate the amount of GHG emissions resulting from a project. A lead agency shall have discretion to determine, in the context of a particular project, whether to:
 - 1) Use a model or methodology to quantify greenhouse gas emissions resulting from a project, and which model or methodology to use. The lead agency has discretion to select the model or methodology it considers most appropriate provided it supports its decision with substantial evidence. The lead agency should explain the limitations of the particular model or methodology selected for use; and/or
 - 2) Rely on a qualitative analysis or performance-based standards.

While the amendments require calculation of a project's contribution, they clearly do not establish a standard by which to judge a significant effect or a means to establish such a standard.

5.5.3 Local

5.5.3.1 San Diego Forward: The Regional Plan

Refer to Section 5.1.5 for a discussion of SANDAG's San Diego Forward: The Regional Plan.

5.5.3.2 City of San Diego General Plan (2008)

The City General Plan includes several climate change-related policies aimed at reducing GHG emissions from future development and City operations. For example, Conservation Element policy CE-A.2 aims to "reduce the City's carbon footprint" and to "develop and adopt new or amended regulations, programs, and incentives as appropriate to implement the goals and policies set forth" related to climate change. The Land Use and Community Planning Element, the Mobility Element, the Urban Design Element, and the Public Facilities, Services, and Safety Element also identify GHG reduction and climate change adaptation goals. These elements contain policy language related to sustainable land use patterns, alternative modes of transportation, energy efficiency, water conservation, waste reduction, and greater landfill efficiency. The overall intent of these policies is to support climate protection actions, while retaining flexibility in the design of implementation measures, which could be influenced by new scientific research, technological advances, environmental conditions, or state and federal legislation.

One specific concept introduced in the General Plan is the City of Villages Strategy, which proposes growth to be directed into pedestrian-friendly mixed-use activity centers linked to an improved regional transit system. The City of Villages Strategy shifts the focus of land use policies to encourage infill development and reinvest in existing communities. Locating different land uses

types near one another can decrease mobile emissions. Thus, the development of dense urban "villages" would generate less GHG emissions. The City of Villages Strategy can be seen as an effort to avoid what is commonly referred to as "urban sprawl".

Cumulative impacts of GHG emissions were qualitatively analyzed and determined to be significant and unavoidable in the PEIR for the General Plan. A PEIR Mitigation Framework was included that indicated that "for each future project requiring mitigation (measures that go beyond what is required by existing programs, plans, and regulations), project-specific measures will [need to] be identified with the goal of reducing incremental project-level impacts to less than significant; or the incremental contributions of a project may remain significant and unavoidable where no feasible mitigation exists".

5.5.3.3 Climate Action Plan

In December 2015, the City adopted its Climate Action Plan (CAP). The CAP identifies measures to meet GHG reduction targets for 2020 and 2035. The CAP consists of a 2010 inventory of GHG emissions, a BAU projection for emissions at 2020 and 2035, state targets, and emission reductions with implementation of the CAP. The City identifies GHG reduction strategies focusing on energyand water-efficient buildings; clean and renewable energy; bicycling, walking, transit, and land use; zero waste; and climate resiliency. Accounting for future population and economic growth, the City projects GHG emissions will be approximately 15.9 MMT CO₂E in 2020 and 16.7 MMT CO₂E in 2035. To achieve its proportional share of the state reduction targets for 2020 (AB 32) and 2050 (EO S-3-05), the City would need to reduce emissions below the 2010 baseline by 15 percent in 2020 and 50 percent by 2035. To meet these goals, the City must implement strategies that reduce emissions to approximately 11.0 MMT CO₂E in 2020 and 6.5 MMT CO₂E in 2035. Through implementation of the CAP, the City is projected to reduce emissions even further below targets by 1.2 MMT CO₂E by 2020 and 205,462 MTCO₂E by 2035.

5.6 Noise

5.6.1 State

5.6.1.1 California Code of Regulations

Title 24, Chapter 12, Section 1207, of the California Building Code (CBC) requires that interior noise levels, attributable to exterior sources, not exceed to 45 dB CNEL in any habitable room. A habitable room in a building is used for living, sleeping, eating or cooking; bathrooms, closets, hallways, utility spaces, and similar areas are not considered habitable spaces. An acoustical study is required for proposed single-family, multiple-unit residential and hotel/motel structures within areas where the noise contours exceeds 60 dB CNEL. The studies must demonstrate that the design of the building will reduce interior noise to 45 dB CNEL or lower in inhabitable rooms. If compliance requires windows to be inoperable or closed, the structure must include ventilation or air-conditioning (24 California Code of Regulations [CCR] 1207 2010).

Title 24, Chapter 11 of CalGreen provides mandatory measures for residential and non-residential buildings. Section 5.507, Environmental Comfort, addresses interior noise control in non-residential buildings. This section provides the minimum Sound Transmission Class and Outdoor–Indoor Sound Transmission Class for wall, roof-ceiling assemblies, and windows for buildings located within the 65 A-weighted decibels (dB(A)) CNEL contour of an airport, freeway, expressway, railroad, industrial source, or fixed guideway source as determined by the Noise Element of the General Plan. As indicated, buildings shall be constructed to provide an interior noise environment attributable to exterior sources that does not exceed an hourly average equivalent level of 50 dB(A) Leq. Exterior features such as sound walls or earth berms may be utilized as appropriate to the building, addition, or alteration project to mitigate sound migration to the interior. An acoustical analysis documenting complying interior sound levels shall be prepared by personnel approved by the architect or engineer of record.

5.6.2 Local

5.6.2.1 City of San Diego General Plan

a. Exterior Noise

The City specifies compatibility standards for different categories of land use in the Noise Element of the General Plan. Table 5-3 provides the allowable noise levels by land use as identified in the City's General Plan (City of San Diego 2015).

As shown, the "compatible" noise level for noise sensitive receptors, including single- and multifamily residential, is 60 CNEL. Compatibility indicates that standard construction methods will attenuate exterior noise to an acceptable indoor noise level and people can carry out outdoor activities with minimal noise interference.

Exterior noise levels ranging between 65 and 70 CNEL are considered "conditionally compatible" for multiple units, mixed-use commercial/residential, live work, and group living accommodations. The Noise Element also states (Section B, Motor Vehicle Traffic Noise) that although not generally considered compatible, the City conditionally allows multi-family and mixed-use residential uses up to 75 dB(A) CNEL with a requirement to include attenuation measures to ensure an interior noise level of 45 dB(A) CNEL where a Community Plan allows multi-family and mixed-use.

For single-family units, mobile homes, and senior housing, exterior noise levels ranging between 60 and 65 dB(A) CNEL are considered "conditionally compatible." Conditionally compatible uses are permissible, provided interior noise levels will not exceed 45 dB(A) CNEL. Therefore, projects sited on land that falls into the "conditionally compatible" noise environment require an acoustical study.

Park uses are considered compatible in areas up to 70 dB(A) CNEL and conditionally compatible in areas between 70 and 75 dB(A) CNEL.

Table 5-3								
City of San Diego Land Use - Noise Compatibility Guidelines								
		(Table NE-3)					
				Ext	erior	Noise	Expos	ure
					(a	IBA CN	IEL)	75
	1	and Use Categor	∼v		00	05	70	75
Parks and Rec	reational		<i>,</i>					
Parks, Active a	and Passive Recreation	า						
Outdoor Spec Facilities	Dutdoor Spectator Sports, Golf Courses; Water Recreational Facilities; Indoor Recreation Facilities							
Agricultural	Agricultural Erop Baicing & Farming: Community Gardons, Aguacultura, Dairias: Horticultu							
Nurseries & G	ireenhouses; Animal F	aising, Maintain	& Keeping; Commercial Stables					
Single Dwellin	g Units; Mobile Home	S			45			
Multiple Dwe NE-D.3.	lling Units *For uses a	affected by aircra	aft noise, refer to Policies NE-D.2. &		45	45*		
Institutional			н Н					
Hospitals; Nu 12Educationa	rsing Facilities; Intern I Facilities; Libraries; N	nediate Care Fac Iuseums;; Child C	ilities; Kindergarten through Grade Care Facilities	·	45			
Other Educational Facilities including Vocational/Trade Schools and Colleges and Universities					45	45		
Cemeteries								
Retail Sales	nling/Equipments For	d Dovoragos 9	Crocorios, Data & Dat Suppliant			1		
Sundries, Pha	rmaceutical, & Conver	nience Sales; Wea	aring Apparel & Accessories			50	50	
Building Ser	vices Business Sur	nort: Fating &	Drinking: Financial Institutions:					
Maintenance & Repair; Personal Services; Assembly & Entretrainment (includes public and religious assembly); Radio & Television Studios; Golf Course Support						50	50	
Visitor Accommodations					45	45	45	
Offices						1		
Business & Pr Corporate He	ofessional; Governme adquarters	ent; Medical, Den	tal & Health Practitioner; Regional &			50	50	
Vehicle and Vehicle	ehicular Equipment Sa	ales and Services	Use					
Sales & Renta	ls; Vehicle Equipment	& Supplies Sales	& Rentals; Vehicle Parking					
Wholesale, Di	stribution, Storage Us	e Category	9 Storage Facilities: Warehouse					
Wholesale Dis	stribution	rarus, woving	& Storage Facilities, Warehouse,					
Howy Mapuf	acturing: Light Manuf	acturing: Marino	Industry: Trucking & Transportation					
Terminals; Mi	ning & Extractive Indu	stries					50	
Research & D	evelopment						50	
	Compatible	Indoor Uses	to an acceptable indoor noise level.	Refer	ttenu to Se	ate exection	terior I.	noise
		Outdoor Uses	Activities associated with the land us	se ma	y be	carrie	d out.	
45 50	Conditionally	Indoor Uses	Building structure must attenuate noise level indicated by the numb areas. Refer to Section I.	exteri ber (4	ior na IS or	oise to 50) f	o the in or occ	ndoor upied
Compatible Outdoor Uses Feasible noise mitigation techniques should be analyzed a incorporated to make the outdoor activities acceptable. Refer Section I.						d and efer to		
	Incompatible	Indoor Uses	New construction should not be unc	dertak	ken.			
	псотрацые	Outdoor Uses	Severe noise interference m unacceptable.	nakes	οι	utdoor	act	ivities
Source: City o	f San Diego, General F	Plan Amendment	to the Noise Element 2015.					

b. Interior Noise

Noise-sensitive residential/habitable interior spaces have an interior standard of 45 CNEL, as stated in the City's 2011 Significance Determination Thresholds and the California Noise Insulation Standards. The Significance Determination Thresholds indicate that for multi-family development, exterior noise levels would be considered significant if future projected traffic would result in noise levels exceeding 65 dB(A) CNEL at exterior usable areas or interior noise levels exceeding 45 dB(A) CNEL.

The City assumes that standard construction techniques will provide a 15 dB reduction of exterior noise levels to an interior receiver. Given this assumption, standard building construction could be assumed to result in interior noise levels of 45 dB CNEL or less when exterior noise sources are 60 dB(A) CNEL or less. When exterior noise levels are greater than 60 dB(A) CNEL, consideration of specific non-standard building construction techniques is required.

Proposed new construction and major renovations must demonstrate compliance with the current interior noise standards through submission and approval of a Title 24 Compliance Report.. In the case of ministerial projects for single family, there is no procedure to ensure that noise is adequately attenuated outside of the Airport Influence Area.

c. Policies

The General Plan Noise Element contains the following policies regarding the preparation of acoustical studies and interior noise guidelines:

- NE-A.4. Require an acoustical study consistent with Acoustical Study Guidelines (Table NE-4) for proposed developments in areas where the existing or future noise level exceeds or would exceed the "compatible" noise level thresholds as indicated on the Land Use Noise Compatibility Guidelines (Table NE-3), so that noise mitigation measures can be included in the project design to meet the noise guidelines.
- NE-I.1. Require noise attenuation measures to reduce the noise to an acceptable noise level for proposed developments to ensure an acceptable interior noise level, as appropriate, in accordance with California's noise insulation standards (CCR Title 24) and Airport Land Use Compatibly Plans.
- NE-1.2. Apply CCR Title 24 noise attenuation measures requirements to reduce the noise to an acceptable noise level for proposed single-family, mobile homes, senior housing, and all other types of residential uses not addressed by CCR Title 24 to ensure an acceptable interior noise level, as appropriate.
- NE-E.5. Implement night and daytime on-site noise level limits to address noise generated by commercial uses where it affects abutting residential and other noise-sensitive uses.

5.6.2.2 Noise Abatement and Control Ordinance

Section 59.5.0101 et seq. of the City Municipal Code, the Noise Abatement and Control Ordinance, regulates the sources of disturbing, excessive, or offensive noises within the City limits. Sound level limits are established for various types of land uses and are measured in one-hour averages. The 1-hour, A-weighted equivalent sound level, $L_{eq}(1)$, is the energy average of the A-weighted sound levels occurring during a 1-hour period. The Ordinance states that it is unlawful for any person to cause noise by any means to the extent that the 1-hour average sound level exceeds the applicable limit given for that land use. The sound level limit at a location on a boundary between two zoning districts is the arithmetic mean of the respective limits for the two districts. Table 5-4, shows the exterior noise limits specified in the City's Noise Control Ordinance.

Construction noise is regulated by Section 59.5.0404 of the City Municipal Code, that states:

- It shall be unlawful for any person, between the hours of 7:00 P.M. of any day and 7:00 A.M. of the following day, or on legal holidays as specified in Section 21.04 of the San Diego Municipal Code, with exception of Columbus Day and Washington's Birthday, or on Sundays, to erect, construct, demolish, excavate for, alter or repair any building or structure in such a manner as to create disturbing, excessive or offensive noise...
- . . . it shall be unlawful for any person, including the City of San Diego, to conduct any construction activity so as to cause, at or beyond the property lines of any property zoned residential, an average sound level greater than 75 decibels during the 12hour period from 7:00 A.M. to 7:00 P.M.

Table 5-4 San Diego Property Line Noise Level Limits							
Noise Level [dB(A)]							
	7:00 A.M. to 7:00 P.M. to 10:00 P.M. t						
Receiving Land Use Category	7:00 р.м.	10:00 р.м.	7:00 a.m.				
Single-family Residential	50	45	40				
Multi-family Residential (up to a maximum density of 1 dwelling unit/2,000 square feet)	55	50	45				
All Other Residential	60	55	50				
Commercial	65	60	60				
Industrial or Agricultural	75	75	75				
SOURCE: City of San Diego Municipal Code Section 5	59.5.0401						

5.6.2.3 Airport Land Use Compatibility Plan

As discussed in Section 5.1.4, the San Diego County Regional Airport Authority, prepared an ALUCP for the SDIA. The Uptown CPU area is within the Review Area 1 AIA for SDIA. The AIA serves as the boundary for the ALUCP. In addition to the policies and criteria addressing land use compatibilities, including building heights and densities, the ALUCP contains policies and criteria concerning noise.

The adopted ALUCP for SDIA contains policies that place conditions on residential uses at and above the 60 dB CNEL contour (Review Area 1). Table 5-5 provides the allowable noise levels by land use.

Table 5-5 Airport Noise Compatibility Criteria						
Land Use Category ^a		Exterior Noise	e Exposure (CNEL	.)		
Note: Multiple categories may apply to a project	60-65	65-70	70-75	75+		
Residential						
Single-family, Multi-family	45	45 ¹	45 ^{1,2}	45 ^{1,2}		
Single Room Occupancy (SRO) Facility	45	45 ¹	45 ^{1,2}	45 ^{1,2}		
Group Quarters	45	45 ¹	45 ^{1,2}	45 ^{1,2}		
Commercial, Office, Service, Transient Lodging				I		
Hotel, Motel, Resort	45/50	45/50	45/50	45/50		
Office – Medical, Financial, Professional Services, Civic			50	50		
Retail (e.g., Convenience Market, Drug Store, Pet Store)			50	50		
Service – Low Intensity (e.g., Gas Station, Auto Repair, Car			50	50		
Wash)						
Service – Medium Intensity (e.g., Check-cashing,			50	50		
Veterinary Clinics, Kennels, Personal Services)			50	50		
Service – High Intensity (e.g., Eating, Drinking			50	50		
Establishment, Funeral Chapel, Mortuary)			50	50		
Sport/Fitness Facility			50	50		
Theater – Movie/Live Performance/Dinner		45	45	45		
Educational, Institutional, Public Services				I		
Assembly – Adult (Religious, Fraternal, Other)	45	45 ¹	45 ¹	45 ¹		
Assembly – Children (Instructional Studios, Cultural						
Heritage Schools, Religious, other)	45					
Cemetery						
Child Day Care Center/Pre-K						
Convention Center						
Fire and Police Stations			50	50		
Jail, Prison		45/50	45/50	45/50		
Library, Museum, Gallery		45	45	45		
Medical Care – Congregate Care Facility, Nursing and						
Convalescent Home	45					
Medical Care – Hospital	45					
Medical Care – Out-Patient Surgery Centers	45					
Schools for Adults – College, University, Vocational/Trade	45	45 ¹	45 ¹			
School						
Schools– Kindergarten through Grade 12 (includes	45					
Charter Schools)						
Industrial						
Junkyard, Dump, Recycling Center, Construction Yard						
Manufacturing/Processing – General						
Manufacturing/Processing of Biomedical Agents.						
Biosafety Levels 3						
and 4 Only						
Manufacturing/Processing of Hazardous Materials ⁴						
Mining/Extractive Industry						
Research and Development – Scientific. Technical						
Sanitary Landfill						
Self-Storage Facility						
Warehousing/Storage – General						
Warehousing/Storage of Biomedical Agents, Biosafety						
Levels 3 and 4 Only						
Warehousing/Storage of Hazardous Materials ⁴						

Table 5-5 Airport Noiso Compatibility Critoria						
	Exterior Noise Exposure (CNEL)					
Note: Multiple categories may apply to a project		60-65	65-70	70-75	-/ 75+	
Transportation. Communication. Utilities			0370	,0,0	13.	
Auto Parking						
Electrical Power Generation Plant						
Electrical Substation						
Emergency Communications Facilities						
Marine Cargo Terminal						
Marine Passenger Terminal						
Transit Center, Bus/Rail Station						
Transportation, Communication, Utilities – General						
Truck Terminal						
Water, Wastewater Treatment Plant						
Recreation, Park, Open Space						
Arena, Stadium						
Golf Course						
Golf Course Clubhouse						
Marina						
Park, Open Space, Recreation						
Agriculture						
Aquaculture						
Agriculture						
	Compatible: Use is permitted.					
	Conditionally Compatible: Use is permitted subject to stated conditions.					
	Incompatible: Use is not permitted under any circumstances.					
45	Indoor uses: building must be capable of attenuating exterior noise to 45 CNEL					
50	Indoor uses: building must be capable of attenuating exterior noise to 50 CNEL					
45/50	Sleeping rooms must be attenuated to 45 CNEL any other indoor areas must be attenuated to 50 CNEL					
1	Avigation easement must be dedicated to the Airport owner/operator.					
2	New residential use is permitted above 70 CNEL contour only if current General/Community Plan					
designation allows for residential use. General/Community Plan amendments from a nonres						
	designation to a residential designation are not permitted.					
3 Refer to Appendix A of the San Diego International Airport Land Compatiblity Plan for definition Assembly – Children.						
						4
	manufacturing, processing and storage of hazardous materials.					
a Land uses not specifically listed shall be evaluated, as determined by Airport Land Use Cor						
using the critiera for similar uses. Refer to Appendix A of the San Diego Interna					al Airport Land	
	Compatiblity Plan.					
b	If this land use would occur within a sing	gle- or multi-f	amily residence,	it must be eval	uated using the	
criteria for single- or multi-family residential.						
SOURCE: San Diego County Regional Airport Authority 2014.						

5.7 Historical Resources

Federal, state, and local criteria have been established for the determination of historical resource significance. The criteria for determining a resource's significance generally focus on a resource's integrity and uniqueness, its relationship to similar resources, and its potential to contribute important information to scholarly research. Some resources that do not meet federal significance criteria may be considered significant under state or local criteria.

5.7.1 Federal

5.7.1.1 National Historic Preservation Act of 1966 and National Register of Historic Places

The National Historic Preservation Act of 1966 established the National Register of Historic Places (NRHP) as the official Federal list of cultural resources that have been nominated by State offices for their significance at the local, State, or Federal level. Listing on the NRHP provides recognition that a property is historically significant to the nation, the state, or the community. Properties listed (or potentially eligible for listing) on the NRHP must meet certain significance criteria and possess integrity of form, location, or setting. Barring exceptional circumstances, resources generally must be at least 50 years old to be considered for listing on the NRHP.

Criteria for listing on the NRHP are stated in Title 36, Part 60 of the Code of Federal Regulations (36 CFR 60). A resource may qualify for listing if there is quality of significance in American history, architecture, archaeology, engineering, and culture present in districts, sites, buildings, structures, and objects that possess integrity of location, design, setting, materials, workmanship, feeling, and association; and where such resources:

- Are associated with events that have made a significant contribution to the broad patterns of history.
- Are associated with the lives of persons significant in the past.
- Embody the distinctive characteristics of a type, period, or method of construction; represent the work of a master; possess high artistic values; or represent a significant and distinguishable entity whose components may lack individual distinction.
- Have yielded, or may be likely to yield, information important in prehistory or history.

Eligible properties must meet at least one of the NRHP criteria and exhibit integrity, measured by the degree to which the resource retains its historical properties and conveys its historical character, the degree to which the original historic fabric has been retained, and the reversibility of changes to the property. The fourth criterion is typically reserved for archaeological and paleontological resources. These criteria have largely been incorporated into the State CEQA Guidelines (Section 15065.5), as well.
5.3.1.2 Native American Involvement

Native American involvement in the development review process is addressed by several federal and state laws. The most notable of these are the California Native American Graves Protection and Repatriation Act (2001) and the federal Native American Graves Protection and Repatriation Act (1990). These acts ensure that Native American human remains and cultural items be treated with respect and dignity. In addition, SB 18 details requirements for local agencies to consult with identified California Native American Tribes during the development process.

At the local level, Policy HP-A.5.e of the Historic Preservation Element in the General Plan states that Native American monitors should be included during all phases of the investigation of archaeological resources. This would include surveys, testing, evaluations, data recovery phases, and construction monitoring.

5.7.2 State

5.7.2.1 California Environmental Quality Act

For the purposes of CEQA, a significant historical resource is one that qualifies for the California Register of Historic Resources (CRHR) or is listed in a local historic register or deemed significant in an historical resources survey, as provided under Section 5025.1(g) of the Public Resources Code. A resource that is not listed in or is not determined to be eligible for listing in the CRHR, is not included in a local register or historic resources, or is not deemed significant in an historical resources survey may nonetheless be deemed significant by a CEQA lead agency.

As indicated above, the California criteria (State CEQA Guidelines Section 15065.5) for the registration of significant architectural, archaeological, and historical resources on the CRHR are nearly identical to those for the NRHP. Furthermore, CEQA Section 21083.2(g) defines the criteria for determining the significance of archaeological resources. These criteria include definitions for a "unique" resource, based on its:

- Containing information needed to answer important scientific research questions and that there is a demonstrable public interest in that information.
- Having a special and particular quality such as being the oldest or best available example of its type.
- Being directly associated with a scientifically recognized important prehistoric or historic event or person.

5.7.2.2 California Register of Historic Resources (Public Resources Code Section 5020 et seq.)

Properties listed, or formally designated eligible for listing, on the NRHP are automatically listed on the CRHR as are State Historical Landmarks and Points of Interest. The CRHR also includes properties designated under local ordinances or identified through local historical resource surveys.

5.7.2.3 Native American Burials (Public Resources Code Section 5097 et seq.)

State law addresses the disposition of Native American burials in archaeological sites and protects such remains from disturbance, vandalism, or inadvertent destruction; establishes procedures to be implemented if Native American skeletal remains are discovered during construction of a project; and designates the NAHC to resolve disputes regarding the disposition of such remains. In addition, the Native American Historic Resource Protection Act makes it a misdemeanor punishable by up to a year in jail to deface or destroy an Indian historic or cultural site that is listed or may be eligible for listing in the CRHR.

5.7.3 Local

5.7.3.1 City of San Diego Municipal Code: Historical Resources Regulations

In January 2000, the City's Historical Resources Regulations (Regulations), part of the SDMC (Chapter 14, Article 3, Division 2: Purpose of Historical Resources Regulations or Sections 143.0201-143.0280), were adopted, providing a balance between sound historic preservation principles and the rights of private property owners. The Regulations have been developed to implement applicable local, state, and federal policies and mandates. Included in these are the City's General Plan, CEQA, and Section 106 of the National Historic Preservation Act of 1966. Historical resources, in the context of the City's Regulations, include site improvements, buildings, structures, historic districts, signs, features (including significant trees or other landscaping), places, place names, interior elements and fixtures designated in conjunction with a property, or other objects historical, archaeological, scientific, educational, cultural, architectural, aesthetic, or traditional significance to the citizens of the city. These include structures, buildings, archaeological sites, objects, districts, or landscapes having physical evidence of human activities. These are usually over 45 years old, and they may have been altered or still be in use.

Historic Resources Guidelines are incorporated in the San Diego LDC by reference. These Guidelines set up a Development Review Process to review projects in the City. This process is composed of two aspects: the implementation of the Historical Resources Regulations and the determination of impacts and mitigation under CEQA.

Compliance with the Historical Resources Regulations begins with the determination of the need for a site-specific survey for a project. Section 143.0212(b) of the Regulations requires that historical

resource sensitivity maps be used to identify properties in the City that have a probability of containing archaeological sites. These maps are based on records maintained by the South Coastal Information Center of the California Historic Resources Information System and San Diego Museum of Man, as well as site-specific information in the City's files. If records show an archaeological site exists on or immediately adjacent to a subject property, the City shall require a survey. In general, archaeological surveys are required when the proposed development is on a previously undeveloped parcel, if a known resource is recorded on the parcel or within a one-mile radius, or if a qualified consultant or knowledgeable City staff member recommends it. A historic property (built environment) survey can be required on a project if the properties are over 45 years old and appear to have integrity of setting, design, materials, workmanship, feeling, and association.

Section 143.0212(d) of the Regulations states that if a property-specific survey is required, it shall be conducted according to the Guidelines criteria. Using the survey results and other available applicable information, the City shall determine whether a historical resource exists, whether it is eligible for designation as a designated historical resource, and precisely where it is located.

5.7.3.2 Historical Resources Register

As compared to CEQA, the City provides a broader set of criteria for eligibility for the City's Historical Resources Register. As stated in the City's Historical Resources Guidelines, "Any improvement, building, structure, sign, interior element and fixture, feature, site, place, district, area, or object may be designated as historic by the City of San Diego Historical Resources Board if it meets any of the following criteria:"

- Exemplifies or reflects special elements of the City's, a community's, or a neighborhood's historical, archaeological, cultural, social, economic, political, aesthetic, engineering, landscaping, or architectural development;
- Is identified with persons or events significant in local, State, or national history;
- Embodies distinctive characteristics of a style, type, period, or method of construction or is a valuable example of the use of indigenous materials or craftsmanship;
- Is representative of the notable work of a master builder, designer, architect, engineer, landscape architect, interior designer, artist, or craftsman;
- Is listed or has been determined eligible by National Park Service for listing on the National Register of Historic Places or is listed or has been determined eligible by the State Historic Preservation Office (SHPO) for listing on the State Register of Historical Resources; or
- Is a finite group of resources related to one another in a clearly distinguishable way or is a geographically definable area or neighborhood containing improvements which have a special character, historical interest, or aesthetic value or which represent one or more architectural periods or styles in the history and development of the City.

5.7.3.2 General Plan Historic Preservation Element

The Historic Preservation Element of the General Plan provides guidance on archaeological and historic site preservation in San Diego, including the roles and responsibilities of the Historical Resources Board (HRB), the status of cultural resource surveys, the Mills Act, conservation easements, and other public preservation incentives and strategies. A discussion of criteria used by the HRB to designate landmarks is included, as is a list of recommended steps to strengthen historic preservation in San Diego. The Element sets a series of goals for the City for the preservation of historic resources, and the first of these goals is to preserve significant historical resources. These goals are realized through implementation of policies that encourage the identification and preservation of historical resources.

City General Plan Policies HP-A.1 through HP-A.5 are associated with the overall identification and preservation of historical resources. This includes policies to provide for comprehensive historic resource planning and integration of such plans within City land use plans, such as the proposed CPU being analyzed within this PEIR. These policies also focus on coordinated planning and preservation of tribal resources, promoting the relationship with Kumeyaay/Diegueño tribes. Historic Preservation policies HP-B.1 through HP-B.4 address the benefits of historical preservation planning and the need for incentivizing maintenance, restoration, and rehabilitation of designated historical resources. This is proposed to be completed through a historic preservation sponsorship program and through cultural heritage tourism.

5.8 Biological Resources

5.8.1 Federal

5.8.1.1 Endangered Species Act

The federal Endangered Species Act (ESA) of 1973, as amended (16 U.S.C. 1531 et seq.), provides for listing of endangered and threatened species of plants and animals and designation of critical habitat for listed animal species. The ESA also prohibits all persons subject to U.S. jurisdiction from "taking" endangered species, which includes any harm or harassment. Section 7 of the ESA requires that federal agencies, prior to project approval, consult the U.S. Fish and Wildlife Service (USFWS) and/or the National Marine Fisheries Service to ensure adequate protection of listed species that may be affected by the project.

5.8.1.2 Migratory Bird Treaty Act

The Migratory Bird Treaty Act (MBTA) (16 U.S.C. 703 et seq.) is a federal statute that implements treaties with several countries on the conservation and protection of migratory birds. The list of bird species covered by the MBTA is extensive and is detailed in 50 CFR 10.13. The regulatory definition of "migratory bird" is broad and includes any mutation or hybrid of a listed species, including any part, egg, or nest of such a bird (50 CFR 10.12). Migratory birds are not necessarily federally listed endangered or threatened birds under the ESA. The MBTA, which is enforced by the USFWS, makes it unlawful "by any means or in any manner, to pursue, hunt, take, capture, [or] kill" any migratory

bird or attempt such actions, except as permitted by regulation. The applicable regulations prohibit the take, possession, import, export, transport, sale, purchase, barter, or offering of these activities, except under a valid permit or as permitted in the implementing regulations (50 CFR 21.11).

5.8.1.3 Clean Water Act

The federal Water Pollution Control Act (also known as the Clean Water Act) (33 U.S.C. 1251 et seq.), as amended by the Water Quality Act of 1987 (PL 1000-4), is the major federal legislation governing water quality. The purpose of the Clean Water Act is to "restore and maintain the chemical, physical, and biological integrity of the nation's waters." Discharges into waters of the United States are regulated under Section 404. Waters of the United States include (1) all navigable waters (including all waters subject to the ebb and flow of tides); (2) all interstate waters and wetlands; (3) all other waters, such as intrastate lakes, rivers, streams (including intermittent streams), mudflats, sand flats, wetlands, sloughs, or natural ponds; (4) all impoundments of waters mentioned above; (5) all tributaries to waters mentioned above; (6) the territorial seas; and (7) all wetlands adjacent to waters mentioned above. In California, the State Water Resources Control Board and the nine Regional Water Quality Control Boards are responsible for implementing the Clean Water Act. Important applicable sections of the Clean Water Act are discussed below:

- Section 303 requires states to develop water quality standards for inland surface and ocean waters and submit to the U.S. Environmental Protection Agency for approval. Under Section 303(d), the state is required to list waters that do not meet water quality standards and to develop action plans, called total maximum daily loads, to improve water quality.
- Section 304 provides for water quality standards, criteria, and guidelines.
- Section 401 requires an applicant for any federal permit that proposes an activity that may result in a discharge to waters of the United States to obtain certification from the state that the discharge will comply with other provisions of the Clean Water Act. Certification is provided by the respective Regional Water Quality Control Board.
- Section 402 establishes the National Pollutant Discharge Elimination System, a permitting system for the discharge of any pollutant (except for dredge or fill material) into waters of the United States. The National Pollutant Discharge Elimination System program is administered by the Regional Water Quality Control Board. Conformance with Section 402 is typically addressed in conjunction with water quality certification under Section 401.
- Section 404 provides for issuance of dredge/fill permits by the U.S. Army Corps of Engineers (ACOE). Permits typically include conditions to minimize impacts on water quality. Common conditions include ACOE review and approval of sediment quality analysis before dredging, a detailed pre- and post-construction monitoring plan that includes disposal site monitoring, and required compensation for loss of waters of the United States.

5.8.1.4 U.S. Army Corps of Engineers

The ACOE has primary federal responsibility for administering regulations that concern waters and wetlands in the project area. In this regard, the ACOE acts under two statutory authorities, the Rivers and Harbors Act (33 U.S.C., Sections 9 and 10), which governs specified activities in navigable waters, and the Clean Water Act (Section 404), which governs specified activities in waters of the United States, including wetlands and special aquatic sites. Wetlands and non-wetland waters (e.g., rivers, streams, and natural ponds) are a subset of waters of the United States and receive protection under Section 404 of the Clean Water Act. The ACOE has primary federal responsibility for administering regulations that concern waters and wetlands in the project area under statutory authority of the Clean Water Act (Section 404). In addition, the regulations and policies of various federal agencies mandate that the filling of wetlands be avoided to the maximum extent feasible. The ACOE requires obtaining a permit if a project proposes placing structures within navigable waters and/or alteration of waters of the United States.

5.8.2 State

5.8.2.1 California Endangered Species Act

Similar to the federal ESA, the California ESA of 1970 provides protection to species considered threatened or endangered by the State of California (California Fish and Game Code, Section 2050 et seq.). The California ESA recognizes the importance of threatened and endangered fish, wildlife, and plant species and their habitats, and prohibits the taking of any endangered, threatened, or rare plant and/or animal species unless specifically permitted for education or management purposes.

5.8.2.2 California Fish and Game Code

The California Fish and Game Code regulates the handling and management of the state's fish and wildlife. Most of the code is administered or enforced by the California Department of Fish and Wildlife (CDFW; before January 1, 2013, California Department of Fish and Game). One section of the code generally applies to public infrastructure projects:

 Section 1602 regulates activities that would divert or obstruct the natural flow or substantially change the bed, channel, or bank of any river, stream, or lake that supports fish or wildlife. CDFW has jurisdiction over riparian habitats associated with watercourses. Jurisdictional waters are delineated by the outer edge of riparian vegetation or at the top of the bank of streams or lakes, whichever is wider. CDFW jurisdiction does not include tidal areas or isolated resources.

5.8.2.3 Porter-Cologne Water Quality Act

The Porter-Cologne Water Quality Act of 1969, updated in 2012 (California Water Code, Section 13000 et seq.), established the principal California legal and regulatory framework for water quality control. The act is embodied in the California Water Code. The California Water Code authorizes the State Water Resources Control Board (SWRCB) to implement the provisions of the federal Clean

Water Act. The state of California is divided into nine regions governed by the Regional Water Quality Control Board (RWQCB). The RWQCBs implement and enforce provisions of the California Water Code and Clean Water Act under the oversight of the SWRCB.

5.8.3 Local

5.8.3.1 Multiple Species Conservation Program

The MSCP is a comprehensive habitat conservation planning program for San Diego County. A goal of the MSCP is to preserve a network of habitat and open space, thereby protecting biodiversity. Local jurisdictions, including the City of San Diego, implement their portions of the MSCP through subarea plans, which describe specific implementing mechanisms.

The City of San Diego's MSCP Subarea Plan was approved in March 1997. The MSCP Subarea Plan is a plan and process for the issuance of permits under the federal and state Endangered Species Act and the California Natural Communities Conservation Planning Act of 1991. The primary goal of the MSCP Subarea Plan is to conserve viable populations of sensitive species and to conserve regional biodiversity while allowing for reasonable economic growth.

In July 1997, the City of San Diego signed an Implementing Agreement (IA) with USFWS and CDFW. The IA serves as a binding contract between the City, USFWS, and CDFW that identifies the roles and responsibilities of the parties to implement the MSCP and subarea plan. The agreement became effective on July 17, 1997, and allows the City to issue Incidental Take Authorizations under the provisions of the MSCP. Applicable state and federal permits are still required for wetlands and listed species that are not covered by the MSCP.

a. Multi-Habitat Planning Area

The Multi-Habitat Planning Area (MHPA) is the area within which the permanent MSCP preserve will be assembled and managed for its biological resources. Input from responsible agencies and other interested participants resulted in adoption of the City's MHPA in 1997. The City's MHPA areas are defined by "hard-line" limits, "with limited development permitted based on the development area allowance of the OR-1-2 zone [open space residential zone]".

Private land wholly within the MHPA is allowed only up to 25 percent development in the least sensitive area per the City's MSCP Subarea Plan. Should more than 25 percent development be desired, an MHPA boundary line adjustment may be proposed. The City's MSCP Subarea Plan states that adjustments to the MHPA boundary line are permitted without the need to amend the City's Subarea Plan, provided the boundary adjustment results in an area of equivalent or higher biological value. To meet this standard, the area proposed for addition to the MHPA must meet the six functional equivalency criteria set forth in Section 5.5.2 of the Final MSCP Plan. All MHPA boundary line adjustments require approval by the Wildlife Agencies and the City.

For parcels located outside the MHPA, "there is no limit on the encroachment into sensitive biological resources, with the exception of wetlands, and listed non-covered species' habitat (which are regulated by state and federal agencies) and narrow endemic species." However, "impacts to

sensitive biological resources must be assessed and mitigation, where necessary, must be provided in conformance" with the City's Biological Guidelines.

The MSCP includes management priorities to be undertaken by the City as part of its MSCP implementation requirements. Those actions identified as Priority 1 are required to be implemented by the City as a condition of the MSCP Take Authorization to ensure that covered species are adequately protected. The actions identified as Priority 2 may be undertaken by the City as resources permit.

b. MHPA Land Use Adjacency Guidelines

To address the integrity of the MHPA and mitigate for indirect impacts to the MHPA, guidelines were developed to manage land uses adjacent to the MHPA. The MHPA adjacency guidelines are intended to be incorporated into the Mitigation Monitoring and Reporting Program and/or applicable permits during the development review phase of a proposed project. These guidelines address the issues of drainage, toxics, lighting, noise, barriers, invasive species, brush management, and grading/ development.

c. MSCP Subarea Plan: Overall Management Policies and Directives for Urban Habitat Areas

The Uptown CPU is part of the Urban Habitat Areas of the MHPA. The MSCP plan describes the Urban Habitat Areas of the MHPA and its vision as a network of open and relatively undisturbed canyons containing a full ensemble of native species and providing functional wildlife habitat and movement capability. Management directives to achieve this vision are provided in the MSCP. The general MHPA guidelines and management directives are presented below (as excerpted from Section 1.5.7 of the City of San Diego MSCP Subarea Plan):

- Intense land uses and activities adjacent to and in covered species habitat;
- Dumping, litter, and vandalism;
- Itinerant living quarters;
- Utility, facility and road repair, construction, and maintenance activities;
- Exotic (non-native), invasive plants and animals;
- Urban runoff and water quality.

5.8.3.2 City of San Diego Environmentally Sensitive Lands Regulations

The purpose of the ESL Regulations is to "protect, preserve, and, where damaged restore, the environmentally sensitive lands of San Diego and the viability of the species supported by those lands. These regulations are intended to assure that development occurs in a manner that protects the overall quality of the resources and the natural and topographic character of the area, encourages a sensitive form of development, retains biodiversity and interconnected habitats, maximizes physical and visual public access to and along the shoreline, and reduces hazards due to flooding in specific areas while minimizing the need for construction of flood control facilities. These

regulations are intended to protect the public health, safety, and welfare while employing regulations that are consistent with sound resources conservation principles and the rights of private property owners". ESL Regulations cover sensitive biological resources, including wetlands, within and outside of the coastal zone and MHPA. Future development proposed in accordance with the Uptown CPU would be required to comply with all applicable ESL regulations.

5.8.3.3 City of San Diego General Plan Policies

The City of San Diego General Plan establishes Citywide policies to be cited in conjunction with a Community Plan. The General Plan presents goals and policies for biological resources in the Conservation Element.

5.9 Geologic Conditions

5.9.1 Earthquake Fault Zoning Act (Alquist-Priolo Act)

The State of California Alquist-Priolo Earthquake Fault Zoning Act (1972) was established to mitigate the hazard of surface faulting to structures for human occupancy. Pursuant to the Act, the State Geologist has established regulatory zones (known as Earthquake Fault Zones) around surface traces of active faults. These have been mapped for affected cities, including San Diego. Application for a development permit for any project within a delineated earthquake fault zone shall be accompanied by a geologic report, prepared by a geologist registered in the State of California, that is directed to the problem of potential surface fault displacement through a project site.

5.9.2 City of San Diego Seismic Safety Study

The San Diego Seismic Safety Study includes geologic hazards and fault maps of the City. Areas of the City are identified by geologic hazard category, which reflect the geologic hazard type and related risks. These are generalized maps, and site-specific geologic/geotechnical investigations may be necessary for proposed development or construction. Land Development Code Section 145.1803 describes when a geotechnical investigation is required, and City of San Diego Development Services Information Bulletin 515 describes the minimum submittal requirements for geotechnical and geological reports that may be required for development permits, subdivision approvals, or grading permits.

5.9.3 City of San Diego General Plan Policies

The City's General Plan presents goals and policies for geologic and soil safety in the Public Facilities, Services, and Safety Element. Relevant excerpts from this element are included below.

Policy PF-Q.1. Protect public health and safety through the application of effective seismic, geologic and structural considerations.

a. Ensure that current and future community planning and other specific land use planning studies continue to include consideration of seismic and other geologic hazards. This

information should be disclosed, when applicable, in the California Environmental Quality Act (CEQA) document accompanying a discretionary action.

- b. Maintain updated Citywide maps showing faults, geologic hazards, and land use capabilities, and related studies used to determine suitable land uses.
- c. Require the submission of geologic and seismic reports, as well as soils engineering reports, in relation to applications for land development permits whenever seismic or geologic problems are suspected.
- d. Utilize the findings of a beach and bluff erosion survey to determine the appropriate rate and amount of coastline modification permissible in the City.
- e. Coordinate with other jurisdictions to establish and maintain a geologic "data bank" for the San Diego area.
- f. Regularly review local lifeline utility systems to ascertain their vulnerability to disruption caused by seismic or geologic hazards and implement measures to reduce any vulnerability.
- g. Adhere to state laws pertaining to seismic and geologic hazards.

Policy PF-Q.2. Maintain or improve integrity of structures to protect residents and preserve communities.

- Abate structures that present seismic or structural hazards with consideration of the desirability of preserving historical and unique structures and their architectural appendages, special geologic and soils hazards, and the socio- economic consequences of the attendant relocation and housing programs.
- Continue to consult with qualified geologists and seismologists to review geologic and seismic studies submitted to the City as project requirements.
- Support legislation that would empower local governing bodies to require structural inspections for all existing pre-Riley Act (1933) buildings, and any necessary remedial work to be completed within a reasonable time.

5.10 Paleontological Resources

Under California law, paleontological resources are protected by CEQA; the CCR, Title 14, Division 3, Chapter 1, Sections 4307 and 4309; and Public Resources Code Section 5097.5. Pursuant to Section 15065 of the CEQA Guidelines (CCR Sections 15000–15387), a lead agency must find that a project would have a significant effect on the environment when the project has the potential to eliminate important examples of the major periods of California prehistory, including significant paleontological resources. The City's Paleontological Guidelines (July 2002) and Significance Determination Thresholds (January 2011) are used to make this determination.

5.11 Hydrology/Water Quality

There are federal, state, and local regulations that impose requirements on new development for erosion control, control of runoff contaminants, and control of direct discharge of pollutants that impact water quality. These laws, regulations, and standards are summarized below.

5.11.1 Federal

5.11.1.1 Clean Water Act

The Clean Water Act (33 U.S.C. §1251 et seq.) (1972) is the primary federal law that protects the nation's waters, including lakes, rivers, aquifers, and coastal areas. The Clean Water Act established basic guidelines for regulating discharges of pollutants into the waters of the U.S. and requires that states adopt water quality standards to protect public health, enhance the quality of water resources, and ensure implementation of the Clean Water Act.

Section 401 of the Clean Water Act requires that any applicant for a Federal permit to conduct any activity, including the construction or operation of a facility which may result in the discharge of any pollutant, must obtain certification from the state. Section 402 of the Clean Water Act established the National Pollutant Discharge Elimination System (NPDES) to regulate the discharge of pollutants from point sources, and Section 404 established a permit program to regulate the discharge of dredged material into Waters of the U.S. In California, the SWRCB and RWQCBs administer the NPDES permitting programs and are responsible for developing waste discharge requirements. The local RWQCB is responsible for developing waste discharge requirements specific to its jurisdiction. General waste discharge requirements that may apply to projects or recommendations contained within the Plans include the SWRCB Construction General Permit and Industrial General Permit and the regional Municipal Separate Storm Sewer System (MS4) Permit administered by the RWQCB.

Under section 303(d) of the Clean Water Act, states, territories, and authorized tribes are required to develop lists of impaired waters. These are waters that are too polluted or otherwise degraded to meet the water quality standards set by states, territories, or authorized tribes. The law requires that these jurisdictions establish priority rankings for waters on the lists and develop Total Maximum Daily Loads for these waters. A Total Maximum Daily Load is a calculation of the maximum amount of a pollutant that a waterbody can receive and still safely meet water quality standards.

5.11.1.2 Executive Order 11988, Floodplain Management

The major requirements of this federal order are to avoid support of floodplain development; to prevent uneconomic, hazardous, or incompatible use of floodplains; to protect and preserve the natural and beneficial floodplain values; and to be consistent with the standards and criteria of the National Flood Insurance Program. The basic tools for regulating construction in potentially hazardous floodplain areas are local zoning techniques. Proper floodplain zoning can be beneficial in the preservation of open space, retention of floodplains as groundwater recharge areas, and directing of development to less flood-prone areas.

5.11.2 State

5.11.2.1 California Department of Fish and Wildlife Code – Streambed Alteration Program

CDFW regulates activities that would divert or obstruct the natural flow or substantially change the bed, channel, or bank of any river, stream, or lake that supports fish or wildlife. CDFW has jurisdiction over riparian habitats (e.g., southern willow scrub) associated with watercourses. CDFW jurisdictional resources are delineated by the outer edge of riparian vegetation or at the top of the bank of streams or lakes, whichever is wider. A Streambed Alteration Agreement is required for a project that would impact CDFW jurisdictional resources. The Agreement with CDFW typically requires mitigation in the form of on-site, off-site, or in-lieu fee mitigation, or combination of all three forms.

5.11.2.2 Porter-Cologne Water Quality Control Act

Refer to Section 5.8 for discussion of the Porter-Cologne Water Quality Control Act.

5.11.2.3 San Diego Regional Water Quality Control Board (Water Board) Order No. R9-2013-0001, as amended by Order No. R9-2015-0001 and Order No. R9-2015-0100, NPDES Permit No. CAS0109266

Under the authority of the Clean Water Act amendments and Federal NPDES Permit regulations, the Water Board issued this order to the Copermittees consisting of San Diego County, the 18 cities within San Diego County, the Port of San Diego, and the San Diego Regional Airport Authority. This order requires that all jurisdictions within the San Diego region prepare Jurisdictional Urban Runoff Management Plans. Each of these jurisdictional plans must contain a component addressing construction activities and a component addressing existing development. The subsequent amendments expanded coverage to portions of Orange County and Riverside County within the San Diego Region (Region 9) and made other modifications.

5.11.3 Local

5.11.3.1 Water Quality Control Plan for the San Diego Basin

The San Diego Basin encompasses approximately 3,900 square miles, including most of San Diego County and portions of southwestern Riverside and Orange counties. The basin is composed of 11 major Hydrologic Units, 54 Hydrologic Areas, and 147 Hydrologic Sub Areas, extending from Laguna Beach southerly to the U.S./Mexico border. Drainage from higher elevations in the east flow to the west, ultimately into the Pacific Ocean. The RWQCB prepared the Basin Plan, which defines existing and potential beneficial uses and water quality objectives for coastal waters, groundwater, surface

waters, imported surface waters, and reclaimed waters in the basin. Water quality objectives seek to protect the most sensitive of the beneficial uses designated for a specific water body.

5.11.3.2 City of San Diego Jurisdictional Runoff Management Program

This document is a total account of how the City of San Diego plans to protect and improve the water quality of rivers, bays and the ocean in the region in compliance with the Water Board permit referenced above. The document describes how the City incorporates storm water best management practices into land use planning, development review and permitting, City capital improvement program project planning and design, and the execution of construction contracts.

5.11.3.3 Water Quality Improvement Plans

The MS4 Permit also requires development of Water Quality Improvement Plans (WQIPs) that guide the Copermittees' jurisdictional runoff management programs towards achieving improved water quality in MS4 discharges and receiving waters. <u>The San Diego River Watershed Management Area</u> <u>WQIP applies to the portion of Uptown Community draining to the San Diego River, while the San</u> <u>Diego Bay Watershed Management Area WQIP applies to the remainder of the community draining</u> <u>to San Diego Bay.</u> The WQIPs further the Clean Water Act's objectives to protect, preserve, enhance, and restore the water quality and designated beneficial uses of waters of the state. The requirement sets forth a collaborative and adaptive planning and management area and implements strategies through the jurisdictional runoff management programs of the respective jurisdictions.

5.11.3.4 Local Drainage Design Manual

Chapter 14, Article 2, Division 2 of the SDMC outlines Storm Water Runoff and Drainage Regulations which apply to all development in the City, regardless of whether or not a development permit or other approval is required. In addition, drainage design policies and procedures are provided in the City's Drainage Design Manual (which is incorporated in the Land Development Manual as Appendix B). The Drainage Design Manual provides a guide for designing drainage, and drainage-related facilities for developments within the City.

5.11.3.5 Storm Water Standards Manual

The City's current Storm Water Standards Manual provides information to project applicants on how to comply with the permanent and construction storm water quality requirements in the City. Significant elements of the Storm Water Standards Manual include:

- 1. Low Impact Develop (LID) Best Management Practice (BMP) Requirements
- 2. Source Control BMPs
- 3. BMPs Applicable to Individual Priority Development Project Categories
- 4. Treatment Control BMPs

Although the footprint of the LID BMPs can often be fit into planned landscaping features, this requires early planning to ensure that the features are located in places where they can intercept the drainage and safely store the water without adverse effects to adjacent slopes, structures, roadways, or other features. The Storm Water Standards Manual also addresses "Hydromodification – Limitations on Increases of Runoff Discharge Rates and Durations." Hydromodification management requirements would dictate design elements in locations where downstream channels are susceptible to erosion from increases in storm water runoff discharge rates and durations. Future development projects proposed within areas draining to San Diego Bay would typically be exempt from hydromodification management requirements because of the location and hardened drainage systems. Projects discharging into underground storm drains discharging directly to bays or the ocean are exempt.

The Storm Water Standards Manual also provides minimum requirements for construction site management, inspection, and maintenance of construction BMPs; monitoring of the weather and implementation of emergency plans as needed; and provides minimum performance standards, including: pollution prevention measures so that there would be no measurable increase of pollution (including sediment) in runoff from the site, no slope erosion, water velocity moving off-site must not be greater than pre-construction levels, and preserve natural hydraulic features and riparian buffers where possible. The City's Storm Water Standards Manual was updated in 2016 for consistency with the Regional Best Management Practices Design Manual.

5.11.3.6 City of San Diego General Plan

The City's General Plan presents goals and policies for storm water infrastructure in the Public Facilities, Services, and Safety Element, and presents goals and policies for open space (including floodplain management) and urban runoff management in the Conservation Element.

5.12 Public Services and Facilities

The City requires payment of Development Impact Fees (DIF) to collect a proportional fair-share cost of capital improvements needed to offset the impact of the development (City of San Diego Municipal Code Section 142.0640). DIF fees are based on community specific financing plans completed when Community Plans are updated. Financing plans were formerly known as Public Facilities Financing Plans (PFFP) and are now referred to as Impact Fee Studies (IFS).

The General Plan Public Facilities Element includes a number of policies that address financing of public facilities and specifies that IFS should be completed concurrent with preparation of Community Plan updates, should set community-level priorities for facility financing, and ensure new development pays its proportional fair-share of public facilities costs through payment of DIFs. Facility types that are eligible for DIF funding include transportation, storm drains, parks and recreation, fire-rescue, police, and libraries.

5.12.1 Police

As specified in the City General Plan, Public Facilities Element, Policy PF-E.2, the City goal is to maintain average response time goals as development and population growth occurs. Average response time guidelines are as follows:

- Priority E Calls (imminent threat to life) within seven minutes.
- Priority 1 Calls (serious crimes in progress) within 12 minutes.
- Priority 2 Calls (less serious crimes with no threat to life) within 30 minutes.
- Priority 3 Calls (minor crimes/requests that are not urgent) within 90 minutes.
- Priority 4 Calls (minor requests for police service) within 90 minutes.

5.12.2 Parks

The General Plan provides standards for population–based parks and Recreation Facilities which include Recreation Centers and Aquatic Complexes. The standard for population-based parks is 2.8 useable acres per 1,000 residents, which can be achieved through a combination of neighborhood and community parks and park equivalencies. The standard for Recreation Center is a minimum of 17,000 square feet per recreation center or a population of 25,000. The standard for Aquatic Complex is one per 50,000 people or within approximately six miles.

5.12.3 Fire

The Fire-Rescue Department has an active program that promotes the clearing of canyon vegetation away from structures in accordance with Section 142.0412 of the San Diego Municipal Code and the San Diego Fire-Rescue Department's Canyon Fire Safety guidelines and policies related to brush management. The City thins brush on city property within 100 horizontal feet of a previously conforming structure unless a site-specific report, which indicates that a greater distance is necessary, is approved by the San Diego Fire-Rescue Department (per SDMC Section 142.0412(i) or a previously recorded entitlement requires a width more or less than the standard 100 feet. Other fire prevention measures include adopting safety codes and an aggressive brush management program. Citywide fire service goals, policies and standards are located in the Public Facilities, Services, and Safety Element of the General Plan and the Fire-Rescue Services Department's Fire Service Standards of Response Coverage Deployment Study.

Response time standards are provided in the General Plan Public Facilities, Services and Safety Element and summarized below:

a. To treat medical patients and control small fires, the first-due unit should arrive within 7.5 minutes, 90 percent of the time from the receipt of the 911 call in fire dispatch. This equates to one-minute dispatch time, 1.5 minutes company turnout time and 5-minute drive time in the most populated areas.

- b. To provide an effective response force for serious emergencies, a multiple-unit response of at least 17 personnel should arrive within 10.5 minutes from the time of 911-call receipt in fire dispatch, 90 percent of the time.
 - This response is designed to confine fires near the room of origin, to stop wildland fires to under 3 acres when noticed promptly, and to treat up to 5 medical patients at once.
 - This equates to 1-minnute dispatch time, 1.5 minutes company turnout time and 8minute drive time spacing for multiple units in the most populated areas.

To direct fire station location timing and crew size planning as the community grows, fire unit deployment performance measures are established based on population density zones and are shown in Table 5-6, below:

Table 5-6							
Deployment Measures to Address Future Growth by Population Density per Square Mile							
	Structure Fire	Structure Fire	Structure Fire	Wildfires			
	Urban Area	Rural Area	Remote Area	Populated Areas			
	>1,000-people/	1,000 to 500	500 to 50	Permanent open			
	sq. mi.	people/sq. mi.	people/sq. mi. *	space areas			
1 st Due Travel Time	5	12	20	10			
Total Reflex Time	7.5	14.5	22.5	12.5			
1 st Alarm Travel	0	16	24	15			
Time	0	10	24	15			
1 st Alarm Total	10 F	10 E		17 5			
Reflex 10.5 18.5 26.5 17.5							
Notes: Reflect time is the total time from receipt of a 9-1-1 call to arrival of the required number of							
emergency units							
SOURCE: City of San Diego General Plan 2008.							

The following population based performance measures are used to plan for needed facilities. Where more than one square mile is not populated at similar densities, and/or a contiguous area with different zoning types aggregates into a population "cluster," these measures guide the determination of response time measures (Table 5-7) and the need for fire stations:

Table 5-7						
Deployment Measu	ires to Address Future Growth b	y Population Clusters				
Area	Aggregate Population	First-Due Unit Travel Time Goal				
Metropolitan	> 200,000 people	4 minutes				
Urban-Suburban	< 200,000 people	5 minutes				
Rural	500 - 1,000 people	12 minutes				
Remote< 500> 15 minutes						
SOURCE: City of San Diego General Plan 2008.						

5.13 Public Utilities

5.13.1 Water Supply

SB 610 requires water suppliers to prepare a Water Supply Assessment (WSA) report for inclusion by land use agencies during the CEQA process for new developments subject to SB 221. SB 221 requires water suppliers to prepare written verification that sufficient water supplies are planned to be available prior to approval of large-scale subdivision of land under the State Subdivision Map Act. Large-scale projects include residential development of more than 500 units, shopping centers or businesses employing more than 1,000 people, shopping centers or businesses having more than 500,000 square feet of floor space, commercial office buildings employing more than 1,000 people, and/or commercial buildings having more than 250,000 square feet of floor space or occupying more than 40 acres of land. SB 221 and SB 610 went into effect January 2002 with the intention of linking water supply availability to land use planning by cities and counties.

5.13.2 Wastewater

Council Policy 400-13 identifies the need to provide maintenance access to all sewers in order to reduce the potential for spills. The policy requires that environmental impacts from access paths in environmentally sensitive areas should be minimized to the maximum extent possible through the use of sensitive access path design, canyon-proficient maintenance vehicles, and preparation of plans that dictate routine maintenance and emergency access procedures.

Council Policy 400-14 outlines a program to evaluate the potential to redirect sewage flow out of canyons and environmentally sensitive areas to an existing or proposed sewer facility located in City streets or other accessible locations. The policy includes an evaluation procedure that requires both a physical evaluation and a cost-benefit analysis. Based on the analysis, if redirection of flow outside the canyon is found to be infeasible, a Long-Term Maintenance and Emergency Access Plan is required. The plan would be specific to the canyon evaluated, and would prescribe long term access locations for routine maintenance and emergency repairs along with standard operating procedures identifying cleaning methods and inspection frequency.

The City's Sewer Design Guide sets forth criteria to be used for the design of sewer systems which may consist of pump stations, gravity sewers, force mains, and related appurtenances. It includes criteria for determining capacity and sizing of pump stations, gravity sewers and force mains, alignment of gravity sewers and force mains, estimating wastewater flow rates, design of bridge crossings, and corrosion control requirements.

5.13.3 Water Distribution

The City's Water Facility Design Guidelines identify general planning, predesign, and design details and approaches to be use for water infrastructure. The guidelines provide uniformity in key concepts, equipment types, and construction materials on facilities built under the Water CIP. These design Guidelines assist in providing professionally sound, efficient, uniform, and workable facilities; whether pipelines, pressure control facilities, pumping stations, or storage facilities.

5.13.4 Communication Facilities

City Council Policy 600-43 established a set of comprehensive guidelines for the review and processing of applications for the placement and design of Wireless Communication Facilities in accordance with the City of San Diego land use regulations. These guidelines are intended to prescribe clear, reasonable, and predictable criteria to assess and process applications in a consistent and expeditious manner, while reducing visual and land use impacts associated with Wireless Communication Facilities. For applicants seeking placement of a Wireless Communication Facility on City-owned land, this policy should be used in conjunction with applicable Council Policies and Land Development Code section 141.0420.

5.13.5 Solid Waste

The California Legislature passed AB 939 to address landfill capacity and solid waste concerns in 1989. The Integrated Waste Management Act mandated that all cities reduce waste disposed in landfills from generators within their borders by 50 percent by the year 2000. The law also required local governments to prepare Source Reduction and Recycling Elements detailing how these reductions would be achieved. In 2011, the State enacted AB 341 which established a policy goal for California of 75 percent recycling, composting, or source reduction of solid waste by 2020. In July 2012, the City updated the Recycling Ordinance to lower the exemption threshold for required recycling, thereby requiring all privately serviced businesses, commercial/institutional facilities, apartments, and condominiums generating four or more cubic yards of trash per week to recycle. The City is currently at a 67 percent diversion rate (City of San Diego 2016). Pursuant to the City's Significance Determination Thresholds, any land development project that may generate approximately 60 tons of waste or more during construction and/or operation is required to prepare a project-specific Waste Management Plan (WMP) to address disposal of waste generated during shot-term project construction and long-term post-construction operation. The WMP is required to identify how the project would reduce waste and achieve target reduction goals.

5.14 Health and Safety

Hazardous materials and hazardous wastes are extensively regulated by federal, state, local regulations, with the major objective of protecting public health and the environment. In general, these regulations provide definitions of hazardous substances; identify responsible parties; establish reporting requirements; set guidelines for handling, storage, transport, remediation, and disposal of hazardous materials and wastes; and require health and safety provisions for both workers and the public, such as emergency response and worker training programs. The major regulations relevant to the Uptown CPU area are summarized below.

5.14.1 State

5.14.1.1 California Code of Regulations Title 22

The CCR Title 22 provides the following definition of hazardous materials:

A hazardous material is a substance or combination of substances which, because of its quantity, concentration or physical, chemical, or infectious characteristics, may either (1) cause or significantly contribute to an increase in mortality or an increase in serious, irreversible or incapacitating irreversible illness; or (2) pose a substantial present or potential hazard to human health and safety, or the environment when improperly treated, stored, transported or disposed of. Hazardous materials include waste that has been abandoned, discarded, or recycled on the property and as a result represents a continuing hazard as the development is proposed. Hazardous materials also include any contaminated soil or groundwater.

Title 22 also provides standards applicable to generators and transporters or hazardous wastes, as well as standards for operators or hazardous waste transfer facilities, among other regulations.

5.14.1.2 Hazardous Materials Release Response Plans and Inventory

Two programs in the California Health and Safety Code (H&SC) Chapter 6.95 are directly applicable to the CEQA issue of risk due to hazardous substance release. In San Diego County, these two programs are referred to as the Hazardous Materials Business Plan (HMBP) program and the California Accidental Releases (CalARP) program. The County of San Diego Department of Environmental Health (DEH) is responsible for the implementation of the HMBP program and the CalARP program in San Diego County. The HMBP and CalARP programs provide threshold quantities for regulated hazards substances. When the indicated quantities are exceeded, an HMBP or Risk Management Plan is required pursuant to the regulations. Congress requires EPA Region 9 to make RMP information available to the public through the EPA's Envirofacts Data Warehouse. The Envirofacts Data Warehouse is considered the single point of access to select EPA environmental data. California H&SC Section 25270, Aboveground Petroleum Storage Act requires registration and spill prevention programs for above ground storage tanks that store petroleum. In some cases, ASTs for petroleum may be subject to groundwater monitoring programs that are implemented by the RWQCBs and the SWRCB.

5.14.1.3 Emergency Response to Hazardous Materials Incidents

California has developed an emergency response plan to coordinate emergency services provided by federal, state, and local governments and private agencies. Response to hazardous material incidents is one part of this plan. The plan is managed by the California Emergency Management Agency, which coordinates the responses of other agencies, including California EPA, the California Highway Patrol, CDFW, and RWQCB.

5.14.1.4 California Department of Toxic Substances Control

Within California EPA, the California Department of Toxic Substances Control (DTSC) has primary regulatory responsibility, with delegation of enforcement to local jurisdictions that enter into agreements with the state agency, for the management of hazardous materials and the generation, transport and disposal of hazardous waste under the authority of the Hazardous Waste Control Law.

Since August 1, 1992, the DTSC has been authorized to implement the state's hazardous waste management program for the California EPA.

The DTSC is responsible for compiling a list of hazardous materials site pursuant to Government Code Section 65962.5, which includes five categories:

- Hazardous waste facilities subject to corrective action pursuant to Section 25187.5 of the health and safety code;
- Land designated as "hazardous waste property" or "border zone property;"
- Properties with hazardous waste disposals on public land;
- Hazardous substance release sites selected for (and subject to) a response action; and
- Sites included in the Abandoned Site Assessment Program.

5.14.2 Local

5.14.2.1 County of San Diego Department of Environmental Health

The Hazardous Materials Division (HMD) of DEH regulates hazardous waste and tiered permitting, USTs, aboveground petroleum storage and risk management plans, hazardous materials business plans and chemical inventory, risk management plans, and medical waste. The HMD's goal is "to protect human health and the environment by ensuring that hazardous materials, hazardous waste, medical waste, and underground storage tanks are properly managed" (County of San Diego 2016).

5.14.2.2 County of San Diego Consolidated Fire Code

The San Diego region is unique within California in having fire protection districts within its boundaries. For the purposes of prescribing regulations in the unincorporated area of San Diego County, the applicable fire code is known as the County Fire Code and includes the Consolidated Fire Code and adopts, by reference, the most current version of the California Fire Code (CCR T- 24 part 9). The Consolidated Fire Code consists of local Fire Protection District ordinances that have modified the Fire Code portion of the State Building Standards Code and any County of San Diego modification to the Fire Districts' amendments. The purpose of the Code is for the protection of the public health and safety, which includes permit and inspection requirements for the installation, alteration, or repair of new and existing fire protection systems, and penalties for violations of the Code. The Code provides the minimum requirements for access, water supply and distribution, construction type, fire protection systems, and vegetation management. Additionally, the Fire Code regulates hazardous materials and associated measures to ensure that public health and safety are protected from incidents to hazardous substance release.

5.14.2.3 California EPA's Unified Program

In 1993, SB 1082 gave California EPA the authority and responsibility to establish a unified

hazardous waste and hazardous materials management and regulatory program, commonly referred to as the Unified Program. The purpose of this program is to consolidate and coordinate six different hazardous materials and hazardous waste programs, and to ensure that they are consistently implemented throughout the state. California EPA oversees the Unified Program with support from the DTSC, RWQCBs, the San Diego County Office of Emergency Services (OES), and the State Fire Marshal.

State law requires county and local agencies to implement the Unified Program. The agency in charge of implementing the program is called the Certified Unified Program Agency (CUPA). The County of San Diego DEH, Hazardous Materials Division is the designated CUPA for the county. In addition to the CUPA, other local agencies help to implement the Unified Program. These agencies are called Participatory Agencies. The HMD is the Participatory Agency for San Diego County.

5.14.2.4 San Diego County Multi-Jurisdictional Hazard Mitigation Plan

Long-term prevention, mitigation efforts and risk-based preparedness for specific hazards within the city are addressed as a part of the 2010 San Diego County Multi-Jurisdictional Hazard Mitigation Plan (HAZMIT), which was finalized in February 2010. The HAZMIT identifies specific risks for San Diego County and provides methods to help minimize damage caused by natural and man-made disasters. The final list of hazards profiled for San Diego County was determined as wildfire/structure fire, flood, coastal storms/erosion/tsunami, earthquake/liquefaction, rain-induced landslide, dam failure, hazardous materials incidents, nuclear materials release, and terrorism. The plan is currently being reviewed and revised to reflect changes to both the hazards threatening San Diego County as well as the programs in place to minimize or eliminate those hazards. This revision will include an evaluation of the impact climate change is having on the natural hazards facing San Diego. The San Diego County OES is responsible for coordinating with local jurisdictions and participating agencies to monitor, evaluate, and update the HAZMIT as necessary.

5.14.2.5 San Diego County Operational Area Emergency Plan

The 2010 San Diego County Operational Area Emergency Plan describes a comprehensive emergency management system which provides for a planned response to disaster situations associated with natural disasters, technological incidents, terrorism and nuclear-related incidents. It delineates operational concepts relating to various emergency situations, identifies components of the Emergency Management Organization, and describes the overall responsibilities for protecting life and property and assuring the overall well-being of the population. The plan also identifies the sources of outside support that might be provided (through mutual aid and specific statutory authorities) by other jurisdictions, state and federal agencies and the private sector.

5.14.2.6 City of San Diego General Plan

The City's General Plan presents goals and policies relating to hazardous materials and disaster preparedness in the Public Facilities, Services, and Safety Element.

5.14.2.7 Brush Management Regulations

The City of San Diego Municipal Code includes general hazardous materials regulations (Sections 42.0801, 42.0901, and 54.0701) as well as regulations regarding specific hazardous materials such as explosives (Section 55.3301).

The City of San Diego Municipal Code includes regulations pertaining to brush management (Section 142.0412) and construction materials for development near open space (Chapter 14, Article 5) to minimize fire risk. Brush management is required in all base zones on publicly or privately owned premises that are within 100 feet of a structure and contain native or naturalized vegetation. The City requires submittal of Brush Management Plans for all new development, which are intended to reduce the risk of significant loss, injury, or death involving wildland fires. Unless otherwise approved by the City Fire Marshal, the brush management plans for all future development would consist of two separate and distinct zones as follows:

- Zone One would consist of the area adjacent to structures where flammable materials would be minimized through the use of pavement and/or permanently irrigated ornamental landscape plantings. This zone would not be allowed on slopes with a gradient greater than 4:1.
- Zone Two would consist of the area between Zone One and any area of native or nonirrigated vegetation and shall consist of thinned native or naturalized vegetation.

6

Chapter 6.0 Environmental Analysis

The following sections in Chapter 6.0 analyze the potential environmental impacts that may occur as a result of implementation of the proposed Uptown Community Plan Update (CPU) and associated discretionary actions. The environmental issues addressed in this chapter include the following:

- Land Use
- Visual Effects and Neighborhood Character
- Transportation/Circulation
- Air Quality
- Greenhouse Gas Emissions
- Noise

- Historical Resources
- Biological Resources
- Geologic Conditions
- Paleontological Resources
- Hydrology/Water Quality
- Public Service and Facilities
- Public Utilities
- Health and Safety

Each issue analysis section is formatted to include a description of existing conditions (or a reference to Chapter 2.0 for existing conditions), the criteria for the determination of impact significance, evaluation of potential project impacts including cumulative impacts, mitigation measures if applicable, and conclusion of significance after mitigation for impacts identified as requiring mitigation.

6.1 Land Use

This section discusses existing land use and the consistency of the proposed Uptown CPU and associated discretionary actions with applicable plans and regulations. This section analyzes the potential that implementation of the Uptown CPU would permit designation or intensity of use that have indirect or secondary environmental impacts.

6.1.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

6.1.2 Significance Determination Thresholds

The determination of significance regarding any inconsistency with development regulations or plan policies is evaluated in terms of the potential for the inconsistency to result in environmental impacts considered significant under California Environmental Quality Act (CEQA). Thresholds used to evaluate potential impacts related to land use are based on applicable criteria in the CEQA Guidelines Appendix G and the City of San Diego CEQA Significance Determination Thresholds (2011). Thresholds are modified from the City's CEQA Significance Determination Thresholds to reflect the programmatic analysis for the proposed Uptown CPU. A significant land use impact would occur if implementation of the proposed Uptown CPU and other associated discretionary approvals would:

- 1) Conflict with the environmental goals, objectives, or guidelines of a General Plan or Community Plan or other applicable land use plan or regulation, and as a result, cause an indirect or secondary environmental impact;
- Lead to development or conversion of General Plan or Community Plan designated open space or prime farmland to a more intensive land use, resulting in a physical division of the community;
- 3) Conflict with the provisions of the City's Multiple Species Conservation Program (MSCP) Subarea Plan or other approved local, regional, or state habitat conservation plan; or
- 4) Result in land uses which are not compatible with an adopted Airport Land Use Compatibility Plan (ALUCP).

Issues addressed in the City's CEQA Significance Thresholds that are not addressed in this document include whether the project would increase the base flood elevation for upstream properties, or construct in a Special Flood Hazard Area (SFHA) or floodplain/wetland buffer zone. During initial

project scoping, it was determined that implementation of the proposed Uptown CPU and associated discretionary actions would not result in significant impacts related to increases in the base flood elevation or construction in an SFHA or floodplain/wetland buffer zone because existing Land Development Code regulations would adequately address potential impacts related to grading within a SFHA (Municipal Code, Chapter 14, Article 2, Division 2 Drainage Regulations and Chapter 14, Article 3, Division 1 Environmentally Sensitive Lands Regulations). Thus, there is no further discussion of this issue area.

6.1.3 Impact Analysis

Issue 1 Conflicts with Applicable Plans

Would the proposed project conflict with the environmental goals, objectives, or guidelines of a General Plan or Community Plan or other applicable land use plan or regulation and as a result, cause an indirect or secondary environmental impact?

a. City Of San Diego General Plan

The proposed Uptown CPU and other associated discretionary actions are intended to further express General Plan policies in the Uptown CPU area through the provision of site-specific recommendations that implement Citywide goals and policies, address community needs, and guide zoning. The proposed Uptown CPU and General Plan work together to establish the framework for growth and development for Uptown. The proposed Uptown CPU contains nine elements, each providing neighborhood-specific goals and policies. These goals and policies are consistent with development design guidelines, other mobility and civic guidelines, incentives, and programs in accordance with the general goals stated in the General Plan. Table 6.1-1, provides a comprehensive list of all proposed Uptown CPU policies for each element to be referenced in the following land use analysis. Additionally, a description of the proposed land uses and allowed densities are included in Table 6.1-2; locations of proposed land uses are shown in Figures 3-1 through 3-4 of Chapter 3.0, Project Description.

Table 6.1-1								
	Proposed Uptown CPU Policies Related to Land Use							
Policy	Policy Description							
Land Use Element								
Land Use Dist	tribution							
<u>LU-1.1</u>	Provide a variety of land use types to accommodate both affordable and market rate housing							
	and commercial opportunities							
<u>LU-1.2</u>	Encourage mixed-use infill development along commercial corridors and in the core village							
Desidential	<u>centers.</u>							
	Drovide a diverse mix of housing types and forms consistent with allowable densities							
LU-2.1	Enable rental and ownership opportunities in all types of bousing including alternative							
LO-2.2	housing units such as companion units live/ work studios and shonkeeper units							
111-23	Develop adequate housing for those with special needs such as the elderly handicapped							
20 2.5	persons, those who need nursing care, low income, and homeless persons. Consideration							
	should be given to accessibility and proximity to transit stops, public facilities, public spaces,							
	safe and pedestrian-oriented streets, etc.							
LU-2.4	Preserve existing single-family homes and neighborhoods as a distinct housing choice as well							
	as for their contribution to the historic character of the community.							
LU-2.5	Preserve and enhance the special character of specific, well-defined, low-density							
	neighborhoods from encroachment by incompatible, higher density residential or							
	commercial development.							
LU-2.6	Locate medium and high density residential development in selected areas with adequate							
	design controls provided to ensure compatibility with existing lower density development.							
LU-2.7	On upper floors as part of mixed use development in commercial areas:							
	 On upper moors as part or mixed use development in commercial areas; Adjacent to commercial areas; 							
	 Aujacent to confine clai aleas, Near or within transit and higher volume traffic corridors 							
111-2.8	Preserve and provide incentives for mixed residential/commercial development at							
20 2.0	appropriate locations.							
LU-2.9	Locate higher density residential development in appropriate areas that are situated to							
	promote safer and livelier commercial districts.							
LU-2.10	Ensure adequate transition between commercial/mixed-use and residential uses.and							
	buffering between potentially incompatible uses.							
LU-2.11	Design and enforce stricter controls and location criteria on Conditional Use Permits in							
	residential neighborhoods to minimize nuisances generated by nonresidential uses, such as							
	offices in historic structures.							
LU-2.12	Maintain the low-scale, intensity multi-family residential designations for parcels and the end							
	of blocks facing the east-west running streets such as Meade Avenue, Monroe Avenue, and							
Madison Avenue to maintain the traditional development pattern.								
	Encourage new mixed use development and active commercial uses on Fourth and Fifth							
LU-2.13	Avenues especially south of Fir Street to create pedestrian activity along these corridors and							
	establish connections between Uptown and Downtown							
LU-2.14	Incorporate office uses as part of mixed-used developments. Locate them to create a buffer							
	between single-family neighborhoods and active evening uses such as bars, restaurants, and							
	breweries. Use retail and office uses as a means of creating a buffer between active evening							
	-uses such as bars, restaurants, breweries, etc. and single-family neighborhoods.							
LU-2.15	Support the intensification on of existing hospital uses on institutionally designated areas							
	rather than expanding into residential or commercial areas. Expansion should not occur							
	beyond institutional and office designations of the land use map.							

	Table 6.1-1					
	Proposed Uptown CPU Policies Related to Land Use					
Policy	Description					
Institutional						
LU-2.16	Evaluate proposed institutional uses for appropriate development intensity and effects on visual quality and neighborhood character. Additional factors, such as those related to mobility, noise and parking demand should also be evaluated as needed.					
LU-2.17	Consider the reuse of the San Diego Unified School District Education Center at Park Boulevard and Normal Street which could include medium-high residential development, the potential for mixed-use development, public space, and/or the rehabilitation and reuse of the buildings such as the Teachers Training Annex. That includes a mixed-use development with medium-high density and public and private space					
LU-2.18	Ensure that new office development with the Medical Complex neighborhood is evaluated for design compatibility (building height, architectural detailing, setbacks, access, lot configuration, and views), relationship to residential development and open space, and potential traffic circulation impacts.					
Villages						
LU-3.1	Expand mixed-use and commercial development at the Neighborhood Center/ Node at Washington and Goldfinch Streets. –					
LU-3.2	Encourage high intensity pedestrian-oriented commercial and mixed-use development in the Community Village - Hillcrest Core West surrounding University and Fifth Avenues.Permit high intensity pedestrian-oriented commercial and mixed-use development in the Hillcrest Neighborhood Center/Node surrounding University and Fifth Avenues.					
LU-3.3	Encourage "active" commercial business on the ground floor level in the <u>Community Village</u> - Hillcrest Core <u>West</u> , especially those that generate pedestrian- oriented activity into the evening.					
LU-3.4	Consider mixed-use options for the redevelopment of the Department of Motor Vehicles (DMV) site.					
LU-3.6	Concentrate office uses on Fourth Avenue north of Maple Street, Third Avenue south of Laurel Street, and First Avenue in the vicinity of Laurel Street. Encourage re-use of historically-oriented residential structures for office use.					
<u>LU-3.7</u>	Identify, prioritize, and implement future park sites and public space within and in close proximity to village areas with input from the public.					
Urban Desigi	n Element					
Public Space						
UD-4.36	Delineate plazas and courtyards through building and landscape design. Ensure that plazas and courtyards are comfortably scaled, landscaped for shade and ornament, furnished with areas for sitting, and lighted for evening use. Courtyards should be surrounded by active façades or landscape treatments.					
UD-4.38	Provide opportunities for public open spaces in neighborhood centers and nodes.					
<u>UD-4.39</u>	Orient public spaces within private development towards the public right-of-way and frame with active building facades (e.g. entrances, windows, balconies, etc.) that help activate the space and provide "eyes on the street" for security.					
<u>UD-4.40</u>	Explore creative ways to create permanent and temporary public spaces from underutilized rights-of-way, vacant parcels, and alleys.					
<u>UN-4.41</u>	Include public spaces and common areas within multifamily residential development that are clearly marked and conditioned for pet use.					
Economic Pro	osperity Element					
Commercial D	Districts and Corridors					
EP-1.1	Improve the pedestrian, bicycle and transit infrastructure in Uptown's commercial districts.					
EP-1.2	allowing commercial shops and service activities.					

Table 6.1-1							
Proposed Uptown CPU Policies Related to Land Use							
Policy	Description						
EP-1.3	Explore opportunities for boutique hotels in Hillcrest as the area is close to the freeway and Mission Valley.						
EP-1.4	Continue to work with the Uptown Parking District to consider locations for a parking garage near central Hillcrest and other multimodal transportation options for this area.						
EP-1.5	Promote growth of Uptown's health sector enhancing the areas reputation for quality care and to support the expected employment growth in this sector.						
EP-1.6	Support incentives for new development of mid-rise office buildings in the east end of Hillcrest, including but not limited to mixed-use office and residential buildingsCreate incentives for new development of office in the east end of Hillcrest.						
EP-1.7	Promote the LGBTQ historic heart of Hillcrest's Entertainment District, which encourages heritage tourism.						
EP-1.8	Request future City Council legislation is considered to define and recognize the boundaries of the City's "Entertainment District"; specifically, recognize the LGBTQ Entertainment District in the Hillcrest core.						
EP-1.9	Promote development of physical space such as shopkeeper units and other types of live/work spacePromote development of shopkeeper units and other types of live/work space.						
Community R	evitalization						
EP-2. <u>4</u> 6	Position and expand University and 5th Avenue's entertainment districts to attract more regional patrons and tourist.						
EP-2. <u>5</u> 7	Market the Downtown and Balboa Park visitor trade and improve convenient transportation linkages from those destinations to Uptown, including a potential street car linkage.						
EP-2. <u>6</u> 8	Utilize economic development tools and programs to attract and retain small businesses, through the maintenance and enhancement of commercial areas.						
Public Facilit	ies, Services, and Safety Element						
Public Facilitie	es & Services						
<u>PF-1.1</u>	Locate and cluster public facilities, such as libraries, post offices, community meeting space, schools, and transit-oriented development to create an active center.						
Recreation Element							
Parks and Red	creation Facilities						
<u>RE-1.12</u>	Encourage development of parks within residential mixed-use developments, and other public facilities.						
<u>RE-1.16</u>	Explore the possibility of providing a public park within the redevelopment of the San Diego Unified School District's Education Center on Normal Street.						
<u>RE-2.7</u>	Provide pocket parks with ecologically sensitive recreational uses as enhanced gateways to open space lands.						

A	T	T	A	С	Η	N	1	E	N	T	7	

6.1 Land Use

Table 6.1-2 Uptown Community Plan Proposed Land Use Designations						
				Inte	nsity	
General Plan Land Use	Community Plan Designation	Specific Use Considerations	Description	Residential Density (dwelling units/acre)	Development Intensity ¹	
ark, Open Space, and Recreation	Open Space	None	Provides for the preservation of land that has distinctive scenic, natural or cultural features; that contributes to community character and form; or that contains environmentally sensitive resources. Applies to land or water areas that are undeveloped, generally free from development, or developed with very low-intensity uses that respect natural environmental characteristics and are compatible with the open space use. Open Space may have utility for: primarily passive park and recreation use; conservation of land, water, or other natural resources; historic or scenic purposes; visual relief; or landform preservation.	1	OR-1-1 zone 0.45 FAR	
<u>م</u>	Population-based Parks	None	Provides for areas designated for passive and/or active recreational uses, such as community parks and neighborhood parks. It will allow for facilities and services to meet the recreational needs of the community as defined by the Community Plan.	N/A	OP-1-1 zone OP-2-1 zone	
	Residential - Low	None	Provides for single-family housing within a low residential density range and limited accessory uses.	5 - 9	RS-1-7 zone FAR varies	
Residential	Residential - Low Medium	None	Provides for both single-family and multifamily housing within a low- medium residential density range.	10 - 15	RM-1-1 zone 0.75 FAR	
	Residential - Medium	None	Provides for both single-family and multifamily housing within a medium residential density range.	16 - 29	RM-2-5 zones 1.35 FAR	

l

ATTACHMENT 7

6.1 Land Use

Table 6.1-2						
	Uptown Community Plan Proposed Land Use Designations					
General Plan Land Use	Community Plan Designation	Specific Use Considerations	Description	Inte Residential Density (dwelling units/acre)	Development Intensity ¹	
Itial	Residential – Medium High	None	Provides for multifamily housing within a medium-high residential density range.	30-44	RM-3-7 zone 1.80 FAR	
Resider	Residential - High	None	Provides for multi-family housing with a high density range.	45-73	RM-3-9 zone 2.70 FAR	
	Residential – Very High	None	Provides for multi-family housing with a very high density range.	74-109	RM-4-10 zone 3.60 FAR	
Commercial Commercial Commercial			Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Permits office uses. Housing may be allowed up to a medium residential density within a mixed-use setting	0-29	CC-1-3 zone 0.75/0.75 ¹ FAR	
	Office- Commercial	Office- Residential	Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Permits office uses. Housing may be allowed up to a medium-high residential density within a mixed- use setting.	0-44	CC-3-6 Zone 2.0/2.0 ¹ FAR	
		Permitted	Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Permits office uses. Housing may be allowed up to a high residential density within a mixed-use setting.	0-73	CC-3-8 zone 2. <u>0</u> 5/2.5 ¹ FAR	
			Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Permits office uses. Housing may be allowed up to a very high residential density within a mixed-use setting.	0-109	CC-3-9 zone 2.0/3.0 ¹ FAR	

ATTACHMENT 7

6.1 Land Use

Table 6.1-2 Uptown Community Plan Proposed Land Use Designations					
				Inte	ensity
General Plan Land Use	Community Plan Designation	Specific Use Considerations	Description	Residential Density (dwelling units/acre)	Development Intensity ¹
ervices (cont.)			Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Housing may be allowed up to a medium residential density within a mixed-use setting.	0-15	CN-1-1 zone 1.0/1.5 ¹ FAR
commercial Employment, Retail & S Commerci	Neighborhood Commercial	hood Residential rcial Permitted	Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Housing may be allowed up to a medium residential density within a mixed-use setting.	0-29	CN-1-3 zone 1.0/0.75 ¹ FAR
			Provides local convenience shopping, civic uses, and services serving an approximate three mile radius. Housing may be allowed up to a medium-high residential density within a mixed-use setting	0-44	CN-1-4 zone 1.0/1.2 ¹ FAR <u>CC-3-6 zone</u> <u>2.0/2.0¹ FAR</u>
Commercial Employment, Retail & Services	Community Commercial	Community Residential Commercial Permitted	Provides for shopping areas with retail, service, civic, and office uses for the community at-large within three to six miles. Housing may be allowed up to a high residential density within a mixed-use setting.	0.29	CC-3- <u>45</u> zone <u>12</u> .0/ <u>02.50</u> 1 FAR
			Provides for shopping areas with retail, service, civic, and office uses for the community at-large within three to six miles. Housing may be allowed up to a high residential density within a mixed-use setting.	0-73	CC-3-8 zone 2.0/2.5 ¹ FAR
			Provides for shopping areas with retail, service, civic, and office uses for the community at-large within three to six miles. Housing may be allowed up to a very high residential density within a mixed-use setting	0-109	CC-3-9 zone 2.0/3.0 ¹ FAR

6.1 Land Use

Table 6.1-2 Uptown Community Plan Proposed Land Use Designations					
				Inte	nsity
General Plan Land Use	Community Plan Designation	Specific Use Considerations	Description	Residential Density (dwelling units/acre)	Development Intensity ¹
Institutional, Public and Semi-Public Facilities	Institutional	None	Provides a designation for uses that are identified as public or semi- public facilities in the Community Plan and which offer public and semi-public services to the community. Uses may include but are not limited to: military facilities, community colleges, communication and utilities, transit centers, schools, libraries, police and fire facilities, post offices, hospitals, park-and-ride lots, government offices and civic centers.	N/A	Based on the underlying zone for non- government owned property
¹ Where residential is permitted a Floor Area Ratio bonus is provided per the zone to encourage residential mixed-use development.					

The **Land Use Element** of the proposed Uptown CPU contains community-specific policies to guide development within the Uptown community. This element establishes the distribution and pattern of land uses throughout the community along with associated residential densities.

Uptown is a community with an established land use pattern that is expected to remain with commercial and mixed-use located along transit corridors, multifamily and single-family uses located adjacent to commercial areas and open space located primarily within single-family neighborhoods. The community has a unique level of complexity due to its long-standing and diverse development history; varied geography; and proximity to Balboa Park, Downtown, and Mission Valley. Policies within the Land Use Element are constructed to promote the overall land use goals of the proposed Uptown CPU, which include residential goals such as provision of a diversity of housing options. Commercial goals include appropriately located commercial and office facilities offering a wide variety of goods, services, and employment to benefit the entire community; continued revitalization of Uptown's business districts that respect potential impacts to adjacent neighborhoods; and diversification of employment opportunities. Mixed-use goals generally include the creation of villages with a lively, walkable, and unique atmosphere that builds upon existing neighborhoods and includes places to live and work; and commercial/residential transition areas that promote compatible development and reinvestment along the community's commercial districts.

As with the General Plan, the proposed Uptown CPU places an emphasis on directing growth into mixed-use activity centers that are pedestrian-friendly and linked to an improved regional transit system. Prior to the adoption of the General Plan, Uptown was already in a position to promote "village-like" development with identified areas for mixed-use development already focused along

major transportation corridors and policies for improving the pedestrian environment by enhancing pedestrian activity in business districts and neighborhoods. Uptown is expected to see an improved level of walkability, bicycling, and transit through the implementation of mobility-related projects and improvements and efforts that are focused within community village areas and commercial corridors.

The proposed Uptown CPU would also be consistent with the General Plan goal of providing diverse and balanced neighborhoods and communities. The land use plan prepared for the proposed Uptown CPU provides for a combination of land uses, which emphasize the existing diversity of the community, as well as a diversity that supports future growth and prosperity within the CPU area.

The existing development within Uptown provides a foundation for achievement of the goals laid out in the General Plan Mobility Element due to the urban character of the community, existing transit connections, and adjacency to major roadways and interstates. The proposed Uptown CPU **Mobility Element** policies support the development of pedestrian-friendly facilities along streets and emphasize a safe bicycle network with provision of bicycle parking facilities for transition to pedestrian use within the commercial areas. The proposed Uptown CPU also includes Intelligent Transportation System policies that promote the application of technology to transportation systems with the goal to maximize efficiency of services while increasing vehicle throughput, reducing congestion, and providing quality information to the commuting public.

The **Urban Design Element** of the proposed Uptown CPU supports and implements the General Plan at the Community Plan level by including specific design guidelines and policies for the proposed Uptown CPU area that are consistent with the community's existing and projected character. The proposed Uptown CPU contains policies that are intended to improve the quality of life through safe and secure neighborhoods and in a manner that respects the natural environment. It addresses existing and planned access to outdoor and active spaces, and identifies active and passive open space areas, recreational facilities, and access via pedestrian and bicycle pathways.

The **Economic Prosperity Element** supports employment growth within the community by increasing small business opportunities and supports a diverse mix of businesses that provide a variety of goods and services. This element identifies the value of successful entertainment districts that appeal to local and regional residents, as well as tourists, as well as vibrant neighborhood commercial districts where residents purchase a significant share of their basic needs and services from within the community. Additionally, the Economic Prosperity Element calls for the expansion of medical-related development and employment, as well as parking and multi-modal transit options for commercial districts.

Consistent with the Public Facilities, Services, and Safety Element of the General Plan, the proposed Uptown CPU **Public Facilities, Services, and Safety Element** includes goals to provide and maintain infrastructure and public services for future growth without diminishing services to existing development. Specific policies regarding public facilities financing include public facilities and services prioritization as well as fire-rescue, police, wastewater, storm water infrastructure, waste management and recycling, libraries, schools, public utilities, and healthcare services and facilities, all included within the proposed Uptown CPU.

In regard to the Recreation Element of the General Plan, the proposed Uptown CPU also provides **Recreation Element** policies that support the pursuit of land acquisition needed for the creation of public parks with a special effort to locate new parkland within the community and promoting connectivity, safety, public health, and sustainability. Strategies to reduce the existing parkland deficit in the CPU area are also included in the Recreation Element. Policies to provide parkland to help meet the needs of the community through plan build-out and provide for preservation, protection, and enhancement of existing and planned parkland facilities are included. The Uptown Community would remain in deficit of nearly 101approximately 97 acres of population-based park space. Uptown has an existing deficit in parkland facilities. As discussed above, the proposed Uptown CPU Recreation Element includes community-specific policies addressing park and recreation guidelines, preservation, and accessibility. As proposed, the proposed Uptown CPU policies regarding parks and recreational facilities are consistent with the General Plan environmental goals, objectives, and guidelines policies; however implementation of the proposed Uptown CPU would still result in a shortfall in the amount of population-based park land, which is adverse. While there are potential environmental impacts from the development of park and recreational facilities as discussed in Section 6.12, Public Services and Facilities, the proposed Uptown CPU community-specific goals and recommendations are intended to support and implement the General Plan environmental goals, objectives, and guidelines policies. Implementation of the proposed Uptown CPU would have an adverse impact because of the population-base park land deficit, but there would be no impact to land use compatibility. As such, the impacts to land use would be less than significant.

The proposed Uptown CPU is consistent with the conservation policies contained within the Conservation Element of the General Plan. The **Conservation Element** of the proposed Uptown CPU addresses the conservation goals and policies that can be effective in managing, preserving, and thoughtfully using the natural resources of the community. Climate change is also addressed in a manner consistent with the General Plan within both the Urban Design Element and Conservation Element. Sustainable energy policies are included that promote development that qualifies for the City's Sustainable Buildings Expedite Program; educate residents and businesses on efficient appliances and techniques for reducing energy consumption; provide for, or retrofit, lighting in the public rights-of-way that is energy efficient; and provide information on programs and incentives for achieving more energy-efficient buildings and renewable energy production.

With respect to the General Plan policies concerning noise and land use compatibility, the **Noise Element** of the proposed Uptown CPU includes goals and policies to guide compatible land uses and require the incorporation of noise attenuation measures for new uses. Additionally, this element provides additional detail to General Plan policies. Please see Section 6.6 for a discussion of noise impacts.

The City of San Diego's General Plan Historic Preservation Element guides the preservation, protection, restoration, and rehabilitation of historical and cultural resources and maintain a sense of the City. The Uptown community is one of the oldest urban neighborhoods in San Diego. The **Historic Preservation Element** of the proposed CPU provides general policies to preserve significant historical resources. This element calls for the identification and preservation of significant historical resources, as well as educational opportunities and incentives relative to

historical resources in Uptown. Impacts relative to historical resources are discussed in Section 6.7, Historical Resources.

As part of the proposed project analyzed within this Program Environmental Impact Report (PEIR), the City is updating the Impact Fee Study (IFS; formerly Public Facilities Financing Plan) for the Uptown community, which was originally adopted in 1988. The IFS sets forth the major public facilities' needs specific to the Uptown community with respect to transportation (streets, storm drains, traffic signals, etc.), libraries, park and recreation facilities, and fire stations. The proposed Uptown CPU is a guide for the future development within the community and serves to determine public facility needs. Revisions to public facility needs, Development Impact Fees (DIFs), or other capital improvement programs, would be included in the updated IFS.

b. Land Development Code Regulations

Implementation of the actions associated with adoption of the proposed Uptown CPU would include specific rezone actions, as well as the repeal of the Mid-City Communities Planned District Ordinance to a zone program that uses Citywide zoning and community-specific zoning through implementation of a Community Plan Implementation Overlay Zone (CPIOZ) to address building height within the mixed-use corridors specified in the Uptown Community Plan. The proposed Planned District/Citywide zone conversions applicable to the proposed Uptown CPU are shown in Table 6.1-3. The implementation program for the proposed Uptown CPU also includes residential zones identified in Table 6.1-3 that would be converted to an open space-residential zone to preserve privately-owned property that is designated in the proposed Uptown CPU as open space with limited development.

Table 6.1-3							
Recommended Zone Designations – Uptown							
Mid-City Communities Planned District Zones	Citywide Zones						
MR-3000	RM-1-1						
MR-1500	RM-2-5						
MR-1000	RM-3-7						
MR-800B	RM-3-9						
MR-400	RM-4-10						
CV-4, CL-5, CL-6	CN-1-3						
CN-3, CN-4, CN-2A, CV-3, CL-2	CN-1-4						
NP-3	CC-1-3 ,CC-3-4						
<u>CN-1</u>	<u>CC-3-4</u>						
CN-1	CC-3-5						
NP-2 <u>, CN-2A</u>	CC-3-6						
CN-1, CN-1A, CL-2, NP-1	CC-3-8						
CN-1, CN-1A, CV-1	CC-3-9						
West Lewis Street Planned District Zone	Citywide Zone						
WLSPD	CN-1-1						
Residential Zones	Citywide Zones						
RS-1-1, RS-1-2, RS-1-4, RS-1-5	OR-1-1						

The proposed use of the CPIOZ would apply to specific geographic areas within the Uptown CPU area per Chapter 13, Article 2, Division 14 of the Municipal Code (refer to Chapter 3.0, Project Description, Figures 3-7 and 3-8). The CPIOZ provides community-specific building heights and supersedes equivalent regulations in the zones applied within the community. CPIOZ is not intended to address use. Use categories are determined by the applicable base zone. The purpose of the CPIOZ is to supplement the Municipal Code by providing development regulations that are tailored to specific circumstances and/or sites within the community and have been adopted as part of the community plan. The CPIOZ would also provide for a discretionary review process to more effectively implement the proposed Uptown CPU policies and recommendations related to building height in the Urban Design Element.

The CPIOZ has two types differentiated by their review process: Type A (ministerial review), and Type B (discretionary review). Both types are applied within the community depending on geographic area. Development proposals subject to CPIOZ Type B would require discretionary review to determine if the development proposal is consistent with the proposed Uptown CPU as well as the applicable regulations listed below. Development proposals subject to CPIOZ Type B would be required to process and obtain approval of a Process Three Site Development Permit in accordance with Chapter 12, Article 6, Division 5 of the Municipal Code. Exceptions from these regulations may be granted per Municipal Code Section 132.1403 for development that is minor, temporary, or incidental and is consistent with the purpose and intent of this CPIOZ. Any development proposals that proposes to deviate from the regulations or the supplemental CPIOZ regulations would be required to obtain a discretionary permit.

ESL Regulations

Environmentally sensitive lands (ESL; e.g., sensitive biological resources, steep hillsides, historical resources) occur within the proposed Uptown CPU area. Any future development proposed on environmentally sensitive lands would be subject to the City's ESL Regulations (Chapter 14, Article 3, Division 1), which require that future projects demonstrate that the proposed development site is physically suitable for the proposed use and that it would minimize disturbance to natural landforms and not increase flood hazards. In the event a future specific project is considered for an ESL Regulations deviation, supplemental findings would be required prior to approval in order to show that development would not result in an additional public safety threat or extraordinary public expense, or create a public nuisance. Adherence to these regulations would avoid significant impacts to environmentally sensitive lands within the proposed Uptown CPU area.

MHPA Land Use Adjacency Guidelines

The Multi-Habitat Planning Area (MHPA) has been designed to maximize conservation of sensitive biological resources, including sensitive species. When land is developed adjacent to the MHPA, there is a potential for secondary impacts that may degrade the habitat value or disrupt animals within the preserve area. These secondary effects of project development may include habitat insularization, drainage/water quality impacts, lighting, noise, exotic plant species, nuisance animal species, and human intrusion. These impacts could be short-term resulting from construction activities, or long-term. Short-term construction impacts could result in disruption of nesting and breeding thus affecting the population of sensitive species. To address these concerns, the Multiple
Species Conservation Program (MSCP) includes a set of MHPA Land Use Adjacency Guidelines that are to be evaluated and implemented at the project level.

c. San Diego Forward – The Regional Plan

The proposed Uptown CPU land use scenario would be consistent with the goals of *San Diego Forward: the Regional Plan*, prepared by San Diego Association of Governments (SANDAG) to develop compact, walkable communities close to transit connections and consistent with smart growth principles, as summarized above. The CPU proposes to establish a pedestrian-oriented, urban, and mixed-use community village that would reduce reliance on the automobile and promote walking and use of alternative transportation. Policies contained within the proposed Uptown CPU Land Use and Mobility Elements serve to promote bus transit use as well as other forms of mobility, including walking and bicycling. These measures are consistent with *San Diego Forward*'s smart growth strategies. The adoption and implementation of the proposed Uptown CPU would not generate any conflict or inconsistencies with *San Diego Forward*: *the Regional Plan*; therefore the potential impacts would be less than significant.

Issue 2 Conversion of Open Space or Farmland

Would the proposed project lead to the development or conversion of general plan or community plan designated open space or prime farmland to a more intensive land use, resulting in a physical division of the community?

The proposed project involves an update to the Uptown Community Plan, a fully built-out community in the City of San Diego, and other associated discretionary actions. The current makeup of the urbanized Uptown CPU area includes a mix of land uses that includes open space. In addition, there is approximately a half acre identified as agricultural land within the Uptown CPU area this is a community garden, not prime farmland. The siting of mixed uses in proximity to each other, the provision of enhanced pedestrian corridors and bicycle amenities, and the planned changes to the street network would additionally serve to foster community connectivity rather than create division.

Goals of the proposed Uptown CPU Land Use Element that address community connectivity include supporting a vibrant, pedestrian-oriented community village within the proposed Uptown CPU area that provides diverse housing opportunities and encourages quality neighborhood and community-supporting institutional and commercial uses. Overall, incorporation of the goals and recommendations of the elements contained in the proposed Uptown CPU would enhance community connectivity. In addition, the proposed Uptown CPU Conservation Element contains polices that preserve open space within the Community Plan area. The proposed Uptown CPU would not result in conversion of open space and there is no proposed change to the community garden. Therefore, the implementation of the proposed Uptown CPU and associated discretionary actions would not result in any policies that would permit the conversion of open space in adjacent communities. Therefore, the implementation of the groupsed Uptown CPU and associated discretionary actions would not lead to the development or conversion of the proposed Uptown CPU and associated discretionary actions would not result in any policies that would permit the conversion of open space or physically divide the communities. Therefore, the implementation of the proposed Uptown CPU and associated discretionary actions would not lead to the development or conversion of identified open space or physically divide the community and would not result in any policies that would permit the conversion of identified open space or physically divide the community and would not result in any policies that would permit the conversion of identified open space or physically divide the community and would not result in any policies that would permit the conversion of identified open space or physically divide the community and would not result in any policies that would permit the conversion of identified open space or physically divide the community and would not result in any policies that would permit the convers

Issue 3 Conflicts with the MSCP Subarea Plan

Would the project conflict with the provisions of the City's Multiple Species Conservation Program (MSCP) Subarea Plan or other approved local, regional, or state habitat conservation plan?

The highly urbanized Community Plan area lies within the City's MSCP Subarea Plan, and contains preserve areas designated as MHPA in the northern portion of the project area. Because the proposed Uptown CPU area contains MHPA lands, the ESL Regulations limit development encroachment into sensitive biological resources. As concluded in Section 6.8, Biological Resources, the project would be consistent with the MSCP Subarea Plan, and impacts would be less than significant.

Issue 4 Conflicts with an Adopted ALUCP

Would the project result in land uses which are not compatible with an adopted Airport Land Use Compatibility Plan (ALUCP)?

The project site is located within San Diego International Airport's (SDIA) Airport Influence Area (AIA). The AIA is "the area in which current or future airport-related noise, overflight, safety, or airspace protection factors may significantly affect land uses or necessitate restrictions on those uses." To facilitate implementation and reduce unnecessary referrals of projects to the Airport Land Use Commission, the AIA is divided into Review Area 1 and Review Area 2. The project site is located within both of them (Figure 6.1-1). The composition of each area is determined as follows:

- Review Area 1 is defined by the combination of the 60 dB CNEL noise contour, the outer boundary of all safety zones, and the airspace Threshold Siting Surfaces (TSSs) (Figure 6.1-2). All policies and standards apply within Review Area 1.
- Review Area 2 is defined by the combination of the airspace protection and overflight boundaries beyond Review Area 1. Only airspace protection and overflight policies and standards apply within Review Area 2.

The ALUCP contains four principal compatibility concerns: noise (exposure to aircraft noise), safety (land use factors that affect safety both for people on the ground and occupants of aircraft), airspace protection (protection of airport airspace), and overflight (annoyance or other general concerns related to aircraft overflights). The ALUCP policies and standards are only applicable to new uses. A portion of the Uptown CPU is located within the SDIA noise contours of the ALUCP, which range from 60- to 75+ -decibel (dB) community noise equivalent level (CNEL) (Figure 6.1-3 and Figure 6.1-4). Noise impacts are fully evaluated in Section 6.6, Noise, of this PEIR. As discussed in Section 6.6 of this PEIR, the proposed Uptown CPU would not result in adverse airport noise impacts to existing uses because the proposed Uptown CPU would not result in a change to existing uses or a change in SDIA operations. New development would be required to provide noise attenuation consistent with the ALUCP for the San Diego International Airport; thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant exposure to noise from aircraft.

Map Source: SDIA - ALUCP



FIGURE 6.1-1 SDIA Airport Influence Area – Uptown

Map Source: SDIA - ALUCP



FIGURE 6.1-2 SDIA Threshold Siting Surface Profile – Uptown





FIGURE 6.1-3 SDIA Noise Contour Map – Uptown



FIGURE 6.1-4 Residential Use Allowed Under the General Community Plans Inside SDIA 70 dB CNEL – Uptown A portion of the Uptown community is located within ALUCP Safety Compatibility Zones 3NE, 2E, and 3SE (Figure 6.1-5, and Figure 6.1-6). Safety compatibility standards of the ALCUP restrict certain uses and provide maximum residential density and non-residential intensity limits that are allowable within the safety zones. These standards are shown in Table 3-1 of the SDIA ALUCP. Future development within the ALUCP Safety Compatibility Zones would be required to comply with these standards or request a City Council overrule; as such, impacts relative to ALUCP safety zones would be less than significant.

The airspace protection boundary (Figure 6.1-7) for SDIA establishes the area where the policies and standards of the ALUCP apply. Additional boundaries at the ends of the runway represent the TSSs within which specific height limitations apply. A TSS defines critical airspace that must be protected to allow for safe approaches to runways. Any objects penetrating the TSSs would cause the runway thresholds to be further displaced, reducing available landing distances.

The airspace protection boundary is based on the outermost edge of the following airspace surfaces:

- Part 77, Subpart B, 100:1 notification surface boundary
- Part 77 civil airport imaginary airspace surfaces
- The approach surfaces for both runway ends defined by the criteria in Federal Aviation Administration Order 8260.3B, United States Standard for Terminal Instrument Procedures (TERPS)

The Uptown community is located within the Federal Aviation Regulations Part 77 Notification Surfaces, and the southern portion of the community is additionally located within the Threshold Siting Surface. The City requires a Federal Aviation Administration determination of no hazard to air navigation for both ministerial and discretionary projects prior to approving or recommending approval as addressed in Development Services Department Information Bulletin 520. Additionally, future projects located within the TSS would be required to comply with ALUCP criteria relative to this airspace protection area. As such, impacts to airspace protection would be less than significant.

Overflight compatibility concerns apply to the Uptown CPU area. The Uptown community is located within the Overflight Notification Area (Figure 6.1-8). An overflight notification agreement must be recorded with the Office of the County Recorder for any new dwelling unit within the overflight area. The recordation of an overflight notification agreement is not necessary where the dedication of an avigation easement is required. Alternative methods of providing overflight notification are acceptable if approved by the Airport Land Use Commission.

Thus, as described in this section, implementation of the proposed Uptown CPU and associated discretionary actions would be consistent with the adopted ALUCP as future development within the Uptown CPU area would be subject to the requirements of the ALUCP and associated Federal Aviation Administration and City requirements. Thus, impacts related to conflicts with an adopted ALUCP would be less than significant.

Map Source: SDIA - ALUCP



FIGURE 6.1-5 SDIA Safety Compatibility Zones – Uptown



FIGURE 6.1-6

Residential Use Allowed Under the General/Community Plans Inside the SDIA Safety Compatibility Zones – Uptown

Map Source: SDIA - ALUCP



FIGURE 6.1-7 SDIA Airspace Protection Boundary – Uptown



FIGURE 6.1-8 SDIA Overflight Notification Area – Uptown

Cumulative Impacts

As discussed in this section, the proposed Uptown CPU contains 9 core elements providing community-specific goals and policies that are consistent with Citywide zoning classifications, development design guidelines, mobility guidelines, and programs in accordance with the goals of the City's General Plan and the implementing regulations of the City's Land Development Code. Both the Uptown CPU along with the North Park and Golden Hill CPUs would accommodate existing development as well as encourage development consistent with community goals and character.

The Uptown CPU combined with the North Park and Golden Hill CPUs are consistent with and also implement the environmental goals or objectives of the SANDAG's *San Diego Forward: the Regional Plan.* The three CPUs are consistent with the City's Multiple Species Conservation Program. Development implemented in accordance with the Uptown, North Park, and Golden Hill CPUs would not result in conflicts with the City's ESL Regulations, which contains policies supporting the goals of these regulations. Any development within the CPU areas that would encroach into ESL would be subject to review in accordance with the ESL Regulations (Land Development Code, Section 143.0101 et seq.). Future development projects within the Airport Influence Area would be submitted to the Airport Authority, acting as the Airport Land Use Commission, to ensure the consistency of future development with the ALUCP for the San Diego International Airport, until the Airport Land Use Commission determines the updated CPU and development regulations consistent with the ALUCP. Based on the compatibility of the proposed CPUs (Uptown, North Park, and Golden Hill) with the General Plan policy framework and other applicable land use plans and regulations, cumulative land use compatibility, impacts associated with build-out of the CPUs would be less than significant.

6.1.4 Significance of Impacts

6.1.4.1 Conflicts with Applicable Plans

The proposed Uptown CPU would be consistent with the General Plan and the City of Villages strategy. Furthermore, the policies developed for the proposed Uptown CPU associated with each of the elements were drafted in a manner that is consistent with the General Plan. Proposed amendments to the Land Development Code and zoning amendments would implement the proposed Uptown CPU and would be consistent with applicable environmental goals, objectives and guidelines of the General Plan. The proposed Uptown CPU would include approval of an amendment to the Land Development Code to repeal the existing Mid-City Communities and West Lewis Street Planned District Ordinances that serve as the community's zoning regulations and replace them with Citywide zoning and as well as amend the Uptown CPIOZ related to building height in specific geographic areas. These proposed amendments are intended to accommodate existing desirable uses and encourage future development consistent with the proposed Update CPU. The proposed change from the PDO to Citywide zone and amendment of the CPIOZ boundary areas would not create any conflicts or inconsistencies with the adopted Land Development Code.

Future development in accordance with the proposed Uptown CPU would be required to comply with ESL regulations. The proposed Uptown CPU incorporates the multi-modal strategy of *San Diego Forward* through the designation of a high-density mixed-use village. In addition, the proposed

Uptown CPU includes policies related to land use, mobility, and circulation/transportation that promote the *San Diego Forward*'s smart growth strategies. As the proposed Uptown CPU and associated discretionary actions would be consistent with applicable environmental goals, objectives, or guidelines of a General Plan and other applicable plans and regulations, no indirect or secondary environmental impact would result and impacts would be less than significant. No mitigation is required.

6.1.4.2 Conversion of Open Space or Farmland

The proposed Uptown CPU and associated discretionary actions would not convert open space or prime farm land. The proposed Uptown CPU and associated discretionary actions would not physically divide an established community. Community connectivity would be enhanced by provisions in the proposed Uptown CPU that improve pedestrian and transit amenities. No significant impacts have been identified; therefore, no mitigation would be required.

6.1.4.3 Conflicts with the MSCP Subarea Plan

The proposed Uptown CPU implementation would not have significant impacts on the MHPA and the project would be consistent with the MSCP. Therefore, no impacts would occur. No mitigation is required.

6.1.4.4 Conflicts with an Adopted ALUCP

Although the Uptown community is within the SDIA AIA, the proposed Uptown CPU and associated discretionary actions would not result in impacts associated with the four compatibility concern areas. Future projects would be required to receive ALUC consistency determinations, as necessary, stating that the project is consistent with the SDIA ALUCP. As a result, the proposed Uptown CPU and associated discretionary actions would not result in land uses that are incompatible with an adopted Airport Land Use Compatibility Plan. Therefore, no impacts would result, and no mitigation is required.

6.1.5 Mitigation Framework

Land use impacts related to build-out of the proposed Uptown CPU and associated discretionary actions would be less than significant. Thus, no mitigation is required.

6.2 Visual Effects and Neighborhood Character

This section addresses visual effects of the proposed Uptown Community Plan Update (CPU), the associated discretionary actions, and potential for impacts on neighborhood character and includes a description of the built and natural visual resources within the CPU area. It also describes relevant existing state and local regulations. In addition, the proposed Uptown CPU's consistency with relevant design regulations is assessed, including the adopted General Plan and the proposed Uptown Community Plan elements, as well as the Land Development Code (LDC).

6.2.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

6.2.2 Significance Determination Thresholds

Thresholds used to evaluate potential impacts related to visual effects and neighborhood character is based on applicable criteria in the California Environmental Quality Act (CEQA) Guidelines Appendix G and the City of San Diego CEQA Significance Determination Thresholds (2011). Thresholds are modified from the City's CEQA Significance Determination Thresholds to reflect the programmatic analysis for the proposed Uptown CPU. A significant visual effect and neighborhood character impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions would:

- 1) Result in a substantial obstruction of a vista or scenic view from a public viewing area as identified in the community plan;
- 2) Result in a substantial adverse alteration (e.g. bulk, scale, materials or style) to the existing or planned (adopted) character of the area;
- 3) Result in the loss of any distinctive or landmark tree(s), or stand of mature trees as identified in the community plan
- 4) Result in a substantial change in the existing landform; or
- 5) Create substantial light or glare which would adversely affect daytime and nighttime views in the area.

6.2.3 Impact Analysis

Potential impacts resulting from implementation of the proposed Uptown CPU and associated discretionary actions were evaluated based on information from existing conditions assessments of

urban design, recreation, and conservation in the Uptown CPU area. The assessment was made using data from observation, spatial analysis, and a photographic inventory.

Issue 1 Scenic Vistas or Views

Would the project result in a substantial obstruction of a vista or scenic view from a public viewing area as identified in the community plan?

The Urban Design Element of the proposed Uptown CPU identifies public view corridors in three primary categories: views looking to the north and west over Mission Valley and to Mission Bay, views looking to the west and south toward San Diego Bay, and views looking east to Balboa Park. These views are largely available from public rights-of-ways, parks, and landmarks. The proposed Uptown CPU contains policies in both the Urban Design and Conservation elements relative to views, and identifies public view corridors and viewsheds in the community (Figure 6.2-1). Proposed Urban Design Element policies would protect against view obstructions to Balboa Park from public vantage points and would provide design criteria for development along canyons that would protect available canyon views from surrounding roadways. Proposed policies of the Conservation Element would provide protection for public views from identified public vantage points such as public right-of-ways, parks, and landmarks. Future development projects within the identified public view corridors would be evaluated at the time of a future development proposal for potential impacts on identified public view corridors.

Due to the built-out nature of the Uptown community, future projects would blend with the existing urban framework through established and regulated set back requirements, and would not result in new obstructions to view corridors along public streets where view opportunities largely exist. Should future projects seek to impede upon existing circulation elements in the form of diminished right-of-way or a street vacation, for example, these projects would be reviewed as discretionary projects and those future project impacts on view corridors that occur within the right-of-way would be evaluated at that time.

An Officially Designated State Scenic Highway runs through the Uptown community for a one-mile stretch of State Route 163 (SR-163) between the north and south boundaries of Balboa Park. The proposed Uptown CPU would not impact this scenic resource, as any future development would be limited to the mesa top of the community and SR-163 is located at the base of the mesas in a canyon that would not be subject to development. Because the scenic highway section of the freeway is bordered to the east and west by Balboa Park, the existing regional park protects the scenic resource. Thus, no impacts to an Officially Designated State Scenic Highway would result.

Implementation of the proposed Uptown CPU would not result in a substantial alteration or blockage of public views from critical view corridors, designated open space areas, public roads, or public parks; new development within the community would take place within the constraints of the existing urban framework and development pattern. Thus, future development would not impact view corridors or viewsheds as viewed from identified public vantage points. Public view impacts would be less than significant, and no mitigation would be required. Map Source: SanGIS

ATTACHMENT 7



FIGURE 6.2-1 Uptown Canyons and Public Views

Issue 2 Neighborhood Character

Would the project result in a substantial alteration (e.g. bulk, scale materials or style) to the existing or planned (adopted) character of the area?

Uptown is a developed, urbanized community although not all lot or building sites are built to their allowable capacity under the adopted Community Plan and zone. <u>The Uptown CPU area has a variety of neighborhoods with different character as described in Section 1.1 of the proposed Uptown CPU.</u> Thus, build-out of the proposed Uptown CPU would generally result in intensification of existing land uses. Any new development or redevelopment would take place on infill sites Implementation of the proposed Uptown CPU would be subject to change primarily where existing properties are undeveloped or underutilized. For example, there are single-family areas of the University Heights neighborhood designated medium residential density (15-29 dwelling units per acre [du/ac]), However, significant land uses changes are not anticipated in these areas because most lots are already completely developed with residences in addition to companion units or duplex units relegated to the rear of the yard. Thus, the focus of development would be limited to underdeveloped properties. An area that is anticipated to result in a change in development intensity includes land designated for Community Commercial 0-109 du/ac along 5th and 6th Avenue south of Upas Street in Bankers Hill/Park West. In this area, taller, higher intensity, mixed-use or multi-family development could be developed, compared to existing conditions.

and would thus blend with the existing urbanized character of the community.

Additionally, <u>fF</u>uture development projects would be undertaken in accordance with the City's General Plan and Land Development Code in addition to proposed Uptown CPU policies. The proposed Uptown CPU includes Urban Design Element policies intended to direct future development in a manner that ensures that the physical attributes of the Uptown community will be retained and enhanced by a design that responds to the community's particular context while acknowledging the potential for growth and change.

The proposed Urban Design Element provides policies relative to streetscape (publicly owned street rights-of-way) and public spaces (publicly accessible open spaces such as parks, squares, plazas, courtyards, and alleys) that would shape the area's character and function as future development occurs. Proposed streetscape policies would address the siting of street furnishings, design character, and provision of plazas and pedestrian nodes to enhance the pedestrian realm. Urban forestry policies are also proposed that would maximize the benefits of trees, including their contribution to the character, identity, and comfort of the community's streets. Trees also contribute to the spatial definition of a street, providing both a comfortable sense of scale and enclosure to the public realm. They may add shade, which contributes to pedestrian comfort, and color, texture, and pattern that contribute to the street's visual quality.

The proposed Urban Design Element of the proposed Uptown CPU would also provide policies addressing commercial and mixed-use development, and residential in-fill development. Policies are related to street wall articulation, ground-level uses, windows, building materials, lighting, signs, corners, architectural projections, rooftop and mechanical screening, public space, public art, street orientation and setbacks, sustainable building design, height and massing, development transitions,

and canyons and open space. Implementation of these policies would provide specific policy support to ensure that the bulk and scale of development is not out of character with the existing environment.

Additionally, the proposed Uptown CPU Urban Design Element includes Design Guidelines by Building Type that would address building height, further protecting public view corridors and regulating the bulk and scale of development. The Design Guidelines by Building Type provides typical height limits for low, mid-, and high-rise buildings within neighborhood districts. Under the high-rise building policies, areas within the Uptown CPU area could be permitted to develop with buildings up to 100 feet in height; however, these areas would be covered by a Community Plan Implementation Overlay Zone (CPIOZ) Type B (see Figure 6.1-3 in Section 6.1, Land Use). Within the CPIOZ areas, projects would be subject to a discretionary review process that would implement the proposed Uptown CPU policies and recommendations, particularly those related to building height consistent with the Urban Design Element. Proposed Uptown CPU Urban Design Element provides design guidelines by building types to control massing and ensure compatible transitions. Building setbacks and upper-story stepbacks are recommended to address massing and compatibility where more intense development is located adjacent to lower height buildings (refer to Urban Design Element policies related to development transitions). These policies and guidelines would ensure taller buildings would not adversely impact surrounding lower intensity properties through neighborhood incompatibility or through creation of excessive shade or shadows. With the implementation of the proposed Urban Design Element policies, zoning, and Land Development Code regulations, new development would be consistent with the existing neighborhood character. Thus, impacts related to substantial alterations to the existing or planned character of the area would be less than significant. No mitigation would be required.

Issue 3 Distinctive or Landmark Trees

Would the project result in the loss of any distinctive or landmark tree(s), or stand of mature trees as identified in the community plan? (Normally, the removal of non-native trees within a wetland as part of a restoration project would not be considered significant.)

The adopted Community Plan identifies one tree within its *Conservation, Cultural and Heritage Resources* section as a historical resource. This tree is identified as:

FLORENCE HOTEL TREE, GRAPE STREET BETWEEN THIRD AVENUE AND FOURTH AVENUE: A large Morton Bay Fig tree planted in the late 19th century, exhibiting an extremely well developed, buttress root system and trunk.

The Florence Hotel tree is identified as a designated historical resource within the proposed Uptown CPU. As discussed in Section 6.7, Historical Resources, this designated historical resource is protected and preserved through existing General Plan policies, the Historic Resource Regulations and guidelines of the Municipal Code. Any alteration or proposed removal of this historical resource would be subject to discretionary review before it could be removed or modified. Thus, existing protective measures for the Florence Hotel Moreton Bay fig are in place that would prevent the loss or alteration of this designated tree, except as required because of tree's health or public safety. As

such, implementation of the proposed Uptown CPU and associated discretionary actions would not result in the loss of any distinctive or landmark trees or any stand of mature trees.

In addition, street trees present within the Uptown CPU area are subject to City Council Policy 900-19, which provides for protection of street trees. The proposed Uptown CPU Urban Design Element, Section 4.3, includes Urban Forestry polices that would augment the Council Policy and includes polices that protect existing trees, promote the planting of new trees, and provide guidance as to the types of trees that should be planted. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to losses of distinctive or landmark trees or mature stands of trees.

Issue 4 Landform Alteration

Would the project result in a substantial change in the existing landform?

It is not anticipated that future development allowed by the proposed Uptown CPU and associated discretionary actions would result in significant landform alteration. As discussed previously, the community is largely developed with existing urban land uses. While the proposed Uptown CPU would intensify some uses, the proposed CPU contains policies to ensure that redevelopment takes into account existing development as well as the landform. Of particular importance are the proposed Uptown CPU Conservation Element and Urban Design Element policies that would support conservation of existing landforms, canyon lands, and open space and would support the design of buildings that respects existing landforms.

Because the proposed Uptown CPU is an adoption of a plan, development would occur in the future over an extended time period and specific grading quantities associated with future development are presently unknown. However, no mass grading is anticipated since the proposed Uptown CPU area is already nearly fully developed with urban uses. As future development proposals come forward pursuant to the proposed Uptown CPU and associated discretionary actions, they would be reviewed to determine whether grading plans demonstrate compliance with the City's significance thresholds for grading or if excavation is required for alternative design features. Therefore, impacts to landform from future development would be less than significant.

Issue 5 Light and Glare

Would the project create substantial light or glare which would adversely affect daytime or nighttime views in the area?

The Uptown community is a built-out urban community. Sources of light currently include those typical of an urban community, such as building lighting for residential and nonresidential land uses, roadway infrastructure lighting, and signage. Future development implemented in accordance with the proposed Uptown CPU and associated discretionary actions would necessitate the use of additional light fixtures and may contribute to existing conditions of light and glare. New light sources may include residential and non-residential interior and exterior lighting, parking lot lighting, commercial signage lighting, and lamps for streetscape and public recreational areas.

Lighting policies within the proposed Uptown CPU Urban Design Element would support pedestrianoriented street lighting with appropriate shielding and low heights to minimize light spillage. Other proposed policies would address glare from photovoltaic panels. These policies would support existing lighting regulations in the LDC. Outdoor lighting is regulated by Section 142.0740 of the LDC. The purpose of the City's outdoor lighting regulations is to minimize negative impacts from light pollution including light trespass, glare, and urban sky glow in order to preserve enjoyment of the night sky and minimize conflict caused by unnecessary illumination. Regulation of outdoor lighting is also intended to promote lighting design that provides for public safety and conserves electrical energy. New outdoor lighting fixtures must minimize light trespass in accordance with the Green Building Regulations, where applicable, or otherwise shall direct, shield, and control light to keep it from falling onto surrounding properties. The regulations prohibit direct-beam illumination from leaving the premises and requires that most outdoor lighting be turned off between 11:00 P.M. and 6:00 A.M. with some exceptions (such as lighting provided for commercial and industrial uses that continue to be fully operational after 11:00 P.M., adequate lighting for public safety).

Section 142.0730 of the City's LDC regulates glare. Section 142.0730 limits a maximum of 50 percent of the exterior of a building to be composed of reflective material that has a light reflectivity factor greater than 30 percent. Additionally, per Section 142.0730(b), reflective building materials are not permitted where the City Manager determines that their use would contribute to potential traffic hazards, diminished quality of riparian habitat, or reduced enjoyment of public open space. Lighting impacts to Multi Habitat Planning Area (MHPA) areas that occur within the Uptown community (within the canyon areas) would be regulated through compliance with Multiple Species Conservation Program (MSCP) Adjacency Guidelines, which requires lighting of all developed areas adjacent to the MHPA to be directed away from the MHPA. With requisite implementation of both the proposed Uptown CPU and General Plan/LDC regulations, as well as requirement of the MHPA Adjacency Guidelines, lighting and glare impacts would be less than significant.

Cumulative Impact Analysis

Future growth within the Uptown CPU area and surrounding communities including North Park and Golden Hill have the potential to cumulatively impact the visual environment through the design and location of future buildings. Changes in visual character and quality resulting from individual development projects within the Uptown CPU area with development within the North Park and Golden Hill CPU areas could contribute incrementally to cumulative impacts with regard to aesthetics. However, the cumulative visual impact of build-out of the three communities would not result in a cumulatively significant impact since the CPU areas are already urbanized and include existing development of the type that would be further developed under the CPUs.

Future development in accordance with the CPU areas is likely to take place on infill sites in previously developed locations. Each proposed CPU (Golden Hill, North Park, and Uptown) contains policies to ensure that any new development is consistent with the existing character and protects public views. The proposed policies address consistency in setbacks, height and bulk, landscaping, design, historic character, and natural features such as canyons and hillsides. The proposed CPUs contain policies to preserve, protect, and restore existing landforms. Compliance with the Municipal Code would ensure that cumulative light and glare impacts are avoided. Based on the existing

urbanized character of the CPU areas and implementation of existing regulations and policies in the proposed CPUs, cumulative impacts would be less than significant.

6.2.4 Determination of Significance

6.2.4.1 Scenic Vistas or Views

Implementation of the proposed Uptown CPU and associated discretionary actions would not result in substantial alteration or blockage of public views from critical view corridors, designated open space areas, public roads, or public parks; new development within the community would take place within the constraints of the existing urban framework and development pattern, thereby not impacting public view corridors and viewsheds along public right-of-ways. Therefore, public view impacts would be less than significant, and no mitigation would be required.

No impacts to Officially Designated State Scenic Highway SR-163 would occur due to the topography and location of the freeway, which is set below the mesa tops where future development could occur. Additionally, Balboa Park provides separation from future development areas, precluding structures from impeding on views to SR-163. Impacts would be less than significant, and no mitigation would be required.

6.2.4.2 Neighborhood Character

The proposed Uptown CPU includes policies that would encourage residential and mixed-use development that would be consistent with the existing neighborhood character, and impacts would be less than significant. No mitigation would be required.

6.2.4.3 Distinctive or Landmark Trees

There are protective measures for the existing Florence Hotel Morton Bay fig, and the implementation of the proposed Uptown CPU would prevent the loss of existing mature trees except as required because of tree health or public safety. The implementation of the proposed Uptown CPU and associated discretionary actions would not result in the loss of any distinctive or landmark trees, or any stand of mature trees; therefore, no impacts would result. No mitigation is required.

6.2.4.4 Landform Alteration

Implementation of the proposed Uptown CPU and associated discretionary actions would result in less than significant impacts related to landform alteration based on implementation of proposed Uptown CPU polices that require building form to be sensitive to topography and slopes, and existing protections for steep slopes (Environmentally Sensitive Lands) and grading regulations within the LDC. Thus, impacts related to landform alteration would be less than significant, and no mitigation would be required.

6.2.4.5 Light and Glare

Impacts relative to lighting and glare would be less than significant. No mitigation is required.

6.2.5 Mitigation Framework

Potential visual effects and neighborhood character impacts resulting from implementation of the Uptown CPU and associated discretionary actions would be less than significant. Thus, no mitigation is required.

6.3 Transportation and Circulation

Kimley-Horn and Associates. Inc. conducted the Uptown, North Park, and Golden Hill Community Plan Update (CPU) Traffic Impact Study (June 2015). The report is included as Appendix B-1 to this EIR. Additionally, a supplemental letter report was prepared for the project to provide an updated analysis that reflected minor updates to the proposed land use map after the June 2015 report was finalized. This supplemental report dated March 15, 2016 prepared by Kimley-Horn and Associates is included as Appendix B-2 of this Program Environmental Impact Report (PEIR) and is titled North Park and Uptown Updated Residential Densities Traffic Evaluation Summary of Findings for the Cluster Community Plan Update. The results of the report pertinent to the Uptown community are presented in this section. Additionally, Kimley-Horn and Associates, Inc. prepared an Uptown, North Park, and Golden Hill Community Plan Update Mobility Study for Build-out Conditions. That report is included in Appendix C to this EIR and discussed in this section, as applicable.

6.3.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively. This existing roadway circulation network, daily and peak-hour traffic volumes, and operations at the study intersections and roadway and freeway segments pertinent to the Uptown CPU area are discussed below.

6.3.1.1 Roadway Network

The following section provides a description of the existing study area streets within the Uptown CPU area. Ultimate roadway classifications are taken from the Uptown Community Plan, last updated in February 1988. The portions of the roadways described are intended to reflect the areas within the community and may not reflect the entirety of the roadway. Functional classifications are based on field observations performed during preparation of the *Traffic Impact Study*. Figure 6.3-1, illustrates the existing roadway classifications for the Uptown community. The City of San Diego Bicycle Master Plan (City BMP) identifies several bicycle facilities in the community, as noted in the roadway descriptions below.

First Avenue functions as a north-south, 2-lane collector, with a curb-to-curb width of 50 feet between Interstate 5 (I-5) and Arbor Avenue. First Avenue is two-way for the majority of its length between Grape Street and Washington Street and one-way northbound otherwise. First Avenue is lined with sidewalks and curbs, with parallel parking <u>generally</u> available on both sides of the street for the entire length of the street. <u>Intermittent angled parking is available on First Avenue between Redwood Street and Palm Street</u>. The posted speed limit <u>is_ranges from 25 miles per hour (mph) to 30 miles per hour (mph)</u>. Access to I-5 northbound is provided at the intersection of First Avenue and Elm Street. The ultimate adopted

Map Source: SanGIS



FIGURE 6.3-1 Existing Roadway Functional Classification – Uptown Community Plan street classification for First Avenue is a 3-lane collector. The City BMP identifies First Avenue as a Class III (Bike Route) facility between downtown and Lewis Street, with the option of a Class II (Bike Lanes) between Upas Street and Washington Street.

Fourth Avenue functions at its adopted Community Plan ultimate classification as a north-south roadway varying between a 2-lane collector and a 3-lane collector. Fourth Avenue is a one-way southbound 3-lane collector with a curb-to-curb width of 50 feet between I-5 and Walnut Avenue; a one-way southbound 2-lane collector with a curb-to-curb width of 45 feet between Walnut Avenue and Washington Street; and a two-way, 2-lane collector with a curb-to-curb width of 50 feet north of Washington Street. Fourth Avenue is lined with sidewalks and curbs, with parallel parking available on both sides of the street. The posted speed limit <u>is-ranges from 25 mph to 30 mph</u>. Currently classified as a Class III bicycle route south of Juniper Street, Fourth Avenue also has a striped enhanced Class II (buffered bicycle lane) between Elm Street and Laurel Street. The City BMP identifies Fourth Avenue as a Class III (Bike Route) facility between downtown and Upas Street, as a Class II (Bike Lanes) facility between Upas Street and Washington Street, and a Class III facility between Washington Street and Lewis Street.

Fifth Avenue functions at its adopted Community Plan ultimate classification as a one-way northbound 3-lane collector, with a curb-to-curb width of 50 feet between I-5 and Washington Street. Fifth Avenue is lined with sidewalks and curbs, with parallel parking available on both sides of the street. The posted speed limit <u>is-ranges from 25 mph to 30 mph</u>. Fifth Avenue is classified as a Class III bicycle route south of Laurel Street and has a striped enhanced Class II (buffered bicycle lane) between Elm Street and Laurel Street. The City BMP identifies Fifth Avenue as a Class III (Bike Lanes) facility between downtown and Washington Street, with the option of a Class III (Bike Route) between University Avenue and Washington Street.

Sixth Avenue functions at its adopted plan ultimate classification as a north-south 4-lane collector, with no center lane and a curb-to-curb width of 60 feet between I-5 and University Avenue. Sixth Avenue provides access to State Route 163 (SR-163) north of University Avenue. From Washington Street to University Avenue, Sixth Avenue functions as a 3-lane collector, with a curb-to-curb width of 65 feet. Sixth Avenue is lined with sidewalks and curbs, with parallel parking available on both sides of the street. Balboa Park runs along the east side of Sixth Avenue. The posted speed limit for Sixth Avenue is 30 mph, and it is classified as a Class III (Bike Route) facility south of Upas Street. The City BMP identifies Sixth Avenue as a Class II (Bike Lanes) facility between downtown and Upas Street.

Ninth Avenue is a short, two-way, north-south roadway. Ninth Avenue has a curb-to-curb width of 50 feet between University Avenue and Washington Street and a SR-163 southbound off-ramp connection. Ninth Avenue is lined with sidewalks and curbs with parallel parking available on both sides of the street. The posted speed limit is 25 mph.

Campus Avenue functions at its adopted Community Plan ultimate classification as a north-south 2lane collector with, a curb-to-curb width of 50 feet between Washington Street and Madison Avenue. It is currently functioning at its adopted plan ultimate classification. Campus Avenue is lined with sidewalks and curbs, with parking available on both sides of the street. Angle parking is available on the west side of the street between Madison Avenue and Monroe Avenue and between Van Buren Avenue and Tyler Avenue. Parallel parking is available along the other sections. The posted speed limit is 25 mph. *Cleveland Avenue* functions at its adopted Community Plan ultimate classification as a 2-lane collector with bike lanes, parallel parking, and sidewalks on both sides of the street. Cleveland Avenue has a curb-to-curb width of 50 feet between Washington Street and Madison Avenue. South of Washington Street, no bike lanes are provided, but parallel parking and sidewalks continue to line the street on both sides. Cleveland Avenue has a posted speed limit of 25 mph. The City BMP identifies Cleveland Avenue as a Class II (Bike Lanes) facility between Madison Avenue and Richmond Street.

Curlew Street functions at its adopted Community Plan ultimate classification as a 2-lane collector, with a curb-to-curb width of 40 feet between Reynard Way and Robinson Avenue. Curlew Street is lined with sidewalks and curbs, with parallel parking available on both sides of the street. The posted speed limit for Curlew Street is 25 mph. The City BMP identifies the entirety of Curlew Street as a Class III (Bike Route) facility.

Elm Street functions at its adopted Community Plan ultimate classification as a two-way 3-lane collector, with a curb-to-curb width of 50 feet from First Avenue to Second Avenue; a one-way westbound 2-lane collector, with a curb-to-curb width of 50 feet from Second Avenue to Third Avenue; and a 3-lane collector, with a curb-to-curb width of 50 feet between Third Avenue and Sixth Avenue. Elm Street is bounded by an I-5 northbound off-ramp on the east and a northbound I-5 on-ramp on the west. Elm Street is lined with sidewalks and curbs with parallel parking available on both sides of the street. The posted speed limit is 25 mph.

Fort Stockton Drive functions at its adopted Community Plan ultimate classification as a 2-lane collector, with a curb-to-curb width of 40 feet between Ampudia Street and Eagle Street. Fort Stockton Drive is lined with sidewalks and curbs and has parallel parking available on both sides of the street. Class III (Bike Lanes) are provided on Fort Stockton Drive between Witherby Street and Hermosa Way. The posted speed limit is 25 mph.

Front Street is non-continuous roadway through the Uptown community, with breaks between Washington Street and University Avenue, Robinson Avenue and Brookes Avenue, Spruce Street and Maple Street, and Fir Street and Date Street. For areas south of Washington Street, Front Street is a two-lane roadway, with parking allowed that serves residential areas. The portion of Front Street north of Washington Street provides access to UCSD Medical Center and is a key circulation roadway. The portion of Front Street located north of Washington Street functions as a north-south two-way 2-lane collector, with a curb-to-curb width of 40 feet between Dickinson Street and Arbor Drive; a one-way southbound 2-lane collector, with a curb-to-curb width of 40 feet between Arbor Drive and Lewis Street; and a one-way southbound 3-lane collector, with a curb-to-curb width of 50 feet between Lewis Street and Washington Street. Based on the adopted Community Plan, the ultimate classification for Front Street is a 3-lane collector between Arbor Drive and Washington Street is a 3-lane collector between Arbor Drive and Washington Street is a a-lane collector between Arbor Drive and Washington Street. The posted speed limit is 25 mph. Front Street is lined with sidewalks and curbs with parallel parking available on both sides of the street.

Grape Street functions as a one-way eastbound, 3-lane collector, with a curb-to-curb width of 50 feet between I-5 and First Avenue; and as a two-way, 2-lane collector, with a curb-to-curb width of 50 feet between First Avenue and Sixth Avenue. The adopted Community Plan shows Grape Street with an ultimate classification as a 3-lane collector between First Avenue and Sixth Avenue. Grape Street is lined with sidewalks and curbs. Angle parking is available on the north side of the street between First Avenue and Fourth Avenue, on both sides of the street between Fourth Avenue and Fifth Avenue, and on the south side between Fifth Avenue and Sixth Avenue. The posted speed limit is 25 mph. The City BMP identifies Grape Street as a Class III (Bike Route) facility between First Avenue and Sixth Avenue.

Hawthorn Street functions as a one-way westbound 3-lane collector, with a curb-to-curb width of 50 feet from Brant Street to First Avenue; and as a two-way, 2-lane collector, with a curb-to-curb width of 50 feet from First Avenue to Sixth Avenue. The adopted Community Plan ultimate classification for Hawthorn Street is as a 3-lane collector for its entirety. Hawthorn Street is lined with sidewalks and curbs, with parking available on both sides of the street. Angle parking is available on the north side of the street between. Third Avenue and Sixth Avenue; parallel parking is available along the other sections. Access is provided to I-5 northbound from Hawthorn Street. The posted speed limit on Hawthorn Street is 30 mph. The City BMP identifies Hawthorn Street as a Class III (Bike Route) facility between First Avenue and Sixth Avenue.

India Street functions as a one-way northbound collector with a varying classification between 2 lanes and 3 lanes and between two-way and one-way between I-5 to San Diego Avenue. North of San Diego Avenue, India Street is a two-way, 2-lane collector until it terminates at Washington Street. The adopted Community Plan ultimate classification for India Street is a 3-lane collector south of Washington Street and a 2-lane collector north of Washington Street. India Street is lined with sidewalks and curbs, with parallel parking available on the east side of the street only. Running parallel to I-5, India Street provides access to I-5 northbound at San Diego Avenue. The posted speed limit on India Street is 35 mph. The City BMP identifies India Street as a Class II (Bike Lanes) facility between Laurel Street and Washington Street.

Juan Street provides access into the adjacent Old Town community and functions as a 2-lane collector, with a curb-to-curb width of 35 feet between Witherby Street and the community boundary. Juan Street was not included in the adopted Community Plan future classifications. Juan Street is lined with sidewalks and curbs, with parallel parking available on both sides of the street. The posted speed limit on Juan Street is 30 mph. The City BMP identifies Juan Street as a Class III (Bike Route) facility between Sunset Boulevard in the Uptown community and Taylor Street in the Old Town community.

Laurel Street functions as an east-west 4-lane collector, with a curb-to-curb width of 50 feet between I-5 and Union Street; and as a 2-lane collector with a two-way left-turn lane, and a curb-to-curb width of 50 feet between Union Street and Sixth Avenue. East of Sixth Avenue, Laurel Street enters Balboa Park and changes name to El Prado. Laurel Street's adopted Community Plan ultimate classification is a 2-lane collector. Laurel Street is lined with sidewalks and curbs, with parallel parking available on both sides of the street; the posted speed limit is 25 mph. The City BMP identifies Laurel Street as a Class III (Bike Route) facility between Reynard Way and Sixth Avenue, joining with the existing bike route in Balboa Park to the east.

Lewis Street functions at its adopted Community Plan ultimate classification as an east-west 2-lane collector, with a curb-to-curb width of 50 feet between Fort Stockton Drive and Hawk Street; and as a one-way, 2-lane eastbound collector, with a curb-to-curb width of 35 feet between Front Street and Fourth Avenue. Natural terrain severs Lewis Street between Goldfinch Street and Albatross Street. Bike lanes are provided between Fort Stockton Drive and Ibis Street. Lewis Street is lined with

sidewalks and curbs, with parallel parking available on both sides of the street between Fort Stockton Drive and Ibis Street; angle parking is available on the south side of the street between Ibis Street and Hawk Street. The posted speed limit is 25 mph.

Normal Street functions at its adopted Community Plan ultimate classification as a 4-lane major arterial with a curb-to-curb width of 110 feet between University Avenue and Washington Street; and as a 6-lane major arterial with a curb-to-curb width of 110 feet between Washington Street and Park Boulevard/El Cajon Boulevard. It is currently functioning at its adopted plan ultimate classification. Normal Street is lined with sidewalks and curbs on both sides of the street, with angled parking available on both sides of the street between University Avenue and Washington Street. The posted speed limit is 30 mph. The City BMP identifies Normal Street as a Class II (Bike Lanes) facility between Washington Street and El Cajon Boulevard.

Park Boulevard changes cross-sections multiple times throughout the Uptown community. It functions as a north-south 2-lane collector, with a two-way left-turn lane and a curb-to-curb width of 65 feet between Upas Street and Cypress Avenue; as a 3-lane collector (2 northbound, 1 southbound), with a curb-to-curb width of 65 feet between Cypress Avenue and Essex Street; as a 4lane major, with a curb-to-curb width of 110 feet between Essex Street and Normal Street/El Cajon Boulevard; as a 3-lane collector, with a curb-to-curb width of 50 feet between Normal Street/El Cajon Boulevard and Meade Avenue; and as a 2-lane collector with a continuous two- way left-turn lane and a curb-to-curb width of 50 feet between Meade Avenue and Adams Avenue. The ultimate classification for Park Boulevard in the adopted Community Plan is a 4-lane major between Upas Street and Washington Street. Park Boulevard is lined with sidewalks and curbs, with parking available on both sides of the street. Angle parking is available on both sides of the street between Normal Street and University Avenue; parallel parking is along the other sections. The posted speed limit for Park Boulevard is 35 mph between Upas Street and Washington Street and 30 mph north of Washington Street. Park Boulevard serves as the community boundary between the Uptown and North Park Community Plan areas. Beyond these communities, Park Boulevard continues into Balboa Park providing access to the attractions within the park including the San Diego Zoo. Park Boulevard is classified as a Class III bicycle facility. The City BMP identifies Park Boulevard as a Class II (Bike Lanes) facility between Adams Avenue and Upas Street and throughout Balboa Park, with the option of keeping Class III (Bike Route) facilities between Upas Street and El Cajon Boulevard/Normal Street and north of Madison Avenue.

Reynard Way functions at its adopted Community Plan ultimate classification as a 2-lane collector, with a continuous left-turn lane and a curb-to-curb width of 55 feet between Torrance Street and Maple Street. Reynard Way becomes Goldfinch Street north of Torrance Street and becomes State Street south of Maple Street. The posted speed limit along Reynard Way is 30 mph. It is currently functioning at its adopted plan ultimate classification. Reynard Way is lined with sidewalks and curbs on both sides of the street. The City BMP identifies the entirety of Reynard Way as a Class III (Bike Route) facility.

Richmond Street functions as a north-south 2-lane collector, with a curb-to-curb width of 50 feet between Upas Street and Washington Street. The adopted Community Plan ultimate classification for Richmond Street is as a 3-lane collector between Cleveland Avenue and Robinson Avenue and a 2-lane collector between Robinson Avenue and Upas Street. Richmond Street is lined with sidewalks and curbs, with parallel parking allowed on both sides of the street. The posted speed limit along Richmond Street is 25 mph. The City BMP identifies Richmond Street as a Class II (Bike Lanes) facility between Upas Street and Cleveland Avenue.

Robinson Avenue functions as an east-west 2-lane collector, with a curb-to-curb width of 35 feet between Curlew Street and Park Boulevard. Between Vermont Street and Park Boulevard, Robinson Avenue functions as a 2-lane collector, with a two-way left-turn lane and a curb-to-curb width of 50 feet. It is currently functioning at its adopted Community Plan ultimate classification. Robinson Avenue is lined with sidewalks and curbs with parallel parking available on both sides of the street, except between Fifth Avenue and Seventh Avenue, where parking is not available. Robinson Avenue provides access to and from SR-163 between Eighth Avenue and Tenth Avenue. The posted speed limit for Robinson Avenue is 25 mph between Curlew Street and Tenth Avenue and 30 mph between Tenth Avenue and Park Boulevard. The City BMP identifies Robinson Avenue as Class III (Bike Route) facility between First Avenue and Park Boulevard, and continuing east of Park Boulevard as a Bicycle Boulevard facility providing connection to Alabama Street.

San Diego Avenue functions at its adopted Community Plan ultimate classification as a 2-lane collector, with a curb-to-curb width of 50 feet between India Street and the community boundary. One segment of San Diego Avenue between McKee Street and Washington Street functions as a 3-lane collector, with a curb-to-curb width of 50 feet. The roadway is one-way northbound between California Street and India Street and provides a connection to the adjacent Old Town community. It is currently functioning at its adopted plan ultimate classification. San Diego Avenue is lined with sidewalks and curbs, with parking available on both sides of the street. Angle parking is available on the east side of the street between Washington Street and India Street; parallel parking is along the other sections. The posted speed limit is 35 mph. The City BMP identifies San Diego Avenue as a Class II (Bike Lanes) facility between India Street and Congress Street.

State Street functions as a 2-lane collector, with a curb-to-curb width of 50 feet, between Juniper Street and Laurel Street. State Street is lined with sidewalks and curbs, with parallel parking available on both sides of the street. The posted speed limit for State Street is 25 mph. The City BMP identifies State Street as a Class III (Bike Route) facility between Laurel Street and downtown.

Sunset Boulevard functions at its adopted Community Plan ultimate classification as an east-west 2lane collector, with bike lanes and a curb-to-curb width of 50 feet, between Witherby Street and Fort Stockton Drive. It is lined with sidewalks and curbs, the street has parallel parking available on both sides. It is currently functioning at its adopted plan ultimate classification. The posted speed limit of 25 mph.

University Avenue functions at its adopted Community Plan ultimate classification as an east-west 2lane collector, with a curb-to-curb width of 45 feet, between Washington Street and Fifth Avenue; as a 4-lane collector between Fifth Avenue and Eighth Avenue (varying between with and without a center lane); as a 4-lane major between Vermont Street and Normal Street; and a 4-lane collector between Normal Street and Park Boulevard. University Avenue has a curb-to-curb width of 60 feet between Fifth Avenue and Tenth Avenue, 110 feet between Tenth Avenue and Normal Street, and 50 feet between Normal Street and Park Boulevard. It is currently functioning at its adopted plan ultimate classification. University Avenue is lined with sidewalks and curbs on both sides of the street. Angle parking is available on both sides of the street between Vermont Street and Normal Street; parallel parking is available along the other sections between Fifth Avenue and Park Boulevard. The posted speed limit for University Avenue is 25 mph between Washington Street and Park Boulevard. University Avenue is classified as a Class III bicycle facility between Goldfinch Street and Third Avenue. The City BMP identifies University Avenue as a Class II (Bike Lanes) facility east of First Avenue beyond the community boundaries, with the option of a Class III (Bike Route) facility between Fifth Avenue and Florida Street.

Upas Street functions as an east-west 2-lane collector, with a curb-to-curb width of 50 feet between Front Street and Sixth Avenue, and provides access to Balboa Park. Upas Street is lined with sidewalks and curbs, with parallel parking available on both sides of the street. The posted speed limit for Upas Street is 25 mph and it is classified as a Class III bicycle facility east of Third Avenue. The City BMP identifies Upas Street as a Class III (Bike Route) facility between First Avenue and Third Avenue as well.

Washington Street functions at its adopted Community Plan ultimate classification as an east-west 4lane major, with a curb-to-curb width of 80 feet, between I-5 and Richmond Street; and as a 6-lane major between Richmond Street and Normal Street. Washington Street does not have sidewalks or curbs between I-5 and Hawk Street and between SR-163 and Lincoln Avenue. It is lined with sidewalks and curbs on both sides of the street throughout the rest of the segment. Parallel parking is available on select segments of Washington Street between Hawk Street and Park Boulevard. The posted speed limit for Washington Street is 45 mph between I-5 and Hawk Street and 35 mph from Hawk Street to Park Boulevard. Washington Street is classified as a Class II (Bike Lanes) facility between University Avenue and India Street. The City BMP identifies the entirety of Washington Street as a Class II (Bike Lanes) facility.

6.3.1.2 Roadway Segment Conditions

In order to determine the impacts on the study area roadway segments, Table 6.3-1 has been developed by the City of San Diego and is used as a reference. The segment traffic volumes under Level of Service (LOS) E as shown in this table are considered at capacity because at LOS E the v/c Ratio is equal to 1.0.

Table 6.3-1 City of San Diego Roadway Segment Capacity and Level of S <u>ervice</u>							
Road Class	Lanes	A	В	С	D	E	
Freeway	8	60,000	84,000	120,000	140,000	150,000	
Freeway	6	45,000	63,000	90,000	110,000	120,000	
Freeway	4	30,000	42,000	60,000	70,000	80,000	
Expressway	6	30,000	42,000	60,000	70,000	80,000	
Prime Arterial (two-way)	6	25,000	35,000	50,000	55,000	60,000	
Major Arterial (two-way)	6	20,000	28,000	40,000	45,000	50,000	
Major Arterial (two-way)	4	15,000	21,000	30,000	35,000	40,000	
Major Arterial (two-way)	3	11,250	15,750	22,500	26,250	30,000	
Major Arterial (one-way)	3	12,500	16,500	22,500	25,000	27,500	
Major Arterial (one-way)	2	10,000	13,000	17,500	20,000	22,500	
Collector (two-way)	4	10,000	14,000	20,000	25,000	30,000	
Collector (No center lane)	4	F 000	7,000	10,000	13,000	15,000	
(Continuous left-turn lane)	2	5,000					
Collector (No fronting property)	2	4,000	5,500	7,500	9,000	10,000	
Collector (two-way)	3	7,500	10,500	15,000	17,500	20,000	
Collector (no center turn lane)	3	4,000	5,500	7,500	10,000	11,500	
Collector (Commercial/Industrial fronting)	2	2,500	3,500	5,000	6,500	8,000	
Collector (Multi-family)	2	2,500	3,500	5,000	6,500	8,000	
Collector (one-way)	3	11,000	14,000	19,000	22,500	26,000	
Collector (one-way with one lane dedicated for bike facility)	3	7,500	9,500	12,500	15,000	17,500	
Collector (one-way)	2	7,500	9,500	12,500	15,000	17,500	
Collector (one-way)	1	2,500	3,500	5,000	6,250	7,500	
Sub-Collector (Single family)	2	-	_	2,200	_	_	
Notes:							

The volumes and the average daily level of service listed above are only intended as a general planning guideline.

Levels of service are not applied to residential streets since their primary purpose is to serve abutting lots, not carry through traffic. Levels of service normally apply to roads carrying through traffic between major trip generators and attractors.

Capacities for any classification not identified in the sources noted below were developed based on interpolation from similar classifications.

SOURCES: City of San Diego Traffic Impact Study Manual, Table 2, Page 8, July 1998. City of San Diego Planning Department Mobility Section.

Based on planning-level analysis using Average Daily Traffic (ADT) volumes, it is estimated that all roadway segments within the Uptown CPU area function at an acceptable LOS D or better, except for the following segments. The segments listed below have volumes near or above their existing capacity, resulting in periods of congestion.

- First Avenue between Washington Avenue and University Avenue (LOS E)
- First Avenue between University Avenue and Robinson Avenue (LOS F)
- First Avenue between Robinson Avenue and Pennsylvania Avenue (LOS E)
- First Avenue between Pennsylvania Avenue and Walnut Avenue (LOS E)
- First Avenue between Laurel Street and Hawthorn Street (LOS E)
- Fourth Avenue between Arbor Drive and Washington Avenue (LOS F)
- Sixth Avenue between University Avenue and Robinson Avenue (LOS F)
- Sixth Avenue between Robinson Avenue and Upas Street (LOS F)
- Sixth Avenue between Upas Street and Laurel Street (LOS F)
- Cleveland Avenue between Lincoln Street and Richmond Street (LOS E)
- Fort Stockton Drive between Hawk Street and Goldfinch Street (LOS F)
- India Street between Glenwood Drive and Sassafras Street (LOS F)
- India Street between Sassafras Street and Redwood Street (LOS E)
- Laurel Street between Columbia Street and Union Street (LOS E)
- Lincoln Avenue between Washington Street and Park Boulevard (LOS F)
- Park Boulevard between Adams Avenue and Mission Avenue (LOS E)
- Park Boulevard between Mission Avenue and El Cajon Boulevard (LOS F)
- Richmond Street between Cleveland Avenue and University Avenue (LOS E)
- Robinson Avenue between Third Avenue and Eighth Avenue (LOS F)
- University Avenue between Ibis Street and Albatross Street (LOS F)
- University Avenue between Albatross Street and First Avenue (LOS F)
- University Avenue between First Avenue and Fourth Avenue (LOS F)
- University Avenue between Fourth Avenue and Fifth Avenue (LOS F)
- University Avenue between Sixth Avenue and Eighth Avenue (LOS F)
- University Avenue between Normal Street and Park Boulevard (LOS F)
- Washington Street between Fifth Avenue and Sixth Avenue (LOS E)
- Washington Street between Sixth Avenue and Richmond Street (LOS F)

Figure 6.3-2 illustrates the existing ADT volumes along the roadway segments in the Uptown CPU study area.

Map Source: Kimley Horn

ATTACHMENT 7



FIGURE 6.3-2 Existing Roadway Segment ADT Volumes – Uptown

6.3.1.3 Intersection Conditions

The Traffic Impact Study (Appendix B-1) includes a LOS analysis for the study intersections within the Uptown CPU area under Existing Conditions. LOS for signalized intersections is defined in terms of delay, which is a measure of driver discomfort, frustration, fuel consumption, and loss of travel time. Specifically, LOS criteria are stated in terms of the average control delay per vehicle for the peak 15-minute period within the hour analyzed. The average control delay includes initial deceleration delay, queue move-up time, and final acceleration time in additional to the stop delay. The level of service for unsignalized intersections is determined by the computed or measured control delay and is defined for each minor movement. The criteria for the various levels of service designations for signalized and unsignalized intersections are given in Table 6.3-2.

Table 6.3-2							
Level of Service Criteria for Intersections							
LOS	Signalized (Control Delay) (sec/veh) ^a	Unsignalized (Control Delay) (sec/veh) ^b	Description				
А	≤10.0	≤10.0	Operations with very low delay and most vehicles do not stop.				
В	>10.0 and ≤20.0	>10.0 and ≤15.0	Operations with good progression but with some restricted movement.				
С	>20.0 and ≤35.0	>15.0 and ≤25.0	Operations where a significant number of vehicles are stopping with some backup and light congestion				
D	>35.0 and ≤55.0	>25.0 and ≤35.0	Operations where congestion is noticeable, longer delays occur, and many vehicles stop. The proportion of vehicles not stopping declines.				
E	>55.0 and ≤80.0	>35.0 and ≤50.0	Operations where these is significant delay, extensive queuing, and poor progression.				
F	>80.0	>50.0	Operations that are unacceptable to most drivers, when the arrival rates exceed the capacity of the intersection.				
SOURCES:							
^a 2000 Highway Capacity Manual, Chapter 16, Page 2, Exhibit 16-2							
^b 2000 Highway Capacity Manual, Chapter 17, Page 2, Exhibit 17-2							

Within the City of San Diego, all signalized and unsignalized intersections are considered deficient if they operate at LOS E or F. All Uptown CPU study area intersections currently operate at LOS D or better during both peak periods, except for the following intersection that operates at LOS F during the p.m. peak period:

• Washington Street & Eighth Ave/SR-163 Off-Ramp (LOS F – PM peak)

At the intersection of Washington Street and SR-163, the eastbound through volumes are over 2,100 during the PM peak period. The existing two eastbound lanes do not have the capacity to adequately handle this demand.

6.3.1.4 Freeway Segments

Table 6.3-3 identifies California Department of Transportation (Caltrans) criteria used to rate freeway segment operations based on a LOS scale from A to F.

Table 6.3-3						
Level of Service Criteria for Freeway Segment Analysis						
LOS	v/c ratio	Congestion/Delay	Traffic Description			
А	<0.41	None	Free Flow			
В	0.41 – 0.62	None	Free to stable flow, light to moderate volumes			
С	0.63 – 0.80	None to minimal	Stable flow, moderate volumes, freedom to maneuver noticeably restricted			
D	0.81 – 0.92	Minimal to substantial	Approaches unstable flow, heavy volumes, and very limited freedom to maneuver			
E	0.93 – 1.00	Significant	Extremely unstable flow, maneuverability and psychological comfort extremely poor			
Fo	1.01 – 1.25	Considerable 0-1 hour delay	Operations that are unacceptable to most drivers, when the arrival rates exceed the capacity of the intersection			
F ₁	1.26 - 1.35	Severe 1-2 hour delay	Forced flow, heavy congestion, long queues form behind breakdown points, stop and go			
F ₂	1.36 - 1.45	Very severe 2-3 hour delay	Extremely heavy congestion, very long queues			
F ₃	>1.46	Extremely severe 3+ hour delay	Gridlock			
Notes: Source: Ca	ltrans Guideline	s 1992				

Freeway volumes were obtained from Caltrans. Table 6.3-4 displays the LOS analysis results for the study freeway segments under existing conditions. As shown in the table, the freeway segments surrounding the Uptown CPU area have volumes that exceed the capacity during peak hours. In general, the failing segments are those that move traffic away from study area in the morning and towards the study area in the afternoon.

6.3.1.5 Freeway Ramp Metering

Ramp volumes were obtained from intersection turning movements data when applicable, or from Caltrans volumes. Table 6.3-5 displays the queuing analysis results for the ramps in the study area that are currently metered. The table compares the peak hour demand at the on-ramp with the current meter rate. As shown in the table, the meter rate adequately controls the expected demand without excess queuing (in excess of 15 minutes) for all ramp meters in the Uptown CPU area.

Table 6.3-4									
Existing Freeway Segment Level of Service									
					2-Way				
					Peak				
					Hour	D			
		Number of	Capacity		Volume	(Directional	Peak-Hour	V/C	
Freeway Segment	Direction	Lanes	(A)	ADT (B)	(B)	Split)	Volume (C)	Ratio	LOS
AM PEAK									
I-5									
Old Town Ave to Washington	NB	4 M + 1 A	9,200	196,000	15,600	0.560	8,736	0.95	E
St	SB	4 M + 1 A	9,200			0.440	6,864	0.75	С
Washington St to Pacific	NB	4 M	8,000	148,000	12,000	0.560	6,720	0.84	D
Highway	SB	4 M	8,000			0.440	5,280	0.66	С
First Ave to Sixth Ave	NB	4 M + 1 A	9,200	201,000	15,500	0.750	11,625	1.26	F ₁
	SB	5 M + 1 A	11,200			0.250	3,875	0.35	А
SR-163 to SR-94	NB	5 M + 1 A	11,200	210,000	16,200	0.750	12,150	1.08	Fo
	SB	5 M + 1 A	11,200			0.250	4,050	0.36	А
SR-94 to Imperial Ave	NB	4 M + 1 A	9,200	164,000	12,700	0.750	9,525	1.04	Fo
	SB	4 M + 1 A	9,200			0.250	3,175	0.35	А
I-8									
Hotel Circle (W) to Hotel	WB	4 M + 1 A	9,200	208,000	16,500	0.570	9,405	1.02	Fo
Circle (E)	EB	4 M	8,000			0.430	7,095	0.89	D
Mission Center Rd to	WB	4 M + 1 A	9,200	224,000	17,900	0.570	10,203	1.11	Fo
Qualcomm Wy	EB	4 M + 1 A	9,200			0.430	7,697	0.84	D
I-805 to SR-15	WB	4 M + 1 A	9,200	242,000	19,100	0.650	12,415	1.35	F ₁
	EB	4 M + 1 A	9,200			0.350	6,685	0.73	С
SR-15									
I-805 to SR-94	NB	3 M + 1 A	7,200	96,000	8,900	0.430	3,827	0.53	В
	SB	2 M + 1 A	5,200			0.570	5,073	0.98	E
I-805	I-805								
I-8 to Adams Ave	NB	4 M + 1 A	9,200	192,000	15,900	0.730	11,607	1.26	F ₁
	SB	5 M + 1 A	11,200			0.270	4,293	0.38	А
El Cajon Blvd to University	NB	4 M	8,000	171,000	14,600	0.330	4,818	0.60	В
Ave	SB	4 M + 1 A	9,200			0.670	9,782	1.06	Fo
University Ave to SR-15	NB	4 M + 1 A	9,200	169,000	13,000	0.330	4,290	0.47	В
	SB	4 M + 1 A	9,200			0.670	8,710	0.95	E
ATTACHMENT 7

Table 6.3-4												
		Existing	g Freeway Se	gment Leve	el of Service	e						
					2-Way							
					Peak							
					Hour	D						
		Number of	Capacity		Volume	(Directional	Peak-Hour	V/C				
Freeway Segment	Direction	Lanes	(A)	ADT (B)	(B)	Split)	Volume (C)	Ratio	LOS			
SR-94	•					•						
25th St to 28th St	WB	4 M	8,000	123,000	10,700	0.730	7,811	0.98	E			
	EB	4 M	8,000			0.270	2,889	0.36	А			
28th St to 30th St	WB	4 M	8,000	130,000	12,000	0.730	8,760	1.10	Fo			
	EB	4 M	8,000			0.270	3,240	0.41	А			
Broadway to SR-15	WB	4 M	8,000	144,000	13,300	0.730	9,709	1.21	Fo			
-	EB	4 M + 1 A	9,200			0.270	3,591	0.39	А			
SR-163			•	•	•	•						
I-8 to Washington St	NB	3 M + 1 A	7,200	126,000	10,100	0.410	4,141	0.58	В			
_	SB	3 M + 1 A	7,200			0.590	5,959	0.83	D			
Washington St to Robinson	NB	2 M	4,000	96,000	7,800	0.410	3,198	0.80	C			
Ave	SB	2 M	4,000			0.590	4,602	1.15	Fo			
Quince Dr to I-5	NB	2 M	4,000	108,000	10,100	0.350	3,535	0.88	D			
	SB	2 M	4,000			0.650	6,565	1.64	F ₂			
			PM	PEAK								
1-5												
Old Town Ave to Washington	NB	4 M + 1 A	9200	196,000	15,600	0.460	7,176	0.78	C			
St	SB	4 M + 1 A	9200			0.540	8,424	0.92	D			
Washington St to Pacific	NB	4 M	8000	148,000	12,000	0.460	5,520	0.69	C			
Highway	SB	4 M	8000			0.540	6,480	0.81	D			
First Ave to Sixth Ave	NB	4 M + 1 A	9200	201,000	15,500	0.640	9,920	1.08	Fo			
	SB	5 M + 1 A	11200			0.360	5,580	0.50	В			
SR-163 to SR-94	NB	5 M + 1 A	11200	210,000	16,200	0.640	10,368	0.93	E			
	SB	5 M + 1 A	11200			0.360	5,832	0.52	В			
SR-94 to Imperial Ave	NB	4 M + 1 A	9200	164,000	12,700	0.640	8,128	0.88	D			
	SB	4 M + 1 A	9200			0.360	4,572	0.50	В			
I-8												
Hotel Circle (W) to Hotel	WB	4 M + 1 A	9200	208,000	16,500	0.450	7,425	0.81	D			
Circle (E)	EB	4 M	8000			0.550	9,075	1.13	Fo			
Mission Center Rd to	WB	4 M + 1 A	9200	224,000	17,900	0.450	8,055	0.88	D			
Qualcomm Wy	EB	4 M + 1 A	9200			0.550	9,845	1.07	Fo			

Uptown Community Plan Update PEIR Page 6.3-15

ATTACHMENT 7

Table 6.3-4												
		Existing	g Freeway Se	gment Leve	el of Service	2						
					2-Way							
					Peak							
					Hour	D						
		Number of	Capacity		Volume	(Directional	Peak-Hour	V/C				
Freeway Segment	Direction	Lanes	(A)	ADT (B)	(B)	Split)	Volume (C)	Ratio	LOS			
I-805 to SR-15	WB	4 M + 1 A	9200	242,000	19,100	0.430	8,213	0.89	D			
	EB	4 M + 1 A	9200			0.570	10,887	1.18	F ₀			
SR-15									-			
I-805 to SR-94	NB	3 M + 1 A	7200	96,000	8,900	0.430	3,827	0.53	В			
	SB	2 M + 1 A	5200			0.570	5,073	0.98	E			
I-805												
I-8 to Adams Ave	NB	4 M + 1 A	9200	192,000	15,900	0.340	5,406	0.59	В			
	SB	5 M + 1 A	11200			0.660	10,494	0.94	E			
El Cajon Blvd to University	NB	4 M	8000	171,000	14,600	0.600	8,760	1.10	Fo			
Ave	SB	4 M + 1 A	9200			0.400	5,840	0.63	С			
University Ave to SR-15	NB	4 M + 1 A	9200	169,000	13,000	0.600	7,800	0.85	D			
	SB	4 M + 1 A	9200			0.400	5,200	0.57	В			
SR-94												
25th St to 28th St	WB	4 M	8000	123,000	10,700	0.300	3,210	0.40	А			
	EB	4 M	8000			0.700	7,490	0.94	E			
28th St to 30th St	WB	4 M	8000	130,000	12,000	0.300	3,600	0.45	В			
	EB	4 M	8000			0.700	8,400	1.05	Fo			
Broadway to SR-15	WB	4 M	8000	144,000	13,300	0.300	3,990	0.50	В			
	EB	4 M + 1 A	9200			0.700	9,310	1.01	Fo			
SR-163												
I-8 to Washington St	NB	3 M + 1 A	7200	126,000	10,100	0.620	6,262	0.87	D			
	SB	3 M + 1 A	7200			0.380	3,838	0.53	В			
Washington St to Robinson	pinson NB 2 M 4000 96,00		96,000	7,800	0.620	4,836	1.21	Fo				
Ave SB 2 M 4000					0.380	2,964	0.74	С				
Quince Dr to I-5	NB	2 M	4000	108,000	10,100	0.540	5,454	1.36	F ₂			
	SB	2 M	4000			0.460	4,646	1.16	Fo			

Bold values indicate freeway segments operating at LOS E or F.

M=Main Lane; A= Auxiliary Lane.

(a) The capacity is calculated as 2,000 ADT per main lane and 1,200 ADT per auxiliary lane

(b) Traffic volumes provided by Caltrans (2008)

(c) Peak-hour volume calculated by: (2-way Peak-Hour Volume)*(D)

Refer to Table 6.3-3 Level of Service Criteria for Freeway Segment Analysis for descriptions of LOS A through F₃.

Uptown Community Plan Update PEIR

6.3 Transportation and Circulation

Table 3.6-5											
	Existing F	reeway Ramp	Metering								
				Excess							
	Peak	Meter Rate ¹	Demand ²	Demand	Average						
On-Ramp	Period	(Veh/Hr)	(Veh/Hr)	(Veh/Hr)	Delay (Min)						
	I	Interstate 5									
Washington St to I-5 NB	AM	996	1020	24	1.4						
	PM	996	1034	38	2.3						
India St to I-5 NB	AM	996	915	0	0.0						
	PM	996	1066	70	4.2						
Hawthorn St to I-5 NB	AM	996	454	0	0.0						
	PM	996	842	0	0.0						
Hancock St to I-5 SB	AM	Ramp not me	tered in the AN	1 peak							
	PM	1140	1287	147	7.7						
Kettner Blvd to I-5 SB	AM	Ramp not me	tered in the AN	1 peak							
	PM	498	269	0	0.0						
Fifth Ave to I-5 SB	AM	Ramp not me	tered in the AN	1 peak							
	PM	996	1087	91	5.5						
	1	Interstate 8									
NB Texas St to I-8 EB	AM	Ramp not me	tered in the AN	1 peak							
	PM	498	465	0	0.0						
SB Texas St to I-8 EB	AM	Ramp not me	tered in the AN	1 peak							
	PM	1140	866	0	0.0						
		Interstate 805									
El Cajon Blvd to I-805 NB	AM	1140	860	0	0.0						
	PM	Ramp not me	tered in the PN	1 peak							
University Ave to I-805 NB	AM	1140	998	0	0.0						
	PM	Ramp not me	tered in the PM	1 peak							
		State Route 94									
28th St to SR-94 WB	AM	534	100	0	0.0						
	PM	Ramp not me	tered in the PN	1 peak							
32nd St/Broadway to SR-94 WB	AM	570	99	0	0.0						
	PM	Ramp not me	tered in the PN	1 peak							
25th St to SR-94 EB	AM	Ramp not me	tered in the AN	1 peak							
	PM	960	785	0	0.0						
28th St to SR-94 EB	AM	Ramp not me	tered in the AM	1 peak							
	PM	960	732	0	0.0						
32nd St/Broadway to SR-94 EB	AM	Ramp not me	tered in the AM	1 peak							
	PM	570	464	0	0.0						
	S	tate Route 16	3								
Washington St to SR-163 SB	AM	498	373	0	0.0						
	PM	Ramp not me	tered in the PM	1 peak							
Notes:											

1) Meter rate is the assumed peak hour capacity expected to be processed through the ramp meter (using Caltrans fast rate)

2) Demand is the peak hour demand using the on-ramp

6.3.1.6 Alternative Transportation Facilities

a. Transit

The Uptown CPU area is served by several local and rapid bus routes, providing several options along Washington Street, University Avenue, Reynard Way, Fort Stockton Drive, First Avenue, Fourth Avenue, Fifth Avenue, and Sixth Avenue, as well as connections to the adjacent communities. <u>Rapid Bus rapid transit (BRT)</u> was recently implemented along Park Boulevard, north of University Avenue. Each of the transit route roadways are popular for vehicles and bicyclists as well, providing a shared-use atmosphere for the different modes of travel. One missing transit connection that the community has expressed interest in providing is connection to the San Diego International Airport.

Planned transit routes within the Uptown CPU area include BRTRapid, light rail transit (LRT), and streetcar improvements as shown on Figure 6.3-3.

b. Bicycle Facilities

The City of San Diego Bicycle Master Plan establishes guidance on achieving an ideal bicycle environment throughout the City. Similarly, a key focus of The San Diego Regional Bicycle Plan (RBP) prepared by San Diego Association of Governments (SANDAG) is to develop an interconnected network of bicycle corridors to improve the connectivity and quality of bicycle facilities and their supporting facilities. While these documents look at citywide and regional goals, the same focuses to develop quality facilities are applied to the local street networks in the community of Uptown. The types of bicycle facilities delegated and applied to local networks include bicycle boulevards, bicycle paths (Class I), bicycle lanes (Class II), bicycle routes (Class III), and cycle tracks (Class IV). SANDAG's regional bicycle facilities planned for the Uptown Community Plan area to the Old Town and Midway communities are shown on Figure 6.3-4.

Uptown's location in the central portion of San Diego makes bicycling an attractive mode of transportation for this community, although geography challenges in the community can result in out of direction travel and steep hills. Canyons limit the ability to provide a continuous grid pattern of streets, limiting bicycle options for short trips within the community. Uptown is located adjacent to downtown San Diego, where many Uptown residents work. Class II (Bicycle Lanes) and III (Bicycle Route) facilities are provided on Fourth, Fifth, and Sixth Avenues, as well as on portions of downtown streets. Recent facility upgrades, such as buffered bicycle lanes, have resulted in a noticeable increase in cyclists along these routes. Uptown sits on a mesa, with the Mission Valley community to the north and the Old Town and Midway communities to the west. There are no connections down to Mission Valley from the Uptown community, and there are limited connections to the west. Class III bicycle routes provide the only existing connections, one on Presidio Drive (to Old Town) and one on Laurel Street (to Midway).

SANDAG's regional bicycle facilities planned for the Uptown Community Plan to the Old Town and Midway communities are shown on Figure 6.3-4. The recommended bicycle facility network for the Uptown Community Plan area that interfaces with the regional bicycle network is shown on Figure 6.3-5.



FIGURE 6.3-3 Planned Transit Services – Uptown

Map Source: KTU+A

ATTACHMENT 7



FIGURE 6.3-4 Regional Bicycle Plan – Uptown Map Source: SanGIS

ATTACHMENT 7



FIGURE 6.3-5 Existing and Planned Bicycle Facilities – Uptown

c. Pedestrian Facilities

Uptown is a large community, several miles wide and long in some places, with some challenging terrain for pedestrians. There are differences in the pedestrian environment throughout the community. Several areas have high pedestrian activity, but there are also large areas with low pedestrian activity. The low pedestrian activity areas are the residential areas challenged with steep terrain on the western side of the community. Additionally, SR-163 impedes pedestrian connectivity within the eastern portion of the community, providing crossings only on University Avenue and Robinson Avenue. There is one other pedestrian crossing farther south within Balboa Park near Laurel Street.

Near the edges of downtown and Balboa Park there is a mix of residential and commercial attractions that instigate a lot of pedestrian activity. People live and work in these areas of the community and the gridded street network helps with pedestrian connectivity. However, portions of that area have steep hills that make it difficult for pedestrians to walk long distances. The terrain encourages people to try to find parking close to their destination even though there are good pedestrian facilities available. Fourth, Fifth, and Sixth Avenues are all designated as Corridor Sidewalks south of Robinson Avenue, and several other streets are Connector Sidewalks.

Further north the terrain flattens out and there is a long stretch of high pedestrian activity area near University Avenue, about a block on either side from Washington Street into Greater North Park. There is also high pedestrian activity near the hospital area adjacent to and north of Washington Street. University Avenue and the adjacent sections of Fourth, Fifth, and Sixth Avenues are all designated as a combination of District and Corridor Sidewalks in this area. Washington Street is designated as a Corridor Sidewalk. Several other streets in the vicinity are Connector Sidewalks.

On the western side of the community, India Street is the main pedestrian attraction with its row of restaurants. It is classified as a Corridor Sidewalk north of Sassafras Street, and a Connector Sidewalk to the south. The section of India Street designated as a Corridor Sidewalk is an isolated pedestrian activity area with steep terrain, busy freeway connections, and wide streets creating barriers from other nearby residential and commercial areas. Washington Street between India Street and Goldfinch Street is a steep section of roadway with high traffic volumes and high speeds and does not provide any pedestrian facilities.

The Uptown Bikeways Project identified in the San Diego Regional Bike Plan includes improvements to enhance the east-west pedestrian connectivity across Fourth and Fifth Avenue along the following streets: Upas, Spruce, Quince, Nutmeg, Laurel, Juniper, Grape, and Elm.

6.3.2 Significance Determination Thresholds

Thresholds used to evaluate potential impacts related to Transportation and Traffic are based on applicable criteria in the California Environmental Quality Act (CEQA) Guidelines Appendix G and the City of San Diego CEQA Significance Determination Thresholds (2011). Thresholds are modified from the City's CEQA Significance Determination Thresholds to reflect the programmatic analysis for the proposed Uptown CPU and associated discretionary actions. A significant impact could occur if implementation of a proposed Uptown CPU and associated discretionary actions would:

- Result in an increase in projected traffic, which is substantial in relation to the existing traffic load and capacity of the street system including roadway segments, intersections, freeway segments, interchanges, or freeway ramps;
- 2) Conflict with adopted policies, plans, or programs supporting alternative transportation.

The City of San Diego and Caltrans have developed acceptable threshold standards to determine the significance of project impacts to intersections, roadway segments, freeway segments, and freeway ramp metering. At intersections, the measurement of effectiveness (MOE) is based on allowable increases in delay. Along roadway segments and freeway segments, the MOE is based on allowable increases in the volume-to-capacity (v/c) ratio. At a freeway ramp meter, the MOE is based on allowable increases in delay, measured in minutes. These thresholds, applicable to the analysis of transportation facilities (Issue 1) are summarized in Table 6.3-6 and further detailed below.

Table 6.3-6 Significance Criteria for Facilities in Study Area										
Facility	Measures of Effectiveness (MOE)	Significance Threshold ¹								
Intersection	Seconds of Delay	> 2.0 seconds at LOS E or > 1.0 second at LOS F								
Roadway Segment	ADT, v/c ratio	> 0.02 at LOS E or > 0.01 at LOS F								
Freeway Segment	v/c ratio	> 0.01 at LOS E or > 0.005 at LOS F								
Freeway Ramp Meter	Minutes of delay per vehicle	 > 2.0 minutes for freeway segments operating at LOS E, and >1.0 minutes for freeway segments operating at LOS F. The criteria only apply for ramp meters where the delay without project is 15 minutes or higher. 								
v/c = volume to c LOS = Level of Se ¹ Applies only wh Source: City of S	capacity ratio ervice en the facilities operate: Gan Diego Significance D	s at LOS E or F Petermination Thresholds 2011; Kimley Horn Traffic Impact								

a. Signalized and Unsignalized Intersections

LOS F is not acceptable for any approach leg except for side streets on an interconnected arterial system. If vehicle trips from a project cause an intersection approach leg to operate at LOS F, except in the cases of side streets on an interconnected arterial system, this would be considered a significant project traffic impact. At intersections that are expected to operate at LOS E or F without the project, the allowable increase in delay is two seconds at LOS E and one second at LOS F with the addition of the project. If vehicle trips from a project cause the delay at an intersection to increase by more than the allowable threshold, this would be considered a significant project impact. Also, if the project causes an intersection that was operating at an acceptable LOS to operate at LOS E or F, this would be considered a significant project impact.

b. Roadway Segments

For roadway segments that are forecasted to operate at LOS E or F with the project, the allowable increase in v/c ratio is 0.02 at LOS E and 0.01 at LOS F. If vehicle trips from a project cause the v/c ratio to increase by more than the allowable threshold, this would be considered a significant project traffic impact. Also, if the project causes a street segment that was operating at an acceptable LOS to operate at LOS E or F, this would be considered a significant impact.

Where the roadway segment operates at LOS E or F, if the intersections at the ends of the segment are calculated to operate at an acceptable LOS with the project; and a peak hour Highway Capacity Manual (HCM) arterial analysis for the same segment shows that the segment operates at an acceptable LOS with the project; then the project impacts would be less than significant. If analysis shows either the intersections or segment under the peak hour HCM analysis do not operate acceptably, the project impacts would be significant.

In certain instances, mitigation may not be required even if a roadway segment operates at LOS E or LOS F. In such cases the following three conditions must all be met:

- 1. The roadway is built to its ultimate classification per the adopted Community Plan;
- 2. The intersections on both ends of the failing segment operate at an acceptable LOS; and
- 3. An HCM arterial analysis indicates an acceptable LOS on the segment.

c. Freeway Segments

For freeway segments that are forecasted to operate at LOS E or F with the project, the allowable increase in v/c ratio is 0.01 at LOS E and 0.005 at LOS F. If vehicle trips from a project cause the v/c ratio to increase by more than the allowable threshold, this would be considered a significant project traffic impact. Also, if the project causes a freeway segment that was operating at an acceptable LOS to operate at LOS E or F, this would be considered a significant impact.

d. Freeway Ramp Metering

Ramp metering is a means of controlling the volume of traffic entering the freeway with the goal of improving the traffic operations and flow on the freeway main lanes. Freeway ramp meter analysis estimates the peak hour queues and delays at freeway ramps by comparing existing volumes to the meter rate at the given location. The excess demand, if any, forms the basis for calculating the maximum queues and maximum delays anticipated at each location. Substantial queues and delays can form where demand significantly exceeds the meter rate. This approach assumes a static meter rate throughout the course of the peak hour. However, Caltrans has indicated that the meter rates are continually adjusted based on the level of traffic using the on-ramp. To the extent possible, the meter rate is set such that the queue length does not exceed the available storage, smooth flows on the freeway mainline is maintained, and there is no interference to arterial traffic.

If vehicle trips from a project cause a metered ramp with a delay of 15 minutes per vehicle or higher to increase its delay by more than two minutes per vehicle, this would be considered a significant project traffic impact if the freeway segment operates at LOS E or F.

6.3.3 Impact Analysis

Issue 1 Traffic Circulation

Would the project result in an increase in projected traffic, which is substantial in relation to the existing traffic load and capacity of the street system including roadway segments, intersections, freeway segments, interchanges, or freeway ramps?

In order to assess potential impacts, this section provides a description of future community buildout conditions for the Uptown CPU area. Due to the nature of the project being an update to the proposed Uptown CPU and associated discretionary actions with no specific development project being proposed at this time, the analysis provided in this section is cumulative in nature. The analysis considers the existing conditions within the Uptown CPU area and evaluates impacts to applicable facilities within the Uptown CPU area after build-out of the proposed Uptown CPU and associated discretionary actions. Since the analysis is looking at impacts over the long term, through 2035, projected traffic volume increases associated with development in neighboring communities (Golden Hill and North Park) is included within the analysis.

a. Build-out Traffic Volumes

The future community build-out conditions were developed based on proposed Uptown CPU and associated discretionary actions build-out land use and network assumptions within the Uptown CPU area and superimposed on SANDAG 2035 regional model. The peak-hour intersection turning movements and roadway segment traffic data for the existing condition were obtained from several sources as detailed in the Traffic Impact Analysis. Where traffic volumes in the City's traffic model were determined not to represent existing conditions (based on comparing data with available count data), new traffic counts were obtained. Where appropriate, traffic counts from recent traffic studies were used, including the Hillcrest Mobility Study and University Avenue Mobility Plan. Traffic counts performed by the City in 2007 to calibrate the traffic planning model were also used. Traffic counts were taken in the remaining areas of the CPU area in 2010 or were obtained through the latest City of San Diego traffic count database (2010). Since traffic counts are now greater than two years old, with counts gathered between 2006 and 2010, validation was required to determine if the counts still represent current traffic conditions. Roadway segment ADT counts were compared to current (i.e., Year 2012 and 2013) City of San Diego and Caltrans machine counts and adjacent freeway ramp facilities to determine if the counts were still valid. It was concluded that traffic volumes were within a 10-percent fluctuation and thus were still valid for use. Thus, the traffic counts provide a good representation of volumes for existing conditions for a planning level study.

Model adjustments were incorporated to provide consistency with vehicular traffic counts collected for the proposed Uptown CPU and expected traffic patterns within the Uptown, North Park, and Golden Hill CPU areas. These adjustments included the following:

• For roadway segments where the difference between the City's calibrated 2008 model and the actual count data collected between 2006 and 2010 exceeded ten percent or 2,000 daily vehicles, the difference was subtracted or added to the Year 2035 forecast model to adjust

the future volume based on the discrepancy noted between the City's traffic model volumes and count data. For roadway segments that have existing daily volumes less than 5,000, no adjustments were applied to the future model volumes.

The resulting daily traffic volumes for the Uptown CPU area at build-out under the proposed Uptown CPU and associated discretionary actions are presented in Figure 6.3-6.

b. Intersection Analysis

Table 6.3-7 displays the LOS analysis results for the study intersections using existing lane configurations and the future peak-hour traffic volumes. As shown in Table 6.3-7 and summarized below, the Uptown CPU would have a cumulative traffic related impact at six of the thirty study intersections.

- **Impact 6.3-1**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the intersection of Washington Street and Fourth Avenue in the PM peak hour.
- **Impact 6.3-2:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the intersection of Washington Street and Eighth Avenue/SR-163 Off-Ramp in the AM peak hour.
- **Impact 6.3-3:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the intersection of Washington Street/Normal Street and Campus Avenue/Polk Avenue in the AM and PM peak hours.
- **Impact 6.3-4:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the intersection of University Avenue and Sixth Avenue in the PM peak hour.
- **Impact 6.3-5:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the intersection of Elm Street and Sixth Avenue in the AM peak hour.
- **Impact 6.3-6:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the intersection of Cedar Street and Second Avenue in the AM and PM peak hours.

c. Roadway Segment Analysis

Table 6.3-8 displays the LOS analysis results for roadway segments within the Uptown community using existing roadway classifications and the future peak-hour traffic volumes based on build-out of the CPU area. As shown in Table 6.3-8, the Uptown CPU would have a cumulative traffic related impact on 52 of the 105 roadway segments within the study area. Where impacts occur on consecutive segments of the same roadway, these impacts have been combined for clarity.



FIGURE 6.3-6 Build-out Proposed Land Use Roadway Segment ADT Volumes – Uptown

ATTACHMENT 7

	Table 6.3-7 Build-out Summary of Intersection Analysis - Untown													
	Du	Traffic	Book	Evict	Analysis -	Build	out							
	Intersection	Control	Hour	DELAY (A)	LOS (B)	DELAY (A)	LOS (B)	Δ	SIGNIFICANT?					
1	Washington St & Hancock St	Signal	AM	24.9	C	33.2	C	8.3	NO					
		- 0 -	PM	28.2	C	51.6	D	23.4	NO					
2	Washington St & San Diego Ave	Signal	AM	19.7	В	15.4	В	-6.3	NO					
		0	PM	17.6	В	21.9	С	6.3	NO					
3	Washington St & India St	Signal	AM	11.7	В	15.8	В	4.1	NO					
		0	PM	14.2	В	20.3	С	6.1	NO					
4	Washington St & Fourth Ave	Signal	AM	25.2	С	31.8	С	6.6	NO					
		0	PM	37.3	D	59.9	Е	22.6	YES					
5	Washington St & Fifth Ave	Signal	AM	15.2	В	14.1	В	-1.1	NO					
	_		PM	16.3	В	19.2	В	2.9	NO					
6	Washington St & Eighth Ave/SR-163	Signal	AM	42.6	D	71.5	E	28.9	YES					
	Off-Ramp		PM	333.0	F	331.7	F	-1.3	NO					
7	Washington St & Richmond St/SR-	Signal	AM	18.6	В	51.4	D	32.8	NO					
	163 On-Ramp		PM	13.2	В	33.9	С	20.7	NO					
8	Washington St/Normal St & Campus	Signal	AM	43.0	D	62.7	E	19.7	YES					
	Ave/Polk Ave		PM	50.0	D	57.3	E	7.3	YES					
9	Normal St/El Cajon Blvd & Park Blvd	Signal	AM	25.2	С	26.6	С	1.4	NO					
			PM	36.3	С	43.8	D	9.5	NO					
10	University Ave & Fourth Ave	Signal	AM	29.1	С	31.8	С	2.7	NO					
			PM	28.2	С	30.3	С	2.1	NO					
11	University Ave & Fifth Ave	Signal	AM	12.9	В	13.7	В	0.8	NO					
			PM	25.3	С	28.0	С	2.7	NO					
12	University Ave & Sixth Ave	Signal	AM	32.9	С	38.7	D	5.8	NO					
			PM	54.8	D	55.3	E	0.5	YES					
13	University Ave & Tenth St	Signal	AM	18.6	В	17.5	В	-1.1	NO					
			PM	20.6	С	37.0	D	16.4	NO					
14	University Ave & Normal St	Signal	AM	5.6	А	6.3	А	0.7	NO					
			PM	10.6	В	13.3	В	2.7	NO					
15	University Ave & Park Blvd	Signal	AM	24.5	С	25.2	С	0.7	NO					
			PM	39.4	D	42.1	D	2.7	NO					
16	Robinson Ave & Fourth Ave	Signal	AM	21.4	С	27.0	С	5.6	NO					
			PM	18.4	В	20.8	С	2.4	NO					
17	Robinson Ave & Fifth Ave	Signal	AM	10.8	В	12.5	В	1.7	NO					
			PM	15.0	В	17.5	В	2.5	NO					

ATTACHMENT 7

	Table 6.3-7												
Build-out Summary of Intersection Analysis - Uptown													
		Traffic	Peak	Exist	ing	Build	lout						
	Intersection	Control	Hour	DELAY (A)	LOS (B)	DELAY (A)	LOS (B)	Δ	SIGNIFICANT?				
18	Robinson Ave & Sixth Ave	Signal	AM	21.6	С	22.7	С	1.1	NO				
			PM	27.6	С	30.9	С	3.3	NO				
19	Vine St & India St	Signal	AM	5.6	А	5.9	А	0.3	NO				
			PM	7.3	А	8.5	А	1.2	NO				
20	Sassafras St & Kettner Blvd	Signal	AM	10.4	В	13.2	В	2.8	NO				
			PM	12.5	В	43.6	D	31.1	NO				
21	Sassafras St & India St	Signal	AM	6.3	А	8.4	А	2.1	NO				
		_	PM	20.9	С	47.4	D	26.5	NO				
22	Laurel St & India St/I-5 NB On-Ramp	Signal	AM	17.0	В	19.7	В	2.7	NO				
		_	PM	21.4	С	29.5	С	8.1	NO				
23	Laurel St & Fourth Ave	Signal	AM	12.2	В	13.8	В	1.6	NO				
		_	PM	14.9	В	23.8	С	8.9	NO				
24	Laurel St & Fifth Ave	Signal	AM	12.3	В	13.3	В	1.0	NO				
		-	PM	12.7	В	17.8	В	5.1	NO				
25	Laurel St & Sixth Ave	Signal	AM	13.7	В	15.8	В	2.1	NO				
		-	PM	20.5	С	27.9	С	7.4	NO				
26	Hawthorn St & Brant St	Two-Way	AM	9.9	A (SB R)	10.0	B (SB R)	0.1	NO				
		Stop	PM	12.9	B (SB R)	12.9	B (SB R)	0.0	NO				
27	Grape St & State St	Signal	AM	15.7	В	12.6	В	-3.1	NO				
		J	PM	18.7	В	41.7	D	23.0	NO				
28	Elm St & First Ave	Signal	AM	13.3	В	17.8	В	4.5	NO				
		U	PM	21.6	С	21.0	С	-0.6	NO				
29	Elm St & Sixth Ave	Signal	AM	54.4	D	153.6	F	99.2	YES				
			PM	14.8	В	18.8	В	4.0	NO				
30	Cedar St & Second Ave	Two-Way	AM	31.8	D (SB R)	459.3	F (SB L)	427.5	YES				
		Stop	РМ	18.0	C (SB R)	43.0	E (SB L)	25.0	YES				

Notes:

Bold values indicate intersections operating at LOS E or F.

ECL = Exceeds Calculable Limit. Reported when delay exceeds 180 seconds.

(a) Delay refers to the average control delay for the entire intersection, measured in seconds per vehicle. At a one-way or two-way stopcontrolled intersection, delay refers to the worst movement.

(b) LOS calculations are based on the methodology outlined in the 2000 Highway Capacity Manual and performed using Synchro 8.

SB R = Southbound right-turning movement.

SB L = Southbound left-turning movement.

Table 6.3-8 Buildout Summary of Roadway Segment Analysis: Uptown											
			Exist	ting		Build	lout				
				V/C			V/C				
	Roadway Functional	LOS E		Ratio			Ratio		Δin	Δin	
Roadway Segment	Classification	Capacity	ADT	(a)	LOS	ADT	(a)	LOS	ADT	V/C	Significant?
Arbor Dr to Washington											
St	2 Lane Collector (one-way)	17,500	5,240	0.299	A	7,500	0.429	В	2260	0.130	NO
Washington St to University Ave	2 Lane Collector (No center lane)	8,000	7,400	0.925	E	9,100	1.138	F	1700	0.213	YES
University Ave to Robinson Ave	2 Lane Collector (No center lane)	8,000	10,100	1.263	F	16,300	2.038	F	6200	0.775	YES
Robinson Ave to Pennsylvania Ave	2 Lane Collector (No center lane)	8,000	7,500	0.938	E	11,500	1.438	F	4000	0.500	YES
Pennsylvania Ave to Walnut Ave	2 Lane Collector (No center lane)	8,000	7,261	0.908	E	12,800	1.600	F	5539	0.692	YES
Walnut Ave to Laurel St	2 Lane Collector (No center lane)	8,000	4,695	0.587	С	11,900	1.488	F	7205	0.901	YES
Laurel St to Hawthorn St	2 Lane Collector (No center lane)	8,000	7,290	0.911	E	8,400	1.050	F	1110	0.139	YES
Hawthorn St to Grape St	2 Lane Collector (No center lane)	8,000	3,810	0.476	С	6,800	0.850	E	2990	0.374	YES
Grape St to Elm St	2 Lane Collector (one-way)	17,500	3,285	0.188	А	4,500	0.257	А	1215	0.069	NO
Fourth Ave											
Arbor Dr to Washington St	2 Lane Collector (No center lane)	8,000	12,390	1.549	F	14,900	1.863	F	2510	0.314	YES
Washington St to University Ave	2 Lane Collector (one-way)	17,500	10,400	0.594	С	10,400	0.594	С	0	0.000	NO
University Ave to Robinson Ave	2 Lane Collector (one-way)	17,500	11,800	0.674	С	12,900	0.737	D	1100	0.063	NO
Robinson Ave to Walnut Ave	2 Lane Collector (one-way)	17,500	6,946	0.397	А	11,400	0.651	С	4454	0.254	NO
Walnut Ave to Laurel St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	8,492	0.485	В	15,100	0.863	E	6608	0.378	YES
Laurel St to Grape St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	7,790	0.445	В	13,700	0.783	D	5910	0.338	NO

Table 6.3-8											
	Buildout Su	mmary of Ro	adway Se	gment A	Analysi	s: Uptown]				
			Exist	ting		Build	lout	-			
				V/C			V/C				
	Roadway Functional	LOS E		Ratio			Ratio		Δin	Δin	
Roadway Segment	Classification	Capacity	ADT	(a)	LOS	ADT	(a)	LOS	ADT	V/C	Significant?
Grape St to Elm St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	7,570	0.433	В	9,700	0.554	с	2130	0.121	NO
Fifth Ave											
Washington St to University Ave	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	11,700	0.669	С	11,800	0.674	с	100	0.005	NO
University Ave to Robinson Ave	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	10,300	0.589	с	14,000	0.800	D	3700	0.211	NO
Robinson Ave to Walnut Ave	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	12,209	0.698	с	15,800	0.903	E	3591	0.205	YES
Walnut Ave to Laurel St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	11,400	0.651	с	14,800	0.846	D	3400	0.195	NO
Laurel St to Hawthorn St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	9,260	0.529	В	14,400	0.823	D	5140	0.294	NO
Hawthorn St to Grape St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	10,045	0.574	С	14,300	0.817	D	4255	0.243	NO
Grape St to Elm St	3 Lane Collector (one-way w/ one lane dedicated for multi- modal)	17,500	9,220	0.527	В	10,100	0.577	с	880	0.050	NO
Sixth Ave											
Washington St to University Ave	4 Lane Collector (no center lane)	15,000	16,877	0.844	D	45,100	3.007	F	28223	2.163	YES
University Ave to Robinson Ave	4 Lane Collector (no center lane)	15,000	24,900	1.660	F	32,600	2.173	F	7700	0.513	YES
Robinson Ave to Upas St	4 Lane Collector (no center lane)	15,000	15,000	1.000	F	29,900	1.993	F	14900	0.993	YES
Upas St to Laurel St	4 Lane Collector (no center lane)	15,000	15,128	1.009	F	25,900	1.727	F	10772	0.718	YES

Uptown Community Plan Update PEIR Page 6.3-31

Table 6.3-8											
	Buildout Sur	nmary of Roa	adway Se	gment A	nalysi	s: Uptown					
			Exist	ing		Build	lout				
				V/C			V/C				
	Roadway Functional	LOS E		Ratio			Ratio		Δin	Δin	
Roadway Segment	Classification	Capacity	ADT	(a)	LOS	ADT	(a)	LOS	ADT	V/C	Significant?
Laurel St to Juniper St	4 Lane Collector (no center lane)	15,000	10,140	0.676	D	16,600	1.107	F	6460	0.431	YES
Juniper St to Grape St	4 Lane Collector (no center lane)	15,000	10,915	0.728	D	18,700	1.247	F	7785	0.519	YES
Grape St to Elm St	4 Lane Collector (no center lane)	15,000	10,650	0.710	D	20,300	1.353	F	9650	0.643	YES
Ninth Ave			-					-			
Washington St to	2 Lana Collector (No contor Jana)	8 000	5 204	0.651	П	8 000	1 000	E	2706	0.240	VEC
University Ave		8,000	3,204	0.051	D	8,000	1.000	F	2790	0.549	TLS
Campus Ave/Polk Ave											
Madison Ave to	2 Lane Collector (No center lane)	8 000	3 1 7 5	0 397	B	5 800	0 725	р	2625	0 328	NO
Washington St		0,000	3,173	0.557		5,000	0.725		2025	0.520	110
Washington St to Park	2 Lane Collector (No center lane)	8,000	5,610	0.701	D	7,400	0.925	Е	1790	0.224	YES
BIVO		-									
			4.005	0.000	6	7 2 0 0	0.000	-	2225	0.000	VEC
Tyler St to Lincoln Ave	2 Lane Collector (No center lane)	8,000	4,865	0.608	C	7,200	0.900	E	2335	0.292	YES
Lincoln Ave to	2 Lane Collector (No center lane)	8,000	7,775	0.972	Е	9,600	1.200	F	1825	0.228	YES
Curiew St											
Robinson Ave to	2 Lane Collector (No center lane)	8,000	1,720	0.215	А	4,600	0.575	С	2880	0.360	NO
EIIII SL	2 Lana Collector (and waw)	17 500	7 0 0 0	0.451	D	9 5 0 0	0.496	Р	611	0.025	NO
Second Ave to Third Ave	2 Lane Collector (one-way)	17,500	7,009	0.451		8,500	0.460		021	0.035	NO
Fifth Ave to Firth Ave	3 Lane Collector (one-way)	26,000	6,179	0.315	A	9,100	0.350	A	921	0.035	NO
	3 Lane Collector (one-way)	26,000	6,720	0.258	А	8,100	0.312	A	1380	0.054	NO
Arista St to Support Plud	2 Lana Collector (No contar Jana)	<u> </u>	2 200	0 411	D	4 000	0.612	C	1610	0 202	NO
Support Divid to Llowik St	2 Lane Collector (No center lane)	8,000	5,290	0.411		4,900	0.015	E	1010	0.202	
Sunset Bivu to Hawk St	2 Lane Collector (No Center Jane)	8,000	6,100	0.763		7,900	0.988	E F	1800	0.225	YES
	2 Lane Collector (No center lane)	8,000	8,450	1.056	F	8,900	1.113	F	450	0.057	YES
St	2 Lane Collector (No center lane)	8,000	2,910	0.364	В	3,300	0.413	В	390	0.049	NO
Front St											
Dickinson St to Arbor Dr	2 Lane Collector (No center lane)	8,000	3,790	0.474	С	4,600	0.575	С	810	0.101	NO
Arbor Dr to Washington St	2 Lane Collector (one-way)	17,500	5,510	0.315	А	7,900	0.451	В	2390	0.136	NO

ATTACHMENT 7 Table 6.3-8 **Buildout Summary of Roadway Segment Analysis: Uptown** Existing Buildout V/C V/C Roadway Functional LOS E Ratio Ratio Δin Δin Classification ADT ADT V/C Significant? Capacity (a) LOS ADT (a) LOS 3 Lane Collector (one-way) 26,000 0.080 0.201 NO 2,082 А 7,300 0.281 А 5218 2 Lane Collector (No center lane) 8,000 0.536 С 7,300 0.913 Ε 0.377 YES 4,289 3011 2 Lane Collector (No center lane) YES 8,000 2,097 0.262 А 9,000 1.125 F 6903 0.863

Brant St to First Ave	3 Lane Collector (one-way)	26,000	11,558	0.445	В	15,000	0.577	С	3442	0.132	NO
First Ave to Third Ave	2 Lane Collector (No center lane)	8,000	3,634	0.454	С	7,300	0.913	E	3666	0.459	YES
Third Ave to Sixth Ave	2 Lane Collector (No center lane)	8,000	3,577	0.447	С	8,700	1.088	F	5123	0.641	YES
India St											
Washington St to Winder St	2 Lane Collector (No center lane)	8,000				11,000	1.375	F	-	-	YES
Winder St to Glenwood Dr	3 Lane Collector (one-way)	26,000	8,345	0.321	A	10,700	0.412	A	2355	0.091	NO
Glenwood Dr to Sassafrass St	2 Lane Collector (one-way)	17,500	26,178	1.496	F	30,000	1.714	F	3822	0.218	YES
Sassafras St to Redwood St	3 Lane Collector (two-way)	20,000	18,676	0.934	Е	21,300	1.065	F	2624	0.131	YES
Redwood St to Palm St	3 Lane Collector (one-way)	26,000	16,705	0.643	С	20,300	0.781	D	3595	0.138	NO
Juan St											
Harney St to Witherby St	2 Lane Collector (No center lane)	8,000	2,345	0.293	А	4,600	0.575	С	2255	0.282	NO
Laurel St											
Columbia St to Union St	4 Lane Collector (no center lane)	15,000	13,691	0.913	Ε	21,100	1.407	F	7409	0.494	YES
Union St to First Ave	2 Lane Collector (continuous left- turn lane)	15,000	11,128	0.742	D	17,900	1.193	F	6772	0.451	YES
First Ave to Third Ave	2 Lane Collector (continuous left- turn lane)	15,000	11,326	0.755	D	16,100	1.073	F	4774	0.318	YES
Third Ave to Sixth Ave	2 Lane Collector (continuous left- turn lane)	15,000	11,516	0.768	D	20,200	1.347	F	8684	0.579	YES
Lewis St						-	-				
Fort Stockton Dr to Goldfinch St	2 Lane Collector (No center lane)	8,000	3,720	0.465	С	4,100	0.513	С	380	0.048	NO

Roadway Segment

Albatross St to First Ave

First Ave to Third Ave

Third Ave to Sixth Ave

Hawthorn St

Grape St

ATTACHMENT 7

Table 6.3-8											
	Buildout Sur	nmary of Roa	adway Se	gment A	nalysi	s: Uptown					
			Exist	ing		Build	out				
				V/C			V/C				
	Roadway Functional	LOS E		Ratio			Ratio		Δin	Δin	
Roadway Segment	Classification	Capacity	ADT	(a)	LOS	ADT	(a)	LOS	ADT	V/C	Significant?
Washington St to Park	2 Lane Collector (No center lane)	8,000	8,155	1.019	F	11,100	1.388	F	2945	0.369	YES
Bivu Madisan Ava											
Cloveland Ave to Dark											
Rivd	2 Lane Collector (No center lane)	8,000	3,750	0.469	С	6,100	0.763	D	2350	0.294	NO
Meade Ave											
Cleveland Ave to Park	2 Lane Collector (continuous left-										
Blvd	turn lane)	15,000	3,290	0.219	А	3,500	0.233	A	210	0.014	NO
Normal St				l			l				
Park Blvd to		50.000	22.200	0.446	5	20,200	0.566	C	6004	0.120	NO
Washington St	6 Lane Major Arterial	50,000	22,296	0.446	В	28,300	0.566	C	6004	0.120	NO
Washington St to	4 Lane Major Arterial	40,000	4,974	0.124	А						
University Ave	2 Lane Collector (No center	8 000				1 97/	0.622	C	0	0.498	NO
	lane)*	8,000				4,574	0.022	C			
Park Blvd			1	1			1	1	1	1	
Adams Ave to Mission	2 Lane Collector (continuous left-	15.000	14.839	0.989	Е	14.060	0.937	Е	-779	-0.052	NO
Ave	turn lane)		,			.,		_			
Mission Ave to El Cajon	3 Lane Collector (no center lane)	11,500	11,806	1.027	F	15,467	1.345	F	3661	0.318	YES
BIVO			-								
El Cajon Bivo to Polk	4 Lane Major Arterial	40,000	11,524	0.288	А	18,600	0.465	В	7076	0.177	NO
Ave Bolk Ave to University											
	4 Lane Major Arterial	40,000	13,936	0.348	А	22,500	0.563	С	8564	0.215	NO
Liniversity Ave to											
Robinson Ave	4 Lane Major Arterial	40,000	14,400	0.360	А	19,800	0.495	В	5400	0.135	NO
Robinson Ave to Upas	2 Lane Collector (continuous left-							-			
St	turn lane)	15,000	12,501	0.833	D	17,200	1.147	F	4699	0.314	YES
Upas St to Zoo Pl	4 Lane Major Arterial	40,000	13,807	0.345	А	17,700	0.443	В	3893	0.098	NO
Reynard Wy				•			•			•	
Torranco St to Curlow St	2 Lane Collector (continuous left-	15 000	1 055	0 1 2 0	٨	5 200	0.252	D	2215	0 222	NO
Tomatice St to Curiew St	turn lane)	15,000	1,955	0.150	А	5,500	0.555	D	5545	0.223	NO

Table 6.3-8											
	Buildout Sur	nmary of Roa	adway Se	gment P	nalysi	s: Uptown					
	Roadway Functional	LOS E	EXIS	V/C Ratio		Build	V/C Ratio		Δin	Δin	
Roadway Segment	Classification	Capacity	ADT	(a)	LOS	ADT	(a)	LOS	ADT	V/C	Significant?
Curlew St to Laurel St	2 Lane Collector (continuous left- turn lane)	15,000	7,200	0.480	С	8,600	0.573	С	1400	0.093	NO
Richmond St				1	1		1	1	1	1	1
Cleveland Ave to University Ave	2 Lane Collector (No center lane)	8,000	7,085	0.886	E	9,000	1.125	F	1915	0.239	YES
University Ave to Robinson Ave	2 Lane Collector (No center lane)	8,000	5,345	0.668	D	6,700	0.838	E	1355	0.170	YES
Robinson Ave to Upas St	2 Lane Collector (No center lane)	8,000	5,015	0.627	D	8,100	1.013	F	3085	0.386	YES
Robinson Ave				•			•			•	
Brant St to First Ave	2 Lane Collector (No center lane)	8,000	1,995	0.249	А	4,600	0.575	С	2605	0.326	NO
First Ave to Third Ave	2 Lane Collector (No center lane)	8,000	5,800	0.725	D	11,500	1.438	F	5700	0.713	YES
Third Ave to Eighth Ave	2 Lane Collector (No center lane)	8,000	11,022	1.378	F	14,400	1.800	F	3378	0.422	YES
Tenth Ave to Richmond St	2 Lane Collector (continuous left- turn lane)	15,000	10,120	0.675	D	12,300	0.820	D	2180	0.145	NO
Richmond St to Park Blvd	2 Lane Collector (continuous left- turn lane)	15,000	7,269	0.485	С	9,200	0.613	С	1931	0.128	NO
San Diego Ave	•			•			•			•	•
Hortensia St to Pringle St	2 Lane Collector (No center lane)	8,000	5,830	0.729	D	10,500	1.313	F	4670	0.584	YES
McKee St to Washington St	3 Lane Collector (one-way)	26,000	13,920	0.535	В	18,200	0.700	С	4280	0.165	NO
Washington St to India St	2 Lane Collector (one-way)	17,500	4,920	0.281	А	7,100	0.406	А	2180	0.125	NO
State St					-		-	-	-	-	
Laurel St to Juniper St	2 Lane Collector (No center lane)	8,000	4,140	0.518	С	8,200	1.025	F	4060	0.507	YES
Sunset Blvd			1	1	1			1	T	1	
Witherby St to Fort Stockton Dr	2 Lane Collector (No center lane)	8,000	2,595	0.324	В	4,600	0.575	С	2005	0.251	NO
University Ave				. <u> </u>	-						
Ibis St to Albatross St	2 Lane Collector (No center lane)	8,000	10,527	1.316	F	14,700	1.838	F	4173	0.522	YES
Albatross St to First Ave	2 Lane Collector (No center lane)	8,000	16,851	2.106	F	20,800	2.600	F	3949	0.494	YES

Table 6.3-8 Buildout Summary of Roadway Segment Analysis: Uptown											
			Exist	ing		Build	out				
				V/C			V/C				
	Roadway Functional	LOS E		Ratio			Ratio		Δin	Δin	
Roadway Segment	Classification	Capacity	ADT	(a)	LOS	ADT	(a)	LOS	ADT	V/C	Significant?
First Ave to Fourth Ave	2 Lane Collector (no fronting property)	10,000	11,750	1.175	F	14,100	1.410	F	2350	0.235	YES
Fourth Ave to Fifth Ave	2 Lane Collector (continuous left- turn lane)	15,000	20,250	1.350	F	21,600	1.440	F	1350	0.090	YES
Fifth Ave to Sixth Ave	4 Lane Collector	30,000	21,184	0.706	D	24,900	0.830	D	3716	0.124	NO
Sixth Ave to Eighth Ave	4 Lane Collector (no center lane)	15,000	24,400	1.627	F	29,300	1.953	F	4900	0.326	YES
Vermont St to Normal St	4 Lane Major Arterial	40,000	23,938	0.598	С	25,600	0.640	С	1662	0.042	NO
Normal St to Park Blvd	4 Lane Collector (no center lane)	15,000	16,275	1.085	F	21,200	1.413	F	4925	0.328	YES
Upas St											
Third Ave to Sixth Ave	2 Lane Collector (no fronting property)	10,000	4,475	0.448	В	8,500	0.850	D	4025	0.402	NO
Washington St											
India St to University Ave	4 Lane Major Arterial	40,000	27,929	0.698	С	34,800	0.870	D	6871	0.172	NO
University Ave to First Ave	4 Lane Major Arterial	40,000	20,477	0.512	В	25,400	0.635	С	4923	0.123	NO
First Ave to Fourth Ave	4 Lane Major Arterial	40,000	25,745	0.644	С	25,745	0.644	С	0	0.000	NO
Fourth Ave to Fifth Ave	4 Lane Major Arterial	40,000	30,900	0.773	D	37,300	0.933	E	6400	0.160	YES
Fifth Ave to Sixth Ave	4 Lane Major Arterial	40,000	38,428	0.961	E	41,100	1.028	F	2672	0.067	YES
Sixth Ave to Richmond St	4 Lane Major Arterial	40,000	41,778	1.044	F	41,778	1.044	F	0	0.000	NO
Richmond St to Normal St	6 Lane Major Arterial	50,000	38,725	0.775	С	47,100	0.942	E	8375	0.167	YES

Notes:

Bold values indicate roadway segments operating at LOS E or F.

*Howard Avenue, Meade Avenue, Orange Avenue/Howard Avenue will be classified as a two-lane collector with no continuous center left turn lane to accommodate future bicycle boulevard pending further project-level analysis.

Capacity for non-standard roadway classifications were provided by City of San Diego staff.

Capacity for non-standard roadway classifications were provided by City of San Diego stan.

(a) The v/c ratio is calculated by dividing the ADT volume by each respective roadway segment's capacity.

- **Impact 6.3-7**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to seven consecutive street segments of First Avenue from Washington Street to Grape Street.
- **Impact 6.3-8:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Fourth Avenue from Arbor Drive to Washington Street.
- **Impact 6.3-9:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Fourth Avenue from Walnut Avenue to Laurel Street.
- **Impact 6.3-10**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Fifth Avenue from Robinson Avenue to Walnut Avenue.
- **Impact 6.3-11:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to seven consecutive street segments of Sixth Avenue from Washington Street to Elm Street.
- **Impact 6.3-12**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Ninth Avenue from Washington Street to University Avenue.
- **Impact 6.3-13**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Campus Avenue/Polk Avenue from Washington Street to Park Boulevard.
- **Impact 6.3-14**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of Cleveland Avenue from Tyler Street to Richmond Street.
- **Impact 6.3-15**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of Fort Stockton Drive from Sunset Boulevard to Goldfinch Street.
- **Impact 6.3-16**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of Grape Street from First Avenue to Sixth Avenue.
- **Impact 6.3-17**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of Hawthorn Street from First Avenue to Sixth Avenue.
- **Impact 6.3-18**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to India Street from Washington Street to Winder Street.

- **Impact 6.3-19**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of India Street from Glenwood Drive to Redwood Street.
- **Impact 6.3-20**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to four consecutive street segments of Laurel Street from Columbia Street to Sixth Avenue.
- **Impact 6.3-21:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Lincoln Avenue from Washington Street to Park Boulevard.
- **Impact 6.3-22:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Park Boulevard from Mission Avenue to El Cajon Boulevard.
- **Impact 6.3-23:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Park Boulevard from Robinson Avenue to Upas Street.
- **Impact 6.3-24:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to three consecutive street segments of Richmond Street from Cleveland Avenue to Upas Street.
- **Impact 6.3-25**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to two consecutive street segments of Robinson Avenue from First Avenue to Eighth Avenue.
- **Impact 6.3-26:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to San Diego Avenue from Hortensia Street to Pringle Street.
- **Impact 6.3-27:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to State Street from Laurel Street to Juniper Street.
- **Impact 6.3-28:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to four consecutive street segments of University Avenue from Ibis Street to Fifth Avenue.
- **Impact 6.3-29:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact University Avenue from Sixth Avenue to Eighth Avenue.
- **Impact 6.3-30:** The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to University Avenue from Normal Street to Park Boulevard.
- Impact 6.3-31:The proposed Uptown CPU and associated discretionary actions would have a
cumulative traffic impact to two consecutive street segments of Washington
Street from Fourth Avenue to Sixth Avenue.

 Impact 6.3-32:
 The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to Washington Street from Richmond Street to Normal Street.

d. Freeway Segments

Table 6.3-9 displays the LOS analysis results for the freeway segments using their existing freeway configuration and the future peak-hour traffic volumes. As shown, the traffic generated by the land use changes associated with the Uptown, North Park and Golden Hill would have a cumulative traffic related impact along all 18 freeway segments within the study area.

The following significant cumulative freeway segment impacts are identified:

- **Impact 6.3-33**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to five segments of I-5 from Old Town Avenue to Imperial Avenue.
- **Impact 6.3-34**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to three consecutive segments of Interstate 8 (I-8) from Hotel Circle West to State Route 15 (SR-15).
- **Impact 6.3-35**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to the segment of SR-15 from Interstate 805 (I-805) to State Route 94 (SR-94).
- **Impact 6.3-36**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to three segments of I-805 from I-8 to SR-15.
- **Impact 6.3-37**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to three segments of SR-94 from 25th Street to SR-15.
- **Impact 6.3-38**: The proposed Uptown CPU and associated discretionary actions would have a cumulative traffic impact to three segments of SR-163 from I-8 to I-5.

e. Ramp Meters

Table 6.3-10 displays the analysis results for the ramp meters using their existing configuration and meter rate and the future peak-hour traffic volumes. As shown, the traffic generated by the land use changes associated with the Uptown, North Park and Golden Hill CPUs would have a cumulative traffic related impact at three ramp meters within the study area as follows:

Impact 6.3-39:	Hancock Street to I-5 southbound on-ramp in the PM peak period.
Impact 6.3-40:	Kettner Boulevard to I-5 southbound on-ramp in the PM peak period.
Impact 6.3-41:	Fifth Avenue to I-5 southbound on-ramp in the PM peak period.

			Table 6.3	-9								
Summary of Freeway Segment Level of Service												
				Existin	ng	Buildo	ut					
Freeway Segment	Direction	Number of Lanes	Capacity ¹	V/C Ratio ²	LOS	V/C Ratio	LOS	Δ^3	Significant?			
AM Peak												
I-5												
Old Town Ave to Washington St	NB	4 M + 1 A	9,200	0.950	E	1.183	F٥	0.234	YES			
	SB	4 M + 1 A	9,200	0.746	С	0.798	С	0.052	NO			
Washington St to Pacific Highway	NB	4 M	8,000	0.840	D	1.096	F₀	0.256	YES			
washington St to Facility righway	SB	4 M	8,000	0.660	С	0.739	С	0.079	NO			
First Ave to Sixth Ave	NB	4 M + 1 A	9,200	1.264	F₁	1.341	F₁	0.078	YES			
	SB	5 M + 1 A	11,200	0.346	A	0.743	С	0.397	NO			
SR-163 to SR-94	NB	5 M + 1 A	11,200	1.085	F₀	1.149	F₀	0.064	YES			
	SB	5 M + 1 A	11,200	0.362	A	0.901	D	0.540	NO			
SP 04 to Imporial Ave	NB	4 M + 1 A	9,200	1.035	F₀	1.064	F₀	0.029	YES			
	SB	4 M + 1 A	9,200	0.345	A	0.835	D	0.490	NO			
I-8	-8											
Hotel Circle (W) to Hotel Circle (E)	WB	4 M + 1 A	9,200	1.022	F₀	1.333	F₁	0.311	YES			
	EB	4 M	8,000	0.887	D	0.763	С	-0.124	NO			
Mission Center Rd to Qualcomm W/v	WB	4 M + 1 A	9,200	1.109	F₀	1.366	F ₂	0.257	YES			
	EB	4 M + 1 A	9,200	0.837	D	0.680	С	-0.157	NO			
L805 to SR-15	WB	4 M + 1 A	9,200	1.349	F₁	1.545	F ₂	0.196	YES			
1-000 10 511-10	EB	4 M + 1 A	9,200	0.727	С	0.766	С	0.040	NO			
SR-15												
L805 to SP 94	NB	3 M + 1 A	7,200	0.532	В	0.772	С	0.241	NO			
1-000 10 511-94	SB	2 M + 1 A	5,200	0.976	E	1.283	F ₁	0.307	YES			
I-805						-			-			
L8 to Adams Ave	NB	4 M + 1 A	9,200	1.262	F₁	1.515	F ₂	0.253	YES			
	SB	5 M + 1 A	11,200	0.383	A	0.458	В	0.074	NO			
El Cajon Blyd to University Ave	NB	4 M	8,000	0.602	В	1.427	F ₂	0.825	YES			
	SB	4 M + 1 A	9,200	1.063	F₀	0.457	В	-0.607	NO			
University Ave to SR-15	NB	4 M + 1 A	9,200	0.466	В	1.207	Fo	0.740	YES			
	SB	4 M + 1 A	9,200	0.947	E	0.421	В	-0.526	NO			

			Table 6.3	-9							
Summary of Freeway Segment Level of Service											
				Existing		Buildout					
Freeway Segment	Direction	Number of Lanes	Capacity ¹	V/C Ratio ²	LOS	V/C Ratio	LOS	Δ^3	Significant?		
SR-94											
25th St to 28th St	WB	4 M	8,000	0.976	E	1.241	Fo	0.264	YES		
2501 50 2601 50	EB	4 M	8,000	0.361	A	0.470	В	0.109	NO		
28th St to 30th St	WB	4 M	8,000	1.095	Fo	1.303	F 1	0.208	YES		
	EB	4 M	8,000	0.405	A	0.494	В	0.089	NO		
Broadway to SR-15	WB	4 M	8,000	1.214	Fo	1.414	F ₂	0.200	YES		
bloadway to SIX-15	EB	4 M + 1 A	9,200	0.390	A	0.466	В	0.075	NO		
SR-163					-						
I-8 to Washington St	NB	3 M + 1 A	7,200	0.575	В	1.121	Fo	0.546	YES		
1-0 to Washington St	SB	3 M + 1 A	7,200	0.828	D	0.950	E	0.122	YES		
Washington St to Robinson Ave	NB	2 M	4,000	0.800	С	0.830	D	0.031	NO		
Washington St to Robinson Ave	SB	2 M	4,000	1.151	Fo	1.846	F ₂	0.696	YES		
Quipce Dr. to 1-5	NB	2 M	4,000	0.884	D	0.914	D	0.030	NO		
	SB	2 M	4,000	1.641	F ₂	2.032	F ₃	0.391	YES		
			PM PEA	K							
1-5					-						
Old Town Ave to Washington St	NB	4 M + 1 A	9,200	0.780	С	1.000	E	0.220	YES		
	SB	4 M + 1 A	9,200	0.916	D	1.187	Fo	0.220	YES		
Washington St to Pacific Highway	NB	4 M	8,000	0.690	С	0.926	E	0.236	YES		
Washington of to Fashio Fighway	SB	4 M	8,000	0.810	D	1.100	F₀	0.290	YES		
First Ave to Sixth Ave	NB	4 M + 1 A	9,200	1.078	Fo	1.133	F ₀	0.055	YES		
	SB	5 M + 1 A	11,200	0.498	В	1.105	F ₀	0.607	YES		
SR-163 to SR-94	NB	5 M + 1 A	11,200	0.926	E	1.091	F ₀	0.166	YES		
	SB	5 M + 1 A	11,200	0.521	В	1.213	F ₀	0.693	YES		
SR-94 to Imperial Ave	NB	4 M + 1 A	9,200	0.883	D	1.011	F ₀	0.204 1 0.109 1 0.208 Y 0.089 1 0.200 Y 0.075 1 0.546 Y 0.031 1 0.696 Y 0.030 1 0.220 Y 0.391 Y 0.236 Y 0.220 Y 0.271 Y 0.236 Y 0.236 Y 0.255 Y 0.607 Y 0.693 Y 0.627 Y 0.627 Y 0.082 1 0.315 Y 0.035 1 0.027 Y	YES		
	SB	4 M + 1 A	9,200	0.497	В	1.124	F₀	0.627	YES		
1-8											
Hotel Circle (W) to Hotel Circle (E)	WB	4 M + 1 A	9,200	0.807	D	0.889	D	0.082	NO		
	EB	4 M	8,000	1.134	Fo	1.449	F ₂	0.315	YES		
Mission Center Rd to Qualcomm Wy	WB	4 M + 1 A	9,200	0.876	D	0.910	D	0.035	NO		
	EB	4 M + 1 A	9,200	1.070	Fo	1.291	F1	0.221	YES		
1-805 to SR-15	WB	4 M + 1 A	9,200	0.893	D	0.920	E	0.027	YES		
	EB	4 M + 1 A	9,200	1.183	F ₀	1.511	F ₂	0.327	YES		
SR-15							1				
I-805 to SR-94	NB	3 M + 1 A	7,200	0.532	В	1.120	Fo	0.589	YES		

Table 6.3-9 Summary of Freeway Segment Level of Service										
Freeway Segment	Direction	Number of Lanes	Capacity ¹	V/C Ratio ²	LOS	V/C Ratio	LOS	Δ^3	Significant?	
	SB	2 M + 1 A	5,200	0.976	E	1.367	F ₂	0.391	YES	
I-805									· •	
1.9 to Adama Ava	NB	4 M + 1 A	9,200	0.588	В	1.063	F₀	0.475	YES	
I-o to Adams Ave	SB	5 M + 1 A	11,200	0.937	E	1.297	F ₁	0.360	YES	
El Cajan Rhyd ta University Ave	NB	4 M	8,000	1.095	Fo	1.001	F₀	-0.094	NO	
El Cajon Bivo lo Oniversity Ave	SB	4 M + 1 A	9,200	0.635	С	1.293	F1	0.659	YES	
Lipivoroity Avo to SP 15	NB	4 M + 1 A	9,200	0.848	D	0.867	D	0.019	NO	
University Ave to SR-15	SB	4 M + 1 A	9,200	0.565	В	1.203	F₀	0.637	YES	
SR-94										
25th St to 29th St	WB	4 M	8,000	0.401	А	0.612	В	0.210	NO	
25111 51 10 26111 51	EB	4 M	8,000	0.936	E	1.482	F ₂	0.545	YES	
28th St to 20th St	WB	4 M	8,000	0.450	В	0.642	С	0.192	NO	
	EB	4 M	8,000	1.050	F ₀	1.556	F ₂	0.506	YES	
Broadway to SR-15	WB	4 M	8,000	0.499	В	0.697	С	0.198	NO	
Bloadway to SIC-15	EB	4 M + 1 A	9,200	1.012	F ₀	1.468	F ₂	0.456	YES	
SR-163										
L8 to Washington St	NB	3 M + 1 A	7,200	0.870	D	1.301	F1	0.431	YES	
1-0 to Washington St	SB	3 M + 1 A	7,200	0.533	В	0.797	С	0.264	NO	
Washington St to Pohinson Avo	NB	2 M	4,000	1.209	F0	1.658	F2	0.449	YES	
Washington St to Robinson Ave	SB	2 M	4,000	0.741	С	1.016	F0	0.275	YES	
Quinco Dr. to 1.5	NB	2 M	4,000	1.364	F2	1.362	F2	-0.001	NO	
	SB	2 M	4,000	1.162	F0	1.160	F0	-0.001	NO	
Notes: Bold values indicate freeway segmen	ts operating at	LOS E or F.								

For descriptions of LOS ratings for freeway segments, refer to Table 6.3-3.

V/C Ratio is the volume to capacity ratio

 Δ = change in v/c ratio between existing and buildout ¹The capacity is calculated as 2,000 ADT per lane and 1,200 ADT per auxiliary lane ² Traffic volumes provided by City of San Diego model ³Peak-hour volume calculated by: (ADT*K*D)/Truck Factor

		Pe	eak Hour Rar	T np Metering	able 6.3-10 Analysis – H	lorizon Year	Conditions				
On-Ramp	Peak Period	Meter Rate ¹ (veh/hr)	Existing Demand ² (veh/hr)	Excess Existing Demand (veh/hr)	Average Existing Delay (min)	Build-out Demand ² (veh/hr)	Excess Build-out Demand (veh/hr)	Average Build-out Delay (min)	D In Delay With Project (min)	Significant ?	Average With Project Queue
Interstate 5	1	1	1	1	1	1	1	1	1	1	r
Washington St to I-5 NB	AM	996	1020	24	1.4	1241	245	14.8	13.3	NO	6,125 ft
	PM	996	1034	38	2.3	1227	231	13.9	11.6	NO	5,775 ft
India St to I-5 NB	AM	996	915	0	0.0	1007	11	0.6	0.6	NO	263 ft
	PM	996	1066	70	4.2	1173	177	10.6	6.4	NO	4,415 ft
Howthorn St to L5 NR	AM	996	454	0	0.0	460	0	0.0	0.0	NO	0 ft
Tawtholl St to 1-5 NB	PM	996	842	0	0.0	825	0	0.0	0.0	NO	0 ft
Hanagak St to LE SD	AM	Ramp not metered in the AM peak								NO	0 ft
Hancock St to 1-5 SB	PM	1140	1287	147	7.7	1542	402	21.2	13.4	YES	10,050 ft
Kattaar Blud ta LE SB	AM	Ramp not metered in the AM peak								NO	0 ft
Kellner Biva to 1-5 SB	PM	498	269	0	0.0	861	363	43.7	43.7	YES	9,070 ft
	AM	Ramp not metered in the AM peak							0.0	NO	0 ft
Fifth Ave to I-5 SB	PM	996	1087	91	5.5	1894	898	54.1	48.6	YES	22,462 ft
Interstate 8											
	AM	Ramp not metered in the AM peak							0.0	NO	0 ft
NB Texas St to I-8 EB	PM	498	465	0	0.0	579	81	9.8	9.8	NO	2,026 ft
	AM		•	Ramp no	t metered in t	he AM peak	•		0.0	NO	0 ft
SB Lexas St to I-8 EB	PM	1140	866	0	0.0	888	0	0.0	0.0	NO	0 ft
Interstate 805											
	AM	1140	860	0	0.0	1118	0	0.0	0.0	NO	0 ft
EI Cajon BIVG to 1-805 NB	PM		ł	Ramp no	t metered in t	he PM peak		•	0.0	NO	0 ft
	AM	1140	998	0	0.0	1132	0	0.0	0.0	NO	0 ft
University Ave to 1-805 NB	PM		•	Ramp no	t metered in t	he PM peak		·	0.0	NO	0 ft

Table 6.3-10 Peak Hour Ramp Metering Analysis – Horizon Year Conditions											
On-Ramp	Peak Period	Meter Rate ¹ (veh/hr)	Existing Demand ² (veh/hr)	Excess Existing Demand (veh/hr)	Average Existing Delay (min)	Build-out Demand ² (veh/hr)	Excess Build-out Demand (veh/hr)	Average Build-out Delay (min)	D In Delay With Project (min)	Significant ?	Average With Project Queue
State Route 94	1	1	1	1				1	1	1	
28th St to SR-94 WB	AM	534	100	0	0.0	205	0	0.0	0.0	NO	0 ft
2011 01 10 01(-94 WD	PM			Ramp no	t metered in t	ne PM peak			0.0	NO	0 ft
32nd St/Broadway to SR-94	AM	570	99	0	0.0	173	0	0.0	0.0	NO	0 ft
WB	PM			Ramp no	t metered in t	ne PM peak			0.0	NO	0 ft
	AM		Ramp not metered in the AM peak								0 ft
25th St to SR-94 EB	PM	960	785	0	0.0	935	0	0.0	0.0	NO	0 ft
	AM	Ramp not metered in the AM peak								NO	0 ft
28th St to SR-94 EB	PM	960	732	0	0.0	870	0	0.0	0.0	NO	0 ft
32nd St/Broadway to SR-94	AM			Ramp no	t metered in t	ne AM peak		1	0.0	NO	0 ft
EB	PM	570	464	0	0.0	558	0	0.0	0.0	NO	0 ft
State Route 163										1	
Washington St to SR-163	AM	498	373	0	0.0	615	117	14.2	14.2	NO	2,936 ft
SB	PM		Ramp not metered in the PM peak								0 ft
NOTES: ¹ Meter rate is the assumed pe ² Demand is the peak hour der	eak hour ca mand using	pacity expe the on-ram	cted to be pro	ocessed throu	ugh the ramp	meter (using (Caltrans fast r	ate)	1		

EB= eastbound, SB = southbound, NB = northbound, WB = westbound, SR = State Route

Issue 2 Alternative Transportation

Would the project conflict with adopted policies, plans, or programs supporting alternative transportation?

a. Transit

Planned transit routes within the Uptown CPU area identified in the 2050 <u>SANDAG's San Diego</u> <u>Forward: The</u> Regional Transportation_Plan (RTP) and discussed in the Uptown, North Park, and Golden Hill Community Plan Update Mobility Study for Build-out Conditions (Appendix C, Kimley-Horn and Associates, 2015) include <u>BRTRapid</u>, LRT, and streetcar improvements as shown on Figure 6.3-3. Definitions of each of these types of service are provided in Chapter 2.0 of this PEIR. The changes in existing transit operations to serve the Uptown community are described below.

- **Route 10** currently travels along University Avenue and Washington Street in the Uptown community corridor. Route 10 will convert to <u>Rapid servicea BRT route</u>, with improvements supported by the Mid-City Rapid. Route 10 is currently a limited stop bus service that provides service from University Avenue at College Avenue to Old Town San Diego. Improvements include expansion of the service to La Mesa and Ocean Beach. The expected year for completion of this improvement is 2020.
- A **new streetcar service**, currently designated as route 554, will provide service from Downtown San Diego to Hillcrest neighborhood. Currently, it is planned that the streetcar service will travel along Fourth and Fifth Avenues, University Avenue, and Park Boulevard in the Uptown community corridor. The expected year for completion of this improvement is 2020 as identified in the RTP. However, additional evaluation completed for this potential service suggested that it will not be in place until beyond 2020.
- **Route 120** currently travels along Fourth and Fifth Avenues and University Avenue in the Uptown community corridor. Route 120 will convert to be <u>a BRT routeRapid service</u> along its current route. Route 120 currently provides local bus service from Downtown San Diego to the Kearny Mesa Transit Center. Improvements include transit priority measures and new transfer opportunities to the Trolley Green Line and <u>BRT Rapid services</u>. The expected year for completion of this improvement is 2030.
- **Route 11** will convert to <u>Rapid Service</u> <u>be a BRT route</u> along its current route. Route 11 currently provides local bus service from the SDSU Transit Center to Skyline Hills and travels along Park Boulevard, University Avenue, and First Avenue in the Uptown community. The expected year for completion of this improvement is 2035.
- Mid-City LRT is currently planned as a service extension from the City College Trolley Station. Construction of Mid-City LRT will be done in two phases. Phase 1 will include a LRT extension from downtown to Mid-City via El Cajon Boulevard and Park Boulevard. Phase 2 will extend the Phase 1 construction efforts to the current SDSU transit center. LRT service will be provided via Park Boulevard in the Uptown community corridor. The expected year for completion of this improvement is 2035.

The proposed Uptown CPU and associated discretionary actions would support implementation of the transit improvements identified in the 2050-RTP by providing policies that support prioritizing the transit system and improving efficiency of transit services. For example, a number of transit focused Mobility Element Policies are included in the proposed Uptown CPU that would support efforts to develop planned transit facilities. Thus, implementation of the project would not interfere with implementation of planned transit improvements and would provide policy support to support their implementation. Thus, impacts related to conflicts with existing or planned transit facilities would be less than significant.

b. Bicycle Facilities

The proposed Uptown CPU and associated discretionary actions would support existing plans and policies relative to the bicycle network. The recommended bicycle facility network for the proposed CPU is shown on Figure 6.3-5. The Mobility Element includes several bicycle-focused policies that support installation of bicycle parking facilities, identification of bicycle priority streets to connect neighboring communities, and increasing the level of bicycle comfort and safety for all levels of bicycle riders. Policies in the proposed plan support coordination with SANDAG on the planning and implementation of regional bicycle facilities, support increased bicycle comfort and safety, repurposing rights-of-way for bicycle facilities, and bike sharing. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would not conflict with adopted policies, plans, or programs supporting bicycle facilities.

bc. Pedestrian Facilities

There are no major planned and funded pedestrian facility improvement projects for the Uptown community. However, the proposed Uptown CPU Mobility Element includes a number of policies that support enhancements to pedestrian travel within the CPU area such as providing corner bulbouts along some of the main pedestrian corridors, enhanced pedestrian crosswalks, and increasing pedestrian safety and safe routes to schools. Implementation of the proposed Uptown CPU and associated discretionary actions would not restrict or impede pedestrian connectivity and would not conflict with any adopted policies or plans addressing pedestrian facilities. Thus, impacts would be less than significant.

In addition, SANDAG's Uptown Bikeways Project incorporates pedestrian improvements to provide pedestrian access along the corridors identified in the Banker's Hill "Walk the Walk" Plan. This project will include improvements to enhance east-west pedestrian connectivity across Fourth and Fifth Avenues along the following streets: Upas, Spruce, Quince, Nutmeg, Laurel, Juniper, Grape, and Elm.

6.3.4 Significance of Impacts

Cumulative impacts to intersections, roadway segments, freeway segments and ramp meters were determined to be significant, as detailed below.

6.3.4.1 Traffic Circulation

a. Intersections

- Washington Street & Fourth Avenue (Impact 6.3-1)
- Washington Street & Eighth Avenue/ SR-163 Off-Ramp (Impact 6.3-2)
- Washington Street/ Normal Street & Campus Avenue/ Polk Avenue (Impact 6.3-3)
- University Avenue & Sixth Avenue (Impact 6.3-4)
- Elm Street & Sixth Avenue (Impact 6.3-5)
- Cedar Street & Second Avenue (Impact 6.3-6)

b. Segments

- First Avenue: Washington Street to University Avenue (Impact 6.3-7)
- First Avenue: University Avenue to Robinson Avenue (Impact 6.3-7)
- First Avenue: Robinson Avenue to Grape Street (Impact 6.3-7)
- Fourth Avenue: Arbor Drive to Washington Street (Impact 6.3-8)
- Fourth Avenue: Walnut Avenue to Laurel Street (Impact 6.3-9)
- Fifth Avenue: Robinson Avenue to Walnut Avenue (Impact 6.3-10)
- Sixth Avenue: Washington Street to University Avenue (Impact 6.3-11)
- Sixth Avenue: University Avenue to Laurel Street (Impact 6.3-11)
- Sixth Avenue: Laurel Street to Elm Street (Impact 6.3-11)
- Ninth Avenue: Washington Street to University Avenue (Impact 6.3-12)
- Campus Avenue/ Polk Avenue: Washington Street to Park Boulevard (Impact 6.3-13)
- Cleveland Avenue: Tyler Street to Richmond Street (Impact 6.3-14)
- Fort Stockton Drive: Sunset Boulevard to Goldfinch Street (Impact 6.3-15)
- Grape Street: First Avenue to Third Avenue (Impact 6.3-16)
- Grape Street: Third Avenue to Sixth Avenue (Impact 6.3-16)
- Hawthorn Street: First Avenue to Third Avenue (Impact 6.3-17)
- Hawthorn Street: Third Avenue to Sixth Avenue (Impact 6.3-17)
- India Street: Washington Street to Winder Street (Impact 6.3-18)
- India Street: Glenwood Drive to Sassafrass Street (Impact 6.3-19)
- India Street: Sassafrass Street to Redwood Street (Impact 6.3-19)
- Laurel Street: Columbia Street to Sixth Avenue (Impact 6.3-20)
- Lincoln Avenue: Washington Street to Park Boulevard (Impact 6.3-21)
- Park Boulevard: Mission Avenue to El Cajon Boulevard (Impact 6.3-22)
- Park Boulevard: Robinson Avenue to Upas Street (Impact 6.3-23)
- Richmond Street: Cleveland Avenue to Upas Street (Impact 6.3-24)
- Robinson Avenue: First Avenue to Third Avenue (Impact 6.3-25)
- Robinson Avenue: Third Avenue to Eighth Avenue (Impact 6.3-25)
- San Diego Avenue: Hortensia Street to Pringle Street (Impact 6.3-26)
- State Street: Laurel Street to Juniper Street (Impact 6.3-27)
- University Avenue: Ibis Street to Fifth Avenue (Impact 6.3-28)
- University Avenue: Sixth Avenue to Eighth Avenue (Impact 6.3-29)
- University Avenue: Normal Street to Park Boulevard (Impact 6.3-30)

- Washington Street: Fourth Avenue to Sixth Avenue (Impact 6.3-31)
- Washington Street: Richmond Street to Normal Street (Impact 6.3-32)

c. Freeway Segments

- I-5 from Old Town Avenue to Imperial Avenue (Impact 6.3-33)
- I-8 from Hotel Circle West to SR-15 (Impact 6.3-34)
- SR-15 from I-805 to SR-94 (Impact 6.3-35)
- I-805 from I-8 to SR-15 (Impact 6.3-36)
- SR-94 from 25th Street to SR-15 (Impact 6.3-37)
- SR-163 from I-8 to I-5 (Impact 6.3-38)

d. Ramp Meters

- Hancock Street to I-5 southbound on-ramp in the PM peak period (6.3-39)
- Kettner Boulevard to I-5 southbound on-ramp in the PM peak period (6.3-40)
- Fifth Ave to I-5 southbound on-ramp in the PM peak period (6.3-41)

6.3.4.2 Alternative Transportation

The proposed Uptown CPU and associated discretionary actions would be consistent with adopted policies, plans, or programs supporting alternative transportation. Additionally, the proposed Uptown CPU and associated discretionary actions would provide policies that support improvements to pedestrian, bicycle, and transit facilities. Thus, the project would have a less than significant impact related to conflicts with adopted policies, plans or programs supporting alternative transportation, and no mitigation is required.

6.3.5 Mitigation Framework

The Traffic Impact Study identified improvements that would mitigate or reduce roadway segment and intersection impacts. The improvements that are ultimately recommended as part of the proposed Uptown CPU are included in the Uptown Impact Fee Study (IFS). However, in most cases, the improvements that would mitigate or reduce vehicular impacts were not recommended as part of the proposed Uptown CPU in order to maintain consistency with the overall mobility vision and other proposed CPU policies.

6.3.5.1 Intersections

While the following intersection mitigation measures would reduce potentially significant impacts, <u>none of the measures are</u>only TRANS 6.3-5 is proposed as part of the Uptown CPU and associated discretionary actions. <u>included within the proposed Uptown IFS.</u>

TRANS 6.3-1: Washington Street & Fourth Avenue (Impact 6.3-1): Widen Fourth Avenue in the southbound direction to add a second left-turn lane. Restripe the southbound approach to be two left-turn lanes, one through lane, and one right-turn lane.

- **TRANS 6.3-2:** Washington Street & Eighth Avenue/ SR-163 Off-Ramp (Impact 6.3-2): Widen Washington Street in the eastbound direction to four lanes and the westbound direction to three lanes. Widen the SR-163 Off-ramp to two lanes.
- **TRANS 6.3-3:** Washington Street/Normal Street & Campus Avenue/ Polk Avenue (Impact 6.3-3): Widen Washington Street in the northeast direction to add an exclusive right-turn lane.
- **TRANS 6.3-4:** University Avenue & Sixth Avenue (Impact 6.3-4): Widen Sixth Avenue in the southbound direction to add a second left-turn lane.
- **TRANS 6.3-5:** Elm Street & Sixth Avenue (Impact 6.3-5): Widen Elm Street in the westbound direction to add a second right-turn lane. This improvement project is identified in the Uptown IFS.
- **TRANS 6.3-6:** Cedar Street & Second Avenue (Impact 6.3-6): Install a traffic signal at this intersection. This intersection is located outside the boundaries of the Uptown CPU area.

6.3.5.2 Segments

While the following intersection mitigation measures would reduce potentially significant impacts, only TRANS 6.3-7d., TRANS 6.3-24a., and TRANS 6.3-27 are are included within the proposed Uptown IFSproposed as part of the proposed Uptown CPU and associated discretionary actions. The remaining measures would be inconsistent with the proposed Uptown CPU and are not proposed as part of the proposed Uptown IFS.

TRANS 6.3-7: First Avenue (Impact 6.3-7)

- a. Washington Street to University Avenue: Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- b. University Avenue to Robinson Avenue: Widen the roadway to a 4 lane collector with continuous left-turn lane.
- c. Robinson Avenue to Laurel Street: Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- d. Laurel Street to Hawthorn Street: Restripe the roadway to a 2 lane collector with continuous left-turn lane. This improvement project is identified in the Uptown IFS.
- e. Hawthorn Street to Grape Street: Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-8:** Fourth Avenue from Arbor Drive to Washington Street (Impact 6.3-8): Widen the roadway to a 4 lane collector with continuous left-turn lane.

- **TRANS 6.3-9:** Fourth Avenue from Walnut Avenue to Laurel Street (Impact 6.3-9): Restore the roadway to a 3 lane one-way collector for vehicles and remove the dedicated multi-modal lane.
- **TRANS 6.3-10:** Fifth Avenue from Robinson Avenue to Walnut Avenue (Impact 6.3-10): Restore the roadway to a 3 lane one-way collector for vehicles and remove the dedicated multi-modal lane.
- **TRANS 6.3-11:** Sixth Avenue (Impact 6.3-11)
 - a. Washington Street to University Avenue: Widen the roadway to a 6 lane prime arterial.
 - b. University Avenue to Laurel Street: Widen the roadway to a 4 lane major arterial.
 - c. Laurel Street to Elm Street: Widen the roadway to a 4 lane collector.
- **TRANS 6.3-12**: Ninth Avenue from Washington Street to University Avenue (Impact 6.3-12): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-13**:Campus Avenue/ Polk Avenue from Washington Street to Park Boulevard (Impact
6.3-13): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-14:** Cleveland Avenue from Tyler Street to Richmond Street (Impact 6.3-14): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-15:** Fort Stockton Drive from Sunset Boulevard to Goldfinch Street (impact 6.3-15): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-16**: Grape Street from First Avenue to Sixth Avenue (Impact 6.3-16): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-17**: Hawthorn Street from First Avenue to Sixth Avenue (Impact 6.3-17): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-18**: India Street from Washington Street to Winder Street (Impact 6.3-18): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- TRANS 6.3-19: India Street (Impact 6.3-19)
 - a. Glenwood Drive to Sassafrass Street: Widen the roadway to a 4 lane one-way collector.
 - b. Sassafrass Street to Redwood Street: Widen the roadway to a 3 lane one-way collector.
- **TRANS 6.3-20:** Laurel Street from Columbia Street to Sixth Avenue (Impact 6.3-20): Widen the roadway to a 4 lane collector.
- **TRANS 6.3-21:** Lincoln Avenue from Washington Street to Park Boulevard (Impact 6.3-21): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-22:** Park Boulevard from Mission Avenue to El Cajon Boulevard (Impact 6.3-22): Widen the roadway to a 4 lane one-way collector.
- **TRANS 6.3-23:** Park Boulevard from Robinson Avenue to Upas Street (Impact 6.3-23): Widen the roadway to a 4 lane one-way collector.
- TRANS 6.3-24: Richmond Street (Impact 6.3-24)
 - a. Cleveland Avenue to Robinson Avenue: Restripe the roadway to a 2 lane collector with continuous left-turn lane. This improvement project is identified in the Uptown IFS.
 - b. Robinson Avenue to Upas Street: Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- TRANS 6.3-25: Robinson Avenue (Impact 6.3-25)
 - a. First Avenue to Third Avenue: Restripe the roadway to a 2 lane collector with continuous left-turn lane.
 - b. Third Avenue to Eighth Avenue: Widen the roadway to a 4 lane collector.
- **TRANS 6.3-26:** San Diego Avenue from Hortensia Street to Pringle Street (Impact 6.3-26): Restripe the roadway to a 2 lane collector with continuous left-turn lane.
- **TRANS 6.3-27:** State Street from Laurel Street to Juniper Street (Impact 6.3-267): Restripe the roadway to a 2 lane collector with continuous left-turn lane. This improvement project is identified in the Uptown IFS.
- **TRANS 6.3-28:** University Avenue from Ibis Street to Fifth Avenue (Impact 6.3-28): Widen the roadway to a 4 lane collector.
- **TRANS 6.3-29:** University Avenue from Sixth Avenue to Eighth Avenue (Impact 6.3-29): Widen the roadway to a 4 lane major arterial and install a raised median.
- **TRANS 6.3-30:** University Avenue from Normal Street to Park Boulevard (Impact 6.3-30): Widen the roadway to a 4 lane collector.
- **TRANS 6.3-31:** Washington Street from Fourth Avenue to Sixth Avenue (Impact 6.3-31): Widen the roadway to a 6 lane major arterial.
- **TRANS 6.3-32:** Washington Street from Richmond Street to Normal Street (Impact 6.3-32): Restripe the roadway to a 6 lane prime arterial and remove on-street parking.

6.3.5.3 Freeway Segments

Mitigation measures are identified for impacts to freeways; however, because freeway improvements are not within the authority of the City they are infeasible and not proposed as part of the Uptown CPU. The improvements identified in SANDAG's RTP would improve operations along the freeway segments and ramps; however, to what extent is still undetermined, as these are future improvements that must be defined more over time. Furthermore, implementation of freeway improvements in a timely manner is beyond the full control of the City since Caltrans has approval authority over freeway improvements. However, tThe City will continue to coordinate with Caltrans and SANGAG on future improvements, as future project-level developments proceed, to develop potential "fair share" multi-modal mitigation strategies for freeway impacts, as appropriate. The following are the freeway mainline improvements identified in SANDAG's RTP:

- **TRANS 6.3-33**: I-5 northbound and southbound from Old Town Avenue to Imperial Avenue: SANDAG's 2050 Revenue Constrained RTP includes operational improvements along I-5 between Old Town Avenue and Imperial Avenue. This project is expected to be constructed by year 2050. This measure provides partial mitigation, since it improves freeway operation in the vicinity of the project<u>No improvements are</u> identified for this segment in SANDAG's San Diego Forward, The Regional Plan (RP). (Impact 6.3-33)
- **TRANS 6.3-34:** I-8 eastbound and westbound from Hotel Circle (W) to SR-15: SANDAG's 2050 Revenue Constrained RTP includes operational improvements along I-8 between Hotel Circle (W) and SR-15I-5 and SR 125. This project is expected to be constructed by year 2050. This measure provides partial mitigation since it improves freeway operation in the vicinity of the project. (Impact 6.3-34)
- **TRANS 6.3-35:** SR-15 northbound and southbound from I-805 to SR-94: SANDAG's <u>2050 Revenue</u> <u>Constrained</u>_RTP proposes the construction of managed lanes along SR-15 <u>between I-805 and SR-94from I-5 to I-805 and from I-8 to SR-163</u>. <u>Between I-8 and</u> <u>SR-163, the project is expected to be constructed by 2035, between SR-94 and I-805, the project is expected to be constructed by 2035, and between I-5 and SR-94, <u>the project is expected to be constructed by 2050</u>. This project is expected to be constructed by year 2035. This measure provides partial mitigation, since it reduces the traffic demand on the freeway general purpose lane. (Impact 6.3-35)</u>
- **TRANS 6.3-36:**I-805 northbound and southbound from I-8 to SR-15: SANDAG's 2050 Revenue
Constrained RTP proposes the construction of managed lanes along I-805 between
 $\frac{128-158}{128}$ and SR-1635. This project is expected to be constructed by year 20530. This
measure provides partial mitigation, since it reduces the traffic demand on the
freeway general purpose lane. Additionally, Caltrans is studying buses on shoulder
options along the I-805 corridor on an interim basis. (Impact 6.3-36)
- **TRANS 6.3-37:**SR-94 eastbound and westbound from 25th Street to SR-15: SANDAG's 2050
Revenue Constrained RTP proposes the construction of managed lanes along SR-
94 between 25th Street and SR-15[-5 and SR-125. Between I-5 and I-805, this

project is expected to be constructed by year 2035. In 2050 the project is expected to be constructed between I-805 and SR-125. Caltrans is evaluating alternatives to this measure as part of the environmental analysis for the SR-94 Express Lanes Project, including bus on shoulders and other multi-modal projects outlined in the Community Based Alternatives of the SR-94 Express Lanes Project. This project is expected to be constructed by year 2020. This measure (or an alternative measure) would provides partial mitigation, since it reduces the traffic demand on the freeway general purpose lanes. (Impact 6.3-37)

TRANS 6.3-38:SR-163 northbound from I-8 to Robinson Avenue and SR-163 southbound from I-8
to I-5: No improvements are identified for this state route segment in SANDAG's
2050-RTP. (Impact 6.3-38)

At the project-level, significant impacts at locations outside of the jurisdiction of the City could be partially mitigated in the form of transportation demand management (TDM) measures that encourage carpooling and other alternative means of transportation consistent with proposed CPU policies. Fair share contributions could also be provided toward the construction of the following projects that are included in the SANDAG's RP:

- Operational improvements along I-8 between I-5 to SR-15 (TRANS 6.3-34)
- Construction of managed lanes along SR-15 between I-805 and SR-94 (TRANS 6.3-35)
- Construction of managed lanes along I-805 between SR-8 to SR-163 (TRANS 6.3-36)
- Construction of managed lanes along SR-94 between I-5 to I-805 (TRANS 6.3-37)

6.3.5.<u>34</u> Ramp Meters

TRANS 6.3-39: The City of San Diego shall coordinate with Caltrans to address ramp capacity at impacted on-ramp locations. Improvements could include additional lanes, interchange reconfiguration, etc.; however, specific capacity improvements are still undetermined, as these are future improvements that must be defined more over time. Furthermore, implementation of freeway improvements in a timely manner is beyond the full control of the City since Caltrans has approval authority over freeway improvements. At the project level, significant impacts at locations outside of the jurisdiction of the City could be partially mitigated in the form of fair share contribution or TDM measures that encourage carpooling and other alternative means of transportation consistent with proposed CPU policies. Fair share contributions may be provided at the project level for impacted ramps where the impacted facility is included in the SANDAG's RP; however, at this time none of the impacted ramps are included in the SANDAG RP. (Impacts 6.3-39 – 6.3-41)

6.3.6 Significance of Impacts after Mitigation

While implementation of the mitigation measures identified above would reduce impacts to less than significant at many of the intersections and roadway segments, only mitigation measures TRANS 6.3-5, TRANS 6.3-7d., TRANS 6.3-24a., and TRANS 6.3-27 are included within the proposed Uptown CPU and IFS. There is no funding mechanism for the remaining measures not included

within the IFS. Additionally, implementation of the roadway segment and intersection measures not included within the proposed IFS would be inconsistent with the mobility goals of the proposed Uptown CPU.

Due to the programmatic nature of the proposed Uptown CPU and associated discretionary actions, there is uncertainty as to the specific phasing of development including actual design and specific location of future projects, and thus, the timing of the proposed mitigation improvements. The design of these mitigation improvements are for the build-out of the CPU and the effectiveness at the project-level is not known at this time. Future development projects' transportation studies would be able to more accurately identify potential transportation impacts and provide the mechanism to address project-specific mitigation including, but not limited to, physical improvements, fair share contribution, or transportation demand management measures, or a combination of these measures. Impacts to intersections and roadway segments would remain significant and unavoidable.

Likewise, impacts to Caltrans facilities (freeway segments and ramps, Impacts 6.3-33 through 6.3-41) would remain significant and unavoidable because the City cannot ensure that the mitigation necessary to avoid or reduce the impacts to a level below significance would be implemented prior to will occuroccurrence of the impact.

6.4 Air Quality

An Air Quality Analysis for the Uptown, North Park, and Golden Hill Community Plan Updates (CPUs) was prepared by RECON (May 16, 2016). This report addresses air quality impacts associated with the proposed Uptown CPU and associated discretionary actions. The report is included as Appendix D to this PEIR and forms the basis for the discussion in this section.

6.4.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

6.4.2 Significance Determination Thresholds

6.4.2.1 CEQA Guidelines

Thresholds used to evaluate potential impacts to air quality are based on applicable criteria in the California Environmental Quality Act (CEQA) Guidelines Appendix G and the City of San Diego Significance Determination Thresholds (2011), and applicable air district standards described below. Thresholds are modified from the City's CEQA Significance Determination Thresholds to reflect the programmatic analysis for the proposed Uptown CPU and associated discretionary actions. A significant impact could occur if implementation of a proposed CPU would:

- 1) Conflicts or obstructs the implementation of the applicable air quality plan;
- 2) Result in a violation of any air quality standard or contributes substantially to an existing or projected air quality violation;
- 3) Expose sensitive receptors to substantial pollutant concentrations, including toxins; or
- 4) Create objectionable odors affecting a substantial number of people.

6.4.2.2 San Diego Air Pollution Control District

a. Air Quality Standards

Regarding question 2 above, the San Diego Air Pollution Control District (APCD) has established trigger levels that determine when a new or modified stationary source would require an air quality analysis. These trigger levels are utilized by the City of San Diego in their Significance Determination Thresholds (City of San Diego 2011) as one of the considerations when determining the potential

significance of air quality impacts for projects within the City. These thresholds would be applicable to future, individual development projects implemented within the proposed Uptown CPU area. The air quality impact screening levels applicable to future development within the proposed Uptown CPU area are shown in Table 6.4-1.

Table 6.4-1								
Aiı	Air Quality Impact Screening Levels							
		Emission Rate						
	Pounds/Hour Pounds/Day Tons/Year							
NO _X	25	250	40					
SO _X	25	250	40					
CO	100	550	100					
PM ₁₀		100	15					
Lead		3.2	0.6					
VOC, ROG ¹	VOC, ROG ¹ 137 ² 15							
PM _{2.5}	PM _{2.5} 100 ³							
SOURCE: APCD, Rule 20.2 (12/17/1998); City of San Diego 2011.								
¹ The terms reactive organic gases (ROG) and volatile organic compounds (VOC)								
are essentially synonymous and are used interchangeably.								
² VOC threshold a	² VOC threshold are based on levels per the South Coast Air Quality							
Management District (SCAQMD) and Monterey Bay Air Pollution Control								
District, which have similar federal and state attainment status as San Diego.								
³ PM _{2.5} threshold developed from the SCAQMD Final Methodology to Calculate								
$PM_{2.5}$ and $PM_{2.5}$ Significance Thresholds (SCAQMD 2006) and the PM_{10} standard								
of the San Diego /	of the San Diego APCD.							

The above thresholds are applicable to individual development projects and not a program level analysis such as the proposed Uptown CPU and associated discretionary actions. The project level thresholds are intended to ensure many individual projects would not obstruct the timely attainment of the national and state ambient air quality standards (AAQS). Generally, discretionary program-level planning activities, such as general plans, community plans, and specific plans, are evaluated for consistency with the local air quality plans as a measure of significance.

b. Toxic Air Emissions

Regarding toxic air emissions (Issue 3), for San Diego APCD permitted projects in general, the APCD does not identify a significant impact if the potential health risks from the proposed project would not exceed the health risk public notification thresholds specified by San Diego APCD Rule 1210. The public notification thresholds are:

- Maximum incremental cancer risks equal to or greater than ten in one million, or
- Cancer burden equal to or greater than 1.0, or
- Total acute non-cancer health hazard index equal to or greater than 1.0, or
- Total chronic non-cancer health hazard index equal to or greater than 1.0.

Therefore, for the purposes of evaluating the potential health risks associated with the air toxics addressed in this assessment, a significant impact would occur if the worst-case incremental cancer

risk is greater than or equal to ten in one million, or if the worst-case total acute or chronic health hazard index is greater than or equal to one.

6.4.3 Impact Analysis

Issue 1 Conflicts with Air Quality Plans

Would the project conflict with or obstruct implementation of the applicable air quality plan?

As described in Chapter 5.0, the California Clean Air Act requires air basins that are designated nonattainment of state AAQS for criteria pollutants prepare and implement plans to attain the standards by the earliest practicable date. The two pollutants addressed in the San Diego Regional Air Quality Strategy (RAQS) are volatile organic compounds (VOC) and oxides of nitrogen (NOx), which are precursors to the formation of ozone. Projected increases in motor vehicle usage, population, and industrial growth create challenges in controlling emissions to maintain and further improve air quality. The RAQS, in conjunction with the Transportation Control Measures, were most recently adopted in 2009 as the air quality plan for the San Diego Air Basin (SDAB).

The basis for the RAQS is the distribution of population in the region as projected by San Diego Association of Governments (SANDAG). The San Diego APCD refers to approved general plans to forecast, inventory, and allocate regional emissions from land use and development-related sources. These emissions budgets are used in statewide air quality attainment planning efforts. As such, projects that propose development at an intensity equal to or less than population growth projections and land use intensity are inherently consistent. Amending the adopted Community Plan to change development potential would not necessarily result in an inconsistency between the current air quality plans (that are based on the adopted Community Plan) and the proposed Uptown CPU and associated discretionary actions. The focus of the RAQS is on emissions from the sources, not the actual land use, projects that propose development that is greater than anticipated in the growth projections warrant further analysis to determine consistency with RAQS and the State Implementation Plan (SIP). The consistency with the RAQS is further evaluated by comparing emissions that would occur under build-out of the proposed Uptown CPU and associated discretionary actions.

The proposed Uptown CPU and associated discretionary actions would change the planned land use mix as follows:

- Decrease the projected number of residential units by approximately six percent;
- Decrease the amount of land designated for commercial development by 0.2 percent, and
- Increase the amount of land designated for institutional development by nine percent.

As presented below, future operational emissions under the proposed Uptown CPU and associated discretionary actions would be less than future operational emissions under the adopted Community Plan. Thus, because the land use changes associated with the proposed Uptown CPU and associated discretionary actions would not result in an effective increase in operational emissions, the proposed Uptown CPU and associated discretionary actions would be consistent with assumptions contained in the RAQS, and impacts would be less than significant.

Issue 2 Air Quality Standards

Would the project result in a violation of any air quality standard or contribute substantially to an existing or projected air quality violation?

Air quality impacts can result from the construction and operation of a project. Construction impacts are short-term and result from fugitive dust, equipment exhaust, and indirect effects associated with construction workers and deliveries. Operational impacts can occur on two levels: regional impacts resulting from development or local effects stemming from sensitive receivers being placed close to roadways or stationary sources. In the case of the proposed Uptown CPU and associated discretionary actions, operational impacts are primarily due to emissions from mobile sources associated with the vehicular travel along the roadways. Construction and operational impacts of the proposed Uptown CPU and associated discretionary actions are discussed below.

a. Construction

Construction-related activities are temporary, short-term sources of air emissions. Sources of construction-related air emissions include:

- Fugitive dust from grading activities;
- Construction equipment exhaust;
- Construction-related trips by workers, delivery trucks, and material-hauling trucks; and
- Construction-related power consumption.

To illustrate the range of potential construction-related air quality impacts from projects that could occur, three hypothetical projects were evaluated. The size and scope of these hypothetical projects was selected to reflect typical projects in heavily developed areas such as the Uptown CPU area. Hypothetical projects include a 1.8-acre multi-family residential project, a 25,000-square-foot commercial project, and a 65,000-square-foot light industrial project. The 1.8-acre multi-family development is assumed to consist of the demolition of an existing 5,000-square-foot structure and the construction of a 29-unit multi-family structure. The commercial development is assumed to consist of the demolitinal development is assumed to consist of the demolition of an existing 5,000-square foot structure and the construction of 25,000 square feet of commercial use. The light industrial development is assumed to consist of the demolition of an existing 5,000-square feet of consist of the structure and the construction of 65,000 square feet of industrial use. Although there are no proposed industrial land use designations in the CPU area, the size and scope of these hypothetical projects was selected to reflect typical projects in heavily developed areas such as the Uptown area and represents a conservative analysis.

Air emissions were calculated using CalEEMod 2013.2.2 (CalEEMod). The CalEEMod program is a tool used to estimate air emissions resulting from land development projects based on California specific emission factors. The model estimates mass emissions from two basics sources: construction sources and operational sources (i.e., area and mobile sources). CalEEMod can estimate the required construction equipment when project specific information is unavailable. The estimates are based on surveys performed by the South Coast Air Quality Management District and the Sacramento Metropolitan Air Quality Management District of typical construction projects which provide a basis for scaling equipment needs and schedule with a project's size. Air emission

estimates in CalEEMod are based on the duration of construction phases; construction equipment type, quantity, and usage; grading area; season; and ambient temperature, among other parameters.

CalEEMod estimates were used to develop construction scenarios based on typical construction that would occur with build-out of the proposed Uptown CPU area. The analysis assumed that standard dust and emission control during grading operations would be implemented to reduce potential nuisance impacts and to ensure compliance with SDAPCD Rule 55.0, Fugitive Dust Control. An architectural coating VOC limit of 150 grams per liter was used for all interior and exterior coatings to reflect the requirements of San Diego APCD, Rule 67.

Table 6.4-2 Sample Daily Construction Emissions (pounds/day)						
	Residential	Commercial	Industrial	Project-level		
Pollutant	Project	Project	Project	Threshold		
ROG	55	70	91	137		
NO _X	29	14	29	250		
CO	22	10	22	550		
SO ₂	0	0	0	250		
PM ₁₀	4	1	4	100		
PM _{2.5}	3	1	3	100		
NOTE: Due to rounding, the total PM emissions indicated in the CalEEMod output files						

A summary of the modeling results for these sample projects is shown in Table 6.4-2.

Emissions summarized in Table 6.4-2 are the maximum emissions for each pollutant and that they may occur during different phases of construction. They would not necessarily occur simultaneously. These are, therefore, the worst-case emissions. For assessing the significance of the air quality emissions resulting during construction of the hypothetical projects, the construction emissions were compared to the thresholds shown in the far right column of Table 6.4-2. As shown, the hypothetical individual projects would not exceed the applicable thresholds. Potential cumulative construction emissions are addressed below.

Typical daily construction emissions are presented to illustrate the potential scope of air impacts for projects that could be constructed under the proposed Uptown CPU and associated discretionary actions. Based on this analysis, individual projects constructed as part of build-out of the proposed Uptown CPU area would not exceed air quality significance thresholds for construction. Additionally, the regulations at the federal, state, and local level provide a framework for developing project-level air quality protection measures for future discretionary projects. The City's process for the evaluation of discretionary projects includes environmental review and documentation pursuant to CEQA, as well as an analysis of those projects for consistency with the goals, policies and recommendations of the General Plan. Based on the hypothetical worst case construction emission analysis, emissions associated with build-out of the proposed Uptown CPU and associated discretionary actions at the project level would be less than significant. Ministerial projects would not require a formal environmental review. Generally, ministerial permits require a public official to

determine only that the project conforms to applicable zoning and building code requirements, and that applicable fees have been paid. These projects are generally smaller in size than those requiring discretionary review and would be smaller than the hypothetical projects evaluated in this analysis. As such, construction related air quality impacts associated with ministerial projects would be less than significant.

b. Operation

Operation emissions are long-term and include mobile and area sources. Sources of operational emissions associated with future projects developed under the proposed Uptown CPU and associated discretionary actions include:

- Traffic generated by the project.
- Area source emissions from the use of natural gas, fireplaces, and consumer products.

Air pollutants generated by all land uses within the proposed Uptown CPU area were modeled based on average emissions from land use types. For the purposes of this analysis, it was assumed that the land use changes contained in the proposed Uptown CPU and associated discretionary actions would be fully constructed in 2035. Actual emissions would vary depending on future projects and regulations within the Uptown CPU area.

Program-level air emissions would exceed the City's project-level thresholds; however, project-level standards are not appropriate for a program-level analysis, as the thresholds are conservative and intended to ensure that multiple simultaneous individual projects would not obstruct the timely attainment of the national and state ambient air quality standards. Generally, discretionary, program-level planning activities, such as general plans, community plans, specific plans, etc., are evaluated for consistency with the local air quality plan. In contrast, project-level thresholds are applied to individual project-specific approvals, such as a proposed development project. Therefore, the analysis of the proposed Uptown CPU and associated discretionary actions is based on the future emissions estimates and related to attainment strategies derived from the adopted Community Plan.

At the program level, the analysis looks at the emissions of the proposed Uptown CPU and associated discretionary actions in relation to the adopted Community Plan to determine if the emissions would exceed the emissions estimates included in the RAQS to determine whether the proposed Uptown CPU would obstruct attainment or result in an exceedance of AAQS that would result in the temporary or permanent exposure of persons to unhealthy concentrations of pollutants. As such, the analysis evaluates the potential for future development within the Uptown CPU area to result in, or contribute to, a violation of any air quality standard based on the change in pollutant emissions that would result from build-out of the adopted Community Plan in the year 2035 compared to the emissions resulting from the proposed Uptown CPU and associated discretionary actions in the year 2035. Table 6.4-3, summarizes the estimated maximum emissions for the proposed Uptown CPU and associated discretionary actions by source. As shown in Table 6.4-3, operational emissions associated with the proposed Uptown CPU and associated discretionary actions would be lower for all pollutants when compared to the adopted Community Plan.

Table 6.4-3 Total Operational Emissions for the Uptown CPU Area								
		Pollutant (pounds per day)						
Condition	Source	ROG	NO _X	CO	SO ₂	PM ₁₀	PM _{2.5}	
	Area	1,367.24	32.85	2,850.68	0.15	57.64	57.20	
Adopted Community	Energy	12.86	110.86	54.04	0.70	8.88	8.88	
Plan	Mobile	1,083.86	1,681.45	9,902.87	37.45	2,535.50	703.36	
	Total	2,463.96	1,825.16	12,807.59	38.30	2,602.02	769.44	
Proposed CPU and Associated Discretionary Actions	Area	1,308.54	30.92	2,683.45	0.14	54.26	53.84	
	Energy	12.42	107.15	52.80	0.68	8.58	8.58	
	Mobile	924.30	1,427.93	8,421.29	31.74	2,148.53	596.04	
	Total	2,245.25	1,566.00	11,157.54	32.56	2,211.36	658.47	
Change	-218.70	-259.15	-1,650.05	-5.74	-390.66	-110.98		

Further, while emissions associated with build-out of the entire CPU area would exceed the City's project-level thresholds, the Uptown CPU would emit fewer pollutants than would occur under the adopted Community Plan. Therefore, the air emissions from build-out of the proposed Uptown CPU and associated discretionary actions would not increase air pollutants in the region, would not further increase the frequency of existing violations of federal or state AAQS, or result in new exceedances. Air quality impacts associated with the adoption of the proposed Uptown CPU and associated discretionary actions would result in less than significant impacts.

Issue 3 Sensitive Receptors

Would the project expose sensitive receptors to substantial pollutant concentrations, including toxins?

a. Localized Carbon Monoxide Hot Spots Impacts

The traffic study concluded that six intersections in the proposed Uptown CPU area would operate at Level of Service (LOS) E or worse. Based on the CO Protocol, the three worst signalized intersections in the Uptown CPU area were selected for a detailed carbon monoxide (CO) hot spot analysis. These intersections are listed in Table 6.4-4. A computer air emission dispersion model, CALINE4, was used to calculate CO concentrations at receivers located at each intersection. These concentrations were derived from inputs including traffic volumes from the CPU traffic analysis and emission factors from EMFAC2014. The results of the modeling for these three intersections in the Uptown CPU area are summarized in Table 6.4-4.

Table 6.4-4 Maximum Build-out CO Concentrations in the Uptown CPU Area						
	1-Hour CO	1-Hour CO Standard CAAOS/	8-Hour	8-Hour CO Standard CAAOS/		
Roadway	ppm	NAAQS	CO ppm ¹	NAAQS		
Washington St & Eighth Ave/ SR-163 Off-Ramp	4.5		3.2			
Washington St/Normal St & Campus Ave/Polk Ave	3.8	9.0/9	2.7	20/35		
Elm St & Sixth Ave.	5.1		3.6			
¹ 8-hour concentrations developed based on a 0.7 persistence factor SR = State Route						

As shown, the maximum 1-hour concentration would be 5.1 ppm. This concentration is below the federal and state 1-hour standards. In order to determine the 8-hour concentration, the 1-hour value was multiplied by a persistence factor of 0.7, as recommended in the CO Protocol. Based on this calculation, the maximum 8-hour concentration would be 3.6 ppm. Thus, increases of CO due to the Uptown CPU would be below the federal and state 8-hour standards. Therefore, there would be no harmful concentrations of CO within the Uptown CPU area, and localized air quality emissions would be less than significant.

b. Toxic Air Emissions

An assessment was completed to evaluate the potential effects associated with placing sensitive land uses in the vicinity of existing sources of air pollution. In the case of the proposed Uptown CPU and associated discretionary actions, this source of air pollution is vehicle traffic on freeways Therefore, this assessment discloses the maximum potential health risks (residential and worker) within the Uptown CPU area due to these existing external sources.

Stationary Sources

The proposed Uptown CPU and associated discretionary actions include land uses which may generate air pollutants affecting adjacent sensitive land uses. In air quality terms, individual land uses that emit air pollutants in sufficient quantities are known as stationary sources. The primary concern with stationary sources is local; however, they also contribute to air pollution in the SDAB. Stationary sources include gasoline stations, power plants, dry cleaners, and other commercial and industrial uses. Stationary sources are regulated by the local air pollution control or management district through the issuance of permits; in this case, the agency is the San Diego APCD.

The California Air Toxics Program establishes the process for the identification and control of toxic air contaminants and includes provisions to make the public aware of significant toxic exposures and for reducing risk. In accordance with Assembly Bill 2588, if adverse health impacts exceeding public notification levels are identified, the facility would provide public notice, and if the facility poses a potentially significant public health risk, the facility must submit a risk reduction audit and

plan to demonstrate how the facility would reduce health risks. Thus, with this regulatory framework, at the program level, impacts associated with stationary sources in the Uptown CPU area would be less than significant.

Mobile Sources

Unlike stationary sources, local agencies, such as the San Diego APCD, do not regulate roadways as emission sources. While the California Air Resources Board (CARB) regulates vehicle emissions and fuel formulations, the source of the majority of diesel particulate matter (DPM) is regulated nationwide by the U. S. Environmental Protection Agency. To determine the exposure of sensitive receptors to DPM within the Uptown CPU area, a single AERMOD run was created for all freeway sources in the Uptown CPU area. The results provide the total average annual DPM concentrations at each modeled grid receiver. The resulting total average annual DPM concentrations were then used to calculate the incremental cancer risk and chronic health hazard index at each receiver. The model, AERMOD, input and output data results are included and summarized below.

Carcinogenic Risk

There is no adopted standard for evaluating the DPM emission impacts due to vehicles traveling on local roadway and freeways. Therefore, the significance threshold of ten in one million was used in evaluating the potential impacts from the vehicular sources. DMP concentrations can be equated to carcinogenic risk to determine significance of an impact. Carcinogenic health risk is determined by calculating lifetime average daily exposure based on a variety of factors such as respiration rate, body weight and pollutant concentration. Specific methodology for determining carcinogenic risk is described in the Air Quality Analysis, Section 5.0 (Appendix D).

The average annual concentration of diesel particulates at each modeled receiver was calculated using air dispersion models as detailed in Section 5.3.2.2 of the Air Quality Analysis (Appendix D). Contours of the particulate matter less than 10 microns in diameter (PM10) annual maximum annual concentrations for the Uptown CPU are shown in Figure 6.4-1.

The results of the assessment indicate that the worst-case residential incremental increase in cancer risk due to DPM emissions associated with increased traffic on local freeways in the Uptown CPU area is 0.08 in one million and occurs in the southwestern portion of the CPU area, west of I-5 near the intersection of Grape Avenue and Front Street. The location of the Uptown maximum exposed individual resident and maximum exposed individual worker are shown in Figure 6.4-1. The maximum concentrations higher than the maximum exposed individual resident and maximum exposed individual vorker for a risk is less than the significance threshold of ten in one million. Exposure associated with the 65th percentile, 80th percentile, and worker incremental cancer risks at this location would be less than the 95th percentile value. Therefore, incremental increase in cancer risks to sensitive receivers would be less than significant.

ATTACHMENT 7

Image source: SanGIS (flown May 2012)







2035 Annual PM Concentrations from Freeway Operations - Uptown

RECON M:\JOBS4\6086\common_gis\Fig6.4-1_EIR_UT.mxd 6/1/2016 fmm

Chronic Risk

Chronic risk is a long-term, non-carcinogenic health risk. Characterization of these risks is performed by comparing the estimated annual air concentrations of the substance (pollutant) to a reference exposure level. A chronic hazard quotient is obtained by dividing the average annual concentration by the reference exposure level. The hazard index provides a measure of total potential chronic non-carcinogenic health effects and is calculated for each receiver by summing the hazard quotients for all individual substances that impact the same toxicological endpoint. The analysis conducted for the proposed Uptown CPU and associated discretionary actions considered inhalation diesel particulate matter. When an individual hazard quotient is less than or equal to one, no adverse chronic non-carcinogenic health effects are expected from that substance. Similarly, if the hazard index is greater than one, chronic non-carcinogenic effects resulting from exposure to the substances emitted may be possible.

An assessment of the potential chronic risk due to DPM was made at the same receivers throughout the Uptown CPU area as discussed above for the carcinogenic risk. The results of the analysis indicate that the worst-case chronic health hazard index due to DPM from the freeways would be approximately 0.1 or less in 2035. The 2035 chronic health hazard index would be less than one at all locations within the Uptown CPU area. Therefore, this represents a less than significant chronic health impact.

Based on the preceding analysis the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to exposure of sensitive receptors to carbon monoxide hot spots and toxic air emissions.

Issue 4 Odors

Would the project create objectionable odors affecting a substantial number of people?

A potential odor impact can occur from two different situations: 1) the proposed Uptown CPU and associated discretionary actions would introduce receptors in a location where they would be affected by an existing or future planned odor source, or 2) proposed uses within the proposed Uptown CPU and associated discretionary actions would generate odors that could adversely affect a substantial number of persons.

The proposed Uptown CPU and associated discretionary actions would allow for development of single-family residential, multi-family residential, commercial, institutional, hotel, and park and open space land uses within the Uptown CPU area. While specific future land uses within the Uptown CPU area are not known at this program level of analysis, planned land uses would not encourage or support uses that would be associated with significant odor generation. The proposed Uptown CPU applies land uses based on the developed nature of the Uptown CPU area that includes residential uses in close proximity to commercial areas. A typical use in the CPU area that would generate odors would be restaurants. Restaurants can create odors from cooking activities, but would not generally be considered adverse. Odors associated with restaurants or other commercial uses would be similar to existing residential and food service uses throughout the Uptown CPU area. Odor generation is generally confined to the immediate vicinity of the source. Thus, implementation

of the proposed Uptown CPU and associated discretionary actions would not create operationalrelated objectionable odors affecting a substantial number of people within the City.

Cumulative Impacts

a. Air Quality Plans

For purposes of Issue 1, the cumulative study area would be considered the SDAB. Since the analysis provided under Issue 1 is a discussion of consistency with the air quality plan for the SDAB (i.e. the RAQS), the analysis provided a cumulative analysis by nature since it considers consistency of the project with a regional air quality plan that relies on the land use plans of jurisdictions within the basin. As discussed above, the proposed Uptown CPU and associated discretionary actions would generated less air emissions than the air emissions associated with build-out of the adopted Community Plan. Thus, the proposed Uptown CPU and associated discretionary actions would result in emissions less than what were anticipated when the RAQS were developed and the proposed Uptown CPU and associated discretion of the air quality plan. Thus, cumulative impacts related to conflicts with air quality plans would be less than significant.

b. Air Quality Standards

Construction

As shown in Table 6.4-2 above, the hypothetical worst case individual projects would not result in air emissions that would exceed the applicable thresholds. If several of these worst case hypothetical projects were to occur simultaneously, there is the potential to exceed significance thresholds. However, in order for exceedance of construction emissions thresholds to occur, more than one large scale project would have to be occurring within close proximity to one another with overlapping construction schedules. While unlikely to occur based on the fact that the Uptown CPU area is largely built out, future environmental review for these larger projects would allow for a sitespecific analysis of construction level air quality emissions to ensure projects are appropriately phased and timed to avoid such cumulative construction emissions. Thus, with implementation of the existing regulatory framework, cumulative construction emissions would be less than significant.

Operation

Regarding operational emissions, for purposes of this program level analysis, consistency with the RAQS was considered the applicable threshold since the City's project specific air quality impact screening levels shown in Table 6.4-1 would not be applicable to a community wide plan update. As discussed, build-out of the Uptown CPU area would result in emissions below what was used in the assumptions used to develop the RAQS; thus, overall build-out of the Uptown CPU area would not result in operational emission impacts. Since the RAQS are established for the SDAB which is the cumulative study area for air quality emissions, build-out of the land uses within the Uptown CPU area would not have the potential to result in a significant cumulative impact. Thus, cumulative operational emissions associated with build-out of the proposed Uptown CPU and associated discretionary actions would be less than significant.

c. Sensitive Receptors

CO Hot Spots

The CO hot spot analysis evaluated three intersections in the Uptown CPU area. The hot spot analysis indicated that the increases of CO due to the implementation of the CPU would be below the federal and state 1-hour and 8-hour standards. Since CO hot spots are a localized phenomenon, development within other community plans would not contribute to a cumulative CO hot spot impact.

Toxic Air Emissions

As discussed under Issue 2 above, the San Diego APCD would require an emissions inventory and health risk assessment in accordance with Assembly Bill 2588 prior to issuance of any permits to construct or operate a stationary emission source. These requirements would extend to land uses within the Uptown CPU area in addition to land uses within the SDAB as a whole. Thus, existing laws are in place that require evaluation and reduction of risks for individual projects developed in accordance with applicable and use plans. Site specific evaluation of health risks associated with stationary sources cannot be conducted at this level of review, as the project does not include specific development proposals. Nevertheless, existing regulations would ensure that cumulative impacts associated with stationary sources of toxic air emissions would be less than significant as build-out of the plan occurs.

As discussed above under Issue 3, the carcinogenic risks associated with diesel-fueled vehicles operating on local freeways would be less than ten in a million within the Uptown CPU area and the non-carcinogenic risks from PM_{10} are measured to have a maximum chronic hazard index below the significance threshold of one. Development of cumulative projects within the SDAB would not exacerbate health effects since the evaluation is location specific considering exposure to contaminants at a specific location. Therefore, the cumulative carcinogenic and non-carcinogenic toxic air emissions from exposure of residents to diesel particulate matter emissions would be less than significant.

d. Odors

For purposes of odor impacts, build-out of the three proposed Community Plans including North Park, Golden Hill, and Uptown is considered within the cumulative analysis. Implementation of the CPUs would not result in a significant cumulative odor impact because the CPUs and associated discretionary actions would result in single-family residential, multi-family residential, commercial, and park and open space land uses. These uses are not associated with generation of substantial odors. Additionally, odors are typically confined to the immediate area surrounding their source and thus, individual odor sources would not combine to produce a cumulative impact. Thus, objectionable odors affecting a substantial number of people within the City would not result, and cumulative odor impacts would be less than significant.

6.4.4 Significance of Impacts

Future operational emissions from the build-out of the proposed Uptown CPU and associated discretionary actions would be less than anticipated for future operational emissions under the adopted Community Plan. Thus, emissions associated with the proposed Uptown CPU and associated discretionary actions are already accounted for in the RAQS, and adoption of the proposed Uptown CPU and associated discretionary actions would not conflict with the RAQS. Thus regarding Issue 1, impacts related to conflicts with applicable air quality plans would be less than significant.

Regarding construction emissions under Issue 2, based on the hypothetical worst case construction emission analysis discussed previously, air emissions associated with build-out of individual projects under the proposed Uptown CPU and associated discretionary actions would be less than significant. Additionally, based on the types and scale of projects that are ministerial, air emissions associated with ministerial projects would not be of a size that would have the possibility of exceeding project level thresholds for air quality. Thus, construction emissions would be less than significant.

Regarding operational emissions under Issue 2, build-out of the CPU area would exceed the City's project-level thresholds for the proposed Uptown CPU; however the proposed Uptown CPU and associated discretionary actions would emit fewer pollutants than would occur under the adopted Community Plan. Therefore, the air emissions from build-out of the proposed Uptown CPU and associated discretionary actions would not increase air pollutants in the region, would not further increase the frequency of existing violations of federal or state AAQS, or would not result in new exceedances. Therefore, operational air quality impacts associated with the adoption of the proposed Uptown CPU and associated discretionary actions would be less than significant.

Regarding impacts to sensitive receptors (Issue 3), implementation of the proposed Uptown CPU and associated discretionary actions would not result in any CO hotspots. Additionally, carcinogenic risks associated with diesel-fueled vehicles operating on local freeways would be less than the applicable threshold, and non-carcinogenic risks from diesel particulate matter would be below the maximum chronic hazard index. Thus, air quality impacts to sensitive receptors would be less than significant and no mitigation is required

Regarding Issue 4, odor impacts would be less than significant, as the proposed Uptown CPU and associated discretionary actions do not propose land uses associated with generation of adverse odors. No mitigation is required.

6.5 Greenhouse Gas Emissions

A Greenhouse Gas Analysis for the Uptown, North Park, and Golden Hill Community Plan Updates (CPUs) was prepared by RECON (September 18, 2015). A Supplemental Analysis to the Greenhouse Gas Analysis for Uptown, North Park, and Golden Hill Community Plan Updates was prepared by RECON dated May 16, 2016. These reports address greenhouse gas (GHG) emissions and impacts associated with the proposed Uptown CPU and associated discretionary actions. The reports are included as Appendix E-1 and E-2, respectively, to this Program Environmental Impact Report (PEIR) and form the basis for the discussion in this section.

6.5.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

6.5.1.1 Methodology and Assumptions

Annual GHG emissions due to the operation of build-out of the Uptown Community Plan area under the adopted and proposed plans were calculated using California Emissions Estimator Model (CalEEMod; CAPCOA 2013). The emissions sources include construction (off-road vehicles), mobile (on-road vehicles), area (fireplaces, consumer products [cleansers, aerosols, and solvents], landscape maintenance equipment, and architectural coatings), water and wastewater, and solid waste sources. Where project-specific data were not available, model inputs were based on information provided in the CalEEMod User's Guide (CAPCOA 2013).

GHG emissions are estimated in terms of metric tons of carbon dioxide equivalent (MT CO_2E). CO_2E emissions are the preferred way to assess combined GHG emissions because they give weight to the global-warming potential (GWP) of different gases. The GWP is the potential of a gas to warm the global climate in the same amount as an equivalent amount of emissions of carbon dioxide (CO_2). As example, CO_2 has a GWP of 1, methane (CH_4) has a GWP of 21, and nitrous oxide (N_2O) has a GWP of 310, which means CH_4 and N_2O have 21 and 310 times greater global warming effect than CO_2 , respectively.

a. Estimating Construction Emissions

At a program level, it would be speculative to estimate the schedule and construction requirements of individual projects that could occur in the Uptown CPU area. Thus, this analysis relies on the methodology used in the San Diego County Updated Greenhouse Gas Inventory (San Diego County 2013), which forecasts that between 2015 and 2035 construction emissions would comprise roughly

2.1 percent of total GHG emissions within the county. Therefore, construction emissions are estimated at 2.1 percent of the total operational GHG emissions associated with the planning area.

b. Estimating Vehicle Emissions

Vehicle emissions are calculated based on the vehicle type, the trip rate, and trip length for each land use. The vehicle emission factors and fleet mix used in CalEEMod are derived from California Air Resources Board's (CARB) Emission Factors 2011 model, which includes GHG reducing effects from the implementation of Pavley I (Clean Car Standards) and the Low Carbon Fuel Standard, and are thus considered in the calculation of emissions. Emission factors that include the effects of the Tire Pressure Program and the Low Emission Vehicles III regulations are not available. Therefore, to account for the effects of the Tire Pressure Program (0.6 percent) and the Low Emission Vehicles III (2.4 percent), a total 3 percent reduction was applied to the vehicle emissions calculated in CalEEMod (CARB 2011a).

The proposed Uptown CPU encourages increased development diversity by increasing commercial land uses in certain areas and decreasing the planned number of single-family and multi-family residences. Locating different land use types near one another can decrease vehicle miles traveled (VMT), as trips between land use types are shorter and may be accommodated by alternative modes of transportation (CAPCOA 2010). This reduction was calculated using methodology from California Air Pollution Control Officers Association's (CAPCOA) Quantifying Greenhouse Gas Mitigation Measures (CAPCOA 2010). By increasing residential density within proximity of transit and commercial services, people's travel distances are affected and greater options for the mode of travel are provided. This can result in a substantial reduction in VMT depending on the change in density compared to a typical urban residential density (CAPCOA 2010). By increasing transit accessibility (e.g., by locating a high-density project near transit), a shift in travel mode is facilitated along with reduced VMT. The effectiveness of these land-use strategies ranges from less than 1 percent up to a maximum 30 percent reduction in communitywide VMT and are not additive (CAPCOA 2010). For example, where high-density mixed use development is located within a 5- to 10-minute walk from a transit station with high-frequency transit or bus service and is combined with walkable and bicyclefriendly neighborhood design, a total VMT reduction up to 24 percent can be achieved (CAPCOA 2010). The proposed Uptown CPU's focus on community walkability and bikeability, diversity of land uses, and development of higher densities near job centers (downtown San Diego) was included in the CPU emission calculations. Based on a review of mapping, the average distance from areas with increased residential density to the nearest major job center, downtown San Diego, is approximately 1.9 miles for the Uptown CPU area. The proposed Uptown CPU and associated discretionary actions propose an increase in multi-family residences in close proximity to transit and existing commercial uses. The VMT from residents of these new developments would be less due to the reduced trip lengths. Although this reduction was only counted for new development under the proposed Uptown CPU, this would reduce overall mobile emissions by 5.9-2 percent in the Uptown CPU area.

c. Estimating Energy Use Emissions

CalEEMod estimates GHG emissions from energy use by multiplying average rates of residential and non-residential energy consumption by the quantities of residential units and non-residential square footage entered in the land use module to obtain total projected energy use. This value is then

multiplied by electricity and natural gas GHG emission factors applicable to the project location and utility provider.

Building energy use is typically divided into energy consumed by the built environment and energy consumed by uses that are independent of the construction of the building such as plug-in appliances. In California, Title 24 governs energy consumed by the built environment, mechanical systems, and some types of fixed lighting. Non-building energy use, or "plug-in energy use," can be further subdivided by specific end-use (refrigeration, cooking, office equipment, etc.).

Energy consumption values are based on the California Energy Commission (CEC) sponsored *California Commercial End Use Survey and Residential Appliance Saturation Survey* studies, which identify energy use by building type and climate zone. Because these studies are based on older buildings, adjustments have been made in CalEEMod to account for changes to Title 24 Building Codes. CalEEMod is based on the 2008 Title 24 energy code (Part 6 of the Building Code).

As identified by the CEC, the Energy Code requires various improvements in the built environment that would achieve a 21.8 percent increase in electricity efficiency and a 16.8 percent increase in natural gas efficiency in non-residential buildings, a 36.4 percent increase in electricity efficiency and a 6.5 percent increase in natural gas efficiency in single-family uses, and a 23.3 percent increase in electricity efficiency and a 3.8 percent increase in natural gas efficiency in multi-family uses (CEC 2013).

The Uptown CPU area would be served by San Diego Gas & Electric (SDG&E). Therefore, SDG&E's specific energy intensity factors (i.e., the amount of CO₂, CH₄, and N₂O per kilowatt-hour) are used in the calculations of GHG emissions. The state mandate for renewable energy is 33 percent by 2020 and 50 percent by 2030 (RECON 2015). However, the energy intensity factors included in CalEEMod by default only represent a 10.2 percent procurement of renewable energy (SDG&E 2011). SDG&E currently has procured 36.4 percent and would achieve 50 percent by 2030. To account for the continuing effects of Renewables Portfolio Standard (RPS) through 2020, the energy intensity factors included in CalEEMod were reduced based on the percentage of renewables reported by SDG&E. SDG&E energy intensity factors that include this reduction are shown in Table 6.5-1.

Table 6.5-1 San Diego Gas & Electric Intensity Factors						
GHG	2009	2016	2020	2035		
	(lbs/MWh)	(lbs/MWh)	(lbs/MWh)	(lbs/MWh)		
Carbon dioxide (CO ₂)	720.49	531.72	531.72	433.73		
Methane (CH ₄)	0.029	0.021	0.021	0.017		
Nitrous oxide (N ₂ O)	0.006	0.004	0.004	0.004		
SOURCE: SDG&E 2011.						
lbs = pounds						
MWh = megawatt hour						

d. Estimating Area Source Emissions

Area sources include GHG emissions that would occur from the use of landscaping equipment. The use of landscape equipment emits GHGs associated with the equipment's fuel combustion. The landscaping equipment emission values were derived from the 2011 In-Use Off-Road Equipment Inventory Model (CARB 2011b).

e. Estimating Water and Wastewater Emissions

The amount of water used and wastewater generated by a project has indirect GHG emissions associated with it. These emissions are a result of the energy used to supply, distribute, and treat the water and wastewater. In addition to the indirect GHG emissions associated with energy use, wastewater treatment can directly emit both CH_4 and N_2O .

The indoor and outdoor water use consumption data for each land use subtype comes from the Pacific Institute's *Waste Not, Want Not: The Potential for Urban Water Conservation in California* 2003 (as cited in CAPCOA 2013). Based on that report, a percentage of total water consumption was dedicated to landscape irrigation, which is used to determine outdoor water use. Wastewater generation was similarly based on a reported percentage of total indoor water use (CAPCOA 2013).

Development would be subject to California Green Building Standards Code (CalGreen), which requires a 20 percent increase in indoor water use efficiency. Thus, in order to demonstrate compliance with CalGreen, a 20 percent reduction in indoor water use was included in the water consumption calculations.

In addition to water reductions under CalGreen, the GHG emissions from the energy used to transport the water are affected by RPS. As discussed previously, to account for the effects of RPS through 2020 and 2030, the energy intensity factors included in CalEEMod were reduced by the values shown in Table 6.5-1.

f. Estimating Solid Waste Emissions

The disposal of solid waste produces GHG emissions from anaerobic decomposition in landfills, incineration, and transportation of waste. To calculate the GHG emissions generated by disposing of solid waste for the project, the total volume of solid waste was calculated using waste disposal rates identified by California Department of Resources Recycling and Recovery. The methods for quantifying GHG emissions from solid waste are based on the Intergovernmental Panel on Climate Change (IPCC) method using the degradable organic content of waste. GHG emissions associated with the project's waste disposal were calculated using these parameters. No solid waste reductions were modeled.

6.5.2 Significance Determination Thresholds

Thresholds used to evaluate potential impacts related to GHG emissions are based on applicable criteria in the California Environmental Quality Act (CEQA) Guidelines Appendix G. A significant impact could occur if implementation of a proposed CPU would:

- 1) Generate GHG emissions, either directly or indirectly, that may have a significant impact on the environment; or
- 2) Conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emission of GHGs.

As stated in the Guidelines, these questions are "intended to encourage thoughtful assessment of impacts and do not necessarily represent thresholds of significance" (Title 14, Division 6, Chapter 3 Guidelines for Implementation of the CEQA, Appendix G, VII Greenhouse Gas Emissions). The CEQA Guidelines require lead agencies to adopt GHG thresholds of significance. When adopting these thresholds, the Guidelines allow lead agencies to develop their own significance threshold and/or to consider thresholds of significance adopted or recommended by other public agencies, or recommended by experts, provided that the thresholds are supported by substantial evidence.

Section 15064.4 of the CEQA Guidelines includes the following requirements for determining the significance of impacts from GHG emissions:

- (a) The determination of the significance of greenhouse gas emissions calls for a careful judgment by the lead agency consistent with the provisions in section 15064. A lead agency should make a good-faith effort, based to the extent possible on scientific and factual data, to describe, calculate, or estimate the amount of GHG emissions resulting from a project. A lead agency shall have discretion to determine, in the context of a particular project, whether to:
 - (1) Use a model or methodology to quantify greenhouse gas emissions resulting from a project, and which model or methodology to use. The lead agency has discretion to select the model or methodology it considers most appropriate provided it supports its decision with substantial evidence. The lead agency should explain the limitations of the particular model or methodology selected for use; and/or
 - (2) Rely on a qualitative analysis or performance-based standards.

While calculation of a project's contribution to greenhouse gas emissions is required, the CEQA Guidelines do not establish a standard by which to judge a significant effect or a means to establish such a standard. In order to determine significance of the impacts associated with implementation of the proposed Uptown CPU and associated discretionary actions, an inventory was developed based on the land use designations associated with the adopted Community Plan. Emissions from the proposed Uptown CPU and associated discretionary actions were then compared to the existing GHG emissions inventory and the GHG emissions inventory for the adopted Community Plan. If emissions from build-out of the Uptown CPU and associated discretionary actions are less than those that would be generated by build-out of the adopted Community Plan, impacts related to GHG emissions would be less than significant provided the proposed Uptown CPU and associated discretionary actions are less than (CAP). If emissions from build-out of the proposed Uptown CPU and associated discretionary actions are greater than those of the adopted Community Plan, impacts related to GHG emissions are greater than those of the adopted Community Plan, impacts related to GHG emissions are greater than those of the adopted Community Plan, impacts related to GHG emissions are greater than those of the adopted Community Plan, impacts related to GHG emissions could still be less than significant if the increase in GHG emissions is a direct result of implementing CAP strategies and the General Plan's City of Villages Strategy.

As discussed in Section 5.5 of this PEIR, implementation of the City's CAP would result in Citywide GHG reductions consistent with its proportionate share of Statewide GHG emissions targets. The CAP assumes future population and economic growth based on the community plans that were in effect at the time the CAP was being developed. Therefore, community plan updates that would result in a reduction in GHG at build-out compared to GHG emissions at build-out under the adopted Community Plan would result in further GHG reductions. However, the CAP is a Citywide program and the General Plan City of Villages Strategy calls for redevelopment, infill, and new growth to be targeted into compact, mixed-use, and walkable villages that are connected to the regional transit system. Concentrating new growth in an area can result in greater GHG emissions than allowing the less intensive land uses to remain. Thus, consistency with the City of Villages Strategy can result in specific areas having an increase in GHG emissions, while Citywide a decrease of GHG emissions may occur. To address this phenomenon, this section takes a two-tiered approach in discussing GHG emissions: 1) a quantitative analysis of the existing conditions, build-out of the adopted Community Plan, and build-out of the proposed North Uptown CPU and implementation of the associated discretionary actions; and 2) a discussion of whether or not the Uptown CPU and associated discretionary actions are consistent with the CAP.

6.5.3 Impact Analysis

Issue 1 Greenhouse Gas Emissions

Would the proposed project generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment?

As compared to the existing land uses, the proposed Uptown CPU and associated discretionary actions would reduce industrial, institutional, hotel, and single-family residential land uses while increasing the development of commercial uses and multi-family dwelling units. This change represents an increase in land use types and density in the Community Plan area. Table 6.5-2 summarizes the land use distribution for the Uptown Community Plan area for existing conditions, the adopted Community Plan, and the proposed Uptown CPU and associated discretionary actions.

Table 6.5-2 Land Use Distribution								
	Adopted Community Proposed Community							
Land Use	Existing Land Use	Plan	Plan					
Residential (dwelling u	units)							
Single-Family	7,540	5,540	5,500					
Multi-Family ¹	15,620	29,060	27,180					
Subtotal ² 23,160 34,600 32,680								
Non-Residential (squa	re feet)							
Commercial	4,184,170	4,783,000	4,785,200					
Industrial	dustrial 19,710							
Institutional	Institutional 2,627,550 2,314,900 2,485,700							
Hotels	366,460	174,000	174,000					
Recreation	31,110	31,100	31,100					
Subtotal ²	Subtotal ² 7,229,000 7,303,000 7,476,000							
¹ All dwelling units that are not single-family were counted as multi-family. This includes dwelling								
units on other land uses such as commercial and institutional.								
² Total area may not match the sum of listed areas due to rounding.								

Based on the methodology summarized above, GHG emissions were calculated for the existing (on the ground) land uses, the land uses at build-out of the adopted Community Plan (in 2035, and the land uses at build-out of the proposed Uptown CPU and associated discretionary actions (in 2035). Table 6.5-3 summarizes the GHG emissions under each scenario.

Table 6.5-3 GHG Emissions for the Uptown Community Plan Area (MT CO₂E per Year)						
				Difference		
		Adopted		(Proposed –		
Emission Source	Existing	Community Plan	Proposed CPU	Adopted)		
Vehicles	382,422	380,530<u>3</u>47,868	372,922<u>340,913</u>	- 7,608<u>6,</u>955		
Energy Use	80,430	85,603<u>74,909</u>	83,533<u>73,046</u>	- 2,070<u>1,863</u>		
Area Sources	16,805	25,105	23,712	-1,393		
Solid Waste Disposal	16,411	17,459	17,488	29		
Water Use	14,339	15,969<u>13,594</u>	15,494<u>13,189</u>	-475 <u>405</u>		
Construction	n/a	11,018<u>10,058</u>	10,776<u>9,835</u>	- <u>242222</u>		
TOTAL	510,407	535,684 488,993	523,925 478,184	- 11,759 10,809		

For the purposes of determining significance, GHG emissions attributable to the proposed Uptown CPU and associated discretionary actions at full build-out were compared to the adopted Community Plan GHG emissions. The reason this comparison is appropriate is because the GHG emissions from the adopted Community Plan were used when developing the City's CAP GHG Inventory. Thus, if calculated future emissions of the proposed Uptown CPU would be consistent with, or less than, the emissions assumed in developing the CAP inventory, build-out of future land uses under the proposed Uptown CPU and associated discretionary actions would be consistent with the City's CAP emissions inventory and would not represent a significant impact related to GHG emissions. As illustrated in Table 6.5-3 the total GHG emissions attributable to build-out of land uses

under the adopted Community Plan equals 535,684488,993 MT CO₂E per year. The total GHG emissions attributable to build-out of the land uses under the proposed Uptown CPU and associated discretionary actions equals 523,925478,184 MT CO₂E per year. As shown in the above table, implementation of the proposed Uptown CPU and associated discretionary actions would result in a decrease in GHG emissions of 11,75910,809 MT CO₂E when compared to the emissions that would occur under build-out of the adopted Community Plan. Because the proposed Uptown CPU and associated discretionary actions when compared with land used currently approved, impacts would be less than significant.

Issue 2 Conflicts with Plans or Policies

Would the proposed project conflict with an applicable plan, policy, or regulation adopted for the purpose of reducing emissions of greenhouse gases?

The regulatory plans and policies discussed in Section 5.5 of this PEIR aim to reduce national, state, and local GHG emissions by primarily targeting the largest emitters of GHGs: the transportation and energy sectors. Plan goals and regulatory standards are, thus, largely focused on the automobile industry and public utilities. For the transportation sector, the reduction strategy is generally three-pronged: to reduce GHG emissions from vehicles by improving engine design; to reduce the carbon content of transportation fuels through research, funding and incentives to fuel suppliers; and to reduce the miles these vehicles travel through land use change and infrastructure investments.

For the energy sector, the reduction strategies aim to: reduce energy demand; impose emission caps on energy providers; establish minimum building energy and green building standards; transition to renewable non-fossil fuels; incentivize homeowners and builders; fully recover landfill gas for energy; and expand research and development.

a. Consistency with State Plans

As discussed earlier, Executive Order S-3-05 establishes GHG emission reduction targets for the state, and Assembly Bill (AB) 32 launched the Climate Change Scoping Plan that outlines the reduction measures needed to reach these targets. Out of the Recommended Actions contained in CARB's Scoping Plan, the actions that are most applicable to the proposed Uptown CPU and associated discretionary actions would be Actions E-1 and GB-1. CARB Scoping Plan Action E-1, together with Action GB-1 (Green Building), aim to reduce electricity demand by increasing the efficiency of Utility Energy Programs and adoption of more stringent building and appliance standards. The new construction associated with the proposed Uptown CPU and associated discretionary actions would be required to include all mandatory green building measures under the CalGreen Code. Therefore, the proposed Uptown CPU and associated discretionary actions would be strong plan measures through incorporation of stricter building and appliance standards.

b. Consistency with Regional Plans

San Diego Association of Government's (SANDAG's) San Diego Forward: The Regional Plan

The proposed Uptown CPU and associated discretionary actions would be consistent with the goals of the Regional Plan to develop compact, walkable and bicycle-friendly communities close to transit connections and consistent with smart growth principles. The proposed Uptown CPU and associated discretionary actions would reinforce transit corridors, bicycle lanes, and establish five pedestrianoriented, urban, and mixed-use Villages that would reduce reliance on the automobile, and promote walking and biking and the use of alternative transportation. Policies contained within the proposed Uptown CPU Land Use and Mobility elements would serve to promote bus transit use as well as other forms of mobility, including walking and bicycling. These measures would be consistent with the Regional Plan's Sustainable Communities Strategy. Thus, no significant adverse environmental effects would result from the adoption of the proposed Uptown CPU and associated discretionary actions in terms of consistency or conflicts with the Regional Plan.

c. Consistency with Local Plans

City of San Diego General Plan

Compared to the existing land uses, the proposed Uptown CPU envisions reducing industrial, institutional, hotel, and single-family residential land uses and increasing commercial space and multi-family dwelling units. This would increase the diversity of land uses within the CPU area by encouraging "village-like" development consistent with the San Diego General Plan. The proposed Uptown CPU and associated discretionary actions also support General Plan concepts including increased walkability, a higher level of alternative transportation use, and sustainable development and green building practices.

Policies within the Land Use Element of the proposed Uptown CPU promote mixed-use development along major transportation corridors, specifically calling out Washington Street, University Avenue, 5th Avenue, 6th Avenue, Laurel Street, and Park Boulevard for a diversity of uses. Policies within the Mobility Element of the proposed Uptown CPU promote multi-modal development, enhanced pedestrian and bicycle facilities, and active storefronts to increase pedestrian engagement. Policies within the Conservation Element of the proposed Uptown CPU promote golar panels and the preservation and planting of street trees. All of these policies correspond with policies set out by the General Plan. Thus, the proposed Uptown CPU and associated discretionary actions would be consistent with the San Diego General Plan.

City of San Diego Climate Action Plan

New land use designations and policies within the proposed Uptown CPU have been designed to reflect and implement the CAP and the GHG reduction recommendations of the General Plan. Specifically, the proposed Uptown CPU includes updated Land Use, Mobility, and Conservation elements that include multiple policies aimed at reducing GHG emissions from target emission sources and adapting to climate change. The proposed policies refine existing General Plan policies

with site-specific recommendations applicable to the individual community. In several cases, these policies are also consistent with state key GHG reduction plans, regulations, and recommended mitigation measures.

The CAP establishes five primary strategies for achieving the goals of the plan. Strategy 1 (Energy & Water Efficient Buildings) includes goals, actions, and targets with the aim of reducing building energy consumption. Energy reduction can be achieved through the continued use or adaptive reuse of the existing building stock along with any needed energy efficiency upgrades. The proposed Uptown CPU includes narrative and policies in the Conservation Element for creation of energy buildings, and more specifically, the retrofitting of public right-of-way lighting with energy efficient lighting to meet the City's energy efficiency goals outlined in the CAP. Another goal in Strategy 1 is to reduce daily per capita water consumption. The proposed Uptown CPU includes discussion and policies to address water usage within the Urban Design Element and Conservation Element, and encourages sustainable building design and incorporation of building features that would reduce water consumption. This is coupled with reducing the dependency on non-renewable energy sources and the maximization of daylight, the minimization of solar heat gain and natural ventilation, and the reduction of emissions. The Conservation Element also includes policies for use of recycled or graywater landscape irrigation systems and the retrofitting of public spaces with low-water vegetation, which would, in turn, reduce water usage.

Regarding CAP Strategy 2 (Clean & Renewable Energy), the Urban Design Element of the proposed Uptown CPU includes a policy to encourage development that incorporates renewable energy, such as small low-impact wind turbines or photo-voltaic panels on roof tops. The Conservation Element of the proposed Uptown CPU also contains an overarching goal to reduce dependence on nonrenewable energy sources, and policies that include the use of sustainable building techniques for construction and operation of buildings that could include solar energy installations, electric vehicle charging stations, and solar water heating.

Strategy 3 (Bicycling, Walking, Transit & Land Use) of the CAP has a number of goals that relate to land use and planning. As discussed in Section 6.1.3 of this PEIR, the proposed Uptown CPU is consistent with the General Plan's Mobility Element and the City of Villages Strategy and is thus consistent with Action 3.1 of the CAP. Consistent with Action 3.2 of the CAP, the proposed Uptown CPU would mixed-use development and would promote pedestrian improvements in Transit Priority Areas to increase commuter walking opportunities. Consistent with Action 3.6 of the CAP, the proposed Uptown CPU would implement transit-oriented development, particularly within and around the two Community Villages and three Neighborhood Villages.

The primary goal of CAP Strategy 4 (Zero Waste – Gas & Waste Management) is to divert solid waste and capture landfill methane gas emissions. This strategy is Citywide in nature; however, the proposed Uptown CPU furthers this strategy by including policies in the Urban Design Element that support the use of recycled materials in public improvements, encourages recycled or rapidly renewable source materials, and recycling of building materials for both public and private new development. The Urban Design Element includes a policy that supports the incorporation of recycling containers into streetscapes as well as public trails, and ensures that the locations are protected from weather and are secure so the containers cannot be removed and do not spill. Strategy 5 (Climate Resiliency) of the CAP calls for further analysis of the resiliency issues that face the various areas of the City. Resiliency is addressed throughout the proposed Uptown CPU as it pertains to water usage, energy efficiency, and sustainable development practices as noted above. Also included within the proposed Uptown CPU are policies supporting and encouraging an increase in the tree canopy within the community to reduce summer heat temperatures and contribute to more inviting business districts for pedestrians. The selection, siting, and management of the planting of street trees within the Uptown CPU area are outlined within the proposed CPU to ensure successful establishment of trees to meet the CAP goals.

As mentioned in Section 5.5 of this PEIR, the CAP's Monitoring and Reporting Program Measure 1.4 calls for City Staff to annually evaluate City policies, plans (including the CAP), and codes as needed to ensure the CAP reduction targets are met. Through monitoring the effectiveness of CAP actions at reducing GHG emissions, the City would be able to make adjustments to the CAP, which could include amending land use plans to reflect more aggressive strategies for GHG reduction. Therefore, the proposed Uptown CPU and associated discretionary actions would be consistent with and would implement the CAP.

Cumulative Impacts

The impact analysis discussed under Issue 1 above is a cumulative analysis by its nature because GHG emissions are a cumulative issue caused by the global greenhouse gas emissions and not an individual project. Cumulatively, there exists a significant impact related to greenhouse gas emissions at the global level. However, as discussed under Issue 1 above, the project's contribution to the cumulative impact from GHG emissions would be less than cumulatively considerable because implementation of the proposed Uptown CPU and associated discretionary actions would result in a reduction in the future GHG emissions compared to the current land use plan. As discussed under Issue 2, City policies, plans, and codes will be evaluated as needed to ensure that CAP GHG emissions reduction targets are met. If implementation of the proposed Uptown CPU and associated discretionary actions cumulatively with other CPUs would be inconsistent with the CAP or other plans/policies for the reduction of GHG, the City could amend land use plans to reflect more aggressive strategies for GHG reduction and to ensure consistency with the adopted CAP. Thus, the contribution of the proposed Uptown CPU and associated discretionary actions to the existing cumulative impact would be less than cumulatively considerable.

6.5.4 Significance of Impacts

Potential impacts related to GHG emissions from implementation of the proposed Uptown CPU and associated discretionary actions would be less than significant, as the GHG emissions from the Uptown CPU would be less than those assumed for the Uptown CPU area in the CAP GHG Inventory. Thus, the proposed Uptown CPU and associated discretionary actions would be consistent with the CAP and would result in a less than significant impact related to GHG emissions.

The proposed Uptown CPU would implement the General Plan's City of Villages Strategy and include policies for the promotion of walkability and bicycle use, polices promoting transit-supportive development, and is thus consistent with the CAP and the General Plan. Impacts related to conflicts

with applicable plans and policies addressing GHG emissions would be less than significant and no mitigation is required.

6.5.5 Mitigation Framework

All impacts related to greenhouse gas emissions would be less than significant. Thus, no mitigation is required.

6.6 Noise

This section addresses the potential noise impacts that would result from implementation of the proposed Uptown Community Plan Update (CPU) and associated discretionary actions. It also discusses the regulations applicable to subsequent projects contemplated by the proposed Uptown CPU and associated discretionary actions and the existing noise setting within the study area. This section is based on the Noise Analysis for the Uptown, North Park, and Golden Hill Community Plan Updates (Noise Report) prepared by RECON (2016) for the project (Appendix F).

6.6.1 Existing Conditions

The existing regional environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively. The specific noise conditions for the Uptown CPU area are discussed in the following sections.

Existing noise sources in the Uptown CPU area are transportation and stationary sources. Transportation noise sources include vehicle traffic, and overflight of aircraft approaching and departing the San Diego International Airport (SDIA). Stationary noise sources include industrial and commercial operations. Noise from these sources conflicts with existing noise sensitive receptors throughout the communities.

6.6.1.1 Noise Measurements

As part of the noise assessment, ambient noise levels were measured in the planning area to provide a characterization of the variability of noise throughout the Uptown CPU area and to assist in determining constraints and opportunities for future development. Ambient noise levels were measured to characterize the variability of noise and to assist in determining constraints and opportunities to avoid noise conflicts. Six 15-minute, daytime noise level measurements were conducted throughout the study area. Noise measurements were taken with two Larson-Davis LxT Type 1 Integrating Sound Level Meters, serial numbers 3827 and 3828. Each measurement location is shown in Figure 6.6-1. A summary of the measurements is provided in Table 6.6-1.

Based on the measurement data shown in Table 6.6-1, daytime noise levels in the Uptown CPU area are typical of an urban environment. Each measurement location and noise source observed during the measurements is discussed below.

Table 6.6-1 Noise Measurements – Uptown						
ID*	Location	Date	Time	L _{eq}		
U-1	Columbia Street	3/03/2015	10:12 а.м. – 10:27 а.м.	77.6		
U-2	San Diego Avenue	3/03/2015	9:25 a.m. – 9:40 a.m.	69.1		
U-3	Washington Street	3/03/2015	10:51 а.м. – 11:06 а.м.	64.5		
U-4	Reynard Way	3/03/2015	1:30 р.м. – 1:45 р.м.	57.7		
U-5	Sixth Avenue	3/03/2015	12:27 р.м. – 12:42 р.м.	63.5		
U-6	Normal Street	3/03/2015	11:46 а.м. – 12:01 р.м.	64.4		
*Measurement locations are shown in Figure 6.6-1 and are represented by the ID provided above. L_{eq} = average noise level.						

Measurement U-1 was taken on Columbia Street adjacent to Interstate 5 (I-5). The main sources of noise at the measurement location were vehicle traffic on I-5 and aircraft arriving at and departing from the SDIA. The average measured noise level was 77.6 A-weighted decibels [dB(A) L_{eq}].

Measurement U-2 was taken adjacent to San Diego Avenue. The measured speed on this portion of San Diego Avenue was 35 miles per hour (mph). The main source of noise at the measurement location was vehicle traffic on I-5, San Diego Avenue, and India Street. The average measured noise level was 69.1 dB(A) L_{eq}.

Measurement U-3 was taken adjacent to Keating Street on top of a slope overlooking Washington Street. The main source of noise at the measurement location was vehicle traffic on Washington Street. The measured speed on this portion of Washington Street was 50 mph. The average measured noise level was $64.5 \text{ dB}(A) \text{ L}_{eq}$.

Measurement U-4 was taken adjacent to Reynard Way. The main source of noise at the measurement location was vehicle traffic on Reynard Way. The measured speed on this portion of Reynard Way was 30 mph. The average measured noise level was 57.7 dB(A) L_{eq} .

Measurement U-5 was taken adjacent to Sixth Avenue. The main source of noise at the measurement location was vehicle traffic on Sixth Avenue. The measured speed on this portion of Sixth Avenue was 30 mph. The average measured noise level was $63.5 \text{ dB}(A) L_{eg}$.

Measurement U-6 was taken adjacent to Normal Street. The main source of noise at the measurement location was vehicle traffic on Normal Street and Polk Avenue. The measured speed on this portion of Normal Street was 30 mph. The average measured noise level was 64.4 dB(A) L_{eq}.

ATTACHMENT 7

Image source: SanGIS (flown May 2012)









Noise Measurements

FIGURE 6.6-1

Noise Measurement Locations

 $M:\JOBS4\6086\common_gis\fig6.6-1_EIR_UT.mxd \ 6/3/2016 \ fmm$

6.6.1.2 Existing Vehicle Traffic Noise

The dominant noise source is vehicle traffic on roadways. Vehicle traffic noise is directly related to the traffic volume, speed, and mix of vehicles. Vehicles traveling on I-5, I-8, State Route 163 (SR-163) are the dominant vehicle noise sources affecting the Uptown CPU area. The streets generating the greatest noise level in the Uptown CPU area are Sixth Avenue, India Street, Park Boulevard, Robinson Avenue, University Avenue, and Washington Street. The noise contour distances represent the predicted noise level for each roadway without the attenuating effects of noise barriers, structures, topography, or dense vegetation. As intervening structures, topography, and dense vegetation would affect noise exposure at a particular location, the noise contours should not be considered site-specific but are rather guides to determine when detailed acoustic analysis should be undertaken.

Figure 6.6-2 shows the existing vehicle traffic noise contours for the Uptown CPU area. As shown, existing noise levels in the community exceed 60 dB(A) community noise equivalent level (CNEL). The freeways are the dominant noise sources affecting the Uptown CPU area and encompass the noise contours from streets in the CPU area.

6.6.1.3 Existing Rail Traffic Noise

Railway noise results from trolley travel, horns, emergency signaling devices, and stationary bells at grade crossings. The rail corridor generally parallels I-5 at the western boundary of the Uptown planning area in the Midway Pacific Highway Community Plan Area. The San Diego Metropolitan Transit System provides trolley service along the rail corridor. Amtrak operates passenger trains and the Coaster operates commuter trains along the rail corridor daily. The Burlington Northern Santa Fe Railway Company also operates freight trains along the corridor daily. The rail traffic noise is less than 60 dB CNEL within the Uptown CPU area.

6.6.1.4 Existing Aircraft Noise

The SDIA is located west of the Uptown CPU area. A majority of aircraft flying over the CPU area are approaching SDIA. Occasionally, aircraft flying over the CPU area are departing SDIA. Aircraft noise is evaluated based on the noise contours developed by the San Diego County Regional Airport Authority and provided in the Airport Land Use Compatibility Plan for San Diego International Airport (2014). The aircraft noise contours are based on year 2030 forecast noise exposure.

6.6.1.5 Existing Stationary Noise

Stationary sources of noise within the Uptown CPU area are due to the normal activities associated with a given land use. For example, within residential areas noise sources include dogs, landscaping activities, and parties. Commercial uses include car washes, fast food restaurants, and auto repair facilities. Sources of noise from commercial uses include machinery and truck loading/unloading. Noises from these types of activities would be considered normal environmental noises that would be expected to occur within these types of land uses and are not typically considered significant sources of noise. The Municipal Code regulates excessive noises resulting from these types of activities.



ATTACHMENT 7



FIGURE 6.6-2 Existing Traffic Noise Contours for the Uptown CPU Area

6.6.2 Significance Determination Thresholds

Thresholds used to evaluate potential impacts to air quality are based on applicable criteria in the California Environmental Quality Act (CEQA) Guidelines Appendix G and the City of San Diego CEQA Significance Determination Thresholds (2011). Thresholds are modified from the City's CEQA Significance Determination Thresholds to reflect the programmatic analysis for the proposed Uptown CPU. A significant impact related to noise would occur if the proposed Uptown CPU and associated discretionary actions would:

- 1) Result in or create a significant increase in the existing ambient noise levels;
- 2) Result in an exposure of people to current or future transportation noise levels which exceed guidelines established in the Noise Element of the General Plan;
- 3) Result in land uses which are not compatible with aircraft noise levels as defined by an adopted Airport Land Use Compatibility Plan (ALUCP);
- 4) Result in the exposure of people to noise levels which exceed property line limits established in the Noise Abatement and Control Ordinance of the Municipal Code; or
- 5) Result in the exposure of people to significant temporary construction noise.

6.6.2.1 Noise

Thresholds used to determine the significance of noise impacts are based on standards in the City General Plan Noise Element and the Noise Abatement and Control Ordinance (Section 59.5.0101 et seq. of Municipal Code) as described in the Regulatory Framework chapter, sections 5.6.2.1 and 5.6.2.2, respectively.

6.6.2.2 Vibration

While the City has not established specific groundborne noise and vibration standards, publications of the Federal Transit Administration (FTA) and California Department of Transportation (Caltrans) provide guidance for the analysis of environmental impacts due to groundborne noise and vibration relating to transportation and construction projects. Based on Caltrans recommended standards, a significant vibration impact would occur where residences would be exposed to an exceedance of 0.2 inch per second peak particle velocity.

6.6.3 Methodology

6.6.3.1 Vehicle Traffic Noise

Existing freeway volumes and traffic mixes were obtained from Caltrans and the San Diego Association of Governments (SANDAG) traffic and truck counts for the I-5, SR-163, and I-8. These
traffic mixes, further detailed in the Noise Analysis (see Appendix F), were used for modeling existing and future freeway noise. For all freeways, Caltrans existing truck counts indicate an approximate traffic mix of 96 percent cars, 3 percent medium trucks, and 1 percent heavy trucks. These traffic mixes were used for modeling existing and future freeway noise. This is consistent with traffic counts taken during the existing noise measurements, and the same as Caltrans truck counts for most area freeways.

The Federal Highway Administration (FHWA) Traffic Noise Model was used to calculate distances to noise contours for freeways and streets. The FHWA model takes into account traffic mix, speed, and volume; roadway gradient; relative distances between sources, barriers, and sensitive receptors; and shielding provided by intervening terrain or structures. The analysis of the noise environment considered that the topography was flat with no intervening terrain between sensitive land uses and roadways. Because no obstructions were assumed in the noise modeling, predicted noise levels used in the analysis are higher than would actually occur. In the actual environment, buildings and other obstructions along the roadways would shield distant receivers from the traffic noise. For example, I-8 and SR-163 and portions of I-5 are at lower elevations than the streets and buildings in the Uptown CPU area and it is likely that the slopes adjacent to the freeways reduce the actual noise levels.

6.6.4 Impact Analysis

Issue 1 Ambient Noise

Would the proposed project result in or create a significant increase in the existing ambient noise level?

As discussed in Section 6.6.1.1, Noise Measurements, existing noise levels were measured in the planning area to identify ambient noise conditions (refer to Table 6.6-1).

The freeways generating the greatest noise levels affecting the Uptown CPU area are I-5, I-8, and SR-163. The streets generating the greatest noise levels within the CPU area are Sixth Avenue, India Street, Park Boulevard, Robinson Avenue, University Avenue, and Washington Street. Increases in traffic noise gradually degrade the ambient noise environment, especially with respect to sensitive receptors. Vehicular traffic on streets in the CPU area would increase due to build-out of the proposed Uptown CPU and associated discretionary actions. Table 6.6-2 summarizes the existing and build-out traffic noise levels along various roadway segments in the Uptown CPU area. Roadway noise is measured in dB(A) CNEL at 50 feet from the roadway centerline.

6.6 Noise

Table 6.6-2								
Increases in Ambient Noise for the Uptown CPU Area								
			Existing	2035	Change			
Roadway	From	То	Noise Level	Noise Level	in dB			
First Avenue	Arbor Drive	Washington Street	62.1	63.7	1.6			
	Washington Street	University Avenue	63.6	64.5	0.9			
	University Avenue	Robinson Avenue	65.0	67.1	2.1			
	Robinson Avenue	Pennsylvania Avenue	63.7	65.6	1.9			
	Pennsylvania Avenue	Walnut Avenue	63.6	66.0	2.4			
	Walnut Avenue	Laurel Street	61.7	65.7	4.0			
	Laurel Street	Hawthorn Street	63.6	64.2	0.6			
	Hawthorn Street	Grape Street	63.6	63.3	-0.3			
	Grape Street	Elm Street	60.1	61.5	1.4			
Fourth Avenue	Arbor Drive	Washington Street	65.9	66.7	0.8			
	Washington Street	University Avenue	65.1	64.9	-0.2			
	University Avenue	Robinson Avenue	65.7	66.1	0.4			
	Robinson Avenue	Walnut Avenue	63.4	65.5	2.1			
	Walnut Avenue	Laurel Street	64.2	66.7	2.5			
	Laurel Street	Grape Street	63.9	66.3	2.4			
	Grape Street	Elm Street	63.7	64.8	1.1			
Fifth Avenue	Washington Street	University Avenue	65.6	65.7	0.1			
	University Avenue	Robinson Avenue	65.1	66.4	1.3			
	Robinson Avenue	Walnut Avenue	65.8	66.9	1.1			
	Walnut Avenue	Laurel Street	65.5	66.6	1.1			
	Laurel Street	Hawthorn Street	64.6	66.5	1.9			
	Hawthorn Street	Grape Street	65.0	66.5	1.5			
	Grape Street	Elm Street	64.6	65.0	0.4			
Sixth Avenue	Washington Street	University Avenue	67.2	71.5	4.3			
	University Avenue	Robinson Avenue	68.9	70.1	1.2			
	Robinson Avenue	Upas Street	66.7	69.7	3.0			
	Upas Street	Laurel Street	66.7	69.1	2.4			
	Laurel Street	luniper Street	65.0	67.1	2.1			
	luniner Street	Grape Street	65.3	67.7	2.4			
	Grape Street	Flm Street	65.2	68.0	2.8			
Ninth Avenue	Washington Street	University Avenue	60.6	62.4	1.8			
	Madison Avenue	Washington Avenue	58.4	61.0	2.6			
campus / wende	Washington Avenue	Polk Avenue	60.9	62.1	1.0			
Cleveland Avenue	Tyler Street	Lincoln Street	60.3	62.0	1.2			
Cicvelana Avenae	Lincoln Street	Richmond Street	62.3	63.2	0.9			
Curlow Street	Pobinson Avenue	Remard Way	55.7	60.0	13			
Elm Street	Second Avenue		62.4	62.7	5 0.3			
	Third Avenue	Fifth Avenue	62.5	62.0	0.5			
	Fifth Avenue	Sixth Avenue	02.5 62.5	62.5	0.5			
Fort Stackton Drive	Arista Streat	Sixui Avenue	62.5	62.5	0.0			
Fort Stockton Drive	Arista Street		58.6	60.3	1.7			
	Sunset Boulevard	Hawk Street	61.2	62.4	1.2			
	Hawk Street	Goldfinch Street	62.7	62.9	0.2			
F C	Goldfinch Street	Falcon Street	58.0	58.6	0.6			
Front Street		Arbor Drive	59.2	60.0	0.8			
	Arbor Drive	Washington Street	60.8	62.4	1.6			
Grape Street	Albatross Street	First Avenue	56.6	62.0	5.4			
	First Avenue	Third Avenue	59.7	62.0	2.3			
	Third Avenue	Sixth Avenue	56.6	62.9	6.3			

6.6 Noise

Table 6.6-2								
	Increases in Amb	ient Noise for the Uptor	wn CPU Area					
			Existing	2035	Change			
Roadway	From	То	Noise Level	Noise Level	in dB			
Hawthorn Street	Brant Street	First Avenue	65.6	66.7	1.1			
	First Avenue	Third Avenue	60.5	63.6	3.1			
	Third Avenue	Sixth Avenue	60.5	64.3	3.8			
India Street	Washington Street	Winder Street	69.7	66.4	-3.3			
	Winder Street	Glenwood Drive	69.7	66.3	-3.4			
	Glenwood Drive	Sassafras Street	70.2	70.8	0.6			
	Sassafras Street	Redwood Street	68.7	69.3	0.6			
	Redwood Street	Palm Street	68.2	69.1	0.9			
Juan Street	Harney Street	Witherby Street	58.6	61.6	3.0			
Laurel Street	Columbia Street	Union Street	64.8	66.6	1.8			
	Union Street	First Avenue	63.9	65.9	2.0			
	First Avenue	Third Avenue	63.9	65.5	1.6			
	Third Avenue	Sixth Avenue	64.0	66.4	2.4			
Lewis Street	Fort Stockton Drive	Goldfinch Street	59.1	59.5	0.4			
Lincoln Avenue	Washington Street	Park Boulevard	62.5	63.8	1.3			
Madison Avenue	Cleveland Avenue	Park Boulevard	59.1	61.2	2.1			
Meade Avenue	Cleveland Avenue	Park Boulevard	58.6	58.8	0.2			
Normal Street	Park Boulevard	Washington Street	68.4	69.5	1.1			
	Washington Street	University Avenue	61.9	61.8	-0.1			
Park Boulevard	Adams Avenue	Mission Avenue	66.7	65.1	-1.6			
	Mission Avenue	El Cajon Boulevard	65.7	67.1	1.4			
	El Cajon Boulevard	Polk Avenue	65.6	67.6	2.0			
	Polk Avenue	University Avenue	67.4	69.5	2.1			
	University Avenue	Robinson Avenue	67.6	69.0	1.4			
	Robinson Avenue	Upas Street	67.0	68.4	1.4			
	Upas Street	Zoo Place	67.4	68.5	1.1			
Reynard Way	Torrance Street	Curlew Street	57.9	62.2	4.3			
	Curlew Street	Laurel Street	63.5	64.3	0.8			
Richmond Street	Cleveland Avenue	University Avenue	61.9	62.9	1.0			
	University Avenue	Robinson Avenue	60.7	61.7	1.0			
	Robinson Avenue	Upas Street	60.4	62.5	2.1			
Robinson Avenue	Brant Street	First Avenue	56.4	60.0	3.6			
	First Avenue	Third Avenue	61.0	64.0	3.0			
	Third Avenue	Eighth Avenue	63.8	65.0	1.2			
	Eighth Avenue	Tenth Avenue	66.7	64.3	-2.4			
	Tenth Avenue	Richmond Street	68.2	65.8	-2.4			
	Richmond Street	Park Boulevard	63.6	64.6	1.0			
San Diego Avenue	Hortensia Street	McKee Street	63.7	66.2	2.5			
	McKee Street	Washington Street	67.4	68.6	1.2			
	Washington Street	India Street	62.9	64.5	1.6			
State Street	Laurel Street	Juniper Street	59.6	62.5	2.9			
Sunset Boulevard	Witherby Street	Fort Stockton Drive	57.5	60.0	2.5			

6.6 Noise

Table 6.6-2							
	Increases in Ambi	ent Noise for the Uptow	n CPU Area				
			Existing	2035	Change		
Roadway	From	То	Noise Level	Noise Level	in dB		
University Avenue	Ibis Street	Albatross Street	63.6	65.1	1.5		
	Albatross Street	First Avenue	65.7	66.6	0.9		
	First Avenue	Fourth Avenue	66.5	64.9	-1.6		
	Fourth Avenue	Fifth Avenue	66.5	66.7	0.2		
	Fifth Avenue	Sixth Avenue	66.7	67.4	0.7		
	Sixth Avenue	Eighth Avenue	67.3	68.1	0.8		
	Eighth Avenue	Vermont Street	67.3	67.5	0.2		
	Vermont Street	Richmond Street	67.2	67.5	0.3		
	Richmond Street	Park Boulevard	65.5	66.7	1.2		
Upas Street	Third Avenue	Sixth Avenue	59.9	62.7	2.8		
Washington Street	India Street	University Avenue	73.0	74.0	1.0		
	University Avenue	First Avenue	69.1	70.1	1.0		
	First Avenue	Fourth Avenue	71.1	69.9	-1.2		
	Fourth Avenue	Fifth Avenue	70.9	71.7	0.8		
	Fifth Avenue	Sixth Avenue	71.8	72.1	0.3		
	Sixth Avenue	Richmond Street	72.2	72.1	-0.1		
	Richmond Street	Normal Street	71.9	72.7	0.8		
Freeways		1	1	I	1		
I-5	Old Town Avenue	Washington Street	85.3	87.0	1.7		
	Washington Street	Sassafras Street	84.0	85.8	1.8		
	Sassafras Street	Pacific Highway	84.2	85.9	1.7		
	Pacific Highway	India Street	85.2	87.0	1.8		
	India Street	Hawthorn Street	85.4	87.1	1.7		
	Hawthorn Street	First Avenue	84.6	86.4	1.8		
	First Avenue	Sixth Avenue	85.6	87.4	1.8		
	Sixth Avenue	SR-163	85.8	88.1	2.3		
	SR-163	Pershing Drive	85.4	87.7	2.3		
	Pershing Drive	SR-94	85.8	87.6	1.8		
	SR-94	Imperial Avenue	84.7	86.5	1.8		
I-8	Hotel Circle (W)	Hotel Circle (E)	85.3	86.2	0.9		
	Hotel Circle (E)	SR-163	85.5	86.1	0.6		
	SR-163	Mission Center Road	85.6	86.1	0.5		
	Mission Center Road	Qualcomm Way	85.9	86.9	1.0		
	Qualcomm Way	I-805	85.4	86.2	0.8		
	I-805	I-15	86.2	86.8	0.6		
SR-163	I-8	Sixth Avenue	84.6	85.6	1.0		
	Sixth Avenue	Washington Street	83.6	84.7	1.1		
	Washington Street	Robinson Avenue	82.4	83.0	0.6		
	Robinson Avenue	Richmond Street	82.8	83.3	0.5		
	Richmond Street	Quince Street	82.9	83.4	0.5		
	Quince Street	1-5	83.0	83.5	0.5		
Bold = 2035 noise level	would exceed the estab	lished exterior compatibil	ity level for the	surrounding la	nd use		
and noise levels would	increase by 3 dB or mor	e or future noise levels wo	uld be below 6	5 dB(A) CNEL b	ut		
ambient noise levels wo	ould increase by more th	nan 5 dB(A) over existing a	mbient noise le	vels.			

SOURCE: RECON 2016 (see Appendix F).

As shown in Table 6.6-2, the following roadway segment currently generates noise levels greater than 65 dB(A) CNEL, and future noise levels would increase by more than 3 dB(A):

• Sixth Avenue from Washington Street to University Avenue

The following street segments in the Uptown CPU currently generate noise levels lower than 65 dB(A) CNEL and would generate future noise levels lower than 65 dB(A) CNEL, but future noise levels would increase by more than 5 dB(A) over existing ambient noise levels:

- Grape Street from Albatross Street to First Avenue
- Grape Street from Third Avenue to Sixth Avenue

a. Existing Noise Sensitive Land Uses

There are existing noise sensitive uses located adjacent to these streets segments and there could be additional future sensitive uses located adjacent to the street segments under the proposed Uptown CPU. The increase in ambient noise levels adjacent to these segments of Sixth Street and Grape Street would result in the exposure of existing sensitive receptors to a significant increase in ambient noise levels, and impacts would be significant. Possible noise-reduction measures would include retrofitting older residential structures with new window and door components with higher Sound Transmission Class (STC) ratings, which is a measure of how well a building wall, windows, and door components attenuate exterior noise.

The Quieter Home Program administered by the San Diego County Regional Airport Authority is intended to attenuate interior noise levels of existing buildings from aircraft noise. Attenuation would also reduce interior noise levels from exterior motor vehicle noise. Some of the existing residences in the Uptown CPU area have already participated in this program and have undergone retrofits to reduce interior noise levels to 45 dB(A) CNEL. However, for existing structures that have not participated in or are not eligible for the Quieter Home Program, it cannot be determined at the program-level whether the existing structures contain adequate attenuation to reduce interior noise to the 45 dB(A) CNEL standard nor what measures would be required to retrofit these structures. Because the significant noise impacts would occur to existing residential structures in an already urbanized area, there is no feasible mitigation at the program-level. Thus, ambient noise increases affecting existing residential structures or other structures with sensitive land uses would be significant and unavoidable.

- **Impact 6.6-1** The increase in ambient noise levels as a result of build-out of the proposed Uptown CPU and associated discretionary actions along the road segments listed below would result in the exposure of existing sensitive receptors to a significant increase in future noise levels, resulting in a significant impact:
 - Sixth Avenue from Washington Street to University Avenue
 - Grape Street from Albatross Street to First Avenue
 - Grape Street from Third Avenue to Sixth Avenue

b. Future Noise Sensitive Land Uses

An existing regulatory mitigation framework and review process exists for new development in areas exposed to high levels of ambient noise. Policies in the proposed Uptown CPU and General Plan related to decibel levels, procedures in the Municipal Code, and regulations (Title 24) would reduce traffic noise exposure, because they set standards for the siting of sensitive land uses. Sitespecific noise analyses that demonstrate that the project would not place sensitive receptors in locations where the exterior existing or future noise levels would exceed the noise compatibility guidelines of the City's General Plan would be required as part of the review process for discretionary projects, to the extent practicable. With implementation of these regulations and proceduresthis framework, noise impacts to new discretionary projects would be less than significant. However, in the case of ministerial projects, there is no procedure to ensure that exterior noise is adequately attenuated. Therefore, exterior noise impacts for ministerial projects located in areas that exceed the applicable land use and noise compatibility level would be significant and unavoidable. Interior noise impacts for all projects including ministerial projects would be less than significant because applicants must demonstrate compliance with the current interior noise standards (45 dB(A) CNEL) through submission and approval of a Title 24 Compliance Report.

The proposed Uptown CPU and associated discretionary actions would result in future noise levels greater than 65 dB(A) CNEL along Sixth Avenue from Washington Street to University Avenue and future noise levels would increase by more than 5 dB(A) over existing ambient noise levels on segments of Grape Street. While future discretionary projects have a framework in place that would ensure exterior noise levels are appropriately attenuated to meet the General Plan Compatibility Standards, there is no similar mechanism in place for ministerial projects, resulting in a significant impact.

- **Impact 6.6-2** The increase in ambient noise levels as a result of build-out of the proposed Uptown CPU and associated discretionary actions along the road segments listed below would result in the exposure of projects that only require approval of a ministerial permit to a significant increase in future noise levels, resulting in a significant impact:
 - Sixth Avenue from Washington Street to University Avenue
 - Grape Street from Albatross Street to First Avenue
 - Grape Street from Third Avenue to Sixth Avenue

For all other street segments in the Uptown CPU area, the increase in ambient noise would be less than significant.

Issue 2 Vehicular Noise

Would the proposed project cause exposure of people to current or future transportation noise levels which exceed standards established in the Noise Element of the General Plan?

a. Freeway and Roadway Noise

A significant impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions would result in an exposure of people to current or future motor vehicle traffic noise levels that exceed standards established in the Noise Element of the General Plan. The General Plan noise and land use compatibility guidelines are presented in Chapter 5.0, Regulatory Framework, Table 5-3. The proposed Uptown CPU and associated discretionary actions propose single-family residential, multi-family residential, commercial, institutional, visitor accommodations, and park and open space land uses, which are compatible with the following noise levels.

- Single-family residential is compatible up to 60 dB(A) CNEL and conditionally compatible up to 65 dB(A) CNEL.
- Multi-family residential and mixed uses are compatible up to 60 CNEL and conditionally compatible up to 70 CNEL.
- Additionally, as stated in Section B of the City's Noise Element, although not generally considered compatible, the City conditionally allows multi-family and mixed-use residential uses up to 75 dB(A) CNEL in areas affected by motor vehicle traffic noise with existing residential uses. Any future residential use exposed to noise levels up to 75 dB(A) CNEL must include attenuation measures to ensure an interior noise level of 45 dB(A) CNEL and be located in an area where a community plan allows multi-family and mixed-use residential uses.
- Sales, commercial services, and office uses are compatible up to 65 dB(A) CNEL and conditionally compatible up to 75 dB(A) CNEL.
- Institutional uses are compatible up to 60 dB(A) CNEL and conditionally compatible up to 65 dB(A) CNEL
- Visitor accommodations (hotel) uses are compatible up to 60 dB(A) CNEL and conditionally compatible up to 75 dB(A) CNEL.
- Neighborhood parks are compatible up to 70 dB(A) CNEL and conditionally compatible up to 75 dB(A) CNEL.

The vehicle traffic from adjacent freeways are the dominant vehicle noise sources affecting the Uptown CPU area. The freeways and streets generating the greatest noise level in the Uptown CPU are I-5, I-8, SR-163, Sixth Avenue, India Street, Park Boulevard, Robinson Avenue, University Avenue, and Washington Street. The distances to the 60 dB, 65 dB, 70 dB, and 75 dB CNEL noise contours in the build-out condition for freeways and major roadways in the Uptown planning area are shown in Table 6.6-3. Distances to the roadway noise contours are based on a hard, flat site with no

intervening barriers or obstructions (worst-case analysis). Future horizon year noise contours for the proposed Uptown CPU area are shown in Figure 6.6-3.

Table 6.6-3							
	Future Vehicle Traffic	Contour Distances for t	he Uptown C	PU Area			
			Distance To (feet)				
			75	70	65	60	
Roadway	From	То	CNEL	CNEL	CNEL	CNEL	
Uptown Planning Ar	ea	•					
First Avenue	Arbor Drive	Washington Street	4	12	37	117	
	Washington Street	University Avenue	4	14	45	141	
	University Avenue	Robinson Avenue	8	26	81	256	
	Robinson Avenue	Pennsylvania Avenue	6	18	57	182	
	Pennsylvania Avenue	Walnut Avenue	6	20	63	199	
	Walnut Avenue	Laurel Street	6	19	59	186	
	Laurel Street	Hawthorn Street	4	13	42	132	
	Hawthorn Street	Grape Street	3	11	34	107	
	Grape Street	Elm Street	2	7	22	71	
Fourth Avenue	Arbor Drive	Washington Street	7	23	74	234	
	Washington Street	University Avenue	5	15	49	155	
	University Avenue	Robinson Avenue	6	20	64	204	
	Robinson Avenue	Walnut Avenue	6	18	56	177	
	Walnut Avenue	Laurel Street	7	23	74	234	
	Laurel Street	Grape Street	7	21	67	213	
	Grape Street	Elm Street	5	15	48	151	
Fifth Avenue	Washington Street	University Avenue	6	19	59	186	
	University Avenue	Robinson Avenue	7	22	69	218	
	Robinson Avenue	Walnut Avenue	8	24	77	245	
	Walnut Avenue	Laurel Street	7	23	72	229	
	Laurel Street	Hawthorn Street	7	22	71	223	
	Hawthorn Street	Grape Street	7	22	71	223	
	Grape Street	Elm Street	5	16	50	158	
Sixth Avenue	Washington Street	University Avenue	22	71	223	706	
	University Avenue	Robinson Avenue	16	51	162	512	
	Robinson Avenue	Upas Street	15	47	148	467	
	Upas Street	Laurel Street	13	41	129	406	
	Laurel Street	Juniper Street	8	26	81	256	
	Juniper Street	Grape Street	9	29	93	294	
	Grape Street	Elm Street	10	32	100	315	
Ninth Avenue	Washington Street	University Avenue	3	9	27	87	
Campus Avenue	Madison Avenue	Washington Avenue	2	6	20	63	
	Washington Avenue	Polk Avenue	3	8	26	81	
Cleveland Avenue	Tyler Street	Lincoln Street	3	8	25	79	
	Lincoln Street	Richmond Street	3	10	33	104	
Curlew Street	Robinson Avenue	Reynard Way	2	5	16	50	
Elm Street	Second Avenue	Third Avenue	3	9	29	93	
	Third Avenue	Fifth Avenue	3	10	32	100	
	Fifth Avenue	Sixth Avenue	3	9	28	89	
Fort Stockton Drive	Arista Street	Sunset Boulevard	2	5	17	54	
	Sunset Boulevard	Hawk Street	3	9	27	87	
	Hawk Street	Goldfinch Street	3	10	31	97	
	Goldfinch Street	Falcon Street	1	4	11	36	

Table 6.6-3							
	Future Vehicle Traffic	Contour Distances for t	he Uptown Cl	PU Area			
				Dis	stance To (feet)	
	_	_	75	70	65	60	
Roadway	From	10	CNEL	CNEL	CNEL	CNEL	
Front Street	Dickinson Street	Arbor Drive	2	5	16	50	
	Arbor Drive	Washington Street	3	9	27	8/	
Grape Street	Albatross Street	First Avenue	3	8	25	79	
	First Avenue	Third Avenue	3	8	25	/9	
	Third Avenue	Sixth Avenue	3	10	31	97	
Hawthorn Street	Brant Street	First Avenue		23	/4	234	
	First Avenue	Third Avenue	4	11	36	115	
	Third Avenue	Sixth Avenue	4	13	43	135	
India Street	Washington Street	Winder Street	/	22	69	218	
	Winder Street	Glenwood Drive	/	21	6/	213	
	Glenwood Drive	Sassafras Street	19	60	190	601	
	Sassafras Street	Redwood Street	13	43	135	426	
	Redwood Street	Palm Street	13	41	129	406	
Juan Street	Harney Street	Witherby Street	2	/	23	/2	
Laurel Street	Columbia Street	Union Street	/	23	/2	229	
	Union Street	First Avenue	6	19	62	195	
	First Avenue	Third Avenue	6	18	56	1//	
	Third Avenue	Sixth Avenue	/	22	69	218	
Lewis Street	Fort Stockton Drive	Goldfinch Street	1	4	14	45	
Lincoln Avenue	Washington Street	Park Boulevard	4	12	38	120	
Madison Avenue	Cleveland Avenue	Park Boulevard	2	7	21	66	
Meade Avenue	Cleveland Avenue	Park Boulevard	1	4	12	38	
Normal Street	Park Boulevard	Washington Street	14	45	141	446	
	Washington Street	University Avenue	2	8	24	76	
Park Boulevard	Adams Avenue	Mission Avenue	5	16	51	162	
	Mission Avenue	El Cajon Boulevard	8	26	81	256	
	El Cajon Boulevard	Polk Avenue	9	29	91	288	
	Polk Avenue	University Avenue	14	45	141	446	
		Robinson Avenue	13	40	126	397	
	Robinson Avenue	Upas Street	11	35	109	346	
D 114/	Upas Street	Zoo Place	11	35	112	354	
Reynard Way	Torrance Street	Curlew Street	3	8	26	83	
	Curiew Street	Laurei Street	4	13	43	135	
Richmond Street	Cleveland Avenue	University Avenue	3	10	31	97	
	University Avenue	Robinson Avenue	2	/	23	/4	
	Robinson Avenue	Upas Street	3	9	28	89	
Robinson Avenue	Brant Street	First Avenue	2	5	16	50	
	First Avenue	Inird Avenue	4	13	40	126	
	Inird Avenue	Eighth Avenue	5	16	50	158	
	Eighth Avenue	Tenth Avenue	4	13	43	135	
	Pietro and Stread	Richmond Street	6	19	60	190	
	Richmond Street	Park Boulevard	5	14	46	144	
San Diego Avenue	Hortensia Street	Weekington Street			66 14 F	208	
	Weekingstreet	wasnington Street		30	115	362	
Chaba Chuc at	washington Street		4	14	45	141	
State Street	Laurei Street	Juniper Street	3	9	28	89	
Sunset Boulevard	witherby Street	Fort Stockton Drive	2	5	16	50	

6.6	Noise

Table 6.6-3							
	Future Vehicle Traffic	Contour Distances for t	he Uptown C	PU Area			
				Dis	stance To (feet)	
			75	70	65	60	
Roadway	From	То	CNEL	CNEL	CNEL	CNEL	
University Avenue	lbis Street	Albatross Street	5	16	51	162	
	Albatross Street	First Avenue	7	23	72	229	
	First Avenue	Fourth Avenue	5	15	49	155	
	Fourth Avenue	Fifth Avenue	7	23	74	234	
	Fifth Avenue	Sixth Avenue	9	27	87	275	
	Sixth Avenue	Eighth Avenue	10	32	102	323	
	Eighth Avenue	Vermont Street	9	28	89	281	
	Vermont Street	Richmond Street	9	28	89	281	
	Richmond Street	Park Boulevard	7	23	74	234	
Upas Street	Third Avenue	Sixth Avenue	3	9	29	93	
Washington Street	India Street	University Avenue	40	126	397	1,256	
_	University Avenue	First Avenue	16	51	162	512	
	First Avenue	Fourth Avenue	15	49	155	489	
	Fourth Avenue	Fifth Avenue	23	74	234	740	
	Fifth Avenue	Sixth Avenue	26	81	256	811	
	Sixth Avenue	Richmond Street	26	81	256	811	
Richmond Street		Normal Street	29	93	294	931	
Freeways	L			1	I	1	
1-5	Old Town Avenue	Washington Street	315	680	1,464	3,155	
	Washington Street	Sassafras Street	262	565	1,218	2,624	
	Sassafras Street	Pacific Highway	266	574	1,237	2,665	
	Pacific Highway	India Street	315	680	1,464	3,155	
	India Street	Hawthorn Street	320	690	1,487	3,204	
	Hawthorn Street	First Avenue	288	620	1,335	2,877	
	First Avenue	Sixth Avenue	335	723	1.557	3,355	
	Sixth Avenue	SR-163	374	805	1.734	, 3,735	
	SR-163	Pershing Drive	351	757	1,630	3,513	
	Pershing Drive	SR-94	346	745	1.606	3,459	
	SR-94	Imperial Avenue	292	629	1.356	2,922	
1-8	Hotel Circle (W)	Hotel Circle (E)	279	601	1.295	2.790	
	Hotel Circle (F)	SR-163	275	592	1.275	2.748	
	SR-163	Mission Center Road	275	592	1.275	2,748	
	Mission Center Road	Qualcomm Way	311	669	1 442	3 107	
	Qualcomm Way	I-805	279	601	1 295	2 790	
	1-805	I-15	306	659	1,235	3 059	
SR-163	1.8	Sixth Avenue	254	5/18	1 1 2 1	2 5/15	
51(-105	Sixth Avenue	Washington Street	234	177	1,101	2,345	
	Washington Street	Rohinson Avenue	171	368	7029	1 707	
		Pichmond Stroot	170	200	820	1 700	
	Richmond Streat	Ouince Street	1/9	201	020 Q12	1,700 1 915	
	Ouince Street		102	207	045 856	1 9/2	
	Quince Sileel	-J	104	531	020	1,040	

ATTACHMENT 7



FIGURE 6.6-3 Future (2035) Traffic Noise Contours for the Uptown CPU Area

Map Source: SanGIS

At any specific location the actual existing noise would depend upon not only the source noise level, but also the nature of the path from the source to the sensitive receptor. Buildings, walls, dense vegetation, and other barriers would block the direct line of sight and reduce noise levels at the receptor. As an example, a first row of buildings would reduce traffic noise levels at receptors by 3 to 5 dB(A) behind those structures depending on the building-to-gap ratio. Large continuous structures can provide substantially greater attenuation of traffic noise.

While the General Plan Noise Element has a compatibility level of 60 dB(A) CNEL or less for residential uses, noise levels up to 65 dB(A) CNEL for single-family residential and up to 70 dB(A) CNEL for multi-family residential are considered conditionally compatible, since interior noise levels can be reduced to 45 dB(A) CNEL through simple means, such as closing/sealing windows and providing mechanical ventilation. Additionally, as stated in Section B of the General Plan Noise Element, although not generally considered compatible, the General Plan conditionally allows multifamily and mixed-use residential uses up to 75 dB(A) CNEL in areas affected by motor vehicle traffic noise with existing residential uses. Any future residential use exposed to noise levels up to 75 dB(A) CNEL must include attenuation measures to ensure an interior noise level of 45 dB(A) CNEL and be located in an area where a community plan allows multi-family and mixed-use residential uses. Passive mitigation such as noise walls adjacent to freeways and roadways can usually reduce exterior noise levels to comply with General Plan Noise Element guidelines. The majority of CPU residential land uses planned for the Uptown community would be located within the conditionally compatible noise level range. Multi-family residential uses located where exterior noise levels range from 65 to 70 dB(A) CNEL are considered conditionally compatible and can generally provide the required structural attenuation to reduce noise levels at interior locations. Multi-family and mixeduse residential uses that meet the requirements of Section B of the General Plan Noise Element would be conditionally compatible up to 75 dB(A) CNEL and would also be required to provide structural attenuation to reduce noise levels at interior locations.

Additionally, due to the provision of common exterior use areas, multi-family residential land uses can generally provide greater shielding to these areas, thus providing exterior use areas that comply with the General Plan Noise Element guidelines. Likewise, backyards of single-family residential uses can be shielded from roadway noise by the residential structure, providing exterior use areas that are compatible with the General Plan Noise Element guidelines.

As shown in Figure 6.6-3, future traffic noise levels with build-out of the proposed Uptown CPU at existing and proposed residential use areas closest to the freeways and heavily traveled roadways would exceed the General Plan Noise Element conditionally compatible thresholds for residential land uses (65 dB(A) CNEL for single-family and conditionally up to 75 dB(A) CNEL for multi-family and mixed-use developments that meet the requirements of Section B of the Noise Element). Noise levels greater than 75 dB(A) CNEL are considered incompatible for all land use types. Uses located adjacent to I-5 and SR-163 in the Uptown CPU area have the potential to be exposed to noise levels greater than 75 dB(A) CNEL. However, the proposed Uptown CPU and associated discretionary actions would not locate new sensitive land uses in areas that are exposed to 75 dB(A) CNEL or greater.

In the Uptown CPU area, noise levels for all land uses would be incompatible (i.e., greater than 75 dB(A) CNEL) at areas located approximately 262 to 374 feet from I-5 and 171 to 254 feet from SR-

163. Noise levels for sensitive land uses would be incompatible (i.e., greater than 70 dB(A) CNEL) at areas located approximately 565 to 805 feet from I-5 and 368 to 548 feet from SR-163 (see Figure 6.6-3). These areas are currently developed, and the proposed Uptown CPU and associated discretionary actions would not change the land use in these areas or introduce new sensitive land uses in these areas. Thus, while land uses in these areas would be exposed to noise levels that exceed General Plan guidelines, this noise exposure would not be a significant noise impact resulting from implementation of the proposed Uptown CPU and associated discretionary actions. Additionally, per Section B of the General Plan Noise Element, any future multi-family and mixed-use residential use exposed to noise levels up to 75 dB(A) CNEL must include attenuation measures to ensure an interior noise level of 45 dB(A) CNEL and be located in an area where a community plan allows multi-family and mixed-use residential uses.

Furthermore, policies in the proposed Uptown CPU and General Plan and California Code of Regulations (CCR) Title 24 would reduce traffic noise exposure, because they set standards for the siting of sensitive land uses. General Plan policy NE-A.4 requires an acoustical study consistent with Acoustical Study Guidelines (Table NE-4) for proposed developments in areas where the existing or future noise level exceeds or would exceed the "compatible" noise level thresholds as indicated on the Land Use – Noise Compatibility Guidelines. Site-specific exterior noise analyses that demonstrate that the project would not place sensitive receptors in locations where the exterior existing or future noise levels would exceed the noise compatibility guidelines of the General Plan would be required as part of future discretionary proposals. Additionally, site-specific interior noise analyses demonstrating compliance with the interior noise compatibility guidelines of the General Plan would be required for land uses located in areas where exterior noise levels exceed the noise and land use compatibility thresholds as defined in the General Plan Noise Element, Table N-3. This requirement is implemented through submission of a Title 24 Compliance Report to demonstrate interior noise levels of 45 dB(A) CNEL. With this framework, exterior traffic noise impacts associated with new development requiring discretionary approvals and interior traffic noise impacts for both ministerial and discretionary projects would be less than significant.

However, in the case of exterior noise impacts associated with ministerial projects, there is no procedure to ensure that exterior noise is adequately attenuated. Therefore, exterior noise impacts for ministerial projects located in areas that exceed the applicable land use and noise compatibility level would be significant and unavoidable.

Impact 6.6-3: A significant impact would occur for ministerial projects exposed to vehicular traffic noise levels in excess of the compatibility levels established in the General Plan Noise Element, based on future (2035) noise contours as shown on Figure 6.6-3 of this PEIR.

b. Rail Noise

Railway noise results from light rail and heavy rail vehicle operations, horns, emergency signaling devices, and stationary bells at grade crossings. The rail corridor generally parallels I-5 at the western boundary of the Uptown planning area in the Midway Pacific Highway Community Plan Area. The San Diego Metropolitan Transit System provides trolley service along the rail corridor. Amtrak operates passenger trains and the Coaster operates commuter trains along the rail corridor

daily. The Burlington Northern Santa Fe Railway Company also operates freight trains along the corridor daily.

Trolley and train noise consists of noise from the trolleys, trains, and emergency signaling devices. Trolley and train vehicles are equipped with horns for use in emergency situations and as a general audible warning to track workers and trespassers within the right-of-way as well as to pedestrians and motor vehicles at road grade crossings. Horns on the moving trolley and train vehicles combined with stationary bells at grade crossings can generate excessive noise levels that can affect noise sensitive receptors.

Noise associated with trolley, Amtrak, Coaster, and freight train operations was modeled using the FTA recommended Chicago Rail Efficiency and Transportation Efficiency railroad noise model. Modeling results are shown in Table 6.6-4. The trolleys were modeled at 25 mph. This is based on the distances between trolley stations and the average timing between stations obtained from published trolley schedules. Noise contour distances were calculated assuming flat-site conditions and no intervening buildings that would provide noise attenuation, which would represent a conservative, worst-case analysis.

The number of Amtrak and Coaster trains operating along the corridor was obtained from existing published schedules. There are approximately 25 Amtrak trains and 22 Coaster trains that travel on the tracks west of the Uptown CPU area daily. Amtrak trains have an average of eight cars per train and travel at a speed of 30 mph. This is based on the distances between stations and the average timing between stations obtained from published schedules. The number of cars and speed of the Coaster were assumed to be the same as the Amtrak train.

There are approximately five freight trains that typically operate through the corridor each day (FHWA 2002). Freight trains have an average of 50 cars per train. Freight trains were modeled at the same speed as Amtrak and Coaster trains. As a conservative analysis, it was assumed that all freight train operations would occur during the nighttime hours.

Table 6.6-4 Railway Noise Levels					
Noise Level at Nearest Uptown					
Source	CPU Boundary				
Trolley	47				
Commuter Train (Amtrak and Coaster)	54				
Freight Train	56				
All Rail Sources Combined	59				
CPU = community plan update					

The western Uptown CPU area boundary is located more than 400 feet from the rail corridor. As shown in Table 6.6-4, noise levels at the nearest Uptown CPU area boundary 400 feet from the rail corridor would not exceed 60 dB(A) CNEL. The nearest sensitive land uses are located on the northeast side of India Street, and buildings with commercial uses are located between the residential uses and the rail corridor and I-5. Trolley and train noise levels at these uses would, therefore, be less than the General Plan Noise Element, Land Use Compatibility Guidelines. Trolley

and train noise would not result in a significant increase in noise over motor vehicle traffic noise from I-5. Therefore, noise level impacts resulting from trolley and train operations would be less than significant.

Issue 3 Airport Compatibility

Would the proposed project result in land uses which are not compatible with aircraft noise levels as defined by an adopted Airport Land Use Compatibility Plan (ALUCP)?

The SDIA is located west of the Uptown CPU area. A majority of aircraft flying over the CPU area are approaching SDIA, with the occasional departure. Aircraft noise is evaluated based on the noise contours developed by the San Diego County Regional Airport Authority and provided in the Airport Land Use Compatibility Plan for San Diego International Airport (2014). The aircraft noise contours are based on year 2030 forecast noise exposure.

A significant impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions would result in land uses that are not compatible with aircraft noise levels as defined by an adopted ALUCP. The SDIA is located approximately one mile west of the Uptown CPU area. As shown in Figure 6.6-4, the 60 to 75 dB CNEL contours for the SDIA extend into the Uptown CPU area. Residential uses located generally in the southwestern portion of the Uptown CPU area have the potential to be exposed to aircraft noise levels exceeding 60 dB CNEL. However, the proposed Uptown CPU and associated discretionary actions would not change the land use designations of the existing residential land uses located within the 65 dB and above CNEL contours for the SDIA. The ALUCP conditionally allows future residential. Future residential buildings would include noise attenuation consistent with the Noise Element of the General Plan and the ALUCP for the SDIA.

The San Diego County Regional Airport Authority has an Airport Noise Mitigation Office and has implemented a number of programs to reduce the aircraft noise impact on the community. Actions include the enforcement of a curfew on departing aircraft and the Quieter Home Program. The Quieter Home Program provides sound insulation retrofits for residences located within the 65 dB(A) CNEL contour with the goal of reducing interior noise levels by at least 5 dB(A). Existing residences located within the 65 dB(A) CNEL contour for the SDIA in the Uptown CPU area are eligible for this program (Note that eligibility to participate in the program is based on the noise exposure maps prepared under 14 Code of Federal Regulations Part 150, which are different than the ALUCP contour maps). Figure 6.6-5 shows a map of the parcels that have participated in the program as of January 2015.

Per the City Significance Determination Thresholds, if a future project implemented under the proposed Uptown CPU and associated discretionary actions is proposed within the 60 dB(A) CNEL and greater contour (as shown in the ALUCP for SDIA), the potential exterior noise impacts from aircraft noise would not constitute a significant environmental impact. However, interior noise



M:\JOBS4\6086\common_gis\fig6.6-4_EIR_UT.mxd 6/7/2016 fmm

ATTACHMENT 7

Uptown Community Plan Boundary

Airport Noise Contours

60-65 dB CNEL
65-70 dB CNEL
70-75 dB CNEL
>75 dB CNEL



FIGURE 6.6-4

Airport Noise Contours

ATTACHMENT 7





Quieter Home Program Participation for the Uptown CPU Area

impacts would be regulated by the requirement for residential development within the 60 dB(A) CNEL and greater (as shown in the ALUCP for SDIA) to reduce interior noise levels attributable to airport noise to 45 dB(A) CNEL. The City currently submits both discretionary and ministerial projects that increase residential units and non-residential floor area and change in use to the Airport Land Use Commission for a consistency determination with the ALUCP. Interior noise levels for new construction are also addressed through implementation General Plan policies NE-I.1 and NE-I.2, and Title 24 of the California Code of Regulations which requires submission of a Title 24 Compliance Report to demonstrate interior noise levels of 45 dB(A) CNEL). With this framework, airport noise impacts to new development would be less than significant.

The proposed Uptown CPU and associated discretionary actions would not result in land use compatibility impacts related to airports because the CPU would not result in a change to these existing uses or a change in SDIA operations. Because future development is required to provide noise attenuation consistent with the Noise Element of the General Plan and the ALUCP for the San Diego International Airport and follow procedures in the Municipal Code, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant exposure to noise from aircraft.

Issue 4 Noise Ordinance Compliance

Would the proposed project result in the exposure of people to noise levels which exceed property line limits established in the Noise Abatement and Control Ordinance of the Municipal Code?

A significant impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions results in the exposure of people to noise levels that exceed property line limits established in the Noise Abatement and Control Ordinance of the Municipal Code. Stationary sources of noise include activities associated with a given land use. For example, noise sources in commercial uses would include car washes, fast food restaurants, auto repair facilities, parking lots, and a variety of other uses. Due to the number of eating and drinking establishments in the CPU area, the Uptown experiences elevated noise levels associated with these uses.

Mixed-use areas would contain residential and commercial interfaces. Mixed-use and areas where residential uses are located in proximity to commercial sites would result in an exposure of sensitive receptors to noise. The interface between commercial and residential uses would be exposed to noise due to traffic, loading docks, mechanical equipment [such as generators and heating, ventilation, and air conditioning (HVAC) units], deliveries, trash-hauling activities, and customer and employee use of commercial facilities. Limiting truck idling time and enclosing external equipment (generators, HVAC units, etc.) that are adjacent to residential uses would reduce stationary noise levels.

While noise-sensitive residential land uses would be exposed to noise associated with the operation of commercial uses, policies are in place to control noise and reduce noise impacts between various land uses. Noise policies, as contained in the General Plan Noise Element, the proposed Uptown CPU, and regulations in the Noise Ordinance are in place to control noise and reduce noise impacts between various land uses. These include the requirement for noise studies, limits on hours of operation for various noise-generating activities, and standards for the compatibility of various land

uses with the existing and future noise environment. In addition, enforcement of the federal, state, and local noise regulations would control impacts. Given implementation of these policies and enforcement of the Noise Abatement and Control Ordinance of the Municipal Code, impacts would be less than significant.

Issue 5 Temporary Construction Noise

Would the proposed project result in the exposure of people to significant temporary construction noise?

a. Construction Noise

A significant impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions resulted in the exposure of people to significant temporary construction noise. Future development as allowed under the Uptown CPU and associated discretionary actions could potentially result in temporary ambient noise increase due to construction activities.

Although no specific construction or development is proposed under the proposed Uptown CPU and associated discretionary actions at this time, construction noise impacts could occur as future development occurs. Due to the developed nature of the Uptown CPU area, there is a high likelihood that construction activities would take place adjacent to existing structures and that sensitive receptors would be located in proximity to construction activities.

Construction noise typically occurs intermittently and varies depending upon the nature or phase of construction (e.g., demolition/land clearing, grading and excavation, erection). Construction noise in any one particular area would be short-term and would include noise from activities such as site preparation, truck hauling of material, pouring of concrete, and use of power tools. Noise would also be generated by construction equipment, including earthmovers, material handlers, and portable generators, and could reach high levels for brief periods. Typical construction noise levels are discussed in Appendix F.

Construction equipment would generate maximum noise levels between 85 and 90 dB at 50 feet from the source when in operation. Hourly average noise levels would be 82 dB(A) at 50 feet from the center of construction activity when assessing the loudest pieces of equipment working simultaneously. Noise levels would vary depending on the nature of the construction including the duration of specific activities, nature of the equipment involved, location of the particular receiver, and nature of intervening barriers. Construction noise levels of 82 dB(A) L_{eq} at 50 feet would attenuate to 75 dB(A) L_{eq} at 110 feet. Therefore, significant impacts would occur if sensitive land uses are located closer than 110 feet of construction activities.

Impact 6.6-4: A significant noise impact due to construction noise would occur is sensitive land uses are located within 110 feet of future construction activities.

The City regulates noise associated with construction equipment and activities through its Noise Abatement and Control Ordinance which puts limits on the days of the week and hours of operation allowed for construction. The City also imposes conditions of approval for building and grading permits related to noise. However, there is also a procedure in place that allows for a permit to deviate from the noise ordinance. Due to the highly developed nature of the Uptown CPU area with sensitive receivers potentially located in proximity to construction sites, there is a potential for construction of future projects to expose existing residences to significant noise levels (see Impact 6.6-4).

b. Vibration - Construction

Construction of projects implemented under a permit to deviate from the noise ordinance would likely be located adjacent to existing structures. Construction activities may include demolition of existing structures, site preparation work, excavation of parking and subfloors, foundation work, and building construction. Demolition for an individual site may last several weeks to months and may produce substantial vibration. Excavation for underground levels could also occur on some project sites, and vibratory pile driving could be used to stabilize the walls of excavated areas. Piles or drilled caissons may also be used to support building foundations.

As with any type of construction, vibration levels during any phase may at times be perceptible. However, non-pile driving or foundation work construction phases that have the highest potential of producing vibration (such as jackhammering and other high power tools) would be intermittent and would only occur for short periods of time for any individual project site. By use of administrative controls, such as scheduling construction activities with the highest potential to produce perceptible vibration to hours with least potential to affect nearby properties, perceptible vibration can be kept to a minimum and as such would result in a less than significant impact with respect to perception.

Pile driving has the potential to generate the highest groundborne vibration levels and is the primary concern for structural damage when it occurs within 95 feet of structures. Past studies have established a peak vertical particle velocity of 0.20 inch per second as the limit where vibration would begin to annoy people in buildings and at which there is a risk of cosmetic damage to normal dwellings. Maximum vibration levels from pile driving would exceed this level at approximately 95 feet. Vibration levels generated by pile-driving activities would vary depending on project conditions, such as soil conditions, construction methods, and equipment used. Pile-driving activities generate vibrations at various frequencies, with the dominant frequency of propagating waves from impact sources ranging between 3 and 60 Hz. Using the middle range for illustration purposes, equipment operating at a frequency range of 30 Hz would exceed the perceptible range at approximately 100 feet. Pile driving within 95 feet of existing structures has the potential to exceed the 0.20 inch per second PPV threshold. Thus, implementation of future land uses under the proposed Uptown CPU and associated discretionary actions would have the potential to result in a significant impact related to construction related vibration.

Impact 6.6-5: If future pile driving occurs within 95 feet of existing structures, a potentially significant impact associated with vibration would result.

c. Vibration – Operation

Commercial operations, on occasion, utilize equipment or processes that have a potential to generate groundborne vibration. However, vibrations found to be excessive for human exposure that are the result of commercial machinery are generally addressed from an occupational health

and safety perspective. The residual vibrations are typically of such low amplitude that they quickly dissipate into the surrounding soil and are rarely perceivable at the surrounding land uses. Additionally, the commercial uses that may be constructed under the proposed Uptown CPU and associated discretionary actions would include uses such as retail, restaurants, and small offices that would not require heavy mechanical equipment that would generate groundborne vibration or heavy truck deliveries. Residential and civic uses do not typically generate vibration. Thus, operational vibration impacts associated with proposed Uptown CPU and associated discretionary actions would be less than significant.

Cumulative Impacts

The analysis provided above for each issue area is cumulative in nature because the analysis considers noise and vibration impacts associated with build-out of the entirety of the Uptown CPU area and the traffic assumptions used in the analysis includes cumulative traffic associated with build-out of neighboring communities (North Park and Golden Hill CPU areas). Noise impacts associated with build-out of neighboring CPUs would be localized in nature. For example, construction of restaurants or commercial uses in North Park or Golden Hill would not affect residences in Uptown with the exception of development that may occur at the boundary of the CPU areas. However, build-out of land uses within each CPU area would be subject to the same General Plan policies, noise ordinance requirements, and Title 24 standards discussed in this document. Thus, cumulative noise impacts would be less than significant.

6.6.5 Significance of Impacts

6.6.5.1 Ambient Noise

An increase in ambient vehicular traffic noise in the Uptown CPU area would result from continued build-out of the proposed Uptown CPU and associated discretionary actions and increases in traffic due to regional growth. A significant increase would occur adjacent to several street segments in the Uptown CPU area that contain existing noise sensitive land uses. The increase in ambient noise levels could result in the exposure of existing noise sensitive land uses to noise levels in excess of the compatibility levels established in the General Plan, and impacts would be significant (Impact 6.6-1).

For new discretionary development, there is an existing regulatory framework in place that would ensure future projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions would not be exposed to ambient noise levels in excess of the compatibility levels in the General Plan. Thus, noise impacts to new discretionary projects would be less than significant.

However, in the case of ministerial projects, there is no procedure to ensure that exterior noise would be adequately attenuated. Therefore, exterior noise impacts for ministerial projects located in areas that exceed the applicable land use and noise compatibility level would be significant and unavoidable (Impact 6.6-2).

6.6.5.2 Vehicular Noise

In the Uptown CPU area, noise levels for all land uses would be incompatible (i.e., greater than 75 dB(A) CNEL) closest to the freeways and specific segments of Sixth Avenue and Grape Street. These areas are currently developed and the proposed Uptown CPU and associated discretionary actions would not change the land use in these areas. Thus, while land uses in these areas would be exposed to noise levels that exceed General Plan standards, this noise exposure would not be a significant noise impact resulting from implementation of the proposed Uptown CPU and associated discretionary actions. No mitigation is required at the program-level.

A<u>n existing regulatory mitigation</u> framework <u>and review process would</u> exist for new discretionary development in areas exposed to high levels of vehicle traffic noise. Implementation of the policies in the proposed Uptown CPU and General Plan would preclude or reduce traffic noise impacts because they would be required to demonstrate that exterior and interior noise levels would be compatible with City standards. Noise compatibility impacts associated with future discretionary projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions would be less than significant with implementation of existing regulations and noise standards. However, in the case of ministerial projects, there is no procedure to ensure that exterior noise is adequately attenuated. Therefore, exterior noise impacts for ministerial projects located in areas that exceed the applicable land use and noise compatibility level would be significant and unavoidable (Impact 6.6-3).

Amtrak, Coaster, and freight train noise levels at the nearest planning area boundary and the nearest sensitive receptors would not exceed 60 dB(A) CNEL. Noise impacts due to trolley and train operations would be compatible with General Plan standards. Impacts would be less than significant, and no mitigation is required.

6.6.5.3 Airport Compatibility

Based on the projected airport noise contours for the SDIA, there are sensitive receptors in the Uptown CPU area that are located where noise levels due to aircraft operations exceed 60 dB(A) CNEL. At the project-level, future development must include noise attenuation consistent with the Noise Element of the General Plan and the Airport Land Use Compatibility Plan for the SDIA; therefore, impacts related to airport noise would be less than significant.

6.6.5.4 Noise Ordinance Compliance

Mixed-use areas would contain residential and commercial interfaces. Mixed-use sites and areas where residential uses are located in proximity to commercial sites would expose sensitive receptors to noise. Although noise-sensitive residential land uses would be exposed to noise associated with the operation of these commercial uses, City policies and regulations would control noise and reduce noise impacts between various land uses. In addition, enforcement of the federal, state, and local noise regulations would control impacts. With implementation of these policies and enforcement of the Noise Abatement and Control Ordinance of the Municipal Code, impacts would be less than significant and no mitigation is required at the program-level.

6.6.5.5 Temporary Construction Noise

a. Construction Noise

Construction activities related to implementation of the Uptown CPU and associated discretionary action would potentially generate short-term noise levels in excess of 75 dB(A) L_{eq} at adjacent properties. While the City regulates noise associated with construction equipment and activities through enforcement of noise ordinance standards (e.g., days of the week and hours of operation) and imposition of conditions of approval for building or grading permits, there is a procedure in place that allows for variance to the noise ordinance. Due to the highly developed nature of the CPU area with sensitive receivers potentially located in proximity to construction sites, there is a potential for construction of future projects to expose existing sensitive land use to significant noise levels. While future development projects would be required to incorporate feasible mitigation measures, due to the close proximity of sensitive receivers to potential construction sites, the program-level impact related to construction noise would remain_be potentially_significant and unavoidable (Impact 6.6-4).

b. Vibration – Construction

By use of administrative controls, such as scheduling construction activities with the highest potential to produce perceptible vibration to hours with least potential to affect nearby properties, perceptible vibration can be kept to a minimum and as such would result in a less than significant impact with respect to perception. However, pile driving within 95 feet of existing structures has the potential to exceed 0.20 inch per second, and would be potentially significant (Impact 6.6-5).

c. Vibration – Operation

Post-construction operational vibration impacts could occur as a result of commercial operations that are implemented in accordance with the proposed Uptown CPU and associated discretionary actions. The commercial uses that would be constructed under the proposed Uptown CPU and associated discretionary actions would include uses such as retail, restaurants, and small offices that would not require heavy mechanical equipment that would generate groundborne vibration or heavy truck deliveries. Residential and civic uses do not typically generate vibration. Thus, operational vibration impacts associated with the proposed Uptown CPU implementation and associated discretionary actions would be less than significant. No mitigation is required.

6.6.6 Mitigation Framework

Increases in ambient noise levels resulting in the exposure of existing noise sensitive land uses to noise levels in excess of the compatibility levels established in the General Plan Noise Element would be significant and unavoidable (Impact 6.6-1). No feasible mitigation has been identified at the program-level to reduce this impact to less than significant.

New noise sensitive land uses that require only a ministerial permit would be subject to significant and unavoidable exterior traffic noise impacts resulting from increases in ambient noise levels generated from build-out of the proposed Uptown CPU and associated discretionary actions (Impact 6.6-2). Additionally, significant and unavoidable impacts would occur for future ministerial projects exposed to vehicular traffic noise levels in excess of the compatibility levels established in the General Plan Noise Element, based on future (2035) noise contours (Impact 6.6-3). No feasible mitigation has been identified at the program-level to reduce these impacts to less than significant as there is no mechanism to require exterior noise analysis and attenuation for these ministerial projects.

In order to mitigate impacts related to construction noise (Impact 6.6-4), the following mitigation measure would be implemented.

- **NOISE 6.6-1:** At the project-level, future development projects will be required to incorporate feasible mitigation measures. Typically, noise can be reduced to comply with City standards when standard construction noise control measures are enforced at the project site and when the duration of the noise-generating construction period is limited to one construction season (typically one year) or less.
 - Construction activities shall be limited to the hours between 7:00 A.M. and 7:00 P.M. Construction is not allowed on legal holidays as specified in Section 21.04 of the San Diego Municipal Code, with exception of Columbus Day and Washington's Birthday, or on Sundays. (Consistent with Section 59.5.0404 of the San Diego Municipal Code).
 - Equip all internal combustion engine-driven equipment with intake and exhaust mufflers that are in good condition and appropriate for the equipment.
 - Locate stationary noise-generating equipment (e.g., compressors) as far as possible from adjacent residential receivers.
 - Acoustically shield stationary equipment located near residential receivers with temporary noise barriers.
 - Utilize "quiet" air compressors and other stationary noise sources where technology exists.
 - The contractor shall prepare a detailed construction plan identifying the schedule for major noise-generating construction activities. The construction plan shall identify a procedure for coordination with adjacent residential land uses so that construction activities can be scheduled to minimize noise disturbance.
 - Designate a "disturbance coordinator" who would be responsible for responding to any complaints about construction noise. The disturbance coordinator will determine the cause of the noise complaint (e.g., bad muffler, etc.) and will require that reasonable measures be implemented to correct the problem.

In order to mitigate impacts relative to Vibration – Construction (Impact 6.6-5), the following mitigation measure would be implemented.

- **NOISE 6.6-2:** For discretionary projects where construction would include vibration-generating activities, such as pile driving, within 95 feet of existing structures, site-specific vibration studies shall be conducted to <u>ensure the development project would not</u> adversely affect adjacent properties to the satisfaction of the Chief Building Official. Such efforts shall be conducted by a qualified structural engineer and could determine the area of impact and to present appropriate mitigation measures that may include the following:
 - Identify sites that would include vibration compaction activities such as pile driving and have the potential to generate groundborne vibration and the sensitivity of nearby structures to groundborne vibration. This task should be conducted by a qualified structural engineer.
 - Develop a vibration monitoring and construction contingency plan to identify structures where monitoring would be conducted; set up a vibration monitoring schedule; define structure-specific vibration limits; and address the need to conduct photo, elevation, and crack surveys to document before and after construction conditions. Construction contingencies would be identified for when vibration levels approach the limits.
 - At a minimum, mMonitor vibration during initial demolition activities and during pile-driving activities. Monitoring results may indicate the need for more or less intensive measurements.
 - When vibration levels approach limits, suspend construction and implement contingencies to either lower vibration levels or secure the affected structures.
 - Conduct post-survey on structures where either monitoring has indicated high levels or complaints of damage have been made. Make appropriate repairs or compensation where damage has occurred as a result of construction activities.

6.6.7 Significance of Impacts after Mitigation

Impacts to existing noise sensitive land uses due to the increase in ambient noise levels associated with build-out of the proposed Uptown CPU and associated discretionary actions would remain significant and unavoidable (Impact 6.6-1). No feasible mitigation measures have been identified to address this impact because there is no mechanism or funded program in place to provide noise attenuation at existing structures that would be exposed to ambient noise increases.

There are no feasible mitigation measures to reduce impacts from ambient noise level increases associated with future ministerial development within the Uptown CPU area (Impact 6.6-2); thus, ambient noise impacts associated with future ministerial projects would remain significant and unavoidable. Similarly, impacts associated with future ministerial projects exposed to vehicular

traffic noise levels in excess of the compatibility levels established in the General Plan Noise Element, based on future (2035) noise contours would be significant and unavoidable (Impact 6.6-3).

Regarding temporary construction noise impacts (Impact 6.6-4), future construction projects would be required to incorporate the standard controls outlined in NOISE 6.6-1, which would reduce construction noise levels emanating from the site, limit construction hours, and minimize disruption and annoyance. With the implementation of these controls, and the limited duration of the noisegenerating construction period, the substantial temporary increase in ambient noise levels would be less than significant.

Regarding vibration impacts during construction (Impact 6.6-5), pile driving within 95 feet of existing structures has the potential to exceed 0.20 inch per second, resulting in a potentially significant impact. Implementation of mitigation measure NOISE 6.6-2 would reduce construction-related vibration impacts; however, at the program-level it cannot be known whether the measures would be adequate to minimize vibration levels to less than significant. Thus, even with implementation of mitigation measure NOISE 6.6-2, construction related vibration impacts would be significant and unavoidable.

6.7 Historical Resources

This section analyzes the potential impacts on historical resources due to implementation of the proposed Uptown Community Plan Update (CPU) and associated discretionary actions. It documents the historical background for the Uptown community and addresses prehistoric, historic, archaeological, and sacred sites. The information in this section is based on the *Community Plan Update for the Community of Uptown Prehistoric Cultural Resources* study (AECOM, January 2015) and the *Uptown Community Plan Area Historic Resources Survey* (Historic Resources Group, June 2014) and other primary and secondary sources. These reports are included in Appendix G1 and G2, respectively, to this Program Environmental Impact Report (PEIR).

6.7.1 Existing Conditions

A general discussion of the environmental setting relative to historical resources and the applicable regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

6.7.1.1 Designated Historical Resources

Uptown is home to 12 National Register-listed properties. These include the George Marston House, listed in 1974 as the home of Progressive San Diegan George Marston; the work of master architect Irving Gill during his formative years; and Park Place Methodist Episcopal Church, listed in 1982 as a remarkable example of a Classical Revival building designed by a master architect Norman Foote Marsh. In addition, as of February 2016, the Uptown community is home to 340 individually designated historical resources (Figure 6.7-1) and two designated historic districts (Figure 6.7-2) – Mission Hills and Fort Stockton Line – containing 209 contributing resources that have been listed on the City's register by the Historical Resources Board.



FIGURE 6.7-1 Location of City Register Designated Historic Resources

ATTACHMENT 7



 $FIGURE \ 6.7-2$ Location of City Register Designated Historic District

Map Source: SanGIS

6.7.2 Methodology

A Prehistoric Cultural Resources Study and a Historic Resources Survey were prepared for the proposed Uptown CPU and associated discretionary actions. The Cultural Resources Study describes the pre-history of the Uptown area, identifies known significant archaeological resources, provides guidance on the identification of possible new significant archaeological resources, and includes recommendations for the treatment of significant archaeological resources. The Historic Resources Survey provides information regarding the significant historical themes in the development of Uptown, the property types that convey those themes in an important way, and the location of potential historical resources within the community, including individual resources, multiple property listings, and districts.

6.7.2.1 Prehistoric Resources

Cultural sensitivity levels for the Uptown Community Plan area are rated low, moderate, or high based on the results of an archival records search using the California Historical Resources Information System (CHRIS), a literature search at the South Coastal Information Center (SCIC) located a San Diego State University, a records update at the San Diego Museum of Man, a Sacred Lands File check by the Native American Heritage Commission (NAHC), and regional environmental factors.

A low sensitivity rating indicates that there are few or no previously recorded resources within the area. Resources at this level would not be expected to be complex, with little to no site structure or artifact diversity. The potential for identification of additional resources in such areas would be low. A moderate sensitivity rating indicates that some previously recorded resources were identified within the area. These are more complex resources consisting of more site structure, diversity of feature types, and diversity of artifact types. The potential for the presence of additional resources in such areas would be moderate. Areas identified as high sensitivity would indicate that the records search identified several previously recorded sites within the area. These resources may range from moderately complex to highly complex, with more-defined living areas or specialized work space areas, and a large breadth of features and artifact assemblages. The potential for identification of additional resources in such areas would be high. Sensitivity ratings may be adjusted based on the amount of disturbance that has occurred, which may have previously impacted archaeological resources.

Because the majority of the community is developed and there is very little undeveloped land within the Community Plan area, with the exception of canyon areas, the cultural sensitivity for the entire Uptown community is considered low due to steepness of the canyon slopes. However, at the base of these canyons, especially leading into the Mission Valley area, there is a potential for cultural resources to be present; therefore, the cultural sensitivity rating for these areas is considered high, specifically when in proximity to the San Diego Presidio and areas bordering Old Town. As such, the community of Uptown contains two sensitivity ratings as illustrated in Figure 6.7-3.



 $FIGURE\ 6.7-3$ Cultural Sensitivity Areas – Prehistoric Resources

6.7.2.2 Historical Resources

The Historic Resources Survey was conducted using a four-step approach, which included research, fieldwork, evaluation, and documentation. The research phase involved review of various relevant City documents (municipal codes and regulations, planning reports, previous historic resources surveys, and various historic nominations), as well as various historical materials (period newspaper articles, photographs, maps).

The fieldwork phase consisted of a property-by-property inspection of the entire Community Plan area. Field teams identified individual properties that appeared eligible for individual designation, as well as geographically definable areas that appeared eligible for designation as historic districts. For districts, boundaries were defined and contributing and non-contributing resources were identified.

All properties identified in the field as potentially eligible for designation were then evaluated using the City of San Diego local designation criteria. Properties determined potentially eligible for designation on the City's Register were then evaluated for the National Register and California Register. All properties identified and evaluated as potentially eligible for listing on the San Diego Register, California Register, and/or National Register designation as part of this survey were then documented in a database.

Included as an appendix to the Historic Resources Survey is the Historic Context Statement prepared for the Uptown community. The Historic Context Statement was developed primarily through archival research. It synthesizes information collected from a variety of primary and secondary materials. In addition to consulting the historical resource files at the City Planning and Community Investment Department and the archives at Save Our Heritage Organisation, research was conducted at the San Diego Public Library, the San Diego Historical Society, and the libraries at the University of California, San Diego. Primary sources included historic maps, photographs and newspapers, and media advertisements. Specifically, subdivision maps, in conjunction with Sanborn Fire Insurance Maps, were used to establish broad patterns of development within Uptown. Historic photographs provided imagery of the community's evolving landscape and predominant architectural styles. Other primary materials included several articles, advertisements, and editorials from the archives of the *Los Angeles Times* and *San Diego Union*. Secondary sources of information were consulted to supplement these primary materials, and included later accounts of history recorded in a variety of books, essays, journals, and master's theses.

6.7.3 Significance Determination Thresholds

Historical resources significance determination, pursuant to the City of San Diego's Significance Determination Thresholds, consists first of determining the sensitivity or significance of identified historical resources and, secondly, determining direct and indirect impacts that would result from project implementation. Based on the City's 2011 Significance Determination Thresholds, which have been adopted to guide a programmatic assessment of the proposed Uptown CPU and associated discretionary actions, impacts related to historical resources would be significant if the project would result in:

- 1) An alteration, including the adverse physical or aesthetic effects and/or the destruction of a historic building (including an architecturally significant building), structure, object or site; or
- 2) A substantial adverse change in the significance of a prehistoric archaeological resource, a religious or sacred use site, or the disturbance of any human remains, including those interred outside of formal cemeteries.

The City of San Diego's California Environmental Quality Act (CEQA) Significance Determination Thresholds define a significant historical resource as one which qualifies for the California Register of Historical Resources or is listed in a local historic register or deemed significant in a historical resource survey, as provided under Section 5024.1(g) of the Public Resources Code, although even a resource that is not listed in, or determined eligible for listing in, the California Register, not included in a local register, or not deemed significant in a historical resource survey may nonetheless be historically significant for purposes of CEQA. The City's Historical Resources Guidelines state the significance of a resource may be determined based on the potential for the resource to address important research questions as documented in a site-specific technical report prepared as part of the environmental review process.

Research priorities for the prehistoric, ethnohistoric, and historic periods of San Diego history are discussed in Appendix A to the City's Historical Resources Guidelines. As a baseline, the City of San Diego has established the following criteria to be used in the determination of significance under CEQA:

- An archaeological site must consist of at least three associated artifacts/ecofacts (within a 50 square meter area) or a single feature and must be at least 45 years of age. Archaeological sites containing only a surface component are generally considered not significant, unless demonstrated otherwise. Such site types may include isolated finds, bedrock milling stations, sparse lithic scatters, and shellfish processing stations. All other archaeological sites are considered potentially significant. The determination of significance is based on a number of factors specific to a particular site including site size, type and integrity; presence or absence of a subsurface deposit, soil stratigraphy, features, diagnostics, and datable material; artifact and ecofact density; assemblage complexity; cultural affiliation; association with an important person or event; and ethnic importance.
- The determination of significance for historic buildings, structures, objects and landscapes is based on age, location, context, association with an important person or event, uniqueness, and integrity.
- A site will be considered to possess ethnic significance if it is associated with a burial or cemetery; religious social or traditional activities of a discrete ethnic population; an important person or event as defined by a discrete ethnic population; or the mythology of a discrete ethnic population.

6.7.4 Impact Analysis

Issue 1 Historic Structures, Objects, or Sites

Would implementation of the proposed Uptown CPU and associated discretionary actions result in an alteration, including the adverse physical or aesthetic effects and/or the destruction of a historic building (including an architecturally significant building), structure, object, or site?

a. Individual Local Historic Resources

Of the 11,104 properties surveyed in 2004–2006, the survey identified 2,192 properties as potentially significant individual resources, 59 of which are also located in Potential Historic Districts. The resources identified can be found in the Uptown Community Plan Area Historic Resources Survey (Appendix G2). Of the resources identified as potentially significant individual resources, approximately 56 percent are single-family properties, 35 percent are multi-family properties, eight percent are commercial properties, and one percent are institutional properties. Thematically, the potentially significant individual resources are distributed as follows:

- The Railroad Boom and Early Residential Development: 1885 to1909: 12 percent
- The Panama-California Exposition and Streetcar Suburbs: 1909 to1929: 44 percent
- Great Depression and World War II: 1929 to 1948: 21.5 percent
- Postwar Development, Suburbanization, the Automobile, and Modernism: 1948 to 1970: 22 percent
- Neighborhood Revitalization and the Lesbian, Gay, Bisexual, and Transgender (LGBT) Community: 1970 to Present: 0 percent * (*The 2004 to 2006 survey only evaluated properties constructed prior to 1961.)

b. Potential Historic Districts Identified in the Historic Resources Survey

The Historical Resources Survey identified 19 new Potential Historic Districts containing a total of approximately 2,600 properties and roughly 2,000 contributing resources. The name, location, size, period of significance and relationship to Uptown's significant development themes are summarized in Table 6.7-1, and their locations shown in Figure 6.7-4. More detailed information, including listings of contributing resources, can be found in the Uptown Historic Survey Report.

			Table 6.7-1		
Pot	ential Historic Dis	tricts Ident	ified in the l	Jptown Historic Resource Survey	
Detential Historia			Deviaded		Possible
Potential Historic	Location	Cizo	Period of	Theme(c)	HKB
	Location	SIZE	Significance	The Deilroad Dears and Farky	Criterion
Arnold &	Barr Street, Dove	313 Droportion	1890-1951	The Railroad Boom and Early	A&C
Choales Detential Historic	Street, University	Properties		Residential Development: 1885-	
	Avenue anu Dandolph Stroot			1909	
DISTICT	Randolph Street			The Panama California Exposition	
				and Streetcar Suburbs: 1909-1929	
				Great Depression and World War II:	
				1929-1948	
				Postwar Development,	
				Suburbanization, the Automobile	
				and Modernism: 1948-1970.	
Dove Street	West Palm Street,	31	1928-1948	The Panama-California Exposition	A & C
Potential Historic	North Arroyo	Properties		and Streetcar Suburbs: 1909-1929	
District	Drive, Arroyo				
	Drive, and			Great Depression and World War II:	
	Reynard Way			1929-1948	
Heart of	Pennsylvania	125	1870-1940	Early History: 1769-1885	A, C & D
Banker's	Avenue,	Properties		The Dellased Deers and Fash	
Hill Potential	First Avenue,			The Railfoad Boom and Early	
	Reuwoou Street and Dove			1000	
	Street and Dove			1909	
	Sileer			The Panama-California Exposition	
				and Streetcar Suburbs: 1909-1929	
				Great Depression and World War II:	
				1929-1948	
Horton's	Laurel Street, 4 th	143	1871-1940	Early History: 1769-1885	C & D
Addition	Avenue, Grape	Properties			
Potential Historic	Street and Brant			The Railroad Boom and Early	
District	Street			Residential Development: 1885-	
				1909	
				The Panama-California Exposition	
				and Streetcar Suburbs: 1909-1929	
				Great Depression and World War II: 1929-1948	

Pot	Table 6.7-1 Potential Historic Districts Identified in the Untown Historic Resource Survey							
					Possible			
Potential Historic			Period of		HRB			
District	Location	Size	Significance	Theme(s)	Criterion			
Inspiration Heights Potential Historic District	Sunset Boulevard, Saint James Place, Putterbaugh	84 Properties	1887 and 1909-1942	The Railroad Boom and Early Residential Development: 1885- 1909	A & C			
	Street and Couts Street			The Panama-California Exposition and Streetcar Suburbs: 1909-1929 Great Depression and World War II: 1929-1948				
Inspiration View	Torrance Street,	24	1925-1936	The Panama-California Exposition	A & C			
Potential Historic District	Ostego Drive, Walnut	Properties		and Streetcar Suburbs: 1909-1929				
	Avenue and Eagle Street			Great Depression and World War II: 1929-1948				
John Sherman Potential Historic District	Grape Street, First Avenue, Fir Street and Front Street	12 Properties	1880-1915	Early History: 1769-1885 The Railroad Boom and Farly	C & D			
				Residential Development: 1885- 1909				
				The Panama-California Exposition and Streetcar Suburbs: 1909-1929				
Marine View Potential Historic District	University Avenue, Eagle Street, Brookes Avenue and	340 Properties	1891-1950	The Railroad Boom and Early Residential Development: 1885- 1909	A & C			
	Winder and Welborn Streets			The Panama-California Exposition and Streetcar Suburbs: 1909-1929				
				Great Depression and World War II: 1929-1948				
				Postwar Development, Suburbanization, the Automobile, & Modernism: 1948-1970				
Marston Family	Brookes Avenue.	11	1904-1918	The Railroad Boom and Early	A. B. C & D			
Potential Historic District	Highway 163, Upas Street and the alley between	Properties		Residential Development: 1885- 1909	.,			
	6 th and 7 th Avenues			The Panama-California Exposition and Streetcar Suburbs: 1909-1929				
Marston Hills	Pennsylvania	88	1924-1940	The Panama-California Exposition	A, B, C & D			
Potential Historic District	Avenue, Highway 163, Upas Street and Richmond	Properties		and Streetcar Suburbs: 1909-1929 Great Depression and World War II:				
	and Vermont Streets			1929-1948				
Table 6.7-1								
--------------------------------------------------------------------------------	---------------------	-----------------	--------------	------------------------------------	-----------	--	--	
Potential Historic Districts Identified in the Uptown Historic Resource Survey								
.					Possible			
Potential Historic		C:	Period of		HRB			
District	Location	Size	Significance	Ineme(s)	Criterion			
MISSION HIIIS	Altamira Place	517 December	1908-1941	The Railroad Boom and Early	C&D			
Historic District	and the bluff	Properties		Residential Development: 1885-				
Expansion Area	north of			1909				
	Hortensia Street			The Panama-California Exposition				
	to the north:			and Streetcar Suburbs: 1909-1929				
	Stephens Street							
	to the east;			Great Depression and World War II:				
	Sunset Boulevard,			1929-1948				
	Torrance Street,							
	Neale Street and							
	Pringle Street to							
	the south; and St.							
	James Place,							
	Trias Street and							
	Hortensia Street							
	to the west							
North Florence	Hunter Street,	96	1890-1940	The Railroad Boom and Early	A, B & C			
Heights Potential	Randolph Street,	Properties		Residential Development: 1885-				
Historic District	Mission			1909				
	Hills/Pioneer							
	Park, and			The Panama-California Exposition				
	Stephens Street			and Streetcar Suburbs: 1909-1929				
				Great Depression and World War II:				
				1929-1948				
Northwest	Arista Street and	301	1908-1950	The Railroad Boom and Early	A, C & D			
Mission Hills	Conde Street to	Properties		Residential Development: 1885-				
Potential Historic	the north; the			1909				
District	Interstate 8 to the			The Panama-California Exposition				
	east: Witherby			and Streetcar Suburbs: 1909-1929				
	Street, Trias							
	Street and			Great Depression and World War II:				
	Hortensia Street			1929-1948				
	to the south; and							
	Juan Street and			Postwar Development,				
	Sunset Boulevard			Suburbanization, the Automobile, &				
	to the west			Modernism: 1948-1970				

Table 6.7-1						
Pot	ential Historic Dis	tricts ident	med in the C	optown Historic Resource Survey	Possible	
Potential Historic			Period of		HRB	
District	Location	Size	Significance	Theme(s)	Criterion	
Park Boulevard Potential Historic District (West)	Robinson Avenue, Park Boulevard, Upas Street, and the alley between	35 Properties	1888-1960	The Railroad Boom and Early Residential Development: 1885- 1909	A & C	
	Park Boulevard and Herbert Street.			and Streetcar Suburbs: 1909-1929		
				Great Depression and World War II: 1929-1948		
				Postwar Development, Suburbanization, the Automobile, & Modernism: 1948-1970		
Park Edge North Potential Historic District	Herbert Place; the alley between Park Boulevard and Herbert	122 Properties	1888-1940	The Railroad Boom and Early Residential Development: 1885- 1909	A & C	
	Street; Upas Street; and Richmond Street			The Panama-California Exposition and Streetcar Suburbs: 1909-1929		
				Great Depression and World War II: 1929-1948		
Presidio Hill Potential Historic District	Arista Street, Presidio Drive, Cosoy Way and	59 Properties	1926-1945	The Panama-California Exposition and Streetcar Suburbs: 1909-1929	A, C & D	
	the bluff edge overlooking Presidio Park			Great Depression and World War II: 1929-1948		
Robinson Place Potential Historic District	Robinson Avenue, Herbert Street, Pennsylvania Avenue and Albert Street	14 Properties	1925-1927	The Panama-California Exposition and Streetcar Suburbs: 1909-1929	A & C	
Second Avenue Potential Historic District	Along Second Avenue between Upas Street and Palm Street	48 Properties	1871-1945	Early History: 1769-1885 The Railroad Boom and Early Residential Development: 1885- 1909	A, C & D	
				The Panama-California Exposition and Streetcar Suburbs: 1909-1929		
				Great Depression and World War II: 1929-1948		

Table 6.7-1							
Potential Historic Districts Identified in the Uptown Historic Resource Survey							
					Possible		
Potential Historic			Period of		HRB		
District	Location	Size	Significance	Theme(s)	Criterion		
West University	Bounded by the	458	1888-1945	The Railroad Boom and Early	A & C		
Heights Potential	bluff facing	Properties		Residential Development: 1885-			
Historic District	Interstate 8 and			1909			
	Lincoln Avenue to						
	the north;			The Panama-California Exposition			
	Cleveland Avenue			and Streetcar Suburbs: 1909-1929			
	to the east;						
	Washington			Great Depression and World War II:			
	Street to the			1929-1948			
	south; and Rhode						
	Island Street and						
	the west side of						
	Vermont Street to						
	the west						
HRB = Historical Resources Boar							

c. Multiple Property Listing

A Multiple Property Listing (MPL) is a group of related significant properties with shared themes, trends, and patterns of history. The Uptown Survey has identified three thematically related property groupings that appear eligible as Multiple Property Listings, the Bungalow and Apartment Court MPL, the Kate Olivia Sessions MPL, and the Victorian Era MPL

Bungalow and Apartment Court Multiple Property Listing. The Residential Court Multiple Property Listing is a discontiguous grouping of approximately 150 residential courts located throughout the Uptown survey area. Eligible under San Diego Criteria A and C, this potential MPL reflects the distinctive characteristics of courtyard design, as well as special elements of the Uptown Community's social history related to multi-family housing, and its architectural development associated with local transportation patterns.

Kate Olivia Sessions Multiple Property Listing. The Kate Olivia Sessions Multiple Property Listing is a discontiguous grouping of four geographic areas located throughout the Uptown survey area, Sixth Avenue/Balboa Park Urban Edge, Lark Street, the Kate Sessions Mission Hills Nursery Site, and the Kate Sessions Balboa Park Nursery Site. Eligible under San Diego Criteria A and D, this potential MPL reflects special elements of the Uptown Community's landscape design and horticultural history, and is significant as the work of noted horticulturalist Kate Olivia Sessions.

Victorian Era Multiple Property Listing. The Victorian Era Multiple Property Listing is a discontiguous grouping of approximately 459 Victorian Era buildings located throughout the Uptown survey area. Eligible under San Diego Criteria A, C and D, this potential MPL reflects the distinctive characteristics of residential, commercial, and institutional Victorian era architecture; the work of Master Architects and Builders; as well as special elements of the Uptown Community's early development history.



FIGURE 6.7-4 Potential Historic Districts Identified by the Historic Resources Survey

d. Resources Identified through Public Outreach

Substantial public outreach with the Uptown Community Planning group, regional and local preservation groups, and members of the community occurred throughout the CPU process. This information was considered and often incorporated into the results and recommendations of the survey. As a result, the Uptown Community Plan Area Historic Resources Survey identifies as potentially significant all individual resources specifically identified as such by the community. The exception is properties that have been identified as potentially significant under the theme "Neighborhood Revitalization and the LGBTQ Community: 1970–Present." When the survey work was conducted in 2004 to 2006, only properties that were 45 years old or older upon completion of the survey in 2006 were evaluated; therefore, no properties constructed after 1961 were evaluated by the survey. Additionally, due to the reconnaissance nature of the survey, properties that were significantly altered from their original appearance were not evaluated further to explore significance related to lesbian, gay, bisexual, transgender, and queer (LGBTQ) history and redevelopment of Hillcrest. In developing the final theme, staff conducted limited research, oral interviews and a walking tour in an effort to identify the location of resources that may be eligible under the LGBTQ Community: 1970–Present theme. Based on the results of this outreach, additional resources have been identified as potentially significant, requiring further site-specific evaluation. These resources are identified in the Historic Resources Survey and the Historic Preservation Element.

In addition to these individual resources, four additional Potential Historic Districts have been identified by the community (Figure 6.7-5). These include Allen Terrace, Avalon Heights, Hillcrest, and San Diego Normal School/San Diego City Schools Education Complex. The San Diego Normal School/San Diego City Schools Education Complex was the subject of a reconnaissance survey commissioned by the University Heights Historical Society and completed by a qualified historic consultant. Staff conducted a windshield survey to verify the presence of a potential historic district in the other three areas and concurred that these areas may be eligible for designation as Potential Historic Districts. However, the windshield survey undertaken in these areas was not as thorough as the Uptown Community Plan Area Historic Resources Survey (2014) and did not include identification of contributing and non-contributing resources. In regard to Hillcrest, it must be noted that the survey work completed in 2004 to 2006 did not initially identify a potential district in the Hillcrest area. However, the date and reconnaissance nature of the survey significantly limited the evaluation of resources associated with the final theme of revitalization and LGBTQ history. Given the fact that many business catering to and run by members of the LGBTQ community are concentrated within the Hillcrest area, along with residential units occupied by individuals and early advocacy groups, it is appropriate to identify Hillcrest as a potential historic district under Historical Resources Board Criterion A. In addition, because the Hillcrest Potential Historic District area includes 55 properties constructed from 1960 to 1975, and because the 2004 to 2006 survey did not consider any properties constructed post-1960 as potential resources, it is appropriate to consider that the district may also be eligible under Historical Resources Board Criterion C. In order to bring these four potential districts forward for designation, additional, intensive-level research would be required to evaluate the district and define a precise boundary, period of significance, significance criteria, and contributing and non-contributing resources.



FIGURE 6.7-5 Potential Historic Districts Identified by the Community

e. Regulatory Framework

Uptown is home to 12 National Register-listed properties, 340 individually designated historic resources (see Figure 6.7-1), and two designated historic districts (see Figure 6.7-2) that are protected and preserved through existing General Plan policies, the historical resources regulations and guidelines of the Municipal Code, and established City practices. These protections require historic review of all projects impacting these resources. Projects that do not comply with the U.S. Secretary of the Interior Standards are required to process a discretionary action with deviations that are subject to review under California Environmental Quality Act (CEQA).

The Historical Resources Survey conducted for Uptown identified 2,192 properties as potentially significant individual resources, 59 of which are also located in Potential Historic Districts. The resources identified can be found in the Uptown Historic Survey Report. Of the resources identified as potentially significant individual resources, approximately 56 percent are single-family properties, 35 percent are multi-family properties, eight percent are commercial properties, and one percent institutional properties. In addition to potentially individually significant resources, the survey identified 19 potential Historic Districts as well as three potential MPLs.

Specifically, San Diego Municipal Code Section 143.0212 requires review of ministerial and discretionary permit applications impacting parcels containing buildings 45 years old or older to determine whether or not the project has the potential to adversely impact a resource that may be eligible for individual listing on the local register. When it is determined that a resource may exist and a proposed project would constitute a significant impact to that resource, a site-specific survey is required and may be forwarded to the Historical Resources Board to consider designation and listing of the property. If designated, a Site Development Permit with deviation findings and mitigation would be required for any substantial modification of the resource. If the property were not designated, modification of the property would not be subject to the Historical Resources Regulations. Potential individual resources and resources identified as part of the MPL, which are evaluated as single resources to the project parcel and individual eligibility, resources identified as potentially contributing to a potential historic district would not be protect unless they were also eligible individually.

The proposed Uptown CPU and associated discretionary actions would have a significant direct impact on historical resources if they result in the demolition, relocation, or substantial alteration of a resource listed in, or formally determined eligible for listing in the National Register of Historic Places (NRHP) or the California Register of Historic Resources (CRHR), including contributors to NRHP and CRHR-eligible Historic Districts, or the San Diego Historical Resources Register, including contributors to San Diego Register Historic Districts, or which otherwise meet CEQA criteria for historic resources, as discussed above. Although the proposed Uptown CPU and associated discretionary actions do not propose specific development, future development and related construction activities facilitated by the proposed Uptown CPU and associated discretionary actions at the project level could result in the alteration of a historic building, structure, object, or site. Direct impacts may include substantial alteration, relocation, or demolition of historic buildings, structures, objects, sites, and districts. Indirect impacts may include the introduction of visual, audible, or

atmospheric effects that are out of character with a historic property or alter its setting, when the setting contributes to the resource's significance.

Section 143.0212 of the San Diego Municipal Code requires review of ministerial and discretionary permit applications for any parcel identified as sensitive on the Historical Resource Sensitivity Maps specifically to determine whether or not the project has the potential to adversely impact an archaeological resource which may be eligible for individual listing on the local register. In these cases, this review is supplemented with a project-specific records search of the California Office of Historic Preservation's California Historical Resources Information System (CHRIS) data by qualified staff and a site-specific survey would be required. In addition to these existing protections on historic resources, tThe Uptown CPU contains a Historic Preservation Element that supports the Historic Preservation Element of the General Plan through goals and policies for identifying and preserving historical, archaeological and tribal cultural resources, and educating citizens about the benefits of, and incentives for, historic preservation. Proposed policies supporting the identification and preservation of historical resources are included in the Urban Design and Conservation elements of the proposed Uptown CPU. Policies seek to preserve and enhance the historic character of the Uptown community and facilitate the identification, designation, and preservation of historically and culturally significant resources throughout the Uptown CPU area. In response to this policy, the proposed CPU identifies Potential Historic Districts. Supplemental development regulations to the Historical Resources Regulations would address how and where modifications can be made on residential properties identified as potentially contributing to specified Potential Historic Districts. Development that does not comply with the regulations of the supplemental development regulations would be subject to a Neighborhood Development Permit with deviation findings and mitigation.

While the Municipal Code does provide for the regulation and protection of designated and potential historical resources, and while supplemental development regulations would provide additional protection for Potential Historic Districts, it is impossible to ensure the successful preservation of all historic built environment resources within the plan area. Therefore, impacts to the Potential Historic Districts and individual historic resources would be considered significant and unavoidable.

Impact 6.7-1: Implementation of the proposed Uptown CPU and associated discretionary actions could result in an alteration of a historic building, structure, object, or site.

Issue 2 Prehistoric Resources, Sacred Sites and Human Remains

The Preshistoric Cultural Resources Study identified 14 recorded archaeological sites and 98 previous investigations conducted within the community of Uptown. Although there is very little undeveloped land within the CPU area, future development and related construction activities facilitated by the proposed Uptown CPU at the project level could result in the alteration or destruction of prehistoric resources, objects, or sites and could impact religious or sacred uses; or disturb human remains, particularly at the base of canyons leading into the Mission Valley area and in proximity to the Presidio and areas bordering Old Town. Direct impacts may include substantial alteration or demolition of archaeological sites from grading, excavation, or other ground-disturbing

activities. Indirect impacts may include the potential for vandalism or destruction of an archaeological resource or traditional cultural property.

Avoiding impacts on religious or sacred places or human remains may be unavoidable in certain circumstances when resources are discovered during construction. Although there are no known religious or sacred uses within the CPU area, there is potential for these to be encountered during future construction activities associated with implementation of the proposed Uptown CPU and associated discretionary actions, particularly given the high cultural sensitivity of canyon areas leading into the Mission Valley area, which has been previously identified as an area of concern to the local Native American community, and in proximity to the Presidio and areas bordering Old Town. Similarly, there are no known human remains interred outside of formal cemeteries. However, there are many areas within the City where previously unknown prehistoric human remains and prehistoric sites have been uncovered during both archaeological investigations and grading activities. Therefore, tribal consultation in accordance with AB 52 and the Public Resources Code has been incorporated into the Mitigation Framework for subsequent projects to ensure that tribal cultural resources are addressed early in the development review process. However, the potential for encountering human remains during construction activities remains a possibility.

The City has developed Historic Resource Sensitivity Maps that provide general locations of where Historical Resources are known to occur or have the potential to occur. These maps were developed in coordination with technical experts and tribal representatives. Upon submittal of ministerial and/or discretionary permit applications, a parcel is reviewed against the Historical Resource Sensitivity Maps specifically to determine whether or not the project has the potential to adversely impact an archaeological resource which may be eligible for individual listing on the local register (SDMC Section 143.0212).

Section 143.0212 of the SDMC requires review of ministerial and discretionary permit applications for any parcel identified as sensitive on the Historical Resource Sensitivity Maps specifically to determine whether or not the project has the potential to adversely impact an archaeological resource which may be eligible for individual listing on the local register. In these cases, tThis review is supplemented with a project specific records search of the NAHC Sacred Lands File by qualified staff, and as stated above, a site specific archaeological survey would be required. For any subsequent projects implemented in accordance with the CPU where a recorded archaeological site or Tribal Cultural Resource (as defined in the Public Resources Code) is identified, the City would be required to initiate consultation with identified California Indian tribes pursuant to the provisions in Public Resources Code Section 21080.3.1 and 21080.3.2., in accordance with Assembly Bill 52. Results of the consultation process will determine the nature and extent of any additional archaeological evaluation or changes to the proposed project and appropriate mitigation measures for direct impacts that cannot be avoided.

The proposed Uptown CPU is designed to support the historic preservation goals of the City's General Plan, and contains policies requiring protection and preservation of significant archaeological resources in the proposed Historic Preservation Element. Native American consultation early in the project review process is also included in the proposed Uptown CPU to identify tribal cultural resources and to develop adequate treatment and mitigation for significant

archaeological sites with cultural and religious significance to the Native American community in accordance with all applicable local, state and federal regulations and guidelines.

While existing regulations, the Municipal Code, and proposed Uptown CPU policies would provide for the regulation and protection of archaeological resources and human remains, it is impossible to ensure the successful preservation of all archaeological resources. Therefore, potential impacts to archaeological resources are considered significant.

Impact 6.7-2: Implementation of the proposed Uptown CPU and associated discretionary actions could adversely impact a prehistoric archaeological resource including religious or sacred use sites and human remains.

6.7.5 Significance of Impacts

Implementation of the Uptown CPU could result in an alteration of a historic building, structure, object, or site (Impact 6.7-1) and could adversely impact a prehistoric archaeological <u>and tribal</u> <u>cultural</u> resources including religious or sacred use sites and human remains (Impact 6.7-2). These impacts would be potentially significant.

6.7.6 Mitigation Framework

The City of San Diego's General Plan, combined with federal, state, and local regulations, provide a regulatory framework for project-level historical resources evaluation/analysis criteria and, when applicable, mitigation measures for future discretionary projects. All development projects with the potential to affect historical resources—such as designated historical resources; historical buildings, districts, landscapes, objects, and structures; important archaeological sites; tribal cultural resources, and traditional cultural properties—are subject to site-specific review in accordance with the City's Historical Resources Regulations and Historical Resources Guidelines, through the subsequent project review process. The following mitigation measures (HIST-6.7-1 and HIST 6.7-2) provide a framework that would be required of all development projects with the potential to impact significant historical resources.

HIST 6.7-1: Historic Buildings, Structures, and Objects

Prior to issuance of any permit for a development project implemented in accordance with the proposed Uptown CPU and associated discretionary actions that would directly or indirectly affect a building/structure in excess of 45 years of age, the City shall determine whether the affected building/structure is historically significant. The evaluation of historic architectural resources shall be based on criteria such as: age, location, context, association with an important person or event, uniqueness, or structural integrity, as indicated in the Guidelines.

Preferred mitigation for historic buildings or structures shall be to avoid the resource through project redesign. If the resource cannot be entirely avoided, all prudent and feasible measures to minimize harm to the resource shall be taken. Depending upon project impacts, measures shall include, but are not limited to:

- Preparing a historic resource management plan;
- Adding new construction which is compatible in size, scale, materials, color and workmanship to the historical resource (such additions, whether portions of existing buildings or additions to historic districts, shall be clearly distinguishable from historic fabric);
- Repairing damage according to the Secretary of the Interior's Standards for Rehabilitation;
- Screening incompatible new construction from view through the use of berms, walls and landscaping in keeping with the historic period and character of the resource; and
- Shielding historic properties from noise generators through the use of sound walls, double glazing and air conditioning.

Specific types of historical resource reports, outlined in Section III of the Historical Resources Guidelines, are required to document the methods to be used to determine the presence or absence of historical resources, to identify potential impacts from a proposed project, and to evaluate the significance of any historical resources identified. If potentially significant impacts to an identified historical resource are identified these reports will also recommend appropriate mitigation to reduce the impacts to below a level of significance, where possible. If required, mitigation programs can also be included in the report.

HIST-6.7-2: Archaeological and Tribal Cultural Resources

Prior to issuance of any permit for a future development project implemented in accordance with the proposed Uptown CPU and associated discretionary actions that could directly affect an archaeological or tribal cultural resource, the City shall require the following steps be taken to determine: (1) the presence of archaeological or tribal cultural resources and (2) the appropriate mitigation for any significant resources which may be impacted by a development activity. Sites may include, but are not limited to, residential and commercial properties, privies, trash pits, building foundations, and industrial features representing the contributions of people from diverse socio-economic and ethnic backgrounds. Sites may also include resources associated with prehistoric Native American activities.

Initial Determination

The environmental analyst will determine the likelihood for the project site to contain historical resources by reviewing site photographs and existing historic information (e.g. Archaeological Sensitivity Maps, the Archaeological Map Book, and the City's "Historical Inventory of Important Architects, Structures, and People in San Diego") and may conduct a site visit, as needed. If there is any evidence that the site contains archaeological or tribal cultural resources, then an archaeological

evaluation consistent with the City Guidelines would be required. All individuals conducting any phase of the archaeological evaluation program must meet professional qualifications in accordance with the City Guidelines.

Step 1:

Based on the results of the Initial Determination, if there is evidence that the site contains a historical resource, preparation of a historic evaluation is required. The evaluation report would generally include background research, field survey, archaeological testing and analysis. Before actual field reconnaissance would occur, background research is required which includes a record search at the SCIC at San Diego State University and the San Diego Museum of Man. A review of the Sacred Lands File maintained by the NAHC must also be conducted at this time. Information about existing archaeological collections should also be obtained from the San Diego Archaeologyical Center and any tribal repositories or museums.

In addition to the record searches mentioned above, background information may include, but is not limited to: examining primary sources of historical information (e.g., deeds and wills), secondary sources (e.g., local histories and genealogies), Sanborn Fire Maps, and historic cartographic and aerial photograph sources; reviewing previous archaeological research in similar areas, models that predict site distribution, and archaeological, architectural, and historical site inventory files; and conducting informant interviews. The results of the background information would be included in the evaluation report.

Once the background research is complete, a field reconnaissance must be conducted by individuals whose qualifications meet the standards outlined in the City Guidelines. Consultants are encouraged to employ innovative survey techniques when conducting enhanced reconnaissance, including, but not limited to, remote sensing, ground penetrating radar, and other soil resistivity techniques as determined on a case-by-case basis. Native American participation is required for field surveys when there is likelihood that the project site contains prehistoric archaeological resources or traditional cultural properties. If through background research and field surveys historical resources are identified, then an evaluation of significance, based on the City Guidelines, must be performed by a qualified archaeologist.

Step 2

Where a recorded archaeological site or Tribal Cultural Resource (as defined in the Public Resources Code) is identified, the City would be required to initiate consultation with identified California Indian tribes pursuant to the provisions in Public Resources Code Section 21080.3.1 and 21080.3.2., in accordance with Assembly Bill 52. It should be noted that during the consultation process tribal representative(s) will be directly involved in making recommendations regarding the significance of a tribal cultural resource which also could be a prehistoric archaeological site. A testing program may be recommended which requires

reevaluation of the proposed project in consultation with the Native American representative which could result in a combination of project redesign to avoid and/or preserve significant resources as well as mitigation in the form of data recovery and monitoring (as recommended by the qualified archaeologist and Native American representative). The archaeological testing program, if required <u>will-shall</u> include evaluating the horizontal and vertical dimensions of a site, the chronological placement, site function, artifact/ecofact density and variability, presence/absence of subsurface features, and research potential. A thorough discussion of testing methodologies, including surface and subsurface investigations, can be found in the City Guidelines. Results of the consultation process will determine the nature and extent of any additional archaeological evaluation or changes to the proposed project.

The results from the testing program shall be evaluated against the Significance Thresholds found in the Guidelines. If significant historical resources are identified within the Area of Potential Effect, the site may be eligible for local designation. However, this process would not proceed until such time that the tribal consultation has been concluded and an agreement is reached (or not reached) regarding significance of the resource and appropriate mitigation measures are identified. When appropriate, the final testing report must be submitted to Historical Resources Board staff for eligibility determination and possible designation. An agreement on the appropriate form of mitigation is required prior to distribution of a draft environmental document. If no significant resources are found, and site conditions are such that there is no potential for further discoveries, then no further action is required. Resources found to be non-significant as a result of a survey and/or assessment will require no further work beyond documentation of the resources on the appropriate Department of Parks and Recreation site forms and inclusion of results in the survey and/or assessment report. If no significant resources are found, but results of the initial evaluation and testing phase indicates there is still a potential for resources to be present in portions of the property that could not be tested, then mitigation monitoring is required.

Step 3:

Preferred mitigation for historical resources is to avoid the resource through project redesign. If the resource cannot be entirely avoided, all prudent and feasible measures to minimize harm shall be taken. For archaeological resources where preservation is not an option, a Research Design and Data Recovery Program is required, which includes a Collections Management Plan for review and approval. When tribal cultural resources are present and also cannot be avoided, appropriate and feasible mitigation will be determined through the tribal consultation process and incorporated into the overall data recovery program, where applicable or project specific mitigation measures incorporated into the project. The data recovery program shall be based on a written research design and is subject to the provisions as outlined in CEQA, Section 21083.2. The data recovery program must be reviewed and approved by the City's Environmental Analyst prior to distribution of a draft

CEQA document and shall include the results of the tribal consultation process. Archaeological monitoring may be required during building demolition and/or construction grading when significant resources are known or suspected to be present on a site, but cannot be recovered prior to grading due to obstructions such as, but not limited to, existing development or dense vegetation.

A Native American observer must be retained for all subsurface investigations, including geotechnical testing and other ground-disturbing activities, whenever a Native American Traditional Cultural Propertytribal cultural resource or any archaeological site located on City property or within the Area of Potential Effect of a City project would be impacted. In the event that human remains are encountered during data recovery and/or a monitoring program, the provisions of Public Resources Code Section 5097 must be followed. In the event that human remains are discovered during project grading, work shall halt in that area and the procedures set forth in the California Public Resources Code (Section 50987.98) and State Health and Safety Code (Section 7050.5), and in the federal, state, and local regulations described above shall be undertaken. These provisions will be outlined in the Mitigation Monitoring and Reporting Program (MMRP) included in a subsequent project-specific -environmental document. The Native American monitor shall be consulted during the preparation of the written report, at which time they may express concerns about the treatment of sensitive resources. If the Native American community requests participation of an observer for subsurface investigations on private property, the request shall be honored.

Step 4:

Archaeological Resource Management reports shall be prepared by qualified professionals as determined by the criteria set forth in Appendix B of the Guidelines. The discipline shall be tailored to the resource under evaluation. In cases involving complex resources, such as traditional cultural properties, rural landscape districts, sites involving a combination of prehistoric and historic archaeology, or historic districts, a team of experts will be necessary for a complete evaluation.

Specific types of historical resource reports are required to document the methods (see Section III of the Guidelines) used to determine the presence or absence of historical resources; to identify the potential impacts from proposed development and evaluate the significance of any identified historical resources; to document the appropriate curation of archaeological collections (e.g. collected materials and the associated records); in the case of potentially significant impacts to historical resources, to recommend appropriate mitigation measures that would reduce the impacts to below a level of significance; and to document the results of mitigation and monitoring programs, if required.

Archaeological Resource Management reports shall be prepared in conformance with the California Office of Historic Preservation "Archaeological Resource Management Reports: Recommended Contents and Format" (see Appendix C of the Guidelines), which will be used by Environmental staff in the review of archaeological resource reports. Consultants must ensure that archaeological resource reports are prepared consistent with this checklist. This requirement will standardize the content and format of all archaeological technical reports submitted to the City. A confidential appendix must be submitted (under separate cover) along with historical resources reports for archaeological sites and tribal cultural resources containing the confidential resource maps and records search information gathered during the background study. In addition, a Collections Management Plan shall be prepared for projects which result in a substantial collection of artifacts and must address the management and research goals of the project and the types of materials to be collected and curated based on a sampling strategy that is acceptable to the City. Appendix D (Historical Resources Report Form) may be used when no archaeological resources were identified within the project boundaries.

Step 5:

For Archaeological Resources: All cultural materials, including original maps, field notes, non-burial related artifacts, catalog information, and final reports recovered during public and/or private development projects must be permanently curated with an appropriate institution, one which has the proper facilities and staffing for insuring research access to the collections consistent with state and federal standards, unless otherwise determined during the tribal consultation process. In the event that a prehistoric and/or historic deposit is encountered during construction monitoring, a Collections Management Plan would be required in accordance with the project MMRP. The disposition of human remains and burial related artifacts that cannot be avoided or are inadvertently discovered is governed by state (i.e., Assembly Bill 2641_[Coto] and California Native American Graves Protection and Repatriation Act of 2001 [Health and Safety Code 8010-8011]) and federal (i.e., Native American Graves Protection and Repatriation Act [U.S.C. 3001-<u>30131</u>) law, and must be treated in a dignified and culturally appropriate manner with respect for the deceased individual(s) and their descendants. Any human bones and associated grave goods of Native American origin shall be turned over to the appropriate Native American group for repatriation.

Arrangements for long-term curation of all recovered artifacts must be established between the applicant/property owner and the consultant prior to the initiation of the field reconnaissance. When tribal cultural resources are present, or non-burial-related artifacts associated with tribal cultural resources area suspected to be recovered, the treatment and disposition of such resources will be determined during the tribal consultation process. This information must then be included in the archaeological survey, testing, and/or data recovery report submitted to the City for review and approval. Curation must be accomplished in accordance with the California State Historic Resources Commission's Guidelines for the Curation of Archaeological Collection (dated May 7, 1993) and, if federal funding is involved, <u>Title 36 of the</u> Code of Federal Regulations, <u>Part 79 of the Federal Register</u>. Additional information regarding curation is provided in Section II of the Guidelines.

6.7.7 Significance of Impacts after Mitigation

6.7.7.1 Historic Structures, Objects or Sites

Development implemented in accordance with the proposed Uptown CPU and associated discretionary actions that would potentially result in impacts to significant historical resources would be required to incorporate feasible mitigation measures adopted in conjunction with the certification of this PEIR and consistent with existing requirements of the Historic Resources Regulations and Historic Resources Guidelines. The mitigation framework combined with the proposed Uptown CPU policies promoting the identification and preservation of historical resources would reduce the program-level impact related to historical resources of the built environment. However, even with implementation of the mitigation framework, the degree of future impacts and applicability, feasibility, and success of future mitigation measures cannot be adequately known for each specific future project at this program level of analysis.

With respect to Potential Historic Districts, while supplemental development regulations would provide some protections, until such time as the Potential Historic Districts are intensively surveyed, verified, and brought forward for designation consistent with City regulations and procedures, potential impacts to the Potential Historic Districts would remain significant and unavoidable. Thus, potential impacts to historical resources including historic structures, objects or sites and historic districts would be significant and unavoidable.

6.7.7.2 Prehistoric Resources, Sacred Sites, and Human Remains

Development implemented in accordance with the proposed Uptown CPU and associated discretionary actions would potentially result in impacts to significant archaeological and tribal cultural resources, and therefore would be required to implement mitigation measure HIST-6.7-2, which addresses measures to minimize impacts to archaeological and tribal cultural resources. This mitigation, combined with the policies of the General Plan and proposed Uptown CPU policies promoting the identification, protection, and preservation of archaeological resources, in addition to compliance with CEQA and Public Resources Code Section 21080.3.1 requiring tribal consultation early in the development review process, and the City's Historic Resources Regulations (San Diego Municipal Code Section 143.0212), which requires review of ministerial and discretionary permit applications for any parcel identified as sensitive on the Historical Resources Sensitivity Maps would reduce the program-level impact related to prehistoric or historical archaeological resources and tribal cultural resources. However, even with application of the existing regulatory framework and mitigation framework, the feasibility and efficacy of mitigation measures cannot be determined at this program level of analysis. Thus, impacts to prehistoric resources, sacred sites, and human remains would be minimized but not to below a level of significance.

6.8 **Biological Resources**

A Biological Resources Report for the Uptown, North Park, and Golden Hill Community Plan Updates (CPUs) was prepared by RECON March 2, 2016. That analysis addresses biological impacts associated with the proposed Uptown CPU and associated discretionary actions. The entire report is included as Appendix H to this draft Program Environmental Impact Report (PEIR) and forms the basis for the discussion in this section.

6.8.1 Existing Conditions

The existing environmental setting which includes discussion and description of the sensitive biological resources and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

A general description of vegetation communities and land cover types mapped within the Uptown CPU area is described in Section 2.3.8. The specific vegetation communities/land cover types that occur within the Uptown community are shown in Figure 6.8-1. Table 6.8-1 lists acreages per vegetation community/land cover type.

Table 6.8-1 Vegetation Communities and Land Cover Types within the Uptown CPU Area						
	Multiple Species					
Vegetation Community/Land Cover	Conservation					
Туре	Program Tier	Acreage				
Coastal sage scrub	Ш	154.8				
Chaparral	=	136.8				
Grassland	III-B	36.1				
Riparian scrub	n/a	3.3				
Eucalyptus woodland	IV	3.8				
Disturbed land	IV	107.0				
Urban/developed	IV	2,215.3				
TOTAL		2,657.1				



Existing Vegetation Communities and Land Cover Types

6.8.2 Methodology

Data on vegetation, Multi Habitat Planning Area (MHPA) boundary corrections, and open space were provided by the City of San Diego. The analysis of biological resources for the Uptown CPU area was performed at the plan-level using the existing base date files and other available data provided by the City of San Diego. Data from the California Natural Diversity Data Base (CNDDB) was used to provide information on potential sensitive plant and wildlife species occurrences. Additional geographical information system (GIS) data were used to provide more detailed information on areas of potential effect within the Uptown CPU area. These additional data included the location of individual private lots that helped identify areas where brush management could occur in the future.

6.8.2.1 Vegetation Communities

The base vegetation community mapping was taken primarily from the San Diego Association of Governments (1995) digital file for the Multiple Species Conservation Program (MSCP). This vegetation mapping was updated using information from an aerial photograph of the area (SanGIS 2012). Field work was conducted to verify the type of vegetation occurring in specific areas within the Uptown CPU boundaries where there were questions about existing vegetation mapped. In particular, some individual lots identified as potentially having greater than one-tenth of an acre of native vegetation where corrections to the MHPA boundary are proposed were field checked.

Vegetation community classifications follow Holland (1986) and Oberbauer (1996). Assessments of the sensitivity of habitats are based primarily on the California Native Plant Society, the CNDDB, City of San Diego, U.S. Fish and Wildlife Service (USFWS), and Holland.

6.8.2.2 Sensitive Plants

The locations of sensitive plant species evaluated are from the CNDDB. Nomenclature for plant species follows the Jepson Online Interchange and assessments of the sensitivity of species are based primarily on California Native Plant Society (CNPS), State of California, City of San Diego, and USFWS.

6.8.2.3 Sensitive Wildlife

The locations of sensitive wildlife species evaluated are from the CNDDB. Zoological nomenclature for birds is in accordance with the American Ornithologists' Union Checklist (2013) and Unitt (2004), for mammals with Jones et al. (1997), for amphibians and reptiles with Crother (2008), and for butterflies with Brown et al. (1992). Assessments of the sensitivity of species are based primarily on State of California and USFWS.

6.8.3 Significance Determination Thresholds

Based on the City's *CEQA Significance Determination Thresholds* (2011), which have been adapted to guide a programmatic analysis for the Uptown CPU and associated discretionary actions, impacts on biological resources would be significant if the project would result in:

- A substantial adverse impact, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in the MSCP or other local or regional plans, policies or regulations, or by the California Department of Fish and Wildlife (CDFW) or USFWS;
- A substantial adverse impact on any Tier I Habitats, Tier II Habitats, Tier IIIA Habitats, or Tier IIIB Habitats as identified in the Biology Guidelines of the Land Development Manual or other sensitive natural community identified in local or regional plans, policies, regulations, or by the CDFW or USFWS;
- 3) A substantial adverse impact on wetlands (including, but not limited to, marsh, vernal pool, riparian, etc.) through direct removal, filling, hydrological interruption, or other means;
- 4) Interfering substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, including linkages identified in the MSCP Plan, or impede the use of native wildlife nursery sites; or
- 5) A conflict with the provisions of an adopted Habitat Conservation Plan, Natural Conservation Community Plan, or other approved local, regional, or State habitat conservation plan, either within the MSCP plan area or in the surrounding region;

6.8.4 Impact Analysis

Issue 1 Sensitive Wildlife Species

Would the project result in a substantial adverse impact, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in the MSCP or other local or regional plans, policies, or regulations, or by the California Department of Fish and Wildlife (CDFW) or U.S. Fish and Wildlife Service (USFWS)?

The proposed Uptown CPU presents goals and policies for biological resources in the Conservation Element. The purpose of the Conservation Element is to provide for the long-term conservation and sustainable management of natural resources. As part of the proposed Uptown CPU, areas designated as open space were reconfigured to remove areas of existing development to better correlate with the actual location of sensitive biological resources intended for conservation. The open space boundary was reconfigured consistent with the General Plan Land Use and Community Planning Element policies for designation of open space, and the General Plan and Community Plan Conservation Element policies regarding the protection of natural habitats and rare plants and animals. The locations of designated open space areas for the Uptown CPU area are shown on Figure 6.8-2, and acreages summarized by habitat are shown in Table 6.8-2. By locating all remaining sensitive natural resources within the Uptown CPU area within the open space designation and/or MHPA, impacts to sensitive species would be minimized. See also the discussion under Issue 5, which discusses the proposed MHPA corrections.







FIGURE 6.8-2

Location of Open Space

Proposed Open Space

Table 6.8-2 Proposed Open Space for Uptown					
Vegetation Community/Land Cover Type	Open Space				
Coastal Sage Scrub	154.8				
Chaparral	136.8				
Grassland	36.0				
Riparian Scrub	3.3				
Eucalyptus Woodland	3.8				
Disturbed Land	105.6				
Developed	3.3				
TOTAL	443.6				

There is a small potential that wildlife would be displaced and some small mammals, amphibians, and reptiles with low mobility may be inadvertently harmed during future project activities (e.g., Brush Management Zone 1 or re-development of a lot). However, any impacts to these wildlife species would be less than significant, as these common wildlife species are not considered sensitive by the City. As detailed in Section 2.3.8.4 of this PEIR, the sensitive wildlife species that may occur in the Uptown CPU area are the coastal cactus wren and Mexican long-tongued bat. However, implementation of the proposed Uptown CPU and associated discretionary actions would have a low potential to result in impacts to either the coastal cactus wren or Mexican long-tongued bat because they would be located within the canyon portions of the Uptown CPU area within areas of undisturbed native habitats. The coastal cactus wren occupies coastal sage scrub with opuntia thickets, which may be present within the Uptown CPU area. However, opuntia thickets large enough to support cactus wren are not present along the rims of the canyons at the urban interface. In the case of Mexican long-tongued bat, suitable caves or mines are not present along the rims of the canyons at the urban interface. There would be no development potential within the canyon areas due to the open space designation and/or MHPA designations. Potentially occurring sensitive species would be conserved in accordance with Environmentally Sensitive Lands (ESL) regulations, the City's Biology Guidelines, and the provisions of the MSCP Subarea Plan. Thus, impacts to sensitive species resulting from build-out of the Uptown CPU area would be less than significant.

Issue 2 Sensitive Habitats

Would the project result in a substantial adverse impact on any Tier I Habitats, Tier II Habitats, Tier IIIA Habitats, or Tier IIIB Habitats, as identified in the Biology Guidelines of the Land Development manual, or other sensitive natural community identified in local or regional plans, policies, regulations, or by the CDFW or USFWS?

a. Sensitive Vegetation Communities

As detailed in Chapter 2.0 Environmental Setting, the Uptown CPU area has sensitive vegetation communities (Tier II – coastal sage scrub, chaparral; Tier IIIB – grassland; wetland-riparian scrub) primarily within the canyons and some native upland habitat remnants along the canyon rims. The remainder of the Uptown CPU area is built out and supports very few sensitive vegetation communities. Implementation of the proposed Uptown CPU and associated discretionary actions would impact primarily disturbed land and urban/developed land, which are not considered sensitive vegetation communities.

A relatively small acreage of sensitive vegetation is currently located outside of the MHPA or designated open space. It occurs along the edges of the canyons and within areas that could be subject to Brush Management Zone 1 clearing or re-development of a parcel or its existing structures. Potential impacts to sensitive vegetation communities could include the loss of coastal sage scrub and chaparral habitat (Figure 6.8-3). However, the plan-level analysis showed that these potential impacts would occur over numerous individual private lots and impacts on any single lot would not exceed the 0.10-acre significance threshold contained in the City's significance guidelines; therefore, these potential impacts would not be considered significant. Furthermore, all projects with sensitive biological resources would require subsequent environmental review under the City of San Diego ESL regulations prior to disturbance of those lands. Additionally, these small losses would not significantly affect the regional distribution of affected vegetation communities. Implementation of the proposed Uptown CPU policies and future compliance with established development standards contained in the City's ESL Regulations and Biology Guidelines as well as the MSCP Subarea Plan and Land Use Adjacency Guidelines would ensure that impacts to sensitive vegetation communities remain below a level of significance.

b. Sensitive Plants

Implementation of the proposed Uptown CPU and associated discretionary actions have a low potential to impact any of the five sensitive plant species previously recorded in the Uptown CPU area (refer to Figure 6.8-3). Sensitive species documented within the CPU area include San Diego barrel cactus, variegated dudleya, Nuttall's scrub oak, San Diego thornmint, and San Diego goldenstar. San Diego thornmint is federally listed as threatened and state listed as endangered. It is considered a narrow endemic under the MSCP, has a CNPS Rare Plant Ranking of 1B.1 and it can be found in friable or cracked clay soil in grassy openings within chaparral and coastal scrub. San Diego barrel cactus (a covered species under the MSCP and has a CNPS Rare Plant Ranking of 2B.1), variegated dudleya (a narrow endemic species under the MSCP and has a CNPS Rare Plant Ranking of 1B.2), and San Diego goldenstar (a MSCP-covered species and has a CNPS Rare Plant Ranking of







FIGURE 6.8-3

()

1,800

Feet

Location of Sensitive Biological Resource Impacts

1B.1) all occur within coastal sage scrub and chaparral habitats. Nuttall's scrub oak is not covered under the MSCP, but is considered rare and has a CNPS Rare Plant Ranking of 1B.1. This species occurs within coastal sage scrub and chaparral vegetation. As described previously, implementation of the proposed Uptown CPU and associated discretionary actions would result in land use changes that would affect primarily developed areas. The potential for sensitive plant species to still occur is low due to the extent of development that has taken place within the Uptown CPU area and along the urban-canyon interface. Alhough focused surveys for sensitive plant species were not conducted in support of the proposed Uptown CPU consistent with a program analysis, it is anticipated that these species, if they occur, would be located within the canyon portions of the Uptown CPU area.

As described previously, future build-out of the proposed Uptown CPU and associated discretionary actions could impact a relatively small acreage of sensitive vegetation that is outside the MHPA or designated open space that occurs along the edges of the canyons and within areas that could be subject to Brush Management Zone 1 clearing or re-development of a parcel or its existing structures. These areas potentially support very small areas of native habitat (less than 0.10 acre per lot) with a low potential for sensitive plant species to occur. Thus, the implementation of the proposed Uptown CPU and associated discretionary actions is not anticipated to result in impacts to sensitive plant species. Furthermore, implementation of the Uptown CPU and associated discretionary actions of sensitive plant species. In addition, because the area is already highly developed, it is anticipated that only small populations of sensitive plant species. Thus, impacts to sensitive plant significantly impact any regional populations of sensitive plant species. Thus, impacts to sensitive plant species and discretionary actions would not significantly impact any regional populations of sensitive plant species. Thus, impacts to sensitive plans resulting from build-out of the proposed Uptown CPU and associated discretionary actions would be less than significant.

Issue 3 Wetlands

Would the project result in a substantial adverse impact on wetlands (including, but not limited to, marsh, vernal pool, riparian, etc.) through direct removal, filling, hydrological interruption, or other means?

Wetland habitats in the Uptown CPU area consist of riparian scrub. Riparian scrub habitat is located in a canyon bottom within the north-central portion of Uptown CPU area (see Figure 6.8-3). This habitat is within canyon areas that would remain within open space and/or the MHPA and would be further protected from disturbance through the City ESL regulations. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in in less than significant impacts to wetland habitats including riparian scrub.

Issue 4 Wildlife Corridors and Nursery Sites

Would the project interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, including linkages identified in the MSCP Plan, or impede the use of native wildlife nursery sites?

Within the Uptown CPU area, canyons provide for local wildlife movement for birds and small mammals. However, these canyons are isolated by development and are not part of a major wildlife corridor system. Nonetheless, the canyons serve as a stepping-stone for wildlife species movement between other local canyon systems and into major off-site habitat areas. The proposed Uptown CPU would designate canyon areas as open space which would provide protections from future development. The MHPA designation for canyon areas further protects canyon areas from development. The project includes MHPA boundary line corrections to add habitat to the MHPA areas and remove developed areas from the MHPA as described below under Issue 5. These changes would increase the amount of protected open space in canyons, which would be beneficial for wildlife movement in canyon areas. Thus, no impact to wildlife corridors would occur.

Implementation of future projects consistent with the proposed Uptown CPU and associated discretionary actions has the potential to result in direct impacts to migratory or nesting birds. As discussed in Chapter 2.0, Section 2.3.8.4 of this PEIR, there is low potential for occurrence of sensitive bird species. However, where future development areas contain trees or are located adjacent to trees that could serve as nesting habitat for migratory birds, there is a potential for adverse impacts to wildlife nursery sites if construction occurs during the typical bird breeding season (February 1 to September 15).

The Migratory Bird Treaty Act (MBTA), which is enforced by the USFWS, makes it unlawful "by any means or in any manner, to pursue, hunt, take, capture, [or] kill" any migratory bird or attempt such actions, except as permitted by regulation. Thus, there is an existing regulatory framework in place to prevent adverse impacts to migratory birds. Additionally, future discretionary development occurring within the Uptown CPU area that has the potential to impact migratory birds would be required to conduct pre-construction surveys if construction occurs during the typical bird breeding season to determine the presence or absence of breeding birds and to ensure that no impacts occur to any nesting birds or their eggs, chicks, or nests. Within the Uptown CPU area, development adjacent to the MHPA would be subject to additional protections that would avoid impacts to wildlife nursery sites in adjacent habitat areas as detailed further under Issue 5 below. Thus, with the existing regulatory framework in place, potential impacts to wildlife nursery sites would be less than significant.

Issue 5 Multiple Species Conservation Program

Would the project result in a conflict with the provisions of an adopted Habitat Conservation Plan, Natural Conservation Community Plan, or other approved local, regional, or State habitat conservation plan, either within the MSCP plan area or in the surrounding region?

As designated in the City's MSCP Subarea Plan, the MHPA is the permanent preserve area for habitat conservation. There are no remaining lands within the MHPA that have not already been preserved

as open space within this CPU area. All projects with sensitive biological resources would require subsequent environmental review under the City of San Diego ESL regulations.

A comprehensive communitywide MHPA boundary line correction is proposed. The MHPA boundary line correction was considered in coordination with the Wildlife Agencies and is consistent with the goals of the MSCP to conserve biological resources and to exclude legally developed and required uses (i.e., structures, streets, brush management zone 1). As shown in Table 6.8-3, the comprehensive MHPA boundary correction for the Uptown CPU area would result in a net addition of 28.8 acres to the MHPA. Preservation of sensitive habitat is consistent with the goals of the MSCP, the Conservation Element for the Community Plan, and the City's ESL regulations. The MHPA correction removes existing development (i.e., structures and streets), as well as the 35-foot Brush Management Zone 1 area, as required in accordance with the City's Land Development Code, Section 142.0412.

Table 6.8-3 Modifications to Versitation Communities and Lond Course Types of a Booult of the MUDA								
Boundary Line Correction at Uptown (acres)								
	Existing							
Vegetation Community/	Acreage in	MHPA	MHPA	Change in	Total Acreage			
Land Cover Type	MHPA	Addition	Deletion	MHPA	in MHPA			
Coastal sage scrub	154.8	30.7	1.0	+29.7	184.5			
Chaparral	136.8	35.8	2.4	+33.4	170.2			
Grassland	36.1	4.5	0	+4.5	40.6			
Riparian scrub	3.3	0.6	0	+0.6	3.9			
Eucalyptus woodland	3.8	0.6	0.7	-0.1	3.7			
Disturbed land	107.0	4.8	3.5	+1.4	108.4			
Developed	2,215.3	0	40.7	-40.7	2,174.6			
TOTAL	2,657.1	77.1	48.3	+28.8	2,685.9			

As shown in Figure 6.8-4, a majority of the corrections remove developed and disturbed land while adding sensitive habitats, which include coastal sage scrub, chaparral, grasslands, and riparian scrub. City-owned lands within designated community plan open space areas adjacent to the existing MHPA have also been added to the MHPA. In a few cases, sensitive habitat located within designated Community Plan open space on private land was added to the MHPA in order to expand the local wildlife corridor and increase the viability and connectivity of sensitive habitat within the existing MHPA. Regardless of the MHPA boundary line correction, these addition areas are regulated through ESL (see below) for sensitive biological resources and steep slopes. The MPHA boundary line correction does not add or increase any regulations associated with City projects, such as sewer line repairs within the canyons. These projects would continue to be conducted in accordance with the Canyon Sewer Cleaning Program (LDR No. 6020), Council Policies 400-13 and 400-14, and Community Plan policies related to this program. Correcting the MHPA boundary also does not relieve projects from having to otherwise comply with the City's MHPA Land Use Adjacency Guidelines, described below. The MHPA correction results in an overall benefit to the MHPA and is consistent with the goals and policies of the MSCP and the proposed Uptown CPU.

Image source: SanGIS (flown May 2012)









Location of MHPA Boundary Line Correction

Development adjacent to MHPA lands would be subject to the City's MHPA Land Use Adjacency Guidelines, which address indirect effects on the MHPA from adjacent development. Indirect effects can occur wherever development and human activity are adjacent to natural areas. These effects include those due to increased runoff, trampling, and removal of plant cover due to hiking, biking and other human activities, increased presence of toxins, increased nighttime light levels, and redirection or blockage of wildlife movement, and increased levels of non-native and invasive plants. These indirect effects could reduce the quality of the MHPA. However, the City's Land Use Adjacency Guidelines require certain measures to be incorporated in the design of projects adjacent to the MHPA to reduce indirect impacts to a level that is less than significant.

Future development proposals located adjacent to the MHPA would be required to address potential indirect impacts through compliance with the City's MHPA Land Use Adjacency Guidelines. Projects adjacent to the MHPA would incorporate features into the project and/or permit conditions that demonstrate compliance with the MHPA Land Use Adjacency Guidelines. Adherence to these guidelines would avoid any future significant indirect impacts.

The City's Land Use Adjacency Guidelines of the MSCP address requirements for grading and land development: drainage; toxic substances in runoff; lighting, barriers, invasive plant species, brush management; and noise. Furthermore, proposed policies in the Conservation Element of the Uptown CPU would support existing protections for MHPA lands. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would not result in a conflict with the provisions of an adopted Habitat Conservation Plan, Natural Conservation Community Plan, or other approved local, regional, or state habitat conservation plan or local policy protecting biological resources. Impacts would be less than significant.

Cumulative Impacts

Preservation of the region's biological resources has been addressed through the implementation of regional habitat conservation plans. Impacts to biological resources in the City of San Diego are managed through the adopted MSCP Subarea Plan, which is incorporated by reference in the City's adopted General Plan.

As discussed above, the Uptown CPU area currently supports a number of sensitive resources including coastal sage scrub, chaparral, wetlands, grassland, and sensitive plans and wildlife. Surrounding communities such as Golden Hill and North Park contain similar resources that are limited to canyon areas. However, these resources located in canyon areas are protected through the proposed open space designation and/or their location within MHPA in addition to protections provided by the City's ESL regulations. The proposed Uptown CPU and CPUs of surrounding communities incorporate policies related to the protection of biological resources focusing primarily on the CPUs' consistency with the City's ESL Regulations, the Biology Guidelines, and MSCP Subarea Plan Management Policies to protect the area's sensitive plants and animals.

Cumulative development that would occur within the Uptown CPU area combined with development within surrounding communities including the North Park and Golden Hill CPU areas would result in less than significant cumulative impacts to biological resources due to the developed nature of these communities combined with the existing regulatory framework that would ensure

that impacts to sensitive biological resources are avoided. Although each individual future project may contribute to incremental biological resource impacts, compliance with proposed CPU policies, the MSCP Subarea Plan, ESL Regulations, and the Biology Guidelines would ensure that cumulative impacts from future development would be less than significant.

6.8.5 Significance of Impacts

6.8.5.1 Sensitive Wildlife Species

Implementation of the proposed Uptown CPU and associated discretionary actions would result in land use changes that would affect primarily developed areas. Thus impacts to sensitive species would not be anticipated to occur since any sensitive species that could occur within the Uptown CPU area are likely to occupy canyon bottoms and would not be subject to development due to their designation as Open Space and/or MHPA. Additionally, any impact to sensitive vegetation communities would be subject to the City's ESL regulations, which would ensure any impacts to vegetation communities and potential sensitive species that may occupy those communities would addressed. Thus, based on the lack of sensitive species anticipated to occur in the developable areas of the CPU area in addition to the regulatory framework in place that protects sensitive species, impacts to wildlife species would be less than significant and no mitigation would be required.

6.8.5.2 Sensitive Habitats

Implementation of the proposed Uptown CPU and associated discretionary actions has a low potential to impact any of the five sensitive plant species previously recorded in the Uptown CPU area. As described previously, implementation of the proposed Uptown CPU and associated discretionary actions would result in land use changes that would affect primarily developed areas. The potential for sensitive plant species to still occur is low due to the extent of development that has taken place within the CPU area and along the urban-canyon interface. Impacts to sensitive plant species would be less than significant and no mitigation would be required.

6.8.5.3 Wetlands

Implementation of the proposed Uptown CPU and associated discretionary actions would not result in impacts to wetlands (riparian scrub), as area where this habitat occurs would remain within open space and/or the MHPA. No impacts to riparian scrub are expected; therefore, impacts would be less than significant and no mitigation would be required.

6.8.5.4 Wildlife Corridors and Nursery Sites

The proposed MHPA boundary line corrections would increase the amount of protected open space in canyons, which would be beneficial for wildlife movement in canyon areas. Thus, no impact to wildlife corridors would occur.

Impacts to wildlife nursery sites, particularly migratory birds, would be avoided through compliance with the MBTA in addition to compliance with protections afforded to lands within and adjacent to

MHPA lands. Development on lands adjacent to MHPA lands would be required to avoid impacts to wildlife nursery sites in adjacent habitat areas as detailed further under Issue 5. Thus, with the existing regulatory framework in place, potential impacts to wildlife nursery sites would be less than significant.

6.8.5.5 Multiple Species Conservation Program

The proposed Uptown CPU and associated discretionary actions would be consistent with the City's MHPA Land Use Adjacency Guidelines and Municipal Code (Section 142.0740) requirements relative to lighting adjacent to the MHPA. Additionally, in complying with the MHPA Land Use Adjacency Guidelines requirements, landscape plans for future projects would require that grading would not impact environmental sensitive land, that potential runoff would not drain into MHPA land, that toxic materials used on a development do not impact adjacency sensitive land, that development includes barriers that would reduce predation by domestic animals, and that landscaping does not contain exotic plants/invasive species. In addition, the MHPA Land Use Adjacency Guidelines directs development so that any brush management activities are minimized within the MHPA and contains requirements to reduce potential noise impacts to listed avian species. Compliance with the City's MHPA Land Adjacency Guidelines and adherence to the policies in the Conservation Element of the Uptown CPU would reduce potential impacts of the proposed Uptown CPU and associated discretionary actions to less than significant.

6.8.6 Mitigation Framework

All biological resources impacts would be less than significant; thus, no mitigation measures are required.

6.9 Geologic Conditions

GEOCON Inc. prepared the Program EIR-Level Geotechnical Report – Uptown, North Park, and Golden Hill Planning Areas (June 10, 2015). That analysis addresses geotechnical impacts associated with the three proposed Community Plan Updates (CPUs) including the proposed Uptown CPU and associated discretionary actions. The Geotechnical Report is included as Appendix I to this Program Environmental Impact Report (PEIR). This section presents a summary of the findings made in the report and the associated analysis of potential impacts.

6.9.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

Soil and geologic conditions are described in detail in Section 2.3.9 of this PEIR. In summary, the area of the Uptown CPU area is underlain by four surficial soil deposits and three geologic formations. The surficial soils include artificial fill (unmapped), topsoil/colluvium, alluvium (unmapped), and very old terrace deposits (formerly Lindavista Formation). The geologic formations include San Diego Formation, Pomerado Conglomerate, and Mission Valley Formation. Figure 6.9-1 illustrates the location of the geologic formations located within Uptown.

The 2008 City of San Diego Seismic Safety Study, Geologic Hazards and Faults, maps the Uptown CPU area as low risk (Geologic Hazard Category 52), the northern boundary is mapped as low to moderate risk (Geologic Hazard Category 53), and the south end is mapped within the downtown special fault zone, Geologic Hazard Category 13. Figure 6.9-2 shows the Uptown Community Plan area boundary superimposed on the 2008 City of San Diego Seismic Safety Study. Figure 6.9-3 provides a map of geologic hazards for the Uptown CPU area as identified in the Geotechnical Report (Appendix I)



FIGURE 6.9-1 Geology – Uptown





FIGURE 6.9-2 City of San Diego Geologic Hazards and Faults – Uptown



FIGURE 6.9-3 Geologic Hazards – Uptown

6.9.2 Significance Determination Thresholds

Thresholds used to evaluate potential impacts to air quality are based on applicable criteria in the California Environmental Quality Act (CEQA) Guidelines Appendix G and the City of San Diego's CEQA Significance Determination Thresholds (2011). Thresholds are modified from the City's CEQA Significance Determination Thresholds to reflect the programmatic analysis for the proposed Uptown CPU and associated discretionary actions. For impacts related to geologic conditions, a significant impact could occur if implementation of the proposed Uptown CPU and associated discretionary actions of the proposed Uptown CPU and associated discretionary actions for the proposed Uptown CPU and associated discretionary actions would:

- 1) Expose people or structures to potential substantial adverse effects, including the risk of loss, injury, or death involving:
 - Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault,
 - Strong seismic ground shaking,
 - Seismic-related ground failure, including liquefaction,
 - Landslides;
- 2) Result in substantial soil erosion or the loss of topsoil;
- 3) Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse; or
- 4) Be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risks to life or property.

This section does not include analysis related to the capacity of soils to support septic tanks or alternative waste water disposal systems, since sewers are available throughout the Uptown CPU area.
6.9.3 Impact Analysis

Issue 1 Seismic Hazards

Would the proposed project expose people or structures to potential substantial adverse effects, including the risk of loss, injury, or death involving rupture of a known earthquake fault, strong seismic ground shaking, seismic-related ground failure, including liquefaction, or landslides?

Future development associated with the implementation of the proposed Uptown CPU and associated discretionary actions could result in the exposure of more people, structures, and infrastructure to seismic hazards.

The nearest known active fault is the Rose Canyon Fault Zone, which is identified in the GEOCON report as separate from the Newport–Inglewood/Rose Canyon Connected Fault (Table 6.9-1). Both are located approximately one mile to the west of the Community Plan area. Major earthquakes occurring on the Rose Canyon Fault Zone, or other regional active faults located in the southern California area, could subject the site to moderate to severe ground shaking. The Uptown CPU area is located on the east margin of the Rose Canyon Fault Zone (RCFZ) with two faults traversing the area (Figure 6.9-2). These are identified as the Old Town and Mission Bay fault segments of the RCFZ. The *City of San Diego Seismic Safety Study, Geologic Hazards and Faults* (2008) Grid Tiles 20 and 21 describes the Old Town and Mission Bay fault segments of the RCFZ as "potentially active, inactive, presumed inactive, or activity unknown".

A geotechnical investigation that specifically addresses surface fault-rupture hazard is required for proposed projects located in the fault buffer zones. The southern portion of the Uptown CPU area south of Laurel Street is located within the City of San Diego Downtown Special Study Zone. Permitting of projects within the Downtown Special Study Zone requires that a site-specific fault investigation be performed.

The Uptown CPU area will be subjected to hazards caused by ground shaking during seismic events on regional active faults. According to the computer program EZ-FRISK (Version 7.62), six known active faults are located within a search radius of 50 miles from the Uptown CPU area. The nearest known active faults are the Newport-Inglewood/Rose Canyon Connected Fault and Rose Canyon Fault (see Table 6.9-1), located approximately one mile west of the site, and are the dominant source of potential ground motion. Table 6.9-1, lists the estimated maximum earthquake magnitude and peak ground acceleration for faults in relationship to the Uptown CPU area.

As part of the geotechnical update, it was determined that the Uptown CPU area could be subject to moderate to severe ground shaking in the event of an earthquake along any of the faults listed in Table 6.9-1 or other faults in the Southern California/Northern Baja California region.

Table 6.9-1 Deterministic Seismic Hazard Parameters – Uptown Community						
			Peak (Ground Accele	ration	
			Boore	Campbell	Chiou	
		Maximum	and	and	and	
	Distance	Earthquake	Atkinson	Bozorgnia	Youngs	
	from Site	Magnitude	2008	2008	2008	
Fault Name	(miles)	(Mw)	(g)	(g)	(g)	
Newport-Inglewood/Rose Canyon Connected	0.9	7.5	0.49	0.41	0.56	
Rose Canyon	0.9	6.9	0.47	0.41	0.53	
Coronado Bank	13	7.4	0.23	0.18	0.22	
Palos Verde/Coronado Bank Connected	13	7.7	0.25	0.19	0.25	
Elsinore	41	7.85	0.14	0.09	0.12	
Earthquake Valley	45	6.8	0.08	0.06	0.05	

The computer program EZ-FRISK was used to perform a probabilistic seismic hazard analysis. The computer program EZ-FRISK operates under the assumption that the occurrence rate of earthquakes on each mapped Quaternary fault is proportional to the faults slip rate. The program accounts for earthquake magnitude as a function of fault length, and site acceleration estimates are made using the earthquake magnitude and distance from the site to the rupture zone. The program also accounts for uncertainty in each of following: (1) earthquake magnitude, (2) rupture length for a given magnitude, (3) location of the rupture zone, (4) maximum possible magnitude of a given earthquake, and (5) acceleration at the site from a given earthquake along each fault. By calculating the expected accelerations from considered earthquake sources, the program calculates the total average annual expected number of occurrences of site acceleration greater than a specified value.

Table 6.9-2 presents the site-specific probabilistic seismic hazard parameters including accelerationattenuation relationships and the probability of exceedance.

Table 6.9-2 Probabilistic Seismic Hazard Parameters – Uptown Community							
Peak Ground Acceleration							
	Boore	Campbell	Chiou				
	and	and	and				
	Atkinson	Bozorgnia	Youngs				
	2008	2008	2008				
Probability of Exceedance	(g)	(g)	(g)				
2% in a 50-Year Period	0.60	0.51	0.64				
5% in a 50-Year Period	0.40	0.35	0.41				
10% in a 50-Year Period	0.27	0.24	0.26				

While listing peak accelerations is useful for comparison of potential effects of fault activity in a region, other considerations are important in seismic design, including frequency and duration of motion and soil conditions underlying the site.

Severe ground shaking is most likely to occur during an earthquake on one of the regional active faults in the area. The Newport-Inglewood/Rose Canyon Connected faults, located to the northwest, is the active fault considered having the most significant effect from a design standpoint due to the close proximity. Based on a deterministic analysis, a maximum credible earthquake of moment magnitude M7.5 on the Newport-Inglewood/Rose Canyon Connected fault could produce an estimated peak horizontal ground acceleration of 0.56g within the proposed Uptown CPU area. Based on this analysis, damage from earthquake ground shaking could occur. Structural design in accordance with the current Building Code is intended to reduce the impact of earthquake shaking on buildings to an acceptable level of risk. Seismic design of future structures would be evaluated in accordance with the 2013 California Building Code (CBC) guidelines or those currently adopted by the City of San Diego. Design in accordance with the CBC would reduce potentially significant impacts to future structures from strong seismic ground shaking to a less than significant level. San Diego Municipal Code (SDMC) Section 145.1803(a)(2) indicates that no building permit shall be issued for construction where the geotechnical investigation report establishes that construction of buildings or structures would be unsafe because of the geologic hazards. All new development and redevelopment would be required to comply with the SDMC and the CBC, which include design criteria for seismic loading and other geologic hazards and require that a geotechnical investigation be conducted for all new structures, additions to existing structures, or whenever the occupancy classification of a building changes to a higher relative hazard category (SDMC Section 145.1803).

Liquefaction or seismically induced settlement typically occurs when a site is located in a zone with seismic activity; on-site soils are relatively cohesionless with relative densities less than about 70 percent, and groundwater within 50 feet of the surface. If these criteria are met, a seismic event could result in soil liquefaction. The potential for liquefaction and seismically induced settlement occurring for the mesa top areas is very low due to the very dense cemented condition of the geologic formations and lack of groundwater. Building construction in accordance with the SDMC and CBC will reduce this potential hazard to an acceptable level of risk. Thus, while the Uptown CPU area would be subject to seismic events, potential hazards associated with ground shaking and seismically induced hazards such as ground failure, liquefaction, or landslides would be reduced to a less than significant level through implementation of site-specific geotechnical report recommendations associated with future development within the Uptown CPU area.

Issue 2 Erosion or Loss of Topsoil

Would the project result in a substantial erosion or loss of topsoil?

The Uptown CPU area consists of developed and previously graded land and undeveloped land predominantly in the form of canyons and other open space areas. Implementation of the proposed Uptown CPU and associated discretionary actions would allow for the intensification of some land uses that could lead to construction and grading activities that could temporarily expose topsoil and increase soil erosion from water and wind. Development of parcels within the proposed Uptown CPU area could remove the existing pavement and cover, thereby exposing soils to potential runoff and erosion during construction if protective measures are not taken.

SDMC Section 142.0146 requires grading work to incorporate erosion and siltation control measures in accordance with Chapter 14, Article 2, Division 4 (Landscape Regulations) and the standards established in the Land Development Manual. The regulations prohibit sediment and pollutants from leaving the work site and requires the property owner to implement and maintain temporary and permanent erosion, sedimentation, and water pollution control measures. Controls shall include measures outlined in Chapter 14, Article 2, Division 2 Storm Water Runoff Control and Drainage Regulations that address the development's potential erosion and sedimentation impacts.

Conformance to such mandated City grading requirements would ensure that proposed grading and construction operations would avoid significant soil erosion impacts. Furthermore, any development involving clearing, grading, or excavation that causes soil disturbance of one or more acres, or any project involving less than one acre that is part of a larger development plan, is subject to National Pollutant Discharge Elimination System General Construction Storm Water Permit provisions. Additionally, any development of significant size within the City would be required to prepare and comply with an approved Storm Water Pollution Prevention Plan that would consider the full range of erosion control Best Management Practices, including any additional site-specific and seasonal conditions. Project compliance with National Pollutant Discharge Elimination System requirements would significantly reduce the potential for substantial erosion or topsoil loss to occur in association with new development. Impacts would be less than significant.

Issue 3 Geologic Instability

Would the project be located on a geologic unit or soil that is unstable or that would become unstable as a result of the project, and potentially result in an on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse?

The majority of the Uptown CPU area is mapped as Geologic Hazard Category 52, characterized as low risk with favorable geologic structure. Other smaller hazard categories are mapped within the CPU area with low to moderate risk. The southern portion of the CPU area is mapped within the downtown special fault zone, Geologic Hazard Category 13. Refer to Figure 6.9-2 for the location of these Hazard Categories.

No large landslides are mapped in the Uptown CPU area; however, small surficial instability could be present on steep slopes. Areas of known and potential, non-conforming slopes (i.e., slopes steeper than 2:1 horizontal to vertical) are shown on Figure 6.9-3. These areas are generally along Interstate 5 and Interstate 8, in Reynard Canyon, Maple Canyon, Arroyo Drive, and Washington Street.

Future projects built in accordance with the proposed Uptown CPU and associated discretionary actions would be required to prepare a geotechnical investigation that specifically addresses slope stability if located on landslide-prone formations or slopes steeper than 25 percent (slope ratio of 4:1 horizontal to vertical) (SDMC Table 145.1803). Additionally, as discussed in the Geotechnical Report, based on the subsurface soil conditions encountered during the field investigation and the lack of groundwater extraction that would be associated with future development, the risk associated with ground subsidence hazard is low. Potential hazards associated with slope instability would be addressed by the site-specific recommendations contained within geotechnical

investigations as required by the CBC and SDMC. Thus, impacts related to landslide and slope instability would be less than significant.

Issue 4 Expansive Soils

Would the project be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial risks to life or property?

Relative to soil expansion, the highly expansive Normal Heights Mudstone is mapped in the northeastern portion of the Uptown CPU area (Figure 6.9-3). Mudstone can be highly expansive and within the Uptown CPU area could range from a few feet thick to approximately 10 feet thick, or greater, in localized areas. The presence of highly expansive materials, especially if near finish proposed grade, is potentially damaging to foundations surface improvements such as sidewalks and pavements. Special measures would be necessary during design and construction to mitigate the effects of expansive soil.

Site-specific measures based on results of a Geotechnical Investigation would be necessary during design and construction of future projects to remedy the effects of expansive soil. A site-specific Geotechnical Investigation required for future projects within the Uptown CPU area would be required by the SDMC to identify the presence of expansive soils and provide recommendations to be implemented during grading and construction to ensure that potential hazards associated with expansive soils are minimized. Thus, with implementation of the recommendations included in site-specific geotechnical investigations required under the CBC and SDMC, potential impacts associated with expansive soils would be less than significant.

Cumulative Impact Analysis

Cumulative impacts related to geologic hazards within the Uptown CPU area and surrounding CPU areas such as North Park and Golden Hill would be less than significant with implementation of recommendations included in site-specific geotechnical investigations required under the CBC and SDMC, as discussed in the previous analysis. Geologic hazards occur from mapped faulting and site-specific soil or geologic conditions. Development of the Uptown CPU in combination with surrounding CPU areas would not compound or worsen potential geologic hazards. Geologic hazard conditions are site-specific and do not compound or increase in combination with projected development elsewhere in the county. Thus, as each individual development would be required to comply with remedial measures identified in a site-specific geotechnical investigation, as required by the SDMC and CBC, cumulative impacts related to geologic hazards would be less than significant.

6.9.4 Significance of Impacts

Based on the Geotechnical Report prepared by GEOCON, Inc., the proposed Uptown CPU and associated discretionary actions would not have direct or indirect significant environmental impacts with respect to geologic hazards because future development would be required to occur in accordance with the SDMC and CBC. This regulatory framework includes a requirement for site-specific geologic investigations to identify potential geologic hazards or concerns that would need to

be addressed during grading and/or construction of a specific development project. Adherence to the SDMC grading regulations and construction requirements and implementation of the recommendations and standards of the City's Geotechnical Study Requirements would preclude significant impacts related to erosion or loss of topsoil. Thus, impacts would be less than significant and no mitigation is required.

6.9.5 Mitigation Framework

Impacts of build-out of the proposed Uptown CPU and associated discretionary actions related to geologic conditions would be less than significant with implementation of existing SDMC requirements for preparation of geotechnical investigations prior to grading and construction and implementation of applicable measures identified in project specific geotechnical investigations. Thus, no mitigation is required.

6.10 Paleontological Resources

The analysis presented in this section evaluates the potential for impacts to paleontological resources based on existing geologic formations that underlay the Uptown Community Plan Update (CPU) area. Refer to Section 6.9, Geologic Conditions, for a discussion of the geologic formations that could be affected by the project (see Figure 6.9-1). The following analysis is based on a review of available literature, including the City's General Plan, Kennedy maps, the City's Paleontological Guidelines, and the publication of Paleontological Resources, County of San Diego by Deméré and Walsh (1994).

6.10.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively. As described in the Chapter 2.0, Environmental Setting (Section 2.3.9, Geology and 2.3.10, Paleontology) of this draft Program Environmental Impact Report (PEIR), the Uptown CPU area is underlain by the San Diego, Pomerado Conglomerate, and Mission Valley Formations, which are assigned high resource sensitivity. Refer to Section 2.3.10 for additional discussion of the existing setting for paleontological resources and sensitivity ratings.

6.10.2 Significance Determination Thresholds

The City of San Diego's California Environmental Quality Act (CEQA) Significance Thresholds provides guidance to determine potential significance to paleontological resources. Based on the City's thresholds, a significant impact related to paleontological resources would occur if the proposed Uptown CPU and associated discretionary actions would:

- 1) Result in development that requires:
 - Over 1,000 cubic yards of excavation in a high resource potential geologic deposit/formation/rock unit.
 - Over 2,000 cubic yards of excavation in a moderate resource potential geologic deposit/formation/rock unit.

The City's CEQA Significance Thresholds includes a Paleontological Determination Matrix to support the City's significance thresholds that is included in Section 2.3.10 of this PEIR. Additionally, the significance thresholds provide the following additional guidance for determining significance:

- If there are sedimentary rocks such as those found in the coastal areas, they usually contain fossils.
- If there are granitic or volcanic rocks such as those found in the inland areas, they usually will not contain fossils.

6.10.3 Impact Analysis

Issue 1 Paleontological Resources

Would the project result in development that requires over 1,000 cubic yards of excavation in a high resource potential geologic deposit/formation/rock unit or over 2,000 cubic yards of excavation in a moderate resource potential geologic deposit/formation/rock unit?

Because human understanding of history is obtained, in part, through the discovery and analysis of paleontological resources, impacts of activities that excavate or grade geologic formations that could contain fossil resources would be significant. The proposed Uptown CPU area is underlain by the San Diego Formation, Pomerado Conglomerate, and Mission Valley Formations, which are considered to be of high sensitivity for fossil resources, whereas the Uptown CPU area is not underlain by any moderate resource potential formations. Therefore, no impacts relative to moderate resource potential formations would occur.

Grading associated with future development projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions that involve excavation into the underlying geological formations could expose these formations and associated fossil remains. These development projects could destroy paleontological resources if the fossil remains are not recovered and salvaged. In addition, future projects proposing shallow grading where formations are exposed and where fossil localities have already been identified would also result in a potentially significant impact. Thus, impacts resulting from future discretionary development into the high sensitivity San Diego, Pomerado Conglomerate, and Mission Valley Formations would be potentially significant (Impact 6.10-1).

Build-out of future ministerial projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions would likely result in a certain amount of disturbance to the native bedrock within the Uptown CPU area. Since ministerial projects are not subject to a discretionary review process, there would be no mechanism to screen for grading quantities and geologic formation sensitivity and apply appropriate requirements for paleontological monitoring. Thus, impacts related to future ministerial development that would occur with build-out of the proposed Uptown CPU and associated discretionary actions would be potentially significant (Impact 6.10-2)

- **Impact 6.10-1:** Grading activities associated with the future discretionary projects that require grading in excess of 1,000 cubic yards, extending to a depth of 10 feet or greater into high sensitivity formations, could result in significant impacts to paleontological resources.
- **Impact 6.10-2:** Grading activities associated with the future ministerial projects that require grading in excess of 1,000 cubic yards, extending to a depth of 10 feet or greater into high sensitivity formations, could result in significant impacts to paleontological resources.

Cumulative Impacts

Development allowed pursuant to the proposed Uptown CPU and development within surrounding CPUs could involve excavation of previously undeveloped areas, some of which may consist of unique paleontological resources with fossil-bearing potential. Potential cumulative impacts to paleontological resources were evaluated in the General Plan PEIR. The analysis concluded that there is potential for the cumulative loss of paleontological resources throughout the county, as the county continues to develop in response to projected population growth. Likewise, development of the Uptown CPU area may result in the loss of unique paleontological resources or geologic formations with fossil-bearing potential. Certification of the General Plan PEIR included the adoption of mitigation measures that attempt to reduce significant project-level impacts from future development. However, there is only a mechanism to apply the mitigation framework to discretionary projects, not ministerial projects. Thus, within the Uptown CPU area and surrounding communities, significant impacts to paleontological resources could occur associated with grading for ministerial projects. Similar to the General Plan PEIR, build-out of ministerial projects within the Uptown CPU area would result in significant cumulative impacts to paleontological resources (Impact 6.10-2).

6.10.4 Significance of Impacts

Because of high sensitivity for paleontological resources within the San Diego, Pomerado Conglomerate, and Mission Valley Formations, grading into these formations could potentially destroy fossil resources. Therefore, implementation of future discretionary and ministerial projects within the proposed Uptown CPU area within these formations has the potential to result in significant impacts to paleontological resources.

6.10.5 Mitigation Framework

In order to reduce the potential adverse impact to paleontological resources associated with discretionary projects, the project would incorporate the mitigation measure identified in the General Plan PEIR addressing paleontological resource impacts.

The following measure would apply to any discretionary project that proposes subsurface disturbance within a high sensitivity formation. If no subsurface disturbance is planned, then paleontological resources would not be impacted and development of a project-specific paleontological monitoring and discovery treatment plan would not be necessary. The following mitigation measure would reduce impact 6.10-1 to a less than significant level.

PALEO 6.10-1 Paleontological Review and Monitoring

Prior to the approval of subsequent discretionary development projects implemented in accordance with the proposed Uptown CPU, the City shall determine the potential for impacts to paleontological resources within a high sensitivity formation based on review of the project application submitted and recommendations of a project-level analysis completed in accordance with the steps presented below. Future projects shall be sited and designed to minimize impacts on paleontological resources in accordance with the City's Paleontological Resources Guidelines and CEQA Significance Thresholds. Monitoring for paleontological resources required during construction activities shall be implemented at the project level and shall provide mitigation for the loss of important fossil remains with future subsequent development projects that are subject to environmental review.

- I. Prior to Project Approval
 - A. The environmental analyst shall complete a project-level analysis of potential impacts on paleontological resources. The analysis shall include a review of the applicable United States Geological Survey Quad maps to identify the underlying geologic formations, and shall determine if construction of a project would:
 - Require over 1,000 cubic yards of excavation and/or a 10-foot, or greater, depth in a high resources potential geologic deposit/formation/rock unit.
 - Require over 2,000 cubic yards of excavation and/or 10-foot, or greater, depth in a moderate resource potential geologic deposit/formation/rock unit.
 - Require construction within a known fossil location or fossil recovery site. Resource potential within a formation is based on the Paleontological Monitoring Determination Matrix.
 - B. If construction of a project would occur within a formation with a moderate to high resource potential, monitoring during construction would be required and any identified resources shall be recovered.
 - Monitoring is always required when grading on a fossil recovery site or a known fossil location.
 - Monitoring may also be needed at shallower depths if fossil resources are present or likely to be present after review of source materials or consultation with an expert in fossil resources (e.g., the San Diego Natural History Museum).
 - Monitoring may be required for shallow grading (<10 feet) when a site has previously been graded, and/or unweathered geologic deposits/formations/rock units are present at the surface.
 - Monitoring is not required when grading documented artificial fill. When it has been determined that a future project has the potential to impact a geologic formation with a high or moderate fossil sensitivity rating, a Paleontological Mitigation Monitoring and Report Program shall be implemented during construction grading activities.

6.10.6 Significance of Impacts after Mitigation

All future discretionary projects that would occur as a result of the proposed Uptown CPU and associated discretionary actions would be required to comply with PALEO 6.10-1. Implementation of mitigation measure PALEO 6.10-1 would reduce paleontological impacts associated with future discretionary development to below a level of significance.

Build-out of future ministerial projects proposed in conformance with the proposed Uptown CPU and associated discretionary actions would also likely result in a certain amount of disturbance to the native bedrock within the study area. Since ministerial projects are not subject to a discretionary review process, there would be no mechanism to screen for grading quantities and geologic formation sensitivity and apply appropriate requirements for paleontological monitoring. Thus, impacts related to future ministerial development that would occur with build-out of the proposed Uptown CPU and associated discretionary actions would remain significant and unavoidable.

6.11 Hydrology/Water Quality

This section addresses the potential hydrology and surface and groundwater quality impacts that would result from the project. It relies on secondary source information and policies contained within the proposed Uptown Community Plan Update (CPU). This section also details applicable regulations, receiving waters, flood hazards, and other relevant existing conditions within the study area.

6.11.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

6.11.2 Significance Determination Thresholds

Based on the City's Significance Determination Thresholds, which have been adapted to guide a programmatic analysis of the proposed Uptown CPU and associated discretionary actions, a significant hydrology impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions would:

- 1) Result in flooding due to an increase in impervious surfaces, changes in absorption rates, drainage patterns, or the rate of surface runoff;
- 2) Result in a substantial increase in pollutant discharge to receiving waters and increase discharge of identified pollutants to an already impaired water body; or
- 3) Deplete groundwater supplies, degrade groundwater quality, or interfere with groundwater recharge.

6.11.3 Impact Analysis

Issue 1 Flooding and Drainage Patterns

Would the project result in flooding due to an increase in impervious surfaces, changes in absorptions rates, drainage patterns, or the rate of surface runoff?

The Uptown community is an urban community within the City, and the majority of the Uptown CPU area is developed. Large areas of impervious surfaces (buildings, roadways, and surface parking) are mixed with a smaller amount of pervious (landscaping, parks) areas.

Future projects that could occur in the Uptown CPU area would result in an increase in impervious areas due to the new buildings, hardscape, and parking areas. Landscaping, as well as pervious pavements used in lieu of standard pavement, diminish a project's increase in impervious areas and, therefore, diminish a project's increase in urban pollutants. Implementation of the proposed Uptown CPU and associated discretionary actions would also have the potential to change surface runoff characteristics, including the volume of runoff, rate of runoff, and drainage patterns. An increase in the volume or rate of runoff or change in drainage patterns could result in flooding and/or erosion.

Future projects would be required to comply with the National Pollutant Discharge Elimination System (NPDES) and Hydromodification Management Plan (HMP) requirements as described in the City of San Diego Stormw Water Standards Manual. Storm water detention and HMP facilities would be implemented to accommodate the potential increase in storm water runoff rates due to the proposed increase in impervious areas. To fulfill the HMP requirements, projects would need to be designed so that runoff rates and durations are controlled to maintain or reduce pre-project downstream erosion conditions and protect stream habitat. Projects would typically manage the increase in runoff by implementing a series of storm water Best Management Practices (BMPs) and detention facilities that have been specifically designed for Hydromodification Management.

The proposed Uptown CPU Elements include policies that address hydrology and water quality. The Conservation Element of the proposed Uptown CPU contains a goal related to the improvement of the hydrology and drainage within the proposed Uptown CPU area – specifically the application of sustainable urban runoff management techniques applied to support the surrounding landscape and reduce impacts on the surrounding canyons. Other proposed Conservation Element policies address urban runoff management and maintenance and cleaning of canyons.

All development in the City is subject to drainage regulations through the San Diego Municipal Code, which requires that the existing flows of a property proposed for development be maintained to ensure that the existing structures and systems handling the flows are sufficient. Since future development would be required to adhere to existing drainage regulations, development would not result in alterations to existing drainage patterns in a manner that would result in flooding or erosion on- or off-site. Adherence to the requirements of the City's Drainage Design Manual and Storm Water Standards Manual, which require installation of Low Impact Development (LID) practices, such as bioretention areas, pervious pavements, cisterns, and/or rain barrels, would improve surface drainage conditions or, at a minimum, not exacerbate flooding or cause erosion. Furthermore, future development would be required to comply with NPDES permit requirements, which would result in a reduction in the volume and rate of surface runoff compared to the existing condition. The quantity of runoff reduction would depend on the actual design of open space, pervious areas, run-off retention, and the manner of implementation of these LID practices. Thus, impacts would be less than significant.

Issue 2 Water Quality

Would the project result in an increase in pollutant discharge to receiving waters and increase discharge of identified pollutants to an already impaired water body?

Future development projects that could occur in the Uptown community under the proposed Uptown CPU and associated discretionary actions would have the potential to change pollutant discharges. However, as future development in accordance with the proposed Uptown CPU and associated discretionary actions occurs, applicable NPDES permit requirements would require the retention and/or treatment of storm water through the implementation of BMPs. Future development would be required to demonstrate how pollutants such as various trace metals (e.g., copper, lead, zinc, and mercury), fecal coliform, low dissolved oxygen, phosphorus, and total discolved solids that could be associated with future development would be treated to prevent discharge into receiving waters. Much of the existing development in the area was constructed before current storm water regulations were adopted. Thus, future development and redevelopment would be subject to current, more stringent requirements, which would likely improve water quality.

Under current storm water regulations in the City, all projects requiring approvals are subject to certain minimum storm water requirements to protect water quality. Types of storm water BMPs required for new developments include site design, source control, and treatment control practices, many of which overlap with LID practices. Storm water BMPs would reduce the amount of pollutants transported from a future proposed development project to receiving waters. <u>Subsequent projects implemented in accordance with the Uptown CPU would be subject to existing regulations in place at the time projects are implemented.</u> Thus, impacts of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to water quality.

Issue 3 Groundwater

Would the project deplete groundwater supplies, degrade groundwater quality, or interfere with ground water recharge?

Based on the Water Quality Control Plan for the San Diego Basin (April 2011), most of the groundwaters in the region have been extensively developed; the availability of potential future uses of groundwater resources is limited. Further development of ground water resources would probably necessitate groundwater recharge programs to maintain adequate groundwater table elevations. Groundwater within the San Diego Mesa is exempt from municipal and domestic supply beneficial use, as it was determined by the 1989 Regional Water Quality Control Board's Resolution No. 89-33 that this area does not support municipal and domestic supply. Groundwater within the San Diego portion of the San Diego Hydrologic Unit has a potential beneficial use for municipal and domestic supply and existing beneficial uses for agricultural supply, industrial service supply, and industrial process supply.

As discussed under Issues 1 and 2 above, current storm water regulations encourage infiltration of storm water runoff and protection of water quality which would also protect the quality of groundwater resources and support infiltration where appropriate. Thus, implementation of the

proposed Uptown CPU and associated discretionary actions would result in a less than significant impact on groundwater supply and quality.

Cumulative Impacts

Future projects within the Uptown CPU area and surrounding areas including projects within the North Park and Golden Hill CPUs, could have a cumulative impact on hydrology and water quality, including downstream problems with flooding, sizing of drainage facilities, erosion, and sedimentation. However, all future development within the CPU areas would be required to comply with all NPDES permit requirements, including the development of a SWPPP if the disturbed area covers one acre or more or a Water Quality Control Plan if the disturbed area is less than one acre. Future projects would also be required to follow the City's Storm Water Standards Manual for drainage design and BMPs for treatment. Thus, cumulative impacts would be less than significant.

6.11.4 Significance of Impacts

6.11.4.1 Flooding and Drainage Patterns

All development is subject to drainage and floodplain regulations in the San Diego Municipal Code, and would be required to adhere to the City's Drainage Design Manual and Storm Water Standards Manual. Therefore, with future development, the volume and rate of overall surface runoff within the proposed Uptown CPU area would be reduced when compared to the existing condition. Impacts would be less than significant and mitigation is not required.

6.11.4.2 Water Quality

New development under the proposed Uptown CPU and associated discretionary actions would be required to implement LID and storm water BMPs into project design to address the potential for transport of pollutants of concern through either retention or filtration. The implementation of LID design and storm water BMPs would reduce the amount of pollutants transported from the Uptown CPU area to receiving waters. Impacts would be less than significant, and no mitigation would be required.

Future development would adhere to the requirements of the MS4 permit for the San Diego Region and the City's Storm Water Standards Manual, water quality conditions—both surface and groundwater—are not expected to have an adverse effect on water quality. Additionally, the City has adopted the Master Storm Water Maintenance Program to address flood control issues by cleaning and maintaining the channels to reduce the volume of pollutants that enter the receiving waters. Impacts would be less than significant, and no mitigation would be required.

6.11.4.3 Groundwater

Groundwater within the San Diego Mesa is exempt from municipal and domestic supply beneficial use and does not support municipal and domestic supply. Groundwater within the Mission San Diego area of the Lower San Diego portion of the San Diego Hydrologic Unit has a potential beneficial use for municipal and domestic supply. Storm water regulations that encourage infiltration of storm water runoff and protection of water quality would also protect the quality of groundwater resources and support infiltration where appropriate. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact on groundwater supply and quality.

6.11.5 Mitigation Framework

Implementation of the proposed Uptown CPU and associated discretionary actions would not result in significant impact to the environment. No mitigation is required.

6.12 **Public Services and Facilities**

Public services are those functions that serve residents on a communitywide basis. These functions include police protection, parks and recreation centers, fire protection, libraries, and schools. The following provides a discussion of these services and facilities as they relate to the proposed Uptown Community Plan Update (CPU) and associated discretionary actions. This section is based on communication from service providers, which are included in Appendix J of this draft Program Environmental Impact Report (PEIR).

6.12.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively. Existing conditions specifically applicable to the Uptown CPU area are discussed below. Figure 6.12-1 illustrates the location of the public services discussed below.

6.12.1.1 Police Protection

The Uptown community is served by the Western Division of the Police Department. The Western area station is located at 1222 Gaines Street within the Mission Valley community planning area (Figure 6.12-1). The average response times for the Western Division for 2014 were 6.4 minutes for emergency calls, 10.8 minutes for Priority 1 calls, 25.4 minutes for Priority 2 calls, 62.8 minutes for Priority 3 calls, and 69.7 minutes for Priority 4 calls. The San Diego police Department's Citywide response time goals are 7 minutes for emergency calls, 14 minutes for Priority 1 calls, 27 minutes for Priority 2 calls, 68 minutes for Priority 3 calls, and 70 minutes for Priority 4 calls.

6.12.1.2 Parks and Recreation Facilities

The Uptown community is currently served by a number of parks, and joint-use facilities. Mission Hills Park (includes Pioneer Memorial Park), provides passive recreation amenities, such as multipurpose turf areas, parking lot, a children's play area, seating, picnicking, walkways, and landscaping. Similarly, Old Trolley Barn Park provides multi-purpose turf areas, a children's play area, seating, picnicking, walkways, and landscaping. West Lewis Street, a pocket park, provides passive recreation amenities, a trail, public art, interpretive signage, and seating. There are two joint-use facilities within Uptown, which are Birney Elementary School and Roosevelt Middle School.

At full community development, the projected population for the Uptown community is 55,700. Therefore, according to General Plan standards for population-based parks, the community should be served by a minimum of 155.96 useable acres of park land at full community development. Additionally, at full community development, the Uptown CPU population warrants approximately



 $FIGURE \ 6.12\mbox{-}1$ Location of Public Services and Facilities – Uptown

two and one-quarter recreation centers equivalent to 37,910 total square feet and approximately one aquatic complex. Of the total of 155.96 acres of population-based parks needed to serve Uptown at full community development, <u>18.2114.66</u> acres currently exist. This includes the following parks: Mission Hills Park (including Pioneer Memorial Park); Old Trolley Barn Park; West Lewis Street Pocket Park; Birney Elementary and Roosevelt Middle Schools Joint Use Areas; and the Sixth Avenue Children's Play Area located within Balboa Park. Currently, the Uptown CPU area does not have a recreation center or aquatic complex.

6.12.1.3 Fire/Life Safety Protection

The City provides fire services through geographic service areas. The Fire Department provides emergency/rescue services, hazard prevention, and safety education to ensure the protection of life, property, and the environment, including education about vegetation management to protect properties from wildfires in canyon areas. Fire protection for the community is provided primarily by three fire stations. Station 8 is located at Goldfinch and Washington Street, Station 5 is located at Ninth and University Avenue, and Station 3 is located at State and Kalmia Street (see Figure 6.12-1). There are plans in the near term that Fire Station 5 will be rebuilt. In addition, expansion plans for Fire Station 8 include new quarters and parking for fire staff that will occupy the Mission Hills Library site, once the library is relocated.

a. Response Standards General

To treat medical patients and control small fires, the first-due unit should arrive within 7.5 minutes, 90 percent of the time from the receipt of the 911 call in fire dispatch. This equates to 1-minute dispatch time, 1.5-minute company turnout time, and 5-minute drive time in the most populated areas.

b. Response Force for Serious Emergencies

To confine fires near the room of origin, to stop wildland fires to under 3 acres when noticed promptly and to treat up to five medical patients at once (Citygate 2011), a multiple-unit response of at least 17 personnel should arrive within 10.5 minutes from the time of 911-call receipt in fire dispatch, 90 percent of the time. This equates to 1-minute dispatch time, 1.5-minute company turnout time and 8-minute drive time spacing for multiple units in the most populated areas.

c. Adopted Fire Station Location Measures

To direct fire station location timing and crew size planning as the community grows, the adopted fire unit deployment performance measures based on population density zones are listed in Table 6.12-1, below:

Table 6.12-1 Deployment Measures for San Diego City Growth (by Population Density per Square Mile)							
	Structure Fire	Structure Fire	Structure Fire	Wildfires,			
	Urban Area	Rural Area	Remote Area	Populated Areas,			
	(>1,000-people/	(1,000 to 500	(500 to 50	Permanent Open			
	sq. mi.) people/sq. mi.) people/sq. mi.) Space A						
1 st Due Travel Time	5	12	20	10			
Total Reflex Time	7.5	14.5	22.5	12.5			
1 st Alarm Travel Time	8	16	24	15			
1 st Alarm Total Reflex	10.5	18.5	26.5	17.5			

d. Aggregate Population Definitions

Where more than one square mile is not populated at similar densities, and/or a contiguous area with different zoning types aggregates into a population "cluster," these measures guide the determination of response time measures (Table 6-12.2) and the need for fire stations:

Table 6.12-2 Response Time Measures					
Area	Aggregate Population	First-Due Unit Travel Time Goal			
Metropolitan	> 200,000 people	4 minutes			
Urban-Suburban	< 200,000 people	5 minutes			
Rural	500 - 1,000 people	12 minutes			
Remote	< 500	> 15 minutes			

6.12.1.4 Libraries

The Uptown community is served by the Mission Hills and University Heights libraries (Figure 6.12-1). A new 25,000-square-foot facility will replace the current 3,850-square-foot Mission Hills Branch Library located at 925 West Washington Street and built in 1961 prior to the minimum standard of 15,000 square feet for branch libraries. The new library facility site, which the City has acquired, will be located at the southwest corner of Washington and Front streets. General Plan policies PF-J.3 and PF-J.5 support libraries that serve larger areas to maximize capital efficiencies.

6.12.1.5 Schools

The Uptown community is served by <u>three_two_public</u> elementary schools (Florence<u>and</u>, Alice Birney, <u>and Grant</u>_Elementary Schools), <u>Grant K-8 School</u>, Roosevelt Middle School, and San Diego High School (Figure 6.12-1). In addition, there are charter schools, private schools, and neighboring community schools that help serve the community.

6.12.2 Significance Determination Thresholds

Based on the City's Significance Determination Thresholds, which have been adapted to guide a programmatic analysis of the proposed Uptown CPU and associated discretionary actions, a significant public services and facilities impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions would:

 Promote growth patterns resulting in the need for and/or provision of new or physically altered public facilities (including police protection, parks, or other recreational facilities, fire/life safety protection, libraries, schools, or maintenance of public facilities including roads), the construction of which could cause significant environmental impacts in order to maintain service ratios, response times, or other performance objectives.

6.12.3 Impact Analysis

Issue 1 Public Facilities

Would the project promote growth patterns resulting in the need for and/or provision of new or physically altered public facilities (including police protection, parks or other recreational facilities, fire/life safety protection, libraries, schools, or maintenance of public facilities including roads), the construction of which could cause significant environmental impacts in order to maintain service ratios, response times, or other performance objectives?

a. Police Protection

Within the Uptown CPU area, the Western Division of the San Diego Police Department operates under the Citywide response time goals detailed in Chapter 5.0, Regulatory Framework (Section 5.12.1.1) of this PEIR and responds to emergency and Priority 1 through Priority 4 calls. There are no current plans for additional police substations in the Uptown CPU area. Correspondence with the San Diego Police Department identified that police response times within Uptown will continue to increase with the build-out of the proposed Uptown CPU, which could ultimately result in the need for new or expanded police services. However, as future development is proposed within the Uptown CPU area, individual projects would be subject to applicable Development Impact Fees (DIF) for public facilities financing in accordance with Municipal Code Section 142.0640. The Uptown CPU includes a comprehensive Impact Fee Study that will define applicable DIF fees for future development, including fees for police facilities funding.

Proposed Uptown CPU policies support provision of police services within the CPU area by providing guidelines to reduce incidence of criminal activity within the Uptown neighborhoods, including support for Neighborhood Watch and Community Alert Programs, increased foot and bicycle patrols, exchange of information with patrol officers, and development projects that provide adequate lighting, visibility for surveillance, and gradations between public and private space.

The proposed Uptown CPU and associated discretionary actions do not include construction of new police facilities. As population growth occurs and the need for new facilities are identified, any future

construction of police facilities would be subject to a separate environmental review at the time design plans are available. Thus, while build-out of the CPU could result in the demand for new or altered police services, the existing DIF framework in place would require future projects within the CPU area to pay fees for future facility needs. Additionally, no police facilities are currently proposed and any future facility would require a site-specific environmental review. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in less than significant environmental impacts associated with the construction of new facilities in order to maintain service ratios, response times, or other performance objectives related to police services.

b. Parks and Recreation

Based on the projected population for the Uptown Community, 55,700, General Plan standards for population-based parks and recreation facilities would require the community to be served by a minimum of 155.96 useable acres of park land at full community development. Additionally, at full community development, the projected population warrants approximately two and one-quarter recreation centers equivalent to 37,910 total square feet and approximately one aquatic complex.

Opportunities for additional park land and recreation facilities within the Uptown Community are anticipated to come primarily through redevelopment of private and public properties and through the application of park equivalencies as detailed below. Facilities that may be considered as population-based park equivalencies include:

- Joint use facilities;
- Trails through open space;
- Portions of resource-based parks;
- Privately owned, publicly used parks;
- Non-traditional parks, such as rooftop or indoor recreation facilities; and
- Facility or building expansion or upgrades.

The General Plan allows park equivalencies to be used when vacant land is limited, unavailable or is cost-prohibitive. The application of park equivalencies is determined by the community and City staff through a set of guidelines. The community and City identified and evaluated population-based park and recreation opportunities, as well as potential park equivalency sites, for their recreational value, possible uses and functions, public accessibility, consistency with General Plan policies and guidelines, and other land use policy documents (e.g., Balboa Park Master Plan and Balboa Park East Mesa Precise Plan). Tables 6.12-3 and 6.12-4 summarize the existing and proposed parks and equivalencies that have been selected by the Uptown Community to supplement their existing population-based park inventory. The table also includes recommendations contained in the Balboa Park Master Plan, including the Sixth Avenue Area, where appropriate, as well as recommendations generated by the community and City staff for facilities outside Balboa Park. Figure 6.12-2 shows the locations of park facilities.



FIGURE 6.12-2 Parks, Recreation Facilities, and Open Space – Uptown

			Table 6.12-3	
	P	opulatio	n-Based Parks and Recreatio	n Facilities
Parks/	Existing	Future	Parks and Recreation	
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations
Major Parks				
None				
Community Park	(S			
None				
Neighborhood P	arks			
Mission Hills Park (includes Pioneer Memorial Park)	8.34		Existing park consisting of passive recreation amenities, such as multi- purpose turf areas, parking lot, a children's play area, seating, picnicking, walkways, and landscaping.	
Mystic Park		7.58	Proposed park site located east of State Route 163 (SR- 163), north or Washington Street and south of Pascoe Street on City and California Department of Transportation (Caltrans) right-of-way. Portions of the site are designated as a California Historic Parkway and Scenic Highway, and are a State Historic Resource and City Historic Landmark.	Prepare a park feasibility study; The study may address historic resource issues, vehicular, pedestrian and traffic circulation, reconfiguration of freeway on- ramps, new traffic signalization, community recreation needs, other issues to be determined, and a preliminary cost analysis. An agreement with Caltrans may be required. Based on results of the study, acquire, design, and construct park amenities for active and passive uses that could include multi-purpose turf areas, children's play areas, an amphitheater and performance opportunities, picnicking, seating, exercise areas, and an off-leash dog area.
Old Trolley Barn Park	2.92		Existing park consisting of passive recreation amenities, such as multi- purpose turf areas, a children's play area, seating, picnicking, walkways, and landscaping.	

Table 6.12-3					
	P	opulatio	n-Based Parks and Recreatio	n Facilities	
Parks/	Existing	Future	Parks and Recreation		
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities	
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations	
Reynard Way		4.72	Proposed park site on	Acquire, design, and construct park	
Neighborhood			undeveloped property,	amenities for active and passive	
Park			consisting of multiple,	recreation, such as informal multi-	
			privately owned parcels,	purpose sports field, children's play	
			located at 3532 Reynard	areas, seating, picnicking,	
			Way. The site consists of	walkways, and landscaping.	
			varied topography and a		
			potentially historically		
			significant building, which		
			would present some		
			developmental challenges,		
			but would yield many		
			recreational opportunities.		
			Adaptive reuse of the		
			building for recreational		
			purposes is a possibility.		
Mini-Parks					
None					
Pocket Parks/Pl	azas				
Bandini Street		0.18	Proposed pocket park on	Acquire, design, and construct park	
Pocket Park			vacant, privately owned	amenities to include passive	
			property located on the east	recreation, such as a children's play	
			side of Bandini Street at	area, seating, picnicking, walkways,	
			Mergho Impasse.	and landscaping.	
<u>Clark Street</u>		<u>0.24</u>	<u>Proposed pocket park on</u>	Acquire, design and construct park	
<u>Pocket Park</u>			<u>vacant, privately-owned</u>	amenities to include passive	
			<u>property located at the</u>	recreation, such as seating, picnic	
			<u>terminus of Clark Street,</u>	facilities, an overlook and a	
			<u>north of Alameda Terrace.,</u>	trailhead to the adjacent Robyn's	
			and adjacent to the Mission	<u>Egg Trail.</u>	
			<u>Hills Open Space.</u>		
Fir Street		0.23	Proposed pocket park on	Acquire, demolish existing	
Pocket Park			two privately owned parcels,	improvements, design, and	
			located on the southwest	construct park amenities to include	
			corner of Fir Street and Sixth	passive recreation, such as a	
			Avenue. The site is currently	children's play area, seating,	
			developed with a small	picnicking, walkways, and	
			structure and associated	landscaping.	
			parking area.		
First &		0.28	Proposed pocket park on	Acquire, design, and construct park	
Robinson			vacant, privately owned	amenities to include passive	
Pocket Park			property located on the	recreation, such as a children's play	

Table 6.12-3				
	P	opulatio	n-Based Parks and Recreatio	n Facilities
Parks/	Existing	Future	Parks and Recreation	
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations
			northwest corner of First	area, seating, picnicking, walkways,
Calden Cata		0.22	and Robinson avenues.	and landscaping.
Golden Gate		0.33	Proposed pocket park on	Design and construct park
Drive Pocket			city-owned open space land	amenities to include passive
Park			Open Space area directly	interpretive signage, improved
			adjacent to Golden Gate	trails overlook/seating
			Drivo	Landscaping atc
Ealcon Street		0.10	Proposed packet park op	Acquire design and construct park
Pocket Park		0.19	vacant, privately-owned	Acquire, design, and construct park
<u>r ocket r ark</u>			property located on the	recreation such as seating
			southwest side of Falcon	nichicking and landscaping that
			Street between Goldfinch	ontimize views towards Downtown
			and W. Thorn Streets	
Front & W		0.46	Proposed pocket park on	Acquire design and construct park
luniper Streets		0110	Port District property	amenities to include passive
Pocket Park			located on the southwest	recreation, such as a children's play
<u> </u>			corner of Front and	area, seating, picnicking, walkways
			W. Juniper Streets, currently	and landscaping: continuation of
			developed as a community	the community garden use may
			garden.	also be considered.
Goldfinch		<u>0.32</u>	Proposed pocket park on	Acquire, design and construct park
<u>Street &</u>			<u>vacant, privately-owned</u>	amenities to include passive
<u>Pennsylvania</u>			property located on the west	recreation, such as a children's play
<u>Avenue Pocket</u>			side of Goldfinch	area, seating, picnicking, walkways
<u>Park</u>			<u>Street/Reynard Way, north</u>	and landscaping.
			<u>of W. Pennsylvania Avenue.</u>	
<u>Goldfinch &</u>		<u>0.12</u>	<u>Proposed pocket park on</u>	Design and construct park
<u>WSpruce</u>			<u>undeveloped City-owned</u>	<u>amenities to include passive</u>
<u>Streets Pocket</u>			<u>Open Space located on the</u>	recreation, such as a children's play
<u>Park</u>			west side of Goldfinch	area, walkways, seating, picnicking,
			Street, south of the	and landscaping. Pursue inclusion
			W. Spruce Street right-of-	of the W. Spruce Street right-of-way
			way (paper street) and	(paper street) in the planning and
		0.10	W. Thorn Street.	development of the pocket park.
<u>Guy & Henry</u>		0.12	Proposed pocket park on	Acquire, design, and construct park
Streets Pocket			vacant, privately-owned	amenities to include passive
Park			property located on the	recreation, such as seating, picnic
			Southern corner of GUY &	<u>facilities, all overlook and IT</u>
			the Mission Hills Open	<u>reasible, a trainedu to trie aujacent</u> Robyn's Egg Trail
				NUDYITS Egg ITall.
			<u>space.</u>	

Table 6.12-3					
	P	opulatio	n-Based Parks and Recreatio	n Facilities	
Parks/	Existing	Future	Parks and Recreation		
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities	
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations	
<u>Hawk Street</u>		<u>0.24</u>	<u>Proposed pocket park on</u>	Acquire, design, and construct park	
<u>Pocket Park</u>			<u>2 privately-owned, vacant</u>	<u>amenities to include passive</u>	
			<u>parcels, located on the east</u>	<u>recreation, such as seating,</u>	
			<u>side of Hawk Street between</u>	picnicking and landscaping that	
			<u>W. Thorn Street and Horton</u>	<u>optimize easterly views. Pursue</u>	
			<u>Avenue.</u>	inclusion of the W. Spruce Street	
				ROW (paper street) in the planning	
				and development of the pocket	
				<u>park.</u>	
<u>Hawk Street</u>		<u>0.19</u>	<u>Proposed pocket park on</u>	Acquire, design, and construct park	
and Court Way			<u>vacant, privately-owned</u>	<u>amenities to include passive</u>	
<u>Pocket Park</u>			property located on the west	recreation, such as seating,	
			<u>side of Hawk Street at the</u>	<u>picnicking, walkways and</u>	
			intersection with Court Way.	landscaping.	
Horton Avenue		<u>0.33</u>	<u>Proposed pocket park on</u>	Acquire, desig,n and construct park	
<u>& Ibis Street</u>			<u>vacant, privately-owned</u>	amenities to include passive	
<u>Pocket Park</u>			<u>property located on the</u>	recreation, such as a children's play	
			<u>southwest corner of Horton</u>	<u>area, seating, picnicking, walkways</u>	
			<u>Avenue & Ibis Street.</u>	and landscaping.	
<u>lbis Lane</u>		<u>0.10</u>	<u>Proposed pocket park on a</u>	Acquire, design, and construct park	
<u>Pocket Park</u>			<u>vacant, privately-owned</u>	amenities to include passive	
			<u>parcel, located on the west</u>	recreation, such as seating,	
			<u>side of Ibis Street north of</u>	<u>picnicking, walkways and</u>	
			<u>Ibis Lane.</u>	landscaping.	
<u>lbis Street</u>		<u>0.12</u>	<u>Proposed pocket park on a</u>	Acquire, design, and construct park	
<u>Pocket Park</u>			<u>vacant, privately-owned</u>	amenities to include passive	
			parcel, located on the west	recreation, such as a children's play	
			<u>side of Ibis Street, between</u>	<u>area, seating, picnicking, walkways</u>	
			<u>W. Lewis Street and</u>	and landscaping.	
			<u>W. Montecito Way.</u>		
<u>La Callecita</u>		<u>0.11</u>	<u>Proposed pocket park on a</u>	Acquire, design, and construct park	
<u>Street Pocket</u>			<u>vacant, privately-owned</u>	amenities to include passive	
<u>Park</u>			<u>parcel, located on the south</u>	recreation, such as seating,	
			side of La Callecita Street,	picnicking, walkways and	
			between Sunset Road and	landscaping.	
			<u>Witherby Street.</u>		
Laurel Street		0.11	Proposed pocket park on	Design and construct park	
Pocket Park			undeveloped City-owned	amenities to support passive	
			park land.	recreation, such as children's play	
				area, seating, picnicking, walkways,	
				and landscaping.	

	Table 6.12-3					
	P	opulatio	n-Based Parks and Recreatio	n Facilities		
Parks/	Existing	Future	Parks and Recreation			
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities		
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations		
Maryland		<u>0.21</u>	Proposed pocket park on	Acquire, design, and construct park		
<u>Street Pocket</u>			<u>2 vacant privately-owned</u>	amenities to include passive		
<u>Park</u>			<u>parcels, located on the east</u>	recreation, such as a children's play		
			<u>side of Maryland Street,</u>	area, seating, picnicking, walkways		
			<u>between Tyler Avenue,</u>	and landscaping.		
			Morrow Way, and an alley.			
Mission Valley		<u>0.10</u>	<u>Proposed pocket park on</u>	Design and construct park		
<u>Overlook</u>			city-owned open space land	amenities to include passive		
			within the University Heights	recreation, such as interpretive		
			Open Space located on the	signage, overlook/seating, and		
			<u>north side of Golden Gate</u>	landscaping.		
			Drive east of Cleveland			
			<u>Avenue.</u>			
Olive Street		0.60	Proposed pocket park on	Design and construct park		
Park			undeveloped City-owned	amenities to include passive		
			park property located on	recreation, such as a children's play		
			Olive Street.	area, walkways, seating, picnicking,		
				and landscaping.		
Pringle &		<u>0.24</u>	<u>Proposed pocket park on</u>	Acquire, design, and construct park		
<u>Puterbaugh</u>			<u>2 vacant privately-owned</u>	amenities to include passive		
<u>Streets Pocket</u>			<u>parcels, located on the</u>	recreation, such as a children's play		
<u>Park</u>			<u>southern corner of the</u>	<u>area, seating, picnicking, walkways</u>		
			intersection of Pringle and	and landscaping that optimize		
			Puterbaugh Streets.	views towards Downtown.		
Sixth Avenue		0.45	Proposed pocket park	Acquire, demolish existing		
Pocket Park			located on privately owned	improvements, and design and		
			property on the west side of	construct park amenities to include		
			Sixth Avenue between	passive recreation, such as a		
			University and Robinson	children's play area, seating,		
			avenues. The site is	picnicking, walkways and		
			currently developed with the	landscaping.		
			"Pernicano's" restaurant and			
			associated parking lot.			
<u>State and</u>		<u>0.12</u>	<u>Proposed pocket park on a</u>	Acquire, design, and construct park		
<u>W. Thorn</u>			<u>vacant, privately-owned</u>	amenities to include passive		
Streets Pocket			parcel, located on the	recreation, such as seating,		
<u>Park</u>			northern corner of the	picnicking, walkways and		
			Intersection of State and	landscaping.		
	0.05	0.00	<u>vv. Thorn Streets.</u>			
West Lewis	0.35	0.03	Existing park, located	Construct Phase II Improvements,		
Street Pocket			between Falcon and	including the trail connection with		
Park			Goldfinch streets,	the existing Phase I, in accordance		

	Table 6.12-3					
	P	opulatio	n-Based Parks and Recreatio	n Facilities		
Parks/	Existing	Future	Parks and Recreation			
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities		
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations		
	0	U	comprising passive	with the approved General		
			recreational amenities, a	Development Plan.		
			trail, public art, interpretive	·		
			signage, and seating.			
West Maple		0.25	Proposed pocket park on	Construct passive park amenities		
Canyon Pocket			undeveloped City-owned	including seating, interpretive		
Park			land adjacent to the Maple	signage, landscaping, and a		
			Canyon Open Space area.	trailhead, in accordance with the		
			5 1 1	approved General Development		
				Plan.		
Special Activity	Park					
None						
Recreation Cen	ters					
Grant K-8	N/A	N/A	Proposed gymnasium	Pursue an agreement with SDUSD		
School			located within the Grant K-8	for joint use of the proposed		
Gymnasium			School site on San Diego	10,454-square-foot gymnasium.		
			Unified School District	The facility would be designed and		
			(SDUSD) land.	constructed by SDUSD.		
Redwood	N/A	N/A	Proposed recreation facility	Design and construct an		
Recreation			located in the vicinity of the	approximately 10,643 sq. ft.		
Center (within			existing Redwood Bridge	recreation center including		
Balboa Park)			Club, between Sixth Avenue,	community meeting and multi-		
			Balboa Drive, Quince Street,	purpose rooms, arts & crafts, and		
			and Spruce Street.	fitness rooms.		
Uptown	NA	NA	Proposed recreation facility	Expand/replace the existing		
Recreation			located in the southern	building with a 17,000 sq. ft.		
Center (within			portion of the community.	recreation center including a		
Balboa Park)				gymnasium, community meeting		
				and multi-purpose rooms, arts &		
				crafts, and fitness rooms.		
				Incorporate the existing Chess Club		
				and Horseshoe Club uses into the		
				new uses, as appropriate		
Aquatics Compl	ex NI/A	N1/A	Dropood any sting and the	A courier land if the langting is not		
Uptown	N/A	N/A	Proposed aquatics complex	Acquire land if the location is not		
Aqualics			determined within the	and construct an aquatics complex		
Complex				and construct an aquatics complex,		
			optown community.	sized to meet community needs,		
				universal access and water		
				universal access and Water		
				and a therapeutic pool, and a pool		
				and a therapeutic pool, and a pool		

	Table 6.12-3					
	P	opulatio	n-Based Parks and Recreatio	on Facilities		
Parks/	Existing	Future	Parks and Recreation			
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities		
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations		
				house including locker rooms, staff		
				offices, and equipment storage		
				facilities.		
Joint Use Facilit			Evicting joint use facilities			
Elementary	0.00		consisting of turf multi			
School			purpose playfield, multi			
SCHOOL			purpose playfield, filuiti-			
			bardasana far saurt samas			
			naruscape for court games			
			pursuant to long-term lease			
			agreement. (Facility totals			
			1.82 acres and is shared			
			with North Park (0.96 acres)			
Florenco		1 2	Branaged igint use facility at	Design and construct is intuge		
Florence		1.2	Proposed joint use facility at	Design and construct joint use		
Elementary			the school site.	facilities, including multi-purpose		
School				courts. Pursue a pedestrian		
				connection between the joint use		
				area and the future Mission Hills		
				Library site. Enter into a joint Use		
<u> </u>		1.00		Agreement with the SDUSD.		
Grant K-8		1.00	Proposed joint use facility at	Design and construct joint use		
School			school site.	facilities, including multipurpose		
				playfield, hard courts, and a		
				gymnasium. Enter into a Joint Use		
Deservelt	2.10			Agreement with the School District.		
Roosevelt	2.19		Existing joint use facilities			
Middle School			consisting of turf multi-			
			purpose playfields and			
			perimeter running track			
			pursuant to long-term			
			agreement.			
Irails: Useable	acres crec	lit for trail	s was determined by multiplyi	ng the linear footage of trail by 12° 0"		
Bankors Hill	eu by one	0 20	Proposed trail amonities for	Design and construct trail		
		0.59	the existing trails 1 400	amonition such as bonchos		
			line existing u dils, 1,400	interpretive signs, protective		
IIdli			Hill Open Space A partian	foncing nativo landecaning trach		
			of the proposed trail is	and recycling containers, and		
			or the proposed trail is	and recycling containers, and		
			nucated on undeveloped	overlooks, where needed and		
			public right-of-way.	appropriate for the trail type, as		
				determined and approved by City.		

Table 6.12-3					
Population-Based Parks and Recreation Facilities					
Parks/	Existing	Future	Parks and Recreation		
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities	
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations	
Buchannan		0.41	Proposed trail amenities for	Design and construct trail	
Canyon Open			the existing trails, 1,500	amenities, such as benches,	
Space Trail			lineal feet, in the Buchannan	interpretive signs, protective	
			Canyon Open Space. A	fencing, native landscaping, trash	
			portion of the proposed trail	and recycling containers, overlooks,	
			is located on undeveloped	etc., where needed and	
			public right-of-way.	appropriate for the trail type, as	
				determined and approved by City.	
Curlew Canyon		0.14	Proposed trail amenities for	Design and construct trail	
Open Space			the existing trails, 500 lineal	amenities, such as benches,	
Trail			feet, in the Curlew Canyon	interpretive signs, protective	
			Open Space.	fencing, native landscaping, trash	
				and recycling containers, and	
				overlooks, where needed and	
				appropriate for the trail type, as	
				determined and approved by City.	
Cypress		1.16	Proposed trail amenities for	Design and construct trail	
Canyon /			the existing trails, 4,200	amenities such as benches,	
Marston Open			lineal feet, in the Cypress	interpretive signs, protective	
Space			Canyon/Marston Open	fencing, native landscaping, trash	
			Space.	and recycling containers, overlooks,	
				etc., where needed and	
				appropriate for the trail type, as	
Llagnica Daint		0.20	Dranged trail amonities for	determined and approved by City.	
		0.30	Proposed trail amenities for		
			lineal fact in the Hespice	amenicies, such as such as	
Irdii			Reint Open Space	protostivo foncing, nativo	
			Point Open space.	protective rending, native	
				containers overlooks at where	
				needed and appropriate for the	
				trail type, as determined and	
				approved by City	
Manle Canvon		0 771 32	Approximately 2 800 linear	Design and construct new trails	
Open Space		0.77 1.52	feet of existing and 2 020	approximately 2 020 linear feet	
Trail			linear feet of new trails	that will connect to public right-of-	
			located in the Manle Canyon	ways and design and construct	
			Open SpaceProposed trail	trail amenities along the existing	
			amenities for the existing	and new trails such as protective	
			trails. 2.800 lineal feet in the	fencing, native landscaping trash	
			Maple Canvon Open Snace	and recycling containers.	
				interpretive signs, overlooks, etc.	

Table 6.12-3					
Population-Based Parks and Recreation Facilities					
Parks/	Existing	Future	Parks and Recreation		
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities	
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations	
				the trail type, as determined and approved by the City. Design and construct trail amenities , such as benches, interpretive signs,	
				protective fencing, native landscaping, trash and recycling containers, and overlooks, where needed and appropriate for the trail type, as determined and approved by City.	
Mission Hills Open Space Trail		0.41	Proposed trail amenities for the existing trails, 1,480 lineal feet, in the Mission Hills Open Space. A small portion of the proposed trail is located on privately owned property.	Design and construct trail amenities, such as protective fencing, native landscaping, trash and recycling containers, overlooks, etc., where needed and appropriate for the trail type, as determined and approved by City. Acquire a recreation easement for public use of the privately owned portion of the trail.	
University Heights Open Space Trail		0.08	Proposed trail amenities for the existing trails, 300 lineal feet, in the Buchannan Canyon Open Space.	Design and construct trail amenities such as benches, interpretive signs, protective fencing, native landscaping, trash and recycling containers, overlooks, etc., where needed and appropriate for the trail type, as determined and approved by City.	
Portion of Resource-Based Parks					
Freedom Park (within Balboa Park)		2.29	Proposed park located on the north side of the War Memorial Building on Park Boulevard.	Design and construct active and passive recreation amenities and support facilities, such as seating/picnicking, security lighting, walkways, and landscaping.	
Nate's Point Off-leash Dog Area (within Balboa Park)		2.75	Off-leash dog area at Laurel Street and Balboa Drive.	Design and construct off-leash dog area upgrades, such as drinking fountains, site furniture, security lighting, walkways, and landscaping.	
Pershing Recreation		3.45	Proposed community park/sports complex located	Design and construct a community park/ sports complex with active	

6.12 Public Services and Facilities

Table 6.12-3					
Population-Based Parks and Recreation Facilities					
Parks/	Existing	Future	Parks and Recreation		
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities	
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations	
Complex			at the corner of Pershing	recreation facilities consistent with	
(within Balboa			Drive and 26th Street. This	the recommendations in the	
Park)			site is currently used by City	Balboa Park East Mesa Precise	
			Central Operations Station	Plan, subsequent to relocation of	
			facilities. This 15-acre facility	non-park, City facilities.	
			will be shared with		
			Downtown, Greater North		
			Park, Greater Golden Hill,		
Dracidia		2.04	And Optown.	Design and construct additional	
Presidio		5.64	an Casey Way and Presidie	regreation amonities such as a	
Bark (within			Drive within Presidio Park	nichic sholtor, accossible wallways	
Prosidio Park)			Existing uses include a	interpretive signs etc	
			children's play area picnic		
			areas and a comfort station		
Ouince Street		2.30	Proposed mini-park located	Design and construct passive	
Mini-Park			at the southeast corner of	recreation amenities, such as	
(within Balboa			the intersection of Balboa	seating/picnicking, security lighting,	
Park)			Drive and the Quince Street/	walkways, and landscaping.	
			SR-163 northbound exit		
			ramp.		
Sixth Avenue	3.55	<u>3.55</u>	Proposed linear park located	Design and construct amenities	
Linear Park –			between Sixth Avenue and	consistent with the approved	
North (within			Balboa Drive, from Upas	General Development Plan for the	
Balboa Park)			Street to Quince Street.	Sixth Avenue Playground located	
				between Thorn and Spruce streets,	
				as well as additional passive	
				recreational amenities in the	
				adjacent areas such as seating,	
				picnicking, drinking fountains,	
				security lighting, walkways and	
Driverte ha Orange				landscaping.	
Privately Owned Park Sites					
Non-Traditional Park Sites					
Normal Street		1 60	Proposed linear park located	Design and construct a variety of	
Linear Park		1.00	within the Normal Street	passive recreational and	
			right-of-way. including the	community uses, including a	
			medians.	children's play area and flexible	
				opportunities for the weekly	
				farmer's market and other	
				community events. Coordinate	

Table 6.12-3 Population-Based Parks and Recreation Facilities				
Parks/	Existing	Future	Parks and Recreation	
Recreation	Useable	Useable	Facilities	Parks and Recreation Facilities
Facilities	Acreage	Acreage	Locations and Descriptions	Recommendations
				with the State's redevelopment
				efforts of the Department of Motor
				Vehicles office site, where
				appropriate.
Facility or Building Expansion or Upgrade				
None				

Table 6.12-4 Summary of Existing and Proposed Population-Based Parks and Recreation Facilities					
Population-Based Parks	Useable Acres				
Existing Population-based Parks and Park Equivalencies	18.21<u>14.66</u> acres				
Proposed Population-based Parks and Park Equivalencies	36.85<u>35.31</u> acres				
Total Existing and Proposed Population-based Parks and Equivalencies	55.06-<u>58.82</u> acres				
Population-based Park Requirements at full community development	155.96 acres				
Population-based park deficit at full community development	100.90 <u>97.14</u> acres				
Recreation Centers	Square Feet				
Existing Recreation Centers:	0				
Proposed Recreation Center: Grant K-8 School Gymnasium	10,454 SF				
Proposed Recreation Center: Redwood Recreation Center	10,456 SF				
Proposed Recreation Center: Uptown Recreation Center	17,000 SF				
Total Existing and Proposed Recreation Centers	37,910 SF				
Recreation Center Requirement at full community development	37,910 SF				
Aquatic Complex	Percentage				
Existing Aquatic Complexes	0				
Proposed Aquatic Complexes: Uptown Aquatic Complex	1.11				
Total Existing and Proposed Aquatic Complexes	1.11				
Aquatic Complexes Requirement at full community development	1.11				

A total of 155.96 acres of population-based parks would be needed to serve Uptown at full community development, of which <u>18.2114.66</u> acres currently exist. Through the proposed Uptown CPU effort, City staff and community members have identified <u>3635.31.85</u> acres of proposed new population-based park land and park equivalency sites within and adjacent to the Uptown community that, when implemented, would reduce the existing population-based park deficit to <u>100.9097.14</u> acres.

Build-out of the proposed Uptown CPU would add additional population to the CPU area and the CPU area would continue to have a deficit of population-based parks at build-out, which would be an adverse impact. Similarly, until the identified recreational facilities such as the aquatic complex are constructed, the deficit and the associated adverse impact would continue. Future development proposed within the Uptown CPU area would be subject to payment of DIF for public facilities

financing in accordance with Municipal Code Section 142.0640. The Uptown GPU includes an Impact Fee Study that would define applicable DIF fees for future development including fees for park funding. However, fees would not be adequate to address the extent of the parkland deficit. Payment and receipt of DIF funds is contingent on future development, and proposed fees are not designed to fully fund and address the parkland deficit.

The proposed Uptown CPU Recreation Element provides a policy framework that supports acquisition and development of new public parks and park equivalencies and encourages new private development to include recreational facilities.

Thus, although the existing and projected deficit in population-based parks is adverse, impacts associated with the construction of park facilities would be less than significant at the program level. Implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of population-based parks in the CPU area, but does not propose construction of new facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact associated with the construction of new facilities in order to maintain performance objectives for parks.

c. Fire/Life Safety Protection

With the implementation of the proposed Uptown CPU and associated discretionary actions there would be an increase in overall population, which could result in a change in response times. However, future facilities would be planned based on adopted General Plan Public Facilities Element standards detailed in Chapter 5.0, Regulatory Framework (Section 5.12.1.3) of this PEIR. The proposed Uptown CPU and associated discretionary actions do not propose the construction of fire/life safety facilities. However, the proposed Uptown CPU contains a policy framework that addresses maintaining the high level of fire protection throughout the Uptown community. Additionally, as future development is proposed within the Uptown CPU area, individual projects would be subject to payment of DIF, which would provide facilities financing in accordance with Municipal Code Section 142.0640. The Uptown GPU-CPU includes a comprehensive update to the existing-Impact Fee Study that will define applicable DIF fees for future development, including funding for fire/life safety facilities.

At the program level the proposed increase in population would not require that the Fire-Rescue Department construct new facilities. Any expansion construction of existing facilities or the development of a new facility would be subject to separate environmental review at the time design plans are available. Therefore, at the program-level of analysis provided in this PEIR, impacts related to the expansion <u>of new facilities</u> would be less than significant.

As noted in the proposed Uptown CPU, although no additional fire stations are planned within the community, a new replacement facility at the Station 5 location would be undertaken to meet the current needs of the local neighborhood and the station's personnel.

d. Libraries

As identified above, two libraries currently serve the Uptown community. Correspondence with the Library Department (Appendix J) confirms that the City does not require the construction of any additional facilities to meet library service requirements of the proposed Uptown CPU. While not required, there are plans to build an approximately 25,0000-square-foot new library, which would result in an exceedance of the recommended minimum branch library size requirement of 15,000 square feet. The new library would proceed as a separate action from the proposed Uptown CPU and associated discretionary actions and would be required to undergo its own environmental review. The proposed CPU Public Facilities, Services, and Safety Element policy framework supports expanded library facilities, which the new Mission Hills/Hillcrest Branch Library would address. Any expansion construction of existing facilities or the development of a new facility would be subject to separate environmental review at the time design plans are available. Therefore, since the proposed Uptown CPU and associated discretionary actions do not include the construction of library facilities and facility needs would be met within the Uptown CPU area, impacts related to library facilities would be less than significant.

e. Schools

Student generation is based on housing units. For the Uptown community, based on 2010 Census data from San Diego Association of Governments (SANDAG), there are 25,124 existing units. An additional 12,581 residential units are proposed with the CPU. Per correspondence with San Diego Unified School District (SDUSD; April 2014), student generation rates vary based on the type of project, number of units, bedroom mix, affordable or senior housing component, proximity to schools and other amenities, neighborhood, and other factors. There are no district standard or school-specific rates.

Typically, to provide student generation rates for a new project, SDUSD demographers would research similar nearby developments and their student generation rates as a guide for how many students a new project may generate. For the proposed Uptown CPU and associated discretionary actions, however, many factors are not yet determined, such as the specific type of housing and bedroom mix that may be constructed with the potential increase in housing stock at some future point in time. To estimate the number of students potentially generated by future build-out of the Uptown Community Plan, SDUSD demographers referenced the number of existing housing units in the Uptown community and the current number of students who reside in Uptown (based on SDUSD data), to determine the current Uptown communitywide student generation rates. This information is summarized in Table 6.12-5.

Table 6.12-5 Uptown Student Generation Rates from Existing Housing Units				
	2013-2014 Students	Student Generation Rate		
Number of Existing Units	(K-5, 6-8, 9-12, and K-12 total)	(per unit)		
	K-5: 803	K-5: 0.032		
	6-8: 695<u>299</u>	6-8: 0.012		
25,125	9-12: 377	9-12: 0.015		
	K-12: 1,479	K-12: 0.059		
Based on the number of additional units proposed by the proposed Uptown CPU and associated discretionary actions and student generation rates included in Table 6.12-5, potential student generation for future build-out of Uptown is shown in Table 6.12-6. The generation rates are shown as a range. The current generation rate is the low range and the high range is double the low range (current generation rate). A key assumption is that future additional housing units will generate students at a rate similar to current housing units; this is represented by the low range. If future additional housing units are significantly different from the current units in terms of student generation, the number of students could be higher, as indicated by the high range.

Table 6.12-6				
Uptown Potential Student Generation Rates from Future Additional Housing Units				
Number of Additional Units	Potential Student Generation Rates	Number of Potential Students		
12,581	K-5: 0.032-0.064	K-5: 403-805		
	6-8: 0.012-0.024	6-8: 151-302		
	9-12: 0.015-0.030	9-12: 189-377		
	K-12: 0.059-0.118	K-12: 743-1,484		

SDUSD demographers indicated that the cumulative potential increase in students from the number of future additional housing units suggested in the proposed Uptown CPU and associated actions would likely impact district schools to the point of reaching or exceeding capacity. Therefore, new or expanded school facilities would likely be needed.

Government Code Section 65995 and Education Code Section 53080 authorize school districts to impose facility mitigation fees on new development to address any increased enrollment that may result. Senate Bill (SB) 50, enacted on August 27, 1998, significantly revised developer fee and mitigation procedures for school facilities as set forth in Government Code Section 65996. The legislation holds that an acceptable method of offsetting a project's effect on the adequacy of school facilities is payment of a school impact fee prior to issuance of a building permit. Once paid, the school impact fees would serve as mitigation for any project-related impacts to school facilities. As such, the City is legally prohibited from imposing any additional mitigation. The school district will be responsible for potential expansion or development of new facilities, which would undergo a separate environmental review when specific facilities are planned. Therefore, impacts to schools resulting from future development would be less than significant through implementation of SB 50 (City of San Diego 2011).

f. Maintenance of Public Facilities

The proposed Uptown CPU Public Facilities, Services, and Safety Element contains a policy framework related to the maintenance of public facilities. Proposed policies support maintenance assessment district programs, and road and water facility improvements. Additionally, as future development is proposed within the Uptown CPU area, individual projects would be subject to payment of DIF, which would provide facilities financing in accordance with Municipal Code Section 142.0640. The Uptown CPU includes a comprehensive update to the existing Impact Fee Study that will define applicable DIF fees for future development. The proposed Uptown CPU and associated discretionary actions do not propose any construction of specific facilities. When future facilities are

constructed they would require a separate environmental review. Thus, public facilities impacts would be less than significant.

Cumulative Impact Analysis

Some of the City's existing built areas have existing infrastructure deficiencies and would require capacity improvements to serve additional population. Therefore, it is anticipated that new or improved public services and facilities infrastructure would be required to meet the needs of the City's future growth occurring through infill and redevelopment as well as remaining on vacant and developable lands. However, as discussed in this section, implementation of the proposed Uptown CPU and associated discretionary actions do not include construction of any specific public facilities or services. The proposed Uptown CPU includes policies that would support improvements to public facilities and includes a proposed Impact Fee Study that would specify the DIF applicable to future development within the CPU area. Similarly, the proposed North Park and Golden Hill CPUs do not propose specific facility improvements.

The specific public facilities improvements that would be constructed in the cumulative area of Uptown, North Park, and Golden Hill and the degree of future impacts and applicability, feasibility, and success of future mitigation measures cannot be adequately known at this program level of analysis. However, each future facility improvement would undergo a separate environmental review and is not intended to be analyzed for purposes of this proposed Uptown CPU. Thus, cumulative impacts related to public facilities would be less than significant.

6.12.4 Significance of Impacts

Regarding police protection, the proposed Uptown CPU and associated discretionary actions do not include construction of new police facilities. As population growth occurs and the need for new facilities is identified, any future construction of police facilities would be subject to a separate environmental review at the time design plans are available. Therefore, implementation of the proposed Uptown CPU and associated discretionary actions would result in less than significant environmental impacts associated with the construction of new facilities in order to maintain service ratios, response times, or other performance objectives related to police services, and no mitigation is required.

Regarding park and recreational facilities, there is an existing and projected deficit in populationbased parks, which is an adverse impact but not considered significant at the program level. Implementation of the proposed Uptown CPU and associated discretionary actions would provide policy support for increasing the acreage of population-based parks in the Uptown CPU area but do not propose construction of new facilities. Thus, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to parks and recreation, and no mitigation is required.

Regarding fire/life safety protection, implementation of the proposed Uptown CPU and associated discretionary actions would result in an increase in overall population which could result in a change in fire-rescue response times and a demand for new or expanded facilities. However, any expansion construction of existing facilities or the development of a new facility would be subject to separate

environmental review at the time design plans are available. Therefore, at the impacts associated with police/life safety facilities would be less than significant, and no mitigation is required.

Although a new library is planned for the Uptown CPU area, the proposed Uptown CPU and associated discretionary actions do not include construction of library facilities. Development of a new facility would be subject to separate environmental review at the time design plans are available. Therefore, impacts related to library facilities would be less than significant, and no mitigation is required.

Regarding school facilities, future residential development that occurs in accordance with the proposed Uptown CPU and associated discretionary actions would be required to pay school fees as outlined in Government Code Section 65995, Education Code Section 53080, and Senate Bill 50 to mitigate any potential impact on district schools. The City is legally prohibited from imposing any additional mitigation related to school facilities through implementation of Senate Bill 50, and the school district would be responsible for potential expansion or development of new facilities. Therefore, impacts to schools would be less than significant, and no mitigation is required.

The proposed Uptown CPU contains policies to address the maintenance and improvement of public facilities. Impacts would therefore be less than significant, and no mitigation is required.

6.12.5 Mitigation Framework

No mitigation is required for police protection, parks and recreation facilities, fire services, library services, schools, and maintenance of public facilities. While the implementation of the proposed Uptown CPU and associated discretionary actions would result in the continuation of a park deficit, which is an adverse impact it is less than significant. No mitigation is required.

6.13 **Public Utilities**

This section analyzes the impacts of the proposed Uptown Community Plan Update (CPU) and associated discretionary actions on existing public utilities, including those for water, sewer, storm water communications systems, solid waste, and energy. This section includes a discussion of the Water Supply Assessment (WSA) prepared by the City's Public Utilities Department (PUD) (May 2015), which is included as Appendix K to this Program Environmental Impact Repot (PEIR).

6.13.1 Existing Conditions

A discussion of existing conditions for water supply, sewer, storm water, solid waste, energy, and communications in the Uptown CPU area is provided in Chapter 2.0. The existing regulatory framework is summarized in Chapter 5.0. Specific discussion relating to the water supply assessment for Uptown is presented below.

6.13.2 Significance Determination Thresholds

Based on the City's Significance Determination Thresholds, which have been adapted to guide a programmatic analysis of the proposed Uptown CPU and associated discretionary actions, impacts related to water, sewer, solid waste, energy, and communications would be significant if the proposed Uptown CPU and associated discretionary actions would:

- 1) Result in the use of excessive amounts of water beyond projected available supplies;
- 2) Promote growth patterns resulting in the need for and/or provision of new or physically altered utilities, the construction of which could cause significant environmental impacts in order to maintain service ratios, or other performance objectives;
- 3) Result in impacts to solid waste management, including the need for construction of new solid waste landfills; or result in a land use plan that would not promote the achievement of a 75 percent target for waste diversion and recycling as required under AB 341.

6.13.3 Impact Analysis

Issue 1 Water Supply

Would the project use excessive amounts of water beyond projected available supplies?

A WSA was prepared for the proposed Uptown CPU and associated discretionary actions to assess whether sufficient water supplies are, or will be, available to meet the projected water demands of the proposed Uptown CPU and associated discretionary actions. Because no subdivision of land is proposed as part of this project, this WSA was prepared in compliance with the requirements of Senate Bill 610. The WSA includes, among other information, identification of existing water supply entitlements, water rights, water service contracts, or agreements relevant to the identified water supply for the proposed CPU; and quantities of water received in prior years pursuant to those entitlement, rights, contracts, and agreements. The WSA evaluated water supplies that are, or will be, available during a normal, single- dry year, and multiple-dry year (20-year) period, to meet the estimated demands of the proposed Uptown CPU and associated discretionary actions.

Metropolitan Water District (MWD) and the San Diego County Water Authority (SDCWA) have developed water supply plans to improve reliability and reduce dependence upon existing imported supplies. MWD's Regional Urban Water Management Plan (RUWMP) and Integrated Water Resources Plan, and the SDCWA's 2010 Urban Water Management Plan (UWMP) and annual water supply report include water infrastructure projects that meet long-term supply needs through securing water from the State Water Project, Colorado River, local water supply development, and recycled water.

As discussed in the WSA, the City's 2010 UWMP demonstrates that there will be sufficient water supplies available to meet demands for existing and planned future developments that are projected to occur by 2035. Based on a normal water supply year, the estimated water supply projected in five-year increments for a 20-year projection will meet the City's projected water demand of 240,472 acre-feet in 2015; 260,211 acre-feet in 2020; 276,375 acre-feet in 2025; 288,481 acre-feet in 2030; and 298,860 acre-feet in 2035. Based on a single-dry year forecast, the estimated water supply will meet the projected water demand of 255,040 acre-feet in 2015; 276,526 acre-feet in 2020; 293,895 acre-feet in 2025; 307,230 acre-feet in 2030; and 318,586 acre-feet in 2035. Based on a multiple-dry year, third year supply, the estimated water supply will meet the projected demands of 281,466 acre-feet in 2015; 303,004 acre-feet in 2020; 322,166 acre-feet in 2025; 334,720 acre-feet in 2030; and 346,823 acre-feet in 2035.

As demonstrated in the WSA (Appendix K to this PEIR), there is sufficient water planned to supply the proposed Uptown CPU's estimated annual average usage. The projected water demands of the project are 7,879,499 gallons per day (gpd) or 8,825 acre feet per year. In the City's 2010 UWMP, the planned water demands of this project site are 7,879,499 gpd or 8,825 acre feet per year. As a result, the water demand resulting from the proposed Uptown CPU would result in no unforeseen demands.

In summary, the WSA concluded that the proposed Uptown CPU is consistent with the water demands assumptions included in the regional water resource planning documents of the SDCWA

and MWD. Current and future water supplies, as well as the actions necessary to develop these supplies, have been identified in the water resources planning documents of the PUD, the SDCWA, and MWD to serve the projected demands of the proposed Uptown CPU area, in addition to existing and planned future water demand of the PUD. Therefore, impacts related to water supply would be less than significant.

Issue 2 Utilities

Would the project promote growth patterns resulting in the need for and/or provision of new or physically altered utilities, the construction of which could cause significant environmental impacts in order to maintain service ratios, or other performance objectives?

The City's General Plan calls for future growth to be focused into mixed-use activity centers linked to the regional transit system. Implementation of the proposed Uptown CPU would result in infill and redevelopment occurring in selected areas within the CPU area. The City's existing built areas are currently served by storm water, wastewater, and water infrastructure, and various communications systems; however, some of the City's built areas, including those within the Uptown community, have existing infrastructure deficiencies and would require capacity improvements to serve the existing and projected population. The following is a program-level analysis of the significance of impacts for each applicable utility.

a. Storm Water

Because the Uptown CPU area is highly impervious, the volume or rates of runoff are not likely to be increased by new development. It is more likely that the volume and rate of runoff could be slightly decreased due to implementation of new storm water quality regulations as new development occurs, which requires implementation of LID practices that retain a portion of storm water on-site for infiltration, re-use, or evaporation.

No storm drains, or other community-wide drainage facilities, are proposed for construction in conjunction with adoption of the proposed Uptown CPU and associated discretionary actions. However, plans and programs are in place Citywide to maintain and upgrade the storm water system. As individual development projects are implemented in accordance with the proposed Uptown CPU and associated discretionary actions, localized improvements to the storm water system would be required as part of the project design and review. All storm water facilities constructed in conjunction with future development would be reviewed for consistency with the City's Storm Water Standards and other applicable requirements.

All future projects would be required to adhere to General Plan and proposed Uptown CPU policies and implementing regulations, and are required to comply with the City's Storm Water Standards<u>in</u> <u>place at the time future development is proposed</u>. Proposed Uptown CPU policies include those implementing Best Management Practices (BMPs) and Low Impact Development (LID) strategies to manage storm water and urban runoff, as well as those promoting proper maintenance of existing storm water infrastructure, thus reducing potential strains on the City's storm water system and ensuring the long-term viability of existing facilities. While the details of storm water infrastructure improvements would depend on the actual design of a future project, strict adherence to existing storm water regulations, conformance with General Plan and proposed Uptown CPU policies, and project-specific review under California Environmental Quality Act (CEQA) for discretionary projects would assure that significant adverse effects to the City's storm water system, as well as significant impacts associated with the installation of new storm water infrastructure, would be avoided.

b. Sewer

The proposed Uptown CPU and associated discretionary actions does not propose any specific development but provides the framework for future growth. No new sewer collection or wastewater treatment facilities are proposed in conjunction with the proposed Uptown CPU and associated discretionary actions. Any future development would be required to comply with the City's Municipal Code regulations regarding sewers and wastewater facilities (Chapter 6, Article 4) and would be required to follow the City's Sewer Design Guidelines. Adherence to existing regulations and standards would ensure that flows from new projects would not adversely affect downstream conveyance systems and that previous studies have accounted for those flows in the design of the downstream conveyance system.

Given ongoing and planned improvements to the system, existing regulations and guidelines to ensure adequate capacity, and proposed Uptown CPU policies to support capital improvements, impacts associated with the wastewater system would be less than significant.

c. Water Distribution

The potable water distribution system is continually upgraded and repaired on an ongoing basis through the City's Capital Improvements Program. These improvements are determined based on continued monitoring by the Public Works Department (PWD) Engineering Division to determine remaining levels of capacity. The PWD Engineering Division plans its capital improvement projects several years prior to pipelines actually reaching capacity. Such improvements would be required of the water system regardless of the implementation of the proposed Uptown CPU and associated discretionary actions.

As future development takes place in the Uptown CPU area, demand for water is likely to increase and create a potential need to increase sizing of existing pipelines and mains. This would be reviewed on a project-by-project basis. Additionally, the proposed Uptown CPU contains policies supporting water conservation and water-wise practices. All proposed public water facilities would be required to be designed and constructed in accordance with established criteria in the City's Water Facility Design Guidelines, Land Development Code, and any other applicable regulations, standards, or practices. Future development under the proposed Uptown CPU and associated discretionary actions would be generally consistent with the existing urban growth patterns and the necessary infrastructure improvements to the water system would be consistent with what is necessary for new development and to maintain the existing system. The proposed Uptown CPU contains a policy (PF-1.9) to support the systematic improvement and gradual replacement of water facilities. Given that future improvements to water facilities in accordance with the proposed Uptown CPU would be consistent with existing development and capital improvements planning, impacts would be less than significant.

d. Communications

Private utility companies currently provide communications systems within the Uptown CPU area. Future siting of communications infrastructure would be in accordance with the Land Development Code, including section 141.0420 regulating wireless communications facilities, as well as the City's Wireless Communications Facilities Guidelines, which seek to minimize visual impacts. Adhering to General Plan policies supporting the City's undergrounding program would also ensure that visual impacts of new facilities are minimized. Similarly, the proposed Uptown CPU contains policies supporting utility undergrounding and undergrounding is currently underway in the Uptown community. Any construction of communications systems associated with future development would occur in accordance with the City's permitting processes and construction standards to avoid or minimize impacts on environmentally sensitive habitat areas and landforms through siting, grading or excavation, and erosion. Thereby, impacts associated with communications facilities from build-out of the proposed Uptown CPU and associated discretionary actions would be less than significant.

Issue 3 Solid Waste and Recycling

Would the proposed project result in impacts to solid waste management, including the need for construction of new solid waste landfills; or result in a land use plan that would not promote the achievement of a 75 percent waste diversion as targeted in AB 341 and the City's Climate Action Plan?

The California Department of Resources Recycling and Recovery (CalRecycle) provides estimates of solid waste generation rates for different types of land uses. These rates estimate the amount of solid waste created by residences or businesses over a certain amount of time (day, year, etc.). Waste generation rates include all materials discarded, whether or not they are later recycled or disposed of in a landfill, since under state law the total amount of waste "generated" is considered to be the sum of the waste "disposed of" plus the waste "diverted" from disposal. Waste generation rates can be used to estimate the impact of new development on the local solid waste infrastructure, although it should be noted that impacts to solid waste infrastructure are not necessarily the amount of waste but whether any increase would require the development of new facilities. Since the majority of waste is managed through waste diversion, solid waste facilities include those necessary to provide composting, recycling, and other collection, separation, and diversion services. Furthermore, it is specifically the amount of waste remaining for disposal that is considered for compliance with the City's Climate Action Plan and has the greatest potential for impacts associated with greenhouse gas emissions.

Future projects that could occur in the Uptown community with the implementation of the proposed Uptown CPU and associated discretionary actions would be required to comply with City regulations, including the City's Recycling Ordinance (updated July 2015). In addition, a Waste Management Plan (WMP) would be required for any project which exceeds the City's threshold, currently the generation of 60 or more tons of solid waste for projects of 40,000 square feet or more. The WMP shall include measures to provide sufficient interior and exterior storage space for refuse and recyclable materials, and measures to handle landscaping and green waste materials associated with the occupancy of the proposed development. In tandem with the WMP, all new development projects must comply with the City's Construction and Demolition Ordinance and Section 142.0801 et seq. of the Land Development Code, which outlines the requirements for refuse and recyclable materials storage.

The General Plan addresses waste management in Policies PF-I.1 through PF-I.5, focusing on waste recycling and diversion of materials in PF-I.2. The proposed Uptown CPU would result in a less than significant impact to existing recycling operations within the proposed Uptown CPU area and surrounding areas, and would not affect the City's overall ability to attain a 75 percent recycling target as required under Assembly Bill 341. Additionally, the City has adopted a Zero Waste Plan, which would result in 70 percent waste diversion by 2020, 90 percent waste diversion by 2035, and 100 percent diversion by 2040. Furthermore, mandatory compliance with the San Diego Municipal Code and Recycling Ordinance for all new development projects would continue to reduce solid waste generation and increase recycling efforts, thereby resulting in a less than significant impact.

Cumulative Impacts

a. Water Supply

The WSA prepared for the proposed Uptown CPU and associated discretionary actions concluded that the proposed Uptown CPU and associated discretionary actions would be consistent with the water demand assumptions included in the regional water resource planning documents of the SDCWA and the MWD. Furthermore, current and future water supplies, as well as the actions necessary to develop these supplies, have been identified in the water resources planning documents of the PUD, the SDCWA, and MWD to serve the projected demands of the proposed Uptown CPU area, in addition to existing and planned future water demand of the City. Additionally, the proposed Uptown CPU contains policies intended to ensure that no excessive water use takes place, encourage water conservation and reclamation, and ensure the continued operability of existing infrastructure. Thus, cumulative impacts related to water supply would be less than significant.

b. Utilities

Implementation of the General Plan and proposed Uptown CPU policies and compliance with federal, state, and local regulations would preclude incremental impacts associated with new construction of, or improvements to, public utilities infrastructure. These requirements would apply to development within the Uptown CPU area and surrounding communities to ensure that adverse impacts related to the provision of utilities does not occur. Mandatory compliance with City standards for the design, construction, and operation of storm water, water, and wastewater infrastructure (including environmental review) would preclude significant cumulative environmental impacts. As a result, the proposed Uptown CPU and associated discretionary actions would result in a less than significant cumulative impact associated with storm water, water, water, water, and communication systems.

c. Solid Waste

Build-out of the proposed Uptown CPU and associated discretionary actions combined with buildout of surrounding communities would generate solid waste through demolition/construction and ongoing operations that would increase the amount of solid waste generated within the region. Future projects within the Uptown CPU area and Citywide, would be required to comply with City regulations regarding solid waste, including those intended to divert solid waste from the Miramar Landfill to preserve capacity. Compliance with the Municipal Code and consistency with the General Plan and applicable Community Plan policies promoting waste diversion would serve to preserve solid waste capacity. Discretionary projects generating more than 60 tons of waste would be required to develop and implement WMPs targeting 75 percent waste diversion. Therefore, cumulative solid waste impacts would be less than significant.

6.13.4 Significance of Impacts

6.13.4.1 Water Supply

Based on the findings of the WSA, there is sufficient water supply to serve existing and projected demands of the proposed Uptown CPU and associated discretionary actions, and future water demands within the PUD's service area in normal and dry year forecasts during a 20-year projection. Therefore, no significant impacts to water supply are anticipated for the implementation of the proposed Uptown CPU and associated discretionary actions.

6.13.4.2 Utilities

a. Storm Water

Future projects would be required to exercise strict adherence to existing storm water regulations and conformance with General Plan and proposed Uptown CPU policies. Project-specific review under the Municipal Storm Water Permit and CEQA would assure that significant adverse effects related to the storm water system and the installation of storm water infrastructure would be avoided. Thus, impacts related to storm water facilities would be less than significant.

b. Sewer and Water Distribution

The proposed Uptown CPU acknowledges that upgrades to sewer lines are an ongoing process. These upgrades are administered by the PWD and are handled on project-by-project basis. Because future development of properties under the proposed Uptown CPU and associated discretionary actions would likely increase demand, there may be a need to increase sizing of existing pipelines and mains for both wastewater and water. The proposed Uptown CPU takes into consideration the existing patterns of development, and the update is a response to the community's needs and goals for the future. The necessary infrastructure improvements to storm water, wastewater, and water infrastructure would be standard practice for new development to maintain or improve the existing system in adherence to sewer and water regulations and conformance with General Plan and proposed Uptown CPU policies. Additionally, future discretionary projects would be required to

undergo project-specific review under CEQA that would assure that impacts associated with the installation of storm water infrastructure would be reduced to below a level of significance. Therefore, impacts to sewer and water utilities would be less than significant.

c. Communications

Given the number of private utility providers available to serve the proposed Uptown CPU area, there is capacity to serve the area. Impacts would be less than significant.

6.13.4.3 Solid Waste and Recycling

To ensure that waste generation and recycling efforts during construction and post-construction future land use occupancy and operation (i.e., residential, commercial, industrial, mixed-use, etc.) are addressed, a WMP shall be prepared for any project proposed under the proposed Uptown CPU and associated discretionary actions exceeding the threshold of 40,000 square feet or more. Implementation of these WMPs would ensure that future development project impacts would be considered less than significant. Non-discretionary projects proposed under the proposed Uptown CPU and discretionary actions, and discretionary projects that would fall below the 60 ton thresholds, would be required to comply with the San Diego Municipal Code sections addressing construction and demolition debris, waste and recyclable materials storage, and recyclable materials (and in the future organic materials) collection. Therefore, at this program level of review, the proposed Uptown CPU and associated discretionary actionary actionary actions would not require increased landfill capacity, and impacts associated with solid waste would be less than significant.

6.13.5 Mitigation Framework

All public utilities impacts would be less than significant; thus, no mitigation is required.

6.14 Health and Safety

This section describes potential human health and public safety issues related to the presence of hazardous materials and other hazards within the Uptown Community Plan Update (CPU) area, identifies pertinent regulatory standards, and evaluates potential impacts and associated mitigation requirements related to implementation of the proposed Uptown CPU and associated discretionary actions. KLR Planning conducted a GeoTracker search (May 2016) within the proposed Uptown CPU area. The results of that search are summarized in this section and included in Appendix L of this Program Environmental Impact Report (PEIR). Additionally, KLR Planning conducted a California Environmental Protection Agency (Cal EPA) search (May 2016) of Cortese List Data Resources, the results of which are included in this section as Table 6.14-1.

6.14.1 Existing Conditions

The existing environmental setting and regulatory framework are summarized in Chapters 2.0 and 5.0, respectively.

A search of federal, state, and local environmental regulatory agency databases was conducted in order to identify sites within the Uptown area that may have been impacted by hazardous materials or wastes. The search identified 68 documented release cases within Uptown, of which, only three cases are open (see Table 6.14-1). All of the identified sites were the site of either leaking underground storage tanks (LUSTs) or a cleanup program. LUST systems can pose a significant threat to groundwater quality.

The Site Cleanup Program regulates and oversees the investigation and cleanup of "non-federally owned" sites where recent or historical unauthorized releases of pollutants to the environment, including soil, groundwater, surface water, and sediment, have occurred. Sites in the program include, but are not limited to, pesticide and fertilizer facilities, rail yards, ports, equipment supply facilities, metals facilities, industrial manufacturing and maintenance sites, dry cleaners, bulk transfer facilities, refineries, and some brownfields. These releases are generally not from strictly petroleum underground storage tanks (USTs). The types of pollutants encountered at the sites are diverse and include solvents, pesticides, heavy metals, and fuel constituents to name a few. Properties with open cases represent a moderate to high risk of encountering contaminated soils or groundwater during potential future redevelopment. Closed release cases represent a low to moderate risk of encountering contaminated soils or groundwater during potential future redevelopment. However, cases which were closed in the 1990s may not meet current standards and may require additional investigation and/or remediation prior to redevelopment.

Table 6.14-1				
Hazardous Materials Sites in Uptown				
Site	Address	Program/Site Type	Status	
William W. Newkirk (Mobil Oil)	1809 Washington Street	LUST	Closed	
J M A N at the Charmer LLC	3625 India Street	LUST	Closed	
UNOCAL Serv Station #5738	3585 India Street	LUST	Closed	
India Chevron	3535 India Street	LUST	Closed	
7-Eleven Food Store #19974	3070 Reynard Way	LUST	Closed	
Streicher & Seeman Inc.	2553 State Street	LUST	Closed	
Harbor View Medical Center	120 Elm Street	LUST	Closed	
New Palace Associates	1814 5 th Avenue	LUST	Closed	
First International Bank/Land	2201 4 th Avenue	LUST	Closed	
Resident	135 W Kalmia Street	LUST	Closed	
Fifth & Laurel St. Apts	2400 5 th Avenue	LUST	Closed	
Fifth & Laurel St. Apts	2400 5 th Avenue	Cleanup Program Site	Closed	
Fifth Avenue Financial Center	2550 5 th Avenue	LUST	Closed	
Park Laurel Condominiums	2500 6 th Avenue	LUST	Closed	
Albert H Bennett	2961 1 st Avenue	LUST	Closed	
leff & Cindy Cavignac	4314 Altamirano Way	LUST	Closed	
925 Fort Stockton	925 Fort Stockton	LUST	Closed	
SDCITY-Fire Station #8	3974 Goldfinch Street	LUST	Closed	
SDCITY-Fire Station #8	3974 Goldfinch Street	Cleanup Program Site	Closed	
Sam Lee Auto Repair	724 W Washington Street	IUST	Closed	
Sam Lee Auto Repair	724 W Washington Street	Cleanup Program Site	Closed	
Mission Hills Automotive	308 W Washington Street	IUST	Closed	
Mission Hills Automotive	308 W Washington Street	Cleanup Program Site	Closed	
Hillcrest Smog & Repair	3864 1 st Avenue	LUST	Closed	
UCSD Medical Center	200 W Arbor Drive	LUST	Closed	
UCSD Medical Center	200 W Arbor Drive	LUST	Open	
San Diego Hospice	4311 3 rd Avenue	LUST	Closed	
San Diego Hospice	4311 3 rd Avenue	Cleanup Program Site	Closed	
Mission Dry Cleaners	105 Washington Street	LUST	Closed	
Great Western Bank – Mission Hills	333 Washington Street	LUST	Closed	
Hillcrest Shell	330 Washington Street	LUST	Closed	
7-Eleven Food Store #19526	3792 4 th Avenue	LUST	Closed	
Phil & Daniels Properties	5 th Avenue	LUST	Closed	
UNOCAL Service Stn #2196	3795 6 th Avenue	LUST	Closed	
UNOCAL Service Stn #2196	3795 6 th Avenue	Cleanup Program Site	Closed	
Chevron Standard Stations	3806 6 th Avenue	LUST	Closed	
AT&T Communications Inc.	3940 4 th Avenue	LUST	Closed	
Aladrays A-N-A	711 University Avenue	LUST	Closed	
Aladrays A-N-A	711 University Avenue	Cleanup Program Site	Closed	
SDCITY-Fire Station #05	3902 9 th Avenue LUST		Closed	
William/Sidney Price	1060 University Avenue	LUST	Closed	
7-Eleven Food Store #20551	1602 University Avenue	LUST	Closed	
Phil Maestas Chev-SVC #91362	4180 Park Boulevard	LUST	Closed	
Phil Maestas Chev-SVC #91362	4180 Park Boulevard	Cleanup Program Site	Closed	
Wells Fargo Bank	3830 Park Boulevard	LUST	Closed	
Wells Fargo Bank	3830 Park Boulevard	Cleanup Program Site	Closed	
Thao Auto Repair	3752 Park Boulevard	LUST	Closed	

Table 6.14-1 Hazardous Materials Sites in Uptown			
Site	Address	Program/Site Type	Status
Exclusive Cleaners	3740 Park Boulevard	LUST	Closed
Exclusive Cleaners	3740 Park Boulevard	Cleanup Program Site	Closed
SDUSD, Print Services	4100 Normal Street	LUST	Closed
San Diego Smog and Auto Repair	4664 Park Boulevard	LUST	Closed
Sharp Rees-Stealy Downtown Medical Center	1900 4 th Avenue	Cleanup Program Site	Closed
Mr. Robinson	3752 Park Boulevard	Cleanup Program Site	Closed
Sears Roebuck & Co.	1290 Cleveland Avenue	Cleanup Program Site	Closed
Uptown Cleaners	1020 University Avenue	Cleanup Program Site	Closed
Uptown District Shopping Center	940-1092 University Avenue	Cleanup Program Site	Open
Jerry Pinto	4014 Couts Street	Cleanup Program Site	Closed
Private Residence	1819 Sheridan Avenue	Cleanup Program Site	Closed
Capital Service	1111 Fort Stockton Drive	Cleanup Program Site	Closed
The Fort	1011 Fort Stockton Drive	Cleanup Program Site	Open
Residence	4066 Albatross Street	Cleanup Program Site	Closed
Liberty Cleaners	333 East University Avenue	Cleanup Program Site	Closed
Dry Cleaner	335 University Avenue	Cleanup Program Site	Closed
Hillcrest Square Ltd	500 University Avenue	Cleanup Program Site	Closed
Broadstone Balboa Park	3235 4 th Avenue	Cleanup Program Site	Closed
Cambridge Square	2720 4 th Avenue	Cleanup Program Site	Closed
Merrill Gardens at Bankers Hill, LLC	2567 2 nd Avenue	Cleanup Program Site	Closed
Calcare-Western/St Pauls Villa	2340 4 th Avenue	Cleanup Program Site	Closed
LUST = leaking underground storage tank.			
Bold text = open status.			

6.14.2 Significance Determination Thresholds

Based on the City's Significance Determination Thresholds, which have been adapted to guide a programmatic analysis of the proposed Uptown CPU and associated discretionary actions, a significant health and safety impact would occur if implementation of the proposed Uptown CPU and associated discretionary actions would:

- 1) Expose people or structures to a significant risk of loss, injury or death involving wildland fires, including when wildlands are adjacent to urbanized areas or where residents are intermixed with wildlands;
- 2) Result in hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within a quarter mile of an existing or proposed school;
- 3) Impair implementation of, or physically interfere with an adopted emergency response plan or emergency evacuation plan;

- 4) Be located on a site which is included on a list of hazardous materials sites complied pursuant to Government Code Section 65962.5 and, as a result, creates a significant hazard to the public or environment; or
- 5) Expose people or structures to a significant risk of loss, injury, or death from off- airport aircraft operations accidents.

6.14.3 Impact Analysis

Issue 1 Wildfire Hazards

Would the project expose people or structures to a significant risk of loss, injury, or death involving wildland fires, including when wildlands are adjacent to urbanized areas or where residences are intermixed with wildlands?

The City of San Diego receives limited precipitation; therefore, the potential for wildland fires represents a hazard, particularly on undeveloped properties or where development exists or can occur adjacent to open space or within close proximity to wildland fuels. As the proposed Uptown CPU and associated discretionary actions would maintain natural open space within undeveloped canyons, any development adjacent to this open space would be subject to a risk of fire hazards. Existing City policies and regulations would help reduce, but not eliminate, risks from wildfires. The City's General Plan contains goals and policies aimed at reducing the risks of wildfires that would be implemented by the City's Fire-Rescue Department.

The proposed Uptown CPU Public Facilities Services and Safety Element includes policies intended to reduce the risk of wildfire hazards. Policies are included that would prioritize the maintenance of a high level of fire protection throughout the community, particularly in the neighborhoods adjacent to natural open space and would emphasize modernization and/or replacement of facilities and equipment to meet the needs of the community or as firefighting technology becomes available. Policies would also support efforts by the City to educate and inform the community regarding fire prevention technique, particularly those related to brush management and wildland fires.

Regulations regarding brush management are summarized in Chapter 5.0 Regulatory Framework (Section 5.14) of this PEIR. Future development proposals would be reviewed for compliance with all City and Fire Code requirements aimed at ensuring the protection of people or structures from potential wildland fire hazards. Brush management regulations (San Diego Municipal Code Section 142.0412) would ensure that brush management is completed within 100 feet of a structure and requires new development to prepare brush management plans to demonstrate a 100-foot brush management area and identification of Zone 1 and Zone 2 brush management zones. Zone 1 is the area adjacent to structures where pavement or irrigated vegetation occurs and flammable materials are limited. Zone 2 is the area that extends beyond Zone 1 and typically consists of thinned native or naturalized vegetation. Implementation of the City's Brush Management Regulations would ensure future development that occurs within the Uptown CPU area would not expose people or structures to a risk of loss, injury or death involving wildfires. Impacts due to wildland fires would be less than significant.

Issue 2 Schools

Would the project result in hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within a quarter-mile of and existing or proposed school?

A GeoTracker search was undertaken (May 2016) to determine what, if any, potential exposure to hazardous materials occurs within one-quarter-mile of the existing schools. Four public schools are located within the Uptown community:

- Grant Elementary (K-8) located at 1425 Washington Place
- Florence Elementary (K-5) located at 3914 First Avenue
- Birney Elementary (K-5) located at 4345 Campus Avenue
- Roosevelt Junior High (6-8) located at 3366 Park Boulevard

The GeoTracker search identified one hazardous materials site in the Uptown CPU area which falls within one-quarter-mile of one of the community schools. Cleanup on the site is complete, and is marked as closed (see Appendix L for detailed GeoTracker site information). The identified and closed site is a private residence located near Grant Elementary School. In addition, the GeoTracker search identified Roosevelt Junior High as a Small Generator by Resource Conservation and Recovery Act Information (RCRAInfo). Hazardous waste information is contained in the RCRAInfo, a national program management and inventory system about hazardous waste handlers. In general, all generators, transporters, treaters, storers, and disposers of hazardous waste are required to provide information to regional and national Environmental Agencies. These agencies, in turn, pass on the information to regional and national Environmental Protection Agency (EPA) offices. This regulation is governed by the Resource Conservation and Recovery Act (RCRA), as amended by the Hazardous and Solid Waste Amendments of 1984.

There are three "open" sites within the Uptown CPU area, and none are within one-quarter mile of any schools. However, in addition to the public schools within the Uptown CPU area, there are numerous other private schools, charter schools, and day cares that could be located within oneguarter mile of a potentially contaminated site. However, any existing contaminated site identified within the hazardous materials database search would be required to undergo cleanup in accordance with applicable regulatory oversight agencies and any new development that involves contaminated property would necessitate the cleanup and/or remediation of the property in accordance with applicable requirements and regulations of local, state, and or federal requirements. No construction would be permitted to occur at such sites until a "no further action" clearance letter from the County Department of Environmental Health, or similar determination is issued by the City's Fire Rescue Department, California Department of Toxic Substances Control, Regional Water Quality Control Board, or other responsible agency. The current regulatory environment of City, state, and federal requirements provides a high level of protection from new hazardous uses that may be sited near schools or other sensitive receptors. Additionally, existing conditions in the Uptown CPU area show no conflict between existing school sites and open hazardous materials sites. Therefore, implementation of the proposed Uptown CPU and associated discretionary actions would result in a less than significant impact related to hazardous emissions or contaminated sites near schools.

Issue 3 Emergency Evacuation and Response Plans

Would the project impair implementation of, or physically interfere with, an adopted emergency response plan or emergency evacuation plan?

There are no objectives or policies contained in the proposed Uptown CPU or associated discretionary actions that would interfere with or impair implementation of an adopted emergency response or evacuation plan. The Unified San Diego County Emergency Services Organization Operational Area Emergency Plan, Annex Q, Evacuation (County of San Diego 2014) identifies a broad range of potential hazards and a response plan for public protection. The plan identifies major interstates and highways within the County as primary transportation routes for evacuation. The land uses identified in the proposed Uptown CPU would not physically interfere with any known adopted emergency plans because development would occur on infill sites and the community is largely built-out with existing major roads and means for emergency evacuation.

Local Emergency Operations Plans are intended to help local jurisdictions respond to emergency situations with a coordinated system of emergency service providers and facilities. San Diego recently updated its 1995 Multi-Hazard Functional Plan and modernized its Emergency Operations Center. The City will continue to make regular modifications to the Multi-Hazard Functional Plan and Emergency Operations Center as hazards, threats, population and land use, or other factors change. The Multi-Hazard Functional Plan identifies resources available for emergency response and establishes coordinated action plans for specific emergency situations including earthquake, fire, major rail and roadway accidents, flooding, hazardous materials incidents, terrorism and civil disturbances (City of San Diego 2008). Implementation of the proposed Uptown CPU and associated discretionary actions would not impede implementation of existing emergency plans and would not impede future updates to emergency plans. Thus, impacts related to emergency evacuation and emergency response plans as a result of implementation of the proposed Uptown CPU and associated discretionary actions would be less than significant.

Issue 4 Hazardous Materials Sites and Health Hazards

Would the project be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, creates a significant hazard to the public or environment?

Hazardous materials are typically utilized by land uses such as industrial, retail/office, commercial, residential, agriculture, medical, and recreational uses, among other activities. According to a search of federal, state, and local regulatory databases, 61 documented hazardous material release cases were identified within Uptown, of which three are open, as shown in Table 6.14-1. Development of sites with existing contamination within the Uptown CPU area could potentially pose a hazard to the public or environment by placing sensitive receptors on, or adjacent to, known hazardous materials sites.

Federal and state regulations require adherence to specific guidelines regarding the use, transportation, disposal, and accidental release of hazardous materials. In accordance with local City and County, state, and federal requirements, any new development that involves contaminated

property would necessitate the cleanup and/or remediation of the property in accordance with applicable requirements and regulations. No construction would be permitted at such locations until a "no further action" clearance letter from the County Department of Health, or similar determination is issued by the City's Fire Rescue Department, California Department of Toxic Substances Control, Regional Water Quality Control Board, or other responsible agency. Because the Uptown CPU area does not historically have a large quantity of hazardous materials sites, and because the proposed Land Use Plan would not support a significant increase in land uses that have potential to result in hazardous emissions or contamination, there are no policies in the proposed Uptown CPU relative to hazardous materials. However, the General Plan includes policies to protect the health, safety, and welfare of residents relating to industrial land uses, documentation of hazardous materials investigations, and requires a site investigation for potential contaminants and soil remediation, if needed, if existing land uses change from industrial or heavy commercial to residential or mixed residential development. In addition, pesticide use would not pose a significant hazards as there are no major agricultural uses within the Uptown CPU area. Uptown is a built-out community located in the urbanized area of the City. Nominal amounts of pesticides and/or herbicides may be used by residents and other establishments for gardening or landscaping activities. These uses would not introduce significant risk of exposure to people in the Uptown CPU area. Therefore, impacts related to hazardous materials sites and health hazards would less than significant.

Issue 5 Aircraft Related Hazards

Would the project expose people or structures to a significant risk of loss, injury or death from off-airport aircraft operational accidents?

As concluded in Section 6.1 of this PEIR, impacts relative to safety hazards for people residing in or working in a designated airport influence area would be less than significant. Additionally, there are no private airports or heliport facilities within or near the Uptown CPU area. Thus impacts related to exposure of people or structures to aircraft hazards would be less than significant.

Cumulative Impact Analysis

As discussed in this section, compliance with federal, state, regional, and local health and safety laws and regulations would address potential health and safety impacts. Potential health and safety impacts associated with wildfire, hazardous substances, emergency response and evacuation plans, and aircraft hazards would not combine to create cumulative impacts when viewed together with the potential growth that could occur within the Uptown, North Park, and Golden Hill CPU areas. Wildlife impacts in these communities are limited to the canyon areas which are localized and would not be exacerbated by cumulative development in adjacent communities. Additionally, future projects implemented in accordance with the CPUs are required to follow the City's Brush Management regulations and the City and Fire Code requirements. Similarly, potential hazards associated with hazardous material sites are site specific and would not combine with hazards in other CPU areas to create a cumulative impact. Therefore, implementation of the proposed Uptown CPU and associated discretionary actions would not result in a cumulatively significant impact related to health and safety issues.

6.14.4 Significance of Impacts

Existing policies and regulations would help reduce, but not completely abate, the potential risks of wildland fires. The General Plan and proposed Uptown CPU contain goals and policies to be implemented by the City's Fire-Rescue Department, and through land use compatibility, training, sustainable development, and other measures, these goals and policies are aimed at reducing the risk of wildland fires.

Public education, firefighter training, and emergency operations efforts would reduce the potential impacts associated with wildfire hazards. Additionally, future development would be subject to conditions of approval that require adherence to the City's Brush Management Regulations and requirements of the California Fire Code. As such, impacts relative to wildland fire hazard would be less than significant

The proposed Uptown CPU and associated discretionary actions would not result in hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within a quarter-mile of an existing or proposed school. Impacts to schools would be less than significant. No mitigation is required.

The proposed Uptown CPU and associated discretionary actions would not impair implementation of, or physically interfere with, an adopted emergency response plan or emergency evacuation plan; therefore, impacts are less than significant, and no mitigation would be required.

Although there are closed LUST and Cleanup Program sites and there is one open LUST and two open Cleanup Program sites within the Uptown CPU area, there are local, state, and federal regulations and programs in places that minimize the risk to sensitive receptors on or adjacent to hazardous materials sites. Adherence to these regulations would result in less than significant impacts relative to hazardous materials sites and no mitigation is required.

Impacts relative to safety hazards related to being located within an airport influence area are less than significant. No mitigation is required.

6.14.5 Mitigation Framework

All impacts related to health and safety would be less than significant; including, wildfire hazards, exposure of schools or other sensitive receptors to hazardous materials, emergency evacuation, and aircraft related hazards. Thus, no mitigation is required.



Chapter 7.0 Effects Found Not to be Significant

California Environmental Quality Act (CEQA) Guidelines §15128 requires that an Environmental impact Report (EIR) contain a brief statement disclosing the reasons why various possible significant effects of a proposed project were found not to be significant and therefore would not be discussed in detail in the EIR. The impacts associated with the following environmental issue areas were found to not be significant as a result of the proposed Uptown Community Plan Update (CPU) and associated discretionary actions: Agricultural Resources, Mineral Resources, and Population and Housing.

7.1 Agricultural Resources

7.1.1 Farmland Mapping and Monitoring Program

Based on the farmland maps prepared by the California Department of Conservation (2010), the proposed Uptown Community Plan area is not identified as containing Prime Farmland, Unique Farmland, or Farmland of Statewide Importance. While there is a 0.5-acre community garden, which is identified as an agricultural use, this property is not identified as Prime Farmland, Unique Farmland, or Farmland of Statewide Importance. Therefore there would be no impact to Prime Farmland, Unique Farmland, Unique Farmland, or Farmland of Statewide Importance.

7.1.2 Agricultural Zoning/Williamson Act

The Uptown CPU area is not zoned for agriculture and there are no lands under a Williamson Act contract. Therefore, no impact is identified for this issue area.

7.1.3 Forest, Timberland, Timberland Production Zone

The Uptown CPU area is located within an urbanized area. There are no existing forestlands, timberlands, or timberland for Timberland Production Zone either within the CPU area or in the immediate vicinity that would conflict with existing zoning or the proposed rezoning (Forest Service 2007). Therefore, no impact is identified for this issue area.

7.1.4 Loss of Forest Land

The proposed Uptown CPU area is located within an urbanized area. There are no existing forestlands either within the CPU area or in the immediate vicinity (USDA 2016). The implementation of proposed CPU and associated discretionary actions would not result in the loss of forestland or conversion of forestland to non-forest use. Therefore, no impact is identified for this issue area.

7.1.5 Natural Conversion of Farmland or Forest

The proposed Community Plan area is located within an urbanized area; there are no existing forestland uses either on-site or in the immediate vicinity (Forest Service 2007). While there is a 0.5-acre community garden, which is identified as an agricultural use, this area would not be impacted by the proposed Uptown CPU and associated discretionary actions in the Uptown CPU area. The implementation of proposed CPU and associated discretionary actions would not involve any other changes that could result in conversion of farmland to non-agricultural use (i.e., increase in population) or conversion of forestland to non-forest use. Therefore, no impact is identified for this issue area.

7.2 Mineral Resources

According to the California Department of Conservation (CDC), Division of Mines and Geology, the area of the proposed Uptown CPU is designated as the following Mineral Land Classification:

MRZ-3: Areas containing mineral deposits the significance of which cannot be evaluated from available data (CDC 1996).

According to the California Geological Survey Open File Report 96-04, areas mapped as Mineral Resource Zone 1, 2, 3, and 4 (MRZ-1 through MRZ-4) have been mapped for the City of San Diego. MRZ-1 areas are locations in San Diego County that have been identified as having no significant mineral deposits. Areas mapped in MRZ-2 are considered to have extractable aggregate deposits. Areas mapped in MRZ-3 contain mineral deposits that may qualify as mineral resources. MRZ-4 areas are those where geologic information does not rule out either the presence or absence of mineral resources. Based on a review of referenced data, the proposed CPU area is in an urban area where the potential for loss of mineral deposits due to further development is considered low (CDC 2010).

In addition, the proposed CPU area is located entirely within a developed urban area and does not require the acquisition of additional land. There are no identified mineral resources that would be affected or "lost" as a result of the implementation of the proposed Uptown CPU and associated discretionary actions. Thus, the buildout of the proposed Uptown CPU and associated discretionary actions would not result in a loss of availability of a locally important mineral resource recovery site delineated on any local or general plan. Therefore, no impact is identified for this issue area.

7.3 Population and Housing

While population projections for the Uptown Community Plan area indicate that population will increase over time, the population growth would not introduce an impact. The proposed Uptown CPU would serve as a comprehensive, long-term plan for the physical development of the Uptown Community Plan area and is intended to manage and address future growth in it. The proposed Uptown CPU and associated discretionary actions would not displace people or existing housing, as the CPU and associated discretionary actions would designate planned land uses and zoning that would accommodate future development within the CPU area. Therefore no impact would occur.

80

Chapter 8.0 Growth Inducement

This Program Environmental Impact Report (PEIR) must examine the potential growth-inducing impacts of the proposed Uptown CPU and associated discretionary actions. More specifically, California Environmental Quality Act (CEQA) Guidelines Section 15126.2(d) requires that an EIR:

Discuss ways in which the proposed project could foster economic or population growth, or the construction of additional housing, either directly or indirectly, in the surrounding environment. Included in this are projects which would remove obstacles to population growth (a major expansion of a waste water treatment plant might, for example, allow for more construction in service areas). Increases in the population may tax existing community services facilities, requiring construction of new facilities that could cause significant environmental effects. It must not be assumed that growth in any area is necessarily beneficial, detrimental, or of little significance to the environment.

According to the City's Significance Determination Thresholds, growth inducement "is usually associated with those projects that foster economic or population growth, or the construction of additional housing, either directly or indirectly which may result in the construction of major new infrastructure facilities. Also, a change in land use policy or projects that provide economic stimulus, such as industrial or commercial uses, may induce growth. Accelerated growth may further strain existing community facilities or encourage activities that could significantly affect the surrounding environment." In addition, the Thresholds state that "the analysis must avoid speculation and focus on probable growth patterns or projects."

The General Plan PEIR (2008a) notes that "population in San Diego will grow whether or not the Draft General Plan is adopted..." and although a number of the General Plan policies are in place to "...encourage business, education, employment and workforce development...preserve and protect valuable employment land, especially prime industrial land, from conversion to other uses...and facilitate expansion and new growth of high quality employment opportunities in the City." The General Plan incorporates the previously adopted City of Villages strategy, which notes that a "village" is a place where residential, commercial, employment, and civic uses are present and integrated, and are characterized by compact mixed-use area, that are pedestrian-friendly and linked to the regional transit system (City of San Diego 2008b). Based on Government Code Section 65300, the General Plan serves as a comprehensive, long-term plan for physical development of the City and, by definition, is intended to manage and address future growth in the City. Implementation of the City of Villages strategy relies on the future designation and development of village sites through comprehensive community plan updates.

As detailed in the Project Description, Table 3-9, there is a current estimate of 36,750 residents in the Uptown CPU area. Under the adopted Uptown Community Plan the City estimates that the forecasted population would be 58,870 in year 2035; under the proposed CPU, by the year 2035, this population is projected to be 55,700 residents. While the expected build-out of the Uptown Community would be less than projected under the adopted Community Plan, the proposed CPU and associated discretionary actions serve as a comprehensive long-term plan for the physical development of the CPU area, and are intended to manage and address future growth through to build-out of the community.

The proposed Uptown CPU incorporates the City of Villages Strategy by designating a Community Village in the Hillcrest neighborhood, generally centered around Second, Third, Fourth, Fifth, and Sixth avenues between Front Street, Washington Street, State Route 163, and Walnut Street. Four neighborhood villages are also proposed at the following locations: one at Goldfinch Street and Washington Street in Mission Hills; one centered around Laurel Street and Third, Fourth, Fifth, and Sixth avenues between Nutmeg Street and Kalmia Street; a third in the Five Points area centered around India Street and Washington Street, and one in the eastern portion of the Hillcrest neighborhood within the vicinity of Park Boulevard, Washington Street, Richmond Street, and Essex Street. The community and neighborhood village concept draw upon the character and strength of the proposed CPU area setting, commercial centers, transit corridors, institutions, and employment centers. These areas are planned to be vibrant pedestrian neighborhoods with enhanced connectivity that reflect the types of public spaces, structures, public art, connections, and land uses that are influenced by the heterogeneous character of the community's population. The proposed Uptown CPU policy directs housing growth to areas suitable for infill and redevelopment that are integrated into the mixed-use cores of the community.

The proposed Uptown CPU and associated discretionary actions are intended to provide guidance on orderly growth and redevelopment in accordance with smart growth principles. Through the placement of higher density residential development in areas in and around transit and commercial corridors, the proposed CPU would reinforce a mixed-use urban environment that supports transit and pedestrian activity. The proposed Uptown CPU would designate land uses to accommodate growth, although additional housing units would not be built without demand. The proposed CPU includes an Impact Fee Study that would allow the maintenance and improvements in infrastructure capacity and public services to coincide with future development. Other potential environmental impacts associated with population growth in the proposed CPU area (e.g., transportation/traffic, air quality, noise, greenhouse gas emissions) are addressed in the relevant sections of this PEIR. As stated above, the population in the proposed Uptown CPU area will grow whether or not the proposed Uptown CPU and associated discretionary actions are adopted. The proposed Uptown CPU promotes infill residential, commercial, and office development and encourages the use of local and state programs to incentivize business retention and expansion. Additional policies are intended to facilitate economic wellbeing of locally owned and operated businesses and create ample job opportunities for residents in the proposed Uptown CPU area. These policies serve to facilitate expansion and new growth of high-quality employment opportunities. Therefore, the proposed Uptown CPU and associated discretionary actions would provide comprehensive planning for the management of population growth and necessary economic expansion to support the development efforts and allow an appropriate balance of managed population, housing, and economic growth to accommodate community development while maintaining related community and environmental standards.

9

Chapter 9.0 Significant Unavoidable Impacts/Significant Irreversible Environmental Changes/Energy Conservation

9.1 Significant and Unavoidable Impacts

In accordance with California Environmental Quality Act (CEQA) Guidelines Section 15126.2(b), any significant unavoidable impacts of a project, including those impacts that can be mitigated, but not reduced to below a level of significance despite the applicant's willingness to implement all feasible mitigation measures, must be identified in the Program Environmental Impact Report (PEIR). For the proposed Uptown Community Plan Update (CPU) and associated discretionary actions, impacts related to transportation and circulation (cumulative impacts to roadway segments, intersections, freeway segments and freeway ramps), noise (ambient and vehicle noise impacts), historical resources (historic and archeological resources), and paleontological resources would remain significant and unavoidable effects (refer to Chapter 6.0, Environmental Analysis, of this PEIR for further detail). All other significant impacts identified in Chapter 6.0 of this PEIR can be reduced to below a level of significance with implementation of the Mitigation Framework identified as well as through compliance with adopted General Plan and proposed Uptown CPU policies.

9.2 Significant Irreversible Environmental Impacts

Section 15126.2(c) of the CEQA Guidelines requires an evaluation of significant irreversible environmental changes which would occur should the proposed Uptown CPU and associated discretionary actions be implemented. Irreversible changes typically fall into one of three categories:

- Primary impacts such as the use of nonrenewable resources (i.e., biological habitat, agricultural land, mineral deposits, water bodies, energy resources and cultural resources);
- Primary and secondary impacts such as highway improvements which provide access to previously inaccessible areas; and
- Environmental accidents potentially associated with buildout of the proposed Golden Hill and North Park CPUs and associated discretionary actions.

Section 15126.2(c) of the CEQA Guidelines states that irretrievable commitments of resources should be evaluated to assure that current consumption of such resources is justified.

Implementation of the proposed Uptown CPU and associated discretionary actions would not result in significant irreversible impacts to agricultural land, biological resources, energy, historic resources, mineral deposits, or water bodies. Although some sensitive biological resources are identified within the canyons and areas designated as open space in the CPU area, direct and indirect impacts can be offset through strict compliance with CPU policies, regulatory compliance (Multiple Species Conservation Program [MSCP] and Environmentally Sensitive Lands [ESL] Regulations of the Land Development Code [LDC]). Similarly, future development pursuant to the proposed Uptown CPU and associated discretionary actions could impact important historical or archaeological resources given the presence of known and potential historical and archaeological resources within the communities. The potential archaeological resource impacts can be mitigated through strict adherence to CPU policies, regulatory compliance (LDC Historical Resource Regulations), and implementation of the Mitigation Framework further detailed in Section 6.7 of this PEIR. Impacts to historical and archeological resources would, however, remain significant and unavoidable. As evaluated in Chapter 8.0, Effects Not Found to be Significant, of this PEIR, implementation of the proposed Uptown CPU and associated discretionary actions would not result in significant irreversible impacts to agricultural and forestry or mineral resources. Finally, no water bodies are present within the communities, and no downstream receiving waters would be impacted by buildout of the CPU.

The Uptown community is almost completely built out and is accessible via regional transportation facilities (e.g., Interstates 5 and 8, State Route 163). No new freeways or roadways are proposed that would provide access to currently inaccessible areas. Therefore, implementation of the proposed Uptown CPU and associated discretionary actions would not result in a significant irreversible commitment with regard to unplanned land use.

Construction of development implemented in accordance with the proposed Uptown CPU and associated discretionary actions would require the irreversible consumption of natural resources and energy. Natural resource consumption would include lumber and other forest products, sand and gravel, asphalt, steel, copper, other metals, and water. Building materials, while perhaps recyclable in part at some long-term future date, would for practical purposes be considered permanently consumed. Energy derived from nonrenewable sources, such as fossil fuels, would be consumed during construction and as a result of operational lighting, heating, cooling, and transportation uses. The proposed Uptown CPU includes policies aimed at improving energy efficiency, reducing water use, and minimizing impacts on other natural resources. For example, the village/neighborhood center concepts would reduce dependence on fossil fuel energy sources by

integrating housing units in close proximity to transit corridors. These policies would serve to reduce irreversible water, energy, and building materials consumption associated with construction, occupation, and operation. Energy consumption is discussed in greater detail in Section 9.3 below.

With respect to environmental accidents potentially associated with buildout of the proposed Uptown CPU and associated discretionary actions, and as further discussed in Section 6.14 of this PEIR, potential impacts related to hazardous materials and associated health hazards from implementation of the proposed Golden Hill and North Park CPUs and associated discretionary actions would be avoided or reduced to below a level of significance through mandatory conformance with applicable regulatory/industry standard and codes. The North Park and Golden Hill CPU areas contain undeveloped land in the form of canyons that is occupied by a variety of native and non-native plant communities. Due to the amount of natural, unmaintained open space in the Uptown CPU area, there is a high risk for wildfires. Development pursuant to the proposed Uptown CPU and associated discretionary actions, however, would be subject to applicable state and City regulatory requirements related to fire hazards and prevention. Accidents related to flood hazards would not be significant because the CPU area does not contain mapped floodplains.

9.3 Energy Conservation

Section 15126.4 (a)(1) of the CEQA Guidelines states that an EIR shall describe feasible measures, which could minimize significant adverse impacts, including, where relevant, the inefficient and unnecessary consumption of energy. CEQA Guidelines, Appendix F, Energy Conservation, provides guidance for EIRs regarding potential energy impacts of proposed projects, with particular emphasis on avoiding or reducing the inefficient, wasteful, and unnecessary consumption of energy. The Resources Agency amended Appendix F to make it clear that an energy analysis is mandatory. However, the Resources Agency also clarified that the energy analysis is limited to effects that are applicable to the project (Resources Agency 2009). Furthermore, Appendix F is not described as a threshold for determining the significance of impacts. Appendix F merely seeks inclusion of information in the EIR to the extent relative and applicable to the project.

Because the proposed action is the adoption of a community plan and associated discretionary regulatory actions, and does not specifically address any particular development project(s), impacts to energy resources are addressed generally, based on projected buildout of the proposed Uptown CPU and associated discretionary actions. Implementation of the proposed Uptown CPU and associated discretionary actions have the potential to result in impacts to energy supply due to development that is anticipated to occur in response to projected population growth. Depending on the types of future uses, impacts would need to be addressed in detail at the time specific projects are proposed. At a minimum, future projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions would be required to meet the mandatory energy standards of the current California energy code (Title 24 Building Energy Standards of the California Public Resources Code).

Energy resources would be consumed during construction of future development. Energy also would be consumed to provide operational lighting, heating, cooling, and transportation for future development.

9.3.1 Construction-Related Energy Consumption

Grading and construction activities consume energy through the operation of heavy off-road equipment, trucks, and worker traffic. At the program-level, it is too speculative to quantify total construction-related energy consumption of future development, either in total or by fuel type. The majority of energy to be used in conjunction with construction activities would be supplied by San Diego Gas & Electric (SDG&E).

In compliance with the City's Thresholds of Significance, future discretionary projects exceeding the 60-ton solid waste threshold would be required to develop waste management plans targeting at least 75 percent waste reduction, including construction waste. Even though exact details of the projects implemented in accordance with the proposed Uptown CPU and associated discretionary actions are not known at this time, there are no conditions in the CPU areas that would require non-standard equipment or construction practices that would increase fuel-energy consumption above typical rates. Therefore, development pursuant to the proposed Uptown CPU and associated discretionary actionary actions would not result in the use of excessive amounts of fuel or other forms of energy during the construction of future projects.

9.3.2 Long-Term Operation-Related Energy Consumption

Long-term operational energy use associated with the proposed Uptown CPU and associated discretionary actions includes fuel consumption of vehicles and electricity and natural gas consumption by residents and commercial operations, and energy consumption related to obtaining water. However, the use of these resources would still be used daily as essential energy sources and utilities regardless of implementation of the proposed Uptown CPU and associated discretionary actions. As such, although long-term operational energy use would result from future development, such changes would not be considered significant in comparison to the energy use of other cities in the region. The proposed Uptown CPU and associated discretionary actions would not result in any unusual characteristics that would result in excessive long-term operational building energy demand.

At a minimum, development under the proposed Uptown CPU and associated discretionary actions would be required to meet the mandatory energy standards of the current California Energy Code (Title 24 Building Energy Standards of the California Public Resources Code). Some efficiencies associated with the Energy Standards under Title 24 include the building heating, ventilating, and air conditioning (HVAC) mechanical system, water heating system, and lighting system. Additionally, rebate and incentive programs that promote the installation and use of energy-efficient plug-in appliances and lighting would be available, but not covered under Title 24. Development would be required to comply with the proposed Uptown CPU Conservation Element which contains a number of Sustainable Development Policies that focus on designing new development to have a climate, energy efficient, and environmentally oriented site design.

Policies proposed in the Uptown CPU would further address energy consumption. Specifically, proposed Conservation Element and Urban Design Element policies would reduce local dependence on automobile transportation, support incorporation of sustainable building and development

practices, adhering to standardized measures outlined in the City's Climate Action Plan, and encouraging street light retrofits and energy efficient outdoor lighting.

Although these policies would decrease the overall per capita energy use in the CPU areas, they would not ensure that energy supplies would be available when needed. Future projects would be subject to review for measures that would further reduce energy consumption in conformance to existing regulations. Furthermore, the City's Climate Action Plan, adopted by City Council in December 2015, includes 2020 and 2035 targets that are on the trajectory for meeting the 2050 greenhouse gas reduction goals established by Executive Order S-3-05. Future projects would be reviewed for consistency with the Climate Action Plan and applicable implementation measures.

Future operational energy use related to roadways would consist of the transportation fuels consumed to transport area residents, workers, and visitors. The total estimated daily vehicle trips at full buildout are estimated to be 584,112 for the proposed Uptown CPU, as detailed in the traffic impact analysis prepared for the CPUs (Kimley-Horn 2015). The proposed Uptown CPU Mobility Element also contains policies that would reduce vehicle miles travelled and associated fuel consumption. These include policies to improve the pedestrian environment and neighborhood walkability; improving bicycle infrastructure and facilities, encouraging implementation of transportation demand management strategies, and parking management policies that support installation of electric vehicle charging stations and measures to reduce parking demand.

In conclusion, development under the proposed Uptown CPUs and associated discretionary actions would result in increased energy use, in the form of new buildings and transportation. Residential and nonresidential development use electricity, natural gas, and petroleum products for power, lighting, and heating and vehicles use both oil and gas. Use of these types of energy for new development would result in the overall increased use of nonrenewable energy resources. This represents an irreversible environmental change.

As described in this PEIR, the proposed Uptown CPU contains policies aimed at improving energy efficiency, reducing water use, and minimizing impacts to natural resources, which serve to reduce irreversible consumption of building materials, water, and energy use.

10

Chapter 10.0 Alternatives

The California Environmental Quality Act (CEQA) Guidelines Section 15126.6 requires that an Environmental Impact Report (EIR) compare the effects of a "reasonable range of alternatives" to the effects of a project. The CEQA Guidelines further specify that the alternatives selected should attain most of the basic project objectives, and avoid or substantially lessen one or more significant effects of the project. The "range of alternatives" is governed by the "rule of reason," which requires the EIR to set forth only those alternatives necessary to permit an informed and reasoned choice by the lead agency, and to foster meaningful public participation (CEQA Guidelines Section 15126.6[f]). CEQA generally defines "feasible" to mean an alternative that is capable of being accomplished in a successful manner within a reasonable period of time, while also taking into account economic, environmental, social, technological, and legal factors.

As discussed in Chapter 6.0, the proposed Uptown Community Plan Update (CPU) and associated discretionary actions would result in significant and/or cumulative environmental impacts related to transportation, noise, historical resources, and paleontological resources. In developing the alternatives to be addressed in this chapter, consideration was given regarding their ability to meet the basic objectives of the proposed Uptown CPU and associated discretionary actions and the potential to eliminate or substantially reduce significant environmental impacts (as identified in Chapter 6.0 of this Program EIR [PEIR]).

The following specific objectives for the proposed Uptown CPU and associated discretionary actions support the underlying purpose of the project, assist the City as Lead Agency in developing a reasonable range of alternatives to evaluate in this PEIR, and will ultimately aid the Lead Agency in preparing findings and overriding considerations, if necessary. The following primary goals, recommendations, and objectives of the proposed CPU are to:

• Develop a multi-modal transportation network emphasizing active transportation measures for walkable and bicycle-friendly streets, and transit-related measures supporting transit operations and access.

- Maintain or increase the housing supply through the designation of higher residential densities focusing along major transit corridors.
- Provide for increased economic diversification through land use to increase employment and economic growth opportunities.
- Preserve the neighborhood character and design relationships between neighborhoods within each community through the development of transitions and design policies.
- Identify significant historic and cultural resources within the community and provide for their preservation, protection, and enhancement.
- Provide increased recreation opportunities and new public open spaces.
- Preserve, protect, and enhance the community's natural landforms, including canyons and environmentally sensitive lands.
- Include financing strategies that can secure infrastructure improvements concurrent with development.

The alternatives addressed in this PEIR were selected in consideration of one or more of the following factors:

- The extent to which the alternative would feasibly accomplish most or all of the basic objectives of the proposed Uptown CPU;
- The extent to which the alternative would avoid or substantially lessen any of the identified significant environmental effects of the proposed Uptown CPU and associated discretionary actions. The feasibility of the alternative, taking into account site suitability, economic viability, availability of infrastructure, general plan consistency, and consistency with other applicable plans and regulatory limitations;
- The appropriateness of the alternative in contributing to a "reasonable range" of alternatives necessary to permit a reasoned choice; and
- The requirement of the CEQA Guidelines to consider a "no project" alternative; and to identify an "environmentally superior" alternative in addition to the no project alternative (Section 15126.6[e]).

Based on the criteria described above, this PEIR considers the following project alternatives:

- No Project (Adopted Community Plan) Alternative;
- Adopted Community Plan with Removal of the Height Ordinance Alternative;
- Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative;
- Density Redistribution Alternative; and
- Lower-Density Alternative.

General descriptions of the characteristics of each of these alternatives, along with a discussion of their ability to reduce significant environmental impacts associated with the proposed Uptown CPU and associated discretionary actions, are provided in the following subsections. Table 10-1, Matrix Comparison of Project Alternatives and Proposed CPU for Uptown, provides a side-by-side comparison of the potential impacts of the alternatives to the impacts of the proposed Uptown CPU and associated discretionary actions.

ATTACHMENT 7

10.0 Alternatives

Table 10-1						
Matrix Comparison of Project Alternatives and Proposed Uptown CPU						
Environmental Issue Area	Proposed Uptown CPU and Associated Discretionary Actions	No Project (Adopted Community Plan) Alternative	Adopted Community Plan with Removal of Interim Height Ordinance Alternative	Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative	Density Redistribution Alternative	Lower-Density Alternative
Land Use	LS	LS (>)	LS (>)	LS (=)	LS (>)	LS (>)
Visual Effects and Neighborhood Character	LS	LS (>)	LS (>)	LS (=)	LS (=)	LS (=)
Transportation and Circulation	SU – Traffic Circulation LS – Alternative Transportation	SU (>) - Traffic Circulation LS (>) - Alternative Transportation	SU (>) - Traffic Circulation LS (>) - Alternative Transportation	SU (>) - Traffic Circulation LS (=) - Alternative Transportation	SU (<) - Traffic Circulation LS (>) - Alternative Transportation	SU (=) - Traffic Circulation LS (>) - Alternative Transportation
Air Quality	LS	LS (>)	LS (>)	LS (>)	LS (<)	LS (=)
Greenhouse Gas Emissions	LS	SU (>)	SU (>)	LS (>)	LS (>)	LS (>)
Noise	SU	SU (=)	SU (=)	SU (=)	SU (=)	SU (=)
Historical Resources	SU	SU (>) – Historic SU (=) Archeological	SU (>) – Historic SU (=) Archeological	SU (=)	SU (=)	SU (<)<u>(=)</u>
Biological Resources	LS	LS (>)	LS (>)	LS (>)	LS (=)	LS (=)
Geologic Conditions	LS	LS (=)	LS (=)	LS (=)	LS (=)	LS (=)
Paleontological Resources	SU	SU (=)	SU (=)	SU (=)	SU (=)	SU (=)
Hydrology / Water Quality	LS	LS (=)	LS (=)	LS (=)	LS (=)	LS (=)
Public Services and Facilities	LS	LS (=)	LS (=)	LS (=)	LS (=)	LS (=)
Public Utilities	LS	LS (=)	LS (=)	LS (=)	LS (=)	LS (=)
Health and Safety	LS	LS (=)	LS (=)	LS (=)	LS (=)	LS (=)
Notes: SU = Significant and Unavoidable (for the issue that results in the impact); LS = Less than Significant Comparison of Impacts: = Impacts the same/similar to the Proposed Project; < Impact less than the Proposed Project; > Impacts greater than the Proposed Project.						

10.1 No Project (Adopted Community Plan) Alternative

10.1.1 Description

Under the No Project Alternative, the adopted Uptown Community Plan would continue to guide development and would be implemented with the zoning program which includes Mid-City Communities Plan District, West Lewis Plan District, and the Interim Height Ordinance. Last updated in 1988, the current Community Plan identifies the following issues that are the most important to be addressed in the community plan through policies and regulations:

- Provision of a wide variety of housing types for all age, income and social groups.
- Revitalization of certain neighborhood commercial districts.
- Establishment and maintenance of a high level of public facilities and services to meet the needs of the community.
- Promotion of a clean and healthful environment.
- Preservation of significant historic structures.
- Preservation of community character and historical, architectural, and cultural resources.
- Reduction in development that encroaches into open space areas.
- Establishment of urban design standards and criteria for the various neighborhoods.
- Encouragement of mixed land use in appropriate areas to improve land utilization and encourage redevelopment.
- Discrepancies between actual zoning and community plan land use recommendations.
- Preservation of pedestrian-oriented commercial areas.
- Noise generated by air traffic utilizing Lindbergh Field (San Diego International Airport) and by automobile traffic on Interstate 5 and State Route 163.
- Inadequate freeway access.

The No Project Alternative would consist of the current Community Plan land use designations as they apply today, including all amendments to the Community Plan from its original adoption in 1988 to the most recent amendment in 2008. Table 10-2 describes the history of amendments to the Uptown Community Plan that are considered part of the No Project Alternative.

Table 10-2 Amendments to the 1988 Uptown Community Plan			
	Date Adopted by City	Resolution	
Amendment	Council	Number	
Rezone amendment by City Council	May 2, 1989	R-273376	
Uptown Implementation Program	October 3, 1989	R-274502	
Nob Hill Place Amendment	May 7, 2002	R-296458	
Bankers Hill Townhomes	November 29,2005	R-301088	
Scripps Mercy Hospital	May 20, 2008	R-303732	

Adopted community plan land use designations seek to promote a balance of land uses. The majority of the land use is designated as Low-Density Residential at 5 to 10 units per acre. The adopted plan locates higher residential density away from the single family neighborhoods and focuses development on the major transportation corridors: Washington; University; Park Blvd; 4th, 5th and 6th Avenue. Mixed-use development is encouraged in selected areas with residential use over street-level retail use. In Uptown, the Hillcrest and Bankers Hill Neighborhoods are identified for the highest intensity within the community with up to 110 dwelling units per acre (du/ac) along 5th and 6th Avenue and within the Hillcrest core. Institutional and Schools/Public Facilities are designated for City-owned and other public/quasi-public facilities.

Figure 10-1 shows the No Project (Adopted Community Plan) land use map. Areas of proposed land use change are concentrated throughout the community where the proposed Uptown CPU would generally facilitate lower intensity mixed-use development compared to the existing Community Plan. Specifically, as shown in Table 10-3, the proposed Uptown CPU could have approximately 32,700 dwelling units at build-out, while the No Project Alternative could have approximately 34,600 dwelling units at build-out, or 1,900 more units compared to the proposed Uptown CPU. Table 10-3 presents a build-out comparison of the No Project (Adopted Community Plan) Alternative and the proposed Uptown CPU including acreage by generalized land use, dwelling units, floor area, and projected household population at build-out.

With the proposed Uptown CPU, the Mid-City Communities Plan District, West Lewis Plan District and the Interim Height Ordinance would be rescinded and the proposed Uptown CPU would incorporate new building height limits through use of citywide zones and the Community Plan Implementation Overlay Zone (CPIOZ) in geographically specific areas as described in the Project Description, Chapter 3 and shown on Figures 3-7 and 3-8 of this PEIR. The proposed CPIOZ-Type A identifies areas within the community where development up to 50 feet is allowed with a ministerial approval (in Mission Hills) and up to 65 feet (in Hillcrest and Bankers Hill/Park West). The proposed CPIOZ-Type B identifies areas within the community where development up to 150 feet in Bankers Hill/Park West, up to 120 feet in central Hillcrest, and up to 100 feet in Hillcrest east of the SR-163 would be allowed with discretionary approval of a Process 3 Site Development Permit. The CPIOZ would generally allow greater heights in Mission Hills and Hillcrest compared to building heights that would be allowed under the No Project Alternative with the retention of the Interim Height Ordinance. In Bankers Hill/Park west, development under the Interim Height Ordinance of the No Project Alternative of up to 150 feet would be allowed ministerially; however, under the proposed Uptown CPU, the same height limit could be achieved but would require a discretionary approval. Figure 10-2 shows the maximum building heights in areas affected by the Interim Height Ordinance that would apply under this alternative.

ATTACHMENT 7





FIGURE 10-1 No Project Alternative: Adopted Uptown Community Plan
Map Source: City of San Diego

ATTACHMENT 7



0 Feet 600

FIGURE 10-2 Maximum Building Heights under the Existing Interim Height Ordinance

Table 10-3									
Build-out Comparison of the No Project (Adopted Community Plan) Alternative and the Proposed Uptown CPU									
	No Project (Adopted Community Plan) Alternative			Proposed Uptown CPU					
		Dwelling			Dwelling				
Generalized Land Use	Acres	Units	Floor Area	Acres	Units	Floor Area			
Agriculture	0.5			0.5					
Education	29		364,200	29		364,200			
Institutional	90		1,950,700	100		2,121,500			
Multi Family	388	29,060		380	27,180				
Office Commercial	31		1,586,000	32		1,598,700			
Open Space	415			398					
Parks	28			48					
Recreational	3		31,100	3		31,100			
Retail Commercial	184		3,197,000	179		3,186,500			
Roads	761			761					
Single Family	725	5,540		726	5,520				
Visitor Commercial	2		174,000	2		174,000			
Grand Totals	2,656	34,600	7,303,000	2,656	32,700	7,476,000			
Population	58,870			55,700					

10.1.2 Analysis of No Project Alternative

a. Land Use

The No Project (Adopted Community Plan) Alternative would retain the adopted Uptown Community Plan land use map and the current Interim Height Ordinance. Land use impacts under this alternative would be similar or greater than the anticipated impacts of the proposed Uptown CPU and associated discretionary actions because it would not contain the proposed CPU policies and land use changes intended to improve compatibility with and implement the San Diego General Plan. The No Project Alternative would not benefit from the amendments to the Historical Resources Regulations in the Land Development Code that would provide supplemental development regulations pertaining to potential historic districts under the proposed Uptown CPU.

The adopted Community Plan also incorrectly designates parts of residential areas as Multi-Habitat Planning Area (MHPA). The No Project Alternative would not benefit from the proposed communitywide MHPA boundary line correction that would add environmentally sensitive lands to the MHPA and removed existing developed land from the MHPA. Thus, the No Project Alternative would not support the Multiple Species Conservation Program (MSCP) Subarea Plan to the same degree as the proposed Uptown CPU.

Although the adopted community plan would not conflict with adopted land use plans, policies, or ordinances, and would result in a less than significant land use impact overall, the No Project Alternative would less compatible than the proposed Uptown CPU when viewed in relation to

applicable land use plans and policies. Thus, the land use impacts of the No Project Alternative would be slightly greater than the proposed Uptown CPU.

b. Visual Effects and Neighborhood Character

Potential visual effects and impacts to neighborhood character under the No Project Alternative would be similar to those anticipated under the proposed Uptown CPU except that the No Project Alternative would allow reduced building heights when compared to the proposed Uptown CPU and associated discretionary actions. Particularly in Mission Hills and Hillcrest, development under the No Project Alternative would be limited to building heights of 50 and 65 feet, respectively. The reduced building heights could reduce potential neighborhood character impacts; however, the No Project Alternative would also not benefit from the proposed Uptown CPU policies that are intended to ensure compatible development and design that enhances and is sensitivity to neighborhood character.

Additionally, the No Project Alternative would not provide supplemental development regulations to preserve potential historic districts, which could result in additional conflicts with existing historic preservation policies of the General Plan compared to the proposed Uptown CPU and associated discretionary actions. Under the No Project Alternative, all future development would be required to comply with existing regulations regarding grading activities and lighting design. Thus, despite the reduced building heights compared to the proposed Uptown CPU, impacts of the No Project Alternative would be slightly greater than impacts compared to the proposed Uptown CPU due the lack of application of proposed Uptown CPU policies that support protection of public views and neighborhood character.

c. Transportation and Circulation

The No Project Alternative would generate more vehicular trips than the proposed Uptown CPU and associated discretionary actions as it allows for more residential units than the proposed Uptown CPU and associated discretionary actions. <u>MoreoverHowever</u>, the No Project Alternative would not contain the proposed Uptown CPU and associated discretionary actions policies intended to promote a multimodal network that encourage walking, bicycling, and transit. Impacts to individual intersections and roadway segments would be slightly greater under the No Project Alternative due to the increased development potential. Similar to the proposed Uptown CPU, the No Project Alternative would result in significant and unavoidable impact to streets, intersections and freeway segments and ramp meters. Compared to the proposed Uptown CPU, impacts to roadway and freeway facilities would be slightly more severe.

Regarding consistency with applicable plans and policies related to alternative transportation, the No Project Alternative would not include the proposed CPU policies that support increasing multimodal opportunities within the CPU area consistent with the San Diego Association of Governments (SANDAG) Regional Transportation Plan (RTP), the City's General Plan, or the City's Climate Action Plan (CAP). Thus, while the No Project Alternative would not result in significant impacts related to conflicts with plans and policies addressing alternative transportation, the No Project Alternative would not achieve the level of consistency with these applicable plans and policies that the proposed Uptown CPU would achieve. Thus, impacts related to alternative transportation would be slightly greater.

d. Air Quality

The No Project Alternative would retain the existing Uptown Community Plan. Air Quality impacts under this alternative would be slightly greater than the anticipated impacts of the proposed Uptown CPU due to slightly greater density. Like the proposed Uptown CPU, the No Project Alternative would not conflict with or obstruct implementation of the applicable air quality plan nor would it result in a violation of any air quality standard or contribute substantially to an existing or projected air quality violation because the land uses under existing community plan would be consistent with the Regional Air Quality Strategy (RAQS). However, the No Project Alternative's future operational emissions would be greater than those of the proposed Uptown CPU.

e. Greenhouse Gas Emissions

The No Project Alternative would result in 535,684488,993 metric tons of carbon dioxide equivalent (MT CO₂E) GHG emissions, which would be slightly greater than the estimated 523,925478,184 MT CO₂E GHG emissions for the proposed Uptown CPU. The decrease in GHGs associated with the proposed Uptown CPU is a direct result of the proposed land use changes including an increase in commercial uses and decrease in residential, which would implement CAP Strategies and the General Plan's City of Villages Strategy. Since the No Project Alternative would not adjust the land use map or provide policies to implement these strategies, GHG impacts of the No Project Alternative would be significant and unavoidable and greater than the proposed Uptown CPU.

f. Noise

The No Project Alternative would retain the adopted Uptown Community Plan. Noise impacts under this alternative would be similar to the anticipated impacts of the proposed Uptown CPU and associated discretionary actions because, like the proposed Uptown CPU, it would permit development that would be subject to ambient noise increases and traffic noise as the planning area is built out. Although the No Project Alternative would result in slightly greater development potential, the increase in traffic noise would not likely be perceptible compared to the proposed Uptown CPU and associated discretionary actions. While the No Project Alternative does not contain the proposed Uptown CPU policy changes intended to improve compatibility with and implement the General Plan policies, future development implemented under both the No Project Alternative and the proposed Uptown CPU and associated discretionary actions would be required to comply with applicable City and State noise regulations including Title 24 building code requirements. The noise impacts of the No Project Alternative would be similar to the proposed Uptown CPU and both would result in significant and unavoidable impacts related to ambient noise increases, traffic noise exposure, and construction vibration impacts.

g. Historical Resources

The No Project Alternative would retain the existing Uptown Community Plan, with no additional regulations addressing Potential Historic Districts (refer to Section 6.7, Historical Resources for

details on the proposed Uptown CPU protections for historical resources). While the City's Historical Resources Regulations provide for the regulation and protection of historical resources, it is impossible to ensure the successful preservation of all historical resources within the plan area. Therefore, potential impacts to the Historic Districts remain significant and unavoidable. Under the No Project Alternative the proposed supplemental regulations to the Historical Resources Regulations of the Land Development Code addressing potential historic districts would not be implemented. Therefore, the potential loss of historical resources would be the same greater under the No Project Alternative compared to that of the proposed Uptown CPU and associated discretionary actions.

As with the proposed Uptown CPU, future development under the No Project Alternative has the potential to result in significant direct and/or indirect impacts to archaeological resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.7, Historical Resources. The extent of impacts to archaeological resources resulting from implementation of the No Project Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the No Project Alternative would result in potentially significant impacts related to archaeological resources at the program level that would be significant and unavoidable, despite adherence to the existing regulatory framework.

h. Biological Resources

Under the No Project Alternative the MHPA boundary corrections proposed in the proposed Uptown CPU and associated discretionary actions would not be implemented, and it is likely that the amount of preserved open space would be less than with the proposed project. Therefore, the No Project Alternative would result in reduced protections for MHPA lands and environmentally sensitive resources and thus, greater impacts to biological resources than those anticipated under the proposed Uptown CPU. Implementation of the No Project Alternative would be required to adhere to all applicable federal, state, and local regulations regarding the protection of biological resources, as for all subsequent development project submittals under the proposed Uptown CPU. Other than the greater impacts to MHPA lands and environmentally sensitive lands due to lack of implementation of MHPA boundary corrections, all other biological resource impacts of the No Project Alternative would be the same as the proposed Uptown CPU and associated discretionary actions and all impacts would be less than significant.

i. Geologic Conditions

Geologic impacts from implementation of the No Project Alternative would be similar to those of the proposed Uptown CPU and associated discretionary actions. Potential impacts related to seismic and geologic hazards, or to the instability of geological units and soils would be avoided or reduced to less than significant through adherence to existing state and local regulations, including the California Building Code, the San Diego Municipal Code, and the Seismic Hazards Mapping Act. Where required, site-specific geotechnical investigations would be conducted to identify and evaluate seismic hazards and formulate mitigation measures prior to permitting most developments designed for human occupancy. Similarly, project-level compliance with City-mandated grading

requirements, and, compliance with applicable state and/or federal regulations would ensure that future grading and construction activities would avoid significant soil erosion impacts. These requirements would apply equally to both the No Project Alternative and the Proposed Uptown CPU and associated discretionary actions, thus impacts of this alternative would be similar to the project.

j. Paleontological Resources

As with the proposed Uptown CPU and associated discretionary actions, future development under the No Project Alternative has the potential to result in significant direct and/or indirect impacts to paleontological fossil resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.10, Paleontological Resources. The extent of impacts to paleontological resources resulting from implementation of the No Project Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the No Project Alternative would result in potentially significant impacts related to paleontological resources at the program level because adherence to the mitigation framework cannot be guaranteed for ministerial projects that only require a grading permit. Thus impacts of the No Project Alternative would be significant and unavoidable and similar to the proposed Uptown CPU and associated discretionary actions.

k. Hydrology and Water Quality

The land use pattern and distribution for the No Project Alternative is generally the same as for the proposed Uptown CPU. Future development under both the No Project Alternative and the proposed Uptown CPU and associated discretionary actions would be required to comply with existing federal, state, and local regulations relative to runoff and water quality at the project level, which would preclude the potential for hydrology and water quality impacts. Thus impacts of the No Project Alternative would be less than significant and would be similar to the proposed Uptown CPU and associated discretionary.

I. Public Services and Facilities

The No Project Alternative would retain the existing Uptown Community Plan. Impacts to Public Services and Facilities under this alternative would be similar to the anticipated impacts associated with the proposed Uptown CPU because the anticipated population at build-out of the No Project Alternative would be only marginally more than the anticipated population for the build-out of the proposed Uptown CPU and would not have a measurable impact on demand for facilities. For police and fire protection services the difference in population would not impact either the police or fire department in their ability to provide service, nor would the departments require the construction of new facilities. For both the No Project Alternative and the proposed Uptown CPU, future projects would be required to pay required school fees which would mitigate for the potential impacts to schools to less than significant. Similarly both the No Project Alternative and the proposed Uptown CPU would include financing mechanisms to provide for libraries. However, in the case of both the No Project Alternative and the proposed Uptown cPU there would be a deficit in population based parks and the need to build new recreational facilities. For both the No Project Alternative and the

proposed Uptown CPU and associated discretionary actions, any new facilities would require a separate environmental review to identify potential impacts associated with the construction of facilities. Thus, for the No Project Alternative, public facilities and services impacts would be less than significant and the same as the proposed Uptown CPU.

m. Public Utilities

The No Project Alternative would retain the existing Uptown Community Plan. Impacts to Public Utilities under this alternative would be similar to the anticipated impacts of the proposed Uptown CPU. Although the proposed Uptown CPU would have a slightly lower anticipated population than the No Project Alternative (see Table 10-3), implementation of the No Project Alternative would have similar impacts related to storm water, sewer, water, communications, solid waste and recycling, or energy as the proposed Uptown CPU.

n. Health and Safety

Impacts of build-out of the No Project Alternative would be similar to the potential impacts of buildout of the proposed Uptown CPU and associated discretionary actions. Future development under the No Project Alternative has the potential to result in exposure to hazardous materials, wastes, or emissions; airport hazards, and fire hazards. However, land uses under the No Project alternative would be similar to the land uses under the proposed Uptown CPU. Additionally, there would not be any areas of change or land use changes that would increase potential exposure to hazards. Federal, state and local regulations that serve to reduce impacts a less-than-significant level would also reduce impacts for development under the No Project alternative. Overall, impacts would be less than significant and similar to those anticipated under the proposed Uptown CPU.

10.2 Adopted Community Plan with Removal of the Interim Height Ordinance Alternative

10.2.1 Description

Under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative, the adopted Uptown Community Plan and zoning program which includes Mid-City Communities Plan District and West Lewis Plan District would continue to guide development with the exception that the Interim Height Ordinance (O-20329) that limits structure heights in specific areas to 50 and 65 feet would not be applied. Height limits of the base zones would be applied in these areas as described in Figure 10-3. As a result, those areas now subject to the Interim Height Ordinance would allow buildings up to the height permitted by the Mid-City Communities Plan District. In the case of areas in Mission Hills currently limited to 50 feet, structures would be permitted up to 150 feet. In the areas of Hillcrest limited to 65 feet, structures would be permitted to 200 feet.

ATTACHMENT 7



FIGURE 10-3 Maximum Building Heights of Existing Base Zones (Removal of the Interim Height Ordinance)

Feet

Details regarding the adopted Community Plan provided in section 10.1.1 above would also apply to this alternative, with the exception of the discussion on height limitations. Refer to Table 10-3 for a comparison of acreages, dwelling units, floor area allowances, and projected population by generalize land use categories between the adopted community plan and the proposed Uptown CPU. Compared to the proposed Uptown CPU that would include new structure height regulations in certain areas through implementation of the CPIOZ (depicted on Figures 3-7 and 3-8 of this PEIR), the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would allow taller buildings under ministerial review within the Mission Hills, Hillcrest, and Bankers Hill/Park West neighborhoods. The increased building height allowance combined with slightly higher density under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would have the potential to increase the intensity of development with taller buildings compared to the proposed Uptown CPU and associated discretionary actions.

The proposed Uptown CPU would maintain land use designations generally consistent with the adopted Community Plan. with the exception of height limitations. Figure 10-1 shows the Adopted Community Plan land use map that would apply to this Alternative. Areas of proposed land use change are concentrated throughout the community where the proposed Uptown CPU would generally facilitate lower intensity mixed-use development compared to the adopted Community Plan. Specifically, as shown in Table 10-3, the proposed Uptown CPU would support 32,700 dwelling units at build-out, while the Adopted Community Plan would accommodate 34,600 dwelling units at build-out or 1,900 more units compared to the proposed Uptown CPU.

10.2.2 Analysis of Adopted Community Plan with Removal of the Interim Height Ordinance Alternative

a. Land Use

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would retain the adopted Uptown Community Plan land use map but would not apply the existing Interim Height Ordinance. As a result, height limits as defined by the base zone would apply. Land use impacts under this alternative would be similar or greater than the anticipated impacts of the proposed Uptown CPU and associated discretionary actions because this alternative would not contain the proposed CPU policies and land use changes intended to improve compatibility with and implement the San Diego General Plan. The No Project Alternative would not benefit from the amendments to the Historical Resources Regulations in the Land Development Code that would provide supplemental development regulations pertaining to potential historic districts under the proposed Uptown CPU. Additionally, this alternative would allow greater building heights in certain areas compared to the height limits that would be applied with the CPIOZs proposed with the Uptown CPU and associated discretionary actions. Increased building heights could create additional land use conflicts compared to the project. Thus, land use compatibility impacts would be slightly greater compared to the project.

b. Visual Effects and Neighborhood Character

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in potentially greater impacts related to visual effects and neighborhood character because the alternative would retain the higher densities under the adopted Community Plan and would allow building heights up to 150 feet in areas of Mission Hills and 200 feet in areas of Hillcrest. These increased building heights combined with the absence of the proposed Uptown CPU policies that address visual effects and neighborhood character would result in the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative having greater impacts that the proposed Uptown CPU and associated discretionary actions.

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not benefit from the amendments to the Historical Resources Regulations in the Land Development Code for supplemental development regulations pertaining to potential historic districts and would thus, not provide the same level of protections to the historic neighborhood character of some communities. Thus, visual and neighborhood character impacts would be slightly greater compared to the project.

c. Transportation and Circulation

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would generate more vehicular trips than the proposed Uptown CPU and associated discretionary actions as it allows for more residential units than the proposed Uptown CPU and associated discretionary actions. However, the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not contain the proposed Uptown CPU and associated discretionary actions policies intended to promote a multimodal network that encourage walking, bicycling, and transit. Impacts to individual intersections and roadway segments would be slightly greater under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative due to the increased development potential. Similar to the proposed Uptown CPU, the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and unavoidable impact to streets, intersections and freeway segments and ramp meters. Compared to the proposed Uptown CPU, impacts to roadway and freeway facilities would be slightly more severe.

Regarding consistency with applicable plans and policies related to alternative transportation, the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not include the proposed CPU policies that support increasing multi-modal opportunities within the CPU area consistent with the SANDAG RTP, the City's General Plan, or the City's CAP. Thus, while the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not result in significant impacts related to conflicts with plans and policies addressing alternative transportation, it would not achieve the level of consistency with these applicable plans and policies that the proposed Uptown CPU would achieve. Thus, impacts related to alternative transportation would be slightly greater.

d. Air Quality

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would retain the existing Uptown Community Plan land uses. Air Quality impacts under this alternative would be slightly greater than the anticipated impacts of the proposed Uptown CPU due to slightly greater density. Like the proposed Uptown CPU, the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not conflict with or obstruct implementation of the applicable air quality plan nor would it result in a violation of any air quality standard or contribute substantially to an existing or projected air quality violation because the land uses under existing community plan would be consistent with the RAQS. However, the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative's future operational emissions would be greater than those of the proposed Uptown CPU.

e. Greenhouse Gas Emissions

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in <u>535,684488,993</u> MT CO₂E GHG emissions which would be slightly greater than the estimated <u>523,925478,184</u> MT CO₂E GHG emissions for the proposed Uptown CPU. The decrease in GHGs associated with the proposed Uptown CPU is a direct result of the proposed land use changes including an increase in commercial uses and decrease in residential, which would implement CAP Strategies and the General Plan's City of Villages Strategy. Since the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would not adjust the land use map or provide policies to implement these strategies, GHG impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be significant and unavoidable and greater than the proposed Uptown CPU.

f. Noise

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would retain the adopted Uptown Community Plan. Noise impacts under this alternative would be similar to the anticipated impacts of the proposed Uptown CPU and associated discretionary actions because, like the proposed Uptown CPU, it would permit development that would be subject to ambient noise increases and traffic noise as the planning area is built out. Although the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in slightly greater development potential, the increase in traffic noise would not likely be perceptible compared to the proposed Uptown CPU and associated discretionary actions. While the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative does not contain the proposed Uptown CPU policy changes intended to improve compatibility with and implement the General Plan policies, future development implemented under both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the proposed Uptown CPU and associated discretionary actions would be required to comply with applicable City and State noise regulations including Title 24 building code requirements. The noise impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be similar to the proposed Uptown CPU and both would result in significant and unavoidable impacts related to ambient noise increases, traffic noise exposure, and construction vibration impacts.

g. Historical Resources

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would retain the existing Uptown Community Plan, with no additional regulations addressing Potential Historic Districts (refer to Section 6.7, Historical Resources for details on the proposed Uptown CPU protections for historical resources). While the City's Historical Resources regulations provide for the regulation and protection of historical resources, it is impossible to ensure the successful preservation of all historical resources within the plan area. Therefore, potential impacts to the Historic Districts would remain significant and unavoidable. Under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative the proposed supplemental regulations to the Historical Resources Regulations of the Land Development Code addressing potential historic districts would not be implemented. Therefore, the potential loss of historical resources would be greater under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative Code addressing potential historic districts would not be implemented. Therefore, the potential loss of historical resources would be greater under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative CPU and associated discretionary actions.

As with the proposed Uptown CPU, future development under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative has the potential to result in significant direct and/or indirect impacts to archaeological resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.7, Historical Resources. The extent of impacts to archaeological resources resulting from implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in potentially significant impacts related to archaeological resources at the program level that would be significant and unavoidable, despite adherence to the existing regulatory framework.

h. Biological Resources

Under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative the MHPA boundary corrections proposed in the proposed Uptown CPU and associated discretionary actions would not be implemented, and it is likely that the amount of preserved open space would be less than with the proposed project. Therefore, the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in reduced protections for MHPA lands and environmentally sensitive resources and thus, greater impacts to biological resources than those anticipated under the proposed Uptown CPU. Implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be required to adhere to all applicable federal, state, and local regulations regarding the protection of biological resources, as for all subsequent development project submittals under the proposed Uptown CPU. Other than the greater impacts to MHPA lands and environmentally sensitive lands due to lack of implementation of MHPA boundary corrections, all other biological resource impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be the same as the proposed Uptown CPU and associated discretionary actions.

i. Geologic Conditions

Geologic impacts from implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be similar to those of the proposed Uptown CPU and associated discretionary actions. Potential impacts related to seismic and geologic hazards, or to the instability of geological units and soils would be avoided or reduced to less than significant through adherence to existing state and local regulations, including the California Building Code, the San Diego Municipal Code, and the Seismic Hazards Mapping Act. Where required, site-specific geotechnical investigations would be conducted to identify and evaluate seismic hazards and formulate mitigation measures prior to permitting most developments designed for human occupancy. Similarly, project-level compliance with City-mandated grading requirements, and, compliance with applicable State and/or Federal regulations would ensure that future grading and construction activities would avoid significant soil erosion impacts. These requirements would apply equally to both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the Proposed Uptown CPU and associated discretionary actions, thus impacts of this alternative would be similar to the project.

j. Paleontological Resources

As with the proposed Uptown CPU and associated discretionary actions, future development under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative has the potential to result in significant direct and/or indirect impacts to paleontological fossil resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.10, Paleontological Resources. The extent of impacts to paleontological resources resulting from implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would result in potentially significant impacts related to paleontological resources at the program level because adherence to the mitigation framework cannot be guaranteed for ministerial projects that only require a grading permit. Thus impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be significant and unavoidable and similar to the proposed Uptown CPU and associated discretionary actions.

k. Hydrology and Water Quality

The land use pattern and distribution for the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative is generally the same as for the proposed Uptown CPU. Future development under both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the proposed Uptown CPU and associated discretionary actions would be required to comply with existing federal, state, and local regulations relative to runoff and water quality at the project level, which would preclude the potential for hydrology and water quality impacts. Thus impacts of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be less than significant and would be similar to the proposed Uptown CPU and associated discretionary actions.

I. Public Services and Facilities

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would retain the existing Uptown Community Plan. Impacts to Public Services and Facilities under this alternative would be similar to the anticipated impacts associated with the proposed Uptown CPU because the anticipated population at build-out of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be only marginally more than the anticipated population for the build-out of the proposed Uptown CPU and would not have a measurable impact on demand for facilities. For police and fire protection services the difference in population would not impact either the police or fire department in their ability to provide service, nor would the departments require the construction of new facilities. For both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the proposed Uptown CPU, future projects would be required to pay required school fees which would mitigate for the potential impacts to schools to less than significant. Similarly both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the proposed Uptown CPU would include financing mechanisms to provide for libraries. However, in the case of both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the proposed Uptown CPU there would be a deficit in population based parks and the need to build new recreational facilities. For both the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative and the proposed Uptown CPU and associated discretionary actions, any new facilities would require a separate environmental review to identify potential impacts associated with the construction of facilities. Thus, for the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative, public facilities and services impacts would be less than significant and the same as the proposed Uptown CPU.

m. Public Utilities

The Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would retain the existing Uptown Community Plan. Impacts to Public Utilities under this alternative would be similar to the anticipated impacts of the proposed Uptown CPU. Although the proposed Uptown CPU would have a slightly lower anticipated population than the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative (see Table 10-3), implementation of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative similar impacts related to storm water, sewer, water, communications, solid waste and recycling, or energy as the proposed Uptown CPU.

n. Health and Safety

Impacts of build-out of the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative would be similar to the potential impacts of build-out of the proposed Uptown CPU and associated discretionary actions. Future development under the Adopted Community Plan with Removal of the Interim Height Ordinance Alternative has the potential to result in exposure to hazardous materials, wastes, or emissions; airport hazards, and fire hazards. However, land uses under the Adopted Community Plan with Removal of the Interim Height Ordinance alternative would be similar to the land uses under the proposed Uptown CPU. Additionally, there would not be any areas of change or land use changes that would increase potential exposure to hazards. Federal, state and local regulations that serve to reduce impacts a less-than-significant level would also reduce impacts for development under the Adopted Community Plan with Removal of the Interim Height Ordinance alternative. Overall, impacts would be less than significant and similar to those anticipated under the proposed Uptown CPU.

10.3 Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative

10.3.1 Description

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would use the adopted Uptown community plan land use map. The alternative would addresses neighborhood character issues by implementing the new proposed urban design policies that address objectives such as creating development transitions between new development and existing neighborhoods, increasing the urban tree canopy, and supporting sustainable development. Under this alternative, the current zoning program which includes the Mid-City Communities Plan District and the West Lewis Plan District would be retained with the exception of the Interim Height Ordinance (O-20329) which would be rescinded. Figure 10-2 shows the maximum building heights in areas affected by the Interim Height Ordinance that would apply under this alternative. The proposed project CPIOZ would reduce heights in areas of Mission Hills and Hillcrest compared to building heights that would be allowed under the Proposed CPU Policies with the Adopted Community Plan Land Use Map Alternative.

The build-out assumptions and land use map would be identical to the No Project (Adopted Community Plan) Alternative as detailed in Table 10-3 and Figure 10-1. Similar to the proposed project, this alternative would also address potential historical resource impacts by amending the Historical Resources Regulations in the Land Development Code to provide supplemental development regulations pertaining to potential historic districts. Application of the proposed Uptown CPU policies related to urban design and mobility under this alternative would provide design guidance including development transitions to new development, and would support multimodal transportation choices.

10.3.2 Analysis of Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative

a. Land Use

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would retain the adopted Community Plan land uses, would apply proposed CPU policies, and retain the zoning program which includes the Mid-City Communities Plan District and the West Lewis Plan District with the exception of the Interim Height Ordinance which would be rescinded. Application of the proposed CPU policies under this alternative would ensure consistency with the City's General Plan City of Villages Strategy, the City's CAP policies, and other applicable land use plans and policies because the proposed Uptown CPU was designed to provide consistency with and implement existing land use plans.

This alternative would result in a slight increase in development potential within areas served by transit, which would ensure consistency with the City of Villages strategy. The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would facilitate transit-oriented development and mixed use development similar to the proposed Uptown CPU. Implementation of the proposed CPU policies would improve compatibility with and implement the San Diego General Plan. Thus, potential land use impacts would be less than significant and would be similar to the project.

b. Visual Effects and Neighborhood Character

Potential visual effects and impacts to neighborhood character under the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be similar to those anticipated under the No Project Alternative (Adopted Community Plan). Like the proposed Uptown CPU, the Proposed CPU Policies Alternative with Adopted Community Plan Land Use Map would allow development that would be subject to proposed CPU policies addressing view sheds, neighborhood character and preservation of potential historic districts. The Proposed CPU Policies with Adopted Community Plan Land Use Map with would allow increased height structures in certain areas of Hillcrest and Mission Hills compared to the proposed Uptown CPU. However, overall the bulk and scale of development under this alternative would be similar to the proposed No Project Alternative (Adopted Community Plan) and with implementation of the proposed CPU policies, impacts related to the visual environment and neighborhood character would be similar.

c. Transportation and Circulation

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would generate slightly more vehicular trips than the proposed Uptown CPU due to increased development potential. While the Adopted Community Plan with Proposed CPU Policies Alternative would contain the proposed Uptown CPU policies intended to promote a multimodal network that encourage walking, bicycling, and taking transit, the impacts to individual intersections and roadway segments would be greater than the proposed Uptown CPU, and like the proposed Uptown CPU these impacts would remain significant and unmitigated.

Similar to the Uptown CPU, this Alternative would incorporate polices that would support the goal of creating a multi-modal transportation network in the community that supports all users by facilitating transit-oriented development, improving pedestrian amenities to address natural challenges in topography and the existing vehicular dominated environment, and the creation of an integrated bicycle network that will facilitate bicycling and help meet the travel needs of the community. Thus, potential impacts related to alternative transportation would be less than significant and the same as the project.

d. Air Quality

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would slightly increase the amount of traffic generated. Thus, air quality impacts under this alternative would be slightly greater than the anticipated impacts to the proposed Uptown CPU. Like the proposed Uptown CPU, the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would not conflict with or obstruct implementation of the applicable air quality plan nor would it result in a violation of any air quality standard or contribute substantially to an existing or projected air quality violation. However the Higher Density Alternative's future operational emissions would be greater than those of the proposed Uptown CPU.

e. Greenhouse Gas Emissions

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would result in 535,684488,993 MT CO₂E GHG emissions which would be slightly greater than the estimated 523,925478,184 MT CO₂E GHG emissions for the proposed Uptown CPU. The decrease in GHGs associated with the proposed Uptown CPU is a direct result of the proposed land use changes including an increase in commercial uses and decrease in residential, which would implement CAP Strategies and the General Plan's City of Villages Strategy. Since the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would not adjust the land use map, but would include the proposed CPU policies to implement associated CAP strategies, GHG impacts of the No Project Alternative would be less than significant, but would be greater than the proposed Uptown CPU.

f. Noise

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would result in increased densities along certain commercial corridors. Noise impacts under this alternative would be similar to the anticipated impacts to the proposed community plan because like the proposed Uptown CPU the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would permit development that could impact sensitive noise receptors. There is no mechanism in place to mitigate the noise impacts of existing noise sensitive uses or to address exterior noise impacts associated with future ministerial development. Both the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative and the proposed Uptown CPU would follow City noise regulations as well as state regulations such as the Code of Regulations Title 24; however, impacts would remain significant and unmitigated. Impacts of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be similar to the proposed Uptown CPU.

g. Historical Resources

The Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would retain the proposed implementation of supplemental development regulations to preserve the integrity and eligibility of potential historic districts. Like the proposed Uptown CPU, the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would identify Potential Historic Districts that would be subject to supplemental development regulations that limit how and where modifications can be made to residential properties that could contribute to specified potential historic districts.

Like the proposed Uptown CPU, implementation of supplemental development regulations Protecting Potential Historic Districts would provide additional protection for potential historic districts, but would not ensure the successful preservation of all historical resources within the CPU area. Therefore, potential impacts to the Historic Districts from implementation of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be significant and unavoidable and would be similar to the proposed Uptown CPU.

As with the proposed Uptown CPU, future development under the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative has the potential to result in significant direct and/or indirect impacts to archaeological resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.7, Historical Resources. The extent of impacts to archaeological resources resulting from implementation of this Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would result in potentially significant impacts related to archaeological resources at the program level and impacts would be similar to the project.

h. Biological Resources

Under the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative the boundary adjustments proposed in the proposed Uptown CPU would not be in place, and it is likely that the amount of preserved open space would be less than with the proposed project. Therefore, the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would result in greater impacts to biological resources than those anticipated under the proposed Uptown CPU. Implementation of the Proposed CPU Policies with Adoptet to adhere to all applicable federal, state, and local regulations regarding the protection of biological resources, as for all subsequent development project submittals under the proposed Uptown CPU. Therefore, impacts under this alternative would be similar, but slightly greater than those identified for the CPUs, because less developable land would be converted to open space/MHPA.

i. Geologic Conditions

Impacts from the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be similar to those of the proposed Uptown CPU. Potential impacts related to seismic and geologic hazards, or to the instability of geological units and soils would be avoided or reduced to less than significant through adherence to existing state and local regulations, including the California Building Code, the San Diego Municipal Code, and the Seismic Hazards Mapping Act. Where required, site-specific geotechnical investigations would be conducted to identify and evaluate seismic hazards and formulate mitigation measures prior to permitting most developments designed for human occupancy. Similarly, project-level compliance with City-mandated grading requirements, and, if necessary, National Pollutant Discharge Elimination System (NPDES) General

Construction Storm Water Permit provisions and a prepared site-specific Storm water Pollution Prevention Plan would ensure that future grading and construction activities would avoid significant soil erosion impacts. Thus, impacts of this alternative would be similar to the proposed Uptown CPU.

j. Paleontological Resources

As with the proposed Uptown CPU, future development under the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative has the potential to result in significant direct and/or indirect impacts to paleontological fossil resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 7.10, Paleontological Resources. The extent of impacts to paleontological resources resulting from implementation of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Lower-Density Alternative would result in potentially significant paleontological resource impacts associated with future ministerial development. Strict adherence to the Mitigation Framework would still be required to reduce potential impacts to below a level of significance; however, impacts associated with future ministerial development would remain significant and unavoidable, the same as the proposed Uptown CPU.

k. Hydrology and Water Quality

The land use pattern and distribution for the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative is generally the same as the proposed Uptown CPU. Because the alternative would not implement the boundary adjustments proposed in the Uptown CPU, it is likely that less open space would be preserved under the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative when compared to the proposed Uptown CPU. Future development would be required to comply with existing federal, state, and local regulations relative to runoff and water quality at the project level, which would preclude the potential for impacts under both the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative and the proposed Uptown CPU. Thus, impacts of this alternative would be similar to the proposed Uptown CPU.

I. Public Services and Facilities

Impacts to Public Services and Facilities under the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be similar to the anticipated impacts of the proposed Uptown CPU. While the anticipated population at build-out of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be more than the anticipated population for the build-out of the proposed Uptown CPU, the change would not affect availability of services. For police and fire protection services the difference in population would not impact either the police or fire department in their ability to provide service, nor would the departments require the construction of new facilities. For both the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative and the proposed Uptown CPU future projects would be required to pay for any potential impacts to schools reducing these impacts to less than significant. Similarly both the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative and the proposed Uptown CPU include financing mechanisms to provide for libraries. However, in the case of both the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative and the proposed Uptown CPU there results in a deficit in population based parks and the need to build new recreational facilities. However, like the proposed Uptown CPU and associated discretionary actions, any future construction of a public facility would require a separate environmental review to ensure impacts of construction and operation of the facility are addressed. Thus, public facilities and services impacts of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be less than significant and similar to the proposed Uptown CPU.

m. Public Utilities

Impacts to Public Utilities under this alternative would be similar to the anticipated impacts to the proposed Uptown CPU. Like the proposed Uptown CPU, the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative contains the proposed Uptown CPU policies and land use changes intended to improve compatibility with and implement the San Diego General Plan, the anticipated population at build-out of the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative is greater than the anticipated population of the proposed Uptown CPU. As discussed in section 6.13, Public Utilities of this PEIR, implementation of the proposed Uptown CPU would not result in significant impacts to storm water, sewer, water, communications, solid waste and recycling, or energy. While it is anticipated that the population will increase, the impacts related to providing storm water, sewer, water, communications, solid waste and recycling, or energy to serve future development under this alternative would be similar to the proposed Uptown CPU and would be less than significant.

n. Health and Safety

Impacts from the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative would be similar to the proposed Uptown CPU. Future development under the Higher Density Alternative has the potential to result in exposure to hazardous materials, wastes, or emissions; airport hazards, and fire hazards. Federal, state and local regulations that serve to reduce impacts a less-than-significant level would also cover the Proposed CPU Policies with Adopted Community Plan Land Use Map Alternative. Overall, impacts would be less than significant and similar to the impacts anticipated under the proposed Uptown CPU.

10.4 Density Redistribution Alternative

10.4.1 Description

The Density Redistribution Alternative uses land uses proposed in June 2015 Draft Community Plan without the corresponding density bonus incentives originally proposed with this land use scenario, which is lower than the adopted community plan. Under this alternative, the density of future development would be lower along transit commercial nodes except for the transit corridor along Park Boulevard between University Avenue and Washington Street and Normal Street. Under this alternative, the reduction in density would be redistributed resulting in the same overall development potential as the proposed Uptown CPU. The locations and associated density decreases from the proposed Uptown CPU are shown on Figure 10-4 and are noted below:

- 1. India Street (Neighborhood Commercial 0-29 du/ac)
- 2. Reynard Way (Residential Medium 16-29 du/ac and Neighborhood Commercial 0-29 du/ac)
- 3. 4th Avenue between Upas and Spruce (Office Commercial 0-29 du/ac)
- 4. 4th Avenue between Laurel and Grape (Office Commercial 0-29 du/ac)
- 5. Bankers Hills/Park West Neighborhood west of 1st Ave (Residential Medium 16-29 du/ac)
- 6. Medical Center Complex (Neighborhood Office Commercial 0-44 du/ac)
- 7. Washington Street near Dove (Community Commercial 0 44 du/ac)
- 8. Central Hillcrest (Community Commercial 0-44 du/ac)
- 9. South of Pennsylvania in Hillcrest (Community Commercial 0-73 du/ac)

When compared to the proposed Uptown CPU, the Density Redistribution Alternative reduces residential density development potential along India Street, Reynard Way, the 4th Avenue Commercial Office areas, and Bankers Hills/Park West Neighborhood from 44 du/ac to 29 du/ac. The Density Redistribution Alternative reduces areas of the Medical Center Complex, Washington Street near Dove Street, and areas within Central Hillcrest from 73 du/ac to 44 du/ac. Additionally, the core Central Hillcrest area is reduced from 109 du/ac to 44 du/ac and Hillcrest South of Pennsylvania is reduced from 109 du/ac.

There are a few areas where the Density Redistribution Alternative includes higher density than the proposed Uptown CPU. The Normal Street corner lot along Park Blvd is reduced to Community Commercial 0-44 du/ac. The Density Redistribution Alternative increases transit corridor density along Park Boulevard between University Avenue and Washington Street and Normal Street from 73 du/ac to 109 and 145 du/ac. Figure 10-4 shows the proposed Density Redistribution Alternative land use map_and Table 10-4 shows the differences between dwelling units and commercial square footage between the Density Redistribution Alternative and the proposed Uptown CPU.

Table 10-4								
Build-out Comparison of the Density Redistribution Alternative and the Proposed Uptown CPU								
Density Redistribution Alternative			Proposed Uptown CPU					
		Dwelling				Dwelling		
Land Use	<u>Acres</u>	<u>Units</u>	Floor Area	Land Use	<u>Acres</u>	<u>Units</u>	Floor Area	
<u>Agriculture</u>	<u>0.5</u>	-	-	<u>Agriculture</u>	<u>0.5</u>	- 1	-	
Education	<u>29</u>	<u>-</u>	364,200	Education	<u>29</u>	- 1	<u>364,200</u>	
Institutional	<u>100</u>	-	<u>2,121,500</u>	Institutional	<u>100</u>	-	<u>2,121,500</u>	
Multi family	<u>380</u>	<u>26,505</u>	<u>-</u>	Multi family	<u>380</u>	<u>27,180</u>	<u>-</u>	
Office Commercial	<u>32</u>	<u>-</u>	<u>1,598,700</u>	Office Commercial	<u>32</u>	-	<u>1,598,700</u>	
Open Space	<u>398</u>	<u>-</u>	<u>-</u>	Open Space	<u>398</u>	-	<u>-</u>	
Parking	<u>3</u>	-	<u>-</u>	Parking	3	-	<u>-</u>	
<u>Parks</u>	<u>48</u>	-	<u>-</u>	<u>Parks</u>	<u>48</u>	-	<u>-</u>	
Recreational	<u>3</u>	-	<u>31,100</u>	Recreational	3	-	<u>31,100</u>	
Retail Commercial	<u>175</u>	-	<u>3,186,500</u>	Retail Commercial	<u>175</u>	-	<u>3,186,500</u>	
<u>Roads</u>	<u>761</u>	-	<u>-</u>	<u>Roads</u>	761	-	<u>-</u>	
Single Family	<u>726</u>	<u>5,495</u>	<u>-</u>	Single Family	726	<u>5,520</u>	<u>-</u>	
Visitor Commercial	2	-	174,000	Visitor Commercial	2	-	174,000	
Grand Totals	<u>2,656</u>	32,000	<u>7,476,000</u>	Grand Totals	<u>2,656</u>	<u>32,700</u>	7,476,000	
Estimated Future Population = 53,500			Estimated Future Population = 55,700					



FIGURE 10-4 Density Redistribution Alternative

10.4.2 Analysis of Density Redistribution Alternative

a. Land Use

The Density Redistribution Alternative would retain the proposed Uptown CPU land uses, but further lowers density throughout the community with the exception of the Park Boulevard transit corridor between Washington Street, University Avenue and Normal Street. Land use impacts under this alternative would be similar to the anticipated impacts to the proposed Uptown CPU. The Density Redistribution Alternative would facilitate transit-oriented development and mixed use development but to a lesser degree than the proposed Uptown CPU due to reduced density near areas accessible to transit with the exception of the Park Boulevard transit corridor. Land use changes would be compatible with the implementation of the San Diego General Plan, but to a lesser degree. Like the proposed Uptown CPU, this alternative would not conflict with adopted land use plans, policies, or ordinances; however it would achieve consistency with the General Plan City of Villages strategy to a lesser extent. Thus, land use impacts of this alternative would be slightly greater than the proposed Uptown CPU.

b. Visual Effects and Neighborhood Character

Potential visual effects and impacts to neighborhood character under the Density Redistribution Alternative would be similar to those anticipated under the proposed Uptown CPU. The Density Redistribution Alternative is similar the proposed Uptown CPU densities, and like the proposed Uptown CPU would generally produce similar bulk and scale development. The Density Redistribution Alternative would also include proposed Uptown CPU policies that reduce the impact of future development on community character and related visual effects so that the overall impact in the community would be similar to the proposed Uptown CPU.

c. Transportation and Circulation

The Density Redistribution Alternative would generate fewer vehicular trips than the proposed Uptown CPU. The Uptown Community Planning Group Proposed Residential Densities Traffic Evaluation Memo dated March 30th, 2016 produced by Kimley-Horn (included as Appendix B-3 of this PEIR), summarizes the results of the traffic evaluation to reflect residential densities proposed by this Alternative.

Specific roadway segments and intersections were selected for assessment based on the percent difference in traffic volume that would result from the updated residential densities. The selected segments and intersections and the reduction in volume are provided below and described in Appendix B-3.

Roadway Segments

- First Avenue between Laurel Street and Elm Street: 9-11% decrease
- Fourth Avenue between Washington Street and Elm Street: 5-7% decrease
- Fifth Avenue between Walnut Street and Elm Street: 5-6% decrease
- University Avenue between First Avenue and Park Boulevard: 7-12% decrease

Intersections

- Washington Street and Fourth Avenue Decrease movements to and from the south leg of intersection by 5%
- University Avenue and Sixth Avenue Decrease movements to and from the east and west legs of intersection by 7%

The following intersections were found to operate at LOS D or better during both peak periods in the original traffic evaluation. Volumes at these locations would be decreased with the proposed residential densities and these intersections would be expected to operate similar or better then what was previously evaluated. No further analysis was performed at these locations as there would not be a change to the conclusions or number of impacts to the vehicle network by reducing volumes at these locations.

- University Avenue and Fourth Avenue
- University Avenue and Fifth Avenue
- University Avenue and Tenth Street
- University Avenue and Normal Street
- University Avenue and Park Boulevard
- Fourth Avenue and Robinson Avenue
- Fourth Avenue and Laurel Street
- Fifth Avenue and Laurel Street
- First Avenue and Elm Street

The traffic memo (Appendix B-3) provides a comparison of roadway segment analysis between the proposed Uptown CPU land use assumptions and the CPG-preferred residential densities, which reflects the Density Redistribution Alternative. Additionally, the memo provides an updated roadway segment impact analysis when compared to existing conditions. The decrease in volumes on First Avenue would result in similar roadway operations between Laurel Street and Elm Street as the proposed Uptown CPU. The volume reductions would remove a potential impact for the segment of First Avenue between Hawthorn Street and Grape Street. The decrease in volumes on Fourth Avenue would result in similar or improved roadway operations between Washington Street and Elm Street. The volume reductions would remove a potential impact for the segment of Avenue would result in similar or improved roadway operations between Washington Street and Elm Street. The volume reductions would remove a potential impact for the segment of Fourth Avenue between Walnut Street and Laurel Street.

The decrease in volumes on Fifth Avenue would result in similar roadway operations between Walnut Street and Elm Street compared to the proposed Uptown CPU. The volume reductions would not remove any potential impacts on Fifth Avenue. The decrease in volumes on University Avenue would result in similar roadway operations between First Avenue and Park Boulevard. The volume reductions would not remove any potential impacts on University Avenue. The proposed change in residential densities would remove two potential impacts to roadway segments compared to the proposed Uptown CPU.

According analysis, the decrease in volumes at Washington Street and Fourth Avenue would result in similar operations to the proposed Uptown CPU. There would continue to be a significant impact during the afternoon peak. The decrease in volumes at University Avenue and Sixth Avenue would remove the significant impact during the afternoon peak. This intersection would no longer have a

potentially significant impact. The proposed change in residential densities would remove one potential impact to intersections.

The Density Redistribution Alternative would contain the proposed Uptown CPU policies intended to promote a multimodal network that encourage walking, bicycling, and taking transit; however these goals would be achieved to a lesser extent due to the reductions in development potential within areas accessible to transit. While the impacts to individual intersections and roadway segments would be less than the proposed Uptown CPU, like the proposed Uptown CPU these impacts would remain significant and unmitigated.

d. Air Quality

The Density Redistribution Alternative would decrease the amount of traffic generated. Air Quality impacts under this alternative would be less than the anticipated impacts resulting from the proposed Uptown CPU. Like the proposed Uptown CPU, the Density Redistribution Alternative would not conflict with or obstruct implementation of the applicable air quality plan nor would it result in a violation of any air quality standard or contribute substantially to an existing or projected air quality violation. The Density Redistribution Alternative's future operational emissions would be lower than those of the proposed Uptown CPU.

e. Greenhouse Gas Emissions

The Density Redistribution Alternative would decrease development potential over the proposed Uptown CPU, which could reduce GHG emissions; however, the GHG efficiencies of providing development in proximity to transit would be lost, with the exception of the density provided along Park Boulevard. The decrease in development potential would accommodate approximately 1585 fewer units within the Density Redistribution Alternative in areas where residents would have convenient access to transit and commercial services. This would result in a potential conflict with the implementation of CAP Strategies and the General Plan's City of Villages Strategy. Decreasing residential and commercial density in transit corridors and Community Villages within a Transit Priority Area (TPA) would not support the City of San Diego in achieving the GHG emissions reduction targets of the CAP since these residents would need to find housing or employment elsewhere that may not have accessibility to transit. While speculative as to the availability and use of transit elsewhere, it is likely that the residents not within a transit priority area would make greater use of autos – either through not having transit available or longer commute distances – thereby increasing vehicle miles traveled as compared to the proposed Uptown CPU, and thus, GHG emission impacts associated with the Density Redistribution Alternative would be greater than the project.

f. Noise

The Density Redistribution Alternative would result in decreased densities along certain commercial corridors. Noise impacts under this alternative would be similar or lower than the anticipated impacts under the proposed Uptown CPU because like the proposed Uptown CPU, the Density Redistribution Alternative would permit development that could impact existing and future sensitive noise receptors. Both the Density Redistribution Alternative and the proposed Uptown CPU would

follow City noise regulations as well as state regulations such as the Code of Regulations Title 24. However, the resulting noise impacts for both the Density Redistribution Alternative and the proposed Uptown CPU would remain significant and unmitigated.

g. Historical Resources

The Density Redistribution Alternative would retain the proposed implementation of supplemental development regulations to preserve the integrity and eligibility of potential historic districts. Like the proposed Uptown CPU, the Density Redistribution Alternative would identify Potential Historic Districts that would be subject to supplemental development regulations that limit how and where modifications can be made to residential properties that could contribute to specified potential historic districts.

Like the proposed Uptown CPU, implementation of supplemental development regulations Protecting Potential Historic Districts would provide additional protection for potential historic districts, but would not ensure the successful preservation of all historical resources within the CPU area. Therefore, potential impacts to the Historic Districts from implementation of the Density Redistribution Alternative would be significant and unavoidable and would be similar to the proposed Uptown CPU.

As with the proposed Uptown CPU, future development under the Density Redistribution Alternative has the potential to result in significant direct and/or indirect impacts to archaeological resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.7, Historical Resources. The extent of impacts to archaeological resources resulting from implementation of the Density Redistribution Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Density Redistribution Alternative would result in potentially significant impacts related to archaeological resources at the program level and impacts would be similar to the project.

h. Biological Resources

Like the proposed Uptown CPU, the Density Redistribution Alternative would include MHPA boundary adjustments. The Density Redistribution Alternative would result in similar impacts to biological resources as those anticipated under the proposed Uptown CPU. Implementation of the Density Redistribution Alternative would also be required to adhere to all applicable federal, state, and local regulations regarding the protection of biological resources, as for all subsequent development project submittals under the proposed Uptown CPU. Therefore, impacts under this alternative would be similar to those identified for the proposed Uptown CPU and would be less than significant.

i. Geologic Conditions

Impacts from the Density Redistribution Alternative would be similar to those of the proposed Uptown CPU. Potential impacts related to seismic and geologic hazards, or to the instability of

geological units and soils would be avoided or reduced to less than significant through adherence to existing state and local regulations, including the California Building Code, the San Diego Municipal Code, and the Seismic Hazards Mapping Act. Where required, site-specific geotechnical investigations would be conducted to identify and evaluate seismic hazards and formulate mitigation measures prior to permitting most developments designed for human occupancy. Similarly, project-level compliance with City-mandated grading requirements, and, if necessary, NPDES General Construction Storm Water Permit provisions and a prepared site-specific Storm water Pollution Prevention Plan would ensure that future grading and construction activities would avoid significant soil erosion impacts. Since the regulatory framework would apply to the Density Redistribution Alternative, impacts would be the same as the proposed Uptown CPU.

j. Paleontological Resources

As with the proposed Uptown CPU, future development under the Density Redistribution Alternative has the potential to result in significant direct and/or indirect impacts to paleontological fossil resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.10, Paleontological Resources. The extent of impacts to paleontological resources resulting from implementation of the Density Redistribution Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Density Redistribution Alternative would result in potentially significant impacts related to paleontological resources at the program level. Strict adherence to the Mitigation Framework would still be required to reduce potential impacts; however impacts associated with future ministerial development would remain significant and unavoidable, the same as the proposed Uptown CPU.

k. Hydrology and Water Quality

The land use pattern and distribution for the Density Redistribution Alternative is similar to the proposed Uptown CPU and the same areas would ultimately be developed. Like the Proposed Community Plan the Density Redistribution Alternative would implement the boundary adjustments proposed in the Community Plan it is likely that the same amount of open space would be preserved under the Density Redistribution Alternative when compared to the proposed Uptown CPU. Future development would be required to comply with existing federal, state, and local regulations relative to runoff and water quality at the project level, which would preclude the potential for impacts under both the Density Redistribution Alternative and the proposed Uptown CPU. Thus, impacts of this alternative would be similar as the proposed Uptown CPU.

I. Public Services and Facilities

Impacts to Public Services and Facilities under the Density Redistribution Alternative would be similar to the anticipated impacts to the proposed community plan because the anticipated population at build-out of the Density Redistribution Alternative similar to the population for the build-out of the proposed Uptown CPU. For both the Density Redistribution Alternative and the proposed Uptown CPU future projects would be required to pay for any potential impacts to schools reducing these impacts to less than significant. Similarly, both the Density Redistribution Alternative

and the proposed Uptown CPU include financing mechanisms to provide for libraries. Thus impacts to public facilities and services under this alternative would be similar to the proposed Uptown CPU and associated discretionary actions.

m. Public Utilities

Impacts to Public Utilities under this alternative would be similar to the anticipated impacts to the proposed Uptown CPU. Like the proposed Uptown CPU, the Density Redistribution Alternative contains the proposed CPU policies intended to improve compatibility with and implement the San Diego General Plan. As discussed in section 6.13, Public Utilities, the implementation of the proposed Uptown CPU would not result in significant impacts to storm water, sewer, water, communications, solid waste and recycling, or energy. As the population in the Density Redistribution Alternative would similar to the proposed Uptown CPU, impacts to storm water, sewer, sewer, water, sewer, water, communications, solid waste and recycling, or energy would be similar.

n. Health and Safety

Impacts from the Density Redistribution Alternative would be similar to the proposed Uptown CPU. Future development under the Density Redistribution Alternative has the potential to result in exposure to hazardous materials, wastes, or emissions; airport hazards, and fire hazards. Federal, state and local regulations that serve to reduce impacts a less-than-significant level for the proposed Uptown CPU, would also address impacts under the Density Redistribution Alternative. Overall, impacts would be less than significant and similar to those anticipated under the proposed Uptown CPU.

10.5 Lower-Density Alternative

10.5.1 Description

The Lower-Density Alternative incorporates the land uses proposed in June 2015 Draft Community Plan without the corresponding density bonus incentives originally proposed with this land use scenario. The Lower-Density Alternative would be the same as the Density Redistribution Alternative with the exception that density would not increase along the Park Boulevard generally between Washington Street, University Avenue, and Normal Street.

Table 10-4–<u>5</u> presents a summary comparison of the Proposed Uptown CPU and Lower-Density Alternative for residential capacity and reasonably anticipated non-residential development. As shown, the Lower-Density Alternative would reduce multi-family development potential and result in a slight increase in single family development potential. The total projected population under the Lower-Density Alternative would be 2,650 persons less than under the proposed Uptown CPU. Figure 10-5 shows land use designations under the Lower-Density Alternative.

10.0 Alternatives

Table 10-5 Build-out Comparison of the Lower-Density Alternative and the Proposed Uptown CPU								
Proposed Uptown CPU				Lower-Density Alternative				
		Dwelling				Dwelling		
Land Use	Acres	Units	Floor Area	Land Use	Acres	Units	Floor Area	
Agriculture	0.5	-	-	Agriculture	0.5	-	-	
Education	29	-	364,200	Education	29	-	364,200	
Institutional	100	-	2,121,500	Institutional	100	-	2,121,500	
Multi Family	380	27,180	-	Multi Family	380	25,570	-	
Office Commercial	32	-	1,598,700	Office Commercial	32	-	1,598,700	
Open Space	398	-	-	Open Space	398	-	-	
Parking	3	-	-	Parking	3	-	-	
Parks	48	-	-	Parks	48	-	-	
Recreational	3	-	31,100	Recreational	3	-	31,100	
Retail Commercial	175	-	3,186,500	Retail Commercial	175	-	3,186,500	
Roads	761	-	-	Roads	761	-	-	
Single Family	726	5,520	-	Single Family	726	5,530	-	
Visitor Commercial	2	-	174,000	Visitor Commercial	2	-	174,000	
Grand Totals	2,656	32,700	7,476,000	Grand Totals	2,656	31,100	7,476,000	
Estimated Future Population =55,700			Estimated Future Population =53,050					



FIGURE 10-5 Lower-Density Alternative

a. Land Use

The Lower-Density Alternative would retain the proposed Uptown CPU land uses, but would lower multi-family density throughout the community along transit corridors and nodes. Land use impacts under this alternative would be similar to the anticipated impacts to the proposed Uptown CPU. The Lower-Density Alternative would facilitate transit-oriented development and mixed use development, but to a lesser degree than the proposed Uptown CPU due to reduced density near areas within proximity to transit. Land use changes would be compatible with the implementation of the General Plan, but to a lesser degree. Like the proposed Uptown CPU, it would not conflict with adopted land use plans, policies, or ordinances, but land use impacts of this alternative would be slightly greater than the proposed Uptown CPU due to reduced land use plans.

b. Visual Effects and Neighborhood Character

Potential visual effects and impacts to neighborhood character under the Lower-Density Alternative would be similar to those anticipated under the proposed Uptown CPU. The Lower-Density Alternative would generally produce similar bulk and scale development as the proposed Uptown CPU land uses. The Lower-Density Alternative would also include proposed Uptown CPU policies that reduce the impact of future development on community character and related visual effects so that the overall impact in the community would be similar to the proposed Uptown CPU.

c. Transportation and Circulation

The Lower-Density Alternative would generate fewer vehicular trips than the proposed Uptown CPU due to reduced multi-family residential located along Park Boulevard generally between Washington Street, University Avenue, and Normal Street; however even with these reductions, this alternative would result in significant and unavoidable impacts to roadway segments and intersections, the same as the proposed Uptown CPU. Transportation and circulation impacts would likely be similar to the proposed Uptown CPU.

The Lower-Density Alternative would contain the proposed Uptown CPU policies intended to promote a multimodal network that encourage walking, bicycling, and taking transit; however, these goals would be achieved to a lesser extent due to the reductions in development potential within areas accessible to transit. Thus, alternative transportation impacts of the Lower-Density Alternative would be slightly greater than the proposed Uptown CPU.

d. Air Quality

The Lower-Density Alternative would decrease the amount of traffic generated; however the potential decreases in traffic and associated air quality emissions associated with decreases in development potential, could be cancelled out by the fact that the reductions would occur in areas accessible to transit. Thus, air quality impacts under this alternative would likely be similar to the proposed Uptown CPU. Like the proposed Uptown CPU, the Lower-Density Alternative would not conflict with or obstruct implementation of the applicable air quality plan nor would it result in a violation of any air quality standard or contribute substantially to an existing or projected air quality

violation. The Lower-Density Alternative's future operational emissions would be similar to those of the proposed Uptown CPU.

e. Greenhouse Gas Emissions

The Lower-Density Alternative would decrease development potential over the proposed Uptown CPU, which could reduce GHG emissions; however, the GHG efficiencies of providing development in proximity to transit would be lost. The decrease in development potential would accommodate approximately 1,610 fewer multi-family units within the Lower-Density Alternative in areas where residents would have convenient access to transit and commercial services. This would result in a potential conflict with the implementation of CAP Strategies and the General Plan's City of Villages Strategy. Decreasing residential and commercial density in transit corridors and Community Villages within a TPA would not support the City of San Diego in achieving the GHG emissions reduction targets of the CAP since these residents would need to find housing or employment elsewhere that may not have accessibility to transit. While speculative as to the availability and use of transit elsewhere, it is likely that the residents not within a transit priority area would make greater use of autos – either through not having transit available or longer commute distances – thereby increasing vehicle miles traveled as compared to the proposed Uptown CPU, and thus, GHG emission impacts associated with the Lower-Density Alternative would be greater than the project.

f. Noise

The Lower-Density Alternative would result in decreased densities along commercial mixed use transit corridors and nodes and multifamily areas. Noise impacts under this alternative would be similar to than the anticipated impacts under the proposed Uptown CPU because like the proposed Uptown CPU, the Lower-Density Alternative would permit development that could impact existing and future sensitive noise receptors. There is no mechanism in place to mitigate the noise impacts of existing noise sensitive uses or to address exterior noise impacts associated with future ministerial development. Both the Lower-Density Alternative and the proposed Uptown CPU would follow City noise regulations as well as state regulations such as the Code of Regulations Title 24; however, impacts would remain significant and unmitigated. Impacts of the Lower-Density Alternative would be similar to the proposed Uptown CPU.

g. Historical Resources

The Lower-Density Alternative would retain the proposed implementation of supplemental development regulations to preserve the integrity and eligibility of potential historic districts. Like the proposed Uptown CPU, the Lower-Density Alternative would identify Potential Historic Districts that would be subject to supplemental development regulations that limit how and where modifications can be made to residential properties that could contribute to specified potential historic districts.

Like the proposed Uptown CPU, implementation of supplemental development regulations Protecting Potential Historic Districts would provide additional protection for potential historic districts, but would not ensure the successful preservation of all historical resources within the CPU area. Therefore, potential impacts to the Historic Districts from implementation of the LowerDensity Alternative would be significant and unavoidable and would be similar to the proposed Uptown CPU.

As with the proposed Uptown CPU, future development under the Lower-Density Alternative has the potential to result in significant direct and/or indirect impacts to archaeological resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.7, Historical Resources. The extent of impacts to archaeological resources resulting from implementation of the Lower-Density Alternative would be similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Lower-Density Alternative would result in potentially significant impacts related to archaeological resources at the program level and impacts would be similar to the project.

h. Biological Resources

Like the proposed Uptown CPU, the Lower-Density Alternative would include MHPA boundary adjustments. The Lower-Density Alternative would result in similar impacts to biological resources as those anticipated under the proposed Uptown CPU. Implementation of the Lower-Density Alternative would also be required to adhere to all applicable federal, state, and local regulations regarding the protection of biological resources, as for all subsequent development project submittals under the proposed Uptown CPU. Therefore, impacts under this alternative would be similar to those identified for the proposed Uptown CPU and would be less than significant.

i. Geologic Conditions

Impacts from the Lower-Density Alternative would be similar to those of the proposed Uptown CPU. Potential impacts related to seismic and geologic hazards, or to the instability of geological units and soils would be avoided or reduced to less than significant through adherence to existing state and local regulations, including the California Building Code, the San Diego Municipal Code, and the Seismic Hazards Mapping Act. Where required, site-specific geotechnical investigations would be conducted to identify and evaluate seismic hazards and formulate mitigation measures prior to permitting most developments designed for human occupancy. Similarly, project-level compliance with City-mandated grading requirements, and, if necessary, NPDES General Construction Storm Water Permit provisions and a prepared site-specific Storm water Pollution Prevention Plan would ensure that future grading and construction activities would avoid significant soil erosion impacts. Since the regulatory framework would apply to the Lower-Density Alternative, impacts would be the same as the proposed Uptown CPU.

j. Paleontological Resources

As with the proposed Uptown CPU, future development under the Lower-Density Alternative has the potential to result in significant direct and/or indirect impacts to paleontological fossil resources. Implementation of future projects under this alternative would require adherence to all applicable guidelines further described in Section 6.10, Paleontological Resources. The extent of impacts to paleontological resources resulting from implementation of the Lower-Density Alternative would be

similar to those identified for the proposed Uptown CPU, because the extent and areas of disturbance by development would be generally the same and only the land use designation would change. As with the proposed Uptown CPU, implementation of the Lower-Density Alternative would result in potentially significant paleontological resource impacts associated with future ministerial development. Strict adherence to the mitigation framework would still be required to reduce potential impacts; however, impacts associated with future ministerial development would remain significant and unavoidable, the same as the proposed Uptown CPU.

k. Hydrology and Water Quality

The land use pattern and distribution for the Lower-Density Alternative is generally the same as the proposed Uptown CPU. Like the Proposed Community Plan the Lower-Density Alternative would implement the MHPA boundary adjustments proposed in the Community Plan it is likely that the same amount of open space would be preserved under the Lower-Density Alternative when compared to the proposed Uptown CPU. Future development would be required to comply with existing federal, state, and local regulations relative to runoff and water quality at the project level, which would preclude the potential for impacts under both the Lower-Density Alternative and the proposed Uptown CPU. Thus, impacts of this alternative would be similar to the project.

I. Public Services and Facilities

Impacts to Public Services and Facilities under the Lower-Density Alternative would be similar or less than the anticipated impacts to the proposed community plan because the anticipated population at build-out of the Lower-Density Alternative would be less than the anticipated population for the build-out of the proposed Uptown CPU. For police and fire protection services the difference in population would not impact either the police or fire department in their ability to provide service, nor would the departments require the construction of new facilities. For both the Lower-Density Alternative and the proposed Uptown CPU future projects would be required to pay for any potential impacts to schools reducing these impacts to less than significant. Similarly, both the Lower-Density Alternative and the proposed Uptown CPU include financing mechanisms to provide for libraries. However, in the case of both the Lower-Density Alternative and the proposed Uptown CPU there results in a deficit in population based parks and the need to build new recreational facilities. With the anticipated population of the Lower-Density Alternative would require 7.2 acres less new park land and fewer recreational facilities than the proposed project, but a deficit remains. However, like the proposed Uptown CPU and associated discretionary actions, any future construction of a public facility would require a separate environmental review to ensure impacts of construction and operation of the facility are addressed. Thus, public facilities and services impacts of the Lower-Density Alternative would be less than significant and similar to the proposed Uptown CPU.

m. Public Utilities

Impacts to Public Utilities under this alternative would be similar to the anticipated impacts of the proposed Uptown CPU. Like the proposed Uptown CPU, the Lower-Density Alternative contains the proposed community plan policies and land use changes intended to improve compatibility with and implement the San Diego General Plan, the anticipated population at build-out of the Lower-Density

Alternative is lower than the anticipated population of the proposed Uptown CPU. As discussed in section 7.13, Public Utilities, the implementation of the proposed Uptown CPU would not result in significant impacts to storm water, sewer, water, communications, solid waste and recycling, or energy. It is anticipated that the population in the Lower-Density Alternative would be approximately 2,650 fewer than the proposed project. However, impacts related to providing storm water, sewer, water, communications, solid waste and recycling, and energy to future development under this alternative would be similar to the proposed Uptown CPU and impacts would be less than significant.

n. Health and Safety

Impacts from the Lower-Density Alternative would be similar to the proposed Uptown CPU. Future development under the Lower-Density Alternative has the potential to result in exposure to hazardous materials, wastes, or emissions; airport hazards, and fire hazards. As the would result in a slighter lower population growth than the proposed Uptown CPU, there would be fewer people exposed to these potential hazards. Federal, state and local regulations that serve to reduce impacts a less-than-significant level would also cover the Lower-Density Alternative. Overall, impacts would be less than significant and similar to those anticipated under the proposed Uptown CPU.

10.6 Environmentally Superior Alternative

CEQA Guidelines section 15126.6(e)(2) requires the identification of an environmentally superior alternative among the alternatives analyzed in an EIR. The guidelines also require that if the No Project Alternative is identified as the environmentally superior alternative, then another environmentally superior alternative must be identified.

Based on a comparison of the alternatives' overall environmental impacts and their compatibility with the CPUs' goals and objectives, the Density Redistribution Alternative is the environmentally superior alternative for this Program EIR. While the Density Distribution Alternative would not be able to reduce the significant and unavoidable impacts of the proposed Uptown CPU, it would reduce impacts related to traffic circulation and air quality. At the same time, the Density Redistribution Alternative would not support the full implementation of the General Plan's City of Villages strategy of developing multi-modal centers that encourage walking, bicycling, and taking transit and contain a mixture of commercial and residential development because the density of future development under the Density Redistribution Alternative would be lower along transit commercial nodes except for the transit corridor along Park Boulevard between University Avenue and Washington Street and Normal Street. The Density Redistribution Alternative could also conflict with the implementation of the City's Climate Action Plan since the redistribution of density would result in a likely increase in greenhouse gas emission impacts and vehicle miles traveled.



Chapter 11.0 Mitigation Monitoring and Reporting Program

11.1 Introduction

Section 15097 of the California Environmental Quality Act (CEQA) Guidelines requires that a Mitigation Monitoring and Reporting Program (MMRP) be adopted upon certification of an environmental impact report (EIR; including associated Findings), to ensure that the associated mitigation measures are implemented. The MMRP identifies the mitigation measures, specifies the entity (or entities) responsible for monitoring and reporting, and notes when in the process monitoring and reporting should be conducted. The MMRP for the Uptown Community Plan Update is included as an attachment to the Staff Report to be considered by the decision maker.


Chapter 12.0 References Cited

Introduction

San Diego, City of

- 1988 Uptown Community Plan, as amended.
- 2007 Program Environmental Impact Report for the General Plan.
- 2008a General Plan. March.
- 2008b Municipal Code
- 2013 Housing Element FY2013-FY2020.
- 2015 Uptown Community Plan Update Draft Historic Resources Survey Report. November.

Environmental Setting

American Ornithologist Union

2013 Check-list of North American Birds: The Species of Birds of North America from the Arctic through Panama, Including the West Indies and Hawaiian Islands. 7th ed. Committee on Classification and Nomenclature and the 54th Supplement. Accessed on May 19, 2014 at http://checklist.aou.org/taxa/.

AECOM

- 2015 Community Plan Update for the Community of Uptown Prehistoric Cultural Resources City of San Diego, California. January.
- Baker, R. J., L. C. Bradley, R. D. Bradley, J. W. Dragoo, M. D. Engstrom, R. S. Hoffmann, C. A. Jones, F. Reid, D. W. Rice, and C. Jones
 - 2003 Revised Checklist of North American Mammals North of Mexico. Occasional Papers, Museum of Texas Tech University 229:1-24.

California, State of

- 2014 State and Federally Listed Endangered, Threatened, and Rare Plants of California. The Resources Agency. California Natural Diversity Database. January.
- 2015 California Air Quality Data Statistics. California Air Resources Board Internet Site.http://www.arb.ca.gov/adam/welcome.html. Accessed April 23, 2015.California Air Resources Board (CARB)
- 2007 California 1990 Greenhouse Gas Emissions Level and 2020 Emissions Limit. November 16.
- California Department of Transportation (Caltrans)
 - 2013 Technical Noise Supplement. November.
- California Energy Commission (CEC)
 - 2014 Integrated Energy Policy Report 2013 (IEPR). February.

California Native Plant Society CNPS)

2012 Inventory of Rare and Endangered Plants of California (online edition v8-01a). California Native Plant Society. Sacramento, CA.

Hall, E. R.

1981 The Mammals of North America. 2nd ed. 2 vols. John Wiley & Sons, New York.

Kennedy, M. P., and Tan, S.S.

2008 Geology of the San Diego 30'x60' Quadrangle, California, CGS Regional Geologic Map No. 3.

Reed, L. D.

1990 *A New Upper Pleistocene Marine Sedimentary Unit, San Diego, California,* in Geotechnical Engineering Case Histories in San Diego County, San Diego Association of Geologists, p. 1-27.

Regional Water Quality Control Board

1994 Water Quality Control Plan for the San Diego Basin (9). September 8.

San Diego, City of

- 2008 General Plan. March.
- 2011 California Environmental Quality Act Significance Determination Thresholds. Development Services Department.
- 2012 City of San Diego Municipal Code, Chapter 11, Article 03, Division 01.
- 2013 Final Program Environmental Impact Report for the Barrio Logan Community Plan Update; Project No. 240982. May 2.
- 2015a Draft Greater North Park Community Plan. February.
- 2015b Draft Greater Golden Hill Community Plan. February.

Unitt

2004 San Diego County Bird Atlas No. 39. Proceedings of the San Diego Society of Natural History. October 31.

Project Description

San Diego, City of

2016 Draft Uptown Community Plan Update. January.

Regulatory Framework

California Air Resources Board (CARB)

- 2014 First Update to the Climate Change Scoping Plan. Building on the Framework Pursuant to AB 32 The California Global Warming Solutions Act of 2006. May 2014.
- 2015 California Air Quality Data Statistics. California Air Resources Board Internet Site.http://www.arb.ca.gov/adam/welcome.html. Accessed April 23, 2015.

San Diego County Regional Airport Authority (SDCRAA)

2014 San Diego International Airport Land Use Compatibility Plan. Prepared by Ricondo & Associates, Inc. Adopted April 3, 2014.

San Diego, City of

- 2013 City of San Diego Bicycle Master Plan. Prepared by Alta Planning and Design. December.
- 2015 City of San Diego General Plan Amendments. Adopted by City Council June 29, 2015.
- 2016 City of San Diego Recycling Ordinance https://www.sandiego.gov/environmentalservices/recycling/ro. Accessed on June 8, 2016

San Diego, County of

- 1992 1991/1992 Regional Air Quality Strategies. Air Pollution Control District. June.
- 2016 San Diego County Hazardous Materials Division. http://www.sandiegocounty.gov/content/sdc/deh/hazmat.html accessed on June 8, 2016.

United States Environmental Protection Agency (U.S. EPA)

- 2010 Regulatory Announcement "EPA and NHTSA Finalize Historic National Program to Reduce Greenhouse Gases and Improve Fuel Economy for Cars and Trucks" April.
- 2012 Regulatory Announcement "EPA and NHTSA Set Standards to Reduce Greenhouse Gases and Improve Fuel Economy for Model Years 2017-2025 Cars and Light Trucks. August.

Environmental Analysis – Uptown Community Plan Update

Land Use

City of San Diego

2011 California Environmental Quality Act Significance Determination Thresholds. Development Services Department.

Traffic and Circulation

City of San Diego

- 1998 Traffic Impact Study Manual. July
- 2011 California Environmental Quality Act Significance Determination Thresholds. Development Services Department.

State of California

1992 Department of Transportation. Travel Forecasting Guidelines. November.

Air Quality

California Air Pollution Control Officers Association (CAPCOA)

- 2010 Quantifying Greenhouse Gas Mitigation Measures: A Resource for Local Government to Assess Emission Reductions from Greenhouse Gas Mitigation Measures. August.
- 2013 California Emissions Estimator model (CalEEMod). User's Guide Version 2013.2.2 September.

California Air Resources Board (CARB)

- 2011a EMFAC 2011. California Air Resources Board. Updated January 2013.
- 2011b Final Supplement to the AB 32 Scoping Plan Functional Equivalent Document. August 19. Accessed from the CARB website at http://www.arb.ca.gov/cc/scopingplan/document/final_supplement_to_sp_fed.pdf on May 9, 2012.
- 2011c In-Use Off-Road Equipment (Construction, Industrial, Ground Support, and Oil Drilling) 2011 Inventory Model.

Greenhouse Gas Emissions

California Air Pollution Control Officers Association (CAPCOA)

- 2010 Quantifying Greenhouse Gas Mitigation Measures: A Resource for Local Government to Assess Emission Reductions from Greenhouse Gas Mitigation Measures. August.
- 2013 California Emissions Estimator model (CalEEMod). User's Guide Version 2013.2.2 September.

California Air Resources Board (CARB)

- 2011a Final Supplement to the AB 32 Scoping Plan Functional Equivalent Document. August 19. Accessed from the CARB website at http://www.arb.ca.gov/cc/scopingplan/document/final_supplement_to_sp_fed.pdf on May 9. 2012.
- 2011b In-Use Off-Road Equipment (Construction, Industrial, Ground Support, and Oil Drilling) 2011 Inventory Model.

California Energy Commission (CEC)

2013 2013 Building Energy Efficiency Standards for Residential and Nonresidential Buildings.

RECON

2015 Greenhouse Gas Analysis for the Uptown, North Park, and Golden Hill Community Plan Updates, City of San Diego. Project No. 30330/304032. SCH No. 2004651076. RECON No. 6086. September 18, 2015.

San Diego, County of

2013 Climate Action Plan. Final. November.

San Diego Gas and Electric (SDG&E)

2011 March 2011 Semi-Annual Compliance Report Pursuant to the California Renewables Portfolio Standard. Filed March.

Noise

Federal Highway Administration (FHWA)

2002 Rail with Trails: Lessons Learned. Section II Case Studies. FTA-MA-26-0052-04-1. Published August 2002. Accessed at https://www.fhwa.dot. gov/environment/recreational_trails/publications/rwt/ on June 6, 2014.

RECON

2016 Noise Analysis for the Uptown, North Park, and Golden Hill Community Plan Updates, City of San Diego Project No. 30330/304032 SCH No. 2004651076. May 24.

Historical Resources

Historic Resources Group

- 2014a Greater Golden Hill Community Plan Area Historic Resources Survey. June.
- 2014b Greater North Park Community Plan Area Historic Resources Survey. June.

Biological Resources

American Ornithologist Union

2013 Check-list of North American Birds: The Species of Birds of North America from the Arctic through Panama, Including the West Indies and Hawaiian Islands. 7th ed. Committee on Classification and Nomenclature and the 54th Supplement. Accessed on May 19, 2014 at http://checklist.aou.org/taxa/.

Brown, J. W., H. G. Real, and D. K. Faulkner

1992 Butterflies of Baja California. Lepidoptera Research Foundation, Inc., Beverly Hills, California.

City of San Diego

2011 California Environmental Quality Act Significance Determination Thresholds. Development Services Department.

- Crother, B. I., J. Boundy, J. A. Campbell, K. De Quieroz, D. Frost, D. M. Green, R. Highton, J. B. Iverson, R. W. McDiarmid, P. A. Meylan, T. W. Reeder, M. E. Seidel, J. W. Sites, Jr., S. G. Tilley, and D. B. Wake
 - 2008 Scientific and Standard English Names of Amphibians and Reptiles of North America North of Mexico, with Comments Regarding Confidence in our Understanding, Sixth Edition. Society for the Study of Amphibians and Reptiles Herpetological Circular No. 37.

Holland, R. F.

- 1986 Preliminary Descriptions of the Terrestrial Natural Communities of California. State of California Department of Fish and Wildlife.
- Jones, C., R. S. Hoffman, D. W. Rice, R. J. Baker, M. D. Engstrom, R. D. Bailey, D. J. Schmidly, and C. A. Jones
 - 1997 Revised Checklist of North American Mammals North of Mexico. Occasional Papers, Museum of Texas Tech University, Number 173. December 19.

Oberbauer, T. A.

1996 Terrestrial Vegetation Communities in San Diego County Based on Holland's Descriptions. February.

San Diego Geographic Information Source (SanGIS)

2012 SanGIS/SANDAG Data Warehouse. Obtained from http://www.sangis.org/Download_GIS_Data.htm.

Unitt

2004 San Diego County Bird Atlas No. 39. Proceedings of the San Diego Society of Natural History. October 31.

Geologic Conditions

GEOCON Incorporated

2015 Program EIR-Level Geotechnical Report Uptown, North Park, and Greater Golden Hill Planning Areas San Diego, California. June 10.

City of San Diego

2011 California Environmental Quality Act Significance Determination Thresholds. Development Services Department.

Boore, D. M., and Atkinson, G. M,

2008 Boore-Atkinson NGA Ground Motion Relations for the Geometric Mean Horizontal Component of Peak and Spectral Ground Motion Parameters, Report Number PEER 2007/01. Campbell, K. W., Bozorgnia, Y.

2008 NGA Ground Motion Model for the Geometric Mean Horizontal Component of PGA, PGV, PGD and 5% Damped Linear Elastic Response Spectra for Periods Ranging from 0.01 to 10 s, reprint of version submitted for publication in the NGA Special Volume of Earthquake Spectra, Volume 24, Issue 1, pages 139-171, February.

Chiou, B. S. J., and Youngs, R. R.

2008 A NGA Model for the Average Horizontal Component of Peak Ground Motion and Response Spectra, preprint for article to be published in NGA Special Edition for Earthquake Spectra.

Paleontological Resources

Deméré, T., and S. Walsh

1994 Paleontolgical Resources- County of San Diego. August.

Public Services and Facilites

Citygate

2011 Fire Service Standards of Response Coverage Deployment Study for the City of San Diego Fire Rescue Deparment. February 14.

City of San Diego

2011 California Environmental Quality Act Significance Determination Thresholds. Development Services Department.

Health and Safety

San Diego, City of

2008 General Plan Progam Environmental Impact Report. Available at https://www.sandiego.gov/planning/genplan/documents/peir.

San Diego, County of

2014 Unified San Diego County Emergency Services Organization and County of San Diego Operational Area Emergency Operations Plan (Annex Q – Evacuation). September.

Effects Not Found to be Significant

California Department of Conservation, Division of Mines and Geology (CDC)

- 1996 Probabilistic Seismic Hazard Assessment for the State of California, Open File Report 96-08.
- 2010 Farmland Mapping and Monitoring Program for 2010 San Diego County Maps. http://www.conservation.ca.gov/dlrp/fmmp. Accessed on June 8, 2016.

United States Department of Agriculture (USDA)

2016 Forest Service – FSGeodata Clearinghouse. http://data.fs.usda.gov/geodata/edw/datasets.php. Accessed on June 8, 2016.

Growth Inducement

San Diego, City of 2008a General Plan PEIR. 2008b 2008 General Plan. March.

Significant Unavoidable Impacts/Significant Irreversible Environmental Changes/Energy Conservation

California Natural Resources Agency

2009 California Environmental Quality Act, Appendix F – Energy Conservation. Revised 2009.

Kimley-Horn and Associates, Inc

2015 Uptown, North Park and Golden Hill CPU Traffic Impact Study. June.



Chapter 13.0 Individuals and Agencies Consulted

City of San Diego, Library Department

• Stephanie Gilbert

City of San Diego, Transportation and Storm Water Department

- Ruth Kolb, Program Manager
- Mark G. Stephens, AICP<u>, Associate Planner</u>

City of San Diego, Public Utilities Department

• Seevani Bista, P.E.

City of San Diego, Development Services Department

• James Quinn, Assistant Engineering Geologist

San Diego Fire Rescue

• Larry Trame, Assistant Fire Marshal

San Diego Police Department

• Richard Freedman, Police Lieutenant, Operational Support

San Diego Unified School District, Instructional Facilities Planning Department

• Sarah Hudson, Demographer



Chapter 14.0 Certification

This Program Environmental Impact Report (PEIR) has been completed by the City of San Diego's Planning Department and is based on independent analysis and determinations made pursuant to the San Diego Municipal Code Section 128.0103. The following individuals contributed to the fieldwork and/or preparation of this report.

City of San Diego

Planning Department

- Kurtis Steinert, AICP, Senior Environmental Planner
- Alyssa Muto, Deputy Director, Environment and Policy Analysis Division
- Denise Russell, Environmental Planner
- Rebecca Malone, AICP, Environmental Planner
- Marlon Pangilinan, Senior Planner
- Lara Gates, Community Development Specialist IV
- Bernard Turgeon, Senior Planner
- George Ghossain, Associate Traffic Engineer
- Tait Galloway, Program Manager
- Sara Osborn, AICP, Senior Planner
- Samir Hajjiri, PE, DCE, Senior Traffic Engineer
- Susan Morrison, AICP, Environmental Planner
- Myra Herrmann, Senior Environmental Planner
- Kelley Stanco, Senior Planner Historic Resources
- Howard Greenstein, Park Designer
- Jeanne Krosch, Senior MSCP Planner
- Kristen Forburger, Senior MSCP Planner
- Frank January, Project Manager, Facilities Financing
- Scott Mercer, Supervising Management Analyst, Facilities Financing

Development Services Department

- Jim Quinn, Senior Engineering Geologist
- Anna McPherson, Senior Planner

Office of the City Attorney

• Corrine Neuffer, Deputy City Attorney

Environmental Services Department

• Lisa Wood, Senior Planner

City of San Diego, Transportation and Storm Water Department

• Mark Stephens, Associate Planner

PEIR Preparer

RECON Environmental, Inc.

- Bobbi Herdes, Principal
- Jennifer Domeier, Environmental Analyst/Project Manager
- Eija Blocker, Production Specialist
- Stacey Higgins, Senior Production Specialist
- Frank McDermott, GIS Manager
- Sean Bohac, GIS Specialist
- Chris Nixon, Graphics

KLR Planning

• Karen Ruggels, Principal

Technical Appendices Preparers

Air Quality and Greenhouse Gas Technical Reports – RECON

• William A Maddux, Senior Technical Specialist

Biological Technical Report – RECON

- Gerry Scheid, Senior Biologist
- Beth Procsal, Biologist

Geotechnical Report - GEOCON Incorporated

- Garry W. Cannon, CEG, REC
- Rodney Mikesell, GE
- •

Historic Resources Survey – Historic Resources Group (HRG) and City of San Diego

- Christy McAvoy (HRG)
- Kari Michele Fowler (HRG)
- Christine Lazzaretto (HRG)
- Paul Travis (HRG)
- Kelley Stanco (City of San Diego)
- Jennifer Hirsch (City of San Diego)
- Cathy Winterrowd (City of San Diego)

Noise Analysis Report – RECON

• Jessica Fleming, Acoustical Analyst

Prehistoric Cultural Resource Analysis – AECOM

- Stacey Jordan, Practice Leader
- Cheryl Bowden-Renna Archaeologist
- Stephanie Jow Archaeologist

Traffic Impact and Mobility Studies – Kimley-Horn and Associates

- Mychal Loomis, P.E.
- Dave Sorenson, Senior Vice President
- Leo Espelet, Traffic Engineer